

# Barossa Gas Project Coastal Waters Operations Environmental Management Plan



# **Barossa Gas Project**

# **GEP Coastal Waters Operations Environmental Management Plan**

Document No.: BAS-210 0224

Project / Facility	Barossa GEP
Review interval (months)	No Review Required
Safety critical document	No

Rev	Owner	Reviewer(s) Managerial/Technical/Site	Approver	
	Barossa Production Manager	Senior Manager Environmental Approvals – WA, NA & TL	Barossa Project Director	
0				

Any hard copy of this document, other than those identified above, is uncontrolled. Refer to the Santos Offshore Business Document Management System for the latest revision.

## Contents

Units of measurement		22	
1.	Introdu	ction	23
	<b>1.1</b> Co	pastal Waters OEMP summary	23
	<b>1.2</b> Ac	tivity overview	24
	1.2.1	Primary approvals	24
	1.2.2	Other associated Barossa environment plans and approvals	24
	<b>1.3</b> Pu	rpose of this environmental management plan	27
	<b>1.4</b> Er	vironmental management plan validity	27
	<b>1.5</b> Op	perator and titleholder details	27
	1.5.1	Details for nominated liaison for the activity	28
	1.5.2	Notification procedure in the event of changed details	28
	<b>1.6</b> Er	vironmental management framework	28
	1.6.1	Environmental management system	29
	1.6.2	Workforce training, competency and emergency preparedness	29
	1.6.3	Santos environment, health and safety policy	29
	<b>1.7</b> Re	equirements of the Activity	29
2.	Activity description		
	<b>2.1</b> Ac	tivity summary	30
	<b>2.2</b> Op	perational Area and timing of Activity	31
	<b>2.3</b> Op	peration of the Gas Export Pipeline	33
	2.3.1	Pipeline design	33
	<b>2.4</b> Co	ommissioning to steady-state	33
	2.4.1	Initial start-up	33
	2.4.2	Initial start-up overview	34
	2.4.3	Contingencies	34
	<b>2.5</b> Ins	spection, Maintenance, Monitoring and Repair Activities	34
	2.5.1	Inspection methods	35
	2.5.2	Maintenance and Repairs	35
	2.5.3	In-line inspection contingency	36
	2.5.4	Stuck in-line inspection tool contingency	37
	2.5.5	Environmental monitoring activities	40
	<b>2.6</b> Ve	essel and helicopter operations	41
	2.6.1	Vessel types	41
	2.6.2	Uncrewed surface vessels	43
	2.6.3	Helicopter Support	43
	2.6.4	Remotely operated vehicle and autonomous underwater vehicle operations	43
	2.7 Ch	nemical Selection and Use	44

2.7.1	Ecotoxicity assessment	44
2.7.2	Biodegradation assessment	45
2.7.3	Bioaccumulation assessment	45
Descript	ion of the environment	46
<b>3.1</b> En	vironment that may be affected (EMBA)	48
3.1.1	Determining the EMBA	48
<b>3.2</b> Ex	isting environment	51
3.2.1	Geographical extent	51
3.2.2	Physical environment	53
3.2.3	Climate	55
3.2.4	Oceanography	55
3.2.5	Air quality	57
3.2.6	Shoals and banks	57
3.2.7	Offshore reefs and islands	61
3.2.8	Other seabed features of interest	61
3.2.9	Benthic habitats and communities	62
3.2.10	Shoreline habitats	62
3.2.11	Plankton	64
3.2.12	Protected areas and key ecological features	64
3.2.13	Threatened and migratory fauna	70
3.2.14	Socioeconomic receptors	103
3.2.15	Cultural features	122
Consult	ation	137
<b>4.1</b> Co	nsultation background	137
4.2 Re	lationship with consultation on Barossa Production Operations EP	138
<b>4.3</b> Co	nsultation context	139
4.4 OF	PGGS(E)R consultation requirements	139
<b>4.5</b> Go	vernment and industry guidance	140
<b>4.6</b> Ap	plicable case law and guidance	141
<b>4.7</b> Sa	ntos' consultation methodology	142
4.7.1	Overview	142
4.7.2	Identifying Relevant Persons	143
4.7.3	International persons	148
4.7.4	Public awareness campaign and self-identification opportunities	149
4.7.5	Consultation planning, preliminary and consultation activities	166
4.7.6	Provision of sufficient information	174
4.7.7	Reasonable period for consultation	176
<b>4.8</b> Co	nsultation report	177
4.8.1	Commonwealth Government Agency or Authority	178

3.

4.

	4.8	.2	NT Government Agency or Authority	191
	4.8	.3	WA Government Agency or Authority	198
	4.8	.4	Academic and Research Organisations	201
	4.8	.5	Energy Industry Titleholders / Operators	208
	4.8	.6	Environmental Conservation Organisations	215
	4.8	.7	Commercial Fishing (Commonwealth / NT / WA managed)	276
	4.8	.8	First Nations People and Groups	282
	4.8	.9	Industry Associations	315
	4.8	.10	Infrastructure Operators	324
	4.8	.11	Local Government Authorities and Recognised Community Reference / Liaison Groups	329
	4.8	.12	Tourism Operators	336
	4.8	.13	Other Relevant Persons	363
5.	Impa	ct ar	nd Risk Assessment Methodology	364
	5.1	Impa	act and Risk Assessment Methodology	364
	5.2	Sum	nmary of the environmental impact and risk assessment approach	365
	5.2	.1	Overview	365
	5.2	.2	Describe the Activity and hazards (planned and unplanned events)	366
	5.2	.3	Identify receptors and determine the nature and scale of impacts	366
	5.3	Des	cribe the environmental performance outcomes and control measures	367
	5.4	Dete	ermine the impact consequence level and risk rankings	367
	5.5	Eva	luate if impacts and risks are ALARP	369
	5.6	Eva	luate impact and risk acceptability	369
6.	Plan	ned a	activities impact assessment	370
	6.1	Nois	se emissions	371
	6.1	.1	Description of event	371
	6.1	.2	Nature and scale of environmental impacts	372
	6.1	.3	Environmental performance outcomes and control measures	382
	6.1	.4	Environmental impact assessment	386
	6.1	.5	Demonstration of as low as reasonably practicable	388
	6.1	.6	Acceptability evaluation	388
	6.2	Ligh	t emissions	390
	6.2	.1	Description of event	390
	6.2	.2	Nature and scale of environmental impacts	390
	6.2	.3	Environmental performance outcomes and control measures	396
	6.2	.4	Environmental impact assessment	399
	6.2	.5	Demonstration of as low as reasonably practicable	400
	6.2	.6	Acceptability evaluation	400
	6.3	Atm	ospheric and greenhouse gas emissions	403
	6.3	.1	Description of event	403

6.3.2	Nature and scale of environmental impacts	404
6.3.3	Environmental performance outcomes and control measures	415
6.3.4	Environmental impact assessment	421
6.3.5	Demonstration of as low as reasonably practicable	424
6.3.6	Acceptability evaluation	424
<b>6.4</b> Se	abed and benthic habitat disturbance	428
6.4.1	Description of event	428
6.4.2	Nature and scale of environmental impacts	428
6.4.3	Environmental performance outcomes and control measures	432
6.4.4	Environmental impact assessment	434
6.4.5	Demonstration of as low as reasonably practicable	435
6.4.6	Acceptability evaluation	435
6.5 Int	eractions with other marine users	437
6.5.1	Description of event	437
6.5.2	Nature and scale of environmental impacts	437
6.5.3	Environmental performance outcomes and control measures	438
6.5.4	Environmental impact assessment	440
6.5.5	Demonstration of as low as reasonably practicable	441
6.5.6	Acceptability evaluation	442
<b>6.6</b> Op	perational discharges	443
6.6.1	Description of event	443
6.6.2	Nature and scale of environmental impacts	444
6.6.3	Environmental performance outcomes and control measures	445
6.6.4	Environmental impact assessment	449
6.6.5	Demonstration of as low as reasonably practicable	449
6.6.6	Acceptability evaluation	450
Unplanr	ned events risk assessment	452
7.1 Re	elease of solid objects	453
7.1.1	Description of event	453
7.1.2	Nature and scale of environmental impacts	453
7.1.3	Environmental performance outcomes and control measures	454
7.1.4	Environmental impact and risk assessment	456
7.1.5	Demonstration of as low as reasonably practicable	456
7.1.6	Acceptability evaluation	457
7.2 Int	roduction of invasive marine species	459
7.2.1	Description of event	459
7.2.2	Nature and scale of environmental impacts	459
7.2.3	Environmental performance outcomes and control measures	460
7.2.4	Environmental impact and risk assessment	463
7.2.5	Demonstration of as low as reasonably practicable	463

7.

726	Accentability evaluation	161
7.2.0 7.3 Mai		404
731		400
732	Nature and scale of environmental impacts	400
733	Environmental performance outcomes and control measures	400
7.3.3	Environmental impact and rick assessment	409
735	Demonstration of as low as reasonably practicable	471
7.3.5		472
7.3.0 7.4 Min	or release of hydrocarbons and chemicals	472
7.4		474
7.4.1	Nature and scale of environmental impacts	474
7.4.2	Environmental performance outcomes and control measures	474
7 / /	Environmental impact and risk assessment	473
745	Demonstration of as low as reasonably practicable	470
746	Accentability evaluation	473
75 Sut	see release of dry natural das	481
751	Description of event	481
7.5.2	Nature and scale of environmental impacts	481
7.5.3	Environmental performance outcomes and control measures	482
7.5.4	Environmental impact and risk assessment	485
7.5.5	Demonstration of as low as reasonably practicable	486
7.5.6	Acceptability evaluation	486
<b>7.6</b> Unr	blanned release of MDO	488
7.6.1	Description of event	488
7.6.2	Nature and scale of environmental impacts	489
7.6.3	Spill modelling overview	489
7.6.4	Hydrocarbon and weathering characteristics	490
7.6.5	Hydrocarbon exposure thresholds	492
7.6.6	Spill risk assessment approach	498
7.6.7	Potential hydrocarbon impact pathways and nature and scale of impact	499
7.6.8	Stochastic spill dispersion modelling results	516
7.6.9	Environmental performance outcomes and control measures	516
7.6.10	Environmental impact and risk assessment	519
7.6.11	Demonstration of as low as reasonably practicable	523
7.6.12	Acceptability evaluation	523
7.7 Cor	ntingency spill response operations	526
7.7.1	Spill response strategies	526
7.7.2	Description of event	526
7.7.3	Nature and scale of environmental impacts	526
7.7.4	Environmental performance outcomes and control measures	529

	7.7.5	Environmental impact assessment	530
	7.7.6	Demonstration of as low as reasonably practicable	532
	7.7.7	Acceptability evaluation	532
8.	Impleme	ntation strategy	535
	8.1 Env	vironmental performance outcomes, control measures and performance standards	535
	8.1.1	Environmental performance outcomes and standards	535
	Note 1:	Significant is defined in the Santos Environment Consequence Descriptors (Appendix G).	536
	8.1.2	Control measures and performance standards	536
	8.2 Pla	n	547
	8.2.1	Environmental Management System	547
	8.2.1.1	Barossa Management System	547
	8.2.1.2	Environment, Health and Safety Policy	548
	8.2.1.3	Standards, procedures and tools	549
	8.2.2	Leadership, accountability and responsibility	549
	8.2.3	Adverse weather preparedness	554
	8.2.4	GHG Emissions Management	554
	8.2.5	Santos Decommissioning Strategy	556
	8.2.6	Emergency preparedness and response	558
	8.3 Do		561
	8.3.1	Workforce EP Awareness and Training	561
	8.3.2	Supporting Management Processes and Procedures	564
	8.3.2.1	Risk Management	564
	8.3.2.2	Asset Integrity Management	565
	8.3.2.3	Maintenance Management System	565
	8.3.2.4	Control of Work	565
	8.3.2.5	Vessel operations	566
	8.3.2.6	Aviation operations	567
	8.3.2.7	Lifting operations	567
	8.3.2.8	Waste management	567
	8.3.2.9	Biosecurity management	568
	8.3.2.1	0 Chemical management	571
	8.3.2.1	1 Methane Emissions Management	571
	8.3.2.1	2 Greenhouse Gas Emissions Management Plan	572
	8.3.3	Other Measures	573
	8.4 Ch	eck	573
	8.4.1	Assurance framework	573
	8.4.2	Monitoring, recording of emissions and discharges	574
	8.4.3	Inspections	576
	8.4.4	Audits	576
	8.4.5	Incident Investigations	577

	8.4.6	Marine Assurance	578
	8.4.7	Reporting and Notifications	578
	8.4.8	Document Management	596
	8.4.9	Post acceptance consultation implementation strategy	597
	8.5 A	ct	599
	8.5.1	Non-conformance Management	599
	8.5.2	Environment Plan Management of Change	599
	8.5.3	Performance Review and Continuous improvement	601
	8.5.4	Environmental Impact/Risk Review	602
	8.5.5	Engineering Change Management	602
	8.5.6	Decarbonisation Opportunity Management	602
9.	Refere	nces	604
Appe	ndix A	Santos Environment, Health and Safety policy	618
Appe	ndix B	Requirements applicable to the Activity	619
Appe	ndix C	EPBC Act Protected Matters Searches	659
Appe	ndix D	Relevant Persons Consultation and Advertising Materials	660
Appe	ndix E	Santos' Environment Consequence Descriptors	661
Appe	ndix F	Spill Modelling Results (maximum values across all seasons and water depths)	663
Appe	ndix G	Santos Climate Change Policy	664

# **Tables**

Table 1-2: Barossa Development activities and relevant Northern Territory waters environmental approvals and regulatory submissions       25         Table 1-3: Titleholder details for Barossa activities       26         Table 2-1: Summary of Coastal Waters OEMP activities and corresponding EP reference       30
Table 1-3: Titleholder details for Barossa activities       28         Table 2-1: Summary of Coastal Waters OEMP activities and corresponding EP reference       30
Table 2-1: Summary of Coastal Waters OEMP activities and corresponding EP reference
Table 2-2: Planned Timing of OEMP Activities
Table 2-3: Key Barossa Gas Export Pipeline structural design parameters in NT Coastal Waters
Table 2-4: Barossa Gas Export Pipeline key locations in NT waters         33
Table 2-5: Summary of inspection, monitoring, maintenance and repair activities and equipment, and associated discharges
Table 2-6: Typical IMMR vessel parameters    42
Table 2-7: Typical IMMR repair vessel parameters    42
Table 2-8: Typical large USV parameters    43
Table 2-9: OCNS Chemical hazard and risk management hazard quotient and ranking44
Table 2-10: Initial Offshore Chemical Notification Scheme grouping         44
Table 2-11: Aquatic species toxicity grouping    45
Table 3-1: Summary of the Barossa marine studies         47
Table 3-2: Summary of Barossa additional studies    47
Table 2.2: Hydrocarbon exposure threshold values
Table 5-5. Hydrocarbon exposure threshold values
Table 3-4: Integrated Marine and Coastal Regionalisation of Australia provincial bioregions relevant to the Activity
Table 3-3: Hydrocarbon exposure threshold values
Table 3-3: Hydrocarbon exposure threshold values       48         Table 3-4: Integrated Marine and Coastal Regionalisation of Australia provincial bioregions relevant to the Activity       51         Table 3-5: Distances to the nearest shoals and banks from the OA       58         Table 3-6: Summary of the results of the marine studies program       60
Table 3-3: Hydrocarbon exposure threshold values       48         Table 3-4: Integrated Marine and Coastal Regionalisation of Australia provincial bioregions relevant to the Activity       51         Table 3-5: Distances to the nearest shoals and banks from the OA       58         Table 3-6: Summary of the results of the marine studies program       60         Table 3-7: Habitats within the OA, MEVA and EMBA (IMCRA provincial bioregions)       64
Table 3-3: Hydrocarbon exposure threshold values       48         Table 3-4: Integrated Marine and Coastal Regionalisation of Australia provincial bioregions relevant to the Activity       51         Table 3-5: Distances to the nearest shoals and banks from the OA       58         Table 3-6: Summary of the results of the marine studies program       60         Table 3-7: Habitats within the OA, MEVA and EMBA (IMCRA provincial bioregions)       64         Table 3-8: Protected areas and key ecological features within the OA, MEVA and EMBA and approximate distance from the OA boundary       65
Table 3-3: Hydrocarbon exposure threshold values       44         Table 3-4: Integrated Marine and Coastal Regionalisation of Australia provincial bioregions relevant to the Activity       51         Table 3-5: Distances to the nearest shoals and banks from the OA       58         Table 3-6: Summary of the results of the marine studies program       60         Table 3-7: Habitats within the OA, MEVA and EMBA (IMCRA provincial bioregions)       64         Table 3-8: Protected areas and key ecological features within the OA, MEVA and EMBA and approximate distance from the OA boundary       65         Table 3-9: Marine park values overlapping the EMBA       67
Table 3-3: Hydrocarbon exposure threshold values
Table 3-3: Hydrocarbon exposure threshold values       44         Table 3-4: Integrated Marine and Coastal Regionalisation of Australia provincial bioregions relevant to the Activity       51         Table 3-5: Distances to the nearest shoals and banks from the OA       58         Table 3-6: Summary of the results of the marine studies program       60         Table 3-7: Habitats within the OA, MEVA and EMBA (IMCRA provincial bioregions)       64         Table 3-8: Protected areas and key ecological features within the OA, MEVA and EMBA and approximate distance from the OA boundary       65         Table 3-9: Marine park values overlapping the EMBA       67         Table 3-10: Threatened and migratory marine fauna that may be present in the OA, MEVA and EMBA       77         Table 3-11: Summary of turtle species usage of Tiwi Island beaches and waters       77
Table 3-3: Hydrocarbon exposure threshold values       44         Table 3-4: Integrated Marine and Coastal Regionalisation of Australia provincial bioregions relevant to the Activity       51         Table 3-5: Distances to the nearest shoals and banks from the OA       58         Table 3-6: Summary of the results of the marine studies program       60         Table 3-7: Habitats within the OA, MEVA and EMBA (IMCRA provincial bioregions)       64         Table 3-8: Protected areas and key ecological features within the OA, MEVA and EMBA and approximate distance from the OA boundary       65         Table 3-9: Marine park values overlapping the EMBA       67         Table 3-10: Threatened and migratory marine fauna that may be present in the OA, MEVA and EMBA       77         Table 3-11: Summary of turtle species usage of Tiwi Island beaches and waters       77         Table 3-12: Biologically important areas and habitat critical to the survival of a species identified in the EMBA and MEVA       90
Table 3-3: Hydrocarbon exposure timeshold values       44         Table 3-4: Integrated Marine and Coastal Regionalisation of Australia provincial bioregions relevant to the Activity       51         Table 3-5: Distances to the nearest shoals and banks from the OA       56         Table 3-6: Summary of the results of the marine studies program       60         Table 3-7: Habitats within the OA, MEVA and EMBA (IMCRA provincial bioregions)       64         Table 3-8: Protected areas and key ecological features within the OA, MEVA and EMBA and approximate distance from the OA boundary       65         Table 3-9: Marine park values overlapping the EMBA       67         Table 3-10: Threatened and migratory marine fauna that may be present in the OA, MEVA and EMBA       72         Table 3-11: Summary of turtle species usage of Tiwi Island beaches and waters       77         Table 3-12: Biologically important areas and habitat critical to the survival of a species identified in the EMBA and MEVA       90         Table 3-13: Relevant threats identified in recovery plans, conservation advice and wildlife conservation plans for species that occur or may occur within the EMBA       94
Table 3-3: Hybrocarbon exposure threshold values       44         Table 3-4: Integrated Marine and Coastal Regionalisation of Australia provincial bioregions relevant to the Activity       51         Table 3-5: Distances to the nearest shoals and banks from the OA       56         Table 3-6: Summary of the results of the marine studies program       60         Table 3-7: Habitats within the OA, MEVA and EMBA (IMCRA provincial bioregions)       62         Table 3-8: Protected areas and key ecological features within the OA, MEVA and EMBA and approximate distance from the OA boundary       65         Table 3-9: Marine park values overlapping the EMBA       67         Table 3-10: Threatened and migratory marine fauna that may be present in the OA, MEVA and EMBA       77         Table 3-11: Summary of turtle species usage of Tiwi Island beaches and waters       77         Table 3-12: Biologically important areas and habitat critical to the survival of a species identified in the EMBA and MEVA       90         Table 3-13: Relevant threats identified in recovery plans, conservation advice and wildlife conservation plans for species that occur or may occur within the EMBA       94         Table 3-14: Socioeconomic-related activities/features that occur or may occur in the OA and EMBA       94
Table 3-3: Trybrocarbon exposure inteshold values       44         Table 3-4: Integrated Marine and Coastal Regionalisation of Australia provincial bioregions relevant to the Activity       51         Table 3-5: Distances to the nearest shoals and banks from the OA       56         Table 3-6: Summary of the results of the marine studies program       60         Table 3-7: Habitats within the OA, MEVA and EMBA (IMCRA provincial bioregions)       62         Table 3-8: Protected areas and key ecological features within the OA, MEVA and EMBA and approximate distance from the OA boundary       65         Table 3-9: Marine park values overlapping the EMBA       67         Table 3-10: Threatened and migratory marine fauna that may be present in the OA, MEVA and EMBA       77         Table 3-11: Summary of turtle species usage of Tiwi Island beaches and waters       77         Table 3-12: Biologically important areas and habitat critical to the survival of a species identified in the EMBA and MEVA       90         Table 3-13: Relevant threats identified in recovery plans, conservation advice and wildlife conservation plans for species that occur or may occur within the EMBA       94         Table 3-14: Socioeconomic-related activities/features that occur or may occur in the OA and EMBA       94         Table 3-15: Commonwealth and state fisheries that overlap the EMBA       107
Table 3-4: Integrated Marine and Coastal Regionalisation of Australia provincial bioregions relevant to the Activity       51         Table 3-4: Integrated Marine and Coastal Regionalisation of Australia provincial bioregions relevant to the Activity       51         Table 3-5: Distances to the nearest shoals and banks from the OA       56         Table 3-6: Summary of the results of the marine studies program       60         Table 3-7: Habitats within the OA, MEVA and EMBA (IMCRA provincial bioregions)       64         Table 3-8: Protected areas and key ecological features within the OA, MEVA and EMBA and approximate distance from the OA boundary       65         Table 3-9: Marine park values overlapping the EMBA       67         Table 3-10: Threatened and migratory marine fauna that may be present in the OA, MEVA and EMBA       77         Table 3-12: Biologically important areas and habitat critical to the survival of a species identified in the EMBA and MEVA       90         Table 3-13: Relevant threats identified in recovery plans, conservation advice and wildlife conservation plans for species that occur or may occur within the EMBA       94         Table 3-14: Socioeconomic-related activities/features that overlap the EMBA       104         Table 3-15: Commonwealth and state fisheries that overlap the EMBA       107         Table 3-16: Located UCH protected under UCH Act and Heritage Act 2011 (NT) and distance to the OA       118
Table 3-3: Injurce function exposure times for values       44         Table 3-4: Integrated Marine and Coastal Regionalisation of Australia provincial bioregions relevant to the Activity       51         Table 3-5: Distances to the nearest shoals and banks from the OA       56         Table 3-6: Summary of the results of the marine studies program       60         Table 3-7: Habitats within the OA, MEVA and EMBA (IMCRA provincial bioregions)       64         Table 3-8: Protected areas and key ecological features within the OA, MEVA and EMBA and approximate distance from the OA boundary       65         Table 3-9: Marine park values overlapping the EMBA       67         Table 3-10: Threatened and migratory marine fauna that may be present in the OA, MEVA and EMBA       67         Table 3-11: Summary of turtle species usage of Tiwi Island beaches and waters       77         Table 3-12: Biologically important areas and habitat critical to the survival of a species identified in the EMBA and MEVA       90         Table 3-13: Relevant threats identified in recovery plans, conservation advice and wildlife conservation plans for species that occur or may occur within the EMBA       94         Table 3-15: Commonwealth and state fisheries that overlap the EMBA       104         Table 3-16: Located UCH protected under UCH Act and Heritage Act 2011 (NT) and distance to the OA       116         Table 3-17: Summary of cultural features and heritage values       134
Table 3-3: Tryutocation exposite threshold values       53         Table 3-4: Integrated Marine and Coastal Regionalisation of Australia provincial bioregions relevant to the Activity       51         Table 3-5: Distances to the nearest shoals and banks from the OA       56         Table 3-6: Summary of the results of the marine studies program       66         Table 3-7: Habitats within the OA, MEVA and EMBA (IMCRA provincial bioregions)       64         Table 3-8: Protected areas and key ecological features within the OA, MEVA and EMBA and approximate distance from the OA boundary       65         Table 3-9: Marine park values overlapping the EMBA       67         Table 3-10: Threatened and migratory marine fauna that may be present in the OA, MEVA and EMBA       67         Table 3-12: Biologically important areas and habitat critical to the survival of a species identified in the EMBA and MEVA       97         Table 3-13: Relevant threats identified in recovery plans, conservation advice and wildlife conservation plans for species that occur or may occur within the EMBA       94         Table 3-14: Socioeconomic-related activities/features that overlap the EMBA       107         Table 3-15: Commonwealth and state fisheries that overlap the EMBA       107         Table 3-16: Located UCH protected under UCH Act and <i>Heritage Act 2011</i> (NT) and distance to the OA       118         Table 3-17: Summary of cultural features and heritage values       134         Table 3-17: Summary of cultural features a
Table 3-5: Nyurocarbon exposure threshold values       14         Table 3-4: Integrated Marine and Coastal Regionalisation of Australia provincial bioregions relevant to the Activity       51         Table 3-5: Distances to the nearest shoals and banks from the OA       56         Table 3-6: Summary of the results of the marine studies program       60         Table 3-7: Habitats within the OA, MEVA and EMBA (IMCRA provincial bioregions)       64         Table 3-8: Protected areas and key ecological features within the OA, MEVA and EMBA and approximate distance from the OA boundary       66         Table 3-9: Marine park values overlapping the EMBA       67         Table 3-10: Threatened and migratory marine fauna that may be present in the OA, MEVA and EMBA       77         Table 3-11: Summary of turtle species usage of Tiwi Island beaches and waters       77         Table 3-12: Biologically important areas and habitat critical to the survival of a species identified in the EMBA and MEVA       90         Table 3-13: Relevant threats identified in recovery plans, conservation advice and wildlife conservation plans for species that occur or may occur within the EMBA       94         Table 3-15: Commonwealth and state fisheries that overlap the EMBA       107         Table 3-16: Located UCH protected under UCH Act and Heritage Act 2011 (NT) and distance to the OA       118         Table 3-17: Summary of cultural features and heritage values       134         Table 3-17: Summary of cultural features and
Table 3-5: Hydrocarboli exposure timeshou values

Table 4-5: Relevant Person categories Environmental aspects considered for Relevant Person category           identification	144
Table 4-6: Actions for identifying Relevant Persons by category	145
Table 4-7: Public awareness campaign overview	150
Table 4-8: Targeted advertising campaign details	151
Table 4-9: Summary of Relevant Persons	153
Table 4-10: Advertising and notification of Tiwi and Larrakia Consultation Sessions	173
Table 4-11: Consultation Summary Table - Commonwealth Government Agency or Authority	178
Table 4-12: Consultation Summary Table - NT Government Agency or Authority	191
Table 4-13: Consultation Summary Table - WA Government Agency or Authority	198
Table 4-14: Consultation Summary Table - Academic and Research Organisations	201
Table 4-15: Consultation Summary Table - Energy Industry Titleholders / Operators	208
Table 4-16: Consultation Summary Table – Environmental Conservation Organisations	215
Table 4-17: Consultation Summary Table - Commercial Fishing (Commonwealth / NT / WA managed)	276
Table 4-18: Consultation Summary Table – First Nations People and Groups	282
Table 4-19: Consultation Summary Table – Industry Associations	315
Table 4-20: Consultation Summary Table – Infrastructure Operators	324
Table 4-21: Consultation Summary Table – Local Government Authorities and Recognised Community Refere           Liaison Groups	nce / 329
Table 4-22: Consultation Summary Table – Tourism Operators	336
Table 4-23: Consultation Summary Table – Other Relevant Persons	363
Table 5-1: Impact and risk assessment terms	364
Table 5-2: Summary of the Santos Environmental Consequence Descriptors	368
Table 5-3: Likelihood description	368
Table 5-4: Santos risk matrix	369
Table 6-1: Environmental impact assessment summary	370
Table 6-2: Summary of expected sound source levels during the operations activity	372
Table 6-3: Continuous noise: summary of marine mammal impact thresholds as derived from Southall et al. (20 and National Oceanic and Atmospheric Administration (2019)	)19) 375
Table 6-4: Impulsive noise: unweighted sound pressure level, SEL24h and PK thresholds for acoustic effects of marine mammals	n 375
Table 6-5: Estimated distances to behavioural and physiological thresholds for marine mammals from vessels.	375
Table 6-6: Continuous noise: criteria for vessel noise exposure for marine reptiles	377
Table 6-7: Impulsive noise: criteria for impulsive noise exposure for marine reptiles, adapted from Popper et al.         2014	
Table 6-8: Continuous noise: summary of fish impact thresholds	379
Table 6-9:         Impulsive noise: summary of fish impact thresholds	379
Table 6-10: Control measure evaluation for noise emissions	382
Table 6-11: Artificial light impact potential criteria (marine turtles)	392
Table 6-12: Distance of equivalent moon radiances from the source (source: Pendoley, 2022)	393
Table 6-13: Control measure evaluation for light emissions	397
Table 6-14: Estimated Direct (Scope 1) emissions estimate for the 25 year lifecycle of Barossa GEP operations	s 405

Table 6-15: Indirect (Scope 3) emissions estimates for the 25 year lifecycle of Barossa GEP operations	405
Table 6-16: Total emissions estimate for the 25 year lifecycle of Barossa GEP operations	406
Table 6-17: Barossa GEP operations greenhouse gas emissions in context	407
Table 6-18: Control measures evaluation for atmospheric and greenhouse gas emissions	416
Table 6-19: Description and approximate footprint of seabed disturbance	429
Table 6-20: Control measures evaluation for seabed and benthic habitat disturbance	433
Table 6-21: Control measures evaluation for interaction with other marine users	438
Table 6-22: Control measures evaluation for operational discharges	446
Table 7-1: Environmental risk assessment summary	452
Table 7-2: Control measures evaluation for release of solid objects	454
Table 7-3: Control measures evaluation for introduction of invasive marine species	461
Table 7-4: Control measures evaluation for marine fauna interaction	469
Table 7-5: Control measures evaluation for minor release of hydrocarbons and chemicals	475
Table 7-6: Control measures evaluation for subsea release of dry natural gas	483
Table 7-7: Summary of model settings and assumptions for the vessel collision scenario	489
Table 7-8: Properties of marine diesel oil	490
Table 7-9: Hydrocarbon exposure values for the environment that may be affected	492
Table 7-10: Floating hydrocarbons exposure values	495
Table 7-11: Shoreline hydrocarbon accumulation exposure values	495
Table 7-12: Dissolved aromatic hydrocarbon exposure values	496
Table 7-13: Entrained hydrocarbon exposure values	497
Table 7-14: Physical and chemical pathways for hydrocarbon exposure and potential impacts to receptors	500
Table 7-15: Nature and scale of hydrocarbon spills on environment and socioeconomic receptors within the moderate exposure value area (MEVA)	507
Table 7-16: Control measures evaluation for surface release of MDO from a vessel	517
Table 7-17: Identified high environmental value and hotspot receptors	519
Table 7-18: Control measures evaluation for contingency spill response operations	529
Table 8-1: Environmental performance outcomes	536
Table 8-2: Environmental performance standards and measurement criteria	537
Table 8-3: Chain of command, key leadership roles and responsibilities	550
Table 8-4: Design features and maintenance plans to enable removal of the gas export pipeline and ancillary infrastructure at decommissioning	558
Table 8-5: Emergency response levels	559
Table 8-6: Training Standard, Procedures and Plans	563
Table 8-7: Emergency response training	564
Table 8-8: Biofouling mitigation measures	569
Table 8-9: Monitoring of vessel emissions and discharges	575
Table 8-10: Monitoring of vessel emissions and discharges	575
Table 8-11: OEMP Inspection Schedule and Scope for Vessels	576
Table 8-12: OEMP Audit Schedule	577
Table 8-13: Activity notification and reporting requirements	579

able 8-14: Marine user notification recipients
------------------------------------------------

# **Figures**

Figure 1-1: Barossa GEP location showing detail of the GEP Operational Area relevant to this GEP Coastal Wa OEMP 26	aters
Figure 2-1: Activity Operational Area of the GEP in NT Coastal Waters	32
Figure 2-2: Dual van Veen grab sampler	40
Figure 2-3: Bhagwan Dryden, as an example of a typical IMMR vessel	41
Figure 2-4: Sapura Constructor, as an example of a IMMR repair vessel for pipeline repairs	42
Figure 2-5: Reach Remote, an example of a representative USV	43
Figure 3-1: Location and extent of the EMBA and MEVA	50
Figure 3-2: IMCRA provincial bioregions and marine regions overlapping or proximal to the EMBA and MEVA	52
Figure 3-3: Bathymetry and seabed features overlapping or proximal to the EMBA and MEVA	54
Figure 3-4: Surface currents proximal to the EMBA	55
Figure 3-5: Banks and shoals overlapping or proximal to the OA, EMBAand M 59	EVA
Figure 3-6: Marine parks and protected areas overlapping or proximal to the EMBA and M 66	EVA
Figure 3-7: Key Ecological Features overlapping or proximal to the EMBA and MEVA	69
Figure 3-8: Indo-Pacific humpback dolphin biologically important areas overlapping or proximal to the EMBA	91
Figure 3-9: Marine turtle biologically important and habitat critical areas overlapping or proximal to the EMBA	92
Figure 3-10: Commonwealth managed fisheries overlapping or proximal to the OA and EMBA and MEVA	.109
Figure 3-11: Northern Territory managed fisheries overlapping or proximal to the OA and EMBA and MEVA	.110
Figure 3-12: Existing petroleum infrastructure overlapping or proximal to the EMBA and MEVA	.112
Figure 3-13: Defence training and exercise areas overlapping or proximal to the EMBA and MEVA	.114
Figure 3-14: Regional shipping overlapping or proximal to the EMBA and MEVA	.116
Figure 3-15: Underwater maritime cultural heritage overlapping or proximal to the EMBA and MEVA	.121
Figure 3-16: Representative Aboriginal/Torres Strait Islander Body Areas	.126
Figure 5-1: Hazard identification and assessment guideline	.366
Figure 5-2: Hierarchy of controls	.367
Figure 6-1: Photographs of a typical pipelay vessel at dusk	.391
Figure 7-1: Mass balance plot for an instantaneous 50 m <sup>3</sup> surface release of MDO subjected to a constant 5 km (2.6 m/s) wind, currents and 27°C water temperature	ot .491
Figure 7-2: Mass balance plot for an instantaneous 50 m <sup>3</sup> surface release of marine diesel oil subjected to varia wind speeds (1 to 12 m/s or 2 to 23 knots), currents and 27°C water temperature	able 491
Figure 7-3: Low exposure value contours used to define the EMBA and shoreline accumulation	.493
Figure 7-4: Sensitive receptors within the MEVA	.494
Figure 8-1: Key elements of the Barossa Management System	.548
Figure 8-2: Barossa Offshore Organisation Structure	.549
Figure 8-3: Santos Onshore Organisation Structure	.550
Figure 8-4: Emissions Target Setting Cycle	.555
Figure 8-5: Santos Project Delivery Process for Decommissioning	.557
Figure 8-6: Generic biofouling risk assessment process	.571



## Abbreviations and definitions

Term	Definition	
24/7	24 hours a day, seven days a week	
3D	Three dimensional	
AAPA	Aboriginal Areas Protection Authority	
ABF	Australian Border Force (Cth)	
ACCU	Australian Carbon Credit Units	
ACF	Australian Conservation Foundation	
ACMA	Australian Communications and Media Authority	
AFANT	Amateur Fishermen's Association of the Northern Territory	
AFMA	Australian Fisheries Management Authority (Cth)	
AFZ	Australian Fishing Zone	
АНО	Australian Hydrographic Office (Cth)	
AIMS	Australian Institute of Marine Science (Cth)	
ALAN	Artificial light at night	
ALARP	As low as reasonably practicable	
ALR Act	Aboriginal Land Rights (Northern Territory) Act 1976 (Cth)	
ALT	Aboriginal Land Trust	
AMCS - NT	Australian Marine Conservation Society – NT	
AMOSC	Australian Marine Oil Spill Centre	
AMP	Australian Marine Park	
AMSA	Australian Maritime Safety Authority	
AMSA-NT	Australian Marine Sciences Association Northern Territory	
ANP	Autoridade Nacional Do Petróleo (ANP - National Petroleum Authority)	
ANZG	Australian and New Zealand Guidelines for Fresh and Marine Water Quality (2018)	
APCAD	Australian Parents for Climate Action Darwin and NT	
AR6	Sixth Assessment Report	
ARP	Asset Reference Plan	
ASBTIA	Australian Southern Bluefin Tuna Industry Association	
ASC	Aboriginal Sea Company	
ATSIHP Act	Aboriginal and Torres Strait Islander Heritage Protection Act 1984 (Cth)	
ATRF	Arafura Timor Research Facility	
AUV	Autonomous underwater vehicle	
BAC	Balanggarra Aboriginal Corporation	
BJAC	Bardi and Jawi Niimidiman Aboriginal Corporation	
Barossa Development	The Barossa Development includes a Floating Production Storage and Offloading facility, subsea production system, supporting in-field subsea infrastructure and a gas export pipeline within Commonwealth waters. The Barossa Development is further described in the Barossa Development Offshore Project Proposal (OPP); available from: https://docs.nopsema.gov.au/A598153	
Barossa Gas Project / Barossa Project	The proposed Barossa Gas Project amalgamates both the infrastructure of the Barossa Development and the DPD Project to extract and process natural gas from the Barossa field. Barossa Gas Project and Barossa Project are used interchangeably.	
Barossa GEP	Barossa Gas Export Pipeline	

Term	Definition	
BMS	Barossa Management System	
Booklet	Barossa Production Operations Booklet	
BP	Boiling point	
BTEX	benzene, toluene, ethylbenzene and xylene	
BU	Bayu-Undan (pipeline)	
BWO	BW Offshore	
CAMBA	China-Australia Migratory Bird Agreement	
CCA	Climate Change Authority	
CCNT	Chamber of Commerce Northern Territory	
CCR	Central Control Room	
CCWA	Conservation Council of Western Australia	
CDU	Charles Darwin University	
CEFAS	Centre for Environment, Fisheries and Aquaculture Science	
CEMP	Construction Environmental Management Plan	
CEO	Chief executive officer	
CER	Clean Energy Regulator	
CFA	Commonwealth Fisheries Association	
CH <sub>4</sub>	methane	
CHARM	Chemical hazard assessment and risk management	
Climate Act	Climate Change Act 2022 (Cth)	
СМ	Control measure	
CMID	Common Marine Inspection Document	
CMMS	Computerised Maintenance Management System	
CMP	Crisis Management Plan	
CMT	Crisis management team	
CO <sub>2</sub>	carbon dioxide	
CO <sub>2</sub> -e	carbon dioxide equivalents	
Coastal Waters OEMP	Barossa GEP Coastal Waters Operations Environment Management Plan (i.e., this document)	
COLREGs	International Regulations for Preventing Collisions at Sea	
СР	Cathodic protection	
CSIRO	Commonwealth Scientific and Industrial Research Organisation	
Cth	Commonwealth	
CVI	Close visual inspection	
CWC	Concrete weight coating	
D&C	Drilling and Completions	
DAF	Department of Agriculture and Fisheries (NT)	
DAFF	Department of Agriculture, Fisheries and Forestry (Cth)	
DAH	Dissolved aromatic hydrocarbons	
DAWE	Department of Agriculture, Water and the Environment	
DAC	Dambimangari Aboriginal Corporation	

Term	Definition	
DBCA WA	Department of Biodiversity, Conservation and Attractions Western Australia	
DCCEEW	Department of Climate Change, Energy, the Environment and Water (Cth)	
DEPWS	Department of Environment, Parks and Water Security (NT)	
DF	Demersal Fishery	
DFAT	Department of Foreign Affairs and Trade (Cth)	
DHA	Department of Home Affairs	
DHAC	Darwin Harbour Advisory Committee	
DGV	Default guideline value	
DLI	Department of Logistics and Infrastructure (NT)	
DISR	Department of Industry, Science and Resources	
DIT	Department for Infrastructure and Transport (South Australia)	
DITT	Department of Industry, Tourism and Trade (NT)	
DLNG	Darwin liquefied natural gas plant	
DNV	Det Norske Veritas (Norway)	
DNP	Director of National Parks	
DME	Department of Mining and Energy	
DoD	Department of Defence	
DoEE	Department of the Environment and Energy	
DoT WA	Department of Transport Western Australia	
DP	Dynamic positioning	
DPD Project	Barossa Darwin Pipeline Duplication is a proposal to install, operate and decommission approximately 123 km of pipeline – comprising of approximately 23 km in Commonwealth waters (outside of the scope of this OEMP), approximately 8.26 km in NT Coastal Waters (covered by the scope of this OEMP) and approximately 91.74 km in NT Internal Waters (outside of the scope of this OEMP). The DPD Project is further described in the Environment Protection and Biodiversity Conservation Act 1999 (Cth) (EPBC Act) referral (EPBC 2022/09372)	
DPIRD	Department of Primary Industries and Regional Development (WA)	
DTFHC-NT	Department of Territory Families, Housing and Communities Northern Territory	
EC	Emergency Commander	
ECAP	Environmental Compliance and Assurance Plan	
ECNT	Environment Centre Northern Territory	
EDO	Environmental Defenders Office	
EEZ	Exclusive Economic Zone	
EHS	Environment, health and safety	
EMBA	Environment that may be affected	
EMP	Environmental Management Plan	
ENVID	Environmental hazard identification workshop	
EOFL	Endo of Field Life	
EP	Environment Plan	
EP Act	Environment Protection Act 2019 (NT)	
EP Regulations	Environment Protection Regulations 2020 (NT)	
EPA	Environment Protection Authority (NT)	

Term	Definition	
EPBC Act	Environment Protection and Biodiversity Conservation Act 1999 (Cth)	
EPO	Environmental performance outcomes	
EPS	Environmental performance standard	
ERP	Emergency Response Plan	
ERT	Emergency Response Team	
ESD	Ecologically sustainable development	
EVA	Environmental Value Area	
FPSO	Floating production storage and offloading vessel	
FRDC	Fisheries Research Development Council	
GDA	Gwalwa Daraniki Association	
GEP	Gas Export Pipeline	
GEP NT waters OPEP	Gas Export Pipeline (NT waters) Oil Pollution Emergency Plan (Operations Phase)	
GHG	Greenhouse gas	
GHS	Globally harmonised system	
GMDSS	Global Maritime Distress Safety System	
GVI	General visual inspection	
GWP	global warming potential	
HC	Habitat critical	
HEV	High environmental value	
HF	high frequency	
HFO	Heavy fuel oil	
HOCNF	Harmonised offshore chemical notification format	
HSE	Health, safety and environment	
HSS	Health, safety and security	
IAPP	International Air Pollution Prevention	
IEEC	International Energy Efficiency Certificate	
IFO	Intermediate fuel oil	
ILI	In-line inspection	
ILT	In-line tee	
ILUA	Indigenous Land Use Agreement	
IMCA	International Maritime Contractors Association	
IMMR	Inspection, maintenance, monitoring and repair	
IMO	International Maritime Organisation	
IMS	Invasive marine species	
IMT	Incident management team	
INMARSAT-C	International Maritime Satellite C	
IOPP	International Oil Pollution Prevention	
IPA	Indigenous Protected Area	
IPCC	Intergovernmental Panel on Climate Change	
ISO	International Organisation for Standardisation	

Term	Definition
ISPP	International Sewage Pollution Prevention
IUCN	International Union for the Conservation of Nature
JAMBA	Japan-Australia Migratory Bird Agreement
JARC	Jubilee Australia Research Centre
JKM	Japan Korea Marker
JRCC	Joint Rescue Coordination Centre
KEF	Key Ecological Feature
KLC	Kimberley Land Council
KMTA	Kimberley Marine Tourism Association
KP	Kilometre point
KPI	Key performance indicator
LAT	Lowest Astronomical Tide
LBL	Long baseline
LDC	Larrakia Development Corporation
LED	Light-emitting diode
LEVA	Low exposure value area
LF	low frequency
LNAC	Larrakia Nation Aboriginal Corporation
LNG	Liquefied natural gas
LTP	Long Term Plan
MAHA	Maritime archaeological heritage assessment
MARPOL	International Convention for the Prevention of Pollution from Ships 1973 (mod by the Protocol 1987)
MARS	Maritime Arrival Reporting System
MBES	Multibeam echosounder
MC	Measurement criteria
MDO	Marine diesel oil
MEG	Mono ethylene glycol
MEVA	Moderate exposure value area
MFL	Magnetic flux leakage
MFO	Marine fauna observer
MGAC	Miriuwung and Gajerrong Aboriginal Corporation
MGO	Marine gas oil
MGPS	Marine Growth Prevention System
MIAC	Mayala Inninalang Aboriginal Corporation
MNES	Matters of National Environmental Significance
МО	Marine Order
MoC	Management of change
MPNMP	Marine Parks Network Management Plan
MSI	Maritime safety information
MTWA	Marine Tourism WA

Term	Definition	
NA	Northern Australia	
N/A	Not applicable	
N <sub>2</sub> O	nitrous oxide	
NAILSMA	North Australia Indigenous Land and Sea Management Alliance	
NAXA	North Australian Exercise Area	
NDC	Nationally Determined Contributions	
NEBA	Net environmental benefit analysis	
NGER	National Greenhouse and Energy Reporting	
NGER Act	National Greenhouse and Energy Reporting Act 2007 (Cth)	
NIAA	National Indigenous Australians Agency	
NLC	Northern Land Council	
NLPGW	National Light Pollution Guidelines for Wildlife	
NMR	North Marine Region	
NOPSEMA	National Offshore Petroleum, Safety and Environmental Management Authority	
NOPTA	National Offshore Petroleum Titles Administrator	
NOx	nitrous oxides	
NPF	Northern Prawn Fishery	
NPFI	Northern Prawn Fishery Industry Pty Ltd	
NPI	National Pollution Inventory	
NT	Northern Territory	
NTGFIA	Northern Territory Guided Fishing Industry Association	
NTASS Act	Northern Territory Aboriginal Sacred Sites Act 1989 (NT)	
NT Coastal Waters	NT waters between the TSB and the boundary of NT and Commonwealth waters 3 nm offshore from the TSB. Petroleum pipeline activities in NT Coastal Waters are governed by the <i>Petroleum</i> ( <i>Submerged Lands Act</i> ) 1981 (NT)	
NT Internal Waters	NT waters inshore of the TSB. Petroleum pipeline activities in NT Internal Waters are governed by the Energy Pipelines Act 2018 (NT)	
NT Police, Fire and Emergency Services	Department of Police, Fire and Emergency Services Northern Territory	
NTSC	Northern Territory Seafood Council	
NWCS	North-West Cable System	
NWMR	North-West Marine Region	
NWSA	Northern Wildcatch Seafood Australia	
OA	Operational Area	
OCIMF	Oil Companies International Marine Forum	
OCNS	Offshore Chemical Notification Scheme	
ODS	Ozone-depleting substance	
OECD	Organisation for Economic Cooperation and Development	
OEMP	Operations Environmental Management Plan	
OFOV FME	Orientation field of view full moon equivalents	
OIM	Barossa Offshore Installation Manager	

Term	Definition	
OIW	oil in water	
ONLF	Offshore Net and Line Fishery	
OPEP	Oil Pollution Emergency Plan	
OPGGS Act	Offshore Petroleum and Greenhouse Gas Storage Act 2006 (Commonwealth)	
OPGGS(E)R	Offshore Petroleum and Greenhouse Gas Storage (Environment) Regulations 2023 (Cth)	
OPP	Offshore Project Proposal	
OPRC	International Convention on Oil Pollution Preparedness, Response and Cooperation 1990	
ORO	Other regional operators	
OSPAR	Convention for the Protection of the Marine Environment of the North-East Atlantic	
OTL	Barossa Operations Team Leader	
OVID	Offshore Vessel Inspection Database	
PAH	Polycyclic aromatic hydrocarbon	
PBC	Prescribed Bodies Corporate	
PFAS	Perfluoroalkyl substances	
PFOS	Perfluorooctane sulfonic acid	
PIMP	Barossa Project Pipeline Integrity Management Plan	
PLET	Pipeline end terminal	
PLONOR	Pose little or no risk	
PMST	Protected matters search tool	
POB	Persons on board	
PPUCH	Protocol for Protecting Underwater Cultural Heritage	
Production Operations	Operation of the FPSO and associated infrastructure in Commonwealth waters (Operational Area 1) and the operation of the GEP in both Commonwealth and NT Coastal Waters (known as Operational Area 2) as defined in the <i>Production Operation Information Booklet</i> .	
Production Operations EP	the Barossa Production Operations EP covers (among other things) operation of the Barossa GEP in Commonwealth waters	
PRT	Pipeline repair tool	
PSL Act	Petroleum (Submerged Lands Act) 1981 (NT)	
PTS	permanent threshold shift	
RAV	Registered Australian Vessel	
RBI	Risk-based inspection	
RBM	Riser Base Manifold	
RBU	Regional business unit	
ROKAMBA	Republic of Korea-Australia Migratory Bird Agreement	
ROV	Remotely operated vehicle	
RPT	regular public transport	
Santos	Santos NA Barossa Pty Ltd	
SBP	Sub-bottom profiler	
SCR	Santos Client Representative	
SDS	Safety data sheet	
SEL	sound exposure level	

Term	Definition	
SEL <sub>cum</sub>	cumulative sound exposure level	
SFR	Statutory Fishing Rights	
SIMAP	Spill Impact Model Application Package	
SITREP	Situation report	
SMC	Safeguard Mechanism Credits	
SMF	Spanish Mackerel Fishery	
SMPEP	Shipboard marine pollution emergency plan	
SOLAS	(International Convention for the) Safety of Life at Sea	
SOPEP	Shipboard oil pollution and emergency plan	
SO <sub>X</sub>	sulphur oxides	
SPA	Sales and Purchase Agreement	
SPL	sound pressure level	
SSS	Side-scan sonar	
STCW	International Convention on Standards of Training, Certification and Watchkeeping for seafarers	
SURF	Subsea umbilicals, risers and flowlines	
TL	Timor Leste	
TLC	Tiwi Land Council	
TPWC Act	Territory Parks and Wildlife Conservation Act 1976 (NT)	
TSB	Territorial Sea Baseline	
TTS	temporary threshold shift	
UCH	Underwater cultural heritage	
UCH Act	Underwater Cultural Heritage Act 2018 (Cth)	
UFP	Unexpected Finds Protocol	
UNFCCC	United Nations Framework Convention on Climate Change	
USBL	Ultra short baseline	
USV	Uncrewed surface vessel	
UV	Ultraviolet light	
UXO	Unexploded ordnance	
VHF	Very high frequency	
VOCs	volatile organic compounds	
WA	Western Australia	
WAFIC	Western Australian Fishing Industry Council	
WAGFA	Western Australian Game Fishing Association	
WAMSI	WA Marine Science Institution	
WG1	Working Group 1	
WG2	Working Group 2	
WG3	Working Group 3	
WGAC	Wunambal Gaambera Aboriginal Corporation	
WWF	World Wildlife Fund	



# **Units of measurement**

Term	Definition	
0	degree	
°C	degrees Celsius	
%	percent	
bara	absolute pressure (ambient pressure + gauge pressure)	
barG	gauge pressure	
cP	centipoise	
dB	decibel	
EC <sub>50</sub>	half maximal effective concentration	
hp	horsepower	
hr	hour	
Hz	hertz	
kg	kilogram	
kHz	kilohertz	
kL	kilolitres	
km	kilometre	
kW	kilowatt	
L	litre	
LC <sub>50</sub>	lethal concentration, 50%	
Log Pow	octanol-water partition coefficient	
m	metre	
m/s	meters per second	
m²	square metre	
m <sup>3</sup>	cubic metre	
mg/L	milligrams per litre	
ml	millilitre	
mm	millimetre	
MMscfd	million standard cubic feet per day	
nm	nanometre	
NM	nautical mile (1.852 km)	
ppm	parts per million	
ppmv	parts per million by volume	
sr	Steradian	
t	Tonne	
µg/L	micrograms/litre	
μPa	micro pascal	
W	Watt	

# 1. Introduction

## **1.1 Coastal Waters OEMP summary**

Offshore Petroleum and Greenhouse Gas Storage (Environment) Regulations 2023 (OPGGS(E)R 2023) requirements

Section 35. Notice of decision on environment plan, publication of accepted plan and submission and publication of summary

#### Submission of summary of accepted plan

(6) Within 10 days after receiving notice that NOPSEMA has accepted an Environment Plan (EP) (whether in full, in part or subject to limitations or conditions), the titleholder must submit a summary of the accepted plan to NOPSEMA for public disclosure.

(7) The summary:

- a. must include the following material from the environment plan for the activity:
  - i. the location of the activity;
  - ii. description of the receiving environment;
  - iii. a description of the activity;
  - iv. details of environmental impacts and risks of the activity;
  - v. a summary of the control measures for the activity;
  - vi. a summary of the arrangements for ongoing monitoring of the titleholder's environmental performance;
  - vii. a summary of the response arrangements in the oil pollution emergency plan;
  - viii. details of consultation already undertaken, and plans for ongoing consultation;
  - ix. details of the titleholder's nominated liaison person for the activity; and
- b. must be to the satisfaction of NOPSEMA.

This Barossa Gas Export Pipeline (GEP) Coastal Waters Operational Environmental Management Plan (Coastal Waters OEMP) is an Environment Plan for the purposes of the Offshore Petroleum and Greenhouse Gas Storage (Environment) Regulations 2023 (OPGGS(E)R) and provided to the Northern Territory (NT) Department of Mining and Energy. The regulatory regime applicable to this GEP Coastal Waters OEMP is described in further detail in Sections 1.2 and 1.3 below.

# The following Coastal Waters OEMP summary has been prepared as required by Section 35(7) of the OPGGS(E)R.

EP summary material requirement	Relevant section of EP containing EP summary material
The location of the Activity	Section 2
A description of the environment	Section 3 and Appendix C
A description of the Activity	Section 2
Details of the environmental impacts and risks	Sections 6 and 7
The control measures for the Activity	Sections 6 and 7
The arrangements for ongoing monitoring of the titleholder's environmental performance	Section 8
Response arrangements in the oil pollution emergency plan	Section 8.2.6 and Gas Export Pipeline (NT waters) Oil Pollution Emergency Plan (Operations Phase) (GEP NT waters OPEP).
Consultation already undertaken and plans for ongoing consultation	Sections 4 and 8
Details of the titleholders nominated liaison person for the Activity	Section 1.5.1



## 1.2 Activity overview

Santos NA Barossa Pty Ltd (Santos), as titleholder and nominated operator for the Barossa joint venture, proposes to conduct pipeline operations activities within Northern Territory Coastal Waters (Pipeline Licence NTC/PL5), as an integral part of the Barossa Gas Project. Unless expressly stated otherwise, a reference to Santos in this GEP Coastal Waters OEMP, is a reference to Santos in its capacity as nominated operator for the Barossa joint venture.

The Barossa Gas Project includes a Floating Production Storage and Offloading (FPSO) facility, a subsea production system, and supporting in-field subsea infrastructure. Gas and condensate produced from six subsea wells, will be treated and processed at the FPSO. Condensate will be loaded and exported directly from the FPSO to offtake tankers. Dry gas will be transported via the Barossa GEP to the DLNG.

The Activity described in this GEP Coastal Waters OEMP includes the transport of dry gas through the Barossa GEP and all support activities associated with inspection, maintenance, monitoring and repair (IMMR) of the infrastructure in the ~8.26 km section of the Barossa GEP located in Northern Territory (NT) Coastal Waters only. This section of pipeline is approximately 80 km north-west of Darwin, NT, and approximately 32 km south-west of the Tiwi Islands (Figure 1-1).

In accordance with the OPGGS(E)R, this GEP Coastal Waters OEMP identifies and evaluates environmental impacts and risks associated with the Activity (Section 2.2). The scope of this document excludes the operation of the remaining ~91.74 km section of the Barossa GEP (PL37) located in NT Internal Waters and the operation of the Barossa GEP (NT/PL5 and NT/PL6) in Commonwealth waters. These activities will be managed under separate NT and Commonwealth approvals.

## **1.2.1 Primary approvals**

Activities within NTC/PL5 for the Barossa GEP (approximately 8.26 km) are authorised pursuant to approvals granted on 15 March 2024 under the *Environment Protection and Biodiversity Conservation Act 1999* (Cth) (EPBC Act) (EPBC 2022/09372) and on 15 December 2023 under the *Environment Protection Act 2019* (NT) (EP Act) (EP2022/022-001). Pursuant to EPBC 2022/09372 and EP2022/022-001, installation, pre-commissioning, operation and decommissioning of the 123 km portion of the Barossa GEP (referred to as the Darwin Pipeline Duplication Project) and associated infrastructure in Commonwealth waters (NT/PL6), NT waters (NTC/PL5, PL37) and on land was authorised. Activities relating to the remainder of the Barossa GEP (connecting the pipeline to the Barossa FPSO facility) located in Commonwealth Waters were the subject of the Barossa Area Development OPP, accepted by NOPSEMA on 13 March 2018 under the OPGGS(E)R.

## 1.2.2 Other associated Barossa environment plans and approvals

The scope of this GEP Coastal Waters OEMP covers the operation of the Barossa pipeline (NTC/PL5; 8.26 km in length) and inspection, maintenance monitoring and repair activities in NT Coastal Waters only (the Activity) within the defined Operational Area (OA), which is in NT Coastal Waters between the NT/Commonwealth boundary and the Territorial Sea Baseline. The Barossa Gas Project production operations activities in Commonwealth waters, which includes operation of the pipeline in Commonwealth waters (~285 km), are covered in the Barossa Production Operations EP (NT/L1, NT/PL5 and NT/PL6). The operation of the pipeline in NT Internal Waters (PL37) and on land (~91.74 km) is covered in the GEP Internal Waters OEMP.

In addition, several other approvals authorise the construction and operations activities for the Barossa Gas Project. Table 1-1 outlines both the existing and pending Commonwealth approvals, detailing the scope of the approval and the length of the Barossa GEP (where relevant).

Relevant Approval	Jurisdiction	Status	Scone
Relevant Approval	ouncarear	otatao	00000
Barossa Development Drilling and Completions Environment Plan	Cwlth	Accepted by NOPSEMA	Campaign for drilling and completing up to eight production wells
Barossa Subsea Infrastructure Installation Environment Plan	Cwlth	Accepted by NOPSEMA	Installation of subsea umbilicals, risers and flowlines (SURF), manifolds and floating production, storage and offloading (FPSO) moorings
Barossa Gas Export Pipeline Installation Environment Plan	Cwlth	Accepted by NOPSEMA	Installation and pre-commissioning activities of 262 km pipeline in Cwlth waters

Table 1-1: Barossa Gas Project activities and relevant Commonwealth environmental approvals

Relevant Approval	Jurisdiction	Status	Scope
Barossa Darwin Pipeline Duplication Environment Plan	Cwlth	Accepted by NOPSEMA	Darwin Pipeline Duplication installation and preservation of 23 km of pipeline in Cwlth waters
Barossa Production Operations Environment Plan	Cwlth	Under assessment	<ul> <li>FPSO arrival, hook-up and commissioning</li> <li>Gas / condensate production and export from six subsea wells</li> <li>Operation of FPSO, subsea infrastructure, 285 km of the GEP in Cwlth waters</li> </ul>

Approvals and regulatory submissions applicable to the construction and operation of the Barossa GEP in NT waters are listed in Table 1-2 (including those to be developed).

# Table 1-2: Barossa Development activities and relevant Northern Territory waters environmental approvals and regulatory submissions

Northern Territory waters Barossa Development	Relevant Approval/Submission		
activity			
Darwin Pipeline Duplication installation	Darwin Pipeline Duplication (DPD) Project Environmental Approval granted under s133 of the <i>Environment Protection and Biodiversity Conservation Act 1999</i> (Cth) (EPBC Act) dated 15 March 2024 (EPBC 2022/09372):		
	Darwin Pipeline Duplication Project Referral (00-2022-09372)		
	Darwin Pipeline Duplication Project Preliminary Documentation Report.		
	DPD Project Environmental Approval (granted under s 69 of the <i>Environment Protection Act</i> (NT) dated 22 December 2023 (EP2022/022-001):		
	DPD Project Environment Protection Act (NT) Referral		
	• DPD Project Environment Protection Act (NT) Supplementary Environmental Report.		
	DPD Project Coastal Waters CEMP for 8.26 km of pipeline (submitted to the Department of Industry, Tourism and Trade [now Department of Mining and Energy] was accepted on 13 November 2024 in relation to construction, pre-commissioning and preservation activities to b undertaken in NT Coastal Waters only).		
	DPD Project Offshore Construction Environmental Management Plan (CEMP) and DPD Project Onshore CEMP (submitted to the Department of Industry, Tourism and Trade [now Department of Mining and Energy <sup>1</sup> ] in relation to construction, pre-commissioning and preservation activities to be undertaken in NT Internal Waters and land only).		
Operation of Gas Export Pipeline	DPD Project Environmental Approval granted under s133 of the EPBC Act dated 15 March 2024 (EPBC 2022/09372):		
	DPD Project Referral (00-2022-09372)		
	Darwin Pipeline Duplication Project Preliminary Documentation Report.		
	DPD Project Environmental Approval (granted under s 69 of the <i>Environment Protection Act</i> (NT) dated 22 December 2023 (EP2022/022-001):		
	DPD Project Environment Protection Act (NT) Referral		
	• DPD Project Environment Protection Act (NT) Supplementary Environmental Report.		
	This document, the Barossa GEP Coastal Waters Operations Environmental Management Plan (OEMP) for 8.26 km of the GEP (to be submitted to the Department of Mining and Energy for acceptance in relation to Barossa GEP operations activities to be undertaken in NT Coastal Waters only).		
	Barossa GEP Internal Waters OEMP for 91.74 km of the GEP (to be developed and submitted to the Department of Mining and Energy in relation to Barossa GEP operations activities to be undertaken in NT Internal Waters and on land only).		

<sup>&</sup>lt;sup>1</sup> An approved environmental management plan (EMP) is not a regulatory requirement for activities in NT Internal Waters covered by the *Energy Pipelines Act 1981 (NT)* and subsidiary *Energy Pipelines Regulations 2001 (NT)*. However, it is Department of Mining and Energy (DME) policy that the environmental management components of a pipeline management plan be made public in an EMP, which demonstrates how environmental risks are controlled and reduced to a level that is as low as reasonably practicable (ALARP) and acceptable, is submitted with the pipeline management plan required under the *Energy Pipelines Regulations 2001 (NT)*.



Figure 1-1: Barossa GEP location showing detail of the GEP Operational Area relevant to this GEP Coastal Waters OEMP



## **1.3 Purpose of this environmental management plan**

#### **OPGGS(E)R 2023 Requirements**

#### Section 34. Criteria for acceptance of environment plan

For the purposes of section 34, the criteria for acceptance of an environment plan (the environment plan acceptance criteria) for an activity are that the plan:

- a. is appropriate for the nature and scale of the activity; and
- b. demonstrates that the environmental impacts and risks of the activity will be reduced to as low as reasonably practicable; and
- c. demonstrates that the environmental impacts and risks of the activity will be of an acceptable level; and
- d. provides for appropriate environmental performance outcomes, environmental performance standards and measurement criteria; and
- e. includes an appropriate implementation strategy and monitoring, recording and reporting arrangements; and
- f. does not involve the activity or part of the activity, other than arrangements for environmental monitoring or for responding to an emergency, being undertaken in any part of a declared World Heritage property; and
- g. demonstrates that:
  - i. the titleholder has carried out the consultations required by section 25; and
  - ii. the measures (if any) that the titleholder has adopted, or proposes to adopt, because of the consultations are appropriate; and
- h. complies with the Act, this instrument and any other regulations made under this Act.

The GEP Coastal Waters OEMP has been prepared in accordance with the OPGGS(E)R for submission to DME and acceptance by the NT Minister for Mining and Energy.

The OPGGS(E)R are Commonwealth regulations that are enacted in NT by the *Petroleum (Submerged Lands)* (*Application of Commonwealth Laws) Regulations 2004* (NT) and apply to coastal waters between the Commonwealth/NT waters boundary and the Territorial Sea Baseline (Figure 2-1).

In accordance with the OPGGS(E)R, this GEP Coastal Waters OEMP details the environmental impacts and risks associated with the Activity and demonstrates how these will be reduced to as low as reasonably practicable (ALARP) and to an acceptable level. The Coastal Waters OEMP implementation strategy will be used to measure and report on environmental performance to demonstrate that impacts and risks are being continuously reduced to ALARP and are at an acceptable level. The environmental management of the Activity described in this GEP Coastal Waters OEMP complies with Santos' Environment, Health and Safety Policy (Appendix A) and with all relevant legislation (Appendix B). This Coastal Waters OEMP documents and considers all Relevant Persons consultation undertaken during the development of the GEP Coastal Waters OEMP (Section 4).

## **1.4** Environmental management plan validity

The operation of this GEP Coastal Waters OEMP commences on the date it is accepted under the OPGGS(E)R, by the NT Minister for Mining and Energy, and continues until submission and acceptance by a notification made pursuant to Section 46 of the OPGGS(E)R.

Santos may revise the GEP Coastal Waters OEMP post-acceptance, using the Management of Change (MoC) process described in Section 8.5.

## **1.5 Operator and titleholder details**

### **OPGGS(E)R 2023 Requirements**

Section 23. Details of titleholder and nominated liaison

(1) The environment plan must include the following details for the titleholder:

- a. name;
- b. business address;
- c. telephone number (if any);
- d. fax number (if any);
- e. email address (if any);
- f. if the titleholder is a body corporate that has an Australian Company Number (ACN) (within the meaning of the *Corporations Act 2001*).
- (2) The environment plan must also include the following details for the titleholder's nominated liaison for the activity
   a. name;

- b. business address;
- c. telephone number (if any);
- d. fax number (if any)
- e. email address (if any).
- (3) The environment plan must include arrangements for notifying NOPSEMA of any of the following:
  - a. a change in the titleholder;
  - b. a change in the titleholder's nominated liaison for the activity;
  - c. a change in the contact details for either the titleholder or the nominated liaison.

The titleholder details are provided in Table 1-3, with the nominated operator shown in bold.

#### Table 1-3: Titleholder details for Barossa activities

Title	Titleholder (nominated operator in bold)	Australian company number	Interest (%)	Contact details	
Coastal and	Santos NA Barossa Pty Ltd	109 974 932	25.0	Business Address: Level 7, 100 St Georges Terrace, Perth, Western Australia, 6000 Telephone number: (08) 6218 7100 Fax number: (08) 6218 7200 Email address: barossa.regulatory@santos.com	
Territorial waters Licence NTC/PL5 (PSL Act)	Santos Offshore Pty Ltd	005 475 589	25.0		
	SK E&S Australia Pty Ltd	158 702 071	37.5	Business Address: Level 27, 152-158 St Georges Terrace, Perth WA 6000 Telephone number: (08) 6186 2320 Fax number: None Email address: upstream@sk.com	
	JERA Barossa Pty Ltd	654 004 387	12.5	Business Address: Level 36, QV1, 250 St Georges Terrace, Perth, Western Australia, 6000 Telephone number: (08) 6311 7610 Fax number: (08) 6311 7613 Email address: barossa@jeraaustralia.com.au	

## 1.5.1 Details for nominated liaison for the activity

Details for Santos' nominated liaison person for the activity are as follows:

Name:Michael MarrenBusiness address:Level 7, 100 St Georges Terrace, Perth, Western Australia, 6000Telephone number:(08) 6218 7100Email address:offshore.consultation@santos.comACN:109 974 932

## 1.5.2 Notification procedure in the event of changed details

If there is a change in the nominated titleholder, the titleholder's nominated liaison, or a change in the contact details for a titleholder or liaison, Santos will notify DME and NOPSEMA and provide updated details.

## **1.6 Environmental management framework**

OPGGS(E)R 2023 Requirements			
Section 21. Environmental assessment			
Requirements			
(4) The environment plan must:			
<ul> <li>a. describe the requirements, including legislative requirements, that apply to the activity and are relevant to the environmental management of the activity; and</li> </ul>			
b. demonstrate how those requirements will be met.			
Section 24. Other information in the environment plan			
The environment plan must contain the following:			



a. a statement of the titleholder's corporate environmental policy.

#### **1.6.1** Environmental management system

Section 8.2.1 contains a description of the Environmental Management System for the Activity, as relevant to ongoing management of environmental impacts and risks for the duration of the Activity.

### **1.6.2** Workforce training, competency and emergency preparedness

Section 8.3.1 addresses measures to ensure that each employee or contractor working on, or in connection with, the Activity is aware of the employee's or contractor's responsibilities in relation to this GEP Coastal Waters OEMP, including appropriate competencies and training. Section 8.2.6 addresses arrangements for emergencies or potential emergencies.

## 1.6.3 Santos environment, health and safety policy

The Activity will be conducted in accordance with the Santos Environment, Health and Safety Policy (Appendix A).

Sections 3, 4, 6, 7 and 8 reflect this policy, detailing and evaluating environmental impacts and risks and providing control measures with set environmental performance outcomes (EPOs) and standards (EPSs).

## **1.7 Requirements of the Activity**

**OPGGS(E)R 2023 Requirements** 

#### Section 21. Environmental Assessment

Requirements

21(4) The environment plan must:

(a) describe the requirements, including legislative requirements, that apply to the activity and are relevant to the environmental management of the activity; and

(b) demonstrate how those requirements will be met.

Relevant requirements, including legislative requirements applicable to the Activity are presented in Appendix B. with reference to relevant Coastal Waters OEMP sections where the legislation may prescribe or control how an activity is undertaken. Australia is a signatory to international conventions and agreements that oblige the Commonwealth government to prevent pollution and protect specified habitats, flora and fauna. Relevant government departments have been consulted during the development of this GEP Coastal Waters OEMP to identify applicable legislation, conventions and agreements, as detailed in Section 4.

# 2. Activity description

#### OPGGS(E)R 2023 Requirements

#### Section 21. Environmental assessment

#### Description of the activity

(1) The environment plan must contain a comprehensive description of the activity including the following:

- a. the location or locations of the activity;
- b. general details of the construction and layout of any facility that is used in undertaking the activity;
- c. an outline of the operational details of the activity (e.g. seismic surveys, exploration drilling or production) and proposed timetables for undertaking the activity; and

d. any additional information relevant to consideration of environmental impacts and risks of the activity.

Note: An environment plan will not be capable of being accepted by NOPSEMA if an activity or part of the activity, other than arrangements for environmental monitoring or for responding to an emergency, will be undertaken in any part of a declared World Heritage property (see section 34 of the OPGGS(E)R 2023 Requirements).

## 2.1 Activity summary

The Activity is the transport of dry gas through the Barossa GEP and the associated support activities, of inspection, maintenance, monitoring and repair (IMMR) of the infrastructure in NTC/PL5 and environmental monitoring activities in the circumstances referred to in Section 2.5.5. Table 2-1 shows the key attributes of the activities covered by this GEP Coastal Waters OEMP. A detailed activity description is provided in Sections 2.4 to 2.7.

The Activity will be undertaken within the NT Coastal Waters operational area (OA), approximately 80 km northwest of Darwin, NT. The OA and pipeline route within the OA, are approximately 31.5 km and 32 km south of the Tiwi Islands (Cape Fourcroy), NT respectively, at their closest points, and approximately 67.5 km south-east of the Oceanic Shoals Marine Park (Figure 1-1).

The locations for activities along the Barossa GEP are described using 'kilometre points' (KP), where KP 0 is the beginning of the pipeline from the southern pipeline end termination (PLET C) in Commonwealth waters (Figure 2-1). The Barossa GEP traverses through NT Coastal and Internal waters where it ends at KP 122.687 at the upstream weld of the beach valve at the Darwin liquefied natural gas (DLNG) facility. The scope of this GEP Coastal Waters OEMP covers the operation of 8.26 km of the Barossa GEP between the NT/Commonwealth boundary at KP 23 and the Territorial Sea Baseline (TSB) at KP 31.265 (Figure 2-1).

Permit areas	Coastal and territorial waters licence NTC/PL5		
Hydrocarbon type	The pipeline will contain dry natural gas. Vessels undertaking activities associated with this GEP Coastal Waters OEMP will use Group II hydrocarbon fuels such as Marine gas oil (MGO) or marine diesel oil (MDO).		
Location	The pipeline within the scope of this GEP Coastal Waters OEMP is limited to activities within Northern Territory (NT) Coastal Waters from KP 23 (NT/Commonwealth boundary) to KP 31.265 (the Territorial Sea Baseline) (Figure 2-1).		
Summary of key activities	Activities within the Operational Area include:		
	<ul> <li>Initial start-up to steady-state and GEP pressurisation (Section 2.4)</li> </ul>		
	<ul> <li>Transporting dry natural gas from the Barossa field to DLNG</li> </ul>		
	Pipeline inspection, maintenance, monitoring and repair (IMMR)		
	• Environmental monitoring/sampling (e.g. sediment and marine growth), if required		
	Vessel and helicopter support operations.		
Subsea infrastructure	Barossa GEP consisting of 26-inch outer diameter carbon steel with an external anti- corrosion coating and sacrificial anodes to maintain pipeline integrity and concrete coating to provide stability and protection (Section 2.3).		
	GEP supporting structures consisting of span rectification structures (includes gravel/rocks, concrete mattresses and/or grout bags).		
Water depths	Water depths range from 47 m at the TSB to 50 m at the NT/Commonwealth boundary.		
Vessels	Typically, a single vessel is used to conduct IMMR and environmental monitoring activities. The size and type of vessel depends on the activities being carried out. If a repair to the pipeline is required then additional vessels may be required (Section 2.5).		

Activities may be undertaken any time of year within the GEP Coastal Waters OEMP period. The operational design life of the Barossa GEP is 25 years.

## 2.2 **Operational Area and timing of Activity**

The Barossa GEP petroleum activities in NT Coastal Waters will occur within the Operational Area (OA) shown in Figure 2-1. The OA extends from the NT/Commonwealth boundary at KP 23 to the TSB at KP 31.265 (Figure 2-1), with a one km wide corridor, 500 m either side of the Barossa GEP centreline. The corridor represents the area within which the Activity as described in section 2.1 will occur.

This Coastal Waters OEMP assumes the IMMR and environmental monitoring activities in the OA may be undertaken at any time of year within the GEP Coastal Waters OEMP validity period.

The GEP infrastructure exporting gas to the DLNG facility will operate for approximately 25 years following commencement of production operations.

Planned timing of vessel activities are summarised in Table 2-2.

#### **Table 2-2: Planned Timing of OEMP Activities**

Vessel Type and #	Indicative Activities	Indicative Timing and Duration (concurrent timing shown unless specified otherwise)		
Initial start-up to steady state ~4 months				
N/A	Start-up and pressurisation of GEP	~ 4 months		
IMMR Vessel (1)	GEP IMMR activities	During start-up to steady state		
		~ 7 days within the OA		
Steady Sate Operatio	ns ~25 years			
N/A	Transmission of dry gas from the FPSO to DLNG	Commencement upon completion of start-up activities and continuous thereafter for ~25 years		
IMMR Vessel (1 per campaign)	GEP inspection, maintenance     and monitoring activities	Approximately every three years after completion of start-up activities at the following indicative frequencies:		
	<ul> <li>USV activities</li> <li>ROV operations</li> <li>Environmental monitoring using routine sampling techniques and</li> </ul>	<ul> <li>GEP inspections / maintenance / monitoring – every three to five years for up to 7 days</li> </ul>		
		<ul> <li>Environmental monitoring as required during operations, up to half a day in the OA.</li> </ul>		
	equipment	Required for the validity period of this GEP Coastal Waters OEMP to support campaigns as required.		
IMMR Repair	GEP repair activities (if required)	GEP repairs – up to 30 days		
Vessel (1)		Required for the validity period of this GEP Coastal Waters OEMP on an as required basis.		
IMMR Repair	Supporting GEP repair activities	GEP repairs – up to 30 days		
Support Vessel (1)	(if required)	Required for the validity period of this GEP Coastal Waters OEMP on an as required basis.		
	<ul> <li>Undertake support activities via ROV (if required)</li> </ul>			
	<ul> <li>Transport of equipment to the IMMR repair vessel</li> </ul>			



Figure 2-1: Activity Operational Area of the GEP in NT Coastal Waters



## 2.3 **Operation of the Gas Export Pipeline**

This Coastal Waters OEMP includes activities associated with operation of the GEP within the OA for transfer of dry natural gas from the Barossa field to DLNG.

## 2.3.1 Pipeline design

Structural design parameters of the GEP are given in Table 2-3. Nominal coordinates of KPs shown in Figure 2-1 are provided in Table 2-4 for NT waters. The GEP is designed to be stable on the seabed up to 100-year cyclonic metocean conditions.

### Table 2-3: Key Barossa Gas Export Pipeline structural design parameters in NT Coastal Waters

Item	Description
Material	Carbon manganese steel linepipe
	Asphalt enamel anti-corrosion coating
	Concrete weight coating
	Heat shrink sleeve field joint coating
	Epoxy internal flowcoat coating
Pipe size	NT Coastal Waters: 26 inches
Internal diameter (mm)	619.8
Design temperature	0°C minimum and 50°C maximum
Design pressure	199 bara at 0 m LAT

### Table 2-4: Barossa Gas Export Pipeline key locations in NT waters

Location	Kilometre point (KP)	Easting*	Northing*
Boundary between NT and Commonwealth waters	~KP 23	618,128.5	8,663,104.1
Territorial Sea Baseline	KP 31.265	625,835.8	8,657,503.1
26x34 inch reducer (out of scope of this GEP Coastal Waters OEMP)	KP 61.8	654,554.3	8,647,162.5
In-line tee (out of scope of this GEP Coastal Waters OEMP)	KP 62.8	655,495.8	8,646,825.4
Upstream weld of the beach valve (out of scope of this GEP Coastal Waters OEMP)	KP 122.687	702,472.3	8,614,655.7

\*Coordinates in GDA 94, MGA zone 52

## 2.4 Commissioning to steady-state

## 2.4.1 Initial start-up

Following completion of cold-commissioning of the FPSO (out of scope of this GEP Coastal Waters OEMP and is to be authorised by NOPSEMA in the Barossa Production Operations Environment Plan) there will be an initial start-up period. This is planned with the objective to establish stable production in a safe and efficient manner whilst minimising FPSO flaring to as low as reasonably practicable (ALARP). The initial start-up of the FPSO (out of scope of this GEP Coastal Waters OEMP) will take into account the requirements of the DLNG facility for commissioning prior to production ramp-up. Major steps undertaken during the initial start-up phase relevant to this GEP Coastal Waters OEMP (i.e. distribution of dry natural gas in the GEP) are as below:

- Gas treatment is established via a single train with dewpointing and carbon dioxide (CO<sub>2</sub>) removal online. Once on-spec gas is available, initial pressurisation of the Barossa GEP can commence.
- Establishment of export gas compression



- Once DLNG has completed its commissioning (expected to take two to four weeks), the Barossa field will be ramped up to full production rates and the GEP pressurised to normal operating pressure
- Once the FPSO is in steady state, FPSO performance testing will commence (which is out of scope of this GEP Coastal Waters OEMP). On completion of the final FPSO performance test, the start-up phase is complete.

## 2.4.2 Initial start-up overview

Key requirements through the initial start-up phase are summarised below:

- The GEP will initially contain Nitrogen with residual MEG.
- Initial start-up of the GEP will commence after the Offshore facilities have been commissioned and started up to the point where Offshore can supply on-specification export gas for the export system pressurisation.
- The GEP will be initially pressurised (from the FPSO) to enable DLNG commissioning.
- The GEP will be initially ramped up to 50% of the normal operation rates.
- GEP ramp-up to full capacity will begin after the completion of the FPSO Blowdown Testing.
- The GEP pressure will be maintained at the point of near equal pack/de-pack during the periods it is operated at 50% or full capacity to allow sufficient Operator response time in the event of a trip.
- The initial start-up phase will be complete after the conclusion of the final FPSO Performance Test. Subsequently, the gas export system will be operated under normal operation controls.

## 2.4.3 Contingencies

While in communication with each other, should either the FPSO or DLNG be delayed or required to stop, the GEP can be isolated and/or maintained in standby at constant pressure and composition.

## 2.5 Inspection, Maintenance, Monitoring and Repair Activities

IMMR is typically conducted by ROVs or AUVs from one or more vessels that have DP capabilities. An ROV may also be launched from a USV. Divers may be used for operations on the rare occasion ROVs or AUVs cannot be used. Details of IMMR activities, including typical equipment required and potential environmental aspects, are described below and summarised in Table 2-7.

IMMR typically includes:

- general visual inspection (GVI) and close visual inspection (CVI) of the Barossa GEP
- cathodic protection (CP) surveys, including readings and GVI to determine condition of anodes
- geophysical and infrastructure surveys in the vicinity of the GEP within the OA, including:
  - multibeam echo sounder (MBES), which uses sound pulses to establish the seabed profile, position or shape of subsea infrastructure. Most MBES systems work by transmitting a broad acoustic pulse from a hull-, pole- or ROV-mounted transducer.
  - side scan sonar (SSS), which detects debris and other obstructions on the seafloor using a towed transducer that transmits high-frequency acoustic pulses.

Maintenance/repair activities may include non-routine replacement, or repair of a section of the Barossa GEP in the OA. Discharges associated with non-routine pipeline repair or replacement are summarised in Table 2-5.In-line inspection (ILI) via pigging for the GEP is not a planned activity but may be undertaken in the case where it is suspected that pipeline integrity has been compromised due to a pipeline feed gas dew point excursion (see Section 2.5.3). Similarly, it is not planned for an inline inspection tool to become stuck or damaged in the pipeline but it is a possibility that may need to be resolved (see Section 2.5.4).

If 100-year cyclonic conditions are exceeded, as part of the GEP integrity management plan, a post-cyclone survey would be triggered to identify any unacceptable movement, lateral displacement, of loss of support/scour.

Marine survey vessels, campaign or other support vessels may be used for activities as outlined in Section 2.5.

It is through implementing the IMMR regime set out in this section that Santos will meet its obligations under section 98(2) of the *Petroleum (Submerged Lands) Act 1981* (NT) to 'maintain in good condition and repair all structures, equipment and other property in an operations area and used in connection with the operations in which he is engaged'.



## 2.5.1 Inspection methods

### 2.5.1.1 Visual inspection

External visual inspections of the Barossa GEP may be undertaken, typically using an ROV. General visual inspections can be used to confirm the results of other inspection methods, and aid in the planning of maintenance and repair activities.

Close visual inspections of the Barossa GEP system may be undertaken by divers where ROVs or AUVs cannot be used. However, due to the relative complexity (based on health and safety risk) and cost of implementing diving operations in comparison with alternative methods (e.g., ROV), other inspection methods are preferred.

## 2.5.1.2 Cathodic Protection Survey

Cathodic Protection (CP) survey may be performed on the external surface of the GEP using the following methods (or combination of it):

- Contact CP survey, which includes a ROV holding a CP probe and then connected to the exposed pipe steel substrate (e.g. on the field joint / girth weld area, where there is no concrete weight coating) or the probe connected to the exposed anode substrate.
- Field gradient CP survey (contactless survey), which includes a ROV holding a CP probe around 0.2 0.5 m above the pipeline, to measure the potential field gradient along the pipeline length without contacting the pipeline.

## 2.5.1.3 Marine growth removal

Removal of marine growth is typically only required for inspection purposes and is conducted at localised areas using high pressure water cleaning or brushing or a combination of these:

- Water jetting conducted by ROV, water is pressurised to above hydrostatic pressure. Generally, waterjetting activities are through small-diameter water jets that act locally on the pipe and structure. Wash-out or induced currents are typically not experienced during this activity due to the nature of the operation.
- Mechanical brushing typically a coarse brush is applied to the pipeline or structure on a localised area only.

## 2.5.1.4 Inspection and monitoring intervals

Initial 'first in-service' inspections will be performed nominally during the first one to two years of operations and, after that, at the intervals required on the basis of a risk based inspection program. The objective of the first in-service inspections shall be to provide operational performance information.

Inspection of the Barossa GEP at any phase of its life will be conducted in accordance with a risk-based inspection (RBI) schedule, as described in the Barossa Project Pipeline Integrity Management Plan (PIMP) - Nearshore GEP.

To determine IMMR frequency, the risk assessment typically uses:

- a threat assessment to determine threats to integrity in operation
- historical IMMR information acquired through inspection, monitoring and repair of similar assets
- nominal inspection frequencies set within the IMMR Plan.

Inspection and monitoring frequencies can range from three to five years, or as needed. The findings of the IMMR campaigns will be used to inform the future frequencies of the IMMR activities. When an IMMR activity occurs, the expected duration may be seven to 30 days depending on whether maintenance and repairs are required. Additional unplanned external or internal inspections may be performed after significant external events (such as extreme weather, sea conditions, seismic activity, third-party interactions), integrity assessments or other triggers that indicate further inspection is required.

## 2.5.2 Maintenance and Repairs

Anomalies identified during planned inspections and condition monitoring are reviewed, risk assessed, and managed. The risk is mitigated by actions such as repair, re-rating, upgrade, or monitoring, as appropriate.

Urgent repairs (e.g., in the event of damage requiring precautionary shutdown) are addressed in the Emergency Repair Strategy – Subsea Pipelines. The strategy outlines various repair options available in the event of a Barossa GEP leak, rupture, or severe mechanical damage, including information on aspects such as material, equipment, and potential support requirements, and repair contractors and timescales (including mobilisation)


associated with various repair options. An Emergency Pipeline Repair Reference Manual – Subsea Pipelines will be used to inform required repair work.

The Barossa Project PIMP identifies that non-urgent repairs can be made at opportune times (e.g., during facility shutdowns). Non-urgent repairs are subject to the Operational Risk Management Procedure. If a change to the Activity to that described in this GEP Coastal Waters OEMP is required as an outcome of the risk assessment, then the Santos Management of Change (MOC) Procedure will be applied.

Maintenance and repair of the pipeline in Coastal Waters may consist of the following activities:

- Excavation of the sediment around the Barossa GEP to establish the extent of any damage, and to provide appropriate access for repairs to be carried out. Typically, a jetting tool or air-lifting tool operated by a ROV or divers would be used to remove sand and rocks from around the Barossa GEP and to excavate beneath the pipeline, as required;
- Removal of concrete weight coating (CWC) and corrosion coating by ROV, divers, or specially designed CWC removal tools, using high pressure water jets or hydraulic saws;
- Free span correction using water jetting, or placement of gravel/rocks, concrete mattresses or grout bags using a ROV from a support vessel;
- In the event of a minor repair (where positive pressure has been maintained within the Barossa GEP and there has not been an ingress of seawater), a clamp repair (diver or diverless clamp) may be implemented. If a minor repair is required, the seabed around the Barossa GEP may need to be excavated to enable access for the clamp. Alternatively, the pipeline may be lifted, and grout-bags placed underneath. The pipeline may also be brought to the surface for the clamp repair.
- In the unlikely event of a major loss of containment where the contents of the line have been released and seawater ingress has occurred, removal of seawater and debris, such as marine growth and sand, that may exacerbate Barossa GEP corrosion is required. This would likely involve pushing the ingress seawater out of the Barossa GEP at the location of the breach, by pig trains being sent from both the DLNG and offshore ends of the pipeline at the FPSO to meet near the breach and force the pipeline contents and debris out of the pipeline. Pig launching and receiving is out of scope of this OEMP. The pig train would be pushed by either an inert gas (e.g., nitrogen), UV treated seawater, or chemically treated seawater (dosed with biocide, corrosion inhibitor and an oxygen scavenger); and
- In the case where a major pipeline repair is required, or an in-line inspection tool is stuck, the damaged/impacted section will be removed using divers or ROV cutting tools, and a prefabricated pipe repair spool would be installed, typically connected to the pipeline by use of either diver or diverless connectors. Once the pipeline has been repaired a hydrotest will be performed, this will use treated seawater injected at DLNG towards the FPSO, with the hydrotest water being discharged at the FPSO (which is out of scope of this OEMP) when the GEP is depressurised. The GEP is then dewatered and reconditioned ready for the reintroduction of hydrocarbons. Dewatering is performed using a series of pigs separated by inhibited MEG and pushed with dry nitrogen, the pigs would be pushed from DLNG with the treated seawater and MEG being discharged at the FPSO (which is out of scope of this OEMP and is addressed in the Barossa Production Operations Environment Plan). The pipeline will then be filled with nitrogen in preparation for recommissioning. Treated seawater (the same treatment would apply to MEG) is seawater conditioned with a hydrotest mixture such as Hydrosure, Roemex Hydro 3 or similar product that is ranked as Gold through the Offshore Chemical Notification Scheme (OCNS) or has a pseudo-ranking of Gold based on aquatic toxicity, biodegradation and bioaccumulation data. Treated seawater is typically a mixture of biocides (to prevent biofouling on the internal surfaces), an oxygen scavenger and corrosion inhibitor (to control corrosion of the pipeline) and sometimes a dye (allows for leaks to be detected through visual inspections).

#### 2.5.3 In-line inspection contingency

In-line inspection (ILI) via pigging for the GEP is not a planned activity but may be undertaken in the case where it is suspected that the pipeline integrity has been compromised due to a pipeline feed gas dew point excursion or other unplanned event.

Internal inspection of the GEP is performed using an ILI tool (tool) equipped with Magnetic Flux Leakage (MFL) measurement technology (or other alternative technology) capable of measuring the GEP wall thickness and detecting significant anomalies. This tool is used to inspect the GEP by pushing the tool between the riser base assembly at the Barossa Field to DLNG. Deployment and retrieval of the tool is out of scope of this GEP Coastal Waters OEMP; however the tool will traverse through the NT Coastal Waters section of the GEP. Bi-directional cleaning/gauging tools are used as part of the ILI campaign prior to sending the ILI tool, to check for internal restrictions within the pipeline and to clean the pipeline.



The tool will be deployed via a temporary subsea pig launcher, which will be temporarily installed and connected to the riser base.

It is expected that up to 5 m<sup>3</sup> of treated seawater or treated MEG would be used per run. The tool and any treated seawater / MEG will be received at DLNG (which is out of scope of this GEP Coastal Waters OEMP). The connection and disconnection of the tool launcher may result in the release of up to nominally 125 m<sup>3</sup> of MEG at the riser base assembly at the Barossa Field (which is out of scope of this GEP Coastal Waters OEMP and is addressed in the Barossa Production Operations Environment Plan).

#### 2.5.4 Stuck in-line inspection tool contingency

If an in-line inspection tool gets stuck or damaged in the GEP during the ILI campaign, it will be:

- a) forced out using a high seal pig or a train of high seal pigs (which may be separated by treated seawater or MEG), resulting in a discharge at the DLNG; or
- b) pushed back to the subsea receiver at the riser base (reverse flow), which may result in an unplanned event of flaring gas at Barossa FPSO; or
- c) removed by sectional removal/ replacement of the GEP (Section 2.5.2).

Discharges associated with the DLNG and Barossa FPSO are out of scope of this GEP Coastal Waters OEMP.

Activity	Equipment	Associated Discharges
<ul> <li>Barossa GEP inspection and cleaning typically includes:</li> <li>Inspection of subsea infrastructure</li> <li>Post-cyclone survey (if required)</li> <li>Marine growth removed during cleaning (refer Section 2.5.1.3)</li> <li>Non-contact and contact CP checks, including field gradient survey technology and laser scanning</li> <li>Non-destructive testing</li> <li>Water jetting</li> </ul>	<ul> <li>Vessel(s)</li> <li>ROV and tooling</li> <li>AUV</li> <li>Remotely operated USV (typically with ROV)</li> <li>Mechanical brushing</li> <li>Ultra-high-definition phototropic equipment</li> <li>SSS</li> <li>MBES imaging sonar</li> <li>Water jetting equipment</li> </ul>	<ul> <li>Discharges released to the marine environment includes:</li> <li>In-situ marine growth (&lt;5 m<sup>3</sup> per campaign) removed from pipeline</li> <li>No gas or liquid discharges</li> </ul>
<ul> <li>Stabilisation of subsea infrastructure with use of remediations typically includes:</li> <li>Placement of gravel/rocks, grout bags and/or concrete mattresses on specific areas of the subsea infrastructure showing scour or movement;</li> <li>Localised seabed excavation around structures</li> </ul>	<ul> <li>Vessel(s)</li> <li>Gravel/rocks and grout bags, mattress</li> <li>ROV and tooling</li> <li>Water jetting equipment</li> </ul>	No gas or liquid discharges Seabed disturbance / sand movement with no associated discharges.

Table 2-5: Summary of inspection, monitoring, maintenance and repair activities and equipment, and associated discharges

Activity	Equipment	Associated Discharges
<ul> <li>Barossa GEP repair and replacement (contingency activity) typically includes:</li> <li>Repair or replacement of a section of the Barossa GEP</li> <li>Placement of infrastructure on the seabed.</li> <li>ROV and tooling – concrete removal, coating removal, weld seam removal, marine growth removal, end preparation, water jetting, mechanical brushing, non contact and contact CP checks, non-destructive testing</li> <li>ROV survey either side and nearby seabed</li> <li>Cutting of the damaged infrastructure using ROV operated tooling</li> <li>Installation and connection of ROV operated pipe clamps or joins</li> <li>Installation of the new spool piece</li> <li>Stabilisation and span rectifications, such as sediment relocation, gravel and grout bags and concrete mattresses use, if required</li> <li>Diver assistance.</li> <li>For large repairs, a pipelay vessel may be required.</li> </ul>	<ul> <li>Vessel(s)</li> <li>ROV and tooling</li> <li>Pipe lift frames and installation aids</li> <li>SSS</li> <li>MBES imaging sonar</li> <li>Remotely operated USV (typically with ROV)</li> <li>Chemical use (e.g., Non-toxic dye, residual hydrocarbon production fluids)</li> </ul>	<ul> <li>Discharges to the marine environment include:</li> <li>In-situ marine growth (&lt;5 m<sup>3</sup> per campaign), concrete coating, anti-corrosion coating, metal swarf/fillings removed from pipeline</li> <li>Gas and liquid discharges may include: <ul> <li>Minor release of hydrocarbon / residual hydrocarbon (&lt;1 L per campaign)</li> <li>Inert gas (&lt;1 L per campaign)</li> <li>Non-toxic dye (&lt;5 L per operation)</li> </ul> </li> </ul>
<ul> <li>Installation of temporary or permanent subsea instruments and retrieval of data from subsea monitoring instrumentation typically includes:</li> <li>ROV installation of temporary or permanent instruments to measure e.g., pipe displacement and strain, or vibration.</li> <li>ROV deployed to interrogate and retrieve data from instruments</li> <li>ROV deployed to retrieve subsea instruments, recover data and re-deploy</li> <li>ROV deployed for seabed sampling</li> <li>Marine growth removal (water jetting or brushing)</li> <li>Mechanical brushing</li> <li>Pigging (contingency)</li> </ul>	<ul> <li>Vessel(s)</li> <li>ROV and tooling</li> <li>Pig train (contingency)</li> </ul>	<ul> <li>Discharges to the marine environment includes:</li> <li>In-situ marine growth (&lt;5 m<sup>3</sup> per campaign) removed from pipeline</li> <li>Removal of untreated seawater and debris (marine growth and sand) in the event of seawater ingress removed by pig train (volumes unknown)</li> <li>No gas discharges.</li> </ul>



#### 2.5.5 Environmental monitoring activities

Environmental monitoring activities, such as sampling of seabed material (i.e. sediment) or investigation/sampling of biological material (i.e., marine growth) for environmental studies may be undertaken to increase Santos' understanding of the environmental impacts and risks as part of planning activities for decommissioning. Sediment sampling may be undertaken along the Barossa GEP to characterise sediment and understand baseline conditions after a loss of containment. Analytes could include total organic carbon, particle sizes, major cations, trace metals, and types of bacteria. This activity will be performed using routine sampling techniques and equipment from a vessel, such as using a dual van Veen grab sampler shown in Figure 2-2 below.



Figure 2-2: Dual van Veen grab sampler



### 2.6 Vessel and helicopter operations

A range of vessels required to carry out IMMR activities and environmental monitoring activities could be present and operate within the Operational Area under this GEP Coastal Waters OEMP. Vessels are selected and onboarded in accordance with procedures to ensure contracted vessels are operated, maintained and crewed in accordance with industry standards (for example, Marine Orders) and regulatory requirements (this GEP Coastal Waters OEMP). All required audits and inspections will assess compliance with the laws of the international shipping industry, which include safety and environmental management requirements, and maritime legislation including International Convention for the Prevention of Pollution from Ships 1973 as modified by the Protocol of 1987 (MARPOL) and other International Maritime Organisation (IMO) standards.

The vessels will display navigational lighting and external lighting, as required for safe operations. Lighting levels will be determined primarily by operational safety and navigational requirements under relevant legislation, specifically the *Navigation Act 2012* and relevant Marine Orders. The vessels will be lit to maintain operational safety on a 24-hour basis.

#### 2.6.1 Vessel types

Vessel based IMMR activities are preferentially undertaken from May to November, outside of cyclone season, to minimise or avoid operational disruptions. However, depending on maintenance requirements, maintenance activities could occur at any time during the year. Smaller vessels will be used for inspections and larger vessels for undertaking pipeline repairs (if required).

Vessels used for inspection activities are expected to range between 30 m and 130 m in length. The vessel type and specifications will depend on availability and specific activity requirements. These vessels typically use a dynamic positioning (DP) system to allow manoeuvrability and to avoid anchoring in proximity to the Barossa GEP. The vessels may be sourced locally or from an international location. Pipeline inspections and freespan rectification activities would typically be undertaken using vessels such as the Bhagwan Dryden (57m long and 40 berths) (Figure 2-3). Environmental monitoring activities would typically be undertaken using smaller vessels.



**Figure 2-3: Bhagwan Dryden, as an example of a typical IMMR vessel** Anticipated, typical IMMR vessel parameters are provided in Table 2-6.

#### Table 2-6: Typical IMMR vessel parameters

Parameter	Description
Draft (typical)	3.25 m
Gross tonnage	1,475 GT
Hull	Twin keel catamaran steel hull
Fuel type	Marine diesel
Total fuel volume (approximately)	130 m <sup>3</sup>
Volume of largest fuel tank (approximately)	31 m <sup>3</sup>
Persons on board	40

The Sapura Constructor is an example of a typical vessel that may be used for undertaking pipeline repair, which is a 117 m Class DNV vessel (Figure 2-4). The Sapura Constructor has berths for up to 120 persons and 15 divers onboard. It would be supported by up to two vessels of the same or lesser class – including a supply/support vessel and/or a hyperbaric rescue vessel. Refer to Table 2-7 for typical IMMR repair vessel parameters.

Bunkering of the repair vessels may take place either at sea or in port. Vessels will use marine diesel oil (MDO) or marine gas oil (MGO).



Figure 2-4: Sapura Constructor, as an example of a IMMR repair vessel for pipeline repairs

Anticipated, typical IMMR repair vessel parameters are provided in Table 2-7.

Table 2-7: Typical IMMR repair vessel parameters

Parameter	Description
Draft (typical)	7.15 m (typical)
Gross tonnage	6,200 t
Hull	Steel hull

Parameter	Description
Fuel type	Marine diesel
Total fuel volume (@ 90% capacity)	1,006 m <sup>3</sup>
Volume of largest fuel tank	185 m <sup>3</sup>
Persons on board	120 and 15 divers

#### 2.6.2 Uncrewed surface vessels

Remotely operated uncrewed surface vessels (USV) may be used for launching electrical ROVs for undertaking inspections. The USV would be operated by a Vessel Master from a remote operations centre and a support vessel would be available in Darwin should any assistance be required. Figure 2-5 shows a representative large USV, Reach Remote 1 (23.9 m long, 8 m wide). They are typically fitted with radars, an emergency anchor, loud speaker, night vision, 360° camera and VHF radio. Refer to Table 2-8 for typical large USV parameters.



Figure 2-5: Reach Remote, an example of a representative USV

#### Table 2-8: Typical large USV parameters

Parameter	Description
Draft (typical)	5.5 m
Gross tonnage	230 t
Volume of largest fuel tank	74 m <sup>3</sup>

#### 2.6.3 Helicopter Support

Helicopters for crew changes usually operate from Darwin. Depending on the specific scope and operational needs, crew changes may occur during the IMMR scopes outlined in Section 2.5. If a crew change were needed, it is assumed it would be a weekly occurrence and may or may not occur in the OA for this GEP Coastal Waters OEMP.

#### 2.6.4 Remotely operated vehicle and autonomous underwater vehicle operations

Remotely operated vehicles (ROVs) or autonomous underwater vehicles (AUVs) may be used to support IMMR and environmental monitoring activities in the OA. ROVs or AUVs can be fitted with various tools and camera systems that can be used to capture permanent records of the operations and immediate surrounding environment. The size of the vessel required to deploy an ROV or AUV, depends on the size of the ROV or AUV and the



associated launch and recovery system. The ROV or AUV is typically deployed from a vessel using a crane or an A-frame and is recovered using a winch or net. In some instances, the ROV may be placed on the seabed.

### 2.7 Chemical Selection and Use

Chemicals may be discharged by vessels during IMMR activities as described in Table 2-5. A risk-based approach to selecting chemical products ranked under the Offshore Chemical Notification Scheme (OCNS) is applied for those chemicals used and discharged to the marine environment. This scheme lists and ranks all chemicals used in exploring, exploiting and associated offshore processing of petroleum on the United Kingdom Continental Shelf.

Chemicals are ranked according to their calculated hazard quotients (HQ) by the chemical hazard assessment and risk management (CHARM) mathematical model (CHARM Implementation Network, 2005). The HQ is converted to a colour banding—gold and silver colour bands represent the least environmentally hazardous chemicals (Table 2-9).

Minimum HQ value	Maximum HQ value	Colour banding	Hazard
>0	<1	Gold	Lowest
≥1	<30	Silver	
≥30	<100	White	
≥100	<300	Blue	
≥300	<1,000	Orange	
≥1,000		Purple	Highest

#### Table 2-9: OCNS Chemical hazard and risk management hazard quotient and ranking

Chemicals not amenable to the CHARM model (i.e. inorganic substances, hydraulic fluids or chemicals used only in pipelines) are assigned an OCNS grouping based on the worst-case ecotoxicity data, with Group E and D representing the least hazard potential (Table 2-10).

Initial grouping	Α	В	С	D	E
Result for aquatic-toxicity data (ppm)	<1	≥1 to 10	>10 to 100	>100 to 1000	>1,000
Result for sediment-toxicity data (ppm)	<10	≥10 to 100	>100 to 1000	>1000 to 10,000	>10,000

Note: Aquatic toxicity refers to the Skeletonema costatum  $EC_{50}$ , Acartia tonsa  $LC_{50}$ , and Scophthalmus maximus (juvenile turbot)  $LC_{50}$  toxicity tests. Sediment toxicity refers to the Corophium volutator  $LC_{50}$  test.

Source: CEFAS Standard Procedure 2022, OCNS 011 NL Protocol Part 1: Core Elements.

Subsea chemicals accepted are CHARM-ranked Gold/Silver, or non-CHARM-ranked E/D chemicals for use and discharge without a detailed environmental risk assessment. The same applies to chemicals that are on the Convention for the Protection of the Marine Environment of the North-East Atlantic (OSPAR) Pose Little or No Risk to the Environment (PLONOR) List. The PLONOR List, agreed upon by the OSPAR Convention, contains a list of substances that will pose little or no risk to the environment in offshore waters. If chemicals do not have a CHARM/non-CHARM ranking under the OCNS, chemicals are assigned a pseudo-ranking based on the available aquatic toxicity, biodegradation and bioaccumulation data (see Sections 2.7.1 to 2.7.3). A risk assessment, informed by the pseudo-ranking, is conducted for non-OCNS listed chemicals to provide technical justification for their use and to show their use and assessment for environmental acceptability for discharge to the marine environment.

#### 2.7.1 Ecotoxicity assessment

Table 2-10 and Table 2-11 provide guidance in assessing the ecotoxicity of chemicals during the investigation of potential alternatives. Table 2-10 is used by the Centre for Environment, Fisheries and Aquaculture Science (CEFAS) to group a chemical based on ecotoxicity results, 'A' representing the highest toxicity and risk to environment and 'E' the lowest. Table 2-11 shows classifications and categories of toxicity against aquatic toxicity results.



#### Table 2-11: Aquatic species toxicity grouping

Category	Species	LC <sub>50</sub> and EC <sub>50</sub> criteria	
Category Acute 1	Fish	LC <sub>50</sub> (96-hr) ≤ 1 mg/L	
Hazard statement – Very	Crustacea	ELC₅₀ (96-hr) ≤ 1 mg/L	
	Algae, other aquatic plant species	ErC₅₀ (72 or 96-hr) ≤ 1 mg/L	
Category Acute 2 – Hazard statement – Toxic to aquatic life	Fish	LC₅₀ (96-hr) >1 mg/L but ≤ 10 mg/L	
	Crustacea	EC₅₀ (48-hr) >1 mg/L but ≤10 mg/L	
	Algae, other aquatic plant species	ErC₅₀ (72 or 96-hr) >1 mg/L but ≤ 10 mg/L	
Category Acute 3 – Hazard statement – Harmful to aquatic life	Fish	LC <sub>50</sub> (96-hr) >10 mg/L but ≤ 100 mg/L	
	Crustacea	EC₅₀ (48-hr) >10 mg/L but ≤ 100 mg/L	
	Algae, other aquatic plant species	$ErC_{50}$ (72 or 96-hr) >10 mg/L but ≤ 100 mg/L	

Source: United Nations (2023) Globally Harmonized System of Classification and Labelling of Chemicals (GHS).

#### 2.7.2 Biodegradation assessment

The biodegradation of chemicals is assessed using the CEFAS biodegradation criteria, which aligns with the categorisation outlined in the United Nations Globally harmonised system of classification and labelling of chemicals, Annex 9 Guidance on Hazards to the Aquatic Environment (UN, 2023). The categorisation is used as a guide when investigating potential chemical alternatives. The preference is to select readily biodegradable chemicals.

CEFAS categorises biodegradation into the groups of:

- readily biodegradable: results of >60% biodegradation in 28 days (OECD 306, 301B -F method), >70% in 28 days (OECD 301A, 301E) to an OSPAR harmonised offshore chemical notification format (HOCNF)accepted ready biodegradation protocol
- Inherently biodegradable: results of >20% and <60% (<70%) to an OSPAR HOCNF-accepted ready biodegradation protocol
- poorly biodegradable: results from OSPAR HOCNF-accepted ready biodegradation protocol or inherent biodegradation protocol are <20%, or half-life derived from aquatic simulation tests indicate persistence.

#### 2.7.3 Bioaccumulation assessment

The bioaccumulation of chemicals is assessed using the CEFAS bioaccumulation criteria, which aligns with the categorisation outlined in the United Nations Globally harmonised system of classification and labelling of chemicals, Annex 9 Guidance on Hazards to the Aquatic Environment (UN, 2023). The preference is to select chemicals that are not bioaccumulative.

The guides used by CEFAS are:

- non-bioaccumulative: Log Pow <3, or BCF ≤100, the molecular weight is ≥700
- bioaccumulative: Log Pow ≥3, or BCF >100, the molecular weight is <700, or if the conclusion of a weightof-evidence expert judgement under OSPAR Agreement 2008-5 is negative.

# 3. **Description of the environment**

#### OPGGS(E)R 2023 Requirements

#### Section 21. Environmental assessment

#### Description of the environment

(2) The environment plan must:

- a. describe the existing environment that may be affected by the activity; and
- b. include details of the relevant values and sensitivities (if any) of that environment.
- Note: definition of environment in section 5 includes its social, economic and cultural features.

(3) Without limiting paragraph (2)(b), relevant values and sensitivities may include any of the following:

- a. the world heritage values of a declared World Heritage property;
- b. the National Heritage values of a National Heritage place;
- c. the ecological character of a declared Ramsar wetland;
- d. the presence of a listed threatened species or listed threatened ecological community;
- e. the presence of a listed migratory species;
- f. any values and sensitivities that exist in, or in relation to, part or all of:
  - i. a Commonwealth marine area; or
    - ii. Commonwealth land.

This section describes the key physical, biological, socioeconomic and cultural features (values and sensitivities) of the existing environment that may be affected by the Activity. The description of the environment applies to the OA, and any areas surrounding the OA that may be affected by the Activity. In this GEP Coastal Waters OEMP the area that may be affected by the impacts and risks of the Activity is described as the environment that may be affected (EMBA), or in the case of a hydrocarbon spill, low exposure value area (LEVA) (which also defines the modelled EMBA) and moderate exposure value area (MEVA). The low and moderate exposure values are listed in Table 3-3 and shown in Figure 3-1.

This section also includes details of the values and sensitivities pertaining to the EMBA. Detailed descriptions of these values and sensitivities are provided in the sections below. The results were informed by the Environment Protection and Biodiversity Conservation Act 1999 (EPBC Act) protected matters reports (all matters potentially relevant to section 21 (3)), stated values in the Marine Bioregional Plans for the North Marine Region (NMR) (CoA, 2012), EPBC Act protected matters reports (Appendix C) and information obtained through consultation and the Barossa environmental studies detailed in Table 3 1.

These values and sensitivities have been identified for the purposes of environmental assessment, identifying potential environmental consequences, and developing spill response plans. More information about the reasons why these exposure values have been included and how their application in defining areas relates to impact and risk assessment and spill response planning is provided in Sections 6 and 7

#### Environmental studies

Extensive environmental and socioeconomic studies have been undertaken to characterise the existing environment for the Barossa Gas Project. Table 3-1 summarises the Barossa marine studies program which involved the collection of detailed baseline data to capture seasonal variability in the region, as well as supplementary surveys and desktop modelling studies to contribute to the understanding of the baseline environment. Santos refers to this description as information previously given under section 56(1) of the OPGGS(E) Regulations. Further detail and copies of the earlier studies are provided in Section 5, Appendix C and Appendix D of the OPP (ConocoPhillips, 2018) as information previously given under section 56(1) of the OPGGS(E) Regulations.

Table 3-2 summarises additional environmental, socioeconomic and cultural features studies undertaken for the Barossa Gas Project to inform the understanding of the environment (including socioeconomic and cultural features) after the initial Barossa marine studies program, including those done specifically for the DPD Project to inform this GEP Coastal Waters OEMP.



#### Table 3-1: Summary of the Barossa marine studies

Study type	Description of study	Reference
Field-based studies		
Metocean data collection	Collection of metocean data on the surface and through the water column from July 2014 to March 2015, within and near the Barossa field, e.g., current, conductivity, wave and wind data.	Fugro, 2015
Underwater noise survey	Collection of baseline data on ambient underwater noise (physical, biological and anthropogenic sources) at 3 locations from July 2014 to July 2015 near the Barossa development and surrounding areas.	JASCO, 2016
Benthic habitat survey	Collection of baseline data to characterise topographic features, benthic habitats and macrofaunal communities near the Barossa field location and surrounding areas, including around Evans Shoal, Tassie Shoal and Lynedoch Bank by using a specialised ROV.	Jacobs, 2016

#### Table 3-2: Summary of Barossa additional studies

Study type	Description of study	Reference
Shoals and shelf survey 2015: benthic habitats and fish communities	A seabed biodiversity survey of 3 shoals to the west of the Barossa field (Evans Shoal, Tassie Shoal and Blackwood Shoal) and 2 mid-continental shelf regions relevant to the pipeline route corridor. The Australian Institute of Marine Science (AIMS) performed the survey in September/October 2015, which involved characterisation of the seabed habitats, associated biota and fish communities (shoals only).	Heyward et al., 2017
Oceanic Shoals Marine Park benthic habitat and fish diversity assessment	An AIMS seabed and fish biodiversity survey conducted in September and October 2017. The survey focused on 6 key sites inside and outside of the Oceanic Shoals Marine Park, including in the Habitat Protection Zone, and Shepparton Shoal. The objective was to use this new data to update the predictive habitat model and statistically compare the proportion and spatial diversity of habitats within and outside the Oceanic Shoals Marine Park.	Radford et al., 2019
Tiwi Islands sensitivity mapping study	Collection of data on environmental, social, cultural and economic sensitivities for the Tiwi Islands. A desktop review of available data (spatial datasets) was followed by workshops with Traditional Owners to identify cultural and environmental sensitivities along the coast of the Tiwi Islands.	Jacobs, 2019
Maritime heritage assessment	A maritime archaeological assessment along the DPD route to identify potential maritime archaeological sites which are defined as wrecks (ship or aircraft) and associated material, dumped material, maritime infrastructure, and associated deposits on or under the seabed below the highest astronomical tide.	Cosmos Archaeology, 2022
Barossa pipelay light modelling	Light modelling assessment of the proposed pipelay and construction vessels, including cumulative impacts to predict the potential light impacts to turtle nesting habitat on the Tiwi Islands and hatchling behaviours.	Pendoley, 2022
Barossa pipelay Darwin Harbour lighting technical note	Desktop assessment of presence and significance of marine turtle nesting activity on beaches surrounding Darwin Harbour and likely level of impact from activity vessel lighting to flatback turtles.	Pendoley, 2022a
Benthic survey for Barossa DPD route	Collection of baseline information on the benthic habitats, sediment composition (including contaminant concentrations), macroinvertebrate (infaunal) assemblages, and water quality along the DPD route.	RPS, 2023
Desktop study Tiwi turtle programs	This desktop report reviewed publicly available literature and research relating to marine turtle activity occurring on, and around, the Tiwi Islands. A total of 19 satellite telemetry studies between 1994-2023 tracked turtles passing through or foraging in waters near the Tiwi Islands.	Pendoley, 2023
Underwater Cultural Heritage (UCH) assessment of the route of the	An independent expert assessment by Dr Brendan Corrigan for the purpose of identifying UCH places along the route of the Barossa GEP west and northwest of the Tiwi Islands ('Corrigan Report').	Corrigan, 2023
Barossa GEP route	reports on archaeology and sedimentology along the GEP route conducted by Wessex Archaeology and Dr Posamentier; and the EDO GEP reports.	
Wessex Archaeology	Archaeological report focusing on the Late Pleistocene and Holocene depositional and erosional history of the Arafura Sea along the GEP corridor.	Wessex Archaeology, 2023



Study type	Description of study	Reference
First Nations spiritual and cultural values in relation to the Darwin Pipeline Duplication Project	An independent expert assessment by Dr Brendan Corrigan for the purpose of identifying spiritual and cultural values relevant to the construction of the Darwin Pipeline Duplication pipeline route (including the 8.26 km route in NT Coastal Waters). The document is available on the Santos website at <a href="https://www.santos.com/wp-content/uploads/2024/09/First-Nations-spiritual-and-cultural-values-in-relation-to-the-Darwin-Pipeline-Duplication-Project.pdf">https://www.santos.com/wp-content/uploads/2024/09/First-Nations-spiritual-and-cultural-values-in-relation-to-the-Darwin-Pipeline-Duplication-Project.pdf</a> .	Corrigan, 2024
First Nations archaeological desktop assessment summary report	An independent expert assessment by Dr Jodie Benton for the purpose of identifying any underwater cultural heritage places along the DPD pipeline route (including the 8.26 km route in NT Coastal Waters). The document is available on the Santos website at <u>https://www.santos.com/wp-content/uploads/2024/09/Desktop-First-Nations-Archaeological-Assessment-Summary-Report-Darwin-Pipeline-Duplication-Project-KP0-31.pdf</u>	Ozark, 2024
Treated seawater discharge dispersion modelling	Treated seawater dispersion modelling report including contingency pipeline dewatering within NT Coastal Waters.	RPS, 2024a
Hydrocarbon spill modelling for Barossa GEP spill scenarios	Hydrocarbon spill scenario modelling for spill scenarios along the Barossa GEP route in NT waters, including a spill scenario in NT Coastal Waters at the NT / Commonwealth waters boundary	RPS, 2024b

### 3.1 Environment that may be affected (EMBA)

This section describes the key physical, ecological, socio-economic and cultural features of the existing environment that may be affected by the Activity and covers the OA and any areas surrounding the OA that may be affected by the Activity.

In this document the area that may be affected by the impacts and risks of the Activity is described as the EMBA, or in the case of a hydrocarbon spill, LEVA (which also defines the modelled EMBA), MEVA and high exposure value area (HEVA). The EMBA and MEVA are shown in Figure 3-1.

#### 3.1.1 Determining the EMBA

Stochastic hydrocarbon dispersion and fate modelling has been applied to worst case credible hydrocarbon spill scenarios relevant to the Activity to inform the spatial extent of the EMBA (defined conservatively by the LEVA) (Section 7.6). Areas potentially contacted by hydrocarbons were determined using stochastic modelling which overlayed hundreds of individual hypothetical spill simulations from a hydrocarbon spill into a single map, with each simulation subject to a different set of metocean conditions drawn from historical records. Stochastic modelling compensates for the uncertainty associated with any single hydrocarbon spill event such that risk assessment and spill response planning are more robust and conservative by covering a wide range of possible scenarios.

Modelling considers key physical and chemical phases of hydrocarbons that pose differing environmental and socioeconomic risks, being surface, entrained, dissolved aromatic and shoreline accumulated hydrocarbons. Defining the areas that may be affected by spilled hydrocarbons depends on the concentrations of the hydrocarbons on the sea surface, in the water column and on the shoreline.

Hydrocarbon exposure threshold values defined by NOPSEMA (2019) for each of these phases were applied to the stochastic modelling outputs to determine the areas affected by the MEVA and the LEVA. The MEVA represents an area wherein contact with hydrocarbons may result in harmful impacts to biota, encompassing the maximum extent of biological impact. The LEVA represents the maximum extent of possible contact with hydrocarbons within the depth range between 0–10 m and reflects the range of socioeconomic considerations for spill response planning and scientific monitoring. For this reason, the LEVA has been used to define the modelled EMBA. Importantly, in terms of impacts to environmental values and sensitivities, the extent of a particular impact and risk may not be relevant to the full extent of the modelled EMBA, therefore, the MEVA is also referred to where relevant in this GEP Coastal Waters OEMP.

The worst-case release scenario identified as relevant to the Activity (see Section 7.6) is considered to be a release of up to 300 m<sup>3</sup> of MDO caused by a vessel collision rupturing a vessel fuel tank, as this represented the largest spatial extent of potential changes to ambient environment conditions. A 'best fit' line is drawn around the outermost limits of the low exposure value contours resulting in a highly conservative EMBA. The MEVA and EMBA are shown in Figure 3-1 and exposure values are provided in Table 3-3. Further information about the reasons why these exposure values have been selected and how their application in defining areas relates to impact and risk assessment and spill response planning is provided in Section 7.7, Section 7.6.5 and Section 7.6.6.



The EMBA is based on stochastic modelling using 300 spill simulations, using low exposure values (Table 3-3). It should be noted that the footprint of an actual spill event is more accurately represented by only one of the simulations from the stochastic modelling, resulting in a much smaller spatial footprint in the event of an actual spill. Modelling of a single simulation, representative of a single spill event, is termed deterministic modelling. This is discussed further in Section 7.6.6 and applied in the risk assessment where relevant.

The primary purpose of the EMBA is to assist with spill response planning and preparedness in the unlikely event of a hydrocarbon spill. The EMBA provides a conservative basis for assessing the range of potential socio-economic impacts and establishes a planning area for scientific monitoring during an unplanned spill event.

#### Table 3-3: Hydrocarbon exposure threshold values

Hydrocarbon phase	Exposure value			
	Low	Moderate	High	
Surface (g/m <sup>2</sup> )	1	10	50	
Shoreline oil accumulation (g/m <sup>2</sup> )	10	100	1,000	
Dissolved hydrocarbon (ppb)	10	50	400	
Entrained (ppb)	10	-N/A	100	



Figure 3-1: Location and extent of the EMBA and MEVA





### 3.2 Existing environment

This section summarises the existing environment that may be affected by the Activity and includes details of the particular values and sensitivities pertaining to the EMBA. A detailed description of the values and sensitivities of the GEP Coastal Waters OEMP was informed by:

- EPBC Act Protected Matters Search Tool (PMST) results (Appendix C),
- Stated values in the Marine Bioregional Plans for the North Marine Region (NMR) (CoA, 2012),
- Barossa Environmental Studies (Section 3),
- publicly available information (such as scientific literature, studies and government databases) and
- Information obtained through consultation.

For the purposes of the environmental assessment, identifying potential environmental consequences and developing spill response plans, the environmental values captured by the moderate hydrocarbon exposure threshold values defined by NOPSEMA (2019), representing the thresholds whereby harmful impacts to biota may result, are also identified within the area referred to as the MEVA in this section. More information about the reasons why these exposure values have been included and how their application in defining areas relates to impact and risk assessment and spill response planning is provided in Section 7.6.

#### 3.2.1 Geographical extent

The OA is located within NT Coastal Waters, approximately 80 km north-west of Darwin, NT, approximately 31.5 km south of the Tiwi Islands, NT and approximately 66.5 km south-east of the Oceanic Shoals Marine Park. The OA and EMBA (Section 2.2 and 3.1.1) are located in the NMR, which is characterised by a monsoonal climate, complex weather patterns and currents driven predominantly by strong winds and tides (CoA, 2012). Key physical characteristics of the NMR that are relevant to the EMBA include:

- A wide continental shelf, with water depths averaging less than 70 m and ranging from approximately 10 m to a maximum known depth of 357 m
- Van Diemen Rise, which forms part of a key ecological feature (KEF) (Section 3.2.12) and includes a range of geomorphic features, such as shelves, shoals, banks, terraces and valleys
- limestone pinnacles, which forms part of a key ecological feature (KEF)—Pinnacles of the Bonaparte Basin (Section 3.2.12), valued for hard substrate in an otherwise soft sediment environment and so are important for sessile species, and
- Cultural features including Sea Country (Section 3.2.15).

#### 3.2.1.1 **Provincial bioregions**

Based on the Integrated Marine and Coastal Regionalisation of Australia, version 4.0 (CoA, 2006), the provincial bioregions relevant to the EMBA and MEVA are outlined in Table 3-4 and shown in Figure 3-2. The OA and EMBA overlap the Northwest Shelf Transition Province, which is the second largest shelf bioregion and primarily features shelf and abyssal plain, and basins, with sandbanks to the west (CoA, 2005b). The EMBA also intersects the Northern Shelf Province, which is characterised as a gently sloping shelf, topped with a number of pinnacles at depths ranging from 5 m to 30 m and tidal eddies (CoA, 2012).

# Table 3-4: Integrated Marine and Coastal Regionalisation of Australia provincial bioregions relevant to the Activity

Bioregion	OA	MEVA	EMBA
Northern Shelf Province	X	X	$\checkmark$
Northwest Shelf Transition	$\checkmark$	$\checkmark$	$\checkmark$



Figure 3-2: IMCRA provincial bioregions and marine regions overlapping or proximal to the EMBA and MEVA



NORTHERN SHELF PROVINCE
nd EMBA MEVA GEP Coastal Waters Operational Area (OA) Barossa GEP - Commonwealth Barossa GEP - NT Coastal Waters Barossa GEP - NT Internal Waters Darwin Harbour region management boundary NT/Commonwealth waters boundary Territorial sea baseline <b>ne regions (DCCEEW)</b> North marine region North-west marine region <b>A Provincial bioregions</b> Northern Shelf Province Northwest Shelf Transition
Date: 20/12/2024 Scale: Map 1:2,500,000 @ A4 AU213015087_GEP_OEMP AU213015087_001_A5_GEP_EMBA_IBRA_v2 IBRA Australian Government Department of the Environment (2015)



#### 3.2.2 Physical environment

#### 3.2.2.1 Geomorphology

#### 3.2.2.1.1 *Formation and history*

About 550 to 160 million years ago, the northern and western parts of Australia formed part of the northern margin of Gondwana. About 300 million years ago, crustal stretching, rifting and breakup-initiated development of an extensive basin where sediments were deposited (Baker et al., 2008 in CoA, 2012). About 135 million years ago the continent broke up, resulting in the separation of greater India and Australia.

#### 3.2.2.1.2 Bathymetry and seabed features

Generally, the EMBA encompasses a wide continental shelf intersected by the carbonate bank and terrace system of the Van Diemen Rise KEF. Water depths within the majority of the EMBA range between 0 and 70 m, with a deep trench approximately 100 km wide to the west of the EMBA, ranging between 100 and 140 m deep (Figure 3-3). Recent surveys have shown that the seabed adjacent to the OA is generally flat, sandy, and featureless (RPS, 2023). Water depths in the OA range from 47 m at the TSB to 50 m at the NT/Commonwealth waters boundary.

The Carbonate bank and terrace system of the Van Diemen Rise KEF is of regional significance (Section 3.2.12.3). The feature consists of banks, terraces, channels, and valleys and the variability in water depth and substrate composition may contribute to the presence of unique ecosystems in the channels. The feature has enhanced biodiversity and productivity relative to surrounding areas and supports relatively high species diversity (DEWHA, 2012a).



Figure 3-3: Bathymetry and seabed features overlapping or proximal to the EMBA and MEVA





#### 3.2.3 Climate

Meteorological data for the region, recorded at the Bureau of Meteorology (BoM) weather station at Melville Island (the closest metrological station to the OA), shows a small seasonal variation in air temperatures. The mean maximum summer and winter air temperatures range between 33.9 °C in October and 31.3°C in July (BoM, 2025). The average tropical cyclone frequency for the Timor and Arafura seas region is one cyclone per year, which occur mostly between November and April (BoM, 2025).

Waters in the northern extent of the EMBA predominantly lie in the arid tropics. Monsoonal conditions usually occur from October to March (wet season), with cooler and drier conditions prevailing from April to September (dry season) (CoA, 2012).

#### 3.2.4 Oceanography

#### 3.2.4.1 Regional current system

Large-scale currents of the Timor and Arafura seas are dominated by the Indonesian Throughflow current system (Figure 3-4). The Indonesian Throughflow brings warm, low-salinity oligotrophic waters through a complex system of currents, linking the Pacific and Indian oceans via the Indonesian Archipelago (DSD, 2010). The strength of the system fluctuates seasonally, reaching maximum strength during the south-east monsoon, and weakening during the north-west monsoon.

The Holloway Current (Figure 3-4), a relatively narrow boundary current that flows along the north-west shelf of Australia between 100 and 200 m depth, also influences the seas in the EMBA. The direction of the current changes seasonally with the monsoon, flowing towards the north-east in summer and the south-west in winter (Fugro, 2015).



Figure 3-4: Surface currents proximal to the EMBA



#### 3.2.4.2 Currents and tides

Water movement within the EMBA is influenced by wind and tidal activity and less by ocean currents. Smaller-scale surface currents reflect seasonal wind activity, flowing easterly to north easterly during the wet season and west to south-west during the dry season (Heyward et al., 1997).

Predicted average monthly surface current speeds are approximately 0.4 m/s within Beagle Bay just outside of Darwin Harbour (outside the Darwin Harbour Marine Management Area) and slightly less within Darwin Harbour (0.33 to 0.36 m/s in the mid-harbour) (RPS, 2024b). Predicted monthly maximum current speeds in these areas exceed 1 m/s (RPS, 2024b).

Tidal activity is typically dominated by semi-diurnal tides, with two daily high tides and two daily low tides. Tidal amplitude varies with location and distance offshore; in the Tiwi region it varies from 2 m offshore to 4 m inshore, 4–6 m in Van Diemen Gulf, while in the Beagle and Bonaparte Gulf regions the tidal range is ~6–8 m and 2–4 m respectively (IMCRATG, 1998).

#### 3.2.4.3 Waves

Wave movements within the EMBA are expected to be composed of locally generated sea waves in response to local wind activity and swell waves created by distant wind activity. Wave height is generally between 0.6 and 0.8 m, coming from the west in the wet season and from the east in the dry season.

Cyclones and tropical storms can greatly increase wave heights by up to 8 m in the outer Timor Sea during the cyclone season (Przeslawski et al., 2011).

The wave climate offshore of the north-west shelf of Australia is normally dominated by the passage of storms over the southern Indian Ocean (Fugro, 2015). However, between October and March, the wave climate is controlled by the south-westerly monsoon winds. This combination of wind directions may lead to concurrent swells approaching from different directions. The sea wave climate also reflects the seasonal wind regime, with waves predominantly from the south-west in summer and from the east in winter.

#### 3.2.4.4 Temperature

Surface water temperatures in the Barossa offshore development area were recorded as generally ranging between 27 °C and 30 °C, while temperatures in the upper water column of the Barossa offshore development area were recorded as reaching a maximum of 30.9 °C in summer and a minimum of 24.7 °C in spring (Fugro 2015). Mean temperatures ranged from 28.1 °C at 34 m below MSL (summer) to 12.6 °C at 253 m below MSL (summer) (Fugro, 2015). Water temperatures within the EMBA are expected to be broadly within the ranges of those observed in the development area.

#### 3.2.4.5 Water Quality

#### 3.2.4.5.1 *Regional overview*

Water quality in the Northwest Shelf Transition provincial bioregion is influenced primarily by the Indonesian Throughflow, which transports warm, low salinity, oligotrophic (low nutrient) waters into the region from Indonesia (CoA, 2012). Offshore waters are generally clear, with the euphotic zone extending down to 100 m across the shelf (CoA, 2012). Localised upwellings of cooler and higher-nutrient content waters occur throughout the Northwest Shelf Transition provincial bioregion; however, the influence and extent of these upwellings are mostly unknown (CoA, 2012).

#### 3.2.4.5.2 Offshore from Darwin Harbour

In 2021, water sampling and analysis along the DPD route in the offshore NT waters of was undertaken (RPS, 2023), which include the OA for this Activity. Concentrations of three metals in water samples were detected above Australian and New Zealand Guidelines (ANZG; 2018) default guideline values (DGVs) (for slightly to moderately disturbed marine offshore ecosystems, at the 95% species protection level). Copper concentrations in samples from three sites at the western end of the offshore pipeline route were above the DGV; one of these exceedances was much higher than the DVG with the other two only slightly greater than the DVG, therefore it is likely an outlier and indicative of a potential contaminant (RPS, 2023).

#### 3.2.4.6 Sediment quality

Large extents of soft substate are interspersed between the described isolated features of the Northwest Shelf Transition provincial bioregion (e.g., ridges and reef) (Przeslawski et al., 2011). Within the offshore NMR, dominant sediment types include sandy silts, very soft-to-soft silts, and very loose-to-loose silty sands with variable shell content and sand fraction ranging from coarse to fine (CoA, 2012).



#### 3.2.5 Air quality

In the offshore areas of the OA and EMBA, there are no permanent sources of air pollution. Consequently, the air quality in this region of the EMBA is expected to be pristine, experiencing only localised and temporary anthropogenic influences (e.g. energy industry and shipping activity).

#### 3.2.6 Shoals and banks

A number of shoals and banks occur within the EMBA (for details of named shoals and banks see Table 3-5 and Figure 3-5 noting that there are also numerous unnamed shoals and banks within the EMBA including one approximately 6 km from the OA). Few historic studies of these features exist, with most of the understanding of shoals and banks in the region derived from the 'big bank shoals' study (Heyward et al. 1997),b PTTEP surveys initiated in response to the Montara incident (Heyward et al., 2010; Heyward et al., 2011) and studies undertaken by the Australian Institute for Marine Science (AIMS) for the Barossa Development (Heyward et al., 2017; Radford et al., 2019).

The biological communities of the shoals and banks within the EMBA are well representative within the broader region (Heyward et al., 2017). AIMS' analysis of survey data showed that the most influential determinants of benthic community composition include depth and light intensity, substrate type and complexity, hydrodynamic environment and position on the continental shelf. 'Mid-shelf' locations, such as those within the EMBA, typically exhibit higher turbidity, resulting in greater light attenuation and the transition between primary producer dominated habitats (such as corals) to those featuring sessile filter feeders (e.g., sponges) is often observed at shallower depths. Consequently, coral reef communities are expected to only be associated with the shallower reefs, shoals and banks, particularly further away from the turbid coastal fringe where sponges, sea fans and to a lesser extent gorgonian soft corals are the dominant contributors to benthic communities (Heyward et al., 2017).

The shoals and banks within the EMBA are expected to support many common species, but to show variation in the abundance and diversity of substrate types and dominant benthic species, with subsets of species featuring more prominently on some shoals and banks than others (Heyward et al., 2017). Shepperton Shoal, west of the OA, is dominated by filter feeder communities (Radford et al., 2019). Other shoals and banks near the OA for which there is data on benthic communities (e.g., Flat Top Bank) show a very high degree of similarity (>90%) to the Goodrich Bank (outside of the EMBA) and Cape Helvetius (sites both inside and outside of the EMBA) sites that were surveyed by AIMS for the Barossa marine studies program. Lesser but still high (>80%) similarity was also reported for some shoals located further offshore, such as Evans Shoal (Heyward et al., 2017). A summary of the results from the 2015 Shoals and shelf survey marine studies is presented in Table 3-6.

#### Table 3-5: Distances to the nearest shoals and banks from the OA

Geomorphic feature	MEVA	ЕМВА	Water depth range (~m) <sup>2</sup>	Approximate distance/direction from OA
Shepparton Shoal	~	~	30–50	23 km W
Afghan Shoal	✓	✓	30–50	25 km N
Flat Top Bank	x	$\checkmark$	60–70	55 km WSW
Jones Bank	x	$\checkmark$	10	43 km SE
Skottowe Shoal	x	✓	20–30	63 km E
Moresby Shoals	x	✓	20	68 km E
Lowry Shoal	x	✓	20	68 km E
Bowra Shoal	x	~	20	78 km SE
Parsons Bank	x	~	10–20	80 km ENE
Hancox Shoal	x	~	10–30	86 km E
Foelsche Bank	x	~	10	87 km E
Marsh Shoal	x	~	10–20	87 km E
Newby Shoal	x	~	30–70	105 km WNW
Parry Shoal	x	~	10-30	106 km NW
Stephens Bank	x	~	10–20	112 km E
Beagle Shoals	x	~	20–30	138 km ENE
Taiyun Shoal	x	~	20–30	143 km ENE
Bill Shoal	x	~	20	144 km ENE
Abbott Shoal	x	~	20	152 km ENE
Renard Shoals	x	~	20	158 km ENE
Ommaney Shoals	x	~	20	163 km ENE
Wells Shoal	x	~	20–30	166 km ENE
Giles Shoal	x	~	20–30	180 km ENE
Mataram Shoal	x	~	20–40	197 km ENE
Fitzpatrick Shoal	x	~	30–40	201 km ENE

<sup>&</sup>lt;sup>2</sup> Note: water depth range provided applies to the entire feature and is not limited to the EMBA.



Figure 3-5: Banks and shoals overlapping or proximal to the OA, EMBA and MEVA



#### Table 3-6: Summary of the results of the marine studies program

Feature	Description		
Oceanic Shoals AMP (within and proximal)	Surveys of benthic habitats and fish communities were undertaken by AIMS within and adjacent to the Oceanic Shoals AMP, in the proximity of the Bayu Undan pipeline, in sites between Goodrich Bank and Bathurst Island as well as Shepparton Shoal (see below). Fish communities were surveyed at five of the sites. Benthic habitats at the six sin unconsolidated sediments such as coarse sand and mud (see photo for example of habitat type). Epibenthic fauna were present at low densities, attached to areas of communities were absent occorrect outcropping, most commonly present around ridges or drop-offs. Light-dependent communities were absent Corals were very rare and outside of bare areas, non-photic filter-feeder communities (notably sponges) were the frequently sparse, with decreasing density with depth, and very little occurrence beyond 50 m water depth (Radfor Fish species richness recorded at the sites surveyed was low compared to other shoals on the north-west shelf occiver and sandy substrate. Fish communities were dominated by bony fish, with sharks and to a lesser extent rat hose recorded at shoals further offshore, such as Tassie and Evans Shoals (both outside the EMBA). Richness, were strongly correlated with habitat characteristics, with greatest numbers linked to increased epibenthic cover (		
Shepparton Shoal	Shepparton Shoal is relatively shallow (~30 m) and differed from most other sites surveyed by having up to medium habitat type) predicted over most (86%) of the shoal (Radford et al., 2019). No hard or soft corals, or Halimeda communities were recorded and areas not supporting non-photic filter feeders to 2019). Fish were not surveyed at this site, but given the depths and habitat types present can be expected to be dominate <i>carponotatus</i> ), rockcod ( <i>Epinephelus</i> spp), sandperch ( <i>Parapercis</i> spp), threadfin bream ( <i>Pentapodus emeryii</i> ) surveyed <i>uboulayi</i> ).		



2017 (Radford et al., 2019). The benthic survey included six tes were dominated by extensive areas of seabed covered solidated pavement covered in fine sediment, or on low relief from most sites and where present were typically sparse. key habitat. However, these filter feeder communities were rd et al., 2019).

Australia, reflecting the greater proportions of bare biotic is also common. Relative abundances were less than half abundance and structure of fish communities across sites Radford et al., 2019).

density filter-feeder communities (see photo for example of

vere expected to comprise bare substrates (Radford et al.,

by bony fishes, likely including stripey snapper (*Lutjanus* eonfish (*Acanthurus* spp) and angelfish (*Chaetodontoplus* 



#### 3.2.7 Offshore reefs and islands

The EMBA does not overlap any of the key offshore reefs and islands in Commonwealth waters of the region.

Several nearshore islands fall within the EMBA, most notably the Tiwi Islands where the EMBA approaches and/or intersects parts of the south-west, south and east coastlines.

The Tiwi Islands are situated about 31.5 km north of the OA, 80 km north of Darwin and are comprised of Melville Island, Bathurst Island and nine smaller uninhabited islands off the northern and southern shores. The islands cover an area of about 8,320 km<sup>2</sup> and support a number of important habitats, including extensive stands of mangroves, tidal mudflats, sandy beaches, seagrass meadows and fringing reef habitats (INPEX, 2010). Many species found on the islands are not recorded anywhere else in the NT, primarily due to their isolation and climatic extremes (high rainfall) (NRETAS, 2009). The Tiwi Islands are Aboriginal freehold land owned by the Tiwi Aboriginal Land Trust (ALT) (NRETAS, 2009). A mapping exercise has been undertaken with the Tiwi Land Council (TLC) to identify environmental and socioeconomic values along the Tiwi Islands coastline (Jacobs, 2019).

The Tiwi Islands, and the small islands nearby, provide important nesting sites for marine turtles, internationally significant seabird rookeries, and some major aggregations of migratory shorebirds (DLRM, 2009). A number of BIAs for turtles are found along the coastlines of the Tiwi Islands (see Section 3.2.13.2.1). The sandy beaches on the Tiwi Islands, specifically the west coast of Bathurst Island and the north coast of Melville Island, are particularly important for marine turtle nesting. Nesting is dominated by flatback and olive ridley turtles (Chatto & Baker, 2008). However, green and hawksbill turtles also nest on the Tiwi Islands. Significant numbers of olive ridley turtles are known to nest on the beaches of Seagull Island and the north-west coast of Melville Island (Chatto & Baker, 2008), but these areas are not within the EMBA.

Five seabird breeding colonies have been reported on small offshore islands surrounding Melville and Bathurst islands (Chatto, 2001) that range in size from 2 to more than 30,000 birds (Chatto 2001). The colony on Seagull Island, off the north-west tip of Melville Island and outside the EMBA, supports a breeding BIA of about 60,000 crested terns (Woinarski et al., 2003). This is thought to be the largest breeding colony of this species and is considered an internationally significant colony (>1% global population) (NRETAS, 2009). A 20 km buffer has been designated around the BIA as a foraging zone for crested terns (see Section 3.2.13.4). The breeding period for the crested tern is from March to July, with most eggs being laid between from late April to early June (Chatto, 2001). In general, colonial seabird breeding in the NT occurs throughout most of the year, though mostly between May and November (Chatto, 2001). The extensive areas of tidal flats, particularly on the south-east of Melville Island, have also been noted as providing important wading and feeding habitats for shorebirds. The highest total count at this site was 40,000 shorebirds in 1993 with the most common species being great knots (Chatto, 2003). Other species recorded in high numbers include red-necked stints, greater and lesser sand plovers and bar-tailed godwits (Chatto, 2003).

#### 3.2.8 Other seabed features of interest

#### 3.2.8.1 Seamounts

Seamounts have been identified ~230 km north of the OA and may be present sporadically within the EMBA. The Barossa environmental baseline studies program (Jacobs, 2016) included sampling sites at seamounts to the west of the field. Seamounts are generally raised up from the seabed to water depths between 50 and 80 m and are characterised by predominantly sand and rubble (Jacobs, 2016). The hard substrate of the seamount slopes support epibenthic communities dominated by sponges and filter feeders such as gorgonians (e.g. sea whips, sea fans and soft corals) and feather stars. Other epibenthic species observed included holothurians (sea cucumbers), sea fans and algae (Jacobs, 2016).

Triggerfish nesting areas were apparent at the seamounts. The triggerfish (family Balistidae) appeared to make depressions in the sand and rubble at the top of the southernmost seamount surveyed, as they were observed in and around these depressions (Jacobs, 2016c). The seamounts also appeared to support schools of fish (predominantly from the families Lutjanidae, Carangidae and Caesionidae, and including larvae or juveniles) both near the top of the seamount and at depth.

#### 3.2.8.2 Scarps

The Barossa environmental baseline studies program (Jacobs, 2016) included sampling sites at 2 scarps in water depths ranging between 160 and 190 m. The substrate of the scarps was similar and characterised by a hard bedrock pavement at the top, with a rocky profile along the ridge and sand habitats at the base (Jacobs, 2016). The scarps provided habitat for gorgonians (e.g. sea whips), feather stars and other filter feeders, sponges, and hydroid/bryozoan turf. A deep-water snapper species (possibly goldband snapper) was also observed in a rocky overhang at the base of the slope and small silver fish and one ray were observed on the sand flat at one of the scarps (Jacobs, 2016).



Scarps may be observed sporadically within the EMBA, if present likely supporting epibenthic communities, such as sponges and filter feeders and schools of fish.

#### 3.2.9 Benthic habitats and communities

The surrounding area of the EMBA supports several types of benthic habitats supporting biota including mangroves, coral, seagrass, macroalgae, filter feeders and soft-bottom benthos. This is primarily driven by depth and seabed characteristics, notably the presence of hard substrates and benthic rugosity (Heyward et al., 2017; Radford et al., 2019; RPS, 2023).

Surveys in and adjacent to the EMBA indicate that the benthos consists mostly of soft, easily re-suspended sediments interspersed with areas of hard substrate (Smit et al., 2000; Przeslawski et al., 2011; Heyward et al., 2017; Radford et al., 2019; RPS, 2023). Overall, the diversity and coverage of epibenthos is low and organisms present are predominantly sponges, gorgonians and soft corals (Kelly & Przeslawski, 2012; Heyward et al., 2017; Radford et al., 2019; RPS, 2023). Areas of soft sediment support infauna communities, with infauna species richness tending to decrease with distance offshore (Przeslawski et al., 2011). Sampling of nearshore sediments in the Beagle Gulf found the infauna to be dominated by crustaceans, molluscs, and echinoderms (Smit et al., 2000), with crustaceans and annelids (polychaete worms) the predominant taxa in sediments along the pipeline route between the Commonwealth waters and Darwin Harbour (RPS, 2023).

Shoals and banks within the EMBA, display biological communities consistent with other similar areas in the broader region (Heyward et al., 2017). 'Mid-shelf' locations, such as those within the EMBA, typically exhibit higher turbidity, resulting in greater light attenuation and the transition between primary producer dominated habitats (such as corals) to those featuring sessile filter feeders (e.g., sponges) is often observed at shallower depths. Shepperton Shoal, is dominated by filter feeder communities (Radford et al., 2019). Other shoals and banks within the EMBA for which there is data on benthic communities (e.g., Flat Top Bank) show a very high degree of similarity (>90%) to the Goodrich Bank and Cape Helvetius sites that were surveyed by AIMS for the Barossa marine studies program.

The hard substrate of the seamount slopes support epibenthic communities dominated by sponges and filter feeders such as gorgonians (e.g., sea whips, sea fans and soft corals) and feather stars. Other epibenthic species observed included holothurians (sea cucumbers), sea fans and algae (Jacobs, 2016). Triggerfish nesting areas were apparent at the seamounts. The triggerfish (family Balistidae) appeared to make depressions in the sand and rubble at the top of the southernmost seamount surveyed, as they were observed in and around these depressions (Jacobs, 2016). The seamounts also appeared to support schools of fish (predominantly from the families Lutjanidae, Carangidae and Caesionidae, and including larvae or juveniles) both near the top of the seamount and at depth.

Scarps may be observed sporadically within the EMBA, if present likely supporting epibenthic communities, such as sponges and filter feeders and schools of fish.

Surveys of mid-shelf benthic habitats of the EMBA indicate that corals are generally rare, predominantly in areas of <30 m water depth and more likely to develop in areas of steeper bathymetry (Heyward et al., 2017; Radford et al., 2019). Corals in turbid waters are likely dominated by members of the genus Turbinaria (IMCRATG, 1998), while Acropora and Montipora species are reported to occur in clearer waters at the Vernon Islands (IMCRATG, 1998; Smit et al., 2000; Calnan, 2006). However, in general extensive hard coral reefs are unlikely to be present in the EMBA.

No seagrasses were recorded during benthic surveys at mid-shelf locations in the EMBA (Heyward et al., 2017; Radford et al., 2019) or at Shepperton Shoal. Seagrasses within NT waters are not well described (Butler and Jernakoff, 1999), but seagrass distribution in the region is disjointed, not common in large open bays and typically found in and around inshore islands, small bays and inlets (Roelof et al., 2005). Seagrass communities are confined to the intertidal area, with high turbidity restricting light penetration in the coastal shelf areas to water depths of up to 20 m (DEWHA, 2008).

#### 3.2.10 Shoreline habitats

Shoreline habitats are defined as those habitats that are adjacent to the water along the mainland and of islands that occur above the lowest astronomical tide (LAT) and most often in the intertidal zone. The EMBA intersects shorelines on the NT mainland, notably around Cape Hotham (excluding Darwin Harbour), at some coastal islands, including the south, south-east and south-western coasts of the Tiwi Islands, and other scattered locations in the NT, including the western tip of Cobourg Peninsula (Figure 3-1). Sections 3.2.10.1 to 3.2.10.4 describe the shoreline habitats within the EMBA.



#### 3.2.10.1 Mangroves

Mangroves are common and widely distributed along coastlines of the NT (Chatto & Baker, 2008), and extensive mangals occur at many, if not most, of the tidal flats, estuaries and tidal creeks along the mainland coast and on islands that fall within the EMBA.

Coastal habitat surveying undertaken following the Montara spill (Duke et al., 2010) estimated mangroves to cover ~90% of the shorelines in Darwin Harbour and ~73% between Darwin Harbour (Mandorah) and Point Blaze. Mangroves also occur less extensively in areas of the EMBA east of Darwin, including Cobourg Peninsula. At the Tiwi Islands, the southern shorelines within the EMBA do not support the more extensive mangroves that occur within tidal creeks that open to the north coast and in Apsley Strait.

Mangroves are important primary producers and have several ecological and economic values. For example, they play a key role in reducing coastal erosion by stabilising sediment with complex root systems (Kathiresan & Bingham, 2001). They are recognised for their capacity to help protect coastal areas from the damaging effects of erosion during storms and storm surge. Mangroves are important in the filtration of runoff from land, which helps maintain water clarity for the coral reefs that are often found offshore in tropical locations (NOAA, 2010).

The muddy sediments that occur in mangrove forests are home to a variety of epibenthic, infaunal and meiofaunal invertebrates (Kathiresan & Bingham, 2001). Crustaceans known to inhabit the mud in mangrove systems include fiddler crabs, mud crabs, shrimps and barnacles. Within the water channels of the mangrove systems, various finfish are found from the smaller fish such as gobies and mudskippers (which are restricted to life in the mangroves) through to larger fish such as barramundi (*Lates calcarifer*) and the mangrove jack (*Lutjanus argentimaculatus*). Mangroves and their associated invertebrate-rich mudflats are an important habitat for migratory shorebirds from the northern hemisphere, as well as some avifauna that are restricted to mangroves as their sole habitat (Garnet & Crowley, 2000).

#### 3.2.10.2 Intertidal mud/sand flats

Intertidal mud/sand flats form when fine sediment carried by rivers and/or the ocean is deposited in a low-energy environment. Due to the large tidal ranges, intertidal flats are common along NT coastlines and often extensive at low tide, frequently occurring adjacent to, or in conjunction with, mangrove communities in the EMBA. Duke et al (2010) indicates that intertidal mud/sand flats occur along >75% of the shore within the Darwin Harbour region and >66% of the coast between Mandorah and Point Blaze. The south-eastern coast of Melville Island also contains reasonably large areas of mud and sand flats that are exposed at low tides (Chatto & Baker, 2008). There is a large amount of intertidal mudflat, backed by extensive mangroves and open saline wetlands, in Fog Bay (southern section) and around parts of the Perron Islands (AMOSC, 2019), with this area of Fog Bay and Darwin Harbour both listed as Nationally Important Wetlands. Section 3.2.12.2 describe the wetlands of international and national importance that intersect the EMBA.

Intertidal flats are highly productive components of shelf ecosystems, responsible for recycling organic matter and nutrients through microbial activity. This microbial activity helps stabilise organic fluxes by reducing seasonal variation in primary productivity providing a more constant food supply. Intertidal sand and mudflats support a wide range of benthic infauna and epifauna which graze on microscopic algae and bivalves, molluscs, polycheate worms and crustaceans (Zell, 2007).

The high abundance of invertebrates found in intertidal sand and mudflats provides an important food source for finfish and rays which swim over the area at high tide. Mudflats have also been shown to be nursery areas for flatfish. During low tide, these intertidal areas are important foraging areas for resident and migratory shorebirds (see Section 3.2.13.4).

#### 3.2.10.3 Sandy beaches

Sandy beaches are those areas within the intertidal zone where unconsolidated sediment has been deposited and eroded by wave and tidal action. Sandy beaches can vary from low to high energy zones, the energy experienced influences the beach profile due to varying rates of erosion and accretion.

Sandy habitats are important for both resident and migratory seabirds and shorebirds (see Section 3.2.13.4). While sand flats and beaches generally support fewer species and numbers of birds than mudflats of similar size; some species such as the beach thick knee (*Esacus giganteus*) are commonly associated with sandy beaches (Garnet & Crowley, 2000). Sandy beaches can also provide important habitat for turtle nesting (see Section 3.2.13.2.1), with female turtles traversing the intertidal beach to lay eggs in the supra-tidal zone (outside the EMBA).

Sandy beaches intersected by the EMBA include part of the extensive stretches along northern Fog Bay up to Point Paterson, at Point Blaze and on many of the islands, including the Tiwi Islands. Turtle nesting on Fog Bay and Tiwi Islands beaches within the EMBA is dominated by flatback and to a lesser extent olive ridley turtles, with the southern beaches of the Tiwi Islands supporting less activity than south-west and northern beaches (Chatto & Baker, 2008).



#### 3.2.10.4 Rocky shorelines

Rocky shores can include pebble/cobble, boulders and rocky cliffs (often at the landward edge of reef platforms). Rocky shorelines can vary from habitats where there is bedrock protruding from soft sediments to cliff–like structures that form headlands. Within the EMBA rocky shorelines occur on the Cobourg Peninsula as well as a number of islands. The Cobourg Peninsula coastlines include numerous rocky headlands and there are intermittent scattered low lateritic cliffs in the Anson-Beagle bioregion (IMCRATG, 1998).

Rocky shorelines are an important foraging area for seabirds and habitat for invertebrates found in the intertidal splash zone.

#### 3.2.11 Plankton

Plankton abundance and distribution is patchy, dynamic and strongly linked to localised and seasonal productivity (Evans et al., 2016). Fluctuations in abundance and distribution occur both vertically and horizontally in response to tidal cycles, seasonal variation (light, water temperature and chemistry, currents and nutrients) and cyclonic events.

In northern Australia, nutrients and detritus (debris) carried by large river outflows combine with sediments and particulate organic matter resuspended by the tides and generally remain trapped within coastal areas to depths of up to ~20 m (or up to 45 nautical miles [Nm] offshore). The coastal waters within this zone generally do not mix with adjacent offshore waters, and as a result support distinctly different and more productive phytoplanktonic communities (made up of small, often microscopic, free-floating plants) than offshore waters, where nutrients are derived primarily from the ocean and atmosphere (DEWHA, 2008d).

Within the EMBA, plankton communities are likely to reflect this regional pattern, varying with depth and distance offshore. Communities of phytoplankton in coastal waters bloom and decay in response to seasonal changes in water flows, resuspension of sediments by cyclones, strong tidal currents, monsoon winds and wind-generated waves (DEWHA, 2008d). In deeper offshore areas, productivity is likely to be more dependent on internal nutrient cycling and upwellings of productive oceanic waters, such as around the shoals and pinnacles associated with KEFs of the region (see Section 3.2.12.3).

	Receptor	OA presence	MEVA presence	EMBA presence	
Category				Northern Shelf Province	Northwest Shelf Transition
Benthic habitats	Coral reefs	X	~	$\checkmark$	$\checkmark$
	Seagrass	X	~	$\checkmark$	$\checkmark$
	Macroalgae	X	~	$\checkmark$	$\checkmark$
	Non-coral benthic invertebrates	~	~	$\checkmark$	~
Shoreline habitats	Mangroves	×	X	$\checkmark$	$\checkmark$
	Intertidal platforms	X	X	$\checkmark$	$\checkmark$
	Sandy beaches	X	X	$\checkmark$	$\checkmark$
	Rocky shorelines	×	X	$\checkmark$	$\checkmark$

#### Table 3-7: Habitats within the OA, MEVA and EMBA (IMCRA provincial bioregions)

#### 3.2.12 Protected areas and key ecological features

Protected areas and key ecological features identified in the OA, MEVA and EMBA are listed in Table 3-8 and are illustrated in Figure 3-6 and Figure 3-7. No threatened ecological communities are overlapped by the OA, MEVA or EMBA.

There are no National heritage places or world heritage property in the OA, MEVA or EMBA.



# Table 3-8: Protected areas and key ecological features within the OA, MEVA and EMBA and approximate distance from the OA boundary

Value/sensitivity name	OA presence	MEVA presence	EMBA presence	Approximate Distance to OA	
Protected areas					
Australian marine parks					
Oceanic Shoals Marine Park	×	x	$\checkmark$	67.5 km	
Joseph Bonaparte Gulf Marine Park	×	x	$\checkmark$	182 km	
Marine national parks					
Garig Gunak Barlu	×	x	$\checkmark$	220 km	
Wetlands of international importance (Ramsar site) and national importance					
Cobourg Peninsula	x	x	$\checkmark$	220 km	
Key ecological features					
North Marine Region					
Carbonate bank and terrace system of the Van Diemen Rise	x	✓	✓	9 km	
Pinnacles of the Bonaparte Basin	×	X	$\checkmark$	173 km	



Figure 3-6: Marine parks and protected areas overlapping or proximal to the EMBA and MEVA





#### 3.2.12.1 Marine parks

The OA and MEVA do not intersect any Australian marine parks or Territory/State marine parks, management areas or reserves. The EMBA overlaps two Australian Marine Parks (AMPs) and a marine national park, which are identified in Table 3-8 and Figure 3-6.

Marine parks are divided into management zones (Figure 3-6) and managed in accordance with the North Marine Parks Network Management Plan (DNP, 2018a), as is the KEF identified in the NMR. The values for these AMPs that overlap the EMBA are summarised in Table 3-9.

In agreement with the states and NT governments, the Australian Government has committed to establish AMPs as a component of the National Representative System of Marine Protected Areas (Director of National Parks, 2012). In November 2012, the Commonwealth Marine Reserves Network was proclaimed with the purpose of protecting the biological diversity and sustainable use of the marine environment. Commonwealth marine reserves were renamed as Australian Marine Parks in October 2017 and there are six marine regions in the Australian Marine Parks Network, namely the Coral Sea, South-west, Temperate East, South-east, North, and North-west.

Management plans for AMPs were developed and enacted on 1 July 2018. Under these plans, AMPs are allocated conservation objectives (International Union for Conservation of Nature [IUCN] Protected Area Category) based on the Australian IUCN Reserve Management Principles in Schedule 8 of the EPBC Regulations 2000. These principles determine what activities are acceptable within the different zones of the AMP network.

Garig Gunak Barlu is managed by the NT Parks and Wildlife Commission and declared under the *Cobourg Peninsula Aboriginal Land, Sanctuary and Marine Park Act 1981* (NT). The Cobourg Marine Park Plan of Management (NT Government, 2011) expired in 2021.

Value sensitivity	Management zone(s)	Values overlapping the EMBA
AMP		
Joseph Bonaparte Gulf	Multiple Use Zone (IUCN VI)     Special Purpose Zone (IUCN VI)	The Joseph Bonaparte Gulf Marine Park values (DNP, 2018a):
		<ul> <li>ecosystems representative of the Northwest Shelf Transition—dynamic environment influenced by strong tidal currents, monsoonal winds, cyclones and wind- generated waves</li> </ul>
		<ul> <li>a range of species, including species listed as threatened, migratory, marine, or cetacean under the EPBC Act</li> </ul>
		<ul> <li>sea country, which is valued for Indigenous cultural identity, health and wellbeing</li> </ul>
		<ul> <li>commercial fishing, tourism, mining, and recreation, including fishing, are important activities in the Marine Park</li> </ul>
Oceanic Shoals	Special Purpose Zone (Trawl)	The Oceanic Shoals Marine Park values (DNP, 2018a):
	(IUCN VI)     Multiple Use Zone (IUCN VI)	ecosystems representative of the Northwest Shelf Transition
	<ul> <li>National Park Zone (IUCN II)</li> <li>Habitat Protection Zone (IUCN IV)</li> </ul>	2 KEFs -carbonate bank and terrace systems of the Van Diemen Rise and the pinnacles of the Bonaparte Basin
		<ul> <li>a range of species, including species listed as threatened, migratory, marine, or cetacean under the EPBC Act</li> </ul>
		BIAs that include foraging habitat for marine turtles
		<ul> <li>sea country, which is valued for Indigenous cultural identity, health and wellbeing</li> </ul>
		<ul> <li>commercial fishing, tourism, and recreation, including fishing, are important activities.</li> </ul>

#### Table 3-9: Marine park values overlapping the EMBA



Value sensitivity	Management zone(s)	Values overlapping the EMBA
Marine National Park		
Garig Gunak Barlu	<ul> <li>Multiple Use A Zone</li> <li>Multiple Use B Zone</li> </ul>	<ul> <li>Multiple Use A and B zones provide for multiple use of the park's resources, including commercial fishing activities. Multiple Use A zone has more intensive fishing, such as prawn trawling and netting. These zones also provide protection of important conservation and scientific values</li> <li>provides BIAs for dolphins, seabirds and marine turtles</li> <li>habitat critical to the survival of flatback, green and olive ridley turtles</li> <li>habitats, feeding areas, dispersal and migratory pathways, and spawning sites for numerous fish and</li> </ul>
		crustacean species of fisheries significance
		Note: The EMBA intersects the perimeter of the marine park with no predicted shoreline or surface oil contact at or above low threshold values.

#### 3.2.12.2 Wetlands of international importance

The Ramsar Convention on Wetlands is an intergovernmental treaty that aims to conserve wetlands of international importance. Ramsar wetlands are recognised as MNES under the EPBC Act (DSEWPaC, 2010). No Ramsar or nationally important wetlands occur within the OA. The EMBA intersects with one Ramsar wetland – Cobourg Peninsula (Table 3-8, Figure 3-6).

The Cobourg Peninsula comprises both coastal and inland wetlands and was declared a Ramsar sites in 1974. Important habitat for seabirds throughout the peninsula includes intertidal forested wetlands and mudflats, seasonal freshwater marshes and permanent freshwater pools. Four coral reefs are located within the Coburg Peninsula Ramsar site: Popham Creek, Kuper Point, Sandy Island No. 1, and Sandy Island No. 2 (AECOM, 2011). A total of 595 marine fish species from 117 families have been recorded from the Cobourg Peninsula area (BMT WMB 2011).

Bird species richness within the Cobourg Peninsula is high, with 236 bird species having been recorded including 89 waterbird species, 21 of which are migratory (MBT WBM, 2011). The Cobourg Peninsula supports habitat and conditions that are important for waterbird breeding. At least six seabird species are known to occupy the Cobourg Peninsula for breeding purposes, with notable breeding colonies found on sandy, coral rubble islands and headlands (BMT WMB 2011).

In addition to providing important habitat for seabirds, the Cobourg Peninsula is known to provide important nesting habitat for six marine turtle species including the green turtle (*Chelonia mydas*), flatback turtle (*Natator depressus*), leatherback turtle (*Dermochelys coriacea*), Olive Ridley Turtle (*Lepidochelys olivacea*), hawksbill turtle (*Eretmochelys imbricata*), loggerhead turtle (*Caretta carreta*). The dugong (*Dugong dugon*) is also known to forage within waters around the site. Additionally, several nationally threatened species are known from the site.

#### 3.2.12.3 Key ecological features

Key ecological features (KEFs) are those components of the marine ecosystem that are important for biodiversity or the ecosystem function and integrity of a Commonwealth marine area. No KEFs overlap the OA. The EMBA overlaps two identified KEFs:

- Carbonate bank and terrace system of the Van Diemen Rise
- Pinnacles of the Bonaparte Basin



Figure 3-7: Key Ecological Features overlapping or proximal to the EMBA and MEVA





#### 3.2.12.3.1 Carbonate Bank and Terrace System of the Van Diemen Rise

The EMBA and MEVA overlap the Carbonate Bank and Terrace System of the Van Diemen Rise KEF (Figure 3-7). The EMBA overlaps approximately 2,467 km<sup>2</sup> (7.89%) of the Carbonate Bank and Terrace System of the Van Diemen Rise KEF (Figure 3-7). The Carbonate Bank and Terrace System of the Van Diemen Rise covers about 31,278 km<sup>2</sup> and forms part of the larger system associated with the Shaul Banks to the north and Londonderry Rise to the east. The value of this KEF is 'unique seafloor feature with ecological properties of regional significance' (DEWHA, 2012a) and it is considered important both for its role in enhancing biodiversity and local productivity relative to its surrounds and for supporting relatively high species diversity. The KEF is characterised by carbonate terrace, banks, channels, and valleys, with variability in water depth and substrate composition contributing to unique ecosystems in the channels.

The carbonate banks and shoals found within the Van Diemen Rise make up 80% of the banks and shoals, 79% of the channels and valleys, and 63% of the terrace found across the NMR. The carbonate banks and shoals rise from depths of 100 to 200 m to within 10 m of the surface (Anderson et al., 2011).

A 2010 survey by Geoscience Australia and AIMS mapped the seabed environments of the Van Diemen Rise (Anderson et al., 2011). The study surveyed 784 km<sup>2</sup> towed video transects at 77 sites including banks, terraces, valleys and plains within the Van Diemen Rise. The shallow banks sampled contained complex benthic features with diverse and often dense epibenthic assemblages. A total of 175 video characterisations were recorded from 13 bank sampling sites in the study area from depths of 11 to 54 m (mean depth of 34 m). The sites were characterised by mostly low-lying rock outcrops with hard corals and octocorals (18% and 99% occurrence, respectively) along with smaller colonies of bryozoa and ascidians. The rocky outcrops were interspersed by small areas of relatively barren coarse-grained soft sediments (Anderson et al., 2011).

The KEF provides habitat for a high diversity of sponges, soft corals, and other sessile filter feeders, epifauna and infauna, along with olive ridley turtles, sea snakes and sharks. Rich sponge gardens and octocorals have been identified on the eastern Joseph Bonaparte Gulf along the banks, ridges, and some terraces. Plains in deep hole/valleys are characterised by scattered epifauna and infauna that include polychaetes and ascidians. Epibenthic communities such as the sponges found in the channels are likely to support fish and second-order consumers. Pelagic fish such as mackerel, red snapper and a distinct gene pool of gold band snapper are found in the Van Diemen Rise.

#### 3.2.12.3.2 Pinnacles of the Bonaparte Basin

The EMBA overlaps approximately 10 km<sup>2</sup> (1.97%) of the Pinnacles of the Bonaparte Basin KEF (Figure 3-7). The limestone pinnacles of the Bonaparte Basin are located in the mid-outer shelf of the western Joseph Bonaparte Gulf and comprise of 61% of the limestone pinnacles in the Northwest Marine Region and 8% of the total limestone pinnacles found within the Australian Exclusive Economic Zone (EEZ) (Baker et al., 2008). The pinnacles are found in waters 30 to 80 m deep and provide hard substrate for sessile species. The pinnacles are thought to be remnants of the calcareous shelf and coastal features from previous low sea-level stands and have been recorded to be up to 50 m in height and range from 50 to 100 km long (Baker et al., 2008; Heyward et al., 1997).

Diverse communities of sessile benthic invertebrates including hard and soft corals, sponges, whips, fans, bryozoans and aggregations of demersal fish species such as snappers, emperors and groupers have been recorded (Brewer et al., 2007). Foraging and general use has been recorded within the pinnacles by marine turtles and the area has also been suggested to be used by freshwater and green sawfish as well as humpback whales (Donovan et al., 2008). The pinnacles have been recognised as a sponge biodiversity hotspot supporting greater diversity and communities than the surrounding seafloor (NERP MBH, 2014).

The Pinnacles of the Bonaparte Basin are defined as a KEF as they are a unique seafloor feature with ecological properties of regional significance. Their biodiversity value relates to both the benthic and pelagic habitats (CoA, 2012). The hard substrate of the pinnacles is likely to support a high number of species, although a better understanding of the species richness and diversity associated with these structures is required.

#### 3.2.12.4 Commonwealth heritage areas

Australia's listed heritage places comprise natural, Indigenous, and historic heritage places which are either entirely within a Commonwealth area, or outside the Australian jurisdiction and owned or leased by the Commonwealth or a Commonwealth Authority. Significant heritage places are identified and grouped (by type) into lists that guide the protection and management of heritage values. Those heritage places located in the EMBA are shown in Figure 3-15 and have been described in Sections 3.2.14 and 3.2.15.

#### 3.2.13 Threatened and migratory fauna

The PMST identified a number of threatened and migratory species (matters of national environmental significance, or MNES) listed under the EPBC Act with the potential to occur in marine or shoreline habitats in the EMBA. Fauna that may be present in the OA, MEVA and EMBA are presented in Table 3-10.

An examination of the species profile and threats (SPRAT) database (DCCEEW, 2022b) showed some threatened species were not expected to occur in significant numbers in the marine and coastal environments (within the EMBA) due to their terrestrial distributions. Species that may occur on shorelines include shorebirds, but terrestrial mammals, reptiles (such as pythons) and bird species that do not have core habitats along shorelines have been excluded. These species are unlikely to come into contact with a hydrocarbon spill and therefore are not discussed further.

An additional species, the grey nurse shark (*Carcharias taurus*; EPBC-listed Vulnerable), has been included in this GEP Coastal Waters OEMP as they were reported as occurring within or near the EMBA during surveys as part of the Barossa Marine Studies Program.

A compilation of tracking data from marine turtle telemetry studies on and around the Tiwi Islands indicates turtle foraging areas and migration pathways do overlap with the EMBA (Pendoley, 2023) (Table 3-11). The report identified that the waters surrounding the Tiwi Islands are traversed by marine turtles nesting in other areas of northern Australia, including olive ridleys from the Wessel Islands, flatback turtles from WA, QLD and the NT, green turtles from WA and from Groote Eylandt, and loggerhead turtles from WA. Collectively, these data indicate that marine turtle migratory pathways are largely restricted to the waters inside of the 100 m depth contour (i.e., waters less than 100 m deep).
## Table 3-10: Threatened and migratory marine fauna that may be present in the OA, MEVA and EMBA

Marine fauna			Territory Parks		OA		MEVA	ЕМВА	
Scientific name	Common name	EPBC Act 1999	and Wildlife Conservation Act 1976	May be present	Values or sensitivities	May be present	Values or sensitivities	May be present	Values or sensitivities
Fishes, sharks, and rays									
Anoxypristis cuspidata	Narrow sawfish	Migratory	×	~	Species or species habitat may occur within area	√	Species or species habitat likely to occur within area	$\checkmark$	Species or species habitat known to occur within area
Carcharhinus longimanus	Oceanic whitetip shark	Migratory	×	~	Species or species habitat may occur within area	√	Species or species habitat may occur within area	$\checkmark$	species or species habitat may occur within area
Carcharodon carcharias	Great white shark	Migratory, Vulnerable	×	~	Species or species habitat may occur within area	√	Species or species habitat may occur within area	$\checkmark$	Species or species habitat may occur within area
Glyphis garricki	Northern river shark	Endangered	Endangered	$\checkmark$	Species or species habitat may occur within area	$\checkmark$	Species or species habitat may occur within area	$\checkmark$	Breeding known to occur within area
Glyphis glyphis	Speartooth shark	Critically Endangered	Vulnerable	X	Species or species habitat may occur within area	√	Species or species habitat may occur within area	$\checkmark$	Species or species habitat known to occur within area
lsurus oxyrinchus	Shortfin mako	Migratory	×	X	Species or species habitat likely to occur within area	√	Species or species habitat likely to occur within area	$\checkmark$	Species or species habitat likely to occur within area
Isurus paucus	Longfin mako	Migratory	×	×	Species or species habitat likely to occur within area	√	Species or species habitat likely to occur within area	$\checkmark$	Species or species habitat likely to occur within area
Mobula alfredi	Reef manta ray	Migratory	×	~	Species or species habitat may occur within area	√	Species or species habitat likely to occur within area	$\checkmark$	Species or species habitat likely to occur within area
Mobula birostris	Giant manta ray	Migratory	×	~	Species or species habitat may occur within area	√	Species or species habitat likely to occur within area	√	Species or species habitat likely to occur within area
Pristis clavata	Dwarf sawfish	Migratory, Vulnerable	Vulnerable	~	Species or species habitat known to occur within area	√	Species or species habitat known to occur within area	$\checkmark$	species or species habitat known to occur within area
Pristis pristis	Freshwater sawfish / largetooth sawfish	Migratory, Vulnerable	Vulnerable	~	Species or species habitat may occur within area	√	Species or species habitat likely to occur within area	√	Species or species habitat likely to occur within area
Pristis zijsron	Green sawfish	Migratory, Vulnerable	Vulnerable	~	Species or species habitat known to occur within area	√	Species or species habitat known to occur within area	$\checkmark$	Species or species habitat known to occur within area
Rhincodon typus	Whale shark	Migratory, Vulnerable	Data deficient	~	Species or species habitat may occur within area	√	Species or species habitat may occur within area	√	species or species habitat may occur within area.
Sphyrna lewini	Scalloped hammerhead	Conservation Dependent	×	$\checkmark$	Species or species habitat likely to occur within area	$\checkmark$	Species or species habitat likely to occur within area	$\checkmark$	Species or species habitat known to occur within area
Carcharias taurus	Grey nurse shark	Migratory, Vulnerable	×	×	Species or species habitat may occur within area	$\checkmark$	Species or species habitat may occur within area	$\checkmark$	Species or species habitat may occur within area
Marine mammals									
Balaenoptera borealis	Sei whale	Migratory, Vulnerable	×	~	Species or species habitat may occur within area	$\checkmark$	Species or species habitat may occur within area	$\checkmark$	Foraging, feeding or related behaviour likely to occur within area.
Balaenoptera edeni	Bryde's whale	Migratory	×	~	Species or species habitat may occur within area	√	Species or species habitat may occur within area	$\checkmark$	Species or species habitat likely to occur within area
Balaenoptera musculus brevicauda	Pygmy blue whale	Migratory, Endangered	×	~	Species or species habitat may occur within area	√	Species or species habitat likely to occur within area	√	Migration route known to occur within area
Balaenoptera physalus	Fin whale	Migratory, Vulnerable	×	~	Species or species habitat may occur within area	√	Species or species habitat may occur within area	$\checkmark$	Foraging, feeding or related behaviour likely to occur within area
Dugong dugon	Dugong	Migratory	×	X	N/A	1	Species or species habitat known to occur within area	√	Breeding known to occur within area
Megaptera novaeangliae	Humpback whale	Migratory	×	1	Species or species habitat likely to occur within area	1	Species or species habitat likely to occur within area	√	Species or species habitat likely to occur within area
Orcaella heinsohni	Australian snubfin dolphin	Migratory, Vulnerable	×	√	Species or species habitat may occur within area	$\checkmark$	Species or species habitat known to occur within area	$\checkmark$	Breeding known to occur within area.

Marine fauna			Territory Parks		OA		MEVA	ЕМВА	
Scientific name	Common name	EPBC Act 1999	and Wildlife Conservation Act 1976	May be present	Values or sensitivities	May be present	Values or sensitivities	May be present	Values or sensitivities
Orcinus orca	Killer whale, orca	Migratory	X	√	Species or species habitat may occur within area	√	Species or species habitat may occur within area	√	Species or species habitat may occur within area
Sousa sahulensis	Australian humpback dolphin	Migratory, Vulnerable	×	~	Species or species habitat may occur within area	1	Species or species habitat known to occur within area	√	Breeding known to occur within area
Tursiops aduncus	Spotted bottlenose dolphin	Migratory	×	~	Species or species habitat likely to occur within area	~	Species or species habitat known to occur within area	√	Species or species habitat known to occur within area
Marine reptiles									
Caretta caretta	Loggerhead turtle	Vulnerable Migratory Listed Marine	Vulnerable	√	Species or species habitat likely to occur within area	√	Species or species habitat known to occur within area	~	Foraging, feeding, or related behaviour known to occur within area
Chelonia mydas	Green turtle	Vulnerable Migratory Listed Marine	X	√	Congregation or aggregation known to occur within area	V	Foraging, feeding or related behaviour known to occur within area	~	Breeding known to occur within area
Crocodylus porosus	Salt-water crocodile	Migratory	×	~	Species or species habitat likely to occur within area	~	Species or species habitat likely to occur within area	√	Species or species habitat likely to occur within area
Dermochelys coriacea	Leatherback turtle	Vulnerable Migratory Listed Marine	Critically Endangered	√	Species or species habitat likely to occur within area	√	Foraging, feeding or related behaviour likely to occur within area	~	Foraging, feeding or related behaviour known to occur within area
Eretmochelys imbricata	Hawksbill turtle	Vulnerable Migratory Listed Marine	Vulnerable	~	Species or species habitat likely to occur within area	√	Species or species habitat known to occur within area	√	Breeding known to occur within area
Lepidochelys olivacea	Olive ridley turtle	Vulnerable Migratory Listed Marine	Vulnerable	1	Congregation or aggregation known to occur within area	~	Breeding known to occur within area	√	Breeding known to occur within area
Natator depressus	Flatback turtle	Vulnerable Migratory Listed Marine	X	1	Congregation or aggregation known to occur within area	√	Foraging, feeding or related behaviour known to occur within area	~	Breeding known to occur within area
Seabirds and shorebirds	•	·	·						
Acrocephalus orientalis	Oriental reed-warbler	Migratory	×	x	N/A	x	N/A	√	Species or species habitat may occur within area
Actitis hypoleucos	Common sandpiper	Migratory	×	~	Species or species habitat may occur within area	~	Species or species habitat may occur within area		Species or species habitat known to occur within area
Anous stolidus	Common noddy	Migratory	×	$\checkmark$	Species or species habitat may occur within area	~	Species or species habitat may occur within area	$\checkmark$	Breeding known to occur within area
Apus pacificus	Fork-tailed swift	Migratory	×	×	N/A	$\checkmark$	Species or species habitat likely to occur within area	$\checkmark$	Species or species habitat likely to occur within area overfly marine area
Arenaria interpres	Ruddy Turnstone	Migratory, Vulnerable	X	x	N/A	X	N/A	$\checkmark$	Foraging, feeding or related behaviour known to occur within area
Calidris acuminata	Sharp-tailed sandpiper	Migratory, Vulnerable	×	~	Species or species habitat may occur within area	~	Species or species habitat may occur within area	√	Roosting known to occur within area
Calidris alba	Sanderling	Migratory	X	X	N/A	X	N/A	$\checkmark$	Foraging, feeding or related behaviour known to occur within area
Calidris canutus	Red knot	Vulnerable Listed Marine Migratory	Endangered	~	Species or species habitat may occur within area	~	Species or species habitat may occur within area	1	Species or species habitat known to occur within area overfly marine area

Marine fauna			<b>Territory Parks</b>		OA		MEVA	ЕМВА	
Scientific name	Common name	EPBC Act 1999	and Wildlife Conservation Act 1976	May be present	Values or sensitivities	May be present	Values or sensitivities	May be present	Values or sensitivities
Calidris ferruginea	Curlew sandpiper	Critically endangered Listed Marine Migratory	Critically Endangered	√	Species or species habitat may occur within area	1	Species or species habitat may occur within area	1	Species or species habitat known to occur within area overfly marine area
Calidris melanotos	Pectoral sandpiper	Migratory	X	~	Species or species habitat may occur within area	~	Species or species habitat may occur within area	1	Species or species habitat known to occur within area overfly marine area
Calidris ruficollis	Red-necked stint	Migratory	×	X	N/A	X	N/A	~	Foraging, feeding or related behaviour known to occur within area
Calidris subminuta	Long-toed stint	Migratory	X	X	N/A	X	N/A	~	Foraging, feeding or related behaviour known to occur within area
Calidris tenuirostris	Great Knot	Migratory, Vulnerable	Critically Endangered	X	N/A	X	N/A	1	Foraging, feeding or related behaviour known to occur within area
Calonectris leucomelas	Streaked shearwater	Migratory	×	√	Species or species habitat known to occur within area	~	Species or species habitat known to occur within area	√	Species or species habitat known to occur within area
Cecropis daurica	Red-rumped Swallow	Migratory	X	X	N/A	x	N/A	1	Species or species habitat may occur within area
Charadrius dubius	Little ringed plover	Migratory	X	X	N/A	X	N/A	~	Foraging, feeding or related behaviour known to occur within area
Charadrius leschenaultii	Greater sand plover	Vulnerable Listed Marine Migratory	Vulnerable	X	N/A	X	N/A	√	Species or species habitat known to occur within area
Charadrius mongolus	Lesser Sand Plover, Mongolian Plover	Migratory, Endangered	Endangered	X	N/A	X	N/A	1	Foraging, feeding or related behaviour known to occur within area
Charadrius veredus	Oriental plover	Migratory	X	×	N/A	x	N/A	1	Roosting known to occur within area overfly marine area
Cuculus optatus	Oriental Cuckoo, Horsfield's Cuckoo	Migratory	X	×	N/A	X	N/A	√	Species or species habitat known to occur within area
Epthianura crocea tunneyi	Alligator Rivers Yellow Chat	Endangered	Endangered	×	N/A	×	N/A	√	Species or species habitat may occur within area
Erythrura gouldiae	Gouldian Finch	Endangered	Vulnerable	×	N/A	×	N/A	√	Species or species habitat likely to occur within area
Erythrotriorchis radiatus	Red goshawk	Endangered	Vulnerable	X	N/A	X	N/A	1	Species or species habitat likely to occur within area
Falco hypoleucos	Grey Falcon	Vulnerable	Vulnerable	X	N/A	x	N/A	1	Species or species habitat likely to occur within area
Fregata ariel	Lesser frigatebird	Migratory	×	√	Species or species habitat likely to occur within area	~	Species or species habitat likely to occur within area	√	Breeding known to occur within area
Fregata minor	Great frigatebird	Migratory	×	~	Species or species habitat likely to occur within area	~	Species or species habitat likely to occur within area	√	Breeding known to occur within area
Gallinago megala	Swinhoe's snipe	Migratory	X	X	N/A	X	N/A	~	Foraging, feeding or related behaviour known to occur within area
Gallinago stenura	Pin-tailed Snipe	Migratory	×	X	N/A	x	N/A	1	Foraging, feeding or related behaviour likely to occur within area

Marine fauna			<b>Territory Parks</b>		OA		MEVA	ЕМВА	
Scientific name	Common name	EPBC Act 1999	and Wildlife Conservation Act 1976	May be present	Values or sensitivities	May be present	Values or sensitivities	May be present	Values or sensitivities
Glareola maldivarum	Oriental pratincole	Migratory	x	X	N/A	X	N/A	√	Roosting known to occur within area overfly marine area
Hirundo rustica	Barn swallow	Migratory	×	X	N/A	X	N/A	√	Species or species habitat likely to occur within area
Limicola falcinellus	Broad-billed sandpiper	Migratory	X	X	N/A	X	N/A	1	Foraging, feeding or related behaviour known to occur within area
Limnodromus semipalmatus	Asian dowitcher	Vulnerable Migratory	×	X	N/A	X	N/A	√	Species or species habitat likely to occur within area
Limosa lapponica baueri	Nunivak bar-tailed godwit	Endangered	Vulnerable	X	N/A	X	N/A	√	Species or species habitat known to occur within area
Limosa lapponica menzbieri	Bar-tailed godwit (northern Siberian)	Migratory Endangered	Critically Endangered	X	N/A	~	Species or species habitat may occur within area	√	Species or species habitat known to occur within area
Limosa limosa	Black-tailed Godwit	Migratory, Endangered	X	X	N/A	X	N/A	1	Foraging, feeding or related behaviour known to occur within area
Motacilla cinerea	Grey Wagtail	Migratory	×	×	N/A	×	N/A	√	Species or species habitat may occur within area
Motacilla flava	Yellow Wagtail	Migratory	×	×	N/A	×	N/A	$\checkmark$	Species or species habitat likely to occur within area
Numenius madagascariensis	Eastern curlew	Migratory, Critically Endangered	Critically Endangered	√	Species or species habitat may occur within area	√	Species or species habitat may occur within area	√	Species or species habitat known to occur within area
Numenius minutus	Little Curlew	Migratory	X	X	N/A	X	N/A	~	Foraging, feeding or related behaviour known to occur within area
Numenius phaeopus	Whimbrel	Migratory	X	X	N/A	X	N/A	1	Foraging, feeding or related behaviour known to occur within area
Pandion haliaetus	Osprey	Migratory	×	×	N/A	×	N/A	$\checkmark$	Species or species habitat known to occur within area
Phaethon lepturus	White-tailed tropicbird	Migratory	×	~	Species or species habitat may occur within area	√	Species or species habitat may occur within area	$\checkmark$	Breeding known to occur within area
Phaethon rubricauda	Red-tailed tropicbird	Migratory	×	×	N/A	×	N/A	$\checkmark$	Species or species habitat may occur within area
Pluvialis fulva	Pacific golden plover	Migratory	X	X	N/A	X	N/A	V	Foraging, feeding or related behaviour known to occur within area
Pluvialis squatarola	Grey plover	Migratory, Vulnerable	X	X	N/A	X	N/A	1	Foraging, feeding or related behaviour known to occur within area
Rostratula australis	Australian painted snipe	Endangered	Endangered	X	N/A	x	N/A	√	Species or species habitat may occur within area overfly marine area
Sternula albifrons	Little tern	Migratory	x	X	N/A	√	Breeding known to occur within area	$\checkmark$	Breeding known to occur within area
Tringa brevipes	Grey-tailed Tattler	Migratory	X	X	N/A	X	N/A	~	Foraging, feeding or related behaviour known to occur within area

Marine fauna			<b>Territory Parks</b>		OA		MEVA		EMBA
Scientific name	Common name	EPBC Act 1999	C Act 1999 Conservation Ma Act 1976 Pre	May be present	Values or sensitivities	May be present	Values or sensitivities	May be present	Values or sensitivities
Tringa glareola	Wood Sandpiper	Migratory	X	X	N/A	X	N/A	V	Foraging, feeding or related behaviour known to occur within area
Tringa incana	Wandering Tattler	Migratory	X	X	N/A	X	N/A	1	Foraging, feeding or related behaviour known to occur within area
Tringa nebularia	Common Greenshank, Greenshank	Endangered, Migratory	X	X	N/A	X	N/A	√	Species or species habitat known to occur within area
Tringa stagnatilis	Marsh Sandpiper, Little Greenshank	Migratory	X	X	N/A	X	N/A	V	Foraging, feeding or related behaviour known to occur within area
Tyto novaehollandiae Kimberli	Masked Owl (northern)	Vulnerable	Vulnerable	X	N/A	X	N/A	√	Species or species habitat known to occur within area
Tyto novaehollandiae melvillensis	Tiwi Masked Owl	Endangered	Endangered	X	N/A	X	N/A	$\checkmark$	Species or species habitat likely to occur within area
Geophaps smithii smithii	Partridge Pigeon (eastern)	Vulnerable	Vulnerable	×	N/A	X	N/A	$\checkmark$	Species or species habitat likely to occur within area
Melanodryas cucullata melvillensis	Tiwi Islands Hooded Robin	Critically Endangered	Critically Endangered	×	N/A	X	N/A	$\checkmark$	Species or species habitat likely to occur within area
Xenus cinereus	Terek Sandpiper	Migratory, Vulnerable	X	X	N/A	X	N/A	√	Foraging, feeding or related behaviour known to occur within area





#### Table 3-11: Summary of turtle species usage of Tiwi Island beaches and waters

Species	Use of Tiwi Isla	nds beaches and	waters			
	Nesting	Inter-nesting	Migration	Foraging	Overlap with OA	Overlap with EMBA
Olive ridley	Yes	Yes	Yes	Unknown	No	Yes (migration)
Flatback	Yes	Yes	Yes	Yes (WA, NT nesting stocks)	Yes (internesting)	Yes (migration)
Green	Yes	Yes*	Yes	Yes (WA, Scott Reef, NW Shelf, Ashmore stock, NT stocks)	No	Yes (migration)
Hawksbill	Yes	Yes*	Yes	Unknown	No	Unknown
Loggerhead	No	No	Yes	Yes	No	Yes (migration)
Leatherback	No	No	Unknown	Unknown	No	Unknown

\* assumed based on nesting

### 3.2.13.1 Marine mammals

#### 3.2.13.1.1 Whales

#### Bryde's whale

Bryde's whales (*Balaenoptera edeni*; Migratory) are distributed across tropical and warm temperate waters with individuals recorded in all Australian states, except the NT (Ceccarelli et al., 2011). The species typically moves between 40 °N and 40 °S, with these movements seeming to be primarily linked to prey availability (Kato, 2002). Bryde's whales are thought to be divided into offshore and onshore forms with the distinction between the 2 based on prey preference (Ceccarelli et al., 2011). The offshore form is found in deeper waters (500 to 1,000 m) and is thought to migrate seasonally in favour of warmer waters in winter months. The onshore form generally inhabits waters over 200 m and displays no distinct migratory movements (Jenner et at., 2001). A noise monitoring study undertaken for the Barossa project detected Bryde's whales almost year-round from January to October (McPherson et al., 2016) and this species has been encountered off Browse Island (Ceccarelli et al., 2011). Bryde's whales may occasionally transit through the EMBA in small numbers.

#### Fin whale

Fin whales (*Balaenoptera physalus*; Vulnerable under the EPBC Act; Migratory) are widely distributed from polar to tropical waters and have been recorded in all Australian states, other than the NT (Bannister et al., 1996). Fin whales feed on planktonic crustacea, such as Antarctic krill, and primarily forage in high latitudes.

The species rarely occupies inshore waters and displays well defined migratory movements (essentially north south) between polar, temperate, and tropical waters (Ceccarelli et al 2011; Bannister et al. 1996). Research by Aulich et al. (2022; 2019) found that fin whales migrate along the WA coast from Cape Leeuwin to as far north as Dampier (19°S). After arriving at Cape Leeuwin in April, the species migrates north along the coast to the Perth Canyon, where they are suggested to aggregate and feed from May to October. This is thought to be a part of the migratory pathway of the species from Eastern Antarctic waters to Australian waters. Furthermore, it has been suggested that there are separate fin whale sub-populations on the east and west coasts of Australia (Aulich et al., 2022; 2019). Within Australian waters, the Bonney Upwelling is thought to be an important foraging ground for this species (TSSC, 2015c; Bannister et al., 1996).

The Australian fin whale distribution is unclear due to limited observations, but the species is thought to be present from Exmouth along the southern coastline to Qld. There are no known mating or calving areas in Australian waters and no BIAs have been developed for fin whales (TSSC, 2015c). Given their distribution and movements, fin whales are unlikely to occur in the EMBA.

#### Humpback whale

The humpback whale (*Megaptera novaeangliae*; Migratory) has a wide distribution with recordings throughout Eastern Antarctic waters and offshore from all Australian states (IUCN-MMPATF, 2023b; Bannister et al., 1996). These whales migrate between summer feeding grounds in Antarctica and winter breeding and calving grounds in the sub-tropical and tropical inshore waters of north-west Australia (Jenner et al., 2001). Although the exact timing of migration varies annually due to a number of factors including water temperature, the northbound migration peaks between late July and early August, and the southbound migration peaks between late August and early September (Jenner et al., 2001).



There has been a steady recovery in the humpback whale population that migrates along the WA coast since the closure of commercial whaling, and as a result the species was removed from the EPBC Act threatened species list in 2022 (TSSC, 2022).

Humpback whales breed and calve in the NWMR between Broome and the northern end of Camden Sound in the months of June to September each year (DCCEEW, 2024j) and a breeding and calving BIA for humpback whales is recognised in nearshore waters adjacent to the northern half of the Dampier Peninsula and encompasses Camden Sound (DCCEEW, 2024j).

Relatively few humpback whales have been known to travel north of Camden Sound (Jenner et al., 2001) and noise monitoring undertaken for the Barossa project did not detect any humpback whale calls in the Timor Sea (McPherson et al., 2016).

There are no BIAs for this species within the EMBA and given the available information on its distribution, it is considered unlikely to occur within the EMBA.

#### Pygmy blue whale

The blue whale (Balaenoptera musculus; Endangered under the EPBC Act; Migratory) has four distinct subspecies, two are found in the southern hemisphere—the pygmy blue whale (Balaenoptera musculus brevicauda; Indo-Australian and Tasman-Pacific populations) and Antarctic blue whale (Balaenoptera musculus intermedia; CoA, 2015a). As southern blue whales occur in waters south of 60°S and pygmy blue whales north of 55°S (CoA, 2015a), only pygmy blue whales are discussed below.

The pygmy blue whale is known to migrate along the WA shelf edge at depths between the 500 m and 1,000 m depth contours from the NW Cape south to Geographe Bay (CoA, 2015a). A biologically important migration corridor is recognised in the deep offshore waters off WA (IUCN-MMPATF, 2023a). The northerly migration toward the calving grounds near the equator occurs in March/April to June (Thums et al., 2021; CoA, 2015a). Noise monitoring for the Barossa project in the Timor Sea detected the presence of blue whales over 400 km north-east of the migration BIA for the species in the months of May to August during their north-bound seasonal migration. No detections of the species were made during the period of their southward migration (McPherson et al., 2016). The southerly migration to the feeding grounds in the high latitudes of the southern hemisphere occurs in September to December (CoA, 2015a). Pygmy blue whales appear to travel as individuals or in small groups when making their migrations (Woodside, 2014). Telemetry data collected from tagged pygmy blue whales indicates their migration patterns, highlighting a journey from the coastal waters of Western Australia to the surrounding marine regions of Indonesia (Sahri et al. 2022)

Generally, this species travels alone or in small groups based on acoustic data. Pygmy blue whale calls from noise loggers deployed around Scott Reef from 2006 to 2009 for the Woodside Browse project found 78% of calls to be from single whales, 18% from whale pairs and 4% from 3 or more whales (Woodside, 2014).

Possible foraging areas for pygmy blue whales in the wider region include Scott Reef off the northern coast of Western Australia (CoA (2015a), and along the Timor Trough (Burton et al. 2023; Ferreira et al. (2024). These areas are likely to provide important feeding grounds for the species, supporting their migratory and ecological needs. The steep gradient features of the Scott Reef location tend to stimulate upwelling and thus increased productivity (seasonally variable) (ConocoPhillips, 2018). There are no known breeding areas of significance to blue whales in the EMBA.

There are no BIAs for pygmy blue whales identified within the EMBA and, if present, are likely to be transient and in low numbers.

#### Sei whale

Sei whales (*Balaenoptera borealis*; Vulnerable under the EPBC Act; Migratory) are thought to have a wide distribution, but their distribution limits are unclear as this species is often confused with Bryde's whales. Sightings are rare, but the species may be seen in coastal and offshore waters throughout Australia (DCCEEW, 2024j; Bannister et al., 1996). The species is able to utilise a diverse range of marine habitats, which has been attributed to a combination of dynamic physical and prey processes (DCCEEW, 2024j).

Sei whale migratory movements are well defined with distinct north-south movements as the species migrates between polar, temperate, and tropical waters for foraging and breeding. The species feeds intensively between the Antarctic and sub-tropical convergences on planktonic crustacea (Ceccarelli et al., 2011; Bannister et al., 1996). There are no known mating or calving areas in Australian waters and the species is thought to infrequently occur in the NW region (Ceccarelli et al., 2011).

There are no BIAs for this species in Australian waters. However, it is possible that individual sei whales may occasionally occur within the EMBA.



#### 3.2.13.1.2 Dolphins

#### Australian humpback dolphin

The Australian humpback dolphin (*Sousa sahulensis*; Migratory and vulnerable, previously/also known as the Indopacific humpback dolphin, *Sousa chinensis*) occurs in water of the Sahul Shelf, from northern Australia to the Kikori Delta in Papua New Guinea, and Bird's Head Seascape in West Papua (Jefferson & Rosenbaum, 2014). Although distribution, life history and habitat preferences of this species are poorly understood, the Australian humpback dolphin is thought to be associated with shallow coastal, estuarine, and tidal river waters less than 20 m in depth (Hanf et al., 2022).

In Australia, humpback dolphins occur along the northern Australian coastline from Shark Bay in WA to southern Queensland (Raudino et al., 2018; Hanf et al., 2022). In the NWMR, this species is thought to inhabit coastal waters up to the 30 m isobath (Hanf et al., 2022), but Australian humpback dolphins have been recorded up to 60 km offshore near Barrow Island, the Montebello Islands (approximately 80 km from the mainland coast and 20 km from Barrow Island), and the western Lowendal Islands (Raudino et al., 2018). Available abundance estimates indicate that this species occurs in small populations with an average of up to 89 individuals and a maximum of 0.19 individuals per km<sup>2</sup> (Parra & Cagnazzi, 2016).

There are breeding BIAs for Australian humpback dolphins that overlap the EMBA.

#### Australian snubfin dolphin

The Australian snubfin dolphin (*Orcaella heinsohni*; Migratory and vulnerable), previously known and only recently differentiated from the closely related Irrawaddy dolphin (*Orcaella brevirostris*), is a poorly known species inhabiting shallow coastal and estuarine waters and tidal rivers. The species typically occurs in water depths of less than 20 m in the vicinity of freshwater outflows but has been recorded up to 23 km offshore (Bouchet et al., 2021). The Australian snubfin dolphin is likely to occur in higher densities in areas of complex habitat type which provide a variety of prey types (Palmer et al., 2014).

In Australia, this species occurs in coastal waters of Qld, NT, and north-western Australia. The population in Australian waters is thought to be continuous with the Papua New Guinea species but separate from populations in Asia. Breeding is thought to occur throughout the year for this species. There are no BIAs that overlap the EMBA.

#### Spotted bottlenose dolphin (Indo-pacific bottlenose dolphin)

The spotted bottlenose dolphin (Arafura/Timor Sea populations; *Tursiops aduncus*; Migratory) is primarily found in nearshore continental shelf waters less than 200 m deep, with rocky or coral reefs, sandy, soft sediments, or seagrass beds (DSEWPaC, 2012a). Small populations also occur in the inshore waters of some oceanic islands (Ceccarelli et al., 2011).

In Australia, migration patterns for the species are variable, including year-round residency in small areas, longrange movements and migration (DCCEEW, 2023j). The species occurs in NT open coastal waters, primarily within the continental shelf and around oceanic islands. Spotted bottlenose dolphins forage in a wide range of habitats and in deeper waters than most dolphins. Groups are resident at Browse Island, Rowley Shoals and other island and reef complexes in offshore waters (Ceccarelli et al., 2011). There are no BIAs for this species within the EMBA

### Orca

The largest member of the dolphin family, killer whales, or orca (*Orcinus orca*; Migratory) are a cosmopolitan species with a vast global distribution across a wide range of habitats. However, they appear to be primarily concentrated in coastal waters and cooler regions of high productivity as they are carnivores with a diet that varies seasonally and regionally (DCCEEW, 2023); Bannister et al., 1996).

Globally, killer whales are known to migrate; however, specific routes and seasonal movement patterns are not known in detail and are thought to relate to prey availability (Bannister et al., 1996).

Killer whales are distributed throughout Australian waters, typically observed moving along the continental slope and shelf, and near seal colonies (Bannister et al., 1996). Migration movements within Australian waters include a summer migration from subantarctic islands to Macquarie Island (DCCEEW, 2023j). While killer whales are known to undertake seasonal migrations and follow regular migratory routes, little is known about these movements (DCCEEW, 2023j).

Killer whales are often observed around seal colonies and may be associated with humpback whale migrations, neither of which occur in the vicinity of the EMBA. No BIAs or migration routes have been identified for this species within the EMBA, although they may occur in low numbers.

#### 3.2.13.1.3 *Dugong*

Dugongs (*Dugong dugon*; Migratory) occur in tropical and sub-tropical coastal and inland waters. They are commonly found in shallow areas to 25 m depth but have been observed in waters up to 37 m deep (Marsh, n.d.). Dugong feeding aggregations tend to occur in large seagrass meadows within wide shallow protected bays,



shallow mangrove channels and in the lee of large inshore islands. Although the movements of most individuals are limited to tens of kilometres in the vicinity of seagrass beds some individuals travel up to 1,000 km (Hobbs & Willshaw, 2015; Whiting, 2008).

Dugongs in the Torres Strait have large home-range sizes compared to other regions, likely due to the vast areas of seagrass, including over 13,000 km<sup>2</sup> of deep-water seagrass, the largest continuous area in Australia (Deutsch et al., 2022). This, along with large seagrass beds in shallow water around reefs, enables dugongs to travel long distances while staying relatively close to accessible food sources (Deutsch et al., 2022).

In northern Australia, the Darwin region supports a dugong population travelling over 300 km between rocky reef habitats (Whiting, 2008), and key sites for dugong conservation have been identified around Cobourg Peninsula, Croker Island and the north coast of the Tiwi Islands (PWSNT, 2003). Aggregations at these sites rank in the top eight dugong populations in Australia (PWSNT, 2003). Dugongs tracked in the INPEX Ichthys Project baseline surveys were recorded around the Vernon Islands, south of Melville Island, and spent time in Darwin Harbour and around the Tiwi Islands (INPEX, 2010).

There are no BIAs for dugong within the EMBA, but the species is likely to occur in suitable habitats.

### 3.2.13.2 Marine reptiles

#### 3.2.13.2.1 Marine turtles

#### Flatback turtle

Flatback turtles (*Natator depressus*; Vulnerable under the EPBC Act; Migratory) are known to occur along the WA, NT and Queensland coastlines, and forage widely across the Australian continental shelf and into the continental waters off Indonesia and Papua New Guinea (CoA, 2017b). Flatback turtles are primarily carnivorous, predominantly feeding on soft-bodied invertebrates. This species breeds in the region, with the highest density rookeries found to be winter at Cape Domett and summer at Eighty Mile Beach, while moderate to lesser density nesting in winter occurred in the North Kimberley offshore islands (Tucker et al., 2021). Flatback turtles that nest within the Pilbara region typically migrate along the continental shelf to foraging grounds as far north as Darwin at the end of the nesting season, returning to breed at varying intervals of a year or more (Thums et al., 2020; CoA, 2017b). Tracking studies have shown individuals migrating from northern WA into Queensland waters and (conversely) from Deliverance Island in Queensland to Kimberley waters, with the waters around the Tiwi Island supporting migrating and foraging flatbacks (Pendoley, 2023).

Flatback turtles nesting within the NT are from the Arafura Sea breeding and genetic stock, with unknown longterm trends for this stock (CoA, 2017b). Nesting has been recorded on the Tiwi Islands, with flatback turtles the predominant nesting species on the southern and south-western beaches that fall within the EMBA (Pendoley, 2023). The greatest proportion of activity occurs on the west coast of Bathurst Island (Chatto & Baker, 2008) with nesting females numbering around 11 to 100 per year, which is comparable to or smaller than other nesting sites of the Arafura Sea genetic stock. Nesting and internesting occurs year-round with a peak from June to September, and hatchling emergence peaking between July and September (CoA, 2017b).

The Recovery plan for marine turtles in Australia defines a 60 km internesting buffer around the Tiwi Islands (CoA, 2017b). Whittock et al. (2016) defined suitable internesting habitat as waters up to 16 m deep within 5 to 10 km of the coastline, and unsuitable internesting habitat as waters over 25 m deep and more than 27 km from the coastline. They also tracked internesting flatback turtles from 5 different mainland and island rookeries and found that these turtles not only stayed in waters less than 44 m deep but were associated with a mean depth of under 10 m (Whittock et al., 2016). To date there is no evidence indicating flatback turtles in deep offshore waters during the internesting period (Pendoley, 2019). There are BIAs for flatback turtle internesting within the OA and EMBA.

#### **Green turtle**

Green turtles (*Chelonia mydas*; Vulnerable under the EPBC Act; Migratory) are predominately found off the WA, NT, and Queensland coastlines (CoA, 2017b). The green turtle is the most common marine turtle breeding in the NWMR, with WA supporting one of the largest remaining populations worldwide (DSEWPaC, 2012b). Green turtles travel up to 3,100 km between nesting and feeding areas (Ferreira et al., 2021; DSEWPaC, 2012b) and forage on algae, seagrass and mangroves, including on offshore coral reefs across northwestern Australia (Ferreira et al., 2021; CoA, 2017b).

In the NT, nesting sites occur mostly from the western end of Melville Island to near the Queensland border (NT Government, n.d). The Cobourg Peninsula green turtle genetic stock is the closest to those on the Tiwi Islands and they nest between October and April, with peak nesting period between December and January. Nesting in the Tiwi Islands includes the beaches within the EMBA on the south-west of Bathurst Island (Chatto & Baker, 2008; Pendoley, 2023). Nesting sites for the species in the Bonaparte or Van Diemen bioregions are Black/Smith Point and Lawson Island, east of the Tiwi Islands near Cobourg Peninsula (Chatto & Baker, 2008).



Green turtles are likely to be encountered within the EMBA, mainly within reef areas, and internesting is expected between October and April (CoA, 2017b). There is a BIA for green turtle foraging within the EMBA and critical habitat for green turtles overlaps the waters of the EMBA.

#### Hawksbill turtle

Hawksbill turtles (*Eretmochelys imbricata*, Vulnerable under the EPBC Act and TPWC Act: Migratory) predominately occur along northern Australian coastlines (WA, NT and Queensland), with 3 recognised stocks: north Queensland stock located in the north Great Barrier Reef and Torres Strait; north-east Arnhem Land stock in the NT; and WA stock located on the North West Shelf. On a global scale, WA provides one of the largest remaining hotspots for this species, and these migrating hawksbill turtles traverse shallow continental-shelf waters less than 200 m deep following the coastline and a migratory corridor along the Pilbara coast (Fossette et al., 2021). Hawksbill turtles are omnivorous and feed on algae, sponges, soft corals, and soft bodied invertebrates foraging in waters ranging from 1.5 to 84 m deep (Fossette et al., 2021). This species is typically associated with rocky and coral reef habitats, often returning to a small foraging area, and is expected to be found within these habitats along the WA coastline, from Shark Bay to the northern extent of the NWMR, migrating over 4,600 km from their nesting site (Crommenacker et al., 2022; Barr et al., 2021; CoA, 2017b). Unlike green turtles, there is little evidence that hawksbill turtles nesting elsewhere in WA, NT, or Queensland migrate to the Tiwi Islands to forage (Pendoley, 2023) and the islands are not listed as an important nesting, foraging, or internesting site for this species (CoA, 2017b).

In the NT, nesting occurs on islands concentrated around north-eastern Arnhem land and Groote Eylandt (NT Government, n.d) and is reported to occur from July to December (DSEWPaC, 2012b). Nesting on the Tiwi Islands has been recorded at Seagull Island and northern Melville Island (Chatto & Baker, 2008). There are no BIAs for the hawksbill turtle within the EMBA.

#### Leatherback turtle

Leatherback turtles (*Dermochelys coriacea*; Endangered under the EPBC Act; Critically Endangered under the TPWC Act; Migratory) are known to forage and migrate throughout the open offshore waters of Australia, with foraging more common along the east coast and Bass Strait. Leatherback turtles are pelagic throughout their life and feed almost exclusively on jellyfish. Records of leatherback turtles nesting in Australia are sparse, and limited to Queensland, NSW, and NT (DCCEEW, 2024j; CoA, 2017b), with scattered isolated nesting (one to 3 nests per year) in Qld and the NT (Limpus & McLachlin, 1994). Due to the lack of significant nesting sites in Australian waters, leatherback turtles are likely migrants from neighbouring countries foraging in Australia (Limpus, 2009). There are no BIAs for the leatherback turtle within the EMBA.

#### Loggerhead turtle

Loggerhead turtles (*Caretta caretta*; Endangered under the EPBC Act; Vulnerable under the TPWC Act; Migratory) range along most of the Australian coastline and throughout the NWMR (CoA, 2017b). This species is carnivorous and mainly feeds on benthic invertebrates in a wide range of habitats from nearshore to waters 55 m deep (CoA, 2017b). Breeding aggregations occur on Australia's east (Queensland, NSW) and west coasts. Loggerhead turtles have one genetic breeding stock within WA, with approximately 3,000 females supporting the third-largest population in the world (CoA, 2017b; Limpus, 2008; Baldwin et al., 2003).

Capable of large migrations, individual loggerhead turtles from both WA and eastern Australian waters have been recorded foraging in the NT, and further afield in Indonesia and Papua New Guinea (Perez et al., 2022; Pendoley, 2023). In the Kimberley region, loggerhead turtles are thought to be transient or end-of-migration foragers with no documented nesting sites in the area (Tucker et al., 2021). Although loggerhead turtles forage in the Oceanic Shoals Marine Park, the Arafura Sea, and the Gulf of Carpentaria, they are not known to breed in the region. Loggerheads found within the EMBA most likely come from the WA population, nesting outside the EMBA (CoA, 2017b). No BIAs for loggerhead turtles intersect the EMBA.

#### Olive ridley turtle

Olive ridley turtles (*Lepidochelys olivacea*; Endangered under the EPBC Act; Vulnerable under the TPWC Act; Migratory) are known to nest in the NT and on western Cape York (Queensland), with low density nesting recorded on the Kimberley coast, in the Dampier Peninsula and along Camden Sound (Tucker et al., 2021; CoA, 2017b). This species is primarily carnivorous and feeds on soft-bodied invertebrates in waters between 15 m and 200 m in depth. Olive ridley turtles migrate through oceanic waters, travelling up to 1,130 km between their nesting and foraging grounds (Cáceres-Farias et al., 2022; CoA, 2017b; Whiting et al., 2005). All reported olive ridley movements were largely restricted to within the 100 m depth contour (Pendoley, 2023).

Olive ridley turtles are known to nest on the Tiwi Islands on the west coast of Bathurst Island and the north coast of Melville Island. These turtles are part of the NT genetic stock, significant at both a national and international level (CoA, 2017b). The NT genetic stock nests throughout the year, with peaks between April and June, and most hatchlings emerge between June and August (CoA, 2017b).



Internesting habitat for this species encompasses nearshore waters along the north, west and east coasts of the Tiwi Islands. Tracking studies showed these turtles remain close to shore in waters less than 55 m deep within 37 km of the nesting beach during the internesting interval (Whiting et al., 2007; 2005). Migrating olive ridley turtles tracked from the Tiwi Islands typically moved in a northeast and west/south-westerly direction, to foraging grounds ~300–400 km to the west in the Joseph Bonaparte Gulf or up to 1,200 km away in the Arafura Sea and Gulf of Carpentaria (Pendoley, 2023). Olive ridley turtles may be encountered in the shallow waters of the Tiwi Islands, with BIAs for foraging and nesting intersecting the EMBA.

### 3.2.13.2.2 Crocodiles

The salt-water crocodile (*Crocodylus porosus*; Migratory) were listed under the EPBC Act to regulate commercial hunting which caused a significant decline in the population (DCCEEW, 2024j). Salt-water crocodiles are found across northern Australia and occur within the nearshore marine and estuarine waters of the Kimberley coast (DCCEEW, 2024j). Larger populations within the major river systems of the Kimberley occur in the rivers draining into the Cambridge Gulf, the Prince Regent and Roe River systems of the east and northwest Kimberley (DCCEEW, 2024j). There is limited availability of nesting habitat for this species within its distribution, with only the Ord, King and Roe River systems typically providing suitable nesting vegetation for the species (DCCEEW, 2024j). There are no BIAs for the salt-water crocodile within the EMBA, but given their widespread distribution, they are likely to be present within the EMBA.

#### 3.2.13.2.3 Sea snakes

Sea snakes are typically distributed in shallow inshore regions and the Tiwi Islands, which provide suitable seabed habitat and clear waters. However, they are also found further offshore at atolls, including the shoals/banks in the Timor Sea (Guinea, 2013).

Most sea snakes are observed in water depths ranging between 10 and 50 m (RPS, 2010) and generally have shallow, benthic feeding patterns. Some species are known to dive deeper than this, but non-pelagic species seldom, if ever, dive deeper than 100 m (Heatwole, 1975). Very few species are known to inhabit deep pelagic environments, given they are air-breathing (Guinea, 2006).

Distribution and movements of sea snakes are largely species-dependent with some species, such as the pelagic yellow-bellied sea snake, known to travel large distances, while others, such as the olive sea snake, usually reside in a particular area.

### 3.2.13.3 Sharks, rays, and other fish

#### 3.2.13.3.1 Sharks

#### Grey nurse shark

The grey nurse shark (*Carcharias taurus*; Vulnerable and migratory under the EPBC Act) has a wide but patchy tropical and temperate distribution in the Indo-West Pacific and Atlantic oceans. There are 2 distinct subpopulations in Australia on the east and west coast. The west coast population inhabits coastal and continental shelf waters from south west Western Australia (WA) (Albany) up to the North West Shelf (FRDC, 2019) and although one aggregation site has been documented, data on their distribution along the WA and NT coastline is lacking (Hoschke et al., 2023). Grey nurse sharks undertake large-scale movements to potentially capitalise on seasonal prey aggregations, with individuals migrating 1,294 km along the WA coast from SW WA to Ningaloo, and 1,500 km on the east coast (Dwyer et al., 2023; DCCEEW, 2024j; Jakobs et al., 2019). Grey nurse sharks are thought to move further north along the coast when from May to December with lower sea temperatures. Individuals have been caught near Browse Island and off Bali, Indonesia (Hoschke et al., 2023; Momigliano & Jaiteh, 2015). During the Barossa marine studies program, 4 grey nurse sharks were observed at seamounts in waters 130 m deep, one possibly pregnant (Jacobs, 2016). This was considered unusual as neither of the subpopulations are known to extend that far north and are generally associated with shallower, more coastal waters (DCCEEW, 2024j). Given grey nurse sharks have been observed at seamounts and oceanic coral reefs in the Timor Sea, the species may be present around reefs, banks, and seamounts in the EMBA.

#### Mako sharks

Shortfin mako (*Isurus oxyrinchus*; Migratory) and longfin mako (*Isurus paucus*; Migratory) sharks are both highly migratory epipelagic species. The shortfin mako is a common shark in tropical and temperate waters above 16 °C (Groeneveld et al., 2014), and as such widespread throughout Australian waters except for the Torres Strait, Arafura Sea and Gulf of Carpentaria (FRDC 2019; Birkmanis et al., 2020; Kyne et al., 2021a). Shortfin mako sharks exhibit sexual and developmental segregation; juveniles spend 90% of their time near the surface whereas adults dive much deeper (Groeneveld et al., 2014). In contrast, the wide but patchy distribution and biology of the rarely encountered longfin mako is less well documented (Kyne et al., 2021a). This epipelagic shark also inhabits tropical and warm-temperature waters. In Australia, longfin mako sharks are found from Geraldton in WA across the Northern Territory and Queensland down to Port Stevens in NSW (FRDC, 2019; Rigby et al., 2019). These species may be rarely encountered within the EMBA.



#### Oceanic whitetip shark

The oceanic whitetip shark (*Carcharhinus longimanus*; Migratory) is a highly mobile globally widespread species found in tropical and warm temperate waters between 18 to 28°C from the surface to at least 180 m, venturing close to shore where the continental shelf is narrow (Kyne et al., 2021a). Within Australian waters, this rarely encountered species is found in warmer waters from Cape Leeuwin in WA across northern Australia down to Sydney (Kyne et al., 2021a). Oceanic whitetip sharks have been globally assessed as Critically Endangered by the IUCN, overfished by SAFS, and listed on CITES Appendix II (FRDC, 2019). It is possible that individuals of this species may be encountered within the EMBA.

#### Northern river shark

Northern river sharks (*Glyphis garricki*; Endangered under the EPBC Act and TPWC Act) are rare and although their distribution is uncertain, they are known to occur in the Ord and King Rivers, King Sound and Joseph Bonaparte Gulf in WA, along with the South and East Alligator Rivers and the Wessel islands in NT (Udyawer et al., 2021; FRDC, 2019; DSEWPaC, 2010). These sharks are thought to segregate during various life stages, occupying rivers, estuarine systems, macrotidal embayments as well as inshore marine habitats (Kyne et al., 2021a; FRDC, 2019; DSEWPaC, 2010). Although the northern river shark has been recorded in offshore waters, the frequency of this occurrence is unknown.

The Sawfish and River Shark Multispecies Recovery Plan (CoA, 2015b) recorded observations of adults and juveniles in marine waters north of Derby, WA while pupping and juveniles occur in King Sound and Cambridge Gulf. Under the recovery plan, all aggregations, and areas of biologically important behaviours such as breeding, foraging, resting, or migrating are considered critical to the survival of the species. Individuals may be encountered in low numbers within the EMBA.

#### Speartooth shark

The speartooth shark (*Glyphis glyphis*; Critically Endangered under the EPBC Act; Vulnerable under the TPWC Act) has been recorded as occurring in macrotidal rivers and estuary environments, with juveniles and sub-adults utilising large tropical river systems as their primary habitat (Kyne et al., 2021b; Stevens et al., 2005). It is thought that their marine distribution may be limited to the coastal marine environment outside of rivers (Udyawer et al., 2021; FRDC, 2019). While the speartooth shark is known to inhabit the Wenlock/Ducie/Port Musgrave river system in Qld and various rivers of the Van Diemen Gulf in the NT, new populations of this species were recently discovered in the Daly River, NT and the Ord River, WA (Kyne et al., 2021b). It has been recorded in tidal rivers and estuaries with turbid waters with fine muddy substrates in temperatures ranging from 27 to 33 °C (Pillans et al., 2009).

Remaining populations throughout Australia are considered isolated with questionable viability. Both species were listed as threatened in 2001 due to their limited geographical distribution and low population estimates, and the population decline is likely to continue (DSEWPaC, 2010).

#### Scalloped hammerhead shark

The scalloped hammerhead shark (*Sphyrna lewini*; Conservation Dependent under the EPBC Act) is a coastal and semi-oceanic species globally distributed in tropical and warm-temperate waters from the intertidal zone to at least 275 m in depth, with newborns found in coastal zones (Kyne et al., 2021a; FRDC, 2019). Recent studies suggest that the Indo-Pacific population (including Australia) is genetically distinct from the Atlantic and Caribbean populations. There is likely to be 2 subpopulations in Australian waters (WA and the rest of Australia), with the non-WA subpopulation connected to Papua New Guinea and Indonesia by shallow water habitats along northern Australia (Green et al., 2022). Across northern Australia, the pupping season peaks from October to January (TSSC, 2018). This mobile species has a broad Australian range from NSW and Qld across the NT to WA (Bartes & Braccini, 2021; Kyne et al., 2021a; FRDC, 2019). Scalloped hammerhead sharks are known to occur within the EMBA.

#### White shark

The white shark (*Carcharodon carcharias*; Vulnerable under the EPBC Act; Migratory) is primarily a temperate species with a wide Australian range and 2 subpopulations; eastern Australasia (from Papua New Guinea along Australia's east coast and Macquarie Island to the south-western Pacific, including waters off New Caledonia, Vanuatu and Tonga) and a southern-western population (from western Victoria across southern Australia and up the WA coast; DSEWPaC, 2013; FRDC, 2019; Kyne et al., 2021a). Although the species has been recorded south from central Qld to up to Ningaloo Reef and may occur further north on both coasts, white sharks are not known to aggregate within the NWMR or NMR and are most likely to be found south of North West Cape (DSEWPaC, 2012a; 2012c). The reasons for movements to north-western WA are unknown and little information is available on their reproduction in Australian waters (McAuley et al., 2016; DSEWPaC, 2012c). White sharks are unlikely to be seen in the EMBA.



#### 3.2.13.3.2 Rays

#### Manta ray

The giant manta ray (*Mobula birostris*; Migratory) and reef manta ray (*Mobula alfredi*; Migratory) are globally distributed in both tropical and temperate waters. Giant manta rays are considered to be the more migratory and oceanic species of the 2, and individuals of this highly mobile species are not expected to be resident in Australian waters (Kyne et al., 2021a; Couturier et al., 2015). While considered more solitary and less frequently sighted than reef manta rays, giant manta rays can be found in large numbers engaging in foraging, mating, or cleaning activities and exhibit seasonal habitat preferences frequenting offshore seamounts and islands (Marshall et al., 2022a).

The reef manta ray typically utilises productive nearshore habitats, including island groups, atolls and continental coastlines (Marshall et al., 2022b), and is coastally distributed across the north of Australia to approximately 30°S on both coasts (Armstrong et al., 2020). While reef manta rays demonstrate a high degree of site fidelity in tropical and subtropical waters, this species has also been shown to travel up to 700 km, undertake seasonal migrations and traverse international waters (Couturier et al., 2015). Reef manta rays are often sighted in high numbers, predominately when undertaking foraging activities or migrating. There are no known foraging or breeding aggregation areas for these species within the EMBA. Based on the habitat preferences of these rays, it is unlikely that either species would occur in large numbers although individuals may transit through the area.

#### 3.2.13.3.3 Other fish

#### **Dwarf sawfish**

The dwarf sawfish (*Pristis clavata*; Vulnerable under the EPBC Act and TPWC Act; Migratory) is primarily found in shallow coastal and estuarine areas, from Cairns in Queensland around the north of Australia to the Pilbara coastline in WA, with juveniles thought to remain in estuarine waters (FRDC, 2019; DEWHA, 2009). Sawfishes feed on a variety of teleost fishes and benthic invertebrates, including cephalopods, crustaceans, and molluscs (Lear et al., 2023; Thorburn et al., 2007; 2008; Pogonoski et al., 2002).

#### Green sawfish

The green sawfish (*Pristis zijsron*; Vulnerable under the EPBC Act and TPWC Act; Migratory) is most common in shallow coastal and estuarine areas, but this species has been recorded in water depths of up to 70 m from Cairns, Queensland across to Broome, WA (FRDC, 2019; DEWHA, 2008a). Green sawfish appear to have limited tidally influenced movements, occupying only a few square kilometres within the coastal fringe, and strongly associated with mangroves and adjacent mudflats (Lear et al., 2023). Baseline surveys for Chevron's Wheatstone project identified green sawfish habitat and juvenile nursery areas within the north-eastern lagoon of the Ashburton Delta and in Hooley Creek near Onslow. Although their spatial and temporal distribution in these creeks is variable with changing tidal and environmental conditions, they typically return to inshore waters to breed and pup during the wet season (Chevron, 2011). Sawfishes feed on a variety of teleost fishes and benthic invertebrates (Lear et al., 2023; Thorburn et al., 2007; 2008; Pogonoski et al., 2002).

#### Largetooth sawfish

The largetooth sawfish (*Pristis pristis*; Vulnerable under the EPBC Act; Migratory) inhabits the sandy or muddy bottoms of river, estuarine and marine environments within north-west Australia and has a patchy distribution including the Fitzroy, Durack, Robinson and Ord rivers in WA. Newborns and juveniles occur primarily in the freshwater areas of rivers and in estuaries, while adults mostly occupy marine and estuarine environments (FRDC, 2019; DSEWPaC, 2012c).

#### Whale shark

The whale shark (*Rhincodon typus*; Vulnerable under the EPBC Act; Migratory) is globally distributed in tropical and warm temperate seas, except the Mediterranean. There are 2 distinct subpopulations, with approximately 75% of the global population in the Indo-Pacific, and the remaining 25% in the Atlantic Ocean (Vignaud et al., 2014 in FRDC, 2019). Ningaloo Reef in WA is a known aggregation site, and whale sharks congregate off Christmas Island from December to January. These aggregations are thought to be linked to seasonal prey fluctuations (TSSC, 2015g). The species is an epipelagic filter feeder with a diet of planktonic and nektonic species, including small crustaceans and smaller schooling fish species (DCCEEW, 2024j). Whale sharks are known to be highly migratory with migrations of over 20,000 km recorded (Guzman et al., 2018). Migration along the northern WA coastline broadly follows the 200 m isobath and typically occurs between July and November (TSSC, 2015g).

Wilson et al. (2006) recorded 6 whale sharks departing Ningaloo Reef and traveling north-east into the Indian Ocean. Meekan and Radford (2010) showed that whale sharks migrated up the coast from Ningaloo Reef and individually dispersed over a broad area; either north-west into the open Indian Ocean, northward towards Sumatra and Java, or north-east towards the Timor Sea; and Thomson et al., (2021) more recently recorded whale sharks tagged in Ningaloo Reef traveling to the North West Shelf. Due to their widespread distribution, highly migratory whale sharks may occur in low numbers within the EMBA.



### 3.2.13.4 Seabirds and shorebirds

#### 3.2.13.4.1 Threatened species

#### Alligator Rivers yellow chat

The Alligator Rivers yellow chat (*Epthianura crocea tunneyi*; Endangered under the EPBC Act and TPWC Act) is a small insectivorous bird that occurs mostly within the Kakadu National Park. The species' range and numbers are thought to have declined after habitat loss from cattle grazing, and habitat degradation caused by feral pigs and water buffalo. Its total population size is now very small, only around 100 individuals. (National Environmental Science Program Threatened Species Research Hub, 2019). Historically this species inhabits coastal grassy floodplains, however sightings have become rare and anecdotal. It is thought likely that there are small, undiscovered groups of chats, but that the overall population is still likely to be very small and to have suffered decline over time (National Environmental Science Program Threatened Species Research Hub, 2019). Given the areas historically observed to be inhabited by this species, it is unlikely to occur within the EMBA.

#### Asian dowitcher

The Asian dowitcher (*Limnodromus semipalmatus*; Vulnerable under the EPBC Act; Migratory) is a large, distinctive wader with a long neck, long legs, and a long, straight, snipe-like bill (DCCEEW, 2024j). In Australia, this bird is only a regular visitor to coastal areas between Broome and Port Hedland and the Port McArthur tidal wetlands in the Gulf of Carpentaria, arriving from August (DCCEEW, 2024f). It roosts in sheltered coastal environments such as estuarine and intertidal mudflats, lagoons, creeks and saltworks, and feeds on inter-tidal mudflats (DCCEEW, 2024f). Only a small proportion of the non-breeding population arrive in Australia, occasionally recorded in the NT and rarely in western and eastern Australia (DCCEEW, 2024j). In the NT, the Asian dowitcher is found in Darwin and Arnhem Land (DCCEEW, 2024j). No sites of international significance are listed in the NT for this species (Birdlife Australia, 2020). The Asian dowitcher typically leaves north-west Australia by the end of April to return to northern hemisphere breeding grounds (DCCEEW, 2024j; DCCEEW, 2024j). Given the areas historically observed to be inhabited by this species, individuals may seasonally occur within the EMBA.

#### Australian painted snipe

The Australian painted snipe (*Rostratula australis*; Endangered under the EPBC Act and TPWC Act) is a wading bird that has been recorded in wetlands of all Australian states, most frequently recorded in the Murray-Darling Basin and in smaller numbers and less frequently at scattered locations in WA and NT (DCCEEW, 2024j; DEPWS, 2021a). The most northerly breeding records are from near Derby and Taylor's Lagoon, near Broome and at Tarrabool Lake on the Barkly Tablelands. Although this species is only occasionally recorded in northern Australia, it has been recorded in northern WA and NT from McMinns Lagoon near Darwin and Yellow Waters in Kakadu (DCCEEW, 2024j; DEPWS, 2021a; Trainor et al., 2017; Knuckey et al., 2013). While this species generally inhabits shallow terrestrial freshwater and occasionally brackish wetlands and other waterlogged areas, the Australian painted snipe requires shallow wetlands with areas of bare wet mud and canopy cover nearby for breeding (DCCEEW, 2022a). The PMST report states that this species or habitat may occur within the area. However, as the Australian painted snipe primarily inhabits freshwater wetlands, it is unlikely to occur in the EMBA.

#### **Black-tailed godwit**

Black-tailed godwits (*Limosa limosa*; Endangered under the EPBC Act; Migratory) are found in all states and territories of Australia during the non-breeding (austral summer) season, with coastal regions supporting the highest densities of the species. This bird usually first arrives in north-west Australia from late August, and most have departed the NT by mid April (DCCEEW, 2024e). The largest populations are found on the north coast between Darwin and Weipa (DCCEEW, 2024e). Roosting usually occurs in sheltered bays, estuaries, and lagoons with large intertidal mudflats and/or sandflats. Feeding habitat includes areas of mud or soft, wet sand within sandflats, intertidal mudflats, saltmarshes, and the beaches of oceanic coastlines, bays, and estuaries (DCCEEW, 2024e). Areas of importance to the species in the NT include Darwin Harbour, North Darwin (the Beagle Gulf coastline), Legune Wetlands and Milingimbi Coast, but none of these are considered to have international significance (Birdlife Australia, 2020). Given the areas historically observed to be inhabited by this species, individuals may seasonally occur within the coastline of the EMBA.

#### Common greenshank

The common greenshank (*Tringa nebularia*; Endangered under the EPBC Act; Migratory) is widespread in coastal regions, occurs in all types of wetlands and has the widest distribution of any shorebird in Australia (DCCEEW, 2024h). The species is sparsely scattered through most of the NT (DCCEEW, 2024h), with important areas in the Kakadu National Park, Milingimbi coast, and the south-west coastline of the Gulf of Carpentaria, but no sites of international significance in the NT (Birdlife Australia, 2020). The common greenshank roosts around wetlands, in shallow pools and puddles, or slightly elevated on rocks, sandbanks or small muddy islets (DCCEEW, 2024h). They occur in estuaries and mudflats, mangrove swamps and lagoons (DCCEEW, 2024h). During feeding, the birds pick from the surface (DCCEEW, 2024h) while wading in shallow water along the edge of tidal estuaries, muddy claypans, saltworks and saltpans (DCCEEW, 2024h). The species arrives in Australia from August, with



most leaving by March and April, but some overwintering also occurs (DCCEEW, 2024h). Given the areas historically observed to be inhabited by this species, individuals may seasonally occur within the EMBA.

#### Curlew sandpiper

The curlew sandpiper (*Calidris ferruginea*; Critically Endangered under the EPBC Act and TPWC Act; Migratory) has a broad distribution and has been recorded along the coasts of all Australian states and territories (DCCEEW, 2024j). In WA, curlew sandpipers occur in large numbers at Port Hedland Saltworks, 80 Mile Beach, Roebuck Bay, and Lake Macleod, but is rarely recorded in the north-west Kimberley. In NT, curlew sandpipers mostly occur around Darwin, north to Melville Island and Cobourg Peninsula, and east and south-east to Gove Peninsula, Groote Eylandt, and Sir Edward Pellew Island (TSSC, 2015e). Although the species prefers intertidal mudflats in sheltered coastal areas to forage in nearshore waters or mud at the edge of wetlands, they are also widespread inland in smaller numbers (TSSC, 2015e). The curlew sandpiper migrates along the East Asian-Australasian Flyway (Flyway) from their breeding grounds in Siberia to Australia, generally arriving from late August/early September and departing by mid-April. Some non-breeding individuals may stay in Australia (TSSC, 2015e). The PMST report states that this species or habitat may occur within the area and individuals may be present within the EMBA based on their known NT distribution.

#### Eastern curlew

The eastern curlew (*Numenius madagascariensis*; Critically Endangered under the EPBC Act and TPWC Act; Migratory) is the world's largest species of shorebird (DCCEEW, 2024j; Menkhorst et al., 2017). Eastern curlews migrate annually to breeding grounds in Russia and north-eastern China before returning to Australia in August to forage primarily on crabs in intertidal mudflats (Menkhorst et al., 2017; Bamford et al., 2008). In Australia, the species has a continuous distribution from Barrow Island and Dampier Archipelago in WA through the Kimberley and along the NT, Qld, NSW coasts including the Torres Strait islands (TSSC, 2015f). There has been an increase at 2 sites in the Darwin region between 2009 and 2015, at Lee Point numbers have increased by 9% per year and 17% per year at East Arm Wharf in Darwin Harbour (Lilleyman et al., 2016). This local increase may be due to changes in roosting behaviour and an increase in suitable high tide roosting habitat. The PMST report states that this species or habitat may occur within the area, and individuals may fly over and be present within the EMBA.

#### **Gouldian finch**

The gouldian finch (*Erythrura gouldiae*; Endangered under the EPBC Act; Vulnerable under the TPWC Act) is a small, brightly coloured bird endemic to northern Australia, with its range extending from the Cape York Peninsula, QLD through to the Kimberly region, WA (O'Malley, 2006) with a northern extent of the Top End of the Northern Territory. The species has suffered substantial population declines throughout its range across Australia, with fewer than 2,500 mature birds estimated. Due to the population decline of the species, small populations of the animals are distributed in parts of the Kimberly and the Northern Territory. Within the NT, the gouldian finch is known to inhabit 6 areas: Yinbirrie Hills, Limmen Gate National Park, Kakadu National Park, Bradshaw Field Training Area, Newry Station, and Keep River National Park. The diet of the gouldian finch consists of seed from a range of grass species and insects (O'Malley, 2006). Due to their habitat and diet, and their restriction to mainland Northern Territory, it is unlikely that they will be present within the EMBA.

### Great knot

The great knot (*Calidris tenuirostris*; Vulnerable under the EPBC Act; Critically Endangered under the TPWC Act; Migratory) is a medium-sized migratory shorebird with relatively short legs, a slender medium-length bill and a wingspan of about 58 cm (DCCEEW, 2024d). The species breeds in north-east Siberia and far north-east Russia and migrates along the East Asia-Australasian Flyway to overwinter in the southern hemisphere (DEPWS, 2021c). Most that reach Australia settle along the northern coastline between north-west WA and the Gulf of Carpentaria, but significant numbers reach eastern Queensland and there are reports of great knots from most Australian coastal areas. The species is common in the NT from Darwin to the south-west Gulf of Carpentaria (DCCEEW, 2024d) with internationally significant numbers recorded in North Darwin (Beagle Gulf coastline) and the Milingimbi Coast (Birdlife Australia, 2020). It prefers sheltered coastal habitats with extensive tidal mudflats or sandflats, including estuaries, lagoons, inlets, and bays. Great knots are gregarious and frequently occur in large flocks with other shorebirds (including red knots), especially when roosting during high tides. They specialise in feeding on bivalves, but also consume other marine invertebrates. Prey are captured on or just below the surface of wet mud or sand (Garnet et al., 2011, DEPWS, 2021c). Given the areas historically observed to be inhabited by this species, individual birds may fly over and feed in coastal zones within the EMBA.

#### Greater sand plover

Greater sand plovers (*Charadrius leschenaultia*; Vulnerable under the EPBC Act and TPWC Act; Migratory) are shorebirds that migrate from breeding areas in Mongolia, Siberia, and China to coastal areas of all Australian states with the area around Darwin an internationally important site. This species occurs in the greatest numbers in northwestern Australia and is widespread between Northwest Cape and Roebuck Bay in WA, with scattered records between Roebuck Bay and Darwin. Greater sand plovers are recorded from most of the coastline of the NT, with significant areas around the Joseph Bonaparte Gulf, from Anson Bay to Murgenella Creek (including the

south coast of the Tiwi Islands), the northern Arnhem coast, and the Port McArthur area (TSSC, 2016). In Australia, greater sand plovers are almost entirely coastal, inhabiting sheltered muddy, sandy, or shelly beaches, large intertidal mudflats, saltmarshes, estuaries, sandbanks, coral reefs, rocky islands rock platforms, tidal lagoons, and coastal dunes. Greater sand plovers feed on molluscs, worms, crustaceans, and insects they find in wet sand or mud on open intertidal flats (DCCEEW, 2023e). The PMST report states that this species or habitat is likely to occur within the area, and individuals may fly over and be present within the EMBA.

#### **Grey Falcon**

Grey falcon (*Falco hypoleucos*; Vulnerable under the EPBC Act and TPWC Act) occur throughout much of the arid and semi-arid zones of Australia, in areas of sparsely timbered lowland plains, typically on inland drainage systems. The species has been recorded across the NT, including on the Tiwi Islands (DEPWS, 2021i). Grey Falcons use nests built by other bird species and prefer those in the tallest trees along watercourses. The Grey Falcon is a specialist predator of birds, particularly parrots and pigeons (TSSC, 2020). Birds may fly over and feed in coastal zones within the EMBA.

#### **Grey plover**

Grey plovers (*Pluvialis squatarola*; Vulnerable under the EPBC Act; Migratory) have been recorded along the coast in all states of Australia, with small numbers regularly recorded in the NT (DCCEEW, 2024g). Migrating birds arrive in northern Australia between August and October with many continuing their migration to southern regions. Plovers which have remained along the northern coastline for the non-breeding season leave between February and April (DCCEEW, 2024g). Some non-breeding individuals may stay in Australia. The species usually roosts in sheltered, sandy areas including unvegetated sandbanks or sand-spits, or other sheltered environments such as estuaries or lagoons, and are often seen in small numbers on mangrove mudflats (DCCEEW, 2024g). Kakadu National Park, Milingimbi coast, and the south-west coastline of the Gulf of Carpentaria have been identified as areas of importance to this species in the NT, but they do not represent sites of international significance (Birdlife Australia, 2020). In Australia, grey plovers feed by pecking and probing for worms, molluscs, and crustaceans mostly in mud or soft, wet sand of sandflats, intertidal mudflats, saltmarshes, and beaches (DCCEEW, 2024g). Given the areas historically observed to be inhabited by this species, individuals may seasonally fly over and be present in coastal zones within the EMBA.

#### Lesser sand plover

The lesser sand plover (*Charadrius mongolus*: Endangered under the EPBC Act and TPWC Act; Migratory) is a small to medium sized shorebird with a short stout bill and short grey legs. The lesser sand plover breeds in central Asia and eastern Russia. Two subspecies occur in Australia as seasonal migrants: Charadrius mongolus mongolus and Charadrius mongolus. stegmanni. In Australia, Charadrius mongolus stegmanni is more common in northern Australia, while Charadrius mongolus. mongolus is more common in eastern Australia (DEPWS, 2021d). After breeding during the northern summer on mountain steppes and tundras of inland eastern Russia (Charadrius mongolus. stegmanni), those that overwinter in Australia migrate southwards along the East Asian-Australasian Flyway. These non-breeding birds occur almost exclusively along the coast, where they forage on sheltered intertidal mudflats and sandflats, sandy beaches, estuaries and mangroves. Inland saline wetlands close to the coast are also used occasionally. They feed on marine worms, molluscs, crustaceans and insects, which are captured on or just below the surface of sand or mud. Given the areas historically observed to be inhabited by this species, individuals may fly over and feed in coastal zones within the EMBA.

### Masked Owl (northern)

Masked owl (northern) (*Tyto novaehollandiae kimberli*; Vulnerable under the EPBC Act and TPWC Act) is distributed widely across northern Australia in tall open eucalypt forests (DEPWS, 2021e). The masked owl (northern) roosts in monsoon rainforests, and also forages in more open vegetation types, including grasslands. Individuals typically roost in tree hollows and may also roost among dense foliage (DCCEEW, 2024j). The diet of the masked owl (northern mainland) mostly comprises mammals up to the size of possums (Garnett & Crowley 2000). Due to their habitat and prey preferences, and their restriction to the Tiwi Islands, it is unlikely that they will be present within the EMBA.

#### Nunivak bar-tailed godwit

Nunivak bar-tailed godwit (*Limosa lapponica baueri;* Vulnerable under the EPBC Act) breeds in west Alaska and north-east Siberia and overwinters mostly in northern and eastern Australia and New Zealand. In the NT, bar-tailed godwits have been reported along almost the entire coastline, including all major islands (DCCEEW, 2024j). After breeding during the northern summer on the arctic tundras of western, migration southwards along the East Asian-Australasian Flyway to overwinter in Australasia. During this non-breeding season (the austral summer), godwits in the NT usually congregate in flocks near the coast. They forage on intertidal mudflats or in shallow water, feeding on worms, molluscs, and crustaceans (DEPWS, 2021b). Birds may fly over and feed in coastal zones within the EMBA.



#### Partridge pigeon (eastern)

Partridge pigeon (eastern) (*Geophaps smithii smithii*; Vulnerable under the EPBC Act and TPWC Act) occur across northern Australia in lowland eucalypt open forests and woodlands, with grassy understoreys. Their diet comprises seeds, mostly of grasses but also from Acacia and other woody plants. The species forages entirely on the ground, and flies infrequently (DEPWS, 2021i). Due to their habitat and diet preferences, it is unlikely that they will be present within the EMBA.

### Red goshawk

The red goshawk (*Erythrotriorchis radiatus*; Endangered under the EPBC Act; Vulnerable under the TPWC Act) occur across northern Australia, from near Broome in the south-west Kimberley to south-eastern Queensland. Within this range it generally occurs in taller forests characteristic of higher rainfall areas, but there are some isolated recent records from central Australia. It appears to be unusually common on the Tiwi Islands (DEPWS, 2021g). The preferred habitat is tall open eucalypt forest and riparian areas (including paperbark forest and gallery forests). The conspicuous basket–shaped stick nest is typically placed in large trees near watercourses (Aumann and Baker-Gabb, 1991). Red goshawks eat mostly birds, especially parrots and pigeons; rarely they also prey on mammals, reptiles, and large insects (Debus et al., 2020). Birds may fly over and feed in coastal zones within the EMBA.

### Red knot

The red knot (*Calidris canutus*; Vulnerable under the EPBC Act; Endangered under the TPWC Act; Migratory) is a migratory omnivorous shorebird which utilises the intertidal mudflats, sandflats and sandy beaches of sheltered coastal areas, estuaries, bays and other similar marine habitats (DCCEEW, 2024c). The red knot is present throughout coastal and offshore Australia, with large numbers regularly recorded in the north-west of Australia (Clarke, 2011; Bamford et al., 2008). The red knot breeds in Siberia and spends the non-breeding season in Australia and New Zealand, arriving in northern Australia in late August to early September and also settles in eastern Australia and New Zealand (DCCEEW, 2024c; Watkins, 1993). During the non-breeding season, the red knot occurs on tidal mudflats or sandflats feeding on invertebrates, especially shellfish (Garnet et al., 2011). Both north-western and south-eastern Australia are key areas for red knots. The Gulf of Carpentaria is an important staging area for migrating birds headed to south-eastern Australia and New Zealand. The NT region between the Daly River and Bynoe Harbour, along with the northern Arnhem Land coast from Boucaut Bay to Buckingham Bay are important areas (Chatto, 2003), with North Darwin (Beagle Gulf coastline) considered to have international significance (Birdlife Australia, 2020). Birds may fly over and feed in coastal zones within the EMBA.

#### **Ruddy turnstone**

The ruddy turnstone (*Arenaria interpres*; Vulnerable under the EPBC Act; Migratory) is a migratory shorebird that leaves its breeding grounds in the northern hemisphere from mid-July to early September (DCCEEW, 2024a) and has an almost cosmopolitan non-breeding distribution, common throughout Australasia and widespread within Australia (DCCEEW, 2024a). This species tends to arrive in the NT and WA from August onwards (DCCEEW, 2024a). Ruddy turnstones typically roost along platforms and shelves of rock, shingle, or gravel beaches, but can also be found along sand, coral, or shell beaches, and along shoals, cays, and dry ridges. In north Australia, they are known to occur in a wide variety of habitats and may prefer wide mudflats (DCCEEW, 2024a). The species feeds mainly on maggots from rotting seaweed in the upper intertidal (DCCEEW, 2024a). Bynoe Harbour and Castlereagh Bay in the NT are reported to be important areas (DCCEEW, 2024a) with the Milingimbi Coast considered to have international significance for this bird (Birdlife Australia, 2020). Given the areas historically observed to be inhabited by this species, individuals may seasonally fly over and be present in coastal zones within the EMBA.

#### Sharp-tailed sandpiper

The sharp tailed sandpiper (*Calidris acuminata*; Vulnerable under the EPBC Act; Migratory) is a small-medium size wader that is widely distributed throughout Australia (DCCEEW, 2024j). The majority (>90%) of the non-breeding population migrates to Australia (DCCEEW, 2024b). They arrive in Australia from mid-August/early September with most birds then moving slowly south to southeast Australia (DCCEEW, 2024j). In the NT, the species mostly occurs in the northern coastal regions (DCCEEW, 2024j), with Darwin Harbour, North Darwin (Beagle Gulf coastline), Kakadu National Park, the Legune Wetlands, Milingimbi coast and Nhulunbuy (Gove Peninsula) considered to be important areas (Birdlife Australia, 2020). Internationally significant numbers have been recorded at Kakadu National Park and Milingimbi coast (Birdlife Australia, 2020). Sharp tailed sandpipers often roost at the edges of wetlands, on wet open mud or sand, in shallow water, or in short sparse grass or saltmarsh, but also occasionally on sandy beaches, stony shores or rocks (DCCEEW, 2024j). They typically feed on seeds, worms, molluscs, crustaceans, and insects (DCCEEW, 2024j), foraging at the edge of the water of wetlands or intertidal mudflats, either on bare wet mud or sand, or in shallow water (DCCEEW, 2024j). The PMST report states that this species is known to occur (roost) within the area, and it is likely to seasonally occur in the EMBA.



#### Tiwi Islands hooded robin

Tiwi Islands hooded robin (*Melanodryas cucullata melvillensis;* Critically Endangered under the EPBC Act and TPWC Act) distribution is restricted to the Tiwi Islands. Tiwi Islands hooded robin occur inhabits more open forests and woodlands and forages on ground-dwelling invertebrates in areas of thinner ground-cover (DEPWS, 2021c). The breeding season (of other subspecies) is spring–summer. The nests are typically placed in the forks of trees, mostly <3 m above ground. The typical foraging behaviour of Tiwi Islands hooded robin is by quietly perching on tree branches, or trunks, and then suddenly pouncing to take prey on the ground (Fitri & Ford 2003; Higgins & Peter 2002). Due to their habitat and prey preferences, and their restriction to the Tiwi Islands, it is unlikely that they will be present within the EMBA.

#### **Terek sandpiper**

The terek sandpiper (*Xenus cinereus*; Vulnerable under the EPBC Act; Migratory) is primarily a coastal species, more common in northern and eastern parts of Australia than southern regions (DCCEEW, 2024j). It is one of the commoner shorebird species in tropical mangrove-lined estuaries, often occurring in small numbers among much larger flocks of other migratory shorebirds (DCCEEW, 2024i). They feed primarily on crustaceans and insects, in the supralittoral or upper littoral zone, where a film of water covers the sand, but may also forage in the lower littoral zone on exposed rock platforms (DCCEEW, 2024i). In the NT, widespread records occur from Darwin, north to Melville Island, and east to the western section of the Gulf of Carpentaria, around Gove Peninsula, Groote Eylandt, Sir Edward Pellew Island and the mouth of the McArthur River (DCCEEW, 2024j). Important areas are considered to include Darwin Harbour, North Darwin (Beagle Gulf coastline), Kakadu National Park, the Legune Wetlands and Milingimbi Coast, with the Kakadu and Milingimbi Coast identified to have international significance (Birdlife Australia, 2020). The preferred roosting habitat for this bird is in or among mangroves (DCCEEW, 2024j). Terek sandpipers migrate south from their Arctic breeding grounds, passing through the Torres Strait and arriving around Cairns and Darwin in August. Most individuals visiting Australia seem to remain on the north coast, leaving by late April (DCCEEW, 2024i). This species is likely to seasonally occur in the EMBA.

#### Tiwi masked owl

The Tiwi masked owl (*Tyto novaehollandiae melvillensis*; Endangered under the EPBC Act) is a subspecies of the masked owl (*Tyto novaehollandiae*) that occurs only on Bathurst and Melville Islands. Tiwi masked owls occur mainly in the forests and woodlands but may roost in monsoon forests or mangroves and may forage over the treeless plains and grasslands (Ward, 2010). Individuals typically roost in tree hollows but may also roost among dense foliage. Masked Owls breed in large tree hollows, which usually form in large rainforest trees. It is likely that individual home ranges are large. The diet of the Tiwi Masked Owl mostly comprises mammals up to the size of possums (DEPWS, 2021f). Due to their habitat and prey preferences, and their restriction to the Tiwi Islands, it is unlikely that they will be present within the EMBA.

#### 3.2.13.4.2 Migratory species

Most migrant birds are expected to fly over the regional area as part of their large-scale transitory movements and are unlikely to land on the sea for significant periods of time (ConocoPhillips, 2018). Considering this, and the general absence of landing areas at a regional offshore scale, the majority of seabird activity is likely to comprise foraging and migration pathways. While seabirds spend much of their lives at sea, migratory shorebirds overfly offshore areas during migratory periods and typically do not interact with the sea surface (ConocoPhillips, 2018; DSEWPaC, 2012d). Migratory wetland species do not interact with open offshore waters but may land on offshore infrastructure while flying between land masses (ConocoPhillips, 2018).

Shorebird migration patterns are seasonal and vary according to species (DSEWPaC, 2012e), but generally shorebirds migrate to northern Australia from August to November. The majority of birds remain in northern Australia, while others disperse southwards (Bennelongia, 2011). On northern beaches migratory shorebirds peak in November then again in March as the majority of birds begin their return to the northern hemisphere between March and May. Most migratory shorebirds do not breed in Australia and juvenile birds may spend several years in Australia before reaching maturity and returning north to breed (DEWHA, 2008c). Species listed as migratory under the EPBC Act that may occur in the EMBA are outlined in Table 3-10.

### 3.2.13.5 Biologically important areas and habitat critical

Table 3-12 and Figure 3-8 to Figure 3-9 show the biologically important areas (BIAs) in relation to the EMBA.

BIAs are spatially defined areas where aggregations of individuals of a species are known to display biologically important behaviour such as breeding, foraging, nesting, internesting or migration. Habitat critical to the survival of marine turtles provides areas for turtle activities, long-term maintenance of the species, maintain genetic diversity and long-term evolutionary development and re-introduction of populations or recovery of the species.

Habitat critical to the survival of three EPBC Act-listed marine turtles (habitat critical – HC) in relation to the EMBA are presented in Table 3-12 and shown in Figure 3-9.



# Table 3-12: Biologically important areas and habitat critical to the survival of a species identified in the EMBA and MEVA

Species	BIA	Distance to the OA (km)	MEVA	ЕМВА	Habitat critical to the survival of a species within the EMBA and distance to OA
Marine mammals	5				
Indo-Pacific humpback dolphin	Breeding	39	×	1	×
Marine reptiles					
Flatback turtle	Internesting	Overlaps	$\checkmark$	√	✓ Overlaps OA & EMBA (nesting June– September)
Green turtle	Foraging	65	$\checkmark$	1	✓ 91 km; Overlaps EMBA (nesting December-January)
Olive ridley	Internesting	20	✓	√	✓ 11 km; Overlaps
turtie	Foraging	60	✓	1	EMBA (nesting May– July)
Leatherback turtle	Reproduction	135	X	×	✓ 185 km (nesting December– January)
Hawksbill turtle	Internesting	100	X	1	X



Figure 3-8: Indo-Pacific humpback dolphin biologically important areas overlapping or proximal to the EMBA





Figure 3-9: Marine turtle biologically important and habitat critical areas overlapping or proximal to the EMBA





### 3.2.13.6 Recovery plans, conservation advice, and wildlife conservation plans

Recovery plans set out the necessary research and management actions to stop the decline of listed Threatened species and support their recovery. Table 3-13 summarises the conservation actions relevant to the activity, with more information about the requirements of the relevant plans of management, including recovery plans, conservation advice and wildlife conservation plans for marine fauna and demonstrates where this GEP Coastal Waters OEMP considers those management requirements.

Further assessment of the activity's consistency with actions and objectives set within the plans is provided throughout Sections 6 and Section 7.

## Table 3-13: Relevant threats identified in recovery plans, conservation advice and wildlife conservation plans for species that occur or may occur within the EMBA

Name	Recovery plan/conservation advice/management plan	Relevant objectives	Threats/strategies identified as relevant to the Activity	Relevant conservation actions	Addressed (where relevant) in OEMP
All					
All vertebrate fauna	Threat Abatement Plan for the Impacts of Marine Debris on Vertebrate Wildlife of Australia's Coasts and Oceans (DoEE, 2018)	<ul> <li>There are 4 main objectives:</li> <li>contribute to the long-term prevention of the incidence of harmful marine debris</li> <li>remove existing harmful marine debris from the marine environment</li> <li>mitigate the impacts of harmful marine debris on marine species and ecological communities</li> <li>monitor the quantities, origins and impacts of marine debris and assess the effectiveness of management arrangements over time for the strategic reduction of debris.</li> </ul>	Marine debris	No explicit management actions for non- fisheries-related industries (note that management actions in the plan relate largely to managing fishing waste (e.g. 'ghost' gear), and state, territory and Commonwealth management through regulation).	Section 7.1 Release of solid objects
Fish and sharks		1		1	
All sawfish and river sharks including: • dwarf sawfish • green sawfish • largetooth sawfish • speartooth shark • northern river shark	Sawfish and River Sharks Multispecies Recovery Plan (CoA, 2015b)	<ul> <li>The primary objective of this recovery plan is to assist the recovery of sawfish and river sharks with a view to:</li> <li>improving the population status leading to the removal of the sawfish and river shark species from the threatened species list of the EPBC Act</li> <li>ensuring that anthropogenic activities do not hinder recovery in the near future, or impact the conservation status of the species in the future.</li> <li>The specific objectives of the recovery plan (relevant to industry) are:</li> <li>Objective 5: Reduce and, where possible, eliminate adverse impacts of habitat degradation and modification on sawfish and river shark species</li> <li>Objective 6: Reduce and, where possible, eliminate any adverse impacts of marine debris on sawfish and river shark species noting the linkages with the Threat Abatement Plan for the impact of marine debris on vertebrate marine life (DoEE, 2018).</li> </ul>	Habitat degradation and modification	Identify risks to important sawfish and river shark habitat and measures needed to reduce those risks.	Section 6.4 Seabed and benthic habitat disturbance Section 6.6 Operational discharges Section 7.1 Release of solid objects Section 7.6 Unplanned release of MDO Section 7.7 Contingency spill response operations
Dwarf sawfish	Approved Conservation Advice for <i>Pristis</i> <i>clavata</i> (Dwarf Sawfish) (DEWHA, 2009)	No explicit relevant objectives.	Habitat degradation and modification	No explicit management actions for industry.	Section 6.4 Seabed and benthic habitat disturbance Section 6.6 Operational discharges Section 7.1 Release of solid objects Section 7.2 Introduction of invasive marine species Section 7.4 Minor release of hydrocarbons and chemicals Section 7.5 Subsea release of dry natural gas Section 7.6 Unplanned release of MDO Section 7.7 Contingency spill response operations
Largetooth sawfish	Approved Conservation Advice for <i>Pristis pristis</i> (largetooth sawfish) (TSSC, 2014b)	No explicit relevant objectives.	Habitat degradation and modification	Implement measures to reduce adverse impacts of habitat degradation and/or modification.	Section 6.4 Seabed and benthic habitat disturbance Section 6.6 Operational discharges Section 7.1 Release of solid objects Section 7.2 Introduction of invasive marine species Section 7.4 Minor release of hydrocarbons and chemicals

Name	Recovery plan/conservation advice/management plan	Relevant objectives	Threats/strategies identified as relevant to the Activity	Relevant conservation actions	Addressed (where relevant) in OEMP
					Section 7.5 Subsea release of dry natural gas Section 7.6 Unplanned release of MDO Section 7.7 Contingency spill response operations
Green sawfish	Approved Conservation Advice for Green Sawfish (DEWHA, 2008c)	No explicit relevant objectives.	Habitat degradation and modification	No explicit relevant management actions.	Sections 6.4 Seabed and benthic habitat disturbance Section 6.6 Operational discharges Section 7.1 Release of solid objects Section 7.2 Introduction of invasive marine species Section 7.4Minor release of hydrocarbons and chemicals Section 7.5 Subsea release of dry natural gas Section 7.6 Unplanned release of MDO Section 7.7 Contingency spill response operations
Northern river shark	Approved Conservation Advice for <i>Glyphis garricki</i> (northern river shark) (TSSC, 2014a)	No explicit relevant objectives.	Habitat degradation and modification	Implement measures to reduce adverse impacts of habitat degradation and/or modification.	Section 6.4 Seabed and benthic habitat disturbance Section 6.6 Operational discharges Section 7.1 Release of solid objects Section 7.2 Introduction of invasive marine species Section 7.4 Minor release of hydrocarbons and chemicals Section 7.5 Subsea release of dry natural gas Section 7.6 Unplanned release of MDO Section 7.7 Contingency spill response operations
			Marine debris	No explicit relevant management actions.	Section 7.1 Release of solid objects
Speartooth shark	Approved Conservation Advice for <i>Glyphis glyphis</i> (speartooth shark) (DoE, 2014a)	No explicit relevant objectives.	Habitat degradation and modification	Implement measures to reduce adverse impacts of habitat degradation and/or modification.	Section 7.1 Release of solid objects Section 7.5 Subsea release of dry natural gas, Section 7.6 Unplanned release of MDO Section 7.7 Contingency spill response operations
			Marine debris	No explicit management actions for industry (note that the responsibility for the action identified is for Commonwealth Government to implement).	Section 7.1 Release of solid objects
Grey nurse shark (west coast population)	Recovery Plan for the Grey Nurse Shark ( <i>Carcharias taurus</i> ) (DoE, 2014b)	The overarching objective of this recovery plan is to assist the recovery of the grey nurse shark in the wild with a view to:	Pollution and disease	Review and assess the potential threat of introduced species, pathogens and pollutants.	Section 7.6 Unplanned release of MDO Section 7.7 Contingency spill response operations
		<ul> <li>Improving the population status</li> <li>ensuring that anthropogenic activities do not hinder the recovery of the grey nurse shark.</li> </ul>	Ecosystem effects as a result of habitat modification	Review the level and spatial extent of protection measures at key aggregation sites to ensure appropriate levels of protection, and a consistent approach to the designation and implementation of protective measures, are applied. Use BIAs to help inform the development of appropriate conservation measures, including applying advice in the marine bioregional plans on the types of actions that are likely to have a	Section 7.6 Unplanned release of MDO Section 7.7 Contingency spill response operations

Name	Recovery plan/conservation advice/management plan	Relevant objectives	Threats/strategies identified as relevant to the Activity	Relevant conservation actions	Addressed (where relevant) in OEMP
				significant impact on the species and updating such conservation measures as new information becomes available.	
White shark	Recovery Plan for the White Shark ( <i>Carcharodon carcharias</i> ) (DSEWPaC, 2013)	<ul> <li>The overarching objective of this recovery plan is to assist the recovery of the white shark in the wild throughout its range with a view to:</li> <li>improving the population status leading to future removal of the white shark from the threatened species list of the EPBC Act</li> <li>ensuring that anthropogenic activities do not hinder recovery in the near future, or impact the conservation status of the species in the future.</li> <li>The specific objective of the recovery plan (relevant to industry) is:</li> <li>Objective 7: Continue to identify and protect habitat critical to the survival of the white shark and minimise the impact of threatening processes within these areas.</li> </ul>	Ecosystem effects as a result of habitat modification and climate change	No explicit relevant management actions.	Section 6.3 Atmospheric and greenhouse gas emissions Section 6.4 Seabed and benthic habitat disturbance Section 6.6 Operational discharges Section 7.1 Release of solid objects Section 7.2 Introduction of invasive marine species Section 7.4 Minor release of hydrocarbons and chemicals Section 7.5 Subsea release of dry natural gas Section 7.6 Unplanned release of MDO Section 7.7 Contingency spill response operations
Whale shark	Conservation Advice for <i>Rhincodon typus</i> (whale shark) (TSSC, 2015g)	To maintain existing levels of protection for the whale shark in Australia while working to increase the level of protection afforded to the whale shark within the Indian Ocean and Southeast Asian region to enable population growth so that the species can be removed from the threatened species list of the EPBC Act.	Boat strike from large vessels	Minimise offshore developments and transit time of large vessels in areas close to marine features likely to correlate with whale shark aggregations along the northward migration route that follows the northern WA coastline along the 200 m isobath (TSSC, 2015g).	Section 7.3 Marine fauna interaction
			Habitat disruption from mineral exploration, production and transportation	Implement measures to reduce adverse impacts of habitat degradation and/or modification.	Section 6.4 Seabed and benthic habitat disturbance Section 6.6 Operational discharges Section 7.2 Introduction of invasive marine species Section 7.4, Minor release of hydrocarbons and chemicals Section 7.5 Subsea release of dry natural gas Section 7.6 Unplanned release of MDO Section 7.7 Contingency spill response operations
			Marine debris	No explicit relevant management actions.	Section 7.1 Release of solid objects
Marine mammals					
Cetaceans and other marine megafauna	National Strategy for Reducing Vessel Strike on Cetaceans and other Marine Megafauna (CoA, 2017a)	<ul> <li>The overarching goal of the strategy is to provide guidance on understanding and reducing the risk of vessel collisions and the impacts they may have on marine megafauna.</li> <li>The specific objective of the strategy (relevant to industry) is:</li> <li>Objective 3: Mitigation – reduce the likelihood and severity of megafauna vessel collision.</li> </ul>	Vessel collision	Encourage innovation and collaboration between research organisations and industry.	Section 7.3 Marine fauna interaction
Blue whale (includes pygmy blue whale)	Conservation Management Plan for the Blue Whale 2015–2025 (CoA, 2015a)	The long-term recovery objective is to minimise anthropogenic threats to allow the conservation status of	Noise interference assess and address anthropogenic noise	Assess and address anthropogenic noise: shipping, industrial and seismic noise.	Section 6.1 Noise emissions
		the blue whale to improve so that it can be removed from the threatened species list under the EPBC Act.	Climate variability and change	Continue to meet Australia's international commitments to reduce greenhouse gas emissions	Section 6.3 Atmospheric and greenhouse gas emissions
			Habitat modification	No explicit relevant management actions.	Section 7.4 Minor release of hydrocarbons and chemicals Section 7.5 Subsea release of dry natural gas

Name	Recovery plan/conservation advice/management plan	Relevant objectives	Threats/strategies identified as relevant to the Activity	Relevant conservation actions	Addressed (where relevant) in OEMP
					Section 7.6 Unplanned release of MDO Section 7.7 Contingency spill response operations
			Vessel disturbance	<ul><li>Minimise vessel collisions:</li><li>develop a national vessel strike strategy that</li></ul>	Section 7.3 Marine fauna interaction
				investigates the risk of vessel strike on blue whales and also identifies potential mitigation measures	
				ensure all vessel strike incidents are reported in the National Ship Strike database <sup>3</sup>	
				<ul> <li>ensure the risk of vessel strikes on blue whales is considered when assessing actions that increase vessel traffic in areas where blue whales occur and, if required, appropriate mitigation measures are implemented.</li> </ul>	
			Marine debris	No explicit relevant management actions.	Section 7.1 Release of solid objects
Fin whale	Conservation Advice for <i>Balaenoptera physalus</i> (fin whale) (TSSC, 2015c)	Determine population abundance, trends and population structure for fin whales, and establish a long-term monitoring program.	Habitat degradation including pollution (increasing port expansion and coastal development)	No explicit relevant management actions.	Section 6.6 Operational discharges Section 7.1 Release of solid objects Section 7.4 Minor release of hydrocarbons and chemicals Section 7.5 Subsea release of dry natural gas Section 7.6 Unplanned release of MDO Section 7.7 Contingency spill response operations
			Climate and oceanographic variability and change	Continue to meet Australia's international commitments to reduce greenhouse gas emissions .	Section 6.3 Atmospheric and greenhouse gas emissions
			Anthropogenic noise	Once the spatial and temporal distribution (including BIAs) of fin whales is further defined, assess the impacts of increasing anthropogenic noise (including seismic surveys, port expansion, and coastal development).	Section 6.1 Noise emissions
			Vessel strike	Develop a national vessel strike strategy that investigates the risk of vessel strikes on fin whales and identifies potential mitigation measures. Ensure all vessel strike incidents are reported in the National Ship Strike database <sup>3</sup> .	Section 7.3 Marine fauna interaction
Sei whale	Conservation Advice for <i>Balaenoptera borealis</i> (sei whale) (TSSC, 2015b)	Determine population abundance, trends and population structure for sei whales, and establish a long-term monitoring program.	Anthropogenic noise	Once the spatial and temporal distribution (including BIAs) of sei whales is further defined, assess the impacts of increasing anthropogenic noise (including seismic surveys, port expansion, and coastal development).	Section 6.1 Noise emissions
			Vessel strike	<ul> <li>Minimise vessel collisions:</li> <li>develop a national vessel strike strategy that investigates the risk of vessel strikes on sei whales and also identifies potential mitigation measures</li> <li>ensure all vessel strike incidents are reported in the National Ship Strike database<sup>3</sup>.</li> </ul>	Section 7.3 Marine fauna interaction

<sup>&</sup>lt;sup>3</sup> <u>https://data.marinemammals.gov.au/report/shipstrike/new</u>



Name	Recovery plan/conservation advice/management plan	Relevant objectives	Threats/strategies identified as relevant to the Activity	Relevant conservation actions	Addressed (where relevant) in OEMP
			Climate and oceanographic variability and change	Continue to meet Australia's international commitments to reduce greenhouse gas emissions.	Section 6.3 Atmospheric and greenhouse gas emissions
			Habitat degradation including pollution	No explicit relevant management actions.	Sections 6.6 Operational discharges , Section 7.1 Release of solid objects Section 7.4 Minor release of hydrocarbons and chemicals Section 7.5 Subsea release of dry natural gas Section 7.6 Unplanned release of MDO Section 7.7 Contingency spill response operations
Reptiles					
All marine turtles (flatback, green, hawksbill, leatherback, loggerhead, olive ridley)	National Light Pollution Guidelines for Wildlife (DCCEEW, 2023c)	Lighting objectives will need to consider the regulatory requirements and Australian standards relevant to the activity, location and wildlife present. Objectives should be described in terms of specific locations and times for which artificial light is necessary. Consideration should be given to whether colour differentiation is required and if some areas should remain dark, either to contrast with lit areas or to avoid light spill. Where relevant, wildlife requirements should form part of the lighting objectives. A lighting installation will be deemed a success if it meets the lighting objectives (including wildlife needs) and areas of interest can be seen by humans clearly, easily, safely and without discomfort.	Light pollution	<ul> <li>Best practice lighting design incorporates these design principles:</li> <li>start with natural darkness and only add light for specific purposes</li> <li>use adaptive light controls to manage light timing, intensity and colour</li> <li>light only the object or area intended – keep lights close to the ground, directed and shielded to avoid light spill</li> <li>use the lowest intensity lighting appropriate for the task</li> <li>use lights with reduced or filtered blue, violet and ultraviolet wavelengths.</li> </ul>	Section 6.2 Light emissions, Section 7.7 Contingency spill response operations
	Recovery Plan for Marine Turtles in Australia 2017–2027 (CoA, 2017b)	<ul> <li>Long-term recovery objective:</li> <li>minimise anthropogenic threats to allow for the conservation status of marine turtles to improve so that they can be removed from the EPBC Act threatened species list.</li> </ul>	Marine debris	<ul> <li>Reduce the impacts from marine debris:</li> <li>support the implementation of the EPBC Act Threat Abatement Plan for the impacts of marine debris on vertebrate marine life (DoEE, 2018).</li> </ul>	Section 7.1 Release of solid objects
		<ul><li>Interim objective 3:</li><li>anthropogenic threats are demonstrably minimised.</li></ul>	Chemical and terrestrial discharge	Minimise chemical and terrestrial discharge.	Sections 6.6 Operational discharges Section 7.4 Minor releases of hydrocarbons and chemicals, Section 7.5 Subsea release of dry natural gas Section 7.6 Unplanned release of MDO
			Vessel disturbance	Vessel interactions identified as a threat. No specific management actions in relation to vessels prescribed in the plan.	Section 7.3 Marine fauna interaction
			Light pollution	<ul> <li>Minimise light pollution:</li> <li>manage artificial light within or adjacent to habitat critical to the survival of marine turtles such that marine turtles are not displaced from these habitats</li> <li>develop and implement best practice light management guidelines for existing and future developments adjacent to marine turtle nesting beaches.</li> <li>identify the cumulative impact on turtles from multiple sources of onshore and offshore light pollution.</li> </ul>	Section 6.2 Light emissions



Name	Recovery plan/conservation advice/management plan	Relevant objectives	Threats/strategies identified as relevant to the Activity	Relevant conservation actions	Addressed (where relevant) in OEMP
			Noise interference	<ul> <li>Assess and address anthropogenic noise:</li> <li>understand the impacts of anthropogenic noise on marine turtle behaviour and biology.</li> </ul>	Section 6.1 Noise emissions
			Habitat modification	Manage anthropogenic activities to ensure marine turtles are not displaced from identified habitat critical to their survival. Manage anthropogenic activities in BIAs to ensure that biologically important behaviour can continue.	Sections 7.1 Release of solid objects Section 7.4 Minor releases of hydrocarbons and chemicals, Section 7.5 Subsea release of dry natural gas Section 7.6 Unplanned release of MDO Section 7.7 Contingency spill response operations
Leatherback turtle	Approved Conservation Advice for	No explicit relevant objectives.	Boat strike	No explicit relevant management actions.	Section 7.3 Marine fauna interaction
	<i>Dermochelys coriacea</i> (Leatherback Turtle) (DEWHA, 2008b)		Habitat degradation (changes to breeding sites and degradation to foraging areas)	Identify and protect migratory corridors between nesting beaches and common foraging areas to facilitate colonisation.	Section 7.5 Subsea release of dry natural gas Section 7.6 Unplanned release of MDO
			Marine debris	No explicit relevant management actions.	Section 7.1 Release of solid objects
			Climate Change	No explicit relevant management actions.	Section 6.3 Atmospheric and greenhouse gas emissions
Seabirds and shorebirds					
All seabirds and shorebirds	National Light Pollution Guidelines for Wildlife (DCCEEW, 2023c)	Lighting objectives will need to consider the regulatory requirements and Australian standards relevant to the activity, location and wildlife present. Objectives should be described in terms of specific locations and times for which artificial light is necessary. Consideration should be given to whether colour differentiation is required and if some areas should remain dark, either to contrast with lit areas or to avoid light spill. Where relevant, wildlife requirements should form part of the lighting objectives. A lighting installation will be deemed a success if it meets the lighting objectives (including wildlife needs) and areas of interest can be seen by humans clearly, easily, safely and without discomfort.	Light pollution	Best practice lighting design incorporates these design principles:	Section 6.2 Light emissions
				<ul> <li>start with natural darkness and only add light for specific purposes</li> </ul>	
				<ul> <li>use adaptive light controls to manage light timing, intensity and colour.</li> </ul>	
				<ul> <li>light only the object or area intended – keep lights close to the ground, directed and shielded to avoid light spill.</li> </ul>	
				<ul> <li>use the lowest intensity lighting appropriate for the task.</li> </ul>	
				• use non-reflective, dark-coloured surfaces.	
				• use lights with reduced or filtered blue, violet and ultraviolet wavelengths.	
Bridled tern Common noddy Great frigatebird Greater crested tern Lesser frigatebird	Wildlife Conservation Plan for Seabirds (CoA, 2020)	Seabirds and their habitats are protected and managed in Australia.	Pollution (marine debris, light, water)	Enhance contingency plans to prevent and/or respond to environmental emergencies that impact seabirds and their habitats.	Section 6.2 Light emissions Section 7.1 Release of solid objects Section 7.6 Unplanned release of MDO , Section 7.7 Contingency spill response operations
Little tern Osprey Streaked shearwater Wedge-tailed shearwater White-tailed tropicbird			Habitat loss and degradation from pollution	No explicit relevant management actions.	Sections 7.4 Minor release of hydrocarbons and chemicals Section 7.5 Subsea release of dry natural gas Section 7.6 Unplanned release of MDO , Section 7.7 Contingency spill response operations
			Invasive species	Manage invasive species at important seabird habitats	Section 7.2 Introduction of invasive marine species
			Climate variability and change	Investigate the impacts of climate variability and change on seabirds and their habitats	Section 6.3 Atmospheric and greenhouse gas emissions

Name	Recovery plan/conservation advice/management plan	Relevant objectives	Threats/strategies identified as relevant to the Activity	Relevant conservation actions	Addressed (where relevant) in OEMP
			Anthropogenic disturbance	Ensure all areas of important habitat for seabirds are considered in the development assessment process. Manage the effects of anthropogenic disturbance to seabird breeding and roosting areas.	Section 7.3 Marine fauna interaction
Asian dowitcher Bar-tailed godwit Black-tailed godwit Common greenshank	Wildlife Conservation Plan for Migratory Shorebirds (CoA, 2015c)	Anthropogenic threats to migratory shorebirds in Australia are minimised or, where possible, eliminated.	Habitat degradation/modification	No explicit relevant management actions.	Sections 7.1 Release of solid objects Section 7.6 Unplanned release of MDO, Section 7.7 Contingency spill response operations
Curlew sandpiper Eastern curlew Great knot Greater sand plover Grey plover Lesser sand plover			Anthropogenic disturbance	Investigate the significance of cumulative impacts on migratory shorebird habitat and populations in Australia. Ensure all areas important to migratory shorebirds in Australia continue to be considered in development assessment processes (specifically for coastal developments).	Section 7.3 Marine fauna interaction
Little curlew Little ringed plover Long-toed stint			Climate change and variability	Investigate the impacts of climate change on migratory shorebird habitat and populations in Australia.	Section 6.3 Atmospheric and greenhouse gas emissions
Marsh sandpiper Oriental plover Pacific golden plover Pectoral sandpiper Red knot Red-necked stint Ruddy turnstone Sanderling Sharp-tailed sandpiper Streaked shearwater Terek sandpiper Whimbrel			Pollution/contamination	No explicit relevant management actions.	Section 7.1 Release of solid objects Section 7.6 Unplanned release of MDO Section 7.7 Contingency spill response operations
Asian dowitcher <sup>4</sup>	Conservation Advice for <i>Limnodromus</i> <i>semipalmatus</i> (Asian dowitcher) (DCCEEW, 2024f)	No explicit relevant objectives.	Pollution/contamination	No explicit relevant management actions.	Section 7.6 Unplanned release of MDO, Section 7.7 Contingency spill response operations
Australian Painted Snipe⁴	Approved Conservation Advice for <i>Rostratula australis</i> (Australian painted snipe) (TSSC, 2013)	No explicit relevant objectives.	Habitat loss, degradation and modification	No explicit relevant management actions.	Section 7.6 Unplanned release of MDO Section 7.7 Contingency spill response operations
	National Recovery Plan for the Australian Painted Snipe ( <i>Rostratula</i> <i>australis</i> ) (DCCEEW, 2022a)	No explicit relevant objectives.	Habitat loss, degradation and modification	No explicit relevant management actions.	Section 7.6 Unplanned release of MDO Section 7.7 Contingency spill response operations
Black-tailed godwit <sup>4</sup>	Conservation Advice for <i>Limosa limosa</i> (black-tailed godwit) (DCCEEW, 2024e)	No explicit relevant objectives.	Pollution	No explicit relevant management actions.	Section 7.1 Release of solid objects Section 7.6 Unplanned release of MDO Section 7.7 Contingency spill response operations
Common greenshank <sup>4</sup>	Conservation Advice for <i>Tringa nebularia</i> (common greenshank) (DCCEEW, 2024h)	No explicit relevant objectives.	Pollution	No explicit relevant management actions.	Section 7.6 Unplanned release of MDO

<sup>&</sup>lt;sup>4</sup> Species or species habitat is not known to be present within planned impact areas (e.g. OA and light assessment boundary), or threats identified are not relevant to the Activity. Therefore, conservation advice or recovery is not evaluated within Section 6 or Sections 7.1–7.5 and 7.8.

Name	Recovery plan/conservation advice/management plan	Relevant objectives	Threats/strategies identified as relevant to the Activity	Relevant conservation actions	Addressed (where relevant) in OEMP	
Curlew sandpiper	Approved Conservation Advice for <i>Calidris ferruginea</i> (Curlew Sandpiper) (TSSC, 2015e)	No explicit relevant objectives.	Habitat loss and degradation from oil pollution	No explicit relevant management actions.	Section 7.6 Unplanned release of MDO	
Eastern curlew	Approved Conservation Advice for <i>Numenius madagascariensis</i> (Eastern Curlew) (TSSC, 2015f)	<ul><li>Australian objectives:</li><li>achieve a stable or increasing population.</li><li>maintain and enhance important habitat.</li></ul>	Habitat loss and degradation from pollution	No explicit relevant management actions.	Section 6.6 Operational discharges Section 7.1 Release of solid objects, Section 7.6 Unplanned release of MDO	
		reduce disturbance at key roosting and feeding sites.	Pollution/contamination	No explicit relevant management actions.	N/A	
Great knot <sup>4</sup>	Conservation Advice for <i>Calidris</i> <i>tenuirostri</i> s (great knot) (DCCEEW, 2024d)	No explicit relevant objectives.	Pollution	No explicit relevant management actions.	Section 7.6 Unplanned release of MDO Section 7.7 Contingency spill response operations	
Greater sand plover <sup>4</sup>	Conservation Advice <i>Charadrius</i> <i>leschenaultii</i> (Greater sand plover) (DCCEEW, 2023e)	No explicit relevant objectives.	Habitat loss and degradation	No explicit relevant management actions.	Section 7.6 Unplanned release of MDO Section 7.7 Contingency spill response operations	
			Pollution/contamination	No explicit relevant management actions.	N/A	
Grey falcon <sup>4</sup>	Conservation Advice <i>Falco hypoleucos</i> (Grey Falcon) (TSSC, 2020)	No explicit relevant objectives.	No explicit relevant threats	No explicit relevant management actions.	N/A	
Grey plover <sup>4</sup>	Conservation Advice for <i>Pluvialis</i> <i>squatarola</i> (grey plover) (DCCEEW, 2024g)	No explicit relevant objectives.	Pollution	No explicit relevant management actions.	Section 7.6 Unplanned release of MDO Section 7.7 Contingency spill response operations	
Lesser Sand Plover, Mongolian Plover <sup>4</sup>	Conservation Advice <i>Charadrius</i> <i>mongolus</i> (Lesser Sand Plover, Mongolian Plover) (DCCEEW, 2024a)	No explicit relevant objectives.	Habitat loss and degradation	No explicit relevant management actions.	Section 7.6 Unplanned release of MDO	
			Pollution/contamination	No explicit relevant management actions.	Section 7.6 Unplanned release of MDO Section 7.7 Contingency spill response operations	
Partridge Pigeon (eastern) <sup>4</sup>	Conservation Advice Geophaps smithii smithii (Partridge Pigeon [eastern]) (TSSC, 2015)	No explicit relevant objectives.	No explicit relevant threats	No explicit relevant management actions.	N/A	
Masked Owl (northern) <sup>4</sup>	Conservation Advice <i>Tyto</i> novaehollandiae kimberli (masked owl [northern]) (TSSC, 2015a)	No explicit relevant objectives.	No explicit relevant threats	No explicit relevant management actions.	N/A	
Nunivak Bar-tailed Godwit, Western Alaskan Bar-tailed Godwit <sup>4</sup>	Conservation Advice <i>Limosa lapponica baueri</i> (Bar-tailed godwit [western Alaska]) (DCCEEW, 2024k)	Conservation Advice <i>Limosa lapponica</i> A baueri (Bar-tailed godwit [western Alaska]) (DCCEEW, 2024k)	ation Advice <i>Limosa lapponica</i> ar-tailed godwit [western (DCCEEW, 2024k)	Habitat loss and degradation	Protect important habitat in Australia.	Section 7.6 Unplanned release of MDO Section 7.7 Contingency spill response operations
			Pollution/contamination	Protect important habitat in Australia.	Section 7.6 Unplanned release of MDO	
Red Goshawk <sup>4</sup>	Conservation Advice for <i>Erythrotriorchis</i> <i>radiatus</i> (Red goshawk) (DCCEEW, 2023f)	No explicit relevant objectives.	No explicit relevant threats	No explicit relevant management actions.	N/A	
Red knot	Conservation Advice for <i>Calidris canutus</i> (red knot) (DCCEEW, 2024c)	Advice for <i>Calidris canutus</i> No explicit relevant objectives. CCEEW, 2024c)	Pollution/contamination impacts	No explicit relevant management actions.	Section 7.1 Release of solid objects Section 7.6 Unplanned release of MDO	
			Habitat loss and degradation	No explicit relevant management actions.	Section 7.1 Release of solid objects Section 7.6 Unplanned release of MDO Section 7.7 Contingency spill response operations	
			Anthropogenic disturbance	No explicit relevant management actions.	Section 7.3 Marine fauna interaction	
Ruddy turnstone <sup>4</sup>	Conservation Advice for <i>Arenaria</i> <i>interpres</i> (ruddy turnstone) (DCCEEW, 2024I)	No explicit relevant objectives.	Pollution	No explicit relevant management actions.	Section 7.6 Unplanned release of MDO Section 7.7 Contingency spill response operations	

Name	Recovery plan/conservation advice/management plan	Relevant objectives	Threats/strategies identified as relevant to the Activity	Relevant conservation actions	Addressed (where relevant) in OEMP
Sharp-tailed sandpiper	Conservation Advice for <i>Calidris</i> <i>acuminata</i> (sharp-tailed sandpiper) (DCCEEW, 2024b)	No explicit relevant objectives.	Pollution	No explicit relevant management actions.	Section 7.1 Release of solid objects Section 7.6 Unplanned release of MDO Section 7.7 Contingency spill response operations
Terek sandpiper <sup>4</sup>	Conservation Advice for <i>Xenus cinereus</i> (terek sandpiper) (DCCEEW, 2024i)	No explicit relevant objectives.	Pollution	No explicit relevant management actions.	Section 7.6 Unplanned release of MDO Section 7.7 Contingency spill response operations
Tiwi Islands Hooded Robin, Hooded Robin (Tiwi Islands) <sup>4</sup>	Conservation Advice <i>Melanodryas</i> <i>cucullata melvillensis</i> (hooded robin [Tiwi Islands]) (TSSC, 2018a)	No explicit relevant objectives.	No explicit relevant threats	No explicit relevant management actions.	N/A
Tiwi Masked Owl, Tiwi Islands Masked Owl <sup>4</sup>	Conservation Advice Tyto novaehollandiae melvillensis (masked owl [Tiwi Islands]) (TSSC, 2015i)	No explicit relevant objectives.	No explicit relevant threats	No explicit relevant management actions.	N/A





## 3.2.14 Socioeconomic receptors

Socioeconomic activities and features that may occur in the OA and EMBA are set out in this section and summarised in Table 3-14.

#### Table 3-14: Socioeconomic-related activities/features that occur or may occur in the OA and EMBA

Value/sensitivity	OA presence	EMBA presence
Commercial fisheries – Commonwealth (Section 3.2.14.1)	<ul> <li>Commonwealth managed fisheries that overlap the OA (see Figure 3-10 and Table 3-15):</li> <li>Northern Prawn Fishery</li> <li>Southern Bluefin Tuna Fishery</li> <li>Western Skipjack Tuna Fishery</li> <li>Western Tuna and Billfish Fishery</li> </ul>	Commonwealth-managed fisheries that overlap the EMBA are described in Table 3-15 and shown in Figure 3-10
Commercial fisheries – NT (Section 3.2.14.1)	<ul> <li>NT managed fisheries that overlap the OA (Figure 3-11):</li> <li>Aquarium Fishery</li> <li>Coastal Line Fishery</li> <li>Demersal Fishery</li> <li>Offshore Net and Line Fishery</li> <li>Spanish Mackerel Fishery</li> </ul>	NT managed fisheries that overlap the EMBA are described in Table 3-15 and shown in Figure 3-11.
Energy industry (Section 3.2.14.2)	Within the OA, there are no established petroleum operations, however there are 2 existing pipelines within the vicinity—Bayu-Undan (greater than 0.1 km from the OA) and Ichthys (46.5 km from the OA).	The nearest offshore operating facility is the Santos-operated Bayu–Undan platform, approximately 375 km west of the OA. Oil and gas exploration permits are operated by other titleholders throughout the EMBA.
Defence activities (Section 3.2.14.3)	The OA intersects a designated defence practice area. During their surveillance, Australian Border Force vessels may transit the OA.	The EMBA intersects a practice area of the North Australian Exercise Area (NAXA). During their surveillance, Australian Border Force vessels may transit the EMBA.
Telecommunications cables (Section 3.2.14.4)	The North-West Cable System is not within the OA.	The North-West Cable System is located within the EMBA though a hydrocarbon spill will not have any impact on submarine cables
Shipping (Section 3.2.14.5)	The closest major commercial port to the OA is Darwin Port, 95 km away. No designated shipping channels intersect the OA.	Figure 3-14 shows the vessels recorded in the Australian Shipping Reporting System in 2021 and shipping density within the region. It shows the main commercial shipping channel tracking to the west of the OAs. Vessel traffic is expected within the EMBA.
Recreational tourism (Section 3.2.14.6)	The OA is in offshore waters that are highly unlikely to be accessed for tourism activities (e.g. recreational fishing and boating and charter boat operations). These activities tend to be centred around nearshore waters, islands and coastal areas.	There are several offshore shoals, banks, coral reefs, shipwrecks within the EMBA. These areas may be visited by recreational fishers, fishing charter vessels, scuba diving, snorkelling and other charter vessels. The Tiwi Islands are a popular tourist destination offering cruises, fishing, sailing and water tours among other cultural activities. Scuba diving, snorkeling and other charter vessels are also a significant tourist attraction, with operators visiting the numerous shipwrecks, coral reefs and artificial reefs and embarking on day or multiday trips out to offshore islands and shoals.

Value/sensitivity	OA presence	EMBA presence
Underwater cultural heritage (see Section 3.2.14.7)	There is one declared protected UCH site within the OA, the wreck of Japanese submarine 1-124, protected under the <i>Underwater Cultural Heritage Act 2018</i> (Cth) (UCH Act). During the design phase of the GEP, the pipeline route was deviated to avoid the I-124 Japanese wreck and its 800m radial exclusion zone, with the pipeline route passing 100m to the east of the exclusion zone at its closest point.	There are multiple sites protected under UCH Act and <i>Heritage Act 2011</i> (NT). Multiple known and unknown locations of shipwrecks, sunken aircraft, and historic (more than 75 years old) aircraft and shipwrecks and other sites occur or may occur within the EMBA (Figure 3-15).



### 3.2.14.1 Commercial fisheries

The NMR supports Commonwealth and NT State managed commercial fisheries that target various shark, demersal and pelagic finfish, molluscs, oyster pearls and crustacean species of commercial importance. Marine aquaculture (mariculture) within the EMBA is mostly associated with pearl oyster (*Pinctada maxima*) production in NT waters, which is focused in 4 main areas (NT Government, 2023):

- Bynoe Harbour
- Beagle Gulf
- Cobourg Peninsula and Croker Island
- around the islands northwest of Nhulunbuy.

The NT Government, via the Darwin Aquaculture Centre, is also encouraging the development of aquaculture of other species, including barramundi, sea cucumber, blacklip oysters, and giant clams. Barramundi is currently grown in ponds on the Adelaide River, and trials on Groote Eylandt and Goulburn Island are looking at growing clams in sea-based cages (NT Government, 2023).

The fisheries overlapping the OA and EMBA are shown in Figure 3-10, and Figure 3-11. Table 3-15 lists and describes the commercial fisheries and Santos' understanding of fishing effort based on publicly available information and consultation with Relevant Persons.

Consultation with the Australian Fisheries Management Authority (AFMA), NT Department of Industry, Tourism and Trade (NT Fisheries) and appropriate fisheries associations and licence holders is discussed in Section 4. A summary report including the outcomes of consultation with Relevant Persons, including any objections or claims and Santos' assessment of them, satisfying the requirements of section 24(b)(i)-(iii) of the OPGGS(E)R, is provided in Section 4.8. The full records of Relevant Persons consultation, as required by section 24(b)(iv) of the OPGGS(E)R, is provided in the Sensitive Information Report.

#### Table 3-15: Commonwealth and state fisheries that overlap the EMBA

Commonaid fishers	Licence Area OA EMBA		Description	Like
Commercial fishery			Description	
Commonwealth-managed				
Northern Prawn Fishery	✓	✓ ✓	Area: extends from Joseph Bonaparte Gulf across the top end to the Gulf of Carpentaria. Most of the Northern Prawn Fishery effort lies in the Gulf of Carpentaria, Joseph Bonaparte Gulf and along the Arnhem Land coast (DoA, 2014).         Gear: trawl.         Key target species: The key target species are banana prawns, tiger prawns and endeavor prawns. There are 2 fishing seasons—the season end date depends on catch rates:         Season 1 (mainly banana prawns caught): 1 April to 15 June         Season 2 (mainly tiger prawns caught): 1 August to end of November.         Fishing for scampi also occurs in deeper waters, with fishing effort spread across 2–3 months of the year (December to February).         Effort (2022): 54 active vessels; total catch 5,526 t (Butler et al. 2023)	Intera and conce south
Southern Bluefin Tuna Fishery	✓	✓	<ul> <li>Area: the Southern Bluefin Tuna Fishery spans the Australian Fishing Zone. However, it is only active in the south and southeastern Australian waters</li> <li>Gear: purse seine and pelagic long line.</li> <li>Key target species: southern bluefin tuna.</li> <li>Effort (2022): 30 active vessels (22 longline, 8 purse seine); total catch 5,972 t (Butler et al. 2023).</li> </ul>	No at there
Western Skipjack Tuna Fishery	~	~	Area: The Western Skipjack Tuna Fishery spans the Australian EEZ and adjacent high seas, from Cape York to the Victoria–South Australia border, including waters around Tasmania and the high seas of the Pacific Ocean. Gear: purse seine Key target species: skipjack tuna Effort (2020): None. There has been no fishing effort since the 2008–2009 season, and in that season, activity was concentrated off South Australia (Butler et al. 2023).	No ad there
Western Tuna and Billfish Fishery	~	1	<ul> <li>Area: Operates in Australia's EEZ and high seas of the Indian Ocean. In recent years, fishing effort has concentrated off south-west WA, with occasional activity off South Australia.</li> <li>Gear: pelagic longline.</li> <li>Key target species: bigeye tuna, yellowfin tuna, striped marlin, swordfish.</li> <li>Effort (2022) : 5 active vessels (2 pelagic longline, 3 minor line); 145 t (Butler et al. 2023).</li> </ul>	No ao OA o
Northern Territory-managed		1		
Aquarium Fishery	✓	✓	<ul> <li>Area: Includes freshwater, estuarine and marine habitats to the outer boundary of the Australian Fishing Zone. Most marine species are collected within 100 km of Nhulunbuy and Darwin. A specimen shell collection enterprise occurs around Ashmore Reef and Cartier Island (outside the EMBA).</li> <li>Gear: handheld, nets, and pots (dive-based).</li> <li>Key target species: fish, invertebrates, and plants for aquariums.</li> <li>Effort: unknown – no restriction on number of licences (NT Government, 2023b).</li> </ul>	No ac is pos
Bait net Fishery	×	~	<ul> <li>Area: Bait fishing is allowed from the high water mark to 3 NM seaward of the low water mark, excluding Darwin Harbour and Shoal Bay.</li> <li>Gear: bait net, cast net or scoop net</li> <li>Key target species: all fish species except barramundi, threadfin salmon, Spanish mackerel or mud crab.</li> <li>Effort: 2 licences are currently allocated (NT Government, 2023b).</li> </ul>	No fis Effort
Barramundi	X	✓	<ul> <li>Area: Barramundi fishing is allowed from the high-water mark to 3 nautical miles seaward of the coast (with exclusion zones and restrictions).</li> <li>Gear: nets are set and retrieved from dinghies and fish are processed onboard motherships.</li> <li>Key target species: barramundi and king threadfin</li> <li>Effort: 14 licences are currently allocated. Fishing effort spread across 8 months of the year (01 February to 30 September; NT Government, 2023b).</li> </ul>	No fis Effort
Coastal Line	×	~	<ul> <li>Area: Fishery is allowed from the high-water mark to 15 nautical miles seaward of the coast.</li> <li>Gear: lines, hooks, cast nets, scoop nets or gaffs.</li> <li>Key target species: black jewfish and golden snapper</li> <li>Effort: 52 licences currently allocated (NT Government, 2023b). Total catch in the western zone is limited to 145 t of jewfish and 4.5 t of golden snapper.</li> </ul>	No fis Effort
Coastal Net Fishery	X	✓ ✓	Area: Fishery is allowed from the high-water mark to 3 Nm seaward of the coast.         Gear: nets.         Key target species: mullet         Effort: 5 licences are currently allocated (NT Government, 2023b).	No fis Effort



#### hood of interaction with fishers

action with this fishery in the OA is possible; however, medium, high fishing effort are outside the OA. The areas of entrated effort are to the north and west of the Tiwi Islands and n of the OA.

active commercial fishing effort reported in the OA or EMBA; efore, interaction with this fishery is unlikely.

active commercial fishing effort reported in the OA or EMBA; efore, interaction with this fishery is unlikely.

ctive commercial fishing effort reported in the Commonwealth r EMBA, therefore interaction with this fishery is unlikely.

ctive commercial fishing effort reported in the OA. Some effort ssible in the EMBA for very limited periods of the year.

shery overlaps the OA. t is expected within the EMBA.

shery overlaps the OA. t is not expected within the EMBA.

shery overlaps the OA. t is not expected within the EMBA.

shery overlaps the OA. t is expected within the EMBA.
O	Licence Area					
Commercial fishery	OA	EMBA	Description			
Demersal Fishery	V	✓	<ul> <li>Area: Demersal fishing is allowed from 15 Nm to the outer boundary of the Australian Fishing Zone, excluding the area of the Timor Reef Fishery.</li> <li>Gear: lines, fish traps and semi-demersal trawl nets.</li> <li>Key target species: snapper (various species).</li> <li>Effort: 16 licences currently issued, with 7 boats nominated for fishing (NT Government, 2023b).</li> </ul>	Intera due t easte 100 n Effort		
Mollusc Fishery	X	✓	<ul> <li>Area: Mollusc harvesting is allowed from the high water mark out to the low water mark.</li> <li>Gear: collected by hand.</li> <li>Key target species: all molluscs and shellfish, except pearl oysters.</li> <li>Effort: 1 licence is currently allocated (NT Government, 2023b).</li> </ul>	No fis Very		
Mudcrab Fishery	✓	1	<ul> <li>Area: Mud crab harvesting is confined to coastal mudflats and estuaries, excluding Darwin Harbour, Kakadu National Park, Leaders Creek and most creeks adjoining Shoal Bay.</li> <li>Gear: pots.</li> <li>Key target species: mud crabs.</li> <li>Effort: 49 licences are currently allocated, each of which is allowed 60 pots (NT Government, 2023b).</li> </ul>	No fis Fishir the E		
Offshore Net and Line Fishery	✓	1	<ul> <li>Area: Operates in NT waters from the low water mark to the boundary of the Australian Fishing Zone. Most fishing is done in the coastal zone within 12 Nm of the coast, and immediately offshore in the Gulf of Carpentaria. The fishery has an area of approximately 522,000 km2.</li> <li>Gear: longlines or pelagic nets (there are restrictions on where certain gear can be used).</li> <li>Key target species: blacktip sharks, grey mackerel.</li> <li>Effort: Unknown – no restriction on number of licences issued (NT Government, 2023b).</li> </ul>	Intera the co the co licenc small		
Pearl Oyster Fishery	✓	~	<ul> <li>Area: high water mark to the outer boundary of the Australian fishing zone, 200 Nm offshore.</li> <li>Gear: harvested by hand only</li> <li>Key target species: pearl oysters</li> <li>Effort: 5 licences are currently allocated. A total of 138,000 oysters can be collected each year (NT Government, 2023b).</li> </ul>	While been High		
Spanish Mackerel Fishery	<i>✓</i>	✓ 	<ul> <li>Area: Commercial fishing for Spanish mackerel is allowed from the high-water mark to the outer boundary of the Australian Fishing Zone, which is 200 Nm offshore.</li> <li>Most fishing effort occurs near reefs, headlands and shoals and includes waters near Bathurst Island, New Year Island, northern and western Groote Eylandt, the Gove Peninsula, the Wessel Islands, the Sir Edward Pellew Group and suitable fishing grounds on the western and eastern mainland coasts.</li> <li>Fishing generally takes place around reefs, headlands, and shoals.</li> <li>Gear: trolling, handline.</li> <li>Key target species: Spanish mackerel.</li> <li>Effort: 15 licences allocated (NT Government, 2023b).</li> </ul>	Intera transi nearb		
Trepang Fishery	X	1	<ul> <li>Area: Trepang fishing is allowed from the high-water mark to 3 NM seaward of the coast. Predominately along the Arnhem Land coast, mainly around the Cobourg Peninsula and Groote Eylandt.</li> <li>Gear: harvested by hand either on foot or by diving, usually on neap tides during the dry season.</li> <li>Key target species: Barramundi and king threadfin</li> <li>Effort: 6 licences currently allocated (NT Government, 2023b).</li> </ul>	No fis Effort		



#### hood of interaction with fishers

action with this fishery in the OA is possible but highly unlikely to the concentration of fishing effort that occurs along the ern boundary of the Timor Reef fishery in water depths of 80m, to the north-east of the OA.

is expected within the EMBA.

shery overlaps the OA.

low effort is expected within the EMBA.

shery overlaps the OA.

ing effort is concentrated in the Gulf of Carpentaria (outside of EMBA); however, very low effort may occur within the EMBA.

action with this fishery in the OA is possible but unlikely due to concentration of fishing effort in coastal areas (within 12 NM of coast) and immediately offshore in the Gulf of Carpentaria. One ce holder may fish off the southwest end of the Tiwi Islands for II pelagic fish.

e there is a licence area that intersects with the OA, there has no active commercial fishing effort reported in the OA. effort is expected within the EMBA.

action with this fishery in the OA is possible with fishers siting within the area. Effort is expected within the EMBA at by shoals and banks, particularly in waters off Bathurst Island.

shery overlaps the OA. t is expected within the EMBA.



Figure 3-10: Commonwealth managed fisheries overlapping or proximal to the OA and EMBA and MEVA





Figure 3-11: Northern Territory managed fisheries overlapping or proximal to the OA and EMBA and MEVA





### 3.2.14.2 Energy industry

No established energy operations are located within or in the immediate surrounds of the OA. However, there are 2 existing pipelines within the vicinity—the Santos-operated Bayu-Undan pipeline (0.1 km or greater distance from the OA) and the INPEX-operated Ichthys pipeline (46.5 km distance from the OA). The closest operational offshore production facilities and in-field subsea infrastructure are the Eni operated Blacktip Gas, approximately 254 km south-west from the OA and the Santos-operated Bayu– Undan platform, approximately 375 km north-west from the OA. Petroleum retention lease area and exploration permit leases within the EMBA are held by various energy operators (and subsidiaries) including INPEX Browse, MEO International, Neptune Energy Bonaparte, Eni, EOG Resources and MBS Oil. The Petroleum retention leases and exploration permit leases within and near to the EMBA are currently held by various energy operators, shown in Figure 3-12.



Figure 3-12: Existing petroleum infrastructure overlapping or proximal to the EMBA and MEVA





#### 3.2.14.3 Defence activities

The OA intersects a practice area, and the EMBA intersects the practice and training areas of the North Australian exercise area and Darwin air weapons range (Figure 3-13). These areas are maritime military zones administered by the Department of Defence and used for offshore naval exercises and onshore weapons-firing training. The Australian Border Force also undertake civil and maritime surveillance (and enforcement) in Australian offshore maritime waters, which include the EEZ. During their surveillance, Australian Border Force vessels may transit the EMBA.



Figure 3-13: Defence training and exercise areas overlapping or proximal to the EMBA and MEVA





### 3.2.14.4 Telecommunications cables

The North-West Cable System (NWCS) is located approximately 2.5 km south of the OA. Extending 2,100 km from Darwin to Port Hedland, the NWCS connects Australia's remote northern and western regions, including offshore energy industry facilities, with onshore locations. Although the NWCS intersects the EMBA, a hydrocarbon spill will not have any impact on submarine cables.

### 3.2.14.5 Shipping

AMSA has established a network of shipping fairways off the north-west coast of Australia to manage traffic patterns. The shipping fairways are designed to keep shipping traffic away from offshore infrastructure to reduce the risk of a vessel collision (AMSA, 2013).

The use of the fairways is strongly recommended, and the International Regulations for Preventing Collisions at Sea 1972 apply to all vessels navigating within or outside the shipping fairways. Under the *Navigation Act 2012* (Cth), certain vessels operating in Australian waters are required to report their location daily to AMSA's Joint Rescue Coordination Centre (JRCC). This Australian Ship Reporting System is an integral part of the Australian Maritime Search and Rescue system.

The OA does not overlap any ports. Darwin Port is a major shipping port in Australia located approximately 90 km south-south east of the OA. In 2022–2023, there were 1,569 vessel calls to port (Landbridge Darwin Port, 2024). Darwin Port is a major port for vessels servicing operations offshore from north-west Australia. The primary shipping channels within the EMBA are between Darwin and Southeast Asian ports. Figure 3-14 illustrates the vessel movement density within the EMBA. Average vessel displacements and speeds for shipping vessels transiting the EMBA and OA include:

- bulk carriers averaging 55,300 t with speeds of 14 knots
- livestock carriers averaging 2,800 t with speeds of 12 knots
- general cargo vessels averaging 4,900 t with speeds of approximately 12 knots.

Although Darwin Port is the primary active port in the region, there is a port, Port Melville, located at the Tiwi Islands (outside of the EMBA), which is approximately 83 km north-east of the OA and 125 km north of Darwin.



Figure 3-14: Regional shipping overlapping or proximal to the EMBA and MEVA





### 3.2.14.6 Recreation and tourism

In the NT, there were 895,000 holiday visitors (for the purposes of tourism) during the year ending March 2024 with ~\$1.5 billion visitor expenditure (NT Tourism, 2023b). While tourism activities (e.g. recreational fishing and boating, charter boat operations) may occur within the OA, they are likely to be transitioning the area to access islands, shoals, and shipwrecks outside the OA.

In the NT, 95% of recreational fishing occurs in in area <5 km from the coastline (West et al., 2022). The peak fishing effort is between October to December and April to June (West et al., 2022). The mainland coastline, several shoals, and banks within the EMBA may be visited by small numbers of recreational fishers/charter vessels targeting fish inhabiting these shallower offshore features. The mainland coastline also offers recreation, and cultural and environmental tourism activities. Scuba diving, snorkelling and other charter vessels are also a significant tourist attraction, with operators visiting the numerous shipwrecks, coral reefs and artificial reefs and embarking on day or multiday trips out to offshore islands and shoals (INPEX Browse, 2010). The peak tourism period occurs between May to October. The Tiwi Islands are a popular tourist destination offering cruises, fishing, sailing and water tours among other cultural activities. Kakadu National Park is also an important visitor attraction which has coastal values that intersect the EMBA. Tourism and recreational activities are likely to be more concentrated within coastal waters of the EMBA, but activities such as deep-water fishing, diving, and snorkelling around offshore shoals and reefs may potentially take place in offshore areas of the EMBA.

#### 3.2.14.7 Underwater cultural heritage

Historic shipwrecks and sunken aircraft, including associated artefacts that have been in Australian waters more than 75 years, are subject to automatic protection under the *Underwater Cultural Heritage Act 2018* (Cth) (UCH Act). Shipwrecks, sunken aircraft, and other types of Underwater Cultural Heritage (UCH) that have been underwater for less than 75 years can be protected through an individual declaration by DCCEEW based on an assessment of heritage significance (DCCEEW, 2024). Underwater cultural heritage artefacts continue to be protected after removal from the water.

There is one declared protected UCH site within the OA, the wreck of Japanese submarine *I-124* (see Figure 3-15). The wreck is located <1 km away from the GEP, in water depths of <42 m. During the design phase of the DPD Project, the pipeline route was deviated to avoid the *I-124* Japanese wreck and its 800 m radial exclusion zone, with the pipeline route passing 100 m to the east of the exclusion zone at its closest point.

Multiple known shipwrecks, sunken aircraft, and historic (more than 75 years old) aircraft, shipwrecks and other sites occur within the EMBA (Figure 3-15). Table 3-16 describes the known and located UCH sites protected under the UCH Act and *Heritage Act 2011* (NT) within the EMBA and lists the distances to the OA. These include a steam ship (*Florence D*) that was sunk to the north-west of Bathurst Island, and a steamer ship (*Don Isidro USAT*) that was sunk adjacent to the west coast of Bathurst Island. These vessels and submarine *I-124* were sunk in 1942 during World War II and are listed under the Historic Shipwrecks Act 1976.

A maritime archaeological heritage assessment was undertaken by Cosmos Archaeology (2022) who reviewed historical sources, databases, and marine geophysical information. The assessment concluded there are no located shipwrecks, aircraft wrecks, dump sites, maritime infrastructure or UXO within the study area, which was defined as a 500 m buffer around the Barossa GEP route.

Santos engaged Cosmos Archaeology to undertake a maritime archaeological heritage assessment (MAHA) (Cosmos Archaeology, 2022) along the DPD pipeline route. The study area of the MAHA is defined as a minimum 1,000 m buffer on either side of the DPD pipeline route (i.e. both Commonwealth and NT waters, including NT Coastal Waters). An archaeological scope of works prepared by the Department of Territory Families, Housing and Communities, NT Heritage branch, in November 2021, informed the Cosmos Archaeology assessment. Cosmos Archaeology analysed data collected during the geophysical survey conducted by Fugro in 2021. The study found three seabed anomalies representing potential cultural objects (i.e. not natural in origin) within the MAHA study area, between pipeline kilometre point (KP) 25 to KP28 (Cosmos Archaeology, 2022). Two of these objects, which could not be determined as natural or cultural, were identified between 143 and 214 m away from the pipeline route and another single high-relief feature was located 68m from the pipeline route. This latter anomaly was considered as having only a remote chance of being associated with the I-124 wreck given its distance of over 2.5 km away from the centre point of the wreck (Cosmos Archaeology, 2022). Given the distance of these anomalies from the pipeline route and the nature of the installation activity, these anomalies were not considered likely to be impacted by the pipeline installation activity, and no further work to further identify them was recommended by Cosmos Archaeology (2022).

Cosmos Archaeology also noted that 29 known but unlocated shipwrecks and 25 known but unlocated aircraft wrecks are believed to have sunk within the MAHA study area vicinity based on recorded historical accounts (Cosmos Archaeology, 2022). Therefore, these unlocated shipwrecks and aircraft wrecks could potentially be located within the EMBA but outside the OA. Cosmos Archaeology identified 17 known shipwrecks, 5 unexploded



ordnance (UXO) and 6 instances of maritime infrastructure (including anti-submarine defences and telegraph cables) within the MAHA study area (outside of the OA) (Cosmos Archaeology, 2022).

Underwater cultural heritage sites that have a declared protected zone prohibit types of conduct within a designated zone. There are three sites that have a declared protected zone within the EMBA, being *I-124* (submarine), *SS Florence* and *SS Macumba*. These declared protected zones prohibit conduct within an 800 m radius, unless authorised by a permit issued under the UCH Act. One additional known shipwreck listed under the UCH Act is located outside the EMBA at the Cartier Island Marine Park: the *Ann Millicent* (wrecked in 1888).

Name	Protected under the UCH ActUnderwater heritage protected zonesProtected under the Heritage Act 2011 (NT)DescriptionProtected zonesHeritage Act 2011 (NT)Description		Site distance to OA (~km)		
B-25D Mitchell N5-140	~	×	×	Aircraft crashed off the coast of Nightcliff, NT in April 1943, cause unknown.	77
Booya	×	<ul> <li>✓ 150 m</li> <li>under Heritage</li> <li>Act 2011 (NT)</li> </ul>	1	Sailing vessel wrecked during Cyclone Tracy in 1974.	70
Brisbane	✓	×	x	Vessel struck Fish Reef near the entrance to Bynoe Harbour, NT in October 1881 where it became permanently stranded.	42
British Motorist	*	×	1	Vessel sunk during attacks by Japanese aircraft in February 1942, while in use by the British Merchant Navy for fuel transportation purposes.	81
Catalina PBY-4 PatWing10 #4 or #8 ("Catalina 6")	*	×	√	One of 3 Catalina aircrafts sunk at mooring in February 1942 by Japanese air raid. Part of USN Patrol Wing 10.	87
Dakota A65- 115 (VH-RGC)	~	×	×	Aircraft crashed off the coast of Mindil Beach, NT in September 1945.	75
Ellengowan	✓	×	1	Vessel sank at its moorings at the Channel Island quarantine station anchorage in 1888.	87
HMAS Kelat	~	×	~	Ship was sunk during attacks by Japanese aircraft in February 1942.	86
HMAS Neptuna	*	×	*	Ship was sunk during attacks by Japanese aircraft during February 1942, while in use by the Allies to transport people, troops and supplies.	82
HMAT Zealandia	✓	×	✓	Ship was sunk during attacks by Japanese aircraft in February 1942, while in use by the Allies to transport people, troops and supplies.	82
I-124 (Submarine)	V	✓ 800 m under the UCH Act	×	The submarine was sunk by multiple attacks by Allied Forces including Australian and US in January 1942. I-124 was an Imperial Japanese Navy minelaying submarine and the sinking resulted in the loss of all 74 crew.	overlaps
RAAF Catalina A24-1 ("Catalina 1")	✓	×	<b>~</b>	Aircraft crashed during takeoff in August 1945.	92
RAAF Catalina A24-206 ("Catalina 3")	~	×	×	Aircraft sunk from accidental depth charge explosion June 1945.	89
RAAF Catalina A24-69 ("Catalina 2")	✓	×	×	Aircraft caught fire by accident in December 1945 while moored in Darwin Harbour, NT.	87

Table 3-16: Located UCH protected under UCH Act and Heritage Act 2011 (NT) and distance to the OA

Name	Protected under the UCH Act	Underwater heritage protected zones	Protected under the <i>Heritage Act</i> 2011 (NT)	Description	Site distance to OA (~km)
Spitfire A58- 372 (ex-JG106)	✓	×	✓	Aircraft crashed into Clarence Strait, NT in July 1945.	120
SS Florence D	~	<ul> <li>✓ 800 m</li> <li>under the UCH</li> <li>Act</li> </ul>	×	Ship was sunk during attacks by Japanese aircrafts in February 1942, while chartered by the US Navy to serve as a blockade runner to transport supplies.	96.5
SS Macumba	~	✓ 800 m under the UCH Act	×	Merchant ship was sunk during an attack by 2 Japanese aircraft in August 1943, while carrying supplies and war materials from Brisbane to Darwin.	496
Subsea telegraph cable – duplicate	×	×	×	Duplicate subsea telegraph cable linking Darwin cable station to Banjoewangi cable station, Java, Indonesia. The duplicate cable was of the same composition as the original 1871 cable.	23
Subsea telegraph cable –replacement	×	×	✓	Replacement subsea telegraph cable linking Darwin cable station to Banjoewangi cable station, Java, Indonesia. Cable is of similar composition to the earlier 2 but contained an additional layer of brass tape around the core to protect the cable from marine borer (namely <i>Teredo navalis</i> ) attack.	46
Subsea telegraph cables landing	×	×	✓ 	First installation of an approximately 1,561 km long subsea telegraph cable linking Darwin cable station to Banjoewangi cable station, Java, Indonesia. The cable consists of seven stranded copper wires, insulated with gutta-percha latex, sheathed in galvanised iron wire armour, and an outside covering of tarred hemp.	81
USAT Don Isidro	~	×	×	Vessel was sunk during Japanese aircraft during February 1942, while in use by the British Merchant Navy for fuel transportation purposes.	42
USAT Mauna Loa	✓	✓ 100 m under <i>Heritage</i> <i>Act 2011</i> (NT)	✓	Ship was sunk during attacks by Japanese aircraft in February 1942, while chartered by the US Navy to transport supplies.	41
USAT Meigs	✓	✓ 100 m under <i>Heritage</i> <i>Act 2011</i> (NT)	✓	Ship was sunk during attacks by Japanese aircraft in February 1942, while chartered by the US Navy to transport supplies.	79
USN Catalina PatWing 10 #41 ("Catalina 4")	✓	×	✓	One of 3 Catalinas sunk at mooring in February 1942 by Japanese air raid. Part of USN Patrol Wing 10.	88.5
USN Catalina PatWing10 #4 or #8 ("Catalina 5")	×	×	×	One of 3 Catalinas sunk at mooring in February 1942 by Japanese air raid. Part of USN Patrol Wing 10.	89
USS Peary	<b>√</b>	✓ 100 m under <i>Heritage</i> <i>Act 2011</i> (NT)	✓	Ship was sunk during attacks by Japanese aircraft in February 1942.	80

During the Last Glacial Maximum, sea level was at its minimum at 125 m below the present-day sea level (Wessex, 2023). A significant portion of the EMBA is within the 125 m depth contour, which represents the furthest extent of historical human habitation and potential for First Nations UCH. Water depths within the OA are between Santos Ltd | Barossa Gas Project BAS-210 0224 Page 119 of 663

approximately 40 m to 50 m; therefore, there is potential for unknown First Nations UCH to exist in the OA. Given the extent of time since sea levels were at these low levels (~20,000 years ago), terrestrial landforms, and any associated heritage artefacts within the EMBA are likely to have been significantly influenced over thousands of years by erosion, sedimentation and deposition as sea levels increased to their present levels (Posamentier, 2023).

Santos engaged OzArk Environment and Heritage (OzArk) to conduct a desktop First Nations archaeological assessment for the DPD Project Area, including the NT Coastal Waters section, based on a detailed geomorphological assessment. This study focussed on the likelihood for deposits associated with the Last Glacial Maximum (LGM) to be impacted by the DPD Project. Only one location was identified where potential sediments associated with the LGM were indicated; this was in the vicinity of KP 36.4 to 37.9 (outside of the OA for this Activity) (OzArk, 2024). At this location, potential sediments are assessed likely to be at a depth of approximately 18 m below the sea floor. At this depth, no activities related to IMMR will have any direct or indirect impact on these potential sediments. In any event, the location of the potential sediments associated with the LGM is outside the OA (OzArk, 2024). No known First Nations UCH sites were identified by OzArk (2024) within the OA.



Figure 3-15: Underwater maritime cultural heritage overlapping or proximal to the EMBA and MEVA





### 3.2.15 Cultural features

Sections 3.2.15.2 to 3.2.15.12 provide detail on cultural features within the EMBA, other than the UCH sites described in Section 3.2.14.7.

### 3.2.15.1 Introduction

First Nations people have occupied the Australian continent for a period in the order of 65,000 years, making them the oldest continuous culture in the world. First Nations Australians' *connection to land is essential to the continued cultural survival of Australia's First Peoples as well as their economic and social development* (AIATSIS, 'Land Rights', Reuters).

Santos acknowledges the tradition of the First Nations people of Australia includes a cultural and spiritual connection to their land and waters, including sea country. These connections are rooted in their traditional communal beliefs and practices. First Nations people view their land and waters as integral to their identity, culture, and spirituality and they have a deep respect for the natural world. First Nations persons and groups that identify as saltwater people/groups have a complex relationship with sea country, based, for the most part, on inherited rights, including totemic affiliation, and ceremonial duties. Santos understands that First Nations groups of Northern Australia are generally aware of the nature and geographic extent of their areas of responsibilities over sea country.

The cultural heritage of First Nations people is defined by Indigenous tradition through traditional laws and customs amongst themselves.

It includes a vast array of cultural artifacts, practices and beliefs. The protected heritage of First Nations peoples is also of cultural value to Australia and the global community. The cultural value of First Nations protected heritage to Australia is evidenced and given force by a range of factors, including the laws, regulations and institutions established across Australia that are designed specifically to protect First Nations rights and interests in relation to sacred sites and other aspects of First Nations cultural heritage, including the *Native Title Act 1993* (Cth) (NT Act), *Aboriginal and Torres Strait Islander Heritage Protection Act 1984* (Cth) (ATSIHP Act), UCH Act, *Aboriginal Land Rights (Northern Territory) Act 1976* (Cth) (ALR Act) and *Northern Territory Aboriginal Sacred Sites Act 1989* (NT) (NTASS Act).

In identifying the cultural features of the OA and EMBA, Santos has considered:

- information shared during consultation for this GEP Coastal Waters OEMP,
- information shared during consultation/engagement for other Barossa EPs, including the Commonwealth Waters DPD EP;
- lay and expert evidence adduced in Munkara, as well as the court's reasoning and findings,
- expert anthropological advice provided by Santos' consultants (some of which was considered by the Court in *Munkara*), and
- other publicly available information.

Information about potential cultural features obtained during consultation/engagement for the D&C EP, SURF EP, and GEP EPs has been considered and included in this GEP Coastal Waters OEMP where potentially relevant, having regard to the recent guidance in *Munkara*.

Further to point 2 above, Santos was provided with information by First Nations people during consultation meetings for the D&C EP and by NOPSEMA in the course of preparing the D&C EP. NOPSEMA provided Santos with 4 separate letters from 4 Tiwi clan members to NOPSEMA in April 2022 requesting the statement of reasons for NOPSEMA's decision to accept Revision 3 of the D&C EP (2022 Statement of Reasons Requests22F<sup>5</sup>), and asked Santos to consider the relevance of the information provided in the letters to the D&C EP. Items raised in the letters from the Tiwi clan members include traditional hunting of marine species, totem species, dreamings, songlines and sacred sites, as well as broad concerns about potential impacts from other Barossa Gas Project activities (e.g. noise and light emissions) on the environment. Santos considered this feedback relevant to this GEP Coastal Waters OEMP given:

- songlines, sea country and sacred sites may occur in the general wider area of the Barossa Gas Project,
- the movement of marine and totemic species may occur within the EMBA for this GEP Coastal Waters OEMP, and

<sup>&</sup>lt;sup>5</sup> 2022 Statement of Reasons Requests asked for copies of statement of reasons to be sent to EDO email addresses. Santos Ltd | Barossa Gas Project BAS-210 0224



• potential environmental impacts associated with this Activity are similar to those associated with the D&C activity (such as noise and light).

Santos also notes that the Tiwi clan members who sent the April 2022 letters attended multiple Tiwi clan consultation sessions for this GEP Coastal Waters OEMP. As described in Section 4, Santos provided Tiwi people (including the authors of the four letters) extensive opportunities for consultation specifically on the activities proposed to be conducted under this GEP Coastal Waters OEMP.

Further to point 4 above, Santos commissioned an independent expert assessment by Dr Brendan Corrigan for the purpose of identifying UCH places along the route of the Barossa Gas Export Pipeline (GEP) west and northwest of the Tiwi Islands ("Corrigan 2023 Report"). As part of his work, Dr Corrigan reviewed extensive ethnographic studies of the Tiwi people in order to gain an historical understanding of their society, culture and hierarchy, and conducted extensive interviews amongst the communities.

Dr Corrigan has also prepared an anthropological survey report ("Corrigan 2024 Report") on cultural and spiritual values in relation to the DPD Project (including the DPD NT Coastal Waters, a link to the report is in Table 3-2. The Corrigan 2024 Report is based on a review of all relevant available ethnographic, linguistic and historical materials and consultations with key First Nations persons identified as having cultural and spiritual knowledge and authority associated with the study area. As far as possible, all persons understood to hold cultural and spiritual rights and interests in the study area, including those who assert relevant cultural knowledge, were identified and invited to participate.

Dr Corrigan concluded that a precise boundary which captures the extent of interests of both the Tiwi Islanders and Larrakia Peoples' in the context of the GEP is unclear at this time. However, cultural, and spiritual values of these groups are understood as extending out into the seas for an indeterminate distance. For example, the spiritual beings *Jirukupai* (crocodile man) and *Ampitji*) are thought by Tiwi Islanders to travel in the surrounding sea, but it is unclear precisely how far. This is also consistent with a range of views put to the Federal Court more recently, in the context of the GEP EP (see for example, Corrigan 2023). Similarly, Tiwi Islanders routinely travel large distances at sea for the purpose of fishing and hunting turtle and dugong. However, there is no settled evidentiary data on the actual extent of these cultural and economic activities in the context of a sea country claim or the like.

There are no native title claims or determinations registered, or sites recorded under the NTASS Act or sites protected under the ATSIHP Act, UCH Act or ALR Act, Aboriginal land rights claimed or granted under the ALR Act or Indigenous Protected Areas (IPAs) within the OA.

There are sacred sites along the mainland and island coastlines and potentially the surrounding waters that overlap the EMBA. These sacred sites may include features which lie both above and below the water (Aboriginal Areas Protection Authority [AAPA], 2022). All sacred sites in the NT are protected in accordance with the NTASS Act. None have been found to exist within the OA.

### 3.2.15.2 Meaning of Cultural features

In its evaluation of potential impacts, Santos has considered the Federal Court's guidance and findings in Munkara v Santos NA Barossa Pty Ltd (No 3) [2024] FCA 9 (Munkara) in identifying the cultural features of the environment. In Munkara, the Court clarified the meaning of 'cultural features' in the definition of 'environment' in section 4 (now section 5) of the OPGGS(E)R:

• The phrase cultural features has a 'communal aspect' to it. This necessitates proof that individual beliefs are broadly representative of the beliefs of other members of the group, although there does not need to be consensus<sup>6</sup>. An idiosyncratic view or belief of an individual may be a manifestation of the culture of that person's society, but if it is not broadly representative of the beliefs of a group, then it will not constitute a cultural feature <sup>7</sup>.

In the context of limb (a) of the definition of 'environment' in the OPGGS(E)R, 'cultural features' attaches to the word 'ecosystem' with all of its constituent parts, including people and communities. The focus must remain on the ecosystem, of which people form a part. This focus is not upon an individual person devoid of the context of the ecosystem<sup>8</sup>.

- In the context of limb (c) of the definition of 'environment', each of the circumstances that:
  - an area is the subject of a spiritual connection to Aboriginal people, provided that the connection is by the laws and customs of *a people*<sup>9</sup>,

<sup>&</sup>lt;sup>6</sup> Munkara at [922] and see also at [194]-[199].

<sup>&</sup>lt;sup>7</sup> Munkara at [204].

<sup>&</sup>lt;sup>8</sup> Munkara at [204].

<sup>&</sup>lt;sup>9</sup> Munkara at [201].

Santos Ltd | Barossa Gas Project



- an 'area' is the country of an Aboriginal person in accordance with Aboriginal traditional laws and customs<sup>10</sup>, and
- there exists in those areas, locations or places cultural heritage in the form of artefacts or other objects evidencing human occupation and activities over the course of human history<sup>11</sup>, may readily be described as a 'cultural feature' of that location, place or area.
- In order for there to be a 'cultural feature' of the environment, there must be a 'sufficiently cogent or coherent belief' that is 'sufficiently accepted' so that it can be described as having normative content for the people or community viewed as a constituent part of an ecosystem, such that a singular perspective will not suffice<sup>12</sup>. The beliefs and values must be held by the relevant people as a people<sup>13</sup>. Further, the question of whether a view is sufficiently cogent or coherent may be answered by reference to the customs and practices of the relevant people, including relevant customs and practices concerning the authority to speak on a topic or relevant customs and practices (if any) concerning the resolution of division<sup>14</sup>.
- The inquiry as to what is 'broadly representative' must be undertaken in the proper cultural context, including by assessing which persons are generally accepted as having authority to speak on the particular topic and excluding those persons who are culturally irrelevant<sup>15</sup>
- Evidence of dissenting views cannot be ignored because they tend against a finding that beliefs have broad acceptance<sup>16</sup>
- Proof that beliefs are broadly representative will be more difficult in the face of discord within the relevant group, and even more so when the discord is among persons of equivalent authority and persons having the same lineage<sup>17</sup>.

### 3.2.15.3 Native title

Native title was first recognised in Australia in *Mabo v Queensland (No 2)* (1992) 175 CLR 1 (Mabo). Consequent to that decision, the NT Act was enacted to provide a statutory mechanism for the recognition of claims for, and protection of, native title.

Native title claims are applications made to the Federal Court under the NT Act for a determination, or decision about native title in a particular area. A claimant application is made by a native title claim group which asserts it holds native title rights and interests in an area of land and/or water, according to its traditional laws and customs. By making a claimant application, the native title claim group seeks a decision that native title exists, so its physical and spiritual rights and interests are recognised by the common law of Australia. This is called a native title determination. A determination is a decision by a recognised body, such as the Federal Court or High Court of Australia, that native title either does or does not exist in relation to a particular area.

A native title claim group must demonstrate that the acknowledgement and observance of traditional laws and customs have continued substantially uninterrupted since sovereignty (capable of being recognised by the common law of Australia) (section 223(1) NT Act). Native title rights and interests are determined as a question of fact. For example, in Western Australia v Ward (2000) 99 FCR 316, [243], the Full Federal Court stated that:

Acknowledgment and observance may be established by evidence that traditional practices and ceremonies are maintained by the community, insofar as that is possible, off the land, and that ritual knowledge including knowledge of the Dreamings which underlie the traditional laws and customs, continue to be maintained and passed down from generation to generation. Evidence of present members of the community, which demonstrates knowledge of the boundaries to their traditional lands, in itself provides evidence of continuing connection through adherence to their traditional laws and customs.

A requirement for obtaining a positive determination of native title in court is proving that there is an organised group that occupied the claimed land and waters at the time of British annexation. The requirement of an 'organised society' is set out in *Mabo*.

From this, it is considered that it is a group of native titleholders that hold communal native title and that native title claims are understood to apply to the area over which First Nations groups are claiming their rights and interests.

<sup>10</sup> Munkara at [855].

<sup>&</sup>lt;sup>11</sup> Munkara at [200].

<sup>&</sup>lt;sup>12</sup> Munkara at [206].

<sup>&</sup>lt;sup>13</sup> Munkara at [208].

<sup>14</sup> Munkara at [206].

<sup>&</sup>lt;sup>15</sup> Munkara at [923].

<sup>&</sup>lt;sup>16</sup> Munkara at [923].

<sup>&</sup>lt;sup>17</sup> Munkara at [924].

Santos Ltd | Barossa Gas Project

A native title determination is where native title has been determined to exist, which may include only part of a native title claim and represents the lands and waters over which the native title group has been recognised to have rights and interests. Where a Court has determined that native title exists, those native title rights and interests will be held (often but not always) in trust by a Registered Native Title Body Corporate designated by the Native Title holders (section 57 NT Act).

Native title is, in any particular case, a collection of rights and interests the content of which varies according to the traditional laws and customs from which they are, in each particular case, derived. For example, these rights may include the right to have access, to camp, hunt, fish, use water, hold meetings, perform ceremony and/or protect cultural sites (see for example, *Akiba v The Commonwealth* (2013) 250 CLR 209).

For the Activity, there are no native title claims or determinations that overlap with the OA or EMBA. The areas of responsibility for regional native title representative bodies that overlap the EMBA are shown in Figure 3-16.



Figure 3-16: Representative Aboriginal/Torres Strait Islander Body Areas





#### 3.2.15.4 Indigenous land use agreements

An Indigenous land use agreement (ILUA) is a voluntary agreement between native title parties and other people or bodies about the use and management of areas of land and/or waters. An ILUA can be made over areas where:

- native title has been determined to exist in at least part of the area
- a native title claim has been made
- no native title claim has been made. .

While registered, ILUAs bind all native title holders to the terms of the agreement. ILUAs also operate as a contract between the parties. A register of ILUAs is maintained by the Native Title Registrar. The register of ILUAs does not disclose the existence of any ILUA which overlaps with the OA or the EMBA.

#### 3.2.15.5 Indigenous protected areas

Indigenous Protected Areas (IPAs) are areas of land and sea managed by First Nations groups as protected areas for biodiversity conservation through voluntary agreements with the Australian Government. IPAs are an essential component of Australia's National Reserve System, which is the network of formally recognised terrestrial parks, reserves and protected areas across Australia's landmass. There are currently 82 dedicated IPAs over 87 million hectares of land. There is also around 5 million hectares of Australia's sea areas in dedicated IPAs. Managing IPAs helps First Nations communities protect the cultural features of their country for future generations.

There are no IPAs that overlap the OA or EMBA.

#### 3.2.15.6 Sacred sites

Santos has applied for, and received on 23 December 2022 an Authority Certificate (C2022-098) from the Aboriginal Areas Protection Authority (AAPA), which covers potential seabed disturbance along the pipeline route in NT waters and a nominal ~1,000m buffer each side of the pipeline route, including the GEP route in the OA. Based on AAPA's research findings, there were no registered or recorded sacred sites, protected under the Northern Territory Aboriginal Sacred Sites Act 1989 (NT) (NTASS Act), identified within or adjacent to the area represented by a ~1,000m buffer each side of the pipeline route within the OA, nor any specific certificate conditions related to activities within the buffer area

There are many NT coastal sites along the mainland and island coastlines and potentially the surrounding waters that overlap the EMBA that are registered under the NTASS Act (whether registered, recorded, or not). These sacred sites may include features which lie both above and below the water (AAPA, 2022).

There are coastal areas (down to the low water mark) that intersect the EMBA which are formally recognised as Aboriginal land under the ALR Act.

Members of the Agalda clan, representing western parts of the Cobourg Peninsula, including coastal areas and adjacent sea country, raised with Santos during consultation on the Commonwealth Waters DPD EP and the Coastal Waters Construction Environmental Management Plan this GEP Coastal Waters OEMP that there are sacred sites around the west and south of Coburg and Croker Island. During consultation meetings for the SURF EP, Santos was also provided with sacred site locations within the EMBA by some members of the Tiwi Island clans. These sacred sites are located on the western coast of Bathurst Island that may also potentially intersect the outer boundary of the EMBA for this Activity.

The Kenbi (Cox Peninsula) Land Claim No. 37 (CoA, 2000) publishes detail on the location and significance of culturally significant First Nations sites within Darwin Harbour and Bynoe Harbour to the east (which are separated by the Cox Peninsula), including registered sacred sites. These sites and areas include those used for hunting, fishing, gathering, camping, ceremonies and associated with dreamings. There are numerous sites identified in this report within the EMBA, including those associated with dreamings of totemic marine fauna species, including Ngalwatnyini (manta ray dreaming), Memarrandjamul-nyini (dugong dreaming), lyn.garrayn-nyini (sea turtle dreaming) (CoA, 2000). The report also identifies 3 sites on the north-eastern side of Darwin Harbour.

All sacred sites in the NT are protected in accordance with the NTASS Act. Sacred sites may be registered in sea country (whether registered, recorded or not). Sacred sites may also be protected under the ATSIHP Act, Heritage Act 2011 (NT), the UCH Act, the ALR Act or the EPBC Act <sup>18</sup>.

<sup>&</sup>lt;sup>18</sup> For completeness Santos notes that on 23 October 2023 it was informed by the DCCEEW that applications had been received under the ATSIHP Act in relation to certain areas of the sea. Santos understands that these areas are outside the OA but overlap the EMBA. Santos understands that no decisions have been made by the Minister in relation to the applications at the time of writing. Santos Ltd | Barossa Gas Project BAS-210 0224



### 3.2.15.7 Land rights

The ALR Act governs Aboriginal land (not native title claims) in the NT. Land that has been granted or recommended for grant under the ALR Act is determined to be held communally by the "traditional Aboriginal owners" of that land. The ALR Act has enabled the establishment of Aboriginal Land Trusts (ALTs) to hold title to Aboriginal land granted in the NT under that Act.

Aboriginal land rights governed under the ALR Act do not extend past the low water mark of tidal waters overlaying the NT coastline. In coastal areas, grants of Aboriginal land under the ALR Act are made to the low water mark. *Northern Territory v Arnhem Land Aboriginal Land Trust* (2008) 236 CLR 24 confirmed that Traditional Owners of First Nations-owned NT coastline have exclusive access rights to the tidal waters overlying their land.

There is no Aboriginal land either claimed or granted under the ALR Act, or sea closures put into effect in accordance with that Act, that overlap with the OA. The EMBA does overlap land and tidal waters (between the low water mark and the high-water mark) granted under the ALR Act. This Aboriginal land is held by the Kenbi ALT, the Cobourg Peninsula Sanctuary ALT, the Tiwi ALT and the Delissaville/ Wagait/ Larrakia ALT (CoA, 2023)

Section 5(2) of the ALR Act provides that ALTs cannot exercise their functions in relation to land they hold except in accordance with directions given to them by the Land Council for the area in which the relevant land is situated. Where any such directions are given, ALTs must comply with them. Accordingly, ALTs cannot act independently of Land Councils. Under the ALR Act, the functions of Land Councils with respect to ALTs involve administering ALTs in their area, including storing their common seals and deeds of grant, maintaining a register of ALT membership, negotiating agreements on behalf of ALTs and receiving moneys on behalf of ALTs. The NLC is the relevant Land Council for the Cobourg Peninsula Sanctuary, Kenbi and Delissaville/Wagait/Larrakia ALTs, while the TLC is the relevant Land Council for the Tiwi ALT.

### 3.2.15.8 Marine parks

The EMBA for this GEP Coastal Waters OEMP overlaps with features of the North Marine Parks Network Management Plan (MPNMP) and the North-West MPNMP, which identify natural, cultural, and spiritual values associated with Australian Marine Parks (AMPs), specifically the Oceanic Shoals AMP and the Joseph Bonaparte Gulf AMP.

Santos acknowledges that Commonwealth and State Marine Park Management Plans have sought to recognise cultural interests of First Nations groups. Australian Marine Parks has described this framework as taking 'values into account' when making decisions and taking action in relation to marine parks. Australian Marine Parks summarises these values into natural, cultural, heritage and socioeconomic categories. Additionally, the Commonwealth and State Marine Park Management Plans state that there could be First Nations groups or native title representative groups who may have responsibility for sea country within marine park areas.

### 3.2.15.9 Cultural fishing and hunting activities

First Nations fishing activity in NT waters predominately occurs within inshore tidal waters. Approximately 80% of NT's coastline is recognised as being under Aboriginal land and sea ownership under the Aboriginal Land Rights Act 1976 (NT) (NT Government, 2022). Almost all traditional fishing effort (~93%) is concentrated within coastal waters (up to 3 Nm beyond the territorial baseline) of the NT coastline and Tiwi Islands (NT Government, 2017).

Darwin Harbour is utilised by Larrakia people for collecting marine resources, including fishing, hunting, crabbing and the collection of shellfish (Corrigan, 2024).

For the Tiwi Island people, traditional fishing effort is greatest near the larger communities of Wurrumiyanga on Bathurst Island, and Pirlangimpi and Milikapiti on Melville Island (DPIF, 2014). Traditional subsistence food sources include fish (mullet, mackerel, barramundi, trevally), mud mussels, mud crabs, long bums shellfish, oysters, yams, mullet, eggs (turtle and bird), chilli worms, mangrove worms, turtles, stingrays, and dugongs. Green turtles are the main species harvested in the water, while eggs of all turtle species are taken periodically (Tiwi Land Council, 2022). Information provided during Tiwi Clan meetings during consultation for the D&C EP indicated that some Tiwi people have a particular interest in turtles as a traditional food source.

Feedback from the 2022 Statement of Reasons letters identified the following First Nations people's use of country for fishing/gathering food (fish, shellfish, turtle/turtle eggs, (mud) mussels, (mud) crabs, yams, mullets, mangrove worm, mackerel, barramundi, trevally, (black lip) oysters, chilli worm, stingray, dugong and seagull eggs.

Traditional subsistence food sources are captured in a culturally appropriate manner learnt from ancestral generations and taught to emerging descendants. This occurs in normal family and community circumstances as well as within the practices of the First Nations groups (Corrigan, 2024).

With the support of the NT Government, Darwin Aquaculture Centre is working with Tiwi People to develop aquacultural enterprises that provide employment and business opportunities (Land Development Corporation,



n.d.). Aquacultural options include Barramundi, Trepang, Mud Crab, Prawns, Oysters and Clams (Tiwi Land Council, 2021).

### 3.2.15.10 Culturally significant marine species

In consultations with Tiwi Clans for the D&C EP, some Tiwi people emphasised that marine turtles are regarded by Tiwi people as totemic and culturally significant species. Therefore, environmental protection measures for marine turtles are important to Tiwi people.

Information about First Nations cultural beliefs and connection with their sea country, within and adjacent to the D&C EMBA, was provided during First Nations consultation meetings for the D&C EP and also from other information provided by NOPSEMA to Santos. As noted above, NOPSEMA provided Santos with four separate letters from Tiwi clans members to NOPSEMA in April 2022 requesting the statement of reasons for NOPSEMA's decision to accept Revision 3 of the D&C EP (2022 Statement of Reasons requests). The 2022 Statement of Reasons requests indicated that Tiwi people also consider fish, dugong and whales to hold cultural significance as totemic species (in addition to marine turtles), and that various marine species are traditional food sources for Tiwi people (refer Section 3.2.15.9). However, the significance of these species was not raised with Santos in its communal consultation sessions with Tiwi people for any of the Barossa OEMPs, noting that the Tiwi clan members who sent the 2022 Statement of Reasons Requests attended multiple Tiwi clan consultation sessions for this GEP Coastal Waters OEMP.

The Northern Land Council (NLC) in a submission as part of the consultation for the D&C EP indicated a number of marine species that are significant to Aboriginal dreamings including birds, crocodiles, crows, whales, manta rays, crabs, dugong, sea turtle, gropers, sea-eagles, octopus and other turtles.

The Corrigan 2024 Report also confirmed that Larrakia people identified turtle, dugong, and sting ray dreamings close to Talc Head (within the EMBA) and noting these have significant importance regarding resources and the spiritual dimensions of Larrakia life. Dreamings were identified as being associated with the sea, winds and stars and regarding the moon and the seasons, mermaid dreaming and dreamings near the Charles Point lighthouse. The term dreaming is used throughout the Corrigan 2024 Report to denote knowledge, songs and narratives associated with Aboriginal religious understandings which set out the origins of the social and physical world and expected behaviours within it.

The Corrigan 2024 Report also identified species important for protection including turtles, crocodiles, dugong, dolphins, whales, and the seagrass beds near Kings Table (within the EMBA).

Terrestrial species of cultural significance are outside the EMBA and therefore are not considered further in this GEP Coastal Waters OEMP.

#### 3.2.15.11 Sea country connection

As outlined in Section 3.2.15.2, Santos acknowledges that the cultural features of the environment include the circumstance that First Nations people have spiritual connections to a particular place within that environment, or that the place forms part of the country of a First Nations group, in accordance with the traditional laws and customs of that group. As such, the circumstance that an area of the environment is part of the sea country of a First Nations group have a spiritual connection, is a cultural feature of that area of the environment.

The North MPNMP (DNP, 2018a) states:

Sea country refers to the areas of the sea that Aboriginal people are particularly affiliated with through their traditional lore and customs. Sea country is valued for Indigenous cultural identity, health and wellbeing. Across Australia, Indigenous people have been sustainably using and managing their sea country for tens of thousands of years.

The nature of sea country was the subject of extensive lay and expert evidence in the Munkara proceeding, to which Santos has had regard in its consideration of cultural features of the environment. Based on this evidence, Santos understands that:

the concept of country is intimately connected with questions of cultural authority. The First Nations group
who is responsible for that area of country has authority to speak in relation to that country, and has
custodian responsibilities in respect of that country. One group's area of sea country will end where the
next group's begins, although groups may share responsibility for particular Dreamings which traverse
different areas of country; and



sea country connections may manifest in the telling of stories about foundational creation myths explaining features of the landscape or particular species<sup>19</sup>.

In order to identify areas of sea country which may be affected by activities under this GEP Coastal Waters OEMP, Santos consulted broadly with First Nations groups and representative organisations both in respect of this GEP Coastal Waters OEMP and its other Barossa EPs. Based on this consultation and Santos' review of publicly available information, Santos has identified that the EMBA likely intersects with sea country interests, although the geographical extent of sea country interests is inherently indeterminate at this time.

#### Features of sea country

During consultation on this GEP Coastal Waters OEMP and previous Barossa EPs, some First Nations Relevant Persons provided additional context as to the manifestation of their sea country connection, being particular stories and creation myths which they believe to be present within the EMBA. Santos acknowledges that expressions of sea country connection may be particular to families and individuals within groups and that there is accordingly divergence in the details of such stories within groups. Notwithstanding this, the information provided is summarised below and has been considered by Santos in the preparation of its EP, including with the benefit of expert anthropological advice.

Dr Corrigan documented a range of views on Tiwi clans' connection with sea country and considered claims for several items to be protected in accordance with Tiwi law and custom (Corrigan, 2023). This included:

- the travels of the Crocodile Man
- the location and existence of a 'Mother Ampitiji' .
- the travels of Ampitji •
- the necessity to look after country in a manner that seeks to ensure no industrial accidents occur which . might affect sea country and marine resources (including spiritual connections to the same)
- the Imunka force present in the seas
- the location of a place under the sea where spirits go to upon people's death and then being moved on • from the world of the living through Pukamani ceremony.

Tiwi Islanders interviewed by Dr Corrigan about the location of the above items expressed a variety of views. This is supported by the observations and findings of the Court in Munkara<sup>20</sup>.

Dr Corrigan's 2024 reporting also documents input from Larrakia people and relevant First Nations persons from Belyuen and Wagait, who also advise the presence of a range of ancestral beings and dreaming stories of relevance to the Darwin Harbour, surrounding seas and the DPD project footprint. None of these cultural features are known to be associated with any specific or particular places in the DPD project footprint, but rather have a more general association with the wider area, as well as having associations with particular and specific places outside of the DPD project footprint.

#### **Spiritual beings**

As part of consultation in the course of preparing the D&C EP some First Nations Relevant Persons expressed cultural connections with sea country in terms of spiritual beings. Information about First Nations cultural beliefs and connection with their sea country, within and adjacent to the D&C EP EMBA, was provided during First Nations consultation meetings for the D&C EP and from other information provided by NOPSEMA to Santos (2022 Statement of Reasons requests).

During Tiwi Clan consultation meetings for the D&C EP, Tiwi people spoke about the importance of their spiritual dreaming which protects the Tiwi Islands from man-made and natural disasters. Santos recognises that some First Nations Relevant Persons fear sickness or other adverse effects from the actions of spiritual beings in response to impacts on the environment of sea country itself. A key Tiwi creation story concerns a spiritual being (or spiritual beings) called Ampitii (sometimes known as a Rainbow Serpent). The Court in Munkara considered lay and anthropological evidence about this creation story at [78]-[81], noting that while there was significant divergence in spiritual beliefs concerning Ampitji, it was not disputed that the spiritual belief in one or more Ampitji is a feature of Tiwi spiritual life and that Ampiti may have a role to play in ensuring compliance with Tiwi law.

During Croker Island consultation meetings in Darwin, Croker Islanders conveyed their affiliation to their land and sea. They advised that their culture is at the coast and includes everything in the water including the marine life.

<sup>&</sup>lt;sup>20</sup> See Munkara at [871], [1003], [1011]–[1014], [1027] and [1212]. Santos Ltd | Barossa Gas Project

Croker Island people informed Santos during D&C EP consultation about their connections to sea country. Sea country was defined as to the north of Cape Croker out to the deep water (referred to as Inigarrka). Inigarrka is considered the most sacred place in the ocean and the Croker Island people are prohibited from the sacred area. Santos recognises the potential for sea country and songlines to extend into the EMBA for the activity the subject of this GEP Coastal Waters OEMP.

In relation to the GEP EP, Dr Corrigan concluded that, in accordance with Indigenous tradition, there were no specific UCH places along the GEP route that may be affected by the activities under the GEP EP: that there are no known sacred sites or some other specific places that are part of well-known sets of ancestral creation stories amongst the Tiwi people.

The Court in *Munkara* reached a similar conclusion on tangible cultural heritage, finding that the evidence was insufficient to show anything other than a negligible chance that there exists one or more objects of archaeological value along the GEP route<sup>21</sup>. Regarding intangible cultural heritage, the Court found that the evidence before the Court was insufficient to prove that the accounts given by the Applicant's witnesses in relation to Ampitji and the Crocodile Man were broadly representative of a belief held by the relevant people as people, such that the belief would constitute a cultural feature<sup>22</sup>. The Court also found that there was insufficient evidence in relation to Imunka<sup>23</sup> to establish that the belief constituted a cultural feature<sup>24</sup>.

Whilst these conclusions of the Court and Dr Corrigan were made in relation to activities covered by the GEP EP, the conclusions are also relevant to this GEP Coastal Waters OEMP due to the spatial proximity with the GEP EP activities. That is, no intangible cultural heritage values and sensitivities constituting a cultural feature have been identified along the GEP and DPD route (on DPD, see Corrigan, 2024).

In its correspondence to Santos of 25 August 2023 in relation to the D&C EP, NOPSEMA drew Santos' attention to 2 reports provided to NOPSEMA by the EDO on behalf of 7 Tiwi Islander clients on 21 July 2023. These reports related to the GEP EP (EDO GEP Reports), which NOPSEMA said may contain information relevant to the EMBA by the Activity covered by this GEP Coastal Waters OEMP. One of the EDO GEP Reports was prepared by Mr Lewis. The Court in Munkara doubted the rigor of Mr Lewis' anthropological work and, as referred to above, ultimately found that his opinions constituted him acting as an advocate rather than assisting the Court to arrive at the correct answer<sup>25</sup>. The other EDO GEP Report was prepared by Dr O'Leary. The Court ultimately placed no weight on this report and dismissed it, along with the subsequent reports prepared by Dr O'Leary, for all purposes<sup>26</sup>.

The EDO GEP Reports claim to provide an assessment of the locations of potential impacts to Indigenous UCH sites along the GEP route. While the locations of these claimed sites of significance are partially within the Activity EMBA, the locations and significance of these claimed sites as put forward in the EDO GEP Reports is disputed by the Corrigan 2023 Report.

The Corrigan 2023 Report included consideration of detailed expert reports on archaeology and sedimentology along the GEP route conducted by Wessex Archaeology and Dr Posamentier; and the EDO GEP reports. The Corrigan 2023 Report concluded there are no specific UCH places along the GEP to which people, in accordance with Indigenous tradition, may have spiritual and cultural connections that may be affected by the GEP EP activities. As the southern section of the GEP extends into the OA of this GEP Coastal Waters OEMP, this conclusion also applies to this Activity.

The Corrigan 2023 Report provided the following independent expert comments on the EDO Reports:

- The EDO Reports come to dramatic conclusions about cultural heritage elements in the vicinity of the GEP which overestimate the consistency of the views of the EDO clients with those held by the wider jural public of the Tiwi Islanders
- Some Tiwi Islanders express views consistent with the EDO Reports, but the authors of those reports failed to consider and take account of other alternative expression
- The narratives contained in the EDO Reports are not anything like the narratives described to Dr Corrigan in the interviews he undertook
- The location or even the existence of a mother Ampitji is not agreed by all relevant parties

<sup>&</sup>lt;sup>21</sup> Munkara at [1306].

<sup>&</sup>lt;sup>22</sup> Munkara at [1003] and [1014].

<sup>&</sup>lt;sup>23</sup> Referred to in Munkara as Yiminga.

<sup>&</sup>lt;sup>24</sup> Munkara at [946].

<sup>&</sup>lt;sup>25</sup> Munkara at [1136]-[1139].

<sup>&</sup>lt;sup>26</sup> Munkara at [879] and [1198].

Santos Ltd | Barossa Gas Project



- Dr O'Leary (the author of one of the EDO Reports) does not mention any qualification he holds for which he might rely upon to undertake detailed and nuanced ethnographic enquiries in the context of a controversial industrial project
- Dr O'Leary incorrectly assumes an accuracy of the advice he received about the location of paleo sub-sea burial places
- The EDO Reports do not correctly identify any specific UCH places along the Barossa GEP Route. .

Dr Corrigan also identified a constant theme in his interviews with the Tiwi Islanders that Ampitji travel within the waterholes of the Tiwi Islands and surrounding the Tiwi Islands and the crocodile man, Jirukupai, is also said by some to traverse the seas towards the OA. Dr Corrigan accepts, this is offset where some senior Tiwi people make the point that the OA is, in their view, a long way away from the Tiwi islands and that Jirukupai and Ampitji do not go out that far into the water. Of direct relevance these sorts of Tiwi cultural and spiritual values were tested in the Federal Court and were found not to be consistently spread amongst relevant Tiwi Islanders and in any event did not represent a particular 'place' of cultural and spiritual significance.

An important outcome of Dr Corrigan's research is that no sacred sites or dreamings are shown to be directly impacted by the DPD project footprint (which includes this Activity OA), although this is not to say that some persons do not have fears that this could be the case in the event of an unplanned event (Corrigan, 2024).

Santos recognises the importance of cultural and spiritual beliefs to First Nations people. Santos recognises that some First Nations remain concerned about the potential for adverse consequences to First Nations people and natural environment, that may arise as a result of disturbance from the Barossa Gas Project to spiritual dreaming and culturally important spiritual beings.

### 3.2.15.12 Summary of cultural features

Cultural features relevant to the Activity, as presented in Section 3.2.15, are summarised in Table 3-17. Table 3-17 also summarises the context for the identified cultural features, sourced information, and an assessment of relevance to the EMBA or OA (if known).

The cultural features presented in Table 3-17 are further assessed in the impact and risk assessment sections (Sections 6 and 7), as applicable. Context for these aspects is described below:

- Cultural heritage protected areas cultural knowledge and the passing down of cultural education to children can occur from performing of ceremonies and rituals and through dreaming narratives and songlines. Traditional laws and customs amongst a group or groups can define indigenous traditions amongst the group or groups. For example, laws and customs can provide a format for social life and ceremonial matters. The transfer of knowledge of traditional law and customs may be integral to a group's<sup>27</sup> intangible cultural heritage (UNESCO, 2003) There may be implications to the transfer of First Nations knowledge if, for example, relevant aspects of the environment disappear. Ongoing observance of First Nations traditional laws and customs can also be recognised through Native Title determinations, and knowledge of and connection with country (land and sea) can be recognised through a range of mechanisms including indigenous land use agreements, indigenous protected areas and Aboriginal land rights claims.
- Sacred sites areas that are traditionally accessed by First Nation people, such as sea country and sacred sites, are important for transferring traditional knowledge and for caring for country. If physical landscapes are altered this could impact the values of sacred sites. Sacred sites and protection of these is a known cultural heritage concern.
- Cultural fishing, hunting and gathering Through consultation it was identified that a number of marine species provide sustenance to some First Nations people and are obtained through cultural fishing, customary hunting (turtles and dugongs) and gathering (turtle eggs and seagull eggs).
- Culturally significant marine species A range of marine species (such as marine turtles, fish, dugongs, whales, sea-eagle, crocodile and manta rays) were raised during consultation as being important for Aboriginal dreaming, or as having totemic status and significance culturally. The First Nations people maintain a continuing spiritual connection with sea country, through caring for sea country and access to cultural food sources.

<sup>&</sup>lt;sup>27</sup> As noted in Munkara v Santos NA Barossa Pty Ltd (No 3) [2024] FCA 9, this cultural heritage must be held communally by the group, although need not be the subject of consensus. Santos Ltd | Barossa Gas Project BAS-210 0224 Page 132 of 663

- Marine Parks Commonwealth and State Marine Park Management Plans have sought to recognise cultural interests of First Nations groups within Marine Parks, and the sea country value of Marine Parks to First Nations people.
- Sea country connection through Songlines Cultural stories and songlines can extend from the shoreline
  to deep water areas, and they tell an important cultural story (Corrigan, 2023). If spiritual injury occurs from
  an activity, some First Nation people believe that songlines can be damaged. It is believed that damaging
  songlines may have the potential to interfere with ability for First Nation people to reproduce cultural
  knowledge and continue to provide cultural education of their children.
- Sea country connection through Dreaming sites and stories, and spiritual beings Some First Nations
  people believe dreamings relate to powerful creative ancestors who left much of the natural and human
  world behind them as they travelled (Corrigan, 2023). It is believed ancestors can travel to areas such as in
  the water or land below the seas, where these ancestors continue to use these areas. Some First Nations
  people are of the opinion that if spiritual injury is caused it can damage dreaming tracks. They believe it is
  their responsibility to look after these dreaming sites to protect the known travels of the spiritual beings.
  Information provided to Santos by First Nations communities during consultation, also highlighted the
  importance of cultural spiritual beings, such as Ampitji, as protectors of First Nations communities, and that
  if spiritual beings are upset or offended it can result in natural disasters or sickness among First Nations
  communities.

### Table 3-17: Summary of cultural features and heritage values

Identified cultural feature	Description	EP Source	OA presence	EMBA presence			
Archaeological heritage							
First Nations UCH	A First Nations archaeological assessment for the DPD Project Area was based on a detailed geomorphological assessment. This study focused on the likelihood for deposits associated with the Last Glacial Maximum (LGM) to be impacted by the DPD Project. Only one location where potential sediments associated with the LGM were indicated was in the vicinity of KP 36.4 to 37.9 (outside of the OA). At this location, potential sediments are assessed likely to be at a depth of approximately 18 m below the sea floor. At this depth, no activities related to the construction of the DPD project will have any direct or indirect impact on these potential sediments. In any event, the location of the potential sediments associated with the LGM is outside the OA. There are no declared protected First Nations UCH sites within the OA.	Desktop First Nations Archaeological Assessment Report: Darwin Pipeline Duplication Project, (OzArk, 2024)	No	Possible (spatial extent undefined)			
Tangible values							
Native title	First Nations people have interests in an area of land and/or water according to their traditional laws and customs, as recognised through cultural heritage legal and regulatory frameworks. There are no native title claims or determinations that overlap with the OA or EMBA (Section 3.2.15.3). The areas of responsibility for regional native title representative bodies that the EMBA are shown in Figure 3-16.	Spatial datasets were downloaded from the National Native Title Tribunal website and confirmed during consultation with First Nations people and representative groups.	No	No			
Indigenous land use agreements	There are no ILUAs within the OA and EMBA (Section 3.2.15.4)		No	No			
Indigenous protected areas	There are no IPAs that overlap the OA or EMBA (Section 3.2.15.5)		No	No			

Identified cultural feature	Description	EP Source	OA presence	EMBA presence
Sacred Sites	<ul> <li>There are no known registered sacred or First Nations UCH sites within the OA.</li> <li>There are many NT coastal sacred sites along the mainland and island coastlines and potentially the surrounding waters that overlap the EMBA. (Section 3.2.15.6):</li> <li>•</li> </ul>		No	Yes
Land rights	There is no Aboriginal land either claimed or granted under the ALR Act, or sea closures put into effect in accordance with that Act, that overlap with the OA. The EMBA does overlap areas of land and tidal waters (between the low water mark and the high-water mark) granted under the ALR Act. This Aboriginal land is held by the Cobourg Peninsula Sanctuary ALT, the Tiwi ALT, the Kenbi ALT, and the Delissaville/Wagait/Larrakia ALT (Section 3.2.15.7).		No	Yes
Marine Parks	The North MPNMP and the North-West MPNMP identify natural, cultural and spiritual values associated with AMP's, specifically the Oceanic Shoals AMP (Section 3.2.15.8).	DNP, 2018a; 2018b	No	Yes
Cultural fishing, hunting, and gathering	Cultural fishing, hunting, and gathering of marine species such as fish, shellfish, worms, stingrays, turtles, dugongs, turtle and bird eggs occur within the EMBA (Section 3.2.15.9). Cultural fishing, hunting and gathering of marine species is possible although not expected within the OA given water depths (>40 m) and distance from nearest shoreline (~30 km).	shing, hunting, and gathering of marine species such as fish, vorms, stingrays, turtles, dugongs, turtle and bird eggs occur within (Section 3.2.15.9). shing, hunting and gathering of marine species is possible although ted within the OA given water depths (>40 m) and distance from noreline (~30 km).		Yes
Culturally significant narine species       First Nations persons and groups that have a deep connection with the sea through totems and dreamings such as marine fauna (marine turtles, whales, dugong) and consider them to be of cultural significance (Section 3.2.15.10).		<ul> <li>2022 Statement of Reasons requests and NLC consultation feedback in relation to the D&amp;C EP</li> <li>Consultation feedback and Dr Corrigan reports (2023, 2024) including a view of extensive ethnographic studies.</li> </ul>	Yes	Yes
Intangible values	-	-		
Sea country connection through SonglinesSonglines can go from land to sea and were identified as important by the Croker Island and Tiwi Islands people, as well as Larrakia people and other First Nations people's with interests in the DPD Project Area (which includes this Activity OA). They ordinarily traverse areas in a manner of travelling from named places to named places (Sections 3.2.15.11).		Consultation feedback and Corrigan reports including a view of extensive ethnographic studies	Possible (spatial extent undefined)	Possible (spatial extent undefined)

Identified cultural feature	Description	EP Source	OA presence	EMBA presence
Sea country connection through Dreaming sites and stories and spiritual beings	<b>Dreaming</b> Dreamings were identified as being in the sea, winds, and stars and regarding the moon and the seasons, mermaid dreaming and dreamings near the lighthouse (Section 3.2.15.10). Several marine species are significant to Aboriginal Dreaming such birds, crocodiles, shellfish, whales, manta rays, crabs, dugong, sea turtle, gropers, sea-eagles and octopus (Sections 3.2.15.11, 0).	<ul> <li>Consultation feedback and Dr Corrigan reports (2023, 2024) including a view of extensive ethnographic studies</li> <li>NLC consultation feedback in relation to the D&amp;C EP.</li> </ul>	Possible (spatial extent undefined)	Possible (spatial extent undefined)
	<b>Spiritual beings</b> Spiritual beings are important to Tiwi Island people, as well as Larrakia people and other First Nations people's with interests in the DPD Project Area (which includes this Activity OA) for their role as protectors of First Nations people and the natural environment. Spiritual beings are believed to be present in the vicinity of the islands (Sections 0)).	Consultation feedback and Dr Corrigan reports (2023, 2024) including a review of extensive ethnographic studies	Possible (spatial extent undefined)	Possible (spatial extent undefined)

# 4. Consultation

### 4.1 Consultation background

Santos has undertaken a comprehensive consultation program for the Barossa Gas Project commencing with the initial primary approval (Barossa Area Development Offshore Project Proposal (OPP). Consultation with stakeholders on the OPP occurred during 2017 and included an eight-week public comment period prior to submission of the OPP to the National Offshore Petroleum Safety and Environmental Management Authority (NOPSEMA) for assessment.

Santos undertook consultation (including formal public comment periods for each) to support the primary approvals for the Darwin Pipeline Duplication (DPD) Project, for the Barossa GEP sections in NT Coastal and Inland Waters to the Darwin LNG plant. The primary approvals include for the construction, installation, operations and decommissioning phases of the project and consist of:

- a referral and subsequent Preliminary Documentation Report under the *Environment Protection and Biodiversity Conservation Act 1999* (Cth) (EPBC Act) - EPBC 2022/09372, granted in March 2024, and
- a referral and subsequent Supplementary Environmental Report under the Northern Territory Environment Protection Act 2019 (NT EP Act), approved in December 2023 by the NT Minister for Environment, Climate Change and Water Security (EP2022/022-001), on the recommendation of the NT Environmental Protection Authority (EPA).

Santos notes that the information contained in the *Production Operations Information Booklet* includes for both the Barossa Production Operations EP activities and this Coastal Waters Operations Environmental Management Plan (OEMP) activities.

Extensive consultation has been undertaken for each of the activity specific EPs and EMPs, as well as other regulatory approvals prepared for different stages of the Barossa Gas Project.

These have included:

- Barossa Gas Export Pipeline (GEP) Installation EP (including through ConocoPhillips, as previous operator of the Barossa Development) accepted by NOPSEMA in March 2020.
- Barossa Development Drilling and Completions EP accepted by NOPSEMA in December 2023.
- Barossa Subsea Infrastructure Installation EP accepted by NOPSEMA in February 2024.
- DPD Environment Plan accepted by NOPSEMA in October 2024.
- Barossa DPD Project Coastal Waters Construction Environmental Management Plan accepted by the NT Department of Mines and Energy (formerly the NT Department of Industry, Tourism and Trade – Energy Division) in November 2024 for construction activities to be undertaken in NT Coastal Waters.
- Barossa Production Operations EP currently under assessment by NOPSEMA. See below at Section 4.2 in relation to consultation on this EP.

Santos has also undertaken consultation on activities managed under the Bayu-Undan to Darwin Gas Export Pipeline Environment Plan. The Bayu-Undan pipeline transports natural gas from the Bayu-Undan offshore platform in the Timor Sea to the Darwin Liquefied Natural Gas (DLNG) facility, and adjacent to the Barossa GEP in NT Coastal Waters. The Bayu-Undan Field is approaching end of field life, at which time production will cease at the Bayu-Undan field. Activities to be managed in the EP are for ongoing operations and preservation activities.

This broader consultation program provides a backdrop to the consultation undertaken for the Coastal Waters OEMP and Barossa Production Operations EP as well as other approvals / regulatory submissions related to operation of the GEP outside of Commonwealth waters (as to which, see Table 1-2 above).

For the Coastal Waters OEMP (and the Barossa Production Operations EP, as explained below in Section 4.2), consultation activities were undertaken in three broad phases:

- Preliminary consultation including to share consultation information and to allow authorities, persons and
  organisations opportunities to self-identify as Relevant Persons and directly contacting potential Relevant
  Persons.
- **Formal consultation** including seeking feedback from Relevant Persons to inform development of the Coastal Waters OEMP (and the Barossa Production Operations EP)



• **Further consultation** with some authorities, persons and organisations following the formal consultation phase given existing relationships, consultation preferences and standing meeting and consultation arrangements.

Consultation for these activities has been undertaken in compliance with the Offshore Petroleum and Greenhouse Gas Storage (Environment) Regulations 2023 (OPGGS(E)R) consultation requirements, applicable case law and applicable guidance (e.g. NOPSEMA guidance issued in May 2023 and subsequent guidance in May 2024 [GL2086 – Consultation in the course of preparing an environment plan]).

A summary report of the consultation carried out under section 25 OPGGS(E)R is included in Section 4.8.

Section 8.4.9 includes Santos' post EP acceptance consultation implementation strategy for activities covered by this OEMP. in accordance with Regulation 22(15) of the OPGGS(E)R.

### 4.2 Relationship with consultation on Barossa Production Operations EP

This Coastal Waters OEMP was prepared concurrently with the Barossa Production Operations EP, submitted to NOPSEMA in August 2024. As set out in section 1.2.2, the Barossa Production Operations EP covers (among other things) operation of the Barossa GEP in Commonwealth waters, while this Coastal Waters OEMP covers the operation of the Barossa GEP in NT Coastal Waters. Santos undertook consultation under s 25 of the OPGGS(E)R regarding activities under this Coastal Waters OEMP concurrently with consultation on activities under the Barossa Production Operations EP. This was done because both plans will authorise the operation of the Barossa GEP, and accordingly:

- there is a high degree of similarity in the activities proposed to be undertaken under the Barossa Production Operations EP (insofar as it relates to the Barossa GEP) and this Coastal Waters OEMP, and therefore in the potential impacts and risks associated with the activities;
- activities under this Coastal Waters OEMP will occur in close proximity and timeframe to activities under the Barossa Production Operations EP;
- the EMBA identified for the Barossa Production Operations EP overlaps the EMBA identified for this Coastal Waters OEMP; and
- for the reasons above:
  - Santos' identification of Relevant Persons for the Barossa Production Operations EP was anticipated to overlap significantly with the identification of Relevant Persons for this Coastal Waters OEMP; and
  - the information required to be given to each Relevant Person in order to allow them to make an informed assessment of the possible consequences of the activities under the Barossa Production Operations EP overlapped significantly with that required in respect of the activities under this Coastal Waters OEMP.

This coordinated approach to consultation enabled Santos to provide the most comprehensive information and widest consultation opportunities to potential Relevant Persons across the larger Barossa Production Operations EMBA, whilst also presenting information in context and seeking to mitigate the impact of Relevant Persons consultation fatigue.

As such, the information contained in the *Production Operations Information Booklet* (Booklet), shared during consultation for the Coastal Waters OEMP, addressed the same activity scope as relevant to the operations phase of the Barossa Gas Project and referenced the larger Barossa Production Operations EMBA.

In the Booklet, Gas Export Pipeline Production Operations Area 2 (OA2) covers the GEP Production Operations activities in both Commonwealth and NT Coastal Waters.

Information provided during consultation on the Barossa Production Operations EP has been considered to determine its relevance to the Coastal Waters OEMP. Where relevant, that information has been addressed in this Coastal Waters OEMP.

Some of the persons identified and consulted by Santos in its concurrent consultation on the Barossa Production Operations EP and Coastal Waters OEMP are anticipated to have functions, interests or activities that may be affected under the EP, but that may not be affected under the OEMP. For example, there is no risk under this Coastal Waters OEMP of a hydrocarbon spill which affects international waters or waters within Western Australia, reducing the likelihood that interests concerning these areas may be affected by activities under this Coastal Waters OEMP.



Notwithstanding this, and noting the similarities between the activities outlined above, Santos adopted a broad approach and has detailed in this section consultation with all persons identified as potentially relevant having regard to the broader Barossa Production Operations EP EMBA (of which the EMBA for the Coastal Waters OEMP is a subset).

Unless otherwise stated, the EMBA referenced in Section 4 of this Coastal Waters OEMP is the larger Barossa Production Operations EP EMBA, of which the EMBA for this Coastal Waters OEMP is a subset.

### 4.3 **Consultation context**

Santos has a long history of regional exploration, appraisal and operations offshore and onshore NT to support safe and reliable operation, including ownership and then operation of the Bayu-Undan Gas Project, which commenced operation in 2006. Santos has also undertaken other exploration and appraisal activities in the region.

Gas from the Bayu-Undan field in Timor-Leste offshore waters has been supplied via a 502 km pipeline to DLNG facility, which was the first LNG production facility in the NT and the second in Australia.

Over the Bayu-Undan Project's almost 20-year operating history, operational impacts that are analogous to the operation of facilities at the Barossa field and the Barossa GEP, as well as the ongoing operation of DLNG facility, have been managed by initial Operator ConocoPhillips and subsequently by Santos.

During Santos' time as Bayu-Undan and Darwin LNG Operator, Santos has consulted a range of regional stakeholders to support environmental approvals for its operations. These consultation activities have provided the foundation on which to build a comprehensive consultation program for the Barossa Gas Project, having regard to the nature and scale of proposed activities and the potential for Relevant Persons functions, interests and activities to be affected by proposed activities managed under respective EPs as well as this OEMP.

Importantly, during this time, Santos has strengthened and developed relationships with a range of regional stakeholders not only through previous consultation, but also through engagements associated with local employment, training, education and enterprise opportunities central to delivering meaningful and long-lasting contributions in NT and Timor-Leste communities.

These engagements have helped Santos anticipate likely issues of interest or concern among Relevant Persons to inform the consultation process, including, for example development of materials to support consultation for this Coastal Waters OEMP (and the Barossa Production Operations EP, as explained above in Section 4.2),. Similarly, these engagements have provided a strong foundation for Relevant Persons to understand the activities proposed in this OEMP and the environmental impacts and risks that may be associated with those activities, so as to support meaningful consultation for this OEMP.

### 4.4 **OPGGS(E)**R consultation requirements

Table 4-1 outlines the applicable OPGGS(E)R requirements for consultation with Relevant Persons for the Coastal Waters OEMP.

#### Table 4-1: Consultation requirements under the OPGGS(E)R

#### OPGGS(E)R 2023 Requirements

#### Section 24. Other information in the environment plan

The environment plan must contain the following:

- b. a report on all consultations under section 25 of any relevant person by the titleholder, that contains:
  - i. a summary of each response made by a relevant person; and
  - ii. an assessment of the merits of any objection or claim about the adverse impact of each activity to which the environment plan relates; and
  - iii. a statement of the titleholder's response, or proposed response, if any, to each objection or claim; and
  - iv. a copy of the full text of any response by a relevant person.

Section 25. Consultation with relevant authorities, persons and organisations, etc

(1) In the course of preparing an environment plan (including a revised environment plan referred to in Division 5) a titleholder must consult each of the following (a *relevant person*):

- a. each Commonwealth, State or Northern Territory agency or authority to which the activities to be carried out under the environment plan may be relevant;
- b. if the plan relates to activities in the offshore area of a State-the Department of the responsible State Minister;
- c. if the plan relates to activities in the Principal Northern Territory offshore area—the Department of the responsible Northern Territory Minister;

- d. a person or organisation whose functions, interests or activities may be affected by the activities to be carried out under the environment plan;
- e. any other person or organisation that the titleholder considers relevant.

(2) For the purpose of the consultation, the titleholder must give each relevant person sufficient information to allow the relevant person to make an informed assessment of the possible consequences of the activity on the functions, interests or activities of the relevant person.

(3) The titleholder must allow a relevant person a reasonable period for the consultation.

(4) The titleholder must tell each relevant person the titleholder consults that:

- a. the relevant person may request that particular information the relevant person provides in the consultation not be published; and
- b. information subject to such a request is not to be published under this Part.

Section 26. Submission of environment plan

#### Form of environment plan

(8) All sensitive information (if any) in an environment plan, and the full text of any response by a relevant person to consultation under section 25 in the course of preparation of the plan, must be contained in the sensitive information part of the plan and not anywhere else in the plan.

Note: Subparagraph 24(b)(iv) requires the plan to contain a copy of the full text of any response by a relevant person to consultation under section 25 in the course of preparation of the plan.

#### Section 28. Publishing environment plan and associated information

(1) If NOPSEMA's provisional decision under section 27 is that the environment plan includes material apparently addressing all the provisions of Division 2 (Contents of an environment plan), NOPSEMA must publish on NOPSEMA's website as soon as practicable:

- a. the plan with the sensitive information part removed; and
- b. the name of the titleholder who submitted the plan; and
- c. a description of the activity or stage of the activity to which the plan relates; and
- d. the location of the activity; and
- e. a link or other reference to the place where the accepted offshore project proposal (if any) is published; and
- f. details of the titleholder's nominated liaison person for the activity.

### 4.5 **Government and industry guidance**

Notwithstanding that the Activity covered by the Coastal Waters OEMP is within NT jurisdiction and will be authorised by the NT Minister for Mining and Energy); but assessed under the OPGGS(E)R 2023, Santos has considered the following NOPSEMA guidance in developing its consultation activities and approach:

- GL2086 Consultation in the course of preparing an environment plan (EP Consultation Guideline) (NOPSEMA, 2023; 2024a)
- GL1887 Consultation with Commonwealth agencies with responsibilities in the marine area (NOPSEMA, 2024b)
- GL1721 Environment plan decision making (NOPSEMA, 2024c)
- GN1344 Environment plan content requirement (NOPSEMA, 2024d)
- GN1488 Oil Pollution Risk Management (NOPSEMA, 2024e)
- Petroleum activities and Australian Marine Parks: A guidance note to support environmental protection and effective consultation (Australian Government, 2024) jointly released by NOPSEMA and Parks Australia.

Santos has also considered other government and industry guidance, including:

- International Standards Organisation
  - ISO14001:2015 Environmental Management Systems
- Australian Fisheries Management Authority
  - Petroleum industry consultation with the commercial fishing industry
- Australian Heritage Commission
  - Ask First A guide to respecting Indigenous heritage places and values
- Department of Agriculture, Fisheries and Forestry
  - Fisheries and the Environment OPGGS Act



- Offshore Installations–Biosecurity Guide (DAFF, 2023a)
- Department of Climate Change, Energy, the Environment and Water
  - Interim Engaging with First Nations People and Communities on Assessments and Approvals under the *Environment Protection and Biodiversity Conservation Act 1999* (DCCEEW, 2023c)
  - Assessing and Managing Impacts to Underwater Cultural Heritage in Australian Waters: *Guidelines on the application of the Underwater Cultural Heritage Act 2018* (DCCEEW, June 2024)
- Commonwealth Ministerial Council on Mineral and Petroleum Resources
  - Principles for Engagement with Communities and Stakeholders
- International Association for Public Participation
  - Quality Assurance Standard for Community and Stakeholder Engagement
- WA Department of Primary Industries and Regional Development
  - Guidance statement for oil and gas industry consultation with the Department of Fisheries
- WA Department of Transport
  - Offshore Petroleum Industry Guidance Note Marine Oil Pollution: Response and Consultation Arrangements
  - WA Incident Management Plan Marine Oil Pollution
- Western Australian Fishing Industry Council
  - Commercial Fishing Consultation Framework for the Offshore Oil and Gas Sector https://www.wafic.org.au/wp-content/uploads/2023/07/Oil-and-Gas-Consultation-Framework.pdf
  - Consultation approach for unplanned events <u>https://www.wafic.org.au/what-we-do/access-</u> sustainability/oil-gas/consultation-approach-for-unplanned-events/

### 4.6 Applicable case law and guidance

In addition to considering the regulatory requirements and guidance set out above, in conducting Relevant Person consultation for the activities covered by this OEMP, Santos has considered the judgments of:

- Justice Bromberg in Tipakalippa v National Offshore Petroleum Safety and Environmental Management Authority (No. 2) [2022] FCA 1121;
- the Full Federal Court in Santos NA Barossa Pty Ltd v Tipakalippa [2022] FCAFC 193 (Appeal Judgment);
- Justice Colvin in Cooper v National Offshore Petroleum Safety and Environmental Management Authority (No 2) [2023] FCA 1158; and
- Justice Charlesworth in Munkara v Santos NA Barossa Pty Ltd (No 3) [2024] FCA 9

The EP Consultation Guideline (NOPSEMA, 2023; 2024a) provides a summary of the Full Federal Court's interpretation of "functions", "activities" and "interests" referenced in section 25(1)(d) of the OPGGS(E)R, adopted by NOPSEMA to assist in informing who may be a Relevant Person and how Relevant Persons may be identified, as defined in Table 4-2.

#### Table 4-2: Relevant Person terms and definitions

Term	Interpretation
Functions	Refers to "a power or duty to do something"
Activities	To be read broadly and is broader than the definition of "activity" in section 5 of the OPGGS(E)R and is likely directed to what the Relevant Person is already doing
Interests	To be construed as conforming with the accepted concept of "interest" in other areas of public administrative law. Includes "any interest possessed by an individual whether or not the interest amounts to a legal right or is a proprietary or financial interest or relates to reputation"

Santos has also had regard to the purpose of consultation as outlined in the Appeal Judgment and EP Consultation Guideline (NOPSEMA, 2024a), the emphasis that superficial or tokenistic consultation is not sufficient and that:

• consultation must be appropriate and adapted to the nature of each Relevant Person;



- for each Relevant Person, the appropriate manner and method of consultation (including the nature of information, time periods for consultation and mode of communication) may differ; and
- there is good reason to adopt pragmatic and practical approaches to consultation conducted in accordance with section 25 of the OPGGS(E)R.

### 4.7 Santos' consultation methodology

### 4.7.1 Overview

Santos consults to ensure that any activity it is proposing under an EP prepared in accordance with the OPGGS(E)R (including this Coastal Waters OEMP) is carried out in a manner:

- consistent with the principles of ecologically sustainable development set out in section 3A of the EPBC Act; and
- by which the environmental impacts and risks of the Activity will be reduced to low as reasonably practicable (ALARP) and to an acceptable level.

The consultation process is designed to assist Santos to further ascertain, understand and assess values and sensitivities of the environment (including ecosystems, including people and communities, natural and physical resources, the qualities and characteristics of locations, places and areas and the heritage value of places) that may be affected by a proposed activity, and the potential environmental impacts and risks, through information obtained during consultations.

Santos may then refine or change its proposed control measures to address potential environmental impacts and risks of the activity based on that information or any claims or objections raised through consultation.

As outlined above at Section 4.2, Santos consulted concurrently on this Coastal Waters OEMP and the Barossa Production Operations EP and identified Relevant Persons for consultation across the broader Production Operations EP EMBA (of which the Coastal Waters OEMP EMBA is a subset).

Santos' consultation methodology and process adopted in developing this OEMP and the Barossa Production Operations EP comprised the following key steps:

- Identifying Relevant Persons, as outlined in Section 4.7.2;
- Implementing a public awareness campaign and providing opportunities for Relevant Persons to identify themselves if they wished to be consulted, as outlined in Section 4.7.4;
- Consultation planning, preliminary consultation and consultation activities, as outlined in Section 4.7.5; and
- Assessing the merits of claims or objections made by Relevant Persons about alleged adverse impacts of each activity to which the EP relates and providing responses to queries, requests and feedback, as summarised in Section 4.8.

As described in Section 3, Santos considered the spatial extent of the environment that may be affected (EMBA) and the particular aspects of the relevant environment as part of its process for identifying Relevant Persons. As outlined in Section 4.2, this process was undertaken by reference to the EMBA for the Barossa Production Operations EP, of which the EMBA set out in Section 3 of this Coastal Waters OEMP is a subset. As the Barossa Production Operations EP EMBA represents the greatest geographical extent that could be affected by hydrocarbons in the event of a spill scenario (see Section 3.1.1), that EMBA is Santos' starting point in identifying potential Relevant Persons.

Santos notes that there is no reasonable possibility that planned impacts from the Activity (undertaken within the Operational Area (OA) will have any consequences on functions, interests or activities concerning areas at the extremities of the EMBA. The only potential consequence for functions, interests or activities concerning these areas is as a result of the risk of an unplanned release of hydrocarbons, described in Section 7.6 of the Coastal Waters OEMP.

The modelling performed to generate the EMBA and MEVA for the Coastal Waters OEMP is based on the worstcase unplanned release scenarios to understand the potential area of influence that could be expected from the worst-case MDO spill event. The likelihood of an unplanned release is unlikely for MDO and, given the mitigation and management controls in place, the residual risk is Low.

There is an even lower likelihood of an unplanned hydrocarbon release affecting a person or organisation's functions, interests or activities where these relate to the extremities of the EMBA. This is because there is significant conservatism associated with the EMBA extents given these:

• are determined from the combination of 300 individual modelled spill scenarios across all seasons



- are based on low exposure values (as described in Section 3.1.1 which represent the maximum potential extent of hydrocarbon contact with environmental receptors, and primarily used to inform Santos preparedness for potential spill response; and
- do not take into account any spill response activities by Santos (as described in the GEP NT Waters Oil Pollution Emergency Plan (OPEP)) which would be implemented in event of a spill and reduce the EMBA extent.

The modelling itself represents the potential extent of detection of a spill in the environment rather than the geographical extent of environmental impact on receptors in the environment. Further, there is no single event that could ever result in the whole EMBA being affected at the same time.

When considering the remote possibility of any major unplanned spill event, and the inherent conservatism of the EMBA, the likelihood of there being persons or organisations along the Northern Territory coastline having an interest that may be affected by the proposed activities becomes increasingly unlikely with increasing distance from the OA where planned activities will occur.

Finally, as outlined in Section 4.2, some of the persons identified and consulted by Santos in its concurrent consultation on the Coastal Waters OEMP and Barossa Production Operations EP have functions, interests or activities that may be affected under the EP, but not under this Coastal Waters OEMP.

In considering this, while Santos has still identified and consulted with Relevant Persons whose functions, interests or activities may only be affected by unplanned events (the likelihood of which is remote), Relevant Persons identification steps and direct consultation effort has tended to focus more closely on those most proximate to the OA. By way of example, Santos held multiple consultation sessions with First Nations Relevant Persons most likely to be affected by activity impacts and risks (e.g. Tiwi and Larrakia people), while Santos held single consultation sessions with those First Nations Relevant Persons most likely to be affected by unplanned events only.

Santos' methodology demonstrates a very broad capture of potential Relevant Persons, including providing ample opportunities, as outlined in Section 4.7.2, for Relevant Persons to self-identify and provide input to the development of the Coastal Waters OEMP if they consider they may be impacted by the activities.

### 4.7.2 Identifying Relevant Persons

As outlined in Section 4.2 and for the reasons set out in that section, Santos identified and consulted Relevant Persons for this Coastal Waters OEMP concurrently with consultation on the Barossa Production Operations EP.

Table 4-3 summarises the preliminary steps adopted by Santos to identify Relevant Persons, noting that the identification of Relevant Persons is an iterative process. The EMBA referenced in Table 4-3 is the larger Barossa Production Operations EP EMBA, unless otherwise stated.

### Table 4-3: Preliminary identification methodology

Pro	ocess steps	OEMP reference	
1.	Identify the impacts of the planned activities and the risks and impacts of unplanned events.	The activity description is described in Section 2.	
		The impacts from planned activities are described in Section 6.	
		The impacts from unplanned events are described in Section 7.	
2.	Consider the spatial extent of the EMBA by the Activity impacts and risks.	The spatial extent of the activity EMBA is described in Section 3.1.1.	
3. be :	Consider and identify aspects of the environment within the environment that may affected, having regard to:	The existing environment is described in Section 3.	
	(a) ecosystems and their constituent parts, including people and communities	Particular aspects of the environment	
	<ul> <li>(b) natural and physical resources</li> <li>(c) the qualities and characteristics of locations, places and areas</li> </ul>	identification are outlined in Table	
	(d) the heritage value of places	4-4.	
	(e) the social, economic and cultural features of the matters mentioned in paragraphs (a), (b), (c) and (d).		
4.	Identify Relevant Person categories, having regard to:	Relevant person categories	
	(a) aspects of the environment identified at Item 3	considered in Relevant Person	
	(b) the departments or agencies of Commonwealth, State and Territory governments that could therefore be relevant		
Pro	cess steps	OEMP reference	
-----	-------------------------------------------------------------------------------------------------------------------------------------------------	-----------------------------------------------------------------	
	(c) the kinds of functions, interests or activities of people or organisations that could therefore be affected		
	(d) submissions received in response to Santos' advertisements asking Relevant Persons to identify themselves if they wished to be consulted		
	(e) any other person or organisation that the titleholder considers relevant.		
Upd	ate during consultation based on new information, if appropriate.		
5.	Identify Relevant Persons within Relevant Person categories, having regard to items 1–4 above.	Actions to identify Relevant Persons are outlined in Table 4-6.	

Table 4-4 outlines the environmental aspects within the EP EMBA (described in detail in Section 3). Santos considered these aspects for the purpose of identifying Relevant Person categories.

### Table 4-4: Environmental aspects considered for Relevant Person category identification

Aspects of the environment	OEMP reference
Physical environment	Section 3.2.2
Provincial bioregions	Section 3.2.1.1
Benthic habitats	Section 3.2.9
Shoreline habitats	Section 3.2.10
Marine parks	Section 3.2.12.1
Wetlands of international and national importance	Section 3.2.12.2
Key ecological features	Section 3.2.12.3
Commonwealth heritage places	Section 3.2.12.4
Threatened and migratory fauna	Section 3.2.13
Biologically important areas and critical habitat	Section 3.2.13.5
Conservation advice, recovery plans and management plans	Section 3.2.13.6
Commercial fisheries	Section 3.2.14.1
Energy industry	Section 3.2.14.2
Defence activities	Section 3.2.14.3
Telecommunications cables	Section 3.2.14.4
Shipping	Section 3.2.14.5
Recreation and tourism	Section 3.2.14.6
Underwater cultural heritage	Section 3.2.14.73.2.14.7
Cultural features	Section 3.2.15

Table 4-5 lists the Relevant Person categories following consideration of the environmental aspects.

# Table 4-5: Relevant Person categories Environmental aspects considered for Relevant Person category identification

Section	25(1)(a)(b)(c) of the OPGGS(E)R
•	Commonwealth Government agency or authority;
•	NT Government agency or authority; and
•	WA Government agency or authority.
0.0	
Section	25(1)(d)(e) of the OPGGS(E)R:
• Section	25(1)(d)(e) of the OPGGS(E)R: academic and research organisations;
•	25(1)(d)(e) of the OPGGS(E)R: academic and research organisations; commercial fishing (Commonwealth-managed);
Section •	25(1)(d)(e) of the OPGGS(E)R: academic and research organisations; commercial fishing (Commonwealth-managed); commercial fishing (NT-managed);
Section • •	25(1)(d)(e) of the OPGGS(E)R: academic and research organisations; commercial fishing (Commonwealth-managed); commercial fishing (NT-managed); commercial fishing (WA-managed);



- environmental conservation organisations;
- First Nations people and groups including recognised community reference/liaison groups;
- infrastructure operators;
- shipping;
- industry associations;
- local government;
- recreational fishing; and
- tourism operators.

Table 4-6 outlines actions used by Santos to identify Relevant Persons within those categories.

<b>Fable 4-6: Actions for identifyin</b>	g Relevant Persons by category
------------------------------------------	--------------------------------

Relevant Person Category	Actions to identify Relevant Persons	
All Relevant Person categories	<ul> <li>Review of relevant regional historical consultation by Santos in the region, including all previous Barossa EPs.</li> </ul>	
	• Review of identified Relevant Persons in publicly available EPs submitted by other Titleholders that may be relevant to proposed activities to be managed under the Coastal Waters OEMP and Barossa Production Operations EP.	
	<ul> <li>Review of persons consulted for Darwin Pipeline Duplication environmental management plans under Northern Territory jurisdiction.</li> </ul>	
	• Reviewing media coverage and associated organisation websites to identify persons and organisations with reasonably ascertainable functions, interests and activities that may be affected by the activities under this OEMP.	
	Public awareness campaign as outlined in Section 4.7.4.	
	<ul> <li>Review advice from authorities, consultants and other Relevant Persons as to potential Relevant Persons.</li> </ul>	
	<ul> <li>Review of information provided by or claims made by persons claiming to be Relevant Persons or made on behalf of organisations who claimed to be Relevant Persons.</li> </ul>	
	Review of published NOPSEMA guidance relevant to consultation.	
Section 25(1)(a) of the OPGGS(E)R		
Commonwealth agency or authority to which the activities to be carried out	<ul> <li>Review of government agency websites and directories to understand agency roles, functions and responsibilities.</li> </ul>	
under the environment plan may be relevant	Review government agency guidance on consultation expectations.	
Section 25(1)(b) and (c) of the OPGGS(E)R		
State and Territory Departments/Agencies	<ul> <li>Review of government agency websites and directories to understand agency roles, functions and responsibilities.</li> </ul>	
	Review government agency guidance on consultation expectations.	
Section 25(1)(d) and (e) of the OPGG	S(E)R	
Academic and research organisations	• Conducting key-word searches of publicly available online search engines, review media coverage and review organisation websites to identify organisations with reasonably ascertainable functions, interests or activities that may be affected, having regard to the region, activities or risks/impacts under the Coastal Waters OEMP and Barossa Production Operations EP.	
Commercial fishing	Review of Commonwealth, NT and WA Government commercial fishing catch and effort data.	
	Review of fisheries entitled to fish in the EMBA.	
Energy industry	Review of EMBA overlap with petroleum, greenhouse gas and any other National Offshore Petroleum Titles Administrator (NOPTA) issued titles.	
Environmental conservation organisations	• Conducting key-word searches of publicly available online search engines, review media coverage and review organisation websites to identify organisations with reasonably ascertainable functions, interests or activities that may be affected, having regard to the region, activities or risks/impacts under this OEMP.	

Relevant Person Category	Actions to identify Relevant Persons
First Nations people and groups	Review of the Judgment and the Appeal Judgment.
including recognised community reference/liaison groups	<ul> <li>Review of EMBA overlap with Native Title determined areas and claims, Indigenous land use agreements (ILUAs), land rights and Indigenous Protected Areas (IPAs).</li> </ul>
	<ul> <li>Review of Representative Aboriginal/Torres Strait Island Bodies (RATSIBs) and Native Title Representative Bodies (NTRBs) on Native Title website.</li> </ul>
	<ul> <li>Review of Prescribed Bodies Corporate on Native Title website, where relevant.</li> </ul>
	<ul> <li>Conducting searches of public cultural heritage databases relevant to the EMBA.</li> </ul>
	Review of marine park management plans relevant to the EMBA.
	<ul> <li>Engagement with government departments/agencies with relevant knowledge or relevant responsibilities.</li> </ul>
	<ul> <li>Engagement with representative bodies under the Native Title Act 1993 (NT Act) and the Aboriginal Land Rights (Northern Territory) Act 1976 (ALR Act).</li> </ul>
	<ul> <li>Engagement with other representative organisations in areas of potential relevance to Barossa Project activities such as liaison committees and First Nations Consultative Committees (FNCCs).</li> </ul>
	<ul> <li>Engagement with other First Nations organisations that may support specific interests of First Nations people, such as economic development.</li> </ul>
	<ul> <li>Engagement with third party consultants to assist with identification of potential First Nations Relevant Persons.</li> </ul>
Infrastructure operators	<ul> <li>Review of EMBA overlap with offshore and onshore infrastructure, such as submarine telecommunications cables or ports.</li> </ul>
Industry associations	Review of industry representation of the following Relevant Person groups:
	commercial fishing
	local industry
	• shipping
	recreational fishing
	tourism operators.
Local government	Review of EMBA overlap with boundaries of local government areas.
Recreational fishing	Review of EMBA overlap with areas of interest to recreational fishing.
	<ul> <li>Review of potential presence of recreational fishing club members in the EMBA.</li> </ul>
	• Review of website information of relevant agencies/organisations that represent recreational fishing interests.
Tourism operators	<ul> <li>Review of EMBA overlap with areas of interest to charter and tourism operators.</li> </ul>
	Review of potential presence in the EMBA.
	• Conducting key-word searches using online search engines and review of website information of relevant operators/organisations that represent commercial tourism interests with reasonably ascertainable functions, interests and activities that may be affected by the activities under the Coastal Waters OEMP and Barossa Production Operations EP.

### 4.7.2.1 Identification of First Nations people and groups

Santos has developed a comprehensive process for identifying First Nations Relevant Persons which was employed in identifying Relevant Persons for consultation on this Coastal Waters OEMP and the Barossa Production Operations EP (as outlined above at Section 4.2)

As with Santos' process for identifying Relevant Persons generally, this is an iterative process with multiple avenues of enquiry including, but not limited to, the following actions:

- Consideration of known cultural features of the environment;
- Active steps to identify First Nations people and groups as per actions outlined in Table 4-6, and further described below, directed to identifying First Nations Relevant Persons with functions, interests or activities that may be affected;



- Asking identified Relevant Persons if there are other persons or organisations who may be a Relevant Person; and
- Advertising broadly to ensure that Relevant Persons that are not otherwise identified by Santos' examination of the EP EMBA are given the opportunity to self-identify.

Santos' process involved engaging a third-party consultant to assist Santos in identifying First Nations groups, clans and/or organisations along the NT/WA coastline in the vicinity of the EP EMBA.

In order to positively identify First Nations Relevant Persons Santos considered the following questions, based on information gathered when taking the steps described in Table 4-6:

- Do any First Nations groups, clans and/or organisations along the NT/WA coastline in the vicinity of the EP EMBA have any native title claims pending<sup>28</sup> or determined, or any ILUA, that extend offshore and cross into the EP EMBA?
- Do any First Nations groups, clans and/or organisations along the NT/WA coastline in the vicinity of the EP EMBA have any reasonably ascertainable responsibilities for sacred sites that extend offshore and cross into the EP EMBA (recognised and protected under the ALR Act, the Northern Territory Aboriginal Sacred Sites Act 1989 (NTASS Act), the Aboriginal and Torres Strait Islander Heritage Protection Act 1984 (ATSIHP Act), the Underwater Cultural Heritage Act (UCH Act), or the EPBC Act)?
- Do any First Nations groups, clans and/or organisations along the NT/WA coastline in the vicinity of the EP EMBA have any land rights (apart from native title claims) pending or determined that extend offshore and cross into the EP EMBA?
- Are there any IPAs that extend offshore and cross into the EP EMBA?

If the answer to any of the above questions was affirmative, this resulted in identification of the particular First Nations group, clan or organisation as a Relevant Person.

Santos recognises that not all relevant functions, interests or activities of First Nations persons or groups will be identified through the four steps above, and that even if the answer to all four of the above questions is negative, First Nations groups in the vicinity of the EP EMBA could still potentially have communal cultural interests (such as connection to sea country) that extend into the EP EMBA.

As is the case for determining whether any person's or organisation's functions, interests or activities may be affected, the context for how the spatial extent of the EP EMBA is determined is also relevant when evaluating whether any First Nations' sea country or other interests could potentially be affected by the activity. The EP EMBA is informed by modelling the maximum potential extent of all major unplanned spill events under all seasonal conditions as further explained in Section 3.1.1.

Having regard to the residual potential for other cultural interests within the EP EMBA, Santos supplemented its consideration of the four questions above by:

- the completion of the other First Nations Relevant Persons identification steps (see Table 4-6);
- making sustained efforts to engage and build relationships with identified NTRBs and PBCs/RNTBCs through a range of appropriate communication methods;
- inviting information from identified First Nations Relevant Persons as to other potential First Nations Relevant Persons; and
- conducting a public awareness and advertising campaign targeted at increasing awareness of the Barossa Gas Project and the activities proposed in the Coastal Waters OEMP.; and encouraging any persons or organisations who consider they have functions, interests or activities that may be affected by the activities in this OEMP. to contact Santos (see Section 4.7.4).

These steps were carried out to further inform Santos' identification of First Nations people or groups with reasonably ascertainable functions, interests or activities that may be affected by the activities to be carried out under the Coastal Waters OEMP. Santos proceeded to consult with a number of First Nations groups identified through the above processes as potential Relevant Persons, with a view to ascertaining during consultation sessions what, if any, functions, interests or activities they had that may be affected by the activities proposed in this OEMP.

With regard to the location of the activities under the Coastal Waters OEMP, there are no impacts from planned activities that may affect the functions, interests or activities of any First Nations groups, clans and/or organisations along the WA coastline.

<sup>&</sup>lt;sup>28</sup> meaning registered claims that are yet to be determined.

The worst-case credible spill scenario modelled for the EMBA for the Coastal Waters OEMP also shows there are no potential impacts from unplanned activities that may affect the functions, interests or activities of any First Nations groups, clans and/or organisations along the WA coastline.

Notwithstanding this, as outlined in Section 4.2 above, Santos adopted a broad approach and considered all the identified NT and WA First Nations people or groups as potential Relevant Persons for the purposes of consultation on the Coastal Waters OEMP (and therefore they have been included in Table 4-9 and Table 4-20) notwithstanding that in some cases, no potentially affected functions, interests or activities were ultimately ascertained.

Santos was not directed to any other First Nations groups or organisations (other than those Santos had identified) in response to Santos' invitation in its consultation materials for Relevant Persons to notify Santos of other potentially Relevant Persons for Santos to consider consulting about the Coastal Waters OEMP.

Santos utilised the public awareness campaign outlined Section 4.7.4 to assist in identification of other First Nations groups with interests (such as connection with sea country) that may be affected by the Activity, that weren't identified through other identification steps described above and in Table 4-6.

While Santos recognises that the obligation to identify Relevant Persons lies with the titleholder, and titleholders cannot rely solely on a process of public notification and self-identification, Santos considers its public awareness campaign to be an appropriate and sufficient measure to provide opportunities for First Nations (and other) Relevant Persons to self-identify, particularly having regard to the remoteness of the activity, the remote possibility of a major unplanned spill event, the inherent conservatism in spill modelling used to inform the EP EMBA and the difficulty in ascertaining whose functions, interests or activities may be affected in remote offshore waters.

### 4.7.3 International persons

With regard to the location of the proposed activities under this Coastal Waters OEMP, there are no impacts from planned activities that may affect the functions, interests or activities of international Relevant Persons.

As outlined above at Section 4.2, Santos consulted concurrently on activities under this Coastal Waters OEMP and activities under the Barossa Production Operations EP (in Commonwealth Waters). With regard to the location of the proposed activities in Commonwealth Waters, while there are no impacts from planned activities that may affect the functions, interests or activities of international Relevant Persons, the worst-case credible spill scenario modelled for the Barossa Production Operations EP indicates a possibility that the Indonesian and Timor-Leste coastline could be in contact with residual entrained hydrocarbons. Notwithstanding this, for the reasons set out in Section 4.2 above, consultation with international Relevant Persons identified as potentially affected by activities under the Barossa Production Operations EP has been included in the Coastal Waters OEMP.

As stated in the Barossa Productions Operations EP, there is significant conservatism associated with the depiction of the EP EMBA based on low exposure values that Santos has applied, and especially given the modelling process (Section 3.1.1) which combines a large number of individual unmitigated spill simulations and the low likelihood of occurrence given the planned engineering prevention measures.

In addition, the modelling at low exposure values is primarily used to inform Santos' preparedness for potential spill response and does not take into account the suite of mitigations described in the applicable OPEP that would be implemented and reduce the EP EMBA extent in the unlikely event of a spill.

Santos also acknowledges the judicial guidance outlined at Section 4.6 above, including that Relevant Persons must be "reasonably capable of ascertainment". Santos further acknowledges the judicial guidance that there is good reason to adopt practical and pragmatic approaches to consultation and that the requirements of Section 25 must be capable of being complied with within a reasonable time.<sup>29</sup>

Santos therefore sought to reasonably ascertain international Relevant Persons in a manner proportionate to the remote likelihood of any effect on the functions, interests or activities of international persons or organisations from a worst-case unmitigated spill. This involved the following steps:

- Conducting key-word searches of publicly available online search engines and review organisation websites to identify environmental conservation organisations with reasonably ascertainable functions, interests or activities that may be affected, having regard to the region, activities or risks/impacts under this OEMP.
- Writing to international persons or organisations that had self-nominated for consultation in respect of previous Barossa Gas Project EPs to:

<sup>&</sup>lt;sup>29</sup> Santos NA Barossa Pty Ltd v Tipakalippa [2022] FCAFC 193 at [88] (per Kenny and Mortimer JJ) and at [136] (per Lee J).



- afford them a direct opportunity to self-nominate for consultation for the Coastal Waters OEMP and Barossa Production Operations EP and to indicate what functions, interests or activities that have been or may be affected by the activities proposed in the OEMP and EP.
- invite them to nominate any other persons or organisations they considered Santos should consider consulting.
- Advertising on social media platforms Facebook, Instagram and Messenger, geotargeting Indonesia and Timor-Leste (during March 2024), inviting Relevant Persons to contact Santos to self-nominate. This was in addition to the general widespread media and advertising campaign (see Table 4-7 and Table 4-8). Both the international and the domestic advertisements contained links to Santos' website with Production Operations consultation information that included information about activities the subject of the Coastal Waters OEMP and Barossa Production Operations EP and a form for self-nominated following the international advertising campaign.

Santos also consulted the Department of Foreign Affairs and Trade (DFAT) which has a function in coordinating and facilitating communication between Australia and the Indonesian or Timor-Leste governments.

During consultation with DFAT, the Department's Timor-Leste desk recommended consultation with the Government of Timor-Leste on Santos' Environment Plan given the proximity of Santos' operations to the territory of Timor-Leste and that the appropriate authority for such consultation is the Autoridade Nacional Do Petróleo (ANP - National Petroleum Authority). Santos proceeded to consult with the ANP accordingly.

DFAT's Indonesia Branch had no comments on the OEMP and EP and offered advice that should Santos wish to consult with the Indonesian Government, the Indonesian Embassy in Canberra should be contacted in the first instance. As Santos has assessed there to be no Activity impacts or risks to internationally held functions, interests or activities, the only matter in respect of which consultation with Indonesian and Timor-Leste persons or organisations might be required is in relation to a hydrocarbon spill that reaches Indonesian or Timor-Leste waters.

With the exception of the ANP, no other Indonesian or Timor-Leste government person or organisation has been identified as having reasonably ascertainable functions, interests or activities that may be affected by the Activity. Santos understands that in the unlikely event of a hydrocarbon spill, communication about such matters is to occur at a Government to Government level, in addition to Santos' notifications to the ANP.

Under the National Plan for maritime environmental emergencies (AMSA, 2020) Australia has entered into mutual aid arrangements and associated cooperation agreements with other countries impacted by maritime environmental emergencies. In this regard, Australian has entered into a bilateral agreement with Indonesia, and a Maritime Boundaries Treaty with Timor-Leste. Any relevant affected government authorities of Indonesia or Timor-Leste would be identified and notified through the domestic arrangements of that Nation State, at a State-to-State level.

Having regard to the above, Santos considered that no further steps were reasonably required to identify international Relevant Persons.

### 4.7.4 Public awareness campaign and self-identification opportunities

In addition to undertaking the process for identification of potential Relevant Persons, as described above, Santos has undertaken a range of activities to promote opportunities for other organisations or individuals to self-identify as potential Relevant Persons if they considered that their functions, interests or activities may be affected.

These promotional activities included a public information campaign using a range of appropriate media, including, radio, print media, targeted social media and drop-in sessions where information about the proposed activities is provided. Drop-in sessions were provided for Darwin community members from 26-28 March 2024 and were supported by advertising in the 23 March 2024 edition of the NT News.

Details of the public information campaign for the Coastal Waters OEMP and Barossa Production Operations EP are described in Table 4-7 and a detailed advertising schedule is described in Table 4-8. In addition, Santos also made additional efforts to promote awareness of the consultation process among First Nations communities considered most likely to have functions, interests or activities that may be affected by the Activity. Such additional efforts are described in Table 4-10. Santos also has an online self-nomination form on its Consultation Hub website, where consultation materials are published and available for download.

The media and advertising campaign had a regional focus, noting the remoteness of First Nations and other communities in Northern Australia. Social media and/or radio advertising were seen as useful tools to raise awareness in First Nations communities about the proposed Activity and associated consultation opportunities and to invite potentially affected persons to contact Santos.

Further, Santos' third-party consultants and supporting cultural advisors, comprising a team of First Nations leaders with extensive knowledge and experience in relation to First Nations cultures of Northern Australia, were active in



raising awareness and opportunities for participation in consultation through activities to support the establishment of First Nations Consultative Committees. See Section 4.7.5.1.4.

Such activities provide a more than reasonable opportunity for organisations and individuals to self-identify as a Relevant Person for the purpose of OPGGS(E)R section 25 consultation, where they consider themselves to have interests, functions or activities that may be affected by the planned activities and for Relevant Persons to provide their input.

In addition to the above opportunities, Santos also wrote to a number of persons and organisations of whom Santos was aware and considered to be potential Relevant Persons where it was unclear what, if any, functions, interests or activities the person or organisation had that may be affected by the activities under the OEMP and EP, Santos shared links to the Barossa Production Operations Information Booklet and invited these persons or organisations to respond confirming if and what functions, interests or activities they had that may be affected by the activities covered by the OEMP and EP in order for Santos to ascertain them as Relevant Persons and consult them accordingly.

Potential Relevant Persons who did not respond to the opportunity provided them to self-identify in the initial email of 9 February 2024, were still included by Santos in the second email on 11 March 2024 advising the formal commencement of the four-week consultation period. Once again, any who did not respond were followed up by Santos via phone and/or email in May and still afforded an opportunity to respond during an additional two-week consultation phase in May 2024.<sup>30</sup>

Santos' process includes opportunities for the self-identification or nomination of others as Relevant Persons, having regard to consultation information and materials shared directly to known and potential Relevant Persons, and/or indirectly during Santos' public awareness campaign.

Table 4-7:	Public	awareness	campaign	overview
	I UDIIC	awareness	campaign	

Preliminary Consultation		
Website: Website content and consultation materials developed and made available at: https://www.santos.com/offshoreconsultation/	<ul> <li>Provide the public with:</li> <li>Information about Santos' consultation obligations and approach.</li> <li>Descriptions of proposed activities, including potential activity impacts and risks, and proposed management measures.</li> <li>Contact details to enable Relevant Persons to provide feedback.</li> <li>Information about how to self-identify as a Relevant Person, including an on-line nomination form.</li> <li>Details about how feedback will be managed, including provision of Santos' offshore WA and NT privacy notice.</li> </ul>	From 9 February 2024
Advertising: Advertisements in the following publications: • The Australian • The West Australian • NT News • Broome Advertiser	Promote awareness of proposed activities and invite Relevant Persons to self-identify.	From 28 January to 8 June 2024

<sup>30</sup> 

In some cases, persons or organisations identified at later stages in the consultation process were contacted by email outside the above timeframes, to inform them about the consultation for the EP and OEMP and share information about the EP and OEMP activity and associated environmental impacts and risks. Such persons or organisations were invited to indicate to Santos if they considered themselves to be a person (or organisation) whose functions, interests or activities may be affected by the activities proposed to be carried out under the EP and OEMP; and, if so, the nature of those potentially affected functions, interests or activities; and to indicate if they wish to be consulted further for the EP and OEMP.



Preliminary Consultation		
<ul> <li>Kimberley Echo</li> <li>Social media</li> <li>Advertisements on the following radio stations:</li> <li>Darwin Hot 100</li> </ul>		
<b>Consultation materials:</b> Email to identified/potential Relevant Persons with a link to the consultation materials for the Barossa Production Operations EP and Coastal Waters OEMP.	Provide Relevant Persons with details on proposed Activities, including potential activity impacts and risks, and proposed management measures, and	From 9 February to 2 May 2024
<b>One-to-one meetings:</b> Meetings held with authorities, persons and organisations	establish consultation expectations.	February to July 2024
Consultation		
<b>Consultation email:</b> Email to identified Relevant Persons and potential Relevant Persons advising of the commencement of the formal consultation period	Reminder to Santos' identified Relevant Persons and potential Relevant Persons of the commencement and closing dates for the formal consultation period.	From 11 March 2024
<ul> <li>Advertising:</li> <li>Advertisements in the following publications: <ul> <li>The West Australian</li> <li>The Australian</li> <li>NT News</li> <li>Advertisements on the following radio stations:</li> <li>Hit 101.3 Broome</li> <li>Darwin Mix 104.9</li> <li>Darwin Hot 100</li> <li>Pilbara and Kimberley Aboriginal Media Radio</li> </ul> </li> </ul>	Promote awareness of proposed Activities and seek feedback from Relevant Persons.	From 11 March 2024
<b>Consultation email:</b> Reminder email to identified Relevant Persons and potential Relevant Persons advising pending closure of consultation period.	Reminder to Santos identified Relevant Persons and potential Relevant Persons of the closing dates for consultation.	From 3 April 2024
<b>Consultation email:</b> Email to identified Relevant Persons and potential Relevant Persons advising of extension of consultation period.	Advice to Santos identified Relevant Persons and potential Relevant Persons of the new closing date for consultation.	From 7 May 2024
Online meetings	Discussions with Relevant Persons who requested an online meeting.	March to July 2024
Community drop-in sessions	Opportunities for Darwin community members to learn more about the Barossa Gas Project.	26-28 March 2024
Meetings (in-person)	Provide Relevant Persons with information about the Barossa Production Operations EP and Coastal Waters OEMP and discussions with Relevant Persons regarding this information.	March to July 2024

### Table 4-8: Targeted advertising campaign details

Publication date	Advertising type	Towns / Communities	Reach	
Preliminary consultation				
March 2024	Social Media post	Facebook, Instagram and Messenger	Geotargeted – Darwin, Tiwi Island, Indonesia and Timor- Leste	

Publication date	Advertising type	Towns / Communities	Reach	
Preliminary consultation				
January - February 2024	Radio ad - Darwin Hot 100	Darwin towns and communities, focusing on remote communities	Ad aired 28 times	
17 February 2024	Press ad NT News	Half page, page 11	Targeted NT with reach of 25,000	
17 February 2024	Press ad, The West Australian	Half page, page 11	Targeted WA with reach of 481,000	
24 February 2024	Press ad NT News	Half page, page 7	Targeted NT with reach of 25,000	
28 February 2024	Press ad NT News	Half page, page 4	Targeted NT with reach of 25,000	
1 March 2024	Press ad NT News	Half page, page 12	Targeted NT with reach of 25,000	
1 March 2024	Press ad The Australian	Half page, page 6	Targeted WA with reach of 398,000	
2 March 2024	Press ad, The West Australian	Half page, page 11	Targeted WA with reach of 481,000	
7 March 2024	Press ad Broome Advertiser	Half page, page 7	Targeted WA with reach of 14,000	
7 March 2024	Press ad Kimberley Echo	Half page, page 5	Targeted WA with reach of 450	
9 March 2024	Press ad NT News	Half page, page 14	Targeted NT with reach of 25,000	
Consultation				
March-April 2024	Radio ad - Hit 101.3 Broome	Broome towns and communities, focusing on remote communities	Ad aired 28 times	
March- April 2024	Radio ad - Darwin Mix 1049	Darwin towns and communities, focusing on remote communities	Ad aired 56 times	
March – April 2024	Radio ad - Darwin Hot 100	Darwin towns and communities, focusing on remote communities	Ad aired 48 times	
March – April 2024	Radio ad - Pilbara and Kimberley Aboriginal Media Radio	Pilbara and Kimberley towns and communities, focusing on remote communities	Ad aired 28 times	
16 March 2024	Press ad NT News	Half page, page 18	Targeted NT with reach of 25,000	
16 March 2024	Press ad, The West Australian	Half page, page 11	Targeted WA with reach of 481,000	
23 March 2024	Press ad NT News	Half page, page 4	Targeted NT with reach of 25,000	
27 March 2024	Press ad NT News	Half page, page 9	Targeted NT with reach of 25,000	
Public Notice				
23 March 2024	Press ad NT News	Quarter page, page 5	Targeted NT with reach of 25,000	

Relevant Persons consulted for the Coastal Waters OEMP and Barossa Production Operations EP are listed below in Table 4-9. As outlined in Section 4.2, the list below includes all Relevant Persons identified and consulted by Santos across the Coastal Waters OEMP and the Barossa Production Operations EP, notwithstanding that some persons may have functions, interests or activities that may only be affected under the Barossa Production Operations EP.

### Table 4-9: Summary of Relevant Persons

Relevant Person Category	Summary of Relevance		
Section 25(1)(a) of the OPGGS(E)R: Departments or agencies of the Commonwealth to which the activities to be carried out under the environment plan may be relevant			
Australian Communications and Media Authority (ACMA)	ACMA is responsible for the regulation of communications and media services in Australia.		
	ACMA is a relevant agency because the Activity has the potential to impact future proposed subsea communications cable installations.		
Australian Fisheries Management Authority (AFMA)	AFMA is responsible for managing Commonwealth fisheries and is a relevant agency because the Activity has the potential to impact on fisheries resources in AFMA managed fisheries. AFMA expects petroleum operators to consult directly with fishing operators about all activities and projects which may affect day to day fishing activities. AFMA also provides industry association contacts for petroleum operators to use when consultation with fishing operators is required.		
Australian Hydrographic Office (AHO)	AHO is responsible for maintaining and disseminating nautical charts, including the distribution of Notices to Mariners.		
Australian Institute of Marine Science (AIMS)	AIMS is Australia's tropical marine research agency and is established under the Australian Institute of Marine Science Act 1972 (AIMS Act).		
Australian Maritime Safety Authority (AMSA) – maritime safety/marine pollution	AMSA is the statutory and control agency for maritime safety and vessel emergencies in Commonwealth Waters. AMSA is a relevant agency because the proposed offshore activities may impact on the safe navigation of commercial shipping in Australian waters. AMSA is also a relevant agency as one of its functions is to prevent and combat ship- sourced pollution in the marine environment.		
Clean Energy Regulator (CER)	CER administers schemes legislated by the Australian Government for measuring, managing, reducing or offsetting Australia's carbon emissions, including the National Greenhouse and Energy Reporting (NGER) Scheme and the Safeguard Mechanism underpinned by the NGER framework.		
Climate Change Authority (CCA)	CCA is a statutory agency responsible for providing independent advice to government on climate change policy. It was established by and operates under the <i>Climate Change Authority Act 2011</i> .		
Commonwealth Science and Industry Research Organisation (CSIRO)	CSIRO's functions include international scientific liaison, training of research workers, publication of research results, technology transfer of other research, provision of scientific services and dissemination of information about science and technology. CSIRO has a division dedicated to oceans and atmosphere research.		
Department of Agriculture, Forestry and Fisheries (DAFF) – Biosecurity	DAFF Biosecurity administers the <i>Biosecurity Act 2015</i> (Cth). DAFF Biosecurity is a relevant agency for consultation because the Activity involves the movement of aircraft or vessels between Australia and offshore petroleum activities either inside or outside Australian territory.		
Department of Agriculture, Forestry and Fisheries (DAFF) – Fisheries	DAFF Fisheries also has primary policy responsibility for promoting the biological, economic and social sustainability of Australian fisheries. DAFF Fisheries is a relevant agency for consultation because the Activity has the potential to impact on fishing operations and/or fishing habitats in Commonwealth waters.		
Department of Climate Change, Energy, the Environment and Water (DCCEEW) – Underwater Cultural Heritage	DCCEEW protects Australia's natural environment and heritage sites, helps Australia respond to climate change and carefully manages water and energy resources. The Underwater Cultural Heritage branch at DCCEEW is responsible for administering the UCH Act.		

Relevant Person Category	Summary of Relevance
Department of Defence (DoD)	DoD is a relevant agency for consultation because:
	<ul> <li>the proposed Activity may impact DoD training and operational requirements, in that the EP EMBA overlaps DoD training areas.</li> </ul>
	<ul> <li>the proposed Activity encroaches on known training areas and/or restricted airspace.</li> </ul>
	<ul> <li>there is a risk of unexploded ordnance in the area where the Activity is taking place.</li> </ul>
Department of Foreign Affairs and Trade (DFAT)	DFAT is a relevant agency for consultation where:
	a proposed activity may cross into or impact on waters outside of Australia's maritime jurisdiction; and/or
	<ul> <li>a proposed activity poses any oil spill or other environmental risks that could result in impacts to other international jurisdictions where persons or organisations that may be impacted by a proposed activity include foreign individuals or governments.</li> </ul>
	DFAT has a role in assisting Oil and Gas operators to liaise with foreign governments in the event waters outside Australian jurisdiction are impacted by an activity.
Department of Home Affairs and Australian Border Force (ABF)	The Department of Home Affairs is responsible for overseeing migration, national security and resilience, and border- related functions. ABF is an operationally independent body within the Home Affairs portfolio. ABF is Australia's border law enforcement agency and customs service. ABF's vessels undertake patrols as part of its surveillance and response activities throughout an offshore maritime area of almost 45.1 million km <sup>2</sup> . This area includes the EP EMBA.
Department of Industry, Science and Resources (DISR)	DISR is a relevant agency for consultation because its responsibilities include offshore oil and gas development and safety and greenhouse gas (GHG) storage.
Director of National Parks (DNP)	DNP is the statutory authority responsible for administration, management and control of Commonwealth marine reserves. The DNP is a Relevant Person for consultation where:
	• the Activity or part of the Activity is within the boundaries of a proclaimed Commonwealth marine reserve;
	<ul> <li>activities proposed to occur outside a reserve may impact on the values within a Commonwealth marine reserve; and / or</li> </ul>
	<ul> <li>an environmental incident occurs in Commonwealth waters surrounding a Commonwealth marine reserve and may impact on the values within the reserve.</li> </ul>
Fisheries Research Development Council (FRDC)	FRDC has a formal role in the planning and investment in fisheries research and development to support the ongoing sustainability of aquatic sectors and aquatic ecosystems. It is a co-funded partnership between the Australian Government and fisheries and aquaculture and a statutory corporation under the <i>Primary Industries Research and Development Act 1989</i> (Cth) responsible to the Minister for Agriculture, Fisheries and Forestry.
National Indigenous Australians Agency (NIAA)	NIAA is an Australian Government agency responsible for whole-of-government coordination of policy development, program design and service delivery for Indigenous Australians.
Section 25(1)(b) of the OPGGS(E)R: Departments or agencie relevant.	es of the Northern Territory to which the activities to be carried out under the environment plan may be
Aboriginal Areas Protection Authority (AAPA)	The AAPA supports development while safeguarding Aboriginal sacred sites. Under the NTASS Act, the AAPA is responsible for overseeing the protection of Aboriginal sacred sites on land and sea across the whole of the NT. The NTASS Act also gives the Authority the power to prosecute people and organisations that damage sacred sites.

Relevant Person Category	Summary of Relevance
Darwin Harbour Advisory Committee (DHAC)	The DHAC provides advice to the NT Government through the Minister for Environment, Parks and Water Security on the effective management of Darwin Harbour and its catchment.
Department of Lands, Planning and the Environment (DLPE) (formerly the Department of Environment, Parks and Water Security (NT) during the initial consultation period for this Coastal Waters OEMP (DEPWS NT).	DEPWS NT combines the functions of the previous Department of Environment and Natural Resources and the Parks and Wildlife Commission from the former Department of Tourism, Sport and Culture (DTSC). The government established the department to combine many of the key functions that foster and protect the environment and natural resources in the NT. This includes water, land resource management, environmental issues and the parks and wildlife functions.
Department of Agriculture and Fisheries (DAF) (formerly the Department of Industry, Tourism and Trade (NT) – Fisheries Division during the initial consultation period for this Coastal Waters OEMP (DITT_NT - Fisheries).	DITT NT Fisheries Division has functions in relation to NT managed fisheries. The OAs overlap NT managed fisheries. The Aquatic Biosecurity Unit of Northern Territory Fisheries monitors and manages the risk of new marine pests arriving in the NT. The unit monitors for early detection of aquatic pests; coordinates inspections and treatment of high-risk vessels entering Darwin; responds to reported sightings of invasive freshwater and marine pests; and educates the public about the impacts, prevention and management of aquatic pests. The Department also operates the Darwin Aquaculture Centre, the NT Government's key aquaculture research and development facility.
Department of Logistics and Infrastructure (DLI) (formerly the Department of Infrastructure, Planning and Logistics (NT) – Transport and Civil during the initial consultation period for this Coastal Waters OEMP (DIPL-NT).	DIPL NT Transport is responsible for all aspects of marine transport in NT waters, including the Port of Darwin which will continue to be the supply base for Barossa offshore activities.
Department of Lands, Planning and the Environment (DLI) (formerly the Department of Territory Families, Housing and Communities (NT) – Heritage branch during the initial consultation period for this Coastal Waters OEMP.(DTFHC NT Heritage)	The DTFHC NT Heritage branch has a role in protecting the maritime heritage of the Northern Territory. Multiple known shipwrecks, sunken aircraft, and historic (more than 75 years old) aircraft and shipwrecks and other sites occur within the EP EMBA. There are multiple sites protected under Commonwealth Underwater Cultural Heritage Act 2018 and NT Heritage Act 2011.
Environmental Protection Authority NT (EPA NT)	The EPA NT is an independent authority established under the <i>NT Environment Protection Authority Act 2012</i> . The EPA NT approves conditions for the DPD in NT Waters and can amend these at any time. Water quality and other environmental aspects of Darwin Harbour and NT waters could be impacted in the event of an unplanned hydrocarbon spill.
NT Fire and Emergency Services (formerly Department of Police, Fire and Emergency Services during the consultation period for this Coastal Waters OEMP.	The Department would be involved in response measures in the event of a spill in NT Waters.
Parks and Wildlife Commission of the Northern Territory	Parks and Wildlife Commission of the Northern Territory is the NT Government agency responsible for tasks including the establishment, management and protection of parks, reserves, sanctuaries and other land, and the protection, conservation and sustainable use of wildlife.
Department of Tourism and Hospitality (DTH) (formerly Tourism NT during the initial consultation period for this Coastal Waters OEMP. (Tourism NT)	Tourism NT is the government statutory authority responsible for promoting tourism in the NT, including potential activity by NT based operators in the EP EMBA.
Section 25(1)(b) of the OPGGS(E)R: Departments or agencie	es of Western Australia to which the activities to be carried out under the environment plan may be relevant.
Department of Biodiversity, Conservation and Attractions Western Australia (DBCA WA)	DBCA WA has functions in relation to the protection of Western Australian flora and fauna, including in relation to the Scott Reef Reserve (which is in WA waters) and works in tandem with Department of Primary Industries and Regional

Relevant Person Category	Summary of Relevance	
	Development (DPIRD) to promote biodiversity and conservation with an interest in sustainable management of species and ecosystems.	
Department of Primary Industries and Regional Development (DPIRD) – Fisheries	DPIRD-WA is responsible for managing West Australian fisheries. Several WA-managed commercial fisheries' zones extend beyond WA Waters and into Commonwealth Waters of the EP EMBA.	
Department of Transport Western Australia (DoT WA)	DoT WA has functions in relation to commercial vessel movements in the navigable waters of the State and seas adjacent to WA. Its interests extend to response to an unplanned spill event through its Maritime Environmental Emergency Response unit.	
Kimberley Ports Authority	The Authority is responsible for the safe and efficient operation of the port and to protect the environment in which the port operates.	
WA Marine Science Institution (WAMSI)	WAMSI is a government consortium of state, government and academic organisations working collaboratively for promotion of science research.	
Section 25(1)(c) of the OPGGS(E)R: Department of the response	onsible Northern Territory Minister.	
Department of Mining and Energy (DME) (formerly the NT Department of Industry, Tourism and Trade– Mines and Energy (DITT-NT – Energy) during the initial consultation period for this Coastal Waters OEMP.(DITT-NT – Energy)	DITT-NT Energy is the department of the responsible Territory Minister and is required to be consulted under regulation 25(1)(c) of the OPGGS(E)R.	
Section 25(1)(d) of the OPGGS(E)R: Persons or organisations whose functions, interests or activities may be affected by the activities to be carried out under the environment plan		
Academic and Research Organisations		
Academic and Nescarch Organisations		
Arafura Timor Research Facility (ATRF)	ATRF is a joint venture between AIMS and the Australian National University. It was developed through a successful Major National Research Facilities grant application with support from the NT government and Charles Darwin University. The facility was established to accommodate world class research into marine and coastal ecosystems of the Arafura and Timor seas and to explore the increasing threats to Australia's fisheries and marine biodiversity in the region. A wide range of research activities are being processed.	
Arafura Timor Research Facility (ATRF) Australian Marine Sciences Association – NT (AMSA-NT)	ATRF is a joint venture between AIMS and the Australian National University. It was developed through a successful Major National Research Facilities grant application with support from the NT government and Charles Darwin University. The facility was established to accommodate world class research into marine and coastal ecosystems of the Arafura and Timor seas and to explore the increasing threats to Australia's fisheries and marine biodiversity in the region. A wide range of research activities are being processed. AMSA-NT is a professional body for marine scientists, with a branch in the NT. Its listed interests and stated activities include promoting all aspects of marine science in the NT and making formal comment on NT marine development assessments and NT Government policies, strategies and plans, and nominations of rare and threatened marine species and habitats in the NT.	
Arafura Timor Research Facility (ATRF)           Australian Marine Sciences Association – NT (AMSA-NT)           AusTurtle Inc	ATRF is a joint venture between AIMS and the Australian National University. It was developed through a successful Major National Research Facilities grant application with support from the NT government and Charles Darwin University. The facility was established to accommodate world class research into marine and coastal ecosystems of the Arafura and Timor seas and to explore the increasing threats to Australia's fisheries and marine biodiversity in the region. A wide range of research activities are being processed. AMSA-NT is a professional body for marine scientists, with a branch in the NT. Its listed interests and stated activities include promoting all aspects of marine science in the NT and making formal comment on NT marine development assessments and NT Government policies, strategies and plans, and nominations of rare and threatened marine species and habitats in the NT.	
Arafura Timor Research Facility (ATRF)         Australian Marine Sciences Association – NT (AMSA-NT)         AusTurtle Inc         Charles Darwin University (CDU)	ATRF is a joint venture between AIMS and the Australian National University. It was developed through a successful Major National Research Facilities grant application with support from the NT government and Charles Darwin University. The facility was established to accommodate world class research into marine and coastal ecosystems of the Arafura and Timor seas and to explore the increasing threats to Australia's fisheries and marine biodiversity in the region. A wide range of research activities are being processed. AMSA-NT is a professional body for marine scientists, with a branch in the NT. Its listed interests and stated activities include promoting all aspects of marine science in the NT and making formal comment on NT marine development assessments and NT Government policies, strategies and plans, and nominations of rare and threatened marine species and habitats in the NT. AusTurtle Inc. is a non-profit organisation that promotes sea turtle conservation and research in northern Australia. The NT's main university is research-intensive with a range of projects and partnerships in indigenous and tropical health, environmental science and public policy. One example is the current investigation of low technology, seabased aquaculture systems for remote coastal communities. The team is sampling wild blacklip oysters from 8 locations across the NT, assessing shellfish quality, heavy metals and vibrio testing. CDU is a member of the Darwin Harbour Advisory Committee	
Arafura Timor Research Facility (ATRF)         Australian Marine Sciences Association – NT (AMSA-NT)         AusTurtle Inc         Charles Darwin University (CDU)         Commercial fishing – Commonwealth managed	ATRF is a joint venture between AIMS and the Australian National University. It was developed through a successful Major National Research Facilities grant application with support from the NT government and Charles Darwin University. The facility was established to accommodate world class research into marine and coastal ecosystems of the Arafura and Timor seas and to explore the increasing threats to Australia's fisheries and marine biodiversity in the region. A wide range of research activities are being processed. AMSA-NT is a professional body for marine scientists, with a branch in the NT. Its listed interests and stated activities include promoting all aspects of marine science in the NT and making formal comment on NT marine development assessments and NT Government policies, strategies and plans, and nominations of rare and threatened marine species and habitats in the NT. AusTurtle Inc. is a non-profit organisation that promotes sea turtle conservation and research in northern Australia. The NT's main university is research-intensive with a range of projects and partnerships in indigenous and tropical health, environmental science and public policy. One example is the current investigation of low technology, seabased aquaculture systems for remote coastal communities. The team is sampling wild blacklip oysters from 8 locations across the NT, assessing shellfish quality, heavy metals and vibrio testing. CDU is a member of the Darwin Harbour Advisory Committee	



Relevant Person Category	Summary of Relevance	
Northern Prawn Fishery		
Southern Bluefin Tuna Fishery		
Western Skipjack Tuna Fishery		
Western Tuna and Billfish Fishery		
North-West Slope Trawl Fishery		
Torres Strait Fishery		
Commercial fishing – Northern Territory managed		
NT-managed fisheries that overlap the EP EMBA:	Santos has consulted via representative organisation, Northern Territory Seafood Council (NTSC).	
Aquarium Fishery		
<ul> <li>Bait Net Fishery (not entitled to fish in NT Coastal Waters OA)</li> </ul>		
<ul> <li>Barramundi Fishery (not entitled to fish in NT Coastal Waters OA)</li> </ul>		
<ul> <li>Coastal Line Fishery (not entitled to fish in NT Coastal Waters OA)</li> </ul>		
<ul> <li>Coastal Net Fishery (not entitled to fish in NT Coastal Waters OA)</li> </ul>		
Demersal Fishery		
<ul> <li>Mollusc Fishery (not entitled to fish in NT Coastal Waters OA)</li> </ul>		
Mudcrab Fishery		
Offshore Net and Line Fishery		
Pearl Oyster Fishery		
Spanish Mackerel Fishery		
<ul> <li>Trepang Fishery (not entitled to fish in NT Coastal Waters OA)</li> </ul>		
Commercial fishing – Western Australian managed		
Licence holders in the following WA-managed fisheries:	Licence holders of these fisheries are entitled to fish within the EP EMBA. Santos has consulted via representative	
Abalone Fishery	organisation, Western Australian Fishing Industry Council (WAFIC), noting WAFIC published guidance on consultation of fishers.	
Kimberley Crab Fishery		
Kimberley Prawn Fishery		
Mackerel Managed Fishery		
Marine Aquarium Fishery		
Northern Demersal Scalefish Managed Fishery		
Pilbara Crab Fishery		

Relevant Person Category	Summary of Relevance
South-West Costal Salmon Fishery	
Specimen Shell Fishery	
West Coast Deep Sea Crustacean Fishery	
Energy Industry – Petroleum titleholders and GHG permit h	olders
Operators:	Operators within the EP EMBA.
Bengal Energy	
Eni Australia Ltd	
EOG Resources	
Finder Energy	
INPEX Ichthys Pty Ltd	
Jadestone	
Melbana Energy	
Neptune	
PTTEP Pty Ltd	
<ul> <li>Shell Development (Australia) Pty Ltd</li> </ul>	
SundaGas Bunda Unipessoal Lda	
Woodside Energy Ltd	
Environmental conservation organisations	
ATSEA-2 Project	According to its website, ATSEA-2 is a Global Environment Facility-funded program, managed and executed under the United Nations Development Program. It has a Regional Steering Committee made up of representatives from national government and lead agencies in Australia, Indonesia, Papua New Guinea and Timor-Leste.
Australia Institute	The Australia Institute is a public policy think tank based in Canberra that carries out research and comments publicly on a broad range of economic, social, and environmental issues.
Australian Conservation Foundation (ACF)	ACF is a national non-government environmental advocacy organisation based in Melbourne.
Australian Marine Conservation Society – NT (AMCS NT)	According to its website, AMCS NT is a grassroots independent environmental conservation organisation and charity that works to protect ocean wildlife along the NT coastline, waters and seas. It advocates for evidence-based solutions to conservation activity and works closely with marine research centres.
Australian Parents for Climate Action Darwin and NT	According to its NT website, Australian Parents for Climate Action Darwin and NT is on a mission to empower parents, carers, families and everyone who cares about kids, to advocate for urgent action on climate. It plans family-friendly, non-partisan activities that engage communities, engage politicians in climate solutions, and amplify positive stories.
Climate Action Darwin	According to its website, Climate Action Darwin influences decision-makers to adopt climate-friendly policies, supports Darwin residents to take climate action and reduce their own climate impact, advocates for a transition to a zero-carbon economy, informs and educates audiences on NT climate change impacts and solutions and supports other local and active groups working for a safe climate.

Relevant Person Category	Summary of Relevance
Conservation Council of Western Australia (CCWA)	According to its website and correspondence dated 12 April 2023, CCWA promotes an interest in the protection and restoration of the WA natural environment, including waters, a marine park and marine life potentially within the EP EMBA.
Doctors for the Environment Australia	Doctors for the Environment Australia (DEA) is a national organisation of medical professionals concerned at the impacts of climate change on human health.
Environment Centre Northern Territory (ECNT)	According to its website, ECNT is a not-for-profit incorporated association whose objects include protection of all aspects of the natural environment, conducting campaigns to protect the natural environment, environmental research, and public education and information about the natural environment. ECNT is involved in the "Stop Barossa Gas" campaign.
Greenpeace	According to its website, Greenpeace's stated goals include the protection of ocean biodiversity and marine life, including campaigning for protection of whales <sup>31</sup> (fauna identified in this Coastal Waters OEMP as potentially affected by the Activity impacts or risks) and sea turtles <sup>32</sup> (also fauna identified in this Coastal Waters OEMP as potentially affected by the Activity impacts or risks).
Jubilee Australia Research Centre	Jubilee Australia Research Centre states that it engages in research and advocacy to promote economic justice for communities in the Asia-Pacific region and accountability for Australian corporations and government agencies operating there. The Centre is involved in the Stop Barossa campaign.
Keep Top End Coasts Healthy	According to its website, Keep Top End Coasts Healthy is part of an alliance of environment groups including the AMCS NT and the ECNT. Keep Top End Coasts Healthy claims to work with stakeholders with respect to coastal preservation and establishment of marine protected areas.
Sea Turtle Foundation	According to its website, the Sea Turtle Foundation <sup>33</sup> is a non-profit, non-government group based in Australia with a stated interest in protecting sea turtles through research, education and action, including specifically the olive ridley turtle, leatherback turtle, loggerhead turtle and flatback turtle, being turtle species cited in this Coastal Waters OEMP as being potentially affected by the impacts or risks of the Activity.
West Timor Care Foundation	According to previous correspondence received from West Timor Care Foundation, the Foundation claims to be an advocacy organisation concerned with the interests and welfare of people who depend on the coast of Timor for their livelihoods and who have been, or may be, impacted by oil spills from petroleum activities in the Timor Sea, including areas within the EP EMBA. Santos has been unable to locate a website for West Timor Care Foundation.
Wilderness Society	According to its website, the Wilderness Society is a peak conservation body with an interest in activities that may affect the marine environment.
WorldFish Timor-Leste	According to its website, WorldFish is a research organisation focusing on sustainable aquatic food systems in Timor- Leste. It has an interest in resilient and sustainable aquaculture projects and small-scale fisheries production, promoting community-based resource management of coastal fisheries to strengthen livelihoods and combat poverty and malnutrition and works in a partnership model with non-government organisations (NGOs) and governments.

<sup>&</sup>lt;sup>31</sup> https://www.greenpeace.org.au/what-we-do/protecting-oceans/whales/

<sup>&</sup>lt;sup>32</sup> https://www.greenpeace.org/international/story/28229/turtle-journey-urgent-protect-the-oceans/; https://www.greenpeace.org/international/publication/28181/turtles-under-threat/

<sup>&</sup>lt;sup>33</sup> https://seaturtlefoundation.org/about

Relevant Person Category	Summary of Relevance
World Wildlife Fund (WWF)	WWF is a peak conservation body with an interest in activities that may affect the marine environment.
First Nations People and groups	
The following groups may have interests that intersect the EP EMBA. Information was also provided to these organisations to help identify and consult groups or individuals whose spiritual or cultural connections to land and sea country in accordance with Indigenous tradition may be affected by proposed activities.	
In addition, targeted regional advertising was conducted to provide opportunity for individuals whose functions, interests and activities may be affected by the proposed activity to self- identify as Relevant Persons.	
Representative organisations – NT	
Larrakia Nation Aboriginal Corporation (LNAC)	LNAC is one of Darwin's leading community service organisations. Larrakia Nation Aboriginal Corporation was set up in 1997 through the Northern Land Council originally to provide a corporate identity for Larrakia people to uphold Native Title claims. In the subsequent 20 years, it has grown to represent the Traditional Owners of the Darwin region and to speak on behalf of Larrakia people, deliver community and outreach services to the broader Darwin community, and operate the Larrakia Land and Sea Ranger services. LNAC board members, six of eight of whom are Larrakia member directors, are responsible for governing LNAC on behalf of all Larrakia members. LNAC has over 600 Larrakia families' members, any Larrakia person is eligible to be a member. Benefits of membership include election of family representatives at the Annual General Meeting and advocacy on a members' behalf. <sup>34</sup>
Northern Land Council (NLC)	The NLC is the Native Title Representative Body for the Northern Region, including sea country. Its functions are prescribed under the NT Act. The NLC also has statutory obligations under the ALR Act and is authorised to perform certain functions under the NT Act. The NLC's area of interest includes sea country where non-exclusive native title rights and interests may exist, including within the EP EMBA. NLC Executive Council members are also the directors of the Top End (Default Prescribed Body Corporate/Community Living Area) Aboriginal Corporation RNTBC (TED PBC) which is responsible for an area of sea country near the Croker Islands. The NLC also provides administrative services to the Corporation. The NLC is also responsible for the administration of Land Trusts. Consultation with Land Trusts also occurs via the NLC.
Tiwi Land Council (TLC)	The TLC is governed under the ALR Act. The Tiwi Aboriginal Land Trust was also established under the ALR Act and the TLC is the only body with authority to direct the Trust. The authority of the TLC does not extend into Commonwealth offshore waters, although the sea country interests of Tiwi Island clans do, including within the EP EMBA.
Wickham Point Deed Liaison Committee (WPDLC)	The objective of the WPDLC is to strengthen the dialogue between Santos and the Larrakia people and support the delivery of the parties' commitments under the Wickham Point Deed entered into between Darwin LNG and the Northern Land Council on 29 April 1999. Santos coordinates quarterly meetings with the Wickham Point Deed liaison committee, which includes representatives from Larrakia family groups, the functions of which are set out in the Wickham Point Deed and include making recommendations to Santos on various matters such as environmental, cultural heritage, employment and business opportunities.
First Nations Consultative Committees and coastal clan groups – NT	

<sup>&</sup>lt;sup>34</sup> Source: https://larrakia.com/

Relevant Person Category	Summary of Relevance
Larrakia People	The Larrakia people are the traditional owners of the Darwin region. Larrakia country runs from Cox Peninsula in the west to Gunn Point in the north, Shoal Bay in the east and down to the Manton Dam area southwards. The Larrakia People consist of between eight to fourteen family groups, depending on how families are grouped. <sup>35</sup>
Cobourg Peninsula Consultative Committee	The Cobourg Peninsula Consultative Committee speaks for all of the Cobourg Peninsula, south to Endyalgout Island and east towards Wauk, and includes the adjacent sea country. The committee includes the Agalda, Murran, Ngaindjagar and Madjunbalmi clans, and includes the Garig Gunak Marine Park (NT)
Kardu Lalingkin Consultative Committee	The Kardu Lalingkin Consultative Committee FNCC speaks for country extending from the Fitzmaurice River, including Wadeye community to north of the Marri-Jabin (Thamururr) Indigenous Protected Area, and including coastal parts of the Joseph Bonaparte Marine Park.
Djulidki (Bradshaw) Consultative Committee	The Djulidki Consultative Committee speaks for the area approximately contiguous with, but larger than the Bradshaw Field Training Area, in the south west coast of the NT. It includes Quoin and Clump Islands and is bordered by the Victoria River to the south and the Fitzmaurice River to the north.
Gapu Maringa Consultative Committee	The Gapu Maringa Consultative Committee speaks for country extending from the Blyth River through to the westernmost part of Elcho Island. It includes coastal areas and islands (Darbada, Crocodile, Milingimbi, Rabuma, Banyan and Mooroongga Islands) and includes the western tip of Elcho Island, including the community of Galiwin'ku and the entirety of the Crocodile Islands Maringa IPA.
Goulburn Island Consultative Committee	The Goulburn Island Consultative Committee speaks for north and south Goulburn Islands. It includes the western section of the Arnhem Marine Park.
Jindiwi Consultative Committee	The Jindiwi Consultative Committee speaks for country extending east from the Adelaide River, through to just south of Endyalgout Island, at the bottom of the Cobourg Peninsula, and including Van Diemen Gulf, Field and Barron Islands. It includes groups living along the coastal areas, of the West, South and East Alligator Rivers, including the Wulna Clan. It also includes the coastal section of the Mary River IPA.
Maningrida Regional Consultative Committee	The Maningrida Regional Consultative Committee speaks for an extensive coastal area beginning south of Wauk and continuing east under the Goulburn Islands past the Nungbalgarri Creek and extending past Maningrida community to the west bank of the Blyth River. It includes the middle and eastern sections of the Arnhem Marine Park, and the Djelk Stage 2 IPA.
Miyarrka Consultative Committee	The Miyarrka Consultative Committee speaks for country around the community of Gapuwiy'ak (Lake Evella) and extends north and west to include the sea country and coastal areas of the Hardy Island Bay and including Inglis, and Cotton Islands, and includes Yolŋu language groups living in these areas. It extends east to the edge of the Dhimurru IPA and includes the south-eastern part of the Marthakal IPA.
Mulyurrud Consultative Committee	The Mulyurrud Consultative Committee speaks for Croker Island, including the Gadura-Minaga, Mangalarra and Mandilarri clan estates, and the adjacent sea country, including several islands to the east and north east of Croker Island and the Ildugidj clan estate located on the mainland coastline (south from Croker Island). This Committee's area includes the southern portion of the Arafura Marine Park.

<sup>&</sup>lt;sup>35</sup> 2012 paper " Larrakia Family Groups" written by Mr Bill Day (https://www.drbilldayanthropologist.com/resources/Larrakia%20Family%20groups.pdf)

Relevant Person Category	Summary of Relevance
Ngoy Garmak Consultative Committee	The Ngoy Garmak Consultative Committee speaks for the Wessel Islands chain, excluding Galiwin'ku, but including the central and northern parts of Elcho Island and small sections of adjacent mainland coastal areas. It includes Yolŋu language groups living in these areas. This Committee also speaks to the northern part of the Marthakal IPA.
Rak Badjalarr Consultative Committee	The Rak Bajalarr Consultative Committee speaks for country extending north from the Daly River to the Cox Peninsula, and adjacent coastal sea country, including the Peron Islands and the Dum In Mirrie, Beer Eeetar, Windirr and Grose, Quail and Indian Islands. It includes the western part of the Darwin harbour and associated waterways, and represents the Kenbi, Emmiyangal, Mendheyangal, Kiyuk, Wadigany, Murranungu, Malak Malak and Marriamu clans located over the coastal areas from the Cox Peninsula to the Daly River.
Tiwi Islands Clan Groups and Individuals	The Appeal Judgment found that "Mr Tipakalippa and the Munupi clan had interests within the meaning of reg 11(A)(d) <sup>36</sup> of the OPGGS(E)R that required them to be consulted <sup>37</sup> . Mr Tipakalippa had claimed that he and the Munupi clan, as well as other Tiwi Island people, have "sea country" in the Timor Sea to the north of the Tiwi Islands. The Tiwi Islands are located approximately 80 km north of Darwin in the Arafura Sea. There are three major communities on the Tiwi Islands. The largest community is Wurrumiyanga (on Bathurst Island), with smaller communities of Milikapiti and Pirlangimpi located on Melville Island. There are eight landowning groups (clans) on the islands, Mantiyupwi, Munupi, Yimpinari, Malawu, Wulirankuwu, Wurankuwu, Mirrikawuyanga and Jikilaruwu (or Tikalaru). Members of the Mantiyupwi clan also speak for the Vernon Islands, which are located between the Tiwi Islands and mainland NT.
Wulna Clan	Wulna Clan are a party to the Mary River ILUA. The Clan has representation through the Wulna members of the Jindiwi Consultative Committee, consistent with their preferences expressed at meetings with Wulna clans prior to this OEMP consultation.
Other First Nations organisations – NT	
Aboriginal Sea Company	Incorporated entity with administrative support provided by the NLC. The Aboriginal Sea Company's area of interest is the entire Top End (sea country and intertidal). The Company facilitates the participation of Traditional Owners in commercial fishing, aquaculture and other opportunities associated with fishing activities in NT waters that could be impacted by planned activities or an unplanned spill. The Company is governed by a board comprising representation from the three land councils with traditional ownership of sea country – Northern, Tiwi and Anindilyakwa land councils.
Gwalwa Daraniki Association (GDA)	Administers the Kalaluk and Minmarama communities near Darwin, including Larrakia people, and is responsible for overseeing a number of developments aimed at improving the long term physical and economic wellbeing of the tenants of those communities.
Kenbi Rangers	The Kenbi Rangers manage the country of the Cox Peninsula - Darwin and Bynoe Harbours and Islands. The Kenbi Rangers' base on Cox Peninsula is administered by the NLC.
Larrakia Development Corporation (LDC)	The LDC aims to create economic opportunity for Larrakia People through the creation and operation of sustainable businesses models, and the maintenance of the Larrakia Development Trust. The Larrakia Development Corporation's core activities include land holdings and development, property development, heritage monitoring, ground

<sup>&</sup>lt;sup>36</sup> Section 25(1)(d) of updated OPGGS(E)R 2023

<sup>&</sup>lt;sup>37</sup> Santos NA Barossa Pty Ltd v Tipakalippa [2022] FCAFC 193 [80]

Relevant Person Category	Summary of Relevance	
	maintenance, and landscaping. LDC is governed by a board of independent directors. Santos has consulted with LDC in its capacity as a commercial organisation that aims to support positive economic outcomes for Larrakia people	
North Australian Indigenous Land and Sea Management Alliance	Darwin-based Native Title Prescribed Body Corporate with administrative services via the NLC. NLC Executive Council members are the directors of the Top End Default Prescribed Body Corporate. Place / Area of Interest (descriptions of land includes adjacent sea country): Entire Top End.	
Representative Organisations – WA		
Kimberley Land Council (KLC)	The KLC is the Native Title Representative Body for the Kimberley region in WA. Its primary role is to provide native title services to Kimberley Aboriginal people. KLC's area of interest includes sea country where non-exclusive native title rights and interests may exist, including within a section of Commonwealth waters within the EP EMBA. The KLC is also named in several Marine Park Management Plans off the Kimberley coast.	
Balanggarra Aboriginal Corporation	The Balanggarra Aboriginal Corporation, based in Wyndham, is the Registered Native Title Body Corporate (RNTBC) for the Balanggarra People and manages their native title determination	
Bardi and Jawi Niimidiman Aboriginal Corporation	The Bardi and Jawi Niimidiman Aboriginal Corporation, based in Broome, is the RNTBC for the Bardi and Jawi Niimidiman People and manages their native title determination.	
Dambimangari Aboriginal Corporation	The Dambimangari Aboriginal Corporation, based in Derby, is the Aboriginal corporation nominated by the Wanjina Wunggurr RNTBC, (which holds the larger native title determination over the area) to manage the southern part of the determination.	
Mayala Inninalang Aboriginal Corporation	The Mayala Inninalang Aboriginal Corporation, based in Broome, is the RNTBC for Mayala Inninalang people, and manages their determination.	
Miriuwung and Gajerrong Aboriginal Corporation	The Miriuwung and Gajerrong Aboriginal Corporation, based in Kununurra, is the RNTBC for the Miriuwung-Gajerrong People and manages their native title determination.	
Wunambal Gaambera Aboriginal Corporation	The Wunambal Gaambera Aboriginal Corporation, based in Wyndham, is Aboriginal Corporation nominated by the Wanjina Wunggurr RNTBC (which holds the larger native title determination over the area) to manage the northern part of the determination.	
Industry Associations – commercial fishing		
Australian Southern Bluefin Tuna Industry Association (ASBTIA)	ASBTIA represents the interests of commercial fishers in the Southern Bluefin Tuna Fishery and Western Skipjack Fishery.	
Commonwealth Fisheries Association (CFA)	CFA represents the interests of commercial fishers with licences in Commonwealth waters.	
Northern Prawn Fishery Industry (NPFI)	NPFI represents the interests of the interests of commercial fishers in the Northern Prawn Fishery.	
Northern Territory Seafood Council (NTSC)	NTSC is the peak representative body for the wild catch, aquaculture and trader/processor seafood sectors in the NT.	
Western Australian Fishing Industry Council (WAFIC)	WAFIC represents the interests of the WA commercial fishing, pearling and aquaculture sector.	
Industry Associations – recreational fishing		
Amateur Fishermen's Association of the Northern Territory (AFANT)	AFANT is the peak body representing NT recreational fishers whose interests may intersect the EP EMBA.	

Relevant Person Category	Summary of Relevance	
Recfishwest	Peak WA body representing the interests of the recreational fishing sector.	
Industry Associations – tourism		
Assosiasaun Turizmu Maritima iha Timor-Leste	The Association is a registered, national industry body that represents the marine tourism sector in Timor-Leste.	
Kimberley Marine Tourism Association	Based in Broome, the Kimberley Marine Tourism Association represents charter boat operators from the Kimberley and wider region	
Marine Tourism WA	Marine Tourism WA is the peak body representing WA charter boat owners and operators.	
Northern Territory Guided Fishing Industry Association (NTGFIA)	NTGFIA is the peak body responsible for promoting, developing, and maintaining the guided fishing industry in the NT. It represents professional fishing guides and operators. Interests may intersect the EP EMBA.	
Tourism Top End	Tourism Top End is the Regional Tourism Association, a non-profit entity serving businesses, individuals and organisations involved in tourism activities in the NT. Interests may intersect the EP EMBA.	
WA Game Fishing Association	Coordinates game fishing activities throughout Western Australia.	
Industry Associations – local industry		
Chamber of Commerce Northern Territory	Regional representative organisation representing the interests of local business.	
Infrastructure operators		
BW Digital	BW Digital is a privately-owned, carrier-neutral and innovative to deliver optimal customer service. It develops, builds and operates a digital ecosystem, specialising in data transport, compute and storage to connect countries across oceans sustainably.	
Darwin Port	Private consortium responsible for the management of shipping and other commercial activities requiring use of Darwin Harbour. Santos-contracted vessels plan to use Darwin Harbour.	
NT Port and Marine	Private consortium that owns and operates the commercial port at Port Melville on the Tiwi Islands.	
Power and Water Corporation (NT)	Power and Water Corporation is a government-owned corporation responsible for the transmission and distribution of electricity and provision of water and sewerage services across the NT. The Corporation's main operating facility relies on the water quality in Darwin Harbour.	
Sun Cable	Privately-owned consortium with plans to install a new submarine cable infrastructure in NT and Commonwealth waters in the EP EMBA.	
Telstra	Telstra Group Limited is an Australian telecommunications company that builds and operates telecommunications networks and markets related products and services.	
Vocus	Operator of the following infrastructure, which is in the EP EMBA: Darwin-Jakarta-Singapore Cable and North West Cable System (NWCS).	
Local Government Authorities – NT		
Belyuen Community Government Council	Represents the Belyuen Community, located approximately 120km from Darwin on the Cox Peninsula. The Council's area includes NT coastline within the EP EMBA.	

Relevant Person Category	Summary of Relevance
City of Darwin	The City of Darwin includes the central business district of the capital, Darwin City, and represents two-thirds of its metropolitan population. The Council's area includes NT coastline within the EP EMBA.
City Of Palmerston Council	The City of Palmerston Council contains the suburbs of Darwin's satellite city, Palmerston, and is situated between the outer industrial areas of Darwin and the rural areas of Howard Springs. The Council's area includes NT coastline within the EP EMBA.
East Arnhem Regional Council	East Arnhem Regional Council services the communities of Milingimbi, Ramingining, Galiwin'ku, Gapuwiyak, Yirrkala, Gunyangara, Umbakumba, Angurugu and Milyakburra. The Council's area includes NT coastline within the EP EMBA.
Litchfield Council	Litchfield Council represents people living in some of Darwin's outer rural suburbs. The Council's area includes NT coastline within the EP EMBA.
Roper Gulf Regional Council	Roper Gulf Regional Council services the communities of Mataranka, Yugul Mangi, Numbulwar Numburindi, Borroloola, Nyirranggulung and Jilkminggan as well as a large amount of unincorporated land in the Gulf, Roper Valley, Stuart Plateau and Southern Arnhem Land. The Council's area includes NT coastline within the EP EMBA.
Tiwi Islands Regional Council	The Tiwi Islands Regional Council provides a range of local government and other services to Bathurst and Melville Islands and the communities of Wurrumiyanga, Wurankuwu, Milikapiti (Snake Bay) and Pirlangimpi (Garden Point), as well as several smaller outstations. The Council's area includes NT coastline within the EP EMBA.
Victoria Daly Regional Council	Victoria Daly Regional Council services the communities of Nauiyu/Daly River, Pine Creek, Timber Creek, Yarralin Walangeri and Kalgkarindji Daguragu. The Council's area includes NT coastline within the EP EMBA.
Wagait Shire Council	The Wagait Shire Council services community on the Cox Peninsula west of Darwin. The Council's area includes NT coastline within the EP EMBA.
West Arnhem Regional Council	West Arnhem Regional Council services the communities of Gunbalanya, Jabiru, Maningrida, Minjilang, Warruwi, as well as outstations. The Council's area includes NT coastline within the EP EMBA.
West Daly Regional Council	West Daly Regional Council services the communities of Wadeye, Palumpi and Peppimenarti. The Council's area includes NT coastline within the EP EMBA.
Tourism Operators – Timor-Leste	
Dreamers Dive Academy Timor	According to its website, the Dreamers Dive Academy is a tourism and diver training business operating from a base near Dili on the north shore of Timor-Leste. Diving activity is undertaken around Atauro Island in locations that may be within or transit the EP EMBA.
Tourism Operators – NT	
Darwin and Tiwi Islands-based operators	Marine tourism operators active within the EP EMBA are listed in Table 4-22.
Other Relevant Persons	
Autoridade Nacional do Petróleo – Timor-Leste (ANP)	ANP is a public institution established by the Timor-Leste Government to manage and regulate petroleum activities in the Timor-Leste area.



### 4.7.5 Consultation planning, preliminary and consultation activities

Santos acknowledges that consultation processes need to have sufficient flexibility to adapt to the "nature of the interests of the Relevant Persons"<sup>38</sup>.

In planning the consultation program for the Coastal Waters OEMP and Barossa Production Operations EP Santos was initially informed by its previous experience in consulting with Relevant Persons about Barossa Project Activities (refer Section 4.1)

For the Coastal Waters OEMP and Barossa Production Operations EP consultation activities were generally undertaken in three broad phases<sup>39</sup>:

- + Preliminary consultation (9 February 10 March 2024) this included:
  - activities to allow authorities, persons and organisations opportunities to self-identify as Relevant Persons and provide feedback about consultation methods and information needs via a portal and form available on its website. [refer to Section 4.7.4]; and
  - directly contacting Relevant Persons and potential Relevant Persons to:
    - inform them about the consultation for the Coastal Waters OEMP and Barossa Production Operations EP, including sharing information about the OEMP and EP activity and associated environmental impacts and risks;
    - seek information to better understand if the person contacted was from a relevant government Department or agency, or was a person (or organisation) whose functions, interests or activities may be affected by the activities proposed to be carried out under the Coastal Waters OEMP and Barossa Production Operations EP; and, if so, the nature of those potentially affected functions, interests or activities; and to
    - share information about titleholder responsibilities and opportunities to provide guidance for consultation expectations.
- Formal consultation (11 March 2024 23 May 2024<sup>40</sup>) this included seeking feedback from Relevant Persons to inform development of the Coastal Waters OEMP and Barossa Production Operations EP during or following exchanges that involved the provision of sufficient information and a reasonable period of time (refer to Section 4.7.6).
- Further consultation (23 May 2024 until submission to the regulator) Santos undertook consultation with some authorities, persons and organisations following the formal consultation phase given existing relationships, consultation preferences and standing meeting and consultation arrangements.

Santos offered and provided information in different formats and via a range of different mediums.

Preferences expressed by Relevant Persons regarding design of the consultation process were considered and accommodated by Santos, where reasonably practicable and appropriate. This approach has included:

- Providing Relevant Persons access to information using different mediums and platforms both at the request of Relevant Persons and of its own volition, having regard to the nature of particular Relevant Persons and their potentially affected functions, interests or activities;
- Consultation methods and platforms including by telephone, email, letters, website, electronic materials including power point presentations, video content, in person and virtual meetings. Santos provided a toll free 1800 number and a dedicated email address for Relevant Person input and feedback;
- Making information about the proposed activities to be managed under the Coastal Waters OEMP and Barossa Production Operations EP available on the Santos website at <u>www.santos.com/offshoreconsultation</u>. Provision of hyperlinks or QR codes to this website were included in consultation emails and in advertising in print media and on social media;
- Recognising NTSC's feedback that information should be provided via post direct to relevant licence holders in addition to being provided to the NTSC which consults directly with the chairs of each fishery;

<sup>&</sup>lt;sup>38</sup> Santos NA Barossa Pty Ltd v Tipakalippa [2022] FCAFC 193 at paragraph [104]

<sup>&</sup>lt;sup>39</sup> In some cases, contact with persons or organisations identified at later stages in the consultation process did not strictly align with the typical phases as set out below.

<sup>&</sup>lt;sup>40</sup> Although initial consultation correspondence advised that the consultation period would close on 9 April 2024, Santos subsequently sent further correspondence providing updated information to account for an additional risk associated with the activities proposed in this OEMP and the EP and extending the consultation period until 16-23 May 2024.



- Recognising NPFI's feedback that it will pass along any information to its members where required and relevant, acknowledging NPFI has advised there is no need for Santos to directly engage with its members;
- Recognising AFANT's feedback that it will respond on an Association level and pass along any information to its members where required and relevant for their own individual feedback;
- Recognising DFAT's feedback to contact DFAT's Indonesian and Timor-Leste desks on consultation matters relevant to Indonesia and Timor-Leste respectively;
- Continuing to respect direction from Tiwi Islands clans and individuals on appropriate consultation methods, which have for some time been a mutually agreed approach to support consultation for other environmental approvals for the Barossa Gas Project, which have been accepted or are under assessment by respective Regulators; and
- Consulting with First Nations communities via consultative committees, or other representative bodies where Santos understands this to have been culturally appropriate (see Section 4.7.5.1). This included consulting with Wulna Clan through the Jindiwi Consultative Committee and Algada Clans through the Cobourg Peninsula Consultative Committee in line with preferences expressed by each of these clans during the course of engagement relating to consultation for the Barossa Darwin Pipeline Duplication Environment Plan.

All authorities, persons and organisations engaged during the preliminary consultation and consultation phases were provided a link to the *NOPSEMA community information brochure: Consultation on offshore petroleum environment plans* and/or had hard copies of the brochure made available during in-person consultation sessions.

Santos also informed each Relevant Person that they may request that particular information they provide during the consultation not be published and that information subject to such a request will not be published under the relevant regulations (and will instead be included in a separate report which will not be published on NOPSEMA's website).

Typically, where Santos did not receive a response from a Relevant Person to its correspondence and/or in person conversations, follow-up attempts were made (usually using different mediums e.g. phone, email or letter) including to confirm receipt of emails/letters and/or to prompt provision of a response. Most cases involved multiple follow-up attempts.

### 4.7.5.1 Consultation with First Nations people and groups

For the Coastal Waters OEMP and Barossa Production Operations EP,, Santos has provided consultation opportunities and supporting information to First Nations communities, clans and groups, representative organisations and other First Nations organisations listed in Table 4-8, acknowledging the use of a highly conservative EP EMBA (as described in Section 3) for the purpose of assisting to identify potentially Relevant Persons.

This conservative approach has ensured a very broad capture of potential Relevant Persons and provided them an opportunity to provide input if they feel their functions, interests or activities may be impacted. The consultation process is further explained below and includes, if and where applicable:

- Consulting First Nations people through existing representative organisations, including Registered Native Title Bodies Corporate, groups associated with Native Title Determinations and groups in active Native Title Claims, Native Title Representative Bodies, and groups who may be parties to Indigenous Protected Areas, or be named in ILUAs;
- Consulting First Nations people through existing liaison committees or reference groups that have been established between Native Title Parties, Native Title Representative Bodies and industry/government;
- Supporting the establishment of First Nations consultative committees or groups that are intended to be
  representative and able to speak on behalf communities where formal structures do not exist and
  consulting such committees or groups;
- Working with First Nations communities, groups and clans to develop culturally appropriate consultation methods reflecting the information needs of First Nations communities, groups and clans. By way of example, Santos held multiple community consultations with Tiwi people at the community's request for previous Barossa EP consultation; and
- Consulting other First Nations organisations that may support specific interests of First Nations people, such as economic development and community well-being.

In addition, Santos has undertaken a range of activities to promote opportunities for First Nations people to provide input and feedback during consultation to support identification and evaluation of environmental impacts and risks for proposed activities and develop appropriate measures to reduce these impacts and risks to ALARP and to an acceptable level.



These activities included a public information campaign using a range of appropriate media, including, radio, print media, targeted social media, drop-in meetings with information about the project activities, provision of consultation materials and availability of Santos staff to answer questions at its central Darwin shopfront and inviting people to self-identify as a Relevant Person in response, where they considered to have interests, functions or activities that may be affected by the planned activities.

Details of the public information campaign for the Coastal Waters OEMP and Barossa Production Operations EP, are included in Section 4.7.4, Table 4-8 and Table 4-10, which outlines advertising and notifications targeting Tiwi and Larrakia clans/communities.

### 4.7.5.1.1 Consultation with existing representative organisations

For the Coastal Waters OEMP and Barossa Production Operations EP, consultation effort in the NT with existing representative organisations has focused on providing input and feedback opportunities for the NLC, TLC, LNAC and WPDLC.

Consultation effort in WA with existing representative organisations has focused on providing input and feedback opportunities for the KLC and six PBCs.

The EMBA modelled for the Barossa Production Operations EP intersects the Kimberley representative Aboriginal/Torres Strait Islander body (RATSIB) area (refer Figure 3-16Figure 3-16) As a RATSIB, the KLC has responsibility for providing services to native title parties in the Kimberley.

While both the Barossa Production Operations EP EMBA and the Coastal Waters OEMP EMBA do not intersect the native title interests of PBCs in the Kimberley region, Santos as a precautionary approach consulted six PBCs given their responsibilities under the *Native Title Act 1993* (Cth) for representing Native Title holders who may have a particular interest in the activity or knowledge that could assist with the consideration of management of environmental impacts and risks.

Santos recognises that native title rights and interests are held by PBCs on behalf of the native title group they represent and reflect the traditional laws and customs of the native title group. These rights and interests may include, among other things, management and protection of cultural values.

#### Wickham Point Deed Liaison Committee

A key mechanism for ongoing consultation by Santos with the Larrakia people is through the Wickham Point Deed Liaison Committee (previously title Wickham Point Deed Reference Group) which includes representation of Larrakia family groups. The Wickham Point Deed was entered into between DLNG and the NLC (which is also identified as a Relevant Person in Table 4-9) on 29 April 1999 and the liaison committee represents a long-running dialogue between Santos and Larrakia families.

Santos coordinates quarterly Wickham Point Deed Liaison Committee meetings and the functions of the committee include making recommendations to Santos on various matters such as environmental matters, cultural heritage, employment and business opportunities.

Santos has discussed the Barossa Gas Project with the Wickham Point Deed Liaison Committee as a regular agenda item for several years, including providing information on Project activities, approval requirements, impacts and risks, the AAPA Authority Certificate process and proposed management measures.

The Wickham Point Deed liaison committee was identified as a Relevant Person for consultation with respect to activities within the Coastal Waters OEMP and Barossa Production Operations EP (refer Table 4-9), with a consultation session held on 7 March 2024 covering the following:

- Recapitulation of the Barossa Project to include a project update on existing activities and Project progress;
- Recapitulation of the regulatory consultation processes and privacy considerations;
- Informing the committee that Relevant Persons may request that particular information they provide during the consultation not be published and that information subject to such a request will not be published under the relevant regulations (and will instead be included in a separate report which will not be published on NOPSEMA's website);
- Discussing the activities covered by the Coastal Waters OEMP and Barossa Production Operations EP, including installation steps and vessel descriptions
- Introducing and discussing the potential environmental impacts and risks involved with the planned activities and planned controls to manage those risks;
- Introducing and discussing the EP EMBA in the event of an unplanned event, the risks and planned controls to manage those risks;



- Providing opportunities for input, including in relation to potential cultural or other environmental impacts and risk of the activities under this OEMP and the EP and proposed control measures; and
- Responding to and closing out any outstanding matters including questions, issues or concerns.

#### Larrakia Nation Aboriginal Corporation

Outside of the Wickham Point Deed Liaison Committee, Santos also sought to consult with the Larrakia Nation Aboriginal Corporation (LNAC). The summary of Santos' consultation efforts is presented in Table 4-18.

The Larrakia Nation Aboriginal Corporation (LNAC) was set up in 1997 through the Northern Land Council to provide a corporate identity for Larrakia people to uphold Native Title claims. The LNAC has grown to represent the Traditional Owners of the Darwin region and to speak on behalf of Larrakia people while delivering community and outreach services to the broader Darwin community.

#### 4.7.5.1.2 Consultation with Larrakia people

For the Coastal Waters OEMP and Barossa Production Operations EP, Santos adopted a multi-faceted approach to providing input and feedback opportunities for Larrakia people. Approximately 2000 Larrakia people live in the urban environment of Darwin and comprise eight family groups.

Consultation effort with Larrakia people focused on providing consultation opportunities through:

- existing representative organisations with a dedicated Larrakia focus, these being NLC, WPDLC and LNAC;
- face-to-face Larrakia People Consultation Sessions coordinated by Santos;
- provision of the Barossa Production Operations Information Booklet, playing of the Barossa project video on a large screen monitor and availability of Santos' staff to answer questions at its central Darwin shopfront;
- Santos Darwin LNG (DLNG) facility site tours;
- other First Nations organisations that support specific interests of First Nations people, such as economic development, these being LDC and GDA; and
- the public awareness campaign.

In addition to the above activities, Santos also met with the LDC to better understand a proposal for the development of a consultation framework allowing project proponents to consult all Larrakia families though the LDC, including the provision of additional and independent advice where needed to meet the information needs of Larrakia people. Santos understands that the framework is still under development and has not yet received confirmation that the framework is supported by all Larrakia family groups.

As such, the framework was not in place for the purpose of consultation for this OEMP. In addition, during Santos' consultation with Larakia people and Larrakia representative bodies, Santos was not advised that the Larrakia people want the LDC to represent them for the purpose of consultation for the Coastal Waters OEMP and Barossa Production Operations EP

Santos will continue to engage with LDC on their proposed consultation framework and if it is prepared and supported by all Larrakia family groups, will have regard to it in relation to ongoing consultation over the life of the Barossa project and for future proposals.

Santos' future consideration of the framework will include consideration of appropriate financial recompense for meeting attendance and input, and consideration of any financial assistance and/or access to independent advice that the framework may propose.

#### Larrakia People Consultation Sessions

To expand the opportunity to reach more Larrakia People outside of the families represented on the Wickham Point Deed Liaison Committee, Santos, has, in consultation unrelated to the Coastal Waters OEMP and Barossa Production Operations EP, requested and received assistance from LNAC.

Santos previously received advice from LNAC on the best way to directly consult with Larrakia People in a culturally sensitive and appropriate way. The consultation process comprised:

- Santos to undertake face-to-face consultation;
- Santos to advertise in the NT News the face-to-face consultation sessions; and
- LNAC to promote the consultation sessions on their social media.



In response, Santos implemented the following tailored consultation approach for Larrakia people:

- A total of four individual Larrakia consultation sessions were held specifically for Larrakia People between April and June 2024, including:
  - o face-to-face consultation sessions, which were held on 23 April 2024 and 12 June 2024
  - two time slots (during and after work hours) were provided on each date to maximise opportunity to attend. There were no Larrakia attendees at the 12 June 2024 session during work hours.
  - the April consultation was held in Darwin CBD however feedback from Larrakia at this session was for future consultations to be held at more accessible and convenient location
  - o the June session was held in the northern suburb of Malak in a community hall with free parking
- Consultation sessions were advertised as described in Table 4-10
- Santos requested LNAC (confirmed) to post the Notice of Consultation on social media channels (Facebook and LinkedIn) for the April and June 2024 Larrakia consultation sessions.

Hardcopy consultation materials were produced and distributed or made available prior to the start of the session for use as a tool to refer to during the consultation session. Material included, but was not limited to, copies of the Production Operations Information Booklet, Santos Privacy Statement and the NOPSEMA community information brochure: Consultation on offshore petroleum environment plans.

Other visual aids such as AO poster sized maps of the project area and EP EMBA, and AO sized posters with photos and images taken as part of the Barossa Project were positioned at each venue to present information regarding operational activity and the project more generally.

Santos also informed attendees that Relevant Persons may request that particular information they provide during the consultation not be published and that information subject to such a request will not be published under the relevant regulations (and will instead be included in a separate report which will not be published on NOPSEMA's website).

The consultation sessions were conducted as an open forum. People were encouraged to ask questions and raise concerns through the presentation. Santos' representatives and subject matter experts (SMEs) explained the activity and associated environmental risks and impacts during the face-to-face presentations, assisted by video content, and PowerPoint slides.

During the sessions Santos provided opportunities for input, including in relation to potential cultural or other environmental impacts and risk of the activities under the Coastal Waters OEMP and Barossa Production Operations EP and proposed control measures. Santos responded to questions and at the completion of each session, Santos' SMEs were also available to speak with individuals who had additional questions, concerns or wished to discuss matters that may not have been raised during the open forum. Table 4-18 includes a chronology of consultation with Larrakia people.

Through this approach Santos has provided sufficient opportunities and a reasonable period of time for Larrakia People to be consulted for the Coastal Waters OEMP and Barossa Production Operations EP and in accordance with the advice provided by LNAC.

#### Darwin shopfront and DLNG site tours

As part of its broader outreach program, Santos operates a shopfront office in central Darwin where members of the public can learn more about Santos' current and proposed activities. The shopfront is open Monday to Thursday, 10am to 3pm and provides resources and consultation materials and Santos' people are available to answer questions on Santos' activities.

During the development of the Barossa Gas Project more than 120 people have visited the shop front, including Larrakia people.

Similarly, Santos has conducted site tours of its DLNG facilities specially for Larrakia people, providing opportunities for Larrakia people to learn more about Santos' activities and ask questions about proposed activities, such as the Barossa Gas Project.

#### Public awareness campaign

The public awareness campaign used a range of appropriate media, including, radio, print media, targeted social media and drop-in sessions through which information about the proposed activities was made available. Table 4-8 and Table 4-10, which outlines advertising and notifications targeting Larrakia people.



### 4.7.5.1.3 Consultation with Tiwi Islands clans and individuals

As a result of specific requests and feedback as to the consultation process and consultation preferences, which Tiwi People have historically expressed during previous consultation on Barossa Project EPs, Santos implemented the following tailored and culturally considered consultation approach for Tiwi people:

- A total of 16 individual clan consultation sessions were held for Tiwi people from March to May 2024;
- Consultation activities were conducted face-to-face in the form of clan sessions held at two locations on Bathurst Island, and one location on Melville Island, collectively known hereon in as the Tiwi Islands;
- Scheduling of consultation sessions was undertaken in conjunction with Clan Trustees/Traditional Owners, the Tiwi Land Council, Tiwi Resourses and Tiwi Enterprises to ensure no clashes with community events, cultural ceremony or "Sorry Business";
- Ensuring appropriate permissions from the Tiwi Land Council were obtained to allow Santos' personnel to visit the Tiwi Islands;
- Clan sessions were scheduled with approximately four weeks' prior written notice (see Table 4-10), ensuring community members were provided sufficient notice. Santos promoted the sessions via public notices at town stores and the Tiwi Islands Facebook page notice board. Santos employees also received and responded to phone calls and drop ins at the Darwin shop front from Tiwi people asking for details about consultation sessions;
- Tiwi Island Consultation sessions were on the following dates: 5–7 March 2024, 8-10 April 2024, 13 May 2024, 15-17 May 2024 and 21-22 May 2024;
- In addition to the above clan specific sessions, consultation sessions were also conducted in Darwin to accommodate Tiwi people who were unable to attend the Tiwi Island based sessions on 22 March 2024 and 8 April 2024. These sessions were open to whomever identified as being a representative from the Tiwi Islands;
- A Welcome/ Acknowledgement to Country was performed at each consultation session by the appropriate senior Clan Traditional Owner/Elder/Senior. This person also opened meetings in language, thanked people for their attendance and encouraged attendees to listen, ask questions and provide feedback;
- Consultation sessions were arranged for clans independent of one another and at a location convenient for that clan. Nevertheless, the attendance and representation at each designated clan session varied for a multitude of reasons. At times clans came together in one meeting in entirety and/or, there was diversified clan representation. Where clans came together in entirety, this was with the agreeance of the clans' trustees. For the latter, this was managed between the individuals present;
- Consultation sessions were attended by qualified interpreters; and
- Consultation sessions were attended by third-party Tiwi Cultural Advisers (individuals and organisations) to Santos who provided advice to Santos to ensure meetings were conducted in a culturally appropriate and respectful way, as well as to provide interpretation support where needed.

On two occasions Santos rescheduled consultation sessions to accommodate 'Sorry Business' on the Tiwi Islands at the request of the impacted clans. On these occasions Santos liaised with the appropriate clan representatives to reschedule the session to an alternate and acceptable date.

Three rounds of meetings, not including the Darwin-based sessions, were held with each individual clan group with the aims of:

- recapitulation of the Barossa Project to include a project update on existing activities and Project progress;
- recapitulation of the regulatory consultation processes and privacy considerations;
- informing attendees that Relevant Persons may request that particular information they provide during the consultation not be published and that information subject to such a request will not be published under the relevant regulations (and will instead be included in a separate report which will not be published on NOPSEMA's website);
- discussing the activities covered by the Coastal Waters OEMP and Barossa Production Operations EP, including installation steps and vessel descriptions;
- introducing and discussing the potential environmental impacts and risks involved with the planned activities and planned controls to manage those risks;
- introducing and discussing the EP EMBA in the event of an unplanned event, the risks and planned controls to manage those risks;



- providing opportunities for community input including in relation to potential cultural or other environmental impacts and risk of the activities under the Coastal Waters OEMP and Barossa Production Operations EP and proposed control measures; and
- responding to and closing out any outstanding matters including questions, issues or concerns.

Hard copy consultation materials were produced and distributed or made available prior to the start of the session to support informed discussion during the consultation session. Material included, but was not limited to, copies of the Production Operations Information Booklet, Santos Privacy Statement and the NOPSEMA community information brochure: Consultation on offshore petroleum environment plans.

Santos also presented information using videos and displayed other visual aids at each venue to present information regarding operational activity and the project more generally, such as iPads with images and additional materials for distribution during meetings and AO-sized maps of the project area and EP EMBA and AO-sized posters with photos and images featuring Barossa Project activities.

The consultation sessions were conducted as an open forum. People were encouraged to ask questions and raise concerns through the presentation. Santos' representatives and subject matter experts (SMEs) explained the activity and associated environmental risks and impacts during the face-to-face presentations, assisted by video content and PowerPoint slides.

During the sessions Santos responded to questions where appropriate. If a matter was raised that required additional information, this was taken on notice.

At the end of each session, Santos SMEs were also available to speak with individuals who had additional questions, concerns or wished to discuss matters that may not have been raised during the open forum.

Matters raised at meetings were captured and responded to in several ways, pending the forum in which it was raised and the nature of the discussion.

For example, open forum matters that were of interest to wider clan representatives were populated into a table and provided at subsequent meetings during the main presentation. Where appropriate, frequently asked questions (FAQ) documents in response to matters were prepared and distributed or made available at subsequent sessions.

Other confidential and/or individually specific matters were managed either in person at subsequent meetings, or via an emailed response pending the nature of the request, and the request of the questionee.

Consultation sessions for Tiwi people were notified and advertised as set out in Table 4-10 which provides a chronology of consultation with Tiwi Islands clans.

Tiwi people have also utilised the Santos shopfront in central Darwin and participated in site tours of DLNG specifically for Tiwi people, providing an opportunity for them to learn more about Santos' activities and ask questions about proposed activities, such as the Barossa Gas Project.

#### 4.7.5.1.4 Consultation with First Nations Consultative Committees

Santos notes that there are remote areas of coastal Northern Australia where formal mechanisms for consultation are few or non-existent.

To support consultation in these areas for the Coastal Waters OEMP and Barossa Production Operations EP, Santos engaged a consultant to support the establishment of First Nations Consultative Committees (FNCCs) with the intention that these be self-nominating and self-governing, and independent of government or industry. The intended purpose of these committees is to provide a forum to allow for culturally appropriate consultation with First Nations peoples represented through FNCCs, and to serve as a means for those peoples to provide feedback to third parties on matters on which the FNCC is consulted.

The FNCC establishment process is led by cultural advisors, comprising a team of First Nations leaders with extensive knowledge and experience in relation to First Nations cultures of Northern Australia, and who possess deep cultural connections to First Nations peoples of the region.

The FNCC establishment process commences with the identification by the cultural advisers of First Nations clans and associated persons who may have functions, interests or activities that may be affected by activities Santos proposes to carry out under an environment plan.

The cultural advisors then contact the identified First Nations persons to discuss the FNCC concept. Santos understands that this includes meetings with Elders and other First Nations leaders who speak for coastal and sea country that may be affected by project activities. Where an interest to participate in the FNCC process is expressed, the cultural advisers support the relevant clan group to establish their own FNCC and to self-determine its functions and operations, including in relation to committee membership, leadership and governance arrangements and desired level and method of consultation.

This process involves the cultural advisors sharing knowledge and experience in relation to their participation on established committees and supporting the identified clan members to determine their own rules and processes for



committee decision-making, membership and the nomination of chairs. Once determined, these matters are formally documented in charters adopted by the FNCCs. Santos has been provided with copies of charters of FNCCs consulted for the Coastal Waters OEMP and Barossa Production Operations EP, which include details about the FNCCs' purposes, membership and procedures.

Once established, and subject to the wishes of FNCC members, the external cultural advisors may provide ongoing support to the FNCCs, including administrative and advisory services. Santos engaged a consultant to support FNCC establishment and operations. This consultant maintains regular contact with FNCCs and Clan groups to facilitate Santos' consultation with these groups.

The activities of these committees are complementary to the functions and responsibilities of representative organisations, such as Land Councils or other formal bodies, with the intention that they be in a position to represent First Nations peoples. FNCC activities are understood to include disseminating consultation information to First Nations community members of relevance.

Santos acknowledges the establishment and operation of these committees in response to the growing need for a means for First Nations voices to be heard and considered. This need is particularly relevant along the NT coastline where formal consultative mechanisms are typically not in existence, in contrast with the WA coastline Prescribed Bodies Corporate (PBCs) which provide an avenue for consulting First Nations people who have been recognised by Australian law as holding rights and interests to traditional land and waters.

Santos has consulted FNCCs with representative functions across the EP EMBA for this Activity, providing a broad coverage of any potential sea country interests within the EP EMBA. Eleven FNCCs were consulted in the preparation of the Coastal Waters OEMP and Barossa Production Operations EP given the expanded geographical coverage of the spill EP EMBA. Consultation sessions for FNCC members were notified via provision of a specific meeting invitation. Table 4-18 includes a chronology of consultation with FNCCs.

#### 4.7.5.1.5 Consultation with other First Nations organisations

Santos has also consulted other First Nations organisations that support specific interests of First Nations people, such as economic development and community well-being.

These organisations were Aboriginal Sea Company, GDA, Kenbi Rangers, and North Australian Indigenous Land and Sea Management Alliance.

Santos also consulted LDC in its capacity as a commercial organisation that supports positive economic outcomes for Larrakia people.

### 4.7.5.1.6 Advertising and notification of Tiwi and Larrakia Consultation Sessions

Date	Advertising/notice	Description	Reach
For Tiwi Island March/April/May 2024 consultation sessions			
March/April/ May 2024	Social media Notice	Facebook, Tiwi Notice Board Facebook Page	Geotargeted Tiwi Islands – 2,800 members
March/April/May 2024	Notice of Consultation	Emailed to several independent stakeholders for sharing across their direct networks, in person, and for posting on Tiwi Island notices boards	Geotargeted Tiwi Islands – 2,800 members
19 February 2024	Press Ad NT News	Page 19 advertising March sessions	Target NT with reach of 25,000
26 February 2024	Press Ad NT News	Page 6 advertising March sessions	Target NT with reach of 25,000
4 March 2024	Press ad – NT News	Page 6 Advertising March sessions	Target NT with reach of 25,000
23 March 2024	Press ad – NT News	Page 5 Advertising March Community drop-in sessions	Geo-targeted Darwin and surrounding areas (e.g. Burrundie and Kakadu, Tiwi Islands and NT)
26 March 2024	Press ad – NT News	Full page, page 6 Advertising April sessions	Target NT with reach of 25,000
2 April 2024	Press ad – NT News	Full page, page 6	Target NT with reach of 25,000

#### Table 4-10: Advertising and notification of Tiwi and Larrakia Consultation Sessions

Date	Advertising/notice	Description	Reach
		Advertising April sessions	
6 April 2024	Press ad – NT News	Full page, page 12 Advertising April sessions	Target NT with reach of 25,000
8 May 2024	Press ad NT News	Full page, page 8 Advertising May sessions	Targeted NT with reach of 25,000
15 May 2024	Press ad NT News	Full page, page 6 Advertising May sessions	Targeted NT with reach of 25,000
20 May 2024	Press ad NT News	Page 6 advertising May sessions	Targeted NT with reach of 25,000
For Larrakia April/Ju	une 2024 sessions		
23 March 2024	Press ad – NT News	Page 5 advertising March Community drop-in sessions	Targeted NT with reach of 25,000
April	Notice of Consultation	Emails to representative organisations for sharing across their direct networks.	Targeted for Larrakia people
April	Notice of Consultation	Emails and phone calls notifying individual Larrakia family representatives	Targeted for Larrakia people
April	Notice of Consultation	Promotion via Santos' Darwin shop front	Targeted for Larrakia people
April	Notice of Consultation	Larrakia Nation social media advertising including Facebook and LinkedIn	Targeted Larrakia people
June	Notice of Consultation	Emailed to representative organisations for sharing across their direct networks.	Targeted for Larrakia people
June	Notice of Consultation	Promotion via Santos' Darwin shop front.	Targeted for Larrakia people
June	Notice of Consultation	Emails and phone calls notifying individual Larrakia family representatives	Targeted for Larrakia people
1 June 2024	Press ad – NT News	Page 30 advertising for June	Target NT with reach of 25,000
5 June 2024	Press ad – NT News	Page 8 advertising for June	Target NT with reach of 25,000
8 June 2024	Press ad – NT News	Page 21 advertising for June	Target NT with reach of 25,000
June	Notice of Consultation	Larrakia Nation social media advertising including Facebook and LinkedIn	Targeted Larrakia people

### 4.7.6 **Provision of sufficient information**

Having regard to the purpose of consultation (described above at Section 4.7.1), Santos provided Relevant Persons with sufficient information so they can make an informed assessment about the possible consequences of the Activity on their functions, interests or activities. As outlined above at Section 4.2 and for the reasons set out in that section, Santos provided Relevant Persons with information related to both the Coastal Waters OEMP and Barossa Production Operations EP together and consulted comprehensively on both activities. In particular, Santos provided Relevant Persons with information regarding:

- The Activity proposed under the Coastal Waters OEMP and Barossa Production Operations EP;
- The environment that may be affected by the Activity, including depictions of the modelled EP EMBA and explaining how the EP EMBA is determined;
- The potential environmental impacts and risks of the Activity and proposed control measures;
- The environmental approval process;



- The purpose of consultation, who may be a Relevant Person and how to self-nominate as a potential Relevant Person;
- The titleholder's obligations during consultation in the course of preparing an environment plan, including informing Relevant Persons that they can request that particular information they provide during consultation not be published and that information subject to such a request will not be published under the relevant regulations (and will instead be included in a separate report which will not be published on NOPSEMA's website);
- The obligation of the titleholder not to publish particular information if so requested by the Relevant Person; and
- How to provide feedback.

Relevant Persons were provided access to information using different mediums and platforms, including by telephone, email, website (<u>https://www.santos.com/barossa/</u>), hard copy and electronic materials, social media, in person and virtual meetings.

At a minimum, this information was available on the Santos website (including in a specific fact sheet providing an overview of pipeline operations in NT Waters) and was included in the Barossa Production Operations Information Booklet, which Santos typically shared with Relevant Persons by mail, email and/or made available during consultation sessions.<sup>41</sup>

Other examples of the consultation materials used are included in Appendix D and included the following:

- A FAQ document, responding to queries and feedback during consultation with Tiwi People provided as part of the consultation process; and
- For particular Relevant Persons or particular groups of Relevant Persons, videos, animations, PowerPoint slides, photos, and maps to convey technical information to different audiences in a clear and accessible way.
- Facebook page translated into Tetum and Bahasa Indonesian

Santos also sent Relevant Persons (and potential Relevant Persons) links to the NOPSEMA community information brochure, *Consultation on offshore petroleum environment plans* and/or made this available during inperson consultation sessions. This brochure contains information for community members to better understand the responsibilities of titleholders to consult Relevant Persons in the development of environment plans, the purpose of consultation and how Relevant Persons can provide feedback.

### 4.7.6.1 First Nations consultation sessions

In addition to the above, to ensure the information provided to First Nation people was culturally appropriate, for each First Nations consultation session, Santos played a short video explaining the purpose of the session and key information relating to the consultation process, how feedback could be provided, privacy considerations and the option for Relevant Persons to request that particular information they provide during consultation not be published and that information subject to such a request will not be published under the relevant regulations (and will instead be included in a separate report which will not be published on NOPSEMA's website).

Santos made available independent, qualified interpreters via the Aboriginal Interpreter Service to assist in the delivery of consultation sessions where appropriate. Santos also used local community members where qualified interpreters were not available.

Santos' representatives and subject matter experts explained the activity, risks and impacts during in person presentations. To improve accessibility and comprehension, they were assisted by visual aids/photos, maps, videos, animations, and PowerPoint slides to present information regarding the activity and the project more generally (including information of a more technical nature).

After each consultation session, Santos' representatives and subject matter experts were available to answer additional questions or provide further information to clan members and individuals. This offered First Nations people the opportunity to speak to Santos' representatives or subject matter experts one-on-one or in a smaller group setting (based on feedback this was a more comfortable format for some people).

<sup>&</sup>lt;sup>41</sup> Between 8 April to 10 May 2024, Santos shared an updated Barossa Production Operations Information Booklet. This was updated to account for an additional risk associated with the Activity, namely a gas release in the unlikely event of an unplanned pipeline loss of containment. A link to the updated booklet was generally shared with Relevant Persons and potential Relevant Persons via email and/or by making a hard copy available during in-person consultation sessions. Santos sent further correspondence linking the updated booklet, highlighting the nature of the additional information in its subsequent emails and indicated where to find that information in the updated version, Santos typically extended the consultation period by approximately two weeks to allow reasonable time for its consideration and for Relevant Persons to provide any consultation inputs having regard to the additional information.



As mentioned at Section 4.7.6 above Santos also provided information about NOPSEMA's community information brochure, *Consultation on offshore petroleum environment plans* and made the brochure available at consultation sessions, as well as making the Barossa Production Operations Information Booklet available at all consultation sessions.

Further detail on First Nations consultation sessions is provided in Table 4-10 in Section 4.8.8.

### 4.7.7 Reasonable period for consultation

Santos is required to allow a Relevant Person a reasonable period for consultation. In considering what constitutes a reasonable period of time for consultation for each Relevant Person, Santos had regard to the nature, extent and likelihood of the potential impact of the Activity on that person's functions, interests or activities.

Santos has undertaken a comprehensive consultation program for the Barossa Gas Project commencing with the OPP. The OPP has been followed by extensive consultation for each of the activity specific EPs and other regulatory approvals prepared for different stages of the Barossa Gas Project.

For the Coastal Waters OEMP and Barossa Production Operations EP, Santos provided 31 days through a 'preliminary consultation' phase for Relevant Persons to consider consultation information, including that shared via a link to the Barossa Production Operations Information Booklet which contained information about the proposed activities and their potential impacts and risks.

Santos then generally provided approximately 70 days during the 'formal consultation' phase for Relevant Persons to respond with feedback about the proposed activities.

In cases where a different period was provided for consultation, Santos considered this to be reasonable having regard to:

- the nature, extent and likelihood of the potential impact of the Activity on that person's functions, interests or activities; and/or
- Santos' understanding of the Relevant Persons' consultation preferences.

Santos directly contacted Relevant Persons notifying them of the consultation process and formal consultation period. Emails or letters were sent to Relevant Persons to invite feedback for the Coastal Waters OEMP and Barossa Production Operations EP, confirming the date by which feedback was sought and outlining how feedback may be provided. In other cases, one or more meetings were arranged, by agreement with the Relevant Person, for the purposes of the consultation.

Following an approximate one month public awareness campaign during the preliminary consultation period to raise awareness of the Coastal Waters OEMP and Barossa Production Operations EP activity and to seek out Relevant Persons for consultation (Table 4-7), Santos also conducted a public awareness campaign from 11 March 2024 to 9 April 2024, specifically reminding Relevant Persons of the consultation opportunity and seeking feedback from Relevant Persons for the Coastal Waters OEMP and Barossa Production Operations EP (Table 4-7).

Where no comments were received from a Relevant Person, Santos generally followed up the Relevant Person during the formal consultation phase to prompt them to consider the information materials previously provided and/or confirm whether the Relevant Person intended to provide feedback. In some cases, Santos extended the formal consultation period to allow Relevant Persons more time to make an informed assessment of the possible consequences of the proposed activity on their functions, interests or activities. Santos also accepted feedback from Relevant Persons at any time prior to the submission of this OEMP, which was more than 12 months after consultation materials were initially provided to most Relevant Persons.

As outlined in Section 4.7.1, Santos notes that there is no reasonable possibility that planned impacts from the Activity will have any consequences on functions, interests or activities concerning areas at the extremities of the EP EMBA. In addition, the likelihood of the unplanned release is assessed as remote given the mitigation and management controls in place, and the residual risk is considered low. There is an even lower likelihood of an unplanned hydrocarbon release affecting a person's or organisation's functions, interests or activities where these relate to the extremities of the EP EMBA.

While Santos has still consulted Relevant Persons whose functions, interests or activities may only be affected by unplanned events (the likelihood of which is remote), consultation tended to focus more closely on those most proximate to the Operational Area and in respect of whom the period reasonably required for consultation is considered to likely be greater.

Considering the above, Santos considers it has provided a more than reasonable period for consultation.



### 4.8 Consultation report

A summary report including the outcomes of consultation with Relevant Persons, including any objections or claims about the adverse impact of the Activity and Santos' assessment of them, satisfying the requirements of section 24(b)(i)-(iii) of the OPGGS(E)R, is provided in Table 4-11 to Table 4-23. The full records of Relevant Persons consultation, as required by section 24(b)(iv) of the OPGGS(E)R, is provided in the Sensitive Information Report.

As outlined above at Section 4.2 and for the reasons set out in that section, the consultation report captures consultation across both the GEP Coastal Waters OEMP and the Barossa Production Operations EP.

Where objections or claims made during consultation were considered relevant to the Coastal Waters OEMP, sections within the Coastal Waters OEMP and GEP NT Waters OPEP have been referenced within the consultation report (refer Table 4-11 to Table 4-23) for each objection or claim, showing where existing information relevant to that objection or claim is located.

Where Santos has received input from Relevant Persons in consultations undertaken in the course of preparing other environment plans, it has considered and applied that input in the course of preparing the Coastal Waters OEMP and included OEMP references where appropriate.

Where a Santos' response provided to a Relevant Person related only to Production Operations activities covered by the Barossa Production Operations Environment Plan, this is indicated in the table.

In addition to including a statement of Santos' response to objections or claims (per section 24(b)(iii)), a statement of the titleholder's response, or proposed response, if any, to each objection or claim; and a summary of responses to Relevant Persons is also included where appropriate.

The EMBA referenced in Table 4-11 to Table 4-23 is the larger Barossa Production Operations EMBA, unless otherwise stated.

#### 4.8.1 **Commonwealth Government Agency or Authority**

#### Table 4-11: Consultation Summary Table - Commonwealth Government Agency or Authority

Section 25(1)(a) of the OPGGS(E)R: Commonwealth agency or authority to which the activities to be carried out under the environment plan may be relevant

#### Australian Communications and Media Authority (ACMA)

Summary of consultation effort:

- On 9 February 2024 Santos emailed ACMA to advise it of preliminary consultation regarding proposed activities for consultation to be managed under the Coastal Waters OEMP and Barossa Production Operations EP. [Con-3787]
- The email advised that Santos was seeking information to better understand:
  - o if you are from a government Department or agency, how the proposed Production Operations activities may be relevant to your department or agency; and
  - what (if any) functions, interests, or activities you or your organisation have that may be affected by the proposed Production Operations activities.
- The email included information on the regulatory process for the activities in Commonwealth and NT jurisdictions and links to a Santos information booklet on the proposed activities and a NOPSEMA brochure on the consultation process and details of how to contact Santos to register as a Relevant Person. The email stated that the consultation phase would commence on 11 March 2024 and close on 9 April 2024.
- On 11 March 2024 Santos emailed ACMA further to the previous correspondence, to advise that it had commenced the consultation phase which would run until 9 April 2024. In addition to the previous information again being provided, Santos provided information on Relevant Persons' entitlements under the regulatory processes, details of how to provide feedback and a reminder of the closing date for consultation. [Con-3793]
- On 27 March 2024 ACMA emailed Santos and advised that Vocus' North-West Cable System is likely to share a cable crossing with the proposed Gas Export Pipeline and is aware of other proposals to install submarine cables landing in Darwin and ACMA recommends engaging with the owners. ACMA advised there are no submarine cable protection zones declared by the ACMA in the vicinity of Santos' proposed activities and does not require additional consultation. [Con-3795]
- On 27 March 2024 Santos emailed ACMA and confirmed Santos is in ongoing engagement with Vocus, BW Digital, Sun Cable, Telstra and NT Power and Water Corporation. [Con-3796]
- On 7 May 2024, Santos emailed ACMA further to ACMA's response on 27 March 2024, to advise it had extended the consultation period until 21 May 2024. In providing this extension of time, Santos advised that the information in the booklet and factsheet had been updated to account for an additional risk associated with the proposed activity. Santos confirmed that consultation will close on the revised date unless Santos hears otherwise from ACMA. [Con-4137]
- On 10 July 2024 Santos emailed ACMA to advise the consultation period had been completed. Santos advised that it considered consultation had now closed for the purpose of Santos finalising and submitting environment plans for these activities to government regulators for assessment. [Con-5086]
- On 15 October 2024 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that the EP had been submitted to NOPSEMA for assessment, provided a link to the full EP on the NOPSEMA website and Santos' contact details for any queries. The Update also stated that the OEMP would be submitted to the NT-DITT for assessment. [Con-5982]
- On 23 January 2025 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that, following a request from NOPSEMA to provide further information, Santos had re-submitted the EP with some amendments in December 2024 for further assessment. The Update also re-stated that the OEMP would be submitted to the NT-DITT for assessment. [Con-6036]
- No further correspondence or feedback was received from ACMA.

Summary of response by Relevant Person	Assessment of merits	Santos' Response Statement	OEMP Reference
ACMA recommended Santos engage with the owners of any submarine cables (existing or planned) within the OA to discuss the activities.	Santos has considered the matters raised in the OEMP and provided a response. Santos notes ACMA's advice and has consulted with the relevant owners of submarine cables (existing or planned) in preparing this OEMP.	Santos confirmed it was consulting with the relevant owners of submarine cables (existing or planned).	Refer to Table 4-20 (Infrastructure Operators) for consultation with submarine cable owners BW Digital, NT Power and Water Corporation, Sun Cable, Telstra and Vocus.
Australian Eisherica Meneroment Authority (AEI			

#### Summary of consultation effort:

- On 9 February 2024 Santos emailed AFMA to advise it of preliminary consultation regarding proposed activities for consultation to be managed under the Coastal Waters OEMP and Barossa Production Operations EP. [Con-3787]
- The email advised that Santos was seeking information to better understand:
  - o if you are from a government Department or agency, how the proposed Production Operations activities may be relevant to your department or agency; and
  - what (if any) functions, interests or activities you or your organisation have that may be affected by the proposed Production Operations activities.
- The email included information on the regulatory process for the activities in Commonwealth and NT jurisdictions and links to a Santos information booklet on the proposed activities and a NOPSEMA brochure on the consultation process and details of how to contact Santos to register as a Relevant Person. The email stated that the consultation phase would commence on 11 March 2024 and close on 9 April 2024.
- On 11 March 2024 Santos emailed AFMA further to the previous correspondence, to advise that it had commenced the consultation phase which would run until 9 April 2024. In addition to the previous information again being provided, Santos provided information on Relevant Persons' entitlements under the regulatory processes, details of how to provide feedback and a reminder of the closing date for consultation. [Con-3793]
- On 14 March 2024, AFMA emailed Santos and advised that it has no specific comments on the proposal as it lies outside the main area of their fisheries operations and encouraged Santos to engage with state fisheries agencies and operators. [Con-3797]
- On 27 March 2024, Santos emailed AFMA to confirm it is consulting with the following organisations on production operations activities: WA Department of Fisheries, NT Department of Fisheries, WAFIC, NTSC, NPFI, ASBTIA, CFA and licence-holders in each fishery through their representative organisations. [Con-3798]
- On 27 March 2024 AFMA thanked Santos for confirmation of which organisations Santos was consulting with. [Con-3799].
- On 7 May 2024, Santos emailed AFMA further to AFMA's response on 27 March 2024, to advise it had extended the consultation period until 21 May 2024. In providing this extension of time, Santos advised that the information in the booklet and factsheet had been updated to account for an additional risk associated with the proposed activity. Santos confirmed that that consultation will close on the revised date unless Santos hears otherwise from AFMA. [Con-4138]



- On 10 July 2024 Santos emailed AFMA to advise the consultation period had been completed. Santos advised that it considered consultation had now closed for the purpose of Santos finalising and submitting environment plans for these activities to government regulators for assessment. [Con-5086]
- On 15 October 2024 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that the EP had been submitted to NOPSEMA for assessment, provided a link to the full EP on the NOPSEMA website and Santos' contact details for any gueries. The Update also stated that the OEMP would be submitted to the NT-DITT for assessment. [Con- 5982]
- On 23 January 2025 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that, following a request from NOPSEMA to provide further information, Santos had re-submitted the EP with some amendments in December 2024 for further assessment. The Update also re-stated that the OEMP would be submitted to the NT-DITT for assessment. [Con-6036]
- No further correspondence or feedback was received from AFMA.

Summary of response by Relevant Person	Assessment of merits	Santos' Response Statement
AFMA advised Santos to consult directly with commercial fishing industry stakeholders, including via representative organisations.	Santos notes AFMA's advice and has consulted with relevant commercial fishing industry stakeholders in preparing the OEMP.	Santos confirmed it was consulting with relevant commercial fishing industry stakeholde
Australian Hydrographic Office (AHO)		

Summary of consultation effort:

- On 9 February 2024 Santos emailed AHO to advise it of preliminary consultation regarding proposed activities for consultation to be managed under the Coastal Waters OEMP and Barossa Production Operations EP. [Con-3787]
- The email advised that Santos was seeking information to better understand:
  - o if you are from a government Department or agency, how the proposed Production Operations activities may be relevant to your department or agency; and
  - what (if any) functions, interests or activities you or your organisation have that may be affected by the proposed Production Operations activities.
- The email included information on the regulatory process for the activities in Commonwealth and NT jurisdictions and links to a Santos information booklet on the proposed activities and a NOPSEMA brochure on the consultation process and details of how to contact Santos to register as a Relevant Person. The email stated that the consultation phase would commence on 11 March 2024 and close on 9 April 2024.
- On 12 February 2024, AHO provided an acknowledgement to Santos that the email has been received and the data will now be registered, assessed, prioritised and validated. AHO advised that standards may result in some data generalisation or filtering due to the scale of existing charts, proximity to other features, and the level of risk a reported feature presents to mariners. [Con-3800]
- On 11 March 2024 Santos emailed AHO further to the previous correspondence, to advise that it had commenced the consultation phase which would run until 9 April 2024. In addition to the previous information again being provided, Santos provided information on Relevant Persons' entitlements under the regulatory processes, details of how to provide feedback and a reminder of the closing date for consultation. [Con-3793]
- On 13 March 2024. AHO provided an acknowledgement to Santos that the email has been received and the data will now be registered, assessed, prioritised and validated, [Con-3801]
- On 7 May 2024, Santos emailed AHO further to emails sent previously, to advise it had extended the consultation period until 21 May 2024. In providing this extension of time, Santos advised that the information in the booklet and factsheet had been updated to account for an additional risk associated with the proposed activity. Santos confirmed that consultation will close on the revised date unless Santos hears otherwise from AHO. [Con-4139]
- On 8 May 2024, AHO provided an acknowledgement to Santos that the email has been received and the data will now be registered, assessed, prioritised and validated. [Con-4141]
- On 10 July 2024 Santos emailed AHO to advise the consultation period had been completed. Santos advised that it considered consultation had now closed for the purpose of Santos finalising and submitting environment plans for these activities to government regulators for assessment. [Con-5086]
- On 23 August 2024 Santos phoned AHO and followed-up with an email on 28 August 2024 advising that, in the absence of any specific response from AHO, Santos has reverted to the standard advice provided by AHO in response to requests for feedback during consultation on other Barossa EPs. In the email Santos provided details of the DAFF information being included in the OEMP and EP and requested any further input by 9 September 2024. [Con-5609]
- On 5 September 2024 AHO responded to Santos' email of 23 August 2024. AHO advised it had no further comment other than requesting that the final positions of any permanent features are sent to the AHO for charting action. [Con-5640] Santos responded via email the same day stating the AHO's charting requirements would be cited in the relevant EPs. [Con-5641]
- On 15 October 2024 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that the EP had been submitted to NOPSEMA for assessment, provided a link to the full EP on the NOPSEMA website and Santos' contact details for any gueries. The Update also stated that the OEMP would be submitted to the NT-DITT for assessment. [Con- 5982]
- On 23 January 2025 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that, following a request from NOPSEMA to provide further information, Santos had re-submitted the EP with some amendments in December 2024 for further assessment. The Update also re-stated that the OEMP would be submitted to the NT-DITT for assessment. [Con-6036]
- No further correspondence or feedback was received from AHO.

Summary of response by Relevant Person	Assessment of merits	Santos' Response Statement	OEMP Reference
Other than its standard advice with respect to maritime safety matters, the AHO advised it had no other comments that would value add to the activity. The AHO requested that, once the activity is fully complete, the final positions of any permanent features are sent to the AHO for charting action.	Santos has followed and actioned the standard advice provided by AHO and AMSA for every EP with respect to maritime safety matters. Santos has considered and applied this standard advice to this OEMP, including activity notifications. Santos notes that activities under the Coastal Waters OEMP do not involve the placement of any permanent features. Santos considers Section 25 consultation requirements to have been met.	<ul> <li>Santos will include all formal notification requirements in the relevant sections of this OEMP, specifically the following:         <ul> <li>Requirement to notify the AHO through datacentre@hydro.gov.au no less than 4 working weeks before operations commence for the promulgation of related notices to mariners.</li> <li>Requirement to notify AMSA's JRCC through rccaus@amsa.gov.au (Phone: 1800 641 792 or +61 2 6230 6811) for promulgation of radio-navigation warnings 24-48 hours before operations commence.</li> </ul> </li> <li>Santos also acknowledges the following standard AHO advice:</li> </ul>	Notification requirements for AHO are included in Table 8-13 and control measure BAO-CM- 6.6.1.



	OEMP Reference	
rs.	Refer to Table 4-17 (Commercial Fishing (Commonwealth / NT managed) for consultation with licence holders.	
	Refer to Table 4-19 (Industry Associations) for consultation with ASBTIA, CFA, NPFI and NTSC.	
	0	Vessel obligations to comply with the International Rules for Preventing Collisions at Sea (COLREGs), in particular, the use of appropriate lights and shapes to reflect the nature of operations (e.g. restricted in the ability to manoeuvre). Vessels should also ensure their navigation status is set correctly in the ship's AIS unit.
--	----------------------	----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------
	0	Evaluation and implementation of adequate anti-collision measures, including the collision risk mitigation measures cited by AMSA, being additional warnings and/or lights to attract attention and offshore guard vessel/s that can monitor traffic and take early action to alert a vessel approaching the area of operations.
	0	Santos' vessel anti-collision measures are in accordance with COLREGs and AMSA requirements.
	Additiona guard ca	ally, Santos will implement cautionary zones around Project vessels and use surveillance vessel to utionary zones.
	Santos c AHO with	onfirmed in its response to AHO for the Production Operations EP that Santos will also provide the n the final positions of any permanent features for charting action.

## Australian Institute of Marine Science (AIMS)

Summary of consultation effort:

- On 9 February 2024 Santos emailed AIMS to advise it of preliminary consultation regarding proposed activities for consultation to be managed under the Coastal Waters OEMP and Barossa Production Operations EP. [Con-3787]
- The email advised that Santos was seeking information to better understand:
  - o if you are from a government Department or agency, how the proposed Production Operations activities may be relevant to your department or agency; and
  - what (if any) functions, interests or activities you or your organisation have that may be affected by the proposed Production Operations activities.
- The email included information on the regulatory process for the activities in Commonwealth and NT jurisdictions and links to a Santos information booklet on the proposed activities and a NOPSEMA brochure on the consultation process and details of how to contact Santos to register as a Relevant Person. The email stated that the consultation phase would commence on 11 March 2024 and close on 9 April 2024.
- On 21 February 2024 AIMS emailed Santos and advised its schedule was indicative only. However, it has planned voyages around the Goodrich Bank area and enquired if Santos anticipated disruption to these operations. [Con-3802]
- On 28 February 2024 Santos emailed AIMS and advised it was waiting on internal feedback and would respond soon. [Con-3803]
- On 29 February 2024 Santos emailed AIMS and requested coordinates and details of its Goodrich Bank interests. [Con-3804]
- On 1 March 2024 AIMS provided Santos with the information requested. [Con-4120]
- On 5 March 2024 Santos thanked AIMS and advised it would revert back. [Con-4056]
- On 11 March 2024 Santos emailed AIMS further to the previous correspondence, to advise that it had commenced the consultation phase which would run until 9 April 2024. In addition to the previous information again being provided, Santos provided information on Relevant Persons' entitlements under the regulatory processes, details of how to provide feedback and a reminder of the closing date for consultation. [Con-3793]
- On 27 March 2024 Santos emailed AIMS with information on potential activities during dates in 2025, in response to AIMS' emails of 21 February 2024 and 1 March 2024. In the email Santos advised it did not believe there would be any impacts from Barossa activities on any AIMS activities during the advised time periods due to the distance from the nearest Barossa Operational Area. [Con-5153]
- On 2 May 2024 Santos emailed AIMS further to its response in March 2024, to advise it had extended the consultation period until 16 May 2024. In providing this extension of time. Santos advised that the information in the booklet and factsheet has been updated to account for an additional risk associated with the proposed activity. In the email, Santos confirmed that consultation will close on the revised date unless Santos hears otherwise from AIMS. [Con-4219]
- On 3 May 2024 AIMS emailed Santos requesting information on the infrastructure it had in the approaches to Darwin Harbour. The email related to Barossa DPD activities in NT Internal waters and was not relevant to this OEMP. [Con-4945] Santos subsequently responded to AIMS separate to the consultation process for this OEMP.
- On 10 July 2024 Santos emailed AIMS to advise the consultation period had been completed. Santos advised that it considered consultation had now closed for the purpose of Santos finalising and submitting environment plans for these activities to government regulators for assessment. [Con-5086]
- On 15 October 2024 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that the EP had been submitted to NOPSEMA for assessment, provided a link to the full EP on the NOPSEMA website and Santos' contact details for any gueries. The Update also stated that the OEMP would be submitted to the NT-DITT for assessment. [Con- 5982]
- On 23 January 2025 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that, following a request from NOPSEMA to provide further information, Santos had re-submitted the EP with some amendments in December 2024 for further assessment. The Update also re-stated that the OEMP would be submitted to the NT-DITT for assessment. [Con-6036]
- No further correspondence or feedback was received from AIMS.

Summary of response by Relevant Person	Assessment of merits	Santos' Response Statement
----------------------------------------	----------------------	----------------------------



**OEMP** Reference

Discussion was held on indicative voyages planned by AIMS to Goodrich Bank.	Santos has considered the matters raised in the OEMP and provided a response. Planned activities to be managed under this OEMP are unlikely to affect AIMS' field activities given the distance from the OA to Goodrich Bank. As a result, no credible impacts to AIMS' potential field activities are expected from planned activities. While impacts to AIMS' functions, interests and activities are possible in the event of an unplanned hydrocarbon spill, Santos considers appropriate controls are in place to prevent a hydrocarbon spill. Santos considers the measures and controls described within the GEP NT Waters OPEP adequately address oil spill planning and response in the event of a spill. AIMS is kept updated on Santos' activities in the OA via Notice to Mariners issued by the Australian Hydrographic Office, Santos' pre-activity notifications to marine users and Santos' Barossa Quarterly Project Update.	Santos thanked AIMS for provision of details on its Goodrich Bank interests. Santos advised AIMS that it did not believe there would be any impacts from Barossa a voyages during the advised time periods due to the distance from the nearest Barossa of Santos advised AIMS that it would be kept updated on Santos' activities via Notice to M Australian Hydrographic Office, Santos' pre-activity notifications to marine users and Sa Quarterly Project Update.
AIMS asked for information on Santos' infrastructure in the approaches to Darwin Harbour.	Santos has considered the matters raised in the OEMP and provided a response. The information request from AIMS with respect to Darwin Harbour infrastructure is outside the scope of this OEMP. Santos' infrastructure in the approaches to Darwin Harbour are located in NT Internal waters. Petroleum activities in NT Internal waters are outside the scope of this OEMP.	Santos responded to AIMS and provided requested information, separate to the activitie the Coastal Waters OEMP.
Australian Maritime Safety Authority (AMSA)		
Cummers of equalitation offerts		

- On 9 February 2024 Santos emailed AMSA to advise it of preliminary consultation regarding proposed activities for consultation to be managed under the Coastal Waters OEMP and Barossa Production Operation
- The email advised that Santos was seeking information to better understand:
  - o if you are from a government Department or agency, how the proposed Production Operations activities may be relevant to your department or agency; and
  - what (if any) functions, interests or activities you or your organisation have that may be affected by the proposed Production Operations activities.
- The email included information on the regulatory process for the activities in Commonwealth and NT jurisdictions and links to a Santos information booklet on the proposed activities and a NOPSEMA brochure on Santos to register as a Relevant Person. The email stated that the consultation phase would commence on 11 March 2024 and close on 9 April 2024.
- On 7 March 2024 AMSA emailed Santos and advised it would like to register as a relevant person for further consultation on the development of the Coastal Waters OEMP and Barossa Production Operations EP.
- On 11 March 2024 Santos emailed AMSA further to the previous correspondence, to advise that it had commenced the consultation phase which would run until 9 April 2024. In addition to the previous information Relevant Persons' entitlements under the regulatory processes, details of how to provide feedback and a reminder of the closing date for consultation. [Con-3793]
- On 12 March 2024 AMSA's marine safety division emailed Santos with an auto-response providing further information on relevant Maritime Safety Information (MSI). [Con-3806]
- On 9 April 2024 AMSA's marine safety division emailed Santos to advise that AMSA's Joint Rescue Coordination Centre (JRCC) should be notified for promulgation of radio-navigation warnings 24-48 hours before appropriate lights and shapes to reflect the nature of operations and comply with the International Rules for Preventing Collisions at Sea (COLREGs). In particular, the use of appropriate lights and shapes to reflect manoeuvre). Vessels should also ensure their navigation status is set correctly in the vessel's AIS unit. Collision risk mitigation measures may include but are not limited to:
- Additional warnings and/or lights to attract attention
- Installation of Automatic Identification System (AIS) units
- Offshore guard vessel/s that can monitor traffic and take early action to alert a vessel approaching the area of operations. [Con-3807]
- On 8 May 2024, Santos emailed AMSA further to emails previously sent, to advise it had extended the consultation period until 22 May 2024. In providing this extension of time, Santos advised that the information for an additional risk associated with the proposed activity. Santos confirmed that consultation will close on the revised date unless Santos hears otherwise from AMSA. [Con-4142]
- On 10 July 2024 Santos emailed AMSA to advise the consultation period had been completed. Santos advised that it considered consultation had now closed for the purpose of Santos finalising and submitting environmentation for assessment. [Con-5086]
- On 15 October 2024 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The NOPSEMA for assessment, provided a link to the full EP on the NOPSEMA website and Santos' contact details for any queries. The Update also stated that the OEMP would be submitted to the NT-DITT for assessment.
- On 23 January 2025 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. Th
  NOPSEMA to provide further information, Santos had re-submitted the EP with some amendments in December 2024 for further assessment. The Update also re-stated that the OEMP would be submitted to the N
- No further correspondence or feedback was received from AMSA's marine safety division. AMSA's marine pollution division did not provide any response.

Summary of response by Relevant Person	Assessment of merits	Santos' Response Statement
AMSA's marine safety division advised Santos of the required formal notifications process prior to and during activities.	Santos has considered the matters raised in the OEMP. Santos notes AMSA's advice and has included requirements in the relevant sections of this OEMP, specifically the following:	No response required.



ctivities on any AIMS Operational Area. ariners issued by the ntos' Barossa	Shoals and banks area described in Section 3.2.4.5 Control measures for unplanned events are described in Section 7.	
s to be managed under	Not applicable.	
ıs EP. [Con-3787]		
the consultation process	and details of how to contact	
[Con-3805]		
again being provided, S	antos provided information on	
e operations commence. Vessels should exhibit t the nature of operations (e.g. restricted in the ability to		
in the booklet and factsheet had been updated to account		
vironment plans for these activities to government		
e Update advised that the EP had been submitted to ssment. [Con- 5982]		
e Update advised that, following a request from NT-DITT for assessment. [Con-6036]		
	OEMP Reference	
	Notification requirements for	
Notification requirements for AHO and AMSA JRCC are included in Table 8-13 and		

	Requirement to notify the Australian Hydrographic Office through datacentre@hydro.gov.au no less than four working weeks before activities commence for the promulgation of related notices to mariners.	
	Requirement to notify AMSA's JRCC through rccaus@amsa.gov.au (Phone: 1800 641 792 or +61 2 6230 6811) for promulgation of radio-navigation warnings 24-48 hours before operations commence.	
AMSA's marine safety division advised Santos of	Santos has considered the matters raised in the OEMP.	No response required.
the required maritime safety measures.	Santos notes AMSA's advice and has included requirements in the relevant sections of this OEMP, specifically the following:	
	• Vessels to comply with COLREGs, in particular, the use of appropriate lights and shapes to reflect the nature of operations (e.g. restricted in the ability to manoeuvre).	
	Vessels to ensure their navigation status is set correctly in the ship's AIS unit.	
	• Evaluation and implementation of adequate anti-collision measures, including the collision risk mitigation measures cited by AMSA, being additional warnings and/or lights to attract attention and offshore guard vessel/s that can monitor traffic and take early action to alert a vessel approaching the area of operations.	
Clean Energy Regulator (CER)		

- On 9 February 2024 Santos emailed CER to advise it of preliminary consultation regarding proposed activities for consultation to be managed under the Coastal Waters OEMP and Barossa Production Operations
- The email advised that Santos was seeking information to better understand:
  - o if you are from a government Department or agency, how the proposed Production Operations activities may be relevant to your department or agency; and
  - what (if any) functions, interests or activities you or your organisation have that may be affected by the proposed Production Operations activities.
- The email included information on the regulatory process for the activities in Commonwealth and NT jurisdictions and links to a Santos information booklet on the proposed activities and a NOPSEMA brochure on Santos to register as a Relevant Person. The email stated that the consultation phase would commence on 11 March 2024 and close on 9 April 2024.
- On 11 March 2024 Santos emailed CER further to the previous correspondence, to advise that it had commenced the consultation phase which would run until 9 April 2024. In addition to the previous information a
  Relevant Persons' entitlements under the regulatory processes, details of how to provide feedback and a reminder of the closing date for consultation. [Con-3793]
- On 28 March 2024, CER emailed Santos and provided detail on the schemes legislated by the Australian Government for measuring, managing, reducing or offsetting Australia's carbon emissions. It confirmed nor
  regulatory approval to progress. CER advised that during the course of its activities Santos will need to meet any reporting requirements that apply under the National Greenhouse and Energy Reporting Act 2007 (
  Gas project exceed covered 'scope 1' emissions of 100,000 tonnes of carbon dioxide equivalent (tCO<sub>2e</sub>), it will have obligations under the Safeguard Mechanism. CER provided a case number and additional conta
- On 30 March 2024, Santos emailed CER and requested a meeting in mid-April 2024 to discuss CER's response and other queries. [Con-3809]
- On 4 April 2024, CER emailed Santos and advised the NGER section can provide advice on the requirements for companies to report and that it appears the consultation relates primarily to safety and environmen
  information on. CER requested some questions or points of discussion for the meeting in mid-April, so it can determine attendees to support provision of advice on these matters. [Con-3810]
- On 18 June 2024, Santos sent CER an email requesting a meeting via Teams [Con-4946].
- On 28 June 2024 CER emailed Santos to accept the meeting request. [Con- 4947]
- On 3 July 2024 Santos met with CER to discuss how Santos will present the following information in the OEMP and EP:
- The role of the CER in administering the Safeguard Mechanism.
- Application of the Safeguard Mechanism to regulate GHG emissions in support of Australia meeting its emissions reduction targets.
- Application in principle of the Safeguard Mechanism to regulate GHG emissions from Barossa production operations. [Con-5036]
- At the meeting CER did not raise any concerns with the information presented by Santos.CER agreed to review the information provide within the next week. [Con-5036] Santos emailed the presentation slides to C
- On 9 July 2024 CER emailed Santos to advise it was satisfied with the information describing the Safeguard Mechanism regulations administered by the CER that will be included in the OEMP and EP, as present
- On 10 July 2024 Santos emailed CER to advise the consultation period had been completed. Santos advised that it considered consultation had now closed for the purpose of Santos finalising and submitting envir for assessment. [Con-5086]
- On 15 October 2024 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. Th
  NOPSEMA for assessment, provided a link to the full EP on the NOPSEMA website and Santos' contact details for any queries. The Update also stated that the OEMP would be submitted to the NT-DITT for asse
- On 23 January 2025 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. Th
  NOPSEMA to provide further information, Santos had re-submitted the EP with some amendments in December 2024 for further assessment. The Update also re-stated that the OEMP would be submitted to the N
- No further correspondence or feedback was received from CER.

Summary of response by Relevant Person	Assessment of merits	Santos' Response Statement
----------------------------------------	----------------------	----------------------------



	control measure BAO-CM-6.6.1 for notifying AHO.	
	Vessel anti-collision measures in accordance with COLREGs and AMSA requirements are included in a control measures (refer to BAO-CM-6.1.2 and BAO-CM-6.6.1) and associated performance standards.	
EP. [Con-3787]		
the consultation process	and details of how to contact	
again being provided, Sar	ntos provided information on	
ne of these schemes currently required Santos to obtain (NGER Act). Additionally, if the activities of the Barossa act details. [Con-3808]		
t matters, which they were not able to provide any further		
CER the same day for its further review. [Con-4948]		
ed by Santos at the meeting on 3 July 2024. [Con-5013]		
ronment plans for these activities to government regulators		
e Update advised that the EP had been submitted to ssment. [Con- 5982]		
e Update advised that, following a request from IT-DITT for assessment. [Con-6036]		

CER advised Santos that it was satisfied with information to be included in the EP on its regulation of GHG emissions from Barossa production operations.	The CER was included in consultation for the EP and OEMP, but they only provided feedback specific to the EP. Therefore, the specific feedback is not relevant to the OEMP. The GHG Emissions section of the OEMP describes the Safeguard Mechanism regulations, and that the Activity will comply with the requirements.	No response required.
Climate Change Authority (CCA)		

- On 9 February 2024 Santos emailed CCA to advise it of preliminary consultation regarding proposed activities for consultation to be managed under the Coastal Waters OEMP and Barossa Production Operations
- The email advised that Santos was seeking information to better understand:
  - o if you are from a government Department or agency, how the proposed Production Operations activities may be relevant to your department or agency; and
  - what (if any) functions, interests or activities you or your organisation have that may be affected by the proposed Production Operations activities.
- The email included information on the regulatory process for the activities in Commonwealth and NT jurisdictions and links to a Santos information booklet on the proposed activities and a NOPSEMA brochure on Santos to register as a Relevant Person. The email stated that the consultation phase would commence on 11 March 2024 and close on 9 April 2024.
- On 11 March 2024 Santos emailed CCA further to the previous correspondence, to advise that it had commenced the consultation phase which would run until 9 April 2024. In addition to the previous information a
  Relevant Persons' entitlements under the regulatory processes, details of how to provide feedback and a reminder of the closing date for consultation. [Con-3793]
- On 3 April 2024 Santos phoned CCA and spoke to a team member and advised it would send a request for a meeting.
- On 6 May 2024 Santos emailed CCA further to the phone call on 3 April 2024 to advise it had extended the consultation period until 20 May 2024. Santos repeated its request for a meeting. In providing this extense
  booklet and factsheet had been updated to account for an additional risk associated with the proposed activity. Santos confirmed that consultation will close on the revised date unless Santos hears otherwise from
- On 10 July 2024 Santos emailed CCA to advise the consultation period had been completed. Santos advised CCA that it considered consultation had now closed for the purpose of Santos finalising and submitting regulators for assessment. [Con-5086]
- On 15 October 2024 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. Th
  NOPSEMA for assessment, provided a link to the full EP on the NOPSEMA website and Santos' contact details for any queries. The Update also stated that the OEMP would be submitted to the NT-DITT for asse
- On 23 January 2025 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. Th
  NOPSEMA to provide further information, Santos had re-submitted the EP with some amendments in December 2024 for further assessment. The Update also re-stated that the OEMP would be submitted to the Northern assessment.
- Notwithstanding the consultation information provided and the steps described above, no comments or input were received on this OEMP. from CCA.

Summary of response by Relevant Person	Assessment of merits	Santos' Response Statement
No response was received from CCA.	Santos considers it has provided sufficient information and a reasonable period of time for consultation. Santos considers Section 25 consultation requirements to have been met.	No response required.

- On 9 February 2024 Santos emailed CSIRO to advise it of preliminary consultation regarding proposed activities for consultation to be managed under the Coastal Waters OEMP and Barossa Production Operatio
- The email advised that Santos was seeking information to better understand:
  - o if you are from a government Department or agency, how the proposed Production Operations activities may be relevant to your department or agency; and
  - what (if any) functions, interests or activities you or your organisation have that may be affected by the proposed Production Operations activities.
- The email included information on the regulatory process for the activities in Commonwealth and NT jurisdictions and links to a Santos information booklet on the proposed activities and a NOPSEMA brochure on Santos to register as a Relevant Person. The email stated that the consultation phase would commence on 11 March 2024 and close on 9 April 2024.
- On 13 March 2024 Santos emailed CSIRO further to the previous correspondence, to advise that it had commenced the consultation phase which would run until 9 April 2024. In addition to the previous information Relevant Persons' entitlements under the regulatory processes, details of how to provide feedback and a reminder of the closing date for consultation. [Con-3794]
- On 3 and 4 April 2024 Santos phoned and spoke to the general enquiries line and left a message regarding consultation for Production Operations activities.
- On 2 May 2024, Santos emailed CSIRO further to previous correspondence to advise it had extended the consultation period until 16 May 2024. In providing this extension of time, the Santos advised that the infor account for an additional risk associated with the proposed activity. Santos confirmed that consultation will close on the revised date unless Santos hears otherwise from the CSIRO. [Con-3855]
- On 10 July 2024 Santos emailed CSIRO to advise the consultation period had been completed. Santos advised that it considered consultation had now closed for the purpose of Santos finalising and submitting er
  regulators for assessment. [Con-5086]
- On 15 October 2024 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. Th
  NOPSEMA for assessment, provided a link to the full EP on the NOPSEMA website and Santos' contact details for any queries. The Update also stated that the OEMP would be submitted to the NT-DITT for asse
- On 23 January 2025 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. Th
  NOPSEMA to provide further information, Santos had re-submitted the EP with some amendments in December 2024 for further assessment. The Update also re-stated that the OEMP would be submitted to the Northern assessment.
- Notwithstanding the consultation information provided and the steps described above, no comments or input were received on the Coastal Waters OEMP from CSIRO.



	Not applicable.		
EP. [Con-3787]			
the consultation process	and details of how to contact		
again being provided, Sar	ntos provided information on		
sion of time, Santos advis I CCA. [Con-4132]	ed that the information in the		
g environment plans for th	nese activities to government		
e Update advised that th ssment. [Con- 5982]	e EP had been submitted to		
e Update advised that, fo NT-DITT for assessment.	e Update advised that, following a request from NT-DITT for assessment. [Con-6036]		
	OEMP Reference		
	Not applicable.		
ns EP. [Con-3787]			
	and details of how to contact		
the consultation process and details of how to contact			
again being provided, Santos provided information on			
mation in the booklet and factsheet had been updated to			
vironment plans for these activities to government			
ne Update advised that th ssment. [Con- 5982]	e EP had been submitted to		
e Update advised that, fo	ollowing a request from [Con-6036]		

Summary of response by Relevant Person	Assessment of merits	Santos' Response Statement
No response was received from CSIRO.	Santos considers it has provided sufficient information and a reasonable period of time for consultation. Santos considers Section 25 consultation requirements to have been met.	No response required.
Department of Agriculture, Forestry and Fisherie	s (DAFF) – Biosecurity (marine pests) and Fisheries	
Summary of consultation effort:		
On 9 February 2024 Santos emailed DAFF to a	dvise it of preliminary consultation regarding proposed activities for consult	ation to be managed under the Coastal Waters OEMP and Barossa Production Operation
The email advised that Santos was seeking info	rmation to better understand:	
<ul> <li>if you are from a government Departm</li> </ul>	ent or agency, how the proposed Production Operations activities may be r	elevant to your department or agency; and
<ul> <li>what (if any) functions, interests, or ac</li> </ul>	tivities you or your organisation have that may be affected by the proposed	Production Operations activities.
The email included information on the regulator Santos to register as a Relevant Person. The end	y process for the activities in Commonwealth and NT jurisdictions and links mail stated that the consultation phase would commence on 11 March 2024	to a Santos information booklet on the proposed activities and a NOPSEMA brochure on and close on 9 April 2024.
On 11 March 2024 Santos emailed DAFF furthe Relevant Persons' entitlements under the regula	er to the previous correspondence, to advise that it had commenced the cor atory processes, details of how to provide feedback and a reminder of the c	nsultation phase which would run until 9 April 2024. In addition to the previous information losing date for consultation. [Con-3793]
On 11 March 2024, DAFF's Conveyance Policy	(Maritime) team (formerly Seaports Team) provided an automated response	se. [Con-3811]
On 8 May 2024, Santos emailed DAFF further to account for an additional risk associated with the account for a second back as a for a second back as a for a second back as a	o the previous correspondence, to advise it has extended the consultation perposed activity. Santos confirmed that consultation will close on the rev	period until 22 May 2024. In providing this extension of time, Santos advised that the infor vised date unless Santos hears otherwise from DAFF. [Con-4143]
On 8 May 2024, DAFF's Conveyance Policy (M	aritime) team (formerly Seaports Team) provided an automated response.	[Con-4950]
<ul> <li>On 10 July 2024 Santos emailed DAFF to advis regulators for assessment. [Con-5086]</li> </ul>	e the consultation period had been completed. Santos advised DAFF that i	t considered consultation had now closed for the purpose of Santos finalising and submit
On 23 August 2024 Santos phoned DAFF and f consultation on other Barossa EPs. In the email	ollowed-up with an email on 28 August 2024 advising that, in the absence of Santos provided details of the DAFF information being included in the OE	of any specific response from DAFF, Santos has reverted to the standard advice provided MP and EP and requested any further input by 9 September 2024. [Con-5608]
On 28 August 2024 DAFF emailed an auto-resp	oonse to Santos' email of 28 August 2024 [Con-5610]	
On 11 September 2024 Santos met with DAFF's roadmap for the FPSO, hull biofouling and clear	s Conveyance Policy Biosecurity Operations Division to discuss preparation ning, topside biosecurity & inspections, voyage preparations, ballast water t	n of the Barossa FPSO Biosecurity Management Plan (relevant to activities under the Pro reatment system exemption, FPSO arrival in Australia and the timing/ checklists, reportin
<ul> <li>On 15 October 2024 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. T NOPSEMA for assessment, provided a link to the full EP on the NOPSEMA website and Santos' contact details for any queries. The Update also stated that the OEMP would be submitted to the NT-DITT for ass</li> </ul>		
On 23 January 2025 Santos emailed the Baross NOPSEMA to provide further information, Santo	sa Project Quarterly Update to all stakeholders on its distribution list, includ to had re-submitted the EP with some amendments in December 2024 for f	ing all those consulted during preparation of the Production Operations EP and OEMP. The further assessment. The Update also re-stated that the OEMP would be submitted to the
Notwithstanding the consultation information pro	ovided and the steps described above, no comments or input were received	d on the Coastal Waters OEMP from DAFF.
Summary of response by Relevant Person	Assessment of merits	Santos' Response Statement
DAFF Biosecurity (Conveyance Policy Biosecurity Operations Division) provided input & guidance in	Santos considers it has provided sufficient information and a reasonable period of time for consultation.	All relevant DAFF biosecurity requirements are understood and referenced in commitme OEMP.
the preparation of the FPSO Biosecurity Management Plan. It had no comments on any aspects of the OEMP.	In the absence of any specific response, Santos has reverted to standard advice provided by DAFF Biosecurity with respect to	Santos will report and engage directly with DAFF for the management of biosecurity risk acceptance as stated in the cited offshore biosecurity guidelines and other associated of
	advice to the Coastal Waters OEMP, including activity notifications.	Santos confirmed in its response to DAFF for the Production Operations EP that Santos DAFF informed and incorporate DAFF's assistance offer into relevant management play
	Santos considers Section 25 consultation requirements to have been	
	met.	
No response was received from DAFF Fisheries.	reasonable period of time for consultation.	No response required.
	In the absence of any specific response, Santos has reverted to standard advice provided by DAFF Fisheries with respect to fishery matters. Santos has considered and applied this standard advice to this	
	OEMP., including activity notifications.	
	Santos considers Section 25 consultation requirements to have been met.	



OEMP Reference
Not applicable.

ns EP. [Con-3787]

the consultation process and details of how to contact

n again being provided, Santos provided information on

prmation in the booklet and factsheet has been updated to

tting environment plans for these activities to government

d by DAFF in response to requests for feedback during

oduction Operations EP). Topics included the biosecurity ng requirements and arrival schedules. [Con-6019]

The Update advised that the EP had been submitted to essment. [Con- 5982]

The Update advised that, following a request from NT-DITT for assessment. [Con-6036]

	OEMP Reference
ents documented in this post EP and OEMP ocumentation. will continue to keep ns.	Notifications to DAFF Biosecurity are included in Table 8-13 Santos' environmental management framework relevant to biosecurity risk for OEMP activities is outlined in Section 8.3.2.9 and is consistent with DAFF Biosecurity requirements. Adopted control measures are listed in Section 8.1.2
	Refer to Table 4-17 (Commercial Fishing/ Commonwealth / NT managed) for consultation with licence holders. Refer to Table 4-19 (Industry Associations) for consultation with industry association relevant to Commonwealth fisheries - ASBTIA, CFA and NPFI.

## Department of Climate Change, Energy, the Environment and Water (DCCEEW) - Underwater Cultural Heritage Branch

Summary of consultation effort:

- On 9 February 2024 Santos emailed DCCEEW's Underwater Cultural Heritage Branch to advise it of preliminary consultation regarding proposed activities for consultation to be managed under the Coastal Waters 3787]
- The email advised that Santos was seeking information to better understand:
  - o if you are from a government Department or agency, how the proposed Production Operations activities may be relevant to your department or agency; and
  - what (if any) functions, interests or activities you or your organisation have that may be affected by the proposed Production Operations activities.
- The email included information on the regulatory process for the activities in Commonwealth and NT jurisdictions and links to a Santos information booklet on the proposed activities and a NOPSEMA brochure on Santos to register as a Relevant Person. The email stated that the consultation phase would commence on 11 March 2024 and close on 9 April 2024.
- On 13 March 2024 Santos emailed DCCEEW's Underwater Cultural Heritage Branch further to the previous correspondence, to advise that it had commenced the consultation phase which would run until 9 April 2
  provided, Santos provided information on Relevant Persons' entitlements under the regulatory processes, details of how to provide feedback and a reminder of the closing date for consultation. [Con-3794]
- On 3 April 2024 Santos phoned the DCCEEW's Underwater Cultural Heritage Branch and left a message with reception regarding consultation for Production Operations activities.
- On 5 April 2024, a representative from DCCEEW's Underwater Cultural Heritage Branch phoned Santos and left a voicemail. [Con-4955]
- On 9 April 2024, DCCEEW's Underwater Cultural Heritage Branch emailed Santos advising Santos of the UCH Act requirements, including a summary of the UCH Act protections, key responsibilities and obligation [Con-3814]
- On 8 May 2024, Santos emailed DCCEEW's Underwater Cultural Heritage Branch further to recent correspondence to acknowledge the advice provided in the Branch's 9 April 2024 email and advise that Santos h providing this extension of time, Santos advised that the information in the booklet and factsheet had been updated to account for an additional risk associated with the proposed activity. Santos confirmed that con hears otherwise from DCCEEW's Underwater Cultural Heritage Branch. [Con-4144]
- On 10 July 2024 Santos emailed DCCEEW's Underwater Cultural Heritage Branch to advise the consultation period had been completed. Santos advised it considered that consultation had now closed for the pur for these activities to government regulators for assessment. [Con-5086]
- On 15 October 2024 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The NOPSEMA for assessment, provided a link to the full EP on the NOPSEMA website and Santos' contact details for any queries. The Update also stated that the OEMP would be submitted to the NT-DITT for asse
- On 23 January 2025 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. Th
  NOPSEMA to provide further information, Santos had re-submitted the EP with some amendments in December 2024 for further assessment. The Update also re-stated that the OEMP would be submitted to the N
- No further correspondence or feedback was received from DCCEEW Underwater Cultural Heritage Branch.

Summary of response by Relevant Person	Assessment of merits	Santos' Response Statement
DCCEEW Underwater Cultural Heritage Branch provided advice to Santos on its obligations under the UCH Act.	Santos has considered the matters raised in the OEMP Santos acknowledged the advice from DCCEEW with respect to obligations under the UCH Act, including following DCCEEW guidance if UCH is detected during planned activities or as a result of an unplanned event.	No response required.

## Department of Defence (DoD)

- On 9 February 2024 Santos emailed DoD to advise it of preliminary consultation regarding proposed activities for consultation to be managed under the Coastal Waters OEMP and Barossa Production Operations
- The email advised that Santos was seeking information to better understand:
  - o if you are from a government Department or agency, how the proposed Production Operations activities may be relevant to your department or agency; and
  - what (if any) functions, interests or activities you or your organisation have that may be affected by the proposed Production Operations activities.
- The email included information on the regulatory process for the activities in Commonwealth and NT jurisdictions and links to a Santos information booklet on the proposed activities and a NOPSEMA brochure on Santos to register as a Relevant Person. The email stated that the consultation phase would commence on 11 March 2024 and close on 9 April 2024.
- On 11 March 2024 Santos emailed DoD further to the previous correspondence, to advise that it had commenced the consultation phase which would run until 9 April 2024. In addition to the previous information a
  Relevant Persons' entitlements under the regulatory processes, details of how to provide feedback and a reminder of the closing date for consultation. [Con-3793]
- On 4 April 2024 Santos phoned DoD and left a voice mail message with two DoD contacts regarding consultation for Production Operations activities.
- On 6 May 2024 Santos emailed DoD further to previous correspondence to advise it had extended the consultation period until 20 May 2024. In providing this extension of time, Santos advised that the information
  for an additional risk associated with the proposed activity. Santos confirmed that consultation will close on the revised date unless Santos hears otherwise from DoD. [Con-4133]
- On 10 July 2024 Santos emailed DoD to advise the consultation period had been completed. Santos advised DoD that it considered that consultation had now closed for the purpose of Santos finalising and submi regulators for assessment. [Con-5086]
- On 15 October 2024 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. Th
  NOPSEMA for assessment, provided a link to the full EP on the NOPSEMA website and Santos' contact details for any queries. The Update also stated that the OEMP would be submitted to the NT-DITT for asse
- On 23 January 2025 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. Th
  NOPSEMA to provide further information, Santos had re-submitted the EP with some amendments in December 2024 for further assessment. The Update also re-stated that the OEMP would be submitted to the N



s OEMP and Barossa Pro	oduction Operations EP. [Con-
the consultation process	and details of how to contact
2024. In addition to the pr	evious information again being
ons, management conside	erations and recommendations.
nad extended the consultation period until 22 May 2024. In sultation will close on the revised date unless Santos	
pose of Santos finalising	and submitting environment plans
ne Update advised that the EP had been submitted to essment. [Con- 5982]	
ne Update advised that, following a request from NT-DITT for assessment. [Con-6036]	
	OFMP Reference
	Underwater cultural heritage is described in Section 3.2.14.7.7 Notifications to DCCEEW Underwater Cultural Heritage Branch are included in Table 8-13
	010
EP. [Con-3787]	
the consultation process	and details of how to contact
again being provided, San	tos provided information on
n in the booklet and factsh	neet had been updated to account
itting environment plans f	or these activities to government
ne Update advised that the essment. [Con- 5982]	e EP had been submitted to
ne Update advised that, fo NT-DITT for assessment.	ollowing a request from [Con-6036]

•	Notwithstanding the consultation information	provided and the steps described above	e, no comments or input were received on this OEMP from DoD.
---	----------------------------------------------	----------------------------------------	--------------------------------------------------------------

• In the absence of any specific response, Santos has reverted to standard advice provided by DoD with respect to defence matters. Santos has considered and applied this standard advice to this OEMP., includin

Summary of response by Relevant Person	Assessment of merits	Santos' Response Statement
No response was received from DoD.	Santos considers it has provided sufficient information and a reasonable period of time for consultation. Santos considers Section 25 consultation requirements to have been met.	No response required.
Department of Foreign Affairs and Trade (DEAT)		

Summary of consultation effort:

- On 9 February 2024 Santos emailed DFAT to advise it of preliminary consultation regarding proposed activities for consultation to be managed under the Coastal Waters OEMP and Barossa Production Operation
- The email advised that Santos was seeking information to better understand:
  - o if you are from a government Department or agency, how the proposed Production Operations activities may be relevant to your department or agency; and
  - what (if any) functions, interests, or activities you or your organisation have that may be affected by the proposed Production Operations activities.
- The email included information on the regulatory process for the activities in Commonwealth and NT jurisdictions and links to a Santos information booklet on the proposed activities and a NOPSEMA brochure on Santos to register as a Relevant Person. The email stated that the consultation phase would commence on 11 March 2024 and close on 9 April 2024.
- On 11 March 2024 Santos emailed DFAT further to the previous correspondence, to advise that it had commenced the consultation phase which would run until 9 April 2024. In addition to the previous information Relevant Persons' entitlements under the regulatory processes, details of how to provide feedback and a reminder of the closing date for consultation. [Con-3793]
- On 4 April 2024 Santos phoned DFAT regarding consultation for Production Operations activities. A DFAT representative requested Santos follow-up by email.
- On 15 April 2024 DFAT wrote to Santos and advised that given the location of the activity, there are several areas within DFAT that may need to provide views. DFAT asked if there was still an opportunity for DFA the end of the email. [Con-3815]
- On 8 May 2024 Santos emailed DFAT further to its response of 15 April 2024 to advise it has extended the consultation period until 22 May 2024. In providing this extension of time, Santos advised that the inform
  account for an additional risk associated with the proposed activity. Santos confirmed that consultation will close on the revised date unless Santos hears otherwise from DFAT. [Con-4146]
- On 24 May 2024 DFAT's Timor-Leste Branch emailed Santos and recommended it consult with the Government of Timor-Leste on Santos' Environment Plan given the proximity of Santos' operations to the territor consultation is the Autoridade Nacional Do Petróleo Timor-Leste (ANP National Petroleum Authority). [Con-4215]
- On 24 June 2024 Santos responded via email to DFAT's Timor-Leste Branch confirming that it would be consulting with ANP. [Con-4956]. A separate email was sent to the Indonesia Branch of DFAT asking whet
  [Con-4957]
- On 2 July 2024 Santos followed up one of the emails sent on 24 June 2024 with a phone call to DFAT's Indonesia Branch. The Indonesia Branch representative asked that the email be re-sent which Santos did the sent of the email be re-sent which Santos did the sent of th
- On 10 July 2024 DFAT's Indonesia Desk emailed Santos to advise that it had no comment on the EP. [Con-5083]
- On 15 October 2024 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The NOPSEMA for assessment, provided a link to the full EP on the NOPSEMA website and Santos' contact details for any queries. The Update also stated that the OEMP would be submitted to the NT-DITT for assessment.
- On 23 January 2025 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The NOPSEMA to provide further information, Santos had re-submitted the EP with some amendments in December 2024 for further assessment. The Update also re-stated that the OEMP would be submitted to the I
- No further correspondence or feedback was received from DFAT.

Summary of response by Relevant Person	Assessment of merits	Santos' Response Statement
DFAT recommended, via its Timor-Leste Branch, that Santos consult with the Autoridade Nacional Do Petróleo Timor-Leste (ANP - National Petroleum Authority).	DFAT's response is not relevant to the Coastal Waters OEMP. DFAT's response is addressed in the Barossa Production Operations EP, in which the activity in Commonwealth waters is more proximate to Timor-Leste and the EMBA for which extends into international waters. With regard to the location of the proposed GEP operations activities in NT Coastal Waters, there are no impacts from planned activities that may affect the functions, interests or activities of any international persons. The worst-case credible spill scenario modelled for the EMBA for the Coastal Waters OEMP also shows there are no impacts from unplanned activities that may affect the functions, interests or activities of any international persons.	No response required.

## Department of Home Affairs (DHA) / Australian Border Force (ABF)

- On 9 February 2024 Santos emailed DHA/ABF to advise it of preliminary consultation regarding proposed activities for consultation to be managed under the Coastal Waters OEMP and Barossa Production Operation
- The email advised that Santos was seeking information to better understand:
  - o if you are from a government Department or agency, how the proposed Production Operations activities may be relevant to your department or agency; and



g activity notifications.		
	OEMP Reference	
	Defence activities are described in Section 3.2.14.3. Notifications to DoD are included in Table 8-13	
ns EP. [Con-3787]		
the consultation process	and details of how to contact	
again being provided, Santos provided information on		
AT to provide information,	including the specific questions at	
nation in the booklet and factsheet had been updated to		
bry of Timor-Leste. The appropriate authority for such		
ther it had any similar advice re the consultation process.		
he same day. [Con-4958]		
he Update advised that th essment. [Con- 5982]	e EP had been submitted to	
he Update advised that, fo NT-DITT for assessment.	ollowing a request from [Con-6036]	
	OEMP Reference	
	Not applicable.	
ations EP. [Con-3787]		

what (if any) functions, interests or activities you or your organisation have that may be affected by the proposed Production Operations activities.

- The email included information on the regulatory process for the activities in Commonwealth and NT jurisdictions and links to a Santos information booklet on the proposed activities and a NOPSEMA brochure on the consultation process and details of how to contact
  Santos to register as a Relevant Person. The email stated that the consultation phase would commence on 11 March 2024 and close on 9 April 2024.
- On 11 March 2024 Santos emailed DHA/ABF further to the previous correspondence, to advise that it had commenced the consultation phase which would run until 9 April 2024. In addition to the previous information Relevant Persons' entitlements under the regulatory processes, details of how to provide feedback and a reminder of the closing date for consultation. [Con-3793]
- On 3 April 2024 Santos phoned DHA/ABF and spoke to a representative from the branch responsible critical infrastructure. The representative advised that its interest in Barossa Production Operations activities lives storage and offloading (FPSO) facility is constructed and Santos' requirement to hold and maintain a security plan. The representative advised DHA/ABF had been separately in contact with the Health and Safety project.
- On 6 May 2024, Santos emailed DHA/ABF and acknowledged ABF's guidance in relation to the Maritime Transport and Offshore Facilities Security Act 2003 and accompanying Regulations. Details of DHA/ABF's email Santos also advised that it had extended the consultation period until 20 May 2024. In providing this extension of time, Santos advised that the information in the booklet and factsheet had been updated to a activity. Santos confirmed that consultation will close on the revised date unless Santos hears otherwise from DHA/ABF. [Con-4134]
- On 10 July 2024 Santos emailed DHA/ABF to advise the consultation period had been completed. Santos advised it considered that consultation had now closed for the purpose of Santos finalising and submitting
  regulators for assessment. [Con-5086]
- On 15 October 2024 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The NOPSEMA for assessment, provided a link to the full EP on the NOPSEMA website and Santos' contact details for any gueries. The Update also stated that the OEMP would be submitted to the NT-DITT for assessment.
- On 23 January 2025 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The NOPSEMA to provide further information, Santos had re-submitted the EP with some amendments in December 2024 for further assessment. The Update also re-stated that the OEMP would be submitted to the I
- No further correspondence or feedback was received from DHA/ABF.

Summary of response by Relevant Person	Assessment of merits	Santos' Response Statement
DHA/ABF advised that its interest in proposed production operations activities related to offshore security matters, specifically for Santos to hold and maintain a security plan following construction of the FPSO.	This response relates to the activities to be managed under the Barossa Production Operations EP. Issues related to Production Operations activities in the Barossa Field, including operation of the FPSO, are outside the scope of the Coastal Waters OEMP and are addressed in the Barossa Production Operations EP.	No response required.

## Department of Industry, Science and Resources (DISR)

- On 9 February 2024 Santos emailed DISR to advise it of preliminary consultation regarding proposed activities for consultation to be managed under the Coastal Waters OEMP and Barossa Production Operations
- The email advised that Santos was seeking information to better understand:
  - o if you are from a government Department or agency, how the proposed Production Operations activities may be relevant to your department or agency; and
  - o what (if any) functions, interests or activities you or your organisation have that may be affected by the proposed Production Operations activities.
- The email included information on the regulatory process for the activities in Commonwealth and NT jurisdictions and links to a Santos information booklet on the proposed activities and a NOPSEMA brochure on Santos to register as a Relevant Person. The email stated that the consultation phase would commence on 11 March 2024 and close on 9 April 2024.
- On 9 February 2024, DISR emailed Santos and advised DISR has no comment on the environmental management of the proposed activity. [Con-3816]
- On 15 February 2024, Santos emailed DISR thanking it for its response and advised it will continue to keep DISR updated on Barossa activities. [Con-3817]
- On 7 May 2024, Santos emailed DISR further to the previous correspondence, to advise it had extended the consultation period until 21 May 2024. In providing this extension of time, Santos advised that the inforr
  account for an additional risk associated with the proposed activity. Santos confirmed that consultation will close on the revised date unless Santos hears otherwise from DISR. [Con-4140]
- On 10 July 2024 Santos emailed DISR to advise the consultation period had been completed. Santos advised DISR that it considered consultation had now closed for the purpose of Santos finalising and submittin
  regulators for assessment. [Con-5086]
- On 15 October 2024 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The NOPSEMA for assessment, provided a link to the full EP on the NOPSEMA website and Santos' contact details for any queries. The Update also stated that the OEMP would be submitted to the NT-DITT for assessment.
- On 23 January 2025 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that, following a re
  NOPSEMA to provide further information, Santos had re-submitted the EP with some amendments in December 2024 for further assessment. The Update also re-stated that the OEMP would be submitted to the NT-DITT for assessment. [Con-6036]
- Notwithstanding the consultation information provided and the steps described above, no comments or input were received on this OEMP. from DISR.

Summary of response by Relevant Person	Assessment of merits	Santos' Response Statement
No response was received from DISR.	Santos considers it has provided sufficient information and a reasonable period of time for consultation. Santos considers Section 25 consultation requirements to have been met.	No response required.



-	
ation again being providec	d, Santos provided information on
ies in offshore security ma section of Santos re the r	atters once the floating production, requirements for the Barossa
s request and Santos' resp account for an additional r	ponse are listed below. In the isk associated with the proposed
g environment plans for th	ese activities to government
he Update advised that th essment. [Con- 5982]	e EP had been submitted to
he Update advised that, fo NT-DITT for assessment.	ollowing a request from [Con-6036]
	OFMP Reference
	Not applicable.
s EP. [Con-3787]	
the consultation process	and details of how to contact
mation in the booklet and	factsheet had been updated to
ng environment plans for	these activities to government
he Update advised that th essment. [Con- 5982]	e EP had been submitted to
he Update advised that, fo	ollowing a request from

EP Reference
Not applicable.

### **Director of National Parks (DNP)**

- On 9 February 2024 Santos emailed DNP to advise it of preliminary consultation regarding proposed activities for consultation to be managed under the Coastal Waters OEMP and Barossa Production Operations EP. [Con-3787]
- The email advised that Santos was seeking information to better understand:
  - o if you are from a government Department or agency, how the proposed Production Operations activities may be relevant to your department or agency; and
  - what (if any) functions, interests or activities you or your organisation have that may be affected by the proposed Production Operations activities.
- The email included information on the regulatory process for the activities in Commonwealth and NT jurisdictions and links to a Santos information booklet on the proposed activities and a NOPSEMA brochure on the consultation process and details of how to contact Santos to register as a Relevant Person. The email stated that the consultation phase would commence on 11 March 2024 and close on 9 April 2024.
- On 13 March 2024 Santos emailed DNP further to the previous correspondence, to advise that it had commenced the consultation phase which would run until 9 April 2024. In addition to the previous information again being provided, Santos provided information on Relevant Persons' entitlements under the regulatory processes, details of how to provide feedback and a reminder of the closing date for consultation. [Con-3794]
- On 3 April 2024 Santos phoned DNP and left a voice mail.
- On 12 April 2024, DNP emailed Santos seeking an extension to comment until the week beginning 22 April. [Con-3812] Santos subsequently agreed to the request in a phone discussion with DNP on 18 April 2024.
- On 26 April 2024, DNP emailed Santos to advise that based on the information provided it had no objections and claims. However, as part of the ongoing inspection, maintenance, monitoring and repair (IMMR) of the Barossa GEP, Parks Australia would like to discuss the provision of a report, or similar, that outlines the findings of these activities in relation to the Barossa GEP which traverses parts of the Oceanic Shoals Marine Park. [Con-4122]
- On 30 April 2024, Santos emailed DNP to acknowledge the guidance note provided and confirm it will consider this information in the course of preparing the EP and OEMP and suggested dates for a meeting. [Con-4123]
- On 1 May 2024 DNP emailed Santos and advised that 24 May 2024 was suitable for a meeting. [Con-4124]
- On 1 May 2024, Santos emailed DNP further to the previous correspondence, to advise it has extended the consultation period until mid-May 2024 and that in providing this extension of time, the information in the booklet and factsheet has been updated to account for an additional risk associated with the proposed activity. [Con-4125]
- On 2 May 2024, DNP emailed Santos and advised that the preventative and mitigation measures appeared appropriate for the additional risk. Parks Australia requested a brief summary at the meeting and provided initial guestions that could also be answered during the meeting. [Con-4127]
- On 24 May 2024 Santos met with DNP. At the meeting it was agreed that Santos would include a measure in the Barossa Production Operations EP to provide a report on outcomes of IMMR activities in the Oceanic Shoals Marine Park multiple use zone (for 30km) and the habitat protection zone (for 31km). If additional raw data is requested by DNP for the remainder of the Barossa GEP, Santos would also provide this. Santos also responded to questions asked by DNP at the meeting (see table entries below). [Con-4952]
- On 9 July 2024 Santos emailed DNP minutes of the meeting held with Parks Australia on 24 May 2024. [Con-5018]
- On 10 July 2024 Santos emailed DNP to advise the consultation period had been completed. Santos advised that it considered consultation had now closed for the purpose of Santos finalising and submitting environment plans for these activities to government regulators for assessment. [Con-5086]
- On 15 October 2024 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that the EP had been submitted to NOPSEMA for assessment, provided a link to the full EP on the NOPSEMA website and Santos' contact details for any queries. The Update also stated that the OEMP would be submitted to the NT-DITT for assessment. [Con- 5982]
- On 23 January 2025 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that, following a request from NOPSEMA to provide further information, Santos had re-submitted the EP with some amendments in December 2024 for further assessment. The Update also re-stated that the OEMP would be submitted to the NT-DITT for assessment. [Con-6036]
- No further correspondence or feedback was received from DNP.

Summary of response by Relevant Person	Assessment of merits	Santos' Response Statement	OEMP Reference
DNP asked for information on methods of pipeline monitoring to detect potential gas releases.	Santos has considered the matters raised in the OEMP and provided a response. Santos presented further information on the detection methods including live pressure monitoring and detection controls and ROV inspections. These are discussed in Section 7.5 of the OEMP and reflected in adopted management controls.	<ul> <li>Santos confirmed:</li> <li>Leaks could be long term or short term. Short term leaks / release could potentially occur in areas that are more at risk of dropped objects.</li> <li>Safety procedures are strictly adhered to during lifting operations over the pipeline.</li> <li>Long term leaks may potentially occur at flanges and would be detected during IMMR activities.</li> </ul>	Management controls include BAO-CM-7.6.3.
DNP asked for information on timeliness to detect potential gas releases.	Santos has considered the matters raised in the OEMP and provided a response. Santos discussed the leak detection timeliness in respect to the cause and size of the leak. The detection methods include FPS process monitoring via pressure detection controls and ROV inspections. These are discussed in Section 7.5 of the OEMP and reflected in adopted management controls.	<ul> <li>Santos confirmed:</li> <li>It will send out a remotely operated vehicle (ROV) as soon as possible if a leak occurred due to a dropped object. Santos would then need to obtain equipment. The earliest response is within hours if equipment is available in Darwin.</li> <li>The inspection of the leak and repair may take longer, e.g. up to a couple of months if the equipment is not available locally.</li> <li>The FPSO and DLNG facility will pick up any pressure changes in the pipeline for a larger release. If a larger breach occurs, it could lose the full contents of the pipeline.</li> </ul>	Management controls include BAO-CM-7.6.1 and BAO-CM- 7.6.4.
DNP asked for information on response times following detection of gas releases.	Santos has considered the matters raised in the OEMP and provided a response. Santos discussed the scenarios and associated response times to a gas release. These are discussed in Section 7.5 of the OEMP and reflected in adopted management controls.	<ul> <li>Santos confirmed:</li> <li>Santos will respond immediately once a gas release is identified.</li> <li>Response will be in accordance with emergency response plan, similar to the response information in Bayu Undan EP (which is also gas), which will also be provided in the OEMP.</li> <li>For releases from flowlines/ wells, Santos will activate the Oil Pollution Emergency Plan which has a first strike response.</li> </ul>	Management controls include BAO-CM-7.6.4.



			Danco	
DNP asked for information on proposed repair activities to be undertaken in proximity of the Barossa GEP.	Santos has considered the matters raised in the OEMP and provided a response. Santos discussed what potential repair activities could take place in the event of a subsea release. These are included in Section 7.5 of the OEMP and reflected in adopted management controls.	<ul> <li>Santos confirmed:</li> <li>A large release may require replacement of a component or section of pipeline.</li> <li>Santos would need to mobilise a pipelay vessel and may take months to fix and would need environmental and safety approvals as well.</li> <li>The gas flow would be turned off and the pipeline would be repaired and tested again before resuming operations.</li> <li>Small leaks are generally from old equipment or poor installation. Newly installed pipelines have a rigorous regime of Quality Assurance.</li> <li>Pipeline components have been tested onshore and tested again as part of full Barossa GEP system commissioning.</li> <li>Leaks could potentially occur in areas where flanges are or areas where the pipeline may move over time. These areas will be priority areas during surveys and subject to further investigations during IMMR activities.</li> </ul>	Management controls include BAO-CM-7.6.3 and BAO-CM- 7.6.5.	
that outlines the findings of Santos' IMMR activities in relation to the section of the Barossa GEP that traverses the Oceanic Shoals Marine Park.	Oceanic Shoals Marine Park. Santos has provided a response to DNP and considered the matters raised by DNP in the Production Operations EP.	<ul> <li>Santos commed in its response to DNP for the Production Operations EP that.</li> <li>Santos will provide a report on outcomes of IMMR activities in the Oceanic Shoals Marine Park multiple use zone (for 30km) and the habitat protection zone (for 31km).</li> <li>Santos will also provide raw data if requested by DCCEEW Parks Australia Branch for the remainder of the Barossa GEP.</li> </ul>		
Fisheries Research and Development Corporation	on (FRDC)			
Summary of consultation effort:				
On 9 February 2024 Santos emailed FRDC to a	advise it of preliminary consultation regarding proposed activities for consul-	tation to be managed under the Coastal Waters OEMP and Barossa Production Operations EP. [Con-3787]		
The email advised that Santos was seeking info	ormation to better understand:			
<ul> <li>if you are from a government Department or agency, how the proposed Production Operations activities may be relevant to your department or agency; and</li> </ul>				
<ul> <li>what (if any) functions, interests or activities you or your organisation have that may be affected by the proposed Production Operations activities.</li> </ul>				
<ul> <li>The email included information on the regulatory process for the activities in Commonwealth and NT jurisdictions and links to a Santos information booklet on the proposed activities and a NOPSEMA brochure on the consultation process and details of how to contact Santos to register as a Relevant Person. The email stated that the consultation phase would commence on 11 March 2024 and close on 9 April 2024.</li> </ul>				
<ul> <li>On 13 March 2024 Santos emailed FRDC further to the previous correspondence, to advise that it had commenced the consultation phase which would run until 9 April 2024. In addition to the previous information again being provided, Santos provided information on Relevant Persons' entitlements under the regulatory processes, details of how to provide feedback and a reminder of the closing date for consultation. [Con-3794]</li> </ul>				
On 4 April 2024 Santos phoned FRDC and was advised the Communication Program Team is the right contact and a message was left for that team.				
• On 6 May 2024, Santos emailed FRDC further to the previous correspondence, to advise it has extended the consultation period until 20 May 2024 and that in providing this extension of time, the information in the booklet and factsheet has been updated to account for an additional risk associated with the proposed activity. In the email Santos stated that, if input is not received by this date Santos will infer this means you do not want Santos to consult with you further on the EP and OEMP. [Con-4131]				
<ul> <li>On 10 July 2024 Santos emailed FRDC to advise the consultation period had been completed. Santos advised FRDC that it considered that consultation had now closed for the purpose of Santos finalising and submitting environment plans for these activities to government regulators for assessment. [Con-5086]</li> </ul>				
On 15 October 2024 Santos emailed the Baros NOPSEMA for assessment, provided a link to the second sec	<ul> <li>On 15 October 2024 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that the EP had been submitted to NOPSEMA for assessment, provided a link to the full EP on the NOPSEMA website and Santos' contact details for any queries. The Update also stated that the OEMP would be submitted to the NT-DITT for assessment. [Con- 5982]</li> </ul>			
On 23 January 2025 Santos emailed the Baros NOPSEMA to provide further information, Santo	sa Project Quarterly Update to all stakeholders on its distribution list, includ os had re-submitted the EP with some amendments in December 2024 for	ing all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that, f further assessment. The Update also re-stated that the OEMP would be submitted to the NT-DITT for assessment.	ollowing a request from . [Con-6036]	
Notwithstanding the consultation information provided and the steps described above, no comments or input were received on Coastal Waters OEMP from FRDC.				
Summary of response by Relevant Person	Assessment of merits	Santos' Response Statement	OEMP Reference	
No response was received from FRDC.	Santos considers it has provided sufficient information and a reasonable period of time for consultation. Santos considers Section 25 consultation requirements to have been met.	No response required.	Not applicable	
National Indigenous Australians Agency (NIAA)				
Summary of consultation effort:				
On 9 February 2024 Santos emailed NIAA to ad	dvise it of preliminary consultation regarding proposed activities for consulta	ation to be managed under the Coastal Waters OEMP and Barossa Production Operations EP. [Con-3787]		

Summary of response by Relevant Person	Assessment of merits	Santos' Response Statement
No response was received from FRDC.	Santos considers it has provided sufficient information and a reasonable period of time for consultation. Santos considers Section 25 consultation requirements to have been met.	No response required.
National Indigenous Australians Agency (NIAA)		

- The email advised that Santos was seeking information to better understand: •
  - o if you are from a government Department or agency, how the proposed Production Operations activities may be relevant to your department or agency; and
  - what (if any) functions, interests or activities you or your organisation have that may be affected by the proposed Production Operations activities.
- The email included information on the regulatory process for the activities in Commonwealth and NT jurisdictions and links to a Santos information booklet on the proposed activities and a NOPSEMA brochure on the consultation process and details of how to contact • Santos to register as a Relevant Person. The email stated that the consultation phase would commence on 11 March 2024 and close on 9 April 2024.
- On 13 March 2024 Santos emailed NIAA further to the previous correspondence, to advise that it had commenced the consultation phase which would run until 9 April 2024. In addition to the previous information again being provided, Santos provided information on • Relevant Persons' entitlements under the regulatory processes, details of how to provide feedback and a reminder of the closing date for consultation. [Con-3794]
- On 4 April 2024 Santos phoned the NIAA and left a message with reception regarding consultation on Production Operations activities with reception.



- On 5 April 2024, the NIAA emailed Santos to advise that it does not, as a general practice, make comments on proponent's environmental management plans and only responds to requests for comment under arra Energy, the Environment and Water (through the Environmental Protection and Biodiversity Conservation Act 1999 public consultation process) and the Department of Industry, Science and Resources (through the
- On 2 May 2024, Santos emailed NIAA to acknowledge its email of 5 April 2024 and advise it has extended the consultation period until 16 May 2024 and that in providing this extension of time, the information in th
  an additional risk associated with the proposed activity. [Con-4128]
- On 10 July 2024 Santos emailed NIAA to advise the consultation period had been completed. Santos advised it considered that consultation had now closed for the purpose of Santos finalising and submitting environment regulators for assessment. [Con-5086]
- On 15 October 2024 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The NOPSEMA for assessment, provided a link to the full EP on the NOPSEMA website and Santos' contact details for any queries. The Update also stated that the OEMP would be submitted to the NT-DITT for assess
- On 23 January 2025 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The NOPSEMA to provide further information, Santos had re-submitted the EP with some amendments in December 2024 for further assessment. The Update also re-stated that the OEMP would be submitted to the N
- Notwithstanding the consultation information provided and the steps described above, no comments or input were received on Coastal Waters OEMP from NIAA.

Summary of response by Relevant Person	Assessment of merits	Santos' Response Statement
NIAA advised Santos that it does not, as a general practice, make comments on proponent's environmental management plans	Santos notes the responses provided. Santos considers it has provided sufficient information and a reasonable period of time for consultation. Santos considers Section 25 consultation requirements to have been met.	No response required.



angements with the Department of Climate Change, e Major Projects Facilitation Agency). [Con-3818]				
e booklet and factsheet h	has been updated to account for			
ironment plans for these	ronment plans for these activities to government			
e Update advised that the EP had been submitted to ssment. [Con- 5982]				
e Update advised that, following a request from IT-DITT for assessment. [Con-6036]				
OEMP Reference				
Not applicable				

### 4.8.2 NT Government Agency or Authority

## Table 4-12: Consultation Summary Table - NT Government Agency or Authority

Section 25(1)(b) of the OPGGS(E)R: Northern Territory agency or authority to which the activities to be carried out under the environment plan may be relevant

## **Aboriginal Areas Protection Authority (AAPA)**

### Summary of consultation effort:

- On 9 February 2024 Santos emailed AAPA to advise it of preliminary consultation regarding proposed activities for consultation to be managed under the Coastal Waters OEMP and Barossa Production Operations EP. [Con-3787]
- The email advised that Santos was seeking information to better understand:
  - o if you are from a government Department or agency, how the proposed Production Operations activities may be relevant to your department or agency; and
  - what (if any) functions, interests or activities you or your organisation have that may be affected by the proposed Production Operations activities.
- The email included information on the regulatory process for the activities in Commonwealth and NT jurisdictions and links to a Santos information booklet on the proposed activities and a NOPSEMA brochure on the consultation process and details of how to contact Santos to register as a Relevant Person. The email stated that the consultation phase would commence on 11 March 2024 and close on 9 April 2024.
- On 11 March 2024 Santos emailed AAPA further to previous response, to advise that it had commenced the consultation phase which would run until 9 April 2024. In addition to the previous information again being provided, Santos provided information on Relevant Persons' entitlements under the regulatory processes, details of how to provide feedback and a reminder of the closing date for consultation. [Con-3793]
- On 21 March 2024, the AAPA emailed Santos providing comments from the AAPA on the Production Operations activities and advising that it considers itself a Relevant Person. [Con-3819]
- On 30 April 2024, Santos emailed AAPA a letter of response to its comments. AAPA's comments and Santos' responses are summarised below. In the letter Santos also advised it had extended the consultation period until 13 May 2024. In providing this extension of time, Santos advised that the information in the booklet and factsheet had been updated to account for an additional risk associated with the proposed activity. Santos confirmed that consultation will close on the revised date unless Santos hears otherwise from AAPA. [Con-4364]
- On 10 July 2024 Santos emailed AAPA to advise the consultation period had been completed. Santos advised it consultation had now closed for the purpose of Santos finalising and submitting environment plans for these activities to government regulators for assessment. [Con-5086]
- On 15 October 2024 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that the EP had been submitted to NOPSEMA for assessment, provided a link to the full EP on the NOPSEMA website and Santos' contact details for any gueries. The Update also stated that the OEMP would be submitted to the NT-DITT for assessment. [Con- 5982]
- On 23 January 2025 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that, following a request from NOPSEMA to provide further information, Santos had re-submitted the EP with some amendments in December 2024 for further assessment. The Update also re-stated that the OEMP would be submitted to the NT-DITT for assessment. [Con-6036]
- No further correspondence or feedback was received from AAPA.

Summary of response by Relevant Person	Assessment of merits	Santos' Response Statement	OEMP reference
AAPA stated that its interests and activities may be affected by the Production Operations activities as Santos spill modelling indicated potential impacts to sacred sites in the event of a spill.	Santos considers AAPA's claim has merit. However, there are no registered sacred sites in the Operational Area. As a result, no credible impacts to known sites are expected from planned activities. While impacts to sacred sites are possible in the event of an unplanned hydrocarbon spill, Santos considers appropriate controls are in place to prevent a hydrocarbon spill. Santos also has controls to respond in the highly unlikely event of a hydrocarbon spill. Santos notes access restrictions to sacred sites under the NTASS Act and the need for appropriate permissions in the event that access to sacred sites is required to support and	No response required.	Sacred sites are described in Section 3.2.15.6 Control measures for unplanned hydrocarbon spill events are described in Section 7.
AAPA requested that Santos speak to the Australian Energy Producer's Oil Spill Working Group which had held recent discussions with the NT Government.	Santos has considered the matters raised in the OEMP and provided a response. Santos notes that it has conferred with a member of the Australian Energy Producer's Oil Spill Working Group.	Santos confirmed it had conferred with a representative of the Working Group.	No reference required.
Darwin Harbour Advisory Committee (DHAC)			

- On 9 February 2024 Santos emailed DHAC to advise it of preliminary consultation regarding proposed activities for consultation to be managed under the Coastal Waters OEMP and Barossa Production Operations EP. [Con-3787]
- The email advised that Santos was seeking information to better understand:
  - o if you are from a government Department or agency, how the proposed Production Operations activities may be relevant to your department or agency; and
  - what (if any) functions, interests or activities you or your organisation have that may be affected by the proposed Production Operations activities.
- The email included information on the regulatory process for the activities in Commonwealth and NT jurisdictions and links to a Santos information booklet on the proposed activities and a NOPSEMA brochure on the consultation process and details of how to contact Santos to register as a Relevant Person. The email stated that the consultation phase would commence on 11 March 2024 and close on 9 April 2024.
- On 11 March 2024 Santos emailed DHAC further to the previous correspondence, to advise that it had commenced the consultation phase which would run until 9 April 2024. In addition to the previous information again being provided, Santos provided information on Relevant Persons' entitlements under the regulatory processes, details of how to provide feedback and a reminder of the closing date for consultation. [Con-3793]
- On 2 May 2024, Santos emailed DHAC further to the previous correspondence, to advise it had extended the consultation period until 16 May 2024. In providing this extension of time Santos advised that the information in the booklet and factsheet had been updated to account for an additional risk associated with the proposed activity. Santos confirmed that consultation will close on the revised date unless Santos hears otherwise from DHAC. [Con-4374]
- On 3 May 2024 a representative from DEPWS NT emailed Santos on behalf of the Chairperson of DHAC and advised that Santos' email had been forwarded to DHAC committee members who will respond separately if they have questions. [Con-4369]
- On 10 July 2024 Santos emailed DHAC to advise the consultation period had been completed. Santos advised DHAC that it considered consultation had now closed for the purpose of Santos finalising and submitting environment plans for these activities to government regulators for assessment. [Con-5086]



- On 15 October 2024 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that the EP had been submitted to NOPSEMA for assessment, provided a link to the full EP on the NOPSEMA website and Santos' contact details for any queries. The Update also stated that the OEMP would be submitted to the NT-DITT for assessment. [Con- 5982]
- On 23 January 2025 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that, following a request from ٠ NOPSEMA to provide further information, Santos had re-submitted the EP with some amendments in December 2024 for further assessment. The Update also re-stated that the OEMP would be submitted to the NT-DITT for assessment. [Con-6036]
- No further correspondence or feedback was received from DHAC or DEPWS NT on behalf of the Chairperson of DHAC.
- Notwithstanding the consultation information provided and the steps described above, no comments or input were received on this OEMP from DHAC committee members.

Summary of response by Relevant Person	Assessment of merits	Santos' Response Statement
No response was received from DHAC.	Santos considers it has provided sufficient information and a reasonable period of time for consultation.	No response required.
	Santos considers Section 25 consultation requirements to have been met.	



OEMP reference
Not applicable.

## Department of Lands, Planning and the Environment (DLI) formerly the Department of Environment, Parks and Water Security (DEPWS-NT) during the initial consultation period for this OEMP.

## Summary of consultation effort:

- On 9 February 2024 Santos emailed DEPWS-NT to advise it of preliminary consultation regarding proposed activities for consultation to be managed under the Coastal Waters OEMP and Barossa Production Operations EP. [Con-3787]
- The email advised that Santos was seeking information to better understand:
  - o if you are from a government Department or agency, how the proposed Production Operations activities may be relevant to your department or agency; and
  - what (if any) functions, interests or activities you or your organisation have that may be affected by the proposed Production Operations activities. 0
- The email included information on the regulatory process for the activities in Commonwealth and NT jurisdictions and links to a Santos information booklet on the proposed activities and a NOPSEMA brochure on the consultation process and details of how to contact Santos to register as a Relevant Person. The email stated that the consultation phase would commence on 11 March 2024 and close on 9 April 2024.
- On 9 February 2024 DEPWS-NT emailed Santos in response to the email of 11 March 2024 and raised a question related to produced water discharge during Production Operations activities. [Con-3823]
- On 11 March 2024 Santos emailed DEPWS-NT further to the previous correspondence, to advise that it had commenced the consultation phase which would run until 9 April 2024. In addition to the previous information again being provided, Santos provided information on Relevant Persons' entitlements under the regulatory processes, details of how to provide feedback and a reminder of the closing date for consultation. [Con-3793]
- On 3 May 2024, Santos emailed DEPWS-NT further to the previous correspondence, to advise it has extended the consultation period until 16 May 2024. In providing this extension of time, Santos advised that the information in the booklet and factsheet had been updated to account for an additional risk associated with the proposed activity. Santos confirmed that consultation will close on the revised date unless Santos hears otherwise from DEPWS-NT. [Con-4966]
- On 10 July 2024 Santos emailed DEPWS-NT to advise the consultation period had been completed. Santos advised it consultation had now closed for the purpose of Santos finalising and submitting environment plans for these activities to government regulators for assessment. [Con-5086]
- On 15 October 2024 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that the EP had been submitted to NOPSEMA for assessment, provided a link to the full EP on the NOPSEMA website and Santos' contact details for any queries. The Update also stated that the OEMP would be submitted to the NT-DITT for assessment. [Con- 5982]
- On 23 January 2025 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that, following a request from NOPSEMA to provide further information, Santos had re-submitted the EP with some amendments in December 2024 for further assessment. The Update also re-stated that the OEMP would be submitted to the NT-DITT for assessment. [Con-6036]
- No further correspondence or feedback was received from DEPWS-NT.

Summary of response by Relevant Person	Assessment of merits	Santos' Response Statement	OEMP reference
DEPWS-NT did not provide any comments on the Coastal Waters OEMP.	Santos has considered the matters raised in the OEMP A request from DEPWS-NT for some technical information was outside the scope of the Coastal Waters OEMP.	No response required.	Not applicable.

Department of Agriculture and Fisheries (DAF) formerly the Department of Industry, Tourism and Trade – Fisheries Division (DITT-NT Fisheries) during the initial consultation period for this OEMP

Summary of consultation effort:

- On 9 February 2024 Santos emailed DITTNT Fisheries to advise it of preliminary consultation regarding proposed activities for consultation to be managed under the Coastal Waters OEMP and Barossa Production Operations EP. [Con-3787]
- The email advised that Santos was seeking information to better understand:
  - o if you are from a government Department or agency, how the proposed Production Operations activities may be relevant to your department or agency; and
  - what (if any) functions, interests or activities you or your organisation have that may be affected by the proposed Production Operations activities.
- The email included information on the regulatory process for the activities in Commonwealth and NT jurisdictions and links to a Santos information booklet on the proposed activities and a NOPSEMA brochure on the consultation process and details of how to contact Santos to register as a Relevant Person. The email stated that the consultation phase would commence on 11 March 2024 and close on 9 April 2024.
- On 15 February 2024, the DITT NT Fisheries emailed Santos and provided contact details for appropriate persons within DITT NT Fisheries for consultation. It also nominated the NT Seafood Council, NT Guided Fishing Association and Amateur Fishing Association as other organisations that should be consulted and provided contact details. [Con-3826]
- On 15 February 2024, Santos emailed DITT NT Fisheries and confirmed it was consulting with those organisations and that it would add the Chief Scientist from DITT NT Fisheries as advised. Santos offered to meet in Darwin during 11-15 March 2024. [Con-3827]
- On 15 February 2024, DITT NT Fisheries advised it would meet with Santos while in Darwin and to please advise of specific questions to discuss. [Con-3828]
- On 11 March 2024 Santos emailed DITT NT Fisheries further to the previous correspondence, to advise that it had commenced the consultation phase which would run until 9 April 2024. In addition to the previous information again being provided. Santos provided information on Relevant Persons' entitlements under the regulatory processes, details of how to provide feedback and a reminder of the closing date for consultation. [Con-3793]
- On 3 April 2024 Santos met with DITT NT Fisheries. The Barossa Production Operations video was shown and potential impacts from planned activities and control measures to reduce impacts to ALARP was discussed. DITT NT Fisheries sought further information on seabed disturbance, planned discharges from the floating production system, and water discharge modelling. Santos confirmed that vessels would comply with MARPOL requirements and advised on fishery exclusion zones. No objections or claims were raised by DITT NT Fisheries. [Con-3832]
- On 17 April 2024 Santos emailed DITT NT Fisheries to provide a copy of the Minutes from the meeting of 3 April 2024 regarding the OEMP and EP. [Con-3833]
- On 10 July 2024 Santos emailed DITT NT Fisheries to advise the consultation period had been completed. Santos advised it considered that consultation had now closed for the purpose of Santos finalising and submitting environment plans for these activities to government regulators for assessment. [Con-5086]
- On 15 October 2024 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that the EP had been submitted to NOPSEMA for assessment, provided a link to the full EP on the NOPSEMA website and Santos' contact details for any queries. The Update also stated that the OEMP would be submitted to the NT-DITT for assessment. [Con- 5982]
- On 23 January 2025 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that, following a request from NOPSEMA to provide further information, Santos had re-submitted the EP with some amendments in December 2024 for further assessment. The Update also re-stated that the OEMP would be submitted to the NT-DITT for assessment. [Con-6036]
- No further correspondence or feedback was received from DITT NT Fisheries.

Summary of response by Relevant Person	Assessment of merits	Santos' Response Statement
----------------------------------------	----------------------	----------------------------



**OEMP** reference

DITT NT Fisheries requested information on seabed disturbance, planned discharges from the floating production system, and water	Santos has considered the matters raised in the OEMP. Santos provided the information requested at the meeting.	No additional response required.
discharge modelling.	Santos will comply with MARPOL requirements to reduce impacts and risks from planned discharges to ALARP.	
Department of Logistics and Infrastructure (DLI), formerly the	Department of Infrastructure, Planning and Logistics – Transport and Civil (DIPL-NT) du	uring the initial consultation period for this OEMP.
Summary of consultation effort:		
On 9 February 2024 Santos emailed DIPL NT to advise it of pr	eliminary consultation regarding proposed activities for consultation to be managed under the	Coastal Waters OEMP and Barossa Production Operation
• The email advised that Santos was seeking information to bette	er understand:	
o if you are from a government Department or agency, how the proposed Production Operations activities may be relevant to your department or agency; and		
<ul> <li>what (if any) functions, interests or activities you or yo</li> </ul>	ur organisation have that may be affected by the proposed Production Operations activities.	
• The email included information on the regulatory process for the activities in Commonwealth and NT jurisdictions and links to a Santos information booklet on the proposed activities and a NOPSEMA brochure of Santos to register as a Relevant Person. The email stated that the consultation phase would commence on 11 March 2024 and close on 9 April 2024		

- On 12 February 2024, the DIPL NT identified itself as an interested party for all marine infrastructure projects in the NT and specifically around the Darwin Harbour. [Con-3820]
- On 15 February 2024, Santos emailed DIPL NT and advised it is planning to be in Darwin the week of March 11-15 and seeking a meeting on respective works in Darwin Harbour during 2024, including Darwin Pipeline Duplication, Mandorah Marine Facilities upgrade and Barossa Production Operations and reminded that DIPL provided input during the EPA assessment process. Santos sought a meeting during that week or alternatively earlier on Teams. [Con-3830]
- On 15 February 2024 DIPL-NT confirmed via email that the meeting arrangements were convenient. [Con-3831]
- On 11 March 2024 Santos emailed DIPL NT further to the previous correspondence, to advise that it had commenced the consultation phase which would run until 9 April 2024. In addition to the previous information again being provided, Santos provided information on Relevant Persons' entitlements under the regulatory processes, details of how to provide feedback and a reminder of the closing date for consultation. [Con-3793]
- On 20 March 2024, held a meeting with DIPL NT at which DIPL-NT advised it had read the information provided on the Coastal Waters OEMP and Barossa Production Operations EP and did not require a briefing. The meeting instead focused on operational matters associated with coming Santos and DIPL-NT activities in Darwin Harbour. [Con-5632].
- On 10 July 2024 Santos emailed DIPL-NT to advise the consultation period had been completed. Santos advised it consultation had now closed for the purpose of Santos finalising and submitting environment plans for these activities to government regulators for assessment. [Con-5086]
- On 15 October 2024 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that the EP had been submitted to NOPSEMA for assessment, provided a link to the full EP on the NOPSEMA website and Santos' contact details for any queries. The Update also stated that the OEMP would be submitted to the NT-DITT for assessment. [Con- 5982]
- On 23 January 2025 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that, following a request from NOPSEMA to provide further information, Santos had re-submitted the EP with some amendments in December 2024 for further assessment. The Update also re-stated that the OEMP would be submitted to the NT-DITT for assessment. [Con-6036]
- No further correspondence or feedback was received from DIPL-NT.

Summary of response by Relevant Person	Assessment of merits	Santos' Response Statement
DIPL-NT provided a response on Santos' proposed activities in NT waters stating it had read the information provided on the Coastal Waters OEMP and Barossa Production Operations EP and did not require a briefing and instead discussed operational matters associated with coming Santos and DIPL-NT activities in Darwin Harbour.	Santos has considered the matters raised The information request from DIPL-NT with respect to NT waters is outside the scope of this OEMP.	No response required.

NT Fire and Emergency Services, formerly the Department of Police, Fire and Emergency Services – NT (NT Police, Fire and Emergency Services) during the initial consultation period for this OEMP.

Summary of consultation effort:

- On 9 February 2024 Santos emailed NT Police, Fire and Emergency Services to advise it of preliminary consultation regarding proposed activities for consultation to be managed under the Coastal Waters OEMP and Barossa Production Operations EP. [Con-3787]
- The email advised that Santos was seeking information to better understand:
  - o if you are from a government Department or agency, how the proposed Production Operations activities may be relevant to your department or agency; and
  - what (if any) functions, interests or activities you or your organisation have that may be affected by the proposed Production Operations activities. 0
- The email included information on the regulatory process for the activities in Commonwealth and NT jurisdictions and links to a Santos information booklet on the proposed activities and a NOPSEMA brochure on the consultation process and details of how to contact Santos to register as a Relevant Person. The email stated that the consultation phase would commence on 11 March 2024 and close on 9 April 2024.
- On 11 March 2024 Santos emailed NT Police, Fire and Emergency Services further to the previous correspondence, to advise that it had commenced the consultation phase which would run until 9 April 2024. In addition to the previous information again being provided, Santos provided information on Relevant Persons' entitlements under the regulatory processes, details of how to provide feedback and a reminder of the closing date for consultation. [Con-3793]
- On 3 April 2024 Santos phoned NT Police, Fire and Emergency Services and spoke to a team member left a message regarding consultation for Production Operations activities.
- On 6 May 2024, Santos emailed NT Police, Fire and Emergency Services further to previous emails to advise it had extended the consultation period until 20 May 2024. In providing this extension of time, Santos advised that the information in the booklet and factsheet had been updated to account for an additional risk associated with the proposed activity. Santos confirmed that consultation will close on the revised date unless Santos hears otherwise from NT Police, Fire and Emergency Services. [Con-4383]
- On 10 July 2024 Santos emailed NT Police, Fire and Emergency Services to advise the consultation period had been completed. Santos advised NT Police, Fire and Emergency Services that it considered consultation had now closed for the purpose of Santos finalising and submitting environment plans for these activities to government regulators for assessment. [Con-5086]



Commercial fisheries are described in Section 3.2.14.1.
Control measures for planned activities are described in Section 6.
Control measures for unplanned events are described in Section 7.

ations EP. [Con-3787]

n the consultation process and details of how to contact

OEMP reference
Not applicable.

- On 15 October 2024 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that the EP had been submitted to NOPSEMA for assessment, provided a link to the full EP on the NOPSEMA website and Santos' contact details for any queries. The Update also stated that the OEMP would be submitted to the NT-DITT for assessment. [Con- 5982]
- On 23 January 2025 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that, following a request from NOPSEMA to provide further information. Santos had re-submitted the EP with some amendments in December 2024 for further assessment. The Update also re-stated that the OEMP would be submitted to the NT-DITT for assessment. [Con-6036]

Notwithstanding the consultation information provided and the steps described above, no comments or input were received on this OEMP, from NT Police, Fire and Emergency Services

Summary of response by Relevant Person	Assessment of merits	Santos' Response Statement	OEMP reference	
No response was received from NT Police, Fire and Emergency Services.	Santos considers it has provided sufficient information and a reasonable period of time for consultation. Santos considers Section 25 consultation requirements to have been met.	No response required.	Not applicable.	

## Department of Lands Planning and the Environment (DLI), formerly within the Department of Territory Families, Housing and Communities, NT Heritage branch (DTFHC NT Heritage) during the course of initial consultation for this OEMF

- On 9 February 2024 Santos emailed DTFHC NT Heritage to advise it of preliminary consultation regarding proposed activities for consultation to be managed under the Coastal Waters OEMP and Barossa Production Operations EP. [Con-3787]
- The email advised that Santos was seeking information to better understand:
  - o if you are from a government Department or agency, how the proposed Production Operations activities may be relevant to your department or agency; and
  - what (if any) functions, interests or activities you or your organisation have that may be affected by the proposed Production Operations activities. 0
- The email included information on the regulatory process for the activities in Commonwealth and NT jurisdictions and links to a Santos information booklet on the proposed activities and a NOPSEMA brochure on the consultation process and details of how to contact Santos to register as a Relevant Person. The email stated that the consultation phase would commence on 11 March 2024 and close on 9 April 2024.
- On 13 February 2024, the DTFHC NT Heritage advised it should be consulted throughout this process under the Heritage Act 2011 and the Underwater Cultural Heritage Act 2018 and asked that the generic email heritage.branch@nt.gov.au be used and signed off with a name. [Con-3821]
- On 13 February 2024, Santos emailed DTFHC NT Heritage to arrange a meeting to provide a more detailed briefing on how Santos is approaching the requirements and seeks to meet during the week of March 11- 15 in Darwin or via Teams another week. [Con-3822]
- On 14 February 2024, Santos emailed DTFHC NT Heritage and advised it will send an invitation via Teams. [Con-3825]
- On 15 February 2024, Santos emailed the DTFHC NT Heritage and confirmed it would send a meeting request. [Con-3829]
- On 11 March 2024 Santos emailed DTFHC NT Heritage further to the previous correspondence, to advise that it had commenced the consultation phase which would run until 9 April 2024. In addition to the previous information again being provided, Santos provided information on Relevant Persons' entitlements under the regulatory processes, details of how to provide feedback and a reminder of the closing date for consultation. [Con-3793]
- On 12 March 2024 Santos met with DTFHC NT Heritage. At the meeting DTFHC NT Heritage stated that it did not require a separate full consultation session on the Coastal Waters OEMP and Barossa Production Operations EP as its requirements will have already been met during the Project's DPD construction activities. As a result, the meeting instead focused on how the requirements will be met during Darwin Pipeline Duplication construction activities in NT and Commonwealth waters and is not relevant to this OEMP. [Con-4970]
- On 13 March 2024, Santos emailed the DTFHC NT Heritage minutes from the meeting held on 12 March 2024, [Con-4970]
- On 19 March 2024, the NT Heritage Branch confirmed the minutes from the meeting held on 12 March 2024. [Con-4972]
- On 10 July 2024 Santos emailed DTFHC NT Heritage to advise the consultation period had been completed. Santos advised it considered that consultation had now closed for the purpose of Santos finalising and submitting environment plans for these activities to government regulators for assessment. [Con-5086]
- On 15 October 2024 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that the EP had been submitted to NOPSEMA for assessment, provided a link to the full EP on the NOPSEMA website and Santos' contact details for any queries. The Update also stated that the OEMP would be submitted to the NT-DITT for assessment. [Con- 5982]
- On 23 January 2025 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that, following a request from NOPSEMA to provide further information, Santos had re-submitted the EP with some amendments in December 2024 for further assessment. The Update also re-stated that the OEMP would be submitted to the NT-DITT for assessment. ICon-60361
- No further correspondence or feedback was received from DTFHC NT Heritage.



Summary of response by Relevant Person	Assessment of merits	Santos' Response Statement	OEMP reference	
DTFHC-NT-Heritage provided a response that its requirements will have already been met during the project's DPD construction activities in NT and Commonwealth waters.	The project's construction activities are not within the scope of the Coastal Waters OEMP.DPD construction activities in NT waters are already approved under NT legislation. DPD construction activities in Commonwealth waters were approved under a separate EP assessed by NOPSEMA. Santos considers it has provided sufficient information and a reasonable period of time for consultation.	No response required.	Not applicable.	

- On 9 February 2024 Santos emailed EPA NT to advise it of preliminary consultation regarding proposed activities for consultation to be managed under the Coastal Waters OEMP and Barossa Production Operations EP. [Con-3787]
- The email advised that Santos was seeking information to better understand:
  - o if you are from a government Department or agency, how the proposed Production Operations activities may be relevant to your department or agency; and
  - what (if any) functions, interests or activities you or your organisation have that may be affected by the proposed Production Operations activities. 0
- The email included information on the regulatory process for the activities in Commonwealth and NT jurisdictions and links to a Santos information booklet on the proposed activities and a NOPSEMA brochure on the consultation process and details of how to contact Santos to register as a Relevant Person. The email stated that the consultation phase would commence on 11 March 2024 and close on 9 April 2024.
- On 11 March 2024 Santos emailed EPA NT further to the previous correspondence, to advise that it had commenced the consultation phase which would run until 9 April 2024. In addition to the previous information again being provided, Santos provided information on Relevant Persons' entitlements under the regulatory processes, details of how to provide feedback and a reminder of the closing date for consultation. [Con-3793]
- On 3 April 2024 Santos phoned EPA NT and spoke to a team member regarding consultation for Production Operations activities. The team member advised that the emails sent on 9 February 2024 and 11 March 2024 had been forwarded to the assessments team which had advised that engagement with Santos had concluded.
- On 3 May 2024, Santos emailed EPA NT further to the previous correspondence, to advise it had extended the consultation period until 16 May 2024. In providing this extension of time, Santos advised that the information in the booklet and factsheet had been updated to account for an additional risk associated with the proposed activity. Santos confirmed that consultation will close on the revised date unless Santos hears otherwise from the EPA NT. [Con-4966]
- On 10 July 2024 Santos emailed EPA NT to advise the consultation period had been completed. Santos advised it consultation had now closed for the purpose of Santos finalising and submitting environment plans for these activities to government regulators for assessment. [Con-5086]
- On 15 October 2024 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that the EP had been submitted to NOPSEMA for assessment, provided a link to the full EP on the NOPSEMA website and Santos' contact details for any queries. The Update also stated that the OEMP would be submitted to the NT-DITT for assessment. [Con- 5982]
- On 23 January 2025 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that, following a request from NOPSEMA to provide further information. Santos had re-submitted the EP with some amendments in December 2024 for further assessment. The Update also re-stated that the OEMP would be submitted to the NT-DITT for assessment. [Con-6036]
- Notwithstanding the consultation information provided and the steps described above, no comments or input were received on the Coastal Waters OEMP from EPA NT.

Summary of response by Relevant Person	Assessment of merits	Santos' Response Statement	OEMP reference
EPA NT advised that engagement with Santos had concluded.	Santos considers it has provided sufficient information and a reasonable period of time for consultation. Santos considers Section 25 consultation requirements to have been met.	No response required.	Not applicable.

## NT Parks and Wildlife Commission

- On 9 February 2024 Santos emailed NT Parks and Wildlife Commission to advise it of preliminary consultation regarding proposed activities for consultation to be managed under the Coastal Waters OEMP and Barossa Production Operations EP. [Con-3787]
- The email advised that Santos was seeking information to better understand:
  - o if you are from a government Department or agency, how the proposed Production Operations activities may be relevant to your department or agency; and
  - what (if any) functions, interests or activities you or your organisation have that may be affected by the proposed Production Operations activities.
- The email included information on the regulatory process for the activities in Commonwealth and NT jurisdictions and links to a Santos information booklet on the proposed activities and a NOPSEMA brochure on the consultation process and details of how to contact Santos to register as a Relevant Person. The email stated that the consultation phase would commence on 11 March 2024 and close on 9 April 2024.
- On 11 March 2024 Santos emailed NT Parks and Wildlife Commission further to the previous correspondence, to advise that it had commenced the consultation phase which would run until 9 April 2024. In addition to the previous information again being provided, Santos provided information on Relevant Persons' entitlements under the regulatory processes, details of how to provide feedback and a reminder of the closing date for consultation. [Con-3793]
- On 4 April 2024 Santos phoned NT Parks and Wildlife Commission and left a message with a team member regarding consultation for Production Operations activities.
- On 6 May 2024, Santos emailed NT Parks and Wildlife Commission further to the previous correspondence, to advise it had extended the consultation period until 20 May 2024. In providing this extension of time, Santos advised that the information in the booklet and factsheet had been updated to account for an additional risk associated with the proposed activity. Santos confirmed that consultation will close on the revised date unless Santos hears otherwise from the NT Parks and Wildlife Commission. [Con-4377]
- On 10 July 2024 Santos emailed NT Parks and Wildlife Commission to advise the consultation period had been completed. Santos advised NT Parks and Wildlife Commission that it considered consultation had now closed for the purpose of Santos finalising and submitting environment plans for these activities to government regulators for assessment. [Con-5086]
- On 15 October 2024 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that the EP had been submitted to NOPSEMA for assessment, provided a link to the full EP on the NOPSEMA website and Santos' contact details for any queries. The Update also stated that the OEMP would be submitted to the NT-DITT for assessment. [Con- 5982]
- On 23 January 2025 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that, following a request from NOPSEMA to provide further information, Santos had re-submitted the EP with some amendments in December 2024 for further assessment. The Update also re-stated that the OEMP would be submitted to the NT-DITT for assessment. [Con-6036]
- Notwithstanding the consultation information provided and the steps described above, no comments or input were received on this OEMP from NT Parks and Wildlife Commission.



builtos						
Summary of response by Relevant Person	Assessment of merits Santos' Response Statement OEMP reference					
No response was received from NT Parks and Wildlife Commission.	Santos considers it has provided sufficient information and a reasonable period of time for consultation.	No response required.	Not applicable.			
Santos considers Section 25 consultation requirements to have been met.						
Department of Tourism and Hospitality (DTH), formerly Tourism	NT during the course of initial consultation for the Coastal Waters OEMP.					
Summary of consultation effort:						
On 9 February 2024 Santos emailed Tourism NT to advise it of	preliminary consultation regarding proposed activities for consultation to be managed under t	he Coastal Waters OEMP and Barossa Production Operations EP.	[Con-3787]			
The email advised that Santos was seeking information to bette	er understand:					
$\circ$ if you are from a government Department or agency, h	now the proposed Production Operations activities may be relevant to your department or age	ncy; and				
<ul> <li>what (if any) functions, interests or activities you or you</li> </ul>	ur organisation have that may be affected by the proposed Production Operations activities.					
<ul> <li>The email included information on the regulatory process for the Santos to register as a Relevant Person. The email stated that</li> </ul>	e activities in Commonwealth and NT jurisdictions and links to a Santos information booklet or the consultation phase would commence on 11 March 2024 and close on 9 April 2024.	n the proposed activities and a NOPSEMA brochure on the consulta	tion process and details of how to contact			
<ul> <li>On 11 March 2024 Santos emailed Tourism NT further to the pr on Relevant Persons' entitlements under the regulatory process</li> </ul>	revious correspondence, to advise that it had commenced the consultation phase which would ses, details of how to provide feedback and a reminder of the closing date for consultation. [C	d run until 9 April 2024. In addition to the previous information again on-3793]	being provided, Santos provided information			
On 4 April 2024 Santos phoned Tourism NT and left a message	with a team member regarding consultation for Production Operations activities.					
<ul> <li>On 6 May 2024, Santos emailed Tourism NT further to the prevupdated to account for an additional risk associated with the pro-</li> </ul>	ious correspondence, to advise it had extended the consultation period until 20 May 2024. In oposed activity. Santos confirmed that consultation will close on the revised date unless Santo	providing this extension of time, Santos advised that the information be hears otherwise from the Tourism NT. [Con-4375]	in the booklet and factsheet had been			
<ul> <li>On 10 July 2024 Santos emailed Tourism NT to advise the cons activities to government regulators for assessment. [Con-5086]</li> </ul>	sultation period had been completed. Santos advised Tourism NT Commission that it conside	red consultation had now closed for the purpose of Santos finalising	and submitting environment plans for these			
<ul> <li>On 15 October 2024 Santos emailed the Barossa Project Quart NOPSEMA for assessment, provided a link to the full EP on the</li> </ul>	erly Update to all stakeholders on its distribution list, including all those consulted during prep NOPSEMA website and Santos' contact details for any queries. The Update also stated that	aration of the Production Operations EP and OEMP. The Update ac the OEMP would be submitted to the NT-DITT for assessment. [Co	dvised that the EP had been submitted to n- 5982]			
<ul> <li>On 23 January 2025 Santos emailed the Barossa Project Quart NOPSEMA to provide further information, Santos had re-submit</li> </ul>	erly Update to all stakeholders on its distribution list, including all those consulted during prep tted the EP with some amendments in December 2024 for further assessment. The Update a	aration of the Production Operations EP and OEMP. The Update ac so re-stated that the OEMP would be submitted to the NT-DITT for	dvised that, following a request from assessment. [Con-6036]			
<ul> <li>Notwithstanding the consultation information provided and the s</li> </ul>	steps described above, no comments or input were received on the Coastal Waters OEMP fro	om Tourism NT.				
Summary of response by Relevant Person	Assessment of merits	Santos' Response Statement	OEMP reference			
No response was received from Tourism NT.	received from Tourism NT. Santos considers it has provided sufficient information and a reasonable period of time for No response required Not applicable					
	consultation.					
	Santos considers Section 25 consultation requirements to have been met.					
Section 25(1)(c) of the OPGGS(E)R: Department of the responsi	ble Northern Territory Minister					
Department of Mining and Energy (DME), formerly the Departme	ent of Industry, Tourism and Trade, NT – Mines & Energy (DITT NT Energy) during the o	course of initial consultation for the Coastal Waters OEMP.				
Summary of consultation effort:						
On 9 February 2024 Santos emailed DITT NT Energy to advise	it of preliminary consultation regarding proposed activities for consultation to be managed un	der the Coastal Waters OEMP and Barossa Production Operations	EP. [Con-3787]			
The email advised that Santos was seeking information to bette	r understand:					
<ul> <li>if you are from a government Department or agency, how the proposed Production Operations activities may be relevant to your department or agency; and</li> </ul>						
• what (if any) functions, interests or activities you or your organisation have that may be affected by the proposed Production Operations activities.						
• The email included information on the regulatory process for the activities in Commonwealth and NT jurisdictions and links to a Santos information booklet on the proposed activities and a NOPSEMA brochure on the consultation process and details of now to contact Santos to register as a Relevant Person. The email stated that the consultation phase would commence on 11 March 2024 and close on 9 April 2024.						
<ul> <li>On 11 March 2024 Santos emailed DITT NT Energy further to the previous correspondence, to advise that it had commenced the consultation phase which would run until 9 April 2024. In addition to the previous information again being provided, Santos provided information on Relevant Persons' entitlements under the regulatory processes, details of how to provide feedback and a reminder of the closing date for consultation. [Con-3793]</li> </ul>						
On 4 April 2024 Santos phoned DITT NT Energy and spoke to the nominated contact person who advised that DITT NT Energy input on Barossa Production Operations would be via assessment of Operations Environmental Management Plans under NT legislation.						
<ul> <li>On 3 May 2024, Santos emailed DITT NT Energy further to previous correspondence to advise it had extended the consultation period until 16 May 2024. In providing this extension of time, Santos advised that the information in the booklet and factsheet had been updated to account for an additional risk associated with the proposed activity. Santos confirmed that consultation will close on the revised date unless Santos hears otherwise from DITT NT Energy. [Con-4382]</li> </ul>						
On 7 May 2024 DITT NT Energy emailed Santos acknowledging the email sent on 3 May 2024. [Con-4968]						
<ul> <li>On 10 July 2024 Santos emailed DITT NT Energy to advise the government regulators for assessment. [Con-5086]</li> </ul>	consultation period had been completed. Santos advised it considered that consultation had	now closed for the purpose of Santos finalising and submitting envir	onment plans for these activities to			
On 15 October 2024 Santos emailed the Barossa Project Quart NOPSEMA for assessment, provided a link to the full EP on the	erly Update to all stakeholders on its distribution list, including all those consulted during prep NOPSEMA website and Santos' contact details for any queries. The Update also stated that	aration of the Production Operations EP and OEMP. The Update ac the OEMP would be submitted to the NT-DITT for assessment. [Co	lvised that the EP had been submitted to on- 5982]			

On 23 January 2025 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that, following a request from • NOPSEMA to provide further information, Santos had re-submitted the EP with some amendments in December 2024 for further assessment. The Update also re-stated that the OEMP would be submitted to the NT-DITT for assessment. [Con-6036]

• No further correspondence or feedback was received from DITT-NT-Energy.

|--|



OEMP reference
Not applicable.

No objections or claims were raised by DITT NT Energy which advised it will provide comments on Production Operations activities via its assessment of Operations Environmental Management Plans under NT legislation.	Santos considers it has provided sufficient information and a reasonable period of time for consultation. Santos considers Section 25 consultation requirements to have been met.	No response required.	Not applicable.
4.8.3 WA Government Agency or Authority			

## Table 4-13: Consultation Summary Table - WA Government Agency or Authority

Section 25(1)(b) of the OPGGS(E)R: Western Australia agency or authority to which the activities to be carried out under the environment plan may be relevant

## Department of Biodiversity, Conservation and Attractions (DBCA WA)

Summary of consultation effort:

- On 9 February 2024 Santos emailed DBCA WA to advise it of preliminary consultation regarding proposed activities for consultation to be managed under the Coastal Waters OEMP and Barossa Production Operations EP. [Con-3787]
- The email advised that Santos was seeking information to better understand:
  - o if you are from a government Department or agency, how the proposed Production Operations activities may be relevant to your department or agency; and
  - what (if any) functions, interests or activities you or your organisation have that may be affected by the proposed Barossa Production Operations activities.
- The email included information on the regulatory process for the activities in Commonwealth and NT jurisdictions and links to a Santos information booklet on the proposed activities and a NOPSEMA brochure on the consultation process and details of how to contact Santos to register as a Relevant Person. The email stated that the consultation phase would commence on 11 March 2024 and close on 9 April 2024.
- On 11 March 2024 Santos emailed DBCA WA further to the previous correspondence, to advise that it had commenced the consultation phase which would run until 9 April 2024. In addition to the previous information again being provided, Santos provided information on Relevant Persons' entitlements under the regulatory processes, details of how to provide feedback and a reminder of the closing date for consultation. [Con-3793]
- On 4 April 2024 Santos phoned DBCA WA and left a voice mail message regarding consultation for Production Operations activities.
- On 4 April 2024 DBCA-WA emailed Santos with advice that based on the documentation provided for review and other readily available information, DBCA WA has no comments in relation to its responsibilities under the Conservation and Land Management Act 1984 and Biodiversity Conservation Act 2016. [Con-3836]
- On 7 May 2024. Santos emailed DBCA WA further to its response on 4 April 2024, to advise it had extended the consultation period until 21 May 2024. In providing this extension of time. Santos advised that the information in the booklet and factsheet had been updated to account for an additional risk associated with the proposed activity. Santos confirmed that consultation will close on the revised date unless Santos hears otherwise from DBCA WA. [Con-4367]
- On 10 July 2024 Santos emailed DBCA WA to advise the consultation period had been completed. Santos advised it consultation had now closed for the purpose of Santos finalising and submitting environment plans for these activities to government regulators for assessment. [Con-5086]
- On 15 October 2024 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that the EP had been submitted to NOPSEMA for assessment, provided a link to the full EP on the NOPSEMA website and Santos' contact details for any queries. The Update also stated that the OEMP would be submitted to the NT-DITT for assessment. [Con- 5982]
- On 23 January 2025 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that, following a request from NOPSEMA to provide further information, Santos had re-submitted the EP with some amendments in December 2024 for further assessment. The Update also re-stated that the OEMP would be submitted to the NT-DITT for assessment. [Con-6036]
- No further correspondence or feedback was received from DBCA-WA.

Summary of response by Relevant Person	Assessment of merits	Santos' Response Statement	OEMP	
DBCA WA responded that it had no comments in relation to its responsibilities under the <i>Conservation and Land Management Act 1984</i> and <i>Biodiversity Conservation Act 2016</i> .	Santos notes DBCA WA's response is not relevant to the Coastal Waters OEMP as the OEMP EMBA does not extend beyond Commonwealth Waters off the Northern Territory.	No response required.	Not ap	
Department of Primary Industries and Regional Development – Fisheries (DPIRD Fisheries)				

Summary of consultation effort:

- On 9 February 2024 Santos emailed DPIRD Fisheries to advise it of preliminary consultation regarding proposed activities for consultation to be managed under the Coastal Waters OEMP and Barossa Production Operations EP. [Con-3787]
- The email advised that Santos was seeking information to better understand:
  - o if you are from a government Department or agency, how the proposed Production Operations activities may be relevant to your department or agency; and
  - what (if any) functions, interests or activities you or your organisation have that may be affected by the proposed Production Operations activities.
- The email included information on the regulatory process for the activities in Commonwealth and NT jurisdictions and links to a Santos information booklet on the proposed activities and a NOPSEMA brochure on the consultation process and details of how to contact Santos to register as a Relevant Person. The email stated that the consultation phase would commence on 11 March 2024 and close on 9 April 2024.
- On 11 March 2024 Santos emailed DPIRD Fisheries further to the previous correspondence, to advise that it had commenced the consultation phase which would run until 9 April 2024. In addition to the previous information again being provided, Santos provided information on Relevant Persons' entitlements under the regulatory processes, details of how to provide feedback and a reminder of the closing date for consultation. [Con-3793]
- On 4 April 2024 Santos phoned DPIRD Fisheries and left a message with a team regarding consultation for Production Operations activities.
- On 8 May 2024. Santos emailed DPIRD Fisheries further to the previous correspondence, to advise it had extended the consultation period until 22 May 2024. In providing this extension of time. Santos advised that the information in the booklet and factsheet had been updated to account for an additional risk associated with the proposed activity. Santos confirmed that consultation will close on the revised date unless Santos hears otherwise from DPIRD Fisheries. [Con-4370]
- On 10 July 2024 Santos emailed DPIRD Fisheries to advise the consultation period had been completed. Santos advised DPIRD-WA-Fisheries that it considered consultation had now closed for the purpose of Santos finalising and submitting environment plans for these activities to government regulators for assessment. [Con-5086]
- On 15 October 2024 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that the EP had been submitted to NOPSEMA for assessment, provided a link to the full EP on the NOPSEMA website and Santos' contact details for any queries. The Update also stated that the OEMP would be submitted to the NT-DITT for assessment. [Con- 5982]

S	a	n	t	D	S
			<u> </u>		

reference

oplicable.

On 23 January 2025 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that, following a request from NOPSEMA to provide further information. Santos had re-submitted the EP with some amendments in December 2024 for further assessment. The Update also re-stated that the OEMP would be submitted to the NT-DITT for assessment. [Con-6036]

## No objections or claims were raised by DPIRD Fisheries.

Summary of response by Relevant Person	Assessment of merits	Santos' Response Statement	OEM
No response was received from DPIRD Fisheries.	The EMBA for the Coastal Waters OEMP does not extend beyond Commonwealth Waters off the Northern Territory.	No response required.	Not a
	There are therefore no impacts from planned GEP operations activities in NT Coastal Waters that may affect the functions, interests or activities of any WA-based persons.		
	The worst-case credible spill scenario modelled for the EMBA for the Coastal Waters OEMP also shows there are no impacts from unplanned activities that may affect the functions, interests or activities of any WA-based persons.		
	Santos considers it has provided sufficient information and a reasonable period of time for consultation.		
	Santos considers Section 25 consultation requirements to have been met.		

## **Department of Transport (DoT WA)**

## Summary of consultation effort:

- On 9 February 2024 Santos emailed DoT WA to advise it of preliminary consultation regarding proposed activities for consultation to be managed under the Coastal Waters OEMP and Barossa Production Operations EP. [Con-3787]
- The email advised that Santos was seeking information to better understand:
  - o if you are from a government Department or agency, how the proposed Production Operations activities may be relevant to your department or agency; and
  - what (if any) functions, interests or activities you or your organisation have that may be affected by the proposed Production Operations activities.
- The email included information on the regulatory process for the activities in Commonwealth and NT jurisdictions and links to a Santos information booklet on the proposed activities and a NOPSEMA brochure on the consultation process and details of how to contact Santos to register as a Relevant Person. The email stated that the consultation phase would commence on 11 March 2024 and close on 9 April 2024.
- On 22 February 2024 DoT WA emailed Santos to advise that if there is a risk of a spill impacting State waters to please ensure that the Department of Transport is consulted. [Con-3834]
- On 11 March 2024 Santos emailed DoT WA further to the previous correspondence, to advise that it had commenced the consultation phase which would run until 9 April 2024. In addition to the previous information again being provided, Santos provided information on Relevant Persons' entitlements under the regulatory processes, details of how to provide feedback and a reminder of the closing date for consultation. [Con-3793]
- On 27 March 2024 Santos emailed DoT WA to confirm that the environment that may be affected (EMBA) modelled and provided for the Coastal Waters OEMP. does not show impact to WA State Waters. [Con-3835]
- On 5 April 2024 DoT WA emailed Santos to thank it for the clarification. [Con-3837]
- On 8 May 2024, Santos emailed DoT-WA further to the previous correspondence, to advise it had extended the consultation period until 22 May 2024. In providing this extension of time, Santos advised that the information in the booklet and factsheet had been updated to account for an additional risk associated with the proposed activity. Santos confirmed that consultation will close on the revised date unless Santos hears otherwise from DoT WA. [Con-4371]
- On 10 July 2024 Santos emailed DoT-WA to advise the consultation period had been completed. Santos advised it consultation had now closed for the purpose of Santos finalising and submitting environment plans for these activities to government regulators for assessment. [Con-5086]
- On 15 October 2024 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that the EP had been submitted to NOPSEMA for assessment, provided a link to the full EP on the NOPSEMA website and Santos' contact details for any queries. The Update also stated that the OEMP would be submitted to the NT-DITT for assessment. [Con- 5982]
- On 23 January 2025 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that, following a request from NOPSEMA to provide further information, Santos had re-submitted the EP with some amendments in December 2024 for further assessment. The Update also re-stated that the OEMP would be submitted to the NT-DITT for assessment. [Con-6036]

## No further correspondence or feedback was received from DoT WA.

Summary of response by Relevant Person	Assessment of merits	Santos' Response Statement	OEMF
DoT WA responded that it sought clarification on spill modelling given its marine pollution response authorities in WA water and that it should be consulted if there is a risk of a spill impacting State water	DoT WA's response is not relevant to the Coastal Waters OEMP as the OEMP EMBA does not extend beyond Commonwealth Waters off the Northern Territory.	No response required.	Not ap
State waters.	There are therefore no impacts from planned GEP operations activities in NT Coastal Waters that may affect the functions, interests or activities of any WA-based persons.		
	The worst-case credible spill scenario modelled for the EMBA for the Coastal Waters OEMP also shows there are no impacts from unplanned activities that may affect the functions, interests or activities of any WA-based persons.		
	DoT WA's response is addressed in the Barossa Production Operations EP		
Kimberley Ports Authority			

## Summary of consultation effort:

On 9 February 2024 Santos emailed Kimberley Ports Authority to advise it of preliminary consultation regarding proposed activities for consultation to be managed under the Coastal Waters OEMP and Barossa Production Operations EP. [Con-3787]



P reference

oplicable.

Preference

oplicable.

•	The email advised that	t Santos was seeking	g information to	better understand:

- o if you are from a government Department or agency, how the proposed Production Operations activities may be relevant to your department or agency; and
- what (if any) functions, interests or activities you or your organisation have that may be affected by the proposed Production Operations activities.
- The email included information on the regulatory process for the activities in Commonwealth and NT jurisdictions and links to a Santos information booklet on the proposed activities and a NOPSEMA brochure on the consultation process and details of how to contact Santos to register as a Relevant Person. The email stated that the consultation phase would commence on 11 March 2024and close on 9 April 2024.
- On 11 March 2024 Santos emailed Kimberley Ports Authority further to the previous correspondence, to advise that it had commenced the consultation phase which would run until 9 April 2024. In addition to the previous information again being provided, Santos provided information on Relevant Persons' entitlements under the regulatory processes, details of how to provide feedback and a reminder of the closing date for consultation. [Con-3793]
- On 4 April 2024 Santos phoned the Kimberley Ports Authority regarding consultation for Production Operations activities. A representative asked that the emails be re-sent and emails previously sent on 9 February and 11 March 2024 were resent the same day. [Con-4372]
- On 7 May 2024. Santos emailed Kimberley Ports Authority further to the previous correspondence, to advise it had extended the consultation period until 21 May 2024. In providing this extension of time, Santos advised that the information in the booklet and factsheet had been updated to account for an additional risk associated with the proposed activity. Santos confirmed that consultation will close on the revised date unless Santos hears otherwise from Kimberley Ports Authority [Con-4057]
- On 10 July 2024 Santos emailed Kimberley Ports Authority to advise the consultation period had been completed. Santos advised Kimberley Ports Authority that it considered consultation had now closed for the purpose of Santos finalising and submitting environment plans for these activities to government regulators for assessment. [Con-5086]
- On 15 October 2024 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that the EP had been submitted to NOPSEMA for assessment, provided a link to the full EP on the NOPSEMA website and Santos' contact details for any queries. The Update also stated that the OEMP would be submitted to the NT-DITT for assessment. [Con- 5982]
- On 23 January 2025 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that, following a request from NOPSEMA to provide further information, Santos had re-submitted the EP with some amendments in December 2024 for further assessment. The Update also re-stated that the OEMP would be submitted to the NT-DITT for assessment. [Con-6036]

Notwithstanding the consultation information provided and the steps described above, no comments or input were received on the Coastal Waters OEMP. from Kimberley Ports Authority.

Summary of response by Relevant Person	Assessment of merits	Santos' Response Statement	OEM
No response was received from Kimberley Ports Authority.	The EMBA for the Coastal Waters OEMP does not extend beyond Commonwealth Waters off the Northern Territory.	No response required.	Not a
	There are therefore no impacts from planned GEP operations activities in NT Coastal Waters that may affect the functions, interests or activities of any WA-based persons.		
	The worst-case credible spill scenario modelled for the EMBA for the Coastal Waters OEMP also shows there are no impacts from unplanned activities that may affect the functions, interests or activities of any WA-based persons.		
	Santos considers it has provided sufficient information and a reasonable period of time for consultation.		
	Santos considers Section 25 consultation requirements to have been met.		

## NA Marine Science Institution (WAMSI)

Summary of consultation effort:

- On 9 February 2024 Santos emailed WAMSI to advise it of preliminary consultation regarding proposed activities for consultation to be managed under the Coastal Waters OEMP and Barossa Production Operations EP. [Con-3787]
- The email advised that Santos was seeking information to better understand:
- if you are from a government Department or agency, how the proposed Production Operations activities may be relevant to your department or agency; and
- what (if any) functions, interests or activities you or your organisation have that may be affected by the proposed Production Operations activities.
- The email included information on the regulatory process for the activities in Commonwealth and NT jurisdictions and links to a Santos information booklet on the proposed activities and a NOPSEMA brochure on the consultation process and details of how to contact Santos to register as a Relevant Person. The email stated that the consultation phase would commence on 11 March 2024and close on 9 April 2024.
- On 13 March 2024 Santos emailed WAMSI further to the previous correspondence, to advise that it had commenced the consultation phase which would run until 9 April 2024. In addition to the previous information again being provided, Santos provided information on Relevant Persons' entitlements under the regulatory processes, details of how to provide feedback and a reminder of the closing date for consultation. [Con-3794]
- On 4 April 2024 Santos phoned WAMSI and left a voice mail message regarding consultation for Production Operations activities.
- On 2 May 2024, Santos emailed WAMSI, further to the previous correspondence, to advise it had extended the consultation period until 16 May 2024. In providing this extension of time, Santos advised that the information in the booklet and factsheet had been updated to account for an additional risk associated with the proposed activity. Santos confirmed that consultation will close on the revised date unless Santos hears otherwise from WAMSI. [Con-4368]
- On 10 July 2024 Santos emailed WAMSI to advise the consultation period had been completed. Santos advised WAMSI that it considered consultation had now closed for the purpose of Santos finalising and submitting environment plans for these activities to government regulators for assessment. [Con-5086]
- On 15 October 2024 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that the EP had been submitted to NOPSEMA for assessment, provided a link to the full EP on the NOPSEMA website and Santos' contact details for any queries. The Update also stated that the OEMP would be submitted to the NT-DITT for assessment. [Con- 5982]
- On 23 January 2025 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that, following a request from NOPSEMA to provide further information, Santos had re-submitted the EP with some amendments in December 2024 for further assessment. The Update also re-stated that the OEMP would be submitted to the NT-DITT for assessment. [Con-6036]

Notwithstanding the consultation information provided and the steps described above, no comments or input were received on the Coastal Waters OEMP from WAMSI.

Summary of response by Relevant Person	Assessment of merits	Santos' Response Statement	OEMP
----------------------------------------	----------------------	----------------------------	------



P reference

oplicable.

reference

			-
No response was received from WAMSI.	The EMBA for the Coastal Waters OEMP does not extend beyond Commonwealth Waters off the Northern Territory.	No response required.	Not a
	There are therefore no impacts from planned GEP operations activities in NT Coastal Waters that may affect the functions, interests or activities of any WA-based persons.		
	The worst-case credible spill scenario modelled for the EMBA for the Coastal Waters OEMP also shows there are no impacts from unplanned activities that may affect the functions, interests or activities of any WA-based persons.		
	Santos considers it has provided sufficient information and a reasonable period of time for consultation.		
	Santos considers Section 25 consultation requirements to have been met.		

## 4.8.4 Academic and Research Organisations

## Table 4-14: Consultation Summary Table - Academic and Research Organisations

Section 25 (1)(d) of the OPGGS(E)R: Persons or organisations whose functions, interests or activities may be affected by the activities to be carried out under the environment plan **Arafura Timor Research Facility** Summary of consultation effort: On 9 February 2024 Santos emailed AIMS, in its capacity as operator of the Arafura Timor Research Facility, to advise it of preliminary consultation regarding proposed activities for consultation to be managed under the Coastal Waters OEMP and Barossa Production Operations EP. [Con-3787] The email advised that Santos was seeking information to better understand: o if you are from a government Department or agency, how the proposed Production Operations activities may be relevant to your department or agency; and what (if any) functions, interests or activities you or your organisation have that may be affected by the proposed Production Operations activities. The email included information on the regulatory process for the activities in Commonwealth and NT jurisdictions and links to a Santos information booklet on the proposed activities and a NOPSEMA brochure on the consultation process and details of how to contact Santos to register as a Relevant Person. The email stated that the consultation phase would commence on 11 March 2024 and close on 9 April 2024. On 11 March 2024 Santos emailed AIMS further to the previous correspondence, to advise that it had commenced the consultation phase which would run until 9 April 2024. In addition to the previous information again being provided, Santos provided information on Relevant Persons' entitlements under the regulatory processes, details of how to provide feedback and a reminder of the closing date for consultation. [Con-3793] On 4 April 2024 Santos phoned AIMS, in its capacity as operator of the Arafura Timor Research, regarding consultation for Production Operations activities but was unable to leave a voice message. On 2 May 2024, Santos emailed AIMS, in its capacity as operator of the Arafura Timor Research facility, further to the previous correspondence, to advise it had extended the consultation period until 16 May 2024. In providing this extension of time, Santos advised that the information in the booklet and factsheet had been updated to account for an additional risk associated with the proposed activity. Santos confirmed that consultation will close on the revised date unless Santos hears otherwise from the Arafura Timor Research facility [Con-3854] On 10 July 2024 Santos emailed AIMS, in its capacity as operator of the Arafura Timor Research facility, to advise the consultation period had been completed. Santos advised it considered that consultation had now closed for the purpose of Santos finalising and submitting environment plans for these activities to government regulators for assessment. [Con-5130] On 15 October 2024 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that the EP had been submitted to NOPSEMA for assessment, provided a link to the full EP on the NOPSEMA website and Santos' contact details for any queries. The Update also stated that the OEMP would be submitted to the NT-DITT for assessment. [Con- 5982] On 23 January 2025 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that, following a request from NOPSEMA to provide further information, Santos had re-submitted the EP with some amendments in December 2024 for further assessment. The Update also re-stated that the OEMP would be submitted to the NT-DITT for assessment. [Con-6036] Notwithstanding the consultation information provided and the steps described above, no comments or input were received on the Coastal Waters OEMP from Arafura Timor Research Facility. Summary of response by Relevant Person Assessment of merits Santos' Response Statement **OEMP** reference Santos considers it has provided sufficient information and a No response was received from Arafura Timor Research Facility. No response required. Not applicable. reasonable period of time for consultation. Santos considers Section 25 consultation requirements to have been met. Australian Marine Sciences Association - NT (AMSA-NT) Summary of consultation effort: On 9 February 2024 Santos emailed AMSA-NT to advise it of preliminary consultation regarding proposed activities for consultation to be managed under the Coastal Waters OEMP and Barossa Production Operations EP. [Con-3787]

• The email advised that Santos was seeking information to better understand:

o if you are from a government Department or agency, how the proposed Production Operations activities may be relevant to your department or agency; and

what (if any) functions, interests or activities you or your organisation have that may be affected by the proposed Production Operations activities.

## applicable.



- The email included information on the regulatory process for the activities in Commonwealth and NT jurisdictions and links to a Santos information booklet on the proposed activities and a NOPSEMA brochure on the consultation process and details of how to contact Santos to register as a Relevant Person. The email stated that the consultation phase would commence on 11 March 2024 and close on 9 April 2024.
- On 11 March 2024 Santos emailed AMSA-NT further to the previous correspondence, to advise that it had commenced the consultation phase which would run until 9 April 2024. In addition to the previous information again being provided, Santos provided information on Relevant Persons' entitlements under the regulatory processes, details of how to provide feedback and a reminder of the closing date for consultation. [Con-3793]
- On 3 April 2024 Santos attempted to phone AMSA NT on three occasions without success. On 3 April 2024, Santos resent the emails sent on 9 February and 11 March 2024 and advised that the consultation is closing on April 9. [Con-3838]
- On 11 April 2024, AMSA-NT advised that it wished to be considered as 'Relevant Person' for the Coastal Waters OEMP and Barossa Production Operations EP. [Con-3839]
- On 8 May 2024. Santos emailed AMSA-NT further to previous correspondence, to advise it had extended the consultation period until 22 May 2024. In providing this extension of time. Santos advised that the information in the booklet and factsheet had been updated to account for an additional risk associated with the proposed activity. Santos confirmed that consultation will close on the revised date unless Santos hears otherwise from AMSA-NT.[Con-3927]
- On 25 June 2024 Santos returned a phone message left by AMSA-NT on 24 June 2024. AMSA-NT requested an extension until the end of that week for submittal of its comments. Santos followed-up with an email to AMSA-NT the same day agreeing to the extension. [Con-5053]
- On 30 June 2024 AMSA-NT emailed a letter to Santos with comments on Production Operations activities. A summary of the comments is provided below. [Con-5054]On 13 August 2024 Santos emailed AMSA-NT a letter in response to AMSA-NT's letter of 30 June. [Con-5351]
- On 15 October 2024 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that the EP had been submitted to NOPSEMA for assessment, provided a link to the full EP on the NOPSEMA website and Santos' contact details for any queries. The Update also stated that the OEMP would be submitted to the NT-DITT for assessment. [Con- 5982]
- On 23 January 2025 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that, following a request from NOPSEMA to provide further information, Santos had re-submitted the EP with some amendments in December 2024 for further assessment. The Update also re-stated that the OEMP would be submitted to the NT-DITT for assessment. [Con-6036]
- No further correspondence or feedback was received from AMSA-NT.

Summary of response by Relevant Person	Assessment of merits	Santos' Response Statement	OEMI
AMSA-NT correspondence to Santos on 30 June 2024 AMSA-NT raised concerns that the Barossa Production Operations information booklet was insufficient to assess the potential environmental risks and impacts of the proposed activity and control measures. (paragraphs 4-5)	Santos does not agree with the AMSA-NT's assertion that the information booklet does not provide sufficient information to enable the AMSA-NT to make an informed assessment of any potential consequences of the Activity on its functions, interests or activities. The information booklet provides a comprehensive description of the environment that may be affected by the Activity, identification of impacts and risks from planned activities and unplanned events and associated proposed control measures.	Santos' correspondence to AMSA-NT on 13 August 2024 in response to AMSA-NT's letter of 30 June 2024 Santos considers the Production Operations information booklet and NT Waters GEP Operations Overview Factsheet provide AMSA-NT with sufficient information to allow it to make an informed assessment of the possible consequences of the activity on any of AMSA's functions, interests or activities.	Not a
AMSA-NT noted their opinion that the regulatory framework is flawed for assessing the cumulative impact of oil and gas industry. (para 7)	The regulatory framework is out of scope of the OEMP.	NOPSEMA's regulatory framework is outside the scope of consultation for activity under the OEMP.	Not a
AMSA-NT stated that it reiterated concerns raised in previous Barossa Offshore Gas Project submissions regarding the project's environmental impact and risk assessment. (para 8)	Santos noted the feedback.	Santos notes your previous correspondence and has responded accordingly for these, a copy of which was included in the relevant approvals document submitted to the regulator, NOPSEMA.	Not a
<ul> <li>AMSA-NT suggests Santos and the regulator are not adequately considering globally significant environmental, fisheries and megafauna values of Darwin Harbour and the region and the international and transboundary issues, when assessing major development activities in the Arafura and Timor Seas region including:</li> <li>ecological connectivity, shared species, shared resources of the region and failure to assesses potential transboundary species, resources and impact in the EMBA and MEVA (para 9(a)).</li> <li>Failure to consult with relevant stakeholders in Indonesia and Timor-Leste that meets the requirements of international law (para 9(b)).</li> <li>Failure to assess potential transboundary environmental harm (para 9(c)).</li> </ul>	Santos noted this request and has provided details on how the impacts associated with the proposed activity are assessed, including the potential for transboundary environmental harm. AMSA-NT's response relates primarily to the EMBA for the Production Operations EP which extends beyond Australia's EEZ. The Barossa Production Operations EP addresses those aspects of AMSA-NT's response. The GEP Coastal Waters EMBA does not extend beyond Australia's EEZ. Relevant information regarding the presence of marine mammals, marine reptiles and birds in Darwin Harbour was provided in the Northern Territory Waters GEP Operations Overview Factsheet	<ul> <li>[9(a)] The Barossa Production Operations information booklet provides a summary of the existing environment (Regional Existing Environment Summary) against which impacts were assessed.</li> <li>Santos has assessed the full potential spatial extent of a worst-case spill event with consideration for biological impacts within the MEVA and socio-economic impacts within the EMBA, including beyond Australia's EEZ into parts of Indonesian and Timor-Leste sovereign waters and impacts to marine users such as commercial and subsistence fishing activities (pages 11, 12, 28 - 36 of the information booklet).</li> <li>Santos has assessed potential impacts on known migratory, rare, threatened, endangered, and protected marine species in the Timor Sea – particularly cetaceans, sea turtles and sharks/rays (Figure 10 and pages 28 - 36 of the information booklet).</li> <li>Potential impacts and risks to marine fauna have been assessed as environmentally acceptable and ALARP.</li> <li>[9(b)] Santos considers its consultation for the EP meets the requirements of s 25 of the Environment Regulations and is consistent with NOSPEMA guideline 'Consultation in the course of preparing an environment plan' (N-04750-GL2086 A900179; 12/05/2023). This has included consultation with the Autoridade Nacional Do Petróleo Timor-Leste (ANP - National Petroleum Authority) and Department of Foreign Affairs and Trade's Indonesia Branch.</li> <li>[9(c)] Internationally significant fauna, wetlands of international importance, internationally significant habitats and internationally</li> </ul>	No ad

# santos

P reference

oplicable

pplicable

applicable

ditional measures adopted.

		significant marine parks are described in the OPP and will be described in the EP. Potential impacts associated with unplanned releases of hydrocarbons that may enter international waters are also described in the OPP and will be described in the Barossa Production Operations EP.	
<ul> <li>AMSA-NT stated that it reiterated concerns regarding data gaps, data analysis and integrity and independence of data, including:</li> <li>lack of assessment of cumulative impacts (para 10(a)).</li> <li>data gaps in baseline information (para 10(b)).</li> <li>issues associated with monitoring, impact and risk assessments in a 'data-poor' setting (para 10(c)).</li> <li>assessing potential impacts from the seabed through the water column (para 10(d)).</li> <li>impact detection and monitoring of marine megafauna populations (para 10(e)).</li> <li>failure to use appropriate data particular for matters of national environmental significance (MNES) species (para 10(f)).</li> <li>lack of scientific independence and peer review of studies (para 10(g)).</li> </ul>	<ul> <li>Questions related to Production Operations activities in the Barossa Field, including operation of the FPSO, are outside the scope of the Coastal Waters OEMP and are addressed in the Barossa Production Operations EP.</li> <li>Notwithstanding this, Santos has considered AMSA-NT's concerns and notes that:</li> <li>Santos performed detailed field and desktop environmental studies and considers adequate data is available;</li> <li>potential cumulative impacts will be assessed and evaluated through the Activity specific' sediment and water quality monitoring program, and managed (if required) through the Produced Water adaptive management plan;</li> <li>Santos has a chemical selection process (BAO-CM-6.7.5);</li> <li>concurrent activities and cumulative impacts area assessed in the OEMP;</li> <li>the NOPSEMA EP content requirement guidance note relevant to protected matters is presented in the OEMP;</li> <li>assessments are undertaken on the full spatial extent of the MEVA and EMBA, based on worst-case credible spills;</li> <li>interactions with marine fauna are recorded and reported (BAO-CM-6.1.1);</li> <li>for the OEMP concurrent activities are not planned to occur in the NT Coastal Waters OA.</li> </ul>	<ul> <li>[10(a)] As noted in the Production Operations information booklet, all planned discharges will be managed in accordance with maritime industry standards and MARPOL requirements to reduce the potential for significant cumulative impacts.</li> <li>Potential for longer term cumulative impacts will be assessed through water and sediment quality monitoring during production operations and need for any additional mitigations assessed.</li> <li>[10(b), (f) and (g)] The information utilised in the development of the OEMP is appropriate to identify risks and impacts arising from production and operations activities and for informing risk mitigation and controls. Santos has followed the NOPSEMA EP content requirement guidance note relevant to matters protected under Part 3 of the EPBC Act, which will be presented in the OEMP for assessment by the regulator.</li> <li>[10(c)] Santos considers adequate data is available and appropriate environmental studies have been undertaken to characterise the existing marine environment within and surrounding the Operational Area.</li> <li>[10(d)] Santos has assessed the full potential spatial extent of a worst case spill event with consideration for biological impacts within the MEVA and socio-economic impacts within the EMBA, including those that extend beyond Australia's Exclusive Economic Zone.</li> <li>[10(e)] Interactions with marine fauna will be recorded and reported as per the requirements of the Protected fauna Interaction and Sighting Procedure (page 26 of the Production Operations information booklet).</li> <li>Incident reporting, investigation and follow-up is monitored (page 37 of the Production Operations information booklet).</li> </ul>	Table Section Section No action
<ul> <li>AMSA-NT raised concerns regarding Santos' proposed control measures for marine fauna interactions (para 11).</li> <li>AMSA-NT noted its ongoing concern regarding Santos' non-inclusion of five voluntary (non-legislated) control measures in the Drilling EP including:</li> <li>further measures to those outlined in EPBC Regulations 2000 <ul> <li>Part 8 Division 8.1 during peak periods of ecological sensitivity, for vessels outlined in the Australian national guidelines National Guidelines for Whale and Dolphin Watching (2017) (para 12(a));</li> <li>avoiding peak periods of ecological sensitivity (para 12(b));</li> <li>restricting vessel speeds in the OA (para 12(c));</li> <li>having a dedicated marine marmal observer (MMO) on vessels (EPBC Policy Statement 2.1 Part B) (para 12(d)); and</li> <li>activities occurring in daylight hours only (para 12(e)).</li> </ul> </li> <li>AMSA-NT criticised Santos' decision to not incorporate these control measures in previous Barossa approval documents, including:</li> <li>Barossa Development Drilling and Completions Environment Plan (para 13(a)); and</li> <li>Barossa Area Development Offshore Project Proposal (para 13(b).</li> </ul> AMSA-NT recommended findings from acoustic baseline studies by JASCO for the Barossa OPP and asserted that noise impacts are not well understood Bryde's, Omura's and Pygmy Blue Whales within the Barossa field area (para 14). AMSA-NT recommended Santos: <ul> <li>undertake further analysis of existing acoustic data (JASCO 2016) to identify habitat use of to better understand and evaluate noise impacts (nara 15): and</li> </ul>	<ul> <li>Santos notes AMSA-NT feedback regarding Santos proposed control measures for marine fauna interaction and concern regarding non-inclusion of non-legislated control measures in the Drilling EP.</li> <li>Although drilling is outside the scope of the Coastal Waters OEMP, Santos has evaluated the control measures from the Drilling EP which are addressed for the purposes of this OEMP in Section 6.1.3 and Section 7.3.3 of the EP and no additional control measures have been adopted for this OEMP</li> <li>Control measures evaluated at the request of AMSA NT include:</li> <li>further measures to those outlined in 'EPBC Regulations 2000 — Part 8 Division 8.1 during peak periods of ecological sensitivity, for vessels outlined in the Australian National Guidelines for Whale and Dolphin Watching (2017) BAO-CM-6.1.1</li> <li>•Application of EPBC Policy Statement 2.1 Interaction between offshore seismic exploration and whales and having a dedicated MMO.</li> <li>Activities occurring in daylight hours only. Santos' assessment is that it has reduced impacts and risks from Activity to ALARP and acceptable levels regarding marine fauna interaction.</li> </ul>	<ul> <li>[11] See response to #12(a)-(e) below.</li> <li>[12(a)] The Production Operations information booklet contains proposed control measures that will be adopted as relevant to potential impacts to species, including marine fauna, and in relation to biologically important areas. All considered control measures (adopted and not adopted) will be presented in the EP for assessment by the Regulator/s.</li> <li>Santos' planned vessel activities overlap the internesting buffer for flatback turtles.</li> <li>Specific to marine fauna interactions, the Production Operations information booklet describes control measures at pages 9, 10 and 26. Santos adopts the following specific controls to protect marine fauna from vessel activities for Barossa Operations in accordance with:</li> <li>(a) Part 8 of the Environment Protection and Biodiversity Conservation Regulations 2000: <ul> <li>i) avoid collision;</li> <li>ii) reduce speed to 6 knots &amp; steer away within caution zone (300m of whales / whale sharks and 150m of dolphins);</li> <li>iii) operate vessel at constant speed (6 knots);</li> <li>iv) do not drift or approach marine fauna;</li> <li>v) do not restrict fauna pathway or pursue fauna; and</li> <li>(b) Australian National Guidelines for Whale and Dolphin Watching 2017: Adopting cautionary and no approach distances for whales, whale sharks (100m no approach and 300m cautionary distances).</li> <li>[12(b)] The risk of interactions with marine fauna in the OA is very low. Ongoing inspection, maintenance, monitoring and repair (IMMR) activities will be temporary in nature and performed</li> </ul></li></ul>	Section No ac



e 3-1 and Table 3-2 for marine and additional studies ion 3.2.2 for Physical environment ion 3.2.13 for Threatened and migratory fauna

ion 7.3 and Section 7.6.

dditional measures adopted.

ion 7.3 Idditional measures adopted.

<ul> <li>accept the five voluntary control measures based on:         <ul> <li>threatened marine fauna have been detected acoustically in the Barossa area (para 16(a))</li> </ul> </li> </ul>	according to a planned inspection and maintenance schedule, or at other intervals if unplanned inspections or repairs are required (page 5 of the Production Operations information booklet).
<ul> <li>uncertainty of extent of habitat use by Bryde's, Omura's and Pygmy Blue Whales (para 16(b))</li> <li>vessel shipping impacts such as fauna displacement and</li> </ul>	It is not practical to operate the activity to avoid 'sensitive periods'. The acceptability evaluation of environmental risks is described in the OEMP for assessment by the regulator.
<ul> <li>avoidance (para 16(c))</li> <li>ability to detect marine mammals and potential impacts (by restricting to daylight hours and have MMOs on baard) (para 16(d))</li> </ul>	[12(c)] Operational area speed restrictions refer to limits on vessel speeds within the OA to maintain safe operations (see page 26 of Information Booklet).
<ul> <li>board) (para 16(d))</li> <li>having consistency with the National Vessel Stike Strategy (reduce speed to &lt; 12 knots in the OA) (para 16(e))</li> <li>a lack of uncertainty of habit use by threated marine fauna (para 16(f))</li> </ul>	[12(d)] Dedicated MMOs on IMMR vessels have been assessed during the risk assessment process and based on the nature and scale of the activity have been determined as not required. Further details will be provided in the OEMP for assessment by the regulator.
	[12(e)] Activities to occur during daylight hours only has been assessed during the risk assessment process and based on the nature and scale of the activity, these restrictions have not been
	adopted. [13(a)] The Barossa Development Drilling and Completions Environment Plan was accepted by NOPSEMA on 15 December 2023.
	Control measures for the production operations activity are separately considered and evaluated by Santos with a view to reducing impacts and risks to ALARP and acceptable levels. Santos' evaluations will be documented in the EP and will be reviewed by NOPSEMA during its assessment of the EP. Refer also to #12.
	[13(b)] The Barossa Development Drilling and Completions Environment Plan was accepted by NOPSEMA on 15 December 2023.
	Control measures for the production operations activity are separately considered and evaluated by Santos with a view to reducing impacts and risks to ALARP and acceptable levels. Santos' evaluations will be documented in the OEMP and will be reviewed by NOPSEMA during its assessment of the OEMP. Refer also to #12.
	[14(a)-(h) and 15] The Production Operations information booklet refers to noise sources and identified proposed control measures for managing potential impacts on marine mammals (e.g. whales) from noise. It presents a summary of the results of underwater acoustic assessments for noise sources relevant to the scope of the Coastal Waters OEMP.
	Santos' evaluation of impacts and risks to marine mammals from noise emissions will be a matter for the Regulator to assess against the requirements of the Regulations.
	[Para 16(a) and (b)] Santos' proposed control measures associated with interaction with marine fauna are outlined on page 26 of the Production Operations information booklet and are designed to align with management actions outlined in government-published fauna recovery plans, the Environment Protection and Biodiversity Conservation Regulations 2000 and include speed restrictions.
	Refer to #12(a)-(e) above.
	[Para 16(c)] Page 26 of the Production Operations information booklet outlines that the highest potential for interactions with marine fauna will be during temporary IMMR vessel operations. The risk of interactions with marine fauna is very low.
	The ALARP and acceptability evaluation of environmental risks in the OEMP for assessment by the regulator.
	[Para 16(d)] Refer to #12(a)-(e) above.
	[Para 16(e)] Operational area speed restrictions refer to limits on vessel speeds within the operational area/s to maintain safe operations (page 26 of the Production Operations information booklet). Vessel speeds will be addressed in the EP for
	assessment by the regulator, which includes vessel speed of 8

# **Santos**

		knots or less within 500m safety zone around the FPSO and campaign vessels. [Para 16(f)] Santos will address the precautionary principles in the ALARP and acceptability evaluation of environmental risks in the OEMP for assessment by the regulator.	
<ul> <li>AMSA-NT stated that it reiterated and highlighted Santos' possible contravention or non-alignment with Blue Whale Conservation Management Plan (CMP) including:</li> <li>vessel strike and underwater noise are key threats (para 17(a).</li> <li>recent research providing evidence of Pygmy Blue Whales aggregation and foraging in the Timor Sea and within the defined EMBA and MEVA for this Barossa Offshore Gas project, i.e. see 'aggregation' and 'foraging/reproduction' areas identified in the Timor Trough region (Sahri et al. 2022, Burton et al. 2023, Ferreira et al. 2024). (para 17(b)).</li> <li>recent major sightings of Pygmy Blue Whales in close proximity (approximately 40 km) to the Barossa OA (para 17(c)).</li> <li>uncertainty regarding habitat usage by Pygmy Blue Whales in the Barossa field area (para 17(d)).</li> <li>the OPP acoustic studies (JASCO 2016) should be reassessed (para 17(e).</li> <li>BIAs should be updated (para 17(f)).</li> <li>AMSA NT noted that temporal control measures were adopted by another operator in the Timor Sea to deal with uncertainty in blue whale presence (para 17(g))</li> </ul>	Santos has considered the recommendation from AMSA-NT to review published research papers on the Pygmy Blue Whale by Sahri et al. 2022, Burton et al. 2023, Ferreira et al. 2024. These papers provide evidence of foraging, migration, and aggregation activity, which has been incorporated into Section 3.2.13.1.1. Santos' assessment is that Pygmy Blue Whale foraging, migration and aggregation activity occurs outside the OA therefore no additional control measures are required. Santos also notes the feedback from AMSA-NT regarding temporal control measures adopted by another operator. Santos has reviewed and evaluated the relevant controls adopted by the operator in Section 6.1.3 of the OEMP with regard to environmental benefits gained versus risk/cost of implementation. No additional control measures were adopted.	<ul> <li>[17(a) and (d)] Figure 10 in the Production Operations information booklet depicts the blue whale biologically important area (page 9). Operational areas 1 and 2 do not overlap with the blue whale BIA.</li> <li>Potential impacts to marine mammals (e.g. whales) from vessel strike an underwater noise have been assessed and control measures presented in the Production Operations information booklet on page 16 (noise) and page 26 (interaction with marine fauna).</li> <li>The nature and scale of environmental impacts, taking into account the Blue Whale CMP, will be provided in the OEMP for assessment by the regulator.</li> <li>[17(b)] The information utilised in the development of the OEMP is appropriate to identify risks and impacts arising from the activities and for informing risk mitigation and controls. Santos has followed the NOPSEMA environment plan content requirement guidance note for matters of national environmental significance protected under Part 3 of the EPBC Act, which will be presented in the OEMP for assessment by the regulator.</li> <li>(In further response to AMSA-NT recommendations, published research studies on the Pygmy Blue Whale by Sahri et al.2022, Burton et al. 2023 and Ferreira et al. 2024 have been reviewed and Section 3.4.3.2 has been updated)</li> <li>[17(c)] The information utilised in the development of the OEMP is appropriate to identify risks and impacts arising from the activities and for informing risk mitigation and controls. Santos has followed the NOPSEMA environment plan content requirement guidance note for matters of national environmental significance protected under Part 3 of the EPBC Act, which will be presented in the OEMP is appropriate to identify risks and impacts arising from the activities and for informing risk mitigation and controls. Santos has followed the NOPSEMA environment plan content requirement guidance note for matters of national environmental significance protected under Part 3 of the EPBC Act, which will be presented in the OEMP for assessmen</li></ul>	Secti No ad
Concerns were raised regarding lack of biologically important areas (BIAs) for Threatened Marine Megafauna, Baseline Marine Megafaunal Surveys, noting that additional marine megafauna surveys are required and the need to develop BIAs for other known conservation priority marine species occurring in the EMBA, MEVA and the region including Bryde's Whale, Omura Whales, Fin Whales, Sei Whales and Sperm Whales (para 18)	Santos has considered the matters raised and provided a response. The development of BIAs is out of scope of the OEMP.	Undertaking surveys and developing BIAs is outside the scope of consultation for this activity.	Not a
Austurtie Inc			

On 9 February 2024 Santos emailed AusTurtle Inc to advise it of preliminary consultation regarding proposed activities for consultation to be managed under the Coastal Waters OEMP and Barossa Production Operations EP. [Con-3787] •

- The email advised that Santos was seeking information to better understand:
  - o if you are from a government Department or agency, how the proposed Production Operations activities may be relevant to your department or agency; and
  - what (if any) functions, interests or activities you or your organisation have that may be affected by the proposed Production Operations activities.
- The email included information on the regulatory process for the activities in Commonwealth and NT jurisdictions and links to a Santos information booklet on the proposed activities and a NOPSEMA brochure on the consultation process and details of how to contact • Santos to register as a Relevant Person. The email stated that the consultation phase would commence on 11 March 2024 and close on 9 April 2024.
- On 11 March 2024 Santos emailed AusTurtle Inc further to the previous correspondence, to advise that it had commenced the consultation phase which would run until 9 April 2024. In addition to the previous information again being provided, Santos provided information on Relevant Persons' entitlements under the regulatory processes, details of how to provide feedback and a reminder of the closing date for consultation. [Con-3793]
- On 4 April 2024 Santos phoned AusTurtle Inc regarding consultation for Production Operations activities and spoke to the nominated person who advised that AusTurtle would be sending Santos some information. •



ions 3.2.13 and 6.1.3. dditional measures adopted.

applicable

- On 7 April 2024, AusTurtle emailed Santos and provided a submission with detailed information on sea turtles, sea snakes and sea birds within the area, including an assessment of nesting, internesting and migration patterns. AusTurtle advised that the number of nests laid on the island during the standard fortnightly survey appears independent of anthropogenic offshore activities such as petroleum or fishing activities. [Con-4006]
- On 27 May 2024, Santos emailed AusTurtle to thank it for the feedback provided to assist Santos' preparation of the OEMP and EP. Santos noted that AusTurtle did not raise any objections or claims and referenced information provided on AusTurtles' ongoing research at Bare Sand Island. Santos advised it was interested in the Olive Ridley and Flatback species, to be referenced in the OEMP and EP with respect to the presence and behaviour of sea turtles within the Environment that May Be Affected. Santos also referenced the information on the presence of Crested Sea Terns on Bare Sand Island and sea snake behaviour in the proposed Operational Areas, which correlated with comments previously provided. Santos asked if AusTurtle research papers or details could be provided to enable it to be referenced in the appropriate sections of the OEMP and EP. [Con-4211]
- On 28 May 2024, AusTurtle emailed Santos and advised that it would get some papers together and that most of the information has come from AusTurtle internal annual reports. [Con-4213]
- On 10 July 2024 Santos emailed AusTurtle to advise the consultation period had been completed. Santos advised it consultation had now closed for the purpose of Santos finalising and submitting environment plans for these activities to government regulators for assessment. [Con-5130]
- On 15 October 2024 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that the EP had been submitted to NOPSEMA for assessment, provided a link to the full EP on the NOPSEMA website and Santos' contact details for any queries. The Update also stated that the OEMP would be submitted to the NT-DITT for assessment. [Con- 5982]
- On 23 January 2025 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that, following a request from NOPSEMA to provide further information, Santos had re-submitted the EP with some amendments in December 2024 for further assessment. The Update also re-stated that the OEMP would be submitted to the NT-DITT for assessment. [Con-6036]
- No further correspondence or feedback was received from AusTurtle.

S	Summary of response by Relevant Person	Assessment of merits	Santos' Response Statement	OEMF
A s	usTurtle provided the following information in relation to flatback ea turtles (on 11 March 24):	Santos has considered the matters raised in the OEMP and provided a response	Santos responded to AusTurtle thanking it for the information provided.	Marin Sea b
•	AusTurtle has monitored nesting flatback sea turtles since 1996 on Bare Island which is located at the edge of the MEVA.	Santos notes information provided on 11 March 24 by AusTurtle. The reference by AusTurtle to Bare (Sand) Island being located at the edge of the MEVA is relevant to the Productions Operations EP only. Bare Sand Island is not located in the OEMP EMBA.		
•	During construction of the Bayu-Undan to Darwin gas pipeline from 2004-2006 and the INPEX Ichthys gas pipeline in 2014- 2016 there was no detectable impact on the numbers of nesting turtles.	Santos notes that AusTurtle has not provided the papers as per correspondence 28 May 2024.		
•	The DPD section will pass through the flatback turtle internesting area where gravid females will dive to depths of 40 m and rest on the bottom to surface every hour or so to breathe.			
•	The previous pipelines had no detectable impact as is expected with this pipeline.			
•	Any impact, including attraction to lights, is likely to be on individuals rather than the population.			
C	Charles Darwin University (CDU)			·

- On 9 February 2024 Santos emailed CDU to advise it of preliminary consultation regarding proposed activities for consultation to be managed under the Coastal Waters OEMP and Barossa Production Operations EP. [Con-3787]
- The email advised that Santos was seeking information to better understand:
  - o if you are from a government Department or agency, how the proposed Production Operations activities may be relevant to your department or agency; and
    - what (if any) functions, interests or activities you or your organisation have that may be affected by the proposed Production Operations activities.
- The email included information on the regulatory process for the activities in Commonwealth and NT jurisdictions and links to a Santos information booklet on the proposed activities and a NOPSEMA brochure on the consultation process and details of how to contact Santos to register as a Relevant Person. The email stated that the consultation phase would commence on 11 March 2024 and close on 9 April 2024.
- On 11 March 2024 Santos emailed CDU further to the previous correspondence, to advise that it had commenced the consultation phase which would run until 9 April 2024. In addition to the previous information again being provided, Santos provided information on Relevant Persons' entitlements under the regulatory processes, details of how to provide feedback and a reminder of the closing date for consultation. [Con-3793]
- On 4 April 2024 Santos phoned CDU and left a voice mail message regarding consultation for Production Operations activities.
- On 2 May 2024, Santos emailed CDU to advise further to its previous correspondence, to advise it had extended the consultation period until 16 May 2024. In providing this extension of time, Santos advised that the information in the booklet and factsheet had been updated to account for an additional risk associated with the proposed activity. Santos confirmed that consultation will close on the revised date unless Santos hears otherwise from CDU. [Con-3856]
- On 10 July 2024 Santos emailed CDU to advise the consultation period had been completed. Santos advised CDU that it considered consultation had now closed for the purpose of Santos finalising and submitting environment plans for these activities to government regulators for assessment. [Con-5130]
- On 15 October 2024 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that the EP had been submitted to NOPSEMA for assessment, provided a link to the full EP on the NOPSEMA website and Santos' contact details for any queries. The Update also stated that the OEMP would be submitted to the NT-DITT for assessment. [Con- 5982]
- On 23 January 2025 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that, following a request from NOPSEMA to provide further information, Santos had re-submitted the EP with some amendments in December 2024 for further assessment. The Update also re-stated that the OEMP would be submitted to the NT-DITT for assessment. [Con-6036]

Notwithstanding the information provided and the steps described above, no comments or input were received on the Coastal Waters OEMP from CDU.

Summary of response by Relevant Person	Assessment of merits	Santos' Response Statement	OEMP
----------------------------------------	----------------------	----------------------------	------



**Reference** 

e turtles are described in Section 3.2.13.2.1. irds are described in Section 3.2.13.4.

No response was received from CDU.	Santos considers it has provided sufficient information and a reasonable period of time for consultation.	No response required.	Not ap
	Santos considers Section 25 consultation requirements to have been met.		

pplicable.



### 4.8.5 Energy Industry Titleholders / Operators

## Table 4-15: Consultation Summary Table - Energy Industry Titleholders / Operators

Section 25(1)(d) of the OPGGS(E)R: Persons or organisations whose functions, interests or activities may be affected by the activities to be carried out under the environment plan

## **Bengal Energy**

### Summary of consultation effort:

- On 9 February 2024 Santos emailed Bengal Energy to advise it of preliminary consultation regarding proposed activities for consultation to be managed under the Coastal Waters OEMP and Barossa Production Operations EP. [Con-3787]
- The email advised that Santos was seeking information to better understand:
  - o if you are from a government Department or agency, how the proposed Production Operations activities may be relevant to your department or agency; and
  - what (if any) functions, interests or activities you or your organisation have that may be affected by the proposed Production Operations activities.
- The email included information on the regulatory process for the activities in Commonwealth and NT jurisdictions and links to a Santos information booklet on the proposed activities and a NOPSEMA brochure on the consultation process and details of how to contact Santos to register as a Relevant Person. The email stated that the consultation phase would commence on 11 March 2024 and close on 9 April 2024.
- On 11 March 2024 Santos emailed Bengal Energy further to the previous correspondence, to advise that it had commenced the consultation phase which would run until 9 April 2024. In addition to the previous information again being provided, Santos provided information on Relevant Persons' entitlements under the regulatory processes, details of how to provide feedback and a reminder of the closing date for consultation. [Con-3793]
- On 4 April 2024 Santos phoned Bengal Energy and left a voice message regarding consultation for Production Operations activities.
- On 2 May 2024, Santos emailed Bengal Energy further to previous correspondence, to advise it had extended the consultation period until 16 May 2024. In providing this extension of time, Santos advised that the information in the booklet and factsheet had been updated to account for an additional risk associated with the proposed activity. Santos confirmed that consultation will close on the revised date unless Santos hears otherwise from Bengal Energy. [Con-3984]
- On 10 July 2024 Santos emailed Bengal Energy to advise the consultation period had been completed. Santos advised Bengal Energy that it considered consultation had now closed for the purpose of Santos finalising and submitting environment plans for these activities to government regulators for assessment. [Con-5130]
- On 15 October 2024 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that the EP had been submitted to NOPSEMA for assessment, provided a link to the full EP on the NOPSEMA website and Santos' contact details for any queries. The Update also stated that the OEMP would be submitted to the NT-DITT for assessment. [Con- 5982]
- On 23 January 2025 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that, following a request from NOPSEMA to provide further information. Santos had re-submitted the EP with some amendments in December 2024 for further assessment. The Update also re-stated that the OEMP would be submitted to the NT-DITT for assessment. [Con-6036]
- Notwithstanding the consultation information provided and the steps described above, no comments or input were received on the Coastal Waters OEMP from Bengal Energy.

Summary of response by Relevant Person	Assessment of merits	Santos' Response Statement	OEMP
No response was received from Bengal Energy.	Santos considers it has provided sufficient information and a reasonable period of time for consultation. Santos considers Section 25 consultation requirements to have been met.	No response required.	Not ap

## Eni Australi

Summary of consultation effort:

- On 9 February 2024 Santos emailed Eni Australia to advise it of preliminary consultation regarding proposed activities for consultation to be managed under the Coastal Waters OEMP and Barossa Production Operations EP. [Con-3787]
- The email advised that Santos was seeking information to better understand:
  - o if you are from a government Department or agency, how the proposed Production Operations activities may be relevant to your department or agency; and
  - what (if any) functions, interests or activities you or your organisation have that may be affected by the proposed Production Operations activities. 0
- The email included information on the regulatory process for the activities in Commonwealth and NT jurisdictions and links to a Santos information booklet on the proposed activities and a NOPSEMA brochure on the consultation process and details of how to contact Santos to register as a Relevant Person. The email stated that the consultation phase would commence on 11 March 2024 and close on 9 April 2024.
- On 11 March 2024 Santos emailed Eni Australia further to the previous correspondence, to advise that it had commenced the consultation phase which would run until 9 April 2024. In addition to the previous information again being provided, Santos provided information on Relevant Persons' entitlements under the regulatory processes, details of how to provide feedback and a reminder of the closing date for consultation. [Con-3793]
- On 3 April 2024 Santos phoned Eni Australia and spoke to a company representative and left a message regarding consultation for Production Operations activities.
- On 6 May 2024, Santos emailed Eni Australia further to previous correspondence, to advise it had extended the consultation period until 20 May 2024. In providing this extension of time, Santos advised that the information in the booklet and factsheet had been updated to account for an additional risk associated with the proposed activity. Santos confirmed that consultation will close on the revised date unless Santos hears otherwise from Eni Australia. [Con-3989]
- On 10 July 2024 Santos emailed Eni Australia to advise the consultation period had been completed. Santos advised Eni Australia that it considered consultation had now closed for the purpose of Santos finalising and submitting environment plans for these activities to government regulators for assessment. [Con-5130]
- On 15 October 2024 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that the EP had been submitted to NOPSEMA for assessment, provided a link to the full EP on the NOPSEMA website and Santos' contact details for any queries. The Update also stated that the OEMP would be submitted to the NT-DITT for assessment. [Con- 5982]
- On 23 January 2025 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that, following a request from NOPSEMA to provide further information, Santos had re-submitted the EP with some amendments in December 2024 for further assessment. The Update also re-stated that the OEMP would be submitted to the NT-DITT for assessment. [Con-6036]
- Notwithstanding the consultation information provided and the steps described above, no comments or input were received on the Coastal Waters OEMP from Eni Australia

Assessment of merits	Santos' Response Statement	OEMP
Santos considers it has provided sufficient information and a	No response required.	Not ap
	Assessment of merits Santos considers it has provided sufficient information and a reasonable period of time for consultation.	Assessment of merits     Santos' Response Statement       Santos considers it has provided sufficient information and a reasonable period of time for consultation.     No response required.



Reference

plicable

Reference

plicable.

	Santos considers Section 25 consultation requirements to have been met.	
EOG Resources		

- On 9 February 2024 Santos emailed EOG Resources to advise it of preliminary consultation regarding proposed activities for consultation to be managed under the Coastal Waters OEMP and Barossa Production Operations EP. [Con-3787]
- The email advised that Santos was seeking information to better understand:
  - o if you are from a government Department or agency, how the proposed PO activities may be relevant to your department or agency; and
  - what (if any) functions, interests or activities you or your organisation have that may be affected by the proposed Production Operations activities.
- The email included information on the regulatory process for the activities in Commonwealth and NT jurisdictions and links to a Santos information booklet on the proposed activities and a NOPSEMA brochure on the consultation process and details of how to contact Santos to register as a Relevant Person. The email stated that the consultation phase would commence on 11 March 2024 and close on 9 April 2024.
- On 11 March 2024 Santos emailed EOG Resources further to the previous correspondence, to advise that it had commenced the consultation phase which would run until 9 April 2024. In addition to the previous information again being provided, Santos provided information on Relevant Persons' entitlements under the regulatory processes, details of how to provide feedback and a reminder of the closing date for consultation. [Con-3793]
- On 5 April 2024 Santos phoned EOG Resources and left a voice mail message regarding consultation for Production Operations activities.
- On 2 May 2024, Santos emailed EOG Resources further to previous correspondence, to advise it had extended the consultation period until 16 May 2024. In providing this extension of time, Santos advised that the information in the booklet and factsheet had been updated to account for an additional risk associated with the proposed activity. Santos confirmed that consultation will close on the revised date unless Santos hears otherwise from EOG Resources [Con-3986]
- On 3 May 2024, EOG Resources emailed Santos to advise it does not have any input for Production Operations activities and asked to be kept updated if there are any material changes. [Con-3987]
- On 10 July 2024 Santos emailed EOG Resources to advise the consultation period had been completed. Santos advised EOG Resources that it considered consultation had now closed for the purpose of Santos finalising and submitting environment plans for these activities to government regulators for assessment. [Con-5130]
- On 15 October 2024 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that the EP had been submitted to NOPSEMA for assessment, provided a link to the full EP on the NOPSEMA website and Santos' contact details for any queries. The Update also stated that the OEMP would be submitted to the NT-DITT for assessment. [Con- 5982]
- On 23 January 2025 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that, following a request from NOPSEMA to provide further information, Santos had re-submitted the EP with some amendments in December 2024 for further assessment. The Update also re-stated that the OEMP would be submitted to the NT-DITT for assessment. [Con-6036] Notwithstanding the consultation information provided and the steps described above, no comments or input were received on the Coastal Waters OEMP from EOG Resources.

Summary of response by Relevant Person	Assessment of merits	Santos' Response Statement	OEM
No response was received from EOG Resources.	Santos considers it has provided sufficient information and a reasonable period of time for consultation. Santos considers Section 25 consultation requirements to have been met.	No response required.	Not a

## Finder Energy

Summary of consultation effort:

- On 9 February 2024 Santos emailed Finder Energy to advise it of preliminary consultation regarding proposed activities for consultation to be managed under the Coastal Waters OEMP and Barossa Production Operations EP. [Con-3787]
- The email advised that Santos was seeking information to better understand:
  - if you are from a government Department or agency, how the proposed Production Operations activities may be relevant to your department or agency; and
  - what (if any) functions, interests or activities you or your organisation have that may be affected by the proposed Production Operations activities.
- The email included information on the regulatory process for the activities in Commonwealth and NT jurisdictions and links to a Santos information booklet on the proposed activities and a NOPSEMA brochure on the consultation process and details of how to contact Santos to register as a Relevant Person. The email stated that the consultation phase would commence on 11 March 2024 and close on 9 April 2024.
- On 11 March 2024 Santos emailed Finder Energy further to the previous correspondence, to advise that it had commenced the consultation phase which would run until 9 April 2024. In addition to the previous information again being provided, Santos provided information on Relevant Persons' entitlements under the regulatory processes, details of how to provide feedback and a reminder of the closing date for consultation. [Con-3793]
- On 19 March 2024 Finder Energy emailed Santos and advised it has no comment or objection to the Production Operations activities. [Con-3981]
- On 7 May 2024, Santos emailed Finder Energy further to previous correspondence, to advise it had extended the consultation period until 21 May 2024. In providing this extension of time, Santos advised that the information in the booklet and factsheet had been updated to account for an additional risk associated with the proposed activity. Santos confirmed that consultation will close on the revised date unless Santos hears otherwise from Finder Energy. [Con-3994]
- On 10 July 2024 Santos emailed Finder Energy to advise the consultation period had been completed. Santos advised it consultation had now closed for the purpose of Santos finalising and submitting environment plans for these activities to government. regulators for assessment. [Con-5130]
- On 15 October 2024 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that the EP had been submitted to NOPSEMA for assessment, provided a link to the full EP on the NOPSEMA website and Santos' contact details for any queries. The Update also stated that the OEMP would be submitted to the NT-DITT for assessment. [Con- 5982]
- On 23 January 2025 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that, following a request from NOPSEMA to provide further information. Santos had re-submitted the EP with some amendments in December 2024 for further assessment. The Update also re-stated that the OEMP would be submitted to the NT-DITT for assessment. [Con-6036]
- No further correspondence has been received from Finder Energy.

Summary of response by Relevant Person	Assessment of merits	Santos' Response Statement	OEMP
Finder Energy advised it had no comment or objection.	Santos considers it has provided sufficient information and a reasonable period of time for consultation.	No response required.	Not ap
	Santos considers Section 25 consultation requirements to have been met.		

## Santos

P Reference

pplicable.

Reference

nlicable

## **INPEX**

Summary of consultation effort:

- On 9 February 2024 Santos emailed INPEX to advise it of preliminary consultation regarding proposed activities for consultation to be managed under the Coastal Waters OEMP and Barossa Production Operations EP. [Con-3787] •
- The email advised that Santos was seeking information to better understand:
- if you are from a government Department or agency, how the proposed Production Operations activities may be relevant to your department or agency; and
- what (if any) functions, interests or activities you or your organisation have that may be affected by the proposed Production Operations activities. •
- The email included information on the regulatory process for the activities in Commonwealth and NT jurisdictions and links to a Santos information booklet on the proposed activities and a NOPSEMA brochure on the consultation process and details of how to contact Santos to register as a Relevant Person. The email stated that the consultation phase would commence on 11 March 2024 and close on 9 April 2024.
- On 11 March 2024 Santos emailed INPEX further to the previous correspondence, to advise that it had commenced the consultation phase which would run until 9 April 2024. In addition to the previous information again being provided, Santos provided information on Relevant Persons' entitlements under the regulatory processes, details of how to provide feedback and a reminder of the closing date for consultation. [Con-3793]
- On 21 March 2024 INPEX emailed Santos to enquire about the INPEX contacts on Santos' mailing list. [Con-3575]
- On 3 April 2024 Santos phoned INPEX regarding consultation activities and left a message with a company representative.
- On 23 April 2024 Santos emailed INPEX and provided information requested on 21 March 2024. [Con-3786] •
- On 6 May 2024, Santos emailed INPEX to further to previous correspondence, to advise it had extended the consultation period until 20 May 2024. In providing this extension of time, Santos advised that the information in the booklet and factsheet had been updated to account for an additional risk associated with the proposed activity. Santos confirmed that consultation will close on the revised date unless Santos hears otherwise from INPEX. [Con-3990]
- On 10 July 2024 Santos emailed INPEX to advise the consultation period had been completed. Santos advised it consultation had now closed for the purpose of Santos finalising and submitting environment plans for these activities to government regulators for assessment. [Con-5130]
- On 15 October 2024 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that the EP had been submitted to NOPSEMA for assessment, provided a link to the full EP on the NOPSEMA website and Santos' contact details for any queries. The Update also stated that the OEMP would be submitted to the NT-DITT for assessment. [Con- 5982]
- On 23 January 2025 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that, following a request from • NOPSEMA to provide further information, Santos had re-submitted the EP with some amendments in December 2024 for further assessment. The Update also re-stated that the OEMP would be submitted to the NT-DITT for assessment. [Con-6036]
- Notwithstanding the consultation information provided and the steps described above, no comments or input were received on the Coastal Waters OEMP from INPEX.

Summary of response by Relevant Person	Assessment of merits	Santos' Response Statement	OEMP
No correspondence was received from INPEX.	Santos considers it has provided sufficient information and a reasonable period of time for consultation. Santos considers Section 25 consultation requirements to have been met	No response required.	Not ap



reference

oplicable

## Jadestone Energy

Summary of consultation effort:

- On 9 February 2024 Santos emailed Jadestone Energy to advise it of preliminary consultation regarding proposed activities for consultation to be managed under the Coastal Waters OEMP and Barossa Production Operations EP. [Con-3787]
- The email advised that Santos was seeking information to better understand:
  - o if you are from a government Department or agency, how the proposed Production Operations activities may be relevant to your department or agency; and
  - what (if any) functions, interests or activities you or your organisation have that may be affected by the proposed Production Operations activities. 0
- The email included information on the regulatory process for the activities in Commonwealth and NT jurisdictions and links to a Santos information booklet on the proposed activities and a NOPSEMA brochure on the consultation process and details of how to contact Santos to register as a Relevant Person. The email stated that the consultation phase would commence on 11 March 2024 and close on 9 April 2024.
- On 28 February 2024 Jadestone Energy emailed Santos and advised it has no comments regarding Production Operations activities and indicated that if the project outlined in correspondence changed significantly they would like to remain informed. [Con-3190]
- On 11 March 2024 Santos emailed Jadestone Energy further to the previous correspondence, to advise that it had commenced the consultation phase which would run until 9 April 2024. In addition to the previous information again being provided, Santos provided information on Relevant Persons' entitlements under the regulatory processes, details of how to provide feedback and a reminder of the closing date for consultation. [Con-3793]
- On 7 May 2024, Santos emailed Jadestone Energy further to previous correspondence, to advise it had extended the consultation period until 21 May 2024. In providing this extension of time, Santos advised that the information in the booklet and factsheet had been updated to account for an additional risk associated with the proposed activity. Santos confirmed that consultation will close on the revised date unless Santos hears otherwise from Jadestone Energy. [Con-3993]
- On 10 July 2024 Santos emailed Jadestone Energy to advise the consultation period had been completed. Santos advised it considered that consultation had now closed for the purpose of Santos finalising and submitting environment plans for these activities to government regulators for assessment. [Con-5130]
- On 15 October 2024 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that the EP had been submitted to NOPSEMA for assessment, provided a link to the full EP on the NOPSEMA website and Santos' contact details for any queries. The Update also stated that the OEMP would be submitted to the NT-DITT for assessment. [Con- 5982]
- On 23 January 2025 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that, following a request from NOPSEMA to provide further information, Santos had re-submitted the EP with some amendments in December 2024 for further assessment. The Update also re-stated that the OEMP would be submitted to the NT-DITT for assessment. [Con-6036]
- No further correspondence has been received from Jadestone Energy.

Summary of response by Relevant Person	Assessment of merits	Santos' Response Statement	OEM
Jadestone Energy advised it had no comments.	Santos considers it has provided sufficient information and a reasonable period of time for consultation. Santos considers Section 25 consultation requirements to have been met.	No response required.	Not aj

## **Melbana Energy**

Summary of consultation effort:

- On 9 February 2024 Santos emailed Melbana Energy to advise it of preliminary consultation regarding proposed activities for consultation to be managed under the Coastal Waters OEMP and Barossa Production Operations EP. [Con-3787]
- The email advised that Santos was seeking information to better understand:
  - o if you are from a government Department or agency, how the proposed Production Operations activities may be relevant to your department or agency; and
  - what (if any) functions, interests or activities you or your organisation have that may be affected by the proposed Production Operations activities. 0
- The email included information on the regulatory process for the activities in Commonwealth and NT jurisdictions and links to a Santos information booklet on the proposed activities and a NOPSEMA brochure on the consultation process and details of how to contact Santos to register as a Relevant Person. The email stated that the consultation phase would commence on 11 March 2024 and close on 9 April 2024.
- On 11 March 2024 Santos emailed Melbana Energy further to the previous correspondence, to advise that it had commenced the consultation phase which would run until 9 April 2024. In addition to the previous information again being provided, Santos provided information on Relevant Persons' entitlements under the regulatory processes, details of how to provide feedback and a reminder of the closing date for consultation. [Con-3793]
- On 12 March 2024 Melbana Energy emailed Santos and advised it had no feedback or objection to Production Operations activities. [Con-3980]
- On 4 April 2024 Santos phoned Melbana Energy regarding consultation for Barossa Production Operations EP activities and left a voice mail message.
- On 7 May 2024, Santos emailed Melbana Energy further to previous correspondence, to advise it had extended the consultation period until 21 May 2024. In providing this extension of time, Santos advised that the information in the booklet and factsheet had been updated to account for an additional risk associated with the proposed activity. Santos confirmed that consultation will close on the revised date unless Santos hears otherwise from Melbana Energy. [Con-3992]
- On 10 July 2024 Santos emailed Melbana Energy to advise the consultation period had been completed. Santos advised it consultation had now closed for the purpose of Santos finalising and submitting environment plans for these activities to government regulators for assessment. [Con-5130]
- On 15 October 2024 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that the EP had been submitted to NOPSEMA for assessment, provided a link to the full EP on the NOPSEMA website and Santos' contact details for any queries. The Update also stated that the OEMP would be submitted to the NT-DITT for assessment. [Con- 5982]
- On 23 January 2025 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that, following a request from NOPSEMA to provide further information, Santos had re-submitted the EP with some amendments in December 2024 for further assessment. The Update also re-stated that the OEMP would be submitted to the NT-DITT for assessment. [Con-6036]
- No further correspondence has been received from Melbana Energy.

Summary of response by Relevant Person	Assessment of merits	Santos' Response Statement	OEMP
Melbana Energy advised it had no feedback or objection.	Santos considers it has provided sufficient information and a reasonable period of time for consultation.	No response required.	Not ap

## Santos

**Preference** 

oplicable.

reference

oplicable.

	Santos considers Section 25 consultation requirements to have been met.	
Neptune Energy		

- On 9 February 2024 Santos emailed Neptune Energy to advise it of preliminary consultation regarding proposed activities for consultation to be managed under the Coastal Waters OEMP and Barossa Production Operations EP. [Con-3787]
- The email advised that Santos was seeking information to better understand:
  - o if you are from a government Department or agency, how the proposed Production Operations activities may be relevant to your department or agency; and
  - what (if any) functions, interests or activities you or your organisation have that may be affected by the proposed Production Operations activities.
- The email included information on the regulatory process for the activities in Commonwealth and NT jurisdictions and links to a Santos information booklet on the proposed activities and a NOPSEMA brochure on the consultation process and details of how to contact Santos to register as a Relevant Person. The email stated that the consultation phase would commence on 11 March 2024 and close on 9 April 2024.
- On 11 March 2024 Santos emailed Neptune Energy further to the previous correspondence, to advise that it had commenced the consultation phase which would run until 9 April 2024. In addition to the previous information again being provided, Santos provided information on Relevant Persons' entitlements under the regulatory processes, details of how to provide feedback and a reminder of the closing date for consultation. [Con-3793]
- On 4 April 2024 Santos phoned Neptune Energy regarding consultation for Production Operations activities and was not able to leave a message.
- On 2 May 2024. Santos emailed Neptune Energy further to previous correspondence, to advise it had extended the consultation period until 16 May 2024. In providing this extension of time, Santos advised that the information in the booklet and factsheet had been updated to account for an additional risk associated with the proposed activity. Santos confirmed that consultation will close on the revised date unless Santos hears otherwise from Neptune Energy. [Con-3970]
- On 20 May 2024, Neptune Energy emailed Santos and advised that it does not have any input for the Production Operations activities and supports Santos' Barossa Gas Project. Neptune Energy further requested that due to the location of the Petrel field in the Bonaparte Basin, that they continue to be kept updated of Santos' activities in relation to this field and mentioned Eni's acquisition of Neptune Energy. [Con-3979]
- On 10 July 2024 Santos emailed Neptune Energy to advise the consultation period had been completed. Santos advised it consultation had now closed for the purpose of Santos finalising and submitting environment plans for these activities to government regulators for assessment. [Con-5130]
- On 15 October 2024 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that the EP had been submitted to NOPSEMA for assessment, provided a link to the full EP on the NOPSEMA website and Santos' contact details for any queries. The Update also stated that the OEMP would be submitted to the NT-DITT for assessment. [Con- 5982]
- On 23 January 2025 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that, following a request from NOPSEMA to provide further information, Santos had re-submitted the EP with some amendments in December 2024 for further assessment. The Update also re-stated that the OEMP would be submitted to the NT-DITT for assessment. [Con-6036]
- No further correspondence has been received from Neptune Energy.
- Santos notes that Eni Australia's acquisition of Neptune was completed in January 2024. https://www.eni.com/en-IT/media/press-release/2024/01/eni-acquisition-neptune-completed.html

Summary of response by Relevant Person	Assessment of merits	Santos' Response Statement	OEMP
Neptune Energy requested that Santos keep them informed of activities in relation to Barossa.	Santos has considered the matters raised in the OEMP. Santos provides pre-activity notifications and quarterly project updates to Eni, noting that the Neptune assets are now owned by Eni Australia.	No response required.	Quarte

## PTTEP Australia

Summary of consultation effort:

- On 9 February 2024 Santos emailed PTTEP Australia to advise it of preliminary consultation regarding proposed activities for consultation to be managed under the Coastal Waters OEMP and Barossa Production Operations EP. [Con-3787]
- The email advised that Santos was seeking information to better understand:
  - o if you are from a government Department or agency, how the proposed Production Operations activities may be relevant to your department or agency; and
  - what (if any) functions, interests or activities you or your organisation have that may be affected by the proposed Production Operations activities. 0
- The email included information on the regulatory process for the activities in Commonwealth and NT jurisdictions and links to a Santos information booklet on the proposed activities and a NOPSEMA brochure on the consultation process and details of how to contact Santos to register as a Relevant Person. The email stated that the consultation phase would commence on 11 March 2024 and close on 9 April 2024.
- On 11 March 2024 Santos emailed PTTEP Australia further to the previous correspondence, to advise that it had commenced the consultation phase which would run until 9 April 2024. In addition to the previous information again being provided, Santos provided information on Relevant Persons' entitlements under the regulatory processes, details of how to provide feedback and a reminder of the closing date for consultation. [Con-3793]
- On 3 April 2024 Santos phoned PTTEP Australia regarding consultation for Production Operations activities and was not able to leave a message.
- On 2 May 2024, Santos emailed PTTEP Australia further to previous correspondence, to advise it had extended the consultation period until 16 May 2024. In providing this extension of time, Santos advised that the information in the booklet and factsheet had been updated to account for an additional risk associated with the proposed activity. Santos confirmed that consultation will close on the revised date unless Santos hears otherwise from PTTEP. [Con-3982]
- On 2 May 2024, PTTEP Australia emailed Santos and advised it had no input or objections to the proposed Production Operations activities. [Con-3985]
- On 10 July 2024 Santos emailed PTTEP Australia to advise the consultation period had been completed. Santos advised it consultation had now closed for the purpose of Santos finalising and submitting environment plans for these activities to government regulators for assessment. [Con-5130]
- On 15 October 2024 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that the EP had been submitted to NOPSEMA for assessment, provided a link to the full EP on the NOPSEMA website and Santos' contact details for any queries. The Update also stated that the OEMP would be submitted to the NT-DITT for assessment. [Con- 5982]
- On 23 January 2025 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that, following a request from NOPSEMA to provide further information. Santos had re-submitted the EP with some amendments in December 2024 for further assessment. The Update also re-stated that the OEMP would be submitted to the NT-DITT for assessment. [Con-6036] No further correspondence has been received from PTTEP Australia.

## Santos

## reference

erly updates are included in Table 8-13

PTTEP Australia advised it had no input or objections. Santos considers it has provided sufficient information and a reasonable period of time for consultation. Santos considers Section 25 consultation requirements to have	Summary of response by Relevant Person	Assessment of merits	Santos' Response Statement	OEMP
been met.	PTTEP Australia advised it had no input or objections.	Santos considers it has provided sufficient information and a reasonable period of time for consultation. Santos considers Section 25 consultation requirements to have been met.	No response required.	Not ap

## **Shell Development**

Summary of consultation effort:

- On 9 February 2024 Santos emailed Shell Development to advise it of preliminary consultation regarding proposed activities for consultation to be managed under the Coastal Waters OEMP and Barossa Production Operations EP. [Con-3787], The email advised that Santos was seeking information to better understand:
  - o if you are from a government Department or agency, how the proposed Production Operations activities may be relevant to your department or agency; and
  - what (if any) functions, interests or activities you or your organisation have that may be affected by the proposed Production Operations activities. 0
- The email included information on the regulatory process for the activities in Commonwealth and NT jurisdictions and links to a Santos information booklet on the proposed activities and a NOPSEMA brochure on the consultation process and details of how to contact Santos to register as a Relevant Person. The email stated that the consultation phase would commence on 11 March 2024 and close on 9 April 2024.
- On 11 March 2024 Santos emailed Shell Development further to the previous correspondence, to advise that it had commenced the consultation phase which would run until 9 April 2024. In addition to the previous information again being provided, Santos provided information on Relevant Persons' entitlements under the regulatory processes, details of how to provide feedback and a reminder of the closing date for consultation. [Con-3793]
- On 3 April 2024 Santos phoned Shell Development regarding consultation for Production Operations activities and left a voice mail message.
- On 2 May 2024, Santos emailed Shell Development further to previous correspondence, to advise it had extended the consultation period until 16 May 2024. In providing this extension of time, Santos advised that the information in the booklet and factsheet had been updated to account for an additional risk associated with the proposed activity. Santos confirmed that consultation will close on the revised date unless Santos hears otherwise from Shell Development. [Con-3983]
- On 10 July 2024 Santos emailed Shell Development to advise the consultation period had been completed. Santos advised Shell Development that it considered consultation had now closed for the purpose of Santos finalising and submitting environment plans for these activities to government regulators for assessment. [Con-5130]
- On 15 October 2024 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that the EP had been submitted to NOPSEMA for assessment, provided a link to the full EP on the NOPSEMA website and Santos' contact details for any queries. The Update also stated that the OEMP would be submitted to the NT-DITT for assessment. [Con- 5982]
- On 23 January 2025 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that, following a request from NOPSEMA to provide further information, Santos had re-submitted the EP with some amendments in December 2024 for further assessment. The Update also re-stated that the OEMP would be submitted to the NT-DITT for assessment. [Con-6036]
- Notwithstanding the consultation information provided and the steps described above, no comments or input were received on the Coastal Waters OEMP from Shell Development.

Summary of response by Relevant Person	Assessment of merits	Santos' Response Statement	OEMP
No correspondence was received from Shell Development.	Santos considers it has provided sufficient information and a reasonable period of time for consultation.	No response required.	Not ap
	Santos considers Section 25 consultation requirements to have been met.		

## Santos

reference

plicable.

reference

oplicable.

## SundaGas

Summary of consultation effort:

- On 9 February 2024 Santos emailed Sunda Gas to advise it of preliminary consultation regarding proposed activities for consultation to be managed under the Coastal Waters OEMP and Barossa Production Operations EP. [Con-3787]
- The email advised that Santos was seeking information to better understand:
  - o if you are from a government Department or agency, how the proposed Production Operations activities may be relevant to your department or agency; and
  - what (if any) functions, interests or activities you or your organisation have that may be affected by the proposed Production Operations activities. 0
- The email included information on the regulatory process for the activities in Commonwealth and NT jurisdictions and links to a Santos information booklet on the proposed activities and a NOPSEMA brochure on the consultation process and details of how to contact Santos to register as a Relevant Person. The email stated that the consultation phase would commence on 11 March 2024 and close on 9 April 2024.
- On 11 March 2024 Santos emailed Sunda Gas further to the previous correspondence, to advise that it had commenced the consultation phase which would run until 9 April 2024. In addition to the previous information again being provided, Santos provided information on Relevant Persons' entitlements under the regulatory processes, details of how to provide feedback and a reminder of the closing date for consultation. [Con-3793]
- On 5 April 2024 Santos phoned Sunda Gas regarding consultation for Production Operations activities and was not able to leave a message.
- On 6 May 2024. Santos emailed Sunda Gas further to previous correspondence, to advise it had extended the consultation period until 20 May 2024. In providing this extension of time, Santos advised that the information in the booklet and factsheet had been updated to account for an additional risk associated with the proposed activity. Santos confirmed that consultation will close on the revised date unless Santos hears otherwise from Sunda Gas. [Con-3988]
- On 10 July 2024 Santos emailed SundaGas to advise the consultation period had been completed. Santos advised SundaGas that it considered consultation had now closed for the purpose of Santos finalising and submitting environment plans for these activities to government regulators for assessment. [Con-5130]
- On 15 October 2024 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that the EP had been submitted to NOPSEMA for assessment, provided a link to the full EP on the NOPSEMA website and Santos' contact details for any queries. The Update also stated that the OEMP would be submitted to the NT-DITT for assessment. [Con- 5982]
- On 23 January 2025 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that, following a request from NOPSEMA to provide further information, Santos had re-submitted the EP with some amendments in December 2024 for further assessment. The Update also re-stated that the OEMP would be submitted to the NT-DITT for assessment. [Con-6036]
- Notwithstanding the consultation information provided and the steps described above, no comments or input were received on the Coastal Waters OEMP from SundaGas.

Summary of response by Relevant Person	Assessment of merits	Santos' Response Statement	OEMP
No correspondence was received from SundaGas.	Santos considers it has provided sufficient information and a reasonable period of time for consultation.	No response required.	Not ap
	Santos considers Section 25 consultation requirements to have been met.		

## **Woodside Energy**

Summary of consultation effort:

- On 9 February 2024 Santos emailed Woodside Energy to advise it of preliminary consultation regarding proposed activities for consultation to be managed under the Coastal Waters OEMP and Barossa Production Operations EP. [Con-3787]
- The email advised that Santos was seeking information to better understand:
  - o if you are from a government Department or agency, how the proposed Production Operations activities may be relevant to your department or agency; and
  - what (if any) functions, interests or activities you or your organisation have that may be affected by the proposed Production Operations activities.
- The email included information on the regulatory process for the activities in Commonwealth and NT jurisdictions and links to a Santos information booklet on the proposed activities and a NOPSEMA brochure on the consultation process and details of how to contact Santos to register as a Relevant Person. The email stated that the consultation phase would commence on 11 March 2024 and close on 9 April 2024.
- On 11 March 2024 Santos emailed Woodside Energy further to the previous correspondence, to advise that it had commenced the consultation phase which would run until 9 April 2024. In addition to the previous information again being provided, Santos provided information on Relevant Persons' entitlements under the regulatory processes, details of how to provide feedback and a reminder of the closing date for consultation. [Con-3793]
- On 3 April 2024 Santos phoned Woodside Energy and left a message with a company representative regarding consultation for Production Operations activities.
- On 6 May 2024, Santos emailed Woodside Energy further to previous correspondence, to advise it had extended the consultation period until 20 May 2024. In providing this extension of time, Santos advised that the information in the booklet and factsheet had been updated to account for an additional risk associated with the proposed activity. Santos confirmed that consultation will close on the revised date unless Santos hears otherwise from Woodside Energy. [Con-3991]
- On 10 July 2024 Santos emailed Woodside Energy to advise the consultation period had been completed. Santos advised Woodside Energy that it considered consultation had now closed for the purpose of Santos finalising and submitting environment plans for these activities to government regulators for assessment. [Con-5130]
- On 15 October 2024 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that the EP had been submitted to NOPSEMA for assessment, provided a link to the full EP on the NOPSEMA website and Santos' contact details for any queries. The Update also stated that the OEMP would be submitted to the NT-DITT for assessment. [Con- 5982]
- On 23 January 2025 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that, following a request from NOPSEMA to provide further information. Santos had re-submitted the EP with some amendments in December 2024 for further assessment. The Update also re-stated that the OEMP would be submitted to the NT-DITT for assessment. [Con-6036]
- Notwithstanding the consultation information provided and the steps described above, no comments or input were received on the Coastal Waters OEMP from Woodside Energy.

Summary of response by Relevant Person	Assessment of merits	Santos' Response Statement	OEMP
No correspondence was received from Woodside.	Santos considers it has provided sufficient information and a reasonable period of time for consultation.	No response required.	Not ap
	Santos considers Section 25 consultation requirements to have been met.		



reference

oplicable.

reference

plicable.

#### 4.8.6 **Environmental Conservation Organisations**

## Table 4-16: Consultation Summary Table – Environmental Conservation Organisations

Section 25 (1)(d) of the OPGGS(E)R: Persons or organisations whose functions, interests or activities may be affected by the activities to be carried out under the environment plan

### **Conservation Organisations**

### **ATSEA-2** Project

## Summary of consultation effort:

- On 9 February 2024 Santos emailed ATSEA-2 Project to advise it of preliminary consultation regarding proposed activities for consultation to be managed under the Coastal Waters OEMP and Barossa Producti
- The email advised that Santos was seeking information to better understand:
  - o if you are from a government Department or agency, how the proposed Production Operations activities may be relevant to your department or agency; and
  - what (if any) functions, interests or activities you or your organisation have that may be affected by the proposed Production Operations activities. 0
- The email included information on the regulatory process for the activities in Commonwealth and NT jurisdictions and links to a Santos information booklet on the proposed activities and a NOPSEMA brochure on the consultation process and details of how to contact Santos to register as a Relevant Person. The email stated that the consultation phase would commence on 11 March 2024 and close on 9 April 2024.
- On 13 March 2024 Santos emailed ATSEA-2 Project further to the previous correspondence, to advise that it had commenced the consultation phase which would run until 9 April 2024. In addition to the previous information again being provided, Santos provided information on Relevant Persons' entitlements under the regulatory processes, details of how to provide feedback and a reminder of the closing date for consultation. [Con-3794]
- On 4 April 2024 Santos phoned ATSEA-2 Project and left a message regarding consultation for Production Operations activities with a team member.
- On 8 May 2024, Santos emailed ATSEA- 2 Project further to previous correspondence to advise it had extended the consultation period until 22 May 2024. In providing this extension of time, Santos advised that the information in the booklet and factsheet had been updated to account for an additional risk associated with the proposed activity. Santos confirmed that consultation will close on the revised date unless Santos hears otherwise from ATSEA- 2. [Con-4029]
- On 10 July 2024 Santos emailed ATSEA-2 Project to advise the consultation period had been completed. Santos advised ATSEA-2 Project that it considered consultation had now closed for the purpose of Santos finalising and submitting environment plans for these activities to government regulators for assessment. [Con-5130]
- On 15 October 2024 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that the EP had been submitted to NOPSEMA for assessment, provided a link to the full EP on the NOPSEMA website and Santos' contact details for any queries. The Update also stated that the OEMP would be submitted to the NT-DITT for assessment. [Con- 5982]
- On 23 January 2025 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that, following a request from NOPSEMA to provide further information. Santos had re-submitted the EP with some amendments in December 2024 for further assessment. The Update also re-stated that the OEMP would be submitted to the NT-DITT for assessment. [Con-6036]
- Notwithstanding the consultation information provided and the steps described above, no comments or input were received on the Coastal Waters OEMP from ATSEA-2 Project.

Summary of response by Relevant Person	Assessment of merits	Santos' Response Statement	OEMP reference
No response was received from ATSEA-2 Project.	Santos considers it has provided sufficient information and a reasonable period of time for consultation.	No response required.	Not applicable.
	Santos considers Section 25 consultation requirements to have been met.		

## Australia Institute

Summary of consultation effort:

- On 9 February 2024 Santos emailed Australia Institute to advise it of preliminary consultation regarding proposed activities for consultation to be managed under the Coastal Waters OEMP and Barossa Production Operations EP. [Con-3787]
- The email advised that Santos was seeking information to better understand:
  - o if you are from a government Department or agency, how the proposed Production Operations activities may be relevant to your department or agency; and
  - what (if any) functions, interests or activities you or your organisation have that may be affected by the proposed Production Operations activities.
- The email included information on the regulatory process for the activities in Commonwealth and NT jurisdictions and links to a Santos information booklet on the proposed activities and a NOPSEMA brochure on the consultation process and details of how to contact Santos to register as a Relevant Person. The email stated that the consultation phase would commence on 11 March 2024 and close on 9 April 2024.
- On 13 March 2024 Santos emailed Australia Institute further to the previous correspondence, to advise that it had commenced the consultation phase which would run until 9 April 2024. In addition to the previous information again being provided, Santos provided information on Relevant Persons' entitlements under the regulatory processes, details of how to provide feedback and a reminder of the closing date for consultation. [Con-3794]
- On 4 April 2024 Santos phoned Australia Institute and left a voice message regarding consultation for Production Operations activities.
- On 2 May 2024, Santos emailed Australia Institute further previous correspondence to advise it had extended the consultation period until 16 May 2024. In providing this extension of time, Santos advised that the information in the booklet and factsheet had been updated to account for an additional risk associated with the proposed activity. Santos confirmed that consultation will close on the revised date unless Santos hears otherwise from the Australia Institute. [Con-4012].
- On 3 May 2024 the Australia Institute emailed Santos to thank it for the reminder and advised the Australia Institute would be providing input by 16 May. [Con-4016]
- On 10 July 2024 Santos emailed Australia Institute to advise the consultation period had been completed. Santos advised it consultation had now closed for the purpose of Santos finalising and submitting environment plans for these activities to government regulators for assessment. [Con-5130]
- On 15 October 2024 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that the EP had been submitted to NOPSEMA for assessment, provided a link to the full EP on the NOPSEMA website and Santos' contact details for any queries. The Update also stated that the OEMP would be submitted to the NT-DITT for assessment. [Con- 5982]
- On 23 January 2025 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that, following a request from NOPSEMA to provide further information, Santos had re-submitted the EP with some amendments in December 2024 for further assessment. The Update also re-stated that the OEMP would be submitted to the NT-DITT for assessment. [Con-6036]
- No further correspondence has been received from Australia Institute.

|--|



on	Operations	ED	[Con_3787]
On	Operations	EP.	[0011-3707]

**OEMP** reference
The Australia Institute responded that it would be		Santos notes that the institute did not provide further input by 16 May 2024.	No response required.
pro	oviding input by 16 May 2024.	Santos considers it has provided sufficient information and a reasonable period of time for consultation.	
		Santos considers Section 25 consultation requirements to have been met.	
Αι	stralian Conservation Foundation (ACF)		
Su	immary of consultation effort		
	On 9 February 2024 Santos emailed ACE to ad	vise it of preliminary consultation regarding proposed activities for consultation to be managed under the Coastal	Waters OEMP and Barossa Production Operation
	The email advised that Santos was soaking info	rmation to better understand:	
•	if you are from a government Departm	initiation to better understand.	r agoney; and
	o what (if any) functions, interests or act	initias you or your organisation have that may be affected by the proposed Paressa Production Operations activiti	agency, and
	The empiliable disference on the regulator	villes you of your organisation have that may be anected by the proposed balossa Production Operations activiti	so.
•	Santos to register as a Relevant Person. The er	nail stated that the consultation phase would commence on 11 March 2024 and close on 9 April 2024.	proposed activities and a NOPSEMA brochure of
•	On 13 March 2024 Santos emailed the ACF to a section of Santos' website and NOPSEMA's Br	advise that the formal consultation period for Production Operations activities had commenced. Santos provided I ochure: 'Consultation on offshore petroleum environment plans – Information for the Community'. Santos advised	nks to the Barossa Production Operations Activity
•	On 9 April 2024 the ACF wrote to Santos to exp	ress its concerns and request further information on a range of topics related to Production Operations activities.	The ACF's concerns primarily related to sufficience
	project, the project's ability to meet the requirem	nents of the Safeguard Mechanism, potential spill impacts and decommissioning. The ACF also requested a meet	ing with Santos to discuss Production Operations
•	On 13 June 2024 Santos wrote to the ACF in re	sponse to the ACF's letter of 9 April 2024. Santos responded to each of the ACF's concerns and information requ	ests, as well as its concerns regarding the consul
	On 19 June 2024 the ACE wrote to Santos to su	eet. Santos requested the ACF provide its availability for a meeting. [Con-5022]	d in its latter of 13 June 2024, and Santos' corres
•	requirements. [Con-5023]	aggest dates and times for a meeting. The ACT asserted that it was not satisfied with the answers bantos provide	
•	On 26 June 2024, Santos wrote to the ACF to c	onfirm its availability for a meeting at 2pm on 9 July 2024 (WST). Santos requested the ACF send an agenda by	1 July 2024. [Con-5024]
•	On 26 June 2024, the ACF wrote to Santos to c	onfirm its attendees at the meeting [Con-5025]	
•	On 2 July 2024 Santos emailed ACF thanking it	for confirming the meeting attendees and again asked ACF to send an agenda by no later than 4 July 2024. [Con	1-5028]
	On 9 July 2024 Santos emailed ACF to ask the	in to send an agenda of list of topics that they would like to discuss at the upcoming meeting. [Con-5136]	
•	On 9 July 2024, Santos met with ACF via a Mic	rosoft Teams video conference. [Con-5212] At the meeting the ACF:	
	<ul> <li>queried how Santos is assessing clima</li> </ul>	te change impacts attributable to the Activity on matters of national environmental significance and vulnerable ec	osystems and communities;
	<ul> <li>queried how the Activity is aligned with</li> </ul>	the Paris Agreement temperature targets;	
	<ul> <li>queried now Santos is assessing scop</li> <li>requested details regarding the management</li> </ul>	e 3 emissions impacts; ement procedures and/or management plans that will be in place for marine fauna and ecosystems	
•	At the meeting on 9 July 2024 Santos stated that	at:	
	<ul> <li>there are limitations to linking the Activ</li> </ul>	ity's emissions with specific climate change impacts;	
	<ul> <li>it has considered impacts to matters of it cale and ACE's concerns regard</li> </ul>	national environmental significance from global climate change;	and communities
	<ul> <li>It acknowledged ACF's concerns regard</li> <li>the Barossa project will be subject to the</li> </ul>	and chinate change impacts on matters of national environmental significance, as well as vulnerable ecosystems the Safeguard Mechanism, which is the regime that implements Australia's Paris Agreement targets:	and communities,
	<ul> <li>customers of the Barossa project will be</li> </ul>	e subject to their own countries' emissions regulation frameworks, which implement their Paris Agreement Natior	ally Determined Contributions (NDCs);
	<ul> <li>the management procedures and/or planet</li> </ul>	ans for marine fauna and ecosystems align with DCCEEW guidance and EPBC regulations;	
	<ul> <li>It will consider any further information t</li> </ul>	hat ACF provides in writing following the meeting.	to of the Activity on its functions, interacts or activ
	On 11 July 2024 ACE emailed Santos [Con-51]	it will write to Santos to set out its remaining concerns and any runner information it requires to assess the impact	is of the Activity of its functions, interests of activity
-	<ul> <li>reiterated its concerns regarding how \$</li> </ul>	Santos is considering climate change impacts attributable to the Activity;	
	<ul> <li>reiterated its concerns regarding how t</li> </ul>	he Barossa project will comply with the Paris Agreement 1.5-degree scenario; and	
	<ul> <li>request details of the control measures</li> </ul>	and protocols that Santos has developed to protect marine fauna and biologically important areas.	
•	Santos advised ACE that Santos was finalising	onse to ACF's email of 11 July 2024. Santos responded to the matters raised by ACF. Santos thanked ACF for it the Barossa Production Operations EP for submission in coming weeks. [Con-5283]	s comments and submissions in respect of the ris
•	On 9 September 2024 ACF responded to Santo	s' email of 7 August 2024 claiming Santos had not sufficiently addressed a number of the concerns ACF had rais	ed during the consultation process to date. [Con-
•	On 20 September 2024 Santos wrote to ACF in	response to ACF's letter of 9 September 2024. Santos responded to the matters raised or restated by ACF. San	tos thanked ACF for its comments and submissio
	relation to the activity. Santos advised ACF that	Santos was finalising the EP for submission. [Con-5644]	
•	On 30 September 2024 ACF emailed Santos in follow-up correspondence on these matters before	response to Santos' letter of 20 September 2024. In the email the ACF stated it did not agree with Santos' assert	ion that sufficient information has been provided,
•	On 15 October 2024 Santos emailed the Baross	a Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparat	ion of the Production Operations EP and OEMP.
	NOPSEMA for assessment, provided a link to the	he full EP on the NOPSEMA website and Santos' contact details for any queries. The Update also stated that the	OEMP would be submitted to the NT-DITT for ass
•	On 19 December 2024 Santos emailed ACF to	advise that, given that no further correspondence was received by 14 October or since, and having regard to the	elapsing of time, Santos has proceeded on the ba
	feedback and that consultation for preparation c	of environment plans relating to Barossa Production Operations activities is complete. [Con-6023]	ates early payt weak [Can CO24]
•	On 20 December 2024 ACF responded to Santo	os email of 19 December 2024. ACF stated in the email that it was infalising its response and will provide it to Sa	tos eany next week. [Con-6031]
	On 23 January 2025 Santos emailed the Baross	se to Salitos. [Coll-0034] se Project Quarterly I Indate to all stakeholders on its distribution list, including all those consulted during preparat	ion of the Production Operations EP and OEMP
	NOPSEMA to provide further information, Santo On 31 January 2025 Santos emailed ACF and r	is had re-submitted the EP with some amendments in December 2024 for further assessment. The Update also r provided responses to the matters raised by ACF in its email of 23 December 2024. [Con-6038]	e-stated that the OEMP would be submitted to the
•	On 24 March 2025 ACF responded to Santos' le	etter of 31 January 2025 with concerns about the content of the Barossa Production Operations EP. [Con-6071]	
•	On 31 March 2025 Santos responded to the AC	F's letter of 24 March 2025 [Con-6017]	



Not applicable.

ns EP. [Con-3787]

on the consultation process and details of how to contact

y Information Booklet, the Barossa Production Operations [Con-3794]

cy of information in relation to GHG emissions from the s activities. [Con-4007]

ltation process. Santos also provided the ACF with notice of

pondence to date was not sufficient to meet the consultation

vities.

sks, impacts, and potential controls in relation to the activity.

n-5642] ions in respect of the risks, impacts, and potential controls in

it was reviewing the information sent and would provide

The Update advised that the EP had been submitted to sessment. [Con- 5982] asis that the ACF does not intend to provide any further

The Update advised that, following a request from e NT-DITT for assessment. [Con-6036]

Summary of response by Relevant Person	Assessment of merits	Santos' Response Statement	OEMP reference
<ul> <li>ACF correspondence to Santos on 9 April 2024</li> <li>Concern raised that the Information Booklet provided insufficient information in relation to: <ul> <li>GHG emissions from the project;</li> <li>the project's ability to meet the requirements of the Safeguard Mechanism;</li> <li>whether carbon capture and storage (CCS) is considered a component of the project;</li> <li>hydrocarbon and non-hydrocarbon spills; and</li> <li>decommissioning.</li> </ul> </li> </ul>	Santos has provided a response to ACF and considered the matters raised by ACF in the OEMP. CCS is not part of the Barossa Gas Project and thereby not relevant to the Coastal Waters OEMP. Santos does not agree with the ACF's assertion that the information booklet does not provide sufficient information to enable the ACF to make an informed assessment of any potential consequences of the Activity on its functions, interests or activities. The information booklet provides a comprehensive description of the environment that may be affected by the Activity, identification of impacts and risks from planned activities and unplanned events and associated proposed control measures.	Santos' correspondence to ACF on 13 June 2024 in response to ACF's letter of 9 April 2024 <sup>142</sup> Santos notes your comments regarding the sufficiency of information provided by Santos in consultation, and your requests for further information. Santos considers that the information provided to date (in the Production Operations Information Booklet and NT Waters GEP Operations Overview Factsheet) is sufficient for the ACF to make an informed assessment of any potential consequences of the Production Operations Activity (Activity) on any of its functions, interests or activities (FIAs). Notwithstanding this, Santos has provided responses to requests #1-15 below, where possible and reasonable.	Not relevant to the ACF's assertion that the information booklet does not provide sufficient information.
The ACF raised concerns about the project's emissions intensity and availability of offsets and requested further information in relation to how the Barossa project will enter the Safeguard Mechanism system at net zero emissions.	Santos has provided a response to ACF and considered the matters raised by ACF in the OEMP. Santos will meet its Safeguard Mechanism compliance obligations, set by the CER in accordance with Australian Government policy and emissions reduction targets, through the purchase and/or generation of Australian Carbon Credit Units (ACCUs) or Safeguard Mechanism Credits (SMCs). The responsible Commonwealth minister has communicated to the Barossa joint venture partners on a number of occasions that he anticipates there being ACCUs available for this project. Natural gas remains an integral part of the energy mix out to 2050 (AEMO 2024, IEA 2023c) and, with ACCUs being part of the national carbon management framework, it is reasonable to assume the availability of ACCUs for gas projects would have been contemplated in setting Australia's NDCs. In this regard, the projected ACCU demand and issuance for the period 2025-2040 is set out at Figure 17 of Australia's emissions projections 2024 chart data, which was released with the DCCEEW reports released in November 2024, as referenced above. This data has been derived by reference to the emissions projection information in these reports and it is therefore reasonable to assume that the number of ACCUs anticipated by the Barossa Development have been considered with relation to Australia's anticipated ACCU issuance. Santos' reliance on ACCUs will form part of its reporting to the CER and will continue to be monitored by the CER. Section 6.3.2.6.2. further captures Santos' consideration of the availability of ACCUs and SMCs. GHG emissions reporting for this Activity are included in the OEMP. Santos views that ACF has been provided sufficient information to enable the ACF to make an informed assessment of any potential consequences of the Activity on its functions, interests or activities.	[Response (1)] The treatment of the Barossa project under the Safeguard Mechanism is a matter for the Clean Energy Regulator. Santos will abide by the Clean Energy Regulator's final determination. There are various options available to meet a baseline (including direct abatement and acquiring offsets, in addition to mechanisms available under the Safeguard Rules such as borrowing adjustments and multi-year monitoring periods). Santos will be required to comply with the applicable baseline for the Barossa project in each compliance year. It is a matter for Santos to determine how it will achieve this compliance. This information is not necessary for the ACF to make an informed assessment of the possible consequences of the Activity on any of its FIAs.	Section 6.3. BAO-CM-6.3.11
The ACF requested Santos clarify whether CCS is a component of the project.	Santos has provided a response to ACF and considered the matters raised by ACF in the OEMP. It was clarified that CCS is not part of the Activity described in the EP or the Coastal Waters OEMP.	<b>[Response (2)]</b> While Santos has committed to explore CCS opportunities at Bayu-Undan and elsewhere, CCS is not part of the Barossa development and any CCS developments will be subject to a separate environmental approvals process, and is not within the scope of the Production Operations EP.	Not applicable.
The ACF requested a range of further information in relation to CCS development and operation.	Santos has provided a response to ACF and considered the matters raised by ACF in the OEMP. As CCS is not part of the Activity for the EP or the Coastal Waters OEMP, Santos did not provide ACF further information in response to this request.	<b>[Response (3)]</b> As noted in response to #2, CCS is not part of the Barossa development, and any CCS developments will be the subject of a separate environmental approvals process. Therefore, the information you have requested is not relevant to the EP, which is the subject of this consultation.	Not applicable.
The ACF raised concerns about the project's emissions impact on climate change. The ACF requested further information regarding how Santos will address climate change impacts of scope 1 and 3 emissions.	Santos has provided a response to ACF and considered the matters raised by ACF in the OEMP. The Barossa Production Operations EP and this OEMP acknowledge the linear relationship between cumulative anthropogenic CO <sub>2</sub> emissions and global warming. The EP and OEMP references the latest commentary from the IPCC on the causes and impacts of climate change. Climate change is a global problem with the solution being led at the international level. Domestically, GHG emissions are regulated through Australia's NDC and the Barossa JV and its contractors have legal obligations to ensure that the Activity operates in accordance with that framework. The projected emissions for the Activity described in the Coastal Waters OEMP are contextualised against established emissions budgets (national and global). The Barossa Joint Venture (JV) has adopted environmental performance outcomes and control measures directed to minimising the GHG emissions from the Activity. A range of controls have been considered for both direct (Scope 1) and indirect (Scope 3) emissions. Santos views that ACF has been provided sufficient information to enable the ACF to make an informed assessment of any potential consequences of the Activity on its functions, interests or activities.	<ul> <li>[Response (4)] As outlined in the Booklet, as a result of the complex nature of the global emissions system, climate change impacts cannot be meaningfully linked to any one activity or emissions source.</li> <li>The 'GHG emissions' section of the Booklet provides information about Santos' proposed control measures to reduce impacts and risks of Scope 1 and 3 emissions from the Activity to as low as reasonably practicable and an acceptable level.</li> <li>To the extent the ACF believes that there is any potential impact on the ACF's FIAs as a result of GHG emissions, the ACF is able to make an informed assessment of that impact with the information set out in the Booklet, including the total annual estimates provided. Santos welcomes the ACF's input regarding any control measures that it considers may be appropriate to adopt for Santos' consideration when preparing the EP for submission to NOPSEMA, including in relation to GHG emissions. Santos has invited a meeting with the ACF, in part to provide an opportunity to receive any such input.</li> </ul>	Section 6.3. BAO-CM-6.3.1 BAO-CM-6.3.9 BAO-CM-6.3.11 BAO-CM-6.3.12 BAO-CM-6.3.16 BAO-CM-6.3.17 BAO-CM-6.3.18 BAO-CM-6.3.21 BAO-CM-6.3.22 BAO-CM-6.4.10
The ACF requested further information regarding how Santos will address the direct or indirect consequences to the environment, including matters of national environmental significance, from GHG emissions.	Santos has provided a response to ACF and considered the matters raised by ACF in the OEMP. The Barossa JV has adopted environmental performance outcomes and control measures directed to minimising the GHG emissions from the Activity. A range of controls have been considered for both direct (Scope 1) and indirect (Scope 3) emissions.	[Response (5)] See response to #4. Globally, the energy sector needs to simultaneously invest in and develop two energy systems; ensure the current system continues to operate and meet global energy demands, while increasing efforts to build the new system centred on lower carbon alternatives. Maintaining a steady supply of	Section 6.3.

<sup>&</sup>lt;sup>42</sup> Santos has given each ACF response a corresponding response number which is marked in bold. Where Santos has cross-referred to earlier responses within its letter of 13 June 2024, it has not reproduced the text in this consultation table but rather than included the response number.



The ACF raised concerns that providing more gas to the market will make it more difficult for countries to progress renewable developments.	The OEMP describes the role of natural gas in energy transition. Gas plays a critical role in the transition to a lower carbon future, able to flexibly fill market supply gaps as alternative energy sources emerge. Analysis of almost 100 IPCC scenarios, all aligned to the temperature goals of the Paris Agreement, show a range of gas demand profiles and all include a continued role for gas in the global energy mix out to 2050. Santos views that ACF has been provided sufficient information to enable the ACF to make an informed assessment of any potential consequences of the Activity on its functions, interests or activities.	gas allows for the orderly and measured development of the new energy system. For further information on this topic, please refer to Santos's Sustainability and Climate Report 2023, available here: <u>https://www.santos.com/wp- content/uploads/2024/03/Sustainability-and-Climate-Report-2023.pdf</u>	
The ACF raised concerns regarding Santos' proposed scope 3 emission control measure. The ACF asked Santos whether it will be checking its customers' countries performance against their Paris Agreement NDC's and adjust or revoke sales contracts accordingly.	Santos has provided a response to ACF. The ACF's concern relates to a control measure proposed to be implemented under the Production Operations EP. This control measure is not proposed to be implemented under the OEMP.	[Response (6)] See responses to #4 and #5. Santos otherwise thanks the ACF for providing this feedback, which it will consider.	BAO-CM-6.3.21 BAO-CM-6.3.22
<ul> <li>The ACF asked Santos whether it will ensure customers are using its product in a responsible way to reduce emissions. The ACF provided examples, which included ensuring:</li> <li>Santos's customers are meeting best practice standards for LDAR and MRV; and</li> <li>Santos is not selling gas to customers for new (post 2020) gas-fired power stations unless it can unequivocally prove that energy is displacing coal and not renewables.</li> </ul>	Santos has provided a response to ACF. Scope 3 emissions associated with Barossa Gas Project customers and the associated use of sold products are beyond the scope of this document and are addressed in the Production Operations EP.	[Response (7)] See response to #6.	BAO-CM-6.3.21 BAO-CM-6.3.22
<ul> <li>The ACF raised concerns that Santos had provided inconsistent emission estimates for different Barossa approval documents. The ACF requested:</li> <li>further information regarding discrepancies in emissions estimates;</li> <li>confirmation regarding whether Santos will share data on the methodologies used to calculate emissions estimates;</li> <li>confirmation regarding whether Santos will provide ongoing transparency to the public in relation to the quantity of Greenhouse Gases it finds in it's wells and details on the technologies used and uncertainty factors.</li> </ul>	Santos has provided a response to ACF and considered the matters raised by ACF in the OEMP. The discrepancy between emissions estimates provided in the SER and the Activity information booklet are explained by different inputs and assumptions for the respective estimates, based on the level of engineering and design at time of the approval preparation, with the conservative estimates reflecting this. The OEMP includes greenhouse gas estimates for both Scope 1 and Scope 3 emissions including the assumptions that underpin these estimates. Santos will meet applicable regulatory requirements of the National Greenhouse and Energy Reporting Act 2007, as relevant to Barossa production operations.	<ul> <li>[Response (8)]</li> <li>a. The annual emissions estimates included in the Booklet are applicable to the Activity. The annual emissions estimates provided in the Booklet are conservative (when extrapolated for 25 years of production operations) given annual emissions are expected to reduce over the life of the Activity as production rates decline. For the purpose of the Activity, the ACF should focus its review on the information provided in the Booklet. To the extent that the ACF believes there is any potential impact on the ACF's FIAs as a result of GHG emissions, the ACF is able to make an informed assessment of that impact with the information set out in the Booklet.</li> <li>b. Information regarding calculation of the GHG emissions and emissions intensity (as required) will be presented in the EP for consideration and assessment by the Regulator for assessment against the requirements of the Regulations.</li> <li>c. (i) Santos has and will continue to be transparent in both the EP, and other published information, about the quantity of GHG emissions in the Barossa wells.</li> <li>(ii) Santos will meet applicable regulatory requirements of the National Greenhouse and Energy Reporting Act 2007, as relevant to Barossa production operations.</li> <li>In any event, Santos considers that the ACF does not require this additional technical information about reporting of Barossa GHG emissions in order to make an informed assessment of the potential consequences of the Activity on any of its FIAs.</li> </ul>	Section 6.3
The ACF raised concerns regarding the adequacy of information Santos provided in relation to non- hydrocarbon liquid releases. In particular, the ACF requested more detail in relation to the 'suite of procedures, storage, handling and clean-up', and the control measures proposed to be adapted.	Santos has provided a response to ACF and considered the matters raised by ACF in the OEMP. Santos does not agree with the ACF's claim that the information booklet does not provide sufficient information to enable the ACF to make an informed assessment of any potential consequences of the Activity on its functions, interests or activities. Section 7.4 of the OEMP address unplanned impacts and risks of non-hydrocarbon liquid releases, and proposed control measures to reduce impacts and risks to ALARP and acceptable levels. Santos views that ACF has been provided sufficient information to enable the ACF to make an informed assessment of any potential consequences of the Activity on its functions, interests or activities.	[Response (9)] The potential impacts and risks from an unplanned non- hydrocarbon liquid release event are described on page 27 of the Booklet, with consideration for the extent of potential impacts and sensitive receptors that could be affected. The suite of operational procedures referred to in the Booklet to manage risk of non-hydrocarbon liquid releases relates to both prevention of unplanned releases and mitigation of environmental impact if an unplanned release were to occur. Procedures relevant to prevention of unplanned releases include chemical handling and storage, equipment inspection and maintenance and dropped object prevention. Procedures relevant to impact mitigation include chemical selection, drain system management and spill response. These suites of procedures form the basis of the proposed control measures for non-hydrocarbon liquid releases. The adequacy of the procedures as proposed control measures will be a matter for the Regulator to assess against the requirements of the Regulations. To the extent the ACF believes the risk of a non-hydrocarbon liquid release (as outlined in the Booklet) may affect the ACF's FIAs, Santos welcomes the ACF's input regarding this in accordance with the legislative	Section 7.4 BAO-CM-6.1.2 BAO-CM-6.4.5 BAO-CM-6.7.5 BAO-CM-7.1.1 BAO-CM-7.1.2 BAO-CM-7.1.3 BAO-CM-7.1.4 BAO-CM-7.4.1 BAO-CM-7.4.2 BAO-CM-7.4.3



		purpose of s 25 consultation and has invited a meeting with the ACF, in part to provide an opportunity to receive any such input.	
<ul> <li>The ACF raised concerns regarding impact of a potential hydrocarbon spill on surrounding habitats and species, and requested further information, including:</li> <li>copies of condensate spill modelling;</li> <li>the Production Operations Oil Pollution Emergency Plan'; and</li> <li>the Well Operations Management Plan; and</li> <li>other response plans which detail the actions Santos will take to control and manage cleanup activities if a spill occurs.</li> </ul>	Santos has provided a response to ACF and considered the matters raised by ACF in the OEMP. Santos has appropriately addressed the potential risks and impacts associated with a potential hydrocarbon spill. Sections 7.4,7.6 and 7.7 of the OEMP describe the potential impacts for a range of unplanned hydrocarbon release scenarios of relevance to the Activity for the Coastal Waters OEMP; and proposed control measures to reduce impacts and risks to ALARP and acceptable levels. The concerns and requests related to Production Operations activities in the Barossa Field (including regarding the Production Operations OPEP – noting that there is a separate Barossa GEP NT Waters OPEP and the Well Operations Management Plan) are outside the scope of the GEP Coastal Waters OEMP and are addressed in the Barossa Production Operations EP. Santos views that ACF has been provided sufficient information to enable the ACF to make an informed assessment of any potential consequences of the Activity on its functions, interests or activities.	[Response (10)] a) Santos has performed sufficient stochastic spill modelling of credible unplanned spill events and applied exposure value impact thresholds in line with NOPSEMA guidance, to present the entirety of the risk, which has informed the EMBA and MEVA as presented in the Booklet. It is not necessary for the ACF to have the spill modelling in order for the ACF to make an informed assessment of the potential consequences of the Activity on any of its FIAs. To the extent the ACF believes the risk of an unplanned hydrocarbon release (as outlined in the Booklet and Factsheet) may affect the ACF's FIAs, Santos welcomes the ACF's input regarding this in accordance with the legislative purpose of s 25 consultation and has invited a meeting with the ACF, in part to provide an opportunity to receive any such input	Sections 7.4, 7.6 and 7.7. GEP NT Waters Oil Pollution Emergency Plan
		<ul> <li>b) The Production Operations Oil Pollution Emergency Plan (Production OPEP) will be provided to the Regulator for assessment against the requirements of the Regulations. Consistent with the requirements of the Regulations, Santos plans are not finalised at the time of consultation and won't be finalised until plans are submitted to the Regulator for assessment. It is not necessary for the ACF to have the Production OPEP in order for the ACF to make an informed assessment of the potential consequences of the Activity on any of its FIAs however, this consultation offers an opportunity for the ACF to provide input on the kinds of measures that it would like to see included in the OPEP.</li> <li>c) The Well Operations Management Plan (WOMP) will be provided to the Regulator for assessment against the requirements of the ACF to make an informed assessment against the requirements of the ACF to make an informed in the OPEP.</li> <li>d) Reference to response plans on pp. 31 refers to secondary Production OPEP will be provided to the regulator for assessment of the potential consequences of the ACF plans. Further to the response above, the Production OPEP will be provided to the regulator for assessment of the requirements of the Regulations. Such as the requirements of the Regulation of the Regulator for assessment against the requirements of the ACF to make an informed assessment of the potential consequences of the ACF to make an informed assessment of the potential consequences of the ACF to make an informed assessment of the potential consequences of the Activity on any of its FIAs.</li> </ul>	
		plans are not finalised at the time of consultation and won't be finalised until plans are submitted to the Regulator for assessment. It is not necessary for the ACF to have the Production OPEP in order for the ACF to make an informed assessment of the potential consequences of the Activity on any of its FIAs, however, this consultation offers an opportunity for the ACF to provide input on the kinds of measures that it would like to see included in the OPEP.	
The ACF raised concerns regarding methane emissions and suggested that a "strong" methane reduction and mitigation plan, as well as publicly available transparent reporting can help Santos address scope 1 emissions associated with the project.	Santos has provided a response to ACF and considered the matters raised by ACF in the OEMP. Concerns related predominantly to Production Operations activities in the Barossa Field (including methane emissions) and are addressed in the Barossa Production Operations EP. Section 6.3 of the Barossa Production Operations EP details the facility design and operations measures to reduce Scope 1 emissions to ALARP, inclusive of sources of methane emissions. Section 6.3 of the OEMP has considered the potential for fugitive emissions. The Barossa JV has adopted environmental performance outcomes and control measures directed to minimising the GHG emissions from the Activity. A range of controls have been considered for both direct (Scope 1) and indirect (Scope 3) emissions.	[Response (11)] Management of fugitive GHG emissions (inclusive of methane emissions) during production operations will be addressed by the GHG Management Plan which is identified as a proposed control measure in the Booklet. Santos reports its scope 1 emissions transparently through its annual Sustainability and Climate Report, and under the National Greenhouse and Energy Reporting framework. The adequacy of proposed control measures to address impacts/risks from methane emissions will be a matter for the Regulator to assess against the requirements of the Regulations however, this consultation offers an opportunity for the ACF to provide input on the kinds of measures that it would like to see Santos consider in its management plans.	Section 6.3 Section 8.3.2.12 BAO-CM-6.3.12 BAO-CM-6.4.1
The ACF queried how Santos will approach flaring and venting on the FPSO, and whether Santos can commit to no venting and routine flaring.	Flaring and venting activities on the FPSO are out of scope for the Coastal Waters OEMP. Santos has provided a response to ACF and considered the matters raised by ACF in the Production Operations EP.	[Response (12)] Santos has committed to no routine flaring, with the exception of the continuously-lit high pressure flare pilot to ensure the operation of the safety critical flare is not impaired. Santos has not committed to no planned venting for production operations. Some planned venting is unavoidable for safety reasons when undertaking planned maintenance activities, such as tank/vessel integrity inspections. Santos has however designed the FPSO to include a vapour recovery system that captures low pressure vented gas that would otherwise be vented. This system reduces unplanned venting.	Section 6.3 of the Barossa Production Operations EP.
The ACF requested further information regarding, and queried whether Santos would apply, practice leak detection and repair to the Barossa project.	Santos has provided a response to ACF and considered the matters raised by ACF in the Production Operations EP. Concerns related to predominantly to production operations activities in the Barossa Field are addressed in the Barossa Production Operations EP. Section 6.3 of the Barossa Production Operations EP details the facility design and operations measures to reduce Scope 1 emissions to ALARP, inclusive of sources of methane emissions. Section 6.3 of the OEMP has considered the potential for fugitive emissions. The Barossa JV has adopted environmental performance outcomes and control measures directed to minimising the GHG emissions	<b>[Response (13)]</b> Management of fugitive GHG emissions (with consideration for best practice leak detection and repair) during production operations will be addressed by the GHG Management Plan which is identified as a proposed control measure in the Booklet.	Section 6.3 Section 8.3.2.12 BAO-CM-6.3.12



	from the Activity. A range of controls have been considered for both direct (Scope 1) and indirect (Scope 3) emissions.	The adequacy of proposed control measures to address impacts/risks from fugitive emissions will be a matter for the Regulator to assess against the requirements of the Regulations however, this consultation offers an opportunity for the ACF to provide input on the kinds of measures that it would like to see included in Santos' management plans.	
The ACF requested further information in relation to Santos's commitment to use the Oil & Gas Methane Partnership 2.0 (OGMP 2.0) framework level 5 for reporting on its methane emissions and whether Santos would publicly report its OGMP 2.0 findings	Santos has provided a response to ACF and considered the matters by the ACF in the OEMP. The OEMP includes a proposed control measure that commits to compliance with GHG emissions reporting under the NGER Scheme. Control measures adopted have reduced the impacts of GHG emissions including methane to ALARP. Santos will continue to report its Scope 1 and 3 emissions, and specifically its methane emissions, transparently via its annual Climate Report. Santos views that ACF has been provided sufficient information to enable the ACF to make an informed assessment of any potential consequences of the Activity on its functions, interests or activities.	[Response (14)] Santos is not a signatory to the Oil & Gas Methane Partnership 2.0 (OGMP 2.0) reporting framework. Santos is committed to action on methane and will continue to evaluate potential enhancements to our measurement and mitigation programs. Earlier this year, Santos signed up to the OGCI's near zero methane initiative. In any event, Santos will continue to report its scope 1 and 2 emissions, and specifically its methane emissions, transparently via its annual Sustainability and Climate Report. This is in line with regulatory requirements in the countries in which we operate, such as the National Greenhouse and Energy Reporting Framework in Australia.	Santos reports on GHG emissions as per the NGER scheme BAO-CM-6.3.9 Section 6.2.2.2 Section 6.2.2.3
The ACF requested further information in relation to the decommissioning of the Barossa project, including whether Santos could commit to taking responsibility for decommissioning.	Santos has provided a response to ACF and considered the matters raised by ACF in the OEMP. Although decommissioning and removal of the Barossa production facilities are outside the scope of the Coastal Waters OEMP, the GEP has been designed and selected to meet the regulatory base case for full removal. Notwithstanding, Santos will meet all relevant and applicable regulatory requirements at the time of decommissioning. Santos views that ACF has been provided sufficient information to enable the ACF to make an informed assessment of any potential consequences of the Activity on its functions, interests or activities.	[Response (15)] Barossa decommissioning will be the subject of a future Barossa Decommissioning Environment Plan, that will meet the requirements of the Offshore Petroleum and Greenhouse Gas Storage Act 2006 (Cth) and the Regulations, and any additional relevant legislation, policies (such as NOPSEMA Policy 'Section 572 Maintenance and removal of property') and guidelines (such as Department of Industry, Science and Resources [DISER] Guideline 'Offshore Petroleum Decommissioning Guideline') in force at the time (NOPSEMA, 2020; DISER, 2022).	Section 8.2.5
Meeting with ACF on 9 July 2024 The ACF reiterated concerns regarding the Activity's climate change impacts on matters of national environmental significance and vulnerable communities and ecosystems. The ACF suggested Santos should undertake the analysis required to attribute the Barossa project's climate change impact on specific environments of national significance.	Santos has provided a response to ACF and considered the matters raised by ACF in the OEMP . The Barossa Production Operations EP and this OEMP acknowledge the linear relationship between accumulative anthropogenic CO <sub>2</sub> emissions and global warming. The impacts on the climate cannot be attributed to one specific sector or activity. The Activity's projected emissions are contextualised against established emissions budgets (national and global). The EP references the latest commentary from the IPCC on the causes and impacts of climate change. The projected emissions for the Activity described in the Coastal Waters OEMP are contextualised against established emissions budgets (national and global). The Barossa Project's overall GHG emissions and proposed management are addressed in the Barossa Production Operations EP. The Barossa JV has adopted environmental performance outcomes and control measures directed to minimising the GHG emissions from the Activity. A range of controls have been considered for both direct (Scope 1) and indirect (Scope 3) emissions.	Santos explained that there are limitations to linking emissions from the Activity to any specific climate change impacts. Santos invited ACF to provide further information regarding how it is able to undertake an analysis that links climate change impacts from the Activity to specific environments or ecosystems.	Section 6.3
The ACF reiterated concerns that the Barossa project is not aligned with the temperature targets of the Paris Agreement. The ACF also queried whether Santos will be ensuring its customers are abating or offsetting their emissions to align with Paris Agreement targets.	Santos has provided a response to ACF and considered the matters raised by ACF in the Barossa Production Operations EP. Scope 3 emissions associated with Barossa Gas Project customers and the associated use of sold products are addressed in the Production Operations EP.	Santos explained that it will comply with the Safeguard Mechanism, which is the regime that implements Australia's commitment to the Paris Agreement. Santos response to ACF for the Production Operations EP also explained that it will only sell gas to customers of Paris Agreement signatory countries, and Santos is unable to influence how those countries comply with the Paris Agreement temperature targets.	Section 6.3 Section 8.3.2.12
The ACF requested any management plans that Santos has prepared in relation to marine fauna and biologically important areas / ecosystems.	Santos has provided a response to ACF and considered the matters raised by ACF in the OEMP. The Barossa Production Operations EP and this OEMP contain information on the procedures required for the Activity, to protect marine fauna, that aligns with DCCEEW guidance and EPBC regulations. Santos views that ACF has been provided sufficient information to enable the ACF to make an informed assessment of any potential consequences of the Activity on its functions, interests or activities.	Santos explained that the Environment Plan will provide specific information regarding each plan and/or procedure required for the Activity. Santos assured ACF that the management procedures and plans for marine fauna and ecosystems align with DCCEEW guidance and EPBC regulations. Santos invited ACF to provide more information in relation to the specific detail they require on this topic that is not already provided in the Information Booklet and regulatory guidance.	Section 7.3 BAO-CM-6.1.1 BAO-CM-6.3.12 BAO-CM-6.7.5 BAO-CM-7.2.1
The ACF requested further information regarding the type of offsetting methodology Santos intend to use to abate emissions from the Activity.	Santos has provided a response to ACF and considered the matters raised by ACF in the OEMP. Santos will meet its Safeguard Mechanism compliance obligations, set by the CER in accordance with Australian Government policy and emissions reduction targets, through the purchase and/or generation of ACCUs or SMCs. This is described in Section 6.3.2.3.2 National greenhouse gas emissions framework Santos views that ACF has been provided sufficient information to enable the ACF to make an informed assessment of any potential consequences of the Activity on its functions, interests or activities.	Santos explained that it will comply with the rules of the Safeguard Mechanism, which is administered by the Clean Energy Regulator. Santos also stated that it does not decide what the criteria for ACCUs or SMCs are but that it is rather Santos' responsibility to comply with the rules set by the Clean Energy Regulator.	Section 6.3.2.3.2
Noting the Barossa gas's CO <sub>2</sub> composition, the ACF requested information regarding how the gas will be vented in first instance on the FPSO and/or at the DLNG facility.	Santos has provided a response to ACF and considered the matters raised by ACF in the OEMP. Concerns related to Production Operations activities in the Barossa Field (including operation of the FPSO and of the DLNG facility) are outside the scope of the GEP Coastal Waters OEMP and are addressed in the Barossa Production Operations EP.	Santos confirmed that there is approximately 18% CO <sub>2</sub> in the Barossa gas. The DLNG facility is designed to process and vent 6% CO <sub>2</sub> . The balance (approximately 12%) is processed and vented offshore.	Not applicable.
The ACF queried the expected lifespan of the DLNG facility.	Santos has considered the matters raised and provided a response to ACF.	Santos confirmed that the DLNG facility is undergoing minor modifications and maintenance to extend the design life of the facility to process Barossa gas for another 20-25 years.	Not applicable.



	Operation at the DLNG facility is outside the scope of the Coastal Waters OEMP and is controlled by a different joint venture to this Activity, Notwithstanding that, Santos provided information on the design life of DLNG. to ACF in response the Barossa Production Operations EP.		
The ACF queried whether produced water from seabed would contain mercury.	Concerns related to Production Operations activities in the Barossa Field (including produced water from the FPSO) are outside the scope of the Coastal Waters OEMP and are addressed in the Barossa Production Operations EP.	Santos confirmed that produced water from the seabed would likely contain very low levels of mercury, which is not atypical for these types of projects.	Not applicable.
The ACF queried where else Santos is exploring CCS opportunities, other than at Bayu Undan.	Santos has provided a response to ACF. Other CCS opportunities are outside the scope of the Coastal Waters OEMP, notwithstanding this, Santos provided information to ACF in response to this request.	Santos confirmed it has exploration permits located in the Bonaparte Basin for seismic and drilling exploration in the next 12 to 24 months	Not applicable.
The ACF queried how Santos is making a decision about acceptable climate change impacts on MNES without having conducted an analysis regarding direct climate impacts attributable from the project and whether it is able to articulate what an acceptable level of impact is.	Santos has considered the matters raised by ACF in the Coastal Waters OEMP and provided a response to ACF. The Barossa Production Operations EP and this OEMP acknowledge the linear relationship between GHG cumulative anthropogenic CO <sup>2</sup> emissions and global warming and reference the latest commentary from the IPCC on the causes and impacts of climate change. Even assuming that emissions from the Barossa Gas Project (including this Activity) will cause an equivalent net increase in cumulative Australian and global emissions, this increase comprises a nominal amount in the overall scheme of the national and international carbon budgets and will not materially or substantially contribute to Australia's GHG emissions or global emissions levels; and there is no correlation between where GHG emissions are released and where climate change impacts are felt. It is not possible to draw a link between GHG emissions from the Activity and any specific climate related impact on any specific element of the Australian environment which may result from any net increase to cumulative GHG emissions globally. By keeping the Project emissions under the Safeguard Mechanism baseline, the Project's incremental contribution is within accontable limits as determined by reference to Australia's NDC under the	Santos explained that the Paris Agreement is key to the solution of managing climate impacts because the Paris Agreement is designed to limit global temperature increase and by doing so limit climate impacts to the environment to acceptable levels. Santos also stated that the acceptable level will depend specifically on the ecosystem and global cooperation to reduce emissions to limit temperature increase. Santos reiterated that the goal is for no significant impacts on MNES from climate impacts.	Section 6.3
ACF correspondence to Santos on 11 July 2024 The ACF reiterated its concerns regarding how Santos are considering climate change impacts attributable to the Activity. The ACF asserted that Santos must state in the Environment Plan what the environmental impacts from climate change due to the Activity will be. To do this, ACF asserted that Santos must undertake an analysis that measures the temperature increase attributable to the total lifetime emissions of the project and then analyse what this temperature increase would mean for the environment (including MNES).	<ul> <li>contribution to global warming is within acceptable limits as determined by reference to Australia's NDC under the UNFCCC.</li> <li>Santos has provided a response to ACF and considered the matters raised by ACF in the OEMP.</li> <li>The Barossa Production Operations EP and this OEMP acknowledge the linear relationship between GHG cumulative anthropogenic CO2 emissions and global warming and reference the latest commentary from the IPCC on the causes and impacts of climate change.</li> <li>The Activity's projected emissions are contextualised against established emissions budgets (national and global).</li> <li>Climate change is a global problem with the solution being led at the international level. Domestically, GHG emissions are regulated through Australia's NDC and the SGM and the Barossa JV has legal obligations to ensure that the Activity operates in accordance with that framework.</li> <li>It is not possible to draw a link between GHG emissions from the Activity and any specific climate related impact on any specific element of the Australian environment which may result from any net increase to cumulative GHG emissions globally.</li> <li>By keeping the Project emissions under the Safeguard Mechanism baseline, the Project's incremental contribution to global warming is within acceptable limits as determined by reference to Australia's NDC under the UNFCCC.</li> </ul>	Santos correspondence to ACF on 7 August 2024 Santos refers ACF to paragraph 4 of the Annexure to Santos' 13 June 2024 correspondence. The acceptability of environmental impacts of GHG emissions from the Activity will be evaluated in the EP, for assessment by the Regulator against the requirements of the Regulations. Santos' methodology for this evaluation of acceptability will be broadly consistent with the methodology adopted for previous Barossa EPs, as an example refer section 5.1 of the Barossa Subsea Infrastructure Installation Environment Plan.	GHG impact assessment is outlined in Section 6.3.4 GHG ALARP assessment Section 6.3.5 GHG Acceptability evaluation Section 6.3.6
The ACF asserted that Santos must demonstrate in the Environment Plan how the Barossa project is compatible with the 1.5-degree scenario under the Paris agreement. The ACF also requested Santos provide it with more information to demonstrate how it will comply with the 1.5-degree scenario.	Santos has provided a response to ACF and considered the matters raised by ACF in the OEMP. Given that natural gas has a continued role under a range of Paris-aligned scenarios, GHG emissions associated with the Activity are able to be accommodated and managed under Australia's carbon budget and, internationally, under the carbon budget of Paris Agreement signatories and countries with mid-century net-zero commitments. The Barossa Gas Project will meet its Safeguard Mechanism compliance obligations, set by the CER in accordance with Australian Government policy and emissions reduction targets, including through the purchase and/or generation of ACCUs or SMCs. Management of the emissions of the Barossa Development in accordance with the Safeguard Mechanism will ensure that this Activity does not have an unacceptable impact on climate change, as the Scope 1 GHG emissions of the Activity are being considered as part of Australia's NDC and, therefore, also into the global trajectory to limit global warming in line with the Paris Agreement targets. Santos views that ACF has been provided sufficient information to enable the ACF to make an informed assessment of any potential consequences of the Activity on its functions, interests or activities.	Santos reters ACF to paragraphs 4 and 5 of the Annexure to Santos' 13 June 2024 correspondence. Santos will be required to comply with the applicable baseline under the Safeguard Mechanism for the Barossa project in accordance with Australia's Paris Agreement targets and associated emissions budget. Santos understands that ACF has a specialist understanding of the Safeguard Mechanism. Compliance with Santos' obligations under the Safeguard Mechanism may be achieved through (among other things) purchase or surrender of ACCUs or SMCs. Santos refers ACF to page 22 of Santos' 2023 Annual Report, which provides further information on Santos' generation and acquisition of carbon credits as follows: In 2023, Santos executed agreements to build a portfolio of projects supporting the development of five nature-based projects across Queensland, Alaska and Papua New Guinea, to generate carbon credits. Further, in 2023 Santos entered into forward contracts for the purchase of 2.5 million ACCUs at fixed prices to be delivered and paid between December 2023 and January 2027. NOPSEMA, in the exercise of its functions as Regulator, will consider whether Santos has demonstrated how the requirements applicable to the Activity will be met.	Section 6.3 BAO-CM-6.3.9 BAO-CM-6.3.11
and protocols that Santos has developed for the Barossa project to protect marine fauna and biologically important areas.	Santos nas provided a response to ACF and considered the matters raised by ACF in the OEMP. Santos provided the ACF with additional information of the specific protocols adopted in Santos procedures adopted in accordance with guidance from Part 8 of the Environment Protection and Biodiversity Conservation Regulations 2000 and the Australian National Guidelines for Whale and Dolphin Watching 2017. Santos views that ACF has been provided sufficient information to enable the ACF to make an informed assessment of any potential consequences of the Activity on its functions, interests or activities.	I ne Production Operations information booklet contains proposed control measures that will be adopted as relevant to potential impacts to species, including marine fauna, and in relation to biologically important areas. All considered control measures (adopted and not adopted) will be presented in the EP for assessment by the Regulator/s.	Section 7.3



		Santos' planned vessel activities overlap th turtles. Specific to marine fauna interactions, the P booklet describes control measures at page the following specific controls to protect ma for Barossa Operations in accordance with: (a) Part 8 of the Environment Protecti Regulations 2000: (i) avoid collision; (ii) reduce speed to 6 knots & ste (300m of whales / whale sharks ai (iii) operate vessel at constant sp (iv) do not drift or approach marin (v) do not restrict fauna pathway (b) Australian National Guidelines for 2017: (i) Adopting cautionary and no a whale sharks (100 m no approach and dolphins (50 m no approach a
September 2024 The ACF reiterated concerns that Santos has not provided it with sufficient information in relation to the climate change impacts attributable to the project, including an analysis that measures the impact on the environment as a result of an increase in temperatures attributable to the total lifetime emissions of the project.	The Barossa Production Operations EP and this OEMP acknowledge the linear relationship between cumulative anthropologic GHG emissions and global warming. The impacts on the climate cannot be attributed to one specific sector or activity. In the context of evaluating potential impacts and risks that may be associated with GHG emissions from all sources globally, including from this Activity, this OEMP has considered broader climate change issues. The OEMP references the latest commentary from the IPCC on the causes and impacts of climate change. By keeping the Project emissions under the Safeguard Mechanism baseline, the Project's incremental contribution to global warming is within acceptable limits as determined by reference to Australia's NDC under the UNFCCC. The Barossa Project's overall GHG emissions and proposed management are addressed in the Barossa Production Operations EP. Santos considered this claim having regard to the previously exchanged information and correspondence as described above in this table, noting that the ACF's 30 September 2024 correspondence relterated the ACF's claim that it has not been provided with sufficient information, but did not elaborate on it any further. Santos views that ACF has been provided sufficient information and a reasonable period of time to enable the ACF to make an informed assessment of any potential consequences of the Activity on its functions, interests or activities.	Thank you for your letter of 9 September 20 matters raised or restated by ACF in that le your concerns and requests for further infor reasonable, in the attached Annexure. Santos remains of the view that sufficient in reasonable amount of time, has been provia an informed assessment of the possible con- carried out under the EP on any of ACF's fu Santos is finalising the EP for submission to its comments and submissions in respect of controls in relation to this EP. (Note: <u>This en- was referring to the Production Operations</u> Santos will comply with its Safeguard Mech- the Australian Government's approach to rr Determined Contribution under the Paris Ag from Barossa (and a portion of indirect emi- at DLNG) are to be managed under Austral As a result of the complex nature of the glo change impacts cannot be meaningfully lini emissions source. The whole system for ac involves additions in some areas and reduc reducing over time and there are many pati country. Products generated from the Baro. to customers from countries that are signat that have policies for reducing GHG emissi required by the Paris Agreement). Santos confirms that in drafting the EP, it h and relevant CSIRO publications. Otherwise, Santos: • refers ACF to paragraph 4 of the Anne correspondence and Santos' response in th Santos' 7 August 2024 correspondence; ar • is of the view that sufficient informatior allow ACF to make an informed assessment the activity to be carried out under the EP of interests or activities. The acceptability of environmental impacts activity will be evaluated in the EP, for asset the requirements of the Offshore Petroleun
The ACF asserted that it was concerned about marine impacts and has engaged a marine expert to provide specific questions and solutions, which it expected to provide Santos in mid-October.	Santos has provided a response to ACF and considered the matters raised by ACF in the OEMP. Santos has provided information to the ACF regarding the marine impact of the Activity, and these are assessed in Sections 6 & 7 of the OEMP. Santos views that ACF has been provided sufficient information and a reasonable period of time to enable the ACF to make an informed assessment of any potential consequences of the Activity on its functions, interests or activities	(Environment) Regulations 2023 (OPGGS Santos notes the ACF's concern. Santos' c commenced on 9 February 2024. During th has provided information in relation to impa the extensive scientific literature publicly av the ACF in respect of concerns regarding p provided ACF more than a reasonable and
		L

Section 6.3
BAO-CM-6.3.9 BAO-CM-6.3.11
Sections 6 & 7

<ul> <li>The ACF reiterated concerns regarding the activity's impact on marine species and requested further information regarding:</li> <li>a. the precautions Santos were taking to protect precious and vulnerable species;</li> <li>b. crossover or near crossover with Santos' operations and Pygmy Blue Whale feeding/foraging or possible breeding areas; and</li> <li>c. whether Santos is open to evaluating and consulting on additional control measures.</li> </ul>	Santos has provided a response to ACF and considered the matters raised by ACF in the OEMP. Santos has considered potential impacts to threatened, migratory and local fauna from planned activities and unplanned events in preparing this OEMP. Santos views that ACF has been provided sufficient information and a reasonable period of time to enable the ACF to make an informed assessment of any potential consequences of the Activity on its functions, interests or activities. Concerns related to the FPSO and activities in Operational Areas 1 and 2 of the Barossa Production Operations EP, including the Pygmy Blue Whale feeding / foraging areas or possible breeding areas, are outside the scope of the Coastal Waters OEMP and are addressed in the Barossa Production Operations EP.	<ul> <li>assessment of the possible consequences of the activity on its functions, interests or activities. The matters raised by ACF during consultation demonstrate that it has been able to engage comprehensively in the consultation process based on the information provided by Santos.</li> <li>Note: Santos undertook consultation under s 25 of the OPGGS(E)R regarding activities under this Coastal Waters OEMP concurrently with consultation on activities under the Barossa Production Operations EP.</li> <li>a. In addition to the control measures set out in its letter of 7 August 2024, Santos proposes to adopt a control measure that will limit vessel speeds within 500m around the FPSO, IMMR vessels and campaign vessels to further mitigate the risk of marine fauna interaction.</li> <li>b. Operational Areas 1 and 2 do not overlap with pygmy blue whale biologically important areas (BIAs) for migration or foraging. The closest pygmy blue whale BIA (migration) is 173km from Operational Area 1. (Note: The reference to Operational Area 1 and Operational Area 2 refers to activities under the Barossa Production Operational Area 1. (Note: The reference to open during the consultation phase to considering additional measures (and has considered measures where proposed).</li> <li>Based on all information available to Santos about the level of impact and risk to marine fauna from the proposed Activity, Santos does not consider any additional measures (in addition to these already proposed) to be necessary. An ALARP evaluation of all potential control measures (adopted and not adopted) to reduce impacts to marine fauna will be presented in the EP for assessment by the Regulator/s. In relation to the activity proposed under this EP, which does not include build operations.</li> <li>Santos otherwise refers ACF to its response in the third row of the Annexure to Santos' 7 August 2024 correspondence. In accordance with Section 22 (15) of the OPGGS Regulations, the EP will contain an implementation strategy to enable appropriate cons</li></ul>	Sections 6 & 7 Section 7.3 BAO-CM-6.1.1 BAO-CM-6.1.2 BAO-CM-6.6.6
The ACF asserted that compliance with the Safeguard Mechanism does not address Barossa's exported emissions. It also stated that Santos does not have accountability measures in place to ensure end users meet Paris Agreement obligations. The ACF stated that it questioned the legitimacy of any claim that the Barossa Project is compliant with a 1.5 degree aligned pathway under the Paris Agreement.	Santos has considered the matters raised and provided a response to ACF. The Barossa Project's overall GHG emissions and proposed management are addressed in the Barossa Production Operations EP.	<ul> <li>Santos refers ACF to paragraphs 6 and 7 of the Annexure to Santos' 13 June 2024 correspondence and Santos' response in the second row of the Annexure to Santos' 7 August 2024 correspondence.</li> <li>Santos reiterates that: <ul> <li>Santos will be required to comply with the Safeguard Mechanism, which is the regime that implements Australia's commitment to the Paris Agreement in relation to industrial emissions; and</li> <li>Santos will only sell gas to customers from countries that are signatories to the Paris Agreement or that have a net-zero commitment as at the date of the relevant contract of sale, and therefore, Santos' international customers are subject to the requirements their respective governments set to achieve their Paris Agreement commitments.</li> </ul> </li> <li>NOPSEMA, in the exercise of its functions as Regulator, will consider whether Santos has demonstrated how the requirements applicable to the activity will be met.</li> </ul>	EPO 11
<ul> <li>The ACF stated that its functions, interests or activities include:</li> <li>engaging with investors across Australia who have an interest in preserving the environment and safeguarding climate; and</li> <li>monitoring and engaging with government and businesses on climate change and emissions reduction measures.</li> </ul>	Santos has considered the matters raised and provided a response to ACF. Santos acknowledges ACF's functions, interests and activities. Santos views that ACF has been provided sufficient information to enable the ACF to make an informed assessment of any potential consequences of the Activity on its functions, interests or activities. The Barossa Project's overall GHG emissions and proposed management are addressed in the Barossa Production Operations EP.	Santos acknowledges ACF's functions, interests and activities. Santos is of the view that sufficient information has been provided to ACF to allow ACF to make an informed assessment of the possible consequences of the activity to be carried out under the EP on any of ACF's functions, interests or activities. See, in particular, Santos' response in the second row of the Annexure to Santos' 7 August 2024 correspondence. Santos' compliance with the SGM is a matter for Australia's Clean Energy Regulator to determine. Santos confirms that it will comply with the SGM using offsets or direct abatement. Santos continues to pursue carbon capture and storage opportunities which would provide direct abatement.	Not applicable.
<ul> <li>The ACF sought further information on Santos' compliance with the Safeguard Mechanism, including:</li> <li>the Baseline scenarios it will comply with across the life of the project; and</li> <li>specific details of how it intends to comply with the Safeguard Mechanism across the</li> </ul>	Santos has provided a response to ACF and considered the matters raised by ACF in the OEMP. Barossa will meet its Safeguard Mechanism compliance obligations, set by the CER in accordance with Australian Government policy and emissions reduction targets, including through the purchase and/or generation of ACCUs or SMCs. Santos considers that the information requested is not necessary for the ACF to make an informed assessment of the possible consequences of the activity on any of its functions, interests or activities. Santos views that ACF has been provided sufficient information to enable the ACF to make an informed assessment of any potential consequences of the Activity on its functions, interests or activities.	Santos refers ACF to paragraph 1 of the Annexure to Santos' 13 June 2024 correspondence and Santos' response in the second row of the Annexure to Santos' 7 August 2024 correspondence. Santos reiterates that the treatment of the Barossa project under the Safeguard Mechanism is a matter for the Clean Energy Regulator. Notwithstanding this, Santos notes that it will be required to comply with the applicable baseline for the Barossa project in each compliance year, and	Section 6.3 BAO-CM-6.3.9 BAO-CM-6.3.11 BAO-CM-6.3.12

life of the project, including abatement options, purchase of carbon credits, emissions reduction strategies, financial assessments for future credit purchases, and the projected percentage of abatement it expects to achieve through ACCUS/SMCs.		there are various options available to meet a abatement and acquiring offsets, in addition the Safeguard Rules such as borrowing adju monitoring periods). It is a matter for Santos to determine how it Safeguard Mechanism. This information is n make an informed assessment of the possib on any of its functions, interests or activities.
ACF correspondence to Santos on 30 September 2024 The ACF acknowledged Santos' response on 20 September 2024 and expressed its disagreement with Santos' position that it has provided sufficient information, but did not further elaborate on the complaint. The ACF foreshadowed follow up correspondence before 14 October 2024.	As of 19 December 2024 ACF had not provided follow up correspondence. Santos does not agree with the ACF's claim that it has not been provided sufficient information to enable the ACF to make an informed assessment of any potential consequences of the Activity on its functions, interests or activities. Santos considers it has provided sufficient information and a reasonable period of time for consultation. Santos considers Section 25 consultation requirements to have been met.	<ul> <li>Santos correspondence to ACF on 19 Dec Your 30 September correspondence was in September 2024 where we noted that:</li> <li>Santos remained of the view that sufficience reasonable amount of time, had been purmake an informed assessment of the poly activity to be carried out under the EP of interests or activities; and</li> <li>that Santos was finalising the EP for sufficiency and having regard to the elapsing of time, Sa basis that the ACF does not intend to provid consultation for preparation of environment p Production Operations activities is complete.</li> <li>As always, while Reg 11A consultation is con engagement with our stakeholders and any purmay have. We will also continue to engage in implementation consultation plan once the E Santos would like to thank you for your respondence.</li> </ul>
ACF correspondence to Santos on 20 December 2024 ACF stated in the email that it will provide it to Santos early next week.	A total 12 weeks had elapsed since the ACF stated on 30 September 2024 that it would be providing follow-up correspondence to Santos by 14 October 2024. Santos remains of the view that sufficient information, and more than a reasonable amount of time, had been provided to ACF to allow ACF to make an informed assessment of the possible consequences of the activity to be carried out under the OEMP on any of ACF's functions, interests or activities. Santos advised the ACF in the same communication on 19 December 2024 that the Barossa Production Operations EP was being finalised for submission to NOPSEMA. Santos considers it has provided sufficient information and a reasonable period of time for consultation.	No response required.
ACF correspondence to Santos on 23 December 2024. ACF stated that its letter raised concerns about the content of the EP version dated 24 September 2024 and set outs issues Santos should address in an amended EP.	This letter and the issues raised in it relate to the Barossa Production Operations EP, and in particular to the version of the EP published on NOPSEMA's website in September 2024. Santos has included this letter, and its response to this letter, for completeness noting that many (although not all) of the concerns raised by the ACF may be assumed to apply equally to the OEMP. Santos has considered the matters raised by ACF in the Barossa Production Operations EP and provided a response to ACF. Refer below for assessment of matters raised. For completeness, Santos confirms that the ACF's comments were made in relation to Rev 1 of the Barossa Production Operations EP, and Santos' response was based on Rev 2 of the Barossa Production Operations EP. For completeness noting that many (although not all) of the concerns raised by the ACF may be assumed to apply equally to the OEMP, concerns that apply to the OEMP have been considered in the OEMP.	Santos correspondence to ACF on 31 Jar The ACF's 23 December letter sets out issue be addressed by Santos in an amended EP. ACF's comments in that letter refer to the EF 2024 (that version, the 24/9/2024 Version). An updated version of the EP was provided 2024 (that version, the Current Version) The Santos' responses to your letter of 23 Decem
[4] – [5] ACF stated the 23/9/2024 version of the EP is deficient in material respects, resulting in non-compliance with the Offshore Petroleum and Greenhouse Gas Storage (Environment) Regulations 2023	Notwithstanding that this issue was raised in relation to the Production Operations EP, Santos has considered the matters raised by ACF in the OEMP. Santos has provided a response to ACF. Santos does not agree that the EP and OEMP and is deficient or otherwise non-compliant with the Regulations.	Santos notes that whether or not the EP is a compliant with the Regulations, is a matter N determine in the exercise of its functions
<ul> <li>[7] ACF stated that its concerns on impacts and protection of marine fauna, raised during the meeting on 9 July 2024, had not been accurately characterised in the EP.</li> <li>[8] ACF also stated it had not been provided with sufficient information to inform its assessment, specifically alleging:</li> </ul>	Notwithstanding that this issue was raised in relation to the Barossa Production Operations EP, Santos has considered the matters raised by ACF in the OEMP and provided a response to ACF. Santos disagrees with the assertion that ACF's concerns regarding impacts and protection of marine fauna, raised in the 9 July meeting, have not been accurately characterised in the EP and OEMP. These concerns are captured in Con 5212 and 5117 and have been consistently reflected and considered in the Barossa Production Operations EP. Regarding point [8.1], Santos acknowledges that the GHG emissions estimates in the Production Operations EP are lower than the estimates provided in the information booklet for the reasons explained in Santos' response to	Santos has considered the ACF's views as to provided to date and remains of the view that with the ACF to the standard required under Regarding point 1, Santos confirms that the EP are lower than the estimates provided in year lifecycle of Barossa production operation start-up and steady state operations), and a information about the Activity at time of subm



et a baseline {including direct on to mechanisms available under djustments and multi-year	GHGEMP is summarised in Section 8.3.2.12
it will achieve compliance with the s not necessary for the ACF to sible consequences of the activity es.	
December 2024	Not applicable
in response to our letter of 19	
ficient information, and more than a provided to ACF to allow ACF to possible consequences of the on any of ACF's functions,	
submission to NOPSEMA.	
s received by 14 October, or since, Santos has proceeded on the vide any further feedback and that nt plans relating to Barossa ste.	
complete, Santos remains open to ny input on our activities that you le in accordance with our post- le Environment Plan is in force.	
sponses and any input provided to	
	Not applicable.
January 2025. sues that the ACF proposes should EP. In this regard, we note that the EP as it stood on 24 September ). ed to NOPSEMA on 23 December The distinction is made to explain cember at Appendix A	Not Applicable
s deficient or otherwise non- er NOPSEMA will consider and	Not Applicable
is to the sufficiency of information that Santos has properly consulted der section 25 of the Regulations. the GHG emissions estimates in the in the information booklet for the 25 ations (inclusive of commissioning, d are based on best available ubmission of the EP. As the annual	Not Applicable

<ol> <li>Emissions estimates provided changed over time without specification or explanation,</li> <li>Santos refused to model the impact of emissions on MNES;</li> <li>Santos refused to engage regarding compliance with the Safeguard Mechanism, including regarding offsets; and</li> <li>Data provided to inform impacts (particularly on fauna) was insufficient and of low-quality.</li> <li>[9] – [21] ACF stated the following in relation to infrastructure design being responsive to changing</li> </ol>	ACF. In circumstances where the GHG emissions estimates in the EP are lower than the estimates provided in the information booklet, Santos does not consider this has affected the ACF's ability to make an informed assessment of the possible consequences of GHG emissions on the ACF's functions, interests or activities. Santos' assessment relating to ACF's allegations on each of the items listed in [8.2] – [8.4] of the 23 December letter are assessed in relevant sections below. Santos considers it has provided sufficient information to the ACF, and that section 25 consultation requirements have been met, including for the reasons given in response to the ACF.	emissions estimates provided in the information booklet preceded the finalisation of the EP for submission to NOPSEMA, they were conservative (if extrapolated at a flat rate for 25 years of production operations) when considering that annual emissions are expected to reduce over the life of the Activity as production rates decline. For further discussion relating to emissions estimates and to each of the other items listed in [8.2] – [8.4] of the 23 December letter inclusive, please see further discussion in the relevant sections below. As identified in Section 2.3 of the 24/9/2024 Version of the EP, the total duration of FPSO steady state operations is approximately 25 years.	BAO-CM-6.4.10 BAO-CM-7.6.3
<ul> <li>weather:</li> <li>it is unclear from the 23/9/2024 version of the EP how long operations will be undertaken for;</li> <li>The infrastructure design needs to account for environmental changes (including emissions, temperature and climate change scenarios) that will occur at least until 2050.</li> <li>In relation to the FPSO, the EP does not deal with the consequences of a potential mooring line failure;</li> <li>The risk of a condensate spill is not appropriately assessed in light of increasingly extreme weather events</li> <li>The 23/9/2024 Version of the EP does not satisfy the Regulations because it fails to include: <ul> <li>comprehensive consideration of changing weather patterns;</li> <li>design responses and contingency plans in response to such changing weather patterns; and</li> <li>detail regarding use of support vessels in the event of extreme weather under the above scenarios.</li> </ul> </li> </ul>	Santos disagrees that there is any lack of clarity as to how long operations will be undertaken. The EP and the OEMP identify the duration of activities and address the substance of the concerns raised by the ACF. In any case, to avoid any uncertainty, Santos has made consequential amendments to Rev 3 of the Barossa Production Operations EP in order to clarify the matters raised by the ACF. Santos considers this OEMP satisfies regulatory requirements, the OEMP has considered and includes control measures to ensure integrity management of the subsea infrastructure within the Pipeline Integrity Management Plan (BA/C-0K-7.6.3). In addition, the Barossa Nearshore Gas Export Pipeline Operations Pipeline Management Plan includes measures to ensure safe operations and ongoing integrity of all facilities and vessels (including support vessels) during adverse weather conditions.	The design of the Barossa facilities has considered climate change risks and associated changes in weather patterns, and these have been incorporated into the project design across the 25-year facility design life. (Note: the facility design life of 25 years aligns with the currently planned 25 years of operations. Operations beyond this period would be subject to detailed design life extension assessments.) In any case, Santos notes the EP is required to be revised every five years and there will therefore be at least four opportunities for Relevant Person consultation and regulator assessment and acceptance of the revised EPs before the design life of 25 years is reached Cyclonic events are the dominant climatic influence on the design of surface infrastructure. During project definition, a cyclonic event dataset was developed and subjected to a range of modelled sensitivities to account for extreme weather conditions (winds, waves, currents) which resulted in modelled 10,000 year cyclonic event, and a 100 yr return period cyclone event with a single mooring line failure as described in Sections 2.4.4.1 and 2.10 of the 24/9/2024 Version. The conservatism in the FPSO mooring system design survivability acceptance criteria is sufficient to account for changing weather patterns due to climate change, over the design and operational life of the Barossa facilities. In the event there is the potential for offshore operations to be impacted by extreme weather events, the Barossa FPSO Facility Safety Case (Safety Case) includes measures to ensure safe operations and ongoing integrity of all facilities and vessels (including support vessels) during adverse weather conditions. The Safety Case is subject to assessment by NOPSEMA under the Offshore Petroleum and Greenhouse Gas Storage (Safety) Regulations 2009 (Cth).	
[22] – [24] ACF stated the scope of the activity is unclear in relation to condensate offtake and cited contradiction in the offtake process once the offtake tanker is disconnected from the FPSO but still in the Operational Area	Santos has considered the matters raised by ACF in relation to the FPSO and condensate offtakes in the Barossa Production Operations EP and provided a response to ACF. These matters are not relevant to this Activity and not addressed in this OEMP.	<ol> <li>Offtakes will be periodic, approximately one every 3 months (described in Section 2.7.4.1).</li> <li>The EP defines the Operational Areas with coordinates in Section 2.2.3 and Figure 2-1.</li> <li>Section 2.7.4.1 describes the offtake activities. For clarity, the operations of the offtake tanker constitute part of Santos' petroleum activities (for the purposes of the Regulations) under this EP when the tanker is under Santos' navigational control in Operational Area 1 (OA1), and when the offtake tanker is connected to the FPSO.</li> <li>Santos has assessed the planned impacts and unplanned event risks associated with offtake tanker operations within the OA1. (Refer Section 6.1, 6.2, 6.3 and 6.6 of the 24/9/2024 Version in relation to impacts from planned activities and Section 7.7 in relation to impacts and risks from unplanned hydrocarbon release events.)</li> </ol>	Not Applicable
<ul> <li>[25] – [30] ACF stated the following in relation to spill risk:</li> <li>The statement in the 23/9/2024 Version of the EP that the maximum credible release of</li> </ul>	Santos has considered the matters raised by ACF in relation to the FPSO and condensate releases in the Barossa Production Operations EP and provided a response to ACF. These matters are not relevant to this Activity and not addressed in this OEMP.	I his EP does not cover Varanus Island Operations which are regulated by another regulator and different legislation and regulations. The 24/9/2024 Version sets out the basis on which the maximum credible release of condensate from offtake equipment has been calculated. The incident referred to by the ACE has no bearing on this calculation because the offtake	Not Applicable

<ul> <li>condensate due to rupture or leak of offtake equipment is 465m3 is not credible and not compliant with the requirement to identify and evaluate risk.</li> <li>The lack of credibility arises from the details of the 20 March 2022 spill Santos was responsible for at Varanus Island</li> <li>The 23/9/2024 Version of the EP does not explain what "major loss" means nor how the detection would occur or how detection would result in "instantaneous" shutdown.</li> <li>Given the Varanus spill the EP needs to particularise this and address the Varanus situation specifically.</li> <li>The 23/9/2024 Version of the EP indicates production flowlines can be monitored for pressure drops indicating leaks. The EP should consider applying this to offtake lines.</li> </ul>		<ul> <li>operations for the Barossa FPSO are materially different than that at the Varanus Island Operations. The offtake hose on the BW Opal is reeled on the FPSO and deployed for offtakes. In the case of an unplanned event, and the release of condensate from the hose, the release would be on the sea surface, as the hose is floating.</li> <li>In contrast, as identified in the extract cited by the ACF, the release of condensate at Varanus Island occurred from a torn subsea hose (that is, underwater and on the seabed). The extracts at p 730 of the 24/9/2024 Version identified by the ACF are based on the fact that the Barossa offtake hose is floating and clearly visible.</li> <li>In particular, as set out in the EP, offtake hook-up &amp; condensate offloading will commence during daylight hours (control measure BAO-CM-6.8.6), enabling visual observation at the commencement of offtake operations and visual detection of an unplanned release. The volume and flow rates are also metered at both the FPSO and offtake tanker during offtake operations, providing further means of detection of an unplanned release.</li> </ul>	
<ul> <li>[31] – [34] ACF stated Santos needs to explain incongruity in the 23/9/2024 Version of the EP with regard to a Petroleum Safety Zone being used to reduce the potential for vessel interaction and collision.</li> <li>ACF stated the incongruity arises because the EP states that third-party vessels are not permitted to enter the PSZ but, if condensate is to be taken off by third party vessels, they will need to enter the PSZ.</li> </ul>	Santos has considered the matters raised by ACF in relation to the Petroleum Safety Zone in the Barossa Production Operations EP and provided a response to ACF. These matters are not relevant to this Activity and not addressed in this OEMP.	Further to Santos' response above to paragraphs [22]-[24] of the ACF's 23 December letter, whilst the offtake tanker is a third-party vessel, it is permitted access to the PSZ (within OA1), after requirements of the Santos permit to work authorisation process have been met (which are managed via radio communications between the third-party operated offtake tanker and the FPSO). As noted above, Santos has considered the impacts and risks associated with the presence of the offtake tanker within OA1 and the PSZ.	Not Applicable
<ul> <li>[35] – [53] ACF stated there is no detail provided in the 23/9/2024 Version of the EP as to how off-take tankers, as third-party vessels, will be vetted, where and when that will occur and by whom, what the criteria are and who will be the parties to the agreement referred to.</li> <li>ACF stated its assumption is that guidelines referred to are the Oil Companies International Marine Forum Guidelines for Offshore Tanker Operations but it is not clear that the guidelines cover the operation in the way the 23/9/2024 Version of the EP describes.</li> <li>ACF stated it is also concerned that reliance is placed on guidelines written by industry that are not subject to Australian government oversight.</li> <li>ACF stated the movement of offtake activities is a risk of the activity and hence must be detailed, evaluated and control measures specified; at a minimum the offtake tankers when approaching the FPSO and within OA1 should be considered part of the activity and treated accordingly for the purposes of the EP.</li> </ul>	Santos has considered the matters raised by ACF in relation to the FPSO and associated offtake tankers and third party vessels in the Barossa Production Operations EP and provided a response to ACF. These matters are not relevant to this Activity and not addressed in this OEMP.	Santos confirms that the reference on p 939 of the 24/9/2024 Version to 'Guidelines' is a reference to the Oil Companies International Marine Forum Guidelines for Offshore Tanker Operations, referred to immediately above and to the left of the statement extracted by the ACF. The relevant detail (that provides for reducing associated risk of unplanned hydrocarbon release) as contained in the Oil Companies International Marine Forum Guidelines (and, by extension, the vessel vetting process) is included in the 24/9/2024 Version of the EP, this being "tankers with double hulls and fully segregated ballast tanks is a requirement of the vetting process as well as a MARPOL requirement that is monitored by way of regular statutory inspections". (refer to Section 2.7.4.1) The unplanned hydrocarbon release risk associated with condensate offtake operations, inclusive of the presence of the offtake tanker within OA1, is assessed and considered in the 24/9/2024 Version. (Also see Santos' above response to paragraphs [22]-[24] and [31]-[34] of the ACF's 23 December letter.) Santos: • has properly considered the impacts, risks and management measures included in the EP to address the risk of a loss of containment; and • notes this is properly a matter for NOPSEMA to consider and determine in the exercise of its regulatory functions.	Not Applicable
[54] ACF stated it is not obvious where the information on support and offtake vessel interactions is in the 23/9/2024 Version of the EP.	Santos has considered the matters raised by ACF in relation to the FPSO and condensate offtakes in the Barossa Production Operations EP and provided a response to ACF. These matters are not relevant to this Activity and not addressed in this OEMP.	The reference to Section 2.8.6 identified by the ACF is a typographical error. The correct reference is to Section 2.7.4.1 (Offtake Operations), which details the role of support vessels in offtake operations, noting: 'at least one support vessel is on location to provide static tow of the offtake tanker and assisting in berthing and disconnect'.	Not Applicable
[55] – [57] ACF stated the Barossa Terminal Handbook is referenced in the 23/09/2024 Version of the EP in relation to several control measures intended to reduce the risk of hydrocarbon spill and should form part of the EP and be provided to Relevant Persons.	Santos has considered the matters raised by ACF in relation to the FPSO and condensate offtakes in the Barossa Production Operations EP and provided a response to ACF. These matters are not relevant to this Activity and not addressed in this OEMP.	The EP is required to describe how the suite of proposed control measures are sufficient to reduce identified impacts and risks of the Activity to ALARP and acceptable levels. On p 768 of the 24/9/2024 version, the 'environmental benefit' of BAO-CM-071 (ie. the Barossa Terminal Handbook) describes the function of the Handbook in reducing the risk of a vessel collision that may lead to an unplanned release of condensate from the FPSO. This is sufficient for the purpose of NOPSEMA's assessment against the requirements of the Offshore Petroleum and Greenhouse Gas Storage (Environment) Regulations 2023 (the Regulations) for an EP.	Not Applicable

		Santos' obligation is to describe the measure sufficiently in the EP to explain its purpose and function as a control measure for identified impacts and risks.	
		Please note that in addition to information provided on p 768 of the 24/9/2024 Version, the function of the Barossa Terminal Handbook as a control measure (BAO-CM-071) for the EP is further described on p 851 of the EP. Santos considers that the information included in the EP provides sufficient detail in relation to the role of the Handbook in reducing impacts and risks to a level that is ALARP and acceptable. NOPSEMA will have opportunities to assess proper implementation and effectiveness of the Barossa Terminal Handbook as a control measure during compliance inspections of the accepted EP.	
<ul> <li>[58] - [67] ACF stated the following in relation to Greenhouse Gas Emission characterisation in the 23/09/2024 Version of the EP:</li> <li>The categorisation of Scope 3 emissions is assumed by ACF to be based on The Greenhouse Gas Protocol A Corporate Accounting and Reporting Standard Revised Edition but this needs to be clarified.</li> <li>The definition of scope 1 under the "NGER regime" is inaccurate</li> <li>The EP should adopt the Australian definition of scope 1, 2 and 3, applying that support vessels emissions are Scope 1 and the same is true of the transport of condensate and business travel.</li> <li>The EP incorrectly determines emission scope by reference to the concept of "reporting entity", without explanation.</li> <li>The 23/9/2024 Version also incorrectly determines emission scope by reference to ownership and control.</li> </ul>	Santos has considered the matters raised by ACF in relation to the Barossa Production Operations EP and provided a response to ACF. The Activity described in this OEMP has adopted the basis of Scope 1 and Scope 3 GHG emissions definitions as those in the Barossa Production Operations EP.	Santos has calculated the anticipated Scope 1 emissions of the Activity consistent with the methods it uses to calculate and report its reporting entity emissions under the NGER regime. Under that regime, a corporation is required to report on emissions under its operational control (section 19, NGER Act). A person has operational control over a facility if the person has the authority to introduce and implement operating, health and safety, and environmental policies (section 11). Emissions from sources such as "support vessels" and "transport of condensate" are not under the operational control of the Barossa Joint Venture and, therefore, will not be included as Scope 1 (or Scope 2) emissions for NGER reporting purposes. The definition of Scope 3 emissions in the GHG Protocol was adopted because there is no "Australian definition" of this concept under the NGER regime (which relates to Scopes 1 and 2 emissions only).	Not Applicable
The 23/9/2024 Version requires revision so the GHG emissions figures are accurate.			
<ul> <li>[68] – [89] ACF stated the following in relation to evaluation impact of changes in climate:</li> <li>It has not seen any documentation whereby Santos sufficiently addresses the impacts of</li> </ul>	Santos has considered the matters raised by ACF in relation to the Barossa Production Operations EP and provided a response to ACF.	International frameworks, namely the Paris Agreement, have been developed to facilitate an orderly approach to what is a global problem. This framework recognises, and is premised on, the fact that climate change is a global issue – there is no correlation between where GHG emissions are released and where climate change impacts are falt.	Not Applicable
<ul> <li>its emissions on the climate or how it will address the climate impacts of its scope 1 and 3 emissions.</li> <li>No detail has been provided on how Santos will address the direct or indirect consequences on the environment, including matters of national environmental significance, from the release of greenhouse gases.</li> <li>Santos should model the temperature increases that will occur as a result of the GHG emissions from Barossa (scope 1 and 3) and assess the impacts of that temperature increase upon the environment.</li> <li>Santos refuses to engage meaningfully on the matter and its approach to provision of information during the consultation phase was not consistent with the Regulations.</li> <li>Santos failed (despite ACF's requests) to provide an assessment of the consequences on Australia of the pollution that the project will generate, specifically the greenhouse gas emissions.</li> <li>The 23/9/2024 Version of the EP is non-compliant with the Regulations and this needs to be addressed before NOPSEMA makes a</li> </ul>		<ul> <li>Where climate change impacts are relit.</li> <li>For this reason, Australia sets and reports against its emissions reduction targets in 'net' terms, not by individual sectors or projects. The Australian Government is aware of planned production in the Barossa gas field and has considered this in its emissions reduction targets outlined in its Nationally Determined Contribution (NDC) under the Paris Agreement. The Offshore Project Proposal was previously approved by NOPSEMA and a production licence conferring rights to extraction of the gas resource on the Barossa joint venture has already been granted.</li> <li>Domestically, GHG emissions are regulated through Australia's NDC and the Safeguard Mechanism, and Santos has legal obligations to ensure that the Activity operates in accordance with that framework.</li> <li>Physical impacts of climate change on environmental receptors are the result of global GHG emissions from a multitude of sources (minus the GHG sinks) that have accumulated in the atmosphere. The impacts of climate change cannot be attributed to one specific sector or activity.</li> <li>In the context of evaluating potential impacts and risks that may be associated with GHG emissions from all sources globally, including from this Activity, Santos has considered the physical impacts of climate change (Section 6.3.2.6 of the 24/9/2024 Version).</li> <li>Santos considers the ACF's summary of engagement on this issue outlined in the 23 December letter, reproduced in the 'ACF assertion' column to the left, is incomplete and inaccurate. The summary does not refer to all relevant engagement from Santos.</li> </ul>	

It is possible, and required, to evaluate the impact on Australia of the GHG emissions the Activity will cause.			
[90] – [91] ACF stated the activity is inconsistent with the Paris Agreement and with limiting warming to 1.5 degrees and that Santos' reasoning on this issue is flawed, in that compliance with the	Santos has considered the matters raised by ACF in relation to the Barossa Production Operations EP and provided a response to ACF.	Australia has a well-established legislative framework under which certain GHG emissions from Barossa production operations will be regulated or managed to further Australia's transition to net zero emissions by 2050. This includes:	Not Applicable
Safeguard Mechanism does not necessarily ensure compliance with a 1.5 degree scenario. ACF further stated Santos has not expressed any		<ul> <li>GHG emissions reporting under the NGER Act and the National Greenhouse and Energy Reporting (Measurement) Determination 2008</li> <li>the Emissions Reduction Fund (Australian Carbon Credit Units</li> </ul>	
ACF on potential solutions.		Scheme) the Safequard Mechanism to keep net emissions below an	
ACF stated that potential solutions could include: 1. Ensuring Santos' customers are using products to displace coal and not on-selling to		established baseline and require net-zero reservoir emissions for new gas fields that feed LNG projects. The Safeguard Mechanism currently applies to facilities that emit more than 0.1 MtCO2-e per applies	
other nations.		Compliance with the Safeguard Mechanism is of primary importance in	
<ol> <li>Limiting sales to customers using on-site abatement (CCS) in their gas-fired power stations.</li> <li>Not selling to customers prior to ensuring that</li> </ol>		ensuring that GHG emissions associated with the Activity are as low as reasonably practicable and acceptable. GHG emissions at or below the baseline and the Safeguard Mechanism's future decline rates are already	
their countries have legitimate and credible policy		anticipated and thus accounted for under Australia's NDCs under the Paris Agreement.	
pathways to meet their NDCs.		Unlike Scope 1 emissions, there is no compliance framework for Scope 3 emissions management in Australia. This is because Scope 3 emissions are the Scope 1 and Scope 2 emissions of upstream suppliers and downstream users. The international system for GHG regulation, underpinned by the UNFCCC and Paris Agreement, recognises that responsibility lies with the countries in which those emissions are generated. NDCs, including that of Australia, are consistent with this approach. Accordingly, Scope 3 emissions commitments made by companies are generally voluntary.	
		Santos, in its capacity as an independent company, has implemented a multi- faceted corporate plan to address Scope 3 emissions across its operations. The actions to address Scope 3 emissions include:	
		<ul> <li>continue engaging our suppliers to deepen our understanding of their emissions</li> </ul>	
		• provide guidance and education to support our suppliers to improve the accuracy of their emissions calculation	
		<ul> <li>work with suppliers to develop mutually beneficial emissions reduction initiatives</li> </ul>	
		<ul> <li>continue engaging with our industrial customers, LNG customers and regional emitters to provide carbon management solutions, including progressing our current MOUs, and</li> </ul>	
		• investment in carbon capture and storage, and other technologies to advance the development of low carbon fuels.	
		These corporate commitments as an independent company will be applied to suppliers and to Santos' customers in respect of the Barossa Development. This approach ensures that Scope 1 emissions for Barossa will be managed in accordance with legislative compliance obligations (which include consideration of best practice in their formulation) and that a proactive and collaborative voluntary approach is taken to addressing Scope 3 emissions through Santos' broader corporate initiatives as an independent company and not in its capacity as Operator of the Barossa JV.	
		Additional control measures addressing Scope 3 emissions have been considered but not adopted for various reasons including that they are the Scope 1 and 2 emissions of other countries, who have their own nationally determined contributions and targets under the Paris Agreement, in accordance with their own national pathways for emissions reduction.	
<ul> <li>[92] – [97] ACF stated there is a lack of adequate (including independent, peer-reviewed) research informing the impact analysis of the operations upon the marine environment, particularly as it relates to marine megafauna and the claims around distribution of migratory species within OA1 and OA2.</li> <li>ACF further stated this undermines the legitimacy of Santos' claims around the consequences of its</li> </ul>	Notwithstanding that this issue was raised in relation to the Barossa Production Operations EP, Santos has considered the matters raised by ACF in the OEMP, given the proximity of OA2 described in the Barossa Production Operations EP and this Activity. Santos disagrees with ACF's comments for the reasons given in response to ACF. Santos considers its evaluation of impacts and risks to marine species, including in respect of the Dusky Sea Snake, Brydes Whale, Omura Whale and Pygmy Blue Whale is appropriate. Santos' evaluation of impacts and risks is informed by a wide array of peer-reviewed scientific papers, extensive baseline field studies, the EPBC protected matters database, conservation management plans, conservation advice and recovery plans where applicable.	Santos has conducted extensive baseline field studies at the activity location and considered regionally relevant contemporaneous information from a wide array of peer-reviewed scientific papers to inform the impact assessment. The management of potential impacts is aligned with conservation advice, and conservative control measures have been adopted to mitigate any potential impacts. The Barossa FPSO Operational Area is located where there are no known significant feeding, breeding, or aggregation areas for marine mammals, largely attributable to the 250-metre depth. Santos' assessment is that the proposed control measures are sufficient	Section 3 (baseline field studies) Section 3.2.13.5 (Biologically important areas and habitats) Section 6.1 Section 6.1.3 Section 6.3



further reinforces the need for further research and more fulsome cumulative impact analysis to be	As previously responded to ACF (refer above to meeting with ACF on 9 July 2024 and Santos' correspondence to ACF on 7 August 2024), Santos reiterates that scientific information available on biologically important areas for	control measures, including vessel speed restrictions within 500m of the FPSO and other vessel operations within Operational Areas (BAO-CM-028).	Section 7.3.3
undertaken. The ACF objected to the mitigation measures proposed by Santos in relation to marine species, including specifically in relation to Sea Snakes, the Dusky Sea Snake (which is now endangered), Byrdes & Omura whales, and the Pygmy Blue Whale).	protected marine species have been considered in the evaluation of impacts and risks on whales and turtles. BIAs for fish, sharks and rays, marine mammals, marine reptiles, and seabirds and shorebirds are addressed in Section 3.2.13 of the OEMP. Regarding ACF's objection to mitigation measures, Santos maintains that its proposed control measures in relation to marine species, including the Dusky Sea Snake, Brydes Whale, Omura Whale and Pygmy Blue Whale, are sufficient. Santos has evaluated the control measures proposed by ACF (see Section 7.3.3 and also Section 6.1.3) (excluding the control measure regarding installation, which is not an activity authorised under this OEMP).	Section 7 of the EP assesses that potential impacts and risks associated with marine fauna interaction are expected to be limited to individuals and/ or small groups transiting the Operational Areas, without impacts at a population level. The environment performance outcome commits to zero incidents of injury/ mortality of cetaceans/ marine reptiles from collision with project vessels operating within the project area (EPO-03) to ensure impacts are limited to acceptable levels.	
<ul> <li>ACF stated that no additional control measures have been taken since its last consultation with Santos and submitted the following:</li> <li>In the absence of adequate data, installation and operation should not happen during peak migration or periods of ecological sensitivity.</li> <li>Vessel operations should only occur with an observer on board and activities should only be undertaken during daylight hours when migration and peak and a server a serve</li></ul>		To inform evaluation of impacts and risks to marine species, and appropriateness of proposed control measures, Santos has not relied solely on regional desktop literature and the EPBC protected matters database, as demonstrated through the Barossa marine field studies program for the Barossa Development OPP. The Barossa marine field studies program was an extensive and robust environmental baseline studies program, designed to characterise the marine environment values and sensitivities within the Project area. The results of the marine studies program have been critical in supplementing published literature at a regional scale with information that is specific to the activity location.	
mitigation measures applied		With regard to the specific marine fauna species highlighted by the ACF:	
		• The Dusky Sea Snake was identified as potentially occurring within the EMBA (associated with a loss of hydrocarbon event), but not within the Operational Areas where planned activities will occur. Locations where Dusky Sea Snakes may be present are situated several hundred kilometres from the Activity OAs.	
		• The Barossa Marine Studies program did identify presence of Brydes whales, Omura whales and Pygmy Blue whales in the vicinity of OA1, albeit in limited numbers. On the basis of these results, it is acknowledged in the EP that impacts from the Activity, such as underwater noise or unplanned vessel interactions, have the potential to impact individuals that may transit OA1. These impacts and risks have been considered in the EP.	
		The above information has informed the evaluation of impacts and risks from the Activity and appropriateness of proposed control measures. Regarding ACF's request that vessel operations should only occur with an observer on board, proposed control measure BAO-CM-001 provides for vessel crew to act as wildlife observers and record sightings of cetaceans and turtles. The EP evaluates a potential control measure of restricting vessel operations to daylight hours, however this was not adopted due to the disproportionate cost impact relative to the potential environmental benefit when only limited numbers of marine species are expected in OA1 (as evidenced by the number of marine fauna sightings (<10) during drilling and installation activities).	
		Note: BAO-CM_028 is now BAO-CM-6.6.6 and BAO CM 001 is now BAO- CM-6.1.1	
[98] – [105] ACF stated it has repeatedly asked about Barossa's compliance with the Safeguard Mechanism and have been repeatedly told that it is not a matter for ACF. ACF stated the 23/9/2024 Version of the EP provides no evidence that the titleholder can meet	Santos has considered the matters raised by ACF and provided a response to ACF. Notwithstanding that this issue was raised in relation to the Barossa Production Operations EP, the OEMP describes Barossa's compliance with the Safeguard Mechanism within Section 6.3. Santos disagrees with the ACF's comments for the reasons given in response to the ACF. In addition, the responsible Commonwealth minister has communicated to the Barossa joint venture partners on a number of occasions that he anticipates there being ACCUs available for this project	Regarding the ACF's concerns about Santos' failure to adequately attempt to demonstrate compliance with the Commonwealth Safeguard Mechanism, Santos notes that the ACF has previously raised similar claims in correspondence to Santos on 9 April 2024 and 9 September 2024 and in the meeting on 9 July 2024. Santos' obligation is to demonstrate compliance with the Commonwealth	Section 6.3 Appendix C
the requirements of the SGM in terms of the availability and cost of Australian Carbon Credit	Santos considers that the ACF has been provided sufficient information to enable the ACF to make an informed	Safeguard Mechanism to the Clean Energy Regulator.	
Units or Safeguard Mechanism Credits and	assessment of any potential consequences of the Activity on its functions, interests or activities.	Santos re-iterates the following points:	
provides no comfort that the necessary volume of offsets will exist through the 25-year design life.		Barossa Production Operations Scope 1 emissions will be managed in accordance with the applicable baseline under the Safeguard Mechanism, which has been developed by the Australian government with regard to Australia's Paris Agreement targets and associated emissions budget. The NT emissions budget is accounted for in Australia's national emissions budget.	
		• The treatment of the Barossa Project under the Safeguard Mechanism is a matter for the Clean Energy Regulator. Santos will abide by the Clean Energy Regulator's final determination.	
		There are various options available to meet legislative requirements     under the Safeguard Mechanism rules. Compliance with obligations     under the Safeguard Mechanism may be achieved through (among other	

C	-	L	
D	d	IU	5

		things) the purchase or surrender of Australian Carbon Credit Units (ACCUs) or Safeguard Mechanism Credits (SMCs).	
		The availability of ACCUs over the next decade is supported by analysis of the balance between supply, holdings, and cancellation rates (Reputex, 2024. Market Outlook Australian Carbon Price, Supply & Demand Outlook September 2024). Throughout the decade, ACCU supply is projected to steadily increase, surpassing 30 million annually by 2030, alongside rising cancellation rates to meet Safeguard Mechanism requirements. By 2027, supply is forecasted to fall into deficit due to growing annual cancellations, which will force ACCUs to be withdrawn from inventory until new supply comes into the market. Modelling shows ACCU holding inventory will continue to rise from approximately 36 million in 2024 to around 50 million by 2027, providing sufficient coverage for the deficit. Additionally, growing demand will send price signals to incentivise new ACCU supply. The Safeguard Mechanism is administered by the Clean Energy Regulator	
		imposing penalties.	
[106] – [115] ACF stated that while the 23/9/2024 Version of the EP says CCS is not part of the Activity for this EP, it is apparent from other public	Santos has provided a response to ACF. CCS is not part of the Activity for the Coastal Waters OEMP.	Santos notes that the ACF has previously raised similar claims regarding CCS in correspondence on 9 April 2024 and 9 September 2024 and in the meeting on 9 July 2024.	Not Applicable
are cited by ACF) that CCS is part of the Barossa project and the EP activity. ACF further stated the EP needs to address this in		Santos re-iterates that while it is exploring CCS opportunities at Bayu-Undan and elsewhere, these CCS projects are not owned or controlled by the Barossa joint venture, nor Santos in its sole capacity, are not part of the Barossa development, and are not within the scope of the EP.	
order to comply with the Regulations, specifically regulations 21(1)(c) and (d) and 21(4)		It is subject to a different regulatory regime and pursued by a different proponent. Santos is one of the joint venture parties for the Bayu-Undan CCS and has committed funding to it. Santos hopes that the project proceeds, but this is reliant on approvals from a foreign government (Timor-Leste) and final investment decisions by the proponents of that project. If the Bayu-Undan CCS project proceeds, the Barossa Joint Venture hopes to be a customer of it.	
		CCS is not a viable control measure for inclusion in the EP because it is not currently available. Potential CCS developments such as the proposed Bayu-Undan CCS project require regulatory frameworks, policies and approvals (from a different jurisdiction to the Activity) to be in place prior to taking final investment decisions. Therefore, it is not possible to rely on them for GHG emissions abatement.	
ACF correspondence to Santos on 24 March 2025 ACF claimed Santos' correspondence of 31	Santos has considered the matters raised by ACF in the Barossa Production Operations EP and provided a response to ACF. For completeness noting that many (although not all) of the concerns raised by the ACF may be assumed to apply equally to the GEP Coastal Waters OEMP, concerns that apply to the OEMP have been	Santos correspondence to ACF on 31 March 2025 in response to ACF correspondence of 24 March 2025. We refer to your letter of 24 March 2025 (24 March letter), and to all other	
January 2025 had not sufficiently addressed the matters ACF had raised in its letter of 23 December 2024, in particular how Santos will	Santos has considered the matters raised by the ACF and remains satisfied that its response to the ACF's letter of 23 December 2024 is comprehensive and appropriate.	correspondence to date between Santos and the Australian Conservation Foundation (ACF) in respect of the Barossa Production Operations Environment Plan (EP).	
ACF claimed the Production Operations EP has numerous deficiencies in the overall approach to	Santos has provided extensive responses to ACF's queries concerning compliance with the Safeguard Mechanism, including in response to the ACF's letter of 23 December 2024. This matter is also addressed in section 6.3 of the OEMP.	Santos has considered the 24 March letter and notes each of these matters has been considered and appropriately addressed by Santos previously in correspondence with the ACF or in the content of the EP.	
risk assessment and lacks sufficient information on environmental features that would be impacted. ACF claimed the (OPGGSR) Regulations and best practice require that Santos should undertake the following actions before the Production Operations EP can be accepted by NOPSEMA:	Santos acknowledges the ACFs objections and claims regarding the content of the Production Operations EP, noting that they are expressed broadly without detailing specifics and that many (although not all) of the concerns raised by the ACF may be assumed to apply equally to the OEMP. The ACF has not identified any specific information in relation to the environment that has been omitted from the OEMP, and Santos remains satisfied that its description of the environment in the OEMP is comprehensive and appropriate. Santos has assessed the specific objections and claims raised by the ACF below.	For the reasons given previously to the ACF, and set out in the EP, Santos remains satisfied that the EP meets the requirements of the Offshore Petroleum and Greenhouse Gas Storage (Environment) Regulations 2023 (Cth), including that its evaluation of impacts and risks is comprehensive and appropriate.	
<ol> <li>Conduct a comprehensive long-term and cumulative impact assessment regarding potential effects on marine ecosystems</li> </ol>	The OEMP, and associated consultation materials, has assessed the impacts and risks of the full life cycle (25 years) of the proposed Activity. There are no regular operations emissions and discharges from the Activity, with the exception of the potential for fugitive emissions (as detailed in Section 6.3) and . no concurrent planned impacts associated with this Activity.		Not applicable
<ol> <li>More accurately estimate the likelihood or consequence of environmental impacts in the risk assessment</li> </ol>	Santos considers its estimation of risks is appropriate, both in its methodology (section 5) and in how that methodology has been applied to the impacts and risks identified (sections 6 and 7). Santos' evaluation of impacts and risks is informed by a wide array of peer-reviewed scientific papers, extensive baseline field studies, the EPBC protected matters database, conservation management plans, conservation advice and recovery plans where applicable. A number of these data sources are quantitative in nature and are used by Santos wherever possible to inform evaluation of impacts and risks from the proposed Activity.		Sections 5, 6 and 7
3. Evaluate impacts on specific marine species, particularly threatened and migratory species, in greater detail	Santos considers its evaluation of impacts and risks to listed and protected marine species is appropriate. Santos' evaluation of impacts and risks is informed by a wide array of peer-reviewed scientific papers, extensive baseline		Sections 6 and 7

		field studies, the EPBC protected matters database, conservation management plans, conservation advice and recovery plans where applicable.
4.	Create robust long-term plans for monitoring of the project's ecological impacts on the marine environment	Santos has considered the matters raised by ACF in the Barossa Production Operations EP and provided a response to ACF. These matters are not relevant to this Activity and not addressed in this OEMP as there are no regular operations emissions and discharges from this Activity and planned activity impacts are minor to negligible (Section 6).
5.	Improve the Production Operation EP's proposed control measures,	The proposed control measures, and supporting EPSs and measurement criteria, are considered sufficiently detailed and specific and address all identified impacts and risks from the Activity. For completeness, Santos notes that control measures, EPSs and measurement criteria have been revised since Rev 1 of the Barossa Production Operations EP, on which the ACF's comments are based, including to increase specificity. Relevant revised control measures from the EP have been adopted in this OEMP.
6.	Properly assess long-term and cumulative impacts of greenhouse gas and atmospheric emissions	The OEMP, and associated consultation materials, has assessed the impacts and risks of the full life cycle (25 years) of the proposed Activity. Activity GHG emissions have been considered in the context of cumulative global GHG emissions (see section 6.3). The OEMP considers current published literature regarding climate science, Australian government GHG emissions reporting and projections, and other relevant contemporary literature from government and non-government sources. An acceptability evaluation for each identified impact and risk forms part of the OEMP, which considers consistency of the OEMP with principles of ecological sustainable development, which in turn includes consideration of the precautionary principle. The OEMP contains control measures to keep net emissions at or below the legislated Safeguard Mechanism baseline and additional control measures to monitor and further reduce atmospheric and GHG emissions to as low as reasonably practicable.
7.	Revise the assessments for unplanned releases and improve long-term	The OEMP is supplemented by the GEP NT waters Oil Pollution Emergency Plan (OPEP) which addresses spill response activities for identified credible unplanned release scenarios.
	ecological monitoring plans	The OEMP includes a risk assessment for the worst case, unplanned releases based on spill modelling, and assesses the environment that may be affected, areas of high environmental value and hot spots. The nature and scale of hydrocarbon spills on the environment that may be affected from entrained, dissolved and floating hydrocarbons are also assessed. Santos considers that the ACF's claim that the assessment of unplanned releases is inadequate, has no merit.
		The OPEP is also accompanied by the Operational and Scientific Monitoring Bridging Implementation Plan, that describes a comprehensive operational and scientific monitoring program to be implemented in the unlikely event of an unplanned hydrocarbon release event.
		The operational monitoring component provides short term situational awareness of a hydrocarbon spill, enabling a timely assessment of the effectiveness of the spill response. Scientific Monitoring is the principal tool for determining the extent, severity and persistence of environmental impacts from a hydrocarbon spill over the longer term and for informing associated remediation activities (if required).
8.	Conduct more detailed analysis on cumulative impacts on marine fauna from multiple vessels, consider impacts on smaller cetaceans and pinnipeds, and properly consider seasonal variations in migratory bird presence	Santos has considered the matters raised by ACF in the Barossa Production Operations EP and provided a response to ACF. These matters are not relevant to this Activity and not addressed in this OEMP as IMMR vessel presence is infrequent, with vessel presence occurring typically for approximately 7 to 30 days in duration every three to five years, or as needed. There are no planned concurrent activities associated with this Activity.
9.	Improve assessment of impacts from operational discharges, providing more	All planned and contingency emissions and discharges from the proposed Activity have been assessed in the OEMP.
specific targets for emissions reductions and more detailed assessment of long- term climate change impact.		Santos notes that there are no regular operations emissions and discharges in the OEMP, except for the potential for fugitive GHG emission from the pipeline. The OEMP includes for monitoring emissions from the Barossa Gas Project and to evaluate effectiveness of control measures, such as the GHG emissions management plan, to reduce emissions to ALARP over the life of the Activity.
		For completeness, Santos notes that further detail has been added to the Barossa Production Operations EP on this topic since Rev 1 (on which the ACF's comments are based), which have also been included in the OEMP where relevant.

#### Australian Marine Conservation Society - NT branch (AMCS NT)

Summary of consultation effort:

- On 9 February 2024 Santos emailed AMCS NT branch to advise it of preliminary consultation regarding proposed activities for consultation to be managed under the Coastal Waters OEMP and Barossa Production Operations EP. [Con-3787]
- The email advised that Santos was seeking information to better understand:
  - o if you are from a government Department or agency, how the proposed Production Operations activities may be relevant to your department or agency; and
  - what (if any) functions, interests or activities you or your organisation have that may be affected by the proposed Production Operations activities.
- The email included information on the regulatory process for the activities in Commonwealth and NT jurisdictions and links to a Santos information booklet on the proposed activities and a NOPSEMA brochure on the consultation process and details of how to contact • Santos to register as a Relevant Person. The email stated that the consultation phase would commence on 11 March 2024 and close on 9 April 2024.
- On 11 March 2024 Santos emailed AMCS NT branch further to the previous correspondence, to advise that it had commenced the consultation phase which would run until 9 April 2024. In addition to the previous information again being provided, Santos provided • information on Relevant Persons' entitlements under the regulatory processes, details of how to provide feedback and a reminder of the closing date for consultation. [Con-3793]
- On 4 April 2024 Santos called AMCS NT regarding consultation for Production Operations activities and reminded AMCS NT of the deadline for providing input. AMCS NT advised that it had forwarded the previous emails to AMCS head office in Brisbane.



Section 6
Sections 6, 7 and 8
Section 6.3
Section 7.6 GEP NT waters OPEP Operational and Scientific Monitoring Bridging Implementation Plan: Northern Australia
Not applicable
Section 6 Section 6.3

Section 8

- On 8 May 2024, Santos emailed AMCS NT branch further to previous correspondence to advise it had extended the consultation period until 22 May 2024. In providing this extension of time, Santos advised that
  updated to account for an additional risk associated with the proposed activity. Santos confirmed that consultation will close on the revised date unless Santos hears otherwise from AMCS NT. [Con-4021]
- On 10 July 2024 Santos emailed AMCS NT branch to advise the consultation period had been completed. Santos advised AMCS NT branch that it considered consultation had now closed for the purpose of Sar
  activities to government regulators for assessment. [Con-5130]
- On 15 October 2024 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP.
   NOPSEMA for assessment, provided a link to the full EP on the NOPSEMA website and Santos' contact details for any queries. The Update also stated that the OEMP would be submitted to the NT-DITT for assessment.
- On 23 January 2025 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP.
   NOPSEMA to provide further information, Santos had re-submitted the EP with some amendments in December 2024 for further assessment. The Update also re-stated that the OEMP would be submitted to the

Notwithstanding the consultation information provided and the steps described above, no comments or input were received on the Coastal Waters OEMP from Australian Marine Conservation Society.

No response was received from Australian Marine Conservation Society.Santos considers it has provided sufficient information and a reasonable period of time for consultation.No response required.Santos considers Section 25 consultation requirements to have been met.Santos considers areasonable period of time for consultation.No response required.	Summary of response by Relevant Person	Assessment of merits	Santos' Response Statement
	No response was received from Australian Marine Conservation Society.	Santos considers it has provided sufficient information and a reasonable period of time for consultation. Santos considers Section 25 consultation requirements to have been met.	No response required.

#### Australian Parents for Climate Action Darwin and NT (APCAD)

Summary of consultation effort:

- On 9 February 2024 Santos emailed APCAD to advise it of preliminary consultation regarding proposed activities for consultation to be managed under the Coastal Waters OEMP and Barossa Production Opera
- The email advised that Santos was seeking information to better understand:
  - o if you are from a government Department or agency, how the proposed Production Operations activities may be relevant to your department or agency; and
  - what (if any) functions, interests or activities you or your organisation have that may be affected by the proposed Production Operations activities.
- The email included information on the regulatory process for the activities in Commonwealth and NT jurisdictions and links to a Santos information booklet on the proposed activities and a NOPSEMA brochure o
  Santos to register as a Relevant Person. The email stated that the consultation phase would commence on 11 March 2024 and close on 9 April 2024.
- On 13 March 2024 Santos emailed APCAD further to the previous correspondence, to advise that it had commenced the consultation phase which would run until 9 April 2024. In addition to the previous informa
  Relevant Persons' entitlements under the regulatory processes, details of how to provide feedback and a reminder of the closing date for consultation. [Con-3794]
- On 8 May 2024, Santos emailed APCAD further to previous correspondence, to advise it had extended the consultation period until 22 May 2024. In providing this extension of time, Santos advised that the informaccount for an additional risk associated with the proposed activity. Santos confirmed that consultation will close on the revised date unless Santos hears otherwise from APCAD. [Con-4022]
- On 10 July 2024 Santos emailed APCAD to advise the consultation period had been completed. Santos advised APCAD that it considered consultation had now closed for the purpose of Santos finalising and su
  government regulators for assessment. [Con-5130]
- On 15 October 2024 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP.
   NOPSEMA for assessment, provided a link to the full EP on the NOPSEMA website and Santos' contact details for any queries. The Update also stated that the OEMP would be submitted to the NT-DITT for assessment.
- On 23 January 2025 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP.
   NOPSEMA to provide further information, Santos had re-submitted the EP with some amendments in December 2024 for further assessment. The Update also re-stated that the OEMP would be submitted to the

Notwithstanding the consultation information provided and the steps described above, no comments or input were received on the Coastal Waters OEMP from APCAD.

Summary of response by Relevant Person	Assessment of merits	Santos' Response Statement
No response was received from Australian Parents for Climate Action Darwin and NT.	Santos considers it has provided sufficient information and a reasonable period of time for consultation. Santos considers Section 25 consultation requirements to have been met.	No response required.

#### **Climate Action Darwin**

Summary of consultation effort:

- On 9 February 2024 Santos emailed Climate Action Darwin to advise it of preliminary consultation regarding proposed activities for consultation to be managed under the Coastal Waters OEMP and Barossa Pro
- The email advised that Santos was seeking information to better understand:
  - o if you are from a government Department or agency, how the proposed Production Operations activities may be relevant to your department or agency; and
  - what (if any) functions, interests or activities you or your organisation have that may be affected by the proposed Production Operations activities.
- The email included information on the regulatory process for the activities in Commonwealth and NT jurisdictions and links to a Santos information booklet on the proposed activities and a NOPSEMA brochure o
  Santos to register as a Relevant Person. The email stated that the consultation phase would commence on 11 March 2024 and close on 9 April 2024.
- On 11 March 2024 Santos emailed Climate Action Darwin further to the previous correspondence, to advise that it had commenced the consultation phase which would run until 9 April 2024. In addition to the previous formation on Relevant Persons' entitlements under the regulatory processes, details of how to provide feedback and a reminder of the closing date for consultation. [Con-3793]
- On 4 April 2024 Santos called Climate Action Darwin to remind Climate Action Darwin of the deadline for providing input. Santos was advised the message would be passed to the appropriate personnel.
- On 8 May 2024, Santos emailed Climate Action Darwin further to previous correspondence to advise it had extended the consultation period until 22 May 2024. In providing this extension of time, Santos advised updated to account for an additional risk associated with the proposed activity. Santos confirmed that consultation will close on the revised date unless Santos hears otherwise from Climate Action Darwin [Con-4]
- On 10 July 2024 Santos emailed Climate Action Darwin to advise the consultation period had been completed. Santos advised Climate Action Darwin that it considered consultation had now closed for the purpo these activities to government regulators for assessment. [Con-5130]
- On 15 October 2024 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP.
   NOPSEMA for assessment, provided a link to the full EP on the NOPSEMA website and Santos' contact details for any queries. The Update also stated that the OEMP would be submitted to the NT-DITT for assessment.



the information in the booklet and factsheet had been			
ntos finalising and submitting enviro	nment plans for these		
The Update advised that the EP had sessment. [Con- 5982]	been submitted to		
The Update advised that, following a	a request from		
e NT-DITT for assessment. [Con-603	36]		
	OEMP reference		
	Not applicable.		
tions EP. [Con-3787]			
on the consultation process and deta	ils of how to contact		
tion again being provided, Santos p	rovided information on		
mation in the booklet and factsheet	had been updated to		
ubmitting environment plans for thes	e activities to		
The Update advised that the EP had been submitted to sessment. [Con- 5982]			
The Update advised that, following a request from > NT-DITT for assessment. [Con-6036]			
	OEMP reference		
	Not applicable.		
oduction Operations EP. [Con-3787]			
on the consultation process and details of how to contact			
evious information again being provided, Santos provided			
I that the information in the booklet and factsheet had been 023].			
se of Santos finalising and submitting environment plans for			
The Update advised that the EP had been submitted to sessment. [Con- 5982]			

- On 23 January 2025 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP.
   NOPSEMA to provide further information, Santos had re-submitted the EP with some amendments in December 2024 for further assessment. The Update also re-stated that the OEMP would be submitted to the
- Notwithstanding the consultation information provided and the steps described above, no comments or input were received on the Coastal Waters OEMP from Climate Action Darwin.

No response was received from Climate Action Darwin.	Santos considers it has provided sufficient information and a reasonable period of time for consultation. Santos considers Section 25 consultation requirements to have been met.	No response required.
Summary of response by Relevant Person	Assessment of merits	Santos' Response Statement

#### Summary of consultation effort:

- On 9 February 2024 Santos emailed CCWA to advise it of preliminary consultation regarding proposed activities for consultation to be managed under the Coastal Waters OEMP and Barossa Production Operation
- The email advised that Santos was seeking information to better understand:
  - o if you are from a government Department or agency, how the proposed Production Operations activities may be relevant to your department or agency; and
  - o what (if any) functions, interests or activities you or your organisation have that may be affected by the proposed Production Operations activities.
- The email included information on the regulatory process for the activities in Commonwealth and NT jurisdictions and links to a Santos information booklet on the proposed activities and a NOPSEMA brochure o
  Santos to register as a Relevant Person. The email stated that the consultation phase would commence on 11 March 2024 and close on 9 April 2024.
- On 11 March 2024 Santos emailed CCWA further to the previous correspondence, to advise that it had commenced the consultation phase which would run until 9 April 2024. In addition to the previous informati
  Relevant Persons' entitlements under the regulatory processes, details of how to provide feedback and a reminder of the closing date for consultation. [Con-3793]
- On 4 April 2024 Santos called CCWA and left a message regarding consultation for Production Operations activities with a team member.
- On 8 May 2024, Santos emailed CCWA further to previous correspondence to advise it had extended the consultation period until 22 May 2024. In providing this extension of time, Santos advised that the inform
  account for an additional risk associated with the proposed activity. Santos confirmed that consultation will close on the revised date unless Santos hears otherwise from CCWA [Con-4031].
- On 10 July 2024 Santos emailed CCWA to advise the consultation period had been completed. Santos advised it considered that consultation had now closed for the purpose of Santos finalising and submitting
  regulators for assessment. [Con-5130]
- On 15 October 2024 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP.
   NOPSEMA for assessment, provided a link to the full EP on the NOPSEMA website and Santos' contact details for any queries. The Update also stated that the OEMP would be submitted to the NT-DITT for assessment.
- On 23 January 2025 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP.
   NOPSEMA to provide further information, Santos had re-submitted the EP with some amendments in December 2024 for further assessment. The Update also re-stated that the OEMP would be submitted to the
- Notwithstanding the consultation information provided and the steps described above, no comments or input were received on the Coastal Waters OEMP from CCWA.

Summary of response by Relevant Person	Assessment of merits	Santos' Response Statement
No response was received from CCWA.	Santos considers it has provided sufficient information and a reasonable period of time for consultation.	No response required.
	Santos considers Section 25 consultation requirements to have been met.	



The Update advised that, following a request from > NT-DITT for assessment. [Con-6036]		
OEMP reference		
Not applicable.		
ons EP. [Con-3787]		
n the consultation process and details of how to contact		
on again being provided, Santos provided information on		
ation in the booklet and factsheet had been updated to		
environment plans for these activities to government		
The Update advised that the EP had been submitted to sessment. [Con- 5982]		
The Update advised that, following a request from e NT-DITT for assessment. [Con-6036]		
OFMP reference		
Not applicable.		

#### Doctors for the Environment Australia

Summary of consultation effort:

- On 9 February 2024 Santos emailed Doctors for the Environment Australia to advise it of preliminary consultation regarding proposed activities for consultation to be managed under the Coastal Waters OEMP a
- The email advised that Santos was seeking information to better understand:
  - o if you are from a government Department or agency, how the proposed Production Operations activities may be relevant to your department or agency; and
  - what (if any) functions, interests or activities you or your organisation have that may be affected by the proposed Production Operations activities.
- The email included information on the regulatory process for the activities in Commonwealth and NT jurisdictions and links to a Santos information booklet on the proposed activities and a NOPSEMA brochure on Santos to register as a Relevant Person. The email stated that the consultation phase would commence on 11 March 2024 and close on 9 April 2024.
- On 13 March 2024 Santos emailed Doctors for the Environment Australia further to the previous correspondence, to advise that it had commenced the consultation phase which would run until 9 April 2024. In ac Santos provided information on Relevant Persons' entitlements under the regulatory processes, details of how to provide feedback and a reminder of the closing date for consultation. [Con-3794]
- On 4 April 2024 Santos called Doctors for the Environment Australia and left voice mail message regarding consultation for Production Operations activities.
- On 2 May 2024, Santos emailed Doctors for the Environment Australia further to previous correspondence to advise it had extended the consultation period until 16 May 2024. In providing this extension of time, factsheet had been updated to account for an additional risk associated with the proposed activity. Santos confirmed that consultation will close on the revised date unless Santos hears otherwise from Doctors for
- On 10 July 2024 Santos emailed Doctors for the Environment Australia to advise the consultation period had been completed. Santos advised Doctors for the Environment Australia that it considered consultation submitting environment plans for these activities to government regulators for assessment. [Con-5130]
- On 15 October 2024 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP.
   NOPSEMA for assessment, provided a link to the full EP on the NOPSEMA website and Santos' contact details for any queries. The Update also stated that the OEMP would be submitted to the NT-DITT for assessment.
- On 23 January 2025 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP.
   NOPSEMA to provide further information, Santos had re-submitted the EP with some amendments in December 2024 for further assessment. The Update also re-stated that the OEMP would be submitted to the
- Notwithstanding the consultation information provided and the steps described above, no comments or input were received on the Coastal Waters OEMP from Doctors for the Environment Australia.

Summary of response by Relevant Person	Assessment of merits	Santos' Response Statement
No response was received from Doctors for the Environment Australia.	Santos considers it has provided sufficient information and a reasonable period of time for consultation. Santos considers Section 25 consultation requirements to have been met.	No response required.

#### Greenpeace Australia Pacific

Summary of consultation effort:

- On 9 February 2024 Santos emailed Greenpeace to advise it of preliminary consultation regarding proposed activities for consultation to be managed under the Coastal Waters OEMP and Barossa Production Operation
- The email advised that Santos was seeking information to better understand:
  - o if you are from a government Department or agency, how the proposed Production Operations activities may be relevant to your department or agency; and
  - what (if any) functions, interests or activities you or your organisation have that may be affected by the proposed Production Operations activities.
- The email included information on the regulatory process for the activities in Commonwealth and NT jurisdictions and links to a Santos information booklet on the proposed activities and a NOPSEMA brochure or Santos to register as a Relevant Person. The email stated that the consultation phase would commence on 11 March 2024 and close on 9 April 2024.
- On 11 March 2024 Santos emailed Greenpeace further to the previous correspondence, to advise that it had commenced the consultation phase which would run until 9 April 2024. In addition to the previous info on Relevant Persons' entitlements under the regulatory processes, details of how to provide feedback and a reminder of the closing date for consultation. [Con-3793]
- On 4 April 2024 Santos called and spoke to a team member regarding consultation for Production Operations activities who confirmed that previous correspondence had been received.
- On 8 May 2024, Santos emailed Greenpeace further to previous correspondence to advise it had extended the consultation period until 22 May 2024. In providing this extension of time, Santos advised that the ir account for an additional risk associated with the proposed activity. Santos confirmed that consultation will close on the revised date unless Santos hears otherwise from Greenpeace. [Con-4032]
- On 10 July 2024 Santos emailed Greenpeace to advise the consultation period had been completed. Santos advised Greenpeace that it considered consultation had now closed for the purpose of Santos finalising overnment regulators for assessment. [Con-5130]
- On 15 October 2024 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. T NOPSEMA for assessment, provided a link to the full EP on the NOPSEMA website and Santos' contact details for any queries. The Update also stated that the OEMP would be submitted to the NT-DITT for ass
- On 23 January 2025 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP.
   NOPSEMA to provide further information, Santos had re-submitted the EP with some amendments in December 2024 for further assessment. The Update also re-stated that the OEMP would be submitted to the
- Notwithstanding the consultation information provided and the steps described above, no comments or input were received on the Coastal Waters OEMP from Greenpeace.

Summary of response by Relevant Person	Assessment of merits	Santos' Response Statement
No response was received from Greenpeace.	Santos considers it has provided sufficient information and a reasonable period of time for consultation. Santos considers Section 25 consultation requirements to have been met.	No response required.
Jubilee Australia Research Centre (JARC)		
Summary of consultation effort:		

nd Barossa Production Operations	EP. [Con-3787]
on the consultation process and deta	ils of how to contact
ddition to the previous information a	gain being provided,
Santos advised that the information or the Environment Australia. [Con-4	in the booklet and 4013]
n had now closed for the purpose of	Santos finalising and
The Update advised that the EP had sessment. [Con- 5982]	been submitted to
The Update advised that, following a e NT-DITT for assessment. [Con-60	a request from 36]
	OEMP reference
	Not applicable.
perations EP. [Con-3787]	
perations EP. [Con-3787] n the consultation process and deta	ils of how to contact
perations EP. [Con-3787] n the consultation process and deta prmation again being provided, Sant	ils of how to contact os provided information
perations EP. [Con-3787] n the consultation process and deta prmation again being provided, Sant nformation in the booklet and factsh	ils of how to contact os provided information eet had been updated to
perations EP. [Con-3787] n the consultation process and deta ormation again being provided, Sant nformation in the booklet and factsh ng and submitting environment plan	ils of how to contact os provided information eet had been updated to is for these activities to
perations EP. [Con-3787] n the consultation process and deta prmation again being provided, Sant nformation in the booklet and factsh ng and submitting environment plan The Update advised that the EP had sessment. [Con- 5982]	ils of how to contact os provided information eet had been updated to is for these activities to d been submitted to
perations EP. [Con-3787] n the consultation process and deta ormation again being provided, Sant nformation in the booklet and factsh ng and submitting environment plan The Update advised that the EP had sessment. [Con- 5982] The Update advised that, following a e NT-DITT for assessment. [Con-60	ils of how to contact os provided information eet had been updated to is for these activities to d been submitted to a request from 36]
perations EP. [Con-3787] n the consultation process and deta ormation again being provided, Sant nformation in the booklet and factsh ng and submitting environment plan The Update advised that the EP had sessment. [Con- 5982] The Update advised that, following a e NT-DITT for assessment. [Con-60	ils of how to contact os provided information eet had been updated to is for these activities to d been submitted to a request from 36]
perations EP. [Con-3787] n the consultation process and deta ormation again being provided, Sant nformation in the booklet and factsh ng and submitting environment plan The Update advised that the EP had sessment. [Con- 5982] The Update advised that, following a e NT-DITT for assessment. [Con-60	ils of how to contact os provided information eet had been updated to is for these activities to d been submitted to a request from 36] OEMP reference Not applicable.
perations EP. [Con-3787] n the consultation process and deta ormation again being provided, Sant nformation in the booklet and factsh ng and submitting environment plan The Update advised that the EP had sessment. [Con- 5982] The Update advised that, following a e NT-DITT for assessment. [Con-60	ils of how to contact os provided information eet had been updated to is for these activities to d been submitted to a request from 36] OEMP reference Not applicable.

- On 9 February 2024 Santos emailed JARC to advise it of preliminary consultation regarding proposed activities for consultation to be managed under the Coastal Waters OEMP and Barossa Production Operations EP. [Con-3787]
- The email advised that Santos was seeking information to better understand:
  - o if you are from a government Department or agency, how the proposed Production Operations activities may be relevant to your department or agency; and
  - what (if any) functions, interests or activities you or your organisation have that may be affected by the proposed Production Operations activities. 0
- The email included information on the regulatory process for the activities in Commonwealth and NT jurisdictions and links to a Santos information booklet on the proposed activities and a NOPSEMA brochure on the consultation process and details of how to contact Santos to register as a Relevant Person. The email stated that the consultation phase would commence on 11 March 2024 and close on 9 April 2024.
- On 13 March 2024 Santos emailed JARC further to the previous correspondence, to advise that it had commenced the consultation phase which would run until 9 April 2024. In addition to the previous information again being provided, Santos provided information on Relevant Persons' entitlements under the regulatory processes, details of how to provide feedback and a reminder of the closing date for consultation. [Con-3794]
- On 8 May 2024, Santos emailed JARC further to previous correspondence to advise it had extended the consultation period until 22 May 2024. In providing this extension of time, Santos advised that the information in the booklet and factsheet had been updated to account for an additional risk associated with the proposed activity. Santos confirmed that consultation will close on the revised date unless Santos hears otherwise from JARC. [Con-4025]
- On 10 July 2024 Santos emailed JARC to advise the consultation period had been completed. Santos advised JARC that it considered consultation now closed for the purpose of Santos finalising and submitting environment plans for these activities to government regulators for assessment. [Con-5130]
- On 10 September 2024 Jubilee emailed Santos stating a transboundary environmental impact assessment (EIA) for the Barossa Gas Project regarding the Arafura and Timor Seas (ATS), is needed as part of the consultation process for the Productions Operations EP. [Con-5643]
- On 20 September 2024, Santos wrote to Jubilee in response to Jubilee's letter of 10 September 2024. Santos explained the Australian regulatory framework of the Coastal Waters OEMP. and advised Jubilee that the international regulatory framework of the Arafura and Timor Seas is outside the scope of consultation for this activity. [Con-5645]
- On 15 October 2024 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that the EP had been submitted to NOPSEMA for assessment, provided a link to the full EP on the NOPSEMA website and Santos' contact details for any queries. The Update also stated that the OEMP would be submitted to the NT-DITT for assessment. [Con- 5982]
- On 23 January 2025 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that, following a request from NOPSEMA to provide further information, Santos had re-submitted the EP with some amendments in December 2024 for further assessment. The Update also re-stated that the OEMP would be submitted to the NT-DITT for assessment. [Con-6036]

Summary of response by Relevant Person	Assessment of merits	Santos' Response Statement	OEMP reference
Jubilee correspondence to Santos on 10 September 2024	Santos has considered the matters raised by Jubilee in the OEMP and provided a response to Jubilee. The international regulatory framework of the Arafura and Timor Seas is outside the scope of consultation for the	Santos correspondence to Jubilee on 20 September 2024 Thank you for your letter of 10 September 2024 and your organisation's	Not applicable.
September 2024 Jubilee provided information on regulatory and environmental developments concerning the Arafura and Timor Seas. It stated that the Barossa Gas Project warrants the need for a transboundary environmental impact assessment.	The international regulatory framework of the Arafura and Timor Seas is outside the scope of consultation for the GEP Coastal Waters Operations activity.	Thank you for your letter of 10 September 2024 and your organisation's interest in the Barossa Gas Project. Santos has considered the matters you have raised. The Barossa Development Offshore Project Proposal was accepted by the National Offshore Petroleum Safety and Environmental Management Authority (NOPSEMA) in March 2018. Santos complies with all Australian and international laws to the extent that they are applicable to the Barossa Gas Project. The Barossa Production Operations Environment Plan will be submitted to NOPSEMA for assessment in accordance with the Offshore Petroleum and Greenhouse Gas Storage (Environment) Regulations 2023. The EP will detail the environmental impacts and risks associated with Barossa Production Operations activity and demonstrate how these will be reduced to as low as reasonably practicable and to an acceptable level through implementation of a suite of control measures. The international regulatory framework of the Arafura and Timor Seas is outside the scope of consultation for the Barossa Production Operations activity. If Jubilee Australia believes transboundary environmental impact	
		with the relevant sovereign governments.	

#### Keep Top End Coasts Healthy

Summary of consultation effort:

- On 9 February 2024 Santos emailed Keep Top End Coasts Healthy to advise it of preliminary consultation regarding proposed activities for consultation to be managed under the Coastal Waters OEMP and Barossa Production Operations EP. [Con-3787]
- The email advised that Santos was seeking information to better understand:
  - if you are from a government Department or agency, how the proposed Production Operations activities may be relevant to your department or agency; and 0
  - what (if any) functions, interests or activities you or your organisation have that may be affected by the proposed Production Operations activities.
- The email included information on the regulatory process for the activities in Commonwealth and NT jurisdictions and links to a Santos information booklet on the proposed activities and a NOPSEMA brochure on the consultation process and details of how to contact Santos to register as a Relevant Person. The email stated that the consultation phase would commence on 11 March 2024 and close on 9 April 2024.
- On 11 March 2024 Santos emailed Keep Top End Coasts Healthy further to the previous correspondence, to advise that it had commenced the consultation phase which would run until 9 April 2024. In addition to the previous information again being provided, Santos provided information on Relevant Persons' entitlements under the regulatory processes, details of how to provide feedback and a reminder of the closing date for consultation. [Con-3793]
- On 4 April 2024 Santos called Keep Top End Coasts Healthy regarding consultation for Production Operations activities and remind it of the 9 April deadline to provide comments.
- On 9 May 2024, Santos emailed Keep Top End Coasts Healthy further to previous correspondence to advise it had extended the consultation period until 23 May 2024. In providing this extension of time, Santos advised that the information in the booklet and factsheet had been updated to account for an additional risk associated with the proposed activity. Santos confirmed that consultation will close on the revised date unless Santos hears otherwise from Keep Top End Coasts Healthy. [Con-4033]
- On 10 July 2024 Santos emailed Keep Top End Coasts Healthy to advise the consultation period had been completed. Santos advised Keep Top End Coasts Healthy that it considered consultation had now closed for the purpose of Santos finalising and submitting environment plans for these activities to government regulators for assessment. [Con-5130]

- On 15 October 2024 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. T NOPSEMA for assessment, provided a link to the full EP on the NOPSEMA website and Santos' contact details for any queries. The Update also stated that the OEMP would be submitted to the NT-DITT for ass
- On 23 January 2025 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP.
   NOPSEMA to provide further information, Santos had re-submitted the EP with some amendments in December 2024 for further assessment. The Update also re-stated that the OEMP would be submitted to the

<ul> <li>Notwithstanding the consultation information provided and the steps described above, no comments or input were received on the Coastal Waters OEMP from Keep Top End Coasts Healthy.</li> </ul>			
Summary of response by Relevant Person         Assessment of merits         Santos' Response Statement			
No response was received from Keep Top End Coasts Healthy.	Santos considers it has provided sufficient information and a reasonable period of time for consultation. Santos considers Section 25 consultation requirements to have been met.	No response required.	

#### **Sea Turtle Foundation**

Summary of consultation effort:

- On 9 February 2024 Santos emailed Sea Turtle Foundation to advise it of preliminary consultation regarding proposed activities for consultation to be managed under the Coastal Waters OEMP and Barossa Pro
- The email advised that Santos was seeking information to better understand:
  - o if you are from a government Department or agency, how the proposed Production Operations activities may be relevant to your department or agency; and
  - what (if any) functions, interests or activities you or your organisation have that may be affected by the proposed Production Operations activities.
- The email included information on the regulatory process for the activities in Commonwealth and NT jurisdictions and links to a Santos information booklet on the proposed activities and a NOPSEMA brochure of Santos to register as a Relevant Person. The email stated that the consultation phase would commence on 11 March 2024 and close on 9 April 2024.
- On 11 March 2024 Santos emailed Sea Turtle Foundation further to the previous correspondence, to advise that it had commenced the consultation phase which would run until 9 April 2024. In addition to the previous formation on Relevant Persons' entitlements under the regulatory processes, details of how to provide feedback and a reminder of the closing date for consultation. [Con-3793]
- On 4 April 2024 Santos called the Sea Turtle Foundation regarding consultation for Production Operations activities and reminded it of the 9 April deadline to provide comments
- On 8 May 2024, Santos emailed Sea Turtle Foundation further to previous correspondence to advise it had extended the consultation period until 22 May 2024. In providing this extension of time, Santos advised updated to account for an additional risk associated with the proposed activity. Santos confirmed that consultation will close on the revised date unless Santos hears otherwise from the Sea Turtle Foundation. [Contemporation]
- On 10 July 2024 Santos emailed Sea Turtle Foundation to advise the consultation period had been completed. Santos advised Sea Turtle Foundation that it considered consultation had now closed for the purpos these activities to government regulators for assessment. [Con-5130]
- On 15 October 2024 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. 1 NOPSEMA for assessment, provided a link to the full EP on the NOPSEMA website and Santos' contact details for any queries. The Update also stated that the OEMP would be submitted to the NT-DITT for ass
- On 23 January 2025 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP.
   NOPSEMA to provide further information, Santos had re-submitted the EP with some amendments in December 2024 for further assessment. The Update also re-stated that the OEMP would be submitted to the

• Notwithstanding the consultation information provided and the steps described above, no comments or input were received on the Coastal Waters OEMP from Sea Turtle Foundation.

Summary of response by Relevant Person	Assessment of merits	Santos' Response Statement
No response was received from Sea Turtle Foundation.	Santos considers it has provided sufficient information and a reasonable period of time for consultation. Santos considers Section 25 consultation requirements to have been met.	No response required.



he Update advised that the EP had been submitted to essment. [Con- 5982]		
The Update advised that, following a request from		
	50]	
	OEMP reference	
	Not applicable.	
duction Operations EP. [Con-3787]		
n the consultation process and details of how to contact		
evious information again being provided, Santos provided		
that the information in the booklet and factsheet had been on-4026] se of Santos finalising and submitting environment plans for		
The Update advised that the EP had been submitted to sessment. [Con- 5982]		
The Update advised that, following a request from NT-DITT for assessment. [Con-6036]		
	OEMP reference	
	Not applicable.	

#### West Timor Care Foundation

Summary of consultation effort:

- On 9 February 2024 Santos emailed West Timor Care Foundation to advise it of preliminary consultation regarding proposed activities for consultation to be managed under the Coastal Waters OEMP and Barossa Production Operations EP. [Con-3787]
- The email advised that Santos was seeking information to better understand:
  - o if you are from a government Department or agency, how the proposed Production Operations activities may be relevant to your department or agency; and
  - what (if any) functions, interests or activities you or your organisation have that may be affected by the proposed Production Operations activities.
- The email included information on the regulatory process for the activities in Commonwealth and NT jurisdictions and links to a Santos information booklet on the proposed activities and a NOPSEMA brochure on the consultation process and details of how to contact Santos to register as a Relevant Person. The email stated that the consultation phase would commence on 11 March 2024 and close on 9 April 2024.
- On 13 March 2024 Santos emailed West Timor Care Foundation further to the previous correspondence, to advise that it had commenced the consultation phase which would run until 9 April 2024. In addition to the previous information again being provided, Santos provided information on Relevant Persons' entitlements under the regulatory processes, details of how to provide feedback and a reminder of the closing date for consultation, [Con-3794]
- On 8 May 2024, Santos emailed West Timor Care Foundation further to previous correspondence to advise it had extended the consultation period until 22 May 2024. In providing this extension of time, Santos advised that the information in the booklet and factsheet had been updated to account for an additional risk associated with the proposed activity. Santos confirmed that consultation will close on the revised date unless Santos hears otherwise from West Timor Care Foundation. [Con-4028]
- On 10 July 2024 Santos emailed West Timor Care Foundation to advise the consultation period had been completed. Santos advised West Timor Care Foundation that it considered consultation had now closed for the purpose of Santos finalising and submitting environment plans for these activities to government regulators for assessment. [Con-5130]
- On 15 October 2024 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that the EP had been submitted to NOPSEMA for assessment, provided a link to the full EP on the NOPSEMA website and Santos' contact details for any queries. The Update also stated that the OEMP would be submitted to the NT-DITT for assessment. [Con- 5982]
- On 14 November 2024 Santos attempted to phone West Timor Care Foundation but was unable to be connected. Santos followed-up with an email seeking final confirmation that the organisation had no comments on the EP and OEMP. Santos included the previously provided information and stated any comments should be provided by 29 November 2024. [Con-5985]
- On 23 January 2025 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that, following a request from NOPSEMA to provide further information. Santos had re-submitted the EP with some amendments in December 2024 for further assessment. The Update also re-stated that the OEMP would be submitted to the NT-DITT for assessment. [Con-6036]

Notwithstanding the consultation information provided and the steps described above, no comments or input were received on the Coastal Waters OEMP from West Timor Care Foundation.

Summary of response by Relevant Person	Assessment of merits	Santos' Response Statement
No response was received from West Timor Care Foundation.	The EMBA for the Coastal Waters OEMP does not extend beyond Commonwealth Waters off the Northern Territory.	No response required.
	There are therefore no impacts from planned GEP operations activities in NT Coastal Waters that may affect the functions, interests or activities of any international persons.	
	The worst-case credible spill scenario modelled for the EMBA for the Coastal Waters OEMP also shows there are no impacts from unplanned activities that may affect the functions, interests or activities of any international persons.	
	Santos considers it has provided sufficient information and a reasonable period of time for consultation. Santos considers Section 25 consultation requirements to have been met.	
		1

#### **Wilderness Society**

Summary of consultation effort:

- On 9 February 2024 Santos emailed the Wilderness Society to advise it of preliminary consultation regarding proposed activities for consultation to be managed under the Coastal Waters OEMP and Barossa Pro
- The email advised that Santos was seeking information to better understand:
  - o if you are from a government Department or agency, how the proposed Production Operations activities may be relevant to your department or agency; and
    - what (if any) functions, interests or activities you or your organisation have that may be affected by the proposed Production Operations activities.
- The email included information on the regulatory process for the activities in Commonwealth and NT jurisdictions and links to a Santos information booklet on the proposed activities and a NOPSEMA brochure of Santos to register as a Relevant Person. The email stated that the consultation phase would commence on 11 March 2024 and close on 9 April 2024.
- On 11 March 2024 Santos emailed the Wilderness Society further to the previous correspondence, to advise that it had commenced the consultation phase which would run until 9 April 2024. In addition to the pr information on Relevant Persons' entitlements under the regulatory processes, details of how to provide feedback and a reminder of the closing date for consultation. [Con-3793]
- On 4 April 2024 Santos called the Wilderness Society and left a voicemail message regarding consultation for Production Operations activities.
- On 2 May 2024, Santos emailed Wilderness Society further to previous correspondence to advise it had extended the consultation period until 16 May 2024. In providing this extension of time, Santos advised that • updated to account for an additional risk associated with the proposed activity. Santos confirmed that consultation will close on the revised date unless Santos hears otherwise from Wilderness Society. [Con-401
- On 10 May 2024, the Wilderness Society emailed Santos to advise it is a Relevant Person for the purposes of the Offshore Petroleum and Greenhouse Gas Storage Act 2006, however it will not provide feedback requests to be kept updated as this activity progresses and advised it may seek to provide feedback into the future. [Con-4034] The Wilderness Society was added to the distribution list for the Barossa Quarterly
- On 10 July 2024 Santos emailed Wilderness Society to advise the consultation period had been completed. Santos advised it considered that consultation had now closed for the purpose of Santos finalising and government regulators for assessment. [Con-5130]
- On 15 October 2024 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. NOPSEMA for assessment, provided a link to the full EP on the NOPSEMA website and Santos' contact details for any queries. The Update also stated that the OEMP would be submitted to the NT-DITT for ass
- On 23 January 2025 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. NOPSEMA to provide further information, Santos had re-submitted the EP with some amendments in December 2024 for further assessment. The Update also re-stated that the OEMP would be submitted to the
- Notwithstanding the consultation information provided and the steps described above, no additional comments or input were received on the Coastal Waters OEMP from Wilderness Society.

	OEMP reference	
	Not applicable.	
oduction Operations EP. [Con-3787]	]	
n the consultation process and details of how to contact		
evious information again being provided, Santos provided		
at the information in the booklet and 4]	factsheet had been	
on this activity at this time. The Wilderness Society Update.		
I submitting environment plans for th	nese activities to	
The Lindete advised that the ED had	boon submitted to	
sessment. [Con- 5982]		
The Update advised that, following a	a request from	
e NI-DIIT for assessment. [Con-60]	36]	

Summary of response by Re	levant Person	Assessment of merits		Santos' Response Statement	OEMP reference
Wilderness Society advised it of Relevant Person for the purpo <i>Petroleum and Greenhouse G</i> however it will not provide feed at this time.	considered itself a ses of the <i>Offshore</i> <i>as Storage Act 2006</i> , lback on this activity	red itself a he <i>Offshore</i> age <i>Act 2006</i> , h this activity		No response required.	Not applicable.
Wilderness Society requested that Santos keep it updated as the activity progressed as it may wish to provide feedback in the future.	Santos provides quarterly project updates to the Wilderness Society.	No response required.	Quarterly notifications will be sent to organisations (refer to Section 8-12).		
WorldFich Timer Locto					

### Summary of consultation effort:

On 9 February 2024 Santos emailed WorldFish Timor-Leste to advise it of preliminary consultation regarding proposed activities for consultation to be managed under the Coastal Waters OEMP and Barossa Pro

- The email advised that Santos was seeking information to better understand:
  - o if you are from a government Department or agency, how the proposed Production Operations activities may be relevant to your department or agency; and
  - what (if any) functions, interests or activities you or your organisation have that may be affected by the proposed Production Operations activities.
- The email included information on the regulatory process for the activities in Commonwealth and NT jurisdictions and links to a Santos information booklet on the proposed activities and a NOPSEMA brochure of Santos to register as a Relevant Person. The email stated that the consultation phase would commence on 11 March 2024 and close on 9 April 2024.
- On 13 March 2024 Santos emailed WorldFish Timor-Leste further to the previous correspondence, to advise that it had commenced the consultation phase which would run until 9 April 2024. In addition to the pr
  information on Relevant Persons' entitlements under the regulatory processes, details of how to provide feedback and a reminder of the closing date for consultation. [Con-3794]
- On 4 April 2024 Santos called WorldFish Timor-Leste regarding consultation for Production Operations activities but was unable to leave a message.
- On 2 May 2024, Santos emailed WorldFish Timor-Leste further to previous correspondence to advise it had extended the consultation period until 16 May 2024. In providing this extension of time, Santos advised updated to account for an additional risk associated with the proposed activity. Santos confirmed that consultation will close on the revised date unless Santos hears otherwise from WorldFish Timor-Leste. [Con-
- On 10 July 2024 Santos emailed WorldFish Timor-Leste to advise the consultation period had been completed. Santos advised WorldFish Timor-Leste that it considered that consultation had now closed for the plans for these activities to government regulators for assessment. [Con-5130]
- On 15 October 2024 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. T NOPSEMA for assessment, provided a link to the full EP on the NOPSEMA website and Santos' contact details for any gueries. The Update also stated that the OEMP would be submitted to the NT-DITT for ass
- On 14 November 2024 Santos phoned WorldFish seeking final confirmation that the organisation had no comments on the EP and OEMP. The representative confirmed they would not be providing any commen
- On 23 January 2025 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP.
   NOPSEMA to provide further information, Santos had re-submitted the EP with some amendments in December 2024 for further assessment. The Update also re-stated that the OEMP would be submitted to the

#### Notwithstanding the consultation information provided and the steps described above, no additional comments or input were received on the Coastal Waters OEMP from WorldFish Timor-Leste.

Summary of response by Relevant Person	Assessment of merits	Santos' Response Statement
No response was received from WorldFish Timor- Leste.	The EMBA for the Coastal Waters OEMP does not extend beyond Commonwealth Waters off the Northern Territory.	No response required.
	There are therefore no impacts from planned GEP operations activities in NT Coastal Waters that may affect the functions, interests or activities of any international persons.	
	The worst-case credible spill scenario modelled for the EMBA for the Coastal Waters OEMP also shows there are no impacts from unplanned activities that may affect the functions, interests or activities of any international persons.	
	Santos considers it has provided sufficient information and a reasonable period of time for consultation.	
	Santos considers Section 25 consultation requirements to have been met.	

#### World Wildlife Fund (WWF)

Summary of consultation effort:

- On 9 February 2024 Santos emailed WWF to advise it of preliminary consultation regarding proposed activities for consultation to be managed under the Coastal Waters OEMP and Barossa Production Operatio
- The email advised that Santos was seeking information to better understand:
  - o if you are from a government Department or agency, how the proposed Production Operations activities may be relevant to your department or agency; and
  - what (if any) functions, interests or activities you or your organisation have that may be affected by the proposed Production Operations activities.
- The email included information on the regulatory process for the activities in Commonwealth and NT jurisdictions and links to a Santos information booklet on the proposed activities and a NOPSEMA brochure of Santos to register as a Relevant Person. The email stated that the consultation phase would commence on 11 March 2024 and close on 9 April 2024.
- On 11 March 2024 Santos emailed WWF further to the previous correspondence, to advise that it had commenced the consultation phase which would run until 9 April 2024. In addition to the previous information Relevant Persons' entitlements under the regulatory processes, details of how to provide feedback and a reminder of the closing date for consultation. [Con-3793]
- On 4 April 2024 Santos phoned WWF regarding consultation for Production Operations activities and spoke to a team member who confirmed that the emails previously sent had been provided to its marine team.



duction Operations EP. [Con-3787]		
n the consultation process and deta	ils of how to contact	
revious information again being prov	rided, Santos provided	
that the information in the booklet and factsheet had been		
purpose of Santos finalising and sub	omitting environment	
The Update advised that the EP hac sessment. [Con- 5982]	been submitted to	
ts [Con-5986]		
The Update advised that, following a request from NT-DITT for assessment. [Con-6036]		
	50]	
	OEMB reference	
	OEMP reference	
	OEMP reference Not applicable.	
ns EP. [Con-3787]	OEMP reference Not applicable.	
ns EP. [Con-3787]	OEMP reference Not applicable.	
ns EP. [Con-3787]	OEMP reference Not applicable.	
ns EP. [Con-3787]	OEMP reference Not applicable.	

- On 6 May 2024, Santos emailed WWF further to previous correspondence to advise it had extended the consultation period until 20 May 2024. In providing this extension of time, Santos advised that the information in the booklet and factsheet had been updated to account for an additional risk associated with the proposed activity. Santos confirmed that consultation will close on the revised date unless Santos hears otherwise from WWF. [Con-4018]
- On 10 July 2024 Santos emailed WWF to advise the consultation period had been completed. Santos advised WWF that it considered consultation had now closed for the purpose of Santos finalising and submitting environment plans for these activities to government regulators for assessment. [Con-5130]
- On 15 October 2024 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that the EP had been submitted to NOPSEMA for assessment, provided a link to the full EP on the NOPSEMA website and Santos' contact details for any queries. The Update also stated that the OEMP would be submitted to the NT-DITT for assessment. [Con- 5982]
- On 23 January 2025 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that, following a request from NOPSEMA to provide further information. Santos had re-submitted the EP with some amendments in December 2024 for further assessment. The Update also re-stated that the OEMP would be submitted to the NT-DITT for assessment. [Con-6036]

Notwithstanding the consultation information provided and the steps described above, no comments or input were received on the Coastal Waters OEMP from WWF.

Summary of response by Relevant Person	Assessment of merits	Santos' Response Statement	OEMP reference
No response was received from WWF.	Santos considers it has provided sufficient information and a reasonable period of time for consultation.	No response required.	Not applicable.
	Santos considers Section 25 consultation requirements to have been met.		

#### **Environment Centre NT (ECNT)**

#### Summary of consultation effort:

- On 9 February 2024 Santos emailed the ECNT to advise it of preliminary consultation regarding proposed activities for consultation to be managed under the Coastal Waters OEMP and Barossa Production Operations EP. [Con-3787] •
- The email advised that Santos was seeking information to better understand:
- if you are from a government Department or agency, how the proposed Barossa Production Operations activities may be relevant to your department or agency; and
- what (if any) functions, interests or activities you or your organisation have that may be affected by the proposed Barossa Production Operations activities.
- The email included information on the regulatory process for the activities in Commonwealth and NT jurisdictions and links to a Santos information booklet on the proposed activities and a NOPSEMA brochure on the consultation process and details of how to contact Santos to register as a Relevant Person. The email stated that the consultation phase would commence on 11 March 2024 and close on 9 April 2024.
- On 8 March 2024 the ECNT wrote to Santos explaining why it is a Relevant Person for the Coastal Waters OEMP., requesting confirmation that Santos considered the ECNT as a Relevant Person, and requesting a meeting with Santos to discuss information gaps relating to the Coastal Waters OEMP and Barossa Production Operations EP. [Con-4002]
- On 11 March 2024 Santos emailed the ECNT to advise that the formal consultation period had commenced. Santos provided links to the Barossa Production Operations Activity Information Booklet, the Barossa Production Operations section of Santos' website, and NOPSEMA's Brochure: 'Consultation on offshore petroleum environment plans - Information for the Community'. Santos advised that consultation period closes on 9 April 2024. [Con-3793]
- On 25 March 2024 the ECNT wrote to Santos requesting further information on a range of topics related to the Barossa Production Operations EP. The ECNT also raised concerns regarding the consultation period timeline being too short for it to make an informed assessment of the Activity on its functions, interests or activities. [Con-4003]
- On 28 March 2024 Santos wrote to the ECNT in response to the ECNT's letter of 8 March 2024. Santos provided links to the consultation materials and further details regarding the purpose and process of consultation, including the type of information being sought from the ECNT. Santos confirmed that it is willing to meet with the ECNT, and that it is willing to accommodate reasonable requests from the ECNT to consult in an alternative manner. [Con-4004]
- On 9 April 2024 the ECNT wrote to Santos and raised concerns about the consultation process to date, including that it had not been provided with sufficient information to make an informed assessment of the possible consequences of the Activity on its functions, interests, or activities. The ECNT also raised concerns about a range of risks and impacts associated with the OEMP and EP. The ECNT advised that it had commissioned expert reports to assist in identifying the impacts of the Activity on the ECNT's functions, interests or activities, and stated that it would detail a timeline for the provision of these expert reports once it receives the information it requires from Santos. [Con-4008]
- On 30 April 2024 Santos wrote to the ECNT in response to the ECNT's letter of 25 March 2024. Santos responded to each of the ECNT's information requests, as well as its concerns regarding the consultation process. Santos also provided the ECNT with notice of an update to the information booklet and factsheet. Santos requested ECNT provide its availability for a meeting. [Con-4009]
- On 2 May 2024, the ECNT emailed Santos in relation to its availability for a meeting. [Con-4010]
- On 3 May 2024, Santos emailed the ECNT confirming Santos was available to meet in Darwin on 20 May 2024. Santos requested the ECNT circulate an agenda by 13 May 2024 setting out the issues which the ECNT would like to discuss at the meeting. [Con-4017]
- On 7 May 2024, the ECNT emailed Santos to confirm the ECNT were available to meet at 2pm on 20 May 2024, and that it would circulate a meeting agenda by 13 May 2024. [Con-4020]
- On 13 May 2024, the ECNT provided Santos with a list of meeting attendees and an agenda with two items: ECNT's concerns regarding consultation process to date; an outline of the relevance of key information gaps to ECNT's functions, interests, and activities. The ECNT also noted that Santos has not responded to some of the ECNT's correspondence, and the ECNT advised that it proposes to engage technical experts to assist in its consideration of the how the Activity impacts its functions, interests and activities once it receives a substantive reply to its letter of 9 April 2024. [Con-4036]
- On 14 May 2024 Santos emailed ECNT to thank it for the agenda and list of attendees and advised that it will meet ECNT on Monday, 20 May 2024. In the email Santos also responded to ECNT's letter of 9 April 2024, responding to each of the ECNT's concerns, as well as its concerns regarding the consultation process. [Con-4030]
- On 20 May 2024 Santos and the ECNT met at Santos' Darwin Office located at Charles Darwin Centre. The meeting ran for approximately 15 minutes. The ECNT advised that it was still considering responses from Santos' letter dated 14 May 2024. Santos answered the ECNT's guestions and provided the ECNT with another opportunity to ask any further guestions about the Barossa Production Operations EP. No further guestions were asked. [Con-5213]
- On 28 May 2024 the ECNT wrote to Santos advising that it is still preparing a response to Santos's letter dated 14 May 2024, including engaging experts to assist in its assessment of the Barossa Production Operations activity's impact on its functions, interests or activities. The ECNT also noted that no further information was received at the meeting held on Monday 20 May 2024, and raised concerns regarding the consultation process. [Con-4214]
- On 12 June 2024 Santos wrote to the ECNT to acknowledge its letter of 28 May 2024 and requested that if the ECNT wishes to provide any additional input for the OEMP and EP (including, if it considers that there are additional measures to be included) Santos requires this by no later than Thursday, 20 June 2024. [Con-5033]
- On 20 June 2024 the ECNT wrote to Santos to provide further input on the OEMP and EP, which primarily focused on GHG emissions related to the Barossa Production Operations EP. In its letter, the ECNT asserted that Santos has not consulted with the ECNT in a meaningful way and has not discharged its regulatory consultation obligations. The ECNT raised concerns that Santos has not answered all its guestions and is withholding information from the ECNT. [Con-5035]
- On 7 August 2024. Santos wrote to ECNT in response to ECNT's letter of 20 June 2024. Santos responded to the matters raised by ECNT. Santos thanked ECNT for its comments and submissions in respect of the risks, impacts, and potential controls in relation to the activity. Santos advised ECNT that Santos was finalising the EP for submission in coming weeks. [Con- 5282]
- On 15 October 2024 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that the EP had been submitted to NOPSEMA for assessment, provided a link to the full EP on the NOPSEMA website and Santos' contact details for any queries. The Update also stated that the OEMP would be submitted to the NT-DITT for assessment. [Con- 5982]
- On 25 November 2024 ECNT wrote to Santos to again express concerns over the consultation process undertaken by Santos and to raise concerns about inconsistencies in emissions estimates provided and Santos' compliance with the Safeguard Mechanism [Con-60131
- On 27 November 2024 NOPSEMA provided Santos with a letter that ECNT had provided separately to NOPSEMA on 22 October 2024 concerning the Barossa Production Operations EP.
- On 20 December 2024 Santos wrote to ECNT in response to ECNT's correspondence of 25 November 2024. In the letter Santos also addressed comments and assertions made by ECNT in the letter it had separately sent to NOPSEMA on 22 October 2024. [Con-6032]



- On 23 January 2025 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that, following a request from NOPSEMA to provide further information, Santos had re-submitted the EP with some amendments in December 2024 for further assessment. The Update also re-stated that the OEMP would be submitted to the NT-DITT for assessment. [Con-6036]
- On 24 January 2025 the ECNT wrote to Santos in response to Santos' letter of 20 December 2024. The ECNT rejected the matters raised in Santos' letter of 20 December 2024, reiterated its concerns regarding consultation to date, GHG emissions and compliance with the Safeguard Mechanism, and requested information regarding the DLNG facility. The ECNT enclosed a letter dated 4 December 2024 sent from ECNT to NOPSEMA raising ECNT's concerns about the DLNG facility (to which Santos was not copied), as well as a bundle of documents received under FOI relating to that facility. [Con-6037]
- On 6 March 2025 Santos wrote to ECNT in response to the matters raised by ECNT in its correspondence of 24 January 2025. [Con-6039]

In NOPSEMA's Request for Further Written Information (RFFWI) dated 24 March 2025 on the Production Operations EP, NOPSEMA enclosed a letter sent from ECNT to NOPSEMA dated 13 March 2025 regarding the EP and noted that the correspondence was provided to Santos in keeping with procedural fairness 'as it may contain information that is relevant to the EP'. Santos has considered this letter, including aspects which may be assumed to relate to the OEMP, and is satisfied with the content of the OEMP, including Santos' assessments of the merits of the issues which have been previously raised by the ECNT with Santos during consultation. The ECNT's letter to NOPSEMA is not addressed in this consultation report because it does not fall within s 25 or 24(b) of the Regulations.



The Update advised that, following a request from e NT-DITT for assessment. [Con-6036] g consultation to date, GHG emissions and compliance with facility (to which Santos was not copied), as well as a

Summary of response by Relevant Person	Assessment of merits	Santos' Response Statement
ECNT correspondence to Santo	bs on 8 March 2024	·
ECNT identified itself as a Relevant Person and requested	Santos has considered the matters raised by ECNT in the OEMP, where relevant. Santos met with the ECNT on 20 May 2024.	Santos' correspondence to ECNT on 28 March 2024 in response
a meeting with Santos to discuss information gaps in the documentation provided by Santos.		Thank you for your correspondence of 8 March 2024 and 25 Marc interest in the Barossa Project. We set out below our response to will respond separately to your correspondence of 25 March 2024
		Consultation materials
		We confirm that we provided links to the following in our emails day 2024:
		<ol> <li>a booklet containing information about the activities proposed Environment Plan (EP) (Commonwealth waters) and the Ope Plan (EMP) (Northern Territory waters) (Information Booklet</li> <li>an information booklet published by National Offshore Petrol Management Authority (NOPSEMA) titled 'Consultation on o plans' (NOPSEMA information booklet) that provides informat consultation and contains guidance for potential relevant per- offshore petroleum activities.</li> <li>The Information Booklet provides a comprehensive overview of th reader to make an informed assessment of the possible consequi- activities on their functions, interests or activities.</li> </ol>
		Each of these documents is also available on Santos' website, to about the Barossa Gas Project and the proposed activities: <u>https://www.santos.com/barossa/production-operations/</u>
		We understand from your correspondence that the ECNT is awar the Information Booklet.
		For convenience, we reattach a link to the NOPSEMA information
		Santos' consultation process
		Santos has been consulting on the EP and EMP since 11 March
		The commencement of Santos' formal consultation period followe
		1. correspondence on 9 February 2024 to identified potenti ECNT:
		<ul> <li>b. requesting that they contact Santos at the earliest oppor a relevant person;</li> </ul>
		<ul> <li>c. inviting them to direct Santos to any additional persons v should consult: and</li> </ul>
		<ol> <li>extensive consultation on previous environmental approsince 2016, concerning construction and installation actiproject, including with the ECNT.</li> <li>Santos will consult with the ECNT as part of its comprehensive control the EP and EMP. That consultation campaign will run until 9 April email to ECNT dated 9 February 2024. This consultation timefrant regard to Santos' regulatory obligations and scheduling in respect Project.</li> </ol>
		To ensure that consultation is meaningful and transparent, Santo the purpose of consultation and the information sought during con
		Purpose of consultation
		As is set out in the NOPSEMA information booklet, the purpose of ascertain, understand and assess:
		<ol> <li>the values and sensitivities of the environment that may activities;</li> </ol>
		2. the potential environmental impacts and risks of the prop
		proposed activities to as low as reasonably practicable a

<sup>43</sup> <u>https://www.nopsema.gov.au/sites/default/files/documents/Consultation%20in%20the%20course%20of%20preparing%20an%20Environment%20Plan%20guideline\_1.pdf</u>



ponse to ECNT's letter of 8 March	Not Applicable
larch 2024 and your organisation's to your letter of 8 March 2024. Santos 024.	
s dated 9 February 2024 and 13 March	
sed under the Production Operations Operations Environmental Management <b>klet</b> ); and roleum Safety and Environmental n offshore petroleum environment mation regarding the purpose of persons in respect of consultation on	
f the proposed activities to allow the quences (if any) of the proposed	
together with additional information	
vare of this webpage and has reviewed	
tion booklet. <sup>43</sup>	
ch 2024.	
wed:	
entially relevant persons, including the	
//P;	
portunity if they considered they may be	
is with whom they considered Santos	
provals for the Barossa Gas Project activities associated with the same	
e consultation campaign in respect of pril 2024, as previously advised in our rame is fair and reasonable having ect of the broader Barossa Gas	
ntos has included below a summary of consultation.	
e of consultation includes to further	
ay be affected by the proposed	
roposed activities; and	
mental impacts and risks of the e and an acceptable level.	

Consultation provides an opportunity to communicate to Santos risks or impacts to it, including information that Santos would out received will be considered by Santos in the preparation of the its assessment of the EP and EMP. If you consider that you hav preparation of the EP and EMP, please communicate this to Sa Santos has an opportunity to consider this information and ensu that is not already known to, or addressed by Santos, is reflected

#### Information sought

Having regard to the purpose of consultation and consistent with Regulations, Santos is seeking information through consultation

- values or sensitivities of the environment that may be (noting that 'environment' includes social, economic all
- 2. potential impacts to the environment;
- 3. potential risks to the environment;
- particular measures that the ECNT thinks Santos shou ECNT's consultation input; and

5. other persons or organisations with whom the ECNT c The information you provide will be used for the development of

- an Environment Plan for the activity in Commonwealth NOPSEMA; and
- an Operations Environmental Management Plan for the waters, which will be assessed by the Energy Division Department of Industry, Tourism and Trade (DITT).

The information you provide will be included in documentation s assessment. This will include Santos' assessment of, and respo

Santos will handle your information in accordance with our <u>Offs.</u> <u>Territory Consultation Privacy Policy</u>. You may request that par published in the EP. If requested, Santos will include your inforr not be published on NOPSEMA's website.

#### Next steps

Santos will separately respond to your letter of 25 March 2024.

Following that response, Santos' consultation team will make the ECNT in person in Darwin or via Microsoft Teams.

In the meantime, to the extent that the ECNT has any informatic outlined above), Santos requests that the ECNT provide this inf possible.

As identified above, the consultation period in respect of the EP Santos will endeavour to respond to your letter of 25 March 202 to facilitate further consultation during this period. Santos reque information sought and make itself available to meet with Santo consultation period.

This approach to, including the period for consultation is approp Santos' regulatory obligations, Santos' previous consultation in the ECNT and to the ECNT's understanding of the Barossa Gas involvement in the Stop Barossa Gas campaign.<sup>44</sup>

Santos will accommodate reasonable requests by the ECNT to

s any knowledge of the environment, or therwise not be aware of. Information EP and EMP, and by the regulator in ve information which should inform the antos as soon as possible so that ure that any information you may have ed in the EP and/or EMP.	
th the requirements of s 25 of the n as to any:	
affected by the proposed activities nd cultural features);	
uld consider adopting because of the	
considers Santos should consult. If the following documents:	
n waters, which will be assessed by	
ne activity in Northern Territory coastal within the Northern Territory	
submitted to NOPSEMA and DITT for onse to, the information you provide.	
hore Western Australia and Northern rticular information you provide not be mation in a separate report which will	
nemselves available to meet with the	
ion of the type sought by Santos (as formation to Santos as soon as	
P and the EMP will end on 9 April 2024. 24 as expeditiously as possible in order ests that the ECNT provide the ps' consultation team within this	
oriate and reasonable having regard to respect of the Barossa Project with s Project, including through its	
consult in an alternative manner.	

<sup>44</sup> https://stopbarossagas.org/about-us/

ECNT's correspondence to Santos of 25 March 2024				
<ul> <li>ECNT's correspondence to Sam</li> <li>A number of concerns were raised in relation to Santos' consultation process, including that:</li> <li>the consultation deadline of 9 April 2024 set by Santos would not provide a reasonable period for consultation for the ECNT and other Relevant Persons;</li> <li>the Information Booklet does not provide sufficient information for the ECNT to make an informed assessment of the possible consequences of the Production Operations activity on its functions, interests or activities; and</li> <li>the consultation in the course of preparing an environment plan".</li> <li>The ECNT sought confirmation that the 9 April 2024 deadline will not be imposed, and that Santos will engage with ECNT to discuss a reasonable process and timeline to occur with the ECNT</li> </ul>	tos of 25 March 2024 Santos has provided a response to ECNT and considered the matters raised by ECNT in the OEMP where relevant. Santos has provided the ECNT with sufficient information and a reasonable period to assess any possible impacts of the Activity for this OEMP on the ECNT's functions, interests and activities, and to provide input to Santos about the environment that may be affected by the Activity and/or additional proposed control measures for consideration by Santos in the course of preparing the OEMP.	<ul> <li>Santos' correspondence to ECNT on 30 April 2024 in response to ECNT's letter of 25 March 2024</li> <li>We note your comments regarding Santos' consultation process and timeframe for consultation on the Production Operations Environment Plan (EP).</li> <li>As you are aware, Santos commenced its preliminary consultation process with the ECNT in respect of the EP on 9 February 2024 and has been formally consultation process has included:</li> <li>correspondence on 9 February 2024 to identify potentially relevant persons (including the ECNT), and:</li> <li>a. outlining the then-upcoming consultation on the EP;</li> <li>b. providing links to all relevant information booklets;</li> <li>c. requesting that potentially relevant persons contact Santos at the earliest opportunity if they considered they may indeed be a relevant persons; and</li> <li>d. inviting them to direct Santos to any additional persons with whom they considered Santos should consult; and</li> <li>further correspondence on 13 March 2024:</li> <li>e. again explaining the consultation (including the consultation period);</li> <li>b. expressly inviting relevant information booklets, together with a link to NOPSEMA's brochure entitled 'Consultation on offshore petroleum environment plans: Information for the community.'</li> <li>We note that the above has followed extensive consultation (including with the ECNT) commencing in 2016 in respect of previous environmental approvals for activities associated with the Barossa Gas Project, that was presented and assessed in the Barossa Development Offshore Project Proposal (OPP). Consultation with stakeholders on the OPP occurred during 2017 and included an eight-week public comment period prior to submission of the OPP to NOPSEMA for assessment. As such, information about the Production Operations activity has been publicly available in the OPP, Santos' regulatory obligations, Santos' previous consultation in respect of the Barossa Gas Project (including the ECNT) consultation tinformation about the Pr</li></ul>	Not Applicable	
<ul> <li>In light of the concerns that the ECNT raised regarding the lack of detail in the Information Booklet, the ECNT requested drafts of the</li> </ul>	Santos has provided a response to ECNT and considered the matters raised by ECNT in the OEMP. Notwithstanding this, Santos' assessment is that the information booklet is sufficient for the ECNT to assess impacts of the Activity on its functions, interests or activities, and to provide input to Santos about the environment that may be affected by the Activity and/or additional proposed control measures for consideration by Santos in the course of prenaring the OEMP.	for the ECNT to make an informed assessment of any potential consequences of the activity on its functions, interests or activities, and encourages the ECNT to meet with Santos to consult constructively in line with the purpose of s 25 consultation. we note your request at paragraph 21 for: any drafts of the EP or any of its addenda, including specific plans, methodologies, underlying modelling, or raw data that may assist in our assessment of how the Activity relates to our functions, interests, and activities.	Not Applicable	
EP or any of its addenda, including specific plans, methodologies, underlying modelling, or raw data that may assist in its assessment of how the Activity relates to its functions, interests, and activities.		Consultation is undertaken 'in the course of preparing an environment plan' (s 25 of the Offshore Petroleum and Greenhouse Gas Storage (Environment) Regulations 2023 (Regulations)). As we advised in our letter of 28 March 2024, the purpose of consultation is to inform the preparation of the EP. Provision of a draft of the EP is not necessary in order for the ECNT to make an informed assessment of the possible consequences of activities the subject of the proposed EP on its functions, interests, or activities. Once preparation of the EP has been completed, including completion of the consultation process, Santos will submit the EP to NOPSEMA for its assessment. Once NOPSEMA has undertaken its completeness check the EP is published on its website, should the ECNT wish to view it then.		
<ul> <li>The ECNT raised concerns in relation to the information provided in the Information Booklet and requested further information regarding:</li> <li>how the control measures originally proposed in the OPP relating to FPSO processes have been adopted or not adopted for</li> </ul>	Operations at the FPSO are outside the scope of the Coastal Waters OEMP and are addressed in the Barossa Production Operations EP.	<ul> <li>[22(a)] The Environmental Performance Outcomes (EPOs) in respect of FPSO processes are set out in Table 7.1 of the OPP at this <u>link</u> [pp. 454, 457, 458]. In accepting the OPP, NOPSEMA was satisfied that the EPOs are consistent with the principles of ecologically sustainable development (see reg 5D(6)(c)(i) of the Regulations then in force).</li> <li>These EPOs will be carried through to the Production Operations EP, with EP-specific control measures and performance standards also developed and incorporated. Santos has developed the specific control measures presented in a summary format in the Production Operations information booklet which has been provided to the ECNT (see in particular the sections titled 'How will Santos manage impacts' in respect of each identified risk and impact).</li> </ul>	Environmental performance outcomes and control measures are listed in Section 8.1.2	



<ul> <li>the EP (para 22(a) of letter); and</li> <li>the flow processes and technology of the FPSO</li> </ul>		These control measures have been developed in order to achie Santos considers that they are fit for purpose. Consistent with o Santos encourages the ECNT to raise any control measures the adopt for Santos' consideration when preparing the EP for subn
(22(b)).		The EP will contain an assessment of all potentially viable contr impacts and risks, including which measures were not adopted/ Regulator/s.
		[ <b>22(b)</b> ] The Environmental Performance Outcomes (EPOs) in re in Table 7.1 of the OPP at this <u>link</u> [pp. 454, 457, 458]. In accept satisfied that the EPOs are consistent with the principles of eco- reg 5D(6)(c)(i) of the Regulations then in force).
		These EPOs will be carried through to the Production Operation measures and performance standards also developed and inco specific control measures presented in a summary format in the booklet which has been provided to the ECNT (see in particular manage impacts' in respect of each identified risk and impact).
		These control measures have been developed in order to achie Santos considers that they are fit for purpose. Consistent with o Santos encourages the ECNT to raise any control measures the adopt for Santos' consideration when preparing the EP for sub-
		The EP will contain an assessment of all potentially viable contr impacts and risks, including which measures were not adopted/ Regulator/s.
The ECNT raised concerns in relation to the GHG emission information provided in the Information Booklet and requested further information	Santos has provided a response to ECNT and considered the matters raised by ECNT are in relation to the FPSO operations. Operations at the FPSO are outside the scope of the Coastal Waters OEMP and are addressed in the Barossa Production Operations EP. However, this Activity does have Scope 1 and 3 GHG emissions, which are described in Section 6.3 along with the proposed control measures.	[22(c)] The Production Operations information booklet provides Greenhouse Gas (GHG) Management Plan. Santos notes ECN Management Plan but does not consider the ECNT requires a c assessment of the potential consequences of the activity on any (FIAs).
<ul> <li>regarding:</li> <li>Santos' GHG Management Plan (22(c));</li> </ul>	The Barossa Production Operations EP includes a description of the GHGEMP, and relevant components to this OEMP are presented in Section 8.3.2.12 of this OEMP.	The Production Operations information booklet details the poter Santos understands from previous consultation that the ECNT i and impacts.
<ul> <li>the breakdown of emissions by source (22(d));</li> <li>clarification of emissions calculations, including for</li> </ul>	informed assessment of the potential consequences of the activity on any of its functions, interests or activities. The Barossa Production Operations EP and this OEMP do include a breakdown of estimated Scope 1 emissions for the purpose of Regulator Assessment. CCS matters are not part of this Activity and have been excluded from the Barossa Production Operations EP	Please refer to Santos' letter to ECNT of 28 March 2024 for info consultation and information sought from ECNT during consulta raise any control measures that it considers may be appropriate when preparing the EP for submission to NOPSEMA, including
<ul> <li>each year of operation (22(e));</li> <li>confirmation the emissions estimates have been updated since OPP was accepted (22(f));</li> <li>discrepancies in emission</li> </ul>	and this OEMP as the Bayu Undan CCS project is subject to separate framework & approvals by a different joint venture. Santos considers that the further information requests by ECNT are not required for the ECNT to assess impacts of the Activity on its functions, interests or activities, and to provide input to Santos about the environment that may be affected by the Activity and/ or additional proposed control measures for consideration by Santos in the course of preparing the OEMP.	[22(d)] As you have identified, the Production Operations inform emissions estimates similar to that provided for the DPD SER a emissions. Please note that the DPD SER provided emissions of operations sources, while the Production Operations EP does n construction activities as these activities are not the subject of the Barossa EPs which have previously considered these emission
<ul> <li>estimates between Barossa approval documentation (22(g));</li> <li>details of the improvements made to the EPSO (22(h));</li> </ul>		In any event, Santos considers that the ECNT does not require in order to make an informed assessment of the potential conse FIAs. To the extent that the there is any potential impact on the emissions, the ECNT is able to make an informed assessment applied estimates associated with the project.
<ul> <li>how CO<sub>2</sub> will be removed from Barossa gas and amissions (22(i));</li> </ul>		[22(e)] Santos response to the ECNT in relation to this request
<ul> <li>how much CH<sub>4</sub> will leak, be vented, and be combusted at the FPSO and the total associated GHG emissions (22(j));</li> </ul>		<ul> <li>The Production Operations information booklet provide emissions. Scope 2 emissions are not applicable to the FPSO flaring represents approximately 5% of total esti contain a breakdown of estimated Scope 1 emissions assessment by the Regulator/s.</li> </ul>
<ul> <li>details of the marine fuels used for FPSO processes (22(k));</li> <li>details of FPSO design to enable CCS (22(l));</li> </ul>		In any event, Santos considers that the ECNT does not require an annual basis for each year of operation or confirmation of the will be from flaring from the FPSO, in order to make an informed consequences of the activity on any of its FIAs.
<ul> <li>GHG control measures (22(m)).</li> </ul>		To the extent that the there is any potential impact on the ECNT the ECNT is able to make an informed assessment of that impa estimates associated with the Barossa project which are set ou

eve the EPOs set out in the OPP and our 28 March 2024 letter to the ECNT, pat it considers may be appropriate to mission to NOPSEMA.	
rol measures relevant to identified /adopted, for assessment by the	
espect of FPSO processes are set out pting the OPP, NOPSEMA was plogically sustainable development (see	
ns EP, with EP-specific control orporated. Santos has developed the e Production Operations information r the sections titled 'How will Santos	
eve the EPOs set out in the OPP and our 28 March 2024 letter to the ECNT, at it considers may be appropriate to mission to NOPSEMA.	
rol measures relevant to identified /adopted, for assessment by the	
s a description of the purpose of the IT's request for a copy of the GHG copy in order to make an informed y of its functions, interests or activities	Section 6.3 assesses the impact of GHG emissions Section 8.3.2.12
ntial risks and impacts related to GHG. is familiar with these potential risks	Control measure to manage GHG emissions include:
ormation about the purpose of ation. Santos encourages the ECNT to e to adopt for Santos' consideration i in relation to GHG emissions.	BAO-CM-6.3.1 BAO-CM-6.3.9 BAO-CM-6.3.11 BAO-CM-6.3.12
mation booklet contains a breakdown of as relevant to Production Operations estimates for both construction and not provide emissions estimates for this EP and are authorised under other ns.	BAO-CM-6.3.16 BAO-CM-6.3.17 BAO-CM-6.3.18 BAO-CM-6.3.21 BAO-CM-6.3.22
e a breakdown of emissions by source equences of the activity on any of its ECNT's FIAs as a result of GHG of that impact on the basis of the total	DAG-OW-0.3.22
is reproduced as follows:	
es estimates for annual Scope 1 and 3 e Production Operations activity. timated Scope 1 emissions. The EP will (fuel, flare, vent) sources, for	
e a further breakdown of emissions on the proportion of scope 1 emissions that ad assessment of the potential	
T's FIAs as a result of GHG emissions, act on the basis of the total annual It in the information booklet.	

	<b>[22(f)]</b> <i>i.</i> The annual emissions estimate from the OPP, as it relate activities, has been updated for the preparation of the EP and is p information booklet.
	ii. Please see response to 22f(i)
	[22(g)] Please see response to 22d and 22f(i).
	[22(h)] A number of FPSO design features have been adopted th in Scope 1 operational emissions than were estimated in the acce
	<ul> <li>Pilotless low pressure (LP) flare and nitrogen (gas-free) purg</li> <li>Vapour recovery units to prevent planned flaring of low press</li> <li>Full electrification of the facility, with highly efficient combined</li> <li>Supply of process heating via waste heat recovery;</li> <li>Destruction of methane emissions in the CO<sub>2</sub> permeate streat</li> </ul>
	In any event, the adoption of these design features has been fact environmental impact and risk, as presented in the in the Product [22(i)]
	<ul> <li>i. The gas export stream sent to DLNG will contain 6% CC remainder of the CO<sub>2</sub> in the extracted gas (12%) will be the FPSO.</li> <li>ii. The FPSO has adopted two stage membrane CO2 removal associated with CO<sub>2</sub> removal and processing are account estimate provided for in the Production Operations inform Although CCS operations will be the subject of a separate environ not part of the Barossa gas project approvals process, Santos is operational as soon as possible.</li> </ul>
	The Bayu-Undan CCS development will reduce Scope 1 emission low emissions intensity project and a net-zero reservoir emission joint venture is in negotiations with the Bayu Undan CCS joint ver foundation customer of the project.
	[22(j)] The Scope 1 emissions estimate provided under the headi Production Operations information booklet is informed by an anal and includes consideration of combustion sources (planned and u unplanned) and fugitive leaks (unplanned only). Therefore, ECNT addressed in the Scope 1 emissions estimate provided in the Pro Booklet.
	As outlined above in response to 22(d), Santos considers that the assessment of any potential consequence on its FIAs by reference information booklet.
	[22(k)] Fuel gas will be used for normal operations. Marine gas o diesel oil) will be used for startup operations and as backup fuel f unavailable.
	The Scope 1 emissions estimate provided in the Production Oper for emissions from all fuel sources.
	[22(I)] Justification of the future feasibility of exporting CO2 for Corequired for the Production Operations EP.
	Nevertheless, Santos confirms that the FPSO is designed with su compression capacity for export (high pressure liquid transfer) of operations. Future export of additional CO <sub>2</sub> (up to 20%) to a CCS
	[22(m)]
	i. Equivalency of Paris Agreement Policies refers to a net zero co
	ii. Reporting against all Barossa EP commitments will be address Performance reports submitted to the Regulator.

lates to Production Operations is provided in Production Operations	
l that have resulted in >50% reduction ccepted Barossa OPP. These include:	
urge; essure vented gas; ned cycle power generation;	
ream by a thermal oxidiser.	
actored into the assessment of uction Operations information booklet.	
CO <sub>2</sub> and be vented at DLNG. The be removed from the gas and vented at	
moval technology. All emissions ounted for in the Scope 1 emissions formation booklet. ironmental approvals process, and is	
is working towards having CCS	
sions from the Barossa field, making it a ons project from 2028. The Barossa venture, with a view to being a	
ading "GHG Emissions" in the nalysis of all potential sources of CH4 Id unplanned), vented (planned and NT requests 22j(i) and 22j(ii) are Production Operations Information	
the ECNT is able to make an informed ence to the total estimates set out in the	
s oil (or equivalent fuels e.g. marine el for FPSO operations when fuel gas is	
perations information booklet accounts	
CCS is outside the scope of and not	
sufficient gas treatment and of 6% CO2 to DLNG from start of CS facility is within FPSO design limits.	
commitment.	
essed in Annual Environmental	

			1
<ul> <li>The ECNT raised concerns in relation to how Santos would comply with the Safeguard Mechanism and requested further information regarding:</li> <li>how the scope 1 emissions from the Activity fit within in carbon budgets (22(n));</li> <li>whether the Barossa project is a 'new facility' (22(o));</li> <li>how Santos will calculate Barossa's baseline (22(p));</li> <li>production variables applicable to Barossa (22(q));</li> <li>modelling of baseline emissions estimates (22(r));</li> <li>how Santos intends to avoid exceed emission situations (22(s))</li> <li>how it intends to source ACCUs (22(t);</li> <li>the percentage Barossa emissions that are expected to be reduced by CCS (22(u));</li> <li>the percentage of Barossa's estimated excess emissions expected to be reduced by CCS each year (22(v)).</li> </ul>	<ul> <li>Santos has provided a response to ECNT. Santos understands the ECNT to have been referring to emissions from the Activity under the Production Operations EP, which is addressed in that EP.</li> <li>These information requests relate to Santos' compliance with the Safeguard Mechanism. Consistent with Santos' obligations, the Production Operations EP and this OEMP demonstrate how Santos will comply with the requirements of the Safeguard Mechanism, as regulated by the Clean Energy Regulator.</li> <li>This OEMP also includes information below in relation to GHG emissions:</li> <li>Scope 1 emissions will be managed in accordance with the applicable baseline under the Safeguard Mechanism in accordance with Australia's Paris agreement targets and associated emissions is still to be finalised by the CER.</li> <li>the Activity will comply with the Safeguard Mechanism, noting that treatment of Barossa GHG emissions is still to be finalised by the CER. Implementation and enforcement of the Safeguard Mechanism is the responsibility of the CER.</li> <li>Santos routing to the ECNT that CCS is not part of the Activity under the EP. Accordingly, ECNT queries about CCS are outside the scope of this EP and therefore have not been considered further.</li> <li>Santos views that ECNT has been provided sufficient information to enable the ECNT to make an informed assessment of any potential consequences of the Activity on its functions, interests or activities</li> </ul>	<ul> <li>[22(n)] Barossa Production Operations Scope 1 emissions will be managed in accordance with the applicable baseline under the Safeguard Mechanism in accordance with Australia's Paris agreement targets and associated emissions budget. The NT emissions budget is accounted for in Australia's national emissions budget.</li> <li>[22(o)] The treatment of the Barossa project under the Safeguard Mechanism is a matter for the Clean Energy Regulator. Santos will abide by the Clean Energy Regulator's final determination.</li> <li>[22(p)] There are various options available to meet a baseline (including direct abatement and acquiring offsets, in addition to mechanisms available under the Safeguard Rules such as borrowing adjustments and multi-year monitoring periods). Santos is not in a position to indicate now whether in any given year or years (which may be decades in the future) it may apply for a borrowing adjustment, trade-exposed baseline-adjusted facility determination or multi-year monitoring period.</li> <li>Santos will be required to comply with the applicable baseline for the Barossa project in each compliance year. It is a matter for Santos to determine how it will achieve this compliance. This information is not necessary for the ECNT to make an informed assessment of the possible consequences of the activity on any of its FIAs.</li> <li>[22(q)] Please see response to 220. Santos will comply with its Safeguard compliance obligations. Management of Santos' Safeguard compliance obligations, including identification of applicable production variables under the Safeguard compliance obligations will be addressed between Santos and the CER.</li> <li>[22(q)] Please see response to 220. Santos will comply with its Safeguard compliance obligations. Management of Santos' Safeguard compliance obligations will be addressed between Santos and the Clean Energy Regulator.</li> <li>This information is not necessary for the ECNT to make an informed assessment of the possible consoulucesees of the activity on any of its</li></ul>	Section 6.3 EPO 11 BAO-CM-6.3.11
<ul> <li>The ECNT raised concerns in relation to how Santos would manage produced water at the FPSO and requested further information regarding:</li> <li>the standards Santos used for assessing the risks of produced water, including impacts on marine species (22(w));</li> <li>concentrations of hazards within the produced water and processes for treatment (22(x));</li> <li>the 6 kilometre mixing zone (22(y));</li> <li>concentration discharge limits (22(z));</li> <li>chemicals used for dehydration of gas, hydrogen-sulphide removal, and chemicals to inhibit hydrates (22(a));</li> <li>how produced water yolumes may grow (22(bb));</li> </ul>	Santos has provided a response to ECNT. Concerns related to produced water are not relevant to the Activities covered by the Coastal Waters OEMP and are addressed in the Barossa Production Operations EP.	<ul> <li>[22(w)] The Production Operations information booklet provides a summary of assessment of the impacts of produced water discharges.</li> <li>In assessing impacts and risks of produced water discharge, as presented in the Production Operations information booklet, Santos has considered relevant standards including relevant species:</li> <li>protection and environmental value protection thresholds per Australia and New Zealand Water Quality Guidelines; and</li> <li>recovery plans, conservation advice, wildlife conservation plans and management actions such as the Recovery Plan for Marine Turtles in Australia 2017–2027.</li> <li>The adequacy of the impact assessment will be assessed by the Regulator when assessing the EP.</li> <li>[22(y)] As discussed in the Production Operations information booklet, the 6-kilometre mixing zone is based on conservative modelling inputs of up to 20,000 barrels per day of produced water throughput. In contrast, produced water throughput during normal operations will be in the order of 3500 to 5000 barrels per day.</li> <li>As explained in the Production Operations information booklet, due to water depths, the absence of marine turtle biologically important areas within OA1 (the location of produced water discharge), and no significant seabed habitat in the mixing zone, marine turtles would be expected to traverse OA1 very infrequently. As a result, even if individual marine turtles did traverse the mixing zone under</li> </ul>	Not applicable.

	<ul> <li>the standard used to process and treat produced</li> </ul>		these worst case mixing conditions, they will not be exposed to the produced water for enough time for contaminants to accumulate within their body.	
	<ul> <li>water (22(cc));</li> <li>the impact of produced</li> <li>water to the Arefuse Shelf</li> </ul>		[22(z)] As outlined in the Production Operations Information booklet, the adopted concentration discharge limit is 30 mg/l over any 24-hour period.	
	and other biologically important areas (22(dd));		[22(aa)] The impacts and risks of chemicals used in the production process are addressed in the 'Produced Water' section of the Production Operations Information booklet.	
,	<ul> <li>ongoing testing of marine discharges (22(ee));</li> <li>the potential ecotoxicological impact of produced water (22(dd));</li> <li>the potential dispersion and dilution of produced water</li> </ul>		Implementation of Santos' chemical selection process requires that all operational chemicals used on the FPSO (including those that may end up in the produced water discharge) are risk-assessed under the UK based Offshore Chemical Notification Scheme (OCNS). Chemicals are ranked according to their calculated hazard quotients by the chemical hazard assessment and risk management (CHARM) mathematical model, which uses aquatic toxicity, biodegradation and bioaccumulation data. Chemical selection in accordance with Santos' process ensures only environmentally acceptable chemicals are used on the FPSO and discharged with the produced water.	
	<ul> <li>(22(gg));</li> <li>the impacts of produced water to marine turtles (22(hh)).</li> </ul>		The ECNT is able to make an informed assessment of any potential consequences of the activity on its FIAs by reference to the information on impacts and risks already provided. Consistent with our 28 March 2024 letter, Santos invites the ECNT to suggest any particular control measures that may be appropriate to adopt in respect of these risks and impacts.	
			<b>[22(bb)]</b> Best available technology in the form of produced water tertiary treatment has been adopted for the FPSO, which is leading practice for Santos and other comparable industry facilities.	
			[ <b>22(dd)]</b> Although OA1 occurs within the bounds of the 'Shelf break and slope of the Arafura Shelf' KEF, the ecological values associated with this unique seafloor feature (i.e., patch reefs and hard substrate pinnacles) were not observed during the Barossa marine studies program, nor are these topographically distinct features evident from the data derived from multiple surveys undertaken across this area.	
			As per the response to 22y, the predicted mixing zone based on dispersion modelling is conservatively set at 6km. This zone does not overlap any biologically important areas, and the features of the Shelf Break and slope of the Arafura Shelf KEF are not present in the mixing zone.	
			<b>[22(ee)]</b> As noted in the Production Operations Information booklet, a water quality monitoring regime, which will include ongoing testing of produced water discharges, will be implemented under a Produced Water Adaptive Management Plan.	
			This will be managed through a combination of discharge sampling and monitoring, and receiving environment sampling and monitoring, to inform effectiveness of existing mitigations and if any further mitigations are required.	
			<b>[22(ff)]</b> As noted in the response for 22aa, chemicals considered for use (that may form part of the produced water discharge stream) are ranked according to their calculated hazard quotients by the chemical hazard assessment and risk management (CHARM) mathematical model, which uses aquatic toxicity, biodegradation and bioaccumulation data, to limit potential for ecotoxicological impacts.	
			Based on the absence of significant marine fauna habitat or activity within the predicted produced water mixing zone, and the combination of best available produced water treatment technology and Santos' chemical selection process to select chemicals with the least aquatic toxicity, ecotoxicological impacts from produced water are not expected (as presented in the Production Operations information booklet).	
			Notwithstanding the above, the potential for longer term ecotoxicological impacts from produced water discharge at OA1 will be assessed through a combination of discharge sampling and monitoring, and receiving environment sampling and monitoring. The results of this assessment will inform if additional mitigations are required to limit potential ecotoxicological impacts to acceptable levels.	
			[22(gg)] Please see the response to 22w, 22aa and 22ff.	
			[22(hh)] Please see response to 22w and 22y.	
	The ECNT requested further information regarding marine	Santos has provided a response to ECNT and considered the matters raised by ECNT where relevant to the Activity in the OEMP	[22(ii)] The Production Operations information booklet provides a summary of the existing environment (Regional Existing Environment Summary) against which impacts were assessed.	Sections 3.2, 6.3, 6.2 and 7
	<ul> <li>environmental objectives and values for the marine environment (22(ii));</li> </ul>	<ul> <li>The Barossa Production Operations EP and this OEMP describe and include the information below in relation, as relevant, to marine environment:</li> <li>Relevant requirements of the Marine Bioregional Plan for the Northern Marine Plan</li> <li>Water and sediment quality monitoring</li> </ul>	[22(jj)] Where relevant to activity impacts and risks, relevant requirements of the Marine Bioregional Plan for the North Marine Region will be considered and addressed in the EP, to ensure management of the Activity and associated impacts and risks are consistent with requirements of the Marine Bioregional Plan for the North Marine Region.	GEP NT Waters Oil Pollution Emergency Plan
	<ul> <li>compliance with the North Marine Bioregional Plan (22(jj));</li> <li>vessel impacts to marine</li> </ul>	<ul> <li>Impacts from FPSO discharges (in the Production Operations EP only)</li> <li>Potential impacts to listed species associated with light, including light spill modelling from the Barossa FPSO for flaring and non flaring scenarios (Worley, 2025)</li> </ul>	In relation to how the Minister has considered this plan, this consultation does not relate to any decision of a Minister. For clarity, the EP will be assessed by NOPSEMA. There is no ministerial decision in respect of the EP.	
	<ul> <li>fauna (22(kk));</li> <li>cumulative impacts of EPSO discharges (22(ll));</li> </ul>	Proposed control measures to protect listed species associated with planned events.	[22(kk)] Operational area speed restrictions refer to limits on vessel speeds within the operational area/s to maintain safe operations.	

•	potential impacts and controls measures to protect flatback turtles (22(mm)); light emission impacts on marine turtles and hatchings (22(nn)); potential impacts and controls measures to protect migratory and threatened species (22(oo)); potential impacts and proposed control measures	Concerns related to Production Operations activities in the Barossa Field and related with FPSO activities are outside the scope of the GEP Coastal Waters OEMP and are addressed in the Barossa Production Operations EP. Santos views that ECNT has been provided sufficient information to enable the ECNT to make an informed assessment of any potential consequences of the Activity on its functions, interests or activities.	<ul> <li>More generally, as noted in the Production Operations information booklet, requirements of the Environment Protection and Biodiversity Conservation Regulations 2000 are to be complied with in regard to marine fauna approach distances and vessel speeds, reducing the likelihood of unplanned marine fauna interactions.</li> <li>[22(II)] As noted in the Production Operations information booklet, all planned discharges will be managed in accordance with maritime industry standards and MARPOL requirements to reduce the potential for significant cumulative impacts.</li> <li>Potential for longer term cumulative impacts will be assessed through water and sediment quality monitoring during production Operations information booklet considers potential impacts to marine turtles, and more specifically where the BIA flatback turtle overlaps OA2.</li> <li>Please refer to Noise Sources and Light Sources sections within the Production Operations information within the Production Operations</li> </ul>	
•	to protect all species with BIAs (22(pp)); an underwater acoustic		light sources. The potential impacts to marine turtle BIAs that overlap OA2 are greatly reduced by the infrequent (3	
•	assessment (22(qq)); details regarding noise		yearly) IMMR vessel activity in OA2.	
	pollution (22(rr));		[22(IIII)] See response to 22/IIII.	
•	and anti-fouling systems (22(ss)).		Ine Production Operations information booklet considers potential impacts from light emissions to marine turtles in OA2. Impacts and risks to marine turtles from light emissions in OA2 are considered low risk due to infrequent IMMR vessel activity ie. approximate duration of 2-3 weeks once every three years. It is also worth noting that IMMR vessels are smaller than construction vessels and have smaller light emissions, further reducing the risk to marine turtles.	
			All considered control measures (adopted and not adopted) will be presented in the EP for assessment by the Regulator/s.	
			[22(oo)] The Production Operations information booklet contains proposed (adopted) control measures as relevant to potential impacts to migratory and threatened species. All considered control measures (adopted and not adopted) will be presented in the EP for assessment by the Regulator/s.	
			<b>[22(pp)]</b> The Production Operations information booklet contains proposed (adopted) control measures as relevant to potential impacts where BIAs overlap OA2 (no BIA overlap with OA1). To the extent that BIAs overlap parts of the EMBA and/or MEVA, this is considered in the proposed control measures for unplanned events in the Production Operations information booklet and will also be addressed in the Production Operations Oil Pollution Emergency Plan to be submitted to NOPSEMA.	
			In any event, potential risks and impacts in respect of the EMBA and MEVA (as distinct from the OAs) arise predominately by virtue of the remote risk of a hydrocarbon spill during the activity. The ECNT has sufficient information to identify BIAs within the EMBA by reference to the graphics of the EMBA provided in the Production Operations information booklet and publicly available information on BIAs, and Santos considers that the ECNT has sufficient information to understand the potential risks and impacts within the EMBA, to the extent that these risks or impacts are relevant to the ECNT's functions, interest or activities.	
			<b>[22(qq)]</b> The Production Operations information booklet presents a summary of the results of underwater acoustic assessments for noise sources relevant to the scope of this EP.	
			Further details about the underwater acoustic assessments for production operations activities will be provided in the EP for assessment by the Regulator.	
			[22(rr)] The FPSO will be a 'continuous' or 'non-impulsive' noise source, but at lower levels than impulsive noise sources. FPSO noise sources are predominantly from machinery and equipment on the deck and in the hull, therefore not a source of underwater noise.	
			The potential for 'short term' behaviour change is associated with impulsive noise sources e.g. safety flaring, support vessels, helicopters.	
			<b>[22(ss)]</b> All ballast water management and anti-fouling systems for the FPSO and other vessels, will be managed in accordance with maritime industry standards and MARPOL requirements.	
			All marine vessels will be compliant with maritime law. Consistent with regulatory requirements, the EP will set out the requirements applicable to the activity and how Santos will comply with these requirements	
T in	he ECNT requested further	Santos has provided a response to ECNT and considered the matters raised by ECNT where relevant to the Activity in the OEMP. The Coastal Waters OEMP describes and includes the information below on spill impacts:	[22(tt)] The EMBA and MEVA as presented in the Production Operations information booklet is informed by scholastic modelling.	Section 3 and 7.6
in	npacts of the Activity, including:	Details of spill toxicity.	As the Production Operations information booklet explains (in detail under section "Environment that	GEP NT Waters
•	stochastic modelling for hydrocarbon and	<ul> <li>Details on the spill ENIBA.</li> <li>OPEP spill response strategies.</li> </ul>	may be affected (EMBA)"), the EMBA represents a conservative depiction of the greatest geographical extent of an unplanned spill event.	Oil Pollution Emergency Plan
	condensate spill scenarios (22(tt));	Concerns raised by ECNT in relation to the Production Operations activities in the Barossa Field and related with FPSO activities, are outside the scope of the GEP Coastal Waters OEMP and are addressed in the Barossa Production Operations EP.	Santos welcomes input from ECNT about any information regarding values and sensitivities within the EMBA or MEVA that may be affected by the Activity, and associated impacts or risks.	

impacts on particular	Santos considers this provides sufficient detail for ECNT to assess impacts of the Activity on its functions.	[22(uu)] The Production Operations information booklet presents information about values and	
<ul> <li>marine areas (22(uu));</li> <li>whether there is risk of a</li> </ul>	Santos considers this provides sufficient detail for ECNT to assess impacts of the Activity on its functions, interests or activities, and to provide input to Santos about the environment that may be affected by the Activity and/or additional proposed control measures for consideration in the course of preparing the OEMP.	sensitivities that may be affected by unplanned spill events relevant to the proposed Activity, that fall within the EMBA or MEVA.	
<ul> <li>severe oil spill in the area of the Arafura Shelf (22(vv));</li> <li>impacts on traditional fishing practices (22(ww));</li> </ul>		In preparing this EP, Santos has identified the presence of Ashmore Reef and Cartier Island within the EMBA, the Oceanic Shoals Marine Park in the MEVA, the Continental Slop Demersal Fish Communities within the EMBA and NT coastline communities within the MEVA.	
<ul> <li>details of the ecotoxicity of the various substances for</li> </ul>		Santos welcomes input from ECNT about any information about values and sensitivities within the EMBA or MEVA that may be affected by the Activity, and associated impacts or risks.	
which a spill scenario was modelled (22(xx)).		[22(vv)] In preparing the EP, Santos has identified the presence of the Arafura Shelf within the MEVA that could be affected by an unplanned spill event. This receptor, along with other relevant receptors, will be considered when identifying areas that require particular protection when developing appropriate spill response strategies for the OPEP, which will be assessed by the regulator.	
		[22(ww)]	
		In preparing the EP, Santos has identified commercial marine fisheries and traditional fishing practices that overlap with the EMBA and MEVA. Fishing interests within the MEVA are considered when identifying potential priorities for protection when developing appropriate spill response strategies for the OPEP, which will be assessed by the Regulator.	
		In any event, having regard to the ECNT's FIAs, as described by the ECNT in its letter and on the ECNT's website, these impacts do not appear to be relevant to any potential consequence of the activity on the ECNT's FIAs.	
		[22(xx)] The impacts and risks presented in the Production Operations information booklet have taken into account the relevant properties of the various substances for which a spill scenario has been modelled, including ecotox data where available for Barossa-condensate and industry fuels. These properties were considered when assessing potential impacts to values and sensitivities of the environment that may be affected, as presented in the Production Operations information booklet.	
		The ECNT is able to make an informed assessment of any potential consequences of the activity on its FIAs by reference to the information on impacts and risks already provided. As noted previously, Santos invites the ECNT to suggest any particular control measures that may be appropriate to adopt in respect of these risks and impacts.	
		Further, the OPEP spill response strategies as relevant to credible unplanned spill scenarios consider the impacts and risks from use of dispersant, where proposed as an appropriate response strategy for specific unplanned spill scenarios. The information contained in the OPEP will be assessed by the Regulator against the requirements of the Regulations.	
ECNT correspondence to Santo	os on 9 April 2024		
Reiterated concerns in relation	Santos has provided a response to ECNT. Santos has provided the ECNT with sufficient information and a	Santos' correspondence to ECNT on 14 May 2024 in response to ECNT's letter of 9 April 2024	Not Applicable.
including the adequacy of information provided, and manner in which the ECNT is	functions, interests and activities, and to provide any feedback it may have.	Noting that we have already responded to your first and second letters, Santos has provided responses to your concerns and requests for further information in your third letter (dated 9 April 2024), where possible and reasonable, in the attached Annexure.	
being consulted with compared		More broadly, and as was set out in our letters of 28 March and 30 April 2024:	
(paragraphs 1-6 of letter).		The purpose of the consultation is to understand:     the values and consultation of the environment that may be effected:	
		<ul> <li>the values and sensitives of the environment that may be anected,</li> <li>the potential environmental risks and impacts of the proposed activities; and</li> </ul>	
		<ul> <li>any control measures proposed to reduce the environmental impacts and risks of the proposed activities to as low as reasonably practicable and an accordingly lowel</li> </ul>	
		<ul> <li>Santos has provided the ECNT with sufficient information to provide this type of feedback in accordance with regulation 25 of the Offshore Petroleum and Greenhouse Gas Storage (Environment) Regulations 2023 (Regulations);</li> </ul>	
		<ul> <li>Santos has been consulting, and will continue to consult, with the ECNT so that Santos can obtain feedback as to any potential consequences which the proposed activity on the ECNT's functions interests and activities</li> </ul>	
		Santos considers that the information provided in this letter and the Annexure hereto, as well as our letter dated 30 April 2024, together with the previously shared information booklet and factsheet, are sufficient to inform an adequate assessment of the impacts of the activity of the EP on the ECNT's functions, interests and activities.	
		Santos notes the ECNT has expressed concerns that it is being treated in a manner that is distinct from other stakeholders. However, the legislative framework and regulatory guidance makes it clear that the consultation process should be undertaken in a manner appropriate for the person or organisation having regard to their functions, interests and activities that may be affected by our proposed activity. With respect to the ECNT specifically, Santos has:	
		<ul> <li>a) provided ECNT with links to the relevant information booklet and factsheet on 9 February 2024 as part of the preliminary consultation process;</li> <li>b) been formally consulting with the ECNT about this EP and OEMP since 11 March 2024;</li> </ul>	

Reiterated concerns in relation to Santos' discrepancies in GHG emission estimates provided in three separate Barossa approval documents (paras 7-9).	Santos has considered the matters raised by ECNT in the EP and this OEMP and provided a response to ECNT including clarification in relation to their reiterated concerns about apparent discrepancies in emissions estimates in separate approval documents in its letter of 14 May 2024. The Barossa Production Operations EP includes for the GHG emissions forecast including the underpinning assumptions for the Barossa Project. Section 6.3 of this OEMP describes the Scope 1 and Scope 3 emissions pertaining to activities in the Coastal Waters section of the Barossa GEP. Santos considers the information provided in the information booklet sufficient for the ECNT to assess impacts of the Activity on its functions, interests or activities, and to provide input to Santos about the environment that may be affected by the Activity and/or additional proposed control measures for consideration by Santos in the course of preparing the OEMP.	<ul> <li>c) responded to ECNT correspondence on 28 March and 30 April 2024; and</li> <li>d) responded to the ECNT's request for a meeting, now confirmed for 20 May 2024.</li> <li>Santos considers that it has been consulting with the ECNT in a manner which is consistent with the regulatory guidance, including giving the ECNT reasonable time to consider the consultation materials and to provide input for the EP.</li> <li>Santos looks forward to meeting with the ECNT in Darwin on Monday, 20 May 2024. The meeting is an opportunity for the ECNT to provide further input (if any) for Santos to consider in preparing the EP.</li> <li>The scope 1 emissions estimate presented in the Barossa Offshore Project Proposal (OPP) is higher than the estimate provided in the Production Operations Information Booklet (Booklet).</li> <li>This is as a result of reductions in the operational emissions achieved during detailed engineering design undertaken since the OPP was accepted. This is explained in Santos' letter dated 30 April 2024 - refer response to #22g and #22h.</li> <li>The annual emissions estimates included in the Booklet are applicable to the Production Operations Activity (Activity). The annual emissions estimates provided in the Booklet are expected to reduce over the life of the Activity as production operations) given annual emissions are expected to reduce over the life of the Activity as production rates decline.</li> <li>As explained above, discrepancies between emissions estimates can be a product of improved engineering definition over the course of project development, the different metrics that can be used to present emissions estimates, and/or the fact that emissions vary from year to year as production changes over the life of a production facility.</li> <li>For the purpose of the Activity, the ECNT should focus its review on the information provided in the Booklet.</li> <li>As previously advised in Santos' letter dated 30 April 2024, to the extent that there is any potential impact on the ECNT's function, in</li></ul>	Section 6.3
Asserted that Santos has not adopted an OPP control measure of using fuel gas instead of marine diesel and marine gas oil (para 10).	Santos provided a response to ECNT that the ECNT has incorrectly interpreted the content in the information booklet, which was clarified by Santos in its response to the ECNT.	<ul> <li>The reference from pp. 339 of the OPP refers to fuel gas as the preferred fuel for FPSO hydrocarbon processing and utilities operations. This control measure has not been rejected and is consistent with the basis for the EP.</li> <li>The references in the Booklet to marine diesel oil (MDO) or marine gas oil (MGO) are not to be confused with the reference from pp. 339 of the OPP and refer to:</li> <li>use of MDO as a fuel source for support vessels and IMMR vessels; and</li> <li>MGO as a backup or emergency fuel if the EPSO fuel gas system is offline/unavailable.</li> </ul>	Not Applicable
Asserted that Santos has not provided sufficient detail regarding emissions profile of the project (para 11).	Santos has considered the matters raised by ECNT. Santos' assessment is that the information booklet is sufficient for the ECNT to assess impacts of the Activity on its functions, interests or activities, and to provide input to Santos about the environment that may be affected by the Activity and/or additional proposed control measures for consideration by Santos in the course of preparing the OEMP.	Santos has addressed the request for additional emissions estimate detail previously in our letter dated 30 April 2024. Regarding the differences with the DPD SER emissions estimate, see response to #8, which refers to an explanation for this.	Section 6.3
Asserted that Santos should provide the ECNT (and other Relevant Persons) with the details of the full range of GHG emission assessments it has undertaken for the project (para 12).	Santos has considered the matters raised by ECNT. The Barossa Production Operations EP and this OEMP include GHG emissions forecast associated with the Barossa Project and specifically in this OEMP for this Activity, including the underpinning assumptions. Santos' assessment is that the information booklet is sufficient for the ECNT to assess impacts of the Activity on its functions, interests or activities, and to provide input to Santos about the environment that may be affected by the Activity and/or additional proposed control measures for consideration by Santos in the course of preparing the OEMP.	Santos has assessed and defined the full range of potential GHG emissions sources relevant to the activity for this EP, taking account of detailed engineering design undertaken since the OPP, and this is accurately reflected in the Booklet as relevant to the EP.	Section 6.3
Asserted that Santos has failed to identify the impacts of the activity over the lifecycle of the project, relying on a limited period of 5 years of impacts (paras 13-21).	Santos has considered the matters raised by ECNT. The OEMP identifies the impacts of the activity over the lifecycle of the Activity fit for purpose given the OEMP is for the first 5 years of the Activity.	The EP will assess the risks and impacts of the Activity for the lifecycle of the project. Correspondingly, the Booklet sets out the risks and impacts of the Activity that are anticipated to arise for the lifecycle of the project.	The impact and risks of the activity over the life cycle of the project are assessed in Sections 6 and 7 of the OEMP.
Asserted that the ECNT is unable to assess the impacts to the environment from the Activity due to lack of clarity regarding GHG emissions of the project (paras 22-23).	Santos has considered the matters raised by ECNT. The Barossa Production Operations EP and this OEMP includes GHG emissions forecast as relevant to the Activity, including the underpinning assumptions. Santos' assessment is that the information booklet along with the many responses described and provided within this section are sufficient for the ECNT to assess impacts of the Activity on its functions, interests or activities, and to provide input to Santos about the environment that may be affected by the Activity and/or additional proposed control measures for consideration by Santos in the course of preparing the OEMP.	Further to the responses provided to ECNT items #7-21, Santos has clarified the actual extent of greenhouse gas emissions from the Activity and rejects this claim.	Section 6.3



Asserted that Santos has failed to appropriately contextualise and evaluate the GHG emissions of the project (para 24).	Santos has considered the matters raised by ECNT in the Coastal Waters OEMP. The contribution of GHG emissions from the Activity have been contextualised against Australian and global carbon budgets. International agreements, namely the Paris Agreement, have been developed to facilitate an orderly approach to what is a global problem. The nature, quantity and timeframe of each country's contribution and the pathways to achieve UNFCCC obligations vary widely, including having regard to the particular circumstances of each country. This framework recognises, and is premised on, the fact that climate change is a global issue – there is no correlation between where GHG emissions are released and where climate change impacts are felt. Santos' assessment is that the information booklet is sufficient for the ECNT to assess impacts of the Activity on its functions, interests or activities, and to provide input to Santos about the environment that may be affected by the Activity and/or additional proposed control measures for consideration by Santos in the course of preparing the OEMP.	ECNTs claim is noted, as advised in Santos' letter dated 30 Apr potential impact on the ECNT's FIAs as a result of GHG emission informed assessment of that impact with the information set out Santos will provide further definition of the acceptable levels of with consideration for Australia's legislated emissions reduction Regulator against the requirements of the Regulations.
Reiterated concerns regarding Santos' approach of assessing environmental impacts from the project's emissions, including in the national and international context, as well as cumulative impacts (paras 25-28 and 30- 31).	Santos has considered the matters raised by ECNT in the Coastal Waters OEMP. The Barossa Production Operations EP and this OEMP acknowledge the linear relationship between GHG emissions and global warming and reference the latest commentary from the IPCC on the causes and impacts of climate change. It is not possible to draw a link between GHG emissions from the Activity and any specific climate related impact on any specific element of the Australian environment which may result from any net increase to cumulative GHG emissions globally. By keeping the Barossa Gas Project emissions under the Safeguard Mechanism baseline, the Project's incremental contribution to global warming is within acceptable limits as determined by reference to Australia's NDC under the UNFCCC. Santos' assessment is that the information booklet is sufficient for the ECNT to assess impacts of the Activity on its functions, interests or activities, and to provide input to Santos about the environment that may be affected by the Activity and/or additional proposed control measures for consideration by Santos in the course of preparing the OEMP.	<ul> <li>Environmental impacts of GHG emissions from the Activity will a of acceptable impact defined in the EP, for assessment by the F the Regulations.</li> <li>As outlined in the Booklet, as a result of the complex nature of t change impacts cannot be meaningfully linked to any one activit As previously advised in Santos' letter dated 30 April 2024, to the impact on the ECNT's FIAs as a result of GHG emissions, the E assessment of that impact with the information set out in the Bo previous consultation that the ECNT is familiar with these potent emissions. Santos reiterates its invitation to the ECNT to raise a may be appropriate to adopt for Santos' letter dated 30 April 2024.</li> <li>Refer the covering letter and Santos' letter dated 30 April 2024. accordance with the legislative purpose of s 25 consultation and highlighted in our letter of 30 April 2024, Santos considers that a provided in the consultation materials sent to ECNT on 9 Februate ECNT to make an informed assessment of the potential consequence of the potential consequence of the sentential consequence of the sentential consequence of the potential consequence</li></ul>
Concerns were raised in relation to the impacts to listed threatened species from climate change (para 29-30).	Santos has considered the matters raised by ECNT in the Coastal Waters OEMP. The Barossa Production Operations EP and this OEMP acknowledge the linear relationship between GHG emissions and global warming and reference the latest commentary from the IPCC on the causes and impacts of climate change. It is not possible to draw a link between GHG emissions from the Activity and any specific climate related impact on any specific element of the Australian environment which may result from any net increase to cumulative GHG emissions globally. By keeping the Project emissions under the Safeguard Mechanism baseline, the Project's incremental contribution to global warming is within acceptable limits as determined by reference to Australia's NDC under the UNFCCC. The predicted GHG emissions associated with the Activity comprise a nominal amount in the overall scheme of the national and international carbon budgets and will not materially or substantially contribute to existing and future predicted Australian and global GHG emissions. Conservatively the associated potential environmental impacts to Threatened, Migratory or local fauna (e.g. seabirds) is assessed as I – Negligible. Santos' assessment is that the information booklet is sufficient for the ECNT to assess impacts of the Activity on its functions, interests or activities, and to provide input to Santos about the environment that may be affected by the Activity and/or additional proposed control measures for consideration by Santos in the course of preparing the OEMP.	The EP will consider all relevant conservation advice, and threa management plans, in defining acceptable levels of impact and related impacts and risks, for assessment by the Regulator aga Regulations.
Reiterated concerns that Santos has not demonstrated how it will meet its legislative requirements under the Safeguard Mechanism (paras 32-39).	Santos has considered these concerns in the preparation of the Barossa Production Operations EP and this OEMP. Santos can confirm that it will comply with the Safeguard Mechanism, noting that treatment of Barossa GHG emissions is still to be finalised by the CER. Implementation and enforcement of the Safeguard Mechanism is the responsibility of the CER.	Santos will be required to comply with the applicable baseline for compliance year, including net-zero reservoir emissions from fir determine how it will achieve this compliance. The EP will demo to the Activity will be met, which will be considered by NOPSEN Regulator. This information is not necessary for the ECNT to ma possible consequences of the Activity on any of its FIAs.
Reiterated concerns that Santos has not demonstrated that the GHG emissions of the project have been reduced to ALARP and are acceptable (paras 40-45 and to 58).	Santos has considered the matters raised by ECNT in the Coastal Waters OEMP. Santos describes the control measures for the Scope 1 & Scope 3 emissions associated with this Activity, including associated ALARP demonstration. Santos has considered the concerns raised. Santos considers that GHG emissions have been reduced to ALARP and are acceptable for the reasons set out in Section 6.3.5 and Section 6.3.6.	The extent to which the EP demonstrates that the impacts and have been reduced to ALARP and acceptable levels, is a matter the requirements of the Regulations.
Concerns raised regarding whether a future CCS project constitutes a component of the	Santos has considered the matters raised by ECNT. ECNT queries about CCS are not considered further as CCS opportunities at Bayu Undan are outside the scope of the Coastal Waters OEMP.	As previously advised in our letter dated 30 April 2024, while Sa opportunities at Bayu-Undan and elsewhere, CCS is not part of within the scope of the Production Operations EP.



April 2024, to the extent that there is any sions, the ECNT is able to make an out in the Booklet.	Section 6.3	
of impact from GHG emissions in the EP, on targets, for assessment by the		
ill be evaluated against acceptable levels e Regulator against the requirements of	Section 6.3	
of the global emissions system, climate ivity or emissions source.		
e ECNT is able to make an informed Booklet. Santos understands from ential risks and impacts related to GHG e any control measures that it considers preparing the EP for submission to		
4. Santos welcomes the ECNT's input in and is meeting with the ECNT. As at the information about GHG emissions ruary 2024 is sufficient in order for equences of the Activity on any of its		
eatened species recovery and nd evaluation of activity GHG emissions gainst the requirements of the	Section 6.3 Section 3	
e for the Barossa project in each first gas. It is a matter for Santos to monstrate how requirements applicable EMA in the exercise of its functions as make an informed assessment of the	Section 6.3 EPO 11	
d risks from Activity GHG emissions tter for the Regulator to assess against	Section 6.3	
Santos has committed to explore CCS of the Barossa development, and not	Not applicable	
Activity, including whether CCS will be a mitigation and control measure for GHG emissions of the project at some point over its		
---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------	--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------	-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------
lifecycle (paras 46-56).		
Asserted that Santos has no viable control measures capable of meaningfully mitigating its scope 1 GHG emissions for the Activity, without CCS being a component of the Barossa project (para 57).	Santos has considered the matters raised by ECNT in the Coastal Waters OEMP. The Barossa Production Operations EP and this OEMP describe the control measures to reduce Scope 1 GHG emissions for the Activity. Santos considers that GHG emissions have been reduced to ALARP and are acceptable. ECNT's claim about CCS is not considered further as CCS opportunities at Bayu Undan are outside the scope of the Coastal Waters OEMP.	As previously advised in the response to #46, CCS is not part As explained in Santos' letter dated 30 April 2024 (refer responsion significant reductions in Scope 1 operational emissions since the will comply with the requirements of the Safeguard Mechanism In any event, the extent to which the EP demonstrates that the emissions have been reduced to ALARP and acceptable level assess against the requirements of the Regulations.
Asserted that Santos has made no attempt to properly define the impacts of indirect emissions from the project (para 59).	Santos has considered the matters raised by ECNT in the Coastal Waters OEMP. The Barossa Production Operations EP and this OEMP acknowledge the linear relationship between GHG emissions and global warming and references the latest commentary from the IPCC on the causes and impacts of climate change. The Activity's projected emissions are contextualised against established emissions budgets (national and global). Climate change is a global problem with the solution being led at the international level. Domestically, GHG emissions are regulated through Australia's NDC and the SGM and the Barossa JV has legal obligations to ensure that the Activity operates in accordance with that framework. Assuming the emissions from the Barossa Gas Project will cause an equivalent net increase in cumulative Australian and global emissions, this increase is de minimis in the context of Australian and global carbon budgets; and there is no correlation between where GHG emissions are released and where climate change impacts are felt. The Barossa JV has adopted environmental performance outcomes and control measures directed to minimising the GHG emissions from the Activity. A range of controls have been considered for both direct (Scope 1) and indirect (Scope 3) emissions. Santos' considers GHG emissions have been reduced to ALARP and are acceptable.	<ul> <li>Santos rejects ECNT's claim that Santos has made no attemp indirect emissions nor to account for the ways these impacts of of the Booklet identifies:</li> <li>the indirect impacts of climate change on the Australian elemissions (Scope 1 and 3) from the Activity; and</li> <li>a proposed control measure to mitigate impacts from indir Activity.</li> <li>In any event, the extent to which the EP demonstrates that the (Scope 3) GHG emissions have been reduced to ALARP and Regulator to assess against the requirements of the Regulatio See also Santos' response to #8 above.</li> </ul>
Asserted that Santos has not described how scope 3 GHG emissions will be reduced to ALARP and acceptable levels, and that the Activity poses an unacceptable risk to the environment (paras 60-61 and 64).	Santos has considered the matters raised by ECNT in the Coastal Waters OEMP. The Barossa Production Operations EP and this OEMP acknowledge the linear relationship between GHG emissions and global warming and references the latest commentary from the IPCC on the causes and impacts of climate change. The Activity's projected emissions are contextualised against established emissions budgets (national and global). Climate change is a global problem with the solution being led at the international level. Assuming the emissions from the Barossa Gas Project will cause an equivalent net increase in cumulative Australian and global emissions, this increase is de minimis in the context of Australian and global carbon budgets; and there is no correlation between where GHG emissions are released and where climate change impacts are felt. The Barossa JV has adopted environmental performance outcomes and control measures directed to minimising the GHG emissions from the Activity. A range of controls have been considered for both direct (Scope 1) and indirect (Scope 3) emissions. Santos' considers that GHG emissions have been reduced to ALARP and are acceptable.	The extent to which the EP demonstrates that the impacts and GHG emissions have been reduced to ALARP and acceptable assess against the requirements of the Regulations. The EP demonstration that the impacts and risks from Activity have been reduced to ALARP and acceptable levels is perform Government's interim (2030) and longer term (2050) emission regulations such as the Safeguard Mechanism. Ultimately, it is the acceptability of Santos' demonstration against the requirem
Asserted that neither Japan nor South Korea are on track to meet Paris Agreement commitments in the period to 2030 (para 63).	Santos notes ECNT's comment.	Noted.
Asserted that Santos has made no attempt to define the impact of methane emissions (para 65).	Santos has considered the matters raised by ECNT. Methane emissions are accounted for in the Barossa Project GHG emissions estimate.	Although methane emissions represent a minor contribution to are accounted for in the Scope 1 GHG emissions estimate pro The EP demonstration that the impacts and risks from Activity have been reduced to ALARP and acceptable levels is perform Government's interim (2030) and longer term (2050) emissions regulations such as the Safeguard Mechanism. Ultimately, it is the acceptability of Santos' demonstration against the requirem
Asserted that the Activity is inconsistent with Commonwealth Recovery Plan for Marine Turtles (para 66).	Santos has considered the matters raised by ECNT in the Coastal Waters OEMP. Santos has assessed the Activity impacts and risks and does not consider the Activity to be inconsistent with the Recovery Plan for Marine Turtles. Consistency with the recovery plan for marine turtles is demonstrated in the following Sections: 6.1 to 6.6 and 7.1 to 7.7.	Santos notes and refutes ECNT's claim, which we address be
Asserted Santos' spill response plans are not finalised and do	Santos has considered the matters raised by ECNT in the Coastal Waters OEMP.	Page 9 of the Booklet acknowledges the various conservation including recovery plans, that have been considered in develo Pollution Emergency Plan (OPEP)). In assessing the potential



of the Barossa development. nse to #22h), Santos has achieved he Barossa Development OPP. Santos n. impacts and risks from Activity GHG s, is a matter for the Regulator to	Control measures relating to managing GHG emissions are outlined in Section 6.3.3
t to properly define the impacts of these ould be reduced or mitigated. Page 14	Section 6.3
nvironment, associated with GHG	
ect (Scope 3) emissions from the	
impacts and risks from Activity indirect acceptable levels, is a matter for the ns.	
risks from Activity indirect (Scope 3) levels, is a matter for the Regulator to	Section 6.3
GHG emissions (direct and indirect) ned in the context of the Australian s reduction targets, and associated a matter for the Regulator to assess nents of the Regulations.	
	Not applicable
Activity Scope 1 GHG emissions, they vided in the Booklet.	Not applicable
GHG emissions (direct and indirect) ned in the context of the Australian s reduction targets, and associated a matter for the Regulator to assess nents of the Regulations.	
ow.	Sections 6.1, 6.2, 6.3, 6.4, 6.5, 6.6 and Sections 7.1, 7.2, 7.3, 7.4, 7.5, 7.6 and 7.7
management plans and advice, oment of the EP (including the Oil impacts from an unplanned spill event,	Sections 7.4 and 7.6

not provide sufficient details regarding harm to turtles (para 68).	Santos has a number of accepted spill response (thereby finalised) plans for Barossa related activities which include for response strategies to reduce harm to turtles. Santos has considered the requirements of the Recovery Plan for Marine Turtles (2017) to ensure the Coastal Waters OEMP Oil Pollution Emergency Plan is consistent with the requirements of the recovery plan.	<ul> <li>Santos has considered the requirements of the Recovery Plan for Marine Turtles (2017) to ensure the proposed control measures are consistent with the requirements of the recovery plan.</li> <li>The OPEP includes an objective to identify environmental sensitivities at risk and conduct operational Net Environmental Benefit Analysis (NEBA). The NEBA process is used by the Incident Management Team during an oil spill response operation so the most effective response strategies with the least detrimental environmental impacts can be identified. As a component of the incident action planning process, in the event of a spill, a NEBA is applied to achieve the following:</li> <li>Identify sensitivities within the area potentially affected by a spill at that time of the year (noting that the sensitivity of some key receptors, such as birdlife and turtles, varies seasonally).</li> <li>Assist in prioritising and allocating resources to sensitivities with a higher protection and response priority.</li> <li>Assist in determining appropriate response strategies with support of real-time metocean conditions, oil spill tracking and fate modelling.</li> <li>Consistent with the requirements of the Regulations, Santos plans are not finalised at the time of consultation and won't be finalised until plans are submitted to the Regulator for assessment.</li> <li>The extent to which Santos has demonstrated Activity impacts and risks are consistent with the Recovery Plan for Marine Turtles (2017) is a matter for the Regulator to assess against the requirements of the Regulations.</li> </ul>	GEP NT Waters Oil Pollution Emergency Plan
Further information requested to ensure that Barossa operations are not inconsistent with artificial lighting requirements for marine turtles (para 71).	Santos has considered the matters raised by ECNT in the Coastal Waters OEMP. Santos notes that impacts to marine turtles from Activity lighting are not expected to occur, primarily due to the infrequent and short duration of vessel planned inspection activities within the Operational Area. Santos considers that Barossa operations are consistent with artificial lighting requirements for marine turtles. Santos views that ECNT has been provided sufficient information to enable them to make an informed assessment of any potential consequences of the Activity on its functions, interests or activities.	Planned vessel inspection and maintenance activities along the Barossa GEP in OA2 would occur at a frequency of approximately once every 3 years, for a duration of approximately several weeks across the full extent of the pipeline. Vessel presence at any one location (during vessel inspection and maintenance activity periods) would be approximately 2-3 days in any one specific location. As stated in the Booklet, impacts to nesting females or hatchlings are not expected to occur, primarily due to the infrequent and short duration of vessel inspection activities, as explained above.	Section 6.2 BAO-CM-6.1.2 BAO-CM-6.2.1 BAO-CM-6.2.2
Asserted that Santos has failed to identify cumulative water quality impacts on marine turtles (para 72).	Santos has considered the matters raised by ECNT in the Coastal Waters OEMP. Routine vessel discharges (deck drainage/run-off, sewage and greywater, cooling water, bilge water, brine and ballast water) in the OA associated with infrequent (IMMR activities nominally every 3 years) vessel activities will occur in the internesting BIA for flatback turtles. Santos has assessed the potential impacts to marine turtles according to the nature and scale of activities, and the locations of planned activities in the OA, relative to the presence and extent of the flatback turtle BIA. Santos considers the potential water quality impacts to marine turtles from the Activity to have been reduced to ALARP and acceptable levels.	Santos has identified the Recovery Plan for Marine Turtles (2017) as a relevant consideration during preparation of the EP. The chemicals that may be present in the discharged water have properties that are non-bioaccumulative, are biodegradable and breakdown quickly. Given the potential exposure times for turtles that may transit through the area, they are not there long enough to experience acute toxic effects. The adequacy of Santos' evaluation of the impacts and risks to marine turtles from the Activity, is a matter for the Regulator to assess against the requirements of the Regulations.	Section 6.6
Asserted that Santos needs to further assess impacts to important marine turtle foraging grounds, migratory corridors, mating areas and habitat for hatchling dispersal (para 73).	Santos has considered the matters raised by ECNT in the Coastal Waters OEMP. Santos has considered all reasonably ascertainable information of relevance to the assessment of potential impacts to marine turtles from the Activity.	The evaluation of impacts and risks to marine turtles as presented in the Booklet has considered Department of Climate Change, Energy, the Environment and Water published biologically important areas (BIAs) as relevant to foraging, mating, nesting, interesting, and has also considered best available information on migration pathways. As noted in the Booklet, OA2 overlaps a portion of the flatback turtle inter-nesting BIA. Santos reiterates that the information provided in the Booklet is sufficient in order for ECNT to make an informed assessment of the potential consequences of the Activity on any of its FIAs.	Section 6
Asserted that Santos has failed to consider compliance with the Threat Abatement Plan for the impacts of marine debris on vertebrate marine life (para 74).	Santos has considered the matters raised by ECNT in the Coastal Waters OEMP. The Barossa Production Operations EP and this OEMP consider the Threat Abatement Plan for Impacts of Marine Debris on Vertebrate wildlife of Australia's coasts and oceans (CoA, 2018) when evaluating potential threats to marine turtles from the Activity.	Santos has considered the Threat Abatement Plan for Impacts of Marine Debris on Vertebrate wildlife of Australia's coasts and oceans (CoA, 2018) when evaluating potential threats to marine turtles from the Activity, and this has informed the Booklet. The EP assesses the scale of impact associated with dropped objects. Santos will comply with legislation for the prevention of garbage disposal from vessels. Given the limited quantities and likely objects, as well as the control measures proposed, the potential impacts from the Activity to species identified in relevant species recovery plans, conservation advice, wildlife conservation plans and management actions will be minimised. Ultimately, the adequacy of Santos' proposed control measures is a matter for the Regulator to assess against the requirements of the Regulations.	Section 7.1 BAO-CM-7.1.1 BAO-CM-7.1.2
Asserted that Santos has not sufficiently assessed the potential impacts to marine species (para 75).	Santos has considered the matters raised by ECNT in the Coastal Waters OEMP. Santos has considered and assessed potential impacts to marine species, as appropriate to the nature and scale of planned activities in the OA and associated impacts and risks.	ECNT's observations about the difference between the level of marine species information provided between OA1 and OA2 is reflective of the difference in nature and scale of activities between the different operational areas. OA1 will comprise infrastructure with an ongoing presence both below and above the waterline and includes continuous support vessel activities. In contrast, OA2 comprises a subsea pipeline located on the seabed, with non-continuous infrequent inspections (approximately every 3 years) and maintenance activity via a vessel. The EMBA and MEVA described in the EP are associated with potential impacts from an unplanned event – namely a hydrocarbon spill. The likelihood of a hydrocarbon release is unlikely.	Sections 6 & 7
Asserted that Santos failed to provide sufficient information on	Santos has considered the matters raised by ECNT in the Coastal Waters OEMP.	<ul> <li>a. The EP describes the noise emissions associated with the Activity and commissioned a Noise Impacts on Marine Fauna to support the noise emissions impact assessment presented.</li> </ul>	Section 6.1 BAO-CM-6.1.1

potential noise impacts on marine mammals (paras 76-78).	The OEMP describes the noise emissions associated with the Activity and considers potential impacts on marine mammals. The information contained in the information booklet about potential noise impacts on marine mammals is sufficient for the ECNT to assess impacts of the Activity on its functions, interests or activities, and to provide input to Santos about the environment that may be affected by the Activity and/or additional proposed control measures for consideration by Santos in the course of preparing the OEMP.	<ul> <li>b. Support vessels will transit from OA1 and OA2 to Darwin. Given OA2 extends along the Barossa GEP from the FPSO to Darwin, Santos considers noise impacts in the wider area have been considered.</li> <li>Santos' evaluation of impacts and risks to marine mammals from noise emissions will be a matter for the Regulator to assess against the requirements of the Regulations.</li> </ul>	BAO-CM-6.1.2
Asserted that Santos failed to provide sufficient information on potential Vessel collisions and other interactions on marine mammals (para 79).	Santos has considered the matters raised by ECNT in the Coastal Waters OEMP. Santos' assessment is that the information presented in the information booklet about unplanned marine fauna interactions is sufficient for the ECNT to assess impacts of the Activity on its functions, interests or activities, and to provide input to Santos about the environment that may be affected by the Activity and/or additional proposed control measures for consideration by Santos in the course of preparing the OEMP.	Santos has considered relevant management actions, policy advice and legislation when evaluating potential threats to marine mammals from the Activity, and this has informed the information presented in the EP and the information Booklet (including as to control measures proposed to be implemented). The OA does not overlap the migration route of the pygmy blue whale. Santos confirm that vessels will be required to comply with its Protected Marine Fauna Interaction and Sighting Procedure, which ensures compliance with Part 8 of Environment Protection and Biodiversity Regulations 2000 which includes controls for minimising the risk of collision with marine fauna. Ultimately, the adequacy of Santos' proposed control measures is a matter for the Regulator to assess against the requirements of the Regulations.	Section 7.3 BAO-CM-6.1.1 BAO-CM-6.1.2 BAO-CM-6.6.6 BAO-CM-6.4.5
Asserted that Santos has dismissed the potential impacts to migratory seabirds and shorebirds (para 80).	Santos has considered the matters raised by ECNT in the Coastal Waters OEMP. Santos has assessed potential impacts to migratory seabirds and shorebirds appropriate to the nature and scale of activities, and the locations of planned activities in the GEP Coastal Waters OEMP, relative to the presence and extent of seabird or shorebird BIAs.	There are no seabird or shorebird BIAs that overlap with either OA1 or OA2, which has informed the presentation of environmental impacts and risks in the Booklet. EPBC Act listed seabird or shorebird species that could occur in the EMBA (associated with potential impacts from an unplanned event – namely a hydrocarbon spill) will be considered in the EP but represent lower environmental risk due to the low likelihood (unlikely) of an unplanned event.	Sections 6 & 7
Asserted that the Information Booklet failed to provide sufficient information on the project's risks and impacts on bird species, including flaring and venting excess gas and light pollution (para 81).	Santos has considered the matters raised by ECNT in the Coastal Waters OEMP. [Santos has assessed potential impacts to migratory seabirds and shorebirds appropriate to the nature and scale of activities, and the locations of planned activities in the OA, relative to the presence and extent of seabird or shorebird BIAs. Santos considers the information presented in the information booklet is sufficient for the ECNT to assess impacts of the Activity on its functions, interests or activities, and to provide input to Santos about the environment that may be affected by the Activity and/or additional proposed control measures for consideration by Santos in the course of preparing the OEMP. Flaring and venting activities on the FPSO are out of scope for the Coastal Waters OEMP, and these are addressed in the Barossa Production Operations EP.	See response to para #80. The EP will assess all potential impacts to seabirds and migratory birds associated with flaring activity. There are no seabird or shorebird BIAs that overlap with either OA1 or OA2, which has informed the presentation of environmental impacts and risks in the Booklet. The EP will assess the potential attraction of birds to the gas flare.	Sections 6.2
Asserted that Santos missed potential impacts on fish and other marine species with habitats beyond the boundaries of OA1 and OA2, including impacts from support vessels travelling to and from Darwin (para 82).	Santos has considered the matters raised by ECNT in the Coastal Waters OEMP. Santos has assessed potential impacts to fish and other marine species from planned activities within the Coastal Waters OEMP Operational Area and from unplanned events in the broader environment that may be affected.	The EP assesses the potential noise emission impacts on fish and other marine species associated with support vessels. Support vessels will transit from OA1 and OA2 to Darwin. Given OA2 extends along the Barossa GEP from the FPSO to Darwin, Santos considers noise impacts in the wider area have been considered in the EP.	Sections 6 and 7
Asserted that Santos' has failed to adequately assess the potential risks and impacts to fish and other marine life resulting from discharges (para 83).	Santos has considered the matters raised by ECNT in the Coastal Waters OEMP. Santos has assessed potential impacts to fish and other marine species from planned activities within the Operational Area and from unplanned events in the broader environment that may be affected.	The chemicals (process) that may be present in the discharged water have properties that are non- bioaccumulative, biodegradable and breakdown quickly. Santos will undertake a full suite of WET testing of the produced water discharges in accordance with ANZECC & ARMCANZ (2000) guidelines once production commences. Given the potential exposure times for plankton, fish, invertebrates and sharks that may transit through the area, they are not there long enough to experience acute toxic effects. The adequacy of Santos' evaluation of impacts and risks to fish and other marine life and any related control measures will be a matter for the Regulator to assess against the requirements of the Regulations.	Section 6.6 BAO-CM-6.7.5
Asserted that the Information Booklet does not provide sufficient information on the chemical composition of produced water (para 85).	Produced water is associated with the FPSO activities and is not an impact of activities under the OEMP Concerns related to the FPSO and Production Operations activities in the Barossa Field are outside the scope of the Coastal Waters OEMP and are addressed in the Barossa Production Operations EP.	Page 20 of the Booklet does provide a general description of the composition of produced water, and Santos refers ECNT to its response in its letter dated 30 April 2024 (request 22aa).	Not applicable.
<ul> <li>Asserted that the Information Booklet does not provide a sufficient explanation for the:</li> <li>forecast produced water rate (para 86); and</li> <li>produced water treatment stages (para 87).</li> </ul>	Produced water is not an impact of activities under the OEMP, and associated with the FPSO activities. Concerns related to the FPSO and Production Operations activities in the Barossa Field are outside the scope of the Coastal Waters OEMP and are addressed in the Barossa Production Operations EP.	<ul> <li>[86] The forecast produced water rate over time is merely a reflection of the Barossa reservoir properties and is included to emphasise the conservatism in the design capacity of the FPSO to process produced water volumes of up to 20,000 bbl/day.</li> <li>In any event, a further explanation is not required in order for ECNT to make an informed assessment of any potential consequences of the Activity on its FIAs by reference to the impacts and risks already provided.</li> <li>The produced water treatment system is a multi-stage treatment process that progressively removes hydrocarbons and other contaminants to reduce oil in water concentrations that are acceptable for discharge.</li> </ul>	Not applicable.

[87] The produced water to solids removal and pumple         • produced water solids removal and pumple         • floatation vessel         • tertiary produced water tree         Asserted that there is insufficient         Produced water is not an impact of activities under the OEMP and associated with the FPSO activities.	treatment and discharge system consi ng equipment. The system consists of surge drum (induced gas flotation unit) eatment unit – macro-porous polymer r letter dated 30 April 2024 (request 2
• produced water s         • hydrocyclone         • floatation vessel         • tertiary produced water tree         Asserted that there is insufficient         Produced water is not an impact of activities under the OEMP and associated with the FPSO activities.	surge drum (induced gas flotation unit) eatment unit – macro-porous polymer r letter dated 30 April 2024 (request 2
Asserted that there is insufficient       Produced water is not an impact of activities under the OEMP and associated with the FPSO activities.       Santos refers ECNT to out	eatment unit – macro-porous polymer r letter dated 30 April 2024 (request 2
Asserted that there is insufficient Produced water is not an impact of activities under the OEMP and associated with the FPSO activities. Santos refers ECNT to out	r letter dated 30 April 2024 (request 2
information provided regarding the chemical selection process, or details of the chemical compounds used as additives in produced water discharges (para 88).Concerns related to the FPSO and Production Operations activities in the Barossa Field are outside the scope of the Coastal Waters OEMP and are addressed in the Barossa Production Operations EP.	
Asserted that there is insufficient information provided regarding produced water monitoring (para 89). Produced water so the Coastal Waters OEMP and are addressed in the Barossa Production Operations EP. As stated in the Booklet, if is routed to a dedicated system until the concentrative of the Coastal Waters OEMP and are addressed in the Barossa Production Operations EP.	f the produced water stream does not storage tank, for subsequent re-proce ations meet the acceptable limit of 30r (request 22z).
Concerns raised regarding the cumulative impacts of produced water in light of the '30 mg/l over any 24- hour period limit' (para 90).Produced water is not an impact of activities under the OEMP and associated with the FPSO activities. Concerns related to the FPSO and Production Operations activities in the Barossa Field are outside the scope of the Coastal Waters OEMP and are addressed in the Barossa Production Operations EP.Santos refers ECNT to out Santos refers ECNT to out 	r letter dated 30 April 2024 (request 2
Concern raised that the produced water is not an impact of activities under the OEMP and associated with the FPSO activities. The produced water adaptive Concerns related to the FPSO and Production Operations activities in the Barossa Field are outside the scope of the Regulator to assess age to a concerns related to the FPSO and Production Operations activities of the Regulator to assess age to a concerns related to the FPSO and Production Operations activities of the Regulator to assess age to a concerns related to the FPSO and Production Operations activities of the Regulator to assess age to a concerns related to the FPSO and Production Operations activities of the Regulator to assess age to a concerns related to the FPSO and Production Operations activities of the Regulator to assess age to a concerns related to the FPSO and Production Operations activities of the Regulator to assess age to a concerns related to the FPSO and Production Operations activities of the Regulator to assess age to a concerns related to the FPSO and Production Operations activities of the Regulator to assess age to a concerns related to the FPSO and Production Operations activities of the Regulator to assess age to a concerns related to the FPSO and Production Operations activities of the Regulator to assess age to a concerns related to the FPSO and Production Operations activities of the Regulator to assess age to a concerns related to the FPSO and Production Operations activities of the Regulator to assess age to a concerns related to the FPSO and Production Operations activities of the Regulator to assess age to a concerns related to the FPSO and Production Operations activities of the Regulator to assess age to a concerns related to the FPSO and Production Operations activities of the Regulator to a concerns activities of the Regulator to a concerns related to the FPSO and Production Operations activities of the Regulator to a concerns activities of the Regulator to a concerns activities of the Regulator to concerns activities	tive management plan will be describ gainst the requirements of the Regula
made available (para 92). Santos considers the ECN of the Activity on its FIAs b	IT is able to make an informed assess by reference to the information on imp
Concern raised that no lists or Produced water is not an impact of activities under the OEMP and associated with the FPSO activities. Santos refers ECNT to out	r letter dated 30 April 2024 (request 2
tables of chemical species being tested for has been provided (para 93). Concerns related to the FPSO and Production Operations activities in the Barossa Field are outside the scope of the Coastal Waters OEMP and are addressed in the Barossa Production Operations EP.	
Concern raised that no       Drilling activity is outside the scope of the Coastal Waters OEMP. and regulated through a separate EP.       Drilling fluids associated water is not an impact of activities under the OEMP and associated with the FPSO activities.       Drilling fluids associated water addressed in the activities under the OEMP and associated with the FPSO activities.       Drilling fluids associated water addressed in the activities.         into produced water discharges (para 94).       Drilling fluids that will feed       Drilling fluids the scope of the Coastal Waters OEMP and associated with the FPSO activities.       Drilling fluids associated water addressed in the activities.	vith drilling operations will not form pa cepted Drilling and Completions Envir
Concerns raised about certain Produced water is not an impact of activities under the OEMP and associated with the FPSO activities. Responding to each of the	ECNT's numbered paragraphs:
produced water matters in the OPP, which the ECNT asserts should be addressed in the EP,	ovided in the EP as required for the R irements of the Regulations.
including chemical concentrationb. The produced water treatinglevels and cumulative impactimpact/risk will be addressissues (para 95(a)-(f)).the Regulations.	atment system is designed to remove sed in the EP for assessment by the F
c. Santos refers ECNT to a	our letter dated 30 April 2024 (reques
d. Drilling fluids associated were addressed in the acc	d with drilling operations will not form cepted Drilling and Completions Envir
e. Drilling fluids associated were addressed in the acc	d with drilling operations will not form cepted Drilling and Completions Envir
f. The OPP was accepted NOPSEMA assessment p	by NOPSEMA in March 2018. The En rocess and the EP will consider applie
Cumulative impacts are un of the chemicals typically u would begin to breakdown as evaporation, oxidation,	nlikely due to the non-bioaccumulative used in production. In addition, any hy a as soon as they enter the water thro and biodegradation.
Meeting with ECNT on 20 May 2024	
The ECNT asserted that the information Santos has provided to date is not adequate to understand the impacts of the Activity for the Coastal Waters OEMP on the ECNT's functions, interests and activities, and to provide any feedback it may have.	



nsists of multiple stages of de-oiling, of a:	
er extraction (MPPE).	
t 22aa).	Not applicable.
ot meet the specifications for discharge cessing in the produced water treatment 0mg/L over 24-hours. See also Santos'	Not applicable.
t 22gg).	Not applicable.
ibed in the EP and will be a matter for lations.	Not applicable.
essment of any potential consequences npacts and risks already provided.	
t 22aa).	Not applicable.
part of the produced water stream and vironment Plan.	Not applicable.
	Not applicable.
Regulator to assess the Activity impacts	
ve low levels of mercury. This Regulator against the requirements of	
est 22aa).	
n part of the produced water stream and vironment Plan.	
n part of the produced water stream and /ironment Plan.	
EP will be subject to a separate licable cumulative impacts.	
ive and rapid biodegradation properties hydrocarbons from produced water rough a complex mix of processes such	
	No additional measures adopted.

The ECNT queried whether Santos's compliance measures to meet the Safeguard Mechanism requirements will be included in the Environment Plan.	Santos has considered the matters raised by ECNT in the Coastal Waters OEMP. Santos agrees that the OEMP and Barossa Production Operations EP must demonstrate compliance with the Safeguard Mechanism.	Santos explained that the assessment of Safeguard Mechanism is under the remit of the Clean Energy Regulator. However, it agreed that the Environment Plan must demonstrate compliance with the Safeguard Mechanism.	Section 6.3 EPO 11
The ECNT sought confirmation that CCS is not included in the Barossa Production Operations Environment Plan.	Santos has considered the matters raised by ECNT and provided a response to ECNT. Santos confirmed that CCS was not part of the Activity under the EP.	Santos confirmed CCS is not included in the Barossa Production Operations Environment Plan.	Not applicable.



EC	ECNT correspondence to Santos on 28 May 2024					
Th	e ECNT:	Santos has considered the matters raised by ECNT in the Coastal Waters OEMP.	Santos' correspondence to ECNT on 12 June 2024 in respo			
•	reiterated concerns	Santos has provided the ECNT with sufficient information and a reasonable period to assess any possible	Santos acknowledges that the ECNT is still preparing a respons			
	regarding adequacy of information ECNT have received in relation GHG emissions;	Santos about the environment that may be affected by the Activity and/or additional proposed control measures for consideration by Santos in the course of preparing the OEMP.	Santos understands the ECNT is seeking information about the includes management controls for Scope 1 GHG emissions at the flaring management, fugitive emissions management, emissions management, emissions and reduction opportunity identification.			
•	indicated it will only provide detailed information to NOPSEMA in the draft EP, but not to the ECNT, which the ECNT claims is inconsistent with the consultation requirements		There has been extensive correspondence exchanged over the EMP which commenced in February 2024. Santos has responde ECNT on 28 March and 14 May. Since that time, Santos has als 2024. At that meeting, Santos answered the ECNT's questions a opportunity to ask any further questions about the EP. No further In those circumstances, if the ECNT wishes to provide any additional measures to be included.			
	under the Regulations;		Thursday, 20 June 2024, noting this is 30 days following the me			
•	inferred that that Santos is referring the ECNT to information in the OPP, which the ECNT claims does not fulfil Santos' consultation obligations		If and when you provide any further input, please let us know if y provide during consultation not be published. If you make this re published as part of the plan, in accordance with relevant legisla give to the regulator to assess our plan will be provided in a sep published plan. Santos will handle your information in accordance <u>Consultation Privacy Policy</u> .			
•	stated that it was engaging experts to assist in its assessment of how the activity may impact its functions, interests or activities.					
EC	CNT correspondence to Santo	s on 20 June 2024				
Th tha prove EF coo ann reg obb EC noo suu 1. 2. Th	e ECNT reiterated concerns at it has been prevented from oviding necessary input on the P because Santos has not nsulted in a meaningful way d has not discharged its gulatory consultation ligations. For example, the CNT alleged that Santos has t provided the ECNT with fficient: time to effectively engage in consultation; and information to fully understand the potential consequences of the Activity on its functions, interests or activities.	Santos has considered the matters raised by ECNT in the Coastal Waters OEMP. The Barossa Project's overall GHG emissions and proposed management are addressed in the Barossa Production Operations EP. The OEMP includes for impact assessment with regards GHG emissions and noise impacts on marine fauna associated with the Activity. Concerns related to Production Operations activities in the Barossa Field, including operation of the FPSO, are outside the scope of the GEP Coastal Waters OEMP and are addressed in the Barossa Production Operations EP. Santos has provided the ECNT with sufficient information and a reasonable period to assess any possible impacts of the Activity for this OEMP on the ECNT's functions, interests and activities, and to provide any feedback it may have	Santos correspondence to ECNT on 7 August 2024 [Cover letter] Thank you for your letter of 20 June 2024. For convenience, in the attached Annexure, we set out a respon- where and when Santos has previously provided information ad set out in your most recent letter. Santos is of the view that suffil ECNT to allow ECNT to make an informed assessment of the pro- to be carried out under the proposed Production EP on any of E activities. Santos thanks ECNT for its comments and submissions in your previous correspondence and our meeting, in respect of the risk relation to the activity. The matters raised by ECNT demonstrate comprehensively in the consultation process on the basis of the ECNT's complaints in respect of Santos' compliance with the OI Gas Storage (Environment) Regulations 2023 are properly matt its consideration as expert Regulator. Santos is finalising the Production EP for submission in the corr [Annexure] • [Paras [4]-[15]] Santos considers that sufficient information			
•	alleging Santos is withholding information from ECNT in a way that is inconsistent with consultation requirements; alleging Santos has not engaged in the substance of the ECNT's concerns, and the majority of the ECNT's questions remain unanswered;		<ul> <li>[Paras [4]-[15]] Santos considers that sufficient information to allow ECNT to make an informed assessment of the poss activity to be carried out under the Production EP on any of activities. As you would expect, Santos has carefully considered releve Australia and the Consultation Guideline. Santos disagrees information intended to be provided to NOPSEMA is required discharge the consultation requirement. Santos also disagrees has had insufficient time to consider the information provide has otherwise been deficient.</li> <li>[Para [16]] The information set out in those paragraphs of t correspondence identified in paragraph 16 of the ECNT Co estimated annual Scope 1 and Scope 3 emissions associated</li> </ul>			



onse to ECNT's letter of 28 May 2024 se to its letter dated 14 May 2024. GHG Management Plan. That plan the FPSO including but not limited to is monitoring and adaptive in processes. course of consultation for the EP and led to requests for information from the 'so met with the ECNT on 20 May and provided the ECNT with another er questions were asked. 'itional input for this EP (including, if it antos requires this by no later than beeting with Santos on 20 May 2024. 'you request particular information you equest, the information will not be lation. Sensitive information we need to barate report, rather than in the face with our <u>Barossa Gas Project</u>	The operations GHGEMP is described in Section 8.3.2.12 Section 6.3 control measures to manage GHG emissions BAO-CM-6.3.1 BAO-CM-6.3.11 BAO-CM-6.3.12 BAO-CM-6.3.16 BAO-CM-6.3.17 BAO-CM-6.3.17 BAO-CM-6.3.21 BAO-CM-6.3.21 BAO-CM-6.3.22
	Sections 6.1 and
onse to the matters raised, including ddressing the concerns and requests "icient information has been provided to possible consequences of the activity ECNT's functions, interests or " letter of 20 June 2024, and in ks, impacts, and potential controls in the that it has been able to engage the information provided by Santos. "Ifshore Petroleum and Greenhouse ters that will be before NOPSEMA for ming weeks. In has been provided to ECNT to date spible consequences of the proposed of ECNT's functions, interests or evant decisions of the Federal Court of s with your apparent position that all red to be provided to ECNT in order to rees with your contentions that ECNT led, or that the process of consultation the Annexure to Santos' 30 April 2024 prespondence is information about the ted with the proposed activity, and	6.3

their management, in particular the information provided in booklet on that topic. Contrary to paragraph 16, the reason ECNT can, using its specialist knowledge and the informat estimates, make an informed assessment of the potential of level of detail already provided.

[Paras [17-20]] Thank you for this description of ECNT's fa the description makes clear, the central concern of ECNT identification of sources of GHG emissions, advocacy so a and advocacy to describe and address any consequential specialist knowledge in the areas of GHG emissions and c position that Santos has provided sufficient information about associated with the activities to be carried out under the Pl make an informed assessment of the possible consequence Santos will be required to comply with the applicable base for the Barossa project set by the Australian Government in Agreement targets and associated emissions budget. The of GHG emissions from the Activity will be evaluated in the against the requirements of the Regulations. Santos' method acceptability will be broadly consistent with the methodolog as an example refer section 5.1 of the Barossa Subsea Inf Plan.

Further, Santos appreciates that your correspondence is la reasons why it advocates that the Production EP is likely to Project should not proceed at all (for example, see [47]). The Santos. However, contrary to the tenor of ECNT's correspondence in ss 25(1)(d) and 25(2) does not require Santos to respondence ECNT of the merits of the proposal.

- [Para [21]] Santos has provided ECNT with information in items identified specifically in paragraph 21. To the extent information has been prepared or is in the course of prepain those paragraphs of the Annexure to Santos' 30 April 20 paragraph 21 of the ECNT Correspondence, they are matt Santos to present to NOPSEMA and for NOPSEMA to ass
- [Para [22](a)] Santos refers ECNT to paragraph 22c of the correspondence in relation to the matters raised in paragra Correspondence, which refers to the information provided risks and impacts, and describes the GHG management paragra
- [Para [22](b)] Santos refers ECNT to paragraph 22d of the correspondence in relation to the matters raised in paragra Correspondence.

In respect of ECNT's request for emissions breakdown by facilities, raised in paragraph 22(b) of the ECNT Correspondent paragraphs 22d and 22e of the Annexure to Santos' 30 Appresponds to substantially identical requests for an emission

- [Para [22](c)] Santos refers ECNT to paragraph 22aa of the correspondence and Part VIII of the Annexure to Santos' 1 relation to the matters raised in paragraph 22(c) of the ECI
- [Para [22](d)] Santos refers ECNT to paragraph 22ll of the correspondence in relation to the matters raised in paragra Correspondence.
- Santos further refers ECNT to paragraph 83 of the Annexu correspondence.
- [Para [22](e)] Santos refers ECNT to paragraphs 22x and April 2024 correspondence and Part VIII of the Annexure to correspondence in relation to the matters raised in paragra Correspondence.
- [Para [23](a)] Santos refers ECNT to paragraph 22cc of the correspondence in relation to the matters raised in paragra Correspondence.
- [Para [23](b)] Santos refers ECNT to paragraphs 22i, 22p, Santos' 30 April 2024 correspondence in relation to the ma ECNT Correspondence.
- [Para [23](c)] Santos refers ECNT to paragraph 46 of the correspondence in relation to the matters raised in paragra Correspondence.

alleging Santos' letters of 30 April and 14 May 2024

did not provide sufficient

information in response to a

number matters relevant to

the ECNT's functions,

interests or activities -

those matters include:

emissions:

water: and

0

(Paras 6 - 38)

risks, impacts and

control measures

related to GHG

risks and impacts

noise impacts on

marine fauna.

related to produced

n the Production Operations information	
ning is quite clear, which is that the	
tion provided about GHG emission	
consequences with information at the	
unctions, interests and activities. As	
In the present context is the	
as to avoid or reduce those emissions,	
climate change. ECINT evidently has	
currate change. It is certainly the	
out GHG emissions estimated to be	
roduction EP, so as to allow ECNT to	
line under the Safeguard Mechanism	
infurtherance of Australia's Paris	
acceptability of environmental impacts	
EP, for assessment by the Regulator	
odology for this evaluation of	
gy adopted for previous Barossa EPs,	
frastructure Installation Environment	
arealy directed to action and any area	
argery unected to setting out numerous	
hose matters have been considered by	
ondence, the consultation requirement	
nd to all the objections raised, or to	
- '	
sufficient detail about each of the eight	
that additional or more detailed	
ration in respect of the matters set out	
024 correspondence identified in	
ters that (in Santos' view) are for	
Sess.	
e Annexure to Santos' 30 April 2024	
aph 22(a) of the ECNT	
about GHG emissions and associated	
ian.	
e Annexure to Santos' 30 April 2024	
aph 22(b) of the ECN I	
emission-tupe for the EPSO and DLNG	
ndence Santos refers FCNT to	
oril 2024 correspondence which	
ns breakdown.	
he Annexure to Santos' 30 April 2024	
14 May 2024 correspondence in	
NT Correspondence.	
e Annexure to Santos' 30 April 2024	
aph 22(d) of the ECNT	
ure to Santos' 14 May 2024	
22w of the Annovure to Senter 20	
22W UI LITE ATTITEXULE TO SATITOS 30	
20 satilles 14 ividy 2024 and 22(a) of the ECNT	
ADIT 22(G) OF THE LOINT	
ne Annexure to Santos' 30 April 2024	
aph 23(a) of the ECNT	
, 22u, 22ff and 22gg of the Annexure to	
atters raised in paragraph 23(b) of the	
Annexure to Santos' 14 May 2024	
aph 23(c) of the ECNT	

• [Pa cor	ara [23](d)] Santos refers ECNT to paragraph 22p of the Annexure to Santos' 30 April 2024 rrespondence and Part C of the Annexure to Santos' 14 May 2024 in relation to the matters
• [Pa coi	ara [23](e)] Santos refers ECNT to paragraph 22u of the Annexure to Santos' 30 April 2024 rrespondence in relation to the matters raised in paragraph 23(e) of the ECNT
Co	prrespondence.
Sa	antos further refers ECNT to Part C of the Annexure to Santos' 14 May 2024 correspondence.
• [Fa	rrespondence and paragraphs 22g and 22h of the Annexure to Santos' 30 April 2024
coi	rrespondence in relation to the matters raised in paragraph 25 of the ECNT Correspondence.
• [Pa coi	ara [26]] Santos refers ECNT to paragraph 12 of the Annexure to Santos' 14 May 2024 rrespondence in relation to the matters raised in paragraph 26 of the ECNT Correspondence.
• [Pa	ara [27]] Santos refers ECNT to paragraphs 13 and 21 of the Annexure to Santos' 14 May
202 Co	24 correspondence in relation to the matters raised in paragraph 27 of the ECNT prrespondence.
• [Pa	ara [28](a)] Santos refers ECNT to paragraph 12 of the Annexure to Santos' 14 May 2024
cor Co	prrespondence in relation to the matters raised in paragraph 28(a) of the ECN I
Sa	antos further refers ECNT to Santos' response to paragraph [22](b) above.
• [Pa	ara [28](b)] Santos refers ECNT to Santos' response to paragraph [22](a) above.
Ma	ay 2024 correspondence in relation to the matters raised in paragraph 28(c) of the ECNT
Co	prrespondence.
Sa	antos further refers ECNT to Santos' response to paragraph [22](d) and (e) above.
coi	rrespondence in relation to the matters raised in paragraph 28(d) of the ECNT
Co	
• [Pa	ara [28](e)] Santos refers ECNT to Santos' response to paragraph [23](c), (d) and (e) above.
202	24 correspondence in relation to the matters raised in paragraph 28(f) of the ECNT
Co	prrespondence.
(an	mong other things) purchase or surrender of ACCUs or SMCs. Santos refers ECNT to page 22
of S acc	Santos' 2023 Annual Report, which provides further information on our generation and quisition of carbon credits as follows:
uot	In 2023, Santos executed agreements to build a portfolio of projects supporting the
	development of five nature-based projects across Queensland, Alaska and Papua New Guinea, to generate carbon credits, Further, in 2023 Santos entered into forward contracts
	for the purchase of 2.5 million ACCUs at fixed prices to be delivered and paid between
• [Pa	ara [28](g)] Santos refers ECNT to paragraph 22m of the Annexure to Santos' 30 April 2024
cor Co	rrespondence in relation to the matters raised in paragraph 28(g) of the ECNT
• [Pa	ara [28](h)] Santos refers ECNT to paragraph 32 of the Annexure to Santos' 14 May 2024
COI	rrespondence in relation to the matters raised in paragraph 28(h) of the ECNT
Sa	antos further refers ECNT to paragraph 22m of the Annexure to Santos' 30 April 2024
COI	rrespondence.
• [Pa COI	rrespondence in relation to the matters raised in paragraph 28(i) of the ECNT
Co	prrespondence.
Sa	antos turther refers ECNT to paragraphs 221 and 22u of the Annexure to Santos' 30 April 2024 rrespondence.
• [Pa	ara [29]] Santos refers ECNT to paragraph 27 of the Annexure to Santos' 14 May 2024
COI	rrespondence in relation to the matters raised in paragraph 29 of the ECNT Correspondence.
• [Pa 202	24 correspondence in relation to the matters raised in paragraph 30 of the ECNT
• IP=	orrespondence. ara [31]] Santos refers ECNT to paragraphs 40, 60 and 61 of the Annexure to Santos' 14 May
202	24 correspondence in relation to the matters raised in paragraph 31 of the ECNT
Co	prrespondence.
• [Pa Ma	ara [32](a)] Santos refers EUNT to paragraphs 8, 13-21 and 40 of the Annexure to Santos' 14 av 2024 correspondence in relation to the matters raised in paragraph 32(a) of the ECNT
Co	prrespondence.

				•
		•	[Para [32](b)] Santos refers ECNT to paragraphs 32 and 37 of the Annexure to Santos' 14 May 2024 correspondence in relation to the matters raised in paragraph 32(b) of the ECNT	
		•	Correspondence. [Para [32](c)] Santos refers ECNT to paragraph 65 of the Annexure to Santos' 14 May 2024 correspondence in relation to the matters raised in paragraph 32(c) of the ECNT Correspondence.	
			Santos further refers ECNT to Santos' response to paragraph [28](d) above. [Para [32](d)] Santos' consultation materials provide details of GHG emissions over the life of the project. The Australian Government via mechanisms such as the Safeguard Mechanism sets	
			the controls on emissions within Australia, which Santos must comply with. Santos' international customers' home jurisdictions are signatories to the Paris Agreement. Therefore, Santos' international customers must comply with the requirements their respective governments set to achieve their Paris Agreement commitments. Santos has provided the ECNT with sufficient information to allow it to make an informed assessment of the possible consequences of the activity to be carried out under the proposed Production EP on onv of ECNT's functions.	
		•	[Para [32](e)] Santos refers ECNT to paragraph 22m of the Annexure to Santos' 30 April 2024 correspondence and our response to [32](d) above in relation to the matters raised in paragraph	
		•	32(e) of the ECNT Correspondence. <b>[Para [32](f)]</b> Santos refers ECNT to paragraphs 25 and 37 of the Annexure to Santos' 14 May 2024 correspondence and our response to [32](d) above in relation to the matters raised in paragraph 32(f) of the ECNT Correspondence, which respond to substantially identical requests	
		•	in relation to emissions reduction targets. [Para [32](g)] Santos refers ECNT to Santos' response to paragraphs [23](d) and [28](f) above. [Para [32](h)] Santos refers ECNT to paragraph 22u of the Annexure to Santos' 30 April 2024 correspondence and paragraph 32 of the Annexure to Santos' 14 May 2024 correspondence in	
			relation to the matters raised in paragraph 32(h) of the ECNT Correspondence in relation to the matters raised in paragraph [32](h) of the ECNT Correspondence. [Para [32](i)] Santos refers ECNT to Santos' response to paragraph [29] above.	
		•	[Para 33] The GHG emissions associated with the end use of Barossa products are expected to be managed under the emissions framework each customer country has agreed through their Paris Agreement NDCs and/or net zero commitments.	
			Japan are on track to meet Paris Agreement commitments. [Para 34] Santos refers ECNT to Santos' response to paragraph [22](d) above. [Para 35] Santos refers ECNT to paragraphs 22gg and 22rr of the Annexure to Santos' 30 April	
			2024 correspondence and paragraph 78 of the Annexure to Santos' 14 May 2024 correspondence in relation to the matters raised in paragraph 35 of the ECNT Correspondence, which responds to similar requests in relation to noise impacts of the activity. Santos further refers ECNT to the Santos Dorado Development Offshore Project Proposal, published on NOPSEMA's website in July 2021, which contains at Attachment 11 the Noise	
		•	Impacts on Marine Fauna document. [Para 36] Santos refers ECNT to paragraph 22aa of the Annexure to Santos' 30 April 2024 correspondence and paragraph 85 of the Annexure to Santos' 14 May 2024 correspondence in relation to the matters raised in paragraph 36 of the ECNT Correspondence.	
		•	[Para 37] Santos refers ECNT to Santos' response to paragraph [21] above. [Para 38] Santos refers ECNT to Santos' response to paragraph [16] above. [All remaining paragraphs] The remaining paragraphs do not raise any requests for	
			information. As per the response to paragraphs 17 to 20 above, Santos has considered the remaining matters in your correspondence. The consultation requirement in ss 25(1)(d) and 25(2) does not otherwise require Santos to respond to all of the objections raised, or to persuade ECNT of the merits of the proposal.	
The ECNT reiterated concerns	Santos has considered the matters raised by ECNT in the Coastal Waters OEMP.	Sai	ntos has considered [these] matters.	Section 6.3
that, in the ECNT's view, the environmental impacts and risks of the GHG emissions	The Activity's projected emissions are contextualised against established emissions budgets (national and global). Climate change is a global problem with the solution being led at the international level. The impacts on the climate cannot be attributed to one specific sector or activity.			BAO-CM-6.3.9 BAO-CM-6.3.11 BAO-CM 6.2.12
associated with the Activity have not been reduced to ALARP or acceptable levels for the following reasons:	The Barossa Project's overall GHG emissions and proposed management are addressed in the Barossa Production Operations EP.			BAU-UNI-0.3.12
	Santos' assessment is that it has reduced impacts and risks from Activity GHG emissions to ALARP and acceptable levels.			
The Activity is not consistent	Santos has considered the matters raised by ECNT in the Coastal Waters OEMP.	Sai	ntos has considered [these] matters.	Section 6.3
with warming limit scenarios and carbon budgets - in particular, the goals of the Paris	Santos has adopted environmental performance outcomes and control measures directed to minimising the potential of the Activity to contribute to the accumulation of GHG emissions globally. Even assuming that			BAO-CM-6.3.11

Agreement and achievement of net zero by 2050 (paras 41 – 47);	emissions from the Activity will cause an equivalent net increase in cumulative Australian and global emissions, this increase is de minimis in the context of Australian and global carbon budgets; and there is no correlation between where GHG emissions are released and where climate change impacts are felt.	
Santos has not adequately assessed the indirect impacts of GHG emissions to climate change, including cumulative impacts (paras 48 – 55);	Santos has considered the matters raised by ECNT in the Coastal Waters OEMP. The Barossa JV has adopted environmental performance outcomes and control measures directed to minimising the GHG emissions from the Activity. A range of controls have been considered for both direct (Scope 1) and indirect (Scope 3) emissions. Santos considers that GHG emissions have been reduced to ALARP and are acceptable. This OEMP and the Barossa Production Operations EP both address the concerns raised.	Santos has considered [these] matters.
The ECNT asserted that Santos has not considered the physical risks to the project itself from climate change, given physical climate change effects are expected to worsen over the life of the project (para 56);	Santos has considered the matters raised by ECNT in the Coastal Waters OEMP. The OEMP includes details on the operational design life of the Barossa GEP, which is 25 years and is designed to withstand 100-year cyclonic metocean conditions.	Santos has considered [these] matters.
Santos incorrectly interpreted the meaning 'significant impact' in the context of considering the project's GHG emissions impact on the environment (paras 57 – 60);	Santos has considered the matters raised by ECNT in the Coastal Waters OEMP. Santos disagrees with this ECNT statement. The contribution of emissions from the Activity in the GEP Coastal Waters OEMP are contextualised against Australian and global net carbon budgets.	Santos has considered [these] matters.
It is unclear what FPSO facilities have been optimised and how reduction of GHG emissions has occurred (paras 61 – 62);	Concerns related to Production Operations activities in the Barossa Field (including the FPSO facility) are outside the scope of the GEP Coastal Waters OEMP and are addressed in the Barossa Production Operations EP.	<ul> <li>[Para 62] Santos refers ECNT to paragraph 22h of the Anne correspondence in relation to the matters raised in paragraph which responds to substantially identical requests in relation confirms that the Production EP Information Booklet is the mexpected Barossa reservoir emissions (noting that these are pursuant to the Safeguard mechanism obligations Santos is</li> <li>[All remaining paragraphs] The remaining paragraphs do information. As per the response to paragraphs 17 to 20 above, Santos in your correspondence. The consultation requirement in ss otherwise require Santos to respond to all of the objections merits of the proposal.</li> </ul>
Santos has not proposed mitigation measures sufficient for the project's expected scope 1 and 3 GHG emissions. (paras 63 – 67);	Santos has considered the matters raised by ECNT in this OEMP and provided a response to ECNT. The Barossa JV has adopted environmental performance outcomes and control measures directed to minimising the GHG emissions from the Barossa Gas Project. A range of controls have been considered for both direct (Scope 1) and indirect (Scope 3) emissions. The OEMP demonstrates that the proposed mitigation measures do reduce the potential impacts associated with GHG emissions to ALARP. Management of the emissions from the Barossa Gas Project and in accordance with the Safeguard Mechanism and customer country NDC targets will ensure that this Activity does not have an unacceptable impact on climate change, as the Activity emissions have been considered as part of Australia's NDC and, therefore, also into the global trajectory to limit global warming in line with the Paris Agreement targets. In any event, since the ECNT's review of the Barossa Production EP, Santos has proposed additional control measures in relation to GHG emissions.	<ul> <li>[Para 63] Santos refers ECNT to paragraph 25 of the Annex correspondence in relation to the matters raised in paragraph</li> <li>[All remaining paragraphs] The remaining paragraphs do information. As per the response to paragraphs 17 to 20 above, Santos I in your correspondence. The consultation requirement in ss otherwise require Santos to respond to all of the objections merits of the proposal.</li> </ul>
<ul> <li>Santos has not clarified how it intends to comply with the Safeguard Mechanism, including but not limited to:</li> <li>whether the FPSO and DLNG facility will be treated as the same facility or different facilities;</li> <li>which production variables the best practice emissions intensity number or the defaults emissions intensity number will apply to;</li> <li>the quantum of ACCUs it has purchased to date and expects to purchase for the project; (paras 68 – 77);</li> </ul>	Santos can confirm that it will comply with the Safeguard Mechanism, noting that treatment of Barossa GHG emissions is still to be finalised by the CER. Implementation and enforcement of the Safeguard Mechanism is the responsibility of the CER. Concerns related to the FPSO and DLNG are outside the scope of the GEP Coastal Waters OEMP and are addressed in respective approvals for the FPSO (Barossa Production Operations EP) and DLNG.	Santos has considered [these] matters.



	Section 6.3 BAO-CM-6.3.9 BAO-CM-6.3.11 BAO-CM-6.3.12
	Section 2.3.1
	Section 6.3
Annexure to Santos' 30 April 2024 agraph 62 of the ECNT Correspondence, lation to FPSO emissions. Santos the most up to date information on se are required to be net zero from start-up tos is subject to). s do not raise any requests for ntos has considered the remaining matters in ss 25(1)(d) and 25(2) does not ions raised, or to persuade ECNT of the	Not applicable.
Annexure to Santos' 14 May 2024 agraph 63 of the ECNT Correspondence. s do not raise any requests for ntos has considered the remaining matters in ss 25(1)(d) and 25(2) does not ions raised, or to persuade ECNT of the	Section 6.3 BAO-CM-6.3.1 BAO-CM-6.3.9 BAO-CM-6.3.11 BAO-CM-6.3.12 BAO-CM-6.3.16 BAO-CM-6.3.17 BAO-CM-6.3.18 BAO-CM-6.3.21 BAO-CM-6.3.22
	Section 6.3 BAO-CM-6.3.9 BAO-CM-6.3.11

Santos has not clarified the role of CCS/CCUs with regard to the Activity (paras 78 – 85); and	CCS is outside the scope of the Coastal Waters OEMP.	Santos has considered [these] matters.
Santos has not clarified its approach to obtaining offsets under the Safeguard Mechanism, including whether these are removal, abatement or avoidance offsets (paras 86 – 94).	Santos has considered the matters raised by ECNT in this OEMP and provided a response to ECNT. Santos can confirm that it will comply with the Safeguard Mechanism, noting that treatment of Barossa GHG emissions is still to be finalised by the CER. Implementation and enforcement of the Safeguard Mechanism is the responsibility of the CER.	Santos has considered [these] matters.
The ECNT asserted that compliance with the Safeguard Mechanism should not be considered a control measure (para 96).	Santos has considered the matters raised by ECNT in this OEMP and provided a response to ECNT Santos can confirm that it will comply with the Safeguard Mechanism, noting that treatment of Barossa GHG emissions is still to be finalised by the CER. Implementation and enforcement of the Safeguard Mechanism is the responsibility of the CER.	Santos has considered [these] matters.
<ul> <li>The ECNT asserted that the EP should not be submitted until Santos:</li> <li>provides clarity regarding the application of the Safeguard Mechanism on the Barossa project, including:</li> <li>clarification as to how the Safeguard Mechanism applies to the Barossa facilities and baselines for each facilities;</li> <li>the means in which it will manage excess emissions requirements under the scheme;</li> <li>disclosing details regarding the offsets it has already obtained for scope 1 GHG emissions; and</li> <li>clarifies the role of CCS/CCUS.</li> <li>(para 97).</li> </ul>	Santos has considered the matters raised by ECNT in this OEMP and provided a response to ECNT Santos can confirm that it will comply with the Safeguard Mechanism, noting that treatment of Barossa GHG emissions is still to be finalised by the CER. Implementation and enforcement of the Safeguard Mechanism is the responsibility of the CER. CCS is outside the scope of the Coastal Waters OEMP.	Santos has considered [these] matters.
ECNT correspondence to Santo	s on 25 November 2024	I
ECNT re-stated it is the peak community sector environment organisation in the Northern Territory. ECNT re-stated it is a Relevant Person for the purposes of reg 11A(1) of the OPGGS(E)R and its function to advocate for the nature, climate and environment of the NT, part of which is geographically covered by the EMBA of the draft EP. Asserted that it has been unable to make an informed assessment of the possible consequences of the activity on its functions, interests or activities.	<ul> <li>Santos has considered the matters raised by ECNT in the Coastal Waters OEMP.</li> <li>Santos' does not agree with ECNT's assertion that it has been unable to make an informed assessment of the possible consequences of the activity on its functions, interests or activities.</li> <li>Santos has made genuine attempts to consult constructively with the ECNT on the Barossa Gas Project (including GEP Coastal Waters section) since February 2024, notwithstanding that: <ul> <li>the ECNT's stated objective to 'Stop Barossa Gas' and that its interests and activities appear directed to stopping the activity under the EP and the OEMP from occurring at all which is inconsistent with the purpose of the consultation and the object of the legislation.</li> <li>ECNT's action in both delaying the 25 November letter to Santos and in withholding information raised in its 22 October letter to NOPSEMA – indicate that ECNT has failed to engage in a meaningful two-way conversation and consult in good faith in line with the purpose of consultation.</li> </ul> </li> <li>Santos has repeatedly invited the ECNT to provide to provide input and information relevant to the mitigation of environmental impacts and risks, which Santos may not be aware of, and to provide any comments it may have in relation to potential measures that the ECNT wishes Santos to consider adopting in order to mitigate environmental impacts and risks; the ECNT has largely not engaged.</li> <li>Further, the ECNT has not demonstrated how its functions, interests and activities may be affected by the relevant activities.</li> <li>Santos has, nonetheless, continued its efforts to consult and engage constructively with the ECNT regarding the relevant activities, including now engaging with the information that the ECNT has withheld from Santos.</li> </ul>	<ul> <li>Santos' correspondence to ECNT on 20 December 2024 in a November 2024 as well as correspondence sent by ECNT s NOPSEMA.</li> <li>ECNT's letter to NOPSEMA</li> <li>During the course of NOPSEMA's assessment of Revision 1 of with a copy of a letter from the ECNT to NOPSEMA dated 22 O NOPSEMA).</li> <li>We observe as follows: <ol> <li>Prior to the 25 November letter to Santos, Santos' last relation to the EP was on 7 August 2024.</li> <li>During the 3.5 months between 7 August 2024 and 25 response from the ECNT in relation to its letter of 7 Au</li> <li>Santos was not copied on the ECNT's 22 October letter subsequently provide Santos with a copy.</li> <li>More than a month later, the ECNT sent its 25 Novemin 5. The majority of that 25 November letter to Santos cont amendments) from the ECNT's 22 October letter to NO substantively new assertions made by ECNT in the 25 not made in the 22 October letter to NOPSEMA.</li> </ol> </li> </ul>
	impacts of the Barossa Gas Project (including GEP Coastal Waters section) on the ECNT's functions, interests	



	N/A
	Section 6.3 BAO-CM-6.3.9 BAO-CM-6.3.11
	Section 6.3
	Section 6.3
in response to ECNT's letter of 25 T separately on 22 October 2024 to	Not Applicable
of the EP, Santos has been provided 2 October 2024 (22 October letter to	
ast correspondence with the ECNT in 25 November 2024, Santos received no August 2024. etter to NOPSEMA, nor did the ECNT ember letter to Santos.	
NOPSEMA. That is, there are no 25 November letter to Santos that were	

and activities, and to provide input to Santos about the environment that may be affected by the Activity and/or additional proposed control measures for consideration by Santos in the course of preparing the OEMP.	<ul> <li>6. In fact, substantive comments and assertions regarding the EP which were made by the ECNT in the 22 October letter to NOPSEMA were not included by the ECNT in its 25 November letter to Santos.</li> <li>7. Further, the 22 October letter to NOPSEMA contains comments and assertions which have never been made by the ECNT to Santos in any of the consultation across more than 6 months – including five substantive exchanges of letters between the ECNT and Santos regarding the EP. These comments and assertions in the 22 October letter to NOPSEMA include that: <ul> <li>a. Santos has failed to adequately account for fugitive methane emissions at the DLNG plant in the EP, including that Santos:</li> <li>i. has failed to engage with recent research findings regarding the under-repeting of fugitive methane emissions and</li> </ul> </li> </ul>	
	<ul> <li>is required, at a minimum, to commit to both direct (at the offshore facility)</li> <li>and indirect (at DLNG) fugitive emissions monitoring for the activity, as</li> <li>well as become a signatory to the Oil &amp; Gas Methane Partnership 2.0;</li> <li>and</li> <li>b. there must be a case made in the EP that the environmental impacts and risks of</li> </ul>	
	<ul> <li>the activity will not forego the health, diversity and productivity of the environment for future generations, as required by a consideration of ESD principles, and that because Santos has failed to account for climate change impacts which partially result from the GHG emissions generated by the activity, it has failed to do so.</li> <li>8. Additionally, in the 22 October letter to NOPSEMA, the ECNT has referred to various reports and scientific research which it considers relevant to the assessment of GHG emissions</li> </ul>	
	<ul> <li>associated with the activities the subject of the EP. Such materials were not referred to in previous correspondence with Santos, or raised during consultation, despite Santos previously inviting the ECNT to provide information and raise control measures which it considers would be appropriate to adopt when preparing the EP for submission to NOPSEMA.</li> <li>9. The first time that Santos was made aware of the 22 October letter to NOPSEMA was on 27 November 2024, when Santos was provided with a copy by NOPSEMA.</li> </ul>	
	Santos is concerned that the matters above – including most notably the ECNT's action in both delaying the 25 November letter to Santos and in withholding information raised in its 22 October letter to NOPSEMA – indicate that ECNT has failed to engage in a meaningful two-way conversation and consult in good faith in line with the purpose of consultation.	
	As the ECNT has itself identified, a core purpose of consultation under the Offshore Petroleum and Greenhouse Gas Storage (Environment) Regulations 2023 (Cth) (Regulations) is to ensure that relevant persons have an opportunity to provide input and information relevant to the mitigation of environmental impacts and risks, which titleholders and NOPSEMA may otherwise not be aware of. Santos has repeatedly invited the ECNT to provide any such information, and to provide any comments it may have in relation to potential measures that the ECNT wishes Santos to consider adopting in order to mitigate environmental impacts and risks.	
	The ECNT has largely not engaged with this invitation, and asserts in its 25 November letter to Santos that the quality of information provided by Santos during consultation has:	
	'prevented ECNT from fully understanding the potential consequences of the Activity on its FIAs and from effectively engaging in the consultation process and allowing us to provide the necessary input, including regarding additional control measures, needed to fulfil the content requirements of Santos' Production EP.'	
	More than a month before making this assertion to Santos, the ECNT suggested in the 22 October letter to NOPSEMA that Santos ought to commit to direct (at the offshore facility) and indirect (at DLNG) fugitive emissions monitoring. That is, the ECNT has, in effect, suggested to NOPSEMA that an additional control measure be considered. The ECNT has made no effort to raise this potential control measure with Santos.	
	Santos reiterates that throughout the course of consultation with the ECNT on the EP, since February 2024, Santos has made genuine attempts to consult constructively with the ECNT, notwithstanding the ECNT's stated objective to 'Stop Barossa Gas' and that its interests and activities appear directed to stopping the activity under the EP from occurring at all which is inconsistent with the purpose of the consultation and the object of the legislation.	
	In particular the object of the Offshore Petroleum and Greenhouse Gas Storage Act 2006 (Cth) is to provide an effective regulatory framework for petroleum exploration and recovery and the object of the Regulations is to ensure that any petroleum activity carried out in an offshore area is:	
	<ul> <li>carried out in a manner consistent with the principles of ecologically sustainable development set out in section 3A of the EPBC Act: and</li> </ul>	
	<ul> <li>carried out in a manner by which the environmental impacts and risks of the activity will be reduced to as low as reasonably practicable; and</li> </ul>	

		• carried out in a manner by which the environmental impacts and risks of the activity will be of an acceptable level.	
		The ECNT's conduct in relation to consultation and its role in the "Stop Barossa Gas" Campaign is indicative of a broader objective to materially either delay or prevent the Barossa Gas Project from proceeding as opposed to reducing the environmental impacts and risks of the relevant activities to as low as reasonably practicable and acceptable. Further, the ECNT has not demonstrated how its functions, interests and activities (FIA) may be affected by the relevant activities.	
		Santos has, nonetheless, continued its efforts to consult and engage constructively with the ECNT regarding the relevant activities, including now engaging with the information that the ECNT has withheld from Santos.	
	Santos has responded to ECNT.	Consultation in preparation of EP	Not Applicable
ECNT provided a summary of its consultation with Santos: Asserted that publication of the Draft EP on NOPSEMA's website is the first time ECNT has had access to certain information of sufficient particularity informing its ongoing assessment of the possible consequences of the activity and proposed control measures. Asserted that Santos has repeatedly refused to answer its requests for specific information, instead referring ECNT to its information booklet or the Offshore Project Proposal. Asserted that Santos previously indicated it would only consult with ECNT between 11 March and 9 April 2024 and had refused to extend this consultation period at ECNT's requests. Asserted that Santos had, by meeting with ECNT on 20 May 2024 and on 12 June 2024 inviting further feedback by 20 June 2024, unilaterally changed the	Santos has responded to ECNT. Santos remains of the view that Santos has properly consulted with the ECNT to the standard required under section 25 of the Regulations. Santos has made genuine attempts to consult constructively with the ECNT on the Barossa Gas Project (including for the Coastal Waters OEMP) since February 2024 and has repeatedly invited the ECNT to provide to provide input and information relevant to the mitigation of environmental impacts and risks, which Santos may not be aware of, and to provide any comments it may have in relation to potential measures that the ECNT wishes Santos to consider adopting in order to mitigate environmental impacts and risks; the ECNT has largely not engaged. Santos confirms that the ECNT's comments were made in relation to Rev 1 of the Barossa Production Operations EP, and Santos' response was based on Rev 2 of the Barossa Production Operations EP. For completeness, many (although not all) of the concerns raised by the ECNT may be assumed to apply equally to the OEMP. The Barossa Production Operations EP has been publicly available to the ECNT since 1 October 2024 when the ECNT, together with all other recipients of the Barossa Quarterly Project Update, received a link to the Barossa Production Operations EP as published on NOPSEMA's website in the October Quarterly Update circulated on 15 October 2024. It is evident from the content on ENCT's recent letters to Santos and NOPSEMA that he ECNT had considered the substance of the published Barossa Production Operations EP and even with this additional information, the ECNT did not provide Santos any input regarding additional control measures, but in bad faith chose to raise such matters with NOPSEMA. Santos has provided the ECNT with sufficient information and a reasonable period to assess any possible impacts of the Activity for the Barossa Gas Project (which includes for Coastal Waters OEMP) on the ECNT's functions, interests and activities, and to provide input to Santos about the environment that may be affect	<ul> <li>withheld from Santos.</li> <li>Consultation in preparation of EP</li> <li>Each of the ECNT's letters conveys the ECNT's views as to the adequacy of consultation to date. Santos has considered the ECNT's views and remains of the view that Santos has properly consulted with the ECNT to the standard required under section 25 of the Regulations.</li> <li>In particular: <ul> <li>Santos generally disagrees with the matters alleged at paragraphs 5 and 7-10 of the 25 November 2024 letter to Santos, and notes that many of these matters are plainly incorrect or misleading on the face of consultation to date. For example: <ul> <li>the record of consultation set out at paragraph 5 of the 25 November letter to Santos is incomplete and fails to mention various other correspondence and the meeting between Santos and the ECNT on 20 May 2024. For a full summary of consultation up until September 2024, please see pages 350-351 of the EP; and</li> <li>contrary to paragraph 9 of the 25 November letter to Santos, Santos provided further information in response to queries from the ECNT on numerous occasions, including in letters dated 30 April 2024, 14 May 2024 and 7 August 2024.</li> </ul> </li> <li>Santos generally disagrees with the allegations at paragraphs 7 and 8-9 of the 25 November letter to Santos. We note that in any event, the EP has been publicly available to the ECNT since on or around 1 October 2024. The ECNT, together with all other recipients of the Barossa Quarterly Project Update, received a link to the EP as published on NOPSEMA's website in the October Quarterly Update circulated on 15 October 2024.</li> <li>Further, the comments included in the 25 November letter to Santos indicate that the ECNT has engaged with the contents of the EP. Indeed it is evident from the content replicated between the ECNT siz 22 October letter to NORSEMA and 25 November letter to Santos that by at least 22 October 2024, the ECNT had considered the substance of the published EP. Santos notes that even with this additional inf</li></ul></li></ul>	Not Applicable
timeframes in which ECNT could engage in consultation, which was not in line with NOPSEMA's Consultation Guidelines.			
Asserted that, because of Santos' approach, ECNT has been prevented from effectively engaging in consultation and been prevented from being able to seek input from experts.			
Asserted that the primary information ECNT has had available to it is Santos' 36- page booklet, despite			

repeated advice to Santos that the booklet did not contain information of sufficient specificity for ECNT to make an informed assessment about the impacts of the Activity and Santos had refused to provide further information. Asserted that information Santos has provided has been vague, evasive and lacking in sufficient detail for ECNT to be adequately informed of the environmental impacts and risks of the Activity and this had prevented it from effectively engaging in the consultation process			
ECNT asserted that the information provided by Santos during consultation was in many cases inconsistent with that provided in the Draft EP. Asserted that information related to the greenhouse gas emissions of the Activity is contradictory, both within the draft EP and from the information provided in the Information Booklet. (Note: ECNT included a table in its letter which is provided in in entirety in the Sensitive Information Report attached to this OEMP.). Asserted that the table provided indicates the EP contains inconsistent estimates of the	Santos has responded to ECNT. This letter and the issues raised in it relate to the Barossa Production Operations EP, and in particular to the version of the EP published on NOPSEMA's website in September 2024. Santos has included this letter, and its response to this letter, for completeness noting that many (although not all) of the concerns raised by the ECNT may be assumed to apply equally to the OEMP. Santos has considered the matters raised by ECNT in the Barossa Production Operations EP and provided a response to ECNT. Santos confirms that the ECNT's comments were made in relation to Rev 1 of the Barossa Production Operations EP, and Santos' response was based on Rev 2 of the Barossa Production Operations EP. For completeness, many (although not all) of the concerns raised by the ECNT may be assumed to apply equally to the OEMP. Santos does not agree with ECNT's claim that information related to the greenhouse gas emissions of the Barossa Gas Project are contradictory, both within the Draft Barossa Production Operations EP itself and from the information booklet. The annual emissions estimates provided in the information booklet are conservative and prepared for the purpose of impact and risk assessment. The Barossa Production Operations EP and this OEMP include greenhouse gas estimates for both Scope 1 and scope 3 emissions as relevant to each Activity including the assumptions that underpin the estimates. In any event, in circumstances where the GHG emissions estimates in the EP are lower than the estimates provided in the information booklet, the ECNT has been able to make an informed assessment of any potential	<ul> <li>Information provided during consultation</li> <li>Regarding the ECNT's claim that "information provided during consultation is in many cases inconsistent with that provided in the EP", Santos notes that the ECNT has previously raised similar claims in correspondence to Santos on 25 March 2024 and 9 April 2024 to which Santos responded in letters to the ECNT on 30 April 2024 and 14 May 2024 respectively.</li> <li>Further to responses provided in previous correspondence to the ECNT, Santos re-iterates the following points:</li> <li>The annual emissions estimates included in the Production Operations information booklet are applicable to the Production Operations Activity (Activity). The annual emissions estimates provided in the information booklet are conservative (when extrapolated for 25 years of production operations) given annual emissions are expected to reduce over the life of the Activity as production rates decline.</li> <li>Emissions estimates provided in the information booklet and the EP are prepared for the purpose of the impact and risk assessment and control measure evaluation, based on the best available information and assumptions at the time of preparing the EP, but are not a definitive forecast of future actual emissions. Actual emissions during production operations will be reported as per applicable regulatory requirements of the National Greenhouse and Energy Reporting Act 2007 (Cth) (NGER Act).</li> </ul>	Section 6.3
Inconsistent estimates of the greenhouse gas GHG emissions of the activity. Asserted that the Scope 3 emissions estimates are so different as to potentially indicate a substantial flaw or uncertainty in emissions figures calculations. Asserted that this discrepancy reflects a broader failure to appropriately consult us on this matter and a failure to meet the requirements of the EP.	consequences of GHG emissions on its functions, interest or activities because the GHG emissions of the activity are actually less than what ECNT understood them to be. It is not clear what impact any such discrepancy has had on the ECNT's understanding of potential consequences of GHG emissions, and it is notable that the ECNT has not identified the basis of any such impact.	<ul> <li>While the ECNT's observations about apparent discrepancies between GHG emissions in the information booklet and the EP are noted and have been considered by Santos, to the extent that there is any potential impact on the ECNT's FIAs as a result of GHG emissions, the ECNT is able to make an informed assessment of that impact with the information set out in the information booklet, which adopted more conservative assumptions.</li> <li>In any event, the simple fact is that the GHG emissions estimates in the EP are lower than the estimates provided in the information booklet, for the 25 year lifecycle of Barossa production operations (commissioning and steady state operations). We assume that the ECNT is not suggesting that it has not been able to make an informed assessment of any potential consequences of GHG emissions on its FIAs because the GHG emissions of the activity are actually less than what ECNT understood them to be. It is not clear what impact any such discrepancy has had on the ECNT's understanding of potential consequences of GHG emissions from Barossa are the basis of any such impact.</li> <li>In any event, the ECNT explicitly notes at paragraph 40 of its 22 October letter to NOPSEMA that its views as to the acceptability of GHG emissions from Barossa are the same 'regardless of which estimate is used', again indicating that the ECNT has no intention of consulting in good faith or for the proper purpose of consultation under the Regulations.</li> </ul>	
ECNT asserted that it had repeatedly requested a copy of an Operations GHG Emissions Management Plan (GHGEMP) during the course of consultation, but Santos refused to provide it. Further asserted that the discrepancies set out (in its supplied table) underscore the importance of Relevant Persons	Santos responded to previous ECNT requests for the GHGEMP and considers sufficient information has been provided in the Production Operations Information Booklet, the NT Waters GEP Operations Overview Factsheet and the Production Operations EP. The Barossa Production Operations EP includes a description of the GHGEMP and relevant components to this OEMP are presented in Section 8.3.2.12 of this OEMP. Santos does not agree with ECNT's claim that information related to the greenhouse gas emissions contains discrepancies. This OEMP includes greenhouse gas estimates for both Scope 1 and scope 3 emissions including the assumptions that underpin the estimates.	<ul> <li>Regarding the ECNT's request for a copy of the GHG Emissions Management Plan (GHGEMP), Santos notes that the ECNT has previously raised similar claims in correspondence to Santos on 25 March 2024 and 9 April 2024 to which Santos responded in letters to the ECNT on 30 April 2024 and 14 May 2024 respectively.</li> <li>Further to responses provided in previous correspondence to the ECNT, Santos re-iterates the following points:</li> <li>The Production Operations information booklet (and the EP) provide a description of the purpose of the GHGEMP. The EP also describes the measures that will be adopted in the GHGEMP to keep direct emissions to ALARP and acceptable levels and in accordance with Santos' obligations under the Safeguard Mechanism. Accordingly, Santos does not consider</li> </ul>	Section 6.3 Section 8.3.2.12



having access to the proponent's GHGMP.	the ECNT requires a copy in order to make an informed assessment of the potential consequences of the activity on any of its ELAs which remain unspecified	
	The Production Operations information booklet (and the EP) detail the potential risks and impacts related to GHG emissions. The ECNT is necessarily familiar with these potential risks and impacts. through previous consultation with Santos on other activities for Barossa.	
ECNT requested clarity relating to the use of CCS and whether it is expected to be used as an emissions mitigation technology at any point over the lifecycle of the Activity.	Venture, and are outside therefore does notRegarding the ECNT's requests for further information about use of CCS as a control measure, Santos notes that the ECNT has previously raised similar claims in correspondence to Santos on 25 March 2024, 9 April 2024 and 20 June 2024 to which Santos responded in letters to the ECNT on 30 April 2024, 14 May 2024 and 7 August 2024 respectively; and at a meeting between the ECNT and Santos on 20 May 2024. Further to responses provided in previous correspondence to the ECNT, Santos re-iterates that while Santos is exploring CCS opportunities at Bayu-Undan and elsewhere, these CCS projects are not owned or controlled by the Barossa joint venture, are not part of the Barossa development and are not within the scope of the Production Operations EP.Not Applie	icable
ECNT asserted that Santos had failed to demonstrate compliance with relevant legislation, including the Commonwealth Safeguard Mechanism and related rules and regulations. Asserted that the only atempt made by Santos to demonstrate compliance with the Safeguard Mechanism is the claim that DCCEEW has provided and assurance that sufficient ACCUs or SMCs are evailable until 2035. Asserted that Santos has not demonstrate how many offsets will be required to be purchased, the source of these offsets, nor the source of the source and guarantees as to the integrity of these offsets. Asserted that a representative of Santos had representative of Santos had representative of Santos had paid between December 2023 and January 2027. Asserted that ECNT wrote to Santos had paid between December 2023 and January 2027.	<ul> <li>It can term in use adapter of the induction of participations in the induced and induced</li></ul>	5.3.2.3.2

BAS-210 0224



compliance with relevant legislation.			
ECNT requested further consultation with Santos to seek further clarity on the concerns outlined in its letter (of 25 November 2024).	Santos has made genuine attempts to consult constructively with the ECNT on the Barossa Gas Project (including for the GEP Coastal Waters section) since February 2024 and has repeatedly invited the ECNT to provide to provide input and information relevant to the mitigation of environmental impacts and risks, which Santos may not be aware of, and to provide any comments it may have in relation to potential measures that the ECNT wishes Santos to consider adopting in order to mitigate environmental impacts and risks; the ECNT has largely not engaged. Santos has provided the ECNT with sufficient information and a reasonable period to assess any possible impacts of the Activity for the GEP Coastal Waters OEMP on the ECNT's functions, interests and activities, and to provide input to Santos about the environment that may be affected by the Activity and/or additional proposed control measures for consideration by Santos in the course of preparing the GEP Coastal Waters OEMP.	Santos notes the ECNT's request at paragraph 27 of the 25 November letter to Santos for further consultation in order to seek clarity on the concerns set out in the ECNT's letter. For the reasons set out above, Santos disagrees that there is any uncertainty impacting the ECNT's ability to make an informed assessment of any consequences of the activity under the EP on any of its FIAs. Nor is it clear to Santos what further consultation is sought. Indeed, it is a matter of longstanding and extensive public record that the ECNT has already formed an unequivocal assessment about the Barossa Development and associated activities. As just one of many examples, the website stopbarossagas.org clearly states, among other things, in relation to the Barossa Gas Project, that "our planet can't afford another huge gas development." The website also states that the Stop Barossa Gas campaign is a collaboration between the ECNT, Jubilee Australia Research Centre, Solutions for Our Climate (KO) and Japan Centre for a Sustainable Environment and Society. Having regard to the matters outlined above in relation to the ECNT's 22 October letter to NOPSEMA, Santos does not consider that the ECNT's request for further consultation is genuine or required. The ECNT is acutely aware that Santos is targeting commencing operations under the EP in 2025. Santos has been transparent about its (more than reasonable) timeframes for consultation.	Not Applicable
The following are comments a	and assertions made in ECNT's correspondence to NOPSEMA on 22 October 2024 that were not include	ed in ECNT's letter to Santos on 25 November 2024.	
[7] The publication of the EP is the first time the ECNT has been provided with information of sufficient particularity to be able to make an informed assessment of the possible consequences of the activity and proposed control measures.	Santos has responded to ECNT. This letter and the issues raised in it relate to the Barossa Production Operations EP, and in particular to the version of the EP published on NOPSEMA's website in September 2024. Santos has included this letter, and its response to this letter, for completeness noting that many (although not all) of the concerns raised by the ECNT may be assumed to apply equally to the OEMP. Santos has considered the matters raised by ECNT in the Barossa Production Operations EP and provided a response to ECNT. Refer below for assessment of matters raised. Santos confirms that the ECNT's comments were made in relation to Rev 1 of the Barossa Production Operations EP, and Santos' response was based on Rev 2 of the Barossa Production Operations EP. Santos does not agree with ECNT's claim that information related the Barossa Gas Project in the information booklet was insufficient.	Santos notes that in the 22 October letter to NOPSEMA, the ECNT confirms that the ECNT was provided with the information in the EP, and that information is 'of sufficient particularity to be able to make an informed assessment of the possible consequences of the activity and proposed control measures'. The equivalent paragraph in the 25 November letter to Santos has been amended such that it no longer contains this confirmation, with the ECNT maintaining (to Santos) that it still does not have sufficient information. While Santos maintains that information provided to the ECNT during consultation (before the publication of the EP) met Santos' regulatory obligations, Santos notes that in any event, the ECNT by at least 22 October 2024 considered it had sufficient information to make an informed assessment of the potential consequences of the activity on its FIAs. It is a matter of longstanding and extensive public record that the ECNT has already formed an unequivocal assessment about the Barossa Development and associated activities. As just one of many examples, the website stopbarossagas.org clearly states, among other things, that "Our planet can't afford another huge gas development." The website also states that the Stop Barossa Gas campaign is a collaboration between the ECNT, Jubilee Australia Research Centre, Solutions for Our Climate (KO) and Japan Centre for a Sustainable Environment and Society.	Not Applicable
[19] It would be unlawful for NOPSEMA to accept the EP.	This is raised in regards the Barossa Production Operations EP and therefore not considered relevant to this GEP Coastal Waters OEMP. These concerns have been assessed within the Barossa Production Operations EP.	It is not a matter for the ECNT to determine or opine on whether or not it would be unlawful for NOPSEMA to accept the EP. Santos considers such a statement to constitute an implied threat to NOPSEMA (and therefore Santos) that the ECNT may litigate if NOPSEMA decides to accept the EP in any circumstances. Under the law, NOPSEMA must accept the EP if it is reasonably satisfied that the EP meets the EP acceptance criteria. There is no requirement under the law for the ECNT to be satisfied.	Not Applicable
[21]-[24] There are a number of inconsistencies, deficiencies and inadequacies in the EP that demonstrate that the EP cannot be accepted by NOPSEMA.	This is raised in regards the Barossa Production Operations EP and therefore not considered relevant to this GEP Coastal Waters OEMP. These concerns have been assessed within the Barossa Production Operations EP.	It is not a matter for the ECNT to determine or opine on why the EP cannot be accepted by NOPSEMA. Santos considers such a statement to constitute an implied threat to NOPSEMA (and therefore Santos) that the ECNT may litigate if NOPSEMA decides to accept the EP. Under the law, NOPSEMA must accept the EP if it is reasonably satisfied that the EP meets the EP acceptance criteria. There is no requirement under the law for the ECNT to be satisfied. Consultation under section 25 does not require a titleholder to demonstrate to a relevant person that the acceptance criteria in section 34 have been satisfied. Nor is NOPSEMA required to demonstrate this to a relevant person.	Not Applicable
[31] As the existence of the GHGEMP is a key control measure, it is unacceptable that it is not included in the EP itself and is not yet finalised or approved. These factors render the plan unsuitable as a control measure.	Santos responded to previous ECNT requests for the GHGEMP and considers sufficient information has been provided in the Production Operations Information Booklet (which includes Coastal Waters), the NT Waters GEP Operations Overview Factsheet and the Production Operations EP. The Barossa Production Operations EP includes a GHGEMP, and relevant components to this OEMP are presented in Section 8.3.2.12 of this OEMP.	The suitability or otherwise of the GHGEMP is a matter for NOPSEMA to consider in its assessment of the EP. As outlined in the EP at sections 6.3.5.2 and 8.8.9, the GHGEMP governs management of GHGs more broadly than in relation to activities which will be regulated under the EP and accordingly which are within NOPSEMA's jurisdiction. In any event, it is common for titleholders to give effect to control measures under EPs through policies or documents which are not themselves included in the EP. The fact that the GHGEMP itself is not included in the EP does not render it ineffective as a control measure. BAO-CM-6.3.12 sets out the key measures that the GHGEMP will implement (which are described in considerable detail in Section 8.8.9), which Santos will be bound to comply with as a control measure under the EP.	Section 8.3.2.12 Section 6.3.3 BAO-CM-6.3.12
[32] Santos has not adequately accounted for fugitive emissions in the EP. Santos has not	Concerns related to the DLNG are outside the scope of the GEP Coastal Waters OEMP and are addressed in the Production Operations EP and respective approvals for the DLNG.	The EP outlines the sources of fugitive emissions and the basis of estimates in numerous sections of the EP (refer to sections 6.3.1, 6.3.2.1, 6.3.2.1.1, 6.3.2.7.2 and 6.3.2.8). The information provided pertains to the likely sources of fugitive emissions at the FPSO; fugitive emissions as an estimated	Not Applicable



i		1	1
attempted to account for fugitive emissions at DLNG.		total of Scope 1 emissions over a 25 year period; as well as the estimated percentage of total emissions (at steady state operating).	
		With respect to fugitive emissions at DLNG, these are Scope 3 emissions at a facility owned by a different joint venture. The Barossa JV does not own or control DLNG. The DLNG facility, is outside the scope of the EP). The information included in the EP relating to DLNG is based on information provided by DLNG.	
[33] As the GHGEMP is not included in the EP, no attempt has been made to reduce risks and impacts of fugitive emissions.	Santos responded to previous ECNT requests for the GHGEMP and considers sufficient information has been provided in the Production Operations Information Booklet (which includes Coastal Waters), the NT Waters GEP Operations Overview Factsheet and the Production Operations EP. The Barossa Production Operations EP includes a GHGEMP, and relevant components to this OEMP are presented in Section 8.3.2.12 of this OEMP.	Santos refers to its response above. Potential fugitive emissions have been factored into the direct emissions of the Activity. It is incorrect to suggest that "no attempt has been made to reduce risks and impacts of fugitive emissions". Numerous control measures have been proposed that relate to the installation, monitoring and maintenance of plant and equipment, each of which will perform a function in the minimisation of fugitive emissions.	Section 6.3.3 BAO-CM-6.3.12 Section 8.3.2.12
[34] Santos has not engaged with research that industry under-reports fugitive emissions, nor with evidence of uncontrolled methane leakage at its own facilities.	Santos has considered the matters raised by the ECNT in this GEP Coastal Waters OEMP and provided a response to the ECNT. Santos takes into consideration a wide range of research, including the IEEFA research referred to by the ECNT, and to the extent that it is valid and relevant, takes it into account in its management of fugitive emissions. Fugitive emissions from pipeline transmission of dry gas, to the extent those fugitive emissions could occur in the part of the GEP within NT is Coastal Waters, is addressed in Section 6.3.2. Fugitive emissions are reported for all operating assets in accordance with requirements of the NGER Act and the National Greenhouse and Energy Reporting (Measurement) Determination 2008. All emissions (including fugitives) from the Barossa project will also be reported in compliance with these legislative requirements. Loss of containment, such as methane leakage poses a personal safety and facility integrity risk as well as loss of saleable product, as such maintaining equipment and facility integrity, monitoring and repairing any leaks are a top priority for the Barossa JV. See also Santos' consideration of and response to these matters below, in response to ECNT's letter to Santos of 24 January 2025.	Santos reviews and engages extensively with a wide range of research, including the IEEFA research referred to by the ECNT, and to the extent that it is valid and relevant, takes it into account in its management of fugitive emissions. Where there is evidence of methane leakage (for instance, during an unforeseen loss of containment) Santos always takes steps to learn from such incidents and improve its equipment and/ or procedures as relevant. Santos currently reports fugitive emissions from all operated assets in accordance with requirements of the NGER Act and the National Greenhouse and Energy Reporting (Measurement) Determination 2008. All emissions (including fugitives) from the Barossa project will also be reported in compliance with these legislative requirements. Santos notes that the Clean Energy Regulator would also be aware of the IEEFA research referred to by the ECNT. Methane leakage, also known as Loss of Containment, at oil and gas facilities pose a risk to personal safety and facility integrity well as the environment and represents a loss of saleable product. Therefore, maintaining equipment and facility integrity, monitoring and repairing any leaks are a top priority for Santos.	Section 6.3.2 Section 8.3.2.12
[35] At a minimum, the Proponent should be required to commit to direct and indirect fugitive emissions monitoring for the Activity, as well as being a signatory to the Oil & Gas Methane Partnership 2.0.	This is raised in regards the Barossa Production Operations EP and therefore not considered relevant to this GEP Coastal Waters OEMP. These concerns have been assessed within the Barossa Production Operations EP. The Barossa Project has a GHGEMP relevant to both the FPSO and the GEP. This is presented in Section 8.3.2.12 of this OEMP	Santos refers to its response above in relation to fugitive emissions. In any event, signing a corporate initiative is not a control measure and this appears to be inconsistent with other statements the ECNT has made about suitable control measures and performance objectives.	Section 6.3 Section 8.3.2.12 BAO-CM-6.3.12 Not Applicable
[36]-[38] Santos' reasoning as to why GHG emissions from the Activity will be ALARP and acceptable is flawed, including due to reliance on the WAM scenario.	Santos has considered the matters raised by the ECNT in this GEP Coastal Waters OEMP and demonstrated that GHG emissions from the Activity will be ALARP and acceptable. The reference to the Australian carbon budget to 2050 is based on Australia's Emissions Projections 2024 and the Annual Climate Change Statement 2024, published by the Department of Climate Change, Energy, the Environment, and Water (DCCEEW). These documents provide an emissions trajectory forecast to 2050, which outlines Australia's commitment to achieving net zero by 2050, in line with the Paris Agreement (i.e. net zero). This is a matter for the Regulator to determine, not the ECNT.	This is a matter for NOPSEMA to determine, not the ECNT. Notwithstanding this, the reference to the Australian carbon budget to 2050 is <u>based</u> on Australia's Emissions Projections 2023 and the Annual Climate Change Statement 2023, published by the Department of Climate Change, Energy, the Environment, and Water (DCCEEW). These documents provide an emissions trajectory forecast to 2050, which outlines Australia's commitment to achieving net zero by 2050, in line with the Paris Agreement (i.e. net zero). The 'with additional measures' scenario corresponds to the modelled emissions trajectory included in the Emissions Projections 2023. This scenario reflects policies such as the 82% renewable energy target by 2030 and Safeguard Mechanism reforms, which are consistent with Australia's net zero commitments. The linear extrapolation from 2030 to 2050 used to develop the net carbon budget (7966 Mt CO <sub>2</sub> -e) aligns with the framework outlined in these documents, comprising gross economy-wide emissions (additions) less total carbon sequestration volumes (subtractions). Santos considers the 'net' GHG emissions of the Activity to reflect the true contribution of the Activity to Australia's carbon budget, given that reservoir emissions will be required to be net zero in accordance with existing requirements under the NGER Act and the Safeguard Mechanism. Santos has considered both Scope 1 and Scope 3 contributions to this budget. Additionally, when calculating a percentage, it's crucial that the numerator and denominator are consistent in scope. If the denominator is expressed in net terms (i.e., it reflects deductions or adjustments), then the numerator must also represent net values to ensure alignment and maintain logical and mathematical consistency This prevents any "double counting" or inaccuracies.	Not Applicable
[39] It is more appropriate GHG emissions to be put in the context of the relevant sectoral budget. Doing so results in the activity constituting around 4% of the relevant sectoral budget.	Santos has considered the matters raised by the ECNT in this GEP Coastal Waters OEMP and provided a response to ECNT. GHG emissions contribution from the Activity have been contextualised against Australian and global carbon budgets. International frameworks, namely the Paris Agreement, have been developed to facilitate an orderly approach to what is a global problem. This is a matter for he Regulator to determine, not the ECNT.	This is a matter for NOPSEMA to determine, not the ECNT. Santos has sought to contextualise the contribution of emissions from the Activity in the Coastal Waters OEMP. against Australian and global carbon budgets. International frameworks, namely the Paris Agreement, have been developed to facilitate an orderly approach to what is a global problem. The nature, quantity and timeframe of each country's contribution and the pathways to achieve UNFCCC obligations vary widely, including having regard to the particular circumstances of each country. This framework recognises, and is premised on, the fact that climate change is a global issue – there is no correlation between where GHG emissions are released and where climate change impacts are felt.	Not Applicable

		For this reason, Australia sets and reports against its emissions reduction targets in 'net' terms, not by individual sectors or projects. The Australian Government is aware of planned production in the Barossa gas field and has considered this in its emissions reduction targets outlined in its NDC under the Paris Agreement.	
		information provided by Santos. Notably, this calculation could have been performed by the ECNT on the basis of the estimates provided in the information booklet in March 2024.	
[40] Regardless of which estimate is used, it is unacceptable for a single project to compose such a significant amount of Australia's remaining emissions budget.	Santos has considered the matters raised by the ECNT and provided a response to the ECNT. This may be a view of the ECNT, but it is not a fact. The acceptability of GHG emissions related to this Activity is a matter for DME to determine, not the ECNT.	This may be a view of the ECNT, but it is not a fact. The acceptability of GHG emissions is a matter for NOPSEMA to determine, not the ECNT. Such acceptability must be considered in the context that the Offshore Project Proposal was previously approved by NOPSEMA and a production licence conferring rights to extraction of the gas resource on the Barossa joint venture has already been granted. The GHG emissions from the Activity have already been considered in Australia's NDC under the Paris Agreement.	Not Applicable
		The ECNT's response explicitly notes that its view remains the same 'regardless of which estimate is used'. That is, it was open to the ECNT to provide this response on the basis of the original estimates provided in March 2024. This directly contradicts with the ECNT's assertion that it has insufficient information about GHG emissions to make an informed assessment of the consequence of GHG emissions.	
		It is a matter of longstanding and extensive public record that the ECNT has already formed an unequivocal assessment about the Barossa Development and associated activities. As just one of many examples, the website stopbarossagas.org clearly states, among other things, that "Our planet can't afford another huge gas development." The website also states that the Stop Barossa Gas campaign is a collaboration between the ECNT, Jubilee Australia Research Centre, Solutions for Our Climate (KO) and Japan Center for a Sustainable Environment and Society.	
[41] Santos has failed to engage with the Climate Change Authority's 2024 Sector Pathways Review.	Santos has reviewed and engaged with the Sector Pathways Review. However, this review is not relevant to the Activities covered by this GEP Coastal Waters OEMP. Notwithstanding, Santos has considered the matters raised by the ECNT and provided a response.	Santos has reviewed and engaged with the Sector Pathways Review. However, this Review is not relevant to GHG emissions management for the activities covered by this EP. The purpose of consultation for this EP is to seek feedback and input to assist the proponent in minimising environmental risks and impacts in the course of carrying out the activities covered by the EP.	Not Applicable
		In any event, it is acknowledged in the review that "Ultimately, there are many sets of sector pathways that can combine to achieve net zero by 2050. Like nations under the Paris Agreement, sectors can be thought of as having 'common but differentiated responsibilities and respective capabilities' when it comes to meeting Australia's national targets. This report does not recommend specific emissions reduction targets for each sector. Instead, it sets out the abatement potential of each sector on Australia's pathway to net zero emissions by 2050." (page 9).	
		The Sector Pathways Review also acknowledges that 'gas will be required for some time for firming and back-up supply' (page 26).	
[42] Santos has omitted the CCA's scenarios from the EP.	Santos has considered the matters raised by the ECNT in relation to GHG emissions for the Barossa Project and provided a response to the ECNT. The CCA scenarios reflect technology transitions and emission pathways to support Australia's transition to net zero emissions by 2050.	The CCA scenarios reflect technology transitions and emission pathways to support Australia's transition to net zero emissions by 2050. The modelling, undertaken by the CSIRO, presents two scenarios in the review which set out potential technological and operational changes in each sector that if taken together could potentially deliver Australia's reduction targets.	Section 6.3 describes the safeguard mechanism and how Santos will
	Santos confirms that it will comply with the Safeguard Mechanism, noting that treatment of Barossa GHG emissions is still to be finalised by the CER. Implementation and enforcement of the Safeguard Mechanism is the responsibility of the CER.	Australia has a well-established legislative framework under which certain GHG emissions from Barossa production operations will be regulated or managed to further Australia's transition to net zero emissions by 2050. This includes:	comply EPO 11
	Compliance with the Safeguard Mechanism will ensure that net emissions generated from the Activity are congruent with the targets modelled in the CCA scenarios.	GHG emissions reporting under the NGER Act and the National Greenhouse and Energy Reporting (Measurement) Determination 2008	BAU-CIM-6.3.11
		the Emissions Reduction Fund (Australian Carbon Credit Units Scheme)	
		<ul> <li>the Safeguard Mechanism to keep net emissions below an established baseline and require net-zero reservoir emissions for new gas fields that feed LNG projects. The Safeguard Mechanism currently applies to facilities that emit more than 0.1 MtCO2-e per annum.</li> </ul>	
		Compliance with the Safeguard Mechanism is of primary importance in ensuring that GHG emissions associated with the Activity are as low as reasonably practicable and acceptable levels. GHG emissions at or below the baseline and the Safeguard Mechanism's future decline rates are already anticipated and thus accounted for under Australia's NDCs under the Paris Agreement. Thus, the net emissions generated from the Activity are congruent with the targets modelled in the CCA scenarios	
[43] The CCA has suggested CCS and electrification as possible emissions reductions control measures, both of which	This is raised in regards the Barossa Production Operations EP and therefore not considered relevant to this GEP Coastal Waters OEMP. These concerns have been assessed within the Barossa Production Operations EP.	As part of facility design aimed to reduce Scope 1 emissions, full electrification of the FPSO processing equipment, with combined cycle gas turbine power generation has been adopted. Details on this adoption, as well as several other design measures, are outlined in table 6-23 of the Barossa EP.	Not Applicable
have been rejected by Santos. This is an unacceptable inconsistency with best available		Carbon capture and storage has been rejected by Santos because it is not an available option for emissions reduction. Santos considered CCS to reroute reservoir emissions as a control measure, as outlined in table 6-22 of the EP, but it could not be adopted on the basis that these projects do not	



modelling of emissions		exist as vet. Potential CCS developments such as the proposed Bavu-Undan CCS project require	
reductions pathways.		regulatory frameworks, policies and approvals (from a different jurisdiction to the Activity) to be in place prior to taking final investment decisions. Therefore, it is not possible to rely on them for GHG emissions abatement. Santos is currently exploring CCS opportunities at Bayu-Undan and elsewhere. The Darwin Pipeline Duplication for the Barossa Gas Project was undertaken to facilitate the proposed Bayu-Undan CCS project. Front End Engineering and Design is continuing and the operator of that project is actively engaging with a range of stakeholders to progress the development.	
[44] Various research and reports support that emissions from the Activity will undermine the Paris Agreement targets and Australia's net zero emissions by 2050 goal.	The acceptability of GHG emissions is a matter for NOPSEMA to determine, not the ECNT. Notwithstanding, Santos has considered the matters raised by the ECNT and provided a response to the ECNT. It must be acknowledged that there are multiple viable pathways to achieve climate goals, depending on technological developments, policy decisions, economic conditions, and societal choices. These scenarios are tools to explore policy options and inform decision-making. They are not definitive or exclusive blueprints for the global climate response. The Paris Agreement under the UNFCCC framework is the most comprehensive global agreement for climate response.	Santos reviews and engages extensively with a wide range of research, including the IEA Net Zero Roadmap referred to by the ECNT and to the extent that it is valid and relevant, takes it into account in its management of GHG emissions. The acceptability of GHG emissions is a matter for NOPSEMA to determine, not the ECNT. Such acceptability must be considered in the context that the Offshore Project Proposal was previously approved by NOPSEMA and a production licence conferring rights to extraction of the gas resource on the Barossa joint venture has already been granted, and in the context that the GHG emissions from the Activity have already been considered in Australia's NDC under the Paris Agreement. In any event, the scenarios presented by the International Energy Agency (IEA), Intergovernmental Panel on Climate Change (IPCC), and the Climate Council are not prescriptive pathways. Rather, they illustrate a range of potential strategies to meet global emission reduction targets, such as those set out in the Paris Agreement and they rely on a range of assumptions about technology, human behaviour and other factors that must all be met to achieve the outcome of the pathway. There are multiple viable pathways to achieve climate goals, depending on technological developments, policy decisions, economic conditions, and societal choices. These scenarios are tools to explore policy options and inform decision-making. They are not definitive or exclusive blueprints for the global algreement for climate response. As stated by the IEA: " These scenarios are not predictions – the IEA does not have a single view on the future of the energy systemOur scenario analysis is designed to inform decision makers as they consider options, not to predict how they will act, and none of the scenarios should be viewed as a forecast." Similarly, the IPCC emphasizes: "Scientists use computer models to simulate the emissions of greenhouse gases that would be consistent with different levels of warming. The different possibilities ar	Not Applicable
[45]-[48]Santos' assertion that it is neither appropriate nor possible to quantify or attribute any specific impact on climate change to emissions from an individual project is incorrect, including having regard to the definition of environmental impact under the Regulations, outdated and unsound scientific research, and the intergenerational principle.	Santos has considered the matters raised by the ECNT and provided a response to the ECNT. International frameworks, namely the Paris Agreement, have been developed to facilitate an orderly approach to what is a global problem. The nature, quantity and timeframe of each country's contribution (including Australia) and the pathways to achieve UNFCCC obligations vary widely, including having regard to the particular circumstances of each country. This framework recognises, and is premised on, the fact that climate change is a global issue – there is no correlation between where GHG emissions are released and where climate change impacts are felt.	International frameworks, namely the Paris Agreement, have been developed to facilitate an orderly approach to what is a global problem. The nature, quantity and timeframe of each country's contribution and the pathways to achieve UNFCCC obligations vary widely, including having regard to the particular circumstances of each country. This framework recognises, and is premised on, the fact that climate change is a global issue – there is no correlation between where GHG emissions are released and where climate change impacts are felt. For this reason, Australia sets and reports against its emissions reduction targets in 'net' terms, not by individual sectors or projects. The Australian Government is aware of planned production in the Barossa gas field and has considered this in its emissions reduction targets outlined in its NDC under the Paris Agreement. The Offshore Project Proposal was previously approved by NOPSEMA and a production licence conferring rights to extraction of the gas resource on the Barossa joint venture has already been granted. Physical impacts of climate change on environmental receptors are the result of global GHG emissions from a multitude of sources (minus the GHG sinks) that have accumulated in the atmosphere. The impacts of climate change cannot be attributed to one specific sector or activity. In the context of evaluating potential impacts and risks that may be associated with GHG emissions from all sources globally, including from this Activity, Santos has considered the physical impacts of climate change to 6.3.2.2 of the OEMP.	Section 6.3.2.2
[49] There are no appropriate environmental performance outcomes, standards or measurement criteria for BAO- CM-6.3.19 (Barossa products generated from the Activity will only be sold to customers from countries that are signatories to the Paris Agreement, as at the	This is raised in regards the Barossa Production Operations EP and therefore not considered relevant to this GEP Coastal Waters OEMP. These concerns have been assessed within the Barossa Production Operations EP.	This is a matter for NOPSEMA to consider in its assessment of the EP.	Not Applicable



This is raised in regards the Barossa Production Operations EP and therefore not considered relevant to this GEP Coastal Waters OEMP. These concerns have been assessed within the Barossa Production Operations EP.	Santos rejects this assertion in its entirety. Santos has made clear that Barossa would be a potential cust CCS project if the project were to proceed. However, the proje decision and, notwithstanding, this is unrelated to the Activity a in the EP. It is subject to a different regulatory regime and purs one of the joint venture parties for Bayu-Undan CCS and has o project proceeds, but this is reliant on approvals from a foreigr investment decisions by the proponent of that project. If the Ba Barossa JV hopes to be a customer of it. Again, though, this is Barossa JV. Linking the delivery of the Activity to an unrelated, potential fut wholly inappropriate, as it would be grossly disproportionate to of NOPSEMA's Environment Plan Content Requirements Guid result in the Barossa Gas Project not proceeding in circumstar for emissions offsets and reductions
Concerns related to the DLNG are outside the scope of the GEP Coastal Waters OEMP and are addressed in the Production Operations EP and respective approvals for the DLNG.	This is a matter for NOPSEMA to determine, not the ECNT. Santos notes that NOPSEMA does not regulate DLNG and, in should be able to rely on the assessment of another Australiar relevant to its consideration of the activities covered by this EF
This is raised in regards the Barossa Production Operations EP and therefore not considered relevant to this OEMP. These concerns have been assessed within the Barossa Production Operations EP.	Santos has covered risks of a condensate spill at section 7.7 of Previous condensate spills at other facilities have been regular legislation and approvals documents, and, as is always the cas from previous spills and implements continuous improvement spills. The learnings from previous condensate spills have bee for preventing and mitigating condensate spills relevant to the
Santos has considered the matters raised by the ECNT in this GEP Coastal Waters OEM and provided a response to the issue/s raised by ECNT. Santos does not agree that compliance with applicable legislation is not in itself a control measure to mitigate environmental risk, nor is a mere assertion to demonstrate ALARP and acceptability. Santos has considered the matters raised by the ECNT and provided a response.	All approvals processes in Australia are given in anticipation of legislative and common law frameworks of the relevant jurisdic range of assessment processes to determine if the proponent an activity and there are also a range of financial and other ass financial assurance requirements are in place and suitability as production licence was granted. In addition, applicable legislation generally includes reporting, to achieve required outcomes.
bs on 24 January 2025	1
Santos disagrees with the ECNT's characterisation of Santos' response. Santos' letter of 20 December 2024 included the foundations of each of Santos' views.	Santos' correspondence to ECNT on 6 March 2025 in resp 2025 Santos has considered the ECNT's response regarding the ma December 2024. Santos maintains that it has provided sufficie assessment of the impact of the activities on the ECNT's funct
Santos has considered the matters raised by the ECNT and responded.	provide input to the development of the Production Operations made genuine attempts to address the concerns of the ECNT Operations EP.
	This is raised in regards the Barossa Production Operations EP and therefore not considered relevant to this GEP Coastal Waters OEMP. These concerns have been assessed within the Barossa Production Operations EP.         Concerns related to the DLNG are outside the scope of the GEP Coastal Waters OEMP and are addressed in the Production Operations EP and respective approvals for the DLNG.         This is raised in regards the Barossa Production Operations EP and therefore not considered relevant to this OEMP. These concerns have been assessed within the Barossa Production Operations EP and respective approvals for the DLNG.         This is raised in regards the Barossa Production Operations EP and therefore not considered relevant to this OEMP. These concerns have been assessed within the Barossa Production Operations EP.         Santos has considered the matters raised by the ECNT in this GEP Coastal Waters OEM and provided a response to the issue's raised by EONT.         Santos does and argee that compliance with applicable legislation is not in itself a control measure to mitigate environmental risk, nor is a mere assertion to demonstrate ALARP and acceptability.         Santos these with the ECNT's characterisation of Santos' response.         Santos does with the ECNT's characterisation of Santos' response. Santos' letter of 20 December 2024 included the foundations of each of Santos' views.         Santos has considered the matters raised by the ECNT and responded.



omer of the proposed Bayu-Undan ct has not yet taken a final investment and it would be improper to link the two sued by a different proponent. Santos is committed funding to it. It hopes that the government (Timor-Leste) and final nyu-Undan CCS project proceeds, the beyond the control of Santos or the ure project via a control measure is the benefit gained (see section 3.6.3 lance Note) and could unnecessarily rese where there are viable alternatives	Not Applicable
	Not Applicable
an orderly regulatory framework, it regulator to the extent that the issue is	Ποι προιοαυίο
f the EP. Ted in accordance with the relevant se, Santos has implemented learnings measures to prevent and mitigate future n used by Santos to inform processes activities covered by this EP.	Not Applicable
f the proponent's compliance with the tion. Across jurisdictions, there are a is a suitable person to be carrying out surances. In the case of Barossa, ssessments were required when the monitoring and enforcement provisions	Not Applicable
onse to ECNT's letter of 24 January atters raised in Santos' letter of 20 Int information for the ECNT to make an ions, interests and activities, and to Environment Plan. Further, Santos has in the development of the Production	Not Applicable

by NOPSEMA, which is neither unusual nor unreasonable; and • the ECNT made new comments to NOPSEMA because it now had the benefit of information in the Draft EP. ECNT alleged that, contrary to Santos' letter of 20 December 2024, ECNT had described the impact of the EP on its functions, interests and activities in previous correspondence to Santos, and Santos had never previously asserted any failure by ECNT to demonstrate how it	Santos has considered the matters raised by the ECNT. Santos notes that the ECNT has a different view to Santos on the extent to which the ECNT has demonstrated that the Activity has any consequence on any of its functions, interests or activities. In any event, notwithstanding the comments made to ECNT, Santos has consulted with ECNT extensively as a Relevant Person.			
was affected. The ECNT said that its concerns about environmental impacts from GHG emissions are not alleviated by the fact that GHG emissions estimates are lower in the Draft EP than those provided in consultation. In contrast, the ECNT alleged that this raises queries about the veracity of information provided in consultation and how the estimates have been lowered, emphasising the ECNT's concerns about lack of transparency and its calls for the GHGEMP to be published.	Santos has considered the matters raised by the ECNT. Santos responded to previous ECNT requests for the GHGEMP and considers sufficient information has been provided in the Production Operations Information Booklet, the NT Waters GEP Operations Overview Factsheet and the EP. The Barossa Production Operations EP includes a description of the GHGEMP and relevant components to this OEMP are presented in Section 8.3.2.12 of this OEMP The fact that the ECNT's concerns about environmental impacts have not been alleviated is a separate issue. The point is that the ECNT has been provided with sufficient information in order to form those concerns. Otherwise, Santos relies on its previous consideration of these issues, which were raised by the ECNT in its letter of 25 November 2024 and are considered above.	Santos has previously respondent Mechanism and consultation We note for completeness the contextual consideration of the for the latest data published <b>Australian Carbon Budget</b> As reflected in Australia's NH Australia is committed to a s 2005 levels by 2030 and net Based on a 43% reduction be (DCCEEW, 2024). Assuming for this EP), this creates a migross economy wide emission	nded to the ECNT and maintains the GHG emission 3 of GHG emissions wit by DCCEEW. Thi DC under the Paris ingle year target to -zero by 2050. by 2030, the net can g a further linear de et carbon budget o ons (additions) less	's assertions regotes responses. If the Production thin national can s analysis has r Agreement and reduce greenh rbon budget for ecline between 2 f 7,262 Mt CO2 s total carbon se
The ECNT alleged that Santos is required under the Regulations to describe the requirements applying to the project and how those requirements will be met, including the Safeguard Mechanism.	Santos accepts that this is correct and notes that this obligation has been discharged – see section 6.3 and Appendix C. Santos notes that it has previously confirmed to ECNT that the EP and OEMP must demonstrate how Santos will comply with the Safeguard Mechanism – most recently in its letter of 20 December 2024. This is also clear from the Production Operations EP, to which the ECNT has had access since around September 2024.	Stage	Lifecycle gross emissions (MtCO2e)	Lifecycle ne emissions (MtCO2e)
<ul> <li>The ECNT alleged that its interests include:</li> <li>seeing the NT and Commonwealth comply with emissions targets; and</li> <li>ascertaining whether the Activity poses a threat to the NT and Commonwealth's compliance with these legislated emissions reduction commitments.</li> </ul>	<ul> <li>Santos has considered the matters raised by the ECNT in the EP and OEMP and provided a response to the ECNT.</li> <li>Santos notes that the ECNT considers that it has interests which are affected by the Activity in the manner described. As outlined immediately above, Santos agrees that the EP must describe the requirements of the Safeguard Mechanism and demonstrate how those requirements will be met. Also as outlined above, section 6.3 provides this detail and demonstration in relation to the Safeguard Mechanism. For completeness, Santos confirms that:</li> <li>the Safeguard Mechanism is administered by the Clean Energy Regulator; and</li> <li>compliance with the Safeguard Mechanism is itself a proposed control measure under this EP (see BAO-CM-6.3.10 and BAO-CM-6.3.11), in respect of which compliance may be monitored by DME.</li> </ul>	Operations and maintenance <sup>[2]</sup> Operations at DLNG and support operations Product transport and end use Totals [1] Out to 2050 [2] Net-zero reservoir emis [3] End-user combustion w	Scope 1           53.96           Scope 3           32.46           188.52           274.94           sions           ill occur outside Au	19.17 32.46 188.52 <b>240.15</b>
The ECNT enclosed a bundle of documents it had obtained under Freedom of Information (FOI) regarding the DLNG facility and a letter from ECNT to NOPSEMA dated 4 December 2024	Although these assertions were made by the ECNT to NOPSEMA (rather than being raised with Santos directly), Santos has considered and evaluated the matters raised by ECNT, to the extent relevant to the Activity under the Production Operations EP. DLNG emissions are out of scope of this GEP Coastal Waters OEMP. As outlined in section 6.3.2, the DLNG facility is owned by a different joint venture to the Barossa Development and its operation is outside the scope of the Activity and outside the control of the Barossa Joint Venture.	Santos notes ECNT's reque containment'. The DLNG fac Joint Venture and is not regu place and in line with regula program.	st for the disclosur ility is owned and d ulated by NOPSEN tory requirements,	e of further deta operated by a so IA. All regulator tank emissions



garc Op bon esu	ling the GHGEMP, the Safeguard erations EP, Santos has updated its budgets and projections to account lted in the following figures:	Section 6.3 Section 8.3.2.12 Appendix C BAO-CM-6.3.11
this 203 -e. eque	e Climate Change Act 2022, e gas emissions to 43% below period is 4,377 Mt CO2-e 0 and 2050 (the full activity lifecycle The net carbon budget comprises estration volumes (subtractions). by change to Santos' environmental	
et	Barossa Production Operations Net GHG Emissions Contribution (%) Australian carbon budget (Mt CO2e) <sup>[1]</sup>	
	0.3%	
	0.4%	
	NA <sup>[3]</sup>	
	VII /0	

re	egarding the FOI	The framework by which DI NG is regulated is relevant to the evaluation of downstream emissions. That	As outlined in the Production Operations EP, emissions from the DLNG facility are separately
	ocuments The ECNT's	framework includes the reporting and management of fugitive emissions in accordance with requirements of the	managed in accordance with the Safeguard Mechanism including emissions reporting in accordance
		NGER Act and the National Greenhouse and Energy Reporting (Measurement) Determination 2008. That	with the National Greenhouse and Energy Reporting (NGER) determination. The NGER
IE	etter to NOPSEMA made	from and the head of the the NET EDA	determination includes a companyon for furthing emissions. The Sonne 2 emissions estimates
V	arious assertions about the	Tranework also includes regulation by the NT EPA.	determination includes a component for lugitive emissions. The Scope 3 emissions estimates
F	OI documents, the DLNG		Included in the Production Operations EP in respect of the operation of the DLNG facility are based
fa	acility and NOPSEMA's		on a nolistic forecast of DLNG emissions in processing Barossa feed gas, which also includes a
a	ssessment of the		component for fugitive emissions.
	reduction Operations EP		Santas maintains that the amissions astimates included in the Production Operations EP are
			Santos maintains that the emissions estimates included in the Froudction Operations EF are
Ir			appropriate. The matter is subject of origoning regulatory engagement with the relevant regulators and
1	. [5] the FOI documents		being managed in accordance with applicable regulatory requirements and approvals.
	indicate that there is an		
	ongoing methane leak at		
	the DLNG storage tank.		
	with no apparent		
	intention to repair before		
	processing Percess and		
	The size of the leak is		
	unclear but subsequent		
	surveys indicate different		
	and potentially larger		
	amounts of fugitive		
	methane emissions and		
	that mathana amiasiana		
	are larger than has been		
	reported or estimated;		
2	. [6] ECNT's		
	understanding based on		
	publicly available		
	information is that the		
	DI NG tank will continue		
	to look when used to		
	to leak when used to		
	store Barossa gas;		
3	. [7]-[11], [23] atmospheric		
	emissions from the		
	processing of Barossa		
	gas at the DLNG facility		
	are indirect		
	consequences of the		
	A stivity we don the		
	Activity under the		
	Production Operations		
	EP;		
4	. [12(a)], [26]-[27] there		
	may be safety risks at		
	DLNG that are not dealt		
	with in the FP In other		
	jurisdictions leaks have		
	been required to be		
	communities evacuated		
	in the meantime;		
5	. [12(c)-(d)] Santos may		
	have misled NOPSEMA		
	through omitting 'critical		
	relevant information' from		
	the Draft EP and has		
	also failed to provide		
	important information to		
	the public and Delayers		
	the public and Relevant		
	Persons;		
6	. [13]-[15], [31]-[32]		
	NOPSEMA should not		
	accept the EP because		
	as a result of the EP's		
	failure to disclose the		
	information in the FOI		
	documente the ED dece		
1	documents, the EP does		



not meet the EP	
acceptance chiena,	
7. [16] increased methane	
emissions is an indirect	
impact and risk that is	
not disclosed in the EP	
or demonstrated to be	
ALARP and acceptable;	
8 [12/b)] [17] while the	
o. [12(b)], [17]  write the	
calculation of scope 3	
omissions in the ED is	
unclear, atmospheric	
aminaiana fram DLNC	
emissions from DLING	
are likely to exceed the	
estimates included in the	
EP and therefore to	
affect the conclusions	
regarding Australia's	
carbon budget and the	
assessment of negligible	
consequences;	
9. [18]. [24] the EP also	
does not consider	
localised impacts at	
DLING from methane,	
including potential health	
impacts from air quality;	
10, [19], [25], [31], [32(b)-(c)]	
the EP addresses	
various approvals at	
DLNG but these	
approvals do not deal	
with the issue, nor is it	
clear that appropriate	
action has been taken	
under the approvals	
NOPSEMA cannot defer	
to the assessment of an	
NT regulator and in any	
event the NT	
assessment did not	
consider the leak:	
consider the leak,	
11. [20] it is not sufficient for	
monitoring of ongoing	
monitoring of ongoing	
fugitive emissions to be	
dealt with in the	
GHGEMP and not in the	
EP:	
12. [21] the extent of the	
methane leak is unclear	
and the amissions	
and the emissions	
estimate must be	
algoritical by Contact	
clarified by Santos;	
13, [22] the implementation	
attrate and in the CD data	1
strategy in the EP does	
in a forma a foto a	
not meet the	
not meet the	
requirements of the	
requirements of the Regulations because it	
requirements of the Regulations because it	
requirements of the Regulations because it does not provide for	
requirements of the Regulations because it does not provide for sufficient monitoring and	
requirements of the Regulations because it does not provide for sufficient monitoring and	
requirements of the Regulations because it does not provide for sufficient monitoring and recording of fugitive	
requirements of the Regulations because it does not provide for sufficient monitoring and recording of fugitive emissions at DLNG:	
requirements of the Regulations because it does not provide for sufficient monitoring and recording of fugitive emissions at DLNG;	
requirements of the Regulations because it does not provide for sufficient monitoring and recording of fugitive emissions at DLNG; 14. [28]-[30] Santos' failure	
requirements of the Regulations because it does not provide for sufficient monitoring and recording of fugitive emissions at DLNG; 14. [28]-[30] Santos' failure to consult in relation to	
requirements of the Regulations because it does not provide for sufficient monitoring and recording of fugitive emissions at DLNG; 14. [28]-[30] Santos' failure to consult in relation to	
requirements of the Regulations because it does not provide for sufficient monitoring and recording of fugitive emissions at DLNG; 14. [28]-[30] Santos' failure to consult in relation to the methane leak at	





consultation with Relevant Persons is incomplete and unacceptable, and Relevant Persons may not have been identified and consulted by Santos;			
[33]-[35] ECNT is considering next steps including making reports to relevant NT authorities.			
The ECNT noted its belief as to the seriousness of an uncontained methane leakage, and asked Santos to confirm:	Santos has considered the matters raised by the ECNT. Santos refers to and repeats its detailed consideration of the matters raised in the ECNT's letter to NOPSEMA, in the Barossa Production Operations EP. In any event, the ECNT's letter (including the enclosed FOI documents and the letter to NOPSEMA) will be included in the sensitive information part of the EP for NOPSEMA's consideration, to the extent NOPSEMA considers the matters raised by the ECNT are relevant.		
<ul> <li>whether the leak discussed in the documents has been fixed; and</li> <li>what steps were being taken to monitor and prevent future leaks.</li> <li>The ECNT asserted that Santos ought to disclose the matter to NOPSEMA because the matter has consequences for NOPSEMA's assessment of the EP.</li> </ul>			
The ECNT alleged that its consultation to date has been genuine and it has engaged with integrity, and it remains committed to consulting.	Santos notes the ECNT's response. Santos disagrees for the reasons expressed above and in Santos' letter of 20 December 2024. In any event, with the exception of the queries about the DLNG facility, the ECNT has not specified what further consultation is sought.	Santos maintains that the Production Operations EP includes a comprehensive evaluation of impacts and risks from the Activity, and the ECNT has received sufficient information to make an informed assessment of the consequences of the activity on any of its functions, interests or activities. For the avoidance of doubt, Santos maintains that it considers consultation with ECNT is complete and that Santos has discharged its obligations under s 25 in respect of ECNT.	Not applicable.



#### 4.8.7 Commercial Fishing (Commonwealth / NT / WA managed)

### Table 4-17: Consultation Summary Table - Commercial Fishing (Commonwealth / NT / WA managed)

Section 25(1)(d) of the OPGGS(E)R: Persons or organisations whose functions, interests or activities may be affected by the activities to be carried out under the environment plan

Commercial Fishing: Commonwealth-managed fisheries

Northern Prawn Fishery (NPF) Licence Holders

Licence Holders are consulted via their representative body, Northern Prawn Fishery Industry Pty Ltd. Where an NPF Licence Holder is also an NT Commercial Fishing Licence Holder they are also consulted via the NT Seafood Council. Refer to NPFI, NTSC entries in Table 4-20 (Industry associations) and licence holder entries in Table 4-15 (Commercial fishing) for the summary of consultation effort.

Southern Bluefin Tuna/ Western Skipjack Tuna and Western Tuna and Billfish Fisheries Licence Holders

Licence Holders are consulted via their representative body, the ASBTIA.

Refer to ABSTIA entry in Table 4-20 (Industry associations) for the summary of consultation effort.

North-West Slope Trawl Fishery Licence Holders

Licence Holders are consulted via their representative body, CFA.

Refer to CFA entry in Table 4-20 (Industry associations) for the summary of consultation effort.

#### **Torres Strait Fishery**

Licence Holders are consulted via their representative body, the CFA.

Refer to CFA entry in Table 4-20 (Industry associations) for the summary of consultation effort.

**Commercial Fishing:** WA-managed fisheries Licence Holders (entitled to fish in the EMBA) - Abalone Fishery, Kimberley Prawn Fishery, Marine Aquarium Fishery, Northern Demersal Scalefish Managed Fishery, Pilbara Crab Fishery, South-West Costal Salmon Fishery, Specimen Shell Fishery, West Coast Deep Sea Crustacean Fishery

Licence Holders are consulted via their representative body, the WAFIC.

Refer to WAFIC entry in Table 4-20 (Industry associations) for details.

Commercial Fishing: NT-managed fisheries Licence Holders (entitled to fish in the EMBA) - Aquarium Fishery, Development Fishery, Develo Timor Reef Fishery

Licence Holders are consulted via their representative body, NTSC, and directly where a Licence Holder has requested.

In addition to the entries below, also refer to the NTSC entry in this table.

#### A. Raptis and Sons

Summary of consultation effort:

- On 9 February 2024 Santos emailed A. Raptis & Sons to advise it of preliminary consultation regarding proposed activities for consultation to be managed under the Coastal Waters OEMP and Barossa Production Operations EP. [Con-3787]
- The email advised that Santos was seeking information to better understand:
  - o if you are from a government Department or agency, how the proposed Production Operations activities may be relevant to your department or agency; and
  - what (if any) functions, interests or activities you or your organisation have that may be affected by the proposed Production Operations activities.
- The email included information on the regulatory process for the activities in Commonwealth and NT jurisdictions and links to a Santos information booklet on the proposed activities and a NOPSEMA brochure on the consultation process and details of how to contact Santos to register as a Relevant Person. The email stated that the consultation phase would commence on 11 March 2024 and close on 9 April 2024.
- On 11 March 2024 Santos emailed A. Raptis & Sons further to the previous correspondence, to advise that it had commenced the consultation phase which would run until 9 April 2024. In addition to the previous information again being provided, Santos provided information on Relevant Persons' entitlements under the regulatory processes, details of how to provide feedback and a reminder of the closing date for consultation. [Con-3793]
- On 3 April 2024 Santos phoned A. Raptis & Sons which advised that it would not be commenting.
- On 6 May 2024, Santos emailed A. Raptis & Sons further to previous correspondence, to advise it had extended the consultation period until 20 May 2024. In providing this extension of time, Santos advised that the information in the booklet and factsheet had been updated to account for an additional risk associated with the proposed activity. Santos confirmed that consultation will close on the revised date unless Santos hears otherwise from A. Raptis & Sons. [Con-3867]
- On 10 July 2024 Santos emailed A. Raptis & Sons to advise the consultation period had been completed. Santos advised A. Raptis & Sons that it considered consultation had now closed for the purpose of Santos finalising and submitting environment plans for these activities to government regulators for assessment. [Con-5130]
- On 15 October 2024 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that the EP had been submitted to NOPSEMA for assessment, provided a link to the full EP on the NOPSEMA website and Santos' contact details for any queries. The Update also stated that the OEMP would be submitted to the NT-DITT for assessment. [Con- 5982]
- On 23 January 2025 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that, following a request from NOPSEMA to provide further information. Santos had re-submitted the EP with some amendments in December 2024 for further assessment. The Update also re-stated that the OEMP would be submitted to the NT-DITT for assessment. [Con-6036]
- Notwithstanding the information provided and the steps described above, no comments or input were received on the Coastal Waters OEMP by A. Raptis & Sons.

Summary of response by Relevant Person	Assessment of merits	Santos' Response Statement	OEMP
No response was received from A. Raptis & Sons.	Santos considers it has provided sufficient information and a reasonable period of time for consultation. Santos considers Section 25 consultation requirements to have been met.	No response required.	Not ap



reference

plicable.

#### Austfish

Summary of consultation effort:

•	On 9 February 2024 Santos emailed Austfish	to advise it of preliminary consu	Itation regarding proposed activities for	or consultation to be managed under the Coastal W	aters OEMP and Barossa Production Operation

- The email advised that Santos was seeking information to better understand:
  - o if you are from a government Department or agency, how the proposed Production Operations activities may be relevant to your department or agency; and
  - what (if any) functions, interests or activities you or your organisation have that may be affected by the proposed Production Operations activities.
- The email included information on the regulatory process for the activities in Commonwealth and NT jurisdictions and links to a Santos information booklet on the proposed activities and a NOPSEMA brochure on the consultation process and details of how to contact Santos to register as a Relevant Person. The email stated that the consultation phase would commence on 11 March 2024 and close on 9 April 2024.
- On 11 March 2024 Santos emailed Austfish further to the previous correspondence, to advise that it had commenced the consultation phase which would run until 9 April 2024. In addition to the previous information again being provided. Santos provided information on Relevant Persons' entitlements under the regulatory processes, details of how to provide feedback and a reminder of the closing date for consultation. [Con-3793]
- On 3 April 2024 Santos phoned Austfish and a representative advised that the message would be passed to the appropriate personnel.
- On 6 May 2024, Santos emailed Austfish further to previous correspondence, to advise it had extended the consultation period until 20 May 2024. In providing this extension of time, Santos advised that the information in the booklet and factsheet had been updated to account for an additional risk associated with the proposed activity. Santos confirmed that consultation will close on the revised date unless Santos hears otherwise from Austfish. [Con-3868]
- On 10 July 2024 Santos emailed Austfish to advise the consultation period had been completed. Santos advised Austfish that it considered consultation had now closed for the purpose of Santos finalising and submitting environment plans for these activities to government regulators for assessment. [Con-5130]
- On 15 October 2024 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that the EP had been submitted to NOPSEMA for assessment, provided a link to the full EP on the NOPSEMA website and Santos' contact details for any queries. The Update also stated that the OEMP would be submitted to the NT-DITT for assessment. [Con- 5982]
- On 23 January 2025 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that, following a request from NOPSEMA to provide further information. Santos had re-submitted the EP with some amendments in December 2024 for further assessment. The Update also re-stated that the OEMP would be submitted to the NT-DITT for assessment. [Con-6036]
- Notwithstanding the information provided and the steps described above, no comments or input were received on the Coastal Waters OEMP by Austfish.

Summary of response by Relevant Person	Assessment of merits	Santos' Response Statement	OEMP
No response was received from Austfish.	Santos considers it has provided sufficient information and a reasonable period of time for consultation. Santos considers Section 25 consultation requirements to have been met.	No response required.	Not ap

### Austral Fisheries Pty Ltd

Summary of consultation effort:

- On 9 February 2024 Santos emailed Austral Fisheries to advise it of preliminary consultation regarding proposed activities for consultation to be managed under the Coastal Waters OEMP and Barossa Production Operations EP. [Con-3787]
- The email advised that Santos was seeking information to better understand:
  - o if you are from a government Department or agency, how the proposed Production Operations activities may be relevant to your department or agency; and
  - what (if any) functions, interests or activities you or your organisation have that may be affected by the proposed Production Operations activities.
- The email included information on the regulatory process for the activities in Commonwealth and NT jurisdictions and links to a Santos information booklet on the proposed activities and a NOPSEMA brochure on the consultation process and details of how to contact Santos to register as a Relevant Person. The email stated that the consultation phase would commence on 11 March 2024 and close on 9 April 2024.
- On 11 March 2024 Santos emailed Austral Fisheries further to the previous correspondence, to advise that it had commenced the consultation phase which would run until 9 April 2024. In addition to the previous information again being provided, Santos provided information on Relevant Persons' entitlements under the regulatory processes, details of how to provide feedback and a reminder of the closing date for consultation, [Con-3793]
- On 21 March 2024 Santos met with Austral Fisheries to provide further information on the Barossa petroleum safety zones (PSZ). Santos indicated that these are the zones granted for the PSZ application for drilling activities and will be the zones in the PSZ application for subsea installation activities which will be submitted shortly. Austral Fisheries indicated that the zones were acceptable.
- On 22 March 2024 Santos emailed Austral Fisheries and provided presentation slides from the meeting on 21 March 2024, and confirmed that these are the zones granted for our PSZ application for drilling activities and would be the zones in Santos' PSZ application for subsea installation activities. Santos sought re-affirmation that previous feedback that the zones were acceptable to Austral. [Con-3860]
- On 3 April 2024 Austral Fisheries responded via email confirming that the PSZs will not impact its operations. [Con-4993]
- On 3 May 2024, Santos emailed Austral Fisheries further to previous correspondence, to advise it had extended the consultation period until 16 May 2024. In providing this extension of time, Santos advised that the information in the booklet and factsheet had been updated to account for an additional risk associated with the proposed activity. Santos confirmed that consultation will close on the revised date unless Santos hears otherwise from Austral Fisheries. [Con-3865]
- On 10 July 2024 Santos emailed Austral Fisheries to advise the consultation period had been completed. Santos advised it consultation had now closed for the purpose of Santos finalising and submitting environment plans for these activities to government regulators for assessment. [Con-5130]
- On 15 October 2024 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that the EP had been submitted to NOPSEMA for assessment, provided a link to the full EP on the NOPSEMA website and Santos' contact details for any queries. The Update also stated that the OEMP would be submitted to the NT-DITT for assessment. [Con- 5982]
- On 23 January 2025 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that, following a request from NOPSEMA to provide further information, Santos had re-submitted the EP with some amendments in December 2024 for further assessment. The Update also re-stated that the OEMP would be submitted to the NT-DITT for assessment. [Con-6036]
- No further correspondence has been received from Austral Fisheries.



ations EP. [Con-3787]

reference

plicable.

Summary of response by Relevant Person	Assessment of merits	Santos' Response Statement	OEMF
Austral Fisheries responded that the establishment of the PSZs would not affect its operations.	Santos notes Austral Fisheries' response.	No response required.	Not ap

#### Australia Bav Seatoo

Summary of consultation effort:

- On 9 February 2024 Santos emailed Australia Bay Seafoods to advise it of preliminary consultation regarding proposed activities for consultation to be managed under the Coastal Waters OEMP and Barossa Production Operations EP. [Con-3787]
- The email advised that Santos was seeking information to better understand:
  - o if you are from a government Department or agency, how the proposed Production Operations activities may be relevant to your department or agency; and
  - what (if any) functions, interests or activities you or your organisation have that may be affected by the proposed Production Operations activities. 0
- The email included information on the regulatory process for the activities in Commonwealth and NT jurisdictions and links to a Santos information booklet on the proposed activities and a NOPSEMA brochure on the consultation process and details of how to contact Santos to register as a Relevant Person. The email stated that the consultation phase would commence on 11 March 2024 and close on 9 April 2024.
- On 11 March 2024 Santos emailed Australia Bay Seafoods further to the previous correspondence, to advise that it had commenced the consultation phase which would run until 9 April 2024. In addition to the previous information again being provided, Santos provided information on Relevant Persons' entitlements under the regulatory processes, details of how to provide feedback and a reminder of the closing date for consultation. [Con-3793]
- On 3 April 2024 Santos phoned Australia Bay Seafoods which advised it did not have any comments.
- On 8 May 2024, Santos emailed Australia Bay Seafoods further to the phone call on 3 April 2024, to advise it had extended the consultation period until 22 May 2024. In providing this extension of time, Santos advised that the information in the booklet and factsheet had been updated to account for an additional risk associated with the proposed activity. Santos confirmed that consultation will close on the revised date unless Santos hears otherwise from Australia Bay Seafoods. [Con-3928]
- On 10 July 2024 Santos emailed Australia Bay Seafoods to advise the consultation period had been completed. Santos advised Australia Bay Seafoods that it considered consultation had now closed for the purpose of Santos finalising and submitting environment plans for these activities to government regulators for assessment. [Con-5130]
- On 15 October 2024 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that the EP had been submitted to NOPSEMA for assessment, provided a link to the full EP on the NOPSEMA website and Santos' contact details for any queries. The Update also stated that the OEMP would be submitted to the NT-DITT for assessment. [Con- 5982]
- On 23 January 2025 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that, following a request from NOPSEMA to provide further information. Santos had re-submitted the EP with some amendments in December 2024 for further assessment. The Update also re-stated that the OEMP would be submitted to the NT-DITT for assessment. [Con-6036]
- Notwithstanding the information provided and the steps described above, no comments or input were received on the Coastal Waters OEMP by Australia Bay Seafoods.

Summary of response by Relevant Person	Assessment of merits	Santos' Response Statement	OEMP
No response was received from Australia Bay Seafoods.	Santos considers it has provided sufficient information and a reasonable period of time for consultation. Santos considers Section 25 consultation requirements to have been met.	No response required.	Not ap

### Santos

#### reference

oplicable.

reference

oplicable.

#### **Clipper Pearls**

Summary of consultation effort:

- On 9 February 2024 Santos emailed Clipper Pearls to advise it of preliminary consultation regarding proposed activities for consultation to be managed under the Coastal Waters OEMP and Barossa Production Operations EP. [Con-3787]
- The email advised that Santos was seeking information to better understand: •
  - o if you are from a government Department or agency, how the proposed Production Operations activities may be relevant to your department or agency; and
  - what (if any) functions, interests or activities you or your organisation have that may be affected by the proposed Production Operations activities.
- The email included information on the regulatory process for the activities in Commonwealth and NT jurisdictions and links to a Santos information booklet on the proposed activities and a NOPSEMA brochure on the consultation process and details of how to contact Santos to register as a Relevant Person. The email stated that the consultation phase would commence on 11 March 2024 and close on 9 April 2024.
- On 11 March 2024 Santos emailed Clipper Pearls further to the previous correspondence, to advise that it had commenced the consultation phase which would run until 9 April 2024. In addition to the previous information again being provided, Santos provided information on Relevant Persons' entitlements under the regulatory processes, details of how to provide feedback and a reminder of the closing date for consultation. [Con-3793]
- On 3 April 2024 Santos phoned Clipper Pearls who advised they would not be impacted by the proposed activities.
- On 6 May 2024, Santos emailed Clipper Pearls further to its response on 3 April 2024, to advise it had extended the consultation period until 20 May 2024. In providing this extension of time, Santos advised that the information in the booklet and factsheet had been updated to account for an additional risk associated with the proposed activity. Santos confirmed that consultation will close on the revised date unless Santos hears otherwise from Clipper Pearls. [Con-3866]
- On 10 July 2024 Santos emailed Clipper Pearls to advise the consultation period had been completed. Santos advised Clipper Pearls that it considered consultation had now closed for the purpose of Santos finalising and submitting environment plans for these activities to government regulators for assessment. [Con-5130]
- On 15 October 2024 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that the EP had been submitted to NOPSEMA for assessment, provided a link to the full EP on the NOPSEMA website and Santos' contact details for any queries. The Update also stated that the OEMP would be submitted to the NT-DITT for assessment. [Con- 5982]
- On 23 January 2025 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that, following a request from NOPSEMA to provide further information, Santos had re-submitted the EP with some amendments in December 2024 for further assessment. The Update also re-stated that the OEMP would be submitted to the NT-DITT for assessment. [Con-6036]
- Notwithstanding the information provided and the steps described above, no comments or input were received on the Coastal Waters OEMP by Clipper Pearls.

Summary of response by Relevant Person	Assessment of merits	Santos' Response Statement	OEMI	
No response was received from Clipper Pearls.	Santos considers it has provided sufficient information and a reasonable period of time for consultation. Santos considers Section 25 consultation requirements to have been met.	No response required.	Not a	
Fischer Wholesale Ptv I td/H & T Investment Ptv I td/Commercial Catamarans Ptv I td				

Summary of consultation effort:

- On 9 February 2024 Santos emailed Fischer Wholesale/H & T Investment/Commercial Catamarans to advise it of preliminary consultation regarding proposed activities for consultation to be managed under the Coastal Waters OEMP and Barossa Production Operations EP. [Con-3787]
- The email advised that Santos was seeking information to better understand:
  - o if you are from a government Department or agency, how the proposed Production Operations activities may be relevant to your department or agency; and
  - what (if any) functions, interests or activities you or your organisation have that may be affected by the proposed Production Operations activities.
- The email included information on the regulatory process for the activities in Commonwealth and NT jurisdictions and links to a Santos information booklet on the proposed activities and a NOPSEMA brochure on the consultation process and details of how to contact Santos to register as a Relevant Person. The email stated that the consultation phase would commence on 11 March 2024 and close on 9 April 2024.
- On 11 March 2024 Santos emailed Fischer Wholesale/H & T Investment/Commercial Catamarans further to the previous correspondence, to advise that it had commenced the consultation phase which would run until 9 April 2024. In addition to the previous information again being provided. Santos provided information on Relevant Persons' entitlements under the regulatory processes, details of how to provide feedback and a reminder of the closing date for consultation. [Con-3793]
- On 3 April 2024 Santos phoned Fischer Wholesale Pty Ltd/H & T Investment Pty Ltd/Commercial Catamarans Pty Ltd and left a voicemail.
- On 2 May 2024, Santos emailed Fischer Wholesale Pty Ltd/H & T Investment Pty Ltd/Commercial Catamarans Pty Ltd further to previous correspondence to advise it had extended the consultation period until 16 May 2024. In providing this extension of time, Santos advised that the information in the booklet and factsheet had been updated to account for an additional risk associated with the proposed activity. Santos confirmed that consultation will close on the revised date unless Santos hears otherwise from Fischer Wholesale Ptv Ltd/H & T Investment Pty Ltd/Commercial Catamarans Pty Ltd. [Con-3864]
- On 10 July 2024 Santos emailed Fischer Wholesale Pty Ltd/H & T Investment Pty Ltd/Commercial Catamarans Pty Ltd to advise the consultation period had been completed. Santos advised Fischer Wholesale Pty Ltd/H & T Investment Pty Ltd/Commercial Catamarans Pty Ltd that it considered consultation had now closed for the purpose of Santos finalising and submitting environment plans for these activities to government regulators for assessment. [Con-5130]
- On 15 October 2024 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that the EP had been submitted to NOPSEMA for assessment, provided a link to the full EP on the NOPSEMA website and Santos' contact details for any queries. The Update also stated that the OEMP would be submitted to the NT-DITT for assessment. [Con- 5982]
- On 23 January 2025 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that, following a request from NOPSEMA to provide further information, Santos had re-submitted the EP with some amendments in December 2024 for further assessment. The Update also re-stated that the OEMP would be submitted to the NT-DITT for assessment. [Con-6036]
- Notwithstanding the information provided and the steps described above, no comments or input were received on the Coastal Waters OEMP by Fischer Wholesale Pty Ltd/H & T Investment Pty Ltd/Commercial Catamarans Pty Ltd

Summary of response by Relevant Person	Assessment of merits	Santos' Response Statement	OEMP
No response was received from Fischer Wholesale Pty Ltd/H & T Investment Pty Ltd/Commercial Catamarans Pty Ltd.	Santos considers it has provided sufficient information and a reasonable period of time for consultation. Santos considers Section 25 consultation requirements to have been met.	No response required.	Not app
Northern Wildcatch Seafood Australia (NWSA)			



P reference

pplicable.

reference

olicable.

- On 9 February 2024 Santos emailed NWSA to advise it of preliminary consultation regarding proposed activities for consultation to be managed under the Coastal Waters OEMP and Barossa Production Operations EP. [Con-3787] •
- The email advised that Santos was seeking information to better understand:
  - o if you are from a government Department or agency, how the proposed Production Operations activities may be relevant to your department or agency; and
  - what (if any) functions, interests or activities you or your organisation have that may be affected by the proposed Production Operations activities.
- The email included information on the regulatory process for the activities in Commonwealth and NT jurisdictions and links to a Santos information booklet on the proposed activities and a NOPSEMA brochure on the consultation process and details of how to contact • Santos to register as a Relevant Person. The email stated that the consultation phase would commence on 11 March 2024 and close on 9 April 2024.
- On 11 March 2024 Santos emailed NWSA further to the previous correspondence, to advise that it had commenced the consultation phase which would run until 9 April 2024. In addition to the previous information again being provided, Santos provided information on Relevant Persons' entitlements under the regulatory processes, details of how to provide feedback and a reminder of the closing date for consultation. [Con-3793]
- On 3 April 2024 Santos phoned NWSA who advised they would send an email response.
- On 4 April 2024, NWSA emailed Santos and confirmed that NWSA would not be providing feedback. [Con-3863] •
- On 8 May 2024. Santos emailed NWSA, further to its response on 4 April 2024, to advise it had extended the consultation period until 22 May 2024. In providing this extension of time, Santos advised that the information in the booklet and factsheet had been updated to • account for an additional risk associated with the proposed activity. Santos confirmed that consultation will close on the revised date unless Santos hears otherwise from NWSA. [Con-4359]
- On 10 July 2024 Santos emailed NWSA to advise the consultation period had been completed. Santos advised NWSA for its input and advised it consultation had now closed for the purpose of Santos finalising and submitting environment plans for these activities to government regulators for assessment. [Con-5130]
- On 15 October 2024 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that the EP had been submitted to NOPSEMA for assessment, provided a link to the full EP on the NOPSEMA website and Santos' contact details for any queries. The Update also stated that the OEMP would be submitted to the NT-DITT for assessment. [Con- 5982]
- On 23 January 2025 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that, following a request from NOPSEMA to provide further information, Santos had re-submitted the EP with some amendments in December 2024 for further assessment. The Update also re-stated that the OEMP would be submitted to the NT-DITT for assessment. [Con-6036]
- No further correspondence has been received from NWSA.

Summary of response by Relevant Person	Assessment of merits	Santos' Response Statement	OEM
NWSA responded that it would not be participating in the consultation process.	Santos noted NWSA's response. Santos considers it has provided sufficient information and a reasonable period of time for consultation.	No response required.	Not a
	Santos considers Section 25 consultation requirements to have been met.		

### Paspaley Pearling Compar

Summary of consultation effort:

- On 9 February 2024 Santos emailed Paspaley Pearling Company to advise it of preliminary consultation regarding proposed activities for consultation to be managed under the Coastal Waters OEMP and Barossa Production Operations EP. [Con-3787]
- The email advised that Santos was seeking information to better understand:
  - o if you are from a government Department or agency, how the proposed Production Operations activities may be relevant to your department or agency; and
  - what (if any) functions, interests or activities you or your organisation have that may be affected by the proposed Production Operations activities. 0
- The email included information on the regulatory process for the activities in Commonwealth and NT jurisdictions and links to a Santos information booklet on the proposed activities and a NOPSEMA brochure on the consultation process and details of how to contact Santos to register as a Relevant Person. The email stated that the consultation phase would commence on 11 March 2024 and close on 9 April 2024.
- On 16 February 2024 Paspaley Pearling Company emailed Santos to advise that it did not have additional comment in relation to Production Operations activities. In its email, Paspaley Pearling Company reiterated information relevant to Darwin Pipeline Duplication construction activities. Santos has previously provided commitments to Paspaley Pearling Company related to these activities [Con-3840]
- On 8 March 2024 Santos emailed Paspaley Pearling Company to enquire if a meeting in Darwin the following week would be convenient for an update. [Con-3858]
- On 8 March 2024 Santos emailed a further email to Paspaley Pearling Company to provide contact details. [Con-3859]
- On 11 March 2024 Santos emailed Paspaley Pearling Company further to the previous correspondence, to advise that it had commenced the consultation phase which would run until 9 April 2024. In addition to the previous information again being provided, Santos provided information on Relevant Persons' entitlements under the regulatory processes, details of how to provide feedback and a reminder of the closing date for consultation. [Con-3793]
- On 25 March 2024 Santos emailed Paspaley Pearling Company to enquire whether it would like to meet over Teams. In the email Santos provided updated information relevant to DPD construction activities in NT waters. [Con-3861]
- On 25 March 2024 Paspaley Pearling Company thanked Santos for providing further information and advised that it does not require a meeting at this stage. [Con-3862]
- On 7 May 2024, Santos emailed Paspaley Pearling Company further to previous correspondence to advise it had extended the consultation period until 21 May 2024. In providing this extension of time. Santos advised that the information in the booklet and factsheet had been updated to account for an additional risk associated with the proposed activity. Santos confirmed that consultation will close on the revised date unless Santos hears otherwise from Paspaley Pearling. [Con-3926]
- On 10 July 2024 Santos emailed Paspaley Pearling Company to advise the consultation period had been completed. Santos advised it considered that consultation had now closed for the purpose of Santos finalising and submitting environment plans for these activities to government regulators for assessment. [Con-5130]
- On 15 October 2024 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that the EP had been submitted to NOPSEMA for assessment, provided a link to the full EP on the NOPSEMA website and Santos' contact details for any queries. The Update also stated that the OEMP would be submitted to the NT-DITT for assessment. [Con- 5982]
- On 23 January 2025 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that, following a request from NOPSEMA to provide further information. Santos had re-submitted the EP with some amendments in December 2024 for further assessment. The Update also re-stated that the OEMP would be submitted to the NT-DITT for assessment. [Con-6036]
- No further correspondence has been received from Paspaley Pearling Company.

Summary of response by Relevant Person	Assessment of merits	Santos' Response Statement	OEMF
Paspaley Pearling Company responded that it would not be participating in the consultation process.	Santos noted the response	No response required.	Not a

### Santos

**Preference** 

oplicable.

reference

oplicable.

	Santos considers it has provided sufficient information and a reasonable period of time for consultation. Santos considers Section 25 consultation requirements to have been met.	
MA Soctoodo		

- On 9 February 2024 Santos emailed WA Seafoods to advise it of preliminary consultation regarding proposed activities for consultation to be managed under the Coastal Waters OEMP and Barossa Production Operations EP. [Con-3787]
- The email advised that Santos was seeking information to better understand:
  - o if you are from a government Department or agency, how the proposed Production Operations activities may be relevant to your department or agency; and
  - what (if any) functions, interests or activities you or your organisation have that may be affected by the proposed Production Operations activities. 0
- The email included information on the regulatory process for the activities in Commonwealth and NT jurisdictions and links to a Santos information booklet on the proposed activities and a NOPSEMA brochure on the consultation process and details of how to contact Santos to register as a Relevant Person. The email stated that the consultation phase would commence on 11 March 2024 and close on 9 April 2024.
- On 11 March 2024 Santos emailed WA Seafoods further to the previous correspondence, to advise that it had commenced the consultation phase which would run until 9 April 2024. In addition to the previous information again being provided, Santos provided information on Relevant Persons' entitlements under the regulatory processes, details of how to provide feedback and a reminder of the closing date for consultation. [Con-3793]
- On 3 April 2024 Santos phoned WA Seafoods and a representative said the message would be provided to the appropriate person.
- On 6 May 2024, Santos emailed WA Seafoods to advise further to previous correspondence, it had extended the consultation period until 20 May 2024. In providing this extension of time, Santos advised that the information in the booklet and factsheet had been updated to account for an additional risk associated with the proposed activity. Santos confirmed that consultation will close on the revised date unless Santos hears otherwise from WA Seafoods. [Con-3869]
- On 10 July 2024 Santos emailed WA Seafoods to advise the consultation period had been completed. Santos advised WA Seafoods that it considered consultation had now closed for the purpose of Santos finalising and submitting environment plans for these activities to government regulators for assessment. [Con-5130]
- On 15 October 2024 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that the EP had been submitted to NOPSEMA for assessment, provided a link to the full EP on the NOPSEMA website and Santos' contact details for any queries. The Update also stated that the OEMP would be submitted to the NT-DITT for assessment. [Con- 5982]
- On 23 January 2025 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that, following a request from NOPSEMA to provide further information, Santos had re-submitted the EP with some amendments in December 2024 for further assessment. The Update also re-stated that the OEMP would be submitted to the NT-DITT for assessment. [Con-6036]
- Notwithstanding the information provided and the steps described above, no comments or input were received on the Coastal Waters OEMP by WA Seafoods.

Summary of response by Relevant Person	Assessment of merits	Santos' Response Statement	OEM
No response was received from WA Seafoods.	Santos considers it has provided sufficient information and a reasonable period of time for consultation. Santos considers Section 25 consultation requirements to have been met.	No response required.	Not a

### Santos

P reference

pplicable.

#### 4.8.8 **First Nations People and Groups**

### Table 4-18: Consultation Summary Table – First Nations People and Groups

First Nations People and groups: Representative organisations – Northern Territory

### Larrakia Nation Aboriginal Corporation (LNAC)

Summary of consultation effort:

- On 13 February 2024 Santos emailed LNAC to advise it of preliminary consultation regarding proposed activities for consultation to be managed under the Coastal Waters OEMP and Barossa Production Operations EP. [Con-4053]. The email advised that Santos was seeking information to better understand:
  - o if you are from a government Department or agency, how the proposed Production Operations activities may be relevant to your department or agency; and
  - what (if any) functions, interests or activities you or your organisation have that may be affected by the proposed Production Operations activities.
- The email included information on the regulatory process for the activities in Commonwealth and NT jurisdictions, links to a Santos information booklet on the proposed activities and a NOPSEMA brochure on the consultation process and details of how to contact Santos to register as a Relevant Person. The email stated that the consultation phase would commence on 11 March 2024 and close on 9 April 2024.
- In the email Santos asked LNAC if it knew of any other organisations that should be contacted. Santos also advised LNAC that it would be happy to provide consultation sessions for LNAC staff and Board memberships and the santos also advised LNAC that it would be happy to provide consultation sessions for LNAC staff and Board memberships and the santos also advised LNAC that it would be happy to provide consultation sessions for LNAC staff and Board memberships and the santos also advised LNAC that it would be happy to provide consultation sessions for LNAC staff and Board memberships and the santos also advised LNAC that it would be happy to provide consultation sessions for LNAC staff and Board memberships and the santos also advised LNAC that it would be happy to provide consultation sessions for LNAC staff and Board memberships and the santos also advised LNAC that it would be happy to provide consultation sessions for LNAC staff and Board memberships and the santos also advised LNAC that it would be happy to provide consultation sessions for LNAC staff and Board memberships and the santos advised LNAC that it would be happy to provide consultation sessions for LNAC staff and Board memberships and the santos advised LNAC staff and Board memberships and the santos advised LNAC staff and Board memberships and the santos advised LNAC staff and Board memberships and the santos advised LNAC staff and Board memberships advised LNAC staff and Board memberships advised LNAC staff and Board memberships advised LNAC staff advised LNA
- On 12 March 2024 Santos emailed LNAC further to the previous correspondence, to advise that it had commenced the consultation phase which would run until 9 April 2024. In addition to the previous information Relevant Persons' entitlements under the regulatory processes, details of how to provide feedback and a reminder of the closing date for consultation. [Con-4052]
- In the email Santos also advised LNAC that it would be happy to provide consultation sessions for LNAC staff and Board members as well as Larrakia family members. Santos also advised that it would be holdin requested by LNAC, and would send information to the LNAC on these soon.
- Between 25 March 2024 and 2 April 2024 Santos liaised with LNAC on arrangements for Larrakia People attending the planned sessions being held in Darwin on 23 April 2024.
- On 27 March 2024 Santos emailed LNAC details of the OEMP and EP consultation sessions being held for Larrakia People in Darwin on 23 April and asked for the organisation to share the information with its L social media channels. [Con- 6021]
- On 2 April 2024 LNAC confirmed via response email that the information had been posted to its Facebook and LinkedIn social media networks [Con-5234]
- See entry for Larrakia People in this Table for summaries of the consultation sessions held on 23 April 2024.
- On 2 April 2024 Santos emailed LNAC by way of reminder that the consultation is closing on April 9 and asked LNAC to contact Santos as soon as possible if it has any feedback. [Con-4082]
- On 2 May 2024 Santos emailed LNAC further to previous correspondence, to advise it had extended the consultation period until 16 May 2024. In providing this extension of time, Santos advised that the information of the inf account for an additional risk associated with the proposed activity. Santos confirmed that consultation will close on the revised date unless Santos hears otherwise from LNAC. [Con-4994]
- On 17 May 2024 Santos emailed LNAC details of the OEMP and EP consultation sessions being held for Larrakia People in Darwin on 12 June 2024 and asked for the organisation to share the information with social media channels. [Con- 6022]
- See entry for Larrakia People in this Table for summaries of the consultation sessions held on 17 May 2024.
- LNAC was invited (by Larrakia Development Corporation) to a meeting between Larrakia Development Corporation and Santos on 20 June 2024 but did not attend. See LDC entry in this table for a summary of
- On 10 July 2024 Santos emailed LNAC to advise the consultation period had been completed. Santos advised it considered that consultation had now closed for the purpose of Santos finalising and submitting e regulators for assessment. [Con-5119]
- On 15 October 2024 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that the EP had been submitted to NOPSEMA for assessment, provided a link to the full EP on the NOPSEMA website and Santos' contact details for any queries. The Update also stated that the OEMP would be submitted to the NT-DITT for assessment. [Con- 5982]
- On 23 January 2025 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that, following a request from NOPSEMA to provide further information, Santos had re-submitted the EP with some amendments in December 2024 for further assessment. The Update also re-stated that the OEMP would be submitted to the NT-DITT for assessment. [Con-6036]
- Notwithstanding the information provided and the steps described above, no comments or input were received on the Coastal Waters OEMP, from the LNAC.
- Additional to formal consultation on the Coastal Waters OEMP and Barossa Production Operations EP, Santos has continued to make a range of efforts to engage on an ongoing basis with NT First Nations People, both directly and through their representative organisations. See Section 8.4.9 Post-acceptance Consultation Implementation Strategy.

Summary of response by Relevant Person	Assessment of merits	Santos' Response Statement	OEMP reference
No response was received from LNAC except to confirm posting of consultation information to social media pages, as noted above.	Santos considers it has provided sufficient information and a reasonable period of time for consultation. Santos considers Section 25 consultation requirements to have been met.	No response required.	Not applicable.
Northern Land Council (NLC)			

Summary of consultation effort:

- On 13 February 2024 Santos emailed NLC to advise it of preliminary consultation regarding proposed activities for consultation to be managed under the Coastal Waters OEMP and Barossa Production Operations EP. [Con-3971]. The email advised that Santos was seeking information to better understand:
- if you are from a government Department or agency, how the proposed Production Operations activities may be relevant to your department or agency; and
- what (if any) functions, interests or activities you or your organisation have that may be affected by the proposed Production Operations activities.
- The email included information on the regulatory process for the activities in Commonwealth and NT jurisdictions, links to a Santos information booklet on the proposed activities and a NOPSEMA brochure on the consultation process and details of how to contact Santos to register as a Relevant Person. The email stated that the consultation phase would commence on 11 March 2024 and close on 9 April 2024.
- In the email Santos asked NLC if it knew of any other organisations that should be contacted. Santos also advised NLC that it would be happy to provide consultation sessions for NLC staff, Board and members.



bers as well as Larrakia family members. on again being provided, Santos provided information on
ng Larrakia specific consultation sessions, as previously
arrakia networks and post on notice boards and relevant
ation in the booklet and factsheet had been updated to
its Larrakia networks and post on notice boards and relevant
the meeting. environment plans for these activities to government

- On 12 March 2024 Santos emailed NLC further to the previous correspondence, to advise that it had commenced the consultation phase which would run until 9 April 2024. In addition to the previous information again being provided, Santos provided information on Relevant Persons' entitlements under the regulatory processes, details of how to provide feedback and a reminder of the closing date for consultation. [Con-3972]
- On 12 March 2024 NLC emailed Santos to advise that it has forwarded the information to its Principal Legal Officer to follow-up. [Con-3973]
- On 2 April 2024 Santos emailed NLC by way of reminder that the consultation is closing on April 9 and asked NLC to contact Santos urgently if it would like a consultation session or had any questions on the information provided. [Con-3974]
- On 2 May 2024 Santos emailed NLC further to previous correspondence, to advise it had extended the consultation period until 16 May 2024. In providing this extension of time, Santos advised that the information in the booklet and factsheet had been updated to account for an additional risk associated with the proposed activity. Santos confirmed that consultation will close on the revised date unless Santos hears otherwise from NLC [Con-3978]
- During the consultation period for Production Operations activities, Santos also consulted with First Nations Consultative Committees (FNCCs) and/or Clan Groups representing the interests of First Nations people in coastal areas of the NLC regions of East Arnhem, West Arnhem, Darwin/Daly/Wagait and Victoria River District. See the separate entries in this table for the outcomes of consultation with each FNCC/Clan Group.
- On 10 July 2024 Santos emailed NLC to advise the consultation period had been completed. Santos advised it consultation had now closed for the purpose of Santos finalising and submitting environment plans for these activities to government regulators for assessment. Santos advised NLC it would appreciate its support in passing on the information to Council Members/Representatives. [Con-5122]
- On 15 October 2024 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that the EP had been submitted to NOPSEMA for assessment, provided a link to the full EP on the NOPSEMA website and Santos' contact details for any queries. The Update also stated that the OEMP would be submitted to the NT-DITT for assessment. [Con-5982]
- On 14 November 2024 Santos phoned the NLC seeking confirmation that it had no comments on the EP and OEMP. The NLC responded via phone on 15 November 2024 to confirm it had no comments. [Con-5988].
- On 23 January 2025 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that, following a request from NOPSEMA to provide further information, Santos had re-submitted the EP with some amendments in December 2024 for further assessment. The Update also re-stated that the OEMP would be submitted to the NT-DITT for assessment. [Con-6036]
- Notwithstanding the information provided and the steps described above, no comments or input were received on the Coastal Waters OEMP from NLC

Summary of response by Relevant Person	Assessment of merits	Santos' Response Statement	OEMP reference
No response was received from NLC	Santos considers it has provided sufficient information and a reasonable period of time for consultation. Santos considers Section 25 consultation requirements to have been met.	No response required.	Not applicable.
Tiwi Land Council (TLC)			

- On 6 February 2024 Santos met with the TLC as part of a regular series of meetings to provide activity updates, share information and discuss any potential concerns. Santos provided updates on a range of topics including the next round of consultation sessions with Tiwi clan groups. [Con-4668]
- On 13 February 2024 Santos emailed TLC to advise it of preliminary consultation regarding proposed activities for consultation to be managed under the Coastal Waters OEMP and Barossa Production Operations EP. [Con-4058]. The email advised that Santos was seeking information to better understand:
  - o if you are from a government Department or agency, how the proposed Production Operations activities may be relevant to your department or agency; and
  - what (if any) functions, interests or activities you or your organisation have that may be affected by the proposed Production Operations activities.
- The email included information on the regulatory process for the activities in Commonwealth and NT jurisdictions, links to a Santos information booklet on the proposed activities and a NOPSEMA brochure on the consultation process and details of how to contact Santos to register as a Relevant Person. The email stated that the consultation phase would commence on 11 March 2024 and close on 9 April 2024.
- In the email Santos also advised TLC that it would be happy to provide consultation sessions for TLC staff. Board and Council members. Santos also attached the notification for the next round of Tiwi Consultation sessions, at which the Coastal Waters OEMP and Barossa Production Operations EP would be discussed and asked the TLC to post the notification on its noticeboard.
- During the consultation period, the TLC and members who were also Clan Trustees were consulted on proposed meeting dates and confirmed dates and/or when any postponements were required. Some elected members of the TLC were often in attendance at the consultation sessions with their respective Clan Groups.
- On 12 March 2024 Santos emailed TLC further to the previous correspondence, to advise that it had commenced the consultation phase which would run until 9 April 2024. In addition to the previous information again being provided, Santos provided information on Relevant Persons' entitlements under the regulatory processes, details of how to provide feedback and a reminder of the closing date for consultation. [Con-4059]
- In the email Santos also advised TLC that it would be happy to provide a consultation session for TLC staff and will be holding Tiwi Clan meetings in early April to close-out the consultation on Production Operations activities.
- On 2 April 2024 Santos emailed TLC by way of reminder that the consultation is closing on April 9 and asked TLC to contact Santos as soon as possible if TLC would like a consultation session or has any questions on the below information. [Con-4078]
- On 2 May 2024 Santos emailed TLC further to previous correspondence, to advise it had extended the consultation period until 16 May 2024. In providing this extension of time. Santos advised that the information in the booklet and factsheet had been updated to account for an additional risk associated with the proposed activity. Santos confirmed that consultation will close on the revised date unless Santos hears otherwise from TLC. [Con-4110]
- On 10 July 2024 Santos emailed TLC to advise the consultation period had been completed. Santos advised it consultation had now closed for the purpose of Santos finalising and submitting environment plans for these activities to government regulators for assessment. Santos advised TLC it would appreciate its support in passing on the information to the Land Council (Trustee and Directors) for them to share with their clans. [Con-5120]
- On 15 October 2024 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that the EP had been submitted to NOPSEMA for assessment, provided a link to the full EP on the NOPSEMA website and Santos' contact details for any queries. The Update also stated that the OEMP would be submitted to the NT-DITT for assessment. [Con- 5982]
- On 23 January 2025 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that, following a request from NOPSEMA to provide further information, Santos had re-submitted the EP with some amendments in December 2024 for further assessment. The Update also re-stated that the OEMP would be submitted to the NT-DITT for assessment. [Con-6036]
- Notwithstanding the information provided and the steps described above, no comments or input were received on the Coastal Waters OEMP from TLC.
- Additional to formal consultation on the Coastal Waters OEMP and Barossa Production Operations EP, Santos has continued to make a range of efforts to engage on an ongoing basis with NT First Nations People, both directly and through their representative organisations. See Section 8.4.9 Post-acceptance Consultation Implementation Strategy.

Summary of response by Relevant Person	Assessment of merits	Santos' Response Statement	OEMP reference
No response was received from TLC	Santos considers it has provided sufficient information and a reasonable period of time for consultation. Santos considers Section 25 consultation requirements to have been met.	No response required.	Not applicable.
Wickham Point Deed liaison committee (WPDLC)			



- Barossa production operations activities in both Commonwealth and NT waters have been a regular agenda item at guarterly Wickham Point Deed liaison committee meetings since November 2021. As per the entry in this table for the liaison committee, consultation with respect to activities under the Coastal Waters OEMP and Barossa Production Operations EP was held on 7 March 2024.
- On 7 March 2024 Santos held a consultation session with the Wickham Point Deed liaison committee. The following information related to this OEMP was presented and discussed: [Con-4047]
  - 0 The Commonwealth Government and NT Government regulations and approvals required
  - The activities covered by the OEMP 0
  - The environmental impacts and risks involved with the planned activities and planned controls to management those risks 0
  - The EMBA in the event of an unplanned event, the risks and planned controls to management those risks 0
- The presentation also covered the regulatory consultation processes and privacy provisions and provided an overview of Santos the company and the Barossa Project overall.
- The information booklet and NOPSEMA consultation brochure were also provided at the consultation session.
- The activities were conducted in person and visual aids, maps, videos and animations were also to present information regarding the Activity and the project more generally.
- The majority of the engagement with attendees involved discussion to increase their understanding of the proposed activities and the associated risks and impacts, without any objections or claims about the adverse impact of each activity to which the Coastal Waters OEMP relates. [Con-6015] Refer below "Summary of response by Relevant Persons" below for further detail of specific questions answered at the meeting and the general topics/themes discussed.
- In addition, a question on underwater cultural heritage management in NT waters was outside the scope of this OEMP. and a response was separately provided. Questions on how to access Santos' employment and education opportunities or seek funding from Santos for community projects are managed outside the OEMP consultation process.
- On 12 March 2024 Santos emailed Wickham Point Deed liaison committee members thanking them for attending the consultation session. Links to the information that had been provided at the session were included in the email which also advised that consultation is open until Tuesday 9 April 2024. Santos encouraged the committee members to share the information with their Larrakia family and friends. [Con-4995]
- On 27 March 2024 Santos emailed Wickham Point Deed liaison committee members details of the Coastal Waters OEMP and Barossa Production Operations EP consultation sessions being held for Larrakia People in Darwin on 23 April and asked for them to share the information with their Larrakia networks and post on notice boards and relevant social media channels. [Con-6021]
- On 10 July 2024 Santos emailed Wickham Point Deed liaison committee members to advise the consultation period had been completed. Santos advised it considered that consultation had now closed for the purpose of Santos finalising and submitting environment plans for these activities to government regulators for assessment. Santos advised Wickham Point Deed liaison committee members it would appreciate its support in passing on the information to their families. [Con-5123]
- On 15 October 2024 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that the EP had been submitted to NOPSEMA for assessment, provided a link to the full EP on the NOPSEMA website and Santos' contact details for any queries. The Update also stated that the OEMP would be submitted to the NT-DITT for assessment. [Con- 5982]
- On 23 January 2025 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that, following a request from NOPSEMA to provide further information, Santos had re-submitted the EP with some amendments in December 2024 for further assessment. The Update also re-stated that the OEMP would be submitted to the NT-DITT for assessment. [Con-6036]
- No further correspondence or feedback was received from the Wickham Point Deed liaison committee on the Coastal Waters OEMP.
- Additional to formal consultation on the Coastal Waters OEMP and Barossa Production Operations EP, Santos has continued to make a range of efforts to engage on an ongoing basis with NT First Nations People, both directly and through their representative organisations. See Section 8.4.9 Post-acceptance Consultation Implementation Strategy.

Summary of response by Relevant Person	Assessment of merits	Santos' Response Statement	OEMP reference
<ul> <li>Following discussion on consultation material, there were questions and comments received in relation to the matters set out below:</li> <li>Composition of Barossa gas</li> <li>Assessment of water currents and marine mammal movement</li> <li>Management of solid waste on vessels.</li> <li>Santos answered those questions and there were no responses raised for consideration, other than those noted below.</li> </ul>	<ul> <li>Santos considers it has provided sufficient information and a reasonable period of time for consultation.</li> <li>Santos considers Section 25 consultation requirements to have been met.</li> <li>The answers provided were based on: <ul> <li>Composition of Barossa gas described in Section 7.5.1.</li> <li>Water current movements described in Sections 3.2.4.1, 3.2.4.2 and 3.4.2.3</li> <li>Marine mammal migration paths described in Section 3.2.13</li> <li>Management of solid waste on vessels described in Section 6.6</li> </ul> </li> </ul>	Noting that no objection or claims were raised about the adverse impact of each activity to which the Coastal Waters OEMP. relates, no further response required.	Not applicable
A committee representative member asked what residual or by-products will be in the gas coming from the Barossa field to the Darwin LNG facility.	Santos has considered the matters raised in the OEMP and provided details on residual by products. The answers provided were based on the activity description in Section 2.	Santos addressed this question at the meeting., Santos explained that by-product condensate is removed and processed offshore on FPSO. There will be very little residual or by-product coming through the pipeline from Barossa as dry gas is required for processing at DLNG facility. A small amount of Liquid Petroleum Gas is produced and transported from the facility via road transport.	Not applicable
A committee representative member asked if the water currents around the FPSO had been considered when writing the assessment of marine mammal movement and preventative measures.	Questions related to Production Operations activities in the Barossa Field are outside the scope of the Coastal Waters OEMP and are addressed in the Barossa Production Operations EP. At the meeting Santos explained that it had detailed information on water currents and migration paths that has been used to help determine how Santos can reduce risk of vessel interaction.	Not applicable.	Not applicable

A committee representative member asked if the FPSO would be disposing food scraps into the water. Further, the committee representative member asked about Santos' proposed food waste management plan.	Questions related to Production Operations activities in the Barossa Field are outside the scope of the Coastal Waters OEMP and are addressed in the Barossa Production Operations EP.	No response required.
-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------	----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------	-----------------------

#### First Nations People and groups: First Nations Consultative Committees and coastal clan groups - NT

#### Larrakia people

Summary of consultation effort:

- In order to assist with its efforts to reach out to Larrakia people in a culturally sensitive and appropriate way, consistent with NOPSEMA's consultation guidelines (2023, 2024), Santos has previously requested advice and assistance from LNAC, which speaks on behalf of Larrakia people, in relation to appropriate ways to engage with Larrakia people. This is additional to Santos' consultation with LNAC in its own right (see the separate entry for LNAC in Table 4-18).
- The LNAC Board's recommended approach to consultation with Larrakia people, communicated to Santos in December 2023, involved the following:
  - Santos undertake face-to-face consultation
  - o Santos advertise in the NT News the face-to face consultation once venue and time is confirmed
  - Larrakia Nation promote face-to-face consultation on social media including opportunity to provide feedback through Santos' toll-free number on 1800 267 600.
  - o Larrakia Nation email all LNAC staff to ensure they are aware of the consultation session to be conducted.
- On 23 April 2024 Santos held two consultation sessions in Darwin on the Barossa Production Operations EP and Coastal Waters OEMP with Larrakia people. [Con-4249], [Con-4230]
- On 12 June 2024 Santos held two further consultation sessions in Darwin on this OEMP with Larrakia people. [Con-4264], [Con-4263]
- The majority of the engagement with attendees involved discussion to increase their understanding of the proposed activities and the associated risks and impacts, without any objections or claims about the adverse impact of each activity to which the Coastal Waters • OEMP relates. Refer below "Summary of response by Relevant Persons" below for further detail of general topics/themes discussed.
- In addition, guestions on how to access Santos' employment and education opportunities or seek funding from Santos for community projects are managed outside the EP consultation process.
- In addition to the consultation efforts described above, Larrakia families are also represented on the Wickham Point Deed liaison committee, which has been separately consulted in relation to this OEMP see the separate entry for the liaison committee in this Table. Barossa Production Operations activities have been a regular agenda item at quarterly Wickham Point Deed liaison committee meetings since November 2021. As per the entry in this table for the liaison committee, consultation with respect to activities within this OEMP and the OEMP was held on 7 March 2024.
- On 23 September 2024, as part of Santos' ongoing engagement efforts with NT First National people and organisations, Santos hosted representatives of Larrakia Family Groups for a tour of the Darwin LNG Facility during which they were provided with an update on all aspects of the Barossa Project.
- On 15 October 2024 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that the EP had been submitted to NOPSEMA for assessment, provided a link to the full EP on the NOPSEMA website and Santos' contact details for any queries. The Update also stated that the OEMP would be submitted to the NT-DITT for assessment. [Con- 5982]
- On 23 January 2025 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that, following a request from NOPSEMA to provide further information. Santos had re-submitted the EP with some amendments in December 2024 for further assessment. The Update also re-stated that the OEMP would be submitted to the NT-DITT for assessment. [Con-6036]
- No further correspondence or feedback was received on the Coastal Waters OEMP from Larrakia People. •

Additional to formal consultation on the Coastal Waters OEMP and Barossa Production Operations EP, Santos has continued to make a range of efforts to engage on an ongoing basis with NT First Nations People, both directly and through their representative organisations. See Section 8.4.9 Post-acceptance Consultation Implementation Strategy.

Summary of response by Relevant Person	Assessment of merits	Santos' Response Statement
Following discussion on consultation material, there were questions and comments received in relation to the matters set	Committee members did not raise objections or claims about the adverse impact of each activity to which this OEMP relates. Questions related to Production Operations activities in the Barossa Field are outside the scope of the Coastal Waters OEMP and are addressed in the Barossa	Noting that no objection or claims were raised about the adve each activity to which this OEMP relates, no further response
out below:		
Activity		
pipeline route		
pipeline and FPSO maintenance	The answers provided at the meeting related to each activity to which this OEMP relates were based on the following sections of the OEMP:	
<ul> <li>monitoring and alarm process for leaks in pipeline</li> </ul>	The activity described in Section 2, including for the design parameters.	
Consultation & communication	<ul> <li>Consultation and communication described in Section 4 and required notifications in Sections 8.4.7 and 8.4.9.3.</li> <li>The existing environment described in Section 3.</li> <li>The environment risks &amp; impacts as well as the management controls for these described in Sections 6 and 7. Specific risks and impacts pertinent to the matters raised include: <ul> <li>Atmospheric and greenhouse emissions – Section 6.3</li> <li>Physical presence – Section 6.6</li> </ul> </li> <li>Potential impacts to marine life is addressed within a number of subsections in Section 6 and 7, and also in Section 7.3</li> <li>Loss of hydrocarbon management including gas release and spills in Sections 7.4; 7.5 and 7.6</li> <li>The relevant legislative requirements described in Appendix B</li> </ul>	
<ul> <li>Santos' communication with Larrakia people and the general community</li> </ul>		
<ul> <li>consultation process and flow of information to Larrakia People</li> </ul>		
<ul> <li>notification to other marine users, such as fishers, of activities occurring</li> </ul>		
Existing Environment		
<ul> <li>assessment of water currents and marine life movement</li> </ul>		
<ul> <li>nature of currents and tides &amp; influence of infrastructure presence on marine fauna movement</li> </ul>		
<ul> <li>potential for currents to bring environmental hazards &amp; debris that could impact infrastructure</li> </ul>		

Sa	anl	tos

Not applicable

	OEMP reference
erse impact of e required.	Not applicable

<ul> <li>cyclones and other weather events</li> </ul>	Santos considers it has provided sufficient information and a reasonable period of time for consultation	
Environment impacts & risks		
<ul> <li>physical presence of infrastructure &amp; vessels – how will these be detected/identified?</li> </ul>	Santos considers Section 25 consultation requirements to have been met.	
effects of activities on marine life		
effects of activities on the Tiwi Islands		
<ul> <li>ghost nets management</li> </ul>		
<ul> <li>incidence of major spills in the EMBA</li> </ul>		
<ul> <li>impact of GHG emissions, other emissions including chemicals</li> </ul>		
<ul> <li>gas emissions from the pipeline</li> </ul>		
gas supply customers		
<ul> <li>waste management on vessels</li> </ul>		
prevention of gas leaks		
• Other		
the approval processes		
<ul> <li>the AAPA process for identification of sacred sites</li> </ul>		
<ul> <li>drilling campaign details including depths of drilling and timeframe</li> </ul>		
Santos answered those questions and there were no responses raised for consideration, other than those noted below.		
An attendee stated that there were other groups of people around Daly River mouth, including Bungal, Dundee and Crab	Santos has considered the matters raised in the OEMP and provided a response at the meeting.	At the meeting Santos explained that people in the areas nan individual have been consulted for this OEMP.
Claw, and on small islands who should also be consulted.	Santos notes the advice provided by the attendee and has consulted people in these areas via First Nations consultative committees.	
Some attendees stated they and other family members were not aware of the sessions and suggested there be better communication by both Santos and Larrakia representative organisations.	<ul> <li>Santos has considered the matters raised in the OEMP and provided a response at the meeting.</li> <li>The answers provided were based on: <ul> <li>Consultation approach for Larrakia people described in Section 4.7.5.1.2.</li> <li>Public promotion of Larrakia consultation sessions described in Table 4-10.</li> <li>Examples of advertising and public notices in Appendix D.</li> </ul> </li> <li>Santos notes the advice provided by the attendees and considers it has made reasonable attempts to engage Larrakia people directly and via their representative organisations.</li> <li>Santos provided opportunities for Larrakia people to be involved in the consultation process through a range of ways for this OEMP as outlined in Section 4.7.5.1.2 and in the entries in this Table for the Northern L and Council (NLC) L arrakia Nation</li> </ul>	At the meeting Santos explained the activities that had been provide consultation opportunities for Larrakia people, includi participation in the Wickham Point Deed liaison committee.
	Aboriginal Corporation (LNAC), Gwalwa Daraniki Association (GDA) and Wickham Point Deed liaison committee (WPDIC).	
An attendee stated that clan groups were getting confused because of what was in the media, noting differing statements about the environmental impacts of the Barossa Project.	Santos has considered the matters raised in the OEMP and provided a response at the meeting. The answers provided were based on the consultation approach for Larrakia people described in Section 4.7.5.1.2.	At the meeting Santos explained that the Barossa Project is h by government to ensure that what Santos proposes to do wi and the risks are appropriately managed.
	Santos has provided opportunities for Larrakia people to be involved in the consultation process and provide direct input into EP development.	
Cobourg Peninsula Consultative Committee		

• On 29 April 2024 Santos held a consultation session in the morning with Agalda clan #1 in Jabiru. [Con-4250]

• On 29 April 2024 Santos held a consultation session in the afternoon with Agalda clan #2 in Jabiru. [Con-4243]

• On 1 May 2024 Santos held a consultation session in the afternoon with the Murran, Ngaindjagar, Madjunbalmi clan groups in Darwin [Con-4252]

• On 2 May 2024 Santos held a consultation session in the morning with Agalda clan #3 in Wurrumiyanga. [Con-4254]



amed by the	Not applicable
n undertaken to uding through	Not applicable
s heavily regulated will be acceptable	Not applicable

- The following information related to this OEMP was presented and discussed at each consultation session:\
- The Commonwealth Government and NT Government regulations and approvals required
- $\circ \quad \text{The activities covered by this OEMP}$
- The environmental impacts and risks involved with the planned activities and planned controls to management those risk
- $\circ$   $\,$   $\,$  The EMBA in the event of an unplanned event, the risks and planned controls to management those risks
- $\circ$   $\;$  The regulatory consultation processes and privacy provisions
- The presentation also provided an overview of Santos the company and the Barossa Project overall, including a Project status update as per previous Barossa EPs
- The Production Operations information booklet, Santos Privacy Statement and NOPSEMA consultation brochure was also provided at the consultation session.
- The sessions were conducted in person with the use of visual aids such as AO poster sized maps, AO posters with photos and images, PowerPoint presentation and videos to present information regarding the Activity and the project more generally.
- The majority of the engagement with attendees involved discussion to increase their understanding of the proposed activities and the associated risks and impacts, without any objections or claims about the adverse impact of each activity to which the Coastal Waters
  OEMP. relates. Refer below "Summary of response by Relevant Persons" below for further detail of general topics/themes discussed.
- Questions on how to access Santos' employment and education opportunities or seek funding from Santos for community projects are managed outside the OEMP consultation process.
- On 7 June 2024 Santos emailed a member of the Agalda Clan and Cobourg Peninsula Consultative Committee providing additional information in answer to some questions raised during the consultation session. These questions are addressed below. [Con-4997]
- On 10 July 2024 the consultant engaged by Santos to support establishment of the FN Consultative Committees, emailed the Cobourg Peninsula Consultative Committee, on behalf of Santos, to advise the consultation period had been completed. In the email Santos advised it considered that consultation had now closed for the purpose of Santos finalising and submitting environment plans for these activities to government regulators for assessment. Committee members were invited to pass the email on to other people in their network. [Con-5201]
- On 15 October 2024 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that the EP had been submitted to NOPSEMA for assessment, provided a link to the full EP on the NOPSEMA website and Santos' contact details for any queries. The Update also stated that the OEMP would be submitted to the NT-DITT for assessment. [Con- 5982]
- On 23 January 2025 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that, following a request from NOPSEMA to provide further information, Santos had re-submitted the EP with some amendments in December 2024 for further assessment. The Update also re-stated that the OEMP would be submitted to the NT-DITT for assessment. [Con-6036]
- No further correspondence or feedback was received from members of the Cobourg Peninsula Consultative Committee.

Summary of response by Relevant Person	Assessment of merits	Santos' Response Statement
Following discussion on consultation material, there were questions and comments received in relation to the matters set	Committee members did not raise objections or claims about the adverse impact of each activity to which this OEMP relates.	Noting that no objection or claims were raised about the adv each activity to which the Coastal Waters OEMP relates, no required.
<ul> <li>out below:</li> <li>Activity <ul> <li>stability of the FPSO</li> <li>how underwater maintenance is conducted</li> <li>drilling activity</li> <li>FPSO stability</li> </ul> </li> <li>Consultation &amp; communication <ul> <li>Why did it take so long to do consultation with us on our long to a co</li></ul></li></ul>	<ul> <li>Questions related to Production Operations activities in the Barossa Field are outside the scope of the Coastal Waters OEMP and are addressed in the Barossa Production Operations EP.</li> <li>The answers provided at the meeting related to each activity to which this OEMP relates were based on the following sections of the OEMP:</li> <li>The activity described in Section 2, including for the design parameters.</li> <li>Consultation and communication described in Section 4 and required notifications in Sections 8.4.7 and 8.4.9.3.</li> <li>The existing environment described in Section 3 including for the baseline</li> </ul>	
<ul> <li>• Existing Environment</li> <li>• environmental baseline studies</li> <li>• mapping of water currents around the FPSO</li> <li>• There needs to be a base line (for the FPSO), so you know what to do if something was to happen.</li> <li>• Environment impacts &amp; risks</li> <li>• sewage disposal</li> <li>• impacts of drilling or the pipeline to groundwater</li> <li>• how particular areas within the EMBA could be impacted by a spill</li> </ul>	<ul> <li>studies undertaken.</li> <li>The environment risks &amp; impacts as well as the management controls for these described in Sections 6 and 7. Specific risks and impacts pertinent to the matters raised include:</li> <li>Atmospheric and greenhouse gas emissions – Section 6.3</li> <li>Waste management – Section 7.1• Potential impacts to marine life within a number of subsections in Section 6 and 7, and also in Section 7.3</li> <li>Loss of hydrocarbon management including gas release and spills in Sections 7.4; 7.5 and 7.6</li> <li>The Barossa Development Drilling and Completions EP – accepted by NOPSEMA in December 2023.</li> </ul>	
<ul> <li>how is a spill contained and the timeframe involved</li> <li>compensation for environmental impacts due to a spill</li> <li>impacts and regulation of GHG emissions and purchasing of carbon credits</li> <li>how are spills cleaned up and how long does this take</li> <li>Where does the gas go? will it go into the ozone? Will it increase the temperature that will impact our turtles etc.</li> <li>What happens if the gas escapes (from the FPSO)?</li> </ul>	Santos considers it has provided sufficient information and a reasonable period of time for consultation. Santos considers Section 25 consultation requirements to have been met.	



	EP reference			
erse impact of further response	Not applicable			
	•	Sewerage.		
------------------------------	----------------------------------------------------------------------------------------	------------------------------------------------------------------------------------------------	---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------	----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------
	•	Where will spills go? Do they impact our area?		
	•	Spill response timing		
	•	Spill containment		
•	Oth	er		
	•	Historic accidents and past incidence of spills in the region		
	•	Compensation for those affected		
	•	Carbon Capture and Storage process		
	•	Access to employment and education opportunities for FN people (including on the FPSO)		
Sant resp belo	os a ons w.	answered those questions and there were no es raised for consideration, other than those noted		
A co	mm	ittee member asked if Santos benefited financially	Santos has considered the matters raised in the OEMP and provided a response	Following the meeting in an email on 7 June 2024 Santos ex
from	pur	chasing offsets.	Santos provided further details on the potential benefit of purchasing offsets.	The benefit to Santos from offsets is the ability to use them to emissions from our facilities. This is the primary purpose for acquires offsets.
				Purchase of offsets by Santos is at Santos' cost. The Clean B oversees the generation and use of Australian Carbon Credit emissions in accordance with the Safeguard Mechanism und legislation.
A co	mm	ittee member asked what environmental baseline	Santos has considered the matters raised in the OEMP and provided a response	Following the meeting in an email on 7 June 2024 Santos ex
studies had been undertaken.		nad been undertaken.	The answers provided were based on environmental studies described in Section 3 and summarised in Table 3-1 and Table 3-2.	Santos has completed ocean water (wind, wave, current, wat sediment quality, marine fauna, benthic habitat, noise, fish co shoals and shelf and seafloor (geophysical) studies. Santos I undertaken desktop assessments, such as for turtle activity.
				These studies are summarised in the OEMP as part of the as Regulator. The best sources of information are the Appendic Offshore Project Proposal (OPP) which are publicly available NOPSEMA website.
A co	mm	ittee member asked what happens if Santos exceeds	Santos has considered the matters raised in the OEMP and provided a response.	Following the meeting in an email on 7 June 2024 Santos ex
the t San	the total emissions for the year. As a consequence, would Santos shut down operations.		Santos will comply with the Regulatory requirements as defined in control measure BAO-CM-6.3.11 in the OEMP (Table 6-18):	The Clean Energy Regulator will set a baseline (Safeguard M Barossa greenhouse gas emissions, Santos will purchase or
			<ul> <li>Purchase and/or surrender of ACCUs or SMCs required under the NGER</li> <li>(Sector surrender of ACCUs or SMCs required under the NGER)</li> </ul>	Australian Carbon Credit Units (ACCUs) to offset Barossa's r emissions and any emissions above the Safeguard Mechanis
			(Sareguard Mechanism) Regulation Rule 2015 for any emissions from the Barossa facility (comprising the Barossa FPSO and GEP, including the Activity) above the annual baseline, as determined by the Clean Energy Regulator.	Santos will comply with the Regulator's requirements. Compa comply with the Safeguard Mechanism are subject to signific penalties from the Clean Energy Regulator.
A co time	mm s Sa	ittee member asked if there was a limit on how many antos can use the flare on the FPSO.	Questions related to Production Operations activities in the Barossa Field are outside the scope of the Coastal Waters OEMP and are addressed in the Barossa Production Operations EP.	Not applicable.
A co curre	mm ents	ittee member asked if Santos had a map of the around the FPSO.	Questions related to Production Operations activities in the Barossa Field are outside the scope of the Coastal Waters OEMP and are addressed in the Barossa Production Operations EP.	Not applicable.
Karo	lu L	alangkin (Daly River / Port Keats) Consultative Com	mittee	

- On 18 April 2024 Santos held a consultation session with the Kardu Lalingkin (Daly River/Port Keats) Consultative Committee in Wadeye. [Con-4253]
- The following information related to this OEMP. was presented and discussed:
  - The Commonwealth Government and NT Government regulations and approvals required
  - The activities covered by this OEMP.
  - The environmental impacts and risks involved with the planned activities and planned controls to management those risks

olained that: address which Santos Energy Regulator Units to offset er Australian	Not applicable
olained that: er quality), ommunities at has also esessment by the es to the Barossa on the	Not applicable
blained that: lechanism) for generate eservoir CO <sub>2</sub> sm baseline. anies which do not ant enforcement	Not applicable
	Not applicable
	Not applicable

- The EMBA in the event of an unplanned event, the risks and planned controls to management those risks
- The presentation also covered the regulatory consultation processes and privacy provisions and provided an overview of Santos the company and the Barossa Project overall.
- The Production Operations information booklet, Santos Privacy Statement and NOPSEMA consultation brochure was also provided at the consultation session.
- The sessions were conducted in person with the use of visual aids such as AO poster sized maps, AO posters with photos and images, PowerPoint presentation and videos to present information regarding the Activity and the project more generally.
- The majority of the engagement with attendees involved discussion to increase their understanding of the proposed activities and the associated risks and impacts, without any objections or claims about the adverse impact of each activity to which the Coastal Waters OEMP. relates. Refer below "Summary of response by Relevant Persons" below for further detail of general topics/themes discussed.
- A question raised during the session relating to fishing licences and how these may interact with sea country was not relevant to this OEMP. Questions on how to access Santos' employment and education opportunities or seek funding from Santos for community projects are managed outside the OEMP consultation process.
- On 10 July 2024 the consultant engaged by Santos to support establishment of the FN Consultative Committees, emailed the Kardu Lalingkin (Daly River/Port Keats) Consultative Committee, on behalf of Santos, to advise the consultation period had been completed. In the email Santos advised it considered that consultation had now closed for the purpose of Santos finalising and submitting environment plans for these activities to government regulators for assessment. Committee members were invited to pass the email on to other people in their network. [Con-5201]
- On 15 October 2024 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that the EP had been submitted to NOPSEMA for assessment, provided a link to the full EP on the NOPSEMA website and Santos' contact details for any queries. The Update also stated that the OEMP would be submitted to the NT-DITT for assessment. [Con- 5982]
- On 23 January 2025 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that, following a request from NOPSEMA to provide further information, Santos had re-submitted the EP with some amendments in December 2024 for further assessment. The Update also re-stated that the OEMP would be submitted to the NT-DITT for assessment. [Con-6036]
- No further correspondence or feedback was received from the Kardu Lalingkin (Daly River/Port Keats) Consultative Committee.

Summary of response by Relevant Person	Assessment of merits	Santos' Response Statement	OEMP reference
<ul> <li>Following discussion on consultation material, there were questions and comments received in relation to the matters set out below:</li> <li>Activity: <ul> <li>how the FPSO is moved to the field</li> <li>timeframe for FPSO being in the field</li> <li>depth of the pipeline.</li> </ul> </li> <li>Consultation &amp; communication <ul> <li>notification process in the event of a spill</li> </ul> </li> <li>Environment impacts &amp; risks <ul> <li>spills and spill response</li> <li>planning for the possibility of an accident occurring</li> </ul> </li> <li>Santos answered those questions and there were no responses raised for consultative Committee.</li> </ul>	<ul> <li>Committee members did not raise objections or claims about the adverse impact of each activity to which this OEMP relates.</li> <li>Questions related to Production Operations activities in the Barossa Field are outside the scope of the Coastal Waters OEMP and are addressed in the Barossa Production Operations EP.</li> <li>The answers provided at the meeting related to each activity to which this OEMP relates were based on the following sections of the OEMP:</li> <li>The activity described in Section 2, including for the design parameters.</li> <li>Consultation and communication described in Section 4 and required notifications in Sections 8.4.7 and 8.4.9.3.</li> <li>The environment risks &amp; impacts as well as the management controls for thematters raised include loss of hydrocarbon management including gas release and spills in Sections 7.4, 7.5 and 7.6</li> <li>Santos considers it has provided sufficient information and a reasonable period of time for consultation.</li> </ul>	Noting that no objection or claims were raised about the adverse impact of each activity to which the Coastal Waters OEMP relates, no further response required.	Not applicable
Djulidki (Bradshaw) Consultative Committee			

- On 19 April 2024 Santos held a consultation session with the Djulidki (Bradshaw) Consultative Committee in Darwin. [Con-4248]
- The following information related to this OEMP. was presented and discussed:
- The Commonwealth Government and NT Government regulations and approvals required
- The activities covered by this OEMP. 0
- The environmental impacts and risks involved with the planned activities and planned controls to management those risks 0
- The EMBA in the event of an unplanned event, the risks and planned controls to management those risks
- The presentation also covered the regulatory consultation processes and privacy provisions and provided an overview of Santos the company and the Barossa Project overall.
- The Production Operation information booklet, Santos Privacy Statement and NOPSEMA consultation brochure was also provided at the consultation session.
- The sessions were conducted in person with the use of visual aids such as AO poster sized maps, AO posters with photos and images, PowerPoint presentation and videos to present information regarding the Activity and the project more generally.
- The majority of the engagement with attendees involved discussion to increase their understanding of the proposed activities and the associated risks and impacts, without any objections or claims about the adverse impact of each activity to which the Coastal Waters OEMP relates. Refer below "Summary of response by Relevant Persons" below for further detail of general topics/themes discussed.
- A question related to notification in the event of a spill is addressed below. Questions on how to access Santos' employment and education opportunities or seek funding from Santos for community projects are managed outside the OEMP consultation process.
- On 10 July 2024 the consultant engaged by Santos to support establishment of the FN Consultative Committees, emailed the Djulidki (Bradshaw) Consultative Committee, on behalf of Santos, to advise the consultation period had been completed. In the email Santos advised it considered that consultation had now closed for the purpose of Santos finalising and submitting environment plans for these activities to government regulators for assessment. Committee members were invited to pass the email on to other people in their network. [Con-5201]



- On 15 October 2024 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that the EP had been submitted to NOPSEMA for assessment, provided a link to the full EP on the NOPSEMA website and Santos' contact details for any queries. The Update also stated that the OEMP would be submitted to the NT-DITT for assessment. [Con- 5982] On 23 January 2025 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that, following a request from NOPSEMA
- to provide further information, Santos had re-submitted the EP with some amendments in December 2024 for further assessment. The Update also re-stated that the OEMP would be submitted to the NT-DITT for assessment. [Con-6036]

<ul> <li>No further correspondence or feedback was received from the Djulidki (Bradshaw) Consultative Committee.</li> </ul>					
Summary of response by Relevant Person	Assessment of merits	Santos' Response Statement			
<ul> <li>Following discussion on consultation material, there were questions and comments received in relation to the matters set out below:</li> <li>Consultation &amp; communication <ul> <li>notification process in the event of a spill</li> </ul> </li> <li>Environment impacts &amp; risks <ul> <li>biosecurity risk management</li> <li>oil spill management</li> </ul> </li> <li>Santos answered those questions and there were no responses raised for consideration, other than those noted below</li> </ul>	<ul> <li>The answers provided at the meeting related to each activity to which this OEMP relates were based on the following sections of the OEMP:</li> <li>Consultation and communication described in Section 4 and required notifications in Sections 8.4.7 and 8.4.9.3.</li> <li>The environment risks &amp; impacts as well as the management controls for these described in Sections 6 and 7. Specific risks and impacts pertinent to the matters raised include: <ul> <li>Loss of hydrocarbon management including gas release and spills in Sections 7.4, 7.5 and 7.6 and required notifications in Sections 8.4.7 and 8.4.9.3.</li> </ul> </li> <li>Santos considers it has provided sufficient information and a reasonable period of time for consultation.</li> </ul>	Noting that no objection or claims were raised about the adve each activity to which the Coastal Waters OEMP relates, no for required			
A committee member stated that its major concerns were diesel spills and how Santos would let the committee know.	Santos has considered the matters raised in the OEMP and provided a response at the meeting.	At the meeting Santos agreed to notify the Committee Chair of heading towards the Djulidki (Bradshaw) Consultative Commi			
	Santos provided information on how the Djulidki (Bradshaw) Consultative Committee would be notified in the event of a spill.				

### Ngoy Garmak Consultative Committee

Summary of consultation effort:

- On 28 May 2024 Santos held a consultation session with the Ngoy Garmak Consultative Committee in Galiwin'ku, Elcho Island. [Con-4261]
- The following information related to this OEMP. was presented and discussed:
  - The activities covered by this OEMP.
  - The Commonwealth Government regulations and approvals required 0
  - The environmental impacts and risks involved with the planned activities and planned controls to management those risks 0
  - The EMBA in the event of an unplanned event, the risks and planned controls to management those risks 0
  - The regulatory consultation processes and privacy provisions 0
- The presentation also provided an overview of Santos the company and the Barossa Project overall, including a Project status update as per previous Barossa EPs
- The Production Operation information booklet, Santos Privacy Statement and NOPSEMA consultation brochure was also provided at the consultation session.
- The sessions were conducted in person with the use of visual aids such as AO poster sized maps, AO posters with photos, and videos to present information regarding the Activity and the project more generally.
- The majority of the engagement with attendees involved discussion to increase their understanding of the proposed activities and the associated risks and impacts, without any objections or claims about the adverse impact of each activity to which the Coastal Waters OEMP. relates. Refer below "Summary of response by Relevant Persons" below for further detail of general topics/themes discussed. A request was made by an attendee for a follow up meeting to discuss general matters about whale migration separate to the Coastal Waters OEMP.
- On 10 July 2024 the consultant engaged by Santos to support establishment of the FN Consultative Committees, emailed the Ngoy Garmak Peninsula Consultative Committee, on behalf of Santos, to advise the consultation period had been completed. In the email Santos advised it considered that consultation had now closed for the purpose of Santos finalising and submitting environment plans for these activities to government regulators for assessment. Committee members were invited to pass the email on to other people in their network. [Con-5201]
- On 15 October 2024 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that the EP had been submitted to NOPSEMA for assessment, provided a link to the full EP on the NOPSEMA website and Santos' contact details for any queries. The Update also stated that the OEMP would be submitted to the NT-DITT for assessment. [Con- 5982]
- On 17 December 2024 Santos met with two Ngoy Garmak Consultative Committee members to discuss general matters about whale migration separate to the Coastal Waters OEMP. The matters discussed were of a culturally sensitive nature and information unrelated to the Coastal Waters OEMP was provided on a confidential basis. During those meetings, the committee members each confirmed there were no outstanding matters with respect to the Coastal Waters OEMP. [Con-6025]

On 23 January 2025 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that, following a request from NOPSEMA to provide further information. Santos had re-submitted the EP with some amendments in December 2024 for further assessment. The Update also re-stated that the OEMP would be submitted to the NT-DITT for assessment. [Con-6036]

Summary of response by Relevant Person	Assessment of merits	Santos' Response Statement	OEMP reference
Following discussion on consultation material, there were questions and comments received in relation to the matters set out below:	<ul> <li>The answers provided at the meeting related to each activity to which this OEMP relates were based on the following sections of the OEMP:</li> <li>The activity described in Section 2, including for the design parameters.</li> </ul>	Noting that no objection or claims were raised about the adverse impact of each activity to which the Coastal Waters OEMP relates, no further response required.	Not applicable

	OEMP reference
erse impact of further response	Not applicable
of all spills ittee interests.	Notifications are included in Table 8-13

	-		
<ul> <li>Activity <ul> <li>extent of the geographical areas covered by the OEMP</li> <li>how long the pipeline will be in place</li> <li>Will the currents impact safety?</li> </ul> </li> <li>Consultation &amp; communication <ul> <li>notification process in the event of a spill</li> </ul> </li> <li>Existing Environment</li> <li>Protect the sea life and the seabed &amp; protect cultural values.</li> <li>Song lines within the EMBA, song lines record everything, rely on these for territorial boundaries and currents.</li> </ul> <li>Environment impacts &amp; risks <ul> <li>risks of a spill to people and the environment</li> <li>how particular areas within the EMBA could be impacted by a spill</li> <li>timeframe for responding to a spill</li> <li>monitoring of impacts to migrating whales.</li> <li>Can condensate harm people or the environment?</li> <li>Is it going to affect the seabed?</li> <li>Whales are important for this community and the spills would need to be monitored to protect them.</li> </ul> </li> <li>Other <ul> <li>OEMP approval process</li> <li>past incidence of spills in the region</li> </ul> </li>	<ul> <li>Consultation and communication described in Section 4 and required notifications in Sections 8.4.7 and 8.4.9.3.</li> <li>The existing environment described in Section 3 including for the baseline studies undertaken.</li> <li>The environment risks &amp; impacts as well as the management controls for these described in Sections 6 and 7. Specific risks and impacts pertinent to the matters raised include:</li> <li>Seabed disturbance - Section 6.4</li> <li>Potential impacts to marine life within a number of subsections in Section 6 and 7, and also in Section 7.3</li> <li>Loss of hydrocarbon management including gas release and spills in Sections 7.4, 7.5 and 7.6 and required notifications in Sections 8.4.7 and 8.4.9.3</li> <li>Santos considers it has provided sufficient information and a reasonable period of time for consultation.</li> <li>Santos considers Section 25 consultation requirements to have been met.</li> </ul>		
An attendee affirmed the cultural importance of song lines and stories related to whales migrating to and from the South Pacific. The attendee requested that Santos return to speak at another time to talk about these matters.	Santos has considered the matters raised and closed out the request, noting the matters did not related to the OEMP. Answers provided in the 28 May meeting generally relating to whales were based on marine mammal migration paths described in Section 3.2.13.1 and marine fauna interaction control measures described in Table 7-4. Santos has considered impacts to whales from planned activities and unplanned events in preparing the Coastal Waters OEMP.	At the meeting Santos explained that Santos would organise a meeting with the attendee to talk about these matters. Santos met with the attendee on 17 December 2024 and issues discussed were not related to the OEMP	Not applicable
Attendees advised of the importance of stories and song lines that run through the area and how information is recorded in the song lines. Song lines are believed to traverse from the bedrock in the land and out to the sea. Queries were raised about protecting the sea life, seabed and cultural values from potential environmental impacts within the EMBA and wanted to understand the timeframe for a spill response.	<ul> <li>Santos has considered the matters raised in the OEMP and provided a response at the meeting based on:</li> <li>Loss of hydrocarbon management including gas release and spills in Sections 7.4, 7.5 and 7.6</li> <li>Required notifications (including First Nations people and groups) in Sections 8.4.7 and 8.4.9.3.</li> <li>The OPEP outlines timeframes for spill response and includes the arrangements for activating trained Tiwi Rangers to assist with a spill response.</li> </ul>	At the meeting Santos explained spill response times and processes.	Not applicable
Miyarrka Consultative Committee			· · · · · · · · · · · · · · · · · · ·
Summary of concultation offerty			
• On 11 June 2024 Senter held a consultation appoint with th	n Mivarrka Cancultative Committee in Conjunively ICon 50001		
<ul> <li>The following information related to this OEMP was presented</li> <li>The Commonwealth Government regulations and approximation</li> </ul>	ed and discussed: ovals required		

- The activities covered by this OEMP
- The environmental impacts and risks involved with the planned activities and planned controls to management those risks

S	а	n	t	0	S

- The EMBA in the event of an unplanned event, the risks and planned controls to management those risks 0
- The presentation also covered the regulatory consultation processes and privacy provisions and provided an overview of Santos the company and the Barossa Project overall.
- The Production Operation information booklet, Santos Privacy Statement and NOPSEMA consultation brochure was also provided at the consultation session.
- The sessions were conducted in person with the use of visual aids such as AO poster sized maps, AO posters with photos, PowerPoint presentation, and videos to present information regarding the Activity and the project more generally.
- The majority of the engagement with attendees involved discussion to increase their understanding of the proposed activities and the associated risks and impacts, without any objections or claims about the adverse impact of each activity to which this OEMP relates. Refer below "Summary of response by Relevant Persons" below for further detail of general topics/themes discussed.
- Questions on how to access Santos' employment and education opportunities or seek funding from Santos for community projects are managed outside the OEMP consultation process.
- On 10 July 2024 the consultant engaged by Santos to support establishment of the FN Consultative Committees, emailed the Miyarrka Consultative Committee, on behalf of Santos, to advise the consultation period had been completed. In the email Santos advised it considered that consultation had now closed for the purpose of Santos finalising and submitting environment plans for these activities to government regulators for assessment. Committee members were invited to pass the email on to other people in their network. [Con-5201]
- On 15 October 2024 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that the EP had been submitted to NOPSEMA for assessment, provided a link to the full EP on the NOPSEMA website and Santos' contact details for any queries. The Update also stated that the OEMP would be submitted to the NT-DITT for assessment. [Con- 5982]
- On 23 January 2025 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that, following a request from NOPSEMA to provide further information, Santos had re-submitted the EP with some amendments in December 2024 for further assessment. The Update also re-stated that the OEMP would be submitted to the NT-DITT for assessment. [Con-6036]
- No further correspondence or feedback was received from the Miyarrka Consultative Committee.

Summary of	of response by Relevant Person	Assessment of merits	Santos' Response Statement		
Following di questions a out below: Activity risl con Environ risl env ma clir wh atm Santos answ responses r Committee.	iscussion on consultation material, there were nd comments received in relation to the matters set ks posed by natural disasters such as earthquakes d cyclones mposition of the gas and condensate ment impacts & risks ks of the gas and condensate to people and the vironment? anagement of planned discharges to water mate change, the vacuum that comes up to the air en you are doing the testing, does that affect the nosphere wered those questions and there were no raised for consideration, by Miyarrka Consultative	<ul> <li>There is no risk of a condensate spill from activities under this OEMP.</li> <li>The answers provided were based on: <ul> <li>Activity described in Section 2, including for the design parameters.</li> <li>The environment risks &amp; impacts as well as the management controls for these are described in Sections 6 and 7. Specific risks and impacts pertinent to the matters raised include: <ul> <li>Atmospheric and greenhouse emissions – Section 6.3</li> <li>Planned discharges – section 6.6</li> <li>Loss of hydrocarbon management including gas release and spills in Section 7.4, 7.5 and 7.6</li> </ul> </li> <li>Santos considers it has provided sufficient information and a reasonable period of time for consultation.</li> <li>Santos considers Section 25 consultation requirements to have been met.</li> </ul> </li> </ul>	Noting that no objection or claims were raised about the adv each activity to which the Coastal Waters OEMP. relates, no required.		
Maningrida	Appingrida Regional Consultative Committee				

- On 6 June 2024 Santos held a consultation session with the Maningrida Regional Consultative Committee in Maningrida. [Con-4262]
- The following information related to this OEMP was presented and discussed:
  - The activities covered by this OEMP.
  - The Commonwealth Government regulations and approvals required 0
  - The environmental impacts and risks involved with the planned activities and planned controls to management those risks 0
  - The EMBA in the event of an unplanned event, the risks and planned controls to management those risks 0
- The presentation also covered the regulatory consultation processes and privacy provisions and provided an overview of Santos the company and the Barossa Project overall.
- The Production Operation information booklet, Santos Privacy Statement and NOPSEMA consultation brochure was also provided at the consultation session.
- The sessions were conducted in person with the use of visual aids such as AO poster sized maps, AO posters with photos and images, PowerPoint presentation, and videos to present information regarding the Activity and the project more generally.
- The majority of the engagement with attendees involved discussion to increase their understanding of the proposed activities and the associated risks and impacts, without any objections or claims about the adverse impact of each activity to which this OEMP relates. Refer below "Summary of response by Relevant Persons" below for further detail of general topics/themes discussed.
- On 3 July 2024, Santos held a second consultation session with the Maningrida Regional Consultative Committee in Maningrida. [Con-5052]
- The following information related to this OEMP was presented and discussed:
  - The activities covered by this OEMP. 0
  - The Commonwealth Government regulations and approvals required 0
  - The environmental impacts and risks involved with the planned activities and planned controls to management those risks 0



	OEMP reference
erse impact of further response	Not applicable

- The EMBA in the event of an unplanned event, the risks and planned controls to management those risks 0
- The presentation also covered the regulatory consultation processes and privacy provisions and provided an overview of Santos the company and the Barossa Project overall.
- The Production Operation information booklet, Santos Privacy Statement and NOPSEMA consultation brochure was also provided at the consultation session.
- The sessions were conducted in person (with two people attending via teams) with the use of visual aids such as AO posters with photos and images, PowerPoint presentation, and videos to present information regarding the Activity and the project more generally.
- The majority of the engagement with attendees involved discussion to increase their understanding of the proposed activities and the associated risks and impacts, without any objections or claims about the adverse impact of each activity to which the Coastal Waters OEMP relates. Refer below "Summary of response by Relevant Persons" below for further detail of general topics/themes discussed.
- Questions on how to access Santos' employment and education opportunities and seek funding from Santos for community projects are managed outside the OEMP consultation process.
- On 10 July 2024 the consultant engaged by Santos to support establishment of the FN Consultative Committees, emailed the Maningrida Regional Consultative Committee, on behalf of Santos, to advise the consultation period had been completed. In the email Santos advised it considered that consultation had now closed for the purpose of Santos finalising and submitting environment plans for these activities to government regulators for assessment. Committee members were invited to pass the email on to other people in their network. [Con-5201]
- On 15 October 2024 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that the EP had been submitted to NOPSEMA for assessment, provided a link to the full EP on the NOPSEMA website and Santos' contact details for any queries. The Update also stated that the OEMP would be submitted to the NT-DITT for assessment. [Con- 5982]
- On 23 January 2025 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that, following a request from NOPSEMA to provide further information, Santos had re-submitted the EP with some amendments in December 2024 for further assessment. The Update also re-stated that the OEMP would be submitted to the NT-DITT for assessment. [Con-6036]
- No further correspondence or feedback was received from the Maningrida Regional Consultative Committee

Summary of response by Relevant Person	Assessment of merits	Santos' Response Statement	OEMP reference
Following discussion on consultation material, there were questions and comments received in relation to the following	Carbon Capture Storage will not be undertaken as an activity under this OEMP. Santos discussed items raised at the meeting. No further response required.	Not applicable	
matters: <ul> <li>Activity</li> </ul>	outside the scope of the Coastal Waters OEMP and are addressed in the Barossa Production Operations EP.		
<ul> <li>matters:</li> <li>Activity <ul> <li>maintaining stability of the pipeline</li> </ul> </li> <li>Consultation &amp; communication <ul> <li>Santos should talk to the saltwater ranger team and Outback Spirit.</li> </ul> </li> <li>Existing Environment <ul> <li>Marine pests are natural creatures, there are songlines that sing about marine pests. They have the right to live under water.</li> </ul> </li> <li>Environment impacts &amp; risks <ul> <li>regulation of chemical use</li> <li>safety measures for during movement of materials between vessels</li> <li>biosecurity management</li> <li>regulation of GHG emissions</li> <li>process involved in the event of an accident and a spill needing to be cleaned-up</li> <li>modelling of the EMBA</li> <li>how particular areas within the EMBA could be impacted by a spill</li> <li>tidal pattern means communities along the coast will be heavily impacted. It is just nature</li> <li>concerns that fish (their food source) will ingest</li> </ul> </li> </ul>	<ul> <li>Outside the scope of the Coastal Waters OEMP and are addressed in the Barossa Production Operations EP.</li> <li>The answers provided at the meeting related to each activity to which this OEMP relates were based on the following sections of the OEMP:</li> <li>The activity described in Section 2, including for the design parameters.</li> <li>The existing environment described in Section 3 including for the baseline studies undertaken.</li> <li>Consultation and communication described in Section 4 and required notifications in Sections 8.4.7 and 8.4.9.3</li> <li>The environment risks &amp; impacts as well as the management controls for these described in Sections 6 and 7. Specific risks and impacts pertinent to the matters raised include: <ul> <li>Atmospheric and greenhouse gas emissions – Section 6.3</li> <li>Interactions with other marine users – Section 6.5</li> <li>Seabed disturbance - Section 7.2</li> </ul> </li> <li>Potential impacts to marine life within a number of subsections in Section 6 and 7, and also in Section 7.3</li> <li>Potential impacts to fish addressed within a number of subsections in Section 6 and 7</li> <li>Loss of hydrocarbon management including gas release and spills in Sections 7.4, 7.5 and 7.6. With required notifications in Section 25 consultation requirements to have been met.</li> </ul>		
<ul><li>hydrocarbons and the potential for illness or death when the fish is eaten</li><li>Other</li></ul>			
safety of CCS			
<ul> <li>concerns from an historic event where a freighter travelling across the north of Australia from Liverpool coincided with event of a lot of dead fish, uncertain if there was a chemical spill or not. We didn't know, no one told us.</li> </ul>			



Santos answered those questions and there were no responses raised for consideration by the Maningrida Regional Consultative Committee	

### Gapu Maringa Consultative Committee

Summary of consultation effort:

- On 5 June 2024 Santos held a consultation session with the Gapu Maringa Consultative Committee in Milingimbi. [Con-5002]
- The following information related to this OEMP was presented and discussed:
  - 0 The Commonwealth Government regulations and approvals required
  - The activities covered by this OEMP. 0
  - 0 The environmental impacts and risks involved with the planned activities and planned controls to management those risks
  - The EMBA in the event of an unplanned event, the risks and planned controls to management those risks 0
- The presentation also covered the regulatory consultation processes and privacy provisions and provided an overview of Santos the company and the Barossa Project overall.
- The Production Operation information booklet, Santos Privacy Statement and NOPSEMA consultation brochure was also provided at the consultation session.
- The sessions were conducted in person with the use of visual aids such as AO poster sized maps, AO posters with photos and images, and videos to present information regarding the Activity and the project more generally.
- The majority of the engagement with attendees involved discussion to increase their understanding of the proposed activities and the associated risks and impacts, without any objections or claims about the adverse impact of each activity to which the Coastal Waters OEMP. relates. Refer below "Summary of response by Relevant Persons" below for further detail of general topics/themes discussed.
- A question raised during the session relating to the extent of First Nations ownership and control of coastal and inland waters was not relevant to the Coastal Waters OEMP.
- Questions on how to access Santos' employment and education opportunities or seek funding from Santos for community projects are managed outside the OEMP consultation process.
- On 10 July 2024 the consultant engaged by Santos to support establishment of the FN Consultative Committees, emailed the Gapu Maringa Consultative Committee, on behalf of Santos, to advise the consultation period had been completed. In the email Santos advised it considered that consultation had now closed for the purpose of Santos finalising and submitting environment regulators for assessment. Committee members were invited to pass the email on to other people in their network. [Con-5201]
- On 15 October 2024 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that the EP had been submitted to NOPSEMA for assessment, provided a link to the full EP on the NOPSEMA website and Santos' contact details for any queries. The Update also stated that the OEMP would be submitted to the NT-DITT for assessment. [Con- 5982]
- On 23 January 2025 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that, following a request from NOPSEMA to provide further information, Santos had re-submitted the EP with some amendments in December 2024 for further assessment. The Update also re-stated that the OEMP would be submitted to the NT-DITT for assessment. [Con-6036]

### No further correspondence or feedback was received from the Gapu Maringa Consultative Committee.

Summary of response by Relevant Person	Assessment of merits	Santos' Response Statement	OEMP reference
Following discussion on consultation material, there were questions and comments received in relation to the following matters:	Questions related to Production Operations activities in the Barossa Field are outside the scope of the Coastal Waters OEMP and are addressed in the Barossa Production Operations EP.	Noting that no objection or claims were raised about the adverse impact of each activity to which the Coastal Waters OEMP relates, no further response required.	Not applicable
<ul> <li>Activity <ul> <li>composition of the gas and condensate</li> <li>What are those pipes there, will they come off in the wind?</li> </ul> </li> <li>Consultation &amp; communication <ul> <li>Notification if there was a crash or spill</li> </ul> </li> <li>Existing Environment <ul> <li>extent of the geographical areas covered by the Coastal Waters OEMP.</li> </ul> </li> <li>Environment impacts &amp; risks <ul> <li>wastewater management</li> <li>how particular areas within the EMBA could be impacted by a spill</li> <li>how the risk of vessel collisions is managed</li> <li>oil spill management</li> <li>concerns on gas explosions &amp; potential for gas and rubbish to impact their land</li> <li>Santos to protect the land and sea and activity is safe.</li> <li>Concerns on vessel collision and what if spill travels their way</li> </ul> </li> </ul>	<ul> <li>The answers provided at the meeting related to each activity to which this OEMP relates were based on the following sections of the OEMP:</li> <li>The activity described in Section 2, including for the design parameters.</li> <li>The existing environment described in Section 3 including for the baseline studies undertaken.</li> <li>Consultation and communication described in Section 4 and required notifications in Sections 8.4.7 and 8.4.9.3.</li> <li>The environment risks &amp; impacts as well as the management controls for these described in Sections 6 and 7. Specific risks and impacts pertinent to the matters raised include: <ul> <li>Physical presence – Section 6.5</li> <li>Waste management – Section 7.2</li> </ul> </li> <li>Loss of hydrocarbon management including gas release and spills in Sections 7.4, 7.5 and 7.6 and required notifications in Sections 8.4.7 and 8.4.9.3.</li> <li>The relevant legislative requirements described in Appendix B.</li> </ul> <li>Santos considers it has provided sufficient information and a reasonable period of time for consultation.</li>		



• Other
Can the Traditional Owners say yes or no.
Regulator process and penalties for breaches
FN ownership and control of coastal and inland waters
Santos answered those questions and there were no responses raised for consideration by the Gapu Maringa Consultative Committee.
Jindiwi Consultative Committee

- On 30 April 2024 Santos held a consultation session with the Jindiwi Consultative Committee in Jabiru. Representatives at this meeting were expected to include Wulna clan representatives, however they were not able to attend on the day. [Con-4251]
- The following information related to this OEMP was presented and discussed at both consultation sessions:
- The Commonwealth Government and NT Government regulations and approvals required 0
- The activities covered by this OEMP. 0
- 0 The environmental impacts and risks involved with the planned activities and planned controls to management those risks
- The EMBA in the event of an unplanned event, the risks and planned controls to management those risks 0
- The presentation also covered the regulatory consultation processes and privacy provisions and provided an overview of Santos the company and the Barossa Project overall.
- The Production Operations information booklet, Santos Privacy Statement and NOPSEMA consultation brochure was also provided at the consultation session.
- The sessions were conducted in person with the use of visual aids such as AO poster sized maps, AO posters with photos and images, PowerPoint presentation and videos to present information regarding the Activity and the project more generally.
- The majority of the engagement with attendees involved discussion to increase their understanding of the proposed activities and the associated risks and impacts, without any objections or claims about the adverse impact of each activity to which the Coastal Waters OEMP relates. Refer below "Summary of response by Relevant Persons" below for further detail of general topics/themes discussed.
- A question was raised at the meeting on potential access to community project funding for monitoring of waters .At the meeting Santos advised the committee that it would be happy to have further discussions on this and any requests seeking access to funding for community projects.
- Questions on how to access Santos' employment and education opportunities or seek funding from Santos for community projects are managed outside the consultation process.
- On 10 July 2024 the consultation period had been completed. In the email Santos advised it considered that consultation had now closed for the purpose of Santos finalising and submitting environment regulators for assessment. Committee members were invited to pass the email on to other people in their network. [Con-5201]
- Notwithstanding, the communication on 10 July, the consultation period for the Jindiwi Consultative Committee was subsequently extended to allow an opportunity for the Jindiwi Consultative Committee to share consultation information with and obtain input for the OEMP (if any) from the Wulna clan members who had not been in attendance at the earlier consultation meeting with the Jindiwi Consultative Committee on 30 April 2024. On 28 July 2024 the consultant engaged by Santos to support establishment of the First Nations Consultative Committees confirmed that:
  - The Jindiwi Consultative Committee had previously agreed to Wulna clan participation through the Jindiwi Consultative Committee.
  - 0 On 27 July 2024 the consultant, in his role of providing secretariat support to the Jindiwi Consultative Committee, and one of his cultural advisers met with the Wulna clan members who had not been in attendance at the Jindiwi Consultative Committee meeting on 30 April 2024.
  - During the 27 July 2024 meeting the Wulna clan members formally accepted membership into the Jindiwi Consultative Committee. The meeting also involved the consultant discussing and providing the Barossa Production Operations consultation materials to the 0 Wulna clan consultative committee members on behalf of the JCC and inviting any input for the OEMP to pass onto Santos. No concerns were raised by the Wulna clan consultative committee members on the consultation material or the proposed activities. [Con-52801
  - The 27 July meeting occurred as a key role of the consultative committee (as stated in their charters) is for the dissemination to First Nations community members of relevance, which was undertaken in this instance with this meeting and the addition of Wulna clam members into the Jindiwi CC.
- On 15 October 2024 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that the EP had been submitted to NOPSEMA for assessment, provided a link to the full EP on the NOPSEMA website and Santos' contact details for any queries. The Update also stated that the OEMP would be submitted to the NT-DITT for assessment. [Con- 5982]
- On 23 January 2025 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that, following a request from NOPSEMA to provide further information, Santos had re-submitted the EP with some amendments in December 2024 for further assessment. The Update also re-stated that the OEMP would be submitted to the NT-DITT for assessment. [Con-6036]
- No further correspondence or feedback was received from the Jindiwi Consultative Committee.

Summary of response by Relevant Person	Assessment of merits	Santos' Response Statement
<ul> <li>Following discussion on consultation material, there were questions and comments received in relation to the following matters:</li> <li>Activity <ul> <li>Existing risks posed by natural disasters such as earthquakes and cyclones</li> </ul> </li> <li>Environment impacts &amp; risks</li> </ul>	<ul> <li>The answers provided at the meeting related to each activity to which this OEMP relates were based on the following sections of the OEMP:</li> <li>The activity described in Section 2, including for the design parameters. <ul> <li>The environment risks &amp; impacts as well as the management controls for these described in Sections 6 and 7. Specific risks and impacts pertinent to the matters raised include biosecurity management in Section 7.2</li> <li>Potential impacts to marine life within a number of subsections in Section 6 and 7, and also in Section 7.3</li> <li>Loss of hydrocarbon management including gas release and spills in Sections 7.4, 7.5 and 7.6 and required notifications in Sections 8.4.7 and 8.4.9.3.</li> </ul> </li> </ul>	Noting that no objection or claims were raised about the adve each activity to which the Coastal Waters OEMP relates, no required.



	OEMP reference
erse impact of further response	Not applicable

•	how particular areas within the EMBA could be impacted by a spill, oil spill management, which area would be affected near Arnhem land?	Santos considers it has provided sufficient information and a reasonable period of time for consultation.	
•	impact of a spill on marine life -what would happen if the wind changes and will it affect the turtles and dugongs?	Santos considers Section 25 consultation requirements to have been met	
•	biosecurity management – where will the FPSO be checked for marine life in NT waters or Australian waters?		
Santos respons Commi	answered those questions and there were no ses raised for consideration, by the Jindiwi Consultative tee.		
Wulna	Clan (via the Jindiwi Consultative Committee)		

- On 30 April 2024 Santos held a consultation session with the Jindiwi Consultative Committee in Jabiru. Wulna representatives were expected at this meeting, however, were not able to attend on the day. [Con-4251]
- On 27 July 2024, the consultant engaged by Santos to support establishment of the First Nations Consultative Committees, in his role of providing secretariat support to the Jindiwi Consultative Committee, met with Wulna clan members who had not been able to attend the 30 April meeting. During the 27 July 2024 meeting the Wulna clan members formally accepted membership into the Jindiwi Consultative Committee. The meeting also involved the consultant discussing and providing the Barossa Production Operations consultation materials to the Wulna clan consultative committee members on behalf of the Jindiwi Consultative Committee and inviting any input for the OEMP and EP to pass onto Santos. No concerns were raised by the Wulna clan consultative committee members on the consultation material or the proposed activities. [Con-5280]
- On 15 October 2024 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that the EP had been submitted to NOPSEMA for assessment, provided a link to the full EP on the NOPSEMA website and Santos' contact details for any queries. The Update also stated that the OEMP would be submitted to the NT-DITT for assessment. [Con- 5982]
- On 23 January 2025 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that, following a request from NOPSEMA to provide further information. Santos had re-submitted the EP with some amendments in December 2024 for further assessment. The Update also re-stated that the OEMP would be submitted to the NT-DITT for assessment. [Con-6036]

See also the entry in this Table for the summary of consultation with the Jindiwi Consultative Committee including Wulna clan representatives.

Summary of response by Relevant Person	Assessment of merits	Santos' Response Statement
There were no responses raised for consideration, by the Wulna Clan	Santos considers it has provided sufficient information and a reasonable period of time for consultation.	Noting that no objection or claims were raised about the advected activity to which this OEMP. relates, no further respon
	Santos considers Section 25 consultation requirements to have been met.	

### Mulyurrud Consultative Committee (Croker Island)

- On 16 April 2024 Santos held a consultation session with the Mulyurrud Consultative Committee in Darwin. [Con-4234]
- The following information related to the Coastal Waters OEMP was presented and discussed:
  - o The Commonwealth Government and NT Government regulations and approvals required
  - The activities covered by this OEMP.
  - The environmental impacts and risks involved with the planned activities and planned controls to management those risks 0
  - The EMBA in the event of an unplanned event, the risks and planned controls to management those risks
- The presentation also covered the regulatory consultation processes and privacy provisions and provided an overview of Santos the company and the Barossa Project overall.
- The Production Operations information booklet, Santos Privacy Statement and NOPSEMA consultation brochure was also provided at the consultation session.
- The session was conducted in person with the use of visual aids such as AO poster sized maps, AO posters with photos and images, PowerPoint presentation and videos to present information regarding the Activity and the project more generally.
- The majority of the engagement with attendees involved discussion to increase their understanding of the proposed activities and the associated risks and impacts, without any objections or claims about the adverse impact of each activity to which the Coastal Waters OEMP relates. Refer below "Summary of response by Relevant Persons" below for further detail of general topics/themes discussed.
- A copy of the presentation was requested and provided to the committee on 25 June 2024 [Con-5003].
- A cultural advisor helping to facilitate the session noted at the beginning of the consultation session that some words were difficult to understand and that pictures would help. During this meeting, three large (A1 size) maps were adhered to the walls, there were multiple A3 size maps and diagrams handed out, and the presentation which followed made strong use of graphics. At least two FN Consultative Committees cultural advisors were present who were able to translate, and who did so on occasion (i.e during this meeting to address the comment made).
- Separate to the consultation for the Coastal Waters OEMP, the committee requested Santos speak to the committee about purchasing carbon credits and providing funding support for ranger groups. Organisation of this meeting is managed separately to the consultation process.
- Separate to the consultation for the Coastal Waters OEMP, an attendee stated that the committee needed time to discuss Santos proposed community investment activities and requested a follow-up discussion. Questions and requests on how to access Santos' employment and education opportunities or seek funding from Santos for community projects are managed outside the OEMP consultation process.
- On 10 July 2024 the consultant engaged by Santos to support establishment of the FN Consultative Committees, emailed the Mulyurrud Consultative Committee, on behalf of Santos, to advise the consultation period had been completed. In the email Santos advised it considered that consultation had now closed for the purpose of Santos finalising and submitting environment regulators for assessment. Committee members were invited to pass the email on to other people in their network. [Con-5201]
- On 15 October 2024 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that the EP had been submitted to NOPSEMA for assessment, provided a link to the full EP on the NOPSEMA website and Santos' contact details for any queries. The Update also stated that the OEMP would be submitted to the NT-DITT for assessment. [Con- 5982]



	OEMP reference
erse impact of e required.	Not applicable

- During a call on 5 December 2024 to follow up matter not relating to the Coastal Waters OEMP, the Mulyurrud Consultative Committee chair confirmed that there were no outstanding matters with respect to the Coastal Waters OEMP. [Con-6016]
- On 23 January 2025 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that, following a request from NOPSEMA • to provide further information, Santos had re-submitted the EP with some amendments in December 2024 for further assessment. The Update also re-stated that the OEMP would be submitted to the NT-DITT for assessment. [Con-6036]
- No further correspondence or feedback was received from the Mulyurrud Consultative Committee. •

Summary of response by Relevant Person	Assessment of merits	Santos' Response Statement
<ul><li>Following discussion on consultation material, there were questions and comments received in relation to the following matters:</li><li>Activity</li></ul>	Carbon Capture Storage will not be undertaken as an activity under this OEMP. Questions related to Production Operations activities in the Barossa Field are outside the scope of the Coastal Waters OEMP and are addressed in the Barossa Production Operations EP.	Noting that no objection or claims were raised about the adve each activity to which the Coastal Waters OEMP relates, no for required.
<ul> <li>structural integrity of the FPSO during a tsunami, cyclone or natural disaster.</li> </ul>	The answers provided at the meeting related to each activity to which this OEMP relates were based on the following sections of the OEMP:	
<ul> <li>what happens to the gas and condensate</li> </ul>	• The Activity described in Section 2, including for the design parameters.	
Consultation & communication	The existing environment described in Section 3.	
<ul> <li>what is involved in the consultation process</li> </ul>	<ul> <li>Santos Consultation and communication described in Section 4 and required notifications in Sections 8.4.7 and 8.4.9.3</li> </ul>	
Existing Environment	- The environment risks & impacts as well as the management controls for	
<ul> <li>the environmental studies that have been undertaken by Santos</li> </ul>	these described in Sections 6 and 7. Specific risks and impacts pertinent to the matters raised include physical presence described in Section 6.5	
<ul> <li>(in the context of marine life in the environment where the FPSO is located) Is it empty? What about migration?</li> </ul>	• Potential impacts to marine life within a number of subsections in Section 6 and 7, and also in Section 7.3	
Environment impacts & risks	Loss of hydrocarbon management including gas release and splits in Sections 7.4, 7.5 and 7.6 and required notifications in Sections 8.4.7 and 8.4.9.3	
<ul> <li>the environmental impact of the project generally and to marine life</li> </ul>	<ul> <li>The Barossa Development Drilling and Completions EP – accepted by NOPSEMA in December 2023.</li> </ul>	
how far planned discharges extend from the FPSO	Santos considers it has provided sufficient information and a reasonable period of	
<ul> <li>impact of exclusion zones on other activities such as fishing or defence patrols - (in the context of the 500m exclusion zone around the FPSO) – affecting fishing. There are a lot of patrols, military exercises/presences, will they be affected?</li> </ul>	time for consultation. Santos considers Section 25 consultation requirements to have been met	
<ul> <li>Impacts to marine fauna eg bottlenose dolphins, seen fauna washed up land.</li> </ul>		
<ul> <li>Visibility of vessels and physical presence (buoys &amp; lighting safety)</li> </ul>		
<ul> <li>Volumes of carbon dioxide released over the lifetime of the Barossa, where does it go and associated climate change impacts</li> </ul>	me	
Spill EMBA		
<ul> <li>Spill response preparedness and activities &amp; looking after country, the process and timing for responding to a spill</li> </ul>	ng ig to	
<ul> <li>Offsetting of emissions through purchase of carbon credits &amp; Carbon Capture and Storage</li> </ul>		
• Other		
<ul> <li>what is involved in the drilling process, e.g. depth drilled, size of the drill, can earthquakes or tsunamis be caused by drilling?</li> </ul>		
how a permit area for drilling is determined		
<ul> <li>Santos answered those questions and there were no responses raised for consideration, by Mulyurrud Consultative Committee.</li> </ul>		
Rak Badjalarr Consultative Committee		

On 17 April 2024 Santos held a consultation session with the Rak Badjalarr Committee at the Rydges Palmerston in Darwin. [Con-4233]



	OEMP reference
erse impact of further response	Not applicable

- The following information related to this OEMP. was presented and discussed:
  - The Commonwealth Government and NT Government regulations and approvals required
  - The activities covered by this OEMP.
  - The environmental impacts and risks involved with the planned activities; and planned controls to management those risks and 0
  - The EMBA in the event of an unplanned event, the risks and planned controls to management those risks. 0
- The presentation also covered the regulatory consultation processes and privacy provisions and provided an overview of Santos the company and the Barossa Project overall.
- The information booklet and NOPSEMA consultation brochure were also provided at the consultation session.
- The sessions were conducted in person with the use of visual aids such as AO poster sized maps, AO posters with photos and images, PowerPoint presentation and videos to present information regarding the Activity and the project more generally.
- The majority of the engagement with attendees involved discussion to increase their understanding of the proposed activities and the associated risks and impacts, without any objections or claims about the adverse impact of each activity to which the Coastal Waters OEMP. relates. Refer below "Summary of response by Relevant Persons" below for further detail of general topics/themes discussed.
- Santos provided the committee with videos presented at the meeting following a request from a committee member.
- Santos was requested to send through the videos following the meeting. A Santos staff member sent through the link to the videos on 17 April 2024 via SMS via to the contact phone number for the RBCC member [Con-6004]. During the meeting, information was provided by a committee member on a confidential basis related to a Dreamtime story and the protection of a totem species, which was referred to in the minutes as a sacred site. The issue does not relate specifically to the Coastal Waters OEMP and was the subject of separate discussions with the committee member and Santos' third-party cultural advisors, including an anthropological expert who considered this matter as part of their report on other aspects of the Barossa Project (https://www.santos.com/wp-content/uploads/2024/09/First-Nations-spiritual-and-cultural-values-in-relation-to-the-Darwin-Pipeline-Duplication-Project.pdf ) and concluded this matter was not relevant to the Coastal Waters OEMP.
- Questions on how to access Santos' employment and education opportunities or seek funding from Santos for community projects are managed outside the OEMP consultation process.
- On 10 July 2024 the consultant engaged by Santos to support establishment of the FN Consultative Committees, emailed the Rak Badialarr Consultative Committee, on behalf of Santos, to advise the consultation period had been completed. In the email Santos advised it considered that consultation had now closed for the purpose of Santos finalising and submitting environment regulators for assessment. Committee members were invited to pass the email on to other people in their network. [Con-52011
- On 15 October 2024 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that the EP had been submitted to NOPSEMA for assessment, provided a link to the full EP on the NOPSEMA website and Santos' contact details for any queries. The Update also stated that the OEMP would be submitted to the NT-DITT for assessment. [Con- 5982]
- On 23 January 2025 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that, following a request from NOPSEMA to provide further information, Santos had re-submitted the EP with some amendments in December 2024 for further assessment. The Update also re-stated that the OEMP would be submitted to the NT-DITT for assessment. [Con-6036]

### No further correspondence or feedback was received from the Rak Badjalarr Consultative Committee.

Summary of response by Relevant Person	Assessment of merits	Santos' Response Statement
Following discussion on consultation material, there were questions and comments received in relation to the following matters:	Carbon Capture Storage will not be undertaken as an activity under this OEMP. Questions related to Production Operations activities in the Barossa Field are outside the scope of the Coastal Waters OEMP and are addressed in the Barossa Production Operations EP	Noting that no objection or claims were raised about the adve each activity to which the Coastal Waters OEMP. relates, no required.
<ul> <li>Activity <ul> <li>details of the FPSO, e.g. its size, how it gets to the field and anchored</li> <li>What vessels will be coming through the waters</li> <li>structural integrity of the FPSO during a tsunami, cyclone or natural disaster.</li> <li>the pipeline route and nature of the seabed, does the pipeline go through the marine park?</li> </ul> </li> <li>Consultation &amp; communication <ul> <li>how privacy of FN people and information they provide is protected</li> </ul> </li> <li>Environment impacts &amp; risks <ul> <li>how other marine users are notified of facilities and activities</li> <li>presence of vessels and how do TO hunters know where they are?</li> <li>How an EMBA is determined and modelled, is the oil on top on the water</li> <li>GHG emissions management</li> </ul> </li> <li>Other <ul> <li>how carbon capture and storage works</li> <li>previous incidents and how they are recorded</li> </ul> </li> </ul>	<ul> <li>Production Operations EP.</li> <li>The answers provided at the meeting related to each activity to which this OEMP relates were based on the following sections of the OEMP:</li> <li>The activity described in Section 2, including for the design parameters.</li> <li>The existing environment described in Section 3</li> <li>Consultation and communication described in Section 4 and required notifications in Sections 8.4.7 and 8.4.9.3.</li> <li>The environment risks &amp; impacts as well as the management controls for these described in Sections 6 and 7. Specific risks and impacts pertinent to the matters raised include physical presence described in Section 6.5</li> <li>Loss of hydrocarbon management including gas release and spills in Sections 7.4, 7.5 and 7.6 and required notifications in Sections 8.4.7 and 8.4.9.3.</li> <li>Santos considers it has provided sufficient information and a reasonable period of time for consultation.</li> <li>Santos considers Section 25 consultation requirements to have been met.</li> </ul>	



	OEMP reference
erse impact of further response	OEMP reference Not applicable

### **Goulburn Island Consultative Committee**

Summary of consultation effort:

- On 30 May 2024 Santos held a consultation session with the Goulburn Island Consultative Committee at the Warruwi (South Goulburn Island) Yagbani Aboriginal Corporation Meeting Room. [Con-5005]
- The following information related to this OEMP. was presented and discussed:
- The Commonwealth Government regulations and approvals required
- 0 The activities covered by this OEMP.
- The environmental impacts and risks involved with the planned activities and planned controls to management those risks 0
- The EMBA in the event of an unplanned event, the risks and planned controls to management those risks
- The presentation also covered the regulatory consultation processes and privacy provisions and provided an overview of Santos the company and the Barossa Project overall.
- The Production Operation information booklet. Santos Privacy Statement and NOPSEMA consultation brochure was also provided at the consultation session.
- The sessions were conducted in person with the use of visual aids such as AO poster sized maps, AO posters with photos and images, and videos to present information regarding the Activity and the project more generally. The majority of the engagement with attendees involved discussion to increase their understanding of the proposed activities and the associated risks and impacts, without any objections or claims about the adverse impact of each activity to which the Coastal Waters
- OEMP. relates. Refer below "Summary of response by Relevant Persons" below for further detail of general topics/themes discussed.
- Questions on how to access Santos' employment and education opportunities or seek funding from Santos for community projects are managed outside the OEMP consultation process.
- On 10 July 2024 the consultant engaged by Santos to support establishment of the FN Consultative Committees, emailed the Goulburn Island Consultative Committee, on behalf of Santos, to advise the consultation period had been completed. In the email Santos advised it considered that consultation had now closed for the purpose of Santos finalising and submitting environment plans for these activities to government regulators for assessment. Committee members were invited to pass the email on to other people in their network. [Con-5201]
- On 15 October 2024 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that the EP had been submitted to NOPSEMA for assessment, provided a link to the full EP on the NOPSEMA website and Santos' contact details for any queries. The Update also stated that the OEMP would be submitted to the NT-DITT for assessment. [Con- 5982]
- On 23 January 2025 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that, following a request from NOPSEMA to provide further information, Santos had re-submitted the EP with some amendments in December 2024 for further assessment. The Update also re-stated that the OEMP would be submitted to the NT-DITT for assessment. [Con-6036]
- No further correspondence or feedback was received from the Goulburn Island Consultative Committee.

Summary of response by Relevant Person	Assessment of merits	Santos' Response Statement
Following discussion on consultation material, there were questions and comments received in relation to the following matters:	Questions related to Production Operations activities in the Barossa Field are outside the scope of the Coastal Waters OEMP and are addressed in the Barossa Production Operations EP.	Noting that no objection or claims were raised about the adve each activity to which the Coastal Waters OEMP relates, no required.
<ul> <li>Activity</li> <li>pipeline installation process</li> <li>structural integrity of the pipeline during a tsunami, cyclone or natural disaster.</li> <li>how long the FPSO will be located in the field</li> <li>Environment impacts &amp; risks</li> <li>dropped objects</li> <li>process and timing for responding to a spill</li> <li>how particular areas within the EMBA could be impacted by a spill</li> <li>interference of pipeline with fishing</li> <li>how are people advised of facility locations.</li> </ul> Santos answered those questions and there were no responses raised for consideration, by the Goulburn Island Consultative Committee.	<ul> <li>The answers provided at the meeting related to each activity to which this OEMP relates were based on the following sections of the OEMP:</li> <li>The activity described in Section 2, including for the design parameters.</li> <li>The existing environment described in Section 3 including for the baseline studies undertaken.</li> <li>Consultation and communication described in Section 4 and required notifications in Sections 8.4.7 and 8.4.9.3</li> <li>The environment risks &amp; impacts as well as the management controls for these described in Sections 6 and 7. Specific risks and impacts pertinent to the matters raised include: <ul> <li>Physical presence – Section 6.5</li> <li>Dropped objects – Section 7.1</li> </ul> </li> <li>Potential impacts to marine life within a number of subsections in Sections 6 and 7, and also in Section 7.3</li> <li>Loss of hydrocarbon management including gas release and spills in Sections 7.4, 7.5 and 7.6 and required notifications in Sections 8.4.7 and 8.4.9.3.</li> </ul>	
	Santos considers Section 25 consultation requirements to have been met.	

### First Nations People and groups: Other First Nations organisations - Northern Territory

### Aboriginal Sea Company (ASC)

- On 12 March 2024 Santos emailed Aboriginal Sea Company advising that it is consulting on the Coastal Waters OEMP and Barossa Production Operations Environment Plan and asks for consideration as to whether their organisation considers itself a Relevant Person. Santos advised that the consultation is open until Tuesday 9 April 2024. [Con-4049]
- On 15 March 2024 Santos met with the ASC in Darwin. At the meeting Santos explained the Production Operations activities and risks involved. The ASC executive officer advised they had read through the information and did not have any feedback. The officer advised that the ASC typically does not provide comment on environmental matters when approached by oil and gas companies. [Con-6012]
- On 15 March 2024 Aboriginal Sea Company thanked Santos via email for the meeting and provided its Capability Statement. [Con-5235]



	OEMP reference
erse impact of further response	Not applicable

- On 5 June 2024 Santos emailed ASC to thank it for meeting and to advise it had passed on the ASC's information to relevant Santos staff. Santos thanked ASC for taking the time to review the information in the context of ASC typically not providing comment on environment plans, and noted it has no specific feedback on this occasion. [Con-4363]
- On 10 July 2024 Santos emailed ASC to advise the consultation period had been completed. Santos advised it considered that consultation had now closed for the purpose of Santos finalising and submitting environment plans for these activities to government regulators for assessment. [Con-5130]
- On 15 October 2024 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that the EP had been submitted to NOPSEMA for assessment, provided a link to the full EP on the NOPSEMA website and Santos' contact details for any queries. The Update also stated that the OEMP would be submitted to the NT-DITT for assessment. [Con- 5982]
- On 23 January 2025 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that, following a request from NOPSEMA to provide further information, Santos had re-submitted the EP with some amendments in December 2024 for further assessment. The Update also re-stated that the OEMP would be submitted to the NT-DITT for assessment. [Con-6036]

Notwithstanding the information provided and the steps described above, no comments or input were received on the Coastal Waters OEMP. from ASC.		
Assessment of merits		
Santos considers it has provided sufficient information and a reasonable period of time for consultation.		
Santos considers Section 25 consultation requirements to have been met.		

### Gwalwa Daraniki Association (GDA)

Summary of consultation effort:

- On 12 March 2024 Santos emailed GDA to advise that it had commenced the consultation phase consulting for the Coastal Waters OEMP and Barossa Production Operations Environment Plan which would run until 9 April 2024. Santos provided information on Relevant Persons' entitlements under the regulatory processes, details of how to provide feedback and a reminder of the closing date for consultation. [Con-4050]
- On 2 May 2024 Santos emailed GDA further to previous correspondence, to advise it had extended the consultation period until 16 May 2024. In providing this extension of time, Santos advised that the information in the booklet and factsheet had been updated to account for an additional risk associated with the proposed activity. Santos confirmed that consultation will close on the revised date unless Santos hears otherwise from GDA. [Con-4102]
- GDA was invited (by Larrakia Development Corporation) to a meeting between Larrakia Development Corporation and Santos on 20 June 2024 but did not attend.
- On 10 July 2024 Santos emailed GDA to advise the consultation period had been completed. Santos advised it considered that consultation had now closed for the purpose of Santos finalising and submitting environment plans for these activities to government regulators for assessment. [Con-5130]
- On 27 and 28 August 2024 Santos attempted to contact GDA via phone to check whether the Association had intended to comment on Production Operations activities. The calls were not answered and there was no ability to leave message. [Con-5611]
- On 15 October 2024 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that the EP had been submitted to
- NOPSEMA for assessment, provided a link to the full EP on the NOPSEMA website and Santos' contact details for any queries. The Update also stated that the OEMP would be submitted to the NT-DITT for assessment. [Con- 5982]
   On 23 January 2025 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that, following a request from NOPSEMA
- to provide further information, Santos had re-submitted the EP with some amendments in December 2024 for further assessment. The Update also re-stated that the OEMP would be submitted to the NT-DITT for assessment. [Con-6036]
- Notwithstanding the information provided and the steps described above, no comments or input were received on the Coastal Waters OEMP from GDA.

Summary of response by Relevant Person	Assessment of merits
No response was received from GDA.	Santos considers it has provided sufficient information and a reasonable period of time for consultation. Santos considers Section 25 consultation requirements to have been met.

### Kenbi Rangers

Summary of consultation effort:

- On 13 February 2024 Santos emailed the NLC, which also has responsibility for the Kenbi Rangers, to advise it of preliminary consultation regarding proposed activities for consultation to be managed under the Coastal Waters OEMP and Barossa Production Operations
  - EP. [Con-3971] The email advised that Santos was seeking information to better understand:
    - o if you are from a government Department or agency, how the proposed Operations activities may be relevant to your department or agency; and
  - o what (if any) functions, interests or activities you or your organisation have that may be affected by the proposed Operations activities.
- The email included information on the regulatory process for the activities in Commonwealth and NT jurisdictions and links to a Santos information booklet on the proposed activities and a NOPSEMA brochure on the consultation process and details of how to contact
  Santos to register as a Relevant Person. The email stated that the consultation phase would commence on 11 March 2024 and close on 9 April 2024.
- On 12 March 2024 Santos emailed the NLC Kenbi Rangers representative further to advise that it had commenced the consultation phase which would run until 9 April 2024. In addition to the previous information again being provided, Santos provided information on Relevant Persons' entitlements under the regulatory processes, details of how to provide feedback and a reminder of the closing date for consultation. [Con-3975] Santos sought a meeting with Kenbi Rangers to discuss the information provided. [Con-3976]
- On 2 May 2024 Santos emailed the NLC Kenbi Rangers' representative further to previous correspondence, to advise it had extended the consultation period until 16 May 2024. In providing this extension of time, Santos advised that the information in the booklet and factsheet had been updated to account for an additional risk associated with the proposed activity. Santos confirmed that consultation will close on the revised date unless Santos hears otherwise from the NLC Kenbi Rangers' representative. [Con-3977]
- On 10 July 2024 Santos emailed NLC to advise the consultation period had been completed. Santos advised it considered that consultation had now closed for the purpose of Santos finalising and submitting environment plans for these activities to government regulators for assessment. Santos advised NLC it would appreciate its support in passing on the information to Council Members/Representatives. [Con-5122]
- On 15 October 2024 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that the EP had been submitted to NOPSEMA for assessment, provided a link to the full EP on the NOPSEMA website and Santos' contact details for any queries. The Update also stated that the OEMP would be submitted to the NT-DITT for assessment. [Con- 5982]
- On 23 January 2025 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that, following a request from NOPSEMA to provide further information, Santos had re-submitted the EP with some amendments in December 2024 for further assessment. The Update also re-stated that the OEMP would be submitted to the NT-DITT for assessment. [Con-6036]
- Notwithstanding the information provided and the steps described above, no comments or input were received on the Coastal Waters OEMP from the NLC Kenbi Rangers' representative.

Lerrekie Development Corporation (LDC)	
No response was received from Kenbi Rangers.	Santos considers it has provided sufficient information and a reasonable period of time for consultation. Santos considers Section 25 consultation requirements to have been met.
Summary of response by Relevant Person	Assessment of merits

### Larrakia Development Corporation (LDC)

## Santos

Santos' Response Statement **OEMP** reference No response required. Not applicable. Santos' Response Statement **OEMP** reference No response required. Not applicable. Santos' Response Statement **OEMP** reference No response required. Not applicable.

- On 13 February 2024 Santos emailed LDC to advise it of preliminary consultation regarding proposed activities for consultation to be managed under the Coastal Waters OEMP and Barossa Production Operations EP. [Con-4080]
  - The email advised that Santos was seeking information to better understand:
    - o if you are from a government Department or agency, how the proposed Operations activities may be relevant to your department or agency; and
    - what (if any) functions, interests or activities you or your organisation have that may be affected by the proposed Production Operations activities.
- The email included information on the regulatory process for the activities in Commonwealth and NT jurisdictions and links to a Santos information booklet on the proposed activities and a NOPSEMA brochure on the consultation process and details of how to contact Santos to register as a Relevant Person. The email stated that the consultation phase would commence on 11 March 2024 and close on 9 April 2024.
- On 14 February an LDC representative responded to Santos via email advising of staffing changes and to provide appropriate contact details. [Con-4081]
- On 27 March 2024 Santos emailed LDC details of the Coastal Waters OEMP and Barossa Production Operations EP consultation sessions being held for Larrakia People in Darwin on 23 April and asked for the organisation to share the information with its Larrakia networks and post on notice boards and relevant social media channels. [Con- 6021]
- On 2 April 2024 Santos emailed LDC by way of reminder that the consultation is closing on April 9 and asks to be contacted as soon as possible if it has any feedback. [Con-4079]
- On 2 April 2024 LDC emailed Santos stating it would provide a written response to Santos by COB that week. [Con-4083]. Santos acknowledged the LDC email the same day. [Con-4084] The response was not provided by LDC by the date it had nominated.
- On 2 May 2024 Santos emailed LDC further to previous correspondence, to advise it had extended the consultation period until 16 May 2024. In providing this extension of time, Santos advised that the information in the booklet and factsheet had been updated to account for an additional risk associated with the proposed activity. Santos confirmed that consultation will close on the revised date unless Santos hears otherwise from LDC. [Con-4111]
- On 3 May 2024 LDC emailed Santos to advise it would be providing a formal response the following week. [Con-4112]
- On 24 May 2024 LDC emailed Santos and advised that Larrakia people want to have input into the Production Operations activities. LDC seeks to participate in genuine consultation with meetings with Larrakia representatives and access to independent advice about the nature of the proposal. LDC is working to arrange both of these in close partnership with Larrakia Nation Aboriginal Corporation and Gwalwa Daraniki Association and requests a reasonable opportunity to engage in consultations. LDC also requested financial assistance to obtain independent advice and to arrange meetings. [Con-4216]
- On 24 May 2024 Santos emailed LDC to confirm receipt of its email and advised an official response would follow. [Con-4217]
- On 11 June 2024 Santos emailed LDC inviting it to meet on 17 or 18 June 2024. Santos also advised that we were planning to proceed with the next round of Larrakia consultation sessions for the OEMP and EP (which all Larrakia are invited to attend) which are scheduled for Wednesday 12 June at Malak Community Centre. [Con-5007]
- On 12 June 2024 LDC respond to Santos via email stating it could meet on 20 or 21 June 2024. In the email LDC said Larrakia People were not currently resourced to respond to Santos' consultation approach and the consultation meetings Santos had scheduled were not adequate consultation for Larrakia People and should not be relied on as sufficient to meet the requirements for consultation under NOPSEMA guidelines or as discussed in the Tipakalippa court decision. [Con-5008].
- On 20 June 2024 LDC attended a meeting with Santos at which the main topic of discussion was LDC's concerns over engagement with Larrakia people by proponents of all large-scale development in Darwin. [Con-5056]
- At the meeting LDC stated the following:
  - Santos' consultation efforts (for the OEMP and EP) are not being criticised, but LDC is offering to set up a process that can be used from the outset of consultation that will also meet the expectations of the government regulator
  - the NT Government's proposed Middle Arm industrial development has prompted Larrakia organisations to work with the Government on a framework agreement for engagement with Larrakia people 0
  - project proponents have significant resources to spend on consultation but Larrakia organisations are not resourced enough to be able to seek independent advice on proposals and respond accordingly. 0
  - current capacity for LDC to respond is limited and, LDC's priority is Middle Arm.  $\circ$
  - the framework will aim to help redress this imbalance and put Larrakia people on an equal footing with proponents, and ensure they are informed and respected. The framework will set out how Larrakia people will be involved and set out costs 0
  - the framework is under consideration and will be presented to Larrakia families once ready 0
  - regarding the OEMP, it is a complex project and Larrakia have not had access to independent advice because they have not got the resources 0
  - asked whether Santos have been taking advice from Top End Alliance. Santos responded that their scope excludes Larrakia 0
- At the meeting Santos stated the following:
  - Santos has made a range of efforts to consult with Larrakia people on the Coastal Waters OEMP., both through their representative organisations and directly with Larrakia people 0
  - Santos welcomes any input on this OEMP and the EP, which will be submitted to the government regulator for assessment, another meeting can be set up if required. 0
  - Santos is committed to having a long-term relationship with Larrakia people and is happy to work with LDC and other organisations to achieve this 0
  - Santos is supportive in-principle of being involved in such a framework agreement and hearing more about it.
  - regarding this OEMP and the EP, Santos has tried to produce material that is understandable and has taken advice from other First Nations groups about how to present it. Santos wants to hear about possible improvements.
- On 10 July 2024 Santos emailed LDC to advise the consultation period had been completed. Santos advised it consultation had now closed for the purpose of Santos finalising and submitting environment plans for these activities to government regulators for assessment. [Con-5181]
- On 11 July 2024 LDC emailed Santos stating it did not agree that the consultation process was concluded and reiterated its position that proper consultation required the coordinated input of all Larrakia families, the provision of independent expert advice to Larrakia and the resourcing to undertake this entire process. In its email LDC claimed Santos' response did not reflect a genuine commitment to engagement over the life of the project. [Con-5116]
- On 7 August 2024 Santos emailed LDC and clarified that Santos is open to working with LDC to discuss a consultation framework that would help to facilitate future consultation with the Larrakia families, and that it looks forward to receiving a draft framework agreement from LDC once it had concluded consultation with Larrakia families. [Con-5284]
- On 15 October 2024 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that the EP had been submitted to NOPSEMA for assessment, provided a link to the full EP on the NOPSEMA website and Santos' contact details for any queries. The Update also stated that the OEMP would be submitted to the NT-DITT for assessment. [Con- 5982]
- On 25 November 2024 LDC advertised a briefing session which would be run together with LNAC and GDA to present the framework to Larrakia families on 25 November 2024, for the purposes of the NT Government consultation [Con-6017]
- On 23 January 2025 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that, following a request from NOPSEMA to provide further information, Santos had re-submitted the EP with some amendments in December 2024 for further assessment. The Update also re-stated that the OEMP would be submitted to the NT-DITT for assessment. [Con-6036]
- No further correspondence or feedback was received from LDC. A draft framework agreement had not been received from LDC at the time of OEMP submission.
- Additional to formal consultation on the Production Operations EP and OEMP, Santos has continued to make a range of efforts to engage on an ongoing basis with NT First Nations People, both directly and through their representative organisations. See Section 8.4.9 Post-acceptance Consultation Implementation Strategy. No further correspondence or feedback was received from LDC.

Summary of response by Relevant Person	Assessment of merits	Santos' Response Statement	EP reference
An LDC representative stated that consultation on large-scale development in Darwin requires meetings with Larrakia representatives.	Santos has considered the matters raised in the OEMP and provided a response. For the Coastal Waters OEMP, Santos adopted a multi-faceted approach to providing input and feedback opportunities for Larrakia people, including through representative organisations with a dedicated Larrakia focus, other First Nations organisations, Santos-coordinated Larrakia consultation sessions and provision of consultation materials with personnel at Santos' Darwin shopfront available to answer queries.	Santos has consulted with Larrakia people and Larrakia organisations.	Section 4.6.5.1.3.1
	organisation LNAC (see separate entry this section).		



	Santos considers it has provided sufficient information and a reasonable period of time for consultation with LDC.		
	Santos considers Section 25 consultation requirements to have been met.		
An LDC representative stated that LDC required additional time for consultation and communicated a proposal for a consultation framework (which LDC want to undertake in partnership with Larrakia Nation Aboriginal Corporation and Gwalwa Daraniki Association).	Santos has considered the matters raised in the OEMP and provided a response. Santos understands the LDC request for additional time for consultation, which was tied to a proposal for a consultation framework.	Santos provided opportunities for LDC to participate in consultation and provide input regarding these activities, the	Not applicable.
	<ul> <li>Santos' consultation methodology considers and accommodates preferences expressed by Relevant Persons regarding design of the consultation process, where reasonably practicable and appropriate.</li> <li>Consistent with Santos' consultation methodology, Santos has considered this request but this request has not been accommodated for the purposes of s 25 consultation for the Coastal Waters OEMP because it was not reasonably practicable and appropriate in the circumstances.</li> <li>In particular, it was not reasonably practicable to accommodate this request because:</li> <li>Santos has invited further information about the proposed framework and a copy of the framework, but did not receive any further information which it could review and consider during the consultation period for the Coastal Waters OEMP</li> <li>Santos understands that, as of the present date, the proposed framework still does not have endorsement by Larrakia family groups. Further, the purported framework relates to the facilitation of consultation with Larrakia people in circumstances where: <ul> <li>The LDC does not currently have representative authority to speak broadly to the consultation needs of Larrakia People, noting its role as a commercial organisation that aims to support positive economic outcomes for Larrakia People, noting its role as a commercial organisation that aims to support positive economic outcomes for Larrakia People, noting its role as a commercial organisation that aims to support positive economic outcomes for Larrakia People, noting its role as a commercial organisation that aims to support positive and appropriate manner, including (but not limited to) implementing advice from LNAC who represents the Traditional Owners of the Darwin region and speaks on behalf of Larrakia People.</li> <li>During Santos' consultation with Larrakia people and Larrakia representative bodies, Santos was not advised that the Larrakia People want LDC to represent them for the purpose of consultation for the Coastal Waters OE</li></ul></li></ul>	environment that may be affected by the proposed activities, and the environmental impacts and risks associated with the proposed activities. Santos will continue to engage with LDC on the consultation framework, and if prepared and supported by all Larrakia family groups, will have regard to it in relation to ongoing consultation.	
	Santos has provided the LDC with sufficient information and a reasonable period to assess any possible consequences of the Activity in the Coastal Waters OEMP on the LDC's functions, interests and activities, however LDC has not confirmed any consequences for its function, interests and activities. Notwithstanding this, Santos consulted with LDC in its own right, in its capacity as a commercial organisation that supports positive economic outcomes for Larrakia people. Santos included LDC in preliminary and formal consultation commencing in February 2024. LDC was also provided consultation opportunities beyond the formal consultation closing date of 16 May 2024, including in a meeting on 20 June 2024 and follow up email on 7 August 2024. On this basis, Santos considers that it has provided sufficient information and a reasonable period for LDC to participate in the consultation process for the Coastal Waters OEMP. Santos will continue to engage with LDC regarding the Barossa Gas Project. In the event that it delivers a consultation over the life of the Barossa project.		
An LDC representative stated that LDC's preferred mechanism for government and industry proponents to consult with Larrakia people was for Santos to enter into a consultation framework that would support future engagement. The representative noted that the framework was under consideration and would be presented to Larrakia families once ready. The representative stated that LDC was working on these matters in close partnership with Larrakia Nation Aboriginal Corporation and Gwalwa Daraniki Association.	<ul> <li>Santos has considered the matters raised in the OEMP and provided a response.</li> <li>Santos' consultation methodology considers and accommodates preferences expressed by Relevant Persons regarding design of the consultation process, where reasonably practicable and appropriate.</li> <li>Consistent with Santos' consultation methodology, Santos has considered this request but this request has not been accommodated for the purposes of s 25 consultation for the Coastal Waters OEMP because it was not reasonably practicable and appropriate in the circumstances.</li> <li>In particular, it was not reasonably practicable to accommodate this request because: <ul> <li>The request lacked particularity. Santos invited further information and a copy of the consultation framework, but did not receive any further information which it could review and consider during the consultation period for the Coastal Waters OEMP.</li> <li>Santos understands that, as of the present date, the proposed framework still does not have endorsement by Larrakia family groups.</li> </ul> </li> <li>Further, it was not appropriate for Santos to accommodate as it relates to the facilitation of consultation with Larrakia People in circumstances where: <ul> <li>The LDC does not currently have representative authority to speak broadly to the consultation needs of Larrakia People, noting its role as a commercial organisation that aims to support positive economic outcomes for Larrakia People, and that it is governed by a board of independent directors.</li> <li>Santos has consulted with Larrakia People in a culturally sensitive and appropriate manner, including (but not limited to) implementing advice from LNAC who represents the Traditional Owners of the Darwin region and speaks on behalf of</li> </ul> </li> </ul>	Santos understood that LDC would provide Santos a draft consultation framework once LDC had completed their engagement with all Larrakia families. Santos will continue to engage with LDC on the consultation framework, and if prepared and supported by all Larrakia family groups, will have regard to it in relation to ongoing consultation.	Not applicable.



	<ul> <li>Larrakia people, noting its role as a commercial organisation that supports positive economic outcomes for Larrakia People.</li> <li>During Santos' consultation with Larrakia people and Larrakia representative bodies, Santos was not advised that the Larrakia people want LDC to represent them for the purpose of consultation for the Coastal Waters OEMP.</li> <li>Santos will continue to engage with LDC regarding the Barossa Gas Project. In the event that it delivers a consultation framework that is supported by all Larrakia family groups, Santos will then have regard to it in relation to ongoing consultation over the little Barossa project.</li> </ul>
An LDC representative stated that they required financial assistance in order to obtain independent advice and arrange meetings with Larrakia families, as the current capacity for LDC to respond is limited and its priority is Middle Arm.	<ul> <li>Santos has considered the matters raised in the OEMP and provided a response.</li> <li>Santos notes the LDC's request for financial assistance in order to obtain independent advice for Larrakia People and arrangy meetings with all Larrakia families. Santos understands this request was tied to a broader proposal for a consultation framework.</li> <li>Consistent with Santos' consultation methodology, Santos has considered this request but this request has not been accommodated for the purposes of s 25 consultation for the Coastal Waters OEMP. because it was not reasonably practicable and appropriate in the circumstances.</li> <li>In particular, it was not reasonably practicable to accommodate this request because:</li> <li>The request lacked particularity. Santos understands the request was linked to a broader request for Santos to enter a proposed consultation framework about which Santos invited further information and a copy of the framework, but not receive any further information which it could review and consider during the consultation period for the Coastal Waters OEMP.</li> <li>Santos understands that, as of the present date, the proposed framework still does not have endorsement by Larrak family groups.</li> <li>Further, the request was not appropriate for Santos to accommodate as it relates to the facilitation of consultation wit Larrakia People in circumstances where:     <ul> <li>The LDC does not currently have representative authority to speak broadly to the consultation needs of Lar People, noting its role as a commercial organisation that aims to support positive economic outcomes for Larrakia People, and that it is governed by a board of independent directors.</li> <li>Santos has consulted with Larrakia people in a culturally sensitive and appropriate manner, including (but not limited to) implementing advice from LNAC who represents the Traditional Owners of the Darwin region and speaks on behalf of Larrakia people.</li> <li>During Santos' consultation with Larrakia people and Larrakia</li></ul></li></ul>
	Santos will continue to engage with LDC regarding the Barossa Gas Project. In the event that it delivers a consultation frame that is supported by all Larrakia family groups, Santos will then have regard to it in relation to ongoing consultation over the lift the Barossa project. In the event that LDC delivers a framework (as described above), Santos will then consider appropriate financial recompense meeting attendance and input, and/or access to independent advice that the framework may propose.
North Australia Indigenous Land and Sea Management Alliance (NAILSM)	A)

•	On 9 February 2024 Santos ema	iled NAILSMA to advise it of prelimina	ry consultation regarding proposed ac	tivities for consultation to be managed	d under the Coastal Waters OEMP and Bar	ossa Production Ope
---	-------------------------------	----------------------------------------	---------------------------------------	-----------------------------------------	-----------------------------------------	---------------------

- The email advised that Santos was seeking information to better understand:
  - o if you are from a government Department or agency, how the proposed Operations activities may be relevant to your department or agency; and
  - what (if any) functions, interests or activities you or your organisation have that may be affected by the proposed Operations activities.
- The email included information on the regulatory process for the activities in Commonwealth and NT jurisdictions and links to a Santos information booklet on the proposed activities and a NOPSEMA brochure or Santos to register as a Relevant Person. The email stated that the consultation phase would commence on 11 March 2024 and close on 9 April 2024.
- On 13 March 2024 Santos emailed NAILSMA further to the previous correspondence, to advise that it had commenced the consultation phase which would run until 9 April 2024. In addition to the previous inform Relevant Persons' entitlements under the regulatory processes, details of how to provide feedback and a reminder of the closing date for consultation. [Con-3794]
- On 4 April 2024 Santos phoned NAILSMA and left a message with phone contact details.
- On 2 May 2024 Santos emailed NAILSMA to follow-up on previous emails and advised of an additional risk associated with the proposed activity and provided a link to the updated booklet and factsheet. Santos NAILSMA input by 16 May 2024. [Con-4101]
- On 10 July 2024 Santos emailed NAILSMA to advise the consultation period had been completed. Santos advised it considered that consultation had now closed for the purpose of Santos finalising and submittin regulators for assessment. [Con-5130]
- On 15 October 2024 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. 1 NOPSEMA for assessment, provided a link to the full EP on the NOPSEMA website and Santos' contact details for any queries. The Update also stated that the OEMP would be submitted to the NT-DITT for asse
- On 23 January 2025 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. to provide further information, Santos had re-submitted the EP with some amendments in December 2024 for further assessment. The Update also re-stated that the OEMP would be submitted to the NT-DITT for
- Notwithstanding the information provided and the steps described above, no comments or input were received on the Coastal Waters OEMP from NAILSMA.

Assessment of merits Summary of response by Relevant Person



e vork,				
e ork.	Santos understood that LDC, would provide Santos a draft framework once LDC had completed their engagement with all Larrakia families.	Not applicable.		
e	Santos will continue to engage with LDC on the consultation framework and, if prepared and supported by all Larrakia family groups, will have regard to it in			
did	relation to ongoing consultation.			
а				
h				
akia				
ot I				
d				
vork, e of				
for				
rations EP. [Con-3787]				
the consultation process and details of how to contact				
ation again being provided, Santos provided information on				
advised it had extended the consultation period and requested				
g environment plans for these activities to government				
The Up essme The Up	date advised that the EP had beer ent. [Con- 5982] odate advised that, following a requ	n submitted to		
or asse	ssment. [Con-6036]			
	Santos' Response Statement	OFMP reference		

**OEMP** reference

### No response was received from NAILSMA.

Santos considers it has provided sufficient information and a reasonable period of time for consultation. Santos considers Section 25 consultation requirements to have been met.

### Tiwi Islands Clan Groups and Individuals

- Santos continued its staged approach to consultation with Tiwi Islands clan groups and individuals.
- Consultation activities were conducted in person at three locations on the Tiwi Islands, primarily through discussions or presentations.
- The sessions were advertised in advance in accordance with a process agreed with the Clan Groups.
- Some elected members of the TLC were often in attendance at the consultation sessions with their respective Clan Groups.
- At the sessions Santos used visual aids, maps, videos, animations to present information regarding the Activity and the project more generally.
- The presentation also covered the regulatory consultation processes and privacy provisions and provided an overview of Santos the company and the Barossa Project overall.
- The Production Operation information booklet, Santos Privacy Statement and NOPSEMA consultation brochure were provided at the consultation session.
- The following consultations sessions were held on the Tiwi Islands:
- On 5 March 2024 with the Marrikawuyanga, Yimpinari and Wulirankuwu Clans at Milikapiti (Social and Sports Club) [Con-4160]
- On 6 March 2024 with the Mantiyupwi Clan at Wurrumiyanga (Mantiyupwi Motel). [Con-4161]
- On 6 March 2024 with the Jikilaruwu Clan at Wurrumiyanga (Mantiyupwi Motel). [Con-4162]
- On 7 March 2024 with the Wurankuwu Clan at Wurrumiyanga (Mantiyupwi Motel). [Con-4163] -
- On 7 March 2024 with the Malawu Clan at Wurrumiyanga (Mantiyupwi Motel) [Con-4164]
- On 8 April 2024 with the Munupi Clan at Pirlangimpi (Sports and Social Club). [Con-4093] -
- On 9 April 2024 with the Marrikawuyanga and Yimpinari Clans at Milikapiti (Social and Sports Club). [Con-4095]
- On 10 April 2024 with the Jikilaruwu Clan at Wurrumiyanga (Mantiyupwi Motel). [Con-4097]
- On 10 April 2024 with the Mantiyupwi Clan at Wurrumiyanga (Mantiyupwi Motel). [Con-4096]
- -On 13 May 2024 with the Marrikawuyanga, Yimpinari and Wulirankuwu Clans at Milikapiti (Social and Sports Club). [Con-4255]
- On 15 May 2024 with the Jikilaruwu Clan at Wurrumiyanga (Mantiyupwi Motel). [Con-4256]
- On 15 May 2024 with the Mantiyupwi at Wurrumiyanga (Mantiyupwi Motel). [Con-4257]
- On 16 May 2024 with the Wurankuwu and Malawu Clans at Wurrumiyanga (Mantiyupwi Motel). [Con-4258]
- On 17 May 2024 with the Munupi Clan at Pirlangimpi (Sports and Social Club). [Con-4231]
- On 21 May 2024 with the Munupi Clan at Pirlangimpi (Sports and Social Club) [Con-4259]
- On 22 May 2024 with Wurankuwu and Malawu Clans at Wurrumiyanga (Mantiyupwi Motel). [Con-4260]
- Santos also held three consultation sessions with each Tiwi Clan over an extended consultation period to ensure that information regarding the project was understood, that Clan members had ample opportunity to raise questions or concerns, and to ensure that technical information was explained clearly to relevant Clan members.
- In addition to the sessions held on the Tiwi Islands, sessions were also held in Darwin on 22 March 2024 and 8 April 2024 for any Darwin-based Tiwi People. [Con-4844], [Con-4166]
- The following information related to this OEMP. and was presented and discussed at each Tiwi consultation session:
- The Commonwealth Government and NT Government regulations and approvals required
- The activities covered by this OEMP
- The environmental impacts and risks involved with the planned activities and planned controls to management those risks -
- The EMBA in the event of an unplanned event, the risks and planned controls to management those risks -
- The majority of the engagement with attendees involved discussion to increase their understanding of the proposed activities and the associated risks and impacts, without any objections or claims about the adverse impact of each activity to which the Coastal Waters OEMP. relates. Refer below "Summary of response by Relevant Persons" below for further detail of general topics/themes discussed.
- A number of the questions raised at these sessions related to activities covered by other Barossa EPs (being the DPD, D&C, SURF and GEP EPs) that had also been raised and discussed at previous consultation sessions.
- Some requests for further information during Tiwi consultation sessions were answered during the meeting. This includes requests for pictures of the Barossa condensate, which were requested during a Jikilaruwu Clan meeting and were provided to clan members on an iPad at a later stage in the meeting. [Con-4097]. Additionally, with respect to requests for the development of a cultural protocol for consultation which were raised during a consultation session with the Munupi Clan, Santos responded by committing to working better with Tiwi Clans and highlighting approaches Santos already takes to consultation with the Tiwi Clans, including its extended consultation timeframes. [Con-4093]. Where a question could not be answered fully at one session further response and information was provided at the next session. This includes with respect to requests for information on relevant environmental studies during a Wurankuwu Clan and Malawu Clan meeting on 7 March 2024 [Con-4163]. During further meetings, including on 16 May 2024 at a Wurankuwu and Malawu Clans meeting [Con-4258], a Munupi Clan meeting on 17 May 2024 [Con-4231], a Jilikaruwu Clan meeting on 15 May 2024 [Con-4256] and a Wulirankuwu Clan meeting on 13 May 2024] [Con-4255], Santos addressed previous requests for further information on environmental studies relevant to the project by providing attendees with access to relevant studies on an iPad.
- Some requests from individual Tiwi Clan members for more detailed information were followed-up via call or email with them directly. These are summarised below:
  - On 30 April 2024 Santos responded via email to questions on carbon capture and storage and provided links to further information. [Con-4099] 0
    - On 1 May 2024 the individual emailed Santos thanking it for the information provided on 30 April 2024. [Con-4100] 0
    - On 3 May 2024 Santos responded via email to questions on the compensation process in the event of an oil spill. [Con-4114]
    - On 3 May 2024 Santos responded via email to questions on material safety data sheets and the distance from the project activities to a reef formation. [Con-4113] Response to this information was received via email the same day. In their email the 0 individual explained why they were seeking the data sheets and requested images of the reef formation. [Con-4115]
    - On 13 May 2024 Santos emailed a further response to the individual, providing links to images of the reef formation and stating it would be happy to meet with the individual at the next consultation session. [Con-4119] 0
    - On 6 May 2024 Santos responded via email to a question on consultation sessions and request to be advised of any further sessions. [Con-5020] 0
    - On 15 May 2024 Santos emailed a link to a previous Barossa EP (SURF) in response to a request. [Con-5027] 0
    - On 3 July 2024 Santos responded via phone to a question on its permission to conduct clan meetings on the Tiwi Islands and to follow-up on Santos' consultation. In the discussion, with a Senior Elder of the Munupi Clan, no concerns regarding the Munupi 0 Clan consultation sessions were raised. [Con-5026]
  - On 5 July Santos responded by email to a question on carbon credits and offsets [Con-5021]
  - Questions on how to access Santos' employment and education opportunities and seek funding from Santos for community projects are managed outside the OEMP consultation process.
- On 10 July 2024 Santos emailed the Tiwi Land Council to advise the consultation period had been completed. Santos advised it consultation had now closed for the purpose of Santos finalising and submitting environment plans for these activities to government regulators for assessment. Santos advised TLC it would appreciate its support in passing on the information to the Land Council (Trustee and Directors) for them to share with their clans. [Con-5120]

- On 15 October 2024 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that the EP had been submitted to NOPSEMA for assessment, provided a link to the full EP on the NOPSEMA website and Santos' contact details for any queries. The Update also stated that the OEMP would be submitted to the NT-DITT for assessment. [Con- 5982] •
- On 23 January 2025 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that, following a request from NOPSEMA • to provide further information, Santos had re-submitted the EP with some amendments in December 2024 for further assessment. The Update also re-stated that the OEMP would be submitted to the NT-DITT for assessment. [Con-6036]
- No further correspondence or feedback on the Coastal Waters OEMP was received from Tiwi clans
- ٠ Additional to formal consultation on the Coastal Waters OEMP and Barossa Production Operations EP, Santos has continued to make a range of efforts to engage on an ongoing basis with NT First Nations People, both directly and through their representative

organisations. See Section 8.4.9 Post-acceptance Consultation Implementation Str	ategy.		
Summary of response by Relevant Person	Assessment of merits	Santos' Response Statement	OEMP reference
<ul> <li>Following discussion on consultation material, there were questions and comments received in relation to the following matters:</li> <li>Activity <ul> <li>Properties of Barossa condensate</li> <li>Inspection and maintenance of facilities</li> <li>Consultation and Communication</li> <li>Consultation process with Tiwi Clans</li> </ul> </li> <li>Existing Environment <ul> <li>Survey of underwater cultural heritage</li> <li>Whale migratory paths</li> </ul> </li> </ul>	Questions related to Production Operations activities in the Barossa Field are outside the scope of the Coastal Waters OEMP and are addressed in the Barossa Production Operations EP.         The answers provided at the meeting related to each activity to which this OEMP relates were based on the following sections of the OEMP:         The activity described in Section 2, including for the design parameters.         Consultation and communication described in Section 4 and required notifications in Section Sections 8.4.7 and 8.4.9.3.         The existing environment described in Section 3.         The environment risks & impacts as well as the management controls for these described in Sections 6 and 7. Specific risks and impacts pertinent to the matters raised include:         Atmospheric and greenhouse gas emissions – Section 6.3	Noting that no objection or claims were raised about the adverse impact of each activity to which the Coastal Waters OEMP relates, no further response required.	Not applicable.
<ul> <li>Environment impacts &amp; risks</li> <li>Oil spill management.</li> <li>Timeframe for cleaning up an oil spill</li> <li>Compensation process in the event of oil spill impact</li> <li>Produce water discharge</li> <li>Management of lighting impact on marine mammals such as turtles</li> <li>Biosecurity management</li> <li>Safe use of chemicals</li> </ul>	<ul> <li>Interactions with other marine users – Section 6.5</li> <li>Noise emissions – Section 6.1</li> <li>Light emissions – Section 6.2</li> <li>Seabed disturbance - Section 6.4</li> <li>Waste management – Section 6.6</li> <li>Biosecurity management – Section 7.2</li> <li>Marine Fauna interactions – Section 7.3</li> <li>Potential impacts to marine life within a number of subsections in Section 6 and 7, and also in Section 7.3</li> </ul>		
<ul> <li>GHG emissions management &amp; Purchase of carbon credits as offsets</li> <li>Air quality</li> <li>Water quality</li> <li>Noise emissions from the FPSO</li> <li>Use of helicopters and associated noise.</li> <li>Other</li> <li>Commonwealth Government regulatory process</li> <li>Carbon Capture and Storage</li> <li>Santos answered those questions and there were no responses raised for consideration, other than those noted below.</li> </ul>	Loss of hydrocarbon management including gas release and spills in Sections 7.4, 7.5 and 7.6. With required notifications in Sections 8.4.7 and 8.4.9.3 Barossa will meet its Safeguard Mechanism compliance obligations, set by the CER in accordance with Australian Government policy and emissions reduction targets, through the purchase and/or generation of ACCUs or SMCs Santos considers it has provided sufficient information and a reasonable period of time for consultation. Santos considers Section 25 consultation requirements to have been met.		
A clan member asked where further information can be read about Santos' plans regarding CCS.	Santos has considered the matters raised and provided a response. Santos notes the question from the clan member and has provided separately a response via email.	A response has been provided separate to the Coastal Waters OEMP.	Not applicable.
A clan member asked if CCS worked.	Santos has considered the matters raised and provided a response. Santos notes the question from the clan member and has provided separately a response via email.	A response has been provided separate to the Coastal Waters OEMP.	Not applicable.
A clan member asked when Santos will use CCS for Barossa.	Santos has considered the matters raised and provided a response. CCS is not part of the Coastal Waters OEMP and separate approval will be sought. Santos notes the question from the clan member and has provided separately a response via email.	A response has been provided separate to the Coastal Waters OEMP	Not applicable.
A clan member asked how far Barossa activities are located from the largest formation 75m Pavona clavus?	The formation is located 60km west of the Barossa field and OA1 under the Barossa Production Operations EP and is relevant only to activities covered under the Barossa Production Operations EP. Santos notes the question from the clan member and has provided separately a response via email.	A response has been provided separate to the Coastal Waters OEMP	Not applicable.
A clan member asked if an oil spill were to occur, what compensation would the affected Tiwi Island people be entitled to.	Santos notes the question from the clan member and provided a response during the consultation session and further via email.	Santos confirmed via email that: Any compensation would depend on specific circumstances of the incident. As with any claim, assessment	Not applicable.

Several attendees at Tiwi sessions asked about availability of oil spill response training.	Santos has considered the matters raised in the OEMP and provided a response.
One attendee suggested a register should be developed of Tiwi people who undertake the training.	Santos provided information on the availability of oil spill response training.
An attendee at one session asked where oil spill response equipment is stored and whether Port Melville was an option?	Santos has considered the matters raised in the OEMP and provided a response.

and determination would be evidence-based.	
Santos and its Barossa joint venture partners are required to demonstrate a minimum level of financial assurance to be able to cover costs when responding to a spill event. The offshore regulator, NOPSEMA, will not accept the Production Operations Environment Plan without Santos first demonstrating a minimum level of financial assurance for a spill response.	
Santos relies on a combination of its own financial resources and insurance to meet its financial assurance requirements, including third party liability insurance for its activities.	
For each Oil Pollution Emergency Plan there is a comprehensive scientific monitoring program to measure impacts to the physical / biological environment and socio-economic receptors. The results of monitoring inform the extent of impacts.	
Santos confirmed that:	Refer to Other
Spill response arrangements in place that would be activated include mobilisation of people to a spill site.	Measures Section 8.3.3
The first training session with Tiwi Rangers has been held.	
The TLC was nominating additional people to participate in the next round of training.	
Santos confirmed that:	Not applicable
Santos had access to equipment at regional, national and international levels. Stockpiles were located in Australia and around the globe. They are strategically located as they are used not just for oil and gas, but also for shipping companies.	
Santos also had access to equipment stored in Darwin, Geelong and Fremantle. This equipment was owned by companies that provide this access and they decide where it is stored.	
Santos would store three rapid assessment kits on the Tiwi Islands which will be used by rangers to assess spills. The rangers would decide on the	

		best location to store this equipment.	
An attendee at one session asked where Santos will source carbon credits for offsetting its emissions?	Santos has considered the matters raised in the OEMP and provided a response. The control measures Santos will apply to reduce GHG emissions to ALARP and acceptable levels, include the purchase and/or surrender of carbon credits are outlined in Section 6.3.3.	Santos referred them to our latest annual report and comments in that report that outlines further information on our generation and acquisition of carbon credits.	Section 6.3.3
An attendee asked what happens if Santos exceeds the total emissions for the year. As a consequence, would Santos shut down operations.	The answer provided was based on control measure BAO-CM-6.3.11 in Section 6.3. Queries relating to GHG emissions associated with the extraction and processing of Barossa gas (as distinct from GHG emissions specific to the scope of activity under this OEMP) are addressed in the Production Operations EP.	Santos confirmed that the Clean Energy Regulator will set a baseline (Safeguard Mechanism) for Barossa greenhouse gas emissions. Santos will purchase or generate Australian Carbon Credit Units (ACCUs) to offset Barossa's reservoir CO <sub>2</sub> emissions and any emissions above the Safeguard Mechanism baseline.	Section 6.3.2.3.2
		Santos will comply with the Regulator's requirements. Companies which do not comply with the Safeguard Mechanism are subject to significant enforcement penalties from the Clean Energy Regulator.	
An attendee at one session claimed that migrating turtles would be impacted by the activity and asked how Santos will stop turtles migrating to the drill rig location.	The response raised a claim that migrating turtles would be impacted by the activity and asked how Santos would stop turtle migrating to the drill rig location (which is within OA1). Barossa Development drilling activities (i.e. use of a drill rig) are not covered in the Coastal Waters OEMP. Notwithstanding this, Santos recognises that marine turtles, including migrating marine turtles have the potential to be impacted by the activity. Santos has considered the distribution and movements of marine turtles in the vicinity of its operational area and the Tiwi Islands (Section 3.2.13.2.1). A review of turtle movement data indicates that migratory pathways in the vicinity of the Tiwi Islands are largely restricted to the waters inside of the 100 m depth contour (i.e., waters less than 100 m deep) and overlap with the OA (Pendoley, 2023). Although the OA overlaps with Marine Turtle BIAs, Santos considers that the limited frequency and duration of IMMR activities within the OA, combined with the implemented control measures, reduce the impacts and risks to migrating marine turtles to ALARP and acceptable levels Santos does not see merit in stopping marine turtles from migrating and considers this inconsistent with its performance outcomes and control measures which are in place to reduce disturbances to marine turtles.	Santos outlined within the session that OA1 (under the Production Operations EP) is not in or near a biologically important area and it has assessed impacts to marine turtles in its operational areas from light spill using lighting studies.	Santos includes an assessment of migratory marine turtle movements in the vicinity of Tiwi Islands and its operational area in Section 3.2.13.2.1 Santos includes control measures to reduce impacts to marine turtles in the OA to ALARP and acceptable levels in Section 6 and 7
An attendee/s at one session asked questions regarding helicopters flying over Tiwi and Seagull Island, particularly in relation to: number of passengers; clarifying if it was helicopter or plane; purpose of flights.	No helicopter flights over the Tiwi Islands or Seagull Island are planned to occur for activities to which this OEMP relates. Helicopter flights are used to transport workers to and from the IMMR vessels if required. Questions related to Production Operations activities (helicopter flights to the FPSO) in the Barossa Field are outside the scope of the Coastal Waters OEMP and are addressed in the Barossa Production Operations EP.	<ul> <li>Santos responded that:</li> <li>there would be approximately 12 passengers;</li> <li>it would use a helicopter;</li> <li>flights are used to transport workers to and from the FPSO.</li> <li>Santos had also outlined within the sessions that:</li> <li>the use helicopters is required to safely transport our workers to and from the FPSO</li> <li>that helicopters will fly over Tiwi islands on average 3 times per week.</li> <li>planned flight paths are over the eastern end of Melville Island, at its closest point in 22km from Seagull Island</li> </ul>	Not applicable



<ul> <li>An attendee asked where Santos gets its permission to come to the Tiwi Islands and consult with clan group.</li> <li>The attendee stated that the TLC must consult with and have regard of Traditional Owners and: <ul> <li>ensure they understand any proposal;</li> <li>ensure any affected Aboriginal community members, including the whole wider community,</li> <li>ensure they have a chance to say what they think of the proposal and</li> <li>satisfy itself that the Traditional Owners have consented to the proposal.</li> </ul> </li> </ul>	Santos has considered the matters raised in the OEMP and provided a response. The answers provided were based on consultation with Tiwi people and clan groups described in Section 4.7.6.1. Santos notes the statements by the attendee and has responded via a phone call on 3 July 2024 to address statements mad the consultation session, confirming Santos used a culturally appropriate process. In that exchange, no concerns regarding t Munupi Clan consultation sessions were raised. The attendee, who is Senior Elder, thanked Santos for always being "respec and going to the Trustee first, with the Trustee then going to Traditional Owners. This consultation process, agreed with the TLC and clan groups, has been followed for the past three Barossa EPs and inclu consultation in person at specific locations on the Tiwi Islands, primarily through discussions or presentations with all clans.
An attendee asked how Tiwi people could be sure that the FPSO has been cleaned properly as part of biosecurity precautions.	Questions related to Production Operations activities in the Barossa Field are outside the scope of the Coastal Waters OEMF are addressed in the Barossa Production Operations EP.
A clan member advised before sea level rise there was one big land mass and we may find giant serpent bones. They would like to be notified if any bones are found.	Santos has considered the matters raised in the OEMP and provided a response. The answers provided were based on information contained in Section 3.2.15.11 and Section 6.4.2.4.3 Santos engaged Cosmos Archaeology to undertake maritime archaeological heritage assessments. There is no known UCH (including First Nations) within the OA.
A clan member raised a query about water quality and if marine life could be affected from an oil spill, such as turtles and dugongs, which they traditionally hunt. A query was also raised if clams and mussels could be monitored as they are filter feeders and information can be used to determine changes in water quality.	Santos has considered the matters raised in the OEMP and provided a response. The answers were provided based on information contained in Section 7.4 and 7.6.
A clan member raised a query regarding noise emissions from the FPSO affecting turtle hunting.	Santos has considered the matters raised in the OEMP and provided a response. Questions related to Production Operations activities in the Barossa Field are outside the scope of the Coastal Waters OEMP are addressed in the Barossa Production Operations EP.
A clan member raised a query about sea levels, climate and erosion of land particularly from increased shipping movements that may cause accelerated erosion. They stated this erosion in turn can impact turtle nesting beaches and wanted to know how Santos and Tiwi people could work together.	Santos consider the clan member's claim that increased ship movements cause land erosion, and therefore impact turtle nest has no merit. Vessel movements are low during operations. There is no evidence to support the assertion that movements of vessels durin activity would result in erosion of land and impact turtle nesting beaches.

	<ul> <li>to manage any potential noise impacts, the Helicopter will fly between 1.8 km to 2.4km above Tiwi islands.</li> <li>at this height Barossa helicopters are unlikely to be heard any differently than other background air traffic, this is also higher than birds generally fly.</li> </ul>	
	Santos confirmed that:	Not applicable
made at ng the spectful" ncludes ns.	<ul> <li>Santos received the appropriate permits from the TLC.</li> <li>The sessions were notified and advertised in advance in accordance with the usual practice, including via social media.</li> </ul>	
EMP and	No response required.	Not applicable.
ICH	Santos confirmed that no bones have been identified to date and if any bones are identified the clan member would be advised.	Refer to Other Measures Section 8.3.3
	Santos outlined in the session that an Operational and Scientific Monitoring Plan (as part of the GEP NT Waters OPEP) would be implemented in the event of a spill.	Not applicable
EMP and	No response required.	Section 6.1
nesting, luring the	During the session, Santos advised a risk assessment regarding greenhouse gas emissions has been undertaken.	Not applicable
	The Tiwi clan were shown a map of shipping routes in the region (for all shipping and not limited to Santos), based on the Australian Maritime Safety Authority recorded vessel movements.	
	Santos outlined that Activity vessel movements along the pipeline would not be permanent.	
	Santos agreed to talk with the clan member after the session to understand needs, impacts and benefits on working together.	

A clan member requested evidence regarding the environmental studies relevant to the project.	Santos has considered the matters raised in the OEMP. Santos complied with this request.
A clan member requested a photo of Barossa condensate be provided.	Santos has considered the matters raised in the OEMP. Santos complied with this request.
A clan member requested the development of a cultural protocol for consultation, requesting Santos to go through the clan Trustee and work with and respect Tiwi people.	<ul> <li>Santos has considered the matters raised in the OEMP and provided a response.</li> <li>This particular request for a cultural protocol has been considered and assessed in context, having regard to the Santos' broapproach to consultation with Tiwi Clan Groups and Individuals and the development of the existing arrangements. Santos' approach to consultation with Tiwi Clan Groups and Individuals was developed in consultation with Tiwi People on earlier Baters and includes: <ul> <li>obtaining appropriate permissions to be on Tiwi from the Tiwi Land Council,</li> <li>scheduling of consultation sessions in conjunction with Clan Trustees/Traditional Owners, the Tiwi Land Council Recourses and Tiwi Enterprises to ensure no clashes with community events, cultural ceremony or "Sorry Busin providing three consultation sessions per Clan and</li> <li>providing an extended period of consultation for Tiwi Islands Clan Groups and Individuals.</li> </ul> </li> <li>In addition, on 3 July 2024 Santos had a phone discussion with a Senior Elder of the Munupi Clan who made the comments regarding a cultural protocol, to follow-up on Santos' consultation. In that exchange, no concerns regarding the Munupi Clan Consultation Owners. Consistent with Santos' consultation methodology, Santos has considered request but, in the circumstances outlined above, determined it was not appropriate to progress it further (beyond the measu already taken) for the purposes of s 25 consultation for the Coastal Waters OEMP.</li> <li>Santos is committed to continuing to work with Tiwi People to continue to consult with them in a culturally appropriate mannee.</li> </ul>
A clan member raised concerns regarding information provided, stating that some information provided to Tiwi Clan members is technical and needs to be simplified.	Santos has considered the matters raised in the OEMP. Santos notes the statements made by the attendee related to the written material on the power point slides being presented a meeting. The meeting minutes (Con-4259) show that the information on the slides was not being read-out verbatim and an interpreter present to further explain the material in language. The minutes also show there was strong engagement from clan members during the meeting, with clan members asking mai questions, including to clarify technical information, and Santos' representatives providing clear responses. This was the third consultation session held with the Munupi Clan at which the Barossa Production Operations EP was discut the others being on 17 May 2024 (Con-4231) and 8 April 2024 (Con-4093). This consultation sessions were undertaken in line with the consultation process and consultation preferences, which Tiwi Pe have historically expressed during previous consultation on Barossa Project EPs, as outlined in Section 4.7.5.1.3, Consultation with Tiwi Islands Clans and Individuals.



	The answers were provided based on information contained in Sections 2, 3.2.14.5 and 6.3	
	During the session, Santos confirmed it would provide relevant environmental studies and evidence relating to the project environmental impacts and risks.	Environmental studies are referenced throughout the Coastal Waters OEMP.
	At subsequent clan meetings Santos provided further information and had the relevant environmental studies available to review on iPads, and confirmed that they were also available online.	
	During the session, Santos provided an iPad which contained a photo of the condensate.	Not applicable.
ader' ossa	During the session, Santos reiterated that it is committed to working with Tiwi Clans and was consulting with Tiwi Clans for an extended period.	Not applicable.
Tiwi ess",	In addition, on 3 July 2024 Santos had a phone discussion with a Senior Elder of the Munupi Clan who made the comments regarding a cultural protocol, to follow-up on Santos' consultation. In that	
st this es r.	exchange, no concerns regarding the Munupi Clan consultation sessions were raised. The Senior Elder thanked Santos for always being "respectful" and going to the Trustee first with the Trustee then going to Traditional Owners.	
at the was	The Santos representative speaking at the time responded that Santos was trying to share the information with attendees as simply as possible, noting that in the scientific world, some things are not as easy to	Not applicable.
ıy	put things simply. (Con-4259)	
ssed, ople on	thanked the clan member for raising the matter and further stated that Santos attendees would be happy to talk to the clan member after the meeting and to get feedback.	
	On 3 July 2024 Santos followed-up by phone with a Senior Elder of the Munupi Clan. This was the same person who had made the comment on written materials	

1

### First Nations People and groups: Representative organisations – Western Australia

### **Kimberley Land Council (KLC)**

Summary of consultation effort:

- On 22 February 2024 Santos emailed KLC to advise it of preliminary consultation regarding proposed activities for consultation to be managed under the Coastal Waters OEMP and Barossa Production Operations EP. [Con-4037]
- The email advised that Santos was seeking information to better understand:
  - o if you are from a government Department or agency, how the proposed Operations activities may be relevant to your department or agency; and
  - what (if any) functions, interests or activities you or your organisation have that may be affected by the proposed Production Operations activities. 0
- In the email Santos advised it had identified that the KLC - in its capacity as a Representative Body - may have functions, interests or activities that may be affected by these proposed activities and would like to meet with KLC to determine if it wishes to participate in the consultation process. Should KLC be interested Santos can then discuss appropriate consultation methods appropriate to the KLC's information needs and interests.
- The email included information on the regulatory process for the activities in Commonwealth and NT jurisdictions and links to a Santos information booklet on the proposed activities and a NOPSEMA brochure on the consultation process and details of how to contact Santos to register as a Relevant Person. The email stated that the consultation phase would commence on 11 March 2024 and close on 9 April 2024.
- Also included was a map of the Environment that May be Affected (EMBA) and a summary of Santos' understanding of how the EMBA related to Aboriginal heritage sites, native title claims or determinations as well as marine park management. On 2 May 2024 Santos emailed KLC to follow-up on previous emails and advised of an additional risk associated with the proposed activity and provided a link to the updated booklet and factsheet. Santos advised it had extended the consultation period and requested KLC input by 16 May 2024. [Con-4103].
- On 10 July 2024 Santos emailed KLC to advise the consultation period had been completed. Santos advised it consultation had now closed for the purpose of Santos finalising and submitting environment plans for these activities to government regulators for assessment. [Con-5087]
- On 15 October 2024 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that the EP had been submitted to NOPSEMA for assessment, provided a link to the full EP on the NOPSEMA website and Santos' contact details for any queries. The Update also stated that the OEMP would be submitted to the NT-DITT for assessment. [Con- 5982]
- On 14 November 2024 Santos phoned the KLC seeking confirmation that it had no comments on the EP and OEMP. [Con-5987]
- On 23 January 2025 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that, following a request from NOPSEMA to provide further information, Santos had re-submitted the EP with some amendments in December 2024 for further assessment. The Update also re-stated that the OEMP would be submitted to the NT-DITT for assessment. [Con-6036]
- Notwithstanding the information provided and the steps described above, no comments or input were received on the Coastal Waters OEMP from the KLC.

Summary of response by Relevant Person	Assessment of merits	Santos' Response Statement	OEMP reference
No response was received from the KLC.	With regard to the location of the proposed Production Operations activities in NT Coastal Waters, there are no impacts from planned activities that may affect the functions, interests or activities of any First Nations groups, clans and/or organisations along the WA coastline.	No response required.	Not applicable
	The worst-case credible spill scenario modelled for the EMBA for the Coastal Waters OEMP also shows there are no impacts from unplanned activities that may affect the functions, interests or activities of any First Nations groups, clans and/or organisations along the WA coastline.		
	Santos considers it has provided sufficient information and a reasonable period of time for consultation.		
	Santos considers Section 25 consultation requirements to have been met.		
Balanggarra Aboriginal Corporation (BAC)			

- On 22 February 2024 Santos emailed BAC to advise it of preliminary consultation regarding proposed activities for consultation to be managed under the Coastal Waters OEMP and Barossa Production Operations EP. [Con-4060]
- The email advised that Santos was seeking information to better understand:
- if you are from a government Department or agency, how the proposed Operations activities may be relevant to your department or agency; and
- what (if any) functions, interests or activities you or your organisation have that may be affected by the proposed Production Operations activities.
- In the email Santos advised BAC to contact Santos at the earliest opportunity if it considered that it may be a Relevant Person and wished to participate in the consultation process. Should BAC be interested Santos can then discuss appropriate consultation methods appropriate to BAC's information needs and interests.
- The email included information on the regulatory process for the activities in Commonwealth and NT jurisdictions and links to a Santos information booklet on the proposed activities and a NOPSEMA brochure on the consultation process and details of how to contact Santos to register as a Relevant Person. The email stated that the consultation phase would commence on 11 March 2024 and close on 9 April 2024.
- Also included was a map of the Environment that May be Affected (EMBA) and a summary of Santos' understanding of how the EMBA related to Aboriginal heritage sites, native title claims or determinations as well as marine park management.
- On 13 March 2024 Santos emailed BAC further to the previous correspondence, to advise that it had commenced the consultation phase which would run until 9 April 2024. In addition to the previous information again being provided, Santos provided information on Relevant Persons' entitlements under the regulatory processes, details of how to provide feedback and a reminder of the closing date for consultation. In the email Santos also advised it would try to call BAC as well. [Con-4061]



at the 21 May 2024 session.
The Elder did not advise of any
outstanding issues related to
the Munupi consultation
sessions.

- On 14 March 2024 BAC advised Santos via email that the information provided by Santos in the 13 March email had been forwarded to the CEO. [Con-4067] The same day the BAC CEO advised Santos via email that the Chair of the BAC Board would seek instructions from the Board as to how to proceed further. [Con-5014]
- On 2 April 2024 Santos sent a reminder to BAC via email that the consultation period for the EP was closing on April 9 and to contact Santos as soon as possible if BAC had any feedback. The information previously provided was again included. [Con-4085]
- On 2 May 2024 Santos phoned and emailed BAC to follow up on the previous emails and advise it had extended the consultation period until 16 May 2024. In providing this extension of time. Santos advised that the information in the booklet and factsheet had been updated to account for an additional risk associated with the proposed activity. Santos confirmed that consultation will close on the revised date unless Santos hears otherwise from BAC. [Con-4105]
- On 28 May 2024 Santos emailed BAC in relation to its responsibilities under the Australian Marine Parks North Management Plan for sea country in the Joseph Bonaparte Marine Park. The email included a map showing the location of the park in relation to Balanggarra country. [Con-4209].
- On 10 July 2024 Santos emailed BAC to advise the consultation period had been completed. [Con-5127]
- On 15 October 2024 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that the EP had been submitted to NOPSEMA for assessment, provided a link to the full EP on the NOPSEMA website and Santos' contact details for any queries. The Update also stated that the OEMP would be submitted to the NT-DITT for assessment. [Con- 5982]
- On 23 January 2025 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that, following a request from NOPSEMA to provide further information, Santos had re-submitted the EP with some amendments in December 2024 for further assessment. The Update also re-stated that the OEMP would be submitted to the NT-DITT for assessment. [Con-6036]
- No further correspondence or feedback has been received from BAC.

Summary of response by Relevant Person	Assessment of merits	Santos' Response Statement
No response was received from BAC.	With regard to the location of the proposed Production Operations activities in NT Coastal Waters, there are no impacts from planned activities that may affect the functions, interests or activities of any First Nations groups, clans and/or organisations along the WA coastline.	No response required.
	The worst-case credible spill scenario modelled for the EMBA for the Coastal Waters OEMP also shows there are no impacts from unplanned activities that may affect the functions, interests or activities of any First Nations groups, clans and/or organisations along the WA coastline.	
	Santos considers it has provided sufficient information and a reasonable period of time for consultation.	
	Santos considers Section 25 consultation requirements to have been met.	

### Bardi and Jawi Niimidiman Aboriginal Corporation (BJAC)

- On 22 February 2024 Santos emailed BJAC to advise it of preliminary consultation regarding proposed activities for consultation to be managed under the Coastal Waters OEMP and Barossa Production Operations EP. [Con-4038]
- The email advised that Santos was seeking information to better understand:
  - o if you are from a government Department or agency, how the proposed Operations activities may be relevant to your department or agency; and
  - what (if any) functions, interests or activities you or your organisation have that may be affected by the proposed Operations activities.
- In the email Santos advised BJAC to contact Santos at the earliest opportunity if it considered that it may be a Relevant Person and wished to participate in the consultation process. Should Bardi Jawi be interested Santos can then discuss appropriate consultation methods appropriate to BJAC's information needs and interests.
- The email included information on the regulatory process for the activities in Commonwealth and NT jurisdictions and links to a Santos information booklet on the proposed activities and a NOPSEMA brochure on the consultation process and details of how to contact Santos to register as a Relevant Person. The email stated that the consultation phase would commence on 11 March 2024 and close on 9 April 2024.
- Also included was a map of the Environment that May be Affected (EMBA) and a summary of Santos' understanding of how the EMBA related to Aboriginal heritage sites, native title claims or determinations as well as marine park management.
- On 13 March 2024 Santos emailed BAC further to the previous correspondence, to advise that it had commenced the consultation phase which would run until 9 April 2024. In addition to the previous information again being provided. Santos provided information on Relevant Persons' entitlements under the regulatory processes, details of how to provide feedback and a reminder of the closing date for consultation. In the email Santos also advised it would try to call BAC as well.
- On 2 April 2024 Santos sent a reminder to BJAC via email that the consultation period was closing on April 9 and to contact Santos as soon as possible if BJAC had any feedback. The information previously provided was again included. [Con-4039]
- On 15 October 2024 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that the EP had been submitted to NOPSEMA for assessment, provided a link to the full EP on the NOPSEMA website and Santos' contact details for any queries. The Update also stated that the OEMP would be submitted to the NT-DITT for assessment. [Con- 5982]
- On 23 January 2025 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that, following a request from NOPSEMA to provide further information, Santos had re-submitted the EP with some amendments in December 2024 for further assessment. The Update also re-stated that the OEMP would be submitted to the NT-DITT for assessment. [Con-6036]
- Notwithstanding the information provided and the steps described above, no comments or input were received on the Coastal Waters OEMP from BJAC.

Summary of response by Relevant Person	Assessment of merits	Santos' Response Statement
No response was received from BJAC.	Santos considers it has provided sufficient information and a reasonable period of time for BJAC to advise its relevancy to the activities proposed under the Coastal Waters OEMP. With regard to the location of the proposed GEP operations activities in NT Coastal Waters, there are no impacts from planned activities that may affect the functions, interests or activities of any First Nations groups, clans and/or organisations along the WA coastline. The worst-case credible spill scenario modelled for the EMBA for the Coastal	No response required.
	Waters OEMP also shows there are no impacts from unplanned activities that may	



OEMP reference
Not applicable.

OEMP reference
Not applicable.

affect the functions, interests or activities of any First Nations groups, clans and/or organisations along the WA coastline.			
Santos considers Section 25 consultation requirements to have been met.			
Dambimangari Aboriginal Corporation (DAC)			
Summary of consultation effort:			
• On 22 February 2024 Santos emailed DAC to advise it of preliminary consultation regarding proposed activities for consultation to be managed under the Coastal Waters OEMP and Barossa Production Opera			
<ul> <li>The email advised that Santos was seeking information to better understand:</li> </ul>			
<ul> <li>if you are from a government Department or agency, how the proposed Operations activities may be relevant to your department or agency; and</li> <li>what (if any) functions, interacts or activities your experimentian have that may be effected by the proposed Operations, activities</li> </ul>			
<ul> <li>What (II any) functions, interests of activities you of your organisation have that interested by the proposed Operations activities.</li> <li>In the amail Senter advised DAC to contact Senter at the participate operations in the concultation process. Should DAC he interested Set</li> </ul>			
appropriate to DAC's information needs and interests.			
The email included information on the regulatory process for the activities in Commonwealth and NT jurisdictions and links to a Santos information booklet on the proposed activities and a NOPSEMA brochure of the activities activities and a NOPSEMA brochure of the activities activities activities activities activities activities activities activities activities activitities activities activities act			

- tion booklet on the proposed activities and a NOPSEMA brochure on the consultation process and details of how to contact Santos to register as a Relevant Person. The email stated that the consultation phase would commence on 11 March 2024 and close on 9 April 2024.
- Also included was a map of the Environment that May be Affected (EMBA) and a summary of Santos' understanding of how the EMBA related to Aboriginal heritage sites, native title claims or determinations as well as marine park management
- On 12 March 2024 a DAC representative advised Santos via email that the DAC Board would be able to meet with Santos to discuss the information during its meeting to be held on 10/11 April 2024. [Con-4048]
- On 12 March 2024 Santos responded to DAC's offer, stating it would check the dates with the appropriate personnel's availability and revert back as soon as possible. [Con-4054]
- On 13 March 2024 Santos emailed DAC to confirm it would be able to meet at any time on 10/11 April as requested by DAC. The email also advised that Santos had commenced the consultation phase which would run until 9 April 2024. In addition to the previous information again being provided, Santos provided information on Relevant Persons' entitlements under the regulatory processes, details of how to provide feedback and a reminder of the closing date for consultation. [Con-4065]
- On 19 March 2024 Santos re-confirmed its availability for a meeting during DAC's two-day Board meetings and requested details in order to plan attendance. [Con-4071]
- On 19 March 2024 DAC advised Santos via email that it would provide the agenda as soon as possible and also requested that the deadline for feedback be extended to accommodate feedback arising from the meeting. [Con-4072]
- On 19 March 2024 Santos responded to DAC to advise that feedback received at the meeting would be included. [Con-4074]
- On 26 March 2024 DAC emailed Santos the date and time for the meeting and Santos responded via email the same day. [Con-4075]
- On 2 April 2024 Santos sent a reminder to DAC via email that the consultation period was closing on April 9 and to contact Santos as soon as possible if DAC had any feedback. The information previously provided was again included. Santos noted in the email that the 9 April feedback deadline would be adjusted to accommodate the meeting with DAC and provision of its feedback. [Con-5016]
- On 10 April 2024 Santos held a meeting with DAC at the DAC offices in Derby, at which the Productions Operations activities were discussed. During the meeting one question was asked about the composition of Barossa condensate and what a condensate spill would look like. No issues or concerns were raised at the meeting. DAC advised that it would consider the information Santos had provided and get back to Santos if it had any further questions. At the meeting of 10 April 2024 DAC requested a copy of the Santos factsheet providing details of the Barossa Production Operations, and a copy of the information booklet was provided. [Con-4092]
- On 17 April 2024 Santos emailed DAC thanking it for the meeting. [Con-4098]
- On 2 May 2024 Santos emailed DAC to advise it had extended the consultation period until 16 May 2024. In providing this extension of time. Santos advised that the information in the booklet and factsheet had been updated to account for an additional risk associated with the proposed activity. Santos confirmed that consultation will close on the revised date unless Santos hears otherwise from DAC. [Con-5015]
- On 10 July 2024 Santos emailed DAC to advise the consultation period had been completed. Santos advised it consultation had now closed for the purpose of Santos finalising and submitting environment plans for these activities to government regulators for assessment. [Con-5126]
- On 23 January 2025 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that, following a request from NOPSEMA to provide further information, Santos had re-submitted the EP with some amendments in December 2024 for further assessment. The Update also re-stated that the OEMP would be submitted to the NT-DITT for assessment. [Con-6036]
- No further correspondence or feedback was received from DAC.

Summary of response by Relevant Person	Assessment of merits	Santos' Response Statement
An attendee asked about the composition of Barossa condensate and what a condensate spill would look like.	This response does not raise an objection or claim about the adverse impact of each activity to which this OEMP relates. There is no risk of a condensate spill from activities under this OEMP.	No response required.
	With regard to the location of the proposed GEP operations activities in NT Coastal Waters, there are no impacts from planned activities that may affect the functions, interests or activities of any First Nations groups, clans and/or organisations along the WA coastline.	
	The worst-case credible spill scenario modelled for the EMBA for the Coastal Waters OEMP also shows there are no impacts from unplanned activities that may affect the functions, interests or activities of any First Nations groups, clans and/or organisations along the WA coastline	

## Mayala Inninalang Aboriginal Corporation (MIAC)

Summary of consultation effort:

- On 22 February 2024 Santos emailed MIAC to advise it of preliminary consultation regarding proposed activities for consultation to be managed under the Coastal Waters OEMP and Barossa Production Operation
- The email advised that Santos was seeking information to better understand:
  - o if you are from a government Department or agency, how the proposed Operations activities may be relevant to your department or agency; and
    - what (if any) functions, interests or activities you or your organisation have that may be affected by the proposed Operations activities.



ons EP. and the DPD Offshore CEMP. [Con-4040]

intos can then discuss appropriate consultation methods

	OEMP reference
	Not applicable
ons EP, and the DF	PD Offshore CEMP. [Con-4041]

- In the email Santos advised MIAC to contact Santos at the earliest opportunity if it considered that it may be a Relevant Person and wished to participate in the consultation process. Should MAC be interested Santos can then discuss appropriate consultation methods appropriate to MAC's information needs and interests.
- The email included information on the regulatory process for the activities in Commonwealth and NT jurisdictions and links to a Santos information booklet on the proposed activities and a NOPSEMA brochure on the consultation process and details of how to contact Santos to register as a Relevant Person. The email stated that the consultation phase would commence on 11 March 2024 and close on 9 April 2024.
- Also included was a map of the Environment that May be Affected (EMBA) and a summary of Santos' understanding of how the EMBA related to Aboriginal heritage sites, native title claims or determinations as well as marine park management.
- On 14 March 2024 Santos phoned MIAC and, further to the previous correspondence, followed up, later the same day, with an email to advise that it had commenced the consultation phase which would run until 9 April 2024. As reflected in the email, Santos had been instructed on the earlier telephone call to forward the email to another person in the same organisation (forwarding the email was the only matter discussed on that telephone call). [Con- 4068] In addition to the previous information again being provided, Santos provided information on Relevant Persons' entitlements under the regulatory processes, details of how to provide feedback and a reminder of the closing date for consultation. In the email Santos stated it would be happy to speak via phone again or attend a meeting with MIAC. [Con- 4068]
- On 2 April 2024 Santos sent a reminder to MIAC via email that the consultation period was closing on April 9 and to contact Santos as soon as possible if MAC had any feedback. The information previously provided was again included. [Con-4088]
- On 15 October 2024 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that the EP had been submitted to NOPSEMA for assessment, provided a link to the full EP on the NOPSEMA website and Santos' contact details for any queries. The Update also stated that the OEMP would be submitted to the NT-DITT for assessment. [Con- 5982]
- On 23 January 2025 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that, following a request from NOPSEMA to provide further information, Santos had re-submitted the EP with some amendments in December 2024 for further assessment. The Update also re-stated that the OEMP would be submitted to the NT-DITT for assessment. [Con-6036]
- Notwithstanding the information provided and the steps described above, no comments or input were received on the Coastal Waters OEMP from MIAC.

Summary of response by Relevant Person	Assessment of merits	Santos' Response Statement
No response was received from MIAC.	Santos considers it has provided sufficient information and a reasonable period of time for MAC to advise its relevancy to the activities proposed under this OEMP consultation.	No response required.
	With regard to the location of the proposed GEP operations activities in NT Coastal Waters, there are no impacts from planned activities that may affect the functions, interests or activities of any First Nations groups, clans and/or organisations along the WA coastline.	
	The worst-case credible spill scenario modelled for the EMBA for the Coastal Waters OEMP also shows there are no impacts from unplanned activities that may affect the functions, interests or activities of any First Nations groups, clans and/or organisations along the WA coastline.	
	Santos considers Section 25 consultation requirements to have been met.	

### Miriuwung and Gajerrong Aboriginal Corporation (MGAC)

- On 22 February 2024 Santos emailed MGAC to advise it of preliminary consultation regarding proposed activities for consultation to be managed under the Coastal Waters OEMP and Barossa Production Operations EP. [Con-4043]
- The email advised that Santos was seeking information to better understand:
  - o if you are from a government Department or agency, how the proposed Operations activities may be relevant to your department or agency; and
  - what (if any) functions, interests or activities you or your organisation have that may be affected by the proposed Operations activities.
- In the email Santos advised MGAC to contact Santos at the earliest opportunity if it considered that it may be a Relevant Person and wished to participate in the consultation process. Should MGAC be interested Santos can then discuss appropriate consultation methods appropriate to MGAC's information needs and interests.
- The email included information on the regulatory process for the activities in Commonwealth and NT jurisdictions and links to a Santos information booklet on the proposed activities and a NOPSEMA brochure on the consultation process and details of how to contact Santos to register as a Relevant Person. The email stated that the consultation phase would commence on 11 March 2024 and close on 9 April 2024.
- Also included was a map of the Environment that May be Affected (EMBA) and a summary of Santos' understanding of how the EMBA related to Aboriginal heritage sites, native title claims or determinations as well as marine park management.
- On 13 March 2024 Santos followed-up with a phone call to MGAC and then emailed to advise that it had commenced the consultation phase which would run until 9 April 2024. In addition to the previous information again being provided. Santos provided information on Relevant Persons' entitlements under the regulatory processes, details of how to provide feedback and a reminder of the closing date for consultation. [Con-4062]
- On 13 March 2024 MGAC phoned Santos to advise that the Executive Office, the Chair of the Board, and the Board would be in touch with Santos once they have met and considered the issues. [Con- 4066]
- On 14 March 2024 MGAC attempted to phone Santos and followed-up with an email advising that it would call Santos the next day. [Con-4070]
- On 28 March 2024 Santos held a phone call with the Chair of MGAC, where options for meeting were discussed. It was agreed that MGAC was to get back to Santos with an appropriate date for meeting.
- On 28 March 2024, Santos phoned MGAC and followed up with and email to advise there are a number of potential projects of possible interest to MGAC Corporation and offered various option to meet. [Con-4076]
- On 2 April 2024 Santos sent a reminder to MGAC via email that the consultation period was closing on April 9 and to contact Santos as soon as possible if MGAC had any feedback. The information previously provided was again included. [Con-4086]
- On 4 April 2024 the Chair of MGAC emailed Santos to advise the Barossa Production Operations EP would be discussed at a Board meeting the following week, after which an outcome would be provided to Santos. [Con-4090]
- On 4 April 2024 Santos emailed MGAC and advised it looked forward to further communication. [Con-4091]
- On 2 May 2024 Santos emailed MGAC to advise it was following up on previous emails and advised of an additional risk associated with the proposed activity and provided a link to the updated booklet and factsheet. Santos advised it had extended the consultation period and requested MGAC input by 16 May 2024. [Con-4107]
- On 28 May 2024 Santos emailed MGAC in relation to its responsibilities under the Australian Marine Parks North Management Plan for sea country in the Joseph Bonaparte Marine Park. The email included a map showing the location of the park in relation to Miriuwung Gajerrong country. [Con-4210]
- On 28 May 2024 MGAC Chair emailed Santos to advise he would get back to Santos once the matter was reviewed. [Con-5233] No response was received.
- On 10 July 2024 Santos emailed MGAC to advise the consultation period had been completed. Santos advised it consultation had now closed for the purpose of Santos finalising and submitting environment plans for these activities to government regulators for assessment. [Con-5124]



OEMP reference
Not applicable.

- On 15 October 2024 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP.
   NOPSEMA for assessment, provided a link to the full EP on the NOPSEMA website and Santos' contact details for any queries. The Update also stated that the OEMP would be submitted to the NT-DITT for assessment.
- On 23 January 2025 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP.
   NOPSEMA to provide further information, Santos had re-submitted the EP with some amendments in December 2024 for further assessment. The Update also re-stated that the OEMP would be submitted to the
- No further correspondence or feedback on the Coastal Waters OEMP was received from MGAC.

Summary of response by Relevant Person	Assessment of merits	Santos' Response Statement
No response was received from MGAC.	Santos considers it has provided sufficient information and a reasonable period of time for consultation. With regard to the location of the proposed GEP operations activities in NT Coastal Waters, there are no impacts from planned activities that may affect the functions, interests or activities of any First Nations groups, clans and/or organisations along the WA coastline. The worst-case credible spill scenario modelled for the EMBA for the Coastal Waters OEMP also shows there are no impacts from unplanned activities that may affect the functions, interests or activities of any First Nations groups, clans and/or organisations along the WA coastline.	No response required.

### Wunambal Gaambera Aboriginal Corporation (WGAC)

- On 22 February 2024 Santos emailed WGAC to advise it of preliminary consultation regarding proposed activities for consultation to be managed under the Coastal Waters OEMP and Barossa Production Operation
- The email advised that Santos was seeking information to better understand:
  - o if you are from a government Department or agency, how the proposed Operations activities may be relevant to your department or agency; and
  - o what (if any) functions, interests or activities you or your organisation have that may be affected by the proposed Operations activities.
- In the email Santos advised WGAC to contact Santos at the earliest opportunity if it considered that it may be a Relevant Person and wished to participate in the consultation process. Should WGAC be intereste
  appropriate to WAC's information needs and interests.
- The email included information on the regulatory process for the activities in Commonwealth and NT jurisdictions and links to a Santos information booklet on the proposed activities and a NOPSEMA brochure o
  Santos to register as a Relevant Person. The email stated that the consultation phase would commence on 11 March 2024 and close on 9 April 2024.
- Also included was a map of the Environment that May be Affected (EMBA) and a summary of Santos' understanding of how the EMBA related to Aboriginal heritage sites, native title claims or determinations as well as marine park management.
- On 13 March 2024 Santos attempted to call WGAC and then emailed to advise that it had commenced the consultation phase which would run until 9 April 2024. In addition to the previous information again being provided, Santos provided information on Relevant Persons' entitlements under the regulatory processes, details of how to provide feedback and a reminder of the closing date for consultation. In the email Santos provided phone contact details to receive further information or organise a consultation meeting. [Con-4064]
- On 2 April 2024 Santos sent a reminder to WGAC via email that the consultation period was closing on April 9 and to contact Santos as soon as possible if MGAC had any feedback. The information previously provided was again included. [Con-4089]
- On 2 May 2024 Santos phoned and emailed WGAC to follow-up on the previous emails and advised of an additional risk associated with the proposed activity and provided a link to the updated booklet and factsheet. Santos advised it had extended the consultation
  period and requested WAC input by 16 May 2024. [Con-4106]
- On 10 July 2024 Santos emailed WGAC to advise the consultation period had been completed. Santos advised it considered that consultation had now closed for the purpose of Santos finalising and submitting environment plans for these activities to government regulators for assessment. [Con-5125]
- On 15 October 2024 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that the EP had been submitted to NOPSEMA for assessment, provided a link to the full EP on the NOPSEMA website and Santos' contact details for any queries. The Update also stated that the OEMP would be submitted to the NT-DITT for assessment. [Con- 5982]
- On 23 January 2025 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that, following a request from NOPSEMA to provide further information, Santos had re-submitted the EP with some amendments in December 2024 for further assessment. The Update also re-stated that the OEMP would be submitted to the NT-DITT for assessment. [Con-6036]
- Notwithstanding the information provided and the steps described above, no comments or input were received on the Coastal Waters OEMP. from WGAC.

Summary of response by Relevant Person	Assessment of merits	Santos' Response Statement	OEMP reference
No response was received from WGAC.	Santos considers it has provided sufficient information and a reasonable period of time for consultation. With regard to the location of the proposed GEP operations activities in NT Coastal Waters, there are no impacts from planned activities that may affect the functions, interests or activities of any First Nations groups, clans and/or organisations along the WA coastline. The worst-case credible spill scenario modelled for the EMBA for the Coastal Waters OEMP also shows there are no impacts from unplanned activities that may affect the functions, interests or activities of any First Nations groups, clans and/or organisations along the WA coastline.	No response required.	Not applicable.
	Santos considers Section 25 consultation requirements to have been met.		



The Update advised that the EP had been submitted to sessment. [Con- 5982]
The Update advised that, following a request from
e NT-DITT for assessment. [Con-6036]

	OEMP reference	
	Not applicable.	
ations EP. [Con-4044]		
d Santos can then discuss appropriate consultation methods		
n the consultation process and details of how to contact		

### 4.8.9 **Industry Associations**

### Table 4-19: Consultation Summary Table – Industry Associations

Section 25(1)(d) of the OPGGS(E)R: Persons or organisations whose functions, interests or activities may be affected by the activities to be carried out under the environment plan

### Australian Southern Bluefin Tuna Industry Association (ASBTIA)

### Summary of consultation effort:

- On 15 February 2024 Santos emailed ASBTIA to advise it of preliminary consultation regarding proposed activities for consultation to be managed under the Coastal Waters OEMP and Barossa Production Operations EP. [Con-4988]
- The email advised that Santos was seeking information to better understand:
  - o if you are from a government Department or agency, how the proposed Production Operations activities may be relevant to your department or agency; and
  - what (if any) functions, interests or activities you or your organisation have that may be affected by the proposed Production Operations activities.
- The email included information on the regulatory process for the activities in Commonwealth and NT jurisdictions and links to a Santos information booklet on the proposed activities and a NOPSEMA brochure on the consultation process and details of how to contact Santos to register as a Relevant Person. The email stated that the consultation phase would commence on 11 March 2024 and close on 9 April 2024.
- On 11 March 2024, Santos emailed ASBTIA advising that it is now consulting on the Coastal Waters OEMP and Barossa Production Operations EP until Tuesday 9 April 2024 and attached previous correspondence. [Con-3793]
- On 4 April 2024 Santos phoned ASBTIA regarding consultation for Production Operations activities and left a voice mail message.
- On 2 May 2024, Santos emailed ASBTIA, further to previous correspondence to advise it had extended the consultation period until 16 May 2024. In providing this extension of time, Santos advised that the information in the booklet and factsheet had been updated to account for an additional risk associated with the proposed activity. Santos confirmed that consultation will close on the revised date unless Santos hears otherwise from ASBTIA. [Con-4353]
- On 10 July 2024 Santos emailed ASBTIA to advise the consultation period had been completed. Santos advised ASBTIA that it considered consultation had now closed for the purpose of Santos finalising and submitting environment plans for these activities to government regulators for assessment. [Con-5130]
- On 15 October 2024 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that the EP had been submitted to NOPSEMA for assessment, provided a link to the full EP on the NOPSEMA website and Santos' contact details for any queries. The Update also stated that the OEMP would be submitted to the NT-DITT for assessment. [Con-5982]
- On 23 January 2025 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that, following a request from NOPSEMA to provide further information, Santos had re-submitted the EP with some amendments in December 2024 for further assessment. The Update also re-stated that the OEMP would be submitted to the NT-DITT for assessment. [Con-6036]
- Notwithstanding the consultation information provided and the steps described above, no comments or input were received on the Coastal Waters OEMP from ASBTIA

Summary of response by Relevant Person	Assessment of merits	Santos' Response Statement	OEMP reference
No response was received from ASBTIA.	Santos considers it has provided sufficient information and a reasonable period of time for consultation. Santos considers Section 25 consultation requirements to have been met.	No response required.	Not applicable.

### **Commonwealth Fisheries Association (CFA)**

- On 9 February 2024 Santos emailed CFA to advise it of preliminary consultation regarding proposed activities for consultation to be managed under the Coastal Waters OEMP and Barossa Production Operations EP. [Con-3787]
- The email advised that Santos was seeking information to better understand:
  - o if you are from a government Department or agency, how the proposed Production Operations activities may be relevant to your department or agency; and
  - what (if any) functions, interests or activities you or your organisation have that may be affected by the proposed Production Operations activities.
- The email included information on the regulatory process for the activities in Commonwealth and NT jurisdictions and links to a Santos information booklet on the proposed activities and a NOPSEMA brochure on the consultation process and details of how to contact Santos to register as a Relevant Person. The email stated that the consultation phase would commence on 11 March 2024 and close on 9 April 2024.
- On 11 March 2024, Santos emailed CFA advising that it is now consulting on the Coastal Waters OEMP and Barossa Production Operations EP until Tuesday 9 April 2024 and attached previous correspondence. [Con-3793]
- On 4 April 2024 Santos phoned CFA regarding consultation for Production Operations activities and left a voice mail message.
- On 9 May 2024, Santos emailed CFA, further to the previous correspondence, to advise it had extended the consultation period until 23 May 2024. In providing this extension of time, Santos advised that the information in the booklet and factsheet had been updated to account for an additional risk associated with the proposed activity. Santos confirmed that consultation will close on the revised date unless Santos hears otherwise from CFA. [Con-3929]
- On 10 July 2024 Santos emailed CFA to advise the consultation period had been completed. Santos advised CFA that it consultation had now closed for the purpose of Santos finalising and submitting environment plans for these activities to government. regulators for assessment. [Con-5130]
- On 15 October 2024 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that the EP had been submitted to NOPSEMA for assessment, provided a link to the full EP on the NOPSEMA website and Santos' contact details for any queries. The Update also stated that the OEMP would be submitted to the NT-DITT for assessment. [Con-5982]
- On 23 January 2025 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that, following a request from NOPSEMA to provide further information, Santos had re-submitted the EP with some amendments in December 2024 for further assessment. The Update also re-stated that the OEMP would be submitted to the NT-DITT for assessment. [Con-6036]
- Notwithstanding the consultation information provided and the steps described above, no comments or input were received on the Coastal Waters OEMP from CFA.



Summary of response by Relevant Person	Assessment of merits
No response was received from CFA.	Santos considers it has provided sufficient information and a reasonable period of time for consultation.
	Santos considers Section 25 consultation requirements to have been met.

### Northern Territory Seafood Council (NTSC)

Summary of consultation effort:

- On 9 February 2024 Santos emailed NTSC to advise it of preliminary consultation regarding proposed activities for consultation to be managed under the Coastal Waters OEMP and Barossa Production Operation
- The email advised that Santos was seeking information to better understand:
  - o if you are from a government Department or agency, how the proposed Production Operations activities may be relevant to your department or agency; and
  - what (if any) functions, interests or activities you or your organisation have that may be affected by the proposed Production Operations activities.
- The email included information on the regulatory process for the activities in Commonwealth and NT jurisdictions and links to a Santos information booklet on the proposed activities and a NOPSEMA brochure on Santos to register as a Relevant Person. The email stated that the consultation phase would commence on 11 March 2024 and close on 9 April 2024.
- On 11 March 2024, Santos emailed NTSC advising that it is now consulting on the Coastal Waters OEMP and Barossa Production Operations EP until Tuesday 9 April 2024 and attached previous correspondent
- On 4 April 2024 Santos phoned NTSC regarding consultation for Production Operations activities and spoke to a team member who advised that NTSC intended to provide input to the consultation.
- On 8 May 2024 emailed NTSC to advise, further to previous correspondence, it had extended the consultation period until 22 May 2024. In providing this extension of time, Santos advised that the information in an additional risk associated with the proposed activity. Santos confirmed that consultation will close on the revised date unless Santos hears otherwise from NTSC. [Con-4360]
- On 10 July 2024 Santos emailed NTSC to advise the consultation period had been completed. Santos advised NTSC that it considered that consultation had now closed for the purpose of Santos finalising and su
  government regulators for assessment. [Con-5130]
- On 15 October 2024 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. T NOPSEMA for assessment, provided a link to the full EP on the NOPSEMA website and Santos' contact details for any queries. The Update also stated that the OEMP would be submitted to the NT-DITT for ass
- On 23 January 2025 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. T NOPSEMA to provide further information, Santos had re-submitted the EP with some amendments in December 2024 for further assessment. The Update also re-stated that the OEMP would be submitted to the
- Notwithstanding the consultation information provided and the steps described above, no comments or input were received on the Coastal Waters OEMP from NTSC.

Summary of response by Relevant Person	Assessment of merits
No response was received from NTSC.	Santos considers it has provided sufficient information and a reasonable period of time for consultation.
	Santos considers Section 25 consultation requirements to have been met.

### Northern Prawn Fishing Industry (NPFI) Limited

Summary of consultation effort:

- On 9 February 2024 Santos emailed NPFI to advise it of preliminary consultation regarding proposed activities for consultation to be managed under the Coastal Waters OEMP and Barossa Production Operation
- The email advised that Santos was seeking information to better understand:
  - o if you are from a government Department or agency, how the proposed Production Operations activities may be relevant to your department or agency; and
  - what (if any) functions, interests or activities you or your organisation have that may be affected by the proposed Production Operations activities.
- The email included information on the regulatory process for the activities in Commonwealth and NT jurisdictions and links to a Santos information booklet on the proposed activities and a NOPSEMA brochure or Santos to register as a Relevant Person. The email stated that the consultation phase would commence on 11 March 2024 and close on 9 April 2024.
- On 11 March 2024, Santos emailed NPFI advising that it is now consulting on the Coastal Waters OEMP and Barossa Production Operations EP until Tuesday 9 April 2024 and attached previous correspondence
- On 4 April 2024 Santos phoned NPFI regarding consultation for Production Operations activities and left a voice mail message.
- On 2 May 2024, Santos emailed NPFI to advise, further to previous correspondence, it had extended the consultation period until 16 May 2024. In providing this extension of time, Santos advised that the informa
  account for an additional risk associated with the proposed activity. Santos confirmed that consultation will close on the revised date unless Santos hears otherwise from NPFI. [Con-4356]
- On 10 July 2024 Santos emailed NPFI to advise the consultation period had been completed. Santos advised NPFI that it considered consultation had now closed for the purpose of Santos finalising and submitt regulators for assessment. [Con-5130]
- On 15 October 2024 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. T NOPSEMA for assessment, provided a link to the full EP on the NOPSEMA website and Santos' contact details for any queries. The Update also stated that the OEMP would be submitted to the NT-DITT for ass
- On 23 January 2025 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. T NOPSEMA to provide further information, Santos had re-submitted the EP with some amendments in December 2024 for further assessment. The Update also re-stated that the OEMP would be submitted to the

Notwithstanding the consultation information provided and the steps described above, no comments or input were received on the Coastal Waters OEMP from NPFI.

Summary of response by Relevant Person	Assessment of merits
No response was received from NPFI.	Santos considers it has provided sufficient information and a reasonable period of time for consultation.



	Santos' Response Statement	OEMP reference	
	No response required.	Not applicable.	
ons EP. [Con	-3787]		
n the consult	tation process and details of ho	ow to contact	
ce. [Con-379	93]		
the booklet a	and factsheet had been update	d to account for	
ubmitting en	vironment plans for these activ	vities to	
The Update advised that the EP had been submitted to essment. [Con- 5982] The Update advised that, following a request from NT-DITT for assessment. [Con-6036]			
	Santos' Response Statement	OEMP reference	
	No response required.	Not applicable.	
ıs EP. [Con-3787]			
n the consult	tation process and details of ho	ow to contact	
e. [Con-3793]			
ation in the booklet and factsheet had been updated to			
ng environment plans for these activities to government			
The Update advised that the EP had been submitted to essment. [Con- 5982] The Update advised that, following a request from NT-DITT for assessment. [Con-6036]			
	Santos' Response Statement	OEMP reference	
	No response required.	Not applicable.	

	Santos considers Section 25 consultation requirements to have been met.
W	estern Australian Fishing Industry Council (WAFIC)
Su	mmary of consultation effort:
•	On 9 February 2024 Santos emailed WAFIC to advise it of preliminary consultation regarding proposed activities for consultation to be managed under the Coastal Waters OEMP and Barossa Production Operations
•	The email advised that Santos was seeking information to better understand:
	- if you are from a government Department or agency, how the proposed Production Operations activities may be relevant to your department or agency; and
	- what (if any) functions, interests or activities you or your organisation have that may be affected by the proposed Production Operations activities.
•	The email included information on the regulatory process for the activities in Commonwealth and NT jurisdictions and links to a Santos information booklet on the proposed activities and a NOPSEMA brochure on the Santos to register as a Relevant Person. The email stated that the consultation phase would commence on 11 March 2024 and close on 9 April 2024.
•	On 11 March 2024, Santos emailed WAFIC advising that it is now consulting on the Coastal Waters OEMP and Barossa Production Operations EP until Tuesday 9 April 2024 and attached previous correspondence.
•	On 4 April 2024 Santos phoned WAFIC regarding consultation for Production Operations activities and left a voice mail message.
•	On 2 May 2024, Santos emailed WAFIC advise, further to previous correspondence, it had extended the consultation period until 16 May 2024. In providing this extension of time, Santos advised that the information i account for an additional risk associated with the proposed activity. Santos confirmed that consultation will close on the revised date unless Santos hears otherwise from WAFIC. [Con-4357]
•	On 16 May 2024, WAFIC emailed Santos to clarify that WAFIC does not have an interest in activity in NT waters and asked to continue to receive updates on the Barossa Project. [Con-4220]
•	On 10 July 2024 Santos emailed WAFIC to advise the consultation period had been completed. Santos advised it considered that consultation had now closed for the purpose of Santos finalising and submitting environment regulators for assessment. [Con-5130]
•	On 15 October 2024 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The UNOPSEMA for assessment, provided a link to the full EP on the NOPSEMA website and Santos' contact details for any queries. The Update also stated that the OEMP would be submitted to the NT-DITT for assessment

On 23 January 2025 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that, following a request from NOPSEMA to provide further information, Santos had re-submitted the EP with some amendments in December 2024 for further assessment. The Update also re-stated that the OEMP would be submitted to the NT-DITT for assessment. [Con-6036]

Summary of response by Relevant Person	Assessment of merits	Santos' Response Statement	OEMP reference
WAFIC stated that it did not have an interest in activity in NT waters.	Santos acknowledges that WAFIC's stated approach does not require consultation for the activities proposed under the Coastal Waters OEMP. The EMBA for the Coastal Waters OEMP does not extend beyond Commonwealth Waters off the Northern Territory. Therefore, with regard to the location of the proposed GEP operations activities in NT Coastal Waters, there is no consequence from activity on any of WAFIC's interests. The worst-case credible spill scenario modelled for the EMBA for the Coastal Waters OEMP also shows there are no impacts from unplanned activities that may affect the functions, interests or activities of any WA-based persons	Santos confirmed the approach with WAFIC.	No reference required.
WAFIC asked to continue to receive updates on the Barossa project.	Santos provides pre-activity notifications and quarterly project updates to WAFIC.	Santos confirmed the approach with WAFIC.	Section 8.4.7

### Amateur Fishermen's Association of the Northern Territory (AFANT)

Summary of consultation effort:

- On 9 February 2024 Santos emailed AFANT to advise it of preliminary consultation regarding proposed activities for consultation to be managed under the Coastal Waters OEMP and Barossa Production Operations EP. [Con-3787]
- The email advised that Santos was seeking information to better understand:
  - o if you are from a government Department or agency, how the proposed Production Operations activities may be relevant to your department or agency; and
  - what (if any) functions, interests or activities you or your organisation have that may be affected by the proposed Production Operations activities.
- The email included information on the regulatory process for the activities in Commonwealth and NT jurisdictions and links to a Santos information booklet on the proposed activities and a NOPSEMA brochure on the consultation process and details of how to contact Santos to register as a Relevant Person. The email stated that the consultation phase would commence on 11 March 2024 and close on 9 April 2024.
- On 11 March 2024, Santos emailed AFANT advising that it is now consulting on Coastal Waters OEMP and Barossa Production Operations EP until Tuesday 9 April 2024 and attached previous correspondence. [Con-3793]
- On 4 April 2024 Santos phoned AFANT regarding consultation for Production Operations activities and left a voice mail message. On 2 May 2024, Santos emailed AFANT, further to the previous correspondence, to advise it had extended the consultation period until 16 May 2024. In providing this extension of time, Santos advised that the information in the booklet and factsheet had been updated to account for an additional risk associated with the proposed activity. Santos confirmed that consultation will close on the revised date unless Santos hears otherwise from AFANT. [Con-4347]
- On 10 July 2024 Santos emailed AFANT to advise the consultation period had been completed. Santos advised AFANT that it considered consultation had now closed for the purpose of Santos finalising and submitting environment plans for these activities to government regulators for assessment. [Con-5130]
- On 15 October 2024 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that the EP had been submitted to NOPSEMA for assessment, provided a link to the full EP on the NOPSEMA website and Santos' contact details for any gueries. The Update also stated that the OEMP would be submitted to the NT-DITT for assessment. [Con- 5982]
- On 23 January 2025 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that, following a request from NOPSEMA to provide further information, Santos had re-submitted the EP with some amendments in December 2024 for further assessment. The Update also re-stated that the OEMP would be submitted to the NT-DITT for assessment. [Con-6036]
- Notwithstanding the consultation information provided and the steps described above, no comments or input were received on the Coastal Waters OEMP from AFANT.

## Santos

EP. [Con-3787]

consultation process and details of how to contact

[Con-3793]

in the booklet and factsheet had been updated to

conment plans for these activities to government

Jpdate advised that the EP had been submitted to nent. [Con- 5982]

Summary of response by Relevant Person	Assessment of merits
No response was received from AFANT.	Santos considers it has provided sufficient information and a reasonable period of time for consultation.
	Santos considers Section 25 consultation requirements to have been met.

### Recfishwest

Summary of consultation effort:

- On 15 February 2024 Santos emailed Recfishwest to advise it of preliminary consultation regarding proposed activities for consultation to be managed under the Coastal Waters OEMP and Barossa Production O
- The email advised that Santos was seeking information to better understand:
  - o if you are from a government Department or agency, how the proposed Production Operations activities may be relevant to your department or agency; and
  - o what (if any) functions, interests or activities you or your organisation have that may be affected by the proposed Production Operations activities.
- The email included information on the regulatory process for the activities in Commonwealth and NT jurisdictions and links to a Santos information booklet on the proposed activities and a NOPSEMA brochure or Santos to register as a Relevant Person. The email stated that the consultation phase would commence on 11 March 2024 and close on 9 April 2024.
- On 11 March 2024, Santos emailed Recfishwest advising that it is now consulting on the Coastal Waters OEMP and Barossa Production Operations EP until Tuesday 9 April 2024 and attached previous correspondences of the coastal Waters OEMP and Barossa Production Operations EP until Tuesday 9 April 2024 and attached previous correspondences of the coastal Waters OEMP and Barossa Production Operations EP until Tuesday 9 April 2024 and attached previous correspondences of the coastal Waters OEMP and Barossa Production Operations EP until Tuesday 9 April 2024 and attached previous correspondences of the coastal Waters OEMP and Barossa Production Operations EP until Tuesday 9 April 2024 and attached previous correspondences of the coastal Waters OEMP and Barossa Production Operations EP until Tuesday 9 April 2024 and attached previous correspondences of the coastal Waters OEMP and Barossa Production Operations EP until Tuesday 9 April 2024 and attached previous correspondences of the coastal Waters OEMP and Barossa Production Operations EP until Tuesday 9 April 2024 and attached previous correspondences of the coastal Waters OEMP and Barossa Production Operations EP until Tuesday 9 April 2024 and attached previous correspondences of the coastal Waters OEMP and Barossa Production Operations EP until Tuesday 9 April 2024 and attached previous correspondences of the coastal Waters OEMP and Barossa Production Operations EP until Tuesday 9 April 2024 and attached previous correspondences of the coastal Waters OEMP and Barossa Production Operations EP until Tuesday 9 April 2024 and attached previous correspondences of the coastal Waters OEMP and Barossa Production Operations EP until Tuesday 9 April 2024 and attached previous correspondences of the coastal Waters OEMP and Barossa Production Operations EP until Tuesday 9 April 2024 and attached previous correspondences of the coastal Waters OEMP and Barossa Production Operations EP until Tuesday 9 April 2024 and attached previous correspondences of the coastal Wat
- On 4 April 2024 Santos phoned Recfishwest regarding consultation for Production Operations activities and spoke to a team member who advised they would contact Santos if they had any feedback.
- On 6 May 2024, Santos emailed Recfishwest, further to the previous correspondence, to advise it had extended the consultation period until 20 May 2024. In providing this extension of time, Santos advised that
  updated to account for an additional risk associated with the proposed activity. Santos confirmed that consultation will close on the revised date unless Santos hears otherwise from Recfishwest. [Con-4314]
- On 10 July 2024 Santos emailed Recfishwest to advise the consultation period had been completed. Santos advised Recfishwest that it considered consultation had now closed for the purpose of Santos finalisin government regulators for assessment. [Con-5130]
- On 15 October 2024 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. 1 NOPSEMA for assessment, provided a link to the full EP on the NOPSEMA website and Santos' contact details for any queries. The Update also stated that the OEMP would be submitted to the NT-DITT for ass
- On 23 January 2025 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. T NOPSEMA to provide further information, Santos had re-submitted the EP with some amendments in December 2024 for further assessment. The Update also re-stated that the OEMP would be submitted to the
- Notwithstanding the consultation information provided and the steps described above, no comments or input were received on the Coastal Waters OEMP from Recfishwest.

Summary of response by Relevant Person	Assessment of merits
No response was received from Recfishwest.	Santos considers it has provided sufficient information and a reasonable period of time for consultation.
	The EMBA for the Coastal Waters OEMP does not extend beyond Commonwealth Waters off the Northern Territory. Therefore, with regard to the location of the proportions activities in NT Coastal Waters, there are no impacts from planned activities that may affect the functions, interests or activities of any WA-based persons.
	The worst-case credible spill scenario modelled for the EMBA for the Coastal Waters OEMP also shows there are no impacts from unplanned activities that may affect functions, interests or activities of any WA-based persons
	Santos considers Section 25 consultation requirements to have been met.

### Western Australian Game Fishing Association (WAGFA)

- On 9 February 2024 Santos emailed WAGFA to advise it of preliminary consultation regarding proposed activities for consultation to be managed under the Coastal Waters OEMP and Barossa Production Opera
- The email advised that Santos was seeking information to better understand:
  - o if you are from a government Department or agency, how the proposed Production Operations activities may be relevant to your department or agency; and
  - what (if any) functions, interests or activities you or your organisation have that may be affected by the proposed Production Operations activities.
- The email included information on the regulatory process for the activities in Commonwealth and NT jurisdictions and links to a Santos information booklet on the proposed activities and a NOPSEMA brochure or Santos to register as a Relevant Person. The email stated that the consultation phase would commence on 11 March 2024 and close on 9 April 2024.
- On 11 March 2024, Santos emailed WAGFA advising that it is now consulting on the Coastal Waters OEMP and Barossa Production Operations EP until Tuesday 9 April 2024 and attached previous corresponded
- On 8 May 2024, Santos emailed WAGFA, further to the previous correspondence, to advise it had extended the consultation period until 22 May 2024. In providing this extension of time, Santos advised that the
  to account for an additional risk associated with the proposed activity. Santos confirmed that consultation will close on the revised date unless Santos hears otherwise from WAGFA. [Con-4361]
- On 10 July 2024 Santos emailed WAGFA to advise the consultation period had been completed. Santos advised it considered that consultation had now closed for the purpose of Santos finalising and submitting
  regulators for assessment. [Con-5130]
- On 15 October 2024 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. T NOPSEMA for assessment, provided a link to the full EP on the NOPSEMA website and Santos' contact details for any queries. The Update also stated that the OEMP would be submitted to the NT-DITT for assessment.
- On 14 November 2024 Santos phoned the WA Game Fishing Association to seek confirmation that it did not have any comments on the Barossa Production Operations EP or OEMP. An Association representat
  wasn't likely to have any comments. [Con-5983] Santos followed-up via email the same day, advising that any comments should be provided by 29 November 2024 [Con-5984]
- On 23 January 2025 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that, following a re
  NOPSEMA to provide further information, Santos had re-submitted the EP with some amendments in December 2024 for further assessment. The Update also re-stated that the OEMP would be submitted to the NT-DITT for assessment. [Con-6036]
- Notwithstanding the consultation information provided and the steps described above, no comments or input were received on the Coastal Waters OEMP from WAGFA.



		Curreo	
	Santos' Response Statement	OEMP reference	
	No response required.	Not applicable.	
Operations E	P. [Con-4989]		
n the consult	ation process and details of ho	ow to contact	
ondence. [Co	on-3793]		
the informati	on in the booklet and factshee	t had been	
ig and submi	tting environment plans for the	ese activities to	
The Update a sessment. [C	advised that the EP had been s on- 5982]	submitted to	
The Update advised that, following a request from NT-DITT for assessment. [Con-6036]			
	Santos' Response Statement	OEMP reference	
osed GEP t the	No response required.	Not applicable.	
tions EP. [Con-3787]			
n the consultation process and details of how to contact			
ence. [Con-3 information i	793] n the booklet and factsheet ha	d been updated	
environmer	environment plans for these activities to government		
he Update advised that the EP had been submitted to essment. [Con- 5982]			
ve asked for the information to be re-sent and stated it still The Update advised that, following a request from			

Summary of response by Relevant Person	Assessment of merits
No response was received from WAGFA.	Santos considers it has provided sufficient information and a reasonable period of time for consultation.
	The EMBA for the Coastal Waters OEMP does not extend beyond Commonwealth Waters off the Northern Territory. Therefore, with regard to the location of the prop operations activities in NT Coastal Waters, there are no impacts from planned activities that may affect the functions, interests or activities of any WA-based persons.
	The worst-case credible spill scenario modelled for the EMBA for the Coastal Waters OEMP also shows there are no impacts from unplanned activities that may affect functions, interests or activities of any WA-based persons.
	Santos considers Section 25 consultation requirements to have been met.
Northern Territory Guided Fishing Indus	try Association (NTGFIA)
Summary of consultation effort:	
On 9 February 2024 Santos emailed NTGF	IA to advise it of preliminary consultation regarding proposed activities for consultation to be managed under the Coastal Waters OEMP and Barossa Production Opera
• The email advised that Santos was seeking	information to better understand:
<ul> <li>if you are from a government Depart</li> </ul>	artment or agency, how the proposed Production Operations activities may be relevant to your department or agency; and
<ul> <li>what (if any) functions, interests or</li> </ul>	r activities you or your organisation have that may be affected by the proposed Production Operations activities.
The email included information on the regul. Santos to register as a Relevant Person. The second secon	atory process for the activities in Commonwealth and NT jurisdictions and links to a Santos information booklet on the proposed activities and a NOPSEMA brochure on the email stated that the consultation phase would commence on 11 March 2024 and close on 9 April 2024.
• On 11 March 2024, Santos emailed NTGFI/	A advising that it is now consulting on the Coastal Waters OEMP and Barossa Production Operations EP until Tuesday 9 April 2024 and attached previous corresponde
<ul> <li>On 4 April 2024 Santos phoned NTGFIA reg Screamin' Barra Fishing)</li> </ul>	garding consultation for Production Operations activities and left a detailed message with an office-holder. The office-holder was also a representative of a charter fishing
<ul> <li>On 6 May 2024 Santos emailed NTGFIA, fu account for an additional risk associated wit</li> </ul>	in the rot the previous correspondence, to advise it had extended the consultation period until 20 May 2024. In providing this extension of time, Santos advised that the in the proposed activity. Santos confirmed that consultation will close on the revised date unless Santos hears otherwise from NTGFIA. [Con-4312].
<ul> <li>On 10 July 2024 Santos emailed NTGFIA to government regulators for assessment. [Co</li> </ul>	advise the consultation period had been completed. Santos advised NTGFIA that it considered consultation had now closed for the purpose of Santos finalising and s n-5130]
<ul> <li>On 15 October 2024 Santos emailed the Ba NOPSEMA for assessment, provided a link</li> </ul>	arossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. T to the full EP on the NOPSEMA website and Santos' contact details for any queries. The Update also stated that the OEMP would be submitted to the NT-DITT for ass
<ul> <li>On 23 January 2025 Santos emailed the Ba NOPSEMA to provide further information, S</li> </ul>	arossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. <sup>2</sup> Santos had re-submitted the EP with some amendments in December 2024 for further assessment. The Update also re-stated that the OEMP would be submitted to the
Notwithstanding the consultation information	n provided and the steps described above, no comments or input were received on the Coastal Waters OEMP from NTGEIA
Summary of response by Pelevant Person	
Summary of response by Relevant Person	
No response was received from NTGFIA.	Santos considers it has provided sufficient information and a reasonable period of time for consultation.
	Santos considers Section 25 consultation requirements to have been met.
Assosiasaun Turizmu Maritima iha Timo	r-Leste
Summary of consultation effort:	
On 9 February 2024 Santos emailed Assosi	iasaun Turizmu Maritima iha Timor-Leste to advise it of preliminary consultation regarding proposed activities for consultation to be managed under the Coastal Waters
The email advised that Santos was seeking	information to better understand:
<ul> <li>if you are from a government Depair</li> <li>what (if any) functions, interests or</li> </ul>	artment or agency, how the proposed Production Operations activities may be relevant to your department or agency; and r activities you or your organisation have that may be affected by the proposed Production Operations activities.
The email included information on the regul Santos to register as a Relevant Person. Th	atory process for the activities in Commonwealth and NT jurisdictions and links to a Santos information booklet on the proposed activities and a NOPSEMA brochure on the email stated that the consultation phase would commence on 11 March 2024 and close on 9 April 2024.
On 13 March 2024, Santos emailed Assosia Relevant Person, It attaches previous corre	asaun Turizmu Maritima iha Timor-Leste advising that it is now consulting on the Coastal Waters OEMP and Barossa Production Operations EP and asks for consideration spondence and advised that the consultation is open until Tuesday 9 April 2024. [Con-3794]
On 4 April 2024 Santos phoned Assosiasau	In Turizmu Maritima iha Timor-Leste regarding consultation for Production Operations activities but was unable to leave a voice message.

- On 2 May 2024, Santos emailed Assosiasaun Turizmu Maritima iha Timor-Leste, further to the previous correspondence, to advise it had extended the consultation period until 16 May 2024. In providing this exte booklet and factsheet had been updated to account for an additional risk associated with the proposed activity. Santos confirmed that consultation will close on the revised date unless Santos hears otherwise from 4352]
- On 10 July 2024 Santos emailed Assosiasaun Turizmu Maritima iha Timor-Leste to advise the consultation period had been completed. Santos advised Assosiasaun Turizmu Maritima iha Timor-Leste that it consultation period had been completed. • Santos finalising and submitting environment plans for these activities to government regulators for assessment. [Con-5130]
- On 15 October 2024 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that the NOPSEMA for assessment, provided a link to the full EP on the NOPSEMA website and Santos' contact details for any queries. The Update also stated that the OEMP would be submitted to the NT-DITT for assessment. [Con- 5982]



	Santos' Response Statement	OEMP reference	
osed GEP	No response required.	Not applicable.	
t the			
tions EP. [C	on-3787]		
n the consult	ation process and details of ho	ow to contact	
ence. [Con-3 ng operator (	793] also see separate entry in this	Table for Reel	
nformation ir	n the booklet and factsheet had	been updated to	
ubmitting en	vironment plans for these activ	ities to	
The Update a sessment. [C	advised that the EP had been s on- 5982]	submitted to	
The Update advised that, following a request from NT-DITT for assessment. [Con-6036]			
	Santos' Response Statement	OEMP reference	
	No response required.	Not applicable.	
OEMP and Barossa Production Operations EP. [Con-3787]			
n the consultation process and details of how to contact			
ation as to whether their organisation considers itself a			
ension of time, Santos advised that the information in the m Assosiasaun Turizmu Maritima iha Timor-Leste. [Con-			
sidered cons	ultation had now closed for the	e purpose of	
be Undete advised that the CD had been submitted to			

On 23 January 2025 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that, following a request from • NOPSEMA to provide further information, Santos had re-submitted the EP with some amendments in December 2024 for further assessment. The Update also re-stated that the OEMP would be submitted to the NT-DITT for assessment. [Con-6036]

stal Watara OEMD from Apopoiosoup Turizmu Maritima iba Tin the Co

Notwithstanding the consultation information provided and the steps described above, no comments or input were received on the Coastal Waters OEMP from Assosiasaun Turizmu Maritima ina Timor-Leste			
Summary of response by Relevant Person	Assessment of merits	Santos' Response Statement	OEMP reference
No response was received from Assosiasaun Turizmu Maritima iha Timor-Leste.	Santos considers it has provided sufficient information and a reasonable period of time for consultation. The EMBA for the Coastal Waters OEMP does not extend beyond Commonwealth Waters off the Northern Territory. There are therefore no impacts from planned GEP operations activities in NT Coastal Waters that may affect the functions, interests or activities of any international persons. The worst-case credible spill scenario modelled for the EMBA for the Coastal Waters OEMP also shows there are no impacts from unplanned activities that may affect the functions, interests or activities of any international persons. Santos considers Section 25 consultation requirements to have been met.	No response required.	Not applicable.



### Kimberley Marine Tourism Association (KMTA)

Summary of consultation effort:

- On 9 February 2024 Santos emailed KMTA to advise it of preliminary consultation regarding proposed activities for consultation to be managed under the Coastal Waters OEMP and Barossa Production Operations EP. [Con-3787]
- The email advised that Santos was seeking information to better understand:
  - o if you are from a government Department or agency, how the proposed Production Operations activities may be relevant to your department or agency; and
    - o what (if any) functions, interests or activities you or your organisation have that may be affected by the proposed Production Operations activities.
- The email included information on the regulatory process for the activities in Commonwealth and NT jurisdictions and links to a Santos information booklet on the proposed activities and a NOPSEMA brochure on the consultation process and details of how to contact Santos to register as a Relevant Person. The email stated that the consultation phase would commence on 11 March 2024 and close on 9 April 2024.
- On 11 March 2024, Santos emailed KMTA advising that it is now consulting on the Coastal Waters OEMP and Barossa Production Operations EP until Tuesday 9 April 2024 and attached previous correspondence. [Con-3793]
- On 4 April 2024 Santos phoned KMTA regarding consultation for Production Operations activities and left a detailed voice mail message.
- On 2 May 2024, Santos emailed KMTA, further to the previous correspondence, to advise it had extended the consultation period until 16 May 2024. In providing this extension of time, Santos advised that the information in the booklet and factsheet had been updated to account for an additional risk associated with the proposed activity. Santos confirmed that consultation will close on the revised date unless Santos hears otherwise from KMTA. [Con-4355]
- On 10 July 2024 Santos emailed KMTA to advise the consultation period had been completed. Santos advised KMTA that it considered consultation had now closed for the purpose of Santos finalising and submitting environment plans for these activities to government regulators for assessment. [Con-5130]
- On 15 October 2024 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that the EP had been submitted to NOPSEMA for assessment, provided a link to the full EP on the NOPSEMA website and Santos' contact details for any queries. The Update also stated that the OEMP would be submitted to the NT-DITT for assessment. [Con- 5982]
- On 23 January 2025 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that, following a request from NOPSEMA to provide further information, Santos had re-submitted the EP with some amendments in December 2024 for further assessment. The Update also re-stated that the OEMP would be submitted to the NT-DITT for assessment. [Con-6036]

Notwithstanding the consultation information provided and the steps described above, no comments or input were received on the Coastal Waters OEMP from KMTA

Summary of response by Relevant Person	Assessment of merits	Santos' Response Statement	OEMP reference
No response was received from KMTA.	Santos considers it has provided sufficient information and a reasonable period of time for consultation. The EMBA for the Coastal Waters OEMP does not extend beyond Commonwealth Waters off the Northern Territory.	No response required.	Not applicable.
	The EMBA for the Coastal Waters OEMP does not extend beyond Commonwealth Waters off the Northern Territory. There are therefore no impacts from planned GEP operations activities in NT Coastal Waters that may affect the functions, interests or activities of any relevant persons with		
	The worst-case credible spill scenario modelled for the EMBA for the Coastal Waters OEMP also shows there are no impacts from unplanned activities that may affect the functions, interests or activities of any relevant persons with functions, interests or activities in WA waters.		
	Santos considers Section 25 consultation requirements to have been met.		
Marina Tourism WA (MTWA)			

### ISM WA (MTW

Summary of consultation effort:

- On 9 February 2024 Santos emailed MTWA to advise it of preliminary consultation regarding proposed activities for consultation to be managed under the Coastal Waters OEMP and Barossa Production Operations EP. [Con-3787]
- The email advised that Santos was seeking information to better understand:
  - o if you are from a government Department or agency, how the proposed Production Operations activities may be relevant to your department or agency; and
  - what (if any) functions, interests or activities you or your organisation have that may be affected by the proposed Production Operations activities. 0
- The email included information on the regulatory process for the activities in Commonwealth and NT jurisdictions and links to a Santos information booklet on the proposed activities and a NOPSEMA brochure on the consultation process and details of how to contact Santos to register as a Relevant Person. The email stated that the consultation phase would commence on 11 March 2024 and close on 9 April 2024.
- On 11 March 2024, Santos emailed MTWA advising that it is now consulting on the Coastal Waters OEMP and Barossa Production Operations EP until Tuesday 9 April 2024 and attached previous correspondence. [Con-3793]
- On 4 April 2024 Santos phoned MTWA regarding consultation for Production Operations activities which advised that Recfishwest and the WA Department of Fisheries should be contacted instead of MTWA.
- On 6 May 2024, Santos emailed MTWA, further to the previous correspondence, to advise it had extended the consultation period until 20 May 2024. In providing this extension of time, Santos advised that the information in the booklet and factsheet had been updated to account for an additional risk associated with the proposed activity. Santos confirmed that consultation will close on the revised date unless Santos hears otherwise from MTWA. [Con-4316]
- On 10 July 2024 Santos emailed MTWA to advise the consultation period had been completed. Santos advised it consultation had now closed for the purpose of Santos finalising and submitting environment plans for these activities to government regulators for assessment, [Con-5130]
- On 15 October 2024 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that the EP had been submitted to NOPSEMA for assessment, provided a link to the full EP on the NOPSEMA website and Santos' contact details for any queries. The Update also stated that the OEMP would be submitted to the NT-DITT for assessment. [Con- 5982]
- On 23 January 2025 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that, following a request from NOPSEMA to provide further information, Santos had re-submitted the EP with some amendments in December 2024 for further assessment. The Update also re-stated that the OEMP would be submitted to the NT-DITT for assessment. [Con-6036]

Notwithstanding the consultation information provided and the steps described above, no comments or input were received on the Coastal Waters OEMP from MTWA.

Summary of response by Relevant Person	Assessment of merits	Santos' Response Statement	OEMP reference
No response was received from MTWA.	Santos considers it has provided sufficient information and a reasonable period of time for consultation.	No response required.	Not applicable.



The EMBA for the Coastal Waters OEMP does not extend beyond Commonwealth Waters off the Northern Territory.
There are therefore no impacts from planned GEP operations activities in NT Coastal Waters that may affect the functions, interests or activities of any relevant person functions, interests or activities in WA waters.
The worst-case credible spill scenario modelled for the EMBA for the Coastal Waters OEMP also shows there are no impacts from unplanned activities that may affect functions, interests or activities of any relevant persons with functions, interests or activities in WA waters. Santos considers Section 25 consultation requirements to have been met.

### **Tourism Top End**

Summary of consultation effort:

- On 9 February 2024 Santos emailed Tourism Top End to advise it of preliminary consultation regarding proposed activities for consultation to be managed under the Coastal Waters OEMP and Barossa Production Operations EP. [Con-3787]
- The email advised that Santos was seeking information to better understand:
  - o if you are from a government Department or agency, how the proposed Production Operations activities may be relevant to your department or agency; and
  - what (if any) functions, interests or activities you or your organisation have that may be affected by the proposed Production Operations activities. 0
- The email included information on the regulatory process for the activities in Commonwealth and NT jurisdictions and links to a Santos information booklet on the proposed activities and a NOPSEMA brochure on the consultation process and details of how to contact Santos to register as a Relevant Person. The email stated that the consultation phase would commence on 11 March 2024 and close on 9 April 2024.
- On 11 March 2024, Santos emailed Santos emailed KMTA, further to the previous correspondence, to advise it had extended the consultation period until 16 May 2024. In providing this extension of time, Santos advised that the information in the booklet and factsheet had been updated to account for an additional risk associated with the proposed activity. Santos confirmed that consultation will close on the revised date unless Santos hears otherwise from KMTA. advising that it is now consulting on Barossa Production Operations Environment Plan in Commonwealth and Northern Territory waters until Tuesday 9 April 2024 and attached previous correspondence. [Con-3793]
- On 4 April 2024 Santos phoned Tourism Top End regarding consultation for Production Operations activities which stated it would call Santos back.
- On 6 May 2024, Santos emailed Tourism Top End, further to the previous correspondence, to advise it had extended the consultation period until 20 May 2024. In providing this extension of time, Santos advised that the information in the booklet and factsheet had been updated to account for an additional risk associated with the proposed activity. Santos confirmed that consultation will close on the revised date unless Santos hears otherwise from Tourism Top End. [Con-4310]
- On 10 July 2024 Santos emailed Tourism Top End to advise the consultation period had been completed. Santos advised Tourism Top End that it considered consultation had now closed for the purpose of Santos finalising and submitting environment plans for these activities to government regulators for assessment. [Con-5130]
- On 15 October 2024 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that the EP had been submitted to NOPSEMA for assessment, provided a link to the full EP on the NOPSEMA website and Santos' contact details for any gueries. The Update also stated that the OEMP would be submitted to the NT-DITT for assessment. [Con- 5982]
- On 23 January 2025 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that, following a request from NOPSEMA to provide further information, Santos had re-submitted the EP with some amendments in December 2024 for further assessment. The Update also re-stated that the OEMP would be submitted to the NT-DITT for assessment. [Con-6036]
- Notwithstanding the consultation information provided and the steps described above, no comments or input were received on the Coastal Waters OEMP from Tourism Top End.

Summary of response by Relevant Person	Assessment of merits	Santos' Response Statement	OEMP reference
No response was received from Tourism Top End.	Santos considers it has provided sufficient information and a reasonable period of time for consultation. Santos considers Section 25 consultation requirements to have been met.	No response required.	Not applicable.

### Chamber of Commerce Northern Territory (CCNT)

- On 9 February 2024 Santos emailed CCNT to advise it of preliminary consultation regarding proposed activities for consultation to be managed under the Coastal Waters OEMP and Barossa Production Operations EP. [Con-3787]
- The email advised that Santos was seeking information to better understand:
  - o if you are from a government Department or agency, how the proposed Production Operations activities may be relevant to your department or agency; and
  - what (if any) functions, interests or activities you or your organisation have that may be affected by the proposed Production Operations activities. 0
- The email included information on the regulatory process for the activities in Commonwealth and NT jurisdictions and links to a Santos information booklet on the proposed activities and a NOPSEMA brochure on the consultation process and details of how to contact Santos to register as a Relevant Person. The email stated that the consultation phase would commence on 11 March 2024 and close on 9 April 2024.
- On 11 March 2024, Santos emailed CCNT advising that it is now consulting on Coastal Waters OEMP and Barossa Production Operations EP until Tuesday 9 April 2024 and attached previous correspondence. [Con-3793]
- On 4 April 2024 Santos phoned CCNT regarding consultation for Production Operations activities and left a detailed voice message.
- On 2 May 2024, Santos emailed CCNT, further to the previous correspondence, to advise it had extended the consultation period until 16 May 2024. In providing this extension of time, Santos advised that the information in the booklet and factsheet had been updated to account for an additional risk associated with the proposed activity. Santos confirmed that consultation will close on the revised date unless Santos hears otherwise from CCNT. [Con-4354]
- On 10 July 2024 Santos emailed CCNT to advise the consultation period had been completed. Santos advised CCNT that it considered consultation had now closed for the purpose of Santos finalising and submitting environment plans for these activities to government regulators for assessment. [Con-5130]
- On 15 October 2024 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that the EP had been submitted to NOPSEMA for assessment, provided a link to the full EP on the NOPSEMA website and Santos' contact details for any queries. The Update also stated that the OEMP would be submitted to the NT-DITT for assessment. [Con- 5982]
- On 23 January 2025 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that, following a request from NOPSEMA to provide further information. Santos had re-submitted the EP with some amendments in December 2024 for further assessment. The Update also re-stated that the OEMP would be submitted to the NT-DITT for assessment. [Con-6036]
- Notwithstanding the consultation information provided and the steps described above, no comments or input were received on the Coastal Waters OEMP from CCNT.



ons with	
t the	

Summary of response by Relevant Person	Assessment of merits	Santos' Response Statement	OEMP reference
No response was received from CCNT.	Santos considers it has provided sufficient information and a reasonable period of time for consultation. Santos considers Section 25 consultation requirements to have been met.	No response required.	Not applicable.
#### 4.8.10 Infrastructure Operators

# Table 4-20: Consultation Summary Table – Infrastructure Operators

Section 25(1)(d) of the OPGGS(E)R: Persons or organisations whose functions, interests or activities may be affected by the activities to be carried out under the environment plan

# **BW Digital**

Summary of consultation effort:

- On 9 February 2024 Santos emailed BW Digital to advise it of preliminary consultation regarding proposed activities for consultation to be managed under the Coastal Waters OEMP and Barossa Production Operations EP. [Con-3787]
- The email advised that Santos was seeking information to better understand:
  - o if you are from a government Department or agency, how the proposed Production Operations activities may be relevant to your department or agency; and
  - what (if any) functions, interests or activities you or your organisation have that may be affected by the proposed Production Operations activities.
- The email included information on the regulatory process for the activities in Commonwealth and NT jurisdictions and links to a Santos information booklet on the proposed activities and a NOPSEMA brochure on the consultation process and details of how to contact Santos to register as a Relevant Person. The email stated that the consultation phase would commence on 11 March 2024 and close on 9 April 2024.
- On 11 March 2024 Santos emailed BW Digital further to the previous correspondence, to advise that it had commenced the consultation phase which would run until 9 April 2024. In addition to the previous information again being provided, Santos provided information on Relevant Persons' entitlements under the regulatory processes, details of how to provide feedback and a reminder of the closing date for consultation. [Con-3793]
- On 4 April 2024 Santos phoned BW digital regarding consultation for Production Operations activities and left a voice mail message.
- On 2 May 2024, Santos emailed BW Digital, to advise it was calling to follow up on previous emails and to advise it had extended the consultation period until 16 May 2024. In providing this extension of time, Santos advised that the information in the booklet and factsheet had been updated to account for an additional risk associated with the proposed activity. Santos confirmed that consultation will close on the revised date unless Santos hears otherwise from BW Digital. [Con-4346]
- On 10 July 2024 Santos emailed BW Digital to advise the consultation period had been completed. Santos advised BW Digital that it considered consultation had now closed for the purpose of Santos finalising and submitting environment plans for these activities to government regulators for assessment. [Con-5130]
- On 15 October 2024 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that the EP had been submitted to NOPSEMA for assessment, provided a link to the full EP on the NOPSEMA website and Santos' contact details for any queries. The Update also stated that the OEMP would be submitted to the NT-DITT for assessment. [Con- 5982]
- On 23 January 2025 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that, following a request from NOPSEMA to provide further information, Santos had re-submitted the EP with some amendments in December 2024 for further assessment. The Update also re-stated that the OEMP would be submitted to the NT-DITT for assessment. [Con-6036]
- Notwithstanding the consultation information provided and the steps described above, no comments or input were received on the Coastal Waters OEMP from BW Digital

Summary of response by Relevant Person	Assessment of merits	Santos' Response Statement	OEMP reference
No response was received from BW Digital.	Santos considers it has provided sufficient information and a reasonable period of time for consultation.	No response required.	Not applicable.
	Santos considers Section 25 consultation requirements to have been met.		

# **Darwin Port**

Summary of consultation effort:

- On 9 February 2024 Santos emailed Darwin Port to advise it of preliminary consultation regarding proposed activities for consultation to be managed under the Coastal Waters OEMP and Barossa Production Operations EP. [Con-3787]
- The email advised that Santos was seeking information to better understand:
  - o if you are from a government Department or agency, how the proposed Production Operations activities may be relevant to your department or agency; and
  - what (if any) functions, interests or activities you or your organisation have that may be affected by the proposed Production Operations activities. 0
- The email included information on the regulatory process for the activities in Commonwealth and NT jurisdictions and links to a Santos information booklet on the proposed activities and a NOPSEMA brochure on the consultation process and details of how to contact Santos to register as a Relevant Person. The email stated that the consultation phase would commence on 11 March 2024 and close on 9 April 2024.
- On 11 March 2024 Santos emailed Darwin Port further to the previous correspondence, to advise that it had commenced the consultation phase which would run until 9 April 2024. In addition to the previous information again being provided, Santos provided information on Relevant Persons' entitlements under the regulatory processes, details of how to provide feedback and a reminder of the closing date for consultation. [Con-3793]
- On 4 April 2024 Santos phoned Darwin Port and left a message with a team member regarding consultation for Production Operations activities
- On 6 May 2024, Santos emailed Darwin Port to advise it was calling to follow up on previous emails and to advise it had extended the consultation period until 20 May 2024. In providing this extension of time, Santos advised that the information in the booklet and factsheet had been updated to account for an additional risk associated with the proposed activity. Santos confirmed that consultation will close on the revised date unless Santos hears otherwise from Darwin Port. [Con-4348]
- On 10 July 2024 Santos emailed Darwin Port to advise the consultation period had been completed. Santos advised Darwin Port that it considered consultation had now closed for the purpose of Santos finalising and submitting environment plans for these activities to government regulators for assessment, [Con-5130].
- On 15 October 2024 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that the EP had been submitted to NOPSEMA for assessment, provided a link to the full EP on the NOPSEMA website and Santos' contact details for any gueries. The Update also stated that the OEMP would be submitted to the NT-DITT for assessment. [Con- 5982]
- On 23 January 2025 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that, following a request from NOPSEMA to provide further information, Santos had re-submitted the EP with some amendments in December 2024 for further assessment. The Update also re-stated that the OEMP would be submitted to the NT-DITT for assessment. [Con-6036]
- Notwithstanding the consultation information provided and the steps described above, no comments or input were received on the Coastal Waters OEMP from Darwin Port.



Summary of response by Relevant Person	Assessment of merits	Santos' Response Statement
No response was received from Darwin Port.	Santos considers it has provided sufficient information and a reasonable period of time for consultation.	No response required.
	Santos considers Section 25 consultation requirements to have been met.	

# **NT Ports and Marine**

Summary of consultation effort:

- On 9 February 2024 Santos emailed NT Ports and Marine to advise it of preliminary consultation regarding proposed activities for consultation to be managed under the Coastal Waters OEMP and Barossa Prod
- The email advised that Santos was seeking information to better understand:
  - o if you are from a government Department or agency, how the proposed Production Operations activities may be relevant to your department or agency; and
  - what (if any) functions, interests or activities you or your organisation have that may be affected by the proposed Production Operations activities.
- The email included information on the regulatory process for the activities in Commonwealth and NT jurisdictions and links to a Santos information booklet on the proposed activities and a NOPSEMA brochure o Santos to register as a Relevant Person. The email stated that the consultation phase would commence on 11 March 2024 and close on 9 April 2024.
- On 11 March 2024 Santos emailed NT Ports and Marine further to the previous correspondence, to advise that it had commenced the consultation phase which would run until 9 April 2024. In addition to the previous information on Relevant Persons' entitlements under the regulatory processes, details of how to provide feedback and a reminder of the closing date for consultation. [Con-3793]
- On 4 April 2024 Santos phoned NT Ports and Marine and left a voice mail message for message regarding consultation for Production Operations activities.
- On 4 April 2024 NT Ports and Marine emailed Santos and stated that it had no feedback for this consultation process. [Con-3537]
- On 6 May 2024, Santos emailed NT Ports and Marine further to their email of 4 April 2024 to advise it had extended the consultation period until 20 May 2024. In providing this extension of time, Santos advised updated to account for an additional risk associated with the proposed activity. Santos confirmed that consultation will close on the revised date unless Santos hears otherwise from NT Ports and Marine. [Con-4:
- On 10 July 2024 Santos emailed NT Ports and Marine to advise the consultation period had been completed. Santos advised it considered that consultation had now closed for the purpose of Santos finalising ar government regulators for assessment. [Con-5130]
- On 15 October 2024 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. NOPSEMA for assessment, provided a link to the full EP on the NOPSEMA website and Santos' contact details for any queries. The Update also stated that the OEMP would be submitted to the NT-DITT for ass
- On 23 January 2025 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. NOPSEMA to provide further information. Santos had re-submitted the EP with some amendments in December 2024 for further assessment. The Update also re-stated that the OEMP would be submitted to the
- No further correspondence or feedback was received on the Coastal Waters OEMP from NT Ports and Marine.

Summary of response by Relevant Person	Assessment of merits	Santos' Response Statement
NT Ports and Marine responded that it had no feedback for this consultation process.	Santos notes the response from NT Ports and Marine. Santos considers it has provided sufficient information and a reasonable period of time for consultation. Santos considers Section 25 consultation requirements to have been met.	No response required.

# **NT Power and Water Corporation**

Summary of consultation effort:

- On 9 February 2024 Santos emailed NT Power and Water Corporation to advise it of preliminary consultation regarding proposed activities for consultation to be managed under the Coastal Waters OEMP and I
- The email advised that Santos was seeking information to better understand:
  - o if you are from a government Department or agency, how the proposed Production Operations activities may be relevant to your department or agency; and
  - what (if any) functions, interests or activities you or your organisation have that may be affected by the proposed Production Operations activities.
- The email included information on the regulatory process for the activities in Commonwealth and NT jurisdictions and links to a Santos information booklet on the proposed activities and a NOPSEMA brochure o Santos to register as a Relevant Person. The email stated that the consultation phase would commence on 11 March 2024 and close on 9 April 2024.
- On 11 March 2024 Santos emailed NT Power and Water Corporation further to the previous correspondence, to advise that it had commenced the consultation phase which would run until 9 April 2024. In addition provided information on Relevant Persons' entitlements under the regulatory processes, details of how to provide feedback and a reminder of the closing date for consultation. [Con-3793]
- On 3 April 2024 Santos phoned NT Power and Water and spoke to a team member regarding consultation for Production Operations activities. The team member advised that NT Power and Water advised that
- On 6 May 2024, Santos emailed NT Power and Water further to the previous correspondence, to advise it had extended the consultation period until 20 May 2024. In providing this extension of time. Santos advi been updated to account for an additional risk associated with the proposed activity. Santos confirmed that consultation will close on the revised date unless Santos hears otherwise from the NT Power and Wate
- On 10 July 2024 Santos emailed NT Power and Water to advise the consultation period had been completed. Santos advised it considered that consultation had now closed for the purpose of Santos finalising and submitting environment plans for these activities to government regulators for assessment. [Con-5086]
- On 15 October 2024 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that the EP had been submitted to NOPSEMA for assessment, provided a link to the full EP on the NOPSEMA website and Santos' contact details for any queries. The Update also stated that the OEMP would be submitted to the NT-DITT for assessment. [Con- 5982]
- On 23 January 2025 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that, following a request from NOPSEMA to provide further information. Santos had re-submitted the EP with some amendments in December 2024 for further assessment. The Update also re-stated that the OEMP would be submitted to the NT-DITT for assessment. [Con-6036]
- No further correspondence or feedback on the Coastal Waters OEMP was received from NT Power and Water Corporation.



	Santos	
	OEMP reference	
	Not applicable.	
luction Operations EP. [Cc	ın-3787]	
on the consultation process	and details of how to contact	
vious information again be	ing provided, Santos provided	
that the information in the booklet and factsheet had been 381] nd submitting environment plans for these activities to The Update advised that the EP had been submitted to sessment. [Con- 5982] The Update advised that, following a request from a NT-DITT for assessment. [Con-6036]		
	OEMP reference	
	Not applicable.	
Barossa Production Opera	tions EP. [Con-3787]	
on the consultation process	and details of how to contact	
on to the previous information again being provided, Santos		
it would not be contributing	g to the consultation.	
ised that the information in er. [Con-4379]	the booklet and factsheet had	
nd submitting environmen	t plans for these activities to	

Summary of response by Relevant Person	Assessment of merits	Santos' Response Statement
NT Power and Water Corporation responded that it would not be participating in the consultation process.	Santos notes the response by NT Power and Water Corporation. Santos considers it has provided sufficient information and a reasonable period of time for consultation. Santos considers Section 25 consultation requirements to have been met.	No response required.

# Sun Cable

Summary of consultation effort:

- On 9 February 2024 Santos emailed Sun Cable to advise it of preliminary consultation regarding proposed activities for consultation to be managed under the Coastal Waters OEMP and Barossa Production Operation Operation (Consultation Consultation) (Consultation) (Consulta
- The email advised that Santos was seeking information to better understand:
  - o if you are from a government Department or agency, how the proposed Production Operations activities may be relevant to your department or agency; and
  - what (if any) functions, interests or activities you or your organisation have that may be affected by the proposed Production Operations activities.
- The email included information on the regulatory process for the activities in Commonwealth and NT jurisdictions and links to a Santos information booklet on the proposed activities and a NOPSEMA brochure or Santos to register as a Relevant Person. The email stated that the consultation phase would commence on 11 March 2024 and close on 9 April 2024.
- On 11 March 2024 Santos emailed Sun Cable further to the previous correspondence, to advise that it had commenced the consultation phase which would run until 9 April 2024. In addition to the previous inforr Relevant Persons' entitlements under the regulatory processes, details of how to provide feedback and a reminder of the closing date for consultation. [Con-3793]
- On 8 May 2024, Santos emailed Sun Cable further to emails sent previously and to advise it had extended the consultation period until 22 May 2024. In providing this extension of time, Santos advised that the in account for an additional risk associated with the proposed activity. Santos confirmed that consultation will close on the revised date unless Santos hears otherwise from Sun Cable [Con-3996]
- On 23 May 2024, Sun Cable emailed Santos advising that it would like to be consulted on the EP and referred to previous information provided. They also advised they preferred any consultation feedback to ren
- On 10 July 2024 Santos emailed Sun Cable to advise the consultation period had been completed. Santos advised it considered that consultation had now closed for the purpose of Santos finalising and submitti
  regulators for assessment. [Con-5130]
- On 15 October 2024 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP.
   NOPSEMA for assessment, provided a link to the full EP on the NOPSEMA website and Santos' contact details for any queries. The Update also stated that the OEMP would be submitted to the NT-DITT for assessment.
- On 5 December 2024 Santos emailed Sun Cable to request a further confirmation that Sun Cable had no comments or input on the Coastal Waters OEMP. Sun Cable responded with confirmation via email the s
- On 23 January 2025 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP.
   NOPSEMA to provide further information, Santos had re-submitted the EP with some amendments in December 2024 for further assessment. The Update also re-stated that the OEMP would be submitted to the

• Notwithstanding the consultation information provided and the steps described above, no comments or input were received on the Coastal Waters OEMP. from Sun Cable.

Summary of response by Relevant Person	Assessment of merits	Santos' Response Statement
Sun Cable requested their response be kept confidential.	Information provided was outside the scope of the Coastal Waters OEMP. Santos considers it has provided sufficient information and a reasonable period of time for consultation. Santos considers Section 25 consultation requirements to have been met.	No response required.



	OEMP reference	
	Not applicable.	
erations EP. [Con-3787]		
on the consultation process	s and details of how to contact	
mation again being provided, Santos provided information on		
nformation in the booklet and factsheet had been updated to		
nain confidential. [Con-42	18]	
ing environment plans for these activities to government		
The Update advised that the EP had been submitted to sessment. [Con- 5982] same day. [Con-6011] The Update advised that, following a request from e NT-DITT for assessment. [Con-6036]		
	OEMP reference	
	Not applicable.	

# Telstra

Summary of consultation effort:

- On 9 February 2024 Santos emailed Telstra to advise it of preliminary consultation regarding proposed activities for consultation to be managed under the Coastal Waters OEMP and Barossa Production Operations EP. [Con-3787]
- The email advised that Santos was seeking information to better understand:
  - o if you are from a government Department or agency, how the proposed Production Operations activities may be relevant to your department or agency; and
    - what (if any) functions, interests or activities you or your organisation have that may be affected by the proposed Production Operations activities. 0
- The email included information on the regulatory process for the activities in Commonwealth and NT jurisdictions and links to a Santos information booklet on the proposed activities and a NOPSEMA brochure on the consultation process and details of how to contact Santos to register as a Relevant Person. The email stated that the consultation phase would commence on 11 March 2024 and close on 9 April 2024.
- On 11 March 2024 Santos emailed Telstra further to the previous correspondence, to advise that it had commenced the consultation phase which would run until 9 April 2024. In addition to the previous information again being provided, Santos provided information on Relevant Persons' entitlements under the regulatory processes, details of how to provide feedback and a reminder of the closing date for consultation. [Con-3793]
- On 4 April 2024 Santos phoned Telstra regarding consultation for Production Operations activities and spoke to a team member who confirmed it had received the emails.
- On 6 May 2024, Santos emailed Telstra, to advise it was calling to follow up on previous emails and to advise it had extended the consultation period until 20 May 2024. In providing this extension of time, Santos advised that the information in the booklet and factsheet had been updated to account for an additional risk associated with the proposed activity. Santos confirmed that consultation will close on the revised date unless Santos hears otherwise from Telstra. [Con-4349].
- On 10 July 2024 Santos emailed Telstra to advise the consultation period had been completed. Santos advised Telstra that it considered consultation had now closed for the purpose of Santos finalising and submitting environment plans for these activities to government regulators for assessment. [Con-5130]
- On 15 October 2024 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that the EP had been submitted to NOPSEMA for assessment, provided a link to the full EP on the NOPSEMA website and Santos' contact details for any queries. The Update also stated that the OEMP would be submitted to the NT-DITT for assessment. [Con- 5982]
- On 23 January 2025 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that, following a request from NOPSEMA to provide further information. Santos had re-submitted the EP with some amendments in December 2024 for further assessment. The Update also re-stated that the OEMP would be submitted to the NT-DITT for assessment. [Con-6036]
- Notwithstanding the consultation information provided and the steps described above, no comments or input were received on the Coastal Waters OEMP. from Telstra

Summary of response by Relevant Person	Assessment of merits	Santos' Response Statement
No response was received from Telstra.	Santos considers it has provided sufficient information and a reasonable period of time for consultation.	No response required.
	Santos considers Section 25 consultation requirements to have been met.	

# Vocus

Summary of consultation effort:

- On 9 February 2024 Santos emailed Vocus to advise it of preliminary consultation regarding proposed activities for consultation to be managed under the Coastal Waters OEMP and Barossa Production Operations EP. [Con-3787]
- The email advised that Santos was seeking information to better understand:
  - o if you are from a government Department or agency, how the proposed Production Operations activities may be relevant to your department or agency; and
  - what (if any) functions, interests or activities you or your organisation have that may be affected by the proposed Production Operations activities.
- The email included information on the regulatory process for the activities in Commonwealth and NT jurisdictions and links to a Santos information booklet on the proposed activities and a NOPSEMA brochure on the consultation process and details of how to contact Santos to register as a Relevant Person. The email stated that the consultation phase would commence on 11 March 2024 and close on 9 April 2024.
- On 11 March 2024 Santos emailed Vocus further to the previous correspondence, to advise that it had commenced the consultation phase which would run until 9 April 2024. In addition to the previous information again being provided, Santos provided information on Relevant Persons' entitlements under the regulatory processes, details of how to provide feedback and a reminder of the closing date for consultation. [Con-3793]
- On 4 April 2024 Santos phoned Vocus and spoke to a team member regarding Production Operations activities which advised it would check with the relevant Vocus personnel.
- On 4 April 2024 Vocus emailed Santos and advised that the EMBA does cover over the top of the NWCS and if there was to be an incident, Vocus would need to be informed and understand the cleanup process, wanting to ensure there is no disturbance to the sea floor where the cable is located. Vocus noted Santos' documentation already mentioned the NWCS and a crossing Letter of No Objection (LONO) was already in place for the Barossa GEP. Other than receiving the relevant crossing documentation and As Built information upon completion, Vocus stated it had no further concerns about the Barossa Operations. [Con-4315]
- On 8 May 2024, Santos emailed Vocus, to advise it had extended the consultation period until 22 May 2024. In providing this extension of time, Santos advised that the information in the booklet and factsheet had been updated to account for an additional risk associated with the proposed activity. Santos confirmed that consultation will close on the revised date unless Santos hears otherwise from Vocus. [Con-4351]
- On 10 July 2024 Santos emailed Vocus to advise the consultation period had been completed. Santos advised it consultation had now closed for the purpose of Santos finalising and submitting environment plans for these activities to government regulators for assessment. [Con-5130]
- On 15 October 2024 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that the EP had been submitted to NOPSEMA for assessment, provided a link to the full EP on the NOPSEMA website and Santos' contact details for any gueries. The Update also stated that the OEMP would be submitted to the NT-DITT for assessment. [Con- 5982]
- On 6 December 2024 Santos advised Vocus via email that its request to be informed in the event of a spill, if clean-up activities may impact its infrastructure, had been included in Santos' formal notification process. [Con-6014]
- On 23 January 2025 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that, following a request from NOPSEMA to provide further information, Santos had re-submitted the EP with some amendments in December 2024 for further assessment. The Update also re-stated that the OEMP would be submitted to the NT-DITT for assessment. [Con-6036]
- No further correspondence or feedback on the Coastal Waters OEMP was received from Vocus.



OEMP reference
Not applicable.

Summary of response by Relevant Person	Assessment of merits	Santos' Response Statement
Vocus responded that it wished to be informed in the event of a spill if clean-up activities may impact Vocus' infrastructure.	Santos has considered the matters raised in the OEMP and provided a response.	Santos advised Vocus that its request had be formal notification process.
	Santos has included Vocus in its formal spill notification process.	



	OEMP reference
een included in Santos'	Notifications are included in Section 8.4.7

#### Local Government Authorities and Recognised Community Reference / Liaison Groups 4.8.11

# Table 4-21: Consultation Summary Table – Local Government Authorities and Recognised Community Reference / Liaison Groups

Section 25(1)(d) of the OPGGS(E)R: Persons or organisations whose functions, interests or activities may be affected by the activities to be carried out under the environment plan

# **Belyuen Community Government Council**

Summary of consultation effort:

- On 9 February 2024 Santos emailed the Belyuen Community Government Council to advise it of preliminary consultation regarding proposed activities for consultation to be managed under the Coastal Waters OEMP and Barossa Production Operations EP. [Con-3787]
- The email advised that Santos was seeking information to better understand:
  - o if you are from a government Department or agency, how the proposed Production Operations activities may be relevant to your department or agency; and
  - what (if any) functions, interests or activities you or your organisation have that may be affected by the proposed Production Operations activities. 0
- The email included information on the regulatory process for the activities in Commonwealth and NT jurisdictions and links to a Santos information booklet on the proposed activities and a NOPSEMA brochure on the consultation process and details of how to contact Santos to register as a Relevant Person. The email stated that the consultation phase would commence on 11 March 2024 and close on 9 April 2024.
- On 11 March 2024 Santos emailed the Belyuen Community Government Council further to the previous correspondence, to advise that it had commenced the consultation phase which would run until 9 April 2024. In addition to the previous information again being provided, Santos provided information on Relevant Persons' entitlements under the regulatory processes, details of how to provide feedback and a reminder of the closing date for consultation. [Con-3793]
- On 4 April 2024 Santos phoned the Belyuen Community Government Council and left a voice mail message regarding consultation for Production Operations activities.
- On 2 May 2024, Santos emailed the Belyuen Community Government Council further to previous correspondence to advise it had extended the consultation period until 16 May 2024. In providing this extension of time, Santos advised that the information in the booklet and factsheet had been updated to account for an additional risk associated with the proposed activity. Santos confirmed that consultation will close on the revised date unless Santos hears otherwise from Belyuen Community Government Council. [Con-4985]
- On 10 July 2024 Santos emailed Belyuen Community Government Council to advise the consultation period had been completed. Santos advised Belyuen Community Government Council that it considered consultation had now closed for the purpose of Santos finalising and submitting environment plans for these activities to government regulators for assessment. [Con-5130]
- On 11 July 2024 Belyuen Community Government Council emailed Santos advising it had not had a great deal of input with the project and requested Santos be available at some stage to provide a summary of the project relating to the people of the Belyuen region. [Con-5115]
- On 23 July 2024 Santos had a phone discussion with a representative of the Council's Executive during which the representative stated they wanted to ensure the Council was involved in discussions on potential community benefits and employment and training opportunities associated with Santos' activities going forward. The representative did not have any specific questions about the OEMP and EP. [Con-5236]
- On 24 July 2024 Santos emailed Belyuen Community Government Council in follow-up to the phone discussion on 23 July 2024 [Con-5214] Santos is progressing discussions with the Council separate to consultation on the Coastal Waters OEMP.
- On 15 October 2024 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that the EP had been submitted to NOPSEMA for assessment, provided a link to the full EP on the NOPSEMA website and Santos' contact details for any queries. The Update also stated that the OEMP would be submitted to the NT-DITT for assessment. [Con- 5982]
- On 23 January 2025 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that, following a request from NOPSEMA to provide further information. Santos had re-submitted the EP with some amendments in December 2024 for further assessment. The Update also re-stated that the OEMP would be submitted to the NT-DITT for assessment. [Con-6036]
- No further correspondence has been received from the EP from Belyuen Community Government Council.
- Additional to formal consultation on the Coastal Waters OEMP and Barossa Production Operations EP, Santos has continued to make a range of efforts to engage on an ongoing basis with NT First Nations People, both directly and through their representative organisations. See Section 8.4.9 Post-acceptance Consultation Implementation Strategy.

Summary of response by Relevant Person	Assessment of merits	Santos' Response Statement	OEMP I
Belyuen Community Government Council stated it wants to be involved in discussions on potential community benefits and employment and training opportunities associated with Santos' activities going forward.	The matters raised are outside the scope of the Coastal Waters OEMP. Santos considers it has provided sufficient information and a reasonable period of time for consultation. Santos considers Section 25 consultation requirements to have been met.	Santos has committed to ongoing discussions with the Council on matters that are separate to consultation on the Coastal Waters OEMP.	Not app

# **City of Darwin**

Summary of consultation effort:

- On 9 February 2024 Santos emailed City of Darwin to advise it of preliminary consultation regarding proposed activities for consultation to be managed under the Coastal Waters OEMP and Barossa Production Operations EP. [Con-3787]
- The email advised that Santos was seeking information to better understand:
  - o if you are from a government Department or agency, how the proposed Production Operations activities may be relevant to your department or agency; and
  - what (if any) functions, interests or activities you or your organisation have that may be affected by the proposed Production Operations activities.
- The email included information on the regulatory process for the activities in Commonwealth and NT jurisdictions and links to a Santos information booklet on the proposed activities and a NOPSEMA brochure on the consultation process and details of how to contact Santos to register as a Relevant Person. The email stated that the consultation phase would commence on 11 March 2024 and close on 9 April 2024.
- On 11 March 2024 Santos emailed City of Darwin further to the previous correspondence, to advise that it had commenced the consultation phase which would run until 9 April 2024. In addition to the previous information again being provided, Santos provided information on Relevant Persons' entitlements under the regulatory processes, details of how to provide feedback and a reminder of the closing date for consultation. [Con-3793]
- On 3 April 2024 Santos phoned City of Darwin regarding consultation for Community Government Council EP activities and spoke to a team member and was asked to resend the previous emails. Santos sent City of Darwin copies of emails sent on 9 February and 11 March 2024 on the same day. [Con-4149]
- On 5 April 2024 City of Darwin emailed Santos and advised it has no feedback on the OEMP and EP. [Con-4986]



reference

licable.

- On 7 May 2024, Santos emailed City of Darwin further to previous correspondence to advise it had extended the consultation period until 21 May 2024. In providing this extension of time, Santos advised that the information in the booklet and factsheet had been updated to account for an additional risk associated with the proposed activity. Santos confirmed that consultation will close on the revised date unless Santos hears otherwise from the City of Darwin [Con-4159]
- On 10 July 2024 Santos emailed City of Darwin to advise the consultation period had been completed. Santos advised it consultation had now closed for the purpose of Santos finalising and submitting environment plans for these activities to government regulators for assessment. [Con-5130]
- On 15 October 2024 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that the EP had been submitted to NOPSEMA for assessment, provided a link to the full EP on the NOPSEMA website and Santos' contact details for any queries. The Update also stated that the OEMP would be submitted to the NT-DITT for assessment. [Con- 5982]
- On 23 January 2025 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that, following a request from NOPSEMA to provide further information, Santos had re-submitted the EP with some amendments in December 2024 for further assessment. The Update also re-stated that the OEMP would be submitted to the NT-DITT for assessment. [Con-6036]

No further correspondence has been received from the City of Darwin.

Summary of response by Relevant Person	Assessment of merits	Santos' Response Statement	OEMP
City of Darwin responded that it had no feedback on the GEP Coastal Waters OEMP.	Santos considers it has provided sufficient information and a reasonable period of time for consultation.	No response required.	Not ap
	Santos considers Section 25 consultation requirements to have been met.		



reference

plicable.

# **East Arnhem Regional Council**

Summary of consultation effort:

- On 9 February 2024 Santos emailed East Arnhem Regional Council to advise it of preliminary consultation regarding proposed activities for consultation to be managed under the Coastal Waters OEMP and Barossa Production Operations EP. [Con-3787]
- The email advised that Santos was seeking information to better understand:
  - o if you are from a government Department or agency, how the proposed Production Operations activities may be relevant to your department or agency; and
  - what (if any) functions, interests or activities you or your organisation have that may be affected by the proposed Production Operations activities.
- The email included information on the regulatory process for the activities in Commonwealth and NT jurisdictions and links to a Santos information booklet on the proposed activities and a NOPSEMA brochure on the consultation process and details of how to contact Santos to register as a Relevant Person. The email stated that the consultation phase would commence on 11 March 2024 and close on 9 April 2024.
- On 11 March 2024 Santos emailed East Arnhem Regional Council further to the previous correspondence, to advise that it had commenced the consultation phase which would run until 9 April 2024. In addition to the previous information again being provided, Santos provided information on Relevant Persons' entitlements under the regulatory processes, details of how to provide feedback and a reminder of the closing date for consultation. [Con-3793]
- On 3 April 2024 Santos phoned East Arnhem Regional Council and left a voice mail message regarding consultation for Production Operations activities.
- On 2 May 2024, Santos emailed East Arnhem Regional Council further to previous correspondence to advise it had extended the consultation period until 16 May 2024. In providing this extension of time, Santos advised that the information in the booklet and factsheet had been updated to account for an additional risk associated with the proposed activity. Santos confirmed that consultation will close on the revised date unless Santos hears otherwise from the East Arnhem Regional Council. [Con-4152]
- On 10 July 2024 Santos emailed East Arnhem Regional Council to advise the consultation period had been completed. Santos advised East Arnhem Regional Council that it considered consultation had now closed for the purpose of Santos finalising and submitting environment plans for these activities to government regulators for assessment. [Con-5130]
- On 15 October 2024 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that the EP had been submitted to NOPSEMA for assessment, provided a link to the full EP on the NOPSEMA website and Santos' contact details for any queries. The Update also stated that the OEMP would be submitted to the NT-DITT for assessment. [Con- 5982]
- On 23 January 2025 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that, following a request from NOPSEMA to provide further information, Santos had re-submitted the EP with some amendments in December 2024 for further assessment. The Update also re-stated that the OEMP would be submitted to the NT-DITT for assessment. [Con-6036]
- Notwithstanding the consultation information provided and the steps described above, no comments or input were received on the Coastal Waters OEMP from East Arnhem Regional Council.

Summary of response by Relevant Person	Assessment of merits	Santos' Response Statement	OEMP
No response was received from East Arnhem Regional Council.	Santos considers it has provided sufficient information and a reasonable period of time for consultation.	No response required.	Not app
	Santos considers Section 25 consultation requirements to have been met.		

# **Litchfield Council**

Summary of consultation effort:

- On 9 February 2024 Santos emailed Litchfield Council to advise it of preliminary consultation regarding proposed activities for consultation to be managed under the Coastal Waters OEMP and Barossa Production Operations EP. [Con-3787]
- The email advised that Santos was seeking information to better understand:
  - o if you are from a government Department or agency, how the proposed Production Operations activities may be relevant to your department or agency; and
  - what (if any) functions, interests or activities you or your organisation have that may be affected by the proposed Production Operations activities. 0
- The email included information on the regulatory process for the activities in Commonwealth and NT jurisdictions and links to a Santos information booklet on the proposed activities and a NOPSEMA brochure on the consultation process and details of how to contact Santos to register as a Relevant Person. The email stated that the consultation phase would commence on 11 March 2024 and close on 9 April 2024.
- On 11 March 2024 Santos emailed Litchfield Council further to the previous correspondence, to advise that it had commenced the consultation phase which would run until 9 April 2024. In addition to the previous information again being provided, Santos provided information on Relevant Persons' entitlements under the regulatory processes, details of how to provide feedback and a reminder of the closing date for consultation. [Con-3793]
- On 3 April 2024 Santos phoned Litchfield Council regarding consultation for Production Operations activities and spoke to a team member who confirmed previously sent emails had been received.
- On 6 May 2024, Santos emailed Litchfield Council further to previous correspondence to advise it had extended the consultation period until 20 May 2024. In providing this extension of time, Santos advised that the information in the booklet and factsheet had been updated to account for an additional risk associated with the proposed activity. Santos confirmed that consultation will close on the revised date unless Santos hears otherwise from Litchfield Council. [Con-4987]
- On 10 July 2024 Santos emailed Litchfield Council to advise the consultation period had been completed. Santos advised Litchfield Council that it considered consultation had now closed for the purpose of Santos finalising and submitting environment plans for these activities to government regulators for assessment. [Con-5130]
- On 15 October 2024 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that the EP had been submitted to NOPSEMA for assessment, provided a link to the full EP on the NOPSEMA website and Santos' contact details for any queries. The Update also stated that the OEMP would be submitted to the NT-DITT for assessment. [Con-5982]
- On 23 January 2025 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that, following a request from NOPSEMA to provide further information, Santos had re-submitted the EP with some amendments in December 2024 for further assessment. The Update also re-stated that the OEMP would be submitted to the NT-DITT for assessment. [Con-6036]

	•	Notwithstanding the consultation information	provided and the steps described ab	we, no comments or input were received on the Coastal Waters OEMP from Litchfield Council.
--	---	----------------------------------------------	-------------------------------------	--------------------------------------------------------------------------------------------

City of Polymorphon			
	Santos considers Section 25 consultation requirements to have been met.		
No response was received from Litchfield Council.	Santos considers it has provided sufficient information and a reasonable period of time for consultation.	No response required.	Not appli
Summary of response by Relevant Person	Assessment of merits	Santos' Response Statement	OEMP r

### City of Palmersto



reference

licable.

eference

icable.

- On 9 February 2024 Santos emailed the City of Palmerston to advise it of preliminary consultation regarding proposed activities for consultation to be managed under the Coastal Waters OEMP and Barossa Proc
- The email advised that Santos was seeking information to better understand:
  - o if you are from a government Department or agency, how the proposed Production Operations activities may be relevant to your department or agency; and
  - what (if any) functions, interests or activities you or your organisation have that may be affected by the proposed Production Operations activities.
- The email included information on the regulatory process for the activities in Commonwealth and NT jurisdictions and links to a Santos information booklet on the proposed activities and a NOPSEMA brochure on the consultation process and details of how to contact Santos to register as a Relevant Person. The email stated that the consultation phase would commence on 11 March 2024 and close on 9 April 2024.
- On 11 March 2024 Santos emailed the City of Palmerston further to the previous correspondence, to advise that it had commenced the consultation phase which would run until 9 April 2024. In addition to the previous information again being provided, Santos provided information on Relevant Persons' entitlements under the regulatory processes, details of how to provide feedback and a reminder of the closing date for consultation. [Con-3793]
- On 3 April 2024 Santos phoned the City of Palmerston regarding consultation for Production Operations activities and spoke with a team member who asked for the previously sent emails to be resent.
- On 5 April 2024 Santos emailed City of Palmerston copies of emails originally sent on 9 February and 11 March 2024. [Con-4150]
- On 8 April 2024, City of Palmerston emailed Santos and advised they had no feedback on the OEMP and EP [Con-4151]
- On 7 May 2024, Santos emailed the City of Palmerston further to previous correspondence to advise it had extended the consultation period until 21 May 2024. In providing this extension of time, Santos advised that the information in the booklet and factsheet had been updated to account for an additional risk associated with the proposed activity. Santos confirmed that consultation will close on the revised date unless Santos hears otherwise from the City of Palmerston. [Con-4158]
- On 10 July 2024 Santos emailed City of Palmerston to advise the consultation period had been completed. Santos advised it considered that consultation had now closed for the purpose of Santos finalising and submitting environment plans for these activities to government regulators for assessment. [Con-5130]
- On 15 October 2024 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that the EP had been submitted to NOPSEMA for assessment, provided a link to the full EP on the NOPSEMA website and Santos' contact details for any queries. The Update also stated that the OEMP would be submitted to the NT-DITT for assessment. [Con- 5982]
- On 23 January 2025 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that, following a request from NOPSEMA to provide further information. Santos had re-submitted the EP with some amendments in December 2024 for further assessment. The Update also re-stated that the OEMP would be submitted to the NT-DITT for assessment. [Con-6036]
- No further correspondence has been received from the City of Palmerston.

Summary of response by Relevant Person	Assessment of merits	Santos' Response Statement	OEMP r
City of Palmerston responded that it had no feedback on the Production Operations EP.	Santos considers it has provided sufficient information and a reasonable period of time for consultation.	No response required.	Not app
	Santos considers Section 25 consultation requirements to have been met.		

# **Roper Gulf Regional Council**

Summary of consultation effort:

- On 9 February 2024 Santos emailed Roper Gulf Regional Council to advise it of preliminary consultation regarding proposed activities for consultation to be managed under the Coastal Waters OEMP and Barossa Production Operations EP. [Con-3787]
- The email advised that Santos was seeking information to better understand:
  - o if you are from a government Department or agency, how the proposed Production Operations activities may be relevant to your department or agency; and
  - what (if any) functions, interests or activities you or your organisation have that may be affected by the proposed Production Operations activities.
- The email included information on the regulatory process for the activities in Commonwealth and NT jurisdictions and links to a Santos information booklet on the proposed activities and a NOPSEMA brochure on the consultation process and details of how to contact Santos to register as a Relevant Person. The email stated that the consultation phase would commence on 11 March 2024 and close on 9 April 2024.
- On 11 March 2024 Santos emailed Roper Gulf Regional Council further to the previous correspondence, to advise that it had commenced the consultation phase which would run until 9 April 2024. In addition to the previous information again being provided, Santos provided information on Relevant Persons' entitlements under the regulatory processes, details of how to provide feedback and a reminder of the closing date for consultation. [Con-3793]
- On 3 April 2024 Santos phoned Roper Gulf Regional Council and spoke to a team member regarding consultation for Production Operations activities.
- On 8 May 2024, Santos emailed Roper Gulf Regional Council further to previous correspondence to advise it had extended the consultation period until 22 May 2024. In providing this extension of time, Santos advised that the information in the booklet and factsheet had been updated to account for an additional risk associated with the proposed activity. Santos confirmed that consultation will close on the revised date unless Santos hears otherwise from Roper Gulf Regional Council. [Con-4223]
- On 10 July 2024 Santos emailed Roper Gulf Regional Council to advise the consultation period had been completed. Santos advised Roper Gulf Regional Council that it considered consultation had now closed for the purpose of Santos finalising and submitting environment plans for these activities to government regulators for assessment. [Con-5130]
- On 15 October 2024 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that the EP had been submitted to NOPSEMA for assessment, provided a link to the full EP on the NOPSEMA website and Santos' contact details for any queries. The Update also stated that the OEMP would be submitted to the NT-DITT for assessment. [Con- 5982]
- On 23 January 2025 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that, following a request from NOPSEMA to provide further information, Santos had re-submitted the EP with some amendments in December 2024 for further assessment. The Update also re-stated that the OEMP would be submitted to the NT-DITT for assessment. [Con-6036]
- Notwithstanding the consultation information provided and the steps described above, no comments or input were received on the Coastal Waters OEMP from Roper Gulf Regional Council.

Tiwi Islanda Bagianal Caunail			
	Santos considers Section 25 consultation requirements to have been met.		
No response was received from Roper Gulf Regional Council.	Santos considers it has provided sufficient information and a reasonable period of time for consultation.	No response required.	Not appl
Summary of response by Relevant Person	Assessment of merits	Santos' Response Statement	OEMP r

### iwi Islands Regional Council

# Santos

duction	Onorationa	ED	[Con 2797]
auction	Operations	сг.	

eference

licable.

eference

licable.

- On 9 February 2024 Santos emailed Tiwi Islands Regional Council to advise it of preliminary consultation regarding proposed activities for consultation to be managed under the Coastal Waters OEMP and Barossa Production Operations EP. [Con-3787]
- The email advised that Santos was seeking information to better understand:
  - o if you are from a government Department or agency, how the proposed Production Operations activities may be relevant to your department or agency; and
  - what (if any) functions, interests or activities you or your organisation have that may be affected by the proposed Production Operations activities. 0
- The email included information on the regulatory process for the activities in Commonwealth and NT jurisdictions and links to a Santos information booklet on the proposed activities and a NOPSEMA brochure on the consultation process and details of how to contact Santos to register as a Relevant Person. The email stated that the consultation phase would commence on 11 March 2024 and close on 9 April 2024.
- On 11 March 2024 Santos emailed Tiwi Islands Regional Council further to the previous correspondence, to advise that it had commenced the consultation phase which would run until 9 April 2024. In addition to the previous information again being provided, Santos provided information on Relevant Persons' entitlements under the regulatory processes, details of how to provide feedback and a reminder of the closing date for consultation. [Con-3793]
- On 4 April 2024 Santos phoned Tiwi Islands Regional Council and left a voice mail message regarding consultation for Barossa production operations activities.
- . On 2 May 2024, Santos emailed Tiwi Islands Regional Council further to previous correspondence to advise it had extended the consultation period until 16 May 2024. In providing this extension of time, Santos advised that the information in the booklet and factsheet had been updated to account for an additional risk associated with the proposed activity. Santos confirmed that consultation will close on the revised date unless Santos hears otherwise from Tiwi Islands Regional Council. [Con-4155]
- On 10 July 2024 Santos emailed Tiwi Islands Regional Council to advise the consultation period had been completed. Santos advised Tiwi Islands Regional Council that it considered consultation had now closed for the purpose of Santos finalising and submitting environment plans for these activities to government regulators for assessment. [Con-5130]
- On 15 October 2024 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that the EP had been submitted to NOPSEMA for assessment, provided a link to the full EP on the NOPSEMA website and Santos' contact details for any queries. The Update also stated that the OEMP would be submitted to the NT-DITT for assessment. [Con- 5982]
- On 23 January 2025 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that, following a request from NOPSEMA to provide further information, Santos had re-submitted the EP with some amendments in December 2024 for further assessment. The Update also re-stated that the OEMP would be submitted to the NT-DITT for assessment. [Con-6036]
- Notwithstanding the consultation information provided and the steps described above, no comments or input were received on the Coastal Waters OEMP from Tiwi Islands Regional Council.
- Additional to formal consultation on the Production Operations EP and OEMP, Santos has continued to make a range of efforts to engage on an ongoing basis with NT First Nations People, both directly and through their representative organisations. See Section 8.4.9 Post-acceptance Consultation Implementation Strategy.

Summary of response by Relevant Person	Assessment of merits	Santos' Response Statement	OEMP r
No response was received from Tiwi Islands Regional Council.	Santos considers it has provided sufficient information and a reasonable period of time for consultation.	No response required.	Santos' in Secti
	Consultation with Tiwi people was undertaken by Santos directly with Tiwi people and clan groups and organised via Clan Trustees who are also members of the TLC and Tiwi Regional Council. Santos considers Section 25 consultation requirements to have been met.		Tiwi Lan

# Santos

### reference

consultation with Tiwi people and clan groups is described on 4.7.6.1 and summarised in other entries in this Table for nd Council and Tiwi people and clan groups.

# Victoria Daly Regional Council

Summary of consultation effort:

- On 9 February 2024 Santos emailed Victoria Daly Regional Council to advise it of preliminary consultation regarding proposed activities for consultation to be managed under the Coastal Waters OEMP and Barossa Production Operations EP. [Con-3787]
- The email advised that Santos was seeking information to better understand:
  - o if you are from a government Department or agency, how the proposed Production Operations activities may be relevant to your department or agency; and
    - what (if any) functions, interests or activities you or your organisation have that may be affected by the proposed Production Operations activities. 0
- The email included information on the regulatory process for the activities in Commonwealth and NT jurisdictions and links to a Santos information booklet on the proposed activities and a NOPSEMA brochure on the consultation process and details of how to contact Santos to register as a Relevant Person. The email stated that the consultation phase would commence on 11 March 2024 and close on 9 April 2024.
- On 11 March 2024 Santos emailed Victoria Daly Regional Council further to the previous correspondence, to advise that it had commenced the consultation phase which would run until 9 April 2024. In addition to the previous information again being provided, Santos provided information on Relevant Persons' entitlements under the regulatory processes, details of how to provide feedback and a reminder of the closing date for consultation. [Con-3793]
- On 3 April 2024 Santos phoned Victoria Daly Regional Council and left a voice mail message regarding consultation for Production Operations activities.
- On 2 May 2024, Santos emailed Victoria Daly Regional Council further to previous correspondence to advise it had extended the consultation period until 16 May 2024. In providing this extension of time, Santos advised that the information in the booklet and factsheet had been updated to account for an additional risk associated with the proposed activity. Santos confirmed that consultation will close on the revised date unless Santos hears otherwise from Victoria Daly Regional Council. [Con-4153]
- On 10 July 2024 Santos emailed Victoria Daly Regional Council to advise the consultation period had been completed. Santos advised Victoria Daly Regional Council that it considered consultation had now closed for the purpose of Santos finalising and submitting environment plans for these activities to government regulators for assessment. [Con-5130]
- On 15 October 2024 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that the EP had been submitted to NOPSEMA for assessment, provided a link to the full EP on the NOPSEMA website and Santos' contact details for any queries. The Update also stated that the OEMP would be submitted to the NT-DITT for assessment. [Con- 5982]
- On 23 January 2025 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that, following a request from NOPSEMA to provide further information. Santos had re-submitted the EP with some amendments in December 2024 for further assessment. The Update also re-stated that the OEMP would be submitted to the NT-DITT for assessment. [Con-6036]
- Notwithstanding the consultation information provided and the steps described above, no comments or input were received on the Coastal Waters OEMP from Victoria Daly Regional Council.

Summary of response by Relevant Person	Assessment of merits	Santos' Response Statement	OEMP
No response was received from Victoria Daly Regional Council.	Santos considers it has provided sufficient information and a reasonable period of time for consultation. Santos considers Section 25 consultation requirements to have been met.	No response required.	Not app
			1

## **Wagait Shire Council**

Summary of consultation effort:

- On 9 February 2024 Santos emailed Wagait Shire Council to advise it of preliminary consultation regarding proposed activities for consultation to be managed under the Coastal Waters OEMP and Barossa Production Operations EP. [Con-3787]
- The email advised that Santos was seeking information to better understand:
  - o if you are from a government Department or agency, how the proposed Production Operations activities may be relevant to your department or agency; and
  - what (if any) functions, interests or activities you or your organisation have that may be affected by the proposed Production Operations activities.
- The email included information on the regulatory process for the activities in Commonwealth and NT jurisdictions and links to a Santos information booklet on the proposed activities and a NOPSEMA brochure on the consultation process and details of how to contact Santos to register as a Relevant Person. The email stated that the consultation phase would commence on 11 March 2024 and close on 9 April 2024.
- On 11 March 2024 Santos emailed Wagait Shire Council further to the previous correspondence, to advise that it had commenced the consultation phase which would run until 9 April 2024. In addition to the previous information again being provided, Santos provided information on Relevant Persons' entitlements under the regulatory processes, details of how to provide feedback and a reminder of the closing date for consultation. [Con-3793]
- On 3 April 2024 Santos phoned Wagait Shire Council regarding consultation for Production Operations activities and spoke to a team member who asked for previously sent emails to be resent. Santos re-sent the emails the same day. [Con-4147]
- On 7 May 2024, Santos emailed Wagait Shire Council further to previous correspondence to advise it had extended the consultation period until 21 May 2024. In providing this extension of time, Santos advised that the information in the booklet and factsheet had been updated to account for an additional risk associated with the proposed activity. Santos confirmed that consultation will close on the revised date unless Santos hears otherwise from Wagait Shire Council. [Con-4221]
- On 7 May 2024 Wagait Shire Council emailed Santos and advised it would forward the email received to Councillors. [Con-4222]
- On 10 July 2024 Santos emailed Wagait Shire Council to advise the consultation period had been completed. Santos advised Wagait Shire Council that it considered consultation had now closed for the purpose of Santos finalising and submitting environment plans for these activities to government regulators for assessment. [Con-5130]
- On 15 October 2024 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that the EP had been submitted to NOPSEMA for assessment, provided a link to the full EP on the NOPSEMA website and Santos' contact details for any gueries. The Update also stated that the OEMP would be submitted to the NT-DITT for assessment. [Con- 5982]
- On 23 January 2025 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that, following a request from NOPSEMA to provide further information, Santos had re-submitted the EP with some amendments in December 2024 for further assessment. The Update also re-stated that the OEMP would be submitted to the NT-DITT for assessment. [Con-6036]
- Notwithstanding the consultation information provided and the steps described above, no comments or input were received on the Coastal Waters OEMP from Wagait Shire Council

			1
	Santos considers Section 25 consultation requirements to have been met.		
No response was received from Wagait Shire Council.	Santos considers it has provided sufficient information and a reasonable period of time for consultation.	No response required.	Not app
Summary of response by Relevant Person	Assessment of merits	Santos' Response Statement	OEMP r

### Nest Arnhem Regional Council



reference

licable.

reference

licable.

- On 9 February 2024 Santos emailed West Arnhem Regional Council to advise it of preliminary consultation regarding proposed activities for consultation to be managed under the Coastal Waters OEMP and Barossa Production Operations EP. [Con-3787]
- The email advised that Santos was seeking information to better understand:
  - o if you are from a government Department or agency, how the proposed Production Operations activities may be relevant to your department or agency; and
  - what (if any) functions, interests or activities you or your organisation have that may be affected by the proposed Production Operations activities.
- The email included information on the regulatory process for the activities in Commonwealth and NT jurisdictions and links to a Santos information booklet on the proposed activities and a NOPSEMA brochure on the consultation process and details of how to contact Santos to register as a Relevant Person. The email stated that the consultation phase would commence on 11 March 2024 and close on 9 April 2024.
- On 11 March 2024 Santos emailed West Arnhem Regional Council further to the previous correspondence, to advise that it had commenced the consultation phase which would run until 9 April 2024. In addition to the previous information again being provided, Santos provided information on Relevant Persons' entitlements under the regulatory processes, details of how to provide feedback and a reminder of the closing date for consultation. [Con-3793]
- On 3 April 2024 Santos phoned West Arnhem Regional Council regarding consultation for Production Operations activities and spoke to a team member who asked for previously sent emails to be resent. Santos re-sent the emails the same day. [Con-4148]
- On 7 May 2024, Santos emailed West Arnhem Regional Council further to previous correspondence to advise it had extended the consultation period until 21 May 2024. In providing this extension of time, Santos advised that the information in the booklet and factsheet had been updated to account for an additional risk associated with the proposed activity. Santos confirmed that consultation will close on the revised date unless Santos hears otherwise from West Arnhem Regional Council. [Con-4224]
- On 10 July 2024 Santos emailed West Arnhem Regional Council to advise the consultation period had been completed. Santos advised West Arnhem Regional Council that it considered that consultation had now closed for the purpose of Santos finalising and submitting environment plans for these activities to government regulators for assessment. [Con-5130]
- On 15 October 2024 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that the EP had been submitted to NOPSEMA for assessment, provided a link to the full EP on the NOPSEMA website and Santos' contact details for any queries. The Update also stated that the OEMP would be submitted to the NT-DITT for assessment. [Con- 5982]
- On 23 January 2025 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that, following a request from NOPSEMA to provide further information, Santos had re-submitted the EP with some amendments in December 2024 for further assessment. The Update also re-stated that the OEMP would be submitted to the NT-DITT for assessment. [Con-6036]

Notwithstanding the consultation information provided and the steps described above, no comments or input were received on the Coastal Waters OEMP from West Arnhem Regional Council.

	· ·		
	Santos considers Section 25 consultation requirements to have been met.		
	time for consultation.		
No response was received from West Arnhem Regional Council.	Santos considers it has provided sufficient information and a reasonable period of	No response required.	Not an
Summary of response by Relevant Person	Assessment of merits	Santos' Response Statement	OEMP

West Daly Regional Council

Summary of consultation effort:

- On 9 February 2024 Santos emailed West Daly Regional Council to advise it of preliminary consultation regarding proposed activities for consultation to be managed under the Coastal Waters OEMP and Barossa Production Operations EP. [Con-3787]
- The email advised that Santos was seeking information to better understand:
  - o if you are from a government Department or agency, how the proposed Production Operations activities may be relevant to your department or agency; and
  - what (if any) functions, interests or activities you or your organisation have that may be affected by the proposed Production Operations activities. 0
- The email included information on the regulatory process for the activities in Commonwealth and NT jurisdictions and links to a Santos information booklet on the proposed activities and a NOPSEMA brochure on the consultation process and details of how to contact Santos to register as a Relevant Person. The email stated that the consultation phase would commence on 11 March 2024 and close on 9 April 2024.
- On 11 March 2024 Santos emailed West Daly Regional Council further to the previous correspondence, to advise that it had commenced the consultation phase which would run until 9 April 2024. In addition to the previous information again being provided, Santos provided information on Relevant Persons' entitlements under the regulatory processes, details of how to provide feedback and a reminder of the closing date for consultation. [Con-3793]
- On 3 April 2024 Santos phoned West Daly Regional Council and left a voice mail message regarding consultation for Production Operations activities.
- On 2 May 2024, Santos emailed West Daly Regional Council further to previous correspondence to advise it had extended the consultation period until 16 May 2024. In providing this extension of time. Santos advised that the information in the booklet and factsheet had ٠ been updated to account for an additional risk associated with the proposed activity. Santos confirmed that consultation will close on the revised date unless Santos hears otherwise from West Daly Regional Council. [Con-4154]
- On 10 July 2024 Santos emailed West Daly Regional Council to advise the consultation period had been completed. Santos advised West Daly Regional Council that it considered consultation had now closed for the purpose of Santos finalising and submitting environment plans for these activities to government regulators for assessment. [Con-5130]
- On 15 October 2024 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that the EP had been submitted to NOPSEMA for assessment, provided a link to the full EP on the NOPSEMA website and Santos' contact details for any queries. The Update also stated that the OEMP would be submitted to the NT-DITT for assessment. [Con- 5982]
- On 23 January 2025 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that, following a request from NOPSEMA to provide further information. Santos had re-submitted the EP with some amendments in December 2024 for further assessment. The Update also re-stated that the OEMP would be submitted to the NT-DITT for assessment. [Con-6036]

Notwithstanding the consultation information provided and the steps described above, no comments or input were received on the Coastal Waters OEMP from West Daly Regional Council.

Summary of response by Relevant Person	Assessment of merits	Santos' Response Statement	OEMP
No response was received from West Daly Regional Council.	Santos considers it has provided sufficient information and a reasonable period of time for consultation.	No response required.	Not app
	Santos considers Section 25 consultation requirements to have been met.		

# Santos

# reference

plicable.

reference

licable

#### 4.8.12 **Tourism Operators**

# Table 4-22: Consultation Summary Table – Tourism Operators

Section 25(1)(d) of the OPGGS(E)R: Persons or organisations whose functions, interests or activities may be affected by the activities to be carried out under the environment plan

### Alure Fishing Charters

Summary of consultation effort:

- On 9 February 2024 Santos emailed Alure Fishing Charters to advise the start of preliminary consultation regarding proposed activities for consultation to be managed under the Coastal Waters OEMP and Baross
- The email advised that Santos was seeking information to better understand:
  - o if you are from a government Department or agency, how the proposed Production Operations activities may be relevant to your department or agency; and
  - what (if any) functions, interests or activities you or your organisation have that may be affected by the proposed Production Operations activities. 0
- The email included information on the regulatory process for the activities in Commonwealth and NT jurisdictions and links to a Santos information booklet on the proposed activities and a NOPSEMA brochure on the consultation process and details of how to contact Santos to register as a Relevant Person. The email stated that the consultation phase would commence on 11 March 2024 and close on 9 April 2024.
- On 11 March 2024 Santos emailed NT tourism operators further to the previous correspondence, to advise that it had commenced the consultation phase which would run until 9 April 2024. In addition to the previous information again being provided, Santos provided • information on Relevant Persons' entitlements under the regulatory processes, details of how to provide feedback and a reminder of the closing date for consultation. [Con-3793]
- On 3 April 2024 Santos phoned Alure Fishing Charters and left a voice mail message.
- On 2 May 2024, Santos emailed Alure Fishing Charters further to the previous correspondence, to advise it had extended the consultation period until 16 May 2024. In providing this extension of time, Santos advised that the information in the booklet and factsheet had been updated to account for an additional risk associated with the proposed activity. Santos confirmed that consultation will close on the revised date unless Santos hears otherwise from Alure Fishing Charters. [Con-4282]
- On 10 July 2024 Santos emailed Alure Fishing Charters to advise the consultation period had been completed. Santos advised Alure Fishing Charters that it considered consultation had now closed for the purpose of Santos finalising and submitting environment plans for • these activities to government regulators for assessment. [Con-5130]
- On 15 October 2024 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that the EP had been submitted to NOPSEMA for assessment, provided a link to the full EP on the NOPSEMA website and Santos' contact details for any queries. The Update also stated that the OEMP would be submitted to the NT-DITT for assessment. [Con- 5982]
- On 23 January 2025 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that, following a request from NOPSEMA to provide further information, Santos had re-submitted the EP with some amendments in December 2024 for further assessment. The Update also re-stated that the OEMP would be submitted to the NT-DITT for assessment. [Con-6036]
- Notwithstanding the information provided and the steps described above, no comments or input were received on the Coastal Waters OEMP from Alure Fishing Charters.

Summary of response by Relevant Person	Assessment of merits	Santos' Response Statement	OEMP refere
No response was received from Alure Fishing Charters.	Santos considers it has provided sufficient information and a reasonable period of time for consultation. Santos considers Section 25 consultation requirements to have been met.	No response required.	Not applicable



sa	Production	Operations	EP. [Con-3787]	

nce

## Anglers Advantage Fishing Charters

Summary of consultation effort:

- On 9 February 2024 Santos emailed Anglers Advantage Fishing Charters to advise the start of preliminary consultation regarding proposed activities for consultation to be managed under the Coastal Waters OEMP and Barossa Production Operations EP. [Con-3787]
- The email advised that Santos was seeking information to better understand:
  - o if you are from a government Department or agency, how the proposed Production Operations activities may be relevant to your department or agency; and
  - what (if any) functions, interests or activities you or your organisation have that may be affected by the proposed Production Operations activities.
- The email included information on the regulatory process for the activities in Commonwealth and NT jurisdictions and links to a Santos information booklet on the proposed activities and a NOPSEMA brochure on the consultation process and details of how to contact Santos to register as a Relevant Person. The email stated that the consultation phase would commence on 11 March 2024 and close on 9 April 2024.
- On 11 March 2024 Santos emailed Anglers Advantage Fishing Charters further to the previous correspondence, to advise that it had commenced the consultation phase which would run until 9 April 2024. In addition to the previous information again being provided, Santos provided information on Relevant Persons' entitlements under the regulatory processes, details of how to provide feedback and a reminder of the closing date for consultation. [Con-3793]
- On 3 April 2024 Santos phoned Angler's Advantage Fishing Charters and left a voice mail message.
- On 2 May 2024, Santos emailed Angler's Advantage Fishing Charters to advise it had phoned to follow up on previous emails sent and to advise it had extended the consultation period until 16 May 2024. In providing this extension of time, Santos advised that the information in the booklet and factsheet had been updated to account for an additional risk associated with the proposed activity. Santos confirmed that consultation will close on the revised date unless Santos hears otherwise from Angler's Advantage Fishing Charters. [Con-4283]
- On 10 July 2024 Santos emailed Angler's Advantage Fishing Charters to advise the consultation period had been completed. Santos advised Angler's Advantage Fishing Charters that it considered consultation had now closed for the purpose of Santos finalising and submitting environment plans for these activities to government regulators for assessment. [Con-5130]
- On 15 October 2024 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that the EP had been submitted to NOPSEMA for assessment, provided a link to the full EP on the NOPSEMA website and Santos' contact details for any queries. The Update also stated that the OEMP would be submitted to the NT-DITT for assessment. [Con- 5982]
- On 23 January 2025 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that, following a request from NOPSEMA to provide further information, Santos had re-submitted the EP with some amendments in December 2024 for further assessment. The Update also re-stated that the OEMP would be submitted to the NT-DITT for assessment. [Con-6036]
- Notwithstanding the information provided and the steps described above, no comments or input were received on the Coastal Waters OEMP from Anglers Advantage Fishing Charters.

Summary of response by Relevant Person	Assessment of merits	Santos' Response Statement	OEMP refere
No response was received from Anglers Advantage Fishing Charters.	Santos considers it has provided sufficient information and a reasonable period of time for consultation. Santos considers Section 25 consultation requirements to have been met.	No response required.	Not applicabl

# Angler's Choice Fishing Safaris

Summary of consultation effort:

- On 9 February 2024 Santos emailed Angler's Choice Fishing Safaris to advise the start of preliminary consultation regarding proposed activities for consultation to be managed under the Coastal Waters OEMP and Barossa Production Operations EP. [Con-3787]
- The email advised that Santos was seeking information to better understand:
  - o if you are from a government Department or agency, how the proposed Production Operations activities may be relevant to your department or agency; and
  - what (if any) functions, interests or activities you or your organisation have that may be affected by the proposed Production Operations activities.
- The email included information on the regulatory process for the activities in Commonwealth and NT jurisdictions and links to a Santos information booklet on the proposed activities and a NOPSEMA brochure on the consultation process and details of how to contact Santos to register as a Relevant Person. The email stated that the consultation phase would commence on 11 March 2024 and close on 9 April 2024.
- On 11 March 2024 Santos emailed Angler's Choice Fishing Safaris further to the previous correspondence, to advise that it had commenced the consultation phase which would run until 9 April 2024. In addition to the previous information again being provided, Santos provided information on Relevant Persons' entitlements under the regulatory processes, details of how to provide feedback and a reminder of the closing date for consultation. [Con-3793]
- On 3 April 2024 Santos phoned Angler's Choice Fishing Safaris and left a voice mail message.
- On 2 May 2024, Santos emailed Angler's Choice Fishing Safaris to advise it had phoned to follow up on previous emails sent and to advise it had extended the consultation period until 16 May 2024. In providing this extension of time, Santos advised that the information in the booklet and factsheet had been updated to account for an additional risk associated with the proposed activity. Santos confirmed that consultation will close on the revised date unless Santos hears otherwise from Angler's Choice Fishing Safaris. [Con-4284]
- On 10 July 2024 Santos emailed Angler's Choice Fishing Safaris to advise the consultation period had been completed. Santos advised Angler's Choice Fishing Safaris that it considered consultation had now closed for the purpose of Santos finalising and submitting environment plans for these activities to government regulators for assessment. [Con-5130]
- On 15 October 2024 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that the EP had been submitted to NOPSEMA for assessment, provided a link to the full EP on the NOPSEMA website and Santos' contact details for any queries. The Update also stated that the OEMP would be submitted to the NT-DITT for assessment. [Con- 5982]
- On 23 January 2025 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that, following a request from NOPSEMA to provide further information, Santos had re-submitted the EP with some amendments in December 2024 for further assessment. The Update also re-stated that the OEMP would be submitted to the NT-DITT for assessment. [Con-6036]
- Notwithstanding the information provided and the steps described above, no comments or input were received on the Coastal Waters OEMP from Anglers Choice Fishing Safaris.

Summary of response by Relevant Person	Assessment of merits	Santos' Response Statement	OEMP refere
No response was received from Anglers Choice Fishing Safaris.	Santos considers it has provided sufficient information and a reasonable period of time for consultation. Santos considers Section 25 consultation requirements to have been met.	No response required.	Not applicable
Arafura Bluewater Charters			

# Santos

nce

nce

- On 9 February 2024 Santos emailed Arafura Bluewater Charters to advise the start of preliminary consultation regarding proposed activities for consultation to be managed under the Coastal Waters OEMP and Barossa Production Operations EP. [Con-3787]
- The email advised that Santos was seeking information to better understand:
  - o if you are from a government Department or agency, how the proposed Production Operations activities may be relevant to your department or agency; and
  - what (if any) functions, interests or activities you or your organisation have that may be affected by the proposed Production Operations activities.
- The email included information on the regulatory process for the activities in Commonwealth and NT jurisdictions and links to a Santos information booklet on the proposed activities and a NOPSEMA brochure on the consultation process and details of how to contact Santos to register as a Relevant Person. The email stated that the consultation phase would commence on 11 March 2024 and close on 9 April 2024.
- On 11 March 2024 Santos emailed Arafura Bluewater Charters further to the previous correspondence, to advise that it had commenced the consultation phase which would run until 9 April 2024. In addition to the previous information again being provided, Santos provided information on Relevant Persons' entitlements under the regulatory processes, details of how to provide feedback and a reminder of the closing date for consultation. [Con-3793]
- On 4 April 2024 Santos phoned Arafura Bluewater Charters, spoke to a company representative and reminded them of the deadline for comments on the EP and OEMP.
- On 6 May 2024, Santos emailed Arafura Bluewater Charters, to advise it had extended the consultation period until 20 May 2024. In providing this extension of time, Santos advised that the information in the booklet and factsheet had been updated to account for an additional risk associated with the proposed activity. Santos confirmed that consultation will close on the revised date unless Santos hears otherwise from Arafura Charters. [Con-4311]
- On 10 July 2024 Santos emailed Arafura Bluewater Charters to advise the consultation period had been completed. Santos advised Arafura Bluewater Charters that it considered consultation had now closed for the purpose of Santos finalising and submitting environment. plans for these activities to government regulators for assessment. [Con-5130]
- On 15 October 2024 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that the EP had been submitted to NOPSEMA for assessment, provided a link to the full EP on the NOPSEMA website and Santos' contact details for any queries. The Update also stated that the OEMP would be submitted to the NT-DITT for assessment. [Con- 5982]
- On 23 January 2025 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that, following a request from NOPSEMA to provide further information. Santos had re-submitted the EP with some amendments in December 2024 for further assessment. The Update also re-stated that the OEMP would be submitted to the NT-DITT for assessment. [Con-6036]
- Notwithstanding the information provided and the steps described above, no comments or input were received on the Coastal Waters OEMP from Arafura Bluewater Charters.

Summary of response by Relevant Person	Assessment of merits	Santos' Response Statement	OEMP refer
No response was received from Arafura Bluewater Charters.	Santos considers it has provided sufficient information and a reasonable period of time for consultation. Santos considers Section 25 consultation requirements to have been met.	No response required.	Not applicab

### Arnhem Land Safaris

Summary of consultation effort:

- On 10 June 2024 Santos emailed Arnhem Land Safaris regarding proposed activities for consultation to be managed under the Coastal Waters OEMP and Barossa Production Operations EP and was seeking to consult with Relevant Persons. Santos asked Arnhem Land Safaris to advise Santos by 17 June 2024 if it considered it may be a Relevant Person and what functions, interests or activities it has that may be affected by the Production Operations activities. [Con-4975]
- The email included links to the Barossa Production Operations Activity Booklet and NOPSEMA's EP consultation information for the community. Santos advised if it did not hear from Arnhem Land Safaris by 17 June 2024, it would assume that it did not have functions, interests or activities that may be affected by the activities, or didn't wish to be consulted for the OEMP and EP.
- On 10 July 2024 Santos emailed Arnhem Land Safaris to advise the consultation period had been completed. Santos advised it considered that consultation had now closed for the purpose of Santos finalising and submitting environment plans for these activities to government regulators for assessment. [Con-5130]
- On 15 October 2024 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that the EP had been submitted to NOPSEMA for assessment, provided a link to the full EP on the NOPSEMA website and Santos' contact details for any queries. The Update also stated that the OEMP would be submitted to the NT-DITT for assessment. [Con- 5982]
- On 23 January 2025 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that, following a request from NOPSEMA to provide further information, Santos had re-submitted the EP with some amendments in December 2024 for further assessment. The Update also re-stated that the OEMP would be submitted to the NT-DITT for assessment. [Con-6036]

### No further correspondence or feedback was received from Arnhem Land Safaris.

Summary of response by Relevant Person	Assessment of merits	Santos' Response Statement	OEMP refer
No response was received from Arnhem Land Safaris.	Santos considers it has provided sufficient information and a reasonable beriod of time for consultation.	No response required.	Not applicab
	Santos notes that in June 2024, in response to consultation via email and phone on the Darwin Pipeline Duplication EP (Cwth) and DPD Offshore CEMP, Arnhem Land Safaris advised that it did not consider that the activities were relevant to its operations, as it operated on land and inland waters 300km east of Darwin.		
	Santos considers Section 25 consultation requirements to have been met.		

# **Barra Fishing Charters**

Summary of consultation effort:

- On 9 February 2024 Santos emailed Barra Fishing Charters to advise the start of preliminary consultation regarding proposed activities for consultation to be managed under the Coastal Waters OEMP and Barossa Production Operations EP. [Con-3787]
- The email advised that Santos was seeking information to better understand:
  - if you are from a government Department or agency, how the proposed Production Operations activities may be relevant to your department or agency; and 0
  - what (if any) functions, interests or activities you or your organisation have that may be affected by the proposed Production Operations activities. 0

# Santos

ence

le

ence

le l

- The email included information on the regulatory process for the activities in Commonwealth and NT jurisdictions and links to a Santos information booklet on the proposed activities and a NOPSEMA brochure on the consultation process and details of how to contact Santos to register as a Relevant Person. The email stated that the consultation phase would commence on 11 March 2024 and close on 9 April 2024.
- On 11 March 2024 Santos emailed Barra Fishing Charters further to the previous correspondence, to advise that it had commenced the consultation phase which would run until 9 April 2024. In addition to the previous information again being provided, Santos provided information on Relevant Persons' entitlements under the regulatory processes, details of how to provide feedback and a reminder of the closing date for consultation. [Con-3793]
- On 3 April 2024 Santos phoned Barra Fishing Charters and left a voice mail message.
- On 2 May 2024, Santos emailed Barra Fishing Charters to advise it was calling to follow up on previous emails and to advise it had extended the consultation period until 16 May 2024. In providing this extension of time, Santos advised that the information in the booklet and factsheet had been updated to account for an additional risk associated with the proposed activity. Santos confirmed that consultation will close on the revised date unless Santos hears otherwise from Barra Fishing Charters. [Con-4285]
- On 10 July 2024 Santos emailed Barra Fishing Charters to advise the consultation period had been completed. Santos advised Barra Fishing Charters that it considered consultation had now closed for the purpose of Santos finalising and submitting environment plans for these activities to government regulators for assessment. [Con-5130]
- On 15 October 2024 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that the EP had been submitted to NOPSEMA for assessment, provided a link to the full EP on the NOPSEMA website and Santos' contact details for any queries. The Update also stated that the OEMP would be submitted to the NT-DITT for assessment. [Con- 5982]
- On 23 January 2025 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that, following a request from NOPSEMA to provide further information, Santos had re-submitted the EP with some amendments in December 2024 for further assessment. The Update also re-stated that the OEMP would be submitted to the NT-DITT for assessment. [Con-6036]
- Notwithstanding the information provided and the steps described above, no comments or input were received on the Coastal Waters OEMP from Barra Fishing Charters.

No response was received from Barra Fishing Charters.	Santos considers it has provided sufficient information and a reasonable period of time for consultation.	No response required.	Not applicabl
	Santos considers Section 25 consultation requirements to have been met.		
Persilar Mada			

### Bavview Marina

Summary of consultation effort:

- On 9 February 2024 Santos emailed NT tourism operators to advise the start of preliminary consultation regarding proposed activities for consultation to be managed under the Coastal Waters OEMP and Barossa Production Operations EP. [Con-3787]
- The email advised that Santos was seeking information to better understand: •
  - o if you are from a government Department or agency, how the proposed Production Operations activities may be relevant to your department or agency; and
  - what (if any) functions, interests or activities you or your organisation have that may be affected by the proposed Production Operations activities.
- The email included information on the regulatory process for the activities in Commonwealth and NT jurisdictions and links to a Santos information booklet on the proposed activities and a NOPSEMA brochure on the consultation process and details of how to contact Santos to register as a Relevant Person. The email stated that the consultation phase would commence on 11 March 2024 and close on 9 April 2024.
- On 11 March 2024 Santos emailed NT tourism operators further to the previous correspondence, to advise that it had commenced the consultation phase which would run until 9 April 2024. In addition to the previous information again being provided, Santos provided information on Relevant Persons' entitlements under the regulatory processes, details of how to provide feedback and a reminder of the closing date for consultation. [Con-3793]
- On 3 April 2024 Santos phoned Bayview Marina regarding Production Operations activities which advised it had no feedback and did not believe they would be impacted.
- On 6 May 2024 Santos emailed Bayview Marina, and to advise it had extended the consultation period until 20 May 2024. In providing this extension of time, Santos advised that the information in the booklet and factsheet had been updated to account for an additional risk associated with the proposed activity. Santos confirmed that consultation will close on the revised date unless Santos hears otherwise from Bayview Marina. [Con-4317]
- On 10 July 2024 Santos emailed Bayview Marina to advise the consultation period had been completed. Santos advised it considered that consultation had now closed for the purpose of Santos finalising and submitting environment plans for these activities to government regulators for assessment. [Con-5130]
- On 15 October 2024 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that the EP had been submitted to NOPSEMA for assessment, provided a link to the full EP on the NOPSEMA website and Santos' contact details for any queries. The Update also stated that the OEMP would be submitted to the NT-DITT for assessment. [Con- 5982]
- On 23 January 2025 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that, following a request from NOPSEMA to provide further information, Santos had re-submitted the EP with some amendments in December 2024 for further assessment. The Update also re-stated that the OEMP would be submitted to the NT-DITT for assessment. [Con-6036]
- Notwithstanding the information provided and the steps described above, no comments or input were received on the Coastal Waters OEMP from Bayview Marina.

Summary of response by Relevant Person	Assessment of merits	Santos' Response Statement	OEMP refere
No response was received from Bayview Marina.	Santos considers it has provided sufficient information and a reasonable period of time for consultation. Santos considers Section 25 consultation requirements to have been met.	No response required.	Not applicabl

### **Buffalo Boat Hire**

Summary of consultation effort:

- On 10 June 2024 Santos emailed Buffalo Boat Hire regarding proposed activities for consultation to be managed under the Coastal Waters OEMP and Barossa Production Operations EP and was seeking to consult with relevant persons. Santos asked Buffalo Boat Hire to advise Santos by 17 June 2024 if it considered it may be a Relevant Person and what functions, interests or activities it has that may be affected by the production operations activities. [Con-4976]
- The email included links to the Barossa Production Operations Activity Booklet and NOPSEMA's EP consultation information for the community. Santos advised if it did not hear from Buffalo Boat Hire by 17 June 2024, it would assume that it did not have functions, interests or activities that may be affected by the activities, or didn't wish to be consulted for the OEMP and EP.
- On 10 July 2024 Santos emailed Buffalo Boat Hire to advise the consultation period had been completed. Santos advised Buffalo Boat Hire that it considered consultation had now closed for the purpose of Santos finalising and submitting environment plans for these activities to government regulators for assessment. [Con-5130]
- On 15 October 2024 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that the EP had been submitted to NOPSEMA for assessment, provided a link to the full EP on the NOPSEMA website and Santos' contact details for any queries. The Update also stated that the OEMP would be submitted to the NT-DITT for assessment. [Con- 5982]

nce

le.

nce

- On 23 January 2025 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that, following a request from NOPSEMA to provide further information, Santos had re-submitted the EP with some amendments in December 2024 for further assessment. The Update also re-stated that the OEMP would be submitted to the NT-DITT for assessment. [Con-6036]
- Notwithstanding the consultation information provided and the steps described above, no comments or input were received on the Coastal Waters OEMP from Buffalo Boat Hire. •
- Notwithstanding the information provided and the steps described above, no comments or input were received on the Coastal Waters OEMP from Buffalo Boat Hire.

Summary of response by Relevant Person	Assessment of merits	Santos' Response Statement	OEMP refere
No response was received from Buffalo Boat Hire.	Santos considers it has provided sufficient information and a reasonable period of time for consultation. Santos notes that in June 2024, in response to consultation by email and phone on the Darwin Pipeline Duplication EP (Cwth) and DPD Offshore CEMP, Buffalo Boat Hire advised that it did not conduct tours that far from Darwin and had provided no comments. Santos considers Section 25 consultation requirements to have been met.	No response required.	Not applicab

- On 9 February 2024 Santos emailed NT tourism operators to advise the start of preliminary consultation regarding proposed activities for consultation to be managed under the Coastal Waters OEMP and Barossa
- The email advised that Santos was seeking information to better understand:
  - o if you are from a government Department or agency, how the proposed Production Operations activities may be relevant to your department or agency: and
  - what (if any) functions, interests or activities you or your organisation have that may be affected by the proposed Production Operations activities. 0
- The email included information on the regulatory process for the activities in Commonwealth and NT jurisdictions and links to a Santos information booklet on the proposed activities and a NOPSEMA brochure on the consultation process and details of how to contact Santos to register as a Relevant Person. The email stated that the consultation phase would commence on 11 March 2024 and close on 9 April 2024.
- On 11 March 2024 Santos emailed NT tourism operators further to the previous correspondence, to advise that it had commenced the consultation phase which would run until 9 April 2024. In addition to the previous information again being provided, Santos provided information on Relevant Persons' entitlements under the regulatory processes, details of how to provide feedback and a reminder of the closing date for consultation. [Con-3793]
- On 3 April 2024 Santos phoned and left a voice mail message.
- On 2 May 2024, Santos emailed Clearwater Island Lodge to follow up on previous emails and to advise it had extended the consultation period until 16 May 2024. In providing this extension of time, Santos advised that the information in the booklet and factsheet had been updated to account for an additional risk associated with the proposed activity. Santos confirmed that consultation will close on the revised date unless Santos hears otherwise from Clearwater Island Lodge. [Con-4286]
- On 10 July 2024 Santos emailed Clearwater Island Lodge to advise the consultation period had been completed. Santos advised Clearwater Island Lodge that it considered consultation had now closed for the purpose of Santos finalising and submitting environment plans for these activities to government regulators for assessment. [Con-5130]
- On 15 October 2024 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that the EP had been submitted to NOPSEMA for assessment, provided a link to the full EP on the NOPSEMA website and Santos' contact details for any queries. The Update also stated that the OEMP would be submitted to the NT-DITT for assessment. [Con- 5982]
- On 23 January 2025 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that, following a request from NOPSEMA to provide further information, Santos had re-submitted the EP with some amendments in December 2024 for further assessment. The Update also re-stated that the OEMP would be submitted to the NT-DITT for assessment. [Con-6036]

### Notwithstanding the consultation information provided and the steps described above, no comments or input were received on the Coastal Waters OEMP from Clearwater Island Lodge

Summary of response by Relevant Person	Assessment of merits	Santos' Response Statement	• OEMP ret
No response was received from Clearwater Island Lodge.	Santos considers it has provided sufficient information and a reasonable period of time for consultation. Santos considers Section 25 consultation requirements to have been met.	No response required.	Not applicab



ence

le.

a Production Op	erations EP.	[Con-3787]
-----------------	--------------	------------

### erence

# **Cobourg Fishing Safaris/Venture North**

Summary of consultation effort:

- On 10 June 2024 Santos emailed Cobourg Fishing Safaris regarding proposed activities for consultation to be managed under the Coastal Waters OEMP and Barossa Production Operations EP and was seeking to consult with Relevant Persons. Santos asked Cobourg Fishing Safaris to advise Santos by 17 June 2024 if it considered it may be a Relevant Person and what functions, interests or activities it has that may be affected by the production operations activities. [Con-4977]
- The email included links to the Barossa Production Operations Activity Booklet and NOPSEMA's EP consultation information for the community. Santos advised if it did not hear from Cobourg Fishing Safaris by 17 June 2024, it would assume that it did not have functions, interests or activities that may be affected by the activities, or didn't wish to be consulted for the OEMP and EP.
- On 2 July 2024 Santos phoned Cobourg Fishing Safaris and left a message with office staff who advised the owner would call back if they had any comments.
- On 10 July 2024 Santos emailed Cobourg Fishing Safaris to advise the consultation period had been completed. Santos advised it considered that consultation had now closed for the purpose of Santos finalising and submitting environment plans for these activities to government regulators for assessment. [Con-5130]
- On 15 October 2024 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that the EP had been submitted to NOPSEMA for assessment, provided a link to the full EP on the NOPSEMA website and Santos' contact details for any queries. The Update also stated that the OEMP would be submitted to the NT-DITT for assessment. [Con- 5982]
- On 23 January 2025 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that, following a request from NOPSEMA to provide further information. Santos had re-submitted the EP with some amendments in December 2024 for further assessment. The Update also re-stated that the OEMP would be submitted to the NT-DITT for assessment. [Con-6036]
- Notwithstanding the consultation information provided and the steps described above, no comments or input were received on the Coastal Waters OEMP. from Cobourg Fishing Safaris.

Summary of response by Relevant Person	Assessment of merits	Santos' Response Statement	OEMP refere
No response was received from Cobourg Fishing Safaris.	Santos considers it has provided sufficient information and a reasonable period of time for consultation. Santos notes that in June 2024, in response to consultation via email and phone on the Darwin Pipeline Duplication EP (Cwth) and DPD Offshore CEMP, Cobourg Fishing Safaris/Venture North had provided no comments. Santos considers Section 25 consultation requirements to have been met.	No response required.	Not applicabl
Crab Claw Island Posort			

Summary of consultation effort:

On 9 February 2024 Santos emailed NT tourism operators to advise the start of preliminary consultation regarding proposed activities for consultation to be managed under the Coastal Waters OEMP and Barossa Production Operations EP. [Con-3787]

- The email advised that Santos was seeking information to better understand:
  - o if you are from a government Department or agency, how the proposed Production Operations activities may be relevant to your department or agency; and
  - what (if any) functions, interests or activities you or your organisation have that may be affected by the proposed Production Operations activities.
- The email included information on the regulatory process for the activities in Commonwealth and NT jurisdictions and links to a Santos information booklet on the proposed activities and a NOPSEMA brochure on the consultation process and details of how to contact Santos to register as a Relevant Person. The email stated that the consultation phase would commence on 11 March 2024 and close on 9 April 2024.
- On 11 March 2024 Santos emailed NT tourism operators further to the previous correspondence, to advise that it had commenced the consultation phase which would run until 9 April 2024. In addition to the previous information again being provided, Santos provided information on Relevant Persons' entitlements under the regulatory processes, details of how to provide feedback and a reminder of the closing date for consultation. [Con-3793]
- On 4 April 2024 Santos phoned Crab Claw Island Resort and spoke to a company representative who asked Santos to call back the following day.
- On 5 April 2024 Santos phoned Crab Claw Island Resort and left a voice mail message.
- On 2 May 2024, Santos emailed Crab Claw Island Resort further to previous correspondence, to advise it had extended the consultation period until 16 May 2024. In providing this extension of time, Santos advised that the information in the booklet and factsheet had been updated to account for an additional risk associated with the proposed activity. Santos confirmed that consultation will close on the revised date unless Santos hears otherwise from Crab Claw Island Resort. [Con-4301]
- On 10 July 2024 Santos emailed Crab Claw Resort to advise the consultation period had been completed. Santos advised Crab Claw Resort that it considered consultation had now closed for the purpose of Santos finalising and submitting environment plans for these activities to government regulators for assessment. [Con-5130]
- On 15 October 2024 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that the EP had been submitted to NOPSEMA for assessment, provided a link to the full EP on the NOPSEMA website and Santos' contact details for any queries. The Update also stated that the OEMP would be submitted to the NT-DITT for assessment. [Con- 5982]
- On 23 January 2025 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that, following a request from NOPSEMA to provide further information, Santos had re-submitted the EP with some amendments in December 2024 for further assessment. The Update also re-stated that the OEMP would be submitted to the NT-DITT for assessment. [Con-6036]
- Notwithstanding the consultation information provided and the steps described above, no comments or input were received on the Coastal Waters OEMP from Crab Claw Island Resort.

Summary of response by Relevant Person	Assessment of merits	Santos' Response Statement	OEMP refere
No response was received from Crab Claw Island Resort.	Santos considers it has provided sufficient information and a reasonable period of time for consultation. Santos considers Section 25 consultation requirements to have been met.	No response required.	Not applicab
Cullen Bay Fishing Charters			

Summary of consultation effort:

On 9 February 2024 Santos emailed Cullen Bay Fishing Charters to advise the start of preliminary consultation regarding proposed activities for consultation to be managed under the Coastal Waters OEMP and Barossa Production Operations EP. [Con-3787]

# Santos

ence

le.

- The email advised that Santos was seeking information to better understand:
  - o if you are from a government Department or agency, how the proposed Production Operations activities may be relevant to your department or agency; and
  - what (if any) functions, interests or activities you or your organisation have that may be affected by the proposed Production Operations activities.
- The email included information on the regulatory process for the activities in Commonwealth and NT jurisdictions and links to a Santos information booklet on the proposed activities and a NOPSEMA brochure on the consultation process and details of how to contact • Santos to register as a Relevant Person. The email stated that the consultation phase would commence on 11 March 2024 and close on 9 April 2024.
- On 11 March 2024 Santos emailed Cullen Bay Fishing Charters further to the previous correspondence, to advise that it had commenced the consultation phase which would run until 9 April 2024. In addition to the previous information again being provided, Santos provided information on Relevant Persons' entitlements under the regulatory processes, details of how to provide feedback and a reminder of the closing date for consultation. [Con-3793]
- On 4 April 2024 Santos phoned Cullen Bay Fishing Charters and was unable to leave a message.
- On 2 May 2024, Santos emailed Cullen Bay Fishing Charters further to previous correspondence, to advise it had extended the consultation period until 16 May 2024. In providing this extension of time, Santos advised that the information in the booklet and factsheet had been updated to account for an additional risk associated with the proposed activity. Santos confirmed that consultation will close on the revised date unless Santos hears otherwise from Cullen Bay Fishing Charters. [Con-4278]
- On 10 July 2024 Santos emailed Cullen Bay Fishing Charters to advise the consultation period had been completed. Santos advised Cullen Bay Fishing Charters that it considered consultation had now closed for the purpose of Santos finalising and submitting environment plans for these activities to government regulators for assessment. [Con-5130]
- On 15 October 2024 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that the EP had been submitted to NOPSEMA for assessment, provided a link to the full EP on the NOPSEMA website and Santos' contact details for any queries. The Update also stated that the OEMP would be submitted to the NT-DITT for assessment. [Con- 5982]
- On 23 January 2025 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that, following a request from NOPSEMA to provide further information, Santos had re-submitted the EP with some amendments in December 2024 for further assessment. The Update also re-stated that the OEMP would be submitted to the NT-DITT for assessment. [Con-6036]

### Notwithstanding the consultation information provided and the steps described above, no comments or input were received on the Coastal Waters OEMP from Cullen Bay Fishing Charters.

Summary of response by Relevant Person	Assessment of merits	Santos' Response Statement	<b>OEMP</b> refere
No response was received from Cullen Bay Fishing Charters.	Santos considers it has provided sufficient information and a reasonable period of time for consultation. Santos considers Section 25 consultation requirements to have been met.	No response required.	Not applicable



nce

# **Cullen Bay Marina**

Summary of consultation effort:

- On 9 February 2024 Santos emailed Cullen Bay Marina to advise the start of preliminary consultation regarding proposed activities for consultation to be managed under the Coastal Waters OEMP and Barossa Production Operations EP. [Con-3787]
- The email advised that Santos was seeking information to better understand:
  - o if you are from a government Department or agency, how the proposed Production Operations activities may be relevant to your department or agency; and
  - what (if any) functions, interests or activities you or your organisation have that may be affected by the proposed Production Operations activities.
- The email included information on the regulatory process for the activities in Commonwealth and NT jurisdictions and links to a Santos information booklet on the proposed activities and a NOPSEMA brochure on the consultation process and details of how to contact Santos to register as a Relevant Person. The email stated that the consultation phase would commence on 11 March 2024 and close on 9 April 2024.
- On 11 March 2024 Santos emailed Cullen Bay Marina further to the previous correspondence, to advise that it had commenced the consultation phase which would run until 9 April 2024. In addition to the previous information again being provided, Santos provided information on Relevant Persons' entitlements under the regulatory processes, details of how to provide feedback and a reminder of the closing date for consultation. [Con-3793]
- On 4 April 2024 Santos phoned Cullen Bay Marina and left a voice mail message.
- On 2 May 2024, Santos emailed Cullen Bay Marina further to previous correspondence, to advise it had extended the consultation period until 16 May 2024. In providing this extension of time, Santos advised that the information in the booklet and factsheet had been updated to account for an additional risk associated with the proposed activity. Santos confirmed that consultation will close on the revised date unless Santos hears otherwise from Cullen Bay Marina. [Con-4287]
- On 10 July 2024 Santos emailed Cullen Bay Marina to advise the consultation period had been completed. Santos advised Cullen Bay Marina that it considered consultation had now closed for the purpose of Santos finalising and submitting environment plans for these activities to government regulators for assessment. [Con-5130]
- On 15 October 2024 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that the EP had been submitted to NOPSEMA for assessment, provided a link to the full EP on the NOPSEMA website and Santos' contact details for any queries. The Update also stated that the OEMP would be submitted to the NT-DITT for assessment. [Con- 5982]
- On 23 January 2025 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that, following a request from NOPSEMA to provide further information, Santos had re-submitted the EP with some amendments in December 2024 for further assessment. The Update also re-stated that the OEMP would be submitted to the NT-DITT for assessment. [Con-6036]

### Notwithstanding the consultation information provided and the steps described above, no comments or input were received on the Coastal Waters OEMP from Cullen Bay Marina.

Summary of response by Relevant Person	Assessment of merits	Santos' Response Statement	OEMP refere
No response was received from Cullen Bay Marina.	Santos considers it has provided sufficient information and a reasonable period of time for consultation.	No response required.	Not applicabl
	Santos considers Section 25 consultation requirements to have been met.		

# Santos

nce

# **Darwin Bara Fishing Tours**

Summary of consultation effort:

- On 9 February 2024 Santos emailed Darwin Barra Fishing Tours to advise the start of preliminary consultation regarding proposed activities for consultation to be managed under the Coastal Waters OEMP and B
- The email advised that Santos was seeking information to better understand:
  - o if you are from a government Department or agency, how the proposed Production Operations activities may be relevant to your department or agency; and
  - what (if any) functions, interests or activities you or your organisation have that may be affected by the proposed Production Operations activities.
- The email included information on the regulatory process for the activities in Commonwealth and NT jurisdictions and links to a Santos information booklet on the proposed activities and a NOPSEMA brochure on the consultation process and details of how to contact Santos to register as a Relevant Person. The email stated that the consultation phase would commence on 11 March 2024 and close on 9 April 2024.
- On 11 March 2024 Santos emailed Darwin Barra Fishing Tours further to the previous correspondence, to advise that it had commenced the consultation phase which would run until 9 April 2024. In addition to the previous information again being provided, Santos provided information on Relevant Persons' entitlements under the regulatory processes, details of how to provide feedback and a reminder of the closing date for consultation. [Con-3793]
- On 4 April 2024 Santos phoned Darwin Barra Fishing Tours and spoke to a company representative who confirmed that emails sent on 9 February and 11 March had been received and that Darwin Barra Fishing Tours did not have any feedback.
- On 6 May 2024, Santos emailed Darwin Barra Fishing Tours further to previous correspondence, to advise it had extended the consultation period until 20 May 2024. In providing this extension of time, Santos advised that the information in the booklet and factsheet had been updated to account for an additional risk associated with the proposed activity. Santos confirmed that consultation will close on the revised date unless Santos hears otherwise from Darwin Barra Fishing Tours. [Con-4309].
- On 10 July 2024 Santos emailed Darwin Bara Fishing Tours to advise the consultation period had been completed. Santos advised it considered that consultation had now closed for the purpose of Santos finalising and submitting environment plans for these activities to government regulators for assessment. [Con-5130]
- On 15 October 2024 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that the EP had been submitted to NOPSEMA for assessment, provided a link to the full EP on the NOPSEMA website and Santos' contact details for any queries. The Update also stated that the OEMP would be submitted to the NT-DITT for assessment. [Con- 5982]
- On 23 January 2025 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that, following a request from NOPSEMA to provide further information, Santos had re-submitted the EP with some amendments in December 2024 for further assessment. The Update also re-stated that the OEMP would be submitted to the NT-DITT for assessment. [Con-6036]
- No further correspondence or feedback was received from Darwin Barra Fishing Tours.

Summary of response by Relevant Person	Assessment of merits	Santos' Response Statement	OEMP refere
Darwin Bara Fishing Tours responded that it had no feedback on the Production Operations EP.	Santos considers it has provided sufficient information and a reasonable period of time for consultation. Santos considers Section 25 consultation requirements to have been met.	No response required.	Not applicabl

## **Darwin Dive Academy**

Summary of consultation effort:

- On 9 February 2024 Santos emailed Darwin Dive Academy to advise the start of preliminary consultation regarding proposed activities for consultation to be managed under the Coastal Waters OEMP and Barossa Production Operations EP. [Con-3787]
- The email advised that Santos was seeking information to better understand:
  - o if you are from a government Department or agency, how the proposed Production Operations activities may be relevant to your department or agency; and
  - what (if any) functions, interests or activities you or your organisation have that may be affected by the proposed Production Operations activities.
- The email included information on the regulatory process for the activities in Commonwealth and NT jurisdictions and links to a Santos information booklet on the proposed activities and a NOPSEMA brochure on the consultation process and details of how to contact Santos to register as a Relevant Person. The email stated that the consultation phase would commence on 11 March 2024 and close on 9 April 2024.
- On 11 March 2024 Santos emailed Darwin Dive Academy further to the previous correspondence, to advise that it had commenced the consultation phase which would run until 9 April 2024. In addition to the previous information again being provided, Santos provided information on Relevant Persons' entitlements under the regulatory processes, details of how to provide feedback and a reminder of the closing date for consultation. [Con-3793]
- On 4 April 2024 Santos phoned Darwin Dive Academy but was unable to leave a message.
- On 2 May 2024, Santos emailed Darwin Dive Academy to advise it was calling to follow up on previous emails and to advise it had extended the consultation period until 16 May 2024. In providing this extension of time, Santos advised that the information in the booklet and factsheet had been updated to account for an additional risk associated with the proposed activity. Santos confirmed that consultation will close on the revised date unless Santos hears otherwise from Darwin Dive Academy. [Con-4279]
- On 10 July 2024 Santos emailed Darwin Dive Academy to advise the consultation period had been completed. Santos advised Darwin Dive Academy that it considered consultation had now closed for the purpose of Santos finalising and submitting environment plans for these activities to government regulators for assessment. [Con-5130]
- On 15 October 2024 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that the EP had been submitted to NOPSEMA for assessment, provided a link to the full EP on the NOPSEMA website and Santos' contact details for any queries. The Update also stated that the OEMP would be submitted to the NT-DITT for assessment. [Con- 5982]
- On 23 January 2025 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that, following a request from NOPSEMA to provide further information, Santos had re-submitted the EP with some amendments in December 2024 for further assessment. The Update also re-stated that the OEMP would be submitted to the NT-DITT for assessment. [Con-6036]
- No further correspondence or feedback was received from Darwin Dive Academy.

Summary of response by Relevant Person	Assessment of merits	Santos' Response Statement	OEMP refere
No response was received from Darwin Dive Academy.	Santos considers it has provided sufficient information and a reasonable period of time for consultation. Santos considers Section 25 consultation requirements to have been met.	No response required.	Not applicabl
Darwin Fish Seeker Charters			

Summary of consultation effort:

# Santos

arossa Productio	n Operations EP.	[Con-3787]

ence

le

On 9 February 2024 Santos emailed Darwin Fish Seeker Charters to advise the start of preliminary consultation regarding proposed activities for consultation to be managed under the Coastal Waters OEMP and Barossa Production Operations EP. [Con-3787]

- The email advised that Santos was seeking information to better understand:
  - o if you are from a government Department or agency, how the proposed Production Operations activities may be relevant to your department or agency; and
  - what (if any) functions, interests or activities you or your organisation have that may be affected by the proposed Production Operations activities.
- The email included information on the regulatory process for the activities in Commonwealth and NT jurisdictions and links to a Santos information booklet on the proposed activities and a NOPSEMA brochure on the consultation process and details of how to contact
  Santos to register as a Relevant Person. The email stated that the consultation phase would commence on 11 March 2024 and close on 9 April 2024.
- On 11 March 2024 Santos emailed Darwin Fish Seeker Charters further to the previous correspondence, to advise that it had commenced the consultation phase which would run until 9 April 20243. In addition to the previous information again being provided, Santos provided information on Relevant Persons' entitlements under the regulatory processes, details of how to provide feedback and a reminder of the closing date for consultation. [Con-3793]
- On 4 and 5 April 2024 Santos phoned Darwin Fish Seeker Charters and spoke to a company representative who advised that Darwin Fish Seeker Charters did not have any feedback.
- On 8 May 2024, Santos emailed Darwin Fish Seeker Charters to advise it had extended the consultation period until 22 May 2024. In providing this extension of time, Santos advised that the information in the booklet and factsheet had been updated to account for an additional risk associated with the proposed activity. Santos confirmed that consultation will close on the revised date unless Santos hears otherwise from Darwin Fish Seeker Charters. [Con-4323]
- On 10 July 2024 Santos emailed Darwin Fish Seeker Charters to advise the consultation period had been completed. Santos advised Darwin Fish Seeker Charters that it considered consultation had now closed for the purpose of Santos finalising and submitting environment plans for these activities to government regulators for assessment. [Con-5130]
- On 15 October 2024 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that the EP had been submitted to NOPSEMA for assessment, provided a link to the full EP on the NOPSEMA website and Santos' contact details for any queries. The Update also stated that the OEMP would be submitted to the NT-DITT for assessment. [Con- 5982]
- On 23 January 2025 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that, following a request from NOPSEMA to provide further information, Santos had re-submitted the EP with some amendments in December 2024 for further assessment. The Update also re-stated that the OEMP would be submitted to the NT-DITT for assessment. [Con-6036]
- No further correspondence or feedback was received from Darwin Fish Seeker Charters.

Summary of response by Relevant Person	Assessment of merits	Santos' Response Statement	<b>OEMP</b> refere
No response was received from Darwin Fish Seeker Charters.	Santos considers it has provided sufficient information and a reasonable period of time for consultation. Santos considers Section 25 consultation requirements to have been met.	No response required.	Not applicable

### Darwin Harbour Cruises

Summary of consultation effort:

- On 9 February 2024 Santos emailed Darwin Harbour Cruises to advise the start of preliminary consultation regarding proposed activities for consultation to be managed under the Coastal Waters OEMP and Barossa Production Operations EP. [Con-3787]
- The email advised that Santos was seeking information to better understand:
  - o if you are from a government Department or agency, how the proposed Production Operations activities may be relevant to your department or agency; and
  - what (if any) functions, interests or activities you or your organisation have that may be affected by the proposed Production Operations activities.
- The email included information on the regulatory process for the activities in Commonwealth and NT jurisdictions and links to a Santos information booklet on the proposed activities and a NOPSEMA brochure on the consultation process and details of how to contact
  Santos to register as a Relevant Person. The email stated that the consultation phase would commence on 11 March 2024 and close on 9 April 2024.
- On 11 March 2024 Santos emailed Darwin Harbour Cruises further to the previous correspondence, to advise that it had commenced the consultation phase which would run until 9 April 2024. In addition to the previous information again being provided, Santos provided information on Relevant Persons' entitlements under the regulatory processes, details of how to provide feedback and a reminder of the closing date for consultation. [Con-3793]
- On 4 April 2024 Santos phoned Darwin Harbour Cruises and left a message with a team member.
- On 6 May 2024, Santos emailed Darwin Harbour Cruises to advise it had extended the consultation period until 20 May 2024. In providing this extension of time, Santos advised that the information in the booklet and factsheet had been updated to account for an additional risk associated with the proposed activity. Santos confirmed that consultation will close on the revised date unless Santos hears otherwise from Darwin Harbour Cruises. [Con-4308]
- On 10 July 2024 Santos emailed Darwin Harbour Cruises to advise the consultation period had been completed. Santos advised Darwin Harbour Cruises that it considered consultation had now closed for the purpose of Santos finalising and submitting environment plans for these activities to government regulators for assessment. [Con-5130]
- On 15 October 2024 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that the EP had been submitted to NOPSEMA for assessment, provided a link to the full EP on the NOPSEMA website and Santos' contact details for any queries. The Update also stated that the OEMP would be submitted to the NT-DITT for assessment. [Con- 5982]
- On 23 January 2025 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that, following a request from NOPSEMA to provide further information, Santos had re-submitted the EP with some amendments in December 2024 for further assessment. The Update also re-stated that the OEMP would be submitted to the NT-DITT for assessment. [Con-6036]
- No further correspondence or feedback was received from Darwin Harbour Cruises.

Summary of response by Relevant Person	Assessment of merits	Santos' Response Statement	OEMP refere
No response was received from Darwin Harbour Cruises.	Santos considers it has provided sufficient information and a reasonable period of time for consultation.	No response required.	Not applicabl
	Santos considers Section 25 consultation requirements to have been met.		

# Santos

## ence

le.

### ence

### **Darwin Harbour Fishing Charters**

Summary of consultation effort:

- On 9 February 2024 Santos emailed Darwin Harbour Fishing Charters to advise the start of preliminary consultation regarding proposed activities for consultation to be managed under the Coastal Waters OEMP and Barossa Production Operations EP. [Con-3787]
- The email advised that Santos was seeking information to better understand:
  - o if you are from a government Department or agency, how the proposed Production Operations activities may be relevant to your department or agency; and
  - what (if any) functions, interests or activities you or your organisation have that may be affected by the proposed Production Operations activities. 0
- The email included information on the regulatory process for the activities in Commonwealth and NT jurisdictions and links to a Santos information booklet on the proposed activities and a NOPSEMA brochure on the consultation process and details of how to contact Santos to register as a Relevant Person. The email stated that the consultation phase would commence on 11 March 2024 and close on 9 April 2024.
- On 11 March 2024 Santos emailed NT tourism operators further to the previous correspondence, to advise that it had commenced the consultation phase which would run until 9 April 2024. In addition to the previous information again being provided, Santos provided information on Relevant Persons' entitlements under the regulatory processes, details of how to provide feedback and a reminder of the closing date for consultation. [Con-3793]
- On 4 April 2024 Santos phoned Darwin Harbour Fishing Charters regarding consultation for Production Operations activities and spoke to a team member who requested emails sent 9 February and 11 March 2024 be resent. These were re-sent by Santos the same day. [Con-4274]
- On 7 May 2024, Santos emailed Darwin Harbour Fishing Charters to advise it had extended the consultation period until 21 May 2024. In providing this extension of time, Santos advised that the information in the booklet and factsheet had been updated to account for an additional risk associated with the proposed activity. Santos confirmed that consultation will close on the revised date unless Santos hears otherwise from Darwin Harbour Fishing Charters. [Con-4320]
- On 10 July 2024 Santos emailed Darwin Harbour Fishing Charters to advise the consultation period had been completed. Santos advised Darwin Harbour Fishing Charters for its input and advised it considered that consultation had now closed for the purpose of Santos finalising and submitting environment plans for these activities to government regulators for assessment. [Con-5130]
- On 15 October 2024 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that the EP had been submitted to NOPSEMA for assessment, provided a link to the full EP on the NOPSEMA website and Santos' contact details for any queries. The Update also stated that the OEMP would be submitted to the NT-DITT for assessment. [Con- 5982]
- On 23 January 2025 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that, following a request from NOPSEMA to provide further information, Santos had re-submitted the EP with some amendments in December 2024 for further assessment. The Update also re-stated that the OEMP would be submitted to the NT-DITT for assessment. [Con-6036]
- No further correspondence or feedback was received from Darwin Harbour Fishing Charters.

Summary of response by Relevant Person	Assessment of merits	Santos' Response Statement	OEMP refere
No response was received from Darwin Harbour Fishing Charters.	Santos considers it has provided sufficient information and a reasonable period of time for consultation. Santos considers Section 25 consultation requirements to have been met.	No response required.	Not applicable

# Darwin Red Devil Fishing Charters

Summary of consultation effort:

- On 9 February 2024 Santos emailed Darwin Red Devil Fishing Charters to advise the start of preliminary consultation regarding proposed activities for consultation to be managed under the Coastal Waters OEMP and Barossa Production Operations EP. [Con-3787]
- The email advised that Santos was seeking information to better understand:
  - o if you are from a government Department or agency, how the proposed Production Operations activities may be relevant to your department or agency; and
  - what (if any) functions, interests or activities you or your organisation have that may be affected by the proposed Production Operations activities.
- The email included information on the regulatory process for the activities in Commonwealth and NT jurisdictions and links to a Santos information booklet on the proposed activities and a NOPSEMA brochure on the consultation process and details of how to contact Santos to register as a Relevant Person. The email stated that the consultation phase would commence on 11 March 2024 and close on 9 April 2024.
- On 11 March 2024 Santos emailed Darwin Red Devil Fishing Charters further to the previous correspondence, to advise that it had commenced the consultation phase which would run until 9 April 2024. In addition to the previous information again being provided, Santos provided information on Relevant Persons' entitlements under the regulatory processes, details of how to provide feedback and a reminder of the closing date for consultation. [Con-3793]
- On 4 April 2024 Santos phoned Darwin Red Devil Fishing Charters and left a voice mail message regarding consultation for Production Operations activities.
- On 2 May 2024, Santos emailed Darwin Red Devil Fishing Charters to advise it was calling to follow up on previous emails and to advise it had extended the consultation period until 16 May 2024. In providing this extension of time, Santos advised that the information in the booklet and factsheet had been updated to account for an additional risk associated with the proposed activity. Santos confirmed that consultation will close on the revised date unless Santos hears otherwise from Darwin Red Devil Fishing Charters. [Con-4288]
- On 10 July 2024 Santos emailed Darwin Red Devil Fishing Charters to advise the consultation period had been completed. Santos advised Darwin Red Devil Fishing Charters that it considered consultation had now closed for the purpose of Santos finalising and submitting environment plans for these activities to government regulators for assessment. [Con-5130]
- On 15 October 2024 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that the EP had been submitted to NOPSEMA for assessment, provided a link to the full EP on the NOPSEMA website and Santos' contact details for any queries. The Update also stated that the OEMP would be submitted to the NT-DITT for assessment. [Con- 5982]
- On 23 January 2025 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that, following a request from NOPSEMA to provide further information, Santos had re-submitted the EP with some amendments in December 2024 for further assessment. The Update also re-stated that the OEMP would be submitted to the NT-DITT for assessment. [Con-6036]
- No further correspondence or feedback was received from Darwin Red Devil Fishing Charters.

Summary of response by Relevant Person	Assessment of merits	Santos' Response Statement	OEMP refere
No response was received from Darwin Red Devils Fishing Charters.	Santos considers it has provided sufficient information and a reasonable period of time for consultation.	No response required.	Not applicable
	met.		l
Darwin Sailing Club			

# Santos

nce

nce

- On 9 February 2024 Santos emailed Darwin Sailing Club to advise the start of preliminary consultation regarding proposed activities for consultation to be managed under the Coastal Waters OEMP and Barossa Production Operations EP. [Con-3787]
- The email advised that Santos was seeking information to better understand:
  - o if you are from a government Department or agency, how the proposed Production Operations activities may be relevant to your department or agency; and
  - what (if any) functions, interests or activities you or your organisation have that may be affected by the proposed Production Operations activities.
- The email included information on the regulatory process for the activities in Commonwealth and NT jurisdictions and links to a Santos information booklet on the proposed activities and a NOPSEMA brochure on the consultation process and details of how to contact Santos to register as a Relevant Person. The email stated that the consultation phase would commence on 11 March 2024 and close on 9 April 2024.
- On 11 March 2024 Santos emailed Darwin Sailing Club further to the previous correspondence, to advise that it had commenced the consultation phase which would run until 9 April 2024. In addition to the previous information again being provided, Santos provided information on Relevant Persons' entitlements under the regulatory processes, details of how to provide feedback and a reminder of the closing date for consultation. [Con-3793]
- On 4 April 2024 Santos phoned Darwin Sailing Club and was unable to leave a message.
- On 2 May 2024, Santos emailed Darwin Sailing Club to advise it had extended the consultation period until 16 May 2024. In providing this extension of time, Santos advised that the information in the booklet and factsheet had been updated to account for an additional risk associated with the proposed activity. Santos confirmed that consultation will close on the revised date unless Santos hears otherwise from Darwin Sailing Club. [Con-4280]
- On 10 July 2024 Santos emailed Darwin Sailing Club to advise the consultation period had been completed. Santos advised Darwin Sailing Club that it considered consultation had now closed for the purpose of Santos finalising and submitting environment plans for these activities to government regulators for assessment. [Con-5130]
- On 15 October 2024 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that the EP had been submitted to NOPSEMA for assessment, provided a link to the full EP on the NOPSEMA website and Santos' contact details for any queries. The Update also stated that the OEMP would be submitted to the NT-DITT for assessment. [Con- 5982]
- On 23 January 2025 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that, following a request from NOPSEMA to provide further information, Santos had re-submitted the EP with some amendments in December 2024 for further assessment. The Update also re-stated that the OEMP would be submitted to the NT-DITT for assessment. [Con-6036]
- No further correspondence or feedback was received from Darwin Sailing Club.

Summary of response by Relevant Person	Assessment of merits	Santos' Response Statement	OEMP refere
No response was received from Darwin Sailing Club.	Santos considers it has provided sufficient information and a reasonable period of time for consultation. Santos considers Section 25 consultation requirements to have been met.	No response required.	Not applicabl

## Darwin Trailer Boat Club

Summary of consultation effort:

- On 9 February 2024 Santos emailed Darwin Trailer Boat Club to advise the start of preliminary consultation regarding proposed activities for consultation to be managed under the Coastal Waters OEMP and Barossa Production Operations EP. [Con-3787]
- The email advised that Santos was seeking information to better understand:
  - o if you are from a government Department or agency, how the proposed Production Operations activities may be relevant to your department or agency; and
  - what (if any) functions, interests or activities you or your organisation have that may be affected by the proposed Production Operations activities.
- The email included information on the regulatory process for the activities in Commonwealth and NT jurisdictions and links to a Santos information booklet on the proposed activities and a NOPSEMA brochure on the consultation process and details of how to contact Santos to register as a Relevant Person. The email stated that the consultation phase would commence on 11 March 2024 and close on 9 April 2024.
- On 11 March 2024 Santos emailed Darwin Trailer Boat Club further to the previous correspondence, to advise that it had commenced the consultation phase which would run until 9 April 2024. In addition to the previous information again being provided, Santos provided information on Relevant Persons' entitlements under the regulatory processes, details of how to provide feedback and a reminder of the closing date for consultation. [Con-3793]
- On 4 April 2024 Santos phoned Darwin Trailer Boat Club and spoke to a team member who confirmed emails sent on 9 February and 11 March had been received.
- On 6 May 2024, Santos emailed Darwin Trailer Boat Club to advise it was calling to follow up on previous emails and to advise it had extended the consultation period until 20 May 2024. In providing this extension of time, Santos advised that the information in the booklet and factsheet had been updated to account for an additional risk associated with the proposed activity. Santos confirmed that consultation will close on the revised date unless Santos hears otherwise from Darwin Trailer Boat Club. [Con-4306]
- On 10 July 2024 Santos emailed Darwin Trailer Boat Club to advise the consultation period had been completed. Santos advised Darwin Trailer Boat Club that it considered consultation had now closed for the purpose of Santos finalising and submitting environment plans for these activities to government regulators for assessment. [Con-5130]
- On 15 October 2024 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that the EP had been submitted to NOPSEMA for assessment, provided a link to the full EP on the NOPSEMA website and Santos' contact details for any queries. The Update also stated that the OEMP would be submitted to the NT-DITT for assessment. [Con- 5982]
- On 23 January 2025 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that, following a request from NOPSEMA to provide further information. Santos had re-submitted the EP with some amendments in December 2024 for further assessment. The Update also re-stated that the OEMP would be submitted to the NT-DITT for assessment. [Con-6036]
- No further correspondence or feedback was received from Darwin Trailer Boat Club.

Summary of response by Relevant Person	Assessment of merits	Santos' Response Statement	OEMP refere
No response was received from Darwin Trailer Boat Club.	Santos considers it has provided sufficient information and a reasonable period of time for consultation. Santos considers Section 25 consultation requirements to have been met.	No response required.	Not applicabl

# Santos

nce

ence

## **Dinah Beach Cruising Yacht Club**

### Summary of consultation effort:

- On 9 February 2024 Santos emailed Dinah Beach Cruising Yacht Club to advise the start of preliminary consultation regarding proposed activities for consultation to be managed under the Coastal Waters OEMP and Barossa Production Operations EP. [Con-3787]
- The email advised that Santos was seeking information to better understand:
  - o if you are from a government Department or agency, how the proposed Production Operations activities may be relevant to your department or agency; and
  - what (if any) functions, interests or activities you or your organisation have that may be affected by the proposed Production Operations activities. 0
- The email included information on the regulatory process for the activities in Commonwealth and NT jurisdictions and links to a Santos information booklet on the proposed activities and a NOPSEMA brochure on the consultation process and details of how to contact Santos to register as a Relevant Person. The email stated that the consultation phase would commence on 11 March 2024 and close on 9 April 2024.
- On 11 March 2024 Santos emailed Dinah Beach Cruising Yacht Club further to the previous correspondence, to advise that it had commenced the consultation phase which would run until 9 April 2024. In addition to the previous information again being provided, Santos provided information on Relevant Persons' entitlements under the regulatory processes, details of how to provide feedback and a reminder of the closing date for consultation. [Con-3793]
- On 4 April 2024 Santos phoned Dinah Beach Cruising Yacht Club and left a voice mail message regarding consultation for Production Operations activities.
- On 2 May 2024, Santos emailed Dinah Beach Cruising Yacht Club to advise it was calling to follow up on previous emails and to advise it had extended the consultation period until 16 May 2024. In providing this extension of time, Santos advised that the information in the booklet and factsheet had been updated to account for an additional risk associated with the proposed activity. Santos confirmed that consultation will close on the revised date unless Santos hears otherwise from Dinah Beach Cruising Yacht Club. [Con-4289]
- On 2 May 2024, Dinah Beach Cruising Yacht Association emailed Santos and advised it had circulated information received from Santos to its committee and asked interested parties to respond. [Con-4302]
- On 10 July 2024 Santos emailed Dinah Beach Cruising Yacht Association to advise the consultation period had been completed. Santos advised it considered that consultation had now closed for the purpose of Santos finalising and submitting environment plans for these activities to government regulators for assessment. [Con-5130]
- On 15 October 2024 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that the EP had been submitted to NOPSEMA for assessment, provided a link to the full EP on the NOPSEMA website and Santos' contact details for any queries. The Update also stated that the OEMP would be submitted to the NT-DITT for assessment. [Con- 5982]
- On 23 January 2025 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that, following a request from NOPSEMA to provide further information. Santos had re-submitted the EP with some amendments in December 2024 for further assessment. The Update also re-stated that the OEMP would be submitted to the NT-DITT for assessment. [Con-6036]

### No further correspondence or feedback was received from Dinah Beach Cruising Yacht Association.

Summary of response by Relevant Person	Assessment of merits	Santos' Response Statement	OEMP refere
No response was received from Dinah Beach Cruising Yacht Association.	Santos considers it has provided sufficient information and a reasonable period of time for consultation.	No response required.	Not applicab
	Santos considers Section 25 consultation requirements to have been met.		

### Dreamers Dive Academy Timor

Summary of consultation effort:

- On 9 February 2024 Santos emailed Dreamers Dive Academy Timor to advise the start of preliminary consultation regarding proposed activities for consultation to be managed under the Coastal Waters OEMP and Barossa Production Operations EP. [Con-3787]
- The email advised that Santos was seeking information to better understand:
  - if you are from a government Department or agency, how the proposed Production Operations activities may be relevant to your department or agency; and
  - what (if any) functions, interests or activities you or your organisation have that may be affected by the proposed Production Operations activities.
- The email included information on the regulatory process for the activities in Commonwealth and NT jurisdictions and links to a Santos information booklet on the proposed activities and a NOPSEMA brochure on the consultation process and details of how to contact Santos to register as a Relevant Person. The email stated that the consultation phase would commence on 11 March 2024 and close on 9 April 2024.
- On 11 March 2024 Santos emailed Dreamers Dive Academy Timor further to the previous correspondence, to advise that it had commenced the consultation phase which would run until 9 April 2024. In addition to the previous information again being provided, Santos provided information on Relevant Persons' entitlements under the regulatory processes, details of how to provide feedback and a reminder of the closing date for consultation. [Con-3793]
- On 4 April 2024 Santos phoned Dreamers Dive Academy Timor and was unable to leave a message.
- On 2 May 2024, Santos emailed Dreamers Dive Academy further to previous correspondence, to advise it had extended the consultation period until 16 May 2024. In providing this extension of time, Santos advised that the information in the booklet and factsheet had been updated to account for an additional risk associated with the proposed activity. Santos confirmed that consultation will close on the revised date unless Santos hears otherwise from Dreamers Dive Academy. [Con-4281]
- On 10 July 2024 Santos emailed Dreamers Dive Academy Timor to advise the consultation period had been completed. Santos advised it considered that consultation had now closed for the purpose of Santos finalising and submitting environment plans for these activities to government regulators for assessment. [Con-5130]
- On 15 October 2024 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that the EP had been submitted to NOPSEMA for assessment, provided a link to the full EP on the NOPSEMA website and Santos' contact details for any queries. The Update also stated that the OEMP would be submitted to the NT-DITT for assessment. [Con- 5982]
- On 23 January 2025 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that, following a request from NOPSEMA to provide further information, Santos had re-submitted the EP with some amendments in December 2024 for further assessment. The Update also re-stated that the OEMP would be submitted to the NT-DITT for assessment. [Con-6036]

### No further correspondence or feedback was received from Dreamers Dive Academy.

Summary of response by Relevant Person	Assessment of merits	Santos' Response Statement	OEMP refere
No response was received from Dreamers Dive Academy Timor.	Santos considers it has provided sufficient information and a reasonable period of time for consultation.	No response required.	Not applicabl
	met.		

# Santos

ence

le.

### **Dundee Beach Fishing Charters**

Summary of consultation effort:

- On 9 February 2024 Santos emailed Dundee Beach Fishing Charters to advise the start of preliminary consultation regarding proposed activities for consultation to be managed under the Coastal Waters OEMP and Barossa Production Operations EP. [Con-3787]
- The email advised that Santos was seeking information to better understand:
  - o if you are from a government Department or agency, how the proposed Production Operations activities may be relevant to your department or agency; and
  - what (if any) functions, interests or activities you or your organisation have that may be affected by the proposed Production Operations activities.
- The email included information on the regulatory process for the activities in Commonwealth and NT jurisdictions and links to a Santos information booklet on the proposed activities and a NOPSEMA brochure on the consultation process and details of how to contact Santos to register as a Relevant Person. The email stated that the consultation phase would commence on 11 March 2024 and close on 9 April 2024.
- On 11 March 2024 Santos emailed Dundee Beach Fishing Charters further to the previous correspondence, to advise that it had commenced the consultation phase which would run until 9 April 2024. In addition to the previous information again being provided, Santos provided information on Relevant Persons' entitlements under the regulatory processes, details of how to provide feedback and a reminder of the closing date for consultation. [Con-3793]
- On 4 April 2024 Santos phoned Dundee Beach Fishing Charters and spoke to a team member who confirmed emails sent on 9 February and 11 March had been received.
- On 6 May 2024, Santos emailed Dundee Beach Fishing Charters further to previous correspondence, to advise it was calling to follow up on previous emails and to advise it had extended the consultation period until 20 May 2024. In providing this extension of time, Santos advised that the information in the booklet and factsheet had been updated to account for an additional risk associated with the proposed activity. Santos confirmed that consultation will close on the revised date unless Santos hears otherwise from Dundee Beach Fishing Charters. [Con-4305]
- On 10 July 2024 Santos emailed Dundee Beach Fishing Charters to advise the consultation period had been completed. Santos advised it consultation had now closed for the purpose of Santos finalising and submitting environment plans for these activities to government regulators for assessment. [Con-5130]
- On 15 October 2024 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that the EP had been submitted to NOPSEMA for assessment, provided a link to the full EP on the NOPSEMA website and Santos' contact details for any queries. The Update also stated that the OEMP would be submitted to the NT-DITT for assessment. [Con- 5982]
- On 23 January 2025 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that, following a request from NOPSEMA to provide further information, Santos had re-submitted the EP with some amendments in December 2024 for further assessment. The Update also re-stated that the OEMP would be submitted to the NT-DITT for assessment. [Con-6036]
- No further correspondence or feedback was received from Dundee Beach Fishing Charters.

Summary of response by Relevant Person	Assessment of merits	Santos' Response Statement	OEMP reference
No response was received from Dundee Beach Fishing Charters.	Santos considers it has provided sufficient information and a reasonable period of time for consultation. Santos considers Section 25 consultation requirements to have been met.	No response required.	Not applicable.

# **Equinox Fishing Charters**

Summary of consultation effort:

- On 9 February 2024 Santos emailed Equinox Fishing Charters to advise the start of preliminary consultation regarding proposed activities for consultation to be managed under the Coastal Waters OEMP and Barossa Production Operations EP. [Con-3787]
- The email advised that Santos was seeking information to better understand:
  - o if you are from a government Department or agency, how the proposed Production Operations activities may be relevant to your department or agency; and
  - what (if any) functions, interests or activities you or your organisation have that may be affected by the proposed Production Operations activities.
- The email included information on the regulatory process for the activities in Commonwealth and NT jurisdictions and links to a Santos information booklet on the proposed activities and a NOPSEMA brochure on the consultation process and details of how to contact Santos to register as a Relevant Person. The email stated that the consultation phase would commence on 11 March 2024 and close on 9 April 2024.
- On 11 March 2024 Santos emailed Equinox Fishing Charters further to the previous correspondence, to advise that it had commenced the consultation phase which would run until 9 April 2024. In addition to the previous information again being provided, Santos provided information on Relevant Persons' entitlements under the regulatory processes, details of how to provide feedback and a reminder of the closing date for consultation. [Con-3793]
- On 4 April 2024 Santos phoned Equinox Fishing Charters and spoke with a company representative regarding consultation for Production Operations activities who provided feedback during the call. The feedback is summarised below.
- On 3 May 2024, Santos emailed Equinox Fishing Charters to summarise feedback received and to advise it had extended the consultation period until 17 May 2024. In providing this extension of time, Santos advised that the information in the booklet and factsheet had been updated to account for an additional risk associated with the proposed activity. Santos confirmed that consultation will close on the revised date unless Santos hears otherwise from Equinox Fishing Charters. [Con-4303]
- On 10 July 2024 Santos emailed Equinox Fishing Charters to advise the consultation period had been completed. Santos advised it consultation had now closed for the purpose of Santos finalising and submitting environment plans for these activities to government regulators for assessment. [Con-5130]
- On 15 October 2024 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that the EP had been submitted to NOPSEMA for assessment, provided a link to the full EP on the NOPSEMA website and Santos' contact details for any queries. The Update also stated that the OEMP would be submitted to the NT-DITT for assessment. [Con- 5982]
- On 23 January 2025 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that, following a request from NOPSEMA to provide further information, Santos had re-submitted the EP with some amendments in December 2024 for further assessment. The Update also re-stated that the OEMP would be submitted to the NT-DITT for assessment. [Con-6036]
- No further correspondence or feedback was received from Equinox Fishing Charters.

Summary of response by Relevant Person	Assessment of merits	Santos' Response Statement	OEMP Refere
Equinox Fishing Charters were concerned about activity and development across different industries, having a potential impact on	Santos has considered the matters raised in the OEMP.	No response required	An assessme EMBA is prov
fishing.	Santos acknowledges that the response raises a general concern on the potential for cumulative impacts of industry in the region to fishing. Santos has no control over other non-Santos developments and industry in the region but does consider the potential for impacts with		Santos' impac and tourism o

# Santos

### ence

nt of tourism operators within the operational areas and ided in Section 3.2.14.6.

ct and risk assessment, inclusive of impacts and risks to fish perators, , is provided in Section 6 and 7

other marine users within its impact and risk assessment (Section 6 & 7) and considers impacts to fish and tourism operators.         No planned concurrent activities will occur in the OEMP and the vessel Activity is based on IMMR vessel presence occurring once every three to five years for approximately 7 – 30 days.
--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------

# **Estuary Escapes Fishing Charters**

Summary of consultation effort:

- On 9 February 2024 Santos emailed Estuary Escapes Fishing Charters to advise the start of preliminary consultation regarding proposed activities for consultation to be managed under the Coastal Waters OEMP and Barossa Production Operations EP. [Con-3787]
- The email advised that Santos was seeking information to better understand:
  - o if you are from a government Department or agency, how the proposed Production Operations activities may be relevant to your department or agency; and
  - what (if any) functions, interests or activities you or your organisation have that may be affected by the proposed Production Operations activities.
- The email included information on the regulatory process for the activities in Commonwealth and NT jurisdictions and links to a Santos information booklet on the proposed activities and a NOPSEMA brochure on the consultation process and details of how to contact Santos to register as a Relevant Person. The email stated that the consultation phase would commence on 11 March 2024 and close on 9 April 2024.
- On 11 March 2024 Santos emailed Estuary Escapes Fishing Charters further to the previous correspondence, to advise that it had commenced the consultation phase which would run until 9 April 2024. In addition to the previous information again being provided, Santos provided information on Relevant Persons' entitlements under the regulatory processes, details of how to provide feedback and a reminder of the closing date for consultation. [Con-3793]
- On 4 April 2024 Santos phoned Estuary Escapes Fishing Charters regarding consultation for Production Operations activities and spoke to a company representative who indicated they wanted to provide feedback and asked for an extension of time to do so.
- On 8 May 2024, Santos emailed Estuary Escapes Fishing Charters further to previous correspondence, to advise it had extended the consultation period until 22 May 2024. In providing this extension of time, Santos advised that the information in the booklet and factsheet had been updated to account for an additional risk associated with the proposed activity. Santos confirmed that consultation will close on the revised date unless Santos hears otherwise from Estuary Escapes Fishing Charters. [Con-4324]
- On 10 July 2024 Santos emailed Estuary Escapes Fishing Charters to advise the consultation period had been completed. Santos advised Estuary Escapes Fishing Charters that it considered consultation had now closed for the purpose of Santos finalising and submitting environment plans for these activities to government regulators for assessment. [Con-5130]
- On 15 October 2024 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that the EP had been submitted to NOPSEMA for assessment, provided a link to the full EP on the NOPSEMA website and Santos' contact details for any queries. The Update also stated that the OEMP would be submitted to the NT-DITT for assessment. [Con- 5982]
- On 23 January 2025 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that, following a request from NOPSEMA to provide further information. Santos had re-submitted the EP with some amendments in December 2024 for further assessment. The Update also re-stated that the OEMP would be submitted to the NT-DITT for assessment. [Con-6036]

### No further correspondence or feedback was received from Estuary Escapes Fishing Charters.

Summary of response by Relevant Person	Assessment of merits	Santos' Response Statement	OEMP Refer
No response was received from Estuary Escape Fishing Charters.	Santos considers it has provided sufficient information and a reasonable period of time for consultation. Santos considers Section 25 consultation requirements to have been met.	No response required.	Not applicabl

# Fish the Top End Fishing Charters (incorporating Obsession Fishing Safaris and Vision Sport Fishing Adventures)

Summary of consultation effort:

- On 9 February 2024 Santos emailed Fish the Top End Fishing Charters to advise the start of preliminary consultation regarding proposed activities for consultation to be managed under the Coastal Waters OEMP and Barossa Production Operations EP. [Con-3787]
- The email advised that Santos was seeking information to better understand:
  - o if you are from a government Department or agency, how the proposed Production Operations activities may be relevant to your department or agency; and
  - what (if any) functions, interests or activities you or your organisation have that may be affected by the proposed Production Operations activities.
- The email included information on the regulatory process for the activities in Commonwealth and NT jurisdictions and links to a Santos information booklet on the proposed activities and a NOPSEMA brochure on the consultation process and details of how to contact Santos to register as a Relevant Person. The email stated that the consultation phase would commence on 11 March 2024 and close on 9 April 2024.
- On 11 March 2024 Santos emailed Fish the Top End Fishing Charters further to the previous correspondence, to advise that it had commenced the consultation phase which would run until 9 April 2024. In addition to the previous information again being provided, Santos provided information on Relevant Persons' entitlements under the regulatory processes, details of how to provide feedback and a reminder of the closing date for consultation. [Con-3793]
- On 4 April 2024 Santos phoned Fish the Top End Fishing Charters and spoke to a company representative regarding consultation for Production Operations activities who advised that they received the emails and did not have feedback as there would be no impacts from Barossa Production Operations EP activities. They also indicated they also manage Obsession Fishing Safaris and Vision Sport Fishing Adventures.
- On 6 May 2024, Santos emailed Vision Sportfishing Adventures, Obsession Fishing Safaris and Fish the Top End Fishing Charters, to advise it had extended the consultation period until 20 May 2024. In providing this extension of time, Santos advised that the information in the booklet and factsheet had been updated to account for an additional risk associated with the proposed activity. Santos confirmed that consultation will close on the revised date unless Santos hears otherwise from Vision Sportfishing Adventures. Obsession Fishing Safaris and Fish the Top End Fishing Charters. [Con-4307]
- On 10 July 2024 Santos emailed Vision Sportfishing Adventures, Obsession Fishing Safaris and Fish the Top End Fishing Charters to advise the consultation period had been completed. Santos advised it considered that consultation had now closed for the purpose of Santos finalising and submitting environment plans for these activities to government regulators for assessment. [Con-5130]
- On 15 October 2024 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that the EP had been submitted to NOPSEMA for assessment, provided a link to the full EP on the NOPSEMA website and Santos' contact details for any queries. The Update also stated that the OEMP would be submitted to the NT-DITT for assessment. [Con- 5982]
- On 23 January 2025 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that, following a request from NOPSEMA to provide further information, Santos had re-submitted the EP with some amendments in December 2024 for further assessment. The Update also re-stated that the OEMP would be submitted to the NT-DITT for assessment. [Con-6036]
- No further correspondence or feedback was received from Fish the Top End Fishing Charters.

# Santos

Summary of response by Relevant Person	Assessment of merits	Santos' Response Statement	OEMP Reference
Top End Fishing Charters responded that it did not have feedback as there would be no impacts from OEMP activities on its functions, interests or activities.	Santos notes the response provided by Top End Fishing Santos considers it has provided sufficient information and a reasonable period of time for consultation. Santos considers Section 25 consultation requirements to have been met.	No response required.	Not applicable.

## FNA Sports Fishing

Summary of consultation effort:

- On 9 February 2024 Santos emailed FNA Sports Fishing to advise the start of preliminary consultation regarding proposed activities for consultation to be managed under the Coastal Waters OEMP and Barossa Production Operations EP. [Con-3787]
- The email advised that Santos was seeking information to better understand:
  - o if you are from a government Department or agency, how the proposed Production Operations activities may be relevant to your department or agency; and
  - what (if any) functions, interests or activities you or your organisation have that may be affected by the proposed Production Operations activities.
- The email included information on the regulatory process for the activities in Commonwealth and NT jurisdictions and links to a Santos information booklet on the proposed activities and a NOPSEMA brochure on the consultation process and details of how to contact Santos to register as a Relevant Person. The email stated that the consultation phase would commence on 11 March 2024 and close on 9 April 2024.
- On 11 March 2024 Santos emailed FNA Sports Fishing further to the previous correspondence, to advise that it had commenced the consultation phase which would run until 9 April 2024. In addition to the previous information again being provided, Santos provided information on Relevant Persons' entitlements under the regulatory processes, details of how to provide feedback and a reminder of the closing date for consultation. [Con-3793]
- On 4 April 2024 Santos phoned FNA Sports Fishing and left a voice mail message.
- On 2 May 2024, Santos emailed FNA Sports Fishing to advise it had extended the consultation period until 16 May 2024. In providing this extension of time, Santos advised that the information in the booklet and factsheet had been updated to account for an additional risk associated with the proposed activity. Santos confirmed that consultation will close on the revised date unless Santos hears otherwise from FNA Sports Fishing [Con-4290]
- On 10 July 2024 Santos emailed FNA Sports Fishing to advise the consultation period had been completed. Santos advised FNA Sports Fishing that it considered consultation had now closed for the purpose of Santos finalising and submitting environment plans for these activities to government regulators for assessment. [Con-5130]
- On 15 October 2024 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that the EP had been submitted to NOPSEMA for assessment, provided a link to the full EP on the NOPSEMA website and Santos' contact details for any queries. The Update also stated that the OEMP would be submitted to the NT-DITT for assessment. [Con- 5982]
- On 23 January 2025 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that, following a request from NOPSEMA to provide further information, Santos had re-submitted the EP with some amendments in December 2024 for further assessment. The Update also re-stated that the OEMP would be submitted to the NT-DITT for assessment. [Con-6036]
- No further correspondence or feedback was received from FNA Sports Fishing.

Summary of response by Relevant Person	Assessment of merits	Santos' Response Statement	OEMP Refer
No response was received from FNA Sports Fishing.	Santos considers it has provided sufficient information and a reasonable period of time for consultation. Santos considers Section 25 consultation requirements to have been met.	No response required.	Not applicab

### Humbug Fishing

Summary of consultation effort:

- On 9 February 2024 Santos emailed Humbug Fishing to advise the start of preliminary consultation regarding proposed activities for consultation to be managed under the Coastal Waters OEMP and Barossa Production Operations EP. [Con-3787]
- The email advised that Santos was seeking information to better understand:
  - o if you are from a government Department or agency, how the proposed Production Operations activities may be relevant to your department or agency; and
  - what (if any) functions, interests or activities you or your organisation have that may be affected by the proposed Production Operations activities.
- The email included information on the regulatory process for the activities in Commonwealth and NT jurisdictions and links to a Santos information booklet on the proposed activities and a NOPSEMA brochure on the consultation process and details of how to contact Santos to register as a Relevant Person. The email stated that the consultation phase would commence on 11 March 2024 and close on 9 April 2024.
- On 11 March 2024 Santos emailed Humbug Fishing further to the previous correspondence, to advise that it had commenced the consultation phase which would run until 9 April 2024. In addition to the previous information again being provided, Santos provided information on Relevant Persons' entitlements under the regulatory processes, details of how to provide feedback and a reminder of the closing date for consultation. [Con-3793]
- On 4 April 2024 Santos phoned Humbug Fishing and left a voice mail message.
- On 2 May 2024, Santos emailed Humbug Fishing to advise it had extended the consultation period until 16 May 2024. In providing this extension of time, Santos advised that the information in the booklet and factsheet had been updated to account for an additional risk. associated with the proposed activity. Santos confirmed that consultation will close on the revised date unless Santos hears otherwise from Humbug Fishing. [Con-4291]
- On 10 July 2024 Santos emailed Humbug Fishing to advise the consultation period had been completed. Santos advised Humbug Fishing that it considered consultation had now closed for the purpose of Santos finalising and submitting environment plans for these activities to government regulators for assessment. [Con-5130]
- On 15 October 2024 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that the EP had been submitted to NOPSEMA for assessment, provided a link to the full EP on the NOPSEMA website and Santos' contact details for any queries. The Update also stated that the OEMP would be submitted to the NT-DITT for assessment. [Con- 5982]
- On 23 January 2025 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that, following a request from NOPSEMA to provide further information, Santos had re-submitted the EP with some amendments in December 2024 for further assessment. The Update also re-stated that the OEMP would be submitted to the NT-DITT for assessment. [Con-6036]
- Notwithstanding the information provided and the steps described above, no comments or input were received on the Coastal Waters OEMP.

Summary of response by Relevant Person	Assessment of merits	Santos' Response Statement	OEMP Refer



rence

le.

rence

No response was received from Humbug Fishing.	Santos considers it has provided sufficient information and a reasonable period of time for consultation. Santos considers Section 25 consultation requirements to have been met.	No response required.	Not applicabl
Mousies Barra Fishing Charters			

- On 10 June 2024 Santos emailed Mousies Barra Fishing Charters regarding proposed activities for consultation to be managed under the Coastal Waters OEMP and Barossa Production Operations EP and was seeking to consult with Relevant Persons. Santos asked Mousies Barra Fishing Charters to advise Santos by 17 June 2024 if it considered it may be a Relevant Person and what functions, interests or activities it has that may be affected by the production operations activities. [Con-4978]
- The email included links to the Barossa Production Operations Activity Booklet and NOPSEMA's EP consultation information for the community. Santos advised if it did not hear from Mousies Barra Fishing Charters by 17 June 2024, it would assume that it did not have functions, interests or activities that may be affected by the activities, or didn't wish to be consulted for the OEMP and EP.
- On 10 July 2024 Santos emailed Mousies Barra Fishing Charters to advise the consultation period had been completed. Santos advised it considered that consultation had now closed for the purpose of Santos finalising and submitting environment plans for these activities to government regulators for assessment. [Con-5130]
- On 15 October 2024 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that the EP had been submitted to NOPSEMA for assessment, provided a link to the full EP on the NOPSEMA website and Santos' contact details for any queries. The Update also stated that the OEMP would be submitted to the NT-DITT for assessment. [Con- 5982]
- On 23 January 2025 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that, following a request from NOPSEMA to provide further information, Santos had re-submitted the EP with some amendments in December 2024 for further assessment. The Update also re-stated that the OEMP would be submitted to the NT-DITT for assessment. [Con-6036]

No further correspondence or feedback was received from Mousies Barra Fishing Charters.

Summary of response by Relevant Person	Assessment of merits	Santos' Response Statement	OEMP Refer
No response was received from Mousies Barra Fishing Charters.	Santos considers it has provided sufficient information and a reasonable period of time for consultation.	No response required.	Not applicab
	Santos notes that in June 2024, in response to consultation via email and phone on the Darwin Pipeline Duplication EP (Cwth) and DPD Offshore CEMP, Mousies Barra Fishing Charters had provided no comments.		
	Santos considers Section 25 consultation requirements to have been met.		

# **NT Indigenous Tours**

Summary of consultation effort:

- On 9 February 2024 Santos emailed NT Indigenous Tours to advise the start of preliminary consultation regarding proposed activities for consultation to be managed under the Coastal Waters OEMP and Barossa Production Operations EP. [Con-3787]
- The email advised that Santos was seeking information to better understand:
  - o if you are from a government Department or agency, how the proposed Production Operations activities may be relevant to your department or agency; and
  - what (if any) functions, interests or activities you or your organisation have that may be affected by the proposed Production Operations activities.
- The email included information on the regulatory process for the activities in Commonwealth and NT jurisdictions and links to a Santos information booklet on the proposed activities and a NOPSEMA brochure on the consultation process and details of how to contact Santos to register as a Relevant Person. The email stated that the consultation phase would commence on 11 March 2024 and close on 9 April 2024.
- On 11 March 2024 Santos emailed NT Indigenous Tours further to the previous correspondence, to advise that it had commenced the consultation phase which would run until 9 April 2024. In addition to the previous information again being provided, Santos provided information on Relevant Persons' entitlements under the regulatory processes, details of how to provide feedback and a reminder of the closing date for consultation. [Con-3793]
- On 4 April 2024 Santos phoned NT Indigenous Tours regarding consultation for Barossa Production Operations EP activities and spoke to a team member who requested emails sent 9 February and 11 March 2024 be resent. These were resent by Santos the same day. [Con-4275]
- On 7 May 2024, Santos emailed NT Indigenous Tours to advise it had extended the consultation period until 21 May 2024. In providing this extension of time, Santos advised that the information in the booklet and factsheet had been updated to account for an additional risk associated with the proposed activity. Santos confirmed that consultation will close on the revised date unless Santos hears otherwise from NT Indigenous Tours. [Con-4321]
- On 10 July 2024 Santos emailed NT Indigenous Tours to advise the consultation period had been completed. Santos advised NT Indigenous Tours that it considered consultation had now closed for the purpose of Santos finalising and submitting environment plans for these activities to government regulators for assessment. [Con-5130]
- On 15 October 2024 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that the EP had been submitted to NOPSEMA for assessment, provided a link to the full EP on the NOPSEMA website and Santos' contact details for any queries. The Update also stated that the OEMP would be submitted to the NT-DITT for assessment. [Con- 5982]
- On 23 January 2025 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that, following a request from NOPSEMA to provide further information, Santos had re-submitted the EP with some amendments in December 2024 for further assessment. The Update also re-stated that the OEMP would be submitted to the NT-DITT for assessment. [Con-6036]

•	Notwithstanding the information	provided and the steps described	l above, no comments or in	put were received on the Coastal Waters OEM	۰ <u>.</u>
					_

Summary of response by Relevant Person	Assessment of merits	Santos' Response Statement	OEMP Refer
No response was received from NT Indigenous Tours.	Santos considers it has provided sufficient information and a reasonable period of time for consultation.	No response required.	Not applicabl
	Santos considers Section 25 consultation requirements to have been met.		
Offshore Boats Fishing Charters			

Summary of consultation effort:





rence

le

- On 9 February 2024 Santos emailed Offshore Boats Fishing Charters to advise the start of preliminary consultation regarding proposed activities for consultation to be managed under the Coastal Waters OEMP and Barossa Production Operations EP. [Con-3787]
- The email advised that Santos was seeking information to better understand:
  - o if you are from a government Department or agency, how the proposed Production Operations activities may be relevant to your department or agency; and
  - what (if any) functions, interests or activities you or your organisation have that may be affected by the proposed Production Operations activities. 0
- The email included information on the regulatory process for the activities in Commonwealth and NT jurisdictions and links to a Santos information booklet on the proposed activities and a NOPSEMA brochure on the consultation process and details of how to contact Santos to register as a Relevant Person. The email stated that the consultation phase would commence on 11 March 2024 and close on 9 April 2024.
- On 11 March 2024 Santos emailed Offshore Boats Fishing Charters further to the previous correspondence, to advise that it had commenced the consultation phase which would run until 9 April 2024. In addition to the previous information again being provided, Santos provided information on Relevant Persons' entitlements under the regulatory processes, details of how to provide feedback and a reminder of the closing date for consultation. [Con-3793]
- On 4 April 2024 Santos phoned Offshore Boats Fishing Charters and spoke to a company representative who advised they are unlikely to have any interaction with offshore vessels. (CON 5281)
- On 8 May 2024, Santos emailed Offshore Boats Fishing Charters, to advise it had extended the consultation period until 22 May 2024. In providing this extension of time, Santos advised that the information in the booklet and factsheet had been updated to account for an additional risk associated with the proposed activity. Santos confirmed that consultation will close on the revised date unless Santos hears otherwise from Offshore Boats Fishing Charters. [Con-4325]
- On 10 July 2024 Santos emailed Offshore Boats Fishing Charters to advise the consultation period had been completed. Santos advised it consultation had now closed for the purpose of Santos finalising and submitting environment plans for these activities to government regulators for assessment. [Con-5130]
- On 15 October 2024 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that the EP had been submitted to NOPSEMA for assessment, provided a link to the full EP on the NOPSEMA website and Santos' contact details for any queries. The Update also stated that the OEMP would be submitted to the NT-DITT for assessment. [Con- 5982]
- On 23 January 2025 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that, following a request from NOPSEMA to provide further information, Santos had re-submitted the EP with some amendments in December 2024 for further assessment. The Update also re-stated that the OEMP would be submitted to the NT-DITT for assessment. [Con-6036]
- No further correspondence or feedback was received from Offshore Boats Fishing Charters.

Summary of response by Relevant Person	Assessment of merits	Santos' Response Statement	OEMP Refe
Offshore Boats Fishing Charter responded that they are unlikely to have any interaction with offshore vessels.	Santos considers it has provided sufficient information and a reasonable period of time for consultation. Santos considers Section 25 consultation requirements to have been met.	No response required.	Not applicat
Outback Eisbing Charters			

- On 10 June 2024 Santos emailed Outback Fishing Charters regarding proposed activities for consultation to be managed under the Coastal Waters OEMP and Barossa Production Operations EP and was seeking to consult with Relevant Persons. Santos asked Outback Fishing Charters to advise Santos by 17 June 2024 if it considered it may be a Relevant Person and what functions, interests or activities it has that may be affected by the Production Operations activities. [Con-4979]
- The email included links to the Barossa Production Operations Activity Booklet and NOPSEMA's EP consultation information for the community. Santos advised if it did not hear from Outback Fishing Charters by 17 June 2024, it would assume that it did not have functions, interests or activities that may be affected by the activities, or didn't wish to be consulted for the OEMP and EP.
- On 10 July 2024 Santos emailed Outback Fishing Charters to advise the consultation period had been completed. Santos advised it considered that consultation had now closed for the purpose of Santos finalising and submitting environment plans for these activities to government regulators for assessment. [Con-5130]
- On 15 October 2024 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that the EP had been submitted to NOPSEMA for assessment, provided a link to the full EP on the NOPSEMA website and Santos' contact details for any queries. The Update also stated that the OEMP would be submitted to the NT-DITT for assessment. [Con- 5982]
- On 23 January 2025 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that, following a request from NOPSEMA to provide further information, Santos had re-submitted the EP with some amendments in December 2024 for further assessment. The Update also re-stated that the OEMP would be submitted to the NT-DITT for assessment. [Con-6036]
- No further correspondence or feedback was received from Outback Fishing Charters.

Summary of response by Relevant Person	Assessment of merits	Santos' Response Statement	OEMP Refer
No response was received from Outback Fishing Charters.	Santos considers it has provided sufficient information and a reasonable period of time for consultation.	No response required.	Not applicabl
	Santos notes that in June 2024, in response to consultation via email and phone on the Darwin Pipeline Duplication EP (Cwth) and DPD Offshore CEMP, Outback Fishing Charters had provided no comments.		
	Santos considers Section 25 consultation requirements to have been met.		
Palmerston Game Fishing Club			

Summary of consultation effort:

- On 9 February 2024 Santos emailed Palmerston Game Fishing Club to advise the start of preliminary consultation regarding proposed activities for consultation to be managed under the Coastal Waters OEMP and Barossa Production Operations EP. [Con-3787]
- The email advised that Santos was seeking information to better understand:
  - o if you are from a government Department or agency, how the proposed Production Operations activities may be relevant to your department or agency; and
  - what (if any) functions, interests or activities you or your organisation have that may be affected by the proposed Production Operations activities. 0
- The email included information on the regulatory process for the activities in Commonwealth and NT jurisdictions and links to a Santos information booklet on the proposed activities and a NOPSEMA brochure on the consultation process and details of how to contact Santos to register as a Relevant Person. The email stated that the consultation phase would commence on 11 March 2024 and close on 9 April 2024.

rence

ble

ence

- On 11 March 2024 Santos emailed Palmerston Game Fishing Club further to the previous correspondence, to advise that it had commenced the consultation phase which would run until 9 April 2024. In addition to the previous information again being provided, Santos provided information on Relevant Persons' entitlements under the regulatory processes, details of how to provide feedback and a reminder of the closing date for consultation. [Con-3793
- On 4 April 2024 Santos phoned and left a voicemail.
- On 2 May 2024, Santos emailed Palmerston Game Fishing Club to advise it had extended the consultation period until 16 May 2024. In providing this extension of time, Santos advised that the information in the booklet and factsheet had been updated to account for an additional risk associated with the proposed activity. Santos confirmed that consultation will close on the revised date unless Santos hears otherwise from Palmerston Game Fishing Club. [Con-4292]
- On 10 July 2024 Santos emailed Palmerston Game Fishing Club to advise the consultation period had been completed. Santos advised Palmerston Game Fishing Club that it considered consultation had now closed for the purpose of Santos finalising and submitting environment plans for these activities to government regulators for assessment. [Con-5130]
- On 15 October 2024 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that the EP had been submitted to NOPSEMA for assessment, provided a link to the full EP on the NOPSEMA website and Santos' contact details for any queries. The Update also stated that the OEMP would be submitted to the NT-DITT for assessment. [Con- 5982]
- On 23 January 2025 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that, following a request from NOPSEMA to provide further information. Santos had re-submitted the EP with some amendments in December 2024 for further assessment. The Update also re-stated that the OEMP would be submitted to the NT-DITT for assessment. [Con-6036]
- No further correspondence or feedback was received from Palmerston Game Fishing Club.

Summary of response by Relevant Person	Assessment of merits	Santos' Response Statement	OEMP Refer
No response was received from Palmerston Game Fishing Club.	Santos considers it has provided sufficient information and a reasonable period of time for consultation. Santos considers Section 25 consultation requirements to have been met.	No response required.	Not applicab

# Reel Screamin Barra Fishind

Summary of consultation effort:

- On 9 February 2024 Santos emailed Reel Screamin Barra Fishing to advise the start of preliminary consultation regarding proposed activities for consultation to be managed under the Coastal Waters OEMP and Barossa Production Operations EP. [Con-3787]
- The email advised that Santos was seeking information to better understand:
  - o if you are from a government Department or agency, how the proposed Production Operations activities may be relevant to your department or agency; and
  - what (if any) functions, interests or activities you or your organisation have that may be affected by the proposed Production Operations activities.
- The email included information on the regulatory process for the activities in Commonwealth and NT jurisdictions and links to a Santos information booklet on the proposed activities and a NOPSEMA brochure on the consultation process and details of how to contact Santos to register as a Relevant Person. The email stated that the consultation phase would commence on 11 March 2024 and close on 9 April 2024.
- On 11 March 2024 Santos emailed Reel Screamin Barra Fishing further to the previous correspondence, to advise that it had commenced the consultation phase which would run until 9 April 2024. In addition to the previous information again being provided, Santos provided information on Relevant Persons' entitlements under the regulatory processes, details of how to provide feedback and a reminder of the closing date for consultation. [Con-3793]
- On 4 April 2024 Santos phoned Reel Screamin Barra Fishing and spoke to a company representative (also an office-holder with the NT Guided Fishing Industry Association).
- On 6 May 2024, Santos emailed Reel Screamin Barra Fishing to advise it had extended the consultation period until 20 May 2024. In providing this extension of time, Santos advised that the information in the booklet and factsheet had been updated to account for an additional risk associated with the proposed activity. Santos confirmed that consultation will close on the revised date unless Santos hears otherwise from Reel Screamin Barra Fishing. [Con-4313]
- On 10 July 2024 Santos emailed Reel Screamin Barra Fishing to advise the consultation period had been completed. Santos advised Reel Screamin Barra Fishing that it considered consultation had now closed for the purpose of Santos finalising and submitting environment plans for these activities to government regulators for assessment. [Con-5130]
- On 15 October 2024 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that the EP had been submitted to NOPSEMA for assessment, provided a link to the full EP on the NOPSEMA website and Santos' contact details for any queries. The Update also stated that the OEMP would be submitted to the NT-DITT for assessment. [Con- 5982]
- On 23 January 2025 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that, following a request from NOPSEMA to provide further information, Santos had re-submitted the EP with some amendments in December 2024 for further assessment. The Update also re-stated that the OEMP would be submitted to the NT-DITT for assessment. [Con-6036]

### No further correspondence or feedback was received from Reel Screamin Barra Fishing.

Summary of response by Relevant Person	Assessment of merits	Santos' Response Statement	OEMP Refer
No response was received from Reel Screamin Barra Fishing.	Santos considers it has provided sufficient information and a reasonable period of time for consultation. Santos considers Section 25 consultation requirements to have been met.	No response required.	Not applicabl

# **River and Reef**

Summary of consultation effort:

- On 10 June 2024 Santos emailed River and Reef regarding proposed activities for consultation to be managed under the Coastal Waters OEMP and Barossa Production Operations EP and was seeking to consult with Relevant Persons. Santos asked River and Reef to advise Santos by 17 June 2024 if it considered it may be a Relevant Person and what functions, interests or activities it has that may be affected by the production operations activities. [Con-4980]
- The email included links to the Barossa Production Operations Activity Booklet and NOPSEMA's EP consultation information for the community. Santos advised if it did not hear from River and Reef by 17 June 2024, it would assume that it did not have functions, interests or activities that may be affected by the activities, or didn't wish to be consulted for the OEMP and EP.
- On 10 July 2024 Santos emailed River and Reef to advise the consultation period had been completed. Santos advised it considered that consultation had now closed for the purpose of Santos finalising and submitting environment plans for these activities to government regulators for assessment. [Con-5130]
- On 15 October 2024 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that the EP had been submitted to NOPSEMA for assessment, provided a link to the full EP on the NOPSEMA website and Santos' contact details for any queries. The Update also stated that the OEMP would be submitted to the NT-DITT for assessment. [Con- 5982]
- On 23 January 2025 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that, following a request from NOPSEMA to provide further information, Santos had re-submitted the EP with some amendments in December 2024 for further assessment. The Update also re-stated that the OEMP would be submitted to the NT-DITT for assessment. [Con-6036]



### ence

No further correspondence or feedback was received from Ri	ver and Reef.		
Summary of response by Relevant Person	Assessment of merits	Santos' Response Statement	OEMP Refer
No response was received from River and Reef.	Santos considers it has provided sufficient information and a reasonable period of time for consultation.	No response required.	Not applicab
	Santos notes that in June 2024, in response to consultation via email and phone on the Darwin Pipeline Duplication EP (Cwth) and DPD Offshore CEMP, River and Reef had provided no comments.		
	Santos considers Section 25 consultation requirements to have been met.		
Out Description			

### Sail Darwir

Summary of consultation effort:

- On 15 February 2024 Santos emailed Sail Darwin to advise the start of preliminary consultation regarding proposed activities for consultation to be managed under the Coastal Waters OEMP and Barossa Production Operations EP. [Con-4990]
- The email advised that Santos was seeking information to better understand:
  - o if you are from a government Department or agency, how the proposed Production Operations activities may be relevant to your department or agency; and
  - what (if any) functions, interests or activities you or your organisation have that may be affected by the proposed Production Operations activities.
- The email included information on the regulatory process for the activities in Commonwealth and NT jurisdictions and links to a Santos information booklet on the proposed activities and a NOPSEMA brochure on the consultation process and details of how to contact Santos to register as a Relevant Person. The email stated that the consultation phase would commence on 11 March 2024 and close on 9 April 2024.
- On 11 March 2024 emailed Sail Darwin further to the previous correspondence, to advise that it had commenced the consultation phase which would run until 9 April 2024. In addition to the previous information again being provided. Santos provided information on Relevant Persons' entitlements under the regulatory processes, details of how to provide feedback and a reminder of the closing date for consultation. [Con-3793]
- On 4 April 2024 Santos phoned Sail Darwin and left a voice mail message.
- On 2 May 2024, Santos emailed Sail Darwin to advise it was calling to follow up on previous emails to advise it had extended the consultation period until 16 May 2024. In providing this extension of time, Santos advised that the information in the booklet and factsheet had been updated to account for an additional risk associated with the proposed activity. Santos confirmed that consultation will close on the revised date unless Santos hears otherwise from Sail Darwin. [Con-4293]
- On 10 July 2024 Santos emailed Sail Darwin to advise the consultation period had been completed. Santos advised Sail Darwin that it considered consultation had now closed for the purpose of Santos finalising and submitting environment plans for these activities to government regulators for assessment. [Con-5130]
- On 15 October 2024 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that the EP had been submitted to NOPSEMA for assessment, provided a link to the full EP on the NOPSEMA website and Santos' contact details for any queries. The Update also stated that the OEMP would be submitted to the NT-DITT for assessment. [Con- 5982]
- On 23 January 2025 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that, following a request from NOPSEMA to provide further information, Santos had re-submitted the EP with some amendments in December 2024 for further assessment. The Update also re-stated that the OEMP would be submitted to the NT-DITT for assessment. [Con-6036]
- Notwithstanding the information provided and the steps described above, no comments or input were received on the Coastal Waters OEMP from Sail Darwin.

Summary of response by Relevant Person	Assessment of merits	Santos' Response Statement	OEMP Refer
No response was received from Sail Darwin.	Santos considers it has provided sufficient information and a reasonable period of time for consultation. Santos considers Section 25 consultation requirements to have been met.	No response required.	Not applicabl

# Saltwater Cultural Tours

Summary of consultation effort:

- On 9 February 2024 Santos emailed Saltwater Cultural Tours to advise the start of preliminary consultation regarding proposed activities for consultation to be managed under the Coastal Waters OEMP and Barossa Production Operations EP. [Con-3787]
- The email advised that Santos was seeking information to better understand:
  - o if you are from a government Department or agency, how the proposed Production Operations activities may be relevant to your department or agency; and
  - what (if any) functions, interests or activities you or your organisation have that may be affected by the proposed Production Operations activities.
- The email included information on the regulatory process for the activities in Commonwealth and NT jurisdictions and links to a Santos information booklet on the proposed activities and a NOPSEMA brochure on the consultation process and details of how to contact Santos to register as a Relevant Person. The email stated that the consultation phase would commence on 11 March 2024 and close on 9 April 2024.
- On 11 March 2024 Santos emailed Saltwater Cultural Tours further to the previous correspondence, to advise that it had commenced the consultation phase which would run until 9 April 2024. In addition to the previous information again being provided, Santos provided information on Relevant Persons' entitlements under the regulatory processes, details of how to provide feedback and a reminder of the closing date for consultation. [Con-3793]
- On 4 April 2024 Santos phoned Saltwater Cultural Tours and left a voice mail message.
- On 2 May 2024, Santos emailed Saltwater Cultural Tours to advise it was calling to follow up on previous emails and to advise it had extended the consultation period until 16 May 2024. In providing this extension of time, Santos advised that the information in the booklet and factsheet had been updated to account for an additional risk associated with the proposed activity. Santos confirmed that consultation will close on the revised date unless Santos hears otherwise from Saltwater Cultural Tours. [Con-4294]
- On 10 July 2024 Santos emailed Saltwater Cultural Tours to advise the consultation period had been completed. Santos advised Saltwater Cultural Tours that it considered consultation had now closed for the purpose of Santos finalising and submitting environment plans for these activities to government regulators for assessment. [Con-5130]
- On 15 October 2024 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that the EP had been submitted to NOPSEMA for assessment, provided a link to the full EP on the NOPSEMA website and Santos' contact details for any queries. The Update also stated that the OEMP would be submitted to the NT-DITT for assessment. [Con- 5982]
- On 23 January 2025 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that, following a request from NOPSEMA to provide further information, Santos had re-submitted the EP with some amendments in December 2024 for further assessment. The Update also re-stated that the OEMP would be submitted to the NT-DITT for assessment. [Con-6036]



# ence

le.

<ul> <li>Notwithstanding the information provided and the steps described above, no comments or input were received on the Coastal Waters OEMP from Saltwater Cultural Tours.</li> </ul>				
Summary of response by Relevant Person	Assessment of merits	Santos' Response Statement	OEMP Refer	
No response was received from Saltwater Cultural Tours.	Santos considers it has provided sufficient information and a reasonable period of time for consultation. Santos considers Section 25 consultation requirements to have been met.	No response required.	Not applicabl	



rence

ole.

## Sea Darwin

Summary of consultation effort:

- On 15 February 2024 Santos emailed Sea Darwin to advise the start of preliminary consultation regarding proposed activities for consultation to be managed under the Coastal Waters OEMP and Barossa Produc
- The email advised that Santos was seeking information to better understand:
  - o if you are from a government Department or agency, how the proposed Production Operations activities may be relevant to your department or agency; and
  - what (if any) functions, interests or activities you or your organisation have that may be affected by the proposed Production Operations activities.
- The email included information on the regulatory process for the activities in Commonwealth and NT jurisdictions and links to a Santos information booklet on the proposed activities and a NOPSEMA brochure on the consultation process and details of how to contact Santos to register as a Relevant Person. The email stated that the consultation phase would commence on 11 March 2024 and close on 9 April 2024.
- On 11 March 2024 Santos emailed Sea Darwin further to the previous correspondence, to advise that it had commenced the consultation phase which would run until 9 April 2024. In addition to the previous information again being provided, Santos provided information on Relevant Persons' entitlements under the regulatory processes, details of how to provide feedback and a reminder of the closing date for consultation. [Con-3793]
- On 4 April 2024 Santos phoned Sea Darwin and spoke to a company representative regarding email sent on 15 February and 11 March. They requested the emails be resent and this was done the same day. [Con-4276]
- On 5 April 2024, Sea Darwin emailed Santos and acknowledged receipt of emails below and indicated if there was no response by 9 April assume that Sea Darwin had no input. [Con-4277]
- On 7 May 2024, Santos emailed Sea Darwin to advise it had extended the consultation period 21 May 2024. In providing this extension of time, Santos advised that the information in the booklet and factsheet had been updated to account for an additional risk associated with the proposed activity. Santos confirmed that consultation will close on the revised date unless Santos hears otherwise from Sea Darwin. [Con-4318]
- On 10 July 2024 Santos emailed Sea Darwin to advise the consultation period had been completed. Santos advised Sea Darwin that it considered consultation had now closed for the purpose of Santos finalising and submitting environment plans for these activities to government regulators for assessment. [Con-5130]
- On 15 October 2024 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that the EP had been submitted to NOPSEMA for assessment, provided a link to the full EP on the NOPSEMA website and Santos' contact details for any queries. The Update also stated that the OEMP would be submitted to the NT-DITT for assessment. [Con- 5982]
- On 23 January 2025 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that, following a request from NOPSEMA to provide further information. Santos had re-submitted the EP with some amendments in December 2024 for further assessment. The Update also re-stated that the OEMP would be submitted to the NT-DITT for assessment. [Con-6036]

### Notwithstanding the information provided and the steps described above, no comments or input were received on the Coastal Waters OEMP from Sea Darwin.

No response was received from Sea Darwin.Santos considers it has provided sufficient information and a reasonable period of time for consultation.No response required.Not applied	Summary of response by Relevant Person	Assessment of merits	Santos' Response Statement	OEMP Refer
Santos considers Section 25 consultation requirements to have been	No response was received from Sea Darwin.	Santos considers it has provided sufficient information and a reasonable period of time for consultation. Santos considers Section 25 consultation requirements to have been	No response required.	Not applicabl
met.		met.		

# Shoal Bay Sportfishing Tours

Summary of consultation effort:

- On 9 February 2024 Santos emailed Shoal Bay Sportfishing Tours to advise the start of preliminary consultation regarding proposed activities for consultation to be managed under the Coastal Waters OEMP and Barossa Production Operations EP. [Con-3787]
- The email advised that Santos was seeking information to better understand:
  - o if you are from a government Department or agency, how the proposed Production Operations activities may be relevant to your department or agency; and
  - what (if any) functions, interests or activities you or your organisation have that may be affected by the proposed Production Operations activities.
- The email included information on the regulatory process for the activities in Commonwealth and NT jurisdictions and links to a Santos information booklet on the proposed activities and a NOPSEMA brochure on the consultation process and details of how to contact Santos to register as a Relevant Person. The email stated that the consultation phase would commence on 11 March 2024 and close on 9 April 2024.
- On 11 March 2024 Santos emailed Shoal Bay Sportfishing Tours further to the previous correspondence, to advise that it had commenced the consultation phase which would run until 9 April. In addition to the previous information again being provided, Santos provided information on Relevant Persons' entitlements under the regulatory processes, details of how to provide feedback and a reminder of the closing date for consultation. [Con-3793]
- On 4 April 2024 Santos phoned and spoke with a company representative who indicated there was no impact on its activity from Barossa Production Operations EP activities and did not want to respond, requesting to be removed from Santos' contact list.
- On 10 July 2024 Santos emailed Shoal Bay Sportfishing Tours to advise the consultation period had been completed. Santos advised it considered that consultation had now closed for the purpose of Santos finalising and submitting environment plans for these activities to government regulators for assessment. [Con-5130]
- On 15 October 2024 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that the EP had been submitted to NOPSEMA for assessment, provided a link to the full EP on the NOPSEMA website and Santos' contact details for any queries. The Update also stated that the OEMP would be submitted to the NT-DITT for assessment. [Con- 5982]
- On 23 January 2025 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that, following a request from NOPSEMA to provide further information, Santos had re-submitted the EP with some amendments in December 2024 for further assessment. The Update also re-stated that the OEMP would be submitted to the NT-DITT for assessment. [Con-6036]

### Notwithstanding the information provided and the steps described above, no comments or input were received on the Coastal Waters OEMP from Shoal Bay Sportfishing Tours.

Summary of response by Relevant Person	Assessment of merits	Santos' Response Statement	OEMP Refer
No response was received from Shoal Bay Sportfishing Tours.	Santos considers it has provided sufficient information and a reasonable period of time for consultation. Santos considers Section 25 consultation requirements to have been met.	No response required.	Not applicabl

# Skippers at Dundee

On 10 June 2024 Santos emailed Skippers at Dundee regarding proposed activities for consultation to be managed under the Coastal Waters OEMP and Barossa Production Operations EP and was seeking to consult with Relevant Persons. Santos asked Skippers at Dundee to advise Santos by 17 June 2024 if it considered it may be a Relevant Person and what functions, interests or activities it has that may be affected by the production operations activities. [Con-4982]

# Santos

tion	Operations	s FP	[Con-4981]
	oporation		

ence

le.

rence

- The email included links to the Barossa Production Operations Activity Booklet and NOPSEMA's EP consultation information for the community. Santos advised if it did not hear from Skippers by 17 June 2024, it would assume that it did not have functions, interests or activities that may be affected by the activities, or didn't wish to be consulted for the OEMP and EP.
- On 4 July 2024 Santos phoned Skippers at Dundee and left a message with a staff member.
- On 10 July 2024 Santos emailed Skippers at Dundee to advise the consultation period had been completed. Santos advised it consultation had now closed for the purpose of Santos finalising and submitting environment plans for these activities to government regulators for assessment. [Con-5130]
- On 15 October 2024 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that the EP had been submitted to NOPSEMA for assessment, provided a link to the full EP on the NOPSEMA website and Santos' contact details for any queries. The Update also stated that the OEMP would be submitted to the NT-DITT for assessment. [Con- 5982]
- On 23 January 2025 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that, following a request from NOPSEMA to provide further information, Santos had re-submitted the EP with some amendments in December 2024 for further assessment. The Update also re-stated that the OEMP would be submitted to the NT-DITT for assessment. [Con-6036]

No further correspondence or feedback was received from Skippers at Dundee

Summary of response by Relevant Person	Assessment of merits	Santos' Response Statement	OEMP Refe
No response was received from Skippers at Dundee.	Santos considers it has provided sufficient information and a reasonable period of time for consultation. Santos considers Section 25 consultation requirements to have been met.	No response required.	Not applicab

## Spring Tide Safaris

Summary of consultation effort:

- On 9 February 2024 Santos emailed Spring Tide Safaris to advise the start of preliminary consultation regarding proposed activities for consultation to be managed under the Coastal Waters OEMP and Barossa Production Operations EP. [Con-3787]
- The email advised that Santos was seeking information to better understand:
  - o if you are from a government Department or agency, how the proposed Production Operations activities may be relevant to your department or agency; and
  - what (if any) functions, interests or activities you or your organisation have that may be affected by the proposed Production Operations activities. 0
- The email included information on the regulatory process for the activities in Commonwealth and NT jurisdictions and links to a Santos information booklet on the proposed activities and a NOPSEMA brochure on the consultation process and details of how to contact Santos to register as a Relevant Person. The email stated that the consultation phase would commence on 11 March 2024 and close on 9 April 2024.
- On 11 March 2024 Santos emailed Spring Tide Safaris further to the previous correspondence, to advise that it had commenced the consultation phase which would run until 9 April 2024. In addition to the previous information again being provided, Santos provided information on Relevant Persons' entitlements under the regulatory processes, details of how to provide feedback and a reminder of the closing date for consultation. [Con-3793]
- On 4 April 2024 Santos phoned Spring Tide Safaris and left a voice mail message.
- On 2 May 2024, Santos emailed Spring Tide Safaris to advise it was calling to follow up on previous emails and to advise it had extended the consultation period until 16 May 2024. In providing this extension of time, Santos advised that the information in the booklet and factsheet had been updated to account for an additional risk associated with the proposed activity. Santos confirmed that consultation will close on the revised date unless Santos hears otherwise from Spring Tide Safaris. [Con-4295]
- On 10 July 2024 Santos emailed Spring Tide Safaris to advise the consultation period had been completed. Santos advised Spring Tide Safaris for its input and advised it considered that consultation had now closed for the purpose of Santos finalising and submitting environment plans for these activities to government regulators for assessment. [Con-5130]
- On 15 October 2024 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that the EP had been submitted to NOPSEMA for assessment, provided a link to the full EP on the NOPSEMA website and Santos' contact details for any queries. The Update also stated that the OEMP would be submitted to the NT-DITT for assessment. [Con- 5982]
- On 23 January 2025 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that, following a request from NOPSEMA to provide further information, Santos had re-submitted the EP with some amendments in December 2024 for further assessment. The Update also re-stated that the OEMP would be submitted to the NT-DITT for assessment. [Con-6036]
- Notwithstanding the information provided and the steps described above, no comments or input were received on the Coastal Waters OEMP from Spring Tide Safaris

Summary of response by Relevant Person	Assessment of merits	Santos' Response Statement	OEMP Refer
No response was received from Spring Tide Safaris.	Santos considers it has provided sufficient information and a reasonable period of time for consultation. Santos considers Section 25 consultation requirements to have been met.	No response required.	Not applicabl
Streeter Cruises			

# Summary of consultation effort:

- On 9 February 2024 Santos emailed Streeter Cruises to advise the start of preliminary consultation regarding proposed activities for consultation to be managed under the Coastal Waters OEMP and Barossa Production Operations EP. [Con-3787]
- The email advised that Santos was seeking information to better understand:
  - o if you are from a government Department or agency, how the proposed Production Operations activities may be relevant to your department or agency; and
  - what (if any) functions, interests or activities you or your organisation have that may be affected by the proposed Production Operations activities. 0
- The email included information on the regulatory process for the activities in Commonwealth and NT jurisdictions and links to a Santos information booklet on the proposed activities and a NOPSEMA brochure on the consultation process and details of how to contact Santos to register as a Relevant Person. The email stated that the consultation phase would commence on 11 March 2024 and close on 9 April 2024.
- On 11 March 2024 Santos emailed Streeter Cruises further to the previous correspondence, to advise that it had commenced the consultation phase which would run until 9 April 2024. In addition to the previous information again being provided, Santos provided information on Relevant Persons' entitlements under the regulatory processes, details of how to provide feedback and a reminder of the closing date for consultation. [Con-3793]
- On 4 April 2024 Santos phoned Streeter Cruises and left a voice mail message.



rence

le.

ence

le

- On 2 May 2024, Santos emailed Streeter Cruises to advise it was calling to follow up on previous emails and to advise it had extended the consultation period until 16 May 2024. In providing this extension of time, Santos advised that the information in the booklet and factsheet had been updated to account for an additional risk associated with the proposed activity. Santos confirmed that consultation will close on the revised date unless Santos hears otherwise from Streeter Cruises. [Con-4296]
- On 10 July 2024 Santos emailed Streeters Cruises to advise the consultation period had been completed. Santos advised Streeter Cruises that it considered consultation had now closed for the purpose of Santos finalising and submitting environment plans for these activities to government regulators for assessment. [Con-5130]
- On 15 October 2024 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that the EP had been submitted to NOPSEMA for assessment, provided a link to the full EP on the NOPSEMA website and Santos' contact details for any queries. The Update also stated that the OEMP would be submitted to the NT-DITT for assessment. [Con- 5982]
- On 23 January 2025 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that, following a request from NOPSEMA to provide further information, Santos had re-submitted the EP with some amendments in December 2024 for further assessment. The Update also re-stated that the OEMP would be submitted to the NT-DITT for assessment. [Con-6036]
- Notwithstanding the information provided and the steps described above, no comments or input were received on the Coastal Waters OEMP. from Streeter Cruises.

Summary of response by Relevant Person	Assessment of merits	Santos' Response Statement	OEMP Refe
No response was received from Streeter Cruises.	Santos considers it has provided sufficient information and a reasonable period of time for consultation. Santos considers Section 25 consultation requirements to have been met.	No response required.	Not applicat

# Territory Guided Fishing

Summary of consultation effort:

- On 9 February 2024 Santos emailed Territory Guided Fishing to advise the start of preliminary consultation regarding proposed activities for consultation to be managed under the Coastal Waters OEMP and Barossa Production Operations EP. [Con-3787]
- The email advised that Santos was seeking information to better understand:
  - o if you are from a government Department or agency, how the proposed Production Operations activities may be relevant to your department or agency; and
  - what (if any) functions, interests or activities you or your organisation have that may be affected by the proposed Production Operations activities.
- The email included information on the regulatory process for the activities in Commonwealth and NT jurisdictions and links to a Santos information booklet on the proposed activities and a NOPSEMA brochure on the consultation process and details of how to contact Santos to register as a Relevant Person. The email stated that the consultation phase would commence on 11 March 2024 and close on 9 April 2024.
- On 11 March 2024 Santos emailed Territory Guided Fishing further to the previous correspondence, to advise that it had commenced the consultation phase which would run until 9 April 2024. In addition to the previous information again being provided, Santos provided information on Relevant Persons' entitlements under the regulatory processes, details of how to provide feedback and a reminder of the closing date for consultation. [Con-3793]
- On 4 April 2024 Santos phoned Territory Guided Fishing and left a voice mail message.
- On 2 May 2024, Santos emailed Territory Guided Fishing to advise it was calling to follow up on previous emails and to advise it had extended the consultation period until 16 May 2024. In providing this extension of time, Santos advised that the information in the booklet and factsheet had been updated to account for an additional risk associated with the proposed activity. Santos confirmed that consultation will close on the revised date unless Santos hears otherwise from Territory Guided Fishing. [Con-4297]
- On 10 July 2024 Santos emailed Territory Guided Fishing to advise the consultation period had been completed. Santos advised Territory Guided Fishing that it considered consultation had now closed for the purpose of Santos finalising and submitting environment plans for these activities to government regulators for assessment. [Con-5130]
- On 15 October 2024 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that the EP had been submitted to NOPSEMA for assessment, provided a link to the full EP on the NOPSEMA website and Santos' contact details for any queries. The Update also stated that the OEMP would be submitted to the NT-DITT for assessment. [Con- 5982]
- On 23 January 2025 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that, following a request from NOPSEMA to provide further information, Santos had re-submitted the EP with some amendments in December 2024 for further assessment. The Update also re-stated that the OEMP would be submitted to the NT-DITT for assessment. [Con-6036]
- No further correspondence or feedback was received from Territory Guided Fishing.

Summary of response by Relevant Person	Assessment of merits	Santos' Response Statement	OEMP Refer
No response was received from Territory Guided Fishing.	Santos considers it has provided sufficient information and a reasonable period of time for consultation. Santos considers Section 25 consultation requirements to have been met.	No response required.	Not applicab



erence

ble.
#### **Tiwi Island Adventures**

#### Summary of consultation effort:

- On 9 February 2024 Santos emailed Tiwi Island Adventures to advise the start of preliminary consultation regarding proposed activities for consultation to be managed under the Coastal Waters OEMP and Barossa Production Operations EP. [Con-3787]
- The email advised that Santos was seeking information to better understand:
  - o if you are from a government Department or agency, how the proposed Production Operations activities may be relevant to your department or agency; and
  - what (if any) functions, interests or activities you or your organisation have that may be affected by the proposed Production Operations activities.
- The email included information on the regulatory process for the activities in Commonwealth and NT jurisdictions and links to a Santos information booklet on the proposed activities and a NOPSEMA brochure on the consultation process and details of how to contact Santos to register as a Relevant Person. The email stated that the consultation phase would commence on 11 March 2024 and close on 9 April 2024.
- On 11 March 2024 Santos emailed Tiwi Island Adventures further to the previous correspondence, to advise that it had commenced the consultation phase which would run until 9 April 2024. In addition to the previous information again being provided, Santos provided information on Relevant Persons' entitlements under the regulatory processes, details of how to provide feedback and a reminder of the closing date for consultation. [Con-3793]
- On 4 April 2024 Santos phoned Tiwi Island Adventures and left a message with a company representative.
- On 3 May 2024, Santos emailed Tiwi Islands Adventures to advise it was calling to follow up on previous emails and to advise it had extended the consultation period until 17 May 2024. In providing this extension of time, Santos advised that the information in the booklet and factsheet had been updated to account for an additional risk associated with the proposed activity. Santos confirmed that consultation will close on the revised date unless Santos hears otherwise from Tiwi Island Adventures. [Con-4304]
- On 10 July 2024 Santos emailed Tiwi Island Adventures to advise the consultation period had been completed. Santos advised Tiwi Island Adventures that it considered consultation had now closed for the purpose of Santos finalising and submitting environment plans for these activities to government regulators for assessment. [Con-5130]
- On 15 October 2024 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that the EP had been submitted to NOPSEMA for assessment, provided a link to the full EP on the NOPSEMA website and Santos' contact details for any queries. The Update also stated that the OEMP would be submitted to the NT-DITT for assessment. [Con- 5982]
- On 23 January 2025 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that, following a request from NOPSEMA to provide further information, Santos had re-submitted the EP with some amendments in December 2024 for further assessment. The Update also re-stated that the OEMP would be submitted to the NT-DITT for assessment. [Con-6036]
- No further correspondence or feedback was received from Tiwi Island Adventures.

Summary of response by Relevant Person	Assessment of merits	Santos' Response Statement	OEMP Refer
No response was received from Tiwi Island Adventures.	Santos considers it has provided sufficient information and a reasonable period of time for consultation. Santos considers Section 25 consultation requirements to have been met.	No response required.	Not applicabl

#### Tiwi Island Retreat

Summary of consultation effort:

- On 9 February 2024 Santos emailed Tiwi Island Retreat to advise the start of preliminary consultation regarding proposed activities for consultation to be managed under the Coastal Waters OEMP and Barossa Production Operations EP. [Con-3787]
- The email advised that Santos was seeking information to better understand:
  - o if you are from a government Department or agency, how the proposed Production Operations activities may be relevant to your department or agency; and
  - what (if any) functions, interests or activities you or your organisation have that may be affected by the proposed Production Operations activities.
- The email included information on the regulatory process for the activities in Commonwealth and NT jurisdictions and links to a Santos information booklet on the proposed activities and a NOPSEMA brochure on the consultation process and details of how to contact Santos to register as a Relevant Person. The email stated that the consultation phase would commence on 11 March2024 and close on 9 April 2024.
- On 11 March 2024 Santos emailed Tiwi Island Retreat further to the previous correspondence, to advise that it had commenced the consultation phase which would run until 9 April 2024. In addition to the previous information again being provided, Santos provided information on Relevant Persons' entitlements under the regulatory processes, details of how to provide feedback and a reminder of the closing date for consultation. [Con-3793]
- On 4 April 2024 Santos phoned Tiwi Island Retreat and spoke a team member regarding consultation for Production Operations activities who confirmed that emails sent on 9 February and 11 March had been received.
- On 5 April 2024, Tiwi Island Retreat emailed Santos and advised it does not have any comment or input for the Barossa Production Operations EP consultation process. [Con-4983]
- On 7 May 2024, Santos emailed Tiwi Island Retreat to advise it had extended the consultation period until 21 May 2024. In providing this extension of time, Santos advised that the information in the booklet and factsheet had been updated to account for an additional risk associated with the proposed activity. Santos confirmed that consultation will close on the revised date unless Santos hears otherwise from Tiwi Island Retreat. [Con-4322]
- On 10 July 2024 Santos emailed Tiwi Island Retreat to advise the consultation period had been completed. Santos advised it considered that consultation had now closed for the purpose of Santos finalising and submitting environment plans for these activities to government regulators for assessment. [Con-5130]
- On 15 October 2024 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that the EP had been submitted to NOPSEMA for assessment, provided a link to the full EP on the NOPSEMA website and Santos' contact details for any queries. The Update also stated that the OEMP would be submitted to the NT-DITT for assessment. [Con- 5982]
- On 23 January 2025 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that, following a request from NOPSEMA to provide further information, Santos had re-submitted the EP with some amendments in December 2024 for further assessment. The Update also re-stated that the OEMP would be submitted to the NT-DITT for assessment. [Con-6036]
- No further correspondence or feedback was received from Tiwi Island Retreat

Summary of response by Relevant Person	Assessment of merits	Santos' Response Statement	OEMP Refere
Tiwi Island Retreat responded that it had no comment or input for the Coastal Waters OEMP.	Santos considers it has provided sufficient information and a reasonable period of time for consultation. Santos considers Section 25 consultation requirements to have been met.	No response required.	Not applicable
Top End Barra Eishing Tours			

### Santos

ence

le

ence

е

#### Summary of consultation effort:

- On 9 February 2024 Santos emailed Top End Barra Fishing Tours to advise the start of preliminary consultation regarding proposed activities for consultation to be managed under the Coastal Waters OEMP and Barossa Production Operations EP. [Con-3787]
- The email advised that Santos was seeking information to better understand:
  - o if you are from a government Department or agency, how the proposed Production Operations activities may be relevant to your department or agency; and
  - what (if any) functions, interests or activities you or your organisation have that may be affected by the proposed Production Operations activities.
- The email included information on the regulatory process for the activities in Commonwealth and NT jurisdictions and links to a Santos information booklet on the proposed activities and a NOPSEMA brochure on the consultation process and details of how to contact Santos to register as a Relevant Person. The email stated that the consultation phase would commence on 11 March 2024 and close on 9 April 2024.
- On 11 March 2024, Santos emailed Top End Barra Fishing Tours further to the previous correspondence, to advise that it had commenced the consultation phase which would run until 9 April 2024. In addition to the previous information again being provided, Santos provided information on Relevant Persons' entitlements under the regulatory processes, details of how to provide feedback and a reminder of the closing date for consultation. [Con-3793]
- On 4 April 2024 Santos phoned Top End Barra Fishing Tours and left a voice mail message.
- On 2 May 2024, Santos emailed Top End Barra Fishing Tours to advise it was calling to follow up on previous emails and to advise it had extended the consultation period until 16 May 2024. In providing this extension of time, Santos advised that the information in the booklet and factsheet had been updated to account for an additional risk associated with the proposed activity. Santos confirmed that consultation will close on the revised date unless Santos hears otherwise from Top End Barra Fishing Tours. [Con-4984]
- On 10 July 2024 Santos emailed Top End Barra Fishing Tours to advise the consultation period had been completed. Santos advised Top End Barra Fishing Tours that it considered consultation had now closed for the purpose of Santos finalising and submitting environment plans for these activities to government regulators for assessment. [Con-5130]
- On 15 October 2024 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that the EP had been submitted to NOPSEMA for assessment, provided a link to the full EP on the NOPSEMA website and Santos' contact details for any queries. The Update also stated that the OEMP would be submitted to the NT-DITT for assessment. [Con- 5982]
- On 23 January 2025 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that, following a request from NOPSEMA to provide further information, Santos had re-submitted the EP with some amendments in December 2024 for further assessment. The Update also re-stated that the OEMP would be submitted to the NT-DITT for assessment. [Con-6036]
- No further correspondence or feedback was received from Top End Barra Fishing Tours.

	Santos considers Section 25 consultation requirements to have been met.		
No response was received from Top End Barra Fishing Tours.	Santos considers it has provided sufficient information and a reasonable period of time for consultation.	No response required.	Not applicabl
Summary of response by Relevant Person	Assessment of merits	Santos' Response Statement	OEMP Refer

#### **Top End Seafaris**

Summary of consultation effort:

- On 9 February 2024 Santos emailed Top End Seafaris to advise the start of preliminary consultation regarding proposed activities for consultation to be managed under the Coastal Waters OEMP and Barossa Production Operations EP. [Con-3787]
- The email advised that Santos was seeking information to better understand:
  - if you are from a government Department or agency, how the proposed Production Operations activities may be relevant to your department or agency; and
  - what (if any) functions, interests or activities you or your organisation have that may be affected by the proposed Production Operations activities.
- The email included information on the regulatory process for the activities in Commonwealth and NT jurisdictions and links to a Santos information booklet on the proposed activities and a NOPSEMA brochure on the consultation process and details of how to contact Santos to register as a Relevant Person. The email stated that the consultation phase would commence on 11 March 2024 and close on 9 April 2024.
- On 11 March 2024. Santos emailed Top End Seafaris further to the previous correspondence, to advise that it had commenced the consultation phase which would run until 9 April 2024. In addition to the previous information again being provided. Santos provided information on Relevant Persons' entitlements under the regulatory processes, details of how to provide feedback and a reminder of the closing date for consultation. [Con-3793]
- On 4 April 2024 Santos phoned Top End Seafaris and left a voice mail message.
- On 2 May 2024, Santos emailed Top End Seafaris to advise it was calling to follow up on previous emails and to advise it had extended the consultation period until 16 May 2024. In providing this extension of time. Santos advised that the information in the booklet and factsheet had been updated to account for an additional risk associated with the proposed activity. Santos confirmed that consultation will close on the revised date unless Santos hears otherwise from Top End Seafaris, [Con-4299]
- On 10 July 2024 Santos emailed Top End Seafaris to advise the consultation period had been completed. Santos advised Top End Seafaris that it considered consultation had now closed for the purpose of Santos finalising and submitting environment plans for these activities to government regulators for assessment. [Con-5130]
- On 15 October 2024 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that the EP had been submitted to NOPSEMA for assessment, provided a link to the full EP on the NOPSEMA website and Santos' contact details for any queries. The Update also stated that the OEMP would be submitted to the NT-DITT for assessment. [Con- 5982]
- On 23 January 2025 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that, following a request from NOPSEMA to provide further information. Santos had re-submitted the EP with some amendments in December 2024 for further assessment. The Update also re-stated that the OEMP would be submitted to the NT-DITT for assessment. [Con-6036]
- No further correspondence or feedback was received from Top End Seafaris.

Summary of response by Relevant Person	Assessment of merits	Santos' Response Statement	OEMP Refer
No response was received from Top End Seafaris.	Santos considers it has provided sufficient information and a reasonable period of time for consultation. Santos considers Section 25 consultation requirements to have been	No response required.	Not applicabl
	met.		

### Santos

ence

ence

#### **Yknot Fishing Charters**

Summary of consultation effort:

- On 9 February 2024 Santos emailed Yknot Fishing Charters to advise the start of preliminary consultation regarding proposed activities for consultation to be managed under the Coastal Waters OEMP and Barossa Production Operations EP. [Con-3787]
- The email advised that Santos was seeking information to better understand:
  - o if you are from a government Department or agency, how the proposed Production Operations activities may be relevant to your department or agency; and
  - what (if any) functions, interests or activities you or your organisation have that may be affected by the proposed DPD activities. 0
- The email included information on the regulatory process for the activities in Commonwealth and NT jurisdictions and links to a Santos information booklet on the proposed activities and a NOPSEMA brochure on the consultation process and details of how to contact Santos to register as a Relevant Person. The email stated that the consultation phase would commence 11 March 2024 and close on 9 April 2024.
- On 11 March 2024 Santos emailed Yknot Fishing Charters further to the previous correspondence, to advise that it had commenced the consultation phase which would run until 9 April 2024. In addition to the previous information again being provided, Santos provided information on Relevant Persons' entitlements under the regulatory processes, details of how to provide feedback and a reminder of the closing date for consultation. In the email Santos stated that, if input is not received by this date Santos will infer this means you do not want Santos to consult with you further on the Productions Operations EP [Con-3793]
- On 4 April 2024 Santos phoned Yknot Fishing Charters and left a voice mail message
- On 2 May 2024, Santos emailed Yknot Fishing Charters to advise it was calling to follow up on previous emails and to advise it was calling to follow up on previous emails and to advise it had extended the consultation period until 16 May 2024. In providing this extension of time, Santos advised that the information in the booklet and factsheet had been updated to account for an additional risk associated with the proposed activity. Santos confirmed that consultation will close on the revised date unless Santos hears otherwise from Yknot Fishing Charters. [Con-4300]
- On 10 July 2024 Santos emailed Yknot Fishing Charters to advise the consultation period had been completed. Santos advised Yknot Fishing Charters that it considered consultation had now closed for the purpose of Santos finalising and submitting environment plans for these activities to government regulators for assessment. [Con-5130]
- On 15 October 2024 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that the EP had been submitted to NOPSEMA for assessment, provided a link to the full EP on the NOPSEMA website and Santos' contact details for any queries. The Update also stated that the OEMP would be submitted to the NT-DITT for assessment. [Con- 5982]
- On 23 January 2025 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. The Update advised that, following a request from NOPSEMA to provide further information, Santos had re-submitted the EP with some amendments in December 2024 for further assessment. The Update also re-stated that the OEMP would be submitted to the NT-DITT for assessment. [Con-6036]
- No further correspondence or feedback was received from Yknot Fishing Charters.

Summary of response by Relevant Person	Assessment of merits	Santos' Response Statement	OEMP Refer
No response was received from Yknot Fishing Charters.	Santos considers it has provided sufficient information and a reasonable period of time for consultation.	No response required.	Not applicable
	Santos considers Section 25 consultation requirements to have been met.		

### Santos

#### ence

le.

### 4.8.13 Other Relevant Persons

#### Table 4-23: Consultation Summary Table – Other Relevant Persons

Section 25(1)(d) of the OPGGS(E)R: Persons or organisations whose functions, interests or activities may be affected by the activities to be carried out under the environment plan

#### Autoridade Nacional do Petróleo e Minerais (ANP Timor-Leste)

Summary of consultation effort:

- On 26 June 2024 Santos emailed ANP in its capacity as having responsibility for petroleum environmental matters in Timor-Leste. Santos advised that it was currently consulting on the Barossa Production Oper Petroleum and Greenhouse Gas Storage (Environment) Regulations (Cth) and was specifically seeking to clarify the Authority's oil spill notification requirements for inclusion in the appropriate emergency response.
- Santos provided the information booklet on the proposed activities, with specific reference to the section on oil spill risk and management and requested the ANP respond on its oil spill notification requirements a
  information will be used for the development of the EP for production operations activity in Commonwealth waters, which will be assessed by NOPSEMA.
- On 26 June 2024 ANP emailed Santos to acknowledge the email and advised it would be considered. [Con-4963]
- On 1 July 2024 ANP emailed Santos requesting the information that had been provided via email was re-sent as an official letter and, upon receipt, the ANP would respond accordingly. [Con-4964] Santos provid 4965]
- On 1 July 2024 ANP emailed Santos' confirming that it would review the letter provided by Santos earlier the same day. [Con-5084]
- On 17 July 2024 Santos emailed the ANP to request it provide any comments in relation to the environmental management of the Barossa Project by 26 July, if they wished to have them included in this OEMP.
- On 18 July 2024 ANP emailed Santos to advise it was in the process of finalising its response letter and would send in due course. [Con-5141]
- On 2 August 2024 ANP wrote to Santos to advise contact details within Timor-Leste for spill notifications and to request additional information on spill modelling, mitigation measures and the notification process.
- On 7 August Santos wrote to ANP and provided the details requested by the ANP in their letter of 2 August 2024 and advised that it considered consultation had now closed for the purpose of Santos finalising ar government regulators for assessment. [Con--5278]
- On 7 August 2024 ANP emailed and thanked Santos for the information and advised they would contact Santos if further information or clarification was required. [Con 5279
- On 15 October 2024 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP. 1 NOPSEMA for assessment, provided a link to the full EP on the NOPSEMA website and Santos' contact details for any queries. The Update also stated that the OEMP would be submitted to the NT-DITT for ass
- On 23 January 2025 Santos emailed the Barossa Project Quarterly Update to all stakeholders on its distribution list, including all those consulted during preparation of the Production Operations EP and OEMP.
   NOPSEMA to provide further information, Santos had re-submitted the EP with some amendments in December 2024 for further assessment. The Update also re-stated that the OEMP would be submitted to the
- No further correspondence or feedback was received from ANP.

Summary of response by Relevant Person	Assessment of merits
ANP provided contact details within Timor-Leste for spill notifications and requested additional information on spill modelling, spill mitigation measures and the spill notification process	This response does not raise an objection or claim about the adverse impact of each activity to which the Coarelates. With regard to the location of the proposed GEP operations activities in NT Coastal Waters, there are no impact
	activities that may affect the functions, interests or activities of any international persons.
	The worst-case credible spill scenario modelled for the EMBA for the Coastal Waters OEMP also shows there from unplanned activities that may affect the functions, interests or activities of any international persons.
	The response is addressed in the Barossa Production Operations EP which does have an EMBA that extends waters.



ations EP in accordance with section 25 of the Offshore use procedures for the Barossa Project. [Con-4962] at its earliest convenience. Santos advised ANP that the			
led the letter, via emain	ed the letter, via email, to the ANP the same day. [Con-		
[Con-5114]			
[Con- 5258] nd submitting environment plans for these activities to			
The Update advised that the EP had been submitted to sessment. [Con- 5982] The Update advised that, following a request from			
NT-DITT for assessment. [Con-6036]			
	Santos' Response Statement	OEMP Reference	
astal Waters OEMP	No response	Not applicable	
acts from planned			
e are no impacts			
s into international			

# 5. Impact and Risk Assessment Methodology

### **OPGGS(E)R 2023 Requirements**

### Section 21. Environmental assessment

Evaluation of environmental impacts and risks

(5) The environment plan must include:

- a. details of the environmental impacts and risks of the activity; and
- b. an evaluation of all the environmental impacts and risks, appropriate to the nature and scale of each impact or risk; and
- c. details of the control measures that will be used to reduce the impacts and risks of the activity to as low as reasonably practicable and an acceptable level.

(6) To avoid doubt, the evaluation mentioned in paragraph (5)(b) must evaluate all of the environmental impacts and risks arising directly or indirectly from:

- a. all operations of the activity; and
- b. any potential emergency conditions, whether resulting from an accident or any other cause.

Environmental impact and risk assessment refers to a process whereby planned and unplanned events that will or may occur during an activity are assessed for their impacts on the environment (as defined in Section 5 of the OPGGS(E)R) at a defined location and specified period. In addition, unplanned events are assessed based on their likelihood of occurrence, which defines their risk level.

Santos has undertaken environmental impact and risk assessments for the planned events – including any routine, non-routine, and contingency activities – and unplanned events in accordance with the OPGGS(E)R.

This section of the GEP Coastal Waters OEMP provides information on the environmental impact and risk assessment approach, specifically:

- terminology used
- summary of the approach used.

The process used to identify, analyse, and evaluate environmental impacts and risks is documented in Santos' Offshore Division Environmental Hazard Identification and Assessment Guideline.

### 5.1 Impact and Risk Assessment Methodology

Common terms applied during the environmental impact and risk assessment process, and used in this GEP Coastal Waters OEMP, are defined in Table 5-1.

### Table 5-1: Impact and risk assessment terms

Term	Definition
Acceptability	Determined for both impacts and risks. Acceptability of events is in part determined by the consequence of the impact following management controls. Acceptability of unplanned events is in part determined from its risk ranking following management controls. For both impacts and risks, acceptability is also determined from a demonstration of the ALARP principle, consistency with Santos policies, consistency with all applicable legislation, and consideration of information received through consultation when determining management controls.
Activity	Specific tasks and actions undertaken throughout the lifecycle of oil and gas exploration, development, production and decommissioning.
ALARP	As Low as Reasonably Practicable.
	The term refers to reducing impact and risk to a level that is as low as reasonably practicable. In practice, this means showing (through reasoned and supported arguments) that there are no other practicable measures that could reasonably be taken to reduce impacts or risks further (NOPSEMA Guidance Note: ALARP, dated 1/08/2022 (N-04300-GN01660166 A138249); NOPSEMA Guideline: Environment plan decision making guideline, dated 10/01/2024 (N-04750-GL1721 A524696).
Authorised person	Person with the authority to make a decision or take an action. Examples are vessel master, superintendent, supervisor, person-in-charge, company authorised representative, and operations manager.
Control measure	Is defined by the OPGGS(E)R to mean a system, an item of equipment, a person or a procedure that is used as a basis for managing environmental impacts and risks.

Term	Definition
ENVID workshop	Environmental hazard identification workshop.
Environment	<ul> <li>Is under the OPGGS(E)R as:</li> <li>(a) ecosystems and their constituent parts, including people and communities</li> <li>(b) natural and physical resources</li> <li>(c) the qualities and characteristics of locations, places and areas</li> <li>(d) the heritage value of places; and includes</li> <li>(e) the social, economic and cultural features of the matters mentioned in paragraphs (a), (b), (c) and (d).</li> </ul>
Environmental consequence	A consequence is the outcome of an event affecting objectives. Note 1: An event can be one or more occurrences and can have several cases. Note 2: An event can consist of something not happening. ( <i>Reference ISO 73:2009 Risk Vocabulary</i> )
Environmental impact	Defined by the OPGGS(E)R as any change to the environment, whether adverse or beneficial, wholly or partly resulting from an activity.
Environmental objective	An environmental result the company intends to achieve
Environmental risk	Applies to unplanned events. Risk is a function of the likelihood of the unplanned event occurring and the consequence of the environmental impact that arises from that event.
Grossly disproportionate	Where the sacrifice (cost and effort) of implementing a control measure to reduce impact or risk, grossly exceeds the environmental benefit to be gained.
Hazard	A situation with the potential to cause harm.
Impact assessment	The process of determining the consequence of an impact (in terms of the consequence to the environment) arising from a planned or unplanned event over a specified period.
Likelihood	The chance of an unplanned event occurring.
Non-routine planned event	An attribute of the planned activity that may occur or will occur infrequently during the planned activity. A non-routine planned event is intended to occur at the time.
Planned activity	The activity to be undertaken under this GEP Coastal Waters OEMP, including the services, equipment, products, assets, personnel, timing, duration and location and aspect of the activity.
Planned event	An event arising from the activity that is done with intent (i.e., not an unplanned event) and has some level of environmental impact. A planned event could be routine (expected to occur consistently throughout the activity) or non-routine (may occur infrequently if at all). Air emissions and activity discharges are examples of planned events.
Receptor	A feature of the environment that may have values.
Risk	The effect of uncertainty on objectives.
Risk assessment	The process of determining the likelihood of an unplanned event and the consequence of the impact (in terms of economic, human safety and health, or ecological effects) arising from the event over a specified period.
Routine planned event	An attribute of the planned activity that results in some level of environmental impact and will occur continuously or frequently through the duration of the planned activity.
Unplanned event	An event that results in some level of environmental impact and may occur despite preventive safeguards and control measures being in place. An unplanned event is not intended to occur during the activity.

# 5.2 Summary of the environmental impact and risk assessment approach

### 5.2.1 Overview

Santos operates under an overarching Risk Management Policy. The company Risk Procedure underpins the Risk Management Policy and is consistent with the requirements of AS/NZS ISO 31000:2018, Risk Management – Guidelines (ISO, 2018).



The key steps to risk management are illustrated in Figure 5-1, as defined in the Santos Offshore Division Environmental Hazard Identification and Assessment Guideline.



### Figure 5-1: Hazard identification and assessment guideline

These steps are considered in activity-specific environmental assessment workshop(s) (ENVID workshop) and in the development of this GEP Coastal Waters OEMP. The workshop involves participants from Santos' Health, Safety and Environment (HSE), Spill Response and relevant departments, and specialist environmental consultants.

### 5.2.2 Describe the Activity and hazards (planned and unplanned events)

The location, timing and scope of the Activity must be understood to define the hazards and determine the impacts from planned events, as well as the risks and impacts from unplanned events, which together determine the environment that may be affected or indirectly affected (EMBA) by the Activity.

The outcome of this assessment is detailed in Sections 6 and 7.

### 5.2.3 Identify receptors and determine the nature and scale of impacts

A description of the environment within which hazards from the Activity will, or may occur, is required. This constitutes a crucial stage of the risk assessment, as an understanding of the environmental, socioeconomic and cultural features values and sensitivities that will or may be affected is required to determine the type and consequence of impacts from the Activity being assessed.

The environment must be understood with respect to the spatial and temporal limits of the Activity and key resources at risk that will or could be impacted by planned and unplanned events. Section 3 describes the existing environment that may be affected by the Activity and is informed through consultation (refer Section 4). A protected matters search was conducted over the Activity EMBA to identify occurring or potentially occurring receptors. These receptors are detailed in Section 3.

ENVID workshops (as described in Section 5.1) were held on 23 July 2024 and 29 July 2024 to consider the GEP activities and this Activity. New requirements (such as changes in legislation, guidelines or other requirements) were also considered.



The extent of impacts from planned Activities or risks and impacts from unplanned events were assessed using, where required, modelling (e.g. hydrocarbon release) and scientific reports. The expected duration of each event was also defined using subject matter expertise.

Santos assessed the cumulative impacts of the Activity with other marine users. However, due to the relatively remote location of the OA, it is unlikely that there will be a cumulative impact above impact thresholds with other marine users.

# 5.3 Describe the environmental performance outcomes and control measures

As required by the OPGGS(E)R, environmental performance outcomes(s) (EPOs), control measures (CMs), environmental performance standards (EPSs) and measurement criteria (MC) were identified for the identified environmental impacts and risks. All reasonably practicable control measures were considered and either accepted for use or rejected based on whether impacts and risks had been reduced to levels considered to be ALARP and acceptable.

Accepted control measures were allocated in order of preference, as shown in Figure 5-2.

Control	Effective	ness	Example
Eliminate			Removal of the risk. Refueling of vessels at port eliminates the risks of an offshore refueling.
Substitute			Change the risk for a lower one. The use of low-toxicity chemicals that perform the same task as a more toxic additive.
Engineering			Engineer out the risk. The use of oil-in-water separator to minimise the volume of oil discharged.
Isolation			Isolate people or the environment from the risk. The use of bunding for containment of bulk liquid materials.
Administrative			Provide instructions or training to people to lower the risk. The use of Job Hazard Analysis to assess and minimise the environmental risks of an activity.
Protective			Use of protective equipment. Containment and recovery of spilt hydrocarbons.

### Figure 5-2: Hierarchy of controls

### 5.4 Determine the impact consequence level and risk rankings

The consequence level of a potential impact was determined for each planned and unplanned event using Santos' environment consequence descriptors (Appendix E), and assuming that all control measures had been implemented.

These detailed environmental consequence descriptions are based on the consequence of the impact to relevant receptors within the categories of:

- Threatened, Migratory or local fauna
- physical environment and habitat
- threatened ecological communities
- protected areas



- socioeconomic receptors
- cultural features.

Consequence descriptors are based on set criteria for each receptor category and take into consideration the duration and extent of the impact, receptor recovery time, and the effect of the impact at a population, ecosystem or industry level.

When assessing impacts to cultural features that are part of the EMBA for the Activity, Santos considered cultural features of the environment as defined under the OPGGS(E)R):

- (a) ecosystems and their constituent parts, including people and communities
- (b) natural and physical resources
- (c) the qualities and characteristics of locations, places and areas
- (d) the heritage value of places

When assessing the consequence level of impact to cultural features, Santos considers the different types of cultural features and types of impacts. For impacts to cultural features, in the form of impacts to marine species that are either a cultural food source or are considered culturally significant to First Nations people, Santos assesses impacts with reference to the consequence assessment for threatened/migratory/local fauna. Similarly, where cultural features are linked to a specific place, impacts to cultural features are assessed with reference to the consequence assessment for physical environment/threatened ecological communities/protected areas as applicable. Where there are concerns raised by individuals about cultural and spiritual beliefs that do not link to a specific location or place, Santos will evaluate impact and risk acceptability with consideration for assessment of impacts from analogous activities (e.g., historical drilling, trawl fishing activity, industrial shipping) and consider culturally appropriate measures in response to concerns raised by individuals.

As planned events are expected to occur during the Activity, the likelihood of their occurrence was not considered during the environmental assessment, and only a consequence level was assigned (Table 5-2).

Consequence level	Consequence level description
I	Negligible – No impact or negligible impact
II	Minor – Detectable but insignificant change to local population, industry or ecosystem factors
III	Moderate – Significant impact to local population, industry or ecosystem factors
IV	Major – Major long-term effect on local population, industry or ecosystem factors
V	Severe – Complete loss of local population, industry or ecosystem factors and/or extensive regional impacts with slow recovery
VI	Critical – Irreversible impact to regional population, industry or ecosystem factors

### Table 5-2: Summary of the Santos Environmental Consequence Descriptors

For unplanned events, the consequence level of the impact was combined with the likelihood of the impact occurring (Table 5-3) to determine a residual risk ranking using Santos' corporate risk matrix (Table 5-4).

### Table 5-3: Likelihood description

No.	Matrix	Description
f	Almost Certain	Occurs in almost all circumstances OR could occur within days to weeks
е	Likely	Occurs in most circumstances OR could occur within weeks to months
d	Occasional	Has occurred before in Santos OR could occur within months to years
С	Possible	Has occurred before in the industry OR could occur within the next few years
b	Unlikely	Has occurred elsewhere OR could occur within decades
а	Remote	Requires exceptional circumstances and is unlikely to occur even in the long term



### Table 5-4: Santos risk matrix

		Consequence	Consequence							
		I	II	=	IV	V	VI			
	f	Low	Medium	High	Very High	Very High	Very High			
Likelihood	е	Low	Medium	High	High	Very High	Very High			
	d	Low	Low	Medium	High	High	Very High			
	С	Very Low	Low	Low	Medium	High	Very High			
	b	Very Low	Very Low	Low	Low	Medium	High			
	а	Very Low	Very Low	Very Low	Low	Medium	Medium			

### 5.5 Evaluate if impacts and risks are ALARP

For planned and unplanned events, an ALARP assessment was undertaken to demonstrate the standard control measures adopted reduce the impact (consequence level) or risk to ALARP. This process relies on demonstrating further potential control measures would require a disproportionate level of cost and effort to reduce the level of impact or risk. If this cannot be demonstrated, then further control measures are adopted. The level of detail included within the ALARP assessment is based upon the nature and scale of the potential impact or risk (e.g., more detail is required for a risk ranked as 'Medium' compared with a risk ranked as 'Low').

### 5.6 Evaluate impact and risk acceptability

Santos considers an impact or risk associated with the activities to be acceptable if each of the following criteria, where relevant, are satisfied:

- the consequence of a planned event is ranked as I or II; or a risk of impact from an unplanned event is ranked Very Low to Medium
- an assessment has been completed to determine that sufficient information or studies have been considered to validate the consequence assessment
- the activity will be carried out in a manner that is consistent with the principles of ecologically sustainable development (ESD)
- the acceptable levels of impact and risks have been informed by relevant species recovery plans, threat abatement plans and conservation advice
- performance outcomes, control measures and associated performance standards:
  - are consistent with legal and regulatory requirements
  - are consistent with Santos' Environment, Health and Safety Policy (Appendix A)
  - are consistent with industry standards
  - take into consideration Relevant Person feedback
  - have been demonstrated to reduce the impact or risk to ALARP.

# 6. Planned activities impact assessment

### OPGGS(E)R 2023 Requirements

#### Section 21. Environmental assessment

Evaluation of environmental impacts and risks

(5) The environment plan must include:

- a. details of the environmental impacts and risks for the activity; and
- b. an evaluation of all the impacts and risks, appropriate to the nature and scale of each impact or risk; and
- c. details of the control measures that will be used to reduce the impacts and risks of the activity to as low as reasonably practicable and an acceptable level.

(6) To avoid doubt, the evaluation mentioned in paragraph (5)(b) must evaluate all the environmental impacts and risks arising directly or indirectly from:

- a. all operations of the activity; and
- b. potential emergency conditions, whether resulting from accident or any other reason.

Environmental performance outcomes and standards

(7) The environment plan must:

- a. set environmental performance standards for the control measures identified under paragraph (5)(c); and
- b. set out the environmental performance outcomes against which the performance of the titleholder in protecting the environment is to be measured; and
- c. include measurement criteria that the titleholder will use to determine whether each environmental performance outcome and environmental performance standard is being met.

An ENVID workshop (as described in Section 5.2.3) was held on 23 July and 29 July 2024 to assess the impacts associated with activities described in this GEP Coastal Waters OEMP. Santos' environmental assessment identified six sources of environmental and socioeconomic impacts associated with the planned activities to be undertaken within the OA.

A comprehensive assessment for each of the sources, termed planned events, and subsequent control measures adopted to reduce the impacts to ALARP and acceptable levels are detailed in the following subsections. The resultant residual consequence levels associated with the planned events are summarised in Table 6-1 as supported in the following subsections.

Adopted control measures are assigned a reference number (e.g. BAO-CM-6.1.1). The control measure reference numbers are aligned with the Barossa Production Operations EP and therefore are not in numerical order in this GEP Coastal Waters OEMP.

OEMP section	Planned event	Residual consequence level
6.1	Noise emissions	II – Minor
6.2	Light emissions	II – Minor
6.3	Atmospheric and greenhouse gas emissions	I – Negligible
6.4	Seabed and benthic habitat disturbance	II – Minor
6.5	Interactions with other marine users	I – Negligible
6.6	Operational discharges	I – Negligible

#### Table 6-1: Environmental impact assessment summary



### 6.1 Noise emissions

### 6.1.1 Description of event

Event	<ul><li>Underwater noise emissions will be generated from the following sources:</li><li>IMMR vessel operations (e.g. vessel engines, thrusters and other machinery and equipment)</li></ul>
	<ul> <li>non-vessel IMMR activities: including mechanical jetting, ROV and tooling (pipeline cutting and coating removal), AUV, SSS and MBES surveys</li> </ul>
	Helicopter activities
	contingency activities presented in Section 2.4.3
	operation of the pipeline
Extent	<ul> <li>The worst-case distance for potential impact (furthest distance which behavioural impacts may occur) is approximately 9.8 km, associated with a large IMMR vessel on DP. This is a conservative estimate based on sound levels from a pipelay vessel.</li> </ul>
	<ul> <li>A conservative estimate for the use of survey equipment is that noise will propagate horizontally within a few hundred metre radius.</li> </ul>
	<ul> <li>No permanent noise source will be active along the pipeline, however the minor noise from gas moving through the pipeline is expected to fall to ambient levels within 100 m of the pipeline.</li> </ul>
Duration	Noise emissions are expected to occur intermittently every 3-5 years from discrete campaigns for 7 to 30 days (e.g. IMMR vessel presence, ROVs, helicopter movements)

### 6.1.1.1 Introduction

Santos commissioned a technical study into underwater noise impacts on marine fauna (JASCO, 2020) using contemporary criteria and has used the findings to inform the underwater noise emissions impact assessment. Noise sources involved in the activities described in this GEP Coastal Waters OEMP include non-impulsive (continuous e.g. vessel operations) and impulsive (e.g. SSS and MBES surveys) noise sources. Impulsive sounds have the high peak sound pressure and rapid rise and decay time that non-impulsive sounds do not have. The relevant terminology for underwater acoustic levels relevant to non-impulsive sources are sound pressure levels (SPL), and accumulated sound exposure levels (SEL).

ROVs and associated mounted equipment (e.g. cutting device) may be launched from activity vessels and the noise generated from this equipment will have a considerably lower intensity than vessel noise as will the other minor sources of noise generating activities (mechanical jetting and AUV). Underwater sound levels depend on the primary (noisiest) sound source rather than being strictly additive. Thus, these sources will make little contribution to the overall noise emissions associated with vessel activities or SSS and MBES surveys so are not considered further in the impact assessment of noise emissions.

### 6.1.1.2 Noise generated by vessels

Vessel operational noise includes machinery noise (e.g., engine noise) and hydrodynamic noise (e.g., water flowing past the hull, thruster use, and propeller singing). During normal operations, the activity vessels will generate continuous noise from propeller cavitation, thrusters and hydrodynamic flow around the hull. The activity vessels and their activities are described in Section 2.4. Typically, the type of vessel operations that will occur are:

- Vessel steaming at low speed
- Manoeuvring during IMMR activities (vessels under DP).

The typical sound levels generated by vessels are broadband and usually increase with increasing vessel size, with smaller vessels (less than 50 m in length) having source levels 160 to 175 dB (re 1  $\mu$ Pa), medium sized vessels (50 to 100 m in length) 165 to 180 dB (re 1  $\mu$ Pa) and larger sized vessels (greater than 100 m in length) 180 to 190 dB (re 1  $\mu$ Pa) (OSPAR, 2009; Richardson et al., 1995 in Genesis Oil and Gas Consultants, 2011). Gotz et al. (2009) lists tugboats, crew boats, supply ships and many research vessels in the 50 to 100 m size class also having similar levels of 165 to 180 dB re 1  $\mu$ Pa range (221 SELcum [Richardson et al., 1995]).

For activity vessels, the noisiest anticipated activity is when the vessel uses thrusters to maintain its position. McCauley et al. (1998) measured underwater SPLs equivalent to approximately 182 dB re 1  $\mu$ Pa at 1 m with a frequency range of 20 Hz to 10 kHz from a support vessel holding station in the Timor Sea. The thruster noise dropped below 120 dB re 1  $\mu$ Pa within 3–4 km and was audible above ambient noise up to 20 km away (McCauley, 1998). This has been taken as the greatest noise-generating activity for assessment purposes, as other vessel activities will require the vessel to be idle or moving. McCauley et al. (1998) measured underwater sound levels from the Pacific Ariki, a 64 m long support vessel with 6,000 kW main engines during calm conditions in the Timor



Sea in 110 m of water while transiting at 11 knots, and found the distance to 120 dB re 1  $\mu$ Pa to be approximately 1 km.

### 6.1.1.3 Noise generated by a helicopter

Sound travelling from a source in the air (e.g. a helicopter) to a receiver underwater is affected by both in-air and underwater propagation processes, and processes occurring at the air/sea water surface interface (e.g. wind and waves). The level of noise received underwater depends on source altitude and lateral distance, receiver depth, water depth, and other variables.

Helicopter engine noise is emitted at various frequencies; however, the dominant tones are generally of a low frequency below 500 Hz (Richardson et al., 1995). Sound pressure in the water directly below a helicopter is greatest at the surface and diminishes with increasing receiver depth. Noise also reduces with increasing helicopter altitude, but the duration of audibility often increases with increasing altitude, with sound penetrating water at angles less than 13° (Richardson et al., 1995). The noise from the flyover of a Bell 214ST helicopter has been recorded underwater (Richardson et al., 1995), with the maximum recorded sound level for the dominant 22 Hz tone was 109 dB re 1  $\mu$ Pa (SPL) when the helicopter was 152 m from the surface and the hydrophone 3 and 18 m under the surface.

For context, the Bell 214ST uses a single powerful Lycoming LTC4B-8 engine of 2,185 kW (Frawley, 2003), while the modern Bell 412, often used as a rescue helicopter in Australia (Air Services Australia, 2020) uses twin 1,250 hp (930 kW) turboshaft engines (Bell Helicopter, 2012). Typical offshore crew change and medivac helicopters in Australia are Leonardo AW139s (Milne, 2019), which have been measured to be 2 dB(A) quieter than the Bell 412 helicopters (Air Services Australia, 2020).

Helicopter activities produce strong underwater sounds for brief periods when the helicopter takes off/lands on the vessel. Sound from helicopter activities is very localised and infrequent. Further, helicopter operations are expected to result in received underwater noise levels lower than those associated with vessel operations and are not considered further in the impact assessment of noise emissions.

### 6.1.1.4 Noise generated by MBES and SSS

MBES and SSS transmit at high frequencies (approximately 70 to 400 kHz) and produce a highly focused beam of sound down towards the seabed, so there is very limited horizontal sound propagation. Source levels for these survey methods include:

- Multibeam echo-sounder (MBES), such as the Reson SeaBat 7125 transmitting at 400 kHz. At 400 kHz it
  has a 1° beamwidth along the track, and a source level of 220 dB re 1 µPa SPL (Coastal Frontiers, 2017)
- Side scan sonar (SSS), which is generally considered a high acoustic density source and medium-frequency generator. The level of sound pressure ranges from about 200 to 234 dB re 1 µPa SPL. The frequency ranges from about 75 to 900 kHz (Jimenez-Arranz et al., 2020).

### 6.1.1.5 Noise generated by pipeline

The level of noise emitted by subsea infrastructure such as the pipeline are expected to be low levels, similar to ambient noise levels in the region. Based on the measurements of wellhead noise discussed in McCauley (2003), which included flow noise in pipelines, noise produced along a pipeline may be expected to be similar to that described for wellheads, with the radiated noise field falling to ambient levels within 100 m of the pipeline, noting that there are no valves on the pipeline in the OA where noise is expected to be emitted.

### 6.1.1.6 Summary of noise sources and rationale for assessment

The sound source levels expected from the activities are summarised in Table 6-2.

### Table 6-2: Summary of expected sound source levels during the operations activity

Source	Expected Source Levels (dB re 1 µPa)	Reference
Vessels	Support vessel under DP: 182	McCauley et al., 1998
Surveys	MBES: 220 SSS: 200 to 234	Coastal Frontiers, 2017 Jimenez-Arranz et al., 2020

### 6.1.2 Nature and scale of environmental impacts

**Potential receptors**: Threatened, Migratory or local fauna (marine mammals, marine turtles, sharks, fish, rays, and invertebrates), socio-economic and cultural features.



Some of these marine species have cultural significance to First Nations persons either as a traditional food source or for other cultural reasons (see Sections 3.2.15.9 and 3.2.15.10).

A PMST search was undertaken for a 20 km noise assessment boundary around the OA as a conservative buffer to identify any threatened or migratory species that could be affected by noise outside of the OA. Although the estimated behavioural disturbance distance for cetaceans is up to 9.8 km (Table 6-5), a conservative 20 km buffer was chosen based on noise from vessel thrusters that may be audible above ambient noise up to 20 km away (Section 6.1.1.2).

An additional threatened species, the speartooth shark, and three additional migratory species, dugong, longfin mako, and shortfin mako, were identified within the noise assessment boundary compared with the OA (Table 3-14). The 20 km noise assessment boundary intersects the flatback internesting BIA and habitat critical to the survival of the flatback, and the habitat critical to the survival of olive ridley and does not intersect any known marine mammal or bird BIA.

Marine fauna use sound in a variety of functions, including social interactions, foraging, orientation, and response to predators. Underwater noise can affect marine fauna in these ways:

- Attraction
- Disturbance, leading to behavioural changes or displacement to fauna. The occurrence and intensity of disturbance is highly variable and depends on a range of factors relating to the animal and situation
- Disruption to underwater acoustic cues
- Increased stress levels
- Indirectly by inducing behavioural and physiological changes in predator or prey species
- Localised avoidance
- Injury to hearing or other organs; hearing loss may be temporary (temporary threshold shift; TTS) or permanent (permanent threshold shift; PTS)
- Masking or interfering with other biologically important sounds (including vocal communications, echolocation, signals and sounds produced by predators or prey).

The nature and scale of impacts must be considered in the context of the ambient noise environment. Ambient underwater noise levels depend on location, and are often dominated by local wind noise, waves, biological noise (e.g. fish choruses are capable of reaching very high levels, in excess of 130 dB re 1  $\mu$ Pa (McCauley, 2012) and vessel traffic. Wind speed and seabed conditions have a clear influence on the ambient noise level. Anthropogenic underwater noise sources in the region comprise shipping and small vessel traffic, petroleum production and exploration drilling activities and sporadic petroleum seismic surveys.

Marine fauna respond variably when exposed to underwater noise from anthropogenic sources, with effects depending on various factors, including distance from the sound source, water depth and bathymetry, the animal's hearing sensitivity, type and duration of sound exposure and the animal's activity at the time of exposure. Broadly, the effects of sound on marine fauna can be categorised as:

- Acoustic masking anthropogenic sounds may interfere with, or mask, biological signals, therefore
  reducing the communication and perceptual space of an individual. Auditory masking impacts could occur
  when audibility is reduced for one sound (signal) that is caused by the presence of another sound (noise).
  For this to occur, the noise must be loud enough and have a similar frequency to the signal, and both
  signal and noise must occur at the same time.
- Behavioural response behavioural impacts will depend on the audible frequency range of each potential receptor in relation to the noise frequency—marine animals will only respond to acoustic signals they can detect, as well as the noise intensity. The intensity of behavioural responses of marine mammals to sound exposure ranges from subtle responses, which may be difficult to observe and have little implications for the affected animal, to obvious responses, such as avoidance or panic reactions. The context in which an animal receives the sound affects the nature and extent of responses to a stimulus. The threshold for eliciting behavioural responses depends on the received sound level and multiple contextual factors such as the activity state of animals exposed to different sounds, the nature and novelty of a sound, spatial relations between a sound source and receiving animals, and the gender, age, and reproductive status of the receiving animal.
- Physiological impacts auditory threshold shift (temporary and permanent hearing loss) marine fauna exposed to intense sound may experience a loss of hearing sensitivity or even potentially mortal injury. Hearing loss may be temporary (TTS) from which an animal recovers within minutes or hours, or permanent (PTS) from which the animal does not recover.



The levels of acoustic exposure that may result in injury or behavioural changes in marine fauna is an area of increasing research. Because of differences in experimental design, methods and units of measure, comparing studies to determine likely sound exposure thresholds can be difficult. After assessing the available scientific information, thresholds were defined to inform the impact assessment and interpret the estimated sound ranges. These are discussed for each receptor in JASCO (2020).

The assessment compared modelled received underwater sound levels to defined noise effect criteria, as determined by scientific research and academic papers (JASCO, 2020), for the identified environmental and social receptors. Although the relationship between received sound levels and impacts to marine species is the subject of ongoing research, the science underlying noise modelling is well understood (Farcas et al., 2016).

### 6.1.2.1 Marine mammals

There are no known BIAs for marine mammals within the 20 km noise assessment boundary (Table 3-12). Therefore, marine mammals are unlikely to aggregate or occur in large numbers within the noise assessment area, however, cetaceans and sirenians may transit the area. The PMST report for the noise assessment boundary identified a migratory species—dugong— additional to the several whale and dolphin species listed in the PMST report for the OA (Table 3-10). The closest significant feature to the noise assessment boundary are breeding dolphin BIAs—spotted bottlenose (Darwin Harbour stock), Australian humpback (a sub-species of the Indo-Pacific humpback dolphin; Darwin Harbour and Van Diemen Gulf stock) and Australian snubfin (Darwin Harbour and Van Diemen Gulf stock) which are greater than 46 km away from the OA (refer to Table 3-12). The nearest whale (pygmy blue) distribution to the OA is greater than 300 km away.

The PMST report for the 20 km noise assessment boundary identified several threatened marine mammal species, including whales (blue, fin and sei) and migratory marine mammal species, including dolphins (Appendix C). A number of migratory species of whales may also occur within the noise assessment boundary, including humpback and Bryde's. These whales have been classified as low frequency (LF) cetaceans based on their hearing range. A number of odontocetes (including dolphins and killer whales) may also be transiting the noise assessment boundary and have been classified as high frequency (HF) cetaceans.

Dugongs are unlikely to occur within the noise assessment boundary, preferring shallow tidal and subtidal seagrass meadows.

The Conservation Management Plan for the Blue Whale 2015–2025 (CoA, 2015a), Conservation Advice for *Balaenoptera physalus* (fin whale) (TSSC, 2015d) and Conservation Advice for *Balaenoptera borealis* (sei whale) (TSSC, 2015c) list noise disturbance as a threat, specifically relating to impulsive sound sources, such as seismic surveys, and acute industrial noise, such as pile driving. Although seismic surveys and pile driving are outside the scope of this GEP Coastal Waters OEMP, SSS and MBES survey activities are an impulsive sound source. Impulsive sound sources present a greater risk than most continuous sounds because of the high peak levels and frequent repetition (CoA, 2015a). Shipping noise in busy shipping channels is also identified as a potential source of noise emissions, although the risk assessment determines that consequences would be restricted to individuals, and no population-level effects are expected. The Conservation Management Plan for the Blue Whale 2015–2025 requires that anthropogenic noise in BlAs will be managed such that any blue whales may continue to use the area without injury. The OA and the 20km noise assessment boundary used for this assessment is distant from the pigmy blue whale migration BlA (~373 km), such that impacts from vessel noise emissions are not inconsistent with the requirements of the Blue Whale Conservation Management Plan (CoA, 2015a).

To better reflect the auditory similarities between closely related species, but also significant differences between species groups among the marine mammals, Southall et al. (2007) assigned the marine mammal species to functional hearing groups based on their hearing capabilities and sound production. This division into broad categories was intended to provide a realistic number of categories for which individual noise exposure criteria were developed. These groups were revised by National Marine Fisheries Service (NMFS, 2018) and most recently by Southall et al. (2019). The categorisation has proven to be a scientifically justified and useful approach in developing auditory weighting functions and deriving noise exposure criteria for marine mammals. These auditory weighting functions are referred to as frequency weighting. These thresholds that detail receptor noise impacts and behavioural response for continuous noise (vessels and cutting tool), along with the new nomenclature and classifications for marine mammals are summarised in Table 6-3. The table details receptor noise impact and behavioural thresholds for continuous noise (vessel), being:

- Low-frequency (LF) cetaceans: which consists of baleen whales such as humpback whales
- High-frequency (HF) cetaceans: which consists of toothed whales except porpoises and river dolphins.
- Very-High frequency (VHF) cetaceans: which consists of whales such as pygmy sperm whale
- Sirenians: which includes dugong.

For non-impulsive noise such as that expected during operations, NMFS currently uses a step function (all-ornone) threshold of 120 dB re 1 µPa SPL (unweighted) to assess and regulate noise-induced behavioural impacts



for marine mammals (NOAA, 2019). The behavioural disturbance threshold criteria applied summates the most recent scientific literature on the impacts of sound on marine mammal hearing, so is considered the most relevant to this activity. Table 6-3 details cetacean behavioural, TTS and permanent threshold shift (PTS) thresholds for continuous noise (vessels); Table 6-4 details cetacean behavioural, TTS and PTS thresholds for impulsive noise (MBES and SSS survey equipment).

 Table 6-3: Continuous noise: summary of marine mammal impact thresholds as derived from Southall et al.

 (2019) and National Oceanic and Atmospheric Administration (2019)

Hearing group	NOAA (2019)	Southall et al. (2019)			
	Behaviour	PTS onset thresholds (received level)	TTS onset thresholds (received level)		
	SPL (dB re 1 μPa)	Weighted SEL₂₄հ (dB re 1 µPa²⋅s)	Weighted SEL₂₄հ (dB re 1 µPa²⋅s)		
LF cetaceans	120	199	179		
HF cetaceans, including		198	178		
VHF cetaceans		173	153		
Sirenians (dugong)		206	186		

### Table 6-4: Impulsive noise: unweighted sound pressure level, SEL24h and PK thresholds for acoustic effects on marine mammals

Hearing group	NOAA (2019)	NMFS (2018), Southall et al. (2019)				
	Behaviour	PTS onset thresholds (received level)		TTS onse (receiv	t thresholds red level)	
	SPL (dB re 1 μPa)	Weighted SEL <sub>24h</sub> (dB re 1 µPa²⋅s)	PK (L <sub>pk</sub> ; dB re 1 μPa)	Weighted SEL <sub>24h</sub> (dB re 1 µPa²⋅s)	PK (L <sub>pk</sub> ; dB re 1 µPa)	
LF cetaceans	160	183	219	168	213	
HF cetaceans, including	160	185	230	170	224	
Sirenians (dugong)	160	190	226	175	220	

### 6.1.2.1.1 Potential impacts from activity vessels

Using the predicted source levels described above, estimated distances from activity vessels to behavioural and physiological thresholds (as listed in Table 6-5) for cetaceans are provided herein.

Zykov et al. (2013) considers a range of modelling scenarios for pipelay vessels in 23 to 80 m of water, with sea floor surface geology consisting of sand and silt. The depths and geology are similar to those within the OA, and the sound speed profile is similar at the relevant shallow depths to that used in previous work for the Barossa Development (JASCO, 2016). The vessel referenced in Zykov et al. (2013) is the Solitaire, a vessel with substantially greater total thruster power (~55,000 kW) than the support vessel the *Sapura Constructor*, which is representative of the vessel size to be used during IMMR activities (~9,800 kW). Results from this noise modelling are therefore considered highly conservative.

McPherson et al. (2019) considers the most recent criteria for potential physiological effects to marine mammals (Southall, 2019) and the equivalent NMFS (2018) from support vessels with similar total thruster power and length to the *Sapura Constructor*. The water depths for this modelling ranged between 50-150m.

Underwater noise modelling assessed in Inpex (2010) considers modelling scenarios for support vessels in water depths equivalent to the OA (60 m). The modelling is considered to provide a representative estimate of potential underwater noise levels from support vessels in the Barossa (ConocoPhillips, 2018).

These modelling scenarios have been considered where there are similarities to the noise emissions from vessels that may be used for IMMR activities.

### Table 6-5: Estimated distances to behavioural and physiological thresholds for marine mammals from vessels

Potential marine mammal receptor	Estimated distance	Justification/reference
PTS		
HF cetaceans, including dolphins	Not predicted to occur	McPherson et al. (2019), offshore support vessel under DP

Potential marine mammal receptor	Estimated distance	Justification/reference
LF cetaceans	<110 m	McPherson et al. (2019), offshore support vessel under DP
Sirenians (dugong)	Not predicted to occur	McPherson et al. (2019), offshore support vessel under DP
TTS		
HF cetaceans, including dolphins	<120 m	McPherson et al. (2019), offshore support vessel under DP
LF cetaceans	<1.5 km	McPherson et al. (2019), offshore support vessel under DP
Sirenians (dugong)	<1.5 km	McPherson et al. (2019), offshore support vessel under DP
Behaviour		
HF cetaceans, including dolphins	1.3 – 9.8 km	McPherson et al. (2019), offshore support vessel under DP
LF cetaceans		(1.3 km)
Sirenians (dugong)		Zykov et al. (2013), pipelay vessel under DP in 80 m water (9.8 km)
		Inpex (2010), offshore support vessel under DP (3.5 km)

Auditory masking impacts could occur when audibility is reduced for one sound (signal) that is caused by the presence of another sound (noise). For this to occur, the noise must be loud enough and have a similar frequency to the signal, and both signal and noise must occur simultaneously. Therefore, the closer the marine mammal is to the vessel and the more overlap there is with their vocalisation frequencies, the higher the probability of auditory masking. Thus, the potential for masking and communication impacts is classified as high near the vessel (within tens of metres), moderate within hundreds of metres, and low within thousands of metres (Clark et al., 2009).

As outlined in Table 6-5, noise emissions generated from vessel activities can cause behavioural responses, such as avoidance, in marine mammals within 1.3 to 9.8 km of vessels used for IMMR activities.

While it is considered unlikely that transiting individuals would remain in close proximity to the sound source, PTS could occur in LF cetaceans within close proximity (<110 m) of the vessel. TTS could occur up to 1.5 km away for LF cetaceans and within close proximity (<120 m) for HF cetaceans, including sirenians (dugongs). However, it is unlikely that individuals would remain within these distances for periods of time sufficient to be harmful.

### 6.1.2.1.2 Potential impacts from helicopters

Helicopter noise has been measured at a maximum received level of 109 dB re 1 µPa (SPL) and only detectable underwater for 11 to 38 seconds (based on transit speed), depending on water depth (Richardson et al., 1995). Therefore, the only credible impact would be behavioural impacts, limited to short term behavioural responses such as diving or increased swimming speed when the helicopter lands or takes off. Such impacts are considered unlikely to result in substantial effects to marine mammal populations or distribution.

### 6.1.2.1.3 Potential impacts from MBES and SSS survey equipment

Survey geophysical equipment has been modelled at a number of locations, including the coast of Russia, Greenland, California and the Otway Basin (Zykov et al., 2013; Austin et al., 2012; McPherson & Wood, 2017). These studies, along with the example of accumulation provided in McPherson (2020), indicate both peak and frequency-weighted SEL noise emissions from survey equipment, such as MBES operating at 400 kHz, are typically below sound levels that could result in low- and high-frequency marine mammal TTS or PTS from either PK or SEL criteria (Table 6-4) in a horizontal direction. The threshold for behavioural disturbance (Table 6-4) could be exceeded within 120 m (McPherson, 2020). SSS impulses and MBES sound levels are outside the auditory range of LF species and baleen whales (such as humpback and pygmy blue whales), but within the mid-frequency and HF cetacean marine fauna auditory range (such as sperm whales and dolphins). However, PTS and TTS thresholds for these species (Table 6-4) are only expected to be exceeded close to the source. Due to the lack of aggregating areas for these species, individuals are expected to be transitory only, displaying behavioural responses, and moving away from the source before TTS and PTS thresholds are exceeded.

Survey equipment could cause masking of vocalisations of cetaceans due to the overlap in frequency range between signals and vocalisations. Masking will primarily apply to HF cetaceans, with all signals above 2 kHz. Higher frequency sounds have limited propagation, and attenuate rapidly, resulting in a relatively small area of influence. Therefore, the range at which masking impacts could occur would be limited to within hundreds of metres from the sound source.

The risk of impact is further reduced as the survey vessels will be moving along the pipeline when conducting these types of surveys. The likelihood of an individual remaining within the distances above for any length of time is highly unlikely.



### 6.1.2.2 Marine reptiles

The 20 km noise assessment boundary intersects the flatback turtle internesting BIA (>800 km of NT coastline) and habitat critical to the survival of the flatback turtle, and the habitat critical to the survival of olive ridley. The flatback turtle peak internesting period occurs between June to September and low-density nesting occurs during the wet season. Notwithstanding, the OA represents a minute portion of the NT-wide total areas of flatback turtle BIA (internesting) and habitat critical to the survival of flatback turtles (nesting) shown in Figure 3-9. Furthermore, as the OA is located in water depths greater than 50 m and has a lack of foraging habitat, the potential numbers of affected internesting turtles is expected to be limited. The OA may also be traversed by green, olive ridley, loggerhead, leatherback, and hawksbill turtles nesting in other areas of northern Australia as marine turtle migratory pathways are largely restricted to waters less than 100 m deep (Pendoley, 2022).

The Recovery Plan for Marine Turtles in Australia 2017–2027 (CoA, 2017b) highlights noise interference from anthropogenic activities as a threat to marine turtles. The plan refers to vessel noise and the operation of some energy infrastructure as sources of chronic (continuous) noise in the marine environment, exposure to which may lead to the avoidance of important turtle habitat. The recovery plan notes there is limited information available on the impact of noise on marine turtles and that the impact of noise on turtle stocks may vary depending on whether exposure is short (acute) or long term (chronic). Turtles have been shown to respond to LF sound, with indications that they have the highest hearing sensitivity in the frequency range of 100–700 Hz (Bartol and Musick, 2003).

Finneran et al. (2017) presented revised thresholds for marine turtle injury and hearing impairment (TTS and PTS). Their rationale is that marine turtles have better auditory sensitivity at low frequencies and poor auditory sensitivity at other frequencies (Bartol and Ketten, 2006; Dow Piniak et al., 2012; Martin et al., 2012). Accordingly, TTS and PTS thresholds for turtles are likely more similar to those of fish than to marine mammals (Popper et al., 2014).

Studies show that marine turtle behavioural responses occur to received sound levels of approximately 166 dB re 1  $\mu$ Pa and that avoidance responses occur at around 175 dB re 1  $\mu$ Pa (McCauley et al., 2000). These levels overlap with the sound frequencies produced by activity vessels. Based on the limited data regarding noise levels that elicit a behavioural response in turtles, the lower level of 166 dB re 1  $\mu$ Pa from the National Science Foundation (NSF, 2011) is typically applied, both in Australia and by NMFS, as the threshold level at which behavioural disturbance could occur. The recommended criteria for continuous and impulsive sound sources for turtles are listed in Table 6-6 and Table 6-7.

Potential marine fauna receptor	Popper et al. (2014	4)	Finneran Weighted SEL <sub>2</sub>	Finneran et al. (2017) Weighted SEL <sub>24h</sub> (dB re 1 μPa²s)		
	Masking	Behaviour	PTS onset threshold	TTS onset threshold		
Marine reptile	(N) High (I) High	(N) High (I) Moderate	220	200		
	(F) Moderate	(F) Low				

#### Table 6-6: Continuous noise: criteria for vessel noise exposure for marine reptiles

Note: Relative risk (high, moderate, low) is given for animals at three distances from the source defined in relative terms as near (N) – tens of metres, intermediate (I) – hundreds of metres, and far (F) – thousands of metres.

### Table 6-7: Impulsive noise: criteria for impulsive noise exposure for marine reptiles, adapted from Popper et al., 2014

Potential marine fauna receptor	Masking	Behaviour	TTS	Recoverable injury	Mortality and potential mortal injury
Marine reptile	(N) Low	(N) High	(N) High	(N) High	>210 dB SEL <sub>24h</sub>
	(I) Low	(I) Moderate	(I) Low	(I) Low	or
	(F) Low	(F) Low	(F) Low	(F) Low	>207 dB PK

Note: Relative risk (high, moderate, low) is given for animals at three distances from the source defined in relative terms as near (N) – tens of metres, intermediate (I) – hundreds of metres, and far (F) – thousands of metres.

### 6.1.2.2.1 Potential impacts from activity vessels

Based on the criteria detailed within Table 6-6, there is a low risk of any injury to marine turtles from activity vessel noise. Behavioural changes, such as avoidance and diving, are only predicted for individuals near the activity vessels: high risk of behavioural impacts within tens of metres of a vessel and moderate risk of behavioural impacts within hundreds of metres of a vessel. There is a high risk of masking within hundreds of metres of the vessel, and a moderate risk of masking within thousands of metres from the vessel. Little is known regarding masking in marine turtles; behavioural reactions have been found to be highly context-specific, with behavioural sensitisation and habituation affecting the onset threshold for reactions and impacts (Ellison et al., 2012). However, given the relatively low-level increase in sound, it is unlikely vessel noise will cause significant masking impacts in turtles.



### 6.1.2.2.2 Potential impacts from helicopters

Helicopter noise will be intermittent during the Activity and below the behavioural impact threshold (PTS and TTS). Impacts to marine turtles from helicopter noise are expected to be limited to short term behavioural impacts (i.e. diving or swimming rapidly) when the helicopter is taking off or landing, based on measurements of helicopter noise (Richardson et al. 1995). Such impacts are considered unlikely to affect marine turtle populations or distribution substantially.

### 6.1.2.2.3 Potential impacts from survey equipment

The sound levels of the survey equipment are below those associated with the PK criterion for injury (Table 6-7) beyond a few metres (McPherson, 2020) and, due to the low per-pulse SEL (McPherson, 2020), the SEL criterion will also not be exceeded. Recoverable injury and TTS could occur within tens of metres applying the relative risk criteria from Popper et al. (2014) (Table 6-7). Behavioural changes, such as avoidance and diving, are only predicted for individuals in proximity to the survey: high risk of behavioural impacts within tens of metres of the source.

Turtles are unlikely to experience masking, even at close range to the source. This is in part because the sounds from survey equipment are all outside of the hearing frequency range for turtles (approximately 50 to 2000 Hz, with highest sensitivity to sounds between 200 and 400 Hz) (Ridgway et al., 1969; Bartol et al., 1999; Ketten & Bartol, 2005; Bartol & Ketten, 2006; Yudhana et al., 2010; Piniak et al., 2011; Lavender et al., 2012, 2014).

Impacts to marine turtles from underwater noise generated by survey equipment are unlikely to result in significant impacts, given impacts are likely to be limited to physiological impacts in individuals located within tens of metres of the sound source, and behavioural impacts in individuals located within hundreds of metres of the sound source. The risk of impact is further reduced as the vessels will be moving when undertaking surveys and it is highly unlikely any individual would remain within the distances above for any length of time.

In summary, considering the offshore location and water depths of greater than 50 m within the OA, only individual turtles may be affected as they transit the area, and impacts from noise are not considered significant because:

- The 20 km noise assessment boundary intersects a minute portion of the total area of flatback turtle internesting BIA (>800 km of NT coastline) and habitat critical to the survival of the flatback turtle, as well as habitat critical to the survival of the olive ridley turtle.
- There are no flatback turtle or olive ridley nesting sites within the noise assessment boundary
- Vessel noise, and survey and positioning equipment are expected to be below the thresholds for PTS and TTS given the typical size of vessels used for IMMR activities and the slow vessel speeds within the OA; the received levels may result in behavioural impacts, but for a limited time and will not result in significant impacts
- Individual marine turtles may traverse the 20 km noise assessment boundary but are unlikely to aggregate
- following the impact thresholds outlined in Table 6-6 and Table 6-7, marine turtles are at low credible risk of mortality or permanent injury due to continuous noise sources, even near the source
- behavioural responses are expected to occur near the sources but will be limited to avoidance or a temporary change in swimming behaviour.

### 6.1.2.3 Crocodiles and sea snakes

There is limited information on the effects of noise on sea snakes and crocodiles. A current research project investigating the impacts of impulsive noise (based on seismic surveys, noting seismic surveys are outside the scope of this GEP Coastal Waters OEMP) found that the hearing sensitivity of sea snakes is similar to species of fish without a swim bladder. Therefore, it is considered that there is a moderate risk in the near and intermediate distances (which extend hundreds of metres) of behavioural impacts to sea snakes, with the impacts being limited to temporary avoidance of the area.

Crocodiles are considered to hold cultural significance as totemic species (Section 3.2.15.10). There are no known studies that have investigated the effects of noise on crocodiles so the thresholds for turtles shown in Table 6-6 and Table 6-7 are considered applicable. Such impacts are considered unlikely to result in substantial affects to crocodile populations or distribution.

### 6.1.2.4 Sharks, rays, and fish

The PMST report for the noise assessment boundary identified a threatened species—speartooth shark—, and migratory species—longfin mako and shortfin mako — additional to the several sawfish, ray, shark, and other fish species listed in the PMST report for the OA (Table 3-10). There are no known fish spawning or aggregation areas along the GEP route in coastal waters; however, individuals or schools may transit. The closest area that supports site attached fish is the Shepparton Shoal (23 km from the OA) and Afghan Shoal (24 km from the OA).



All fish species can detect noise sources, although hearing ranges and sensitivities vary substantially between species (Dale et al., 2015). Sensitivity to sound pressure in fish seems to be functionally correlated to the presence or absence of gas-filled chambers in the sound transduction system. These chambers enable fish to detect sound pressure and extend their hearing abilities to lower sound levels and higher frequencies (Ladich and Popper, 2004; Braun and Grande, 2008). Based on their morphology, Popper et al. (2014) classified fish into 3 animal groups comprising:

- Fish with swim bladders whose hearing does not involve the swim bladder or other gas volumes
- Fish whose hearing does involve a swim bladder or other gas volume
- Fish without a swim bladder that can sink and settle on the substrate when inactive.

Thresholds for PTS and recoverable injury are between 207 dB peak and 213 dB peak (depending on the presence or absence of a swim bladder), and the threshold for TTS is 186 dB SELcum (Popper et al., 2014). Because there are no exposure criteria for sawfish, sharks and rays, the same criteria are adopted, although these species do not possess a swim bladder.

The criteria defined in Popper et al. (2014) for continuous (Table 6-8) and impulsive (Table 6-9) noise sources were applied when assessing impacts to sharks, rays, and other fish.

Marine fauna group	Mortality and		Impairment	Behaviour	
	potentially fatal injury	Recoverable injury	TTS	Masking	
I Fish: No swim bladder (particle motion detection) includes sharks and rays.	(N) Low (I) Low (F) Low	(N) Low (I) Low (F) Low	(N) Moderate (I) Low (F) Low	(N) High (I) High (F) Moderate	(N) Moderate (I) Moderate (F) Low
II Fish: Swim bladder not involved in hearing (particle motion detection)	(N) Low (I) Low (F) Low	(N) Low (I) Low (F) Low	(N) Moderate (I) Low (F) Low	(N) High (I) High (F) Moderate	<ul><li>(N) Moderate</li><li>(I) Moderate</li><li>(F) Low</li></ul>
III Fish: Swim bladder involved in hearing (primarily pressure detection)	(N) Low (I) Low (F) Low	170 dB SPL for 48 h	158 dB SPL for 12 h	(N) High (I) High (F) High	(N) High (I) Moderate (F) Low
Fish eggs and fish larvae	(N) Low (I) Low (F) Low	(N) Low (I) Low (F) Low	(N) Low (I) Low (F) Low	(N) High (I) Moderate (F) Low	<ul><li>(N) Moderate</li><li>(I) Moderate</li><li>(F) Low</li></ul>

#### Table 6-8: Continuous noise: summary of fish impact thresholds

Source: Adapted from Popper et al., 2014

Note: Relative risk (high, moderate, low) is given for animals at three distances from the source defined in relative terms as near (N) – tens of metres, intermediate (I) – hundreds of metres, and far (F) – thousands of metres.

### Table 6-9: Impulsive noise: summary of fish impact thresholds

Marine fauna group	Mortality and		Impairment		
	fatal injury	Recoverable injury	TTS	Masking	
I Fish: No swim bladder (particle motion detection) includes sharks and rays.	> 219 dB SEL24h or > 213 dB PK	> 219 dB SEL24h or > 213 dB PK	>>186 dB SEL24h	(N) Low (I) Low (F) Low	(N) High (I) Moderate (F) Low
II Fish: Swim bladder not involved in hearing (particle motion detection)	210 dB SEL24h or > 207 dB PK	203 dB SEL24h or > 207 dB PK	>>186 dB SEL24h	(N) Low (I) Low (F) Low	(N) High (I) Moderate (F) Low
III Fish: Swim bladder involved in hearing (primarily pressure detection)	207 dB SEL24h or > 207 dB PK	203 dB SEL24h or > 207 dB PK	186 dB SEL24h	(N) Low (I) Low (F) Moderate	(N) High (I) High (F) Moderate



Fish eggs and fish larvae	> 210 dB	(N) Moderate	(N) Moderate (I)	(N) Low	(N) Moderate (I)
	SEL24h or >	(I) Low	Low	(I) Low	Low
	207 dB PK	(F) Low	(F) Low	(F) Low	(F) Low

Source: Adapted from Popper et al., 2014 Note: Relative risk (high, moderate, low) is given for animals at 3 distances from the source defined in relative terms as near (N) – tens of metres, intermediate (I) – hundreds of metres, and far (F) – thousands of metres.

### 6.1.2.4.1 Potential impacts from vessels

Based on the Popper et al. (2014) review, vessel noise has a low risk of resulting in mortality for all fish types. The risk of recoverable injury to Group I and Group II fish is low; however, it is moderate for TTS and behavioural impacts when fish are within tens of metres of an activity vessel (Popper et al., 2014). For Group III fish, recoverable injury and TTS may occur within 60 m of the source (McPherson et al., 2019), with a high risk of behavioural impacts occurring within tens of metres of an activity vessel (Popper et al., 2014).

Whale sharks are not considered to be particularly vulnerable to noise-related impacts and are categorised as 'fish with no swim bladder' when determining impact thresholds. Whale sharks would be expected to show avoidance to vessel noise, although they are likely to tolerate low-level noise—whale sharks have been observed swimming close to energy industry platforms on WA's North-West Shelf.

Fish within the OA are expected to be highly mobile or a transitory, with no sensitive or significant benthic features known to be present that would cause an aggregation of fauna or site attached species. In addition, impacts to fish are not considered to have the potential to be significant because noise levels from vessels that may cause behavioural responses are expected to be within a radius of a few hundred metres of the noise source.

### 6.1.2.4.2 Potential impacts from survey equipment

The criteria defined in Popper et al. (2014) for impulsive noise sources has been adopted (Table 6-9). Impulsive noises from survey equipment could result in physiological impacts to fish located within metres of the sound source, considering the results presented in McPherson (2020). Given the ability of fish to move away from the sound source, physiological impacts are not expected.

Behavioural impacts to fish from survey equipment noise may occur in individuals located within hundreds of metres of the source. None of the survey equipment has energy below 1 kHz; therefore, it is unable to be heard by most fish, which further reduces the risk of impact (Ladich & Fay, 2013). The impact of masking is low at all ranges, apart from fish who specialise in pressure detection, which can be impacted in a moderate way at thousands of metres. However, as these signals are outside the hearing range of most fish in the region, the risk of impact is reduced.

Impacts to fish from underwater noise generated by survey equipment are unlikely to result in substantial impacts to populations or distribution, given impacts are likely to be limited to behavioural impacts within hundreds of metres and masking within thousands of metres. Fish are considered unlikely to remain in proximity of the sound source for long periods of time and are, therefore, unlikely to be exposed to sound at the above thresholds. Site-attached fish are more at risk of impacts. Given the vessel is constantly moving when undertaking surveys, survey source noise will not impact any one location for an extended duration.

### 6.1.2.5 Invertebrates

### 6.1.2.5.1 Potential impacts from vessels

Benthic invertebrates are unlikely to be negatively impacted from noise generated from vessel activities (i.e., water depth is greater than 50 m). There are no thresholds or guidelines regulating the exposure of marine invertebrates to underwater noise.

Stress responses to non-impulsive sound exposure have been documented for marine invertebrates. The worstcase consequence for individual animals can be expected to be moderate to major, but due to the limited spatial extent of the affected area, population consequences are considered to be minor.

There is no systematic information available if and to what extent marine invertebrates use acoustic cues to communicate with conspecifics or their environment. Anecdotal information indicates no functional relevance of sound for these animals – vibration, such as ground-borne or near-field particle motion – however, sound can be assumed to have functional relevance as it provides information about potential food availability or approaching predators. This information could potentially be masked by the noise and particle motion emitted by the vessels, even though this effect would be limited to the direct vicinity of noise-generating sources. The consequence of (acoustic and vibrational) masking is considered to be, in the worst case, moderate for individuals. Due to an expected limited number of individuals experiencing this masking, it would have a negligible impact on a population level.

There is limited and inconclusive data available about the potential for behavioural responses and noise-induced physical effects on marine invertebrates. Theoretically, behavioural responses and significant sensory impairment



or injury can have moderate consequences for an individual. In the absence of conclusive scientific information about the scope of these effects and the animals' ability to compensate for the effects, it is impossible to assess the consequences of behavioural responses and noise-induced impairment or injury.

Plankton, including fish eggs and larvae, and pelagic invertebrates could drift close to high energy noise sources (for example, vessel thrusters). However, any negative impacts that could occur would be restricted to within metres of the sound source.

### 6.1.2.5.2 Potential impacts from survey equipment

For impulsive noise and benthic invertebrates, the source is an important consideration in the assessment.

Any negative impacts on plankton and invertebrates that could occur would be restricted to within metres of the sound source. At such a localised extent, impacts would be negligible at an ecosystem or population level.

There are no thresholds or information available for assessing the potential impacts from HF sources such as MBES/SBES on either water column or benthic invertebrates. These sources are often used to assess and quantify plankton densities, including within McCauley et al. (2017), who used a Simrad EK60 echosounder operating at 120 kHz.

### 6.1.2.6 Socioeconomic

Impacts to socioeconomic receptors, including commercial fisheries, recreation and tourism are considered to be minor due to the localised, infrequent and temporary noise levels and low socioeconomic activity levels expected within the noise assessment boundary. No commercial fisheries feedback about noise impacts was provided during consultation (refer to Section 4.8).

### 6.1.2.7 Cultural features

No First Nations people feedback was provided about potential noise impacts to any geographically specific cultural features (excluding marine fauna species) during consultation (refer to Section 4.8). The potential direct or indirect impacts to culturally significant marine fauna species (such as dreaming / songs and totem species including whales, dolphins, dugongs, turtles, crocodiles, sawfish, sharks, rays and other fish) are assessed in Sections 6.1.2.1 to 6.1.2.4).

During consultation meetings with Tiwi Clans for the Barossa Development Drilling and Completions EP concerns were raised about the impact of drilling on their dreaming totems (including turtle totems). In the 2022 Statement of Reasons Requests, the Tiwi clan members raised their concern regarding traditional hunting of marine species and totem species. First Nations people maintain a continuing spiritual connection with sea country, including marine fauna species with cultural significance, such as totems or as a cultural food source. The potential impacts to culturally significant marine fauna species are likely to be limited to localised, temporary behavioural impacts (see Sections 6.1.2.1 to 6.1.2.4). It is unlikely to result in significant impacts to marine species at the individual or population level. Consequently, it is anticipated that noise emissions are unlikely to impact traditional hunting practices or resources.

As presented in Section 3.2.15, some First Nations peoples' cultural beliefs place significance on culturally important spiritual beings and the protection they afford First Nations communities from natural disasters and sickness. Santos recognises that some First Nations Relevant Persons fear sickness or other adverse effects from the actions of spiritual beings in response to impacts on the environment of sea country itself. Of direct relevance these sorts of Tiwi cultural and spiritual values were tested in the Federal Court and were found not to be consistently spread amongst relevant Tiwi Islanders and in any event did not represent a particular 'place' of cultural and spiritual significance.

Santos notes that existing subsea infrastructure has previously been placed on the seabed in the region, such as the Bayu-Undan pipeline since approximately 2006, the Ichthys Pipeline since approximately 2016, the North West Cable System since approximately 2016 and the GEP since 2023. The region also has a history of significant historic and ongoing industrial shipping, fish trawling activities and drilling of almost 900 offshore wells. There is no evidence to support actual adverse effects from the actions of spiritual beings in response to impacts on the environment from these activities.

Santos recognises the importance of cultural and spiritual beliefs to First Nations people. Santos recognises that some First Nations people remain concerned about the potential for adverse consequences to First Nations people and natural environment, that may arise as a result of disturbance from the Barossa Gas Project to spiritual dreaming and culturally important spiritual beings. Santos understands the spiritual protection believed to be afforded to the First Nations people is broadly maintained by protecting the features of the natural environment and through ceremonial practices alerting the spiritual beings to the presence of people travelling through country and the like (Corrigan, 2023).

Dr Corrigan (2024) documented input from Larrakia people and relevant First Nations persons from Belyuen and Wagait, who advised the presence of a range of ancestral beings and dreaming stories of relevance to the Darwin



Harbour, surrounding seas and the DPD (now GEP) Project footprint. None of these cultural features are known to be associated with any specific or particular places in the GEP footprint but rather have a more general association with the wider area, as well as having associations with particular and specific places outside of the GEP footprint.

Santos considers that the control measure to provide cultural heritage training (BAO-CM-6.1.4 will reduce impacts and risks to intangible cultural values to ALARP and an acceptable level. Santos has also considered those concerns relating to potential noise impacts relating to other Barossa Gas Project EPs and where applicable additional EPOs, EPSs and CMs have been adopted.

### 6.1.3 Environmental performance outcomes and control measures

The EPOs relating to this event include:

- No injury of, or mortality to, EPBC Act 1999 listed marine fauna or marine fauna listed as threatened species under the Territory Parks and Wildlife Conservation Act 1976 (NT) as a result of the Activity. (EPO-08)
- No displacement of marine turtles from habitat critical during nesting/breeding (including internesting periods for turtles) and ensure biologically important behaviour can continue in biologically important areas. (EPO-15)

An assessment of the environmental benefits and the potential costs or issues associated with control measures for this activity are described in Table 6-10 to demonstrate the potential impacts from this aspect are ALARP. Control measures that are adopted have associated EPSs and measurement criteria and are presented in Table 8-2. Rejected control measures have an ALARP evaluation provided to justify their rejection.

CM reference	Control measure	Environmental benefit	Potential cost/issues	Evaluation
Standard control meas	sures			
BAO-CM-6.1.1	Apply Santos' Protected Marine Fauna Interaction and Sighting Procedure to vessel and helicopter activities when in the vicinity of cetaceans and turtles (isolation control)	Santos implements EPBC Regulations– Part 8 Division 8.1 Interacting with cetaceans (and applied for marine turtles) where vessel crew (or Vessel Master of the USV) act as wildlife observers to reduce the risk of a collision with marine fauna (Section 7.3). This control may result in a minor ancillary reduction in the potential for vessel noise impacts. It also effectively reduces helicopter noise levels received by fauna near the sea surface through minimising interaction with marine fauna. Reduces the potential impacts to culturally significant marine species, including totemic species, such as marine turtles and marine mammals.	Operational costs to adhere to marine fauna interaction restrictions, such as vessel and helicopter speed and direction, are based on legislated requirements and must be accepted.	Adopted – benefits in reducing impacts to marine fauna outweigh the costs incurred by Santos. Control measure drives compliance with EPBC Regulations (Part 8).
BAO-CM-6.1.2	Vessels equipped and crewed in accordance with Australian maritime requirements, including Marine Order	Ensures contracted vessels are operated, maintained, and crewed in accordance with industry standards and	Costs are expected as part of standard procedure.	Adopted – benefit of assuring vessels outweighs procedure compliance costs.

#### Table 6-10: Control measure evaluation for noise emissions

CM reference	Control measure	Environmental benefit	Potential cost/issues	Evaluation
	30 (Prevention of Collisions) and Marine Order 21 (Safety and Emergency Arrangements) (administrative control)	regulatory requirements. Ensures vessels meet Marine Assurance Standards to reduce the likelihood of vessel collision (such as minimum and working lighting for maritime safety).		
Additional control mea	asures		1	1
BAO-CM-6.1.4	Inductions for all site- based workforce will include information on cultural heritage to raise awareness about the cultural and spiritual belief of First Nations people (administrative control)	Addresses concerns raised (during consultation for the Barossa Project construction activities) of some First Nations people about the potential impacts of the Activity on their spiritual beliefs in a culturally appropriate manner.	Cost to engage First Nations representatives to perform cultural ceremony. Administrative cost to deliver cultural heritage training.	Adopted – benefits considered to outweigh costs.
N/A	Avoid activities near cetaceans and turtles) (isolation control)	Reduces noise impacts to internesting flatback turtles during key life stages. Reduces the potential impacts to culturally significant marine species, including totemic species, such as marine turtles and marine mammals.	Reduces the window of opportunity for undertaking the Activity.	Not adopted – the potential for impact is considered acceptable as impacts to marine turtles from underwater noise generated by survey equipment are likely to be limited to physiological impacts in individuals located within tens of metres of the sound source, and behavioural impacts in individuals located within hundreds of metres of the sound source. Activities in HC area will be infrequent (every three to five years, and short in duration).
N/A	Manage the timing of the Activity to avoid sensitive periods, such as migration (whales), spawning (fish), or nesting (turtles) (administrative control)	Reduces potential impacts to fauna during key life stages.	Reduces the window of opportunity for undertaking the Activity.	Not adopted – not considered necessary or feasible as primary noise is from vessel DP thrusters and engines. The OA does not overlap with any whale migration BIAs and therefore seasonal presence of whales is not expected to be higher at certain times of the year. Additionally, given the low potential impacts to individual fauna, including marine turtles significant

CM reference	Control measure	Environmental benefit	Potential cost/issues	Evaluation
				impacts to migratory or nesting behaviours are not expected. Activities in turtle HC area will be infrequent and short in duration. Therefore, no impacts are predicted at a population level that would justify altering the timing of the Activity.
N/A	Verification of noise levels (administrative control)	Allow adaptive management controls to be implemented if impact is greater than expected. May help verify estimated potential noise impact zones.	Costs of deploying noise monitoring equipment and processing data. Field monitoring program not warranted where potential impacts are low risk.	Not adopted – Cost considered disproportionate to the environmental benefit, given the rapid reduction in noise levels from Activity vessels and the low- level behavioural response expected.
N/A	Implement a shutdown zone around MBES and SSS in the OA (elimination control)	Implement an area around survey vessels in which equipment is shut down if marine fauna are observed within a defined zone of potential physiological impact.	Additional costs for suitably trained personnel to observe fauna around the shutdown zone, and potential disruption to the Activity if operations must stop while in progress.	<b>Not adopted</b> – MBES and SSS surveys are infrequently conducted as part of ongoing operations.
N/A	Dedicated marine mammal observers (MMOs) (administrative control)	Improved ability to spot and identify marine fauna.	Additional cost of contracting several specialist MMOs. Even if marine fauna are identified, noise sources cannot be shut down since they are integral to the safe operation of Activity vessels.	Not adopted – cost disproportionate to increase in environmental benefit given that seismic surveys require MMOs and seismic surveys are not being conducted (as per EPBC Policy Statement 2.1 Part B.1). Activity noise is considered negligible
NA	Noise management plan (administrative control)	Impacts are predicted to be minor (e.g. potential temporary and minor behavioural changes); therefore, a noise management plan and associated management controls would have little or no benefit on outcomes (i.e. further reducing impacts).	Personnel costs of preparing and reviewing the management plan.	Not adopted – the OA is not near any resting, foraging, calving or confined migratory pathways for protected cetacean species. Therefore, the cost outweighs the benefit associated with developing a management plan for an Activity of short duration and minor impact (i.e. potential temporary and minor behavioural changes).
N/A	Helicopters will not land or take off if marine megafauna are	Reduces potential impacts to megafauna.	May impact safety during landing or take off.	Not adopted – increased exposure risk to passengers.

CM reference	Control measure	Environmental benefit	Potential cost/issues	Evaluation
	present in the vicinity of an Activity vessel (elimination control)			Risk of exhausting fuel supplies.
N/A	Application of EPBC Policy Statement 2.1 Interaction between offshore seismic exploration and whales	Minimises potential impacts to whales as a result of seismic activities through visual observations, soft starts, stop work, nighttime and low visibility procedures. No seismic activities will be undertaken as part of the proposed activities.	Additional cost of contracting several specialist MMOs. Cost associated with soft starts and stop work procedures.	Not adopted – cost disproportionate to increase in environmental benefit given there are no seismic surveys (as per EPBC Policy Statement 2.1) activity noise generated is considered negligible and no known whale BIAs overlap (or are close to) the OA. Vessels noise is infrequent, and no seismic activities will be undertaken.
N/A	Use of passive acoustic monitoring	Improve detection of some sensitive receptors.	Costs of passive acoustic monitoring operators. Operational costs of shutdowns will potentially prolong the activity.	Not adopted- Cost incurred disproportionate to increase in environmental benefit given the low-level behavioural response expected. Limited ability of passive acoustic monitoring to detect cetaceans would provide little benefit to species expected to be present.
N/A	Manage timing of activities to avoid coinciding with sensitive periods for marine fauna present in the operations area (pygmy blue whale migration period)	Reduces the risk of impacts from noise emissions during environmentally sensitive periods for listed marine fauna	High cost in moving or delaying activity schedule (for the Barossa GEP from the FPSO to DLNG). The risk to all listed marine fauna cannot be reduced due to variability in timing of environmentally sensitive periods and unpredictable presence of some species.	Not adopted - TTS thresholds from underwater noise will be limited to within a few hundred meters of the source and will not overlap the water depths (500m+) that pygmy blue whales are known to use during their migration. Therefore, potential for impacts to migrating whales extremely limited, given close proximity to the source for underwater noise to fall below TTS levels (considered an injury under the Management Plan for Pygmy Blue Whales), in water depths shallower than preferred by migrating pygmy blue whales. Therefore, the activities are not inconsistent with the objectives of the

CM reference	Control measure	Environmental benefit	Potential cost/issues	Evaluation
				Pygmy Blue Whale Management Plan.
N/A	Vessel activities will only occur during daylight hours.	Reduces the risk of impacts from noise emissions due to vessel being stationary at night	Vessel are required to work 24 hours when undertaking operations and maintenance activities to ensure the safe operation of the facility. Limiting the activity to daylight hours would also result in significant financial costs.[	Not adopted-limiting the activity to daylight hours would increase the time required to undertake activities to ensure the safe operation of the facility. The high financial cost would be grossly disproportionate to negligible environmental benefits.

### 6.1.4 Environmental impact assessment

Receptor	Consequence level
Noise emissions	
Threatened, Migratory or local fauna	While the level of noise expected from operational activities has the potential to cause physical injury to marine fauna, most species that may transit through the OA are expected to demonstrate avoidance behaviour if noise levels approach those that could cause pathological effects. Avoidance behaviour is likely to be localised (less than 1 km) within the area of the activity vessels (due to the small spatial extent of elevated noise) and temporary (duration of the activity vessels operating).
	Impacts to marine mammals from underwater noise generated by the Activity are considered unlikely to be substantial given that there are no significant feeding, breeding or aggregation areas in the vicinity of the OA. The closest marine mammal BIA is the dolphin breeding BIA, located approximately 46 km from the OA, outside the area predicted to exceed thresholds for behavioural, masking or physiological impacts. The nearest whale (pygmy blue) distribution is greater than 300 km away. Any responses will be limited to transiting individuals, which is unlikely to result in substantial impacts to marine mammal populations or distribution. Behavioural impacts may include increased swimming speed, changes in dive behaviour or avoidance of the area. Such impacts would be temporary, with no significant impacts predicted to individuals or populations. Potential behavioural impacts from underwater noise will be limited to within 9.8 km of activity vessels. There is potential for TTS to occur within 120 m and 1,500 m from the source for HF cetaceans, including sirenians (dugongs) and LF cetaceans, respectively. The potential for PTS in LF cetaceans is estimated to be within 110 m of the source. Notably, the modelled exposure area for the SEL24 criteria represents an area within which the animals may be exposed to sound levels associated with impairment (PTS or TTS) if they remain within the ensonified area for a duration of 24 hours.
	Behavioural impacts to fish from survey equipment noise may occur in individuals located within hundreds of metres of the source.
	Survey equipment could cause masking of vocalisations of cetaceans but would be limited to within hundreds of metres from the sound source.
	PTS and TTS thresholds for marine mammals are only expected to be exceeded close to the source. Due to the lack of aggregating areas for these species and significant distances to the nearest marine mammal BIA, individuals are expected to be transitory only, displaying behavioural responses, and moving away from the source, before TTS and PTS thresholds are exceeded.
	In the Recovery Plan for Marine Turtles in Australia 2017–2027 (CoA, 2017b), noise interference to marine turtles depends on whether the exposure is short (acute) or long-term (chronic). The noise generated by the Activity is acute, with impacts restricted to localised changes in behaviour within hundreds of metres of the source. The 20 km noise assessment boundary intersects the flatback internesting BIA and habitat critical to the survival of the flatback and habitat critical to the survival of the olive ridley turtle. However, due to the OA water depths, the BIA extending across more than 800 km of coastline, a lack of foraging habitat and that no aggregations are expected, the potential numbers of affected internesting turtles are expected to be limited. Potential impacts to marine turtles from underwater noise are considered unlikely to result in substantial impacts to populations or distribution given that



Receptor	Consequence level
	impacts are likely to be limited to behavioural and masking impacts within a relatively small area of important turtle habitat.
	Potential impacts to threatened or migratory sawfish, sharks, rays or other fish species are limited to the potential for avoidance behavioural responses within hundreds of metres of the source. Although there is the potential for TTS within this range, this is not expected due to noise avoidance behaviour. Impacts to fish are not considered to have the potential to be significant because noise levels from noise sources that may cause avoidance behavioural responses are expected to be within a radius of a few hundred metres of the noise source. Other protected species are not expected to be affected given their wide distribution (in the case of sea snakes, crocodiles and sharks), distances to seabird breeding colonies, and preference for shallow coastal habitats (sawfish). For the above reasons, no substantial change to threatened and migratory species is anticipated that may:
	Lead to a long-term decrease in the size of a population
	Reduce the area of occupancy of the species
	Fragment an existing population into 2 or more populations
	Adversely affect habitat critical to the survival of a species
	Displace threatened and migratory marine fauna from habitat critical to the survival of a species areas
	Disrupt biologically important behaviours of threatened and migratory marine fauna within BIAs
	Disrupt the breeding cycle of a population
	• Modify, destroy, remove, isolate or decrease the availability or quality of habitat to the extent that the species is likely to decline
	Interfere with the recovery of the species.
	The consequence of noise emissions is assessed as I – Negligible
Physical environment/ habitat	Not applicable – Noise will not impact the physical environment itself (including the Carbonate bank and terrace system of the Van Diemen Rise KEF that overlaps the OA). Species associated with the continental slope and patch reefs, and the hard substrate sediments of deep channels that characterise these KEFs – such as demersal fish, whale sharks, sharks, seasnakes and turtles – are unlikely to aggregate within the OA due to the lack of seafloor features. However, potential impacts to these species are described above.
Threatened ecological communities	Not applicable – No threatened ecological communities identified in the area over which noise emissions are expected.
Protected areas	Not applicable – no protected areas have been identified in the area over which noise emissions are expected.
Socioeconomic receptors	The consequence of noise emissions on receptors is assessed as I – Negligible. Noise emissions are not expected to impact commercial fisheries. Impacts to fauna, including fish and other marine species is likely to be limited to temporary behavioural impacts within a 9.8 km radius around activities, and will not result in significant impacts to marine species at the individual or population level. Given the negligible consequence to marine species, subsequent impacts to commercial fish stock are not anticipated. Impacts to socioeconomic receptors, including commercial fisheries, recreation and tourism are considered to be minor due to the localised and temporary noise levels and low socioeconomic activity levels within the noise assessment boundary.
Cultural Features	No First Nations people feedback was provided about potential noise impacts to cultural features during consultation. An important outcome of Dr Corrigan's research is that no sacred sites or dreamings are shown to be directly impacted by the DPD project footprint (now the GEP), however some marine species are known to be associated with dreamings and songs (Corrigan, 2024). It is anticipated that noise emissions are unlikely to impact traditional hunting practices or resources. For assessment of impacts to marine species that are of cultural significance and/or represent a traditional food source for First Nations groups, refer to the assessment for threatened, migratory or local fauna.
	the region, such as the Bayu-Undan pipeline since approximately 2006, the Ichthys Pipeline since approximately 2016, the North West Cable System since approximately 2016 and GEP since 2023. The region also has a history of significant historic and ongoing industrial shipping, fish trawling activities and drilling of almost 900 offshore wells. There is no evidence to support actual adverse effects from the actions of spiritual beings in response to impacts on the environment from these activities.
	Notwithstanding, in response to the concerns raised by some First Nations people during consultation for the D&C EP and the GEP EP (noting no concerns were raised by First Nations

Receptor	Consequence level
	people for this Activity during the development of this GEP Coastal Waters OEMP), a control measure (BAO-CM-6.1.4[) relating to cultural heritage training is proposed. Santos considers the adoption of BAO-CM-6.1.4 and EPO-20 practicable and appropriate.
Overall worst-case consequence	II – Minor

### 6.1.5 Demonstration of as low as reasonably practicable

The use of vessels on DP, survey equipment, and other equipment described in Section 2 for the Activity are unavoidable as there are no other options for undertaking IMMR activities.

The sound levels generated by surveys are medium– to high–frequency and decay rapidly with distance travelled from the source, as demonstrated by Zykov et al. (2013), with the furthest distance survey noise is expected to travel being hundreds of metres. Note that marine fauna affected in varying degrees by acoustic noise (i.e., marine mammals, marine reptiles, sawfish, sharks and fish) are all expected to avoid the source of noise and will be unlikely to remain within the ensonified area for a duration of 24 hours. Avoidance behaviours are likely to be from a small area and to be temporary.

All reasonably practicable control measures were reviewed and those adopted are considered appropriate to manage the impacts such that the residual consequence is assessed to be II – Minor. The proposed management controls are in accordance with Santos' risk management criteria and are considered appropriate to reduce impacts to ALARP.

In relation to spiritual and/or cultural heritage beliefs and connections to sea country and related concerns of some First Nations people, Santos has been implementing cultural heritage training and ceremony while undertaking activities authorised pursuant to the GEP EP since November 2023, with broad support of First Nations communities as a culturally appropriate practice and response to cultural concerns. Santos considers that the adopted control measure (BAO-CM-6.1.4) based on the Corrigan 2024 Report recommendations will reduce environmental impacts and risks to ALARP, as relevant to First Nations individuals who hold these concerns in relation to their beliefs.

The proposed management controls are in accordance with Santos' risk management criteria and are considered appropriate to reduce impacts to ALARP.

### 6.1.6 Acceptability evaluation

Is the consequence ranked as I or II?	Yes – maximum consequence from noise emissions is II – Minor.
Is further information required to validate the consequence assessment?	No – potential impacts and risks are well understood through the information available.
Are the risks and impacts consistent with the principles of ecological sustainable development?	Yes – activity evaluated in accordance with Santos' Offshore Division Environmental Hazard Identification and Assessment Guideline which considers the principles of ESD:
	The impacts associated with noise emissions do not result in 'threats of serious or irreversible harm' as detailed within the EPBC Act and biodiversity and ecological integrity will be maintained.
	Conservative assumptions have been applied to the underwater noise modelling that has been adapted for this assessment.
	There are no identified health, diversity or productivity impacts that may affect the biodiversity or ecological function for future generations.
	The consequence against this aspect is II (Minor) and therefore does not affect the outcomes of the principles of ESD.
Have the acceptable levels of impact and risks been informed by relevant species recovery plans, threat abatement plans, conservation	Yes – controls implemented will minimise the potential impacts from the activity to species identified in recovery plans and conservation advice as having the potential to be impacted by noise emissions.
advice, wildlife conservation plans and Australian marine park zoning objectives)?	Consistent with relevant species recovery plans, conservation advice, wildlife conservation plans and management actions set out in Table 3-13, including:
	<ul> <li>Recovery Plans: The Recovery Plan for Marine Turtles in Australia 2017–2027 (CoA, 2017b)Conservation Management Plan for the Blue Whale 2015–2025 (CoA, 2015a)</li> </ul>
	Conservation Advice:
	<ul> <li>Approved Conservation Advice for Balaenoptera borealis (sei whale) (TSSC, 2015b)</li> </ul>



	<ul> <li>Approved Conservation Advice for Balaenoptera physalus (fin whale) (TSSC, 2015c).</li> <li>For the identified plans, the objectives of those plans are achieved through the adoption of performance outcomes and the control measures outlined in Section 6.1.3. Santos considers that the level of impact from Activity noise emissions is not inconsistent with these plans.</li> </ul>
Are performance outcomes, control measures and associated performance standards consistent with legal and regulatory requirements?	Yes – management consistent with relevant sections of EPBC Regulations Part 8. Through acceptance of this GEP Coastal Waters OEMP, legislative and regulatory requirements will be met as per Appendix B. The GEP Coastal Waters OEMP is also compliant with the conditions of EPBC approval EPBC 2022/09372 (Appendix B).
Are performance outcomes, control measures and associated performance standards consistent with Santos' Environment, Health and Safety Policy?	Yes – aligns with Santos' Environment, Health and Safety Policy (Appendix A).
Are performance outcomes, control measures and associated performance standards consistent with industry standards?	Yes – the most recent and comparable Operations EPs accepted by NOPSEMA have been reviewed for consistency with the performance outcomes, control measures and associated performance standards proposed in this GEP Coastal Waters OEMP.
Have performance outcomes, control measures and associated performance standards taken into consideration Relevant Person feedback	Yes – Relevant Persons feedback for this Activity has been considered when evaluating the EPOs, CMs, or EPSs. An additional performance outcome and control measures (e.g., EPO-19 and BAO-CM-6.1.4) have been adopted based on Relevant Persons feedback on other Barossa Gas Project EPs.
Are performance standards such that the impact or risk is considered to be ALARP?	Yes – ALARP assessment conducted, with no additional control measures adopted.

The consequence of noise emissions is assessed as II – Minor. Based on an assessment of Santos' acceptability criteria and with the control measures in place, potential impacts are considered acceptable.



### 6.2 Light emissions

### 6.2.1 Description of event

Event	Light emissions will occur from Activity vessels and other support vessels. Sources of impacts from lighting on sensitive receptors within the OA may occur as a result of:
	<ul> <li>safety and navigational lighting on vessels and ROVs undertaking IMMR activities</li> </ul>
	<ul> <li>spot lighting used when needed, such as equipment deployment and retrieval</li> </ul>
	Lighting will consist of bright white (as in, metal halide, halogen, fluorescent) lights typical of lighting used in offshore petroleum and other offshore activities in the region, such as fishing and shipping.
	Activity vessels will have external lighting to provide a safe working environment and to comply with relevant maritime navigation requirements at night. Lighting levels will primarily be determined by operational safety and navigational requirements under relevant legislation, specifically the <i>Navigation Act 2012</i> (Cth). Activity vessels will be required to generate and use navigational lighting at night to indicate their position, and they must also indicate their limited ability to manoeuvre during operations under the <i>Navigation Act 2012</i> (Cth).
Extent	No permanent light sources will be required along the pipeline; however, lighting may be required for periodic activities, such as IMMR. Light or light glow from is expected to reduce to levels unlikely to cause biological effects at 3.3 km from the source, either moving slowly along the pipeline or in one area (depending on the type of IMMR).
	The light assessment boundary of 20 km from the source is considered representative of the extent of light exposure, in accordance with National Light Pollution Guidelines for Wildlife (NLPGW) (DCCEEW, 2023c). This additional 20 km buffer around the OA is the extent relevant to the impact assessment for planned light emissions. Cumulative modelling of 2 vessels working together indicates that light is predicted to reduce to below ambient levels at approximately 21.6 km and potential behavioural impacts to turtles is limited to 4.5 km (Pendoley, 2022).
Duration	Navigational and task lighting is required 24 hours a day for the duration of the IMMR activity. IMMR vessels are typically present for 7 to 30 days along the pipeline route during planned inspection campaigns. These campaigns occur every three to five years or as needed.

### 6.2.2 Nature and scale of environmental impacts

**Potential receptors:** threatened, migratory or local fauna (marine mammals, marine reptiles, seabirds, sharks, rays and other fish); socioeconomic; and cultural features.

To humans, light is visible between wavelengths of approximately 380–780 nanometres (nm), between the violet and red regions of the electromagnetic spectrum. In fauna, light is visible between 300 nm and >700 nm, depending on the species. Some fauna do not see long wavelength red light at all, while others see light beyond the blue-violet end of the spectrum and into the ultraviolet (DCCEEW, 2023c). Therefore, the source of light impacts relates to both the amount of artificial light, as well as the types of light and the wavelengths that different lights emit.

The combinations of colour, intensity, closeness, direction and persistence of a light source are key factors in determining the magnitude of environmental impact (DCCEEW, 2023c). Light from Activity vessels may appear from direct, unshielded light sources or through sky glow. Where direct light falls upon the ocean, this area of light is referred to as light spill. Sky glow is the diffuse glow caused by light that is screened from view, but through reflection and refraction creates a glow in the atmosphere. The distance at which direct light and sky glow may be visible from the source is dependent on the lighting on the vessel and environmental conditions.

Activity vessels will have external lighting to provide a safe working environment and to comply with relevant maritime navigation requirements at night. Activity vessels may also periodically require additional lighting for 24-hour operations in short durations at specific points along the pipeline. Construction or pipelay vessels are larger than those proposed for typical IMMR activities and are therefore a conservative worst-case estimate of light emission that may be visible for Activity vessels.

Figure 6-1 shows photographs of a typical pipelay vessel, Audacia, with lights on at dusk. Lights include:

- regular halogen light bulbs (60–75 watt [W]) and fluorescent lights (18–36 W) that illuminate various gangways throughout the vessel and will be on all night for safety reasons
- floodlights of various power ratings (250–500 W) that illuminate working areas
- helideck lights including floodlights (35 W) and light-emitting diodes (LEDs) (3 W) that provide lighting for the helicopter platform during night-time operations. Such lighting is obligatory, but the platform will only be lit for safe helicopter landing and take-off activities (e.g., medivacs or inspections). This lighting will be turned off during normal operations at night that do not involve helicopters



- navigation LEDs, which are installed at various locations around the vessel and are obligatory
- search lights, which are very bright but only used in emergency situations; these are turned off under normal operation.

Unless specifically required to support over the side activities, such as lifting, IMMR, or for navigational purposes, lighting is directed over the work area, which aids in limiting light spill to the marine environment.





Figure 6-1: Photographs of a typical pipelay vessel at dusk

Light modelling was undertaken for installing the pipeline (as described in the Barossa Gas Export Pipeline Installation EP, and the Darwin Pipeline Duplication Offshore Construction Environmental Management Plan. These vessels are larger (and therefore expected to produce a greater level of light spill) than those proposed for typical support and IMMR activities covered by this GEP Coastal Waters OEMP. The light spill modelling conducted is therefore conservative for typical support and IMMR activities (including USV) covered by this GEP Coastal Waters OEMP but representative for larger contingency campaign vessels that are less likely to be required.

Light modelling was undertaken for construction vessels to predict the extent of biologically relevant light spill. Specifics of the respective vessels' lighting design and luminaire specifications were applied to the ILLUMINA Artificial Light At Night (ALAN) model (Aubé et al., 2005). The ILLUMINA model is a three dimensional (3D) model



that accounts for both line of sight and atmospheric scattering, allowing the attenuation of light over distance and extent of light glow to be modelled.

Since light sources (i.e., individual luminaires) can be placed individually within the area of interest, the model can replicate specific lighting designs in terms of light type, spectral distribution, height and orientation of individual luminaires, including any shielding, thus increasing model accuracy. This information was extracted from lighting layout drawings and light manufacturer data sheets for the typical pipelay and construction vessels, *Audacia* and *Fortitude*, respectively. The model assumed that all vessel lights were turned on (apart from search lights, which are only used in an emergency) with no additional shielding other than that provided by the vessel structures. It also assumed vessels were orientated north–south and that cloud cover was zero (no contribution of light from cloud reflectance). Model outputs are provided in radiance ( $W/m^2/sr$ , where W = watts,  $m^2 =$  square metres and sr = steradian). The cumulative impact of the construction and pipelay vessels working together was also modelled. Model results are independent of location-specific variables so are representative along the pipeline route in the OA.

In the absence of any published or generally accepted units of measure, or scale, for measuring the impact of artificial light at night on turtle hatchlings, moonlight was used as a proxy. Output from the light model (radiance, units of W/m²/sr) was converted to units of full moon equivalents to provide biological relevance to the radiance output.

Table 6-11 presents potential impact criteria for marine turtles related to the proportion of radiance of a full moon. This was derived by Pendoley (2022) using their extensive experience observing marine turtles and how they respond to light in field settings. The range of moon brightness across a whole lunar cycle provides a realistic scale representative of ambient light levels to which turtle eyes are adapted. The scale is logarithmic to represent the nature of light decay with distance (a function of the inverse square law). At the lower end of the scale, the radiant output is equivalent to no light in the sky (a new moon) while the upper limit is equivalent to the brightness of 10 full moons.

Impact Leel	OFOV FME <sup>45</sup> ranges*	Impact potential to marine turtles	
5	10 to 100	Light or light glow visible and impact likely	
4	1 to 10	Light or light glow visible and impact likely	
3	0.1 to 1	Light or light glow visible and behavioural impact possible, depending on ambient moon phase	
2	0.01 to 0.1	Light or light glow visible but behavioural impact unlikely (i.e. not biologically relevant)	
1	<0.01	Light or light glow is considered ambient, and no impact expected	

### Table 6-11: Artificial light impact potential criteria (marine turtles)

Source: Pendoley (2022)

\*Where 10 equals the radiance of 10 full moons and 0.01 equals 100th the radiance of one full moon.

The Pendoley (2022) report provides ILLUMINA light modelling for an offshore pipelay vessel and construction vessel, as well as a cumulative assessment (combined light spill) of both vessels side-by-side.

Light emissions were predicted to reduce to below ambient levels (0.01 orientation field of view full moon equivalents [OFOV FME], or 1%, radiance of a full moon) at 14.8 km from the offshore pipelay vessel, 10.9 km from the construction vessel, and 21.6 km when both vessels are together (Pendoley, 2022). It is predicted that behavioural impacts on turtles are unlikely to occur. (0.01–0.1 OFOV FME, or 10%, radiance of a full moon) within 3.3 km of the offshore pipelay vessel, 2.5 km of the construction vessel, and 4.5 km when both vessels are together (see

Table 6-12). The closest regionally significant flatback turtle nesting site is located at Cape Fourcroy on the Tiwi Islands, NT (~31.5 km from the OA).

Therefore, since this modelling of larger pipelay vessels shows that light spill will not impact Cape Fourcroy, the light spill from smaller IMMR vessels is also expected to have no impact.

### Table 6-12: Distance of equivalent moon radiances from the source (source: Pendoley, 2022)

<sup>&</sup>lt;sup>45</sup>orientation field of view full moon equivalents



Proportion of radiance of a	Distance from source (m)			
full moon*	Pipelay vessel	Construction vessel	Cumulative	
10-100	<160	<126	<202	
1-10	160	126	202	
0.1-1	724	557	957	
0.01-0.1	3,274	2,469	4,542	
<0.01	>14,804	>10,949	>21,559	

Source: Pendoley (2022)

\* Where 10 equals the radiance of 10 full moons and 0.01 equals 100th the radiance of one full moon.

Continuous lighting may result in localised alterations to normal marine fauna behaviours that can alter foraging and breeding activity. Marine turtle and seabird species have the greatest sensitivity to light. Receptors within a 20 km buffer of the OA were considered as having potential for interaction with light emissions, as recommended in the NLPGW (DCCEEW, 2023c). The 20 km threshold provides a precautionary limit based on the observed effects of sky glow on marine turtle hatchlings (15 to 18 km) and fledgling seabirds grounded in response to artificial light 15 km away (DCCEEW, 2023c). A PMST search was conducted for a 20 km light assessment boundary around the OA. No additional species were identified within the light assessment boundary compared to the OA (Table 3-10). An internesting BIA and habitat critical area for flatback turtles, and the habitat critical to the survival of olive ridley overlaps the light assessment boundary. The specific impacts on different fauna groups are described below.

### 6.2.2.1 Marine mammals

No marine mammal BIAs occur within the 20 km light assessment boundary. Three threatened and migratory marine mammal species may be present in the area: the blue whale, fin whale, and sei whale. Light is not listed as a threat in the conservation advice or recovery plans for these species, nor in the Conservation Management Plan for the Blue Whale 2015–2025 (CoA, 2015a).

Marine mammals are not known to be attracted to light sources at sea. Cetaceans predominantly use acoustic senses to monitor their environment rather than visual cues (Simmonds et al., 2004). However, light glow may act as an attractant to light-sensitive prey species (e.g., squid and fish) that may alter predator-prey dynamics, particularly in dolphins.

Since mammals use variations in the length of day to anticipate environmental changes and time their reproduction, light emissions that affect the perception of day-length (e.g., 24-hour lighting on the Activity vessels) could impact biological functions. However, marine mammals will likely be transient in the OA; therefore, individuals are unlikely to be exposed to artificial light for durations sufficient to impact biological functions. The impact from light is considered negligible.

### 6.2.2.2 Marine reptiles

### 6.2.2.2.1 Turtles

The NLPGW (DCCEEW, 2023b) state that a 20 km buffer (based on sky glow) to important habitats for turtles should be applied when considering possible impacts. However, the demonstrated impacts on which this buffer is based were in response to light emissions associated with a liquefied natural gas (LNG) plant. The light modelling for GEP installation found that the spatial extent of a measurable change in ambient light from the pipelay and construction vessels is predicted to be approximately 14.8 km and 10.9 km, respectively (Pendoley, 2022). The cumulative impact of these vessels working together is predicted to reduce below ambient levels at approximately 24.6 km. Potential behavioural impacts to turtles is predicted within 3.3 km of the pipelay vessel, 2.5 km of the construction vessel and 4.5 km cumulative impact (Table 6-12) (Pendoley, 2022). These studies indicate that the spatial extent of a change to ambient light is less than the 20 km light assessment boundary used for impact assessment, based on the NLPGW (DCCEEW, 2023b).

An internesting BIA and nesting HC for flatback turtles overlaps the OA, and a nesting HC for olive ridley turtles overlaps the 20 km light assessment boundary. No other turtle BIA or nesting habitat overlap the light assessment boundary. Flatback and olive ridley turtles nest at beaches along the west coast of Bathurst Island approximately 30 km from the OA. Unlike other turtle populations (for example, on the northwest shelf of WA), the flatback and olive ridley turtles on Bathurst Island do not exhibit discrete nesting and hatching seasons. Rather, there is low level nesting year-round, with a peak in nesting, internesting and hatching during winter months. While artificial lighting on or near beaches is known to disrupt nesting behaviour (Witherington & Martin, 2003), the nearest nesting beaches are outside the light assessment boundary, thus behavioural impacts to nesting females at nesting beaches are not expected.

Suitable internesting habitat for flatback turtles is defined as water depths shallower than 16 m (Whittock et al., 2016; Pendoley, 2019). Similarly, internesting olive ridley turtles have been shown to remain relatively close to nesting beaches during the nesting period, in waters typically less than 30 m depth (Hamel et al., 2008). Given that water depths in the OA are between 47 m and 50 m, the flatback turtle BIA and HC areas extend > 800 km coastline, and the lack of foraging habitat in the OA, the number of individual turtles likely to be present is expected to be limited.

No evidence, published or anecdotal, suggests internesting turtles are impacted by light from either natural or anthropogenic sources, as they do not use light as a cue for this behaviour. Furthermore, nothing in their biology would indicate this as a plausible threat (Pendoley, 2019; Witherington and Martin, 2003). Individual turtles (green, olive ridley, loggerhead, leatherback, hawksbill) may transit the OA to forage or migrate to suitable habitat (e.g., nesting beaches and shoals). For the reasons set out above, similarly to interesting turtles, if individual turtles are present, light emissions from Activity vessels are unlikely to be of concern.

Hatchlings emerging from the sand are known to locate the ocean using a combination of topographic and brightness cues (Limpus, 1971; Limpus & Kamrowski, 2013; Pendoley & Kamrowski, 2016; Salmon et al., 1992). Once hatchlings enter the ocean, they are thought to employ a survival strategy that involves rapid dispersal away from predator-rich nearshore habitats to reach deeper waters where they develop into juveniles. An internal compass is set while crawling down the beach, and wave cues are used to reliably guide them offshore (Lohmann & Fittinghoff-Lohmann, 1992; Stapput & Wiltschko, 2005). In the absence of wave cues, however, swimming hatchlings have been shown to orient towards light cues (Lorne & Salmon, 2007; Harewood & Horrocks, 2008), and in some cases, wave cues were overridden by light cues (Thums et al., 2013, 2016). Currents substantially influence the speed and direction of at-sea dispersal. For example, the offshore trajectory of flatback hatchlings at Thevenard Island (WA) was displaced by tidal currents which ran parallel to the beach, an effect that increased as the hatchlings moved further offshore (Wilson et al., 2018).

However, when light was present, this effect was diminished, showing that hatchlings actively swam against currents and towards the light source, which slowed their offshore dispersal from 0.5 m/s when no light was present, to 0.35–0.44 m/s, depending on the type of light (Wilson et al., 2018). The mean swimming speeds of flatback hatchlings under natural light conditions (0.5 m/s) were similar to green turtle hatchlings (0.49 m/s) (Thums et al., 2016). The swimming speed of olive ridley hatchlings has not been measured; however, since they are smaller than flatback and green turtle hatchlings, swimming speeds are expected to be lower (Pendoley et al., 2022).

These results suggest that hatchlings can move in any direction when their swimming speed is greater than the speed of the nearshore current, although the speed at which currents can no longer be overcome is species-specific and related to swimming speeds. Wilson et al. (2018) reported that when flatback hatchlings were within 150 m of the beach, they were able to swim against currents up to 0.3 m/s, although 0.3 m/s was the maximum current speed recorded during the study. Therefore, whether flatback hatchlings can swim against stronger currents is currently untested. If an olive ridley hatchling has a similar response to light cues as flatback hatchlings, their smaller size suggests a reduced capability to swim against currents compared to flatback turtles.

The attraction of dispersing hatchlings to vessel light emissions and spill could result in two main impacts: increased energy expenditure as hatchlings swim against currents towards light sources and when entrapped in light spill, as well as increased risk of predation while silhouetted in areas of light spill.

The Recovery Plan for Marine Turtles in Australia 2017–2027 (CoA, 2017b) highlights artificial light as a threat to marine turtles. Specifically, the plan indicates that artificial light may reduce the overall reproductive output of a stock, and therefore recovery of the species by:

- inhibiting nesting by females
- disrupting hatchling orientation and sea-finding behaviour
- creating pools of light that attract swimming hatchlings and increase their risk of predation.

The most significant risk posed to marine turtles from artificial lighting is the potential disorientation of hatchlings following their emergence from nests by light spill on beaches, although breeding adult turtles can also be disoriented (Longcore and Rich, 2016). The NLPGW (DCCEEW, 2023b) states that within 15 km of the nesting beach, light impacts may affect flatback turtle hatchling behaviours. The nearest turtle nesting beach is approximately 30 km from the OA, and modelling predicts that light spill at an intensity that could lead to turtle behavioural effects are possible at distances less than 3.2 km for the pipelay vessel, 2.4 km for the construction vessel, and 4.5 km for both vessels together (Pendoley, 2022). Therefore, impacts to hatchlings are considered unlikely.

However, there is potential for hatchlings at sea to be attracted to light emissions if they are carried by currents to within these modelled limits. The likelihood of attraction would be lower during periods of full moon, reducing the potential impact of the Activity in these periods. If attraction did occur, it is likely individuals would remain entrapped in light for short periods (Wilson et al., 2018; Thums et al., 2016). At worst case, individuals would be trapped until



dawn. Further, since nesting occurs year-round, there will be a significant proportion of hatchlings originating from Bathurst Island that are not exposed to potential light sources.

In summary, vessel light emissions are not expected to impact nesting females or emerging hatchlings at nesting beaches since modelling predicts that light or light glow will not exceed intensities considered biologically relevant (Pendoley, 2022). Modelling results are conservatively based on light emissions from pipelay and construction vessels, which are assumed to be the worst-case for typical operational and IMMR activities. Additionally, vessel light emissions are not expected to impact individual internesting turtles since there is no evidence, published or anecdotal, to suggest internesting turtles are impacted by light from offshore vessels. Finally, the extent of biologically relevant light will not impact any one location for an extended duration, given the area of disturbance will be based on a vessel slowly moving along a defined pipeline route.

### 6.2.2.2.2 Sea snakes

Studies have shown that sea snakes display varying responses to light. For example, *Hydrophis* species appear to be attracted to light and have been observed floating on the sea surface and swimming up to light (pers. comm. M. Guinea, Charles Darwin University, 2014). However, the *Aipysurus* species of sea snake do not appear to be attracted to light and are not seen on the surface at night (pers. comm. M. Guinea, Charles Darwin University, 2014). Most sea snakes are likely to be associated with the shoals and banks, with the closest being Shepparton Shoal (23 km from the OA) and Afghan Shoal (24 km from the OA). It is recognised that some individuals (*Pelamis* genus) may occur in the OA and may be attracted to the light from Activity vessels; however, it is considered unlikely that they will stay within the area (pers. comm. M. Guinea, Charles Darwin University, 2014).

### 6.2.2.3 Sharks, rays and fish

Fish at the surface of the water have the potential to be impacted by artificial light. The response of fish to light emissions varies according to species and habitat. Experiments using light traps have found that some fish and zooplankton species are attracted to light sources (Meekan et al., 2001), with traps drawing catches from up to 90 m away (Milicich et al., 1992). Lindquist et al. (2005) concluded from a study that artificial lighting associated with offshore energy industry activities resulted in an increased abundance of clupeids (herring and sardines) and engraulids (anchovies). These species are known to be highly photopositive (i.e., tend to move towards light). The artificial light serves to focus their marine plankton prey and consequently leads to greater foraging success.

The area of impact is likely to be restricted to areas of light spill on the ocean surface. Fish may move toward and/or aggregate around light emissions from Activity vessels in the OA; however, the transitory presence of typical Activity vessels and the short duration of the Activities reduce the likelihood of this occurring.

Sharks and rays are not known to be significantly attracted to light sources at sea. However, they may be attracted to fish that are photopositive to light. Given the movement of typical Activity vessels short duration of the Activities and absence of critical habitats within the OA, light impacts will not result in population level effects and will not extend to any areas of biological importance for these species. Light has not been identified as a key threat for the whale shark (Section 3.2.13.6), and although whale sharks may forage around the Activity vessels if prey abundance is increased, this is unlikely to impede migration.

### 6.2.2.4 Seabirds and migratory shorebirds

Seabirds and shorebirds may be directly or indirectly attracted by artificial light. Structures in offshore environments tend to attract marine life at all trophic levels, creating food sources and providing artificial shelter for birds (Surman, 2002). Offshore light sources may also provide enhanced capability for seabirds to forage at night. Artificial light can disorient seabirds, disrupt natural foraging and migratory behaviours, and potentially cause injury through interaction with infrastructure.

The NLPGW recommend using a 20 km threshold, which provides a precautionary limit based on the observed effects of sky glow on fledgling seabirds grounded in response to artificial light 15 km away (DCCEEW, 2023b). No seabird BIAs occur within 20km of the OA. There are three threatened bird species that may occur in the OA: curlew sandpiper, eastern curlew, and red knot (Table 3-10). Conservation advice does not identify light as a threat to these species (Table 3-13); however, light pollution is identified as a low-risk threat in the Wildlife Conservation Plan for Seabirds (CoA, 2020).

Nocturnal species may pass through the area but are not expected in large numbers and are likely to be adults migrating or foraging, which are less vulnerable to light emissions than fledglings. Foraging species are unlikely to be disorientated by light emissions, given the scale of lighting required for typical IMMR vessels and the infrequent and relatively short-term duration of Activities.

### 6.2.2.5 **Protected areas and key ecological features**

The nearest protected area is the Oceanic Shoals AMP Habitat Protection Zone, approximately 67.5 km from the OA. Many of the values and sensitivities of the marine park are described above (e.g., turtles), and are outside of


the 20 km light assessment boundary. The nearest KEF is the Carbonate bank and terrace system of the Van Diemen Rise, approximately 9 km from the OA. However, this is a submerged receptor, and the majority of the KEF is outside the light assessment boundary.

#### 6.2.2.6 Socioeconomic receptors and cultural features

No First Nations people feedback was provided about potential light impacts to any geographically specific cultural features (excluding marine fauna species) during consultation (refer to Section 4.8). Information provided by some Tiwi people during consultation for this GEP Coastal Waters OEMP and the D&C and GEP EPs, raised concerns about the potential impacts of lights on marine turtles from Barossa activities, and potential impacts to marine life generally, and that if totemic species (e.g. turtles) are impacted by the Barossa activities this can impact Tiwi people and make them sick.

Impacts to turtles from the Activity lighting are expected to be restricted to localised attraction and temporary disorientation, with no long-term or residual impact and no impact to nesting beaches given their distance from the OA. It is noted that the closest regionally significant flatback turtle nesting site is located at Cape Fourcroy on the Tiwi Islands, which is ~31.5 km from the OA.

In the 2022 Statement of Reasons Requests, the Tiwi clan members raised their concern regarding traditional hunting of marine species and totem species. First Nations people maintain a continuing spiritual connection with sea country, including marine fauna species with cultural significance, such as totems or as a cultural food source. The potential impacts to culturally significant marine fauna species (such as dreamings and totem species including marine mammals, marine reptiles, sawfish, sharks, rays and other fish) are assessed in Sections 6.2.2.1 to 6.2.2.4. The potential impacts to marine fauna is likely to be limited to localised, temporary behavioural impacts and is unlikely to result in significant impacts to marine species at the individual or population level. Consequently, it is anticipated that light emissions are unlikely to affect traditional hunting practices or resources.

Impacts to culturally significant marine fauna are likely to be limited to localised, temporary behavioural impacts and are unlikely to have long-term consequences at the individual or population levels. Impacts on marine turtles from lighting on Activity vessels are expected to be limited to localised attraction and temporary disorientation. Light emissions are not expected to impact nesting females or emerging hatchlings at nesting beaches, since modelling predicts that light or light glow will not exceed intensities considered biologically relevant (Pendoley, 2022) given their distance from the OA (see Section 6.2.2.2.1).

Other socio-economic receptors include commercial fisheries. The Northern Prawn Fishery (NPF) overlaps the OA and surrounding waters. Light emissions from Activity vessels are not expected to impact the NPF medium and high-intensity fishing areas.

Given the nature of typical IMMR activities and the minor consequence to culturally significant marine fauna and other socioeconomic receptors, impacts from light emissions from Activity vessels are expected to be limited.

#### 6.2.3 Environmental performance outcomes and control measures

The EPOs relating to this event are:

- No injury of, or mortality to, EPBC Act 1999 listed marine fauna or marine fauna listed as threatened species and under the Territory Parks and Wildlife Conservation Act 1976 (NT) as a result of the Activity. (EPO-08)
- No light emissions from the Activity except as required for safe operations and working requirements (EPO-12)
- No significant impact to cultural features from the Activity (EPO-19)

An assessment of the environmental benefits and the potential costs or issues associated with control measures for this Activity are described in Table 6-13 to demonstrate the potential impacts from this aspect are ALARP. Control measures that are adopted have associated EPSs and measurement criteria and are presented in Table 8-2. Rejected control measures have an ALARP evaluation provided to justify their rejection.

CM reference	Control measure	Environmental benefit	Potential cost/issues	Evaluation
Standard control measures				
BAO-CM-6.1.2	Vessels equipped and crewed in accordance with Australian maritime requirements,	Light spill from unnecessary lighting reduced when vessels meet the minimum lighting requirements,	Lighting is required to ensure safe working conditions, and to alert other marine users to vessel presence.	Adopted – requirement to comply with maritime and safety regulations.

#### Table 6-13: Control measure evaluation for light emissions

CM reference	Control measure	Environmental benefit	Potential cost/issues	Evaluation
	including Marine Order 30 (Prevention of Collisions) and Marine Order 21 (Safety and Emergency Arrangements) (administrative control)	further lowering potential additional light pollution to the environment, thus reducing the potential impacts to fauna.		
BAO-CM-6.2.1	Lighting limited to that required for safe work conditions and navigational purposes (isolation control)	Light spill from unnecessary lighting reduced, even further lowering the likelihood of impacts to fauna from vessel lighting. Lighting is assessed to only provide necessary lighting for safety and navigation during the Activity. Reduces the potential for additional light pollution to the environment, thus reducing the potential impacts to marine fauna.	Limited additional cost associated with compliance assurance only.	Adopted - requirement to comply with maritime and safety regulations.
BAO-CM-6.2.2	Additional lighting management (as recommended in the National Light Pollution Guidelines for Wildlife (DCCEEW, 2023c) implemented in the OA when undertaking activities within 3.3 km of turtle BIA or HC, where it does not impact the ability of light to safely illuminate the work area (administrative control)	Reduces light spill when operating vessels within 3.3 km of turtle HC.	Costs associated with lighting assessment and installation of additional shielding (if required).	Adopted – minimises the potential impact to turtles during critical life stages.
Additional control mea	asures		I	
BAO-CM-6.2.3	Vessel searchlights will only be operated when retrieving AUVs at night or in the event of an emergency (administrative control)	Searchlights are the most significant source of light from Activity vessels. Not operating these lights during planned activities will reduce light spill.	Negligible costs	Adopted
N/A	Limit or exclude nighttime IMMR and vessel operations (elimination control)	Would reduce light emissions to the marine environment	Limiting or excluding Activities at night would result in greater impacts in other areas (e.g., waste, air emissions) and risks (e.g., vessel collision). Limiting or excluding typical IMMR Activities at night would significantly increase the Activity schedule and project costs. A minimal level of artificial lighting would	Not adopted - given the minimal risk of impacts to turtles and seabirds occurring, the financial and environmental costs of extending the Activity duration are not considered appropriate, as the extended duration may result in greater impacts from other activities (e.g., operational discharges

CM reference	Control measure	Environmental benefit	Potential cost/issues	Evaluation
			still be required on IMMR vessels on a 24-hour basis for safety reasons.	and physical presence).
N/A	Identify highest intensity lights and replace with luminaire types considered appropriate for use near marine turtle nesting habitat (substitution control)	Would reduce light emissions to the marine environment. Existing luminaries are not expected to impact turtles. Light modelling was conducted assuming all vessel lights were on, with no significant effect on overall light emissions identified.	High cost to change vessel lights. Navigational lighting colours are stipulated by law. Working and egress areas must be lit to meet health and safety requirements.	Not adopted – the high financial cost would be grossly disproportionate to the negligible environmental benefits (if any). Lighting requirements are dictated by maritime regulations, and health and safety considerations.
N/A	Change the wavelength of outdoor lights to avoid wavelengths within the peak sensitivity of turtles and seabirds (substitution control)	Would reduce light emissions to the marine environment. The light modelling predicted that light or light glow will not exceed intensities considered biologically relevant to nesting females or emerging hatchlings at the closest nesting beaches (Pendoley, 2022).	High cost to change vessel lights. Navigational lighting colours are stipulated by law. Working and egress areas must be lit to meet health and safety requirements.	Not adopted – the high financial cost would be grossly disproportionate to the negligible environmental benefits. Lighting requirements are dictated by maritime regulations, and health and safety considerations.
N/A	Use dark, matte surfaces on Activity vessels (substitution control)	Would reduce reflection and scattering of light resulting in skyglow	Additional cost to repaint surfaces. Some areas may require lighter surfaces to manage heat conduction for health and safety. Unlikely to result in a material light reduction.	Not adopted – given the short duration of Activities, the high financial cost would be grossly disproportionate to negligible environmental benefits. May compromise health and safety in some circumstances.
N/A	Avoid any IMMR activities in the OA within peak turtle nesting and hatchling emergence seasons for both flatback and olive ridley turtles (elimination control)	Reduce potential for impacts on turtles during critical life stages caused by light emissions.	Scheduling constraints to avoid peak turtle seasons for planned IMMR activities may impact vessel contracting strategy and implementation.	Not adopted – unlike other turtle populations (for example, on the NWS of WA), the olive ridley and flatback turtles on Bathurst Island do not exhibit discrete nesting and hatching seasons. Rather, there is low-level nesting year-round, with a peak in nesting, internesting and hatching during winter months. Impacts to nesting females, emerging hatchlings and dispersing hatchlings at sea are not expected to result in changes at the individual, population or genetic stock level

CM reference	Control measure	Environmental benefit	Potential cost/issues	Evaluation
				A seasonal exclusion would not avoid all turtle nesting, internesting and hatchling activity but may avoid the known peaks. The impact assessment determined the risk to hatchlings from light emissions is low and not inconsistent with the requirements of the Recovery Plan for Marine Turtles in Australia 2017–2027.
N/A	Restrict lighting to navigation lights only (administrative control)	Reduces light emissions to the marine environment.	Working and egress areas must be lit to meet health and safety requirements	Not adopted – lighting requirements are dictated by maritime regulations, and health and safety considerations.
N/A	Implement a seabird management plan to prevent seabird landings on offshore infrastructure e.g., vessels (administrative control)	A management plan to help manage birds appropriately is recommended in seabird foraging areas during breeding seasons to prevent significant impact on migration and breeding activities (CoA, 2020)	Significant costs associated with implementing controls, monitoring effectiveness, and ensuring appropriate training for personnel involved in implementing measures	Not adopted – the OA does not overlap any seabird BIAs. The closest seabird BIA is approximately 90 km from the OA at Seagull Island (for crested terns), and Activity vessels will typically be present in the OA for short durations only.

### 6.2.4 Environmental impact assessment

Receptor	Consequence level
Light emissions	
Threatened, migratory or local fauna	Sensitive receptors that may be impacted by light emissions include marine turtles, fish at the surface and seabirds.
	The NLPGW (DCCEEW, 2023c) recommends a 20 km threshold as a precautionary limit based on observed effects of sky glow on marine turtle hatchlings and fledgling seabirds.
	The 20 km light assessment boundary intersects the internesting BIA and habitat critical to the survival of the flatback, and the habitat critical to the survival of the olive ridley turtle. However, due to the OA water depths (greater than 50 m), the BIA extending across more than 800 km of coastline, and a lack of foraging habitat, the potential numbers of affected internesting turtles is expected to be limited to a small number of individuals. Internesting female turtles are not expected to be impacted by light emissions from either natural or anthropogenic sources, as they do not use light as a cue for this behaviour. Light modelling predicted a potential for behavioural impacts to turtles within 3.3 km of the pipelay vessel, 2.5 km of the construction vessel and 4.5 km cumulative impact (both vessels working together). In addition, the light from the cumulative impact will reduce to below ambient levels within 21.6 km.
	Therefore, direct light or light glow from Activity vessels in OA does not exceed intensities considered biologically relevant at the closest nesting beaches approximately 30 km distant (Pendoley, 2019). Impacts to nesting habitat or flatback hatchling behaviours are not expected. Impacts to turtles from operational activity lighting are expected to be restricted to localised attraction and temporary disorientation, but with no long-term or residual impact. Considering the distance from the nearest nesting beach, the density of post-dispersal turtle hatchlings in the OA is considered low. It is considered that the Activity will not compromise the objectives set out in the Recovery Plan for Marine Turtles in Australia 2017–2027 (CoA, 2017b). Fish (including sharks) have been shown to be attracted to artificial light sources; however, the Activity is unlikely to lead to large-scale changes in species abundance or distribution. Overall, a short-term localised increase in fish activity is expected to occur because of lighting from the

	Activity vessels, but with minor impacts to the local fish population. These behavioural responses are unlikely to significantly alter critical behaviours such as migration or spawning. Therefore, impacts to transient fish will be limited to short-term behavioural effects with no decrease in local population size or area of occupancy of species, loss or disruption of critical habitat, or disruption to the breeding cycle. There are no known BIA or nesting habitat for birds within the light assessment boundary. Considering the distance from the nearest wedge-tailed shearwater or noddy breeding colony, the density of wedge-tailed shearwater or noddy fledglings in the OA is considered to be low. Therefore, night-time lighting from the Activity is expected to have a negligible impact on breeding or fledging seabirds.
	Marine mammals are not known to be significantly attracted to light sources at sea. There is potential for opportunistic foraging should prey abundance increase; however, due to the nature and duration of typical IMMR activities, individuals are unlikely to be exposed to artificial light for durations sufficient to impact biological functions.
	The consequence level for threatened, migratory, or local fauna is considered to be II – Minor.
Physical environment or habitat	Not applicable – no impacts to physical environments or habitats from light emissions are expected
Threatened ecological communities	Not applicable – no threatened ecological communities identified in the area in which light emissions are expected
Protected areas	Not applicable – the light assessment boundary does not intersect any protected areas
Socioeconomic receptors	Impacts to fish stock are likely to be limited to localised, temporary behavioural impacts and will not result in significant impacts to fish at the individual or population level. Given the negligible consequence to fish, subsequent impacts to commercial fishing (Section 3.2.14.1) are not anticipated.
Socioeconomic receptors	Impacts to fish stock are likely to be limited to localised, temporary behavioural impacts and will not result in significant impacts to fish at the individual or population level. Given the negligible consequence to fish, subsequent impacts to commercial fishing (Section 3.2.14.1) are not anticipated. Lighting from Activity vessels is not expected to cause an impact to other socioeconomic receptors other than to act as a visual cue for avoidance of the area (for safety purposes) by other marine users, including commercial fishers.
Socioeconomic receptors	<ul> <li>Impacts to fish stock are likely to be limited to localised, temporary behavioural impacts and will not result in significant impacts to fish at the individual or population level. Given the negligible consequence to fish, subsequent impacts to commercial fishing (Section 3.2.14.1) are not anticipated.</li> <li>Lighting from Activity vessels is not expected to cause an impact to other socioeconomic receptors other than to act as a visual cue for avoidance of the area (for safety purposes) by other marine users, including commercial fishers.</li> <li>The consequence level for socioeconomic receptors is considered to be I – Negligible.</li> </ul>
Socioeconomic receptors	<ul> <li>Impacts to fish stock are likely to be limited to localised, temporary behavioural impacts and will not result in significant impacts to fish at the individual or population level. Given the negligible consequence to fish, subsequent impacts to commercial fishing (Section 3.2.14.1) are not anticipated.</li> <li>Lighting from Activity vessels is not expected to cause an impact to other socioeconomic receptors other than to act as a visual cue for avoidance of the area (for safety purposes) by other marine users, including commercial fishers.</li> <li>The consequence level for socioeconomic receptors is considered to be I – Negligible.</li> <li>An important outcome of Dr Corrigan's research is that no sacred sites or dreamings are shown to be directly impacted by the DPD project footprint (where this Activity OA is located), however some marine species are known to be associated with dreamings and songs (Corrigan, 2024). It is anticipated that light emissions are unlikely to affect traditional hunting practices or resources.</li> </ul>
Socioeconomic receptors Cultural features	<ul> <li>Impacts to fish stock are likely to be limited to localised, temporary behavioural impacts and will not result in significant impacts to fish at the individual or population level. Given the negligible consequence to fish, subsequent impacts to commercial fishing (Section 3.2.14.1) are not anticipated.</li> <li>Lighting from Activity vessels is not expected to cause an impact to other socioeconomic receptors other than to act as a visual cue for avoidance of the area (for safety purposes) by other marine users, including commercial fishers.</li> <li>The consequence level for socioeconomic receptors is considered to be I – Negligible.</li> <li>An important outcome of Dr Corrigan's research is that no sacred sites or dreamings are shown to be directly impacted by the DPD project footprint (where this Activity OA is located), however some marine species are known to be associated with dreamings and songs (Corrigan, 2024). It is anticipated that light emissions are unlikely to affect traditional hunting practices or resources.</li> <li>For assessment of impacts to marine species that are of cultural significance and/or represent a traditional food source for First Nations groups (including marine mammals, marine reptiles, sawfish, sharks, rays and other fish), refer to the assessment for threatened, migratory or local fauna.</li> </ul>

#### 6.2.5 Demonstration of as low as reasonably practicable

Artificial lighting on Activity vessels is required 24 hours a day for operational and navigational safety (e.g., Marine Order 30 – prevention of collisions). All reasonably practicable control measures were reviewed and those adopted are considered appropriate to manage the impacts such that the residual consequence is assessed to be II – Minor. The proposed management controls are in accordance with Santos' risk management criteria and are considered appropriate to reduce impacts to ALARP.

#### 6.2.6 Acceptability evaluation

The consequence of light emissions is assessed as II – Minor. Based on an assessment of Santos' acceptability criteria, potential impacts are considered acceptable with the control measures in place.

Is the consequence ranked as I or II?	Yes – maximum consequence from light emissions is II – Minor
Is further information required to validate the consequence assessment?	No – potential impacts and risks are well understood through the information available
Are the risks and impacts consistent with the principles of ecologically sustainable development (ESD)?	Yes – activity was evaluated in accordance with Santos' Offshore Division Environmental Hazard Identification and Assessment Guideline, which considers principles of ESD.
	The impacts associated with light emissions do not result in 'threats of serious or irreversible harm' as detailed within the EPBC Act, and biodiversity and ecological integrity will be maintained.
	Conservative assumptions have been applied to the light emissions modelling

	<ul> <li>There are no identified health, diversity or productivity impacts that may affect the biodiversity or ecological function for future generations</li> </ul>
	The consequence against this aspect is II – Minor; therefore, are consistent with the principles of ESD.
Have the acceptable levels of impact and risks been informed by relevant	Yes – consistent with relevant species recovery plans, conservation management plans and management actions set out in Table 3-13, including:
species recovery plans, threat	<ul> <li>National Light Pollution Guidelines for Wildlife (DCCEEW, 2023c)</li> </ul>
advice and Australian marine park	Recovery Plan for Marine Turtles in Australia 2017–2027 (CoA, 2017b)
zoning objectives?	Wildlife Conservation Plan for Seabirds (CoA, 2020).
	For all the plans identified above, the objectives are achieved by adopting EPO- 08 and control measures outlined in Table 6-13. Santos considers the impacts of light emissions as not inconsistent with these recovery plans.
	Recovery plans / conservation advice for other species that may occur in the light assessment boundary do not identify light emissions as a key threat or have explicit relevant objectives or management actions related to light emissions.
	The light assessment boundary does not overlap any AMP or protected place.
	The objectives and actions of these publications were considered during impact and risk assessments. The controls outlined in Table 6-13 are not inconsistent with the objectives of the material listed above and Santos considers the impacts of light emissions to be not inconsistent with these objectives.
Are performance outcomes, control measures and associated performance standards consistent with legal and	Yes – management measures are consistent with SOLAS and the <i>Navigation Act</i> 2012 (Cth). Through acceptance of this GEP Coastal Waters OEMP, legislative and regulatory requirements will be met as per Appendix B.
regulatory requirements?	The GEP Coastal Waters OEMP is also compliant with the conditions of EPBC approval EPBC 2022/09372.
Are performance outcomes, control measures and associated performance standards consistent with Santos' Environment, Health and Safety Policy?	Yes – aligns with Santos' Environment, Health and Safety Policy (Appendix A)
Are performance outcomes, control measures and associated performance standards consistent with industry standards?	Yes – the most recent and comparable EPs accepted by NOPSEMA were reviewed for consistency with the performance outcomes, control measures and associated performance standards proposed in this GEP Coastal Waters OEMP.
Have performance outcomes, control measures and associated performance standards taken into consideration Relevant Person feedback?	Yes – Relevant Person feedback received during consultation for this Activity has been considered when evaluating EPO, CMs or EPSs. Where relevant, CMs implemented based on Relevant Persons feedback for other Barossa EPs have been adopted in this GEP Coastal Waters OEMP.
	Santos has adopted control measures which were informed by Dr Corrigan's recommendations and the suggestions of a number of senior and authoritative Tiwi Islanders about culturally appropriate responses (e.g. EPO-19 and BAO-CM-6.1.4.
Are performance standards such that the impact or risk is considered to be ALARP?	Yes – see ALARP assessment conducted, with no additional control measures adopted.

Lighting on activity vessels is industry standard and is required to meet relevant maritime and safety regulations. The potential consequences of the anthropogenic light sources in the OA are considered to be negligible and restricted to short-term behavioural impacts on individual fauna that may be present in the OA during the Activity. The 20 km light assessment boundary intersects the internesting BIA and habitat critical for the flatback turtle, and the habitat critical to the survival of olive ridley. However, due to the OA water depths (greater than 50 m), the flatback turtle BIA extending across more than 800 km of coastline and a lack of foraging habitat for the species, the potential numbers of affected internesting and nesting turtles are expected be limited to a small number of individuals. Potential behavioural impacts to turtles are predicted within 3.3 km of the Activity vessels, and 4.5 km cumulative impact (multiple vessels working together). Therefore, no nesting habitat will be impacted (closest turtle nesting beach is approximately 31.5 km distant) or flatback hatchling behaviours affected. Light emissions from the activity vessels are unlikely to attract and/or affect the behaviour of large numbers of seabirds, and the impact of lighting from Activity vessels on seabirds is considered minor. The potential consequence of light emissions on receptors is assessed as II – Minor. With the control measures in place, including compliance with navigational safety legislation, no significant impacts are expected. Therefore, the impacts of light emissions from Activity vessels are reduced to ALARP and are considered acceptable.



### 6.3 Atmospheric and greenhouse gas emissions

#### 6.3.1 Description of event

**Event** Santos has adopted a 'whole of project', conservative approach to the assessment of atmospheric and GHG emissions. There is some overlap between the quantification of, and assessment of impacts and risks associated with, atmospheric and GHG emissions across the Barossa Production Operations Environment Plan, the Barossa GEP Coastal Waters OEMP, and the Barossa GEP Internal Waters OEMP. The Barossa Production Operations EP assesses impacts and risks of atmospheric and GHG emissions from the FPSO production operations, including direct GHG (Scope 1) and indirect GHG (Scope 3) emissions. Scope 1 GHG emissions are emissions released into the atmosphere as a direct result of the activities at a facility. Scope 1 GHG emissions from the Barossa production operations (inclusive of production from wells, the FPSO, Barossa GEP through to the upstream weld of the DLNG beach valve) are considered direct emissions and are derived from: fuel combustion - hydrocarbon-based fuels (primarily gas, with diesel used intermittently) are combusted to generate heat and power flaring - a vital safety feature in which hydrocarbons are combusted intermittently (in emergency or planned shutdown or maintenance circumstances) to prevent overpressure and/ or the creation of an explosive atmosphere. Flaring is also expected during commissioning and start up (well clean up and multi-rate testing) venting - reservoir CO<sub>2</sub> extracted from the gas is vented via a thermal oxidiser (primary) or acid gas flare (back-up) fugitive emissions - may occur from pressurised equipment, emitted by infrequent operational activities, or unplanned equipment leaks. The GHG Protocol (World Resources Institute) defines indirect GHG emissions as emissions that are a consequence of the activities of the Activity but occur at sources owned or controlled by another entity. Indirect emissions from the Barossa production operations include those from onshore processing of gas at the Darwin Liquefied Natural Gas (DLNG) facility, vessel activities within the operational areas as well as the transport and end-use of Barossa sales products (condensate and LNG). This OEMP includes Barossa GEP operations for the GEP section located within Coastal waters and assesses the associated fugitives (Scope 1 emissions) from pipeline transmission and the Scope 3 (indirect) emissions resulting from vessel activities along the GEP. The impacts and risks of atmospheric and GHG emissions associated with operating the Coastal Waters section of the Barossa GEP are addressed below, noting there is some overlap between the atmospheric and GHG emissions assessed for the purposes of the Barossa Production Operations EP. Atmospheric emissions Air polluting emissions, such as sulphur oxides (SO<sub>X</sub>), nitrogen oxides (NO<sub>X</sub>) and volatile organic compounds (VOCs) are discharged to the atmosphere during operations, contributing to a localised reduction in air quality. Atmospheric emissions may occur from: GEP fugitive emissions associated with gas transmission combustion emissions from vessel and helicopter activity within the Operational Area operation of vessel incinerators. Activity vessels may use ozone-depleting substances (ODS), but in a closed rechargeable refrigeration system-there is no plan to release ODS to the atmosphere. Greenhouse gas emissions GHG emissions are gases that trap heat within the atmosphere through the absorption of longwave radiation reflected from the earth's surface. The emissions of CO<sub>2</sub>, nitrous oxide (N<sub>2</sub>O), CH<sub>4</sub>, sulphur hexafluoride, hydrofluorocarbons and perfluorocarbons are recognised as GHG emissions. All GHG emissions will add to the atmospheric GHG load which is linked to global warming and climate change. GHG emissions generated from the Activity include direct (Scope 1) and indirect (Scope 3) emissions. Scope 1 emissions include direct emissions from sources that Santos owns or controls. Scope 1 emissions sources associated with the Activity are fugitive emissions from pipeline transmission of dry gas, to the extent those fugitive emissions could occur in the part of the GEP within NT Coastal Waters. Scope 3 emissions include indirect emissions from IMMR vessel activities. The majority of Scope 3 emissions from activities associated with the Barossa Gas Project are emissions from the use of sold products, which is beyond the scope of this document and is addressed in the Barossa Production Operations EP.

The are no indirect (Scope 2) emissions associated with the Activity. Scope 2 emissions are indirect emissions from the generation of energy that Santos purchases for its operations including electricity purchased for ancillary activities such as office buildings.



Extent	The quantities of atmospheric emissions under normal operating conditions will quickly dissipate into the surrounding atmosphere of an open ocean environment. Direct and indirect GHG emissions will be generated within the OA. Indirect GHG emissions will also be generated outside the OA (as described above).
Duration	Generation of direct and indirect GHG emissions and atmospheric emissions will occur over the life of production operations (approximately 25 years). Planned IMMR vessel presence occurs typically for approximately 7 to 30 days in duration ranging every
	three to five years across the life of the Activity. If crew change via helicopter is needed, it will typically happen once per week during IMMR activities.

#### 6.3.2 Nature and scale of environmental impacts

**Potential receptors**: physical environment (air quality); threatened, migratory or local fauna (seabirds); socioeconomic receptors; and cultural features.

The potential impacts from air emissions identified above include:

- Deterioration of local air quality
- Contribution to national and global GHG emissions.

The OA is in an offshore environment where there are no other permanent sources of air pollution.

Hydrocarbon combustion of non-GHG emissions (NOx, SOx and VOCs) may result in a temporary, localised reduction of air quality in the environment immediately surrounding the discharge point. A reduction in local air quality could affect threatened, migratory, or local fauna (seabirds), and the workforce. Atmospheric emissions may be harmful, odoriferous, or aesthetically unpleasing. Interactions with sensitive environmental receptors linked to air pollutants is not expected considering the OA is located offshore and not in the vicinity of populated regions.

VOCs can be harmful to human health and also to the environment, as they can be toxic; however, this is generally relevant to high concentrations of VOCs in closed environments. VOCs are not expected to be in large enough volumes to be harmful and will rapidly disperse, reducing their impacts.

ODSs are used in closed refrigeration systems. ODSs have the potential to contribute to ozone-layer depletion if accidentally released to the atmosphere. ODS air emissions would only occur in the event of damaged or faulty refrigeration equipment, or due to human error.

Possible effects are localised variations in air quality that are restricted to the campaign vessel/s location. It is anticipated that the effects of air emissions will disperse well in advance of the closest populated area (the nearest being the Tiwi Islands (approximately 30 km north from the OA) and will have negligible direct or cumulative effects on environmental receptors or exceed National Environmental Protection (Ambient Air Quality) measures (NEPM) standards. As the Activity will occur in open-ocean offshore waters, the combustion of fuels and incineration in such remote locations will not impact on air quality in coastal towns.

#### Greenhouse gas emissions estimates

To quantify potential GHG emissions, the metric CO<sub>2</sub>-e is used to standardise the various GHGs, such as CO<sub>2</sub>, CH<sub>4</sub>, N<sub>2</sub>O. This involves converting the quantities of emitted gases into an equivalent amount of CO<sub>2</sub> that would have the same global warming potential.

The calculation methodology models GHG emissions based on Activity input data and industry standard data. The methods used in this modelling align with the relevant Australian and international legislation, regulations, standards and guidelines, being:

- National Greenhouse and Energy Reporting (NGER) Act 2007 and associated (Measurement) Determination 2008, and SGM Rule 2015, and has substantially adopted Method 1
- International Organisation for Standardisation (ISO) 14064 Greenhouse gases Part 1: Specification with guidance at the organisation level for quantification and reporting of greenhouse gas emissions and removals
- ISO 14040: 2006 Environmental management Life Cycle Assessment Principles and Framework.

Under the NGER regime, emissions are described as either Scope 1, 2 or 3, which relate to where the emissions occur (Clean Energy Regulator (CER), 2024):

• Scope 1 (direct) GHG emissions are the emissions released to the atmosphere as a direct result of an activity, or series of activities, at a facility level.



- Scope 2 GHG emissions are the emissions released to the atmosphere from the indirect consumption of an energy commodity. For example, 'indirect emissions' come from the use of electricity produced in another facility
- Scope 3 GHG emissions are indirect emissions (other than Scope 2 emissions) that are generated in the wider economy. They occur as a consequence of the activities of a facility, but from sources not owned or controlled by the operator of the facility.

During the operations phase authorised under this GEP Coastal Waters OEMP, the Barossa Joint Venture controls the export of the gas to the DLNG facility via the portion of the GEP covered under this GEP Coastal Waters OEMP.

Emissions from onshore processing is separate to the Activity and is not controlled by the Barossa JV. The DLNG joint venture is responsible for processing and liquefying the gas at the DLNG facility. This activity is not regulated under the OPGGS Act and associated Regulations, and is not part of the Activity authorised by this GEP Coastal Waters OEMP.

For the purposes of this OEMP, Scope 1 emissions from the Barossa GEP as a whole (385 km of pipeline from the FPSO to DLNG) have been estimated in accordance with approved methods in the NGER Measurement Determination. The Scope 1 emissions of the GEP will consist entirely of fugitive emissions associated with transmission of gas in the GEP from the FPSO to DLNG. Fugitive emissions are, by their nature, difficult to quantify and are estimated by application of methods from the NGER Measurement Determination. As much of the safe operation of the pipeline relies on the effective containment of hydrocarbons, the volume of fugitive emissions are expected to be negligible in comparison to GHG emissions from other sources for the Barossa Gas Project.

The section of pipeline which is the subject of this document (NT Coastal Waters) is 8.26 km and therefore comprises ~2% of the total 385km GEP from the FPSO to DLNG. Accordingly, the associated Scope 1 emissions of the Activity represent only a small fraction of total Scope 1 emissions associated with the GEP. Further, the section of GEP covered by this Activity does not have any valves thereby in applying a proportional allocation of emissions to this 8.26km pipeline segment is conservative.

Table 6-14 provides an estimate of the Scope 1 (direct emissions) for the 25 year lifecycle of Barossa GEP steady state operations for the 385 km of pipeline from the FPSO to DLNG.

## Table 6-14: Estimated Direct (Scope 1) emissions estimate for the 25 year lifecycle of Barossa GEP operations

Operations	Scope 1 emissions (MtCO <sub>2</sub> -e)
Steady State Operations and maintenance (total)	
GEP fugitive emissions for the 385km pipeline from FPSO to DLNG (total)	0.11

Australian and International carbon accounting rules mean each country and each emitter is responsible for reporting their own Scope 1 and Scope 2 emissions. The NGER Act does not require reporting of indirect (Scope 3) emissions. Notwithstanding, in order to support Santos' evaluation of potential risks and impacts of the Activity, an estimate of the indirect (Scope 3) emissions is provided in Table 6-15 for the 25 year lifecycle of Barossa GEP steady state operations, for the entirety of the 385km pipeline from the FPSO to DLNG. As for Scope 1 emissions, only a fraction of these GEP emissions are attributable to the Activity, which represents an 8.26 km section (~2%) of a 385 km pipeline.

#### Table 6-15: Indirect (Scope 3) emissions estimates for the 25 year lifecycle of Barossa GEP operations

Subcategory	Scope 3 Emissions (MtCO <sub>2</sub> -e)
IMMR vessel mobilisation and transit for the total length of the Barossa GEP*	0.01

\* Excludes emissions from business travel due to insignificant contribution to total indirect (scope 3) emissions for the Activity. It is noted that the Barossa Production Operations EP accounts for business travel as part of the Scope 3 emissions of the FPSO, which is expected to cover all business travel associated with Barossa operations.

Over the life of production operations, the 385km GEP from the FPSO to DLNG is estimated to be associated with approximately 104,963 t  $CO_{2-e}$  of direct emissions and 10,272 t  $CO_{2-e}$  of indirect emissions. An overview of the direct and indirect emissions for the entirety of the GEP is provided in Table 6-16. All estimates are sensitive to



production rate, which is subject to uncertainty associated with the development of the Barossa Gas Project and will change over the life of the GEP.

Barossa Production Operations	Lifecycle Emissions (MtCO <sub>2</sub> -e)
Direct - Scope 1	0.11
Indirect - Scope 3	0.01
Total	0.12

#### 6.3.2.1 Emissions comparisons

Climate change is a global issue being managed by the international community of states under the United Nations Framework Convention on Climate Change (UNFCCC). Australia is a signatory to the UNFCCC and the Paris Agreement. Under the Paris Agreement, Australia must submit emissions reduction commitments known as Nationally Determined Contributions (NDCs). These-commitments constitute Australia's contribution to global climate efforts and to meeting the temperature objectives of the Paris Agreement (to hold the increase in the global average temperature to well below 2°C above pre-industrial levels and pursuing efforts to limit temperature increase to 1.5°C). Australia has legislated various measures to ensure these commitments are met. This legislation includes the Safeguard Mechanism which will require the Barossa Gas Project to be net-zero reservoir emissions from day one and to have an emissions baseline and reduction trajectory set by Australia's CER.

The international framework has been developed to facilitate an orderly approach to what is a global problem. The nature, quantity and timeframe of each country's contribution and the pathways to achieve UNFCCC obligations vary widely, including having regard to the particular circumstances of each country. This framework recognises, and is premised on, the fact that climate change is a global issue – there is no correlation between where GHG emissions are released and where climate change impacts are felt. For this reason, Australia sets and reports against its emissions reduction targets in 'net' terms, not by individual sectors or projects. The Australian Government is aware of planned production in the Barossa gas field. It has granted property rights for the production of gas from the Barossa field through the award of a production licence and has acknowledged the production of gas from the Barossa field in its latest emissions projections.

In this context, and at NOPSEMA's request, Santos has sought to contextualise the contribution of emissions from the Activity in this GEP Coastal Waters OEMP against Australian and global carbon budgets.

The contribution of estimated annual average CO<sub>2</sub>-e emissions from the GEP to carbon budgets nationally and globally is presented in Table 6-17.

#### Australian Carbon Budget

As reflected in Australia's Nationally Determined Contributions (NDC) under the Paris Agreement and the Climate Change Act 2022, Australia is committed to a single year target to reduce greenhouse gas emissions to 43% below 2005 levels by 2030 and net-zero by 2050 – see Section 6.3.2.3.2 for description of Australia's greenhouse gas emissions framework.

Based on a 43% reduction by 2030, the net carbon budget for this period is 4,377 Mt CO<sub>2</sub>-e (DCCEEW, 2024q). Assuming a further linear decline between 2030 and 2050 (the full activity lifecycle for the Barossa Facilities – as described in the Barossa Production Operations EP and this Coastal Waters OEMP), this creates a net carbon budget of 7262 Mt CO<sub>2</sub>-e. The net carbon budget comprises gross economy wide emissions (additions) less total carbon sequestration volumes (subtractions).

#### Global Carbon Budget

The United Nations Intergovernmental Panel on Climate Change (IPCC) in its Sixth Assessment Report forecast the remaining net carbon budgets (from 1 January 2020) for a 50% likelihood to limit global warming to a specified range of temperature increase based on pre-industrialised levels (i.e. since 1850-1900) (IPCC, 2021). Section 6.3.2.3.1 describes the international framework for management of greenhouse gas emissions.

Global Surface Temperature Change	Estimated carbon budgets (50 <sup>th</sup> percentile) MtCO <sub>2</sub>
1.5°C	500,000
2.0° C	1,350,000



#### Table 6-17: Barossa GEP operations greenhouse gas emissions in context

Stage	Lifecycle emissions (MtCO <sub>2</sub> e)	ecycle GEP Operations Contribution (%)		(%)
		Australian carbon budget (Mt CO₂e) <sup>[1]</sup>	Global carbon budget - 1.5° C (MtCO <sub>2</sub> e) <sup>[1]</sup>	Global carbon budget - 2.0° C (MtCO <sub>2</sub> e) <sup>[1]</sup>
Scope 1				
Fugitive emissions from pipeline transmission	0.11	0.0015%	0.00002%	0.00008%
Scope 3				
IMMR vessel activities	0.01	0.0001%	0.000002%	0.000007%
Totals	0.12	0.0017%	0.00002%	0.00009%

#### [1] Out to 2050

Net GHG emissions (within Australia) from the 385km GEP from the FPSO to DLNG represent 0.0017% of Australia's net carbon budget to 2050.

Net GHG emissions (within Australia) from the 385km GEP from the FPSO to DLNG represent 0.00002% and 0.000009% respectively of net global carbon budgets under 1.5°C and 2°C temperature increase scenarios.

At 0.0017% of Australia's carbon budget to 2050, and 0.00002% and 0.00009% of net global carbon budgets for 1.5°C and 2°C temperature increases respectively, the net emissions from the 385km GEP from the FPSO to DLNG comprise a nominal amount in the overall scheme of the national and international carbon budgets and will not materially or substantially contribute to Australia's net GHG emissions or net global emissions levels. Emissions attributable to the Activity, representing ~2% of the total length of the GEP, are a fractional component of these contributions. In any event, net carbon budgets are made up of both additions and subtractions to cumulative GHG emissions. Accordingly, these are theoretical calculations for indicative purposes only.

#### 6.3.2.2 Risks of climate change to the Australian environment

The IPCC (2021) have reaffirmed the near-linear relationship between cumulative anthropogenic  $CO_2$  emissions and global warming, highlighting the basis for setting net-zero emissions targets. It is known that the physical impacts of climate change on environmental receptors are the result of global GHG emissions from a multitude of sources (minus the GHG sinks) that have accumulated in the atmosphere (IPCC, 2018).

The impacts on the climate cannot be attributed to one specific sector or activity. In the context of evaluating potential impacts and risks that may be associated with GHG emissions from all sources globally, including from this Activity, this GEP Coastal Waters OEMP has considered broader climate change issues. This section outlines the potential environmental impacts that could occur due to global climate change. Santos, in its capacity as an independent company, recognises the scientific consensus on climate change assessed by the IPCC.

The IPCC is the United Nations body for assessing the science related to climate change and finalised the Sixth Assessment Report (AR6) in 2023. This consists of three Working Group contributions and a Synthesis Report. A summary of outcomes of the working group's contributions comprises a range of matters, which amongst others include:

- The AR6 Working Group I (AR6-WG1) report stated that it is unequivocal that there is human-induced warming. It also stated that increased atmospheric carbon dioxide (CO<sub>2</sub>) levels, generated by human activity, are the largest driver of warming over the longer term, and that there are a range of factors, including emissions of methane, which increase warming in the short-term.
- The AR6-Working Group II (AR6-WG2) report stated that human-induced climate change, including more frequent and intense extreme events, has caused widespread adverse impacts and related losses and damages to nature and people, beyond natural climate variability. It stated that global warming, reaching 1.5°C in the near-term, would cause unavoidable increases in multiple climate hazards and present multiple risks to ecosystems and humans. The report noted that societal choices and actions implemented in the next decade will determine the extent to which medium- and long-term pathways will deliver climate resilient development.
- The AR6 Working Group III (AR6-WG3) report provided an updated global assessment of climate change
  mitigation progress and pledges and examined the sources of global emissions. It explained developments
  in emissions reduction and mitigation efforts and assessed the impact of national climate pledges in
  relation to long-term emissions goals. More than 2000 quantitative emissions pathways were submitted to
  the IPCC, of which 1202 scenarios included sufficient information for assessing the associated warming.
  The report found that there are many pathways in the literature that likely limit global warming to 2°C with



no overshoot, or to 1.5°C with limited overshoot. These variations occur because, while climate science is able to calculate a 'carbon budget' of net emissions before any particular temperature outcome is reached, the allocation of this budget between different human activities requires additional judgements about for example technology, economics, consumer preferences and policy choices.

- The State of the Climate 2024 (Commonwealth of Australia, 2024) states that Australia is projected to experience the following in the coming decades:
  - continued warming, with more extremely hot days and fewer extremely cool days.
  - a further decrease in cool season rainfall across many regions of the south and east.
  - continued drying in the south-west of Western Australia, especially during winter and spring.
  - likely increases in the average duration of drought and aridity in regions within the south and east.
  - a longer fire season for the south and east, and an increase in the number of dangerous fire weather days
  - more intense short-duration heavy rainfall events, even in regions where the average rainfall decreases or stays the same.
  - fewer tropical cyclones, but a greater proportion projected to be of high intensity, with ongoing large variations from year to year. The intensity of rainfall associated with tropical cyclones is also expected to increase and, combined with higher sea levels, is likely to amplify the impacts from those tropical cyclones that do occur.
  - fewer east coast lows on average, particularly during the cooler months of the year
  - ongoing sea level rise through this century and beyond, at a rate that varies by region. Recent research on
    potential ice loss from the Antarctic ice sheet suggests that a scenario of larger and more rapid sea level
    rise can't be ruled out.
  - more frequent extreme sea levels linked to coastal inundation and coastal erosion. For most of the Australian coast, extreme sea levels that had a probability of occurring once in a hundred years are projected to become an annual event by the end of this century with lower emissions, and by the mid-21<sup>st</sup> century for higher emissions.
  - continued warming and acidification of surrounding oceans with consequent impacts on biodiversity and ecosystems.
  - increased and longer-lasting marine heatwaves, which will further stress marine environments, such as kelp forests, and increase the likelihood of more frequent and severe bleaching events in coral reefs around Australia, including the Great Barrier Reef and Ningaloo Reef.
  - an increase in the risk of natural disasters from extreme weather, including 'compound extremes', where multiple extreme events occur together or in sequence, thus compounding their impacts.
- The report also provides the following updated projections of Australia's average temperature over the next two decades:
  - the average temperature of each future year is now expected to be warmer than any year prior to the commencement of human-caused climate change.
  - ongoing climate variability means each year will not necessarily be hotter than the last, but the underlying probabilities are changing. This leads to less chance of cool years and a greater chance of repeatedly breaking Australia's record annual average temperature (e.g. record set in 2005 was subsequently broken in 2013 and then again in 2019).
  - while the previous decade was warmer than any other decade in the 20th century, it is likely to be the coolest decade for the 21st century.
  - the average temperature of the next 20 years is virtually certain to be warmer than the average of the past 20 years
  - the amount of climate change expected in the next decade is similar under all plausible global emissions scenarios. However, by the mid-21st century, higher ongoing emissions of greenhouse gases will lead to greater warming and associated impacts, while lower emissions will lead to less warming and fewer impacts
  - warming is generally expected to be greater in the interior of Australia than near the coast.

Ecosystems are particularly susceptible to adverse effects of climate change. The 'loss of climatic habitat caused by anthropogenic emissions of greenhouse gases' has been listed as a key threatening process under the EPBC

Act (DCCEEW, 2021), consisting of reductions in the bioclimatic range within which a given species or ecological community exists due to human induced emissions (DCCEEW, 2021). The process is considered to have a continental distribution, including both terrestrial and marine areas. Ecosystems in which the process occurs include: alpine habitats, coral reefs, wetlands and coastal ecosystems, polar communities, tropical forests, temperate forests, and arid and semi-arid environments (DCCEEW, 2021).

Redistribution and reorganisation of natural systems, driven by climate change, is a major threat to biodiversity (Chapman *et al.*, 2020). A report by Australia's Biodiversity and Climate Change Advisory Group summarises the potential impacts of climate change to marine and terrestrial species, habitats and ecosystems across Australia (Steffen *et al.*, 2009).

Extensive modelling and monitoring studies over the last 20 years provide considerable evidence that global climate change is already affecting and will continue to affect species (Hoegh-Guldberg *et al.*, 2018). However, these impacts are likely to be highly species-dependent and spatially variable. Climate change may not only change species distribution patterns but also life-history traits, such as migration patterns, reproductive seasonality and sex ratios.

Increases in fire regimes will impact Australian ecosystems, altering composition structure, habitat heterogeneity and ecosystem processes. Changes in climate variability and averages could also be important drivers of altered species interactions, both native and invasive species (Dunlop *et al.*, 2012). Climate change could result in significant ecosystem shifts, as well as alterations to species ranges and abundances within those ecosystems (Hoegh-Guldberg *et al.*, 2018).

Climate variability and change has been identified as a threat to some EPBC Act protected species in relevant conservation management plans and recovery plans, including marine turtles mammals, sharks and seabirds and migratory shorebirds as per Table 3-13.

The North-west Marine Parks Network Management Plan 2018 (DNP, 2018a) identifies climate change as a pressure that may impact marine park values. The management plan states that "the impacts of climate change on the marine environment are complex and may include changes in sea temperature, sea level, ocean acidification, sea currents, increased storm frequency and intensity, species range extensions or local extinctions, all of which have the potential to impact on marine park values" (DNP, 2018a).

Within the Marine Bioregional Plan for the North-West Marine Region (NWMR) (DSEWPaC, 2012f), pressures related to climate change are assessed as 'of potential concern' for species of marine turtle, inshore dolphins, sawfish, sea snakes, whale shark, dugong, and seabird and shorebird, as well as the KEFs and shipwrecks known to occur in the NWMR.

Changes to climate can also result in impact to social receptors that have values which include the ecological receptors described above, including KEFs and Australian Marine Parks (AMPs). Climate change may also impact on the functions, interests or activities of other users which rely on these ecological values, including commercial and recreational fisheries and tourism. A temperature change of between 0.9°C to 2.0°C is forecast to reduce fisheries yield as the maximum catch potential around Australia by between 3% and 10% (IPCC 2023).

Impacts to cultural heritage sites and places of spiritual importance in coastal locations may also be experienced due to rising sea levels. Global mean sea level increased by 0.20 [0.15 to 0.25] m over the period 1901 to 2018, with a rate of rise that has accelerated since the 1960s to 3.7 [3.2 to 4.2] mm yr<sup>-1</sup> for the period 2006 –2018. Human activities were very likely the main driver of observed global mean sea level rise since 1971 (IPCC, 2021).

#### 6.3.2.3 Climate Change Management Frameworks

#### 6.3.2.3.1 International greenhouse gas emissions framework

#### Paris Agreement (United Nations Framework Convention on Climate Change, 2021)

The United Nations Framework Convention on Climate Change (UNFCCC) came into force in 1994 and comprise of 198 parties. The convention established a goal of preventing dangerous anthropogenic interference with the climate system. Subordinate treaties and agreements have been ratified by parties to the convention, including the Paris Agreement, which was agreed under the convention at COP21 in 2015 and has been endorsed by 195 parties. The Paris Agreement is currently the world's most comprehensive climate action agreement underpinned by broad international support.

One of the key aspects of the Paris Agreement (the agreement) is Article 2 which, in seeking to strengthen the global response to climate change, reaffirms the goal of limiting global temperature increase to well below 2°C, while pursuing efforts to limit the increase to 1.5°C. This was reaffirmed in December 2023 in the COP28 decision (UNFCCC 2023).

The Paris Agreement also calls on Parties to contribute to global efforts, in a nationally determined manner, taking into account the objectives of the Paris Agreement and their different national circumstances, pathways and approaches, to transition away from fossil fuels in energy systems is to be done in a just, orderly and equitable

manner, accelerating action in this critical decade, so as to achieve net-zero by 2050 in keeping with the science (UNFCCC 2023). The text also recognises that transitional fuels can play a role in facilitating the energy transition while ensuring energy security (UNFCCC 2023).

Australia is a signatory to the agreement. In support of meeting the aims of the agreement, the Australian Government has legislated a target of reducing emissions to 43% below 2005 levels by 2030 and committing to net zero emissions by 2050 (refer to 'Australia's Nationally Determined Contributions' below).

The Paris Agreement requires all parties to put forward their best efforts through NDCs to reduce GHG emissions and to strengthen these efforts in the years ahead. The NDCs represent national action for each country individually. As such, countries will choose to implement their NDCs in a variety of ways, consistent with their domestic policies and strategies. Countries are required to transparently and regularly report their climate actions and support, including whether they have met or are on schedule to meet the goals per their NDCs. As at February 2025, there are 195 parties to the Paris Agreement that have put forward NDCs.

The participating Paris Agreement parties aim to reach global peaking of GHG emissions as soon as possible to achieve a climate-neutral world by 2050, recognising developing country parties' peaking emissions may occur later than developed countries. After the peak in GHG emissions, it is expected there will be rapid reductions in accordance with best available science, to achieve a balance between anthropogenic emissions by sources and removals by sinks of GHG in the second half of this century, on the basis of equity and in the context of sustainable development and efforts to eradicate poverty (UNFCCC, 2021).

The convention recognises that to achieve the Paris Agreement's long-term goals, climate action will need to get more ambitious over time. To sustain this rising ambition, the agreement establishes a continuous improvement cycle through which countries plan and communicate their NDCs, then implement their plans, and finally review individual and collective progress to inform future planning and updates to their next NDCs. This process provides the foundation for countries to fully bring the objectives of the Paris Agreement to fruition (World Resources Institute, 2021).

Effective interaction between climate science and policy underpins the Paris Agreement. Scientific observations, research and assessment continue to inform the international climate regime, as well as national and regional climate policies. The United Nations climate change process, under the Paris Agreement, relies on scientific information about climate change.

This continuous improvement cycle supports the agreement's commitment to comprehensively take stock of collective progress every five years (global stocktake – Article 14 of the Paris Agreement), a key element of the process that is sometimes referred to as the agreement's 'ambition mechanism'. The global stocktake process assesses the collective progress towards achieving the purpose of the agreement and its long-term goals, evaluating both the performance of countries in meeting their NDCs and contemporary climate and environmental scientific literature. The Intergovernmental Panel on Climate Change assesses the scientific, technical and socioeconomic information relevant for understanding the risk of human-induced climate change and prepares comprehensive assessment reports and special reports to support the global stocktake process.

The Paris Agreement is underpinned by the international environmental legal principle of common but differentiated responsibilities. The principle holds that all states are responsible for addressing global environmental degradation yet are not equally responsible. On the one hand, the principle balances the need for all countries to take responsibility for global environmental problems and, on the other hand, the need to recognise the wide differences in levels of economic development between countries. Australia, for example, has a more ambitious target than developing countries because of this principle.

The enhanced transparency framework established within the Paris Agreement (Article 13) requires that, starting in 2024, countries report transparently on actions taken and progress in climate change mitigation, adaptation measures and support provided or received. It also provides international procedures for reviewing submitted performance reports and contemporary climate and environmental scientific literature. The information gathered through the enhanced transparency framework is intended to then feed into five-yearly global stocktakes and review and updates to NDCs.

To facilitate implementation of the Paris Agreement, the Katowice climate change package (UNFCCC, 2018) sets out the essential procedures and mechanisms that bring the Paris Agreement into operation and contains operational guidance on:

- the information about domestic mitigation and other climate goals and activities that governments will provide in their NDCs
- how to communicate about efforts to adapt to climate impacts
- the rules for functioning of the transparency framework for action and support (referred to in Article 13 of the agreement), which will show what countries are doing about climate change



- the need to establish a committee to facilitate implementation of the Paris Agreement and promote compliance with the obligations undertaken under the agreement
- how to conduct the global stocktake (the first stocktake was in 2023) of overall progress towards the aims
  of the Paris Agreement
- how to assess progress on the development and transfer of technology
- how to provide advance information about financial support to developing countries and the process for establishing new targets on finance from 2025 onwards.

The Katowice package provides countries with detailed guidance for performing the continuous improvement cycle of the agreement, guidance on how to prepare their NDCs (clear and transparent information about how GHG emissions are calculated and timeframes for contributions commitments), and what types of information participating countries should share concerning adaptation priorities, plans and actions.

To inform further planning for meeting the global Paris Agreement targets, countries must review their efforts, individually and collectively. The review of individual countries' progress will aim to verify data quality and assess progress against each country's targets, while the global stocktake review will assess the collective progress toward the agreement's long-term goals and identify the remaining gaps, challenges and opportunities for further action. The agreement has also set up an expert committee focused on facilitating implementation and promoting compliance to help countries address barriers to implementation and further climate action.

The countries to which Barossa LNG and condensate will be exported are anticipated to report their associated GHG emissions from processing, refining and use of the Barossa LNG and condensate as their Scope 1 and 2 GHG emissions, within the context of their own NDCs and associated emissions reduction policies and regulations, as parties to the Paris Agreement (or having their own mid-century net-zero commitments). These emissions are described and accounted for within the indirect emissions estimates in the Barossa Production Operations EP.

## Australia's nationally determined contributions (United Nations Framework Convention on Climate Change, 2022)

Australia has ratified the Paris Agreement and has adopted NDCs that can be monitored and reported on as part of the five-year stocktake. At the Paris conference in 2016, Australia announced its first NDC to reduce GHG emissions to 26 to 28% below 2005 levels by 2030. Further commitments were made by the then elected government in 2021 to reach Net Zero Emissions (NZE) by 2050.

In May 2022, the Government announced a goal of reducing Australia's GHG emissions by 43% below 2005 levels by 2030 and reaffirmed Australia's commitment to NZE by 2050. This was lodged with the UNFCCC as an updated NDC as part of Australia's obligations under the Paris Agreement. The procedures around NDCs under the Paris Agreement (obligations to prepare, communicated and maintain NDCs) are legally binding though the NDCs themselves are not. Australia mainly focuses on Article 10, with a low-emissions technology-led approach. Australia's NDCs are implemented through schemes such as the Safeguard Mechanism and the Emissions Reduction Fund, in addition to continuous monitoring and focusing on alternatives to lower overall emissions. The *Climate Change Act 2022* (Cth) (section 6.3.2.3.2) was subsequently enacted to enshrine into law Australia's 2030 emissions reduction target of 43% below 2005 levels.

Under the Paris Agreement, a country must update its NDC every 5 years. Australia's updated NDC is anticipated to be submitted ahead of COP30 in September 2025, and it is expected to include an emissions reduction target to 2035. Each new target is to be more ambitious than the last, to support the global goals under the Paris Agreement.

#### 6.3.2.3.2 National greenhouse gas emissions framework

Australia has a well-established legislative framework under which certain GHG emissions from Barossa production operations will be regulated or managed to further Australia's Paris Agreement commitments. This includes:

- GHG emissions reporting under the National Greenhouse and Energy Reporting Act 2007 (NGER Act) (Cth) and the National Greenhouse and Energy Reporting (Measurement) Determination 2008
- the Emissions Reduction Fund (Australian Carbon Credit Units Scheme)
- the Safeguard Mechanism to keep net emissions below an established baseline and require net-zero
  reservoir emissions for new gas fields that feed LNG projects. The Safeguard Mechanism currently applies
  to facilities that emit more than 0.1 MtCO<sub>2</sub>-e per annum.

Key elements of the mechanism include:

• Safeguard facilities must meet the reporting and record-keeping requirements of the NGER Act, including the Clean Energy Regulator's requirements for audits prior to baseline setting and annual audits for



facilities that emit over 0.1 MtCO<sub>2</sub>-e per annum. In its capacity as the Operator of the Barossa facility, Santos will report audited Scope 1 and 2 GHG emissions from Barossa in accordance with the NGER Act.

- If a safeguard facility is likely to exceed its baseline, the responsible emitter must act, including by purchasing and/or surrendering Australian Carbon Credit Units, to offset excess emissions
- Penalties apply for non-compliance.

#### National Greenhouse and Energy Reporting Act

The NGER Act is a single national framework for reporting and disseminating company information about GHG emissions, energy production, energy consumption and other information otherwise specified under the legislation (DISER, 2022). The objectives of the NGER Act are to:

- inform government policy
- inform the Australian public
- help meet Australia's international reporting obligations
- assist Commonwealth, State and Territory government programmes and activities
- avoid duplication of similar reporting requirements in the states and territories.

Activity Scope 1 emissions will be reported under the NGER Act. There are no Scope 2 GHG emissions related to the Activity. Scope 3 emissions associated with the Activity are not required to be reported, given that these emissions constitute the Scope 1 and 2 emissions of other emitters. This includes the GHG emissions associated with processing at the DLNG facility, which are regulated in accordance with the Safeguard Mechanism and an approved Operations Environmental Management Plan (EMP). It also includes GHG emissions being regulated as Scope 1 and 2 GHG emissions under the frameworks of those countries. All consuming countries will be signatories to the Paris Agreement or have mid-century net-zero emissions commitments. Further detail about the Scope 3 emissions and measures that will be implemented in that regard are detailed in the Barossa Production Operations EP.

The Safeguard Mechanism is also administered under the NGER Act. The Clean Energy Regulator (CER) administers the NGER Act, its legislative instruments, and related policies and processes. The CER administers the scheme by:

- registering and deregistering corporations for reporting
- receiving reports
- monitoring and enforcing compliance
- applying the audit framework
- publishing reported data.

#### **Emissions Reduction Fund (Australian Carbon Credit Units Scheme)**

The purpose of the *Carbon Farming Initiative Amendment Act 2014* (Cth) was to amend the *Carbon Credits* (*Carbon Farming Initiative*) *Act 2011* (Cth) to include and establish the Emissions Reduction Fund (now referred to as the Australian Carbon Credit Units Scheme). The Emissions Reduction Fund (Australian Carbon Credit Units Scheme) is a voluntary scheme that aims to provide incentives for a range of organisations and individuals to adopt new practices and technologies to reduce their emissions. Through the Emissions Reduction Fund, participants in the Australian Carbon Credit Units Scheme can earn Australian Carbon Credit Units (ACCUs) for every tonne of carbon dioxide equivalent they store or avoid emitting. ACCUs can be sold and can generate an income for participants. A number of activities are eligible under the ACCU scheme.

#### Safeguard Mechanism

One of the key statutory instruments for regulating Australia's emissions in line with its NDCs under the Paris Agreement is the *NGER (Safeguard Mechanism) Rule 2015* (Cth) (the Safeguard Mechanism), made under the NGER Act and administered by the Clean Energy Regulator. The Safeguard Mechanism was developed to ensure Australia's largest industrial GHG emitters keep their net emissions below an emissions limit (a baseline). The Safeguard Mechanism currently applies to facilities that emit more than 0.1 Mt CO<sub>2</sub>-e per annum and requires annual emissions to be reported against a designated emissions 'baseline'.

The Safeguard Mechanism is one element of a whole-of-economy approach, implemented to achieve Australia's NDCs, and is complementary to a range of programs that measure, manage, reduce or offset Australia's GHG emissions.

Emissions from the Activity would not, absent the Barossa FPSO, be regulated under the Safeguard Mechanism, given the annual Scope 1 emissions associated with the GEP do not exceed the 0.1 MtCO<sub>2</sub>-e per annum threshold. The GEP, nor that component of the GEP which comprises the Activity, will not be a 'facility' to which the Safeguard Mechanism applies. However, as stated above, the Scope 1 emissions of the GEP are considered to be the Scope 1 emissions of the Barossa Gas Project as a whole, which considers the Barossa FPSO and GEP to be a single facility for the purposes of the Safeguard Mechanism. Therefore, the Safeguard Mechanism will apply to emissions from the Activity through the establishment of a cap (baseline) on Barossa facility emissions. Under the Safeguard Mechanism, annual emissions are reported under the NGER Scheme and compared against the facility baseline. Santos is required to generate or procure and surrender Australian Carbon Credit Units (ACCUs) or Safeguard Mechanism Credits for any emissions above the baseline for the compliance period, to ensure net emissions for the facility remain under the prescribed baseline. In 2022, the Australian government proposed Safeguard Mechanism reforms to require a greater contribution to Australia's climate targets from large industrial facilities. Under these reforms, and as a new-build facility, the Barossa facility Safeguard Mechanism baseline will be set in accordance with global best practice benchmarks by the CER. The emissions baselines for both the Barossa facility and the DLNG facility will gradually decline to limit Scope 1 emissions and achieve net zero by 2050. The decline rates have been set at 4.9% each year to 2030. After 2030, decline rates will be set in predictable five-year blocks, consistent with updates to Australia's NDC under the Paris Agreement. Decline rates for 2030-31 to 2034-35 will be set by 1 July 2027. The process for setting the future decline rates will involve consultation, and advice from the Climate Change Authority (CCA) and the latest Annual Climate Change Statements to Parliament. To assist industry planning for achieving net zero by 2050, an indicative annual decline rate has been set in the Safeguard Rules at 3.3% per annum from 2030-31 to 2049-50. The actual rate will be confirmed through the five-year baseline setting process.

Compliance with the Safeguard Mechanism is of primary importance in ensuring that GHG emissions associated with the Barossa facility, which includes the GEP, are kept to acceptable levels. GHG emissions at or below the baseline and the Safeguard Mechanism's future decline rates will assist in keeping Australia's emissions under its NDCs in accordance with the Paris Agreement.

The coming online of the Activity-has been acknowledged by the Australian Government in its latest emissions projections. For example, the Commonwealth DCCEEW in *Australia's emissions projections 2024* (November 2024) discusses emissions from LNG production and the role of the Safeguard Mechanism in reducing and controlling fugitive and on-site emissions, including the role of safeguard facilities such as the implementation of leak detection and repair measures and reduced flaring (see pages 68 and 69 of the report). The report expressly addresses the proposed resumption of gas production-at the Darwin LNG facility with gas from the 'Barossa field'.

The above report is informed by DCCEEW's *Methodology for Australia's emissions projections* (November 2024). The report identifies major new oil and gas development for which production is anticipated to start in 2025 and beyond, and the Barossa Development is expressly referenced in Table 7 (see pages 25-26) of the report.

Compliance with obligations under the Safeguard Mechanism may be achieved through (among other things) the purchase or surrender of ACCUs or Safeguard Mechanism Credits (SMCs). In 2023, Santos in its capacity as an independent company, entered into forward contracts for the purchase of 2.5 million ACCUs at fixed prices to be delivered and paid between December 2023 and January 2027. Santos is also generating some of its own emissions reduction units (including ACCUs and SMCs) at the portfolio level.

The Safeguard Mechanism baseline will be set in accordance with global best practice benchmarks by the CER, and calculated taking into account a number of production variables which may result in changing baseline requirements from one reporting period to the next.

In 2022, the ACCU scheme was independently reviewed, including the integrity of ACCUs, and recommendations for improvements were made, which are in the process of being implemented, but the scheme was otherwise found to be sound and fundamentally well-designed (*Independent Review of Carbon Credit Units: Final Report*, December 2022).

The responsible Commonwealth minister has communicated to the Barossa joint venture partners on a number of occasions that he anticipates there being ACCUs available for this project. Natural gas remains an integral part of the energy mix out to 2050 (AEMO 2024, IEA 2023c) and, with ACCUs being part of the national carbon management framework, it is reasonable to assume the availability of ACCUs for gas projects would have been contemplated in setting Australia's NDCs.-In this regard, the projected ACCU demand and issuance for the period 2025-2040 is set out in Figure 17 of *Australia's emissions projections 2024 chart data*, which was released with the DCCEEW reports released in November 2024, as referenced above. This data has been derived by reference to the emissions projection information in these reports and it is therefore reasonable to assume that the number of ACCUs anticipated by the Barossa Development have been considered with relation to Australia's anticipated ACCU issuance. Santos' reliance on ACCUs will form part of its reporting to the CER and will continue to be monitored by the CER.

The availability of ACCUs over the next decade is supported by analysis of the balance between supply, holdings, and cancellation rates (RepuTex Energy, 2024). Throughout the decade, ACCU supply is projected to steadily increase, surpassing 30 million annually by 2030, alongside rising cancellation rates to meet Safeguard Mechanism requirements. By 2027, supply is forecasted to fall into deficit due to growing annual cancellations, which will force ACCUs to be withdrawn from inventory until new supply comes into the market. Modelling shows ACCU holding inventory will continue to rise from approximately 36 million in 2024 to around 50 million by 2027, providing sufficient coverage for the deficit. Additionally, growing demand will send price signals to incentivise new ACCU supply. As noted above, Santos is already generating some of its own emissions reduction units (including ACCUs and SMCs) at the portfolio level. The Moomba CCS project completed commissioning in 2024 and is now operating and storing CO<sub>2</sub> in depleted reservoirs of the Cooper Basin. This project has been registered with the CER as an ACCU generating project and is anticipated to commence generating ACCUs within 12 months as at February 2025.

Ultimately, it is a matter for Santos to manage its compliance with its emissions baseline. The surrender of ACCUs is just one available option. The Safeguard Mechanism is administered by the CER and any failure to comply would result in the CED imposing penalties. The CER is constituted under different legislation than NOPSEMA. The CER was established by the <u>Clean Energy Regulator Act 2011</u>. It is a non-corporate Commonwealth entity under the <u>Public Governance, Performance and Accountability Act 2013</u>. It is therefore reasonable for NOPSEMA to rely on the CER's administration of the law, including the Safeguard Mechanism scheme and the ACCU scheme in relation to this EP.

Management of the emissions of the Barossa Gas Project in accordance with the Safeguard Mechanism will ensure that this Activity which is a very small component part, does not have an unacceptable impact on climate change, as the Activity emissions are being considered as part of Australia's NDC and, therefore, also into the global trajectory to limit global warming in line with the Paris Agreement targets. Sectoral pathways to achieve "net-zero" will necessarily differ and this is why this goal is expressed as a net-zero emissions target and not a "zero" emissions target. This reality has been recognised most recently in the Climate Change Authority's Electricity and Energy Sector Pathway Review released on 5 September 2024.

#### Climate Change Act 2022

The Climate Change Act 2022 (Cth) enshrines into law Australia's emissions reduction target of 43% from 2005 levels by 2030 and net-zero emissions by 2050. In addition, this Act ensures accountability through an annual update to Parliament by the Climate Change Minister on the progress made towards the target and empowers the Climate Change Authority to provide advice to government on future targets.

#### 6.3.2.3.3 Corporate greenhouse gas emissions management

The Barossa Gas Project, including the GEP, is structured as a Joint Venture (JV) and emissions management therefore needs to be considered in the context of Santos being not only a standalone corporate entity, but also a Barossa JV partner. As a titleholder and the nominated operator of Barossa, Santos is in a position to apply to the Barossa Development those aspects of its corporate policies, strategies and initiatives relating to emissions management as are appropriate having regard to its joint venture arrangements and Barossa operations.

Scope 1 GHG emissions will be regulated under the NGER regime and the Safeguard Mechanism, which accords with Australia's NDC under the Paris Agreement.

Unlike Scope 1 emissions, there is no compliance framework for Scope 3 emissions management in Australia. This is because Scope 3 emissions are the Scope 1 and Scope 2 emissions of upstream suppliers and downstream users. The international system for GHG regulation, underpinned by the UNFCCC and Paris Agreement, recognises that responsibility lies with the countries in which those emissions are generated. NDCs, including that of Australia, are consistent with this approach. Accordingly, Scope 3 emissions commitments made by companies are generally voluntary.

Santos, in its capacity as an independent company, has set commercial carbon storage and Scope 3 equivalent targets at a corporate level. Santos' commercial carbon storage and Scope 3 equivalent targets at 31 December 2024, as outlined in Santos' Annual Report 2024 including the Climate Report, are:

- **2030**: Reduce customers' emissions, Santos' Scope 3, by at least 1.5 MtCO<sub>2</sub>-e from the supply of low carbon fuels and carbon management services
- **2040**: Build and operate a commercial carbon storage business, safely and permanently storing approximately 14 Mtpa of third-party CO<sub>2</sub>-e per annum
- Long-term: Aspiration to store more carbon than we emit across Scope 1, 2, and 3 equivalent emissions

These targets are supported by a range of strategies and actions, as outlined in Santos' Annual Report 2024 including the Climate Report, and Climate Transition Action Plan. Actions implemented may change year-on-year as pathways to these targets evolve, but in 2024 included, at the corporate level:



- Directly approached approximately 180 Santos suppliers (including Barossa suppliers) and requested Scope 1 and 2 emissions data to verify our Scope 3 emissions estimates and better understand how our suppliers' emissions and emissions reduction targets support our emissions reduction efforts.
- Collaborated with an LNG vessel owner in a trial of additional cooling technology on vessels, which delivered a 16 per cent emissions reduction for the LNG vessel in the initial trial.
- Explored opportunities though engagement with our customers, such as CCS studies with multiple steel and fertiliser manufacturers, low carbon fuel joint studies with international energy consumers, and negotiations for third-party CCS solutions for LNG customers.
- Expanded customer engagement to better understand how our products are used and processed, how our products contribute to reaching customer emissions targets and how customer emissions reduction pathways support Santos' downstream Scope 3 equivalent medium and long-term targets
- Participation in the Climate Leaders Coalition 'Artificial Intelligence for Scope 3' workstream

As discussed in Section 8.5.6, each Santos Regional Business Unit (including the WA NA and TL region where Barossa is located) is required to submit a decarbonisation plan for the business unit as part of the annual corporate Long Term Planning Process. These plans outline decarbonisation opportunities to achieve regulatory compliance, including with Australia's Safeguard Mechanism, and set out pathways that each asset (including Barossa) may follow to contribute to the company-level Scope 1 and 2 emissions reduction targets. The long-term plans, including the decarbonisation opportunities, are consolidated by the corporate planning group and then reviewed and discussed with executive leadership to determine the best way to achieve compliance and targeted emissions reduction. Section 8.5.6 discusses implementation of these plans in greater detail.

The collection of information via the Barossa decarbonisation plan (which outlines emissions sources of the Activity) and through engagement with customers and suppliers will be undertaken for the purposes of corporate climate-related disclosure, under Australia's climate-related financial disclosure regime. The AASB Australian Sustainability Reporting Standard necessitates the disclosure of Scope 3 emissions data that considers the entire value chain, reflects changed circumstances, is based on a robust measurement approach and reasonable assumptions, and is verified (see Australian Sustainability Reporting Standard AASB S2, at B38-B57).

As the operator of the Barossa Development, Santos will apply its corporate targets and adopt its associated strategies and initiatives to Barossa, as appropriate having regard to its joint venture arrangements and Barossa operations. This will support the achievement of the EPOs for GHG emissions outlined in Section 6.3.3. Santos' Scope 3 equivalent targets and associated strategies and initiatives are directed to achieving reductions in Scope 3 emissions via a proactive and collaborative approach which involves engagement with both suppliers and contractors. Specific control measures have been adopted for the Activity in line with this approach (refer to BAO-CM-6.3.21 and 6.3.22, and 6.3.24). These are appropriate and practicable given that Santos does not have operational control over third party GHG emissions. Given the dynamic conditions in which the industry operates, Santos' strategies and targets will change and adapt over time.

#### 6.3.3 Environmental performance outcomes and control measures

Assuming that emissions from the Activity will cause an equivalent net increase in cumulative Australian and global GHG emissions, this increase constitutes a nominal amount in the overall scheme of the national and international carbon budgets. Santos has adopted environmental performance outcomes and control measures directed to minimising the GHG emissions from the Activity and therefore the potential contribution of these emissions to net cumulative GHG emissions globally. A range of controls have been considered for both direct (Scope 1) and indirect (Scope 3) GHG emissions in the design and for the Operations phase, as well as a system of continual review and improvement during operations

In setting the environmental performance outcomes and control measures regarding GHG emissions, it is important to recognise the global consensus of the Paris Agreement under which countries have agreed to manage and reduce their own emissions with the aim to limit the global temperature increase in this century to 2°C, while pursuing efforts to limit the increase even further to 1.5°C. Santos has developed its EPOs and control measures having regard to the UNFCCC framework which sets out the responsibility of each country to manage and reduce its emissions and the autonomy of each country in determining its pathway to achieve its emissions reduction targets.

Accordingly, the acceptable levels of GHG emissions are defined as follows:

- Acceptable levels of direct (Scope 1) emissions from the Activity are set by the Safeguard Mechanism Baseline on Barossa facility emissions, in line with the Australian Government's NDCs under the Paris Agreement.
- Scope 3 emissions associated with the Activity are limited to IMMR vessel activities and are estimated to be nominal to negligible in the overall context of national and international carbon budgets. Acceptable



levels of indirect (Scope 3) emissions from the Activity are those estimated to result from IMMR vessel activities, provided that the control measures relating to indirect emissions as outlined in this OEMP are also implemented

Indirect (Scope 3) emissions that result from the Barossa Gas Project and are emitted outside Australia, caused by the production and sale of LNG and condensate to downstream, international consumers, are beyond the scope of the Activity the subject of this document and are properly addressed in the Barossa Production Operations EP.

The EPOs, consistent with the Barossa Production Operations EP and relating to this Activity are:

- Atmospheric emissions associated with the Activity will meet all regulatory source emission standards [EPO-09]
- Manage indirect GHG emissions associated with the Activity consistent with the temperature objectives of the Paris Agreement, including by implementing company-wide targets and strategies for Scope 3 emissions reduction at the Barossa Gas Project as appropriate (having regard to joint venture arrangements and Barossa operations) and supporting customers and suppliers to reduce their GHG emissions. [EPO-10]
- Undertake the Activity in a manner that is compliant with the requirements of the Safeguard Mechanism (EPO-11)

An assessment of the environmental benefits and the potential costs or issues associated with control measures for this activity are described in Table 6-18 to demonstrate the potential impacts from this aspect are ALARP. Control measures that are adopted have associated EPSs and measurement criteria and are presented in Table 8-2. Rejected control measures have an ALARP evaluation provided to justify their rejection.

It is noted that further control measures regarding engagement with Barossa LNG customers around Scope 3 emissions management are included in the Production Operations EP. As there are no LNG customers for the purposes of this Activity, such control measures were not considered relevant for inclusion in this OEMP.

CM reference	Control measure	Environmental benefit	Potential cost/issues	Evaluation
Standard control n	neasures			
BAO-CM-6.1.2	Vessels equipped and crewed in accordance with Australian maritime requirements, including Marine Order 30 (Prevention of Collisions) and Marine Order 21 (Safety and Emergency Arrangements) (administrative control)	Ensures contracted vessels are operated, maintained, and crewed in accordance with industry standards and regulatory requirements. Ensures vessels meet Marine Assurance Standards to reduce the likelihood of vessel collision (such as minimum and working lighting for maritime safety).	No additional costs, as this is an industry standard requirement.	Adopted – benefit of assuring vessels outweighs procedure compliance costs.
BAO-CM-6.3.1	Monitoring of vessel fuel consumption and vessel speed management to reduce fuel use (administrative control)	Active monitoring of fuel consumption informs opportunities to optimize support vessel fuel use efficiencies to reduce fuel use emissions e.g. vessel speed management depending on operational requirements	Administration costs for monitoring and opportunity evaluation activities.	Adopted - Optimised support vessel fuel consumption has emissions reduction and cost reduction benefits.
BAO-CM-6.3.9	Reporting of GHG emissions as per the NGER Scheme (administrative control)	This is a regulatory requirement under the NGER Act with which Santos and its contractors must comply.	Cost associated with implementing.	Adopted – NGER reporting is a Commonwealth regulatory requirement.

Table 6-18: Control measures evaluation for atmospheric and greenhouse gas emissions

CM reference	Control measure	Environmental benefit	Potential cost/issues	Evaluation
BAO-CM-6.3.11	The purchase and/or surrender of Australian carbon credit units (ACCUs) or Safeguard Mechanism Credits (SMCs) required under the NGER (Safeguard Mechanism) Rule 2015 for any non-reservoir emissions from the Barossa facility above the annual baseline, as determined by the Clean Energy Regulator (administrative control)	Emissions from the Barossa facility (FPSO and GEP) are managed in accordance with baselines set by the Commonwealth government under the Safeguard Mechanism, which supports achievement of the Commonwealth Government's emissions reduction targets under the Climate Change Act 2022 and Australia's Paris Agreement NDCs.	Cost of ACCUs/SMCs. Availability of ACCUs in 2023, Santos, as an independent company, entered into forward contracts for the purchase of 2.5 million ACCUs at fixed prices to be delivered and paid between December 2023 and January 2027. As discussed in Section 6.3.2.3.2, the responsible Commonwealth minister has communicated to the Barossa JV on a number of occasions that he anticipates there being ACCUs available for this project. The availability of ACCUs over the next decade is supported by analysis of the balance between supply, holdings, and cancellation rates undertaken by RepuTex Energy, 2024.	Adopted – in line with regulatory requirements, noting that Santos is committed to reducing emissions to ALARP regardless of regulatory requirements.
BAO-CM-6.3.12	Implement an Operations GHG Emissions Management Plan (GHGEMP) as described in Section 8.3.2.12 to manage facility direct GHG emissions to ALARP over the life of the Activity, inclusive of: • Emissions Performance target setting (Section 8.2.4) • Critical Equipment Maintenance (Section 8.3.2.3.1) • Methane emissions management (Section 8.3.2.11) • Decarbonisation opportunity management (Section 8.5.6) (administrative control)	Managing uncertainty and reduction of GHG emissions to ALARP over the life of the Activity.	Best practice. Opportunity cost between emissions reduction and production impact.	Adopted – benefit outweighs cost. Ongoing reduction of direct (Scope 1) facility GHG emissions to ALARP over the life of the Barossa Gas Project (both the FPSO and the GEP) will be governed under the Operations GHG Emissions Management Plan (GHGEMP).
BAO-CM-6.3.16	MARPOL-compliant (Marine Order 97) fuel oil will be used by vessels to reduce atmospheric emissions (substitution control)	Reduces emissions through use of low sulphur fuel in accordance with MARPOL Annex VI (and Marine Order 97).	None identified.	Adopted – it is a legislated requirement.

CM reference	Control measure	Environmental benefit	Potential cost/issues	Evaluation
BAO-CM-6.3.17	Pursuant to Marine Order 97, relevant vessels will have a current International Air Pollution Prevention (IAPP) Certificate or equivalent and Ship Energy Efficiency Management Plan (SEEMP) (administrative control)	Reduces emissions by ensuring compliance with MARPOL Annex VI (and Marine Order 97).	Cost of maintaining certification.	Adopted – benefit of ensuring vessel is compliant outweighs the minimal costs and it is a legislated requirement.
BAO-CM-6.3.21	<ul> <li>GHG emissions reduction initiatives of suppliers for the Barossa Gas Project will be evaluated in the tender evaluation process via development and implementation of a framework for identifying, assessing and implementing emissions reduction opportunities for all Barossa Gas Project supplier contracts of \$30m+ value.</li> <li>Through the data collection and tender evaluation process, opportunities to collaborate on emissions reduction initiatives and low carbon alternatives will be sought, including the potential to support suppliers in respect of:</li> <li>investments in innovations in technology.</li> <li>research programs.</li> <li>education and training relating to the adoption of emissions reduction policies and processes and/or</li> <li>monitoring programs</li> <li>The tender evaluation framework will be reviewed and refined to ensure it is adaptive to advancements in technology, data collected and other opportunities to encourage reductions in GHG emissions.</li> <li>(administrative control)</li> </ul>	Engaging with suppliers and collaborating-on initiatives will support suppliers to reduce their own GHG emissions. Review and refinement of the approach will continue to encourage entrepreneurialism and the take up of GHG reduction initiatives. The materiality threshold of \$30m+ for supplier contract value has been selected on the basis that it represents approximately 65% of current and budgeted Barossa Gas Project supplier contracts, representing around 85% of the forecast Barossa Gas Project emissions for the first 5 years of Barossa operations. It also aligns with existing executive level delegation procedures that will ensure an additional level of rigour is applied at the procurement phase.	There are potential costs associated with implementing initiatives.	Adopted – Collaborating with suppliers on initiatives will be adopted subject to a feasibility analysis, the willingness of suppliers to collaborate and value to the environment. Ongoing review of the threshold spend will be conducted to ensure the most emissions intensive activities for the Barossa Gas Project are captured.

CM reference	Control measure	Environmental benefit	Potential cost/issues	Evaluation
BAO-CM-6.3.22	<ul> <li>Annual engagement with suppliers with Barossa Gas Project supplier contracts of \$30m+ value to request GHG emissions data for Barossa activities. Data sought would include:</li> <li>quantitative and qualitative climate- related targets (including for Scopes 1, 2 and 3 emissions)</li> <li>information about the supplier's approach to setting, reviewing and monitoring progress against each target</li> <li>information about the supplier's performance against each climate- related target, including GHG emissions data and measurement approach, inputs and assumptions, over the past year</li> <li>the supplier's use over the past year, and planned use, of carbon credits to offset GHG emissions</li> <li>information regarding the supplier's emissions reduction initiatives (if any).</li> <li>Data will be used to verify GHG emissions estimates associated with our suppliers and track performance against Santos' Scope 3 equivalent climate targets and Climate Transition Action Plan. (administrative control)</li> </ul>	Ensures estimates for these emissions are aligned with best practice approach and are within the bounds of this OEMP. The materiality threshold of \$30m+ for supplier contract value has been selected on the basis that it represents approximately 65% of current and budgeted Barossa Gas Project supplier contracts, representing around 85% of the forecast Barossa Gas Project emissions for the first 5 years of Barossa operations It also aligns with existing executive-level delegation procedures that will ensure an additional level of rigour is applied at the procurement phase.	There are potential challenges associated with procuring actual emissions data from third parties.	Adopted – Collaborating with suppliers will be subject to the willingness of suppliers to engage. Ongoing review of the threshold spend will be conducted to ensure the most emissions intensive activities for the Barossa gas project are captured.

CM reference	Control measure	Environmental benefit	Potential cost/issues	Evaluation
BAO-CM-6.3.24	<ul> <li>Supporting Barossa Gas</li> <li>Project suppliers to reduce</li> <li>GHG emissions by:</li> <li>promoting global measurement and reporting standards by participating in relevant industry associations and collaboration initiatives; and</li> <li>advocating for policy frameworks that enable a consistent approach to carbon emissions management.</li> <li>(administrative control)</li> </ul>	There are collective benefits to standardised, transparent and effective measurement and reporting standards, along with stable policy frameworks, which will promote GHG emissions reduction globally.	Nil. Santos will engage in these efforts and initiatives as part of its routine business endeavours.	Adopted
BAO-CM-6.4.1	Ozone depleting substance (ODS) and lower global warming potential (GWP) refrigerants use and handling procedures (administrative control)	Reduces probability of potential impacts to air quality due to ODS emissions.	Personnel cost of maintaining ODS record book/ recording system.	Adopted – benefits of ensuring no ODS release outweighs the minimal costs
BAO-CM-6.4.3	Vessel waste incineration management (engineering control)	Reduces the potential for emissions/ particulates by ensuring only permissible waste is incinerated as per Marine Order 97.	Cost associated with onshore waste disposal.	Adopted – impact on air quality outweighs the costs and impacts associated with transporting waste to shore for landfill.
BAO-CM-6.4.4	National Pollution Inventory (NPI) Reporting (administrative control)	Collects information about emissions across Australia.	Administrative costs of recording and collating information and reports	Adopted – legislated requirement.
BAO-CM-6.4.5	HSE inductions will include applicable environmental requirements (administrative control)	Ensures that crew are aware of the stringent EP, Santos and legislative requirements.	Administrative costs to update existing Santos procedure and induction materials and train personnel.	Adopted – benefits considered to outweigh costs
BAO-CM-6.4.10	Barossa Facilities and vessels planned maintenance system to confirm equipment integrity is maintained in accordance with manufacturers guidelines. (administrative control)	Reduces emissions by ensuring vessels are operated, maintained and crewed in accordance with industry standards and regulatory requirements.	Personnel costs of implementing.	Adopted – benefits of operating equipment within Operational parameters will help control emissions created by equipment.
Additional control	measures			
BAO-CM-6.3.18	Santos' vessel vetting process to include evaluation of vessel emissions and the potential for use of alternative fuels (substitution control)	Reduces total emissions associated with engines.	Consideration to be made but potentially limited by required specifications and vessel availability.	Adopted –. The vessels selected will comply with Santos' vessel vetting process.

CM reference	Control measure	Environmental benefit	Potential cost/issues	Evaluation
N/A	Removal of all ODS containing equipment from contracted vessels (elimination control)	Eliminates potential of ODS emissions occurring.	ODS is rarely found on vessels and there is a low potential for ODS releases. If there is ODS- containing equipment (such as refrigerators), it will be managed as per Marine Order 97: Marine Pollution Prevention – Air Pollution.	Not adopted – based on cost to replace all equipment and the low potential for ODS releases.
N/A	Require vessel contractors to install / use incinerators and engines with higher environmental efficiency (administrative control)	Improves air quality by more efficient burning or fuel combustion.	Significant cost in changing vessel equipment.	Not adopted – cost grossly disproportionate to low environmental benefit (impact rated Negligible).
N/A	Using lower emissions vessels	Reduces total emissions associated with engines.	Not practically feasible at present. The contracted vessels required are specialised and have limited availability. The vessels selected will comply with Santos' vessel vetting process.	Not adopted – Not practically feasible at present. The contracted vessels are specialised and have limited availability. The vessels selected will comply with Santos' vessel vetting process.

### 6.3.4 Environmental impact assessment

Receptor	Consequence level	
Atmospheric and GHG emi	ssions	
Threatened, migratory or local fauna	The following recovery and conservation plans listed in Table 3-13 identify climate change as a threat.	
	Conservation Advice:	
	Approved Conservation Advice for <i>Rhincodon typus</i> (whale shark) (2015)	
	Approved Conservation Advice for Balaenoptera physalus (fin whale) (2015)	
	Approved Conservation Advice for Balaenoptera borealis (sei whale) (2015)	
	<ul> <li>Commonwealth Conservation Advice on <i>Dermochelys coriacea</i> (Leatherback turtle) (2008)</li> <li>Approved Conservation Advice for <i>Calidris canutus</i> (Red knot) (2016)</li> <li>Recovery and management plans:</li> </ul>	
	Recovery Plan for Marine Turtles in Australia 2017–2027 (CoA, 2017)	
	Recovery Plan for the White Shark (Carcharodon carcharias) (CoA, 2013)	
	<ul> <li>Wildlife Conservation Plan for Seabirds (CoA, 2020)</li> </ul>	
	<ul> <li>Wildlife Conservation Plan for Migratory Shorebirds (CoA, 2015c)</li> </ul>	
	<ul> <li>Conservation Management Plan for the Blue Whale 2015–2025 (CoA, 2015a).</li> </ul>	
	Atmospheric emissions	
	Short-term behavioural impacts (e.g. avoidance) to seabirds could be expected if they fly in the vicinity of the location of the Activity in NT Coastal Waters. No decrease in local population size or area of occupancy of species, loss or disruption of critical habitat or disruption to the	



Receptor	Consequence level	
	breeding cycle. The consequence level for threatened migratory or local fauna (seabirds) is considered to be I – Negligible.	
	GHG emissions	
	Considering that:	
	<ul> <li>In any event, even assuming that emissions from the Activity will cause an equivalent net increase in cumulative Australian and global emissions, this increase comprises a nominal amount in the overall scheme of the national and international carbon budgets; and</li> </ul>	
	<ul> <li>In any event, there is no correlation between where GHG emissions are released and where climate change impacts are felt,</li> </ul>	
	It is not possible to draw a link between GHG emissions from the Activity and any specific climate related impact on threatened, migratory or local fauna which may result from any net increase to cumulative GHG emissions globally. By keeping the Project emissions under the Safeguard Mechanism baseline, the Project's incremental contribution to global warming is within acceptable limits as determined by reference to Australia's NDC under the UNFCCC. Conservatively, the associated potential environmental impacts to Threatened, Migratory or local fauna (e.g. seabirds) is assessed as I – Negligible.	
Physical environment or	Atmospheric emissions	
habitat	The activity vessels will generate atmospheric emissions in the open ocean and offshore waters, enabling emissions to dissipate into the surrounding atmosphere quickly. The consequence level for physical environment/habitat is assessed as I – Negligible.	
	GHG emissions	
	The physical environment and associated habitats are susceptible to the effects of climate change. Marine and coastal environments (including those within the EMBA such as shoals and banks, coral reefs, seagrass and algal habitat and mangrove communities) are susceptible to climate change effects such as ocean warming, ocean acidification, rising sea level and changes to ocean current and storm regimes and associated changes to coastal processes. Terrestrial communities are susceptible to changes to temperature, changes to rainfall patterns and environments of fire and environments.	
	Considering that:	
	<ul> <li>In any event, even assuming that emissions from the Activity will cause an equivalent net increase in cumulative Australian and global emissions, this increase comprises a nominal amount in the context of Australian and global carbon budgets; and</li> </ul>	
	<ul> <li>In any event, there is no correlation between where GHG emissions are released and where climate change impacts are felt,</li> </ul>	
	It is not possible to draw a link between GHG emissions from the Activity and any specific climate related impact on the Australian environment which may result from any net increase to cumulative GHG emissions globally. By keeping the Barossa Gas Project's emissions under the Safeguard Mechanism baseline, the Project's (of which the Activity is a small component part) incremental contribution to global warming is within acceptable limits as determined by reference to Australia's NDC under the UNFCCC. Conservatively the associated potential environmental impacts to the physical environment and habitat is assessed as I – Negligible	
Threatened ecological	Atmospheric emissions	
communities	Not applicable – no threatened ecological communities were identified in the area over which air emissions are expected, therefore no impacts associated with atmospheric emissions.	
	GHG emissions	
	Threatened ecological communities are susceptible to the effects of climate change. Coastal / terrestrial communities are susceptible to climate change effects such as rising sea level, changes to ocean current and storm regimes, changes to temperature and rainfall patterns and changes to the frequency and severity of fire events. Considering that:	
	<ul> <li>In any event, even assuming that emissions from the Activity will cause an equivalent net increase in cumulative Australian and global emissions, this increase comprises a nominal amount in the context of Australian and global carbon budgets; and</li> </ul>	
	<ul> <li>In any event, there is no correlation between where GHG emissions are released and where climate change impacts are felt,</li> </ul>	
	It is not possible to draw a link between GHG emissions from the Activity and any specific climate related impact on the Australian environment which may result from any net increase to cumulative GHG emissions globally. By keeping the Barossa Gas Project's emissions under the Safeguard Mechanism baseline, the Project's (of which the Activity is a small component part) incremental contribution to global warming is within acceptable limits as determined by	

Receptor	Consequence level	
	reference to Australia's NDC under the UNFCCC. Conservatively the associated potential environmental impacts to threatened ecological communities is assessed as I – Negligible.	
Protected areas	Atmospheric emissions	
	Not applicable – no protected areas over which atmospheric emissions are expected.	
	GHG emissions	
	The values of protected areas associated with threatened and migratory fauna are described above (Threatened, Migratory or local fauna).	
	Protected areas, include Australian Marine Parks, World Heritage Properties, Commonwealth Heritage Places, Ramsar and nationally important wetlands and key ecological features (KEFs).	
	The Australian Marine Park network supports natural, cultural, heritage and socio-economic values, including habitats (e.g. coral reefs), listed EPBC Act and culturally important species and high species diversity. These values are susceptible to the effects of climate change such as ocean warming, ocean acidification, rising sea level and changes to ocean current and storm regimes. The Australian Marine Parks within the EMBA include the Oceanic Shoals Marine Park and the Joseph Bonaparte Gulf and contain natural, cultural, heritage and socio-economic values (refer Section 3.2.12) that are susceptible to effects of climate change. The Garig Gunak Barlu Marine National Park (refer 3.2.12.1) is within the EMBA and susceptible to the same climate change effects as Australian Marine Parks.	
	KEFs are Commonwealth marine areas important for biodiversity and ecosystem function and integrity. Similar to Australian Marine Parks, KEFs have values that are susceptible to effects of climate change. The KEFs that occur in the EMBA are typically geomorphic features (e.g. tributaries, carbonate bank and terrace system, pinnacles) (refer 3.2.12.3). They can support marine fauna that use the surface waters above the features, including plankton, pelagic invertebrates and fish, marine mammals, marine reptiles and seabirds.	
	Ramsar and nationally important wetlands are critical for biodiversity and ecological balance and provide habitat for EPBC Act listed species (e.g. species of seabirds and shorebirds). Wetlands are susceptible to the effects of climate change such as sea level rise, increased temperatures and increased intensity and frequency of storms and fires. Within the EMBA, Ramsar and nationally important wetlands includes the Coburg Peninsula (refer Section 3.2.12.2). These wetlands provide key habitats that support a high diversity and abundance of migratory birds and various wetland habitats. Considering that:	
	<ul> <li>In any event, even assuming that emissions from the Activity will cause an equivalent net increase in cumulative Australian and global emissions, this increase is de minimis in the context of Australian and global carbon budgets; and</li> </ul>	
	<ul> <li>In any event, there is no correlation between where GHG emissions are released and where climate change impacts are felt,</li> </ul>	
	it is not possible to draw a link between GHG emissions from the Activity and any specific climate related impact on protected area which may result from any net increase to cumulative GHG emissions globally. By keeping the Project emissions under the Safeguard Mechanism baseline, the Project's (of which the Activity is a small component part) incremental contribution to global warming is within acceptable limits as determined by reference to Australia's NDC under the UNFCCC. Conservatively the associated potential environmental impacts to protected areas is assessed as I – Negligible.	
Socioeconomic receptors	Atmospheric emissions	
	Gaseous emissions are relatively small, will quickly dissipate into the surrounding atmosphere, and are not considered to be a potential source of impact to socioeconomic receptors.	
	As the Activity occurs in offshore waters, the air quality in coastal towns or settlements will not be affected.	
	GHG emissions	
	Changes to climate can result in impacts to social receptors that have values which include the ecological receptors described above, including KEFs and Australian Marine Parks (AMPs). Climate change may also impact on the functions, interests or activities of other users which rely on these ecological values, including commercial and recreational fisheries and tourism.	
	Considering that:	
	In any event, even assuming that emissions from the Activity will cause an equivalent net increase in cumulative Australian and global emissions, this increase is de minimis in the context of Australian and global carbon budgets; and	
	<ul> <li>In any event, there is no correlation between where GHG emissions are released and where climate change impacts are felt,</li> </ul>	
	It is not possible to draw a link between GHG emissions from the Activity and any specific climate related impact on the Australian environment which may result from any net increase to	

Receptor	Consequence level
	cumulative GHG emissions globally. By keeping the Barossa Gas Project's emissions under the Safeguard Mechanism baseline, the Project's (of which the Activity is a small component part) incremental contribution to global warming is within acceptable limits as determined by reference to Australia's NDC under the UNFCCC. Conservatively the associated potential environmental impacts to socioeconomic receptors is assessed as I – Negligible.
Cultural features	For assessment of impacts to marine species of cultural significance, refer to the assessment for threatened, migratory or local fauna.
	For assessment of impacts to the physical environment to which First Nations people are connected and have raised concerns, refer to the assessment for the physical environment/ threatened ecological communities / protected areas.
Overall worst-case consequence	I – Negligible

#### 6.3.5 Demonstration of as low as reasonably practicable

Based on the environmental risk assessment outcomes and use of the relevant tools appropriate to the decision, Santos considers the adopted control measures to have reduced the impacts and risks of GHG and atmospheric emissions from this Activity to ALARP through the following measures:

- Facilities design measures to reduce emissions to ALARP.
- Implementation of an Operations GHGEMP (that applies across all Barossa Gas Project Facilities including the GEP) to reduce Scope 1 emissions from facility operations to ALARP over the life of the Project (Section 8.3.2.12). The GHGEMP has been adopted as a specific control measure for the Barossa Gas Project (BAO-CM-6.3.12) and will apply to the Production Operations EP and this OEMP. It is recognised that the majority of Barossa direct (Scope 1) emissions are associated with sources from FPSO processing activities, with the operation of the GEP a minor component. In adopting the GHGEMP the corresponding environmental performance standards are incorporated across the Barossa Production Operations EP and this OEMP. The following protocols, procedures, systems and measures, as detailed further within the implementation strategy at the sections cross-referenced below, will be incorporated in, and form part of, the GHGEMP:
  - Emissions Performance target setting (Section 8.2.4)
  - Critical Equipment Maintenance (Section 8.3.2.3.1)
  - Methane Emissions Management (Section 8.3.2.11)
  - Decarbonisation Opportunity Management (Section 8.5.6)

Adaptive management, to address areas of uncertainty, is integral to emissions performance target setting, methane emissions management and decarbonisation opportunity management.

- Engagement with Barossa Gas Project suppliers at the tender stage and on an annual basis via control measures and performance standards (see BAO-CM-6.3.21, 6.3.22 and 6.3.24). This engagement will:
  - enable Santos to verify its own Scope 3 emissions estimates associated with the Barossa Gas Project via data collection
  - assist Santos in tracking performance against its Scope 3 equivalent climate targets and Climate Transition Action Plan
  - ↔ facilitate support being provided to suppliers to reduce their own GHG emissions, thereby reducing the indirect GHG emissions associated with the Barossa Gas Project
- Promoting global measurement and reporting standards and advocating for policy frameworks that enable a consistent approach to carbon emissions management.

For this Activity, atmospheric and greenhouse gas emissions are largely unavoidable due to operational and health and safety considerations. The use of vessels and thereby fuel emissions cannot be eliminated. Alternative fuels (biofuels) have not been commercially proven at scale.

All reasonably practicable control measures were reviewed and those adopted are considered consistent with maritime/energy industry standards and appropriate to manage the impacts such that the residual consequence is assessed to be I – Negligible. The proposed management controls are in accordance with Santos' risk management criteria and are considered appropriate to reduce impacts to ALARP.

#### 6.3.6 Acceptability evaluation



Is the consequence	Yes – maximum consequence from atmospheric and greenhouse gas emissions is I – Negligible.		
	As outlined in Section 6.3.3 the acceptable levels of GHG emissions are defined as follows:		
	<ul> <li>Acceptable levels of direct (Scope 1) emissions from the Activity are set by the Safeguard Mechanism Baseline on Barossa facility emissions, in line with the Australian Government's NDCs under the Paris Agreement.</li> </ul>		
	• Scope 3 emissions associated with the Activity are limited to IMMR vessel activities and are estimated to be nominal to negligible in the overall context of national and international carbon budgets. Acceptable levels of indirect (Scope 3) emissions from the Activity are those estimated to result from IMMR vessel activities, provided that the control measures relating to indirect emissions as outlined in this OEMP are also implemented.		
	Compliance with the Safeguard Mechanism is of primary importance in ensuring that Scope 1 GHG emissions associated with the Barossa Gas Project are kept to acceptable levels. The emissions associated with the Activity represent an incremental component part of the Scope 1 emissions of the Barossa Gas Project. GHG emissions at or below the baseline and the Safeguard Mechanism's future decline rates are already anticipated and thus accounted for under Australia's NDCs under the Paris Agreement.		
	As outlined in Table 6-17, GHG emissions (within Australia) from the GEP represent 0.0017% of Australia's 2050 net carbon budget.		
	<ul> <li>GHG emissions from the GEP represent 0.00002% and 0.000009% respectively of global net carbon budgets under 1.5°C and 2°C temperature increase scenarios.</li> </ul>		
	At 0.0017% of Australia's carbon budget to 2050, and 0.00002% and 0.000009% of net global carbon budgets for 1.5°C and 2°C temperature increases, the emissions from the GEP, of which the section of the pipeline comprising the Activity represents ~2% of the total GEP, comprise a nominal amount in the overall scheme of the national and international carbon budgets and will not materially or substantially contribute to Australia's net GHG emissions or net global emissions levels. Santos has adopted environmental performance outcomes and control measures directed to minimising GHG emissions from the Activity and therefore the potential contribution of these emissions to net cumulative GHG emissions globally. A range of controls have been considered and adopted for both direct (Scope 1) and indirect (Scope 3) emissions.		
Is further information required to validate the	No – potential impacts and risks are well understood through the information available and Relevant Persons consultation.		
consequence assessment?	The predicted atmospheric and GHG emissions associated with the Activity comprise a nominal amount in the overall scheme of the national and international carbon budgets and will not materially or substantially contribute to existing and future predicted Australian and global GHG emissions, having regard to the acceptability criteria as outlined in the preceding row.		
Are the risks and impacts consistent with the	Yes – activity evaluated in accordance with Santos' Offshore Division Environmental Hazard Identification and Assessment Guideline, which considers principles of ESD.		
principles of ecologically sustainable development (ESD)?	Ensuring that GHG emissions are consistent with the principles of ESD requires balancing economic, social and environmental considerations. The Activity will be regulated under the Safeguard Mechanism under the NGER Act, which plays a critical role in achieving adherence to the principles of ESD. It does this by setting emissions baselines and requiring facilities to implement measures that minimise emissions over time, consistent with Australia's NDCs under the Paris Agreement. Implementation of the proposed control measures will reduce GHG emissions from the Activity to within baseline emissions levels.		
	The GHG emissions of the Barossa Gas Project as a whole are appropriately and comprehensively addressed in the Barossa Production Operations EP, including a detailed assessment of how the impacts and risks of GHG emissions of the whole Development are consistent with the principles of ESD.		
Have the acceptable	Yes –maximum consequence from atmospheric and GHG emissions is I – Negligible.		
levels of impact and risks been informed by relevant species recovery plans, threat abatement plans and conservation advice and Australian marine park zoning objectives?	The Marine Bioregional Plan for the North Marine Region (CoA, 2012) includes consideration of the effects of air quality on species. The implementation of EPO-09 and the control measures outlined in Table 6-18 will ensure the atmospheric emissions from the Activity will not compromise this conservation effort.		
	At 0.0017% of Australia's carbon budget to 2050, and 0.00002% and 0.000009% of net global carbon budgets for 1.5°C and 2°C temperature increases, the emissions from the GEP are de minimis and will not materially or substantially contribute to Australia's net GHG emissions or net global emissions levels. The emissions attributable to the Activity are an even smaller portion of the GEP's emissions. In any event, net carbon budgets are made up of both additions and subtractions to cumulative GHG emissions. Whether or not a net increase in cumulative Australian GHG emissions will occur is subject to multiple variables outside of Barossa JV's' control, other than its responsibility to manage the Activity's emissions to keep them under the Safeguard Mechanism baseline.		
	For all the recovery plans identified in Table 3-13, the objectives are achieved through the adoption of EPO-09, EPO-10 and EPO-11 and the control measures outlined in Table 6-18.		

Are performance outcomes, control measures and associated performance standards consistent with legal and regulatory requirements?	<ul> <li>Yes – management consistent with the Ozone Protection and Synthetic Greenhouse Gas Management Act 1989 (and associated regulations), MARPOL VI/Marine Order 97 and Protection of the Sea (Prevention of Pollution from Ships) Act 1983, MARPOL VI/Marine Order 97, National Environment Protection (Ambient Air Quality) Measure and National Pollutant Inventory, OPGGS Act, OPGGS(E) Regulations, National Environmental Protection Council Act 1994 (Cth) and Environment Protection (National Pollutant Inventory) Objective 2004 (NT).</li> <li>Management measures are consistent with the <i>Climate Change Act 2022</i> (Cth), <i>Ozone Protection and Synthetic Greenhouse Gas Management Act 1989</i> (Cth) (and associated regulations), <i>Protection of the Sea (Prevention of Pollution from Ships) Act 1983</i> (Cth) (and associated regulations), and MARPOL VI/Marine Order 97.</li> <li>Performance outcomes, control measures and associated performance standards to manage the impacts and risks from GHG emissions associated with the Activity, being emissions that are a component of the overall emissions of the Barossa Gas Project, are consistent with relevant global agreements and frameworks and Australian legislation and regulations, including:</li> <li>the NGER (Safeguard Mechanism) Rule 2015, the regulatory mechanism of primary relevance to Barossa production operations in Commonwealth waters GHG emissions, which requires net-zero reservoir emissions, and Scope 1 emissions above a facility-specific baseline to be offset. The Safeguard Mechanism is the appropriate performance standard for Scope 1 emissions from the Activity. This legislation reflects Australia's pathway to meet its obligations under the Paris Agreement.</li> <li>the <i>National</i> Greenhouse <i>and Energy Reporting Act 2007</i></li> <li>GHG emissions are globally managed through the implementation of the Paris Agreement agreed under the United Nations Framework Convention on Climate Change at COP21 in 2015, which has established a global framework under which countries individual sig</li></ul>
	will be met as per Appendix B.
Are performance outcomes, control	Yes – Performance outcomes, control measures and associated performance standards align with relevant Santos policies, including Santos':
measures and associated performance	<ul> <li>Santos' Environment, Health and Safety Policy (Appendix A)</li> </ul>
standards consistent with relevant Santos policies?	<ul> <li>Climate Report 2024 (part of the Santos Annual Report 2024) and the emissions reductions targets outlined therein</li> </ul>
	Climate Transition Action Plan.
	The environmental performance outcomes and the controls that will be implemented are consistent with Santos' internal requirements in its capacity as an independent company.
	Direct emissions from this Activity will be incorporated into the total emissions reporting undertaken by Santos with respect to the Barossa Gas Project once the project becomes operational. Further details regarding Santos' emissions reporting under NGERs and the Safeguard Mechanism are outlined in the Production Operations EP for the Barossa Gas Project. Climate change management is embedded within Santos' business strategy, including lowering operating emissions, in its capacity as an independent company.
Are performance	Yes – Consideration and adoption of controls described in relevant best practice industry
outcomes, control measures and	standards including:
associated performance standards consistent with industry standards?	<ul> <li>Environmental management in the upstream oil and gas industry – IOGP- IPIECA, 2020.</li> <li>OGCI: Aiming for Zero Methane Emissions Initiative</li> </ul>
	IPIECA Flaring Management Guidance
	IPIECA Sustainability Guide
	The GEP Coastal Waters OEMP is also compliant with the conditions of EPBC approval EPBC 2022/09372.
Have performance outcomes, control	Yes – Issues, objections / claims and measures discussed with Relevant Persons during consultation for this activity, including with the ACF, the ECNT and the Clean Energy Regulator,

measures and associated performance standards taken into consideration Relevant Person feedback?	have been considered when evaluating performance outcomes, control measures and associated performance standards (see Section 4.7 for further information). It is acknowledged that GHG emissions associated with the Activity, and the impacts of climate change, were noted as a material issue for Relevant Persons consulted in the course of preparing the Barossa Production Operations EP and this GEP Coastal Waters OEMP. EPOs and control measures have been adopted to ensure that GHG emissions associated with the Activity are ALARP and acceptable. In developing and determining these EPOs and control measures, specific control measures proposed by Relevant Persons (including the ACF and ECNT) were evaluated. Where relevant, control measures implemented based on consultation with Relevant Persons for other Barossa EPs were also evaluated in developing the EPOs and control measures for this GEP Coastal Waters OEMP. The final form of the EPOs and control measures adopted have been determined and adopted for the reasons set out in this GEP Coastal Waters OEMP.
Are performance standards such that the impact or risk is considered to be ALARP?	Yes – ALARP assessment conducted, one additional control measure adopted.

The consequence of atmospheric and GHG emissions on receptors is assessed as I – Negligible. Based on an assessment of Santos' acceptability criteria and with the control measures in place, there is expected to be no substantial contribution to global GHG emissions or change in air quality that may adversely impact the environment and the potential impacts are considered acceptable.



### 6.4 Seabed and benthic habitat disturbance

Event	Disturbance to the seabed in the OA will occur as a result of:		
	<ul> <li>temporary placement and set down of equipment and subsea infrastructure on the seabed (e.g. ROV, tooling baskets and equipment)</li> </ul>		
	<ul> <li>temporary seabed and sediment disturbance during IMMR activities, such as:</li> </ul>		
	<ul> <li>cleaning requiring marine growth removal, which results in resuspension of sediment</li> </ul>		
	<ul> <li>stabilisation of subsea infrastructure requiring the placement of material such as rocks, grout and gravel bags or mattresses on the seabed</li> </ul>		
	<ul> <li>span rectification of subsea infrastructure requiring the placement of material such as rocks, grout and gravel bags on the seabed</li> </ul>		
	<ul> <li>the replacement, maintenance, and repair of subsea equipment components</li> </ul>		
	<ul> <li>subsea infrastructure (including sections of the GEP) repair and replacement</li> </ul>		
	<ul> <li>sediment relocation required to gain access to subsea infrastructure</li> </ul>		
	<ul> <li>environmental monitoring activities such as sampling of seabed material (i.e. sediment) or biotic material (i.e. marine growth) for environmental studies as and if required</li> </ul>		
	The approximate seabed disturbance footprint is detailed in Table 6-19.		
	Seabed disturbance may also cause a localised temporary increase in water turbidity.		
Extent	Seabed disturbance will be localised within the OA, with the worst-case disturbance being the (unplanned) replacement of a section of GEP.		
Duration	Temporary disturbances and placements for the duration of the Activity, being approximately 7 to 30 days, every three to five years, or as needed.		

#### 6.4.1 Description of event

#### 6.4.2 Nature and scale of environmental impacts

**Potential receptors:** physical environment (benthic habitat), threatened, migratory or local fauna (benthic fauna), socioeconomic (commercial fisheries and underwater cultural heritage), and cultural features.

The approximate seabed footprint from the Activity is provided in Table 6-19. Section 2 describes the Activity in detail.

#### 6.4.2.1 Physical environment

The Activity will involve direct and indirect disturbance to the sea floor and will inevitably result in localised impacts to benthic habitats (and associated fauna) within the OA.

The GEP route in the OA is in water depths of approximately 47 – 50 m and in an area of high turbidity, limiting photosynthetically active radiation and benthic primary producer habitats. The seabed within the OA is characterised as featureless silty, shelly sand, with very sparse (<1%) epibiota (mainly soft corals and crinoids) (RPS, 2023). Shepparton Shoal is the closest shoal or bank, 23 km west of the OA (see Section 3.2.9). The benthic habitats and fauna assemblages that are expected to be impacted are considered widespread throughout the region.

Seabed and benthic habitat disturbance will be confined to the OA, within the footprint of subsea infrastructure or materials placed on the seabed. The typical footprint of seabed disturbance is defined in Table 6-19. Materials are placed in localised areas close to the subsea infrastructure and usually over areas of previous disturbance, such as within the GEP corridor. Disturbance to sediment habitat may result in epifauna removal or a reduction in the abundance and diversity of infauna.

The GEP and supporting structures are expected to sink or become partially buried over time, with localised soft sediment accumulation around the pipeline. Localised scouring may also occur around the infrastructure due to strong currents and subsurface waves. Scouring may necessitate IMMR activities, such as span rectification or the placement of grout and gravel bags or mattresses on the seabed. Subsea infrastructure (including replacement segments of the GEP or mattresses and grout bags for span rectification) will be lowered onto the seabed in a controlled manner with minimal disturbance to the sediment. Habitat directly beneath installed infrastructure will most likely be replaced by the hard substrate of the structures.

Given the mobile nature of sediments and high current speeds in the OA, the seabed is expected to infill naturally with sediments and detrital matter, returning to near its original state over a short time (weeks to months). Benthic habitats would remain viable and are expected to recolonise through the recruitment of new colonists from planktonic larvae in adjacent undisturbed areas (Guerra-Garcia & Garcia-Gomez, 2006).



Furthermore, the GEP or any additionally installed IMMR infrastructure will create a hard substrate in an otherwise featureless and soft seabed. It can be reasonably expected that this substrate would serve as an anchor for benthic organisms, such as sponges and gorgonians like those present on hard substrates at equivalent depths within the region. These organisms may create more complex habitats, supporting a localised increase in biodiversity over time.

Activity	Description	Footprint of disturbance
ROV use	Seabed disturbance from temporary placement of ROV, tooling baskets and equipment on the seabed.	4.25 m <sup>2</sup>
	Seabed disturbance from placement of tooling baskets on the seabed, required for maintenance and repair of subsea equipment and infrastructure.	<3 m <sup>2</sup>
Cleaning of infrastructure and equipment requiring marine growth removal	Seabed disturbance from removed marine growth settling on the seabed.	<1 m <sup>2</sup>
Environmental monitoring	Seabed disturbance from removal from small amount of sediment removal.	<1 m <sup>2</sup>
Placement of stabilisation or rectification materials, such as rocks, grout and gravel bags or mattresses, on the seabed	Seabed disturbance is typically limited to areas around subsea infrastructure at small sections of GEP within the OA. The exact details and requirements are made after inspections and surveys. Span rectification and placement of stabilisation materials is typically required very infrequently.	<1 m <sup>2</sup> up to approximately 50 m <sup>2</sup>
Replacement, maintenance and repair of subsea infrastructure	Seabed disturbance from the replacement, maintenance and repair of subsea equipment components, as well as subsea infrastructure, which is typically required very infrequently. Within the OA, the largest infrastructure that could potentially be replaced is a section of the GEP, resulting in seabed disturbance in a localised area of the GEP corridor. In this unplanned scenario, four pipe lift frames are typically deployed along a distance of approximately 100 m, each with a footprint of approximately 400 m <sup>2</sup> .	1,600 m <sup>2</sup>
Sediment relocation	Seabed disturbance from sediment relocation, which may be required to gain access to subsea infrastructure during IMMR. Seabed disturbance is localised to the GEP corridor.	<250 m <sup>2</sup>

#### 6.4.2.2 Water quality

Impacts on water quality are expected to be limited to increased turbidity (suspended sediments) and subsequent sediment deposition resulting from placing materials and infrastructure on the seabed, or from the relocation of sediment to gain access to infrastructure. IMMR activities may create a localised and temporary plume of suspended sediment over the area of seabed disturbance, which subsequently settles on the seabed after a period in the water column. Localised areas of the seabed and the associated fauna may be smothered by these sediments.

Turbidity effects are expected to be localised and short-term while the water column returns to its original state over a period of days. The impact on water quality is predicted to be negligible, with no substantial adverse effects on biodiversity, ecological integrity, social amenity, or human health.

#### 6.4.2.3 Threatened, Migratory or local fauna

Habitat modification is identified as a potential threat to several marine fauna species in relevant recovery plans and conservation advice (Table 3-13), some of which have cultural significance as totems or cultural food sources. However, area of expected seabed disturbance from the Activity represents a negligible portion of the habitat available for threatened, migratory or local fauna, and is not inconsistent with relevant recovery plans and conservation advice.

#### 6.4.2.3.1 Sharks, rays and fish

The seabed in the OA is predominantly bare sediment in water depths of approximately 47 - 50 m and supports a relatively low abundance and diversity of fish assemblages compared to more complex habitats (e.g., reefs). Seabed disturbance from the Activity may temporarily increase the availability of infauna prey for predatory demersal fish, which may result in the short-term attraction of demersal fish to the area.

The GEP and any additionally installed IMMR infrastructure will provide hard substrate to support benthic organisms, such as sponges and gorgonians (see Section 6.4.2.1), creating more complex habitats that may serve as artificial reef. Recent surveys on the North West Shelf have identified increased abundance and species richness of fish associated with subsea pipelines (Bond et al., 2018; McLean et al., 2017). These studies noted that the fish assemblages associated with pipelines tended to have a relatively high portion of large, commercially important fish species that preferred complex habitats (Bond et al., 2018; McLean et al., 2017). The predicted increase in fish assemblages is not expected to have any adverse environmental consequences.

The PMST report identified eleven shark species that may occur in the OA, including seven species listed as threatened under the EPBC Act: the northern river shark (endangered), dwarf sawfish (vulnerable), freshwater sawfish (vulnerable), green sawfish (vulnerable), whale shark (vulnerable, migratory), white shark (vulnerable, migratory), and scalloped hammerhead (conservation dependent) (Appendix C, summarised in Table 3-10). Seabed disturbance from the Activity is unlikely to adversely affect these shark species given their mobile nature, the wide representation of similar habitats in the region, the small size of the OA, and the localised scale of expected seabed disturbance.

Sawfish typically inhabit shallow coastal and estuarine waters and are unlikely to be present in notable numbers given the location and water depth of the OA (47 m - 50 m). However, it is recognised that individuals may occur within the OA, including three species listed as vulnerable under the EPBC Act: the dwarf sawfish, freshwater sawfish, and green sawfish (Appendix C, summarised in Table 3-10). Seabed disturbance from the Activity is unlikely to adversely affect these sawfish species given their mobile nature of sawfish species and preference for shallow habitat, the wide representation of similar habitats in the region, the small size of the OA, and the localised scale of expected seabed disturbance. Additionally, the GEP is expected to become partially or fully buried over time with a low profile above the seabed and is considered unlikely to prevent the movement of sawfish over the pipeline.

#### 6.4.2.3.2 *Marine reptiles*

An internesting BIA and nesting HC for flatback turtles overlaps the OA, while a nesting HC for olive ridley turtles is 11 km from the OA. The southern and western coastline of Bathurst Island hosts flatback, olive ridley and green turtle nesting beaches (Figure 3-9) (Pendoley, 2022). Other species of marine reptiles, such as sea snakes and saltwater crocodiles, are not expected to be present in notable numbers within the OA and are not considered further.

Unlike other turtle populations (for example, on the northwest shelf of WA), the flatback and olive ridley turtles on Bathurst Island do not exhibit discrete nesting and hatching seasons. Rather, there is low level nesting year-round, with a peak in nesting, internesting and hatching during winter months. Flatback turtles forage in soft-bottom sub-tidal environments, feeding opportunistically on a range of benthic invertebrates such as molluscs, crustaceans, soft corals and holothurians, as well as pelagic prey such as jellyfish (Limpus, 2007). Suitable internesting habitat for flatback turtles is defined as water depths shallower than 16 m (Whittock et al., 2016; Pendoley, 2019). Similarly, internesting olive ridley turtles have been shown to remain relatively close to nesting beaches (within 50 km) during the nesting period, in waters typically <30 m depth (Hamel et al., 2008).

As such, internesting flatback and olive ridley turtles are expected to be concentrated in relatively shallow coastal waters (<30 m) around nesting beaches. Given that water depths in the OA are between 47 and 50 m, the wide representation of similar habitats in the region (flatback turtle BIA and HC areas extend > 800 km coastline), and the lack of foraging habitat in the OA, the number of individual turtles likely to be present is expected to be limited. Therefore, seabed disturbance within the OA is unlikely to affect internesting turtles.

#### 6.4.2.3.3 *Marine mammals*

The PMST report identified nine marine mammal species that may occur in the OA, including three species listed as threatened under the EPBC Act: the blue whale (endangered), Sei whale (vulnerable), and fin whale (vulnerable). Several migratory marine mammals may also occur in the OA, including the humpback whale, Bryde's whale, and dolphins (Appendix C, summarised in Table 3-10). The closest significant features are dolphin breeding BIAs (spotted bottlenose, Australian humpback, Australian snubfin), which are over 46 km from the OA (Table 3-12), while the nearest whale BIA (pygmy blue, migration) is 373 km away. Therefore, marine mammals are unlikely to aggregate within the OA. Cetaceans and sirenians may transit the area, but individuals are unlikely to be exposed to the effects of seabed disturbance for durations sufficient to impact biological functions.

Seabed disturbance from the Activity is unlikely to adversely affect marine mammal species given their mobile nature, the wide representation of similar habitats in the region, the small size of the OA, and the localised scale of expected seabed disturbance.



#### 6.4.2.4 Socioeconomic

#### 6.4.2.4.1 Commercial fisheries

Potential impacts to benthic habitats, and subsequently to associated 'fish' species of commercial importance, will be localised and the potential impact to, and displacement of, fish is expected to be insignificant at a stock level. In addition, studies noted that the fish assemblages associated with pipelines tended to have a relatively high portion of large, commercially important fish species that preferred complex habitats (Bond et al., 2018; McLean et al., 2017).

#### 6.4.2.4.2 Underwater cultural heritage

There is one known and protected underwater cultural heritage (UCH) feature within the OA; the *I-124* Japanese submarine wreck. During the design phase of the DPD Project, the pipeline route was deviated to avoid the *I-124* Japanese wreck and its 800 m radial exclusion zone, with the pipeline route passing 100m to the east of the exclusion zone at its closest point.

As per an Archaeological Scope of Works provided by the Heritage Branch of the NT Department of Territory Families, Housing and Communities (DTFHC), Santos engaged the services of a maritime archaeologist to undertake an underwater heritage assessment of the pipeline route, including within NT Coastal Waters (Cosmos Archaeology, 2022).

Along the pipeline corridor, Cosmos Archaeology (2022) analysed data collected during a geophysical survey conducted by Fugro in 2021. The study found three seabed anomalies representing potential cultural objects (i.e. not natural in origin) in the vicinity of the pipeline route, between pipeline kilometre point (KP) 25 to KP 28 (Cosmos Archaeology, 2022). Two of these objects, which could not be determined as natural or cultural, were identified between 143 and 214m away from the pipeline route and another single high-relief feature was located 68m from the pipeline route. This latter anomaly was considered only a remote chance of being associated with the I-124 wreck given its distance of over 2.5km away from the centre point of the wreck (Cosmos Archaeology, 2022). Given the distance of these anomalies from the Barossa GEP route and the nature of IMMR activities, these anomalies are not considered likely to be impacted.

Santos engaged OzArk to conduct a desktop First Nations archaeological assessment for the DPD Project Area, based on a detailed geomorphological assessment, including the NT Coastal Waters (Section 3.2.14.7). No known First Nations UCH sites were identified by OzArk (2024).

#### 6.4.2.4.3 Cultural features

Santos has applied for, and received on 23 December 2022, an Authority Certificate (C2022-098) from the Aboriginal Areas Protection Authority (AAPA), which covers potential seabed disturbance along the pipeline route in NT waters and a nominal 1,000m buffer each side of the pipeline route, including the pipeline route in the NT Coastal Waters OA. There were no registered or recorded sacred sites, protected under the *Northern Territory Aboriginal Sacred Sites Act 1989* (NT) (NTASS Act), identified in the OA, nor any specific certificate conditions related to activities within the OA.

No First Nations people feedback was provided about potential seabed impacts to any geographically specific cultural features during consultation (refer to Section 4.8). The potential impacts to tangible cultural features from seabed disturbance are likely to be associated with any direct or indirect impacts to culturally significant marine fauna habitat and species (refer to Section 6.4.2.3).

Items raised in the 2022 Statement of Reasons Requests from the Tiwi clan members included traditional hunting of marine species and totem species. There is no known traditional hunting or gathering areas within OA, however as the spatial extent is undefined it may be possible traditional hunting and gathering could occur within the OA. Section 6.4.2.3 assesses the potential direct or indirect impacts to culturally significant marine fauna species such as dreaming / songs and totem species (i.e. marine mammals, marine reptiles, sawfish, sharks, rays and other fish). Consequently, it is anticipated that the proposed seabed and benthic habitat disturbance is unlikely to impact traditional hunting practices or resources.

As presented in Section 3.2.15, some First Nations people cultural beliefs place significance on culturally important spiritual beings and the protection they afford First Nations communities from natural disasters and sickness. Dr Corrigan concluded that both the Tiwi Islanders and Larrakia Peoples' cultural and spiritual values within the OA are geographically indeterminate (Corrigan, 2024), based on the materials able to be considered. As part of his study, Dr Corrigan spoke directly with, and obtained information from, many First Nations people including Larrakia, Tiwi Islanders and members of the Belyuen community. Engagement with Tiwi Islanders undertaken by Dr Corrigan also shows that spiritual beings (e.g., crocodile man and Ampitji) are not widely thought to travel to and within the OA due to the distance from the Tiwi Islands, as expressed by some relevant and senior Tiwi people. Of direct relevance to this GEP Coastal Waters OEMP, these sorts of Tiwi cultural and spiritual values were tested in



the Federal Court and were found not to be consistently spread amongst relevant Tiwi Islanders, and in any event, do not represent a particular 'place' of cultural and spiritual significance<sup>46</sup>.

As presented in Section 3.2.15, some First Nations people believe that damaging songlines may have the potential to interfere with ability for First Nations people to reproduce cultural knowledge and continue to provide cultural education of their children.

During consultation for the operation of the Barossa pipeline, the Croker Island people did not identify any sacred sites or songlines within the OA, and no objections or claims were raised.

Santos recognises that some First Nations people remain concerned about the potential for adverse consequences to First Nations people and natural environment, that may arise as a result of disturbance from GEP operations to spiritual dreaming and culturally important spiritual beings. Santos understands the spiritual protection believed to be afforded to the First Nations people is broadly maintained by protecting the features of the natural environment and through ceremonial practices alerting the spiritual beings to the presence of people travelling through country and the like (Corrigan, 2023).

For the DPD project, Dr Corrigan (2024) documented input from Larrakia people and relevant First Nations persons from Belyuen and Wagait, who advised the presence of a range of ancestral beings and dreaming stories of relevance to the Darwin Harbour, surrounding seas and the DPD Project footprint. None of these cultural features are known to be associated with any specific or particular places in the GEP footprint, but rather have a more general association with the wider area, as well as having associations with particular and specific places outside of the GEP footprint.

While Dr Corrigan's assessment concluded that there are no places of specific cultural heritage value in existence in the general area along the GEP route, Santos recognises the general heritage values and cultural and spiritual beliefs and connections to First Nations people.

Santos has determined that the Activity will have low impact and risk to cultural and/or spiritual beliefs because:

- no specific UCH places have been identified by Dr Corrigan, which is consistent with the conclusions of consultation with First Nations people and through the examination of relevant records in the course of preparing this GEP Coastal Waters OEMP
- these intangible cultural and spiritual heritage interests and connections have co-existed with other seabed disturbance activities in the region (including the area surrounding the Tiwi Islands) with no evidence to support actual adverse effects from the actions of spiritual beings in response to impacts on the environment. Regional activities that disturb the seabed include fish trawling, drilling of nearly 900 offshore wells, and subsea infrastructure placement including the Bayu-Undan pipeline (since approximately 2006), the Ichthys Pipeline (since approximately 2016), the North West Cable System (since approximately 2016) and the GEP (since 2023).
- on the views of some Tiwi Islanders who provided information to Dr Corrigan, there were no cultural impediments to the laying of the GEP
- even when considering the most concerned views of Tiwi Islanders (those expressed by the EDO's clients), the impact and risk is expected to be low as the Activity will not meaningfully add to the current level of seabed disturbance in the area.

Santos considers that control measures based on Dr Corrigan's recommendations will allow intangible impacts and risks to be reduced to ALARP and an acceptable level and has adopted these recommendations as BAO-CM-6.1.4.

#### 6.4.3 Environmental performance outcomes and control measures

The EPOs relating to this event include:

- Seabed disturbance to be limited to planned activities and impacts described as part of the Activity and will not occur outside the Operational Area (EPO-04)
- No anchoring or mooring of vessels on shoals/banks (EPO-05)
- No significant impact to cultural features from the Activity (EPO-19)
- No impacts to underwater cultural heritage from the Activity (EPO-20)

An assessment of the environmental benefits and the potential costs or issues associated with control measures for this Activity are shown in Table 6-20 to demonstrate the potential impacts from this aspect are ALARP. Control

<sup>&</sup>lt;sup>46</sup>The concepts of places, sites and similar are used in various places of legislation that contain mechanisms to protect First Nations cultural heritage (including the ATSIHP Act, ALR Act and NTASS Act), to describe specific items or places that should be protected.


measures that are adopted have associated EPSs and measurement criteria that are presented in Table 8-2. Rejected control measures have an ALARP evaluation provided to justify their rejection.

Table 6-20: Control measures	s evaluation for	seabed and ben	thic habitat disturbation	ance
------------------------------	------------------	----------------	---------------------------	------

CM reference	Control measure	Environmental benefit	Potential cost/issues	Evaluation
Standard control mea	sures			
BAO-CM-6.4.5	HSE inductions will include all applicable environmental requirements (administrative control)	Ensures that crew are aware of the stringent Coastal Waters OEMP, Santos and legislative requirements.	Administrative costs to update existing Santos procedure and induction materials, and train personnel.	Adopted – benefits considered to outweigh costs.
BAO-CM-6.5.2	Maintain a subsea infrastructure inventory (administrative control)	Enables Santos to fulfil future decommissioning and removal responsibilities.	Cost of surveys, maintaining equipment and records.	Adopted – benefits considered to outweigh costs.
BAO-CM-6.5.4	Vessels will not anchor under routine operations within the OA however may anchor during emergency conditions	No planned vessel anchoring within the OA reduces seabed disturbance area as no anchor or anchor chain drag/placement will occur.	Cost of contracting activity vessels with DP equipment. Using DP requires continuous engagement of thrusters, which will increase noise emission.	Adopted - environmental benefits of ensuring vessels are compliant outweigh the associated costs.
Additional control me	asures			
BAO-CM-6.1.4	Inductions for all site- based workforce will include information on cultural heritage to raise awareness about the cultural and spiritual belief of First Nations people (administrative control)	Addresses concerns raised (during consultation for the Barossa Project construction activities) of some First Nations people about the potential impacts of the Activity on their spiritual beliefs in a culturally appropriate manner.	Cost to engage First Nations representatives to perform cultural ceremony. Administrative cost to deliver cultural heritage training.	Adopted – benefits considered to outweigh costs.
BAO-CM-6.5.3	Span correction procedures to be developed, if required (administrative control)	Provides clear direction on how spans shall be rectified and surveyed to minimise seabed disturbance.	Costs are expected as part of standard procedure.	Adopted - benefits considered to outweigh costs
N/A	Eliminate IMMR activities, including rectification and stabilisation	Eliminates seabed disturbance from IMMR activities, such as subsea infrastructure replacement, and rectification and stabilisation activities. IMMR is typically undertaken on identified scour and subsea infrastructure movement. Span rectification and stabilisation activities would further limit	Eliminating IMMR may result in more severe environmental impacts (e.g., hydrocarbon leak), compromising safety requirements. Regular IMMR of subsea infrastructure cannot be eliminated, as it is a requirement to maintain subsea equipment and property in good condition, in accordance with PSL Act Section 98.	Not adopted – increased (transferred) risk disproportionate to environmental benefit.

I

CM reference	Control measure	Environmental benefit	Potential cost/issues	Evaluation
		seabed disturbance from scour.		
N/A	Monitor seabed and benthic habitats surrounding subsea infrastructure	Some limited environmental benefit (e.g., data collection) from monitoring benthic habitat.	Costs associated with collecting and reviewing footage.	<b>Not adopted</b> – cost outweighs environmental benefit.
N/A	Use divers for subsea inspections instead of ROV	Reduces seabed disturbance from ROV use and temporary placement of ROV on the seabed.	The use of divers to inspect subsea infrastructure may present unacceptable health and safety risks.	Not adopted – increased (transferred) risk disproportionate to environmental benefit.

### 6.4.4 Environmental impact assessment

Receptors	Consequence level
Seabed and benthic habi	tat disturbance
Physical environment and habitat	Localised sediment (silty, shelly sand) disturbance and turbidity caused by the Activity is expected to be minor in nature and limited to within the OA. Infrastructure placed on the seabed will provide hard substrate for benthic organisms, which may support a localised increase in biodiversity.
	I herefore, the consequence level is considered to be II – Minor.
Threatened, Migratory or local fauna	Given the limited scale of seabed disturbance and location of the OA, potential impacts to threatened, migratory or local fauna species are unlikely. Habitat modification is identified as a potential threat to several marine fauna species in relevant recovery plans and conservation advice (Table 3-13). However, the benthic habitat within the OA is well represented in the wider surrounds and there are no known significant marine fauna BIAs within the OA.
	Marine invertebrates that may inhabit disturbed soft sediment benthic habitats are expected to occur elsewhere within the OA and surrounds. Therefore, the disturbance is not expected to negatively affect prey availability for protected fauna species.
	Seabed disturbance is not expected to cause a significant decrease in local population size, area of occupancy of species, loss or disruption of critical habitat, and disruption to the breeding cycle of any threatened or migratory marine fauna. Therefore, the consequence level is considered to be I – Negligible.
Threatened ecological communities	Not applicable – no threatened ecological communities were identified in the area where seabed disturbance could occur
Protected areas	Not applicable – no protected areas were identified in the area where seabed disturbance could occur
Socioeconomic receptors	The consequence level for Socioeconomic receptors is considered to be I – Negligible.
	Given the relatively small scale of seabed disturbance and knowledge of the existing environment, significant impacts to threatened/migratory/local marine fauna species are not expected. Seabed disturbance is not expected to impact commercial fisheries based on the small size of disturbance compared with the total available fishing area.
	There are no known heritage sites or clear evidence of shipwrecks or aircraft wrecks within the OA.
	There is one UCH site, the Japanese submarine wreck <i>I-124</i> , located <1 km away from the OA and protected under the UCH Act. Santos has deviated the pipeline route around the <i>I-124</i> 800 m radial exclusion zone, with the closest point passing 100 m east of the exclusion zone, to ensure seabed disturbance does not encroach into this zone. A maritime archaeology assessment (Cosmos Archaeology, 2022), assessing geophysical data collected along the route, did not identify any cultural heritage objects likely to be impacted by seabed disturbance associated with the Activity within the OA.

Receptors	Consequence level
Cultural features	There are no sacred sites registered or recorded under the NTASS Act or protected under the ATSIHP Act, UCH Act, ALR Act or EPBC Act that overlap the OA. All of the culturally important sites (including underwater sites) identified by First Nations peoples are outside the OA.
	For the assessment of impacts to marine species of cultural significance, refer to the assessment for threatened, migratory or local fauna.
	In relation to seabed disturbance, Santos notes that existing subsea infrastructure has previously been placed on the seabed in the region, such as the Bayu-Undan pipeline (since approximately 2006), the Ichthys Pipeline (since approximately 2016), and the North West Cable System (since approximately 2016). The region also has a history of significant historic and ongoing industrial shipping, fish trawling and drilling of almost 900 offshore wells. There is no evidence to support actual adverse effects from spiritual beings in response to impacts on people or the environment from these activities. Notwithstanding, a control measure (BAO-CM-6.1.4) relating to cultural heritage training has been adopted. Santos considers the adoption of EPO-19, EPO-20, BAO-CM-6.1.4 and Table 8-2, practicable and appropriate.
Overall worst-case consequence	II – Minor

### 6.4.5 Demonstration of as low as reasonably practicable

There are no reasonably practicable alternatives for conducting regular subsea IMMR activities. All reasonably practicable control measures were reviewed and those adopted are considered appropriate to manage the impacts such that the residual consequence is assessed to be II – Minor. The proposed control measures are in accordance with Santos' risk management criteria and are considered appropriate to manage the impacts to ALARP.

In relation to spiritual and/or cultural heritage beliefs and connections to sea country and related concerns of some First Nations people, Santos has been implementing cultural heritage training and ceremony while undertaking activities authorised pursuant to the GEP EP since November 2023, with broad support of First Nations communities as a culturally appropriate practice and response to cultural concerns. Santos considers that the adopted control measure BAO-CM-6.1.4 will reduce environmental impacts and risks to ALARP, as relevant to First Nations individuals who hold these concerns in relation to their beliefs.

### 6.4.6 Acceptability evaluation

Is the consequence ranked as I or II?	Yes – maximum consequence to seabed and benthic habitats is II – Minor.
Is further information required to validate the consequence assessment?	No – potential impacts and risks are well understood through the information available. Extensive marine studies have been completed within the OA to inform the assessment.
Are the risks and impacts consistent with the principles of ecological sustainable development (ESD)?	Yes – Activity evaluated in accordance with Santos' Offshore Division Environmental Hazard Identification and Assessment Guideline, which considers principles of ESD.

Have the acceptable levels of impact and risks been informed by relevant species recovery plans, threat abatement plans, conservation advice, wildlife conservation plans and Australian marine park zoning objectives?	<ul> <li>Yes – while several plans identify habitat modification as a threat to marine fauna, significant impacts are not predicted for this Activity.</li> <li>Conservation advice: <ul> <li>Approved Conservation Advice for <i>Pristis clavata</i> (Dwarf Sawfish) (DEWHA, 2009)</li> <li>Approved Conservation Advice for Green Sawfish (DEWHA, 2008c)</li> <li>Approved Conservation Advice for <i>Pristis pristis</i> (largetooth sawfish) (TSSC, 2014b)</li> <li>Approved Conservation Advice for <i>Glyphis garricki</i> (northern river shark) (TSSC, 2014a)</li> </ul> </li> </ul>		
	(TSSC, 2015g)		
	Recovery plans:		
	<ul> <li>Recovery Plan for the White Shark (<i>Carcharodon carcharias</i>) (Department of Sustainability, Environment, Water, Population and Communities (DSEWPaC, 2013)</li> </ul>		
	Sawfish and River Sharks Multispecies Recovery Plan (CoA, 2015b)		
	For the identified plans, the objectives of those plans are achieved through the adoption of performance outcomes and the control measures outlined in Section 6.4.3. Santos considers that the level of impact of Activity seabed and benthic habitat disturbance is not inconsistent with these recovery plans.		
Are performance outcomes, control measures and associated performance	Yes – through acceptance of this GEP Coastal Waters OEMP, legislative and regulatory requirements will be met as per Appendix B.		
regulatory requirements?	On 6 December 2023 and 10-11 January 2024, DCCEEW UCH Branch— responsible for administering the UCH Act—was consulted regarding the notification and management of potential UCH for the SURF EP. Feedback on BAO-CM-6.1.4 was affirmative and as a result also adopted for this GEP Coastal Waters OEMP.		
	The GEP Coastal Waters OEMP is also compliant with the conditions of EPBC approval EPBC 2022/09372.		
Are performance outcomes, control measures and associated performance standards consistent with Santos' Environment, Health and Safety Policy?	Yes – aligns with Santos' Environment, Health and Safety Policy (Appendix A).		
Are performance outcomes, control measures and associated performance standards consistent with industry standards?	Yes – the most recent and comparable EPs accepted by NOPSEMA have been reviewed for consistency with the performance outcomes, control measures and associated performance standards proposed in this GEP Coastal Waters OEMP.		
Have performance outcomes, control measures and associated performance standards taken into consideration Relevant Person feedback?	Yes – no objections or claims were specifically raised by Relevant Persons for this Activity. However, feedback received from the Corrigan 2024 Report, GEP EP and D&C EP has been considered, and where applicable, additional EPOs, CMs and EPSs were adopted (EPO-19, EPO-20 and BAO-CM-6.1.4).		
Are performance standards such that the impact or risk is considered to be ALARP?	Yes – ALARP assessment conducted, with additional control measures adopted.		

The consequence of seabed and benthic habitat disturbance is assessed as II – Minor. Based on an assessment of Santos' acceptability criteria and with the control measures in place, potential impacts are considered acceptable.



### 6.5 Interactions with other marine users

	-
Event	Sources of impact to other marine users may occur as a result of:
	• planned vessel activities in the OA
	<ul> <li>unplanned and non-routine vessel activities in the OA (e.g., post major cyclone inspections)</li> </ul>
	ROVs and AUVs operating within the OA
	<ul> <li>the ongoing presence of the GEP and associated seabed infrastructure in the OA which may pose a snag hazard)</li> </ul>
	Activity vessels may include:
	IMMR vessels, which operate 24/7 for a period during the Activity
	<ul> <li>support and supply vessels, which will transit to and from the IMMR vessel</li> </ul>
	<ul> <li>ad-hoc survey and other support vessels including USVs operated by a Vessel Master from a remote operations centre.</li> </ul>
	Other marine users within the OA may include commercial shipping and fishing, tourism (including fishing charters), recreational fishing, defence, and traditional fishing.
Extent	Contained within the OA. A temporary 500 m exclusion zone will be established around the Activity vessels to safeguard them while they are unable to manoeuvre. All Activity vessels will be limited to ≤8 knots within the OA.
Duration	Continuous:
	Continuous presence of the GEP on the seabed throughout field life.
	Intermittent:
	Temporary and intermittent interaction with third-party vessels and helicopters when transiting the OA for the duration of the field life
	<ul> <li>Support vessel presence is required for day-to-day operations during routine IMMR. The duration of IMMR vessel presence will be approximately 7 to 30 days, every three to five years Vessels for unplanned activities, including USV, would be less frequent.</li> </ul>

### 6.5.1 Description of event

### 6.5.2 Nature and scale of environmental impacts

**Potential receptors:** socioeconomic (commercial fisheries, traditional fishing, tourism, recreation, shipping, and defence).

Other marine users may be inhibited by the presence of Activity vessels and a 500 m radius exclusion zone that is maintained around the vessels during temporary operations in the OA, such as IMMR. The GEP inspection frequency will be every three to five years, and findings of the IMMR campaigns will be used to inform the future frequencies of IMMR activities. Activity vessels involved in routine IMMR are expected to operate for a period of weeks during these regularly scheduled periods. Helicopter operations within the OA will be short-term and are unlikely to interfere with other marine users due to the access restrictions around Activity vessels.

The physical presence of the GEP in the OA may also present a hazard to marine users due to the potential for snagging. Potential interactions with other marine users are described below.

### 6.5.2.1 Commercial fishing

The management area of ten commercial fisheries (four Commonwealth, six NT) overlap the OA. Table 3-15 provides a summary of commercial fisheries and Santos' understanding of fishing effort based on publicly available information and consultation with Relevant Persons. However, as described in Section 3.2.14.1 no active commercial fishing occurs in five of the fisheries that overlap the OA. Four commercial fisheries may actively operate in the OA: the Northern Prawn Fishery (NPF), Spanish Mackerel Fishery (SMF), Offshore Net and Line Fishery (ONLF), and the Demersal Fishery (DF). In the NPF, medium and high fishing effort is concentrated to the west and north of the Tiwi Islands, as well as south of the OA. Fishing effort in the SMF is concentrated at nearby shoals and banks within the EMBA, particularly in waters off Bathurst Island. In the ONLF, fishing effort is concentrated to the north-east of the OA, along the eastern boundary of the Timor Reef fishery in water depths of 80-100 m.

### 6.5.2.2 Traditional subsistence fishing

The OA is approximately 30 km south-west of the Tiwi Islands. First Nations fishing activity in NT waters predominately occurs within inshore tidal waters, with approximately 93% of effort concentrated within coastal



waters up to 3 NM (approximately 5.6 km) from the NT coast and the Tiwi Islands (Section 3.2.15.9). In Tiwi Island waters traditional fishing effort is greatest near the larger communities of Wurrumiyanga on Bathurst Island, and Pirlangimpi and Milikapiti on Melville Island (DPIF 2014). Tiwi people continue to undertake the customary harvesting of sea turtles and dugongs. Green turtles are the main species harvested, while eggs of all turtle species are taken periodically (Tiwi Land Council, 2022). Dugongs are occasionally harvested. In 2014, the Blue Mud Bay Settlement Deed was signed by NT Government, Tiwi Land Trust, and the Tiwi Land Council. Tiwi People are proposing to establish a Marine Indigenous Protected Area that extends to 3 NM around the Tiwi Islands (Tiwi Land Council, 2021).

### 6.5.2.3 Tourism and recreation

Tourism and recreational vessels may transit the area infrequently. Tourism, recreation, and traditional fishing are not expected in the OA given the distance to the Tiwi Islands (~30 km) and Darwin (~95 km), the lack of seabed features, and water depths between 47 and 50 m. The seabed within the OA is characterised as silty, shelly sand, with very sparse (<1%) epibiota and no known seabed sites of recreational interest, such as fishing areas, shipwrecks or coral reefs. The closest areas that support site attached fish and a varied benthic community are Shepparton Shoal (23 km from the OA) and Afghan Shoal (24 km from the OA).

### 6.5.2.4 Defence, infrastructure, and commercial shipping

The OA intersects a designated defence practice area. The closest operational offshore production facilities and infield subsea infrastructure are the Eni operated Blacktip Gas, approximately 250 km south-west from the OA, and the Santos operated Bayu–Undan platform, approximately 370 km north-west from the OA. There are two existing pipelines within the vicinity: Bayu-Undan (runs adjacent to the GEP) and Ichthys (18 km distant from the OA).

The presence of Activity vessels may cause temporary disruption to commercial shipping. Darwin Port is a major shipping port located 95 km southeast of the OA. In 2022–2023, there were 1,569 vessel calls to port (Landbridge Darwin Port, 2024). Additionally, Port Melville in the Tiwi Islands is located 83 km north-east of the OA and 125 km north of Darwin. Based on AMSA vessel traffic data, commercial shipping and other marine traffic in the OA is expected to be relatively low (AMSA, 2024).

### 6.5.3 Environmental performance outcomes and control measures

The EPO relating to this event is:

- No vessel collisions or adverse interactions with other marine users (EPO-01).
- No significant impacts to cultural features from the Activity (EPO-19)
- No impacts to underwater cultural heritage from the Activity (EPO-20)

An assessment of the environmental benefits and the potential costs or issues associated with control measures for this activity are described in Table 6-21 to demonstrate the potential impacts from this aspect are ALARP. Control measures that are adopted have associated EPSs and measurement criteria and are presented in Table 8-2. Rejected control measures have an ALARP evaluation provided to justify their rejection.

### Table 6-21: Control measures evaluation for interaction with other marine users

CM reference	Control measure	Environmental benefit	Potential cost/issues	Evaluation
Standard control meas	sures			
BAO-CM-6.1.2	Vessels equipped and crewed in accordance with Australian maritime requirements, including Marine Order 30 (Prevention of Collisions) and Marine Order 21 (Safety and Emergency Arrangements) (administrative control)	Ensures contracted vessels are operated, maintained, and crewed in accordance with industry standards and regulatory requirements. Ensures vessels meet Marine Assurance Standards to reduce the likelihood of vessel collision (such as minimum and working lighting for maritime safety).	Regulatory requirement and therefore the cost is not identified as an issue.	Adopted – it is a regulatory requirement.

CM reference	Control measure	Environmental benefit	Potential cost/issues	Evaluation
BAO-CM-6.4.5	HSE inductions will include applicable environmental requirements (administrative control)	Ensures that crew are aware of the stringent Coastal Waters OEMP, Santos, and legislative requirements.	Administrative costs to update existing Santos procedures and induction materials, and train personnel	Adopted – benefits considered to outweigh costs.
BAO-CM-6.6.1	Notify AHS and AMSA MSI prior to relevant Activity (administrative control)	Maritime notifications ensure identified marine users are informed of the proposed activities, reducing the likelihood of unplanned interactions. Subsea infrastructure will be clearly marked on Australian nautical charts published by the Australian Hydrographic Office (AHO) alerting other marine users to the presence of Activity vessels and exclusion zones and restrictions, thus reducing the likelihood of vessel collision and fishing gear snagging.	Cost and time to perform notifications.	Adopted – benefits considered to outweigh costs.
BAO-CM-6.6.4	Activity undertaken in accordance with Santos HSE management and marine vessel vetting processes (Santos' Offshore Marine Assurance Procedure) (administrative control)	Santos marine vetting process ensures vessel lighting, radios and equipment are inspected and maintained so that other marine users are aware of the vessel's physical presence, thus reducing the potential for interaction and collision.	Standard maritime safety and navigational equipment; regulatory requirement and therefore the cost is not identified as an issue.	Adopted – benefits considered to outweigh costs and regulatory requirements mandate some equipment standards.
Additional control mea	asures			
BAO-CM-6.6.6	Vessel speed restrictions within 500m around the IMMR vessels and campaign vessels (administrative control)	Restricting vessel speeds within the OA to ≤8 knots reduces the likelihood and consequence (causing harm) of vessel-to- vessel collisions by providing vessels with more time to detect and maneuver to avoid each other.	Administrative costs to update existing Santos procedures and induction materials, and train personnel.	Adopted – benefits considered to outweigh costs.
BAO-CM-6.6.7	Communications plan will be implemented for engagement prior to and during the Activity that may impact marine users to raise awareness of the activity (administrative control)	Communications plan will improve awareness of the Activity, encourage engagement with stakeholders, and provide up-to-date information regarding key activities.	Cost associated with implementing procedures	Adopted – benefits considered to outweigh costs.

CM reference	Control measure	Environmental benefit	Potential cost/issues	Evaluation
BAO-CM-6.6.8	Charting of infrastructure on nautical charts	Subsea infrastructure clearly marked on Australian nautical charts published by the AHO alerts other marine users to the presence of activity vessels and exclusion zones and restrictions, thus reducing the likelihood interactions with marine users such as fishing gear snagging.	Cost and time to prepare and distribute information	Adopted - benefits considered to outweigh costs
BAO-CM-6.1.4	Inductions for all site- based workforce will include information on cultural heritage to raise awareness about the cultural and spiritual belief of First Nations people	Addresses concerns raised (during consultation for the Barossa Project construction activities) of some First Nations people about the potential impacts of the Activity on their spiritual beliefs in a culturally appropriate manner.	Cost to engage First Nations representatives to perform cultural ceremony. Administrative cost to deliver cultural heritage training.	Adopted - benefits considered to outweigh costs
N/A	Manage the timing of the Activity to avoid peak marine user periods (e.g., fishing) (elimination control)	Would reduce potential impacts to fisheries and other marine users	Significant costs and increase in Activity duration to demobilise/remobilise the vessels. It also increases the risk profile of the operation.	Not adopted – marine users may be present within the OA at any time of the year. Given this, avoiding peak fishing periods is not considered justified, and is disproportionate to the cost and delay it would cause.
N/A	Installation of protection structures on pipeline infrastructure (engineering control)	Protects trawling vessels against loss of containment from subsea infrastructure.	Installation of protection structures for the pipeline to protect trawling vessels against loss is disproportionate to the cost and delay it would cause.	Not adopted – During design and installation Santos has undertaken trenching and rock dumping along zones of the pipeline route where there is a higher risk of dragged or dropped anchors. An engineering based risk assessment demonstrates that fishing equipment would fail before the pipeline was damaged in the event that trawling gear became

### 6.5.4 Environmental impact assessment

Receptor	Consequence level	
Interaction with other mari	ne users	
Threatened, migratory or local fauna	Not applicable – related to socioeconomic receptors only	



Receptor	Consequence level
Physical environment or habitat	
Threatened ecological communities	
Protected areas	
Socioeconomic receptors	The OA does not overlap any areas of moderate to high fishing effort. This is primarily due to the seabed being characterised as silty, shelly sand, with very sparse (<1%) epibiota and a lack of seabed features such as shoals, reefs and banks. Given the low fishing effort within the OA and the distance from the coastline, interactions with commercial fishers are likely limited to fishers transiting through the region. While 500 m exclusion zones around Activity vessels may impose minor restrictions on where fishing effort can occur, no substantial adverse effects are considered likely given the small OA and the short-term nature of typical IMMR activities. Therefore, the impacts and risks are deemed acceptable.
	The physical presence of subsea infrastructure may present an ongoing hazard to marine users due to the potential for snagging. The control measures outlined in Table 6-21, together with the engineering design of the GEP and support structures (proud to the seabed) prevents snagging of trawling equipment. The risk of snagging was assessed during a fishing interactions survey undertaken for the Darwin Pipeline Duplication (DPD) (Intecsea, 2018). Based on the frequency of trawling vessels crossing the pipeline and the location of snagging hazards (e.g., pipeline spanning structures), it was concluded that there is very low likelihood of trawling equipment becoming snagged on the installed pipeline. On an ongoing basis, it is not credible for any snagging of trawling equipment to result in a loss of containment of the GEP infrastructure (Intecsea, 2018; 2022).
	Based on AMSA vessel traffic data, commercial shipping and other marine traffic in the OA is expected to be relatively low (AMSA, 2024), as the OA is not located in a shipping fairway. Given all shipping and Activity vessels are required to comply with the COLREGS and associated Marine Orders, it is expected navigational and communicative aids are sufficient to prevent any negative interactions beyond basic avoidance of Activity vessels. Activity vessels are typically only present in the OA for infrequent, short-duration IMMR activities. During these periods, should other vessels need to deviate from planned routes to avoid 500 m exclusion zones, it is unlikely to increase transit times and fuel consumption. Therefore, impacts to commercial shipping or fishing vessels are not expected.
	Tourism, recreation, and traditional fishing are not expected in the OA given the distance to the Tiwi Islands (~30 km) and Darwin (~95 km), the lack of existing seabed sites of interest (e.g., fishing areas, shipwrecks, or coral reefs), and water depths between 47 and 50 m. Any interactions with recreational or traditional fishers, scuba diving operators, or tourism vessels are expected to be limited to temporary avoidance of Activity vessels while transiting through the OA. Based on Santos' experience with the Bayu-Undan pipeline, the GEP may become a site of future interest once marine growth becomes established and attracts fish and fishers. Given the small OA and temporary duration of exclusion zones around the Activity vessels, interactions with commercial fishing, shipping, defence, and other incidental marine traffic are expected to be infrequent. Other marine users have previously coexisted in the OA with other nearby restricted areas in place (e.g., defence areas). Communication before and during the Activity will reduce the likelihood of unplanned interactions with other commercial marine users. Therefore, the consequence level for potential interaction with other marine users is considered
Overall worst-case	to be I – Negligible.
consequence	

### 6.5.5 Demonstration of as low as reasonably practicable

There are no alternatives to using vessels to undertake the Activity. Activity vessels must have a 500 m exclusion zone in place to ensure the safety of these vessels and other marine users. Santos' consultation process is described in Section 4. Throughout the consultation period, Relevant Persons were made aware of the proposed exclusion zones around the Activity vessels, and the implications to other marine users including the indicative schedule. No concerns have been raised by Relevant Persons regarding the potential exclusion zone. Notice to Mariners will be issued that detail the location, nature of activities, and confirmation that vessels will maintain navigation aids.

All reasonably practicable control measures were reviewed and those adopted are considered appropriate to manage the impacts such that the residual consequence is assessed to be I – Negligible. The proposed control measures are in accordance with Santos' risk management criteria and are considered appropriate to reduce impacts to ALARP.



### 6.5.6 Acceptability evaluation

Is the consequence ranked as I or II?	Yes – maximum consequence from interaction with other marine users is I – Negligible.		
Is further information required to validate the consequence assessment?	No – potential impacts and risks are well understood through the information available and Relevant Person consultation.		
Are the risks and impacts consistent with the principles of ecologically sustainable development (ESD)?	Yes – Activity evaluated in accordance with Santos' Offshore Division Environmental Hazard Identification and Assessment Guideline, which considers principles of ESD.		
Have the acceptable levels of impact and risks been informed by relevant species recovery plans, threat abatement plans and conservation advice and Australian marine park zoning objectives?	Not applicable. The OA does not intersect any AMP or protected areas. Interaction with other marine users is not a relevant threat identified in the species recovery plans, threat abatement plans, conservation advice and wildlife conservation plans set out in Table 3-13.		
Are performance outcomes, control measures and associated performance standards consistent with legal and regulatory requirements?	Yes – management measures are consistent with the SOLAS and various Commonwealth Acts, such as the <i>Marine Safety (Domestic Commercial</i> <i>Vessel) National Law Act 2012, the Navigation Act 2012.</i> Through acceptance of this GEP Coastal Waters OEMP, legislative and regulatory requirements will be met as per Appendix B. The GEP Coastal Waters OEMP is also compliant with the conditions of EPBC approval EPBC 2022/09372.		
Are performance outcomes, control measures and associated performance standards consistent with Santos' Environment, Health and Safety Policy?	Yes – aligns with Santos' Environment, Health and Safety Policy (Appendix A).		
Are performance outcomes, control measures and associated performance standards consistent with industry standards?	Yes – the most recent and comparable EPs accepted by NOPSEMA were reviewed for consistency with the performance outcomes, control measures and associated performance standards proposed in this GEP Coastal Waters OEMP.		
Have performance outcomes, control measures and associated performance standards taken into consideration Relevant Person feedback?	Yes – Relevant Person feedback indicated no recommendations for revising the EPO, CMs or EPSs.		
Are performance standards such that the impact or risk is considered to be ALARP?	Yes – ALARP assessment conducted, with additional control measures adopted.		

The consequence of interaction with other marine users is assessed as I – Negligible. Based on an assessment of Santos' acceptability criteria and with the control measures in place, potential impacts are considered acceptable.



### 6.6 **Operational discharges**

### 6.6.1 Description of event

**Event** 

Within NT waters, including the NT Coastal Waters OEMP OA, there are restrictions on permissible vessel discharges to the marine environment under the *Navigation Act 2012* (Cth), *Protection of the Sea (Prevention of Pollution from Ships) Act 1983* (Cth) (and subordinate Marine Orders), *Marine Pollution Act 1999* (NT) and *Marine Pollution Regulations 2003* (NT). Collectively, these instruments give effect to various MARPOL Annexes, including Annexes I, II, III, IV and V relating to oil, noxious liquids, harmful packaged substances, sewage and garbage).

Potential impacts may occur in the OA from near-surface IMMR vessel discharges, where permitted under relevant legislation.

Near-surface discharges from Activity vessels

- deck drainage/run-off
- treated sewage and greywater
- cooling water
- bilge water
- brine (if a reverse osmosis unit is used for water treatment)
- ballast water

The Activity vessels are the largest and primary vessels that may operate within the OA. IMMR, support and supply vessels will also operate in the OA. Vessel presence in the OA will be short-term during IMMR campaigns. As a result, it is reasonable to base the worst-case vessel discharge calculations on total maximum persons on board (POB) of 100 across all IMMR vessels operating in the OA concurrently.

#### **Deck drainage**

Deck drainage discharge may potentially contain small residual quantities of oil, grease and detergents if present or used on the decks. Assessment of the unplanned release of hydrocarbons and other environmentally hazardous liquids is discussed in Section 7.

#### Treated sewage and greywater

The volume of sewage and greywater is directly proportional to the number of POB. Up to 30–40 L of sewage/greywater may be generated per person per day and is discharged in accordance with Marine Order 96 and applicable legislation. The estimated maximum sewage and greywater discharged is up to4,000 L/day.

#### **Cooling water**

Sea water will be used as a heat exchange medium for cooling machinery engines. Sea water is drawn from the ocean and flows counter current through closed-circuit heat exchangers, transferring heat from engines and machinery to the sea water. The sea water is then discharged to the ocean (i.e., it is a once-through system). Cooling water temperatures may vary depending on engine workload and activity.

### Bilge water

While in the OA, vessels may discharge oily bilge water after treatment to less than 15 mg/L oil in water content via an approved oily water filter system, in accordance with Marine Order 91 and applicable legislation.

#### Brine

Brine generated from the water supply systems on each vessel will be discharged to the ocean at a salinity of approximately 10% higher than sea water. The volume of the discharge depends on the requirement for fresh (potable) water and will vary between vessels and the number of POB.

The effluent may contain scale inhibitors to control inorganic scale formation, such as the formation of calcium carbonate and magnesium hydroxide in water-making plants. Other water purification and plant cleaning chemicals may be used and discharged to sea after the cleaning process is completed.

#### **Ballast water**

Ballast water could potentially be discharged to the marine environment from vessel ballast tanks. Refer to Section 7.2 for the ballast water risk assessment.

Note: Vessel firefighting systems will not be tested when within the OA.

#### **Discharges from IMMR activities**

Chemicals and residual hydrocarbons may be released during IMMR activities.

Extent The small volumes of vessel discharges may cause localised nutrient enrichment, organic and particulate loading, ecotoxicological effects, and increased water temperature and salinity around discharge points and in the direction of the prevailing current. The environment that may be affected by operational discharges is likely to be within approximately 50 m

of the Activity vessel and contained within the OA.

Duration IMMR vessel discharges will occur periodically across the duration of the Activity. IMMR operations will occur for approximately 7 to 30 days in duration every three to five years, or as needed. Discharges will



result in localised changes to water quality; however, conditions will likely return to normal within minutes to hours after discharges cease.

### 6.6.2 Nature and scale of environmental impacts

**Potential receptors**: physical environment (water quality, benthic habitats), threatened, migratory or local fauna (marine mammals, marine turtles, seabirds, rays, sharks and other pelagic fish); socioeconomic and cultural features.

### 6.6.2.1 Physical environment

Small volumes of vessel discharges will be released to the marine environment and result in a reduction in water quality. Discharges will be temporary (minutes to hours), localised and limited to surface waters. The discharges are expected to disperse and dilute rapidly.

Specifics of potential impacts to water quality from vessel discharges are as follows.

### 6.6.2.1.1 Eutrophication impacts from sewage and greywater

Discharges of treated sewage and greywater can result in localised increases in nutrient concentrations (e.g., ammonia, nitrate, nitrite, and orthophosphate), organics (e.g., volatile and semi-volatile organic compounds, oil and grease, phenols and endocrine-disrupting compounds), and inorganics (e.g. hydrogen sulphide, metals and metalloids, surfactants, phthalates and residual chlorine). Nutrient inputs may facilitate localised increases in phytoplankton, while organic carbon inputs may facilitate localised increases in bacterial activity. This could subsequently impact higher order predators.

However, the discharges are expected to disperse and dilute rapidly in the OA given the water depths, strong currents and wave action. There may be a localised and temporary (minutes to hours) reduction in water quality in the immediate vicinity of the release. However, low volume and short duration discharges are expected to rapidly disperse below levels that would cause adverse impacts.

The organic components of discharges biodegrade through bacterial action, oxidation and evaporation. In a study of sewage discharge in deep ocean waters, Parnell (2003) reported no appreciable differences in the inorganic nutrient levels between the outfall area and background concentrations, suggesting rapid uptake of nutrients and/or rapid dispersion and dilution within hours of discharge.

### 6.6.2.1.2 Salinity increases

The desalination of sea water results in a discharge of brine with a slightly elevated salinity (around 10% higher than sea water). When discharged to the sea, the desalination brine, being of greater density than sea water, is expected to sink and rapidly disperse in the currents. The volume of the discharge depends on the requirement for fresh (potable) water and the number of POB.

Most marine species can tolerate short-term fluctuations in salinity around 20–30% (e.g., Walker and McComb, 1990). It is expected that most pelagic species would be able to tolerate short-term exposure to minor increases in salinity caused by discharged brine.

### 6.6.2.1.3 Changes in temperature

Cooling water will be discharged at a temperature above that of the ambient sea water. Upon discharge it will be subjected to turbulent mixing and transfer of heat to the surrounding waters. Cooling water discharge to the marine environment may result in a localised and temporary increase in the ambient water temperature, which could cause alteration of the physiological processes (particularly enzyme-mediated processes) in marine biota (Wolanski, 1994).

Cooling water discharge points vary for each vessel. However, they are all designed to discharge cooling water to above the water line to help cool and oxygenate the wastewater stream before it mixes with the surrounding sea water.

### 6.6.2.1.4 Contamination from releases of bilge water

Discharged oily water is treated to an OIW limit of 15 ppm in accordance with MARPOL Annex I (as enacted by Australian legislation) and Marine Order 91. Discharges of oily bilge water could result in a localised reduction in water quality and has the potential to create an oil sheen in surface waters, with impacts on protected marine fauna and plankton. Small amounts of dissolved hydrocarbons in treated oily water drainage may be toxic to marine organisms. However, potential impacts are considered unlikely due to the low concentrations of oil and grease residues in oily water drainage, together with the water depths, strong currents and wave action in the OA.



### 6.6.2.1.5 *Toxicity*

Discharges from vessels may include typical chemicals used within standard maritime sewage systems, desalination systems, and residues of those used for cleaning decks. Discharges are expected to be intermittent and similar to other permitted discharges from vessels.

Chemicals and hydraulic fluids will be risk-assessed and selected for their discharge suitability and low toxicity and bioaccumulation potential using the Santos Operations Chemical Selection Evaluation and Approval Procedure. Upon discharge, these types of chemicals (e.g., MEG and methanol) are expected to disperse and dilute rapidly in the OA given the water depths, strong currents and wave action. There may be a localised and temporary (minutes to hours) reduction in water quality in the immediate vicinity of the release. However, low volume and short duration discharges are expected to rapidly disperse below levels that would cause adverse impacts to marine fauna.

Residual hydrocarbons and inert gasses (e.g., methane) may be infrequently discharged during IMMR; however, are not readily water soluble and will rise rapidly through the water column and release into the atmosphere.

### 6.6.2.1.6 Chemical use during contingency GEP repair activities

Leak testing of the GEP may occur if contingency repair activities are required, in which case a small volume (estimated <5 L) of non-toxic dye may also be used to assist in visually detecting leaks in the GEP by ROV. These discharges would only occur in the event of an unplanned repair activity.

As described in Section 2.5.2, residual hydrocarbons and residual inert gas may be present in the GEP and discharged to the marine environment during contingency repair activities. The isolated equipment will be at ambient seabed pressure; therefore, any inert gas will be displaced through natural seawater ingress into the equipment.

All chemicals that are planned for discharge to the environment will be selected in accordance with Santos' Offshore Division Operations Chemical Approval Procedure to ensure that environmentally acceptable products are used, or that risks from the use of other chemicals are demonstrated to be ALARP (Section 2.7).

### 6.6.2.2 Threatened, migratory or local fauna

As discussed in the sections above, the extent of impact for planned discharges is localised, and rapid dilution is predicted to occur within the OA. An internesting buffer for flatback turtles (BIA and habitat critical) overlaps the OA. Due to the water depths in the OA (47 - 50 m), the BIA extending across more than 800 km of coastline, and a lack of foraging habitat, the potential numbers of affected internesting turtles is expected to be limited to a small number of individuals. Marine fauna within the OA, some of which may have cultural significance as totems (such as marine mammals, marine reptiles, rays and fish) or cultural food sources, are likely to be transient. If contact does occur with marine fauna, it will be for a short duration and likely not sufficient to cause a toxic effect.

Discharges may cause changes to the behaviour of marine fauna (i.e., avoidance or attraction). However, such discharges would be infrequent and no prolonged influence on fauna behaviour is expected.

### 6.6.2.3 Cultural features

No First Nations people feedback was provided about potential impacts from vessel discharges to cultural features during consultations for the Activity (refer to Section 4.8). The potential impacts to culturally significant marine fauna species (such as dreaming and totem species including marine mammals, marine reptiles, rays, other fish and birds) are assessed in Section 6.6.2.2.

Tiwi clan members raised concerns—during consultation on the D&C EP and in the 2022 Statement of Reasons Requests—regarding potential impacts from the Drilling Activity on totemic species and culturally significant marine species that provide a food source for traditional fishing and hunting. The potential impact to marine fauna is likely to be limited to localised, temporary behavioural impacts and is unlikely to result in significant impacts to marine species at the individual or population level (Section 6.6.2.2). As a result, Activity discharges are not anticipated to affect traditional hunting practices or resources.

In addition, some Tiwi people informed Santos that impacts to totemic species could also affect Tiwi people by making them sick. First Nations people maintain a continuing spiritual connection with sea country, including marine fauna species with cultural significance, such as totems or as a cultural food source.

### 6.6.3 Environmental performance outcomes and control measures

The EPOs relating to this event include:

 No injury, or mortality to, EPBC Act 1999 listed marine fauna or marine fauna listed as threatened species under the Territory Parks and Wildlife Conservation Act 1976 (NT) as a result of the Activity (EPO-08)



- Planned discharges will meet relevant maritime obligations and Santos chemical assessment and approval process (EPO-14)
- Zero unplanned discharge of hazardous and non-hazardous wastes into the marine environment from the Activity (EPO-16)
- Zero unplanned discharge of hydrocarbons or chemicals to the marine environment from the Activity (EPO-18)

An assessment of the environmental benefits and the potential costs or issues associated with control measures for this Activity are described in Table 6-22 to demonstrate that the potential impacts from this aspect are ALARP. Control measures that are adopted have associated EPSs and measurement criteria and are presented in Table 8-2. Rejected control measures have an ALARP evaluation provided to justify their rejection.

Table 6-22: Control	measures ev	aluation for	operational	discharges

CM reference	Control measure	Environmental benefit	Potential cost/issues	Evaluation
Standard contr	ol measures			
BAO-CM-6.4.5	HSE inductions will include all applicable environmental requirements (administrative control)	Ensures that crew are aware of the stringent Coastal Waters OEMP, Santos and legislative requirements.	Administrative costs to update existing Santos procedure and induction materials, and train personnel.	Adopted - benefits considered to outweigh costs.
BAO-CM- 6.4.10	Barossa Facilities and vessels planned maintenance system to confirm equipment integrity is maintained in accordance with manufacturers guidelines. (administrative control)	Ensures vessel and equipment integrity is maintained through routine checks.	Cost of implementing procedures.	Adopted – benefit of assuring vessels outweighs procedure compliance costs.
BAO-CM-6.7.2	Routine discharges of treated sewage and grey water, in accordance with Marine Order 96 (Marine Pollution Prevention – Sewage)	Managing treated sewage and grey water discharges to applicable legislative requirements ensures no substantial change in water quality will occur.	MARPOL requirement.	Adopted – environmental benefits of ensuring vessels are compliant outweigh the potential costs. Discharge of treated sewage is a permissible maritime discharge.
BAO-CM-6.7.3	Deck cleaning product selection according to MARPOL Annex V (and Marine Order 93: Noxious liquid) (substitution control)	Ensures deck cleaning products are not harmful to the marine environment.	Cost associated with implementing the procedure. Limits deck cleaning products available for use.	Adopted – environmental benefits of ensuring vessels are compliant outweigh the potential costs.
BAO-CM-6.7.5	Apply the Santos chemical selection process for all chemicals planned to be discharged (Section 2.7) (administrative control)	Under the procedure, CHARM-rated gold/silver and non- CHARM Group E/D chemicals managed under the OCNS, or OSPAR PLONOR list, or chemicals risk assessed by Santos and deemed	Cost of implementing procedures. Range of chemicals reduced with potentially higher costs for alternative products.	Adopted – environmental benefits of using environmentally acceptable chemicals outweigh procedural implementation and operational costs.

CM reference	Control measure	Environmental benefit	Potential cost/issues	Evaluation
		environmentally acceptable, will be selected (Section 2.7). Therefore, the chemicals will pose little or no risk to the environment.		
BAO-CM-6.7.6	Routine discharges of treated bilge and deck water from vessels will comply with Marine Order 91 and <i>Marine</i> <i>Pollution Act 1999</i> (NT), as applicable (administrative control)	Managing bilge and deck drainage discharges to applicable legislative requirements ensures no substantial change in water quality will occur. Ensures vessel oily water is treated and discharged in accordance with MARPOL Annex I (and Marine Order 91: Marine pollution prevention – oil).	MARPOL requirement.	Adopted - environmental benefits of ensuring vessels are compliant outweigh the potential costs. The proposed discharge is a permissible maritime discharge.
BAO-CM- 6.7.11	Contractor contingency pipeline preservation procedure and specification (administrative control)	<ul> <li>This control is effective in reducing potential impacts from contingency pipeline preservation activities by:</li> <li>selecting a seawater treatment product that is Gold rated through OCNS CHARM rating or through a pseudo CHARM rating, which is recognised as the least environmentally hazardous chemical rating</li> <li>Minimising the concentration of treatment chemicals required for the required preservation period</li> </ul>	Cost of implementing procedures.	Adopted
Additional cont	trol measures			
BAO-CM- 6.7.12	Contractor contingency pipeline major repair procedure to be developed in the event a major repair is required (administrative control)	This control effectively reduces the likelihood of an unplanned treated seawater release during the pipeline repair activities.	Cost of implementing procedures.	Adopted



CM reference	Control measure	Environmental benefit	Potential cost/issues	Evaluation
N/A	Zero discharge of sewage from vessels (elimination control)	Eliminates treated sewage from being discharged to sea.	Significant health risks from storing sewage onboard. Costs associated with containment and onshore disposal of sewage. Storing sewage would create an additional hazard for working on deck.	<b>Not adopted</b> – health and safety considerations outweigh the environmental benefit. Any discharge will occur in accordance with applicable legislation.
N/A	Zero discharge of bilge water from vessels (elimination control)	Would eliminate treated oily water from being discharged to sea.	Issues include vessel stability comprised, potential fire hazard and flooding risk.	<b>Not adopted</b> – safety and environmental considerations outweigh the environmental benefit. Any discharge of bilge water will comply with applicable legislative requirements
NA	Capture or eliminate use of chemicals used during IMMR (elimination control)	Eliminates or reduces the chemicals discharged to the marine environment.	Capturing chemicals used during IMMR is not practical. Chemicals are assessed to ensure the discharge is environmentally acceptable in accordance with the Santos chemical selection process. Excessive use of chemicals is restricted.	Not adopted – safety and operational considerations outweigh the environmental benefit, given small volumes and low toxicity of the discharges.
N/A	Zero discharge of deck water from vessels (elimination control)	Would eliminate potential contaminants being discharged to sea.	Increased safety risks from inadequate draining of wet decks. Large amounts of water on a vessel's deck can also cause stability issues (free surface effect).	<b>Not adopted</b> – safety considerations outweigh the environmental benefit. Any discharge will occur in accordance with applicable legislation
N/A	Zero discharge of cooling water from vessels (elimination control)	Eliminates potential impacts of cooling water (elevated temperature) being discharged to sea.	Technically not an available option, given the volumes of cooling water that would need to be stored on the vessels to meet operational cooling water needs.	<b>Not adopted</b> – not technically feasible to operate vessels without cooling water. Any discharge will occur in accordance with applicable legislation.
N/A	Restrict use of desalination plant; or zero discharge of brine water from vessels (administrative control)	Would eliminate or reduce brine from being discharged to sea.	Cost associated with transporting freshwater offshore. Health risks associated with limited supply of freshwater. Costs associated with containment and onshore disposal of brine. Storage of brine would create an additional hazard for working on deck.	Not adopted – health and safety considerations outweigh the environmental benefit. Use of 'water making' system and discharge of waste brine will comply with applicable legislative requirements
N/A	Mandatory closed drain system on vessels (administrative control)	Would eliminate untreated deck drainage from being discharged to sea.	Increased cost of the treatment system and vessel modification requirements.	<b>Not adopted</b> – costs significantly outweigh the environmental benefit given the minor impacts expected from planned discharges.



### 6.6.4 Environmental impact assessment

Receptor	Consequence level			
Operational discharges				
Physical environment or habitat	Vessel discharges are predicted to quickly dilute and disperse in the marine environment. Impacts of eutrophication and changes in salinity and temperature are expected to be localised and of short duration. Discharge of chemicals may cause be a localised and temporary (minutes to hours) reduction in water quality in the immediate vicinity of the release. However, low volume and short duration discharges are expected to rapidly disperse below levels that would cause adverse impacts. Any impacts on water quality will be restricted to surface waters only and have no impact on seabed receptors.			
	Impacts from chemicals and fluids discharged near the seabed (for example, from IMMR activities) are also expected to be localised and of short duration. It is expected that discharges would have a negligible impact on benthic habitats.			
	Given the nature and limited volumes of planned discharges, the high levels of dilution, and the nature of the marine environment in the OA and surrounds, the consequence level for physical environment or habitat is considered to be II – Minor.			
Threatened, migratory or local fauna	Sensitive receptors that may be impacted include plankton, fish at the sea surface, marine turtles, marine mammals, and seabirds. Discharges will be temporary and impacts to water quality will be localised, with recovery measured in hours (no sustained impacts).			
	An internesting buffer for flatback turtles (BIA and habitat critical) overlaps the OA. However, marine fauna within the OA are likely to be transient, and if contact does occur, it will likely be for a short duration and not sufficient to cause a toxic effect.			
	Discharges may cause changes to the behaviour of marine fauna (i.e., avoidance or attraction). However, as discharges will be infrequent and small volume, no prolonged influence is expected.			
	Consequently, impacts are expected to be short-term with no decrease in local population size, the area of occupancy of species, loss or disruption of critical habitats, or disruption to breeding cycles.			
	Given the nature and limited volumes of planned vessel discharges, the high levels of dilution, and the nature of the marine environment in the OA and surrounds, the consequence level for threatened, migratory or local fauna is considered to be II – Minor.			
Threatened ecological communities	Not applicable – no threatened ecological communities were identified in the area over which vessel discharges are expected.			
Protected areas	Not applicable – no protected areas were identified in the area over which vessel discharges are expected.			
Socioeconomic receptors	Given the controls in place to manage vessel discharges in accordance with regulatory requirements, impacts to marine fauna (including targeted fishery species) are not expected. Given the minor consequence to marine fauna, subsequent impacts to socioeconomic receptors, including commercial fishing and cultural features, are not anticipated.			
	Vessel discharges will be infrequent, relatively small scale and will become highly diluted. Therefore, the consequence to socioeconomic receptors is assessed as I – Negligible.			
Cultural Features	An important outcome of Dr Corrigan's research is that no sacred sites or dreamings are shown to be directly impacted by the DPD project footprint (where this Activity OA is located), however some marine species are known to be associated with dreamings and songs (Corrigan, 2024). It is anticipated that vessel discharges are unlikely to affect traditional hunting practices or resources.			
	For potential impacts to marine species of cultural significance, or that provide a traditional food sources (such as dreaming and totem species including marine mammals, marine reptiles, rays, other fish and birds), refer to the assessment for threatened, migratory or local fauna.			
Overall worst-case consequence	II – Minor			

### 6.6.5 Demonstration of as low as reasonably practicable

There are no reasonably practicable alternatives to using vessels to undertake the Activity. Onboard treatment of most wastes and their subsequent discharge to the marine environment is consistent with legislative requirements (enacting MARPOL) and is considered environmentally acceptable. Given there are no concurrent activities in the OA and the nature and volumes of expected operational discharges, no cumulative impacts can reasonably be expected.

All reasonably practicable control measures were reviewed and those adopted are considered appropriate to manage the impacts such that the residual consequence is assessed to be II – Minor. The proposed control



measures are in accordance with Santos' risk management criteria and are considered appropriate to reduce impacts to ALARP.

### 6.6.6 Acceptability evaluation

Is the consequence tarked as 1 or II?         Yes – maximum planned vessel discharge consequence is rated II – Minor.           Is further information required to validati         No – potential impacts and risks are well understood through the information available.           Are the risks and impacts consistent with the principles of accologically sustainable development (ESD)?         Yes – activity evaluated in accordance with Santos: Offshore Division Environment Hazard Identification and Assessment Guideline, which considers principles of ESD.           Have the acceptable levels of Impact and constraint with the OA identifies habitat degradation/modification and conservation Advices for Pristis clavata (Dwarf Sawlish) (DEWH 2009)           Approved Conservation Advice for Pristis clavata (Dwarf Sawlish) (DEWH 2009)         Approved Conservation Advice for Pristis pristis (targetooth sawfish) (TSS 2014)           Approved Conservation Advice for Pristis pristis (targetooth sawfish) (TSS 2014a)         Approved Conservation Advice for Balaenoptera physalus (In whale) (TSSC, 2015)           Conservation Advice for Balaenoptera physalus (In whale) (TSSC, 2015)         Conservation Advice for Calidris canutus (ed knot) (DCCEEW, 2024c)           Conservation Advice for Calidris canutus (ed knot) (DCCEEW, 2024c)         Conservation Advice for Calidris canutus (ed knot) (DCCEEW, 2024c)           Conservation Advice for Calidris canutus (ed knot) (DCCEEW, 2024c)         Conservation Advice for Calidris canutus (ed knot) (DCCEEW, 2024c)           Conservation Advice for Calidris canutus (ed knot) (DCCEEW, 2024c)         Conservation Advice for Calidris canutus (ed knot) (DCCEEW, 20		
Eventer information required to validated         No-potential impacts and risks are well understood through the information available.           Are the risks and impacts consistent with the principles of ecologically sustainabilic development (ESD)?         Yes - activity evaluated in accordance with Santos' Offshore Division Environmental Hazard Identification and Assessment Guideline, which considers principles of ESD.           Have the acceptable levels of impact and risks been informed by relevant superiors indice principles of experiments of the santom of conservation Advice of Pristis clavata (Dwarf Sawtish) (DEWH 2009)           Pray the park zoning objectives?         Yes - the following material published in relation to threatened and migratory species that may occur within the OA identifies habitat degradation/modification and conservation Advice for Pristis clavata (Dwarf Sawtish) (DEWH 2009)           Approved Conservation Advice for Pristis pristis (largetooth sawtish) (TSS 2014b)         Approved Conservation Advice for Pristis pristis (largetooth sawtish) (TSSC 2015b)           Approved Conservation Advice for Balaenoptera breakis (withale) (TSSC, 2015c)         Conservation Advice for Balaenoptera breakis (Withale) (TSSC, 2015c)           Conservation Advice for Balaenoptera breakis (Eastern Curlew) (TSSC, 2015)         Approved Conservation Advice for Calidris acuminate (sharp-tailed sandpiper) (DCCEEW, 2024b)           Recovery Plan for the White Shark (Carcharodon carcharias) (Department of Sustinability, Environment, Water, Population and Communities (DSEWPAC, 2015)           Sawtish and River Sharks Multipeeies Recovery Plan for the Environment Protection and Biodiversity Conservation Advice for charge will	Is the consequence ranked as I or II?	Yes – maximum planned vessel discharge consequence is rated II – Minor.
Are the risks and impacts consistent with       Yes – activity evaluated in accordance with Santos' Offshore Division         Have the acceptable levels of impact and risks been informed by relevant species recovery plans, threat abatement plans and conservation advice and Australian mame park zoning objectives?       Yes – the following material published in relation to threatened and migratory risks been informed by relevant species recovery plans, threat abatement plans and conservation advice and Australian mame park zoning objectives?       Yes – the following material published in relation to threatened and migratory risks been informed by relevant species recovery plans, threat abatement plans and conservation advices and Australian mame park zoning objectives?         Approved Conservation Advice for Pristis prists (largetooth sawfish) (DEWH 2009)       Approved Conservation Advice for Pristis prists (largetooth sawfish) (TSS 2014b)         Approved Conservation Advice for Bialenoptera physalus (lin whale) (TSSC, 2015b)       Approved Conservation Advice for Rhincodon typus (whale shark) (TSSC, 2015g)         Conservation Advice for Calidris canutus (red knot) (DCCEEW, 2024c)       Conservation Advice for Calidris acuminate (sharp-tailed sandpiper) (DCCEEW, 2024b)         Recovery Plan for the White Shark (Carcharodon carcharias) (Department of Sustainability, Environment, Water, Population and Communities (DSEWPAC, 2013)         Conservation Advice for Calidris acuminate (sharp-tailed sandpiper) (DCCEEW, 2024b)         Recovery Plan for ther White Shark (Mutispecies Recovery Plan of sustainability, Environment, Water, Population and Communities (DSEWPAC, 2013)         Conservation Advice for Calidris acuminate (sha	Is further information required to validate the consequence assessment?	No – potential impacts and risks are well understood through the information available.
Have the acceptable levels of impact and risks been informed by relevant species recovery plans, threat deatement plans and conservation advice and Australian matrine park zoning objectives?       Yes the following material published in relation to threatened and migratory species that may occur within the OA identifies habitat degradation/modificati and pollution as a threat (Table 3-13):         Conservation Advice       - Approved Conservation Advice for Pristis clavata (Dwarf Sawlish) (DEWH 2009)         - Approved Conservation Advice for Green Sawlish (DEWHA, 2008c)         - Approved Conservation Advice for Bible particki (northern river shark) (TSSC, 2014a)         - Approved Conservation Advice for Bible particki (northern river shark) (TSSC, 2014a)         - Approved Conservation Advice for Bibleonoptera physalus (fit whale) (TSSC, 2015b)         - Approved Conservation Advice for Calidris canutus (fite whale) (TSSC, 2015b)         - Conservation Advice for Calidris canutus (red knot) (DCCEEW, 2024c)         - Conservation Advice for Calidris canutus (red knot) (DCCEEW, 2024c)         - Conservation Advice for Calidris canutus (red knot) (DCCEEW, 2024c)         - Conservation Advice for Calidris canutus (red knot) (DCCEEW, 2024c)         - Conservation Advice for Calidris canutus (red knot) (DCCEEW, 2024c)         - Conservation Advice for Calidris acuminate (sharp-tailed sandpiper) (DCCEW, 2024b)         Recovery Plan for the White Shark (Carcharodon carcharias) (Department of Sustainability, Environment, Water, Population and Conservation Act 1999 2015–2025 (COA, 2015a)         - Recovery Pla	Are the risks and impacts consistent with the principles of ecologically sustainable development (ESD)?	Yes – activity evaluated in accordance with Santos' Offshore Division Environmental Hazard Identification and Assessment Guideline, which considers principles of ESD.
<ul> <li>Approved Conservation Advice for <i>Pristic cardia</i> (Drain Cardins) (DEATI 2008)</li> <li>Approved Conservation Advice for <i>Pristis pristis</i> (largetooth sawfish) (TSS 2014b)</li> <li>Approved Conservation Advice for <i>Biphis garricki</i> (northern river shark) (TSSC, 2014g)</li> <li>Approved Conservation Advice for <i>Balaenoptera physalus</i> (fin whale) (TSSC, 2015g)</li> <li>Conservation Advice for <i>Balaenoptera physalus</i> (fin whale) (TSSC, 2015b)</li> <li>Approved Conservation Advice for <i>Balaenoptera physalus</i> (fin whale) (TSSC, 2015b)</li> <li>Conservation Advice for <i>Balaenoptera borealis</i> (sei whale) (TSSC, 2015b)</li> <li>Conservation Advice for <i>Calidris canutus</i> (red knot) (DCCEEW, 2024c)</li> <li>Conservation Advice for <i>Calidris canutus</i> (red knot) (DCCEEW, 2024c)</li> <li>Conservation Advice for <i>Calidris canutus</i> (red knot) (DCCEEW, 2024c)</li> <li>Conservation Advice for <i>Calidris canutus</i> (red knot) (DCCEEW, 2024c)</li> <li>Conservation Advice for <i>Calidris acuminata</i> (sharp-tailed sandpiper) (DCCEEW, 2024b)</li> <li>Recovery Plans</li> <li>Recovery Plan for the White Shark (<i>Carcharodon carcharias</i>) (Departmen of Sustainability, Environment, Water, Population and Communities (DSEWPaC, 2013)</li> <li>Conservation Management Plan for the Blue Whale - A Recovery Plan under the Environment Protection and Biodiversity Conservation Advice (DSEWPaC, 2015b)</li> <li>Recovery Plan for Marine Turtles in Australia 2017–2027 (CoA, 2017b)</li> <li>Sawfish and River Sharks Multispecies Recovery Plan (coA, 2017b)</li> <li>Sawfish and River Sharks Multispecies Recovery Plan (coA, 2017b)</li> <li>Habitat degradation or modification is identified in a number of plans but pertains to more toxic discharges are allowable in accordance with legislation or are of low toxicity.</li> <li>Recovery plans / conseivation advices or other species that may occur in the OA do not tidentify habitat degradation / modification or are of low toxicity.&lt;</li></ul>	Have the acceptable levels of impact and risks been informed by relevant species recovery plans, threat abatement plans and conservation advice and Australian marine park zoning objectives?	Yes – the following material published in relation to threatened and migratory species that may occur within the OA identifies habitat degradation/modification and pollution as a threat (Table 3-13): <b>Conservation Advice:</b>
<ul> <li>Approved Conservation Advice for Green Sawlish (DEWHA, 2008c)</li> <li>Approved Conservation Advice for <i>Pristis pristis</i> (largetooth sawlish) (TSS 2014b)</li> <li>Approved Conservation Advice for <i>Biphis garricki</i> (northern river shark) (TSSC, 2014a)</li> <li>Approved Conservation Advice for <i>Balaenoptera physalus</i> (fin whale) (TSSC, 2015g)</li> <li>Conservation Advice for <i>Balaenoptera physalus</i> (fin whale) (TSSC, 2015b)</li> <li>Approved Conservation Advice for <i>Balaenoptera physalus</i> (fin whale) (TSSC, 2015b)</li> <li>Conservation Advice for <i>Calidris canutus</i> (red knot) (DCCEEW, 2024c)</li> <li>Conservation Advice for <i>Calidris canutus</i> (red knot) (DCCEEW, 2024c)</li> <li>Conservation Advice for <i>Calidris acuminata</i> (sharp-tailed sandpiper) (DCCEEW, 2024b)</li> <li>Recovery Plan for the White Shark (<i>Carcharodon carcharias</i>) (Departmen of Sustainability, Environment, Water, Population and Communities (DSEWPaC, 2013)</li> <li>Conservation Advice or Shark (Carcharodon carcharias) (Departmen of Sustainability, Environment Protection and Biodiversity Conservation Advice (DSEWPaC, 2015a)</li> <li>Recovery Plan for the White Shark (Carcharodon carcharias) (Departmen of Sustainability, Environment Protection and Biodiversity Conservation Advice (DSEWPaC, 2015a)</li> <li>Conservation Advice Sharks Multispecies Recovery Plan (CoA, 2017b)</li> <li>Sawlish and River Sharks Multispecies Recovery Plan (CoA, 2017b)</li> <li>Sawlish and River Sharks Multispecies no conservation advices however the nature of Advity operational discharges with not result in habitat degradation r. Pollution is identified in a number of plans but pertains to more toxic discharges are allowable in accordance with legislation or are of low toxicity. Recovery plans / conservation advices or ontangements. The collectives of the expective set of the stages on the discharges with the objectives of the materi is tablist degra</li></ul>		
<ul> <li>Approved Conservation Advice for <i>Pristis pristis</i> (largetooth sawlish) (TSS 2014b)</li> <li>Approved Conservation Advice for <i>Glyphis garricki</i> (northern river shark) (TSSC, 2014a)</li> <li>Approved Conservation Advice for <i>Balaenoptera physalus</i> (lin whale) (TSSC, 2015g)</li> <li>Conservation Advice for <i>Balaenoptera physalus</i> (lin whale) (TSSC, 2015b)</li> <li>Approved Conservation Advice for <i>Balaenoptera borealis</i> (sei whale) (TSSC, 2015b)</li> <li>Approved Conservation Advice for <i>Balaenoptera borealis</i> (sei whale) (TSSC, 2015b)</li> <li>Approved Conservation Advice for <i>Calidris acumunata</i> (sharp-tailed sandpiper) (DCCEEW, 2024c)</li> <li>Conservation Advice for <i>Calidris acuminata</i> (sharp-tailed sandpiper) (DCCEEW, 2024b)</li> <li>Recovery Plans</li> <li>Recovery Plans for the White Shark (<i>Carcharodon carcharias</i>) (Departmen of Sustainability, Environment, Water, Population and Communities (DSEWPaC, 2013)</li> <li>Conservation Management Plan for the Blue Whale - A Recovery Plan under the Environment Protection and Biodiversity Conservation Act 1999 2015–2025 (CoA, 2015a)</li> <li>Recovery Plan for Marine Turtles in Australia 2017–2027 (CoA, 2017b)</li> <li>Sawfish and River Sharks Multispecies Recovery Plan (coA, 2015b)</li> <li>Habitat degradation or modification is identified in many conservation advices however the nature of Activity operational discharges will not result in habitat degradation. Pollution is identified in a number of plans but pertains to more toxic discharges are allowable in accordance with legislation or are of low toxicity. Recovery plans / conservation Advice for other species that may occur in the OA do not identify habitat degradation / modification is discharges and therefore is no reases remained actions. The objectives of the materi listed above. Santos considers the potential impacts of the Arbection of the Sea (Prevention of Pollution from Ships) Act 1983 (Cth), Marine Pollution as a key thre: or have expl</li></ul>		Approved Conservation Advice for Green Sawfish (DEWHA, 2008c)
<ul> <li>Approved Conservation Advice for <i>Glyphis garricki</i> (northern river shark) (TSSC, 2014a)</li> <li>Approved Conservation Advice for <i>Rhincodon typus</i> (whale shark) (TSSC, 2015g)</li> <li>Conservation Advice for <i>Balaenoptera physalus</i> (fin whale) (TSSC, 2015b)</li> <li>Approved Conservation Advice for <i>Balaenoptera borealis</i> (sei whale) (TSSC, 2015b)</li> <li>Approved Conservation Advice for <i>Calidris canutus</i> (red knot) (DCCEEW, 2024c)</li> <li>Conservation Advice for <i>Calidris canutus</i> (red knot) (DCCEEW, 2024c)</li> <li>Conservation Advice for <i>Calidris acuminata</i> (sharp-tailed sandpiper) (DCCEEW, 2024c)</li> <li>Conservation Advice for <i>Calidris acuminata</i> (sharp-tailed sandpiper) (DCCEEW, 2024d)</li> <li>Recovery Plans for the White Shark (<i>Carcharodon carcharias</i>) (Departmen of Sustainability, Environment, Water, Population and Communities (DESEWPaC, 2013)</li> <li>Conservation Management Plan for the Blue Whale - A Recovery Plan under the Environment Protection and Biodiversity Conservation Advice 920(15–2025 (CoA, 2015a)</li> <li>Recovery Plan for Marine Turtles in Australia 2017–2027 (CoA, 2017b)</li> <li>Sawfish and River Sharks Multispecies Recovery Plan (coA, 2015b)</li> <li>Habitat degradation or modification is identified in many conservation advices however the nature of Activity operational discharges will not result in habitat degradation. Pollution is identified in a number of plans but pertains to more toxic discharges and therefore is not considered applicable here given the discharges are allowable in a cordance with legislation or are of low toxicity. Recovery Plans / conservation advices for management actions. The objectives or management actisk assessments. The controls or plans tor plans but pertains to</li></ul>		<ul> <li>Approved Conservation Advice for <i>Pristis pristis</i> (largetooth sawfish) (TSSC, 2014b)</li> </ul>
<ul> <li>Approved Conservation Advice for Rhincodon typus (whale shark) (TSSC, 2015g)</li> <li>Conservation Advice for Balaenoptera physalus (fin whale) (TSSC, 2015c)</li> <li>Conservation Advice for Balaenoptera borealis (sei whale) (TSSC, 2015b)</li> <li>Approved Conservation Advice for Numenius madagascariensis (Eastern Curlew) (TSSC, 2015f)</li> <li>Conservation Advice for Calidris canutus (red knot) (DCCEEW, 2024c)</li> <li>Conservation Advice for Calidris canutus (red knot) (DCCEEW, 2024c)</li> <li>Conservation Advice for Calidris acuminate (sharp-tailed sandpiper) (DCCEEW, 2024b)</li> <li>Recovery Plans for the White Shark (Carcharodon carcharias) (Departmen of Sustainability, Environment, Water, Population and Communities (DSEWPaC, 2013)</li> <li>Conservation Management Plan for the Blue Whale - A Recovery Plan under the Environment Protection and Biodiversity Conservation Act 1999 2015–2025 (CoA, 2015a)</li> <li>Recovery Plan for Marine Turtles in Australia 2017–2027 (CoA, 2017b)</li> <li>Sawfish and River Sharks Multispecies Recovery Plan (CoA, 2015b)</li> <li>Habitat degradation or modification is identified in anumber of plans but pertains to more toxic discharges are allowable in accordance with legislation or are of low toxicity. Recovery plans / conservation advices or ontagement actions. The objectives of the satery explicit relevant objectives or management actions. The objectives of these publications were considered during impact and risk assessments. The control soutined in Table 6-22 are consistent with the objectives of the matery of these publications were considered during impact and risk assessments. The control soutined in Table 6-22 are consistent with the objectives of the materie listed above. Santos considers the potential impacts from vessel discharges to be consistent with the objectives of the materie listed above. Santos considers the potential mater of the Protection of the Sea (Prevention of Pollution from Ships) Act 1983 (Cth),</li></ul>		<ul> <li>Approved Conservation Advice for <i>Glyphis garrick</i>i (northern river shark) (TSSC, 2014a)</li> </ul>
<ul> <li>Conservation Advice for Balaenoptera physalus (fin whale) (TSSC, 2015c)</li> <li>Conservation Advice for Balaenoptera borealis (sei whale) (TSSC, 2015b)</li> <li>Approved Conservation Advice for Numenius madagascariensis (Eastern Cultew) (TSSC, 2015)</li> <li>Conservation Advice for Calidris canutus (red knot) (DCCEEW, 2024c)</li> <li>Conservation Advice for Calidris acuminata (sharp-tailed sandpiper) (DCCEEW, 2024b)</li> <li>Recovery Plans:</li> <li>Recovery Plans for the White Shark (Carcharodon carcharias) (Department of Sustainability, Environment, Water, Population and Communities (DSEWPAC, 2013)</li> <li>Conservation Management Plan for the Blue Whale - A Recovery Plan under the Environment Protection and Biodiversity Conservation Act 1999 2015–2025 (CoA, 2015a)</li> <li>Recovery Plan for Marine Turtles in Australia 2017–2027 (CoA, 2017b)</li> <li>Sawfish and River Sharks Multispecies Recovery Plan (COA, 2015b)</li> <li>Habitat degradation or modification is identified in many conservation advices however the nature of Activity operational discharges will not result in habitat degradation. Pollution is identified in a number of plans but pertains to more toxic discharges are allowable in accordance with legislation or are of low toxicity. Recovery plans / conservation advices or management actions. The objectives of these publications were considered auplicable here given the discharges are allowable in Table 6-22 are consistent with the objectives of the materi listed above. Santos considers the potential impacts from vessel discharges to be consistent with the objectives of the materi listed above. Santos considers the potential impacts from vessel discharges to be consistent with these objectives.</li> </ul>		<ul> <li>Approved Conservation Advice for <i>Rhincodon typus</i> (whale shark) (TSSC, 2015g)</li> </ul>
<ul> <li>Conservation Advice for Balaenoptera borealis (sei whale) (TSSC, 2015b)</li> <li>Approved Conservation Advice for Numenius madagascariensis (Eastern Curlew) (TSSC, 2015f)</li> <li>Conservation Advice for Calidris canutus (red knot) (DCCEEW, 2024c)</li> <li>Conservation Advice for Calidris canutus (red knot) (DCCEEW, 2024c)</li> <li>Conservation Advice for Calidris acuminata (sharp-tailed sandpiper) (DCCEEW, 2024b)</li> <li>Recovery Plans:</li> <li>Recovery Plans for the White Shark (Carcharodon carcharias) (Department of Sustainability, Environment, Water, Population and Communities (DSEWPaC, 2013)</li> <li>Conservation Management Plan for the Blue Whale - A Recovery Plan under the Environment Protection and Biodiversity Conservation Act 1999 2015–2025 (CoA, 2015a)</li> <li>Recovery Plan for Marine Turtles in Australia 2017–2027 (CoA, 2017b)</li> <li>Sawfish and River Sharks Multispecies Recovery Plan (CoA, 2015b)</li> <li>Habitat degradation or modification is identified in many conservation advices however the nature of Activity operational discharges will not result in habitat degradation. Pollution is identified in a number of plans but pertains to more toxic discharges are allowable in accordance with legislation or are of low toxicity. Recovery plans / conservation advices for these publications were considered during impacts from vessel discharges to consistent with the objectives of these publication sufficient on a pollution as a key three or have explicit relevant objectives or management actions. The objectives of these publications were considered during impacts from vessel discharges to be consistent with these objectives.</li> </ul>		Conservation Advice for Balaenoptera physalus (fin whale) (TSSC, 2015c)
<ul> <li>Approved Conservation Advice for Numenius madagascariensis (Eastern Curlew) (TSSC, 2015f)</li> <li>Conservation Advice for Calidris canutus (red knot) (DCCEEW, 2024c)</li> <li>Conservation Advice for Calidris acuminata (sharp-tailed sandpiper) (DCCEEW, 2024b)</li> <li>Recovery Plans</li> <li>Recovery Plans</li> <li>Recovery Plans for the White Shark (<i>Carcharodon carcharias</i>) (Departmenn of Sustainability, Environment, Water, Population and Communities (DSEWPaC, 2013)</li> <li>Conservation Management Plan for the Blue Whale - A Recovery Plan under the Environment Protection and Biodiversity Conservation Act 1999 2015–2025 (CoA, 2015a)</li> <li>Recovery Plan for Marine Turtles in Australia 2017–2027 (CoA, 2017b)</li> <li>Sawfish and River Sharks Multispecies Recovery Plan (CoA, 2015b)</li> <li>Habitat degradation or modification is identified in many conservation advices however the nature of Activity operational discharges will not result in habitat degradation. Pollution is identified in a number of plans but pertains to more toxic discharges are allowable in accordance with legislation or are of low toxicity.</li> <li>Recovery plans / conservation advice for other species that may occur in the OA do not identify habitat degradation / modification or pollution as a key three or have explicit relevant objectives or management actions. The objectives of these publications were considered during impact and risk assessments. The controls outlined in Table 6-22 are consistent with the objectives of these publications were considered during impact and risk assessments. The controls outlined in Table 6-22 are consistent with the objectives of these publication of Pollution from Ships) Act 1983 (Cth), Marine Pollution Act 1996 (NT), Navigation Act 2012 (Cth), and the following Marine Orders and is enasted by:</li> <li>Marine Order 91 (Marine pollution prevention – oil)</li> <li>Marine Order 95 (Marine pollution prevention – oil)</li> </ul>		Conservation Advice for <i>Balaenoptera borealis</i> (sei whale) (TSSC, 2015b)
<ul> <li>Conservation Advice for Calidris canutus (red knot) (DCCEEW, 2024c)</li> <li>Conservation Advice for Calidris acuminata (sharp-tailed sandpiper) (DCCEEW, 2024b)</li> <li>Recovery Plans:</li> <li>Recovery Plan for the White Shark (Carcharodon carcharias) (Departmen of Sustainability, Environment, Water, Population and Communities (DSEWPaC, 2013)</li> <li>Conservation Management Plan for the Blue Whale - A Recovery Plan under the Environment Protection and Biodiversity Conservation Act 1999 2015–2025 (CoA, 2015a)</li> <li>Recovery Plan for Marine Turtles in Australia 2017–2027 (CoA, 2017b)</li> <li>Sawfish and River Sharks Multispecies Recovery Plan (coA, 2015b)</li> <li>Habitat degradation or modification is identified in annay conservation advices however the nature of Activity operational discharges will not result in habitat degradation. Pollution is identified in a number of plans but pertains to more toxic discharges are allowable in accordance with legislation or are of low toxicity. Recovery plans / conservation advice for other species that may occur in the OA do not identify habitat degradation / modification or pollution as a key three or have explicit relevant objectives or management actions. The objectives of these publications were considered during impact and risk assessments. The controls outlined in Table 6-22 are consistent with the objectives of the materilised above. Santos considers the potential impacts from vessel discharges to be consistent with these objectives.</li> <li>Vessel discharges comply with the requirements of the <i>Protection of the Sea</i> (<i>Prevention of Pollution Act 2012</i> (Cth), and the following Marine Orders and is enacted by:</li> <li>Marine Order 91 (Marine pollution prevention – oil)</li> <li>Marine Order 95 (Marine pollution prevention – garbage)</li> </ul>		<ul> <li>Approved Conservation Advice for Numenius madagascariensis (Eastern Curlew) (TSSC, 2015f)</li> </ul>
<ul> <li>Conservation Advice for Calidris acuminata (sharp-tailed sandpiper) (DCCEEW, 2024b)</li> <li>Recovery Plan for the White Shark (Carcharodon carcharias) (Departmen of Sustainability, Environment, Water, Population and Communities (DSEWPaC, 2013)</li> <li>Conservation Management Plan for the Blue Whale - A Recovery Plan under the Environment Protection and Biodiversity Conservation Act 1999 2015–2025 (CoA, 2015a)</li> <li>Recovery Plan for Marine Turtles in Australia 2017–2027 (CoA, 2017b)</li> <li>Sawfish and River Sharks Multispecies Recovery Plan (coA, 2015b) Habitat degradation or modification is identified in many conservation advices however the nature of Activity operational discharges will not result in habitat degradation. Pollution is identified in a number of plans but pertains to more toxic discharges are allowable in accordance with legislation or are of low toxicity. Recovery plans / conservation advice for other species that may occur in the OA do not identify habitat degradation / modification or pollution as a key threa or have explicit relevant objectives or management actions. The objectives of these publications were considered during impact and risk assessments. The controls outlined in Table 6-22 are consistent with the objectives of the materilised above. Santos considers the potential impacts from vessel discharges to be consistent with these objectives.</li> </ul>		Conservation Advice for Calidris canutus (red knot) (DCCEEW, 2024c)
Recovery Plans:• Recovery Plan for the White Shark (Carcharodon carcharias) (Departmen of Sustainability, Environment, Water, Population and Communities (DSEWPaC, 2013)• Conservation Management Plan for the Blue Whale - A Recovery Plan under the Environment Protection and Biodiversity Conservation Act 1999 2015–2025 (CoA, 2015a)• Recovery Plan for Marine Turtles in Australia 2017–2027 (CoA, 2017b)• Sawfish and River Sharks Multispecies Recovery Plan (CoA, 2015b) Habitat degradation or modification is identified in many conservation advices however the nature of Activity operational discharges will not result in habitat degradation. Pollution is identified in a number of plans but pertains to more discharges are allowable in accordance with legislation or are of low toxicity. Recovery plans / conservation advices for other species that may occur in the OA do not identify habitat degradation / modification or pollution as a key three of have explicit relevant objectives or management actions. The objectives of these publications were considered during impact and risk assessments. The controls outlined in Table 6-22 are consistent with the objectives of the materi listed above. Santos considers the potential impacts from vessel discharges to be consistent with these objectives.Are performance outcomes, control measures and associated performance standards consistent with legal and regulatory requirements?Vessel discharges comply with the requirements of the Protection of the Sea (Prevention of Pollution from Ships) Act 1983 (Cth), Marine Orders and is enacted by: • Marine Order 91 (Marine pollution prevention – oil) • Marine Order 95 (Marine pollution prevention – oil)		<ul> <li>Conservation Advice for Calidris acuminata (sharp-tailed sandpiper) (DCCEEW, 2024b)</li> </ul>
<ul> <li>Recovery Plan for the White Shark (<i>Carcharodon carcharias</i>) (Departmen of Sustainability, Environment, Water, Population and Communities (DSEWPaC, 2013)</li> <li>Conservation Management Plan for the Blue Whale - A Recovery Plan under the Environment Protection and Biodiversity Conservation Act 1999 2015–2025 (CoA, 2015a)</li> <li>Recovery Plan for Marine Turtles in Australia 2017–2027 (CoA, 2017b)</li> <li>Sawfish and River Sharks Multispecies Recovery Plan (CoA, 2015b)</li> <li>Habitat degradation or modification is identified in many conservation advices however the nature of Activity operational discharges will not result in habitat degradation. Pollution is identified in a number of plans but pertains to more toxic discharges and therefore is not considered applicable here given the discharges are allowable in accordance with legislation or are of low toxicity. Recovery plans / conservation advice for other species that may occur in the OA do not identify habitat degradation / modification or pollution as a key three or have explicit relevant objectives or management actions. The objectives of the materil listed above. Santos considered during impacts from vessel discharges to be consistent with these objectives.</li> </ul>		Recovery Plans:
<ul> <li>Conservation Management Plan for the Blue Whale - A Recovery Plan under the Environment Protection and Biodiversity Conservation Act 1999 2015–2025 (CoA, 2015a)</li> <li>Recovery Plan for Marine Turtles in Australia 2017–2027 (CoA, 2017b)</li> <li>Sawfish and River Sharks Multispecies Recovery Plan (CoA, 2015b) Habitat degradation or modification is identified in many conservation advices however the nature of Activity operational discharges will not result in habitat degradation. Pollution is identified in a number of plans but pertains to more toxic discharges are allowable in accordance with legislation or are of low toxicity. Recovery plans / conservation advices or these publications were considered during impact and risk assessments. The controls outlined in Table 6-22 are consistent with the objectives of these publications were considered during impact and risk assessments. The controls outlined in Table 6-22 are consistent with the objectives of the set of baove. Santos considers the potential impacts from vessel discharges to be consistent with these objectives.</li> </ul>		<ul> <li>Recovery Plan for the White Shark (<i>Carcharodon carcharias</i>) (Department of Sustainability, Environment, Water, Population and Communities (DSEWPaC, 2013)</li> </ul>
<ul> <li>Recovery Plan for Marine Turtles in Australia 2017–2027 (CoA, 2017b)</li> <li>Sawfish and River Sharks Multispecies Recovery Plan (CoA, 2015b)</li> <li>Habitat degradation or modification is identified in many conservation advices however the nature of Activity operational discharges will not result in habitat degradation. Pollution is identified in a number of plans but pertains to more toxic discharges are allowable in accordance with legislation or are of low toxicity. Recovery plans / conservation advice for other species that may occur in the OA do not identify habitat degradation / modification or pollution as a key three or have explicit relevant objectives or management actions. The objectives of these publications were considered during impact and risk assessments. The controls outlined in Table 6-22 are consistent with the objectives of the materil listed above. Santos considers the potential impacts from vessel discharges to be consistent with these objectives.</li> </ul>		<ul> <li>Conservation Management Plan for the Blue Whale - A Recovery Plan under the Environment Protection and Biodiversity Conservation Act 1999 2015–2025 (CoA, 2015a)</li> </ul>
<ul> <li>Sawfish and River Sharks Multispecies Recovery Plan (CoA, 2015b) Habitat degradation or modification is identified in many conservation advices however the nature of Activity operational discharges will not result in habitat degradation. Pollution is identified in a number of plans but pertains to more toxic discharges and therefore is not considered applicable here given the discharges are allowable in accordance with legislation or are of low toxicity. Recovery plans / conservation advice for other species that may occur in the OA do not identify habitat degradation / modification or pollution as a key three or have explicit relevant objectives or management actions. The objectives of these publications were considered during impact and risk assessments. The controls outlined in Table 6-22 are consistent with the objectives of the materi listed above. Santos considers the potential impacts from vessel discharges to be consistent with these objectives.</li> <li>Vessel discharges comply with the requirements of the <i>Protection of the Sea</i> (<i>Prevention of Pollution from Ships</i>) <i>Act</i> 1983 (Cth), <i>Marine Pollution Act</i> 1995 (NT), <i>Navigation Act</i> 2012 (Cth), and the following Marine Orders and is enacted by:         <ul> <li>Marine Order 91 (Marine pollution prevention – oil)</li> <li>Marine Order 95 (Marine pollution prevention – garbage)</li> </ul> </li> </ul>		Recovery Plan for Marine Turtles in Australia 2017–2027 (CoA, 2017b)
Habitat degradation or modification is identified in many conservation advices however the nature of Activity operational discharges will not result in habitat degradation. Pollution is identified in a number of plans but pertains to more toxic discharges and therefore is not considered applicable here given the discharges are allowable in accordance with legislation or are of low toxicity. Recovery plans / conservation advice for other species that may occur in the OA do not identify habitat degradation / modification or pollution as a key three or have explicit relevant objectives or management actions. The objectives of these publications were considered during impact and risk assessments. The controls outlined in Table 6-22 are consistent with the objectives of the materi listed above. Santos considers the potential impacts from vessel discharges to be consistent with these objectives.Are performance outcomes, control measures and associated performance standards consistent with legal and regulatory requirements?Vessel discharges comply with the requirements of the Protection of the Sea (Prevention of Pollution from Ships) Act 1983 (Cth), Marine Pollution Act 1995 (NT), Navigation Act 2012 (Cth), and the following Marine Orders and is enacted by:        		<ul> <li>Sawfish and River Sharks Multispecies Recovery Plan (CoA, 2015b)</li> </ul>
Recovery plans / conservation advice for other species that may occur in the OA do not identify habitat degradation / modification or pollution as a key thre or have explicit relevant objectives or management actions. The objectives of these publications were considered during impact and risk assessments. The controls outlined in Table 6-22 are consistent with the objectives of the materi listed above. Santos considers the potential impacts from vessel discharges to be consistent with these objectives.Are performance outcomes, control measures and associated performance standards consistent with legal and regulatory requirements?Vessel discharges comply with the requirements of the Protection of the Sea (Prevention of Pollution from Ships) Act 1983 (Cth), Marine Pollution Act 1995 (NT), Navigation Act 2012 (Cth), and the following Marine Orders and is enacted by:        		Habitat degradation or modification is identified in many conservation advices, however the nature of Activity operational discharges will not result in habitat degradation. Pollution is identified in a number of plans but pertains to more toxic discharges and therefore is not considered applicable here given the discharges are allowable in accordance with legislation or are of low toxicity.
Are performance outcomes, control measures and associated performance standards consistent with legal and regulatory requirements?Vessel discharges comply with the requirements of the Protection of the Sea (Prevention of Pollution from Ships) Act 1983 (Cth), Marine Pollution Act 1995 (NT), Navigation Act 2012 (Cth), and the following Marine Orders and is enacted by:• Marine Order 91 (Marine pollution prevention – oil)• Marine Order 95 (Marine pollution prevention – garbage)		Recovery plans / conservation advice for other species that may occur in the OA do not identify habitat degradation / modification or pollution as a key threat or have explicit relevant objectives or management actions. The objectives of these publications were considered during impact and risk assessments. The controls outlined in Table 6-22 are consistent with the objectives of the material listed above. Santos considers the potential impacts from vessel discharges to be consistent with these objectives.
Marine Order 96 (Marine pollution prevention – sewage).      These instruments give effect to MARPOL Appeares LILLILLIV and V	Are performance outcomes, control measures and associated performance standards consistent with legal and regulatory requirements?	<ul> <li>Vessel discharges comply with the requirements of the <i>Protection of the Sea</i> (<i>Prevention of Pollution from Ships</i>) <i>Act 1983</i> (Cth), <i>Marine Pollution Act 1999</i> (NT), <i>Navigation Act 2012</i> (Cth), and the following Marine Orders and is enacted by:</li> <li>Marine Order 91 (Marine pollution prevention – oil)</li> <li>Marine Order 95 (Marine pollution prevention – garbage)</li> <li>Marine Order 96 (Marine pollution prevention – sewage).</li> </ul>

	Vessel discharges are consistent with the general duty of the <i>Waste</i> <i>Management and Pollution Control Act 1988</i> (NT) (WMPC Act). The Coastal Waters OEMP is also compliant with the conditions of EPBC approval EPBC 2022/09372.
Are performance outcomes, control measures and associated performance standards consistent with Santos' Environment, Health and Safety Policy?	Yes – aligns with Santos' Environment, Health and Safety Policy (Appendix A).
Are performance outcomes, control measures and associated performance standards consistent with industry standards?	Yes – the most recent and comparable EPs accepted by NOPSEMA were reviewed for consistency with the performance outcomes, control measures and associated performance standards proposed in this GEP Coastal Waters OEMP.
Have performance outcomes, control measures and associated performance standards taken into consideration Relevant Person feedback?	Yes – Relevant Person feedback was considered and indicated no recommendations for revising the EPO, CMs or EPSs. However, feedback received during the development of other Barossa Gas Project EPs has been considered and EPO-20 was adopted. Santos has adopted control measure (BAO-CM-6.4.5) for HSE induction to be conducted that includes environmental requirements.
Are performance standards such that the impact or risk is considered to be ALARP?	Yes – see ALARP above.

The consequence of operational discharges is assessed as II – Minor. Based on an assessment of Santos' acceptability criteria and with the control measures in place, potential impacts are considered acceptable.



# 7. Unplanned events risk assessment

### OPGGS(E)R 2023 Requirements

#### Section 21. Environmental assessment

Evaluation of environmental impacts and risks

(5) The environment plan must include:

- a. details of the environmental impacts and risks for the activity; and
- b. an evaluation of all the impacts and risks, appropriate to the nature and scale of each impact or risk; and
- c. details of the control measures that will be used to reduce the impacts and risks of the activity to ALARP and an acceptable level.

(6) To avoid doubt, the evaluation mentioned in paragraph (5)(b) must evaluate all the environmental impacts and risks arising directly or indirectly from:

- a. all operations of the activity; and
- b. any potential emergency conditions, whether resulting from accident or any other cause.

Environmental performance outcomes and standards

(7) The environment plan must:

- a. set environmental performance standards for the control measures identified under paragraph (5)(c); and
- b. set out the environmental performance outcomes against which the performance of the titleholder in protecting the environment is to be measured; and
- c. include measurement criteria that the titleholder will use to determine whether each environmental performance outcome and environmental performance standard is being met.

An ENVID workshop (as described in Section 5.2.3) was held on 23 July and 29 July 2024 to consider the GEP activities and this Activity. Santos' assessment identified seven environmental risks associated with unplanned events for this Activity.

The results of the impact and risk assessments process undertaken as per Section4 are summarised in Table 7-1. A comprehensive impact and risk assessment for each unplanned event and subsequent control measures proposed by Santos to reduce risk and potential impacts to ALARP and acceptable levels are detailed in the following subsections.

Adopted control measures are assigned a reference number (e.g. BAO-CM-7.1.1). The control measure reference numbers are aligned with the Barossa Production Operations EP and therefore are not in numerical order in this GEP Coastal Waters OEMP.

OEMP section	Unplanned event	Likelihood	Consequence	Residual risk level
7.1	Release of solid objects	B – Unlikely	II – Minor	Very Low
7.2	Introduction of invasive marine species	B – Unlikely	IV – Major	Low
7.3	Marine fauna interaction	B – Unlikely	II – Minor	Very Low
7.4	Minor releases of hydrocarbons and chemicals	C - Possible	I - Negligible	Very Low
7.5	Subsea release of dry natural gas	A – Remote	III – Moderate	Very Low
7.6	Surface release of MDO from a vessel	B – Unlikely	III – Moderate	Low
7.7	Contingency spill response operations	N/A	II – Minor	N/A

### Table 7-1: Environmental risk assessment summary



### 7.1 Release of solid objects

### 7.1.1 Description of event

Event	Solid objects can be accidentally released to the marine environment, including:				
	<ul> <li>non-hazardous solid wastes, such as paper, plastics, and packaging</li> </ul>				
	hazardous solid wastes, such as batteries, fluorescent tubes, medical wastes, and aerosol cans				
	<ul> <li>equipment and materials, such as supplies, hard hats, and tools</li> </ul>				
	infrastructure recovered or replaced during IMMR.				
	Release of these solid objects may occur as a result of:				
	overfull or uncovered bins				
	incorrectly disposed items				
	incidents during transfers of waste or supplies				
	dropped objects and lost equipment through lifting operator error or mechanical failure.				
	Operational area				
	Within the OA, events relate to IMMR activities only.				
Extent	The event will only occur within the OA, and all non-buoyant waste material or dropped objects are expected to sink to the seabed and remain in the OA, with the worst-case disturbance being the loss of a section of GEP, should it be accidentally released during a replacement activity.				
	Buoyant objects could potentially move beyond the OA.				
Duration	Constant:				
	No regular activities are planned which may result in unplanned release of solids to occur within the OA.				
	Infrequent and one-off:				
	IMMR vessel presence occurs typically for approximately 7 to 30 days in duration every three to five years, or as needed and impacts may remain while the solid object is in the environment.				

### 7.1.2 Nature and scale of environmental impacts

**Potential receptors**: physical environment (water quality, benthic habitats, KEF); Threatened, Migratory fauna or local fauna (marine reptiles, whales, sharks, fish, and rays), protected areas (marine parks) and socio-economic receptors (including cultural features).

Solids such as plastics have the potential to affect benthic environments and to harm marine fauna through entanglement or ingestion. Marine turtles and seabirds are particularly at risk from entanglement and ingestion. Marine turtles may mistake plastics for food; once ingested, plastics can damage internal tissues and inhibit physiological processes, which can both potentially result in fauna fatality. Floating, non-biodegradable marine debris has been highlighted as a threat to marine turtles, whales and whale sharks in the relevant recovery plans and approved conservation advice (refer to Table 3-13). The recovery plans, approved conservation advice and wildlife conservation plans, as well as the Threat Abatement Plan for the Impacts of Marine Debris on the Vertebrate Wildlife of Australia's Coasts and Oceans (DoEE, 2018), have specified several recovery actions to help combat this threat. Of relevance is the legislation for the prevention of garbage disposal from vessels.

Release of hazardous solids (for example, wastes such as batteries) may result in pollution of the immediate receiving environment, leading to detrimental health impacts to marine fauna. Physiological damage can occur through ingestion; or absorption may occur in individual fish and sharks, marine mammals, marine reptiles, or seabirds.

The area of potential seabed disturbance due to release of a heavier, non-hydrocarbon solid, such as equipment or infrastructure would be restricted to the OA (for example, accidentally dropped equipment). Damage to substrates within the OA and associated infauna and epifauna may occur; however, such impact is expected to be restricted to the size of the dropped object. While soft sediment benthic habits will not be destroyed, disturbance of the communities on and within them (as in, the epifauna and infauna) will occur in the event of a dropped object, and depressions may remain on the seabed for some time after the dropped object is removed as it gradually infills over time. However, the soft sediment habitat within the OA is not expected to have a particularly high abundance, diversity, or unique composition of benthic invertebrates.

Benthic habitats along the OA consist predominantly of bare sediments, with other benthic habitat types constituting relatively small portions. All of these habitat types are well represented throughout the region (Section 3.2.9).



### 7.1.2.1 Threatened, migratory or local fauna

As discussed above, the impact is not anticipated to significantly affect mobile marine fauna, such as marine mammals, marine reptiles, fish, sharks, and rays. Any impact to an area of seabed disturbed within the OA also represents a negligible portion of the habitat available for Threatened, Migratory or local fauna.

Information provided during consultation identified that if culturally significant species are impacted this can impact First Nations access to food through traditional hunting and fishing, and in accordance with First Nations cultural beliefs if totemic species (e.g., turtles, whales, dugongs, bird and fish) are impacted by the activity, some believe this can in turn can impact Tiwi people and make them sick. Floating, non-biodegradable marine debris has been highlighted as a threat to marine turtles, sharks, seabirds, whales and whale sharks in the relevant recovery plans and approved conservation advice (see Table 3-13). Marine turtles and seabirds are particularly at risk from entanglement and ingestion.

The recognition of the problem of plastic and microplastic debris in the marine environment is a key aspect of the National Plastics Plan (DAWE, 2021). The National Plastics Plan also includes supporting global action to address marine plastic debris, including the implementation of the Threat Abatement Plan for the Impacts of Marine Debris on the Vertebrate Wildlife of Australia's Coasts and Oceans (DoEE, 2018). The Threat Abatement Plan for the Impacts of Marine Debris on the Vertebrate Wildlife of Australia's Coasts and Oceans (DoEE, 2018). The Threat Abatement Plan for the Impacts of Marine Debris on the Vertebrate Wildlife of Australia's Coasts and Oceans (DoEE, 2018). The Threat Abatement Plan for the Impacts of Marine Debris on the Vertebrate Wildlife of Australia's Coasts and Oceans (DoEE, 2018) and Wildlife Conservation Plan for Seabirds (CoA, 2020) have specified various recovery actions to help combat this threat. Floating non-biodegradable marine debris has been highlighted as a threat to marine turtles within the Recovery Plan for Marine Turtles in Australia 2017–2027 (CoA, 2017b).

Santos has considered information contained in relevant recovery plans and approved conservation advice for cetaceans that identify habitat modification as a potential threat (Table 3-13). This includes the objectives and actions with the Sawfish and River Shark Multispecies Recovery Plan (CoA, 2015b) and Recovery Plan for Marine Turtles in Australia 2017–2027 (CoA, 2017b), which relate to habitat degradation and modification. Marine turtles may mistake plastics for food—once ingested, plastics can damage internal tissues and inhibit physiological processes, both of which can potentially result in fauna mortality. Plastics such as microplastics, plastic bags or bottles can cause problems by ingestion or as entanglement in small cetaceans. Entanglement and ingestion of plastics may result in the loss of reproductive fitness or mortality for cetaceans (CoA, 2015a). Given the low level of seabed disturbance and the benthic habitats in the OA being well represented in the wider surrounds, the activities are not inconsistent with the recovery plans and conservation advice.

Of relevance to the Activity is legislation for preventing garbage disposal from vessels, which Santos implements in accordance with MARPOL Annex V, which is implemented in Australia through the *Protection of the Sea* (*Prevention of Pollution from Ships*) Act 1983 (Cth), the Navigation Act 2012 (Cth) and Marine Order 95.

### 7.1.3 Environmental performance outcomes and control measures

The EPO relating to this event is:

- No loss of equipment or cargo overboard from vessels. (EPO-06)
- No injury, or mortality to, EPBC Act 1999 listed marine fauna or marine fauna listed as threatened species under the *Territory Parks and Wildlife Conservation Act 1976* (NT) as a result of the Activity (EPO-08)
- No significant impact to cultural features from the Activity (EPO-19)
- No impacts to underwater cultural heritage from the Activity (EPO-20).

An assessment of the environmental benefits and the potential costs or issues associated with control measures for this activity are shown in Table 7-2 to demonstrate the potential risks are ALARP. Control measures that are adopted have associated EPSs and measurement criteria, which are presented in Table 8-2. Rejected control measures have an ALARP evaluation provided to justify their rejection.

### Table 7-2: Control measures evaluation for release of solid objects

CM reference	Control measure	Environmental benefit	Potential cost/issues	Evaluation
Standard control meas	sures			
BAO-CM-6.4.5	HSE inductions will include applicable environmental requirements (administrative control)	Ensures that crew are aware of the stringent Coastal Waters OEMP, Santos and legislative requirements. Ensures personnel are suitably aware of	Administrative costs to update existing Santos procedure and induction materials and train personnel.	Adopted – benefits considered to outweigh costs.

CM reference	Control measure	Environmental benefit	Potential cost/issues	Evaluation
		cultural features and values		
BAO-CM-7.1.1	Implement standards and procedures for lifting equipment to reduce risk of dropped objects during lifting (administrative control)	Impacts to the environment are reduced by preventing dropped objects and dragged objects during lifting operations. Administrative costs to update induction materials and train personnel.	Cost of implementing procedures.	Adopted – environmental benefits of preventing dropped objects outweigh procedural compliance costs.
BAO-CM-7.1.2	Dropped objects (incident) management (administrative control)	Impacts to the environment are reduced by preventing dropped objects and by retrieving dropped objects unless the environmental consequences of the dropped object are negligible or there are risks to safety.	Cost of implementing procedures.	Adopted – environmental benefits of recovering dropped objects outweigh procedural compliance costs.
BAO-CM-7.1.3	International Maritime Dangerous Goods Code (administrative control)	Regulatory requirement that reduces the risk of an environmental incident, such as an accidental container release to sea or unintended chemical reaction.	Cost of implementing procedures.	Adopted – it is a legislated requirement.
BAO-CM-7.1.4	Chemicals and hydrocarbons will be managed in accordance with SDS to reduce risk of release to the marine environment (administrative control)	Reduces the risk of accidental discharge to sea by controlling the storage, handling and clean-up of chemicals.	Cost of implementing procedures. Regulatory requirement to manage hazardous chemicals.	Adopted – environmental benefits of ensuring procedures are followed outweigh the costs, plus the control is a legislated requirement.
Additional control mea	asures			
N/A	Eliminate lifting in the field (elimination control)	Reduces the risk of dropped objects.	Lifting is an essential activity for IMMR activities.	<b>Not adopted</b> – not feasible to eliminate lifting in the field.
N/A	Cessation of operations until all dropped objects are located and recovered (administrative control)	Would minimise potential for further disturbance due to dropped object potentially moving around on the seabed causing further disturbance or long-term impacts.	Substantial additional cost due to downtime greater than the value of equipment lost. Little benefit, given water depths and sparse distribution of sensitive benthic habitats in the OA.	Not adopted – cost outweighs the benefit.
N/A	Immediate removal of solid waste from the OA (administrative control)	Reduces the risk of release of non-hazardous solids to the marine environment.	Substantial additional fuel cost (emissions increase) and personnel time, as the number of transfers would be increased and is not considered practicable.	Not adopted – cost outweighs the benefit.



Receptors	Physical environment (benthic habitats)
	• Threatened, Migratory or local fauna (marine mammals, marine reptiles, sharks, fish, and rays)
Consequence	II – Minor
Physical enviro	nment (benthic habitats)
An object droppe seabed is limited equipment used of	d into the sea can result in localised and short-term damage to the seabed. The extent of the impact to the to the size of the dropped object (e.g. tools, containers, and installation equipment); given the size of the on the activity vessels.
The unplanned re ocean, which car food chain. Howe pollution is relativ	elease of microplastics has the potential to contribute to the overall amount of marine microplastics in the have various impacts on marine fauna as they are absorbed by plants and animals and accumulate in the ever, given the negligible amounts that may be accidentally released, the overall impact marine microplastic rely limited.
Marine invertebra and surrounds ar	ates that may inhabit disturbed soft sediment benthic habitats are expected to occur elsewhere within the OA and therefore the disturbance is not expected to affect prey availability, or protected fauna species.
No significant sea during the Activity Therefore, the co	abed features or biota have been found in the OA. Therefore, it is highly unlikely that any objects dropped y would cause a significant impact to the ecological values associated with the seabed or benthic habitats. nsequence level is considered I – Negligible.
Threatened, Mig	ratory or local fauna (marine mammals, marine reptiles, sharks, fish, and rays)
Marine debris (increlevant recovery sea are limited by restricted to a sm	cluding plastics and microplastics) is identified as a potential threat to several marine fauna species in plans and conservation advice (Table 3-14). The types of solids and plastics accidentally dropped into the the type of activities planned. If the solid object can be ingested by marine fauna, impacts would be all number of individuals, if any.
Microplastics with ingested. Filter fe consuming prey ( and plastics that shorelines and se controls impleme particles.	hin the ocean come from many sources, and the bioaccumulation potential is high within marine fauna if beders ingest substantial amounts of microplastics by directly swallowing ocean water or indirectly by that have microplastics within the body cavity). Given that the very small volume of unplanned microplastics could potentially be released to the marine environment is relatively small and the distance of the OA to ensitive turtle habitats, it is considered that the consequence of any impacts is considered to be slight. The nted demonstrate that the Activity will be conducted to reduce the release of marine debris and plastic
Recovery Plan fo marine turtles. Th and Oceans (Dol marine turtles and prevention of gar	r Marine Turtles in Australia 2017–2027 (CoA, 2017b) has identified marine debris as a potential threat to the Threat Abatement Plan for the Impacts of Marine Debris on the Vertebrate Wildlife of Australia's Coasts EE, 2018) also identifies marine debris as a threat. These plans identify marine debris as potential threats to d vertebrate wildlife, resulting in potential injury or death, and recommend adherence to legislation for the bage disposal to prevent impacts.
Given the negligil commercial fish s	ble consequence on species, subsequent risks, or significant impacts to socio-economic receptors (including stocks and cultural features relating to species with cultural significance) are not anticipated.
The limited quantities associated with this event indicate, even in a worst-case release of solid waste, impacts to fauna would be limited to individuals and are not expected to result in a decrease of the local population size. The consequence level for marine fauna is therefore considered I – Negligible for both smaller and larger objects of buoyant and non-buoyant materials.	
For potential impacts to marine species of cultural significance or that provide a traditional food source, refer to the above assessment for threatened, migratory or local fauna.	
Socioeconomic	and cultural features
Given the negligil commercial fish s	ble consequence on species, subsequent risks or significant impacts to socioeconomic receptors (including stocks) and cultural features (relating to species with cultural significance) are not anticipated.
Likelihood	B – Unlikely
The proposed co hazardous solid v prevention of gar Marine Debris on occurring over the as it has occurred	ntrol measures will ensure the risks of dropped objects, lost equipment, or release of hazardous and non- vaste to the environment has been reduced. These control measures will also ensure legislation for the bage disposal from vessels is adhered to, as recommended in the Threat Abatement Plan for the Impacts of the Vertebrate Wildlife of Australia's Coasts and Oceans (DoEE, 2018). The likelihood of dropped objects e duration of the activity is considered 'Possible' for larger items that would result in a Minor consequence, d before during other Santos projects.
The risk to socioe	economic receptors and cultural features is considered to be Very Low.
Residual risk	The residual risk is considered Very Low.
15 D-	monstration of as low as reasonably prostigable
. 1.5 Del	monstration of as low as reasonably practicable
II reasonably pr nanage the resid nanagement crit	acticable control measures were reviewed and those adopted are considered appropriate to dual risk to a Low level. The proposed management controls are in accordance with Santos' risk eria and are considered appropriate to reduce the risk to ALARP.



### 7.1.6 Acceptability evaluation

Is the risk ranked between Very Low to Medium?	Yes – residual risk is ranked Very Low.
Is further information required to validate the consequence assessment?	No – potential impacts and risks are well understood through the information available. Extensive marine studies have been completed within the OA to inform the assessment.
Are risks and impacts consistent with the principles of ecologically sustainable development (ESD)?	Yes – activity evaluated in accordance with Santos' Offshore Division Environmental Hazard Identification and Assessment Guideline which considers the principles of ESD:
	The impacts associated with unplanned minor loss of containment do not result in 'threats of serious or irreversible harm' as detailed within the EPBC Act and biodiversity and ecological integrity will be maintained.
	Conservative assumptions on scale of impact have been applied.
	The health, diversity and productivity of the environment will be maintained, including for future generations.
Have the acceptable levels of impact and risks been informed by relevant species recovery plans, threat abatement plans, conservation advice, wildlife conservation plans and Australian marine park zoning objectives)?	Yes – controls implemented will minimise the potential impacts from the Activity to species identified in recovery plans and approved conservation advices as having the potential to be impacted by solid objects. The following material published in relation to threatened and migratory species within the OA identifies marine debris as a threat:
	Management plans
	National Plastics Plan (DAWE, 2021)
	Conservation advice
	<ul> <li>Approved Conservation Advice for <i>Pristis clavata</i> (Dwarf Sawfish) (DEWHA, 2009)</li> </ul>
	<ul> <li>Approved Conservation Advice for <i>Pristis pristis</i> (largetooth sawfish) (TSSC, 2014b)</li> </ul>
	Approved Conservation Advice for Green Sawfish (DEWHA, 2008c)
	<ul> <li>Approved Conservation Advice for <i>Glyphis garricki</i> (northern river shark) (TSSC, 2014a)</li> </ul>
	<ul> <li>Approved Conservation Advice for <i>Glyphis glyphis</i> (speartooth shark) (DoE, 2014a).</li> </ul>
	<ul> <li>Conservation Advice for <i>Rhincodon typus</i> (whale shark) (TSSC, 2015g)</li> </ul>
	<ul> <li>Conservation Advice for <i>Balaenoptera physalus</i> (fin whale) (TSSC, 2015c)</li> </ul>
	<ul> <li>Conservation Advice for <i>Balaenoptera borealis</i> (sei whale) (TSSC, 2015b)</li> </ul>
	<ul> <li>Approved Conservation Advice for <i>Dermochelys coriacea</i> (Leatherback Turtle) (DEWHA, 2008b)</li> </ul>
	<ul> <li>Approved Conservation Advice for Limosa limosa (black-tailed godwit) (DCCEEW, 2024e)</li> </ul>
	<ul> <li>Approved Conservation Advice for Calidris canutus (red knot) (DCCEEW, 2024c)</li> </ul>
	Approved Conservation Advice for Calidris acuminata (sharp-tailed sandpiper) (DCCEEW, 2024b)Approved Conservation Advice for <i>Numenius madagascariensis</i> (Eastern Curlew) (TSSC, 2015f)
	Recovery plans
	Threat Abatement Plan for the impacts of marine debris on vertebrate wildlife of Australia's coasts and oceans (DoEE, 2018)
	Sawfish and River Sharks Multispecies Recovery Plan (CoA, 2015b)
	<ul> <li>Recovery Plan for the White Shark (<i>Carcharodon carcharias</i>) (DSEWPaC, 2013)</li> </ul>
	<ul> <li>Conservation Management Plan for the Blue Whale - A Recovery Plan under the Environment Protection and Biodiversity Conservation Act 1999 2015–2025 (CoA, 2015a)</li> </ul>
	<ul> <li>Recovery Plan for Marine Turtles in Australia 2017–2027 (CoA, 2017b)</li> </ul>
	Wildlife Conservation Plan for Seabirds (CoA, 2020)
	Wildlife Conservation Plan for Migratory Shorebirds (CoA, 2015c)

	Sawfish and River Shark Multispecies Recovery Plan (CoA, 2015b)
	Recovery plans / conservation advice for other species that may occur in the OA do not identify marine debris as a key threat or have explicit relevant objectives or management actions related to marine debris.
	The OA does not intersect any AMP or protected area.
	The objectives of these publications were considered during impact and risk assessments. The controls outlined in Table 7-2 and Section 7.1.3 are consistent with the objectives of the material listed above. Santos considers the Activity is not inconsistent with these objectives.
Are performance outcomes, control measures and associated performance standards consistent with legal and regulatory requirements?	Yes – management measures are consistent with MARPOL Annex V (through the <i>Protection of the Sea (Prevention of Pollution from Ship) Act 1983</i> (Cth), the <i>Navigation Act 2012</i> (Cth) and Marine Order 95: Marine pollution prevention – garbage), Annex X (IMO Marine Litter Action Plan) and International Maritime Dangerous Goods Code).
	Through acceptance of this GEP Coastal Waters OEMP, legislative and regulatory requirements will be met as per Appendix B. The GEP Coastal Waters OEMP is also compliant with the conditions of EPBC approval EPBC 2022/09372.
Are performance outcomes, control measures and associated performance standards consistent with Santos' Environment, Health and Safety Policy?	Yes – aligns with Santos' Environment, Health, and Safety Policy (Appendix A).
Are performance outcomes, control measures and associated performance standards consistent with industry standards?	Yes – the most recent and comparable Operations EPs accepted by NOPSEMA have been reviewed for consistency with the performance outcomes, control measures and associated performance standards proposed in this GEP Coastal Waters OEMP.
Have performance outcomes, control measures and associated performance standards taken into consideration Relevant Person feedback	Yes – no objections or claims raised by Relevant Persons relating to unplanned release of solid objects and waste and potential environmental impacts.
Are performance standards such that the impact or risk is considered to be ALARP?	Yes – ALARP assessment conducted, with no additional control measures adopted.

The residual risk of an unplanned release of solid objects on receptors is assessed as Very Low. Based on an assessment of Santos' acceptability criteria and with the control measures in place, potential risks are considered acceptable.



### 7.2 Introduction of invasive marine species

Event	Invasive marine species (IMS) could potentially be introduced to the OA from IMMR vessels through ineffective ballast water management or through ineffective biofouling management.		
	Potential sources for the transfer and establishment of IMS may include:		
	<ul> <li>biofouling on the vessels' internal niches (such as sea chests, strainers, seawater pipework, anchor cable lockers and bilge spaces)</li> </ul>		
	<ul> <li>biofouling on the vessel's other external niches (such as propulsion units, steering gear and thruster tunnels)</li> </ul>		
	<ul> <li>biofouling on equipment that routinely becomes immersed in water (such as ROVs)</li> </ul>		
	ballast water exchanges		
	cross-contamination between vessels.		
	Ballast water is responsible for up to 30% of all marine pest incursions into Australia, while biofouling (the accumulation of aquatic microorganisms, algae, plants and animals on vessel hulls and submerged surfaces) is also considered a significant pathway for the potential introduction and spread of marine pests (DAWE, 2018).		
	Biofouling organisms may attach to the hull, particularly in areas such as seams and unpainted surfaces which are easy to attach to or where water turbulence is lowest (such as niches, sea chests). Organisms can also be drawn into ballast tanks during cargo unloading, as additional water is required for stabilisation.		
Extent	Localised (seabed and water column) within the OA to widespread if successfully translocated to new areas via ocean currents or equipment transit.		
Duration	Constant		
	Temporary or long-term in the event of successful species translocation		
	Infrequent		
	Introduction of IMS may occur from vessel use. Campaign vessels will be infrequent, as per operational requirements for specific campaigns. Support vessel presence is required for day-to-day operations during routine IMMR. The duration of IMMR vessel presence will be approximately 7 to 30 days, every three to five years Vessels for unplanned activities, including USV, would be less frequent.		

### 7.2.1 Description of event

### 7.2.2 Nature and scale of environmental impacts

**Potential receptors:** physical environment (benthic habitat); threatened, migratory, or local fauna (marine mammals, marine turtles, sharks and rays and other fish); socioeconomic (commercial fisheries, other marine users, tourism); and cultural features (such as totem species and cultural food sources).

IMS are non-native marine plants or animals that harm Australia's marine environment, social amenity or industries that use the marine environment, or have the potential to do so if they were to be introduced, established or spread in Australia's marine environment (DAWE, 2018). Most climatically compatible IMS to northern Australia are found in Southeast Asian countries.

A number of vectors can transport IMS outside their native ranges. Shipping is considered to be the largest contributor for the human-mediated movement of IMS around the world (Ruiz *et al.*1997; Minchin and Gollasch 2002). For instance, ocean-going vessels can transport IMS in ballast water, as biofouling attached to submerged immersible equipment, within internal seawater systems and/or on the exterior of the hull.

Some IMS pose major threat to economy and social amenity by disrupting ecological processes (DAWE, 2018; Wells et al., 2009). When IMS achieve pest status, they are commonly referred to as introduced marine pests. Introduced marine pests can cause a variety of adverse effects in a receiving environment, including:

- Over-predation of native flora and fauna
- Out-competing of native flora and fauna for food
- Human illness through released toxins
- Depletion of viable fishing areas and aquaculture stocks
- Reduction of coastal aesthetics
- Damage to marine and industrial equipment and infrastructure.

The above impacts can result in detrimental flow-on effects to marine parks, tourism, recreation, and cultural features, noting that some native fauna may have cultural significance as dreaming totems or as a traditional food source.



In the 2022 Statement of Reasons Requests, the Tiwi clan members raised their concern regarding traditional hunting of marine species and totem species. First Nations people maintain a continuing spiritual connection with sea country, including marine fauna species with cultural significance, such as totems or as a cultural food source. An important outcome of Dr Corrigan's research is that no sacred sites or dreamings are shown to be directly impacted by the DPD project footprint (where this Activity OA is located), although this is not to say that some persons do not have fears that this could be the case in the event of an unplanned event (Corrigan, 2024).

Species of concern are those that are not native to the region, are likely to survive and establish in the region, and are able to spread by human-mediated or natural means. Species of concern vary from one region to another, depending on various environmental factors such as water temperature, salinity, nutrient levels, and habitat type. These factors dictate invasive species' survival and invasive capabilities.

Artificial, disturbed and polluted habitats in tropical regions are susceptible to introductions, which is why ports are often areas of higher IMS risk (Neil *et al.*, 2005). However, in Australia there are limited records of detrimental impact from IMS compared with other tropical regions (such as the Caribbean).

Once IMS populations have established, they are difficult to eradicate, limiting management options to ongoing control or impact minimisation. However, this depends on the environmental conditions and species. For this reason, increased management requirements have been implemented in recent years by various Australian regulatory agencies.

If an IMS is introduced, species have been known to colonise areas outside of the areas to which it is introduced, but this depends on the diversity and extent of suitable habitat for colonisation.

Potential sources for introducing marine species into the OA include biofouling on the vessels, including both external (e.g propulsion units, steering gear and thruster tunnels) and internal niches (e.g. sea chest, trainers, seawater pipework, anchor cable lockers and bilge spaces). Ballast water is water taken on board vessels from their originating port in order to maintain vessel draft and stability. Unmanaged exchange of this ballast water, which can contain aquatic microbes, plants, and animals from the originating port, can result in the release of a range of IMS to the destination area or port. Biofouling is the accumulation of aquatic micro-organisms, algae, plants and animals on vessel hulls and submerged surfaces. Whilst ballast water exchanges are responsible for up to 30% of all marine pest incursions into Australian waters, research indicates biofouling has been responsible for more foreign marine species introductions (DAWE, 2018).

Equipment that is submerged in water for periods of time, such as ROVs may also acquire marine pest species, which can be spread if the equipment is not cleaned before use in pest-free areas.

IMS are generally unable to successfully establish in deep-water ecosystems (Geiling, 2014), most likely due to a lack of light and suitable habitat to sustain their growth and survival. Therefore, most IMS are found in tidal and subtidal zones, with only a few species known to extend into deeper waters of the continental shelf (Bax et al., 2003). Most species introduced to an area outside of their natural range (such as via ballast water) will not survive to establish or subsequently become invasive or a pest (Wells et al., 2009).

IMS risks are relevant to all maritime activities, including commercial shipping, fishing, military, petroleum, and recreational boating.

### 7.2.3 Environmental performance outcomes and control measures

The EPOs relating to this event include:

- No introduction, establishment or spread of IMS in the natural environment as a result of the Activity (EPO-07)
- No significant impact to cultural features from the Activity (EPO-19)
- No impacts to underwater cultural heritage from the Activity (EPO-20).

An assessment of the environmental benefits and the potential costs or issues associated with control measures for this activity are shown in Table 7-3 to demonstrate potential risks are ALARP. Control measures that are adopted have associated EPSs and measurement criteria that are presented in Table 8-2. Control measures that are not adopted have an ALARP evaluation provided for justification.

Section 8.3.2.9 describes the biosecurity management strategy to be adopted for the Activity, consistent with the Biosecurity Act 2015 and National Biofouling Management Guidance for the Petroleum Production and Exploration Industry (Marine Pest Sectoral Committee, 2009).

IMS are more likely to populate shallower areas with favourable substrates, such as on shoals and reefs. The closest sensitive habitat is an unnamed shoal located 6 km from the OA, which may provide suitable habitat. Shepparton Shoal is approximately 23 km west of the OA.



### Table 7-3: Control measures evaluation for introduction of invasive marine species

CM reference	Control measure	Environmental benefit	Potential cost/issues	Evaluation
Standard control measures				
BAO-CM-6.4.5	HSE inductions will include applicable environmental requirements (administrative control)	Ensures that crew are aware of the stringent Coastal Waters OEMP, Santos and legislative requirements.	Administrative costs to update existing Santos procedure and induction materials and train personnel.	Adopted – cost for implementation proportionate to risk reduction.
BAO-CM-7.2.1	Develop and implement a biosecurity management plan in consultation with and approved by the Department of Agriculture Forestry and Fisheries (DAFF). (administrative control)	Reduces the risk of introducing IMS due to assessment and management procedures	Cost associated with implementing procedures and implementing the mitigation measures.	Adopted – cost for implementation proportionate to risk reduction
BAO-CM-7.2.2	Vessels undertake ballast water management or treatment to achieve low risk ballast water . (administrative control)	The likelihood of introducing IMS via ballasting activities is reduced by managing ballast water exchange and identifying high risk ballast water.	Inconsequential as Australian Ballast Water Management Requirements align to the International Convention for the Control and Management of Ships' Ballast Water and Sediments 2004 (the Ballast Water Management Convention), which entered into force internationally on 8 September 2017.	Adopted – cost for implementation proportionate to risk reduction.
BAO-CM-7.2.3	Vessels equipped with effective anti-fouling coatings (engineering control)	The likelihood of introducing IMS is reduced by in water inspection and cleaning of Activity vessels under the direction of appropriately qualified marine biologist.	Cost associated with implementing procedures and implementing the mitigation measures.	Adopted – cost for implementation proportionate to risk reduction.
BAO-CM-7.2.4	Vessels equipped with Marine Growth Prevention System (MGPS) (engineering control)	The likelihood of introducing IMS is reduced by preventing marine growth on the submerged surfaces of vessels.	Cost associated with implementing procedures and implementing the mitigation measures.	Adopted – cost for implementation proportionate to risk reduction

CM reference	Control measure	Environmental benefit	Potential cost/issues	Evaluation
Additional cont	rol measures			
N/A	Heat treatment of ballast water to eliminate IMS (engineering control)	Would reduce potential for IMS to establish by reducing the potential for IMS present in ballast water.	Compared to traditional ballast treatment methods (e.g., chemical additive), heat treatment has a higher cost and increased energy consumption. Ballast requirements are adequately managed under Australian Ballast Water Management (DAWE, 2020) and the International Convention for the Control and Management of Ships' Ballast Water and Sediments to reduce the risk of IMS introduction.	Not adopted – based on high cost considered disproportionate to the risk.
N/A	Contract vessels only operating in local, state/territory or Commonwealth waters to reduce potential for IMS (substitute control)	Reduce potential for IMS to be transported into area since vessels would not have originated elsewhere.	Vessels and equipment suitable for the Activity may not be available in 'local' waters. Potential significant costs and delays to the Activity schedule by only contracting vessels working in 'local' waters.	Not adopted – potential for significant schedule delays and activity costs if suitable vessels are not 'locally' available. All contracted vessels must be 'low' risk of introducing IMS, regardless of their origin.
N/A	Use an alternative ballast system to avoid uptake or discharge of water (engineering control)	Eliminates need for ballast water exchange, therefore decreasing risk of introducing IMS through ballast water.	Vessels suitable for the Activity may not have options for alternative ballast system, therefore would require modification at significant cost.	Not adopted – costs disproportionately high compared with environment benefit, given the proposed risk- based management framework, which includes potential dry docking and cleaning if justified based on risk assessment.
N/A	Do not discharge ballast water (elimination control)	Would reduce the potential for introducing IMS by implementing a 'no ballast water exchange' policy on vessels.	Ballast water exchange required on the vessels for stability.	Not adopted – on the basis ballast water exchange is a safety-critical activity for marine operations
N/A	Mandatory dry docking of vessels before entering field to clean vessel and equipment and remove biofouling (engineering control)	The risk of IMS being present on vessels or associated equipment is further reduced by the removal of biofouling.	Significant costs and delays to the Activity schedule. Unnecessary docking of vessels and removal of biofouling increases waste and emissions.	Not adopted – costs disproportionately high compared with environmental benefit, given the given the proposed risk-based management framework, which includes potential dry docking and cleaning if justified based on risk assessment.



### 7.2.4 Environmental impact and risk assessment

Receptors	Physical environment (benthic habitats and primary producers)
	<ul> <li>Threatened, Migratory, or local fauna (marine mammals, marine turtles, fish [including sharks and rays])</li> </ul>
	Socio-economic (commercial fisheries, other marine users, tourism and cultural features)
	Cultural features
Consequence	IV – Major

#### Physical environment and habitat (benthic habitats and primary producers)

Within the OA, the water depths range from approximately 40 m to 50 m. The OA does not present a benthic habitat or community structure that is favourable to IMS survival. The seabed within the OA is characterised as silty, shelly sand (Figure 3-3) with very sparse (<1%) epibiota (mainly soft corals and crinoids) (RPS, 2023). The closest shoal is more than 8 km away from the GEP route in the OA and Shepparton Shoal is approximately 23 km away from the OA. Shepparton Shoal has water depths of 30 m to 50 m which limit the amount of light to the shoal.

The consequence level is considered IV – Major.

#### Physical environment and habitat (Key ecological features)

The OA is 9 km from the Carbonate bank and terrace system of the Van Diemen Rise KEF (Table 3-8; Figure 3-7). The values of this KEF includes areas of hard substrate (including patch reefs and pinnacles) that can support ecosystems with high levels of biodiversity. Water depths are greater than 100 m; therefore, the values of this KEF are unlikely to be affected by IMS.

Threatened, Migratory, or local fauna (marine mammals, marine turtles, fish [including sharks and rays])

IMS, if successfully established, can outcompete native species for food or space, prey on native species or change the nature of the environment. The Wildlife Conservation Plan for Migratory Shorebirds identified IMS from ballast water and hull transport as a threat to migratory shorebirds, particularly if the introduction results in the loss of benthic food sources at important intertidal habitat (CoA, 2015c). IMS are also identified as a threat in approved conservation advice for the northern river shark, whale shark, dwarf sawfish, green sawfish, largetooth sawfish, and the recovery plan for the white shark (for further information see Table 3-13).

The consequence level is considered IV - Major.

#### Socioeconomic (commercial fisheries, other marine users, tourism)

The introduction of IMS could have a detrimental effect on commercial fisheries, other marine users, and tourism in the area due to the IMS outcompeting native species for food or space, prey on native species or change the nature of the environment; therefore, the consequence level is considered IV – Major.

#### **Cultural Features**

An important outcome of Dr Corrigan's research is that no sacred sites or dreamings are shown to be directly impacted by the DPD project footprint (which includes this Activity OA), although this is not to say that some persons do not have fears that this could be the case in the event of an unplanned event (Corrigan, 2024). For potential impacts to marine species of cultural significance or that provide a traditional food source (i.e. marine mammals, marine turtles, fish [including sharks and rays]), refer to the assessment for threatened, migratory or local fauna.

Likelihood

B – Unlikely

The pathways for IMS introduction are well known; consequently, standard preventive measures are proposed. The ability for invasive marine species to colonise a habitat depends on several environmental conditions. It has been found that highly disturbed environments (such as marinas) are more susceptible to colonisation than are open-water environments, where the number of dilutions and the degree of dispersal are high (Paulay et al., 2002). IMS are more likely to populate shallower areas with favourable substrates. IMS are more likely to populate shallower areas with favourable substrates. IMS are more likely to populate shallower areas with favourable substrates, such as on shoals and reefs. The closest sensitive habitat (shoal) that may provide suitable habitat is more than 8 km away from the GEP pipeline route in the OA and Shepparton Shoal is approximately 23 km from the OA. The OA is distant from coastal habitats, approximately 30 km from Tiwi Islands. With control measures in place to reduce the risk of introduction of IMS, the likelihood of introducing an IMS is considered unlikely.

**Residual risk** The residual risk is considered Low.

### 7.2.5 Demonstration of as low as reasonably practicable

Vessels and submersible equipment are required for the Activity. There are no alternatives to the use of activity vessels and equipment that are feasible in order to undertake the Activity. The risks from IMS are well understood, and with the proposed control measures, the activity will comply with relevant regulations and guidelines.

All reasonably practicable control measures have been reviewed and those adopted are considered appropriate to manage the residual risk to a Low level. The proposed management controls are in accordance with the Santos risk management criteria and are considered appropriate to manage the risk to ALARP.



### 7.2.6 Acceptability evaluation

Is the risk ranked between Very Low to Medium?	Yes – residual risk is ranked Low.	
Is further information required to validate the consequence assessment?	No – potential impacts and risks well understood through the information available. Extensive marine studies have been completed within the OA to inform the assessment.	
Are risks and impacts consistent with the principles of ESD?	Yes – activity evaluated in accordance with Santos' Offshore Division Environmental Hazard Identification and Assessment Guideline which considers principles of ESD:	
	<ul> <li>While the nature and scale of impacts have the potential to result in lasting change to benthic community dynamics, the controls that will be implemented reduce the risk to an acceptable level.</li> </ul>	
	Conservative assumptions have been applied to the impact assessment, including assuming conditions are conducive for IMS to establish and that vessels mobilised are a vector for IMS.	
Have the acceptable levels of impact and risks been informed by relevant species recovery plans, threat abatement plans, conservation advice, wildlife conservation plans and Australian marine park zoning objectives)?	Yes – Control measures implemented will reduce the risk of IMS introduction to species identified in the following relevant species recovery plans, conservation advice, wildlife conservation plans and other management plans/guidelines, as also set out in Table 3-13. <b>Conservation advice:</b>	
	<ul> <li>Approved Conservation Advice for <i>Pristis clavata</i> (Dwarf Sawfish) (DEWHA, 2009)</li> </ul>	
	• Approved Conservation Advice for <i>Green Sawfish</i> (DEWHA, 2008c)	
	<ul> <li>Approved Conservation Advice for <i>Pristis pristis</i> (largetooth sawfish) (TSSC, 2014b)</li> </ul>	
	<ul> <li>Approved Conservation Advice for <i>Glyphis garricki</i> (northern river shark) (TSSC, 2014a)</li> </ul>	
	<ul> <li>Approved Conservation Advice for <i>Rhincodon typus</i> (whale shark) (TSSC, 2015g)</li> </ul>	
	Recovery plans:	
	<ul> <li>Recovery Plan for the White Shark (<i>Carcharodon carcharias</i>) (Department of Sustainability, Environment, Water, Population and Communities (DSEWPaC, 2013)</li> </ul>	
	<ul> <li>The Wildlife Conservation Plan for Seabirds (COA, 2020) identifies disease, pathogens and invasive species as a threat Santos considers the impacts of IMS to be consistent with this Plan.</li> </ul>	
	Recovery plans / conservation advice for other species that may occur in the OA do not identify invasive species or disease as a key threat or have explicit relevant objectives or management actions related to invasive species or disease. While several plans identify habitat modification (which could occur as a result of IMS establishing) as a threat to marine fauna, significant impacts are not predicted for this Activity and IMS is not identified as a specific threat. The OA does not intersect any AMP	
	The objectives and actions of these publications were considered during impact and risk assessments. The controls outlined in Table 7-3 Section 7.2.3 are consistent with the objectives of the material listed above and Santos considers the risk of introducing IMS to be consistent with these objectives.	
Are performance outcomes, control measures and associated performance standards consistent with legal and regulatory requirements?	Yes – management consistent with the <i>Biosecurity Act 2015</i> , Australian Ballast Water Management Requirements: Version 8 (DAWE, 2020a), Australian biofouling management requirements (DAFF, 2023), Offshore Installations – Biosecurity Guide (DAFF, 2023a), Marine Order 98 (Marine Pollution – anti-fouling systems), International Maritime Organization (IMO) Guidelines for the Control and Management of Ships' Biofouling to Minimize the Transfer of Invasive Aquatic Species (2011), National Biofouling Management Guidelines for The Petroleum Production and Exploration Industry (Marine Pest Sectoral Committee, 2009), and the International Convention for the Control and Management of Ships' Ballast Water and Sediments. Through acceptance of this GEP Coastal Waters OEMP, legislative and	
	regulatory requirements will be met as per Appendix B. The GEP Coastal Waters OEMP is also compliant with conditions of	



	EPBC approval EPBC 2022/09372.
Are performance outcomes, control measures and associated performance standards consistent with Santos' Environment, Health and Safety Policy?	Yes – aligns with Santos' Environment, Health and Safety Policy (Appendix A).
Are performance outcomes, control measures and associated performance standards consistent with industry standards?	Yes – the most recent and comparable EPs were reviewed for consistency with the performance outcomes, control measures and associated performance standards proposed in this GEP Coastal Waters OEMP.
Have performance outcomes, control measures and associated performance standards taken into consideration Relevant Person feedback	Yes – requests from Relevant Persons relating to IMS management and potential environmental impacts have been considered.
Are performance standards such that the impact or risk is considered to be ALARP?	Yes – ALARP assessment conducted, with no additional control measures adopted.

The residual risk of an unplanned introduction of IMS is assessed as Low. Based on an assessment of Santos' acceptability criteria and with the control measures in place, potential risks are considered acceptable.



### 7.3 Marine fauna interaction

### 7.3.1 Description of event

Event	The physical presence of Activity vessels and helicopters/ROVs within the OA results in the potential for marine fauna interactions.
	There is the potential for vessels to interact with marine fauna, including cetaceans, fish, marine reptiles, and seabirds. The main potential for interaction is through vessel collision with large, slow-moving cetaceans.
	Marine fauna interaction may also occur from helicopter collision, during take-off and landing.
	Activity vessel speeds are generally slow due to IMMR requirements and will also be limited to ≤8 knots within the OA.
Extent	Marine fauna interaction will be localised within the OA, in the immediate vicinity of vessels or helicopters, while moving. While impact to individual marine fauna may occur, an impact at a population or ecosystem level is not anticipated.
Duration	Vessel activities will be infrequent, as per operational requirements for specific campaigns within the OA.
	IMMR vessel presence occur infrequently, typically for approximately 7 to 30 days in duration, every three to five years, or as needed.

### 7.3.2 Nature and scale of environmental impacts

**Potential receptors:** threatened, migratory fauna or local fauna (marine mammals, marine turtles, whale sharks, seabirds); socioeconomic receptors (tourism, recreation) and cultural features (such as totem species and cultural food sources). Threatened, Migratory fauna or local fauna (marine mammals, marine turtles, whale sharks, seabirds, migratory shorebirds) and socio-economic receptors via risks to fauna (tourism, recreation, and cultural features).

Movement of vessels in the OA introduces the potential for interaction with marine fauna present at the same location during operations.

Marine fauna in surface waters that would be most at risk from vessel collision include marine mammals, birds, marine turtles, and whale sharks. Other faster-moving species are likely to avoid or not be impacted by the presence of vessels. Consultation on other Barossa EPs identified that some marine fauna may have cultural significance.

Vessel speed has been demonstrated to be a key factor in relation to collision with marine fauna, particularly cetaceans and turtles, with faster moving vessels posing a greater collision risk than slower vessels.

### 7.3.2.1 Marine mammals

Humpback whales are indicated as one of the most frequently reported whale species involved in vessel strikes worldwide (Laist et al., 2001; Jensen and Silber, 2004). This observation is supported by Australian studies referenced in the National Strategy for Reducing Vessel Strike on Cetaceans and other Marine Megafauna (CoA, 2017a). Increased vessel numbers (Silber and Bettridge, 2012) are not only a threat to humpback whales in relation to vessel strikes but also in relation to disturbance and displacement from key habitats. Although there may be individual humpback whales within the OA, the primary migratory route for humpback whales is near the Kimberley coastline and up to Camden Sound, located more than 710 km south-west of the OA. Therefore, it is unlikely that activity vessels will interact with this species.

Similarly, vessel strike is also recognised by the Conservation Advice for *Balaenoptera borealis* (sei whale) (TSSC, 2015b), Conservation Advice for *Balaenoptera physalus* (fin whale) (TSSC, 2015c) and Conservation Management Plan for the Blue Whale 2015–2025 (CoA, 2015a). The blue, sei and fin whales have a wide distribution throughout offshore waters and, therefore, may pass through the OA in low numbers. Vessel speed has been demonstrated to be a key factor in relation to collision with marine fauna, particularly cetaceans, with faster-moving vessels posing a greater collision risk than slower vessels (Laist et.al., 2001; Jensen and Silber, 2003; Hazel, 2009). Laist et al. (2001) suggest that the most severe and lethal injuries to cetaceans are caused by vessels travelling at 14 knots or faster. However, considering the relatively slow vessel speeds within the OA and the mobility of whale species, it is unlikely that activity vessels will adversely interact with any individuals.

Collisions between vessels and cetaceans are most frequent on continental shelf areas where high vessel traffic and cetacean habitat occur simultaneously (Simmonds et al., 2004). There have been recorded instances of cetacean deaths as a result of vessel collisions in Australian waters (e.g. a Bryde's whale in Bass Strait in 1992) (Simmonds et al., 2004), although the data indicate this is likely to be associated with container ships and fast ferries. Some cetacean species, such as humpback whales, can detect and change course to avoid a vessel (Simmonds et al., 2004).

As presented in the National Strategy for Reducing Vessel Strike on Cetaceans and other Marine Megafauna (CoA, 2017a), most reported vessel collisions for whales in Australian waters between 1990 and 2015 have occurred along eastern or south-eastern Australia, with no reported incidents in NT waters. The International Whaling Commission has compiled a database of the worldwide occurrence of vessel strikes to cetaceans, within which Australia constitutes approximately 7% (35 reports) of the reported worldwide (approximately 471 reports) vessel strike records involving large whales (Peel et al., 2018).

Whales' reactions to approaching vessels are variable. Some species remain motionless when close to a vessel, while others are known to be curious and often approach slow-moving or stationary vessels, although they generally do not approach and sometimes avoid faster-moving vessels (Richardson et al., 1995).

While the PMST identified that dugongs or dugong habitat may occur in the OA, dugongs prefer shallow tidal and subtidal seagrass meadows less than 10 m deep, and therefore interaction is expected to be unlikely within the OA, which is ~30 km from the nearest shoreline and >40 m deep, and limited to transiting individuals (Cardno, 2015).

Dolphins (Australian snubfin dolphin, Australian humpback, spotted bottlenose) may transit through the OA; therefore, collisions between activity vessels and dolphin species are possible. However, collisions with dolphins are very infrequent due to the high mobility of these smaller cetaceans, allowing them to avoid vessels. The closest dolphin BIA is the Australian humpback dolphin (breeding), greater than 37 km from the OA. It is noted that dolphins are naturally inquisitive marine mammals, some of which are often attracted to vessels underway (e.g. commonly 'bow ride' with vessels).

### 7.3.2.2 Marine reptiles

The Recovery Plan for Marine Turtles in Australia 2017–2027 recognises increased vessel traffic as one of several key impacts on marine turtles (CoA, 2017b), with vessel disturbance posing a risk to flatback turtles. The plan also notes that while a vessel strike can be fatal for an individual turtle, vessels strike (as a standalone threat) has not been shown to cause declines at a population or stock level and have considered vessel disturbance to be of minor consequence to turtle populations in the NT (DoEE, 2017).

Marine turtle mortality due to vessel strike was identified as an issue in Queensland waters in the Recovery Plan for Marine Turtles in Australia 2017–2027 (CoA, 2017b). However, turtles appear to be more vulnerable to vessel strike in areas of high urban population where incidents with recreational vessel are higher. The approved Conservation Advice for *Dermochelys coriacea* (Leatherback Turtle) (DEWHA, 2008b) listed boat strike as a threat. Turtles spend relatively limited (3 to 6%) time at the surface, with dive times generally lasting 15 to 60 minutes (Milton and Lutz, 2003; cited in Woodside Energy Limited, 2014).

Marine turtles are highly mobile and, given the low speeds of activity vessels, are likely to be able to move from an area where there are vessels. Marine turtles make extensive migrations through the broader region; and it is possible individual turtles of any of the region's species may be encountered in the OA; however, the OA does not contain any significant feeding, breeding or aggregation areas for marine turtles.

There is an internesting BIA for flatback turtles within the OA, which may increase the number of individuals from June to September. Olive ridley turtles are likely to have an increase in the number of individuals from April to August. These periods increase the risk of vessel strikes. The Recovery Plan for Marine Turtles in Australia 2017–2027 defines a 60 km internesting buffer around the Tiwi Islands, which also intersects the OA.

Other turtles, such as green, hawksbill, and loggerhead, may also be within the OA. Internesting olive ridley and flatback turtles are expected to be concentrated in relatively shallow coastal waters (<30 m deep) around nesting beaches. Benthic habitat within the 30 m isobath around the Tiwi Islands is broadly represented and the OA exceeds a water depth of 30 m, ranging from 40–50 m.

Support and supply vessel for the activity will maintain speeds of ≤8 knots within the OA and as such the risk of vessel strike is strongly reduced. The risk of coming into contact with turtles is low as turtles are expected to dive or move away from the activity vessels. Consequently, the likelihood of a vessel strike and injury/mortality to individual turtles within the OA is considered unlikely.

The Recovery Plan for Marine Turtles in Australia 2017–2027 notes that while a vessel strike can be fatal for an individual turtle, vessel strikes (as a standalone threat) have not been shown to cause declines at a population or stock level and have considered vessel disturbance to be of minor consequence to turtle populations in the NT (CoA, 2017b).

Individual sea snakes and crocodiles may transit through the OA. If a vessel strike was to occur, it is unlikely to threaten the overall viability of either population.



### 7.3.2.3 Sharks, rays, and other fish

Most sawfish, sharks, rays and other fish identified as potentially occurring within the OA are not considered at risk of vessel strike as they largely occur on or near the seabed and are not expected to come to the surface, except the giant manta ray and whale shark.

The giant manta ray is oceanic and known to feed on plankton, so it may occasionally be close to the sea surface. However, ~73% of its diet is from deep water sources (Burgess et al., 2016). The giant manta ray is not expected to come to the surface within the OA frequently and is highly mobile (therefore able to avoid vessels). Therefore, vessel collisions with giant manta rays are considered improbable.

The whale shark BIA does not overlap the OA and therefore significant numbers are not expected to be encountered. Conservation Advice for *Rhincodon typus* (whale shark) (TSSC, 2015g) states that vessel strike from large vessels is a threat to whale sharks. Whale sharks are at risk from vessel strikes when feeding at the surface or in shallow waters (where options to dive are limited). Whale sharks have been shown to spend approximately 25% of their time less than 2 m from the surface and more than 40% of their time in the upper 15 m of the water column (Wilson et al., 2006; Gleiss et al., 2013). The OA does not overlap known whale shark foraging areas, and whale shark presence may be transitory and of a short duration. No constraints within the OA (e.g. shallow water or shorelines) would prevent whale sharks from moving away from vessels. Vessel speed has been demonstrated to be a key factor in relation to collision with marine fauna, with faster-moving vessels posing a greater collision risk than slower vessels (Laist et.al., 2001; Jensen and Silber, 2003; Hazel, 2009).

Whale sharks, other pelagic fishes, and demersal fishes are likely to exhibit a short-term avoidance to vessels and ROVs. This is likely be initiated through the vibrations and underwater noise emitted from these activities rather than the physical presence. Such avoidance is likely to be temporary but will further reduce the potential for collisions to occur.

### 7.3.2.4 Seabirds and migratory shorebirds

The OA has no bird BIAs, but several protected species of seabirds and migratory birds may occur at times within the OA (Table 3-10).

Birds may opportunistically rest on a vessel and may be attracted to activity vessels due to lighting and vessel discharges such as macerated food waste.

The Wildlife Conservation Plan for Migratory Shorebirds suggest that disturbance from human activities to shorebirds may compromise energy reserved for migration (CoA, 2015c). Although seabirds may be attracted to activity vessels due to increased feeding opportunities, these behavioural changes are unlikely to alter population dynamics or significantly change the habitat use of birds due to the very short duration of the Activity. The Conservation Advice for *Calidris canutus* (red knot) (DCCEEW, 2024c) indicates that anthropogenic disturbance is a threat, but it relates to disturbance of important sites.

Helicopter noise within the OA is expected to elicit a behavioural response in birds to avoid collision and, given the relatively low speeds helicopters would be flying at during take-off or landing, the helicopter strike is not likely.

### 7.3.2.5 Cultural features

First Nations people maintain a continuing spiritual connection with sea country, including caring for sea country and access to cultural food sources. Sections 7.3.2.1 to 7.3.2.4 above describes the potential impacts to marine species of cultural significance including culturally significant fauna such as dreaming and totem species, including marine mammals (e.g. whales, dolphins, dugongs), marine reptiles (e.g. turtles, crocodiles), sharks, rays, other fish and birds.

An important outcome of Dr Corrigan's research is that no sacred sites or dreamings are shown to be directly impacted by the DPD project footprint, which includes this Activity OA, (although this is not to say that some persons do not have fears that this could be the case in the event of an unplanned event) (Corrigan, 2024). In the 2022 Statement of Reasons Requests, the Tiwi clan members raised their concern regarding traditional hunting of marine species and totem species. The potential impact to marine fauna is likely to be limited to transiting individuals and is unlikely to result in significant impacts to marine species at the individual or population level. As a result, marine fauna interaction is not anticipated to affect traditional hunting practices or resources.

No objections or claims were raised during First Nations people feedback about potential impacts of marine fauna interactions affecting any cultural features (excluding marine fauna species) during consultation for this GEP Coastal Waters OEMP (refer to Section 4.8). Any concerns related to the potential for impacts to cultural features from marine fauna interaction are associated with direct or indirect impacts to culturally significant marine fauna species (refer to Section 3.2.15.10).


## 7.3.3 Environmental performance outcomes and control measures

The EPOs relating to this event are:

- Vessel speeds in operational areas will not exceed applicable restrictions, to reduce the risk of physical interactions between cetaceans / marine reptiles and vessels (EPO-02).
- Zero incidents of injury / mortality of cetaceans / marine reptiles from collision with vessels (EPO-03)
- No injury, or mortality to, EPBC Act 1999 listed marine fauna or marine fauna listed as threatened species under the *Territory Parks and Wildlife Conservation Act 1976* (NT) as a result of the Activity (EPO-08)
- No significant impact to cultural features from the Activity (EPO-19).

An assessment of the environmental benefits and the potential costs or issues associated with control measures for this event are shown in Table 7-4 to demonstrate the potential impacts to this aspect are ALARP. Control measures that are adopted have associated EPSs and measurement criteria that are presented in Table 8-2. Control measures not adopted have an ALARP evaluation provided to justify this outcome.

#### Table 7-4: Control measures evaluation for marine fauna interaction

CM reference	Control measure	Environmental benefit	Potential cost/issues	Evaluation
Standard control	ol measures			
BAO-CM-6.1.1	Apply Santos' Protected Marine Fauna Interaction Procedure to vessel and helicopter activities when in vicinity of cetaceans and turtles (isolation control)	Reduces risk of physical and behavioural impacts to marine fauna from vessels as it implements EPBC Regulations – Part 8 Division 8.1 Interacting with cetaceans. If cetaceans are sighted, vessels can slow down or move away, and helicopters can increase distances from sighted fauna if required. Reduces the potential impacts to culturally significant marine species, including totemic species, such as marine turtles and marine mammals.	Potential delay in vessel and helicopter movement, increasing activity duration and costs to Santos. Cost associated with implementing procedures. Regulatory requirements under EPBC Regulations 2000.	Adopted
BAO-CM-6.1.2	Vessels equipped and crewed in accordance with Australian maritime requirements, including Marine Order 30 (Prevention of Collisions) and Marine Order 21 (Safety and Emergency Arrangements) (administrative control)	Ensures contracted vessels are operated, maintained, and crewed in accordance with industry standards and regulatory requirements. Ensures vessels meet Marine Assurance Standards to reduce the likelihood of vessel collision (such as minimum and working lighting for maritime safety). Demonstrates appropriately trained and competent personnel are used to navigate vessels to reduce interaction with marine fauna.	Costs are expected as part of standard procedure.	Adopted – benefit of assuring vessels outweighs procedure compliance costs.
BAO-CM-6.4.5	HSE inductions will include applicable environmental requirements (administrative control)	Ensures that crew and helicopter operators are aware of the stringent Coastal Waters OEMP, Santos, and legislative requirements.	Administrative costs to update existing Santos procedure and induction materials, and train personnel.	Adopted



CM reference	Control measure	Environmental benefit	Potential cost/issues	Evaluation
Additional cont	rol measures			
BAO-CM-6.6.6	Vessel speed restrictions within 500 m around IMMR vessels and campaign vessels (substitute control)	Restricting vessel speeds within the OA to ≤8 knots reduces the likelihood and consequence (causing harm) of collisions as fauna have longer to detect and avoid the vessel. Reduces the potential impacts to culturally significant marine species, including totemic species, such as marine turtles and marine mammals.	Administrative costs to update existing Santos procedure and induction materials, and train personnel.	Adopted
N/A	Adopt further measures to those outlined in 'EPBC Regulations 2000 — Part 8 Division 8.1 during peak periods of ecological sensitivity; for vessels outlined in the Australian National Guidelines for Whale and Dolphin Watching (DoEE, 2017). (administrative control)	Potentially provides an additional level of protection of marina fauna.	Administrative costs to update existing Santos procedure and induction materials, and train personnel.	Not adopted – the existing controls ensure compliance with legislation. No additional relevant controls have been identified in government or industry guidelines.
N/A	Manage the timing of the Activity to avoid sensitive periods (administration control)	Potential to reduce interaction and disturbance of marine fauna by avoiding sensitive periods or periods of higher activity.	Protected marine fauna species are present in low numbers year-round; therefore, avoidance is not feasible.	Not adopted - the high financial cost would be grossly disproportionate to negligible environmental benefits.
N/A	Restrict vessel operating speeds in the OA (administrative control)	Reduces consequence of collisions and likelihood (causing harm) as fauna have longer to detect and avoid the vessel.	Administrative costs to update existing Santos procedure and induction materials and train personnel. Compliance with EPBC Regulations – Part 8 Division 8.1 Interacting with cetaceans already restricts vessel speeds appropriately when marine fauna is sighted.	Not adopted – not considered necessary, given there are very few marine fauna aggregation areas, migration pathways or BIAs near the OA, noting as per BAO-CM-6.1.1, vessels will comply with EPBC Regulations – Part 8 Division 8.1 Interacting with cetaceans (and applied for marine turtles), through implementation of the Santos Protected Marine Fauna Interaction and Sighting Procedure. As per BAO-CM- 6.6.6 vessel



CM reference	Control measure	Environmental benefit	Potential cost/issues	Evaluation
				speed will be restricted within 500m around the IMMR vessels and campaign vessels.
N/A	Dedicated marine mammal observer on vessels (EPBC Policy Statement 2.1 Part B). (administrative control)	Improved ability to spot and identify marine fauna at risk of collision (that may cause harm).	Additional cost of contracting marine mammal observer on vessels.	Not adopted – likelihood of encountering marine fauna is too low to justify the additional cost of marine mammal observers. Personnel can observe for marine fauna when piloting vessels. Cost would be grossly disproportionate to negligible environmental benefits.
N/A	Activities to only occur during daylight hours. (elimination control)	Potential vessel fauna collisions are decreased while the vessel is stationary when visibility is low at night.	Vessels are required to conduct and support 24-hour operations to meet operational needs (such as IMMR). Restricting operations would increase the duration of the Activity, resulting in significant financial costs. No other maritime industry has such a restriction.	Not adopted – the high financial cost would be grossly disproportionate to negligible environmental benefits.

#### 7.3.4 Environmental impact and risk assessment

Receptors	• Threatened, Migratory fauna or local fauna (marine mammals, marine turtles, whale sharks, seabirds)
	Socioeconomic
	Cultural features
Consequence	II – Minor

In the event of a collision with fauna, there is the potential for injury or death to an individual. The number of receptors present in the OA is expected to be limited to a small number of transient individuals.

#### Marine mammals

Nine species of marine mammals may occur within the OA. Blue, sei, fin, Bryde's and humpback whales may transit through the OA and, while impact to an individual may occur, an impact at a population or ecosystem level is not anticipated.

The Indo-Pacific humpback dolphin breeding, calving, foraging BIA located in Darwin Harbour is the closest BIA to the OA, approximately 20 km away from the boundary. The snubfin dolphin breeding and calving BIA, also located in Darwin Harbour, is approximately 46 km away from the OA boundary. Considering the relatively slow vessel speeds within the OA, and the mobility of these dolphin species, it is highly unlikely activity vessels will adversely interact with any individuals.

While some species of marine mammals may be present in the OA in greater numbers at certain times of the year, the overall numbers are low. Considering this, and the wide distribution of the species, impact at a population is not anticipated.

While dugongs may transit through the OA, they spend most of their time in shallow tidal and subtidal seagrass meadows, which are not present within the OA. If any vessel interaction does occur, it is unlikely to threaten the overall viability of the population.



#### Marine reptiles

Marine turtles make extensive migrations throughout the region, and it is possible individual turtles of any of the species known from the region may be encountered in the OA.

The OA overlaps the internesting buffer HC for flatback turtles, overlaps the internesting BIA for flatback turtles, and is adjacent (distance of 11 km) to the internesting HC for olive ridley turtles. There may be an increase in the number of individual animals in the OA between June to September for flatback turtles and April to August for olive ridley turtles that are at risk from a vessel strike. However, the risk of coming into contact with turtles is low as it is expected turtles will dive or move away from the vessels. While impact to an individual may occur, an impact at a population or ecosystem level is not anticipated.

#### Sharks, rays and fish

Boat strike is recognised by the Approved conservation Advice for *Rhincodon typus* (whale shark) as one of the threats to their recovery (TSSC, 2015g). The OA is more than 400 km away from the nearest BIA for whale sharks and, given the offshore location, large numbers of species are not anticipated. It is possible, however, individuals may transit through the OA and, while impact to an individual may occur, an impact at a population or ecosystem level is not anticipated.

Likelihood	B – Unlikely				
The likelihood of marine fauna interaction resulting in injury or death is considered possible in the OA, given the					
implementation of the Santos procedure for interacting with marine fauna. The OA overlaps areas of increased marine fauna					

abundance; however, there remains a tendency for marine fauna to move away from vessels.

**Residual risk** The residual risk is considered Very Low.

### 7.3.5 Demonstration of as low as reasonably practicable

No alternative options to the use of the vessels are possible in order to undertake the activity.

All reasonably practicable control measures have been reviewed and those adopted are considered appropriate to manage the residual risk to a Very Low level. The proposed management controls are in accordance with the Santos risk management criteria and are considered appropriate to manage the risk to ALARP.

## 7.3.6 Acceptability evaluation

Is the risk ranked between Very Low to Medium?	Yes – residual risk is ranked Very Low.	
Is further information required to validate the consequence assessment?	No – potential impacts and risks are well understood through the information available.	
Are the risks and impacts consistent with the principles of ecological sustainable development?	Yes – activity evaluated in accordance with Santos' Offshore Division Environmental Hazard Identification and Assessment Guideline which considers principles of ESD:	
	The impacts associated with marine fauna interaction have the potential to occur to a small number of an overall population and population-level impacts will not occur so, the event does not result in 'threats of serious or irreversible harm' as detailed within the EPBC Act and biodiversity and ecological integrity will be maintained.	
	Conservative assumptions on scale of impact have been applied including a conservative assumption on marine fauna presence.	
	The health, diversity and productivity of the environment will be maintained, including for future generations.	
Have the acceptable levels of impact and risks been informed by relevant species recovery plans, threat abatement plans, conservation	Yes – consistent with relevant species recovery plans, conservation advice, wildlife conservation plans and management actions set out in Table 3-13.	
advice, wildlife conservation plans and Australian	Conservation Advice:	
marine park zoning objectives)?	<ul> <li>Conservation Advice for Balaenoptera borealis (sei whale) (TSSC, 2015b)</li> </ul>	
	<ul> <li>Conservation Advice for <i>Balaenoptera physalus</i> (fin whale) (TSSC, 2015c)</li> </ul>	
	<ul> <li>Conservation Advice for Calidris canutus (red knot) (DCCEEW, 2024c)</li> </ul>	
	<ul> <li>Conservation Advice for <i>Rhincodon typus</i> (whale shark) (TSSC, 2015g)</li> </ul>	
	<ul> <li>Approved Conservation Advice for <i>Dermochelys coriacea</i> (Leatherback Turtle) (DEWHA, 2008b)</li> </ul>	
	Recovery Plans:	
	<ul> <li>National Strategy for Reducing Vessel Strike on Cetaceans and other Marine Megafauna (CoA, 2017a)</li> </ul>	

	<ul> <li>Conservation Management Plan for the Blue Whale – A Recovery Plan under the Environment Protection and Biodiversity Conservation Act 2015–2025 (CoA, 2015a) identifies vessel collisions as a threat to blue whales: 'Action A4: minimising vessel collisions by ensuring the risk of vessel strikes on blue whales is considered when assessing actions that increase vessel traffic in areas where blue whales occur and, if required appropriate mitigation measures are implemented; and ensure all vessel strike incidents are reported in the National Ship Strike database'. The adoption of EPO-03, BAO-CM-6.1.1 and BAO-CM-6.6.6 reduces potential impacts, hence is considered not inconsistent with the objectives of this management plan.</li> </ul>
	<ul> <li>Recovery Plan for Marine Turtles in Australia 2017–2027 (CoA, 2017b)</li> </ul>
	Wildlife Conservation Plan for Seabirds (CoA, 2020)
	Wildlife Conservation Plan for Migratory Shorebirds (CoA, 2015c).
	Recovery plans / conservation advice for other species that may occur in the OA do not identify vessel or anthropogenic disturbance as a key threat or have explicit relevant objectives or management actions related to vessel or anthropogenic disturbance. The OA does not intersect any AMP or protected area.
	The objectives and actions of these publications were considered during impact and risk assessments. The controls outlined in Table 7-4 Section 7.3.3 are not inconsistent with the objectives of the material listed above and Santos considers the risk of marine fauna interactions to be not inconsistent with these objectives.
Are performance outcomes, control measures and associated performance standards consistent with legal and regulatory	Yes – management is consistent with EPBC Regulations Part 8. Through acceptance of this GEP Coastal Waters OEMP, legislative and regulatory requirements will be met as per Appendix B.
requirements?	The GEP Coastal Waters OEMP is also compliant with the conditions of EPBC approval EPBC 2022/09372 (Appendix B).
Are performance outcomes, control measures and associated performance standards consistent with Santos' Environment, Health and Safety Policy?	Yes – aligns with Santos' Environment, Health, and Safety Policy (Appendix A).
Are performance outcomes, control measures and associated performance standards consistent with industry standards?	Yes – the most recent and comparable Operations EPs accepted by NOPSEMA have been reviewed for consistency with the performance outcomes, control measures and associated performance standards proposed in this GEP Coastal Waters OEMP.
Have performance outcomes, control measures and associated performance standards taken into consideration Relevant Person feedback	Yes – requests from Relevant Persons relating to management of vessel movement and potential environmental impacts have been considered.
Are performance standards such that the impact or risk is considered to be ALARP?	Yes – ALARP assessment conducted, with no additional control measures adopted.

The residual risk of marine fauna interaction is assessed as Very Low. Based on an assessment of Santos' acceptability criteria and with the control measures in place, potential risks are considered acceptable.



## 7.4 Minor release of hydrocarbons and chemicals

## 7.4.1 Description of event

<ul> <li>fuel for portable/deck equipment         <ul> <li>hydraulic fluid (e.g., for ROVs)</li> <li>paints and lubricants</li> <li>miscellaneous chemicals (e.g., cleaning fluids)</li> </ul> </li> <li>An accidental release of minor volumes of chemicals and hydrocarbon liquids into the marine environment can potentially occur from:         <ul> <li>mechanical failure of equipment, such as tank or pipework failure</li> <li>inadequate handling and storage</li> <li>insufficient fastening or inadequate bunding</li> <li>firrefighting foam released during an unplanned incident</li> <li>bunkering</li> </ul> </li> <li>A release of hydrocarbon liquids or chemicals may result in impacts to water quality and hence sensitive environmental receptors.</li> <li>Extent</li> <li>Small spills may occur when the chemicals or hydrocarbons are in use or from leaks within the storage area and can potentially be released into the marine environment.</li> <li>The maximum volume of chemicals that could be released during routine operations is likely to be incidential and minor, with bunding in place to retain substances in the event of a leak. Operational experience indicates typical minor spill volumes are &lt;10 L.</li> <li>Dilution from discharges in open waters is rapid, with h 1 in 1,000 dilution usually occurring within 30 minutes (Costello and Read, 1994). If the spill is not contained on deck, a release to the marine environment (see Section 7.6 for information on unplanned release of MIDO). Hydraulic fluids are medium oils of light to moderate viscosity with a relatively rapid sperading rate, and will dissipate quickly, particularly in high sea states. Lubricating oils are more viscous, with a slightly slower rate of spread following a spill. Hydraulic fluids are used in various equipment, such as Aframes, cranes, winches and ROVs, Failure of hydraulic lines may result in the loss of h</li></ul>	Event	Vessels undertaking the Activity will routinely have a range of chemicals and hydrocarbons onboard, including:
<ul> <li>hydraulic fluid (e.g., for ROVs)</li> <li>paints and lubricants</li> <li>miscellaneous chemicals (e.g., cleaning fluids)</li> <li>An accidental release of minor volumes of chemicals and hydrocarbon liquids into the marine environment can potentially occur from:         <ul> <li>mechanical failure of equipment, such as tank or pipework failure</li> <li>inadequate handling and storage</li> <li>isufficient fastening or inadequate bunding</li> <li>firefighting foam released during an unplanned incident</li> <li>bunkering</li> <li>A release of hydrocarbon liquids or chemicals may result in impacts to water quality and hence sensitive environmental receptors.</li> </ul> </li> <li>Extent</li> <li>Small spills may occur when the chemicals or hydrocarbons are in use or from leaks within the storage area and can potentially be released into the marine environment. The maximum volume of chemicals that could be released during routine operations is likely to be incidental and minor, with bunding in place to retain substances in the event of a leak. Operational experience indicates typical minor spill volumes are &lt;10 L.</li> <li>Dilution from discharges in open waters is rapid, with 1 in 1,000 dilution usually occurring within 30 minutes (Costello and Read, 1994). If the spill is not contained on deck, a release to the marine environment would likely disperse rapidly within the OA.</li> <li>Hydraulic fluids and lubricating fluids behave similarly to marine disel of IMOD). Hydraulic fluids are medium oils of light to moderate viscosity with a relatively rapid spreading rate, and will dissipate quickly, particularly in high sea states. Lubricating oils are more viscous, with a slightly slower rate of spread following a spill. Hydraulic fluids use released into the marine environment. Operational experience indicates typical volumes released tout on than icli line are are 20 L.</li> <ul></ul></ul>		fuel for portable/deck equipment
<ul> <li>paints and lubricants</li> <li>miscellaneous chemicals (e.g., cleaning fluids)</li> <li>An accidental release of minor volumes of chemicals and hydrocarbon liquids into the marine environment can potentially occur from:         <ul> <li>mechanical failure of equipment, such as tank or pipework failure</li> <li>inadequate handling and storage</li> <li>insufficient fastening or inadequate bunding</li> <li>firefighting foam released during an unplanned incident</li> <li>bunkering</li> </ul> </li> <li>A release of hydrocarbon liquids or chemicals may result in impacts to water quality and hence sensitive environmental receptors.</li> <li>Small spills may occur when the chemicals or hydrocarbons are in use or from leaks within the storage area and can potentially be released into the marine environment.</li> <li>The maximum volume of chemicals that could be released during routine operations is likely to be incidental and minor, with bunding in place to retain substances in the event of a leak. Operational experience indicates typical minor spill volumes are &lt;10 L.</li> <li>Dilution from discharges in open waters is rapid, with 1 in 1,000 dilution usually occurring within 30 minutes (Costello and Read, 1994). If the spill is not contained on deck, a release to the marine environment (see Section 7.6 for information on unplanned release of MDO). Hydraulic fluids are medium oils of light to moderate viscosity with a relatively rapid spreading rate, and will dissipate quickly, particularly in high see states. Lubricating oils are more viscous, with a slightly slower rate of spread following a spill. Hydraulic fluids are used in various equipment, such as A-frames, cranes, winches and ROVs. Failure of hydraulic fluids are used in various equipment, such as A-frames, cranes, winches and ROVs. Failure of hydraulic fluids are used in various equipment, such as a the hose or fittings, combined with a failure in proce</li></ul>		hydraulic fluid (e.g., for ROVs)
<ul> <li>miscellaneous chemicals (e.g., cleaning fluids)</li> <li>An accidental release of minor volumes of chemicals and hydrocarbon liquids into the marine environment can potentially occur from:</li> <li>mechanical failure of equipment, such as tank or pipework failure</li> <li>inadequate handling and storage</li> <li>insufficient fastening or inadequate bunding</li> <li>firefighting foam released during an unplanned incident</li> <li>bunkering</li> <li>A release of hydrocarbon liquids or chemicals may result in impacts to water quality and hence sensitive environmental receptors.</li> </ul> Extent Small spills may occur when the chemicals or hydrocarbons are in use or from leaks within the storage area and can potentially be released into the marine environment. The maximum volume of chemicals that could be released during routine operations is likely to be incidental and minor, with bunding in place to retain substances in the event of a leak. Operational experience indicates typical minor spill volumes are <10 L. Dilution from discharges in open waters is rapid, with 1 in 1,000 dilution usually occuring within 30 minutes (Costello and Read, 1994). If the spill is not contained on deck, a release to the marine environment (see Section 7.6 for information on unplanned released oil (MDO) when spilt in the marine environment (see Section 7.6 for information on unplanned release of MDO). Hydraulic fluids are medium oils of light to moderate viscosity with a relatively rapid spreading rate, and will dissipate quickly, particularly in high see states. Lubricating oils are more viscous, with a slightly slower rate of spread following a spill. Hydraulic fluids are used in various equipment, such as A-frames, cranes, winches and ROVs. Failure of hydraulic fluids are used in various equipment, such as A-frames, cranes, winches and ROVs. Failure of hydraulic fluids are used in various equipment, such as A-frames, cranes, winches and ROVs. Failure of hydraulic fl		paints and lubricants
An accidental release of minor volumes of chemicals and hydrocarbon liquids into the marine environment can potentially occur from:         • mechanical failure of equipment, such as tank or pipework failure         • inadequate handling and storage         • insufficient fastening or inadequate bunding         • firefighting foam released during an unplanned incident         • bunkering         A release of hydrocarbon liquids or chemicals may result in impacts to water quality and hence sensitive environmental receptors.         Extent         Small spills may occur when the chemicals or hydrocarbons are in use or from leaks within the storage area and can potentially be released during routine operations is likely to be incidental and minor, with bunding in place to retain substances in the event of a leak. Operational experience indicates typical minor spill volumes are <10 L.         Dilution from discharges in open waters is rapid, with 1 in 1,000 dilution usually occurring within 30 minutes (Costello and Read, 1994). If the spill is not contained on deck, a release to the marine environment would likely disperse rapidly within the OA.         Hydraulic fluids and lubricating fillid is behave similarly to marine diesel oil (MDO) when spilt in the marine environment (see Section 7.6 for information on unplanned release of MDO). Hydraulic fluids are medium oils of light to moderate viscosity with a relatively rapid spreading rate, and will dissipate quickly, particularly in high sea states. Lubricating oils are more viscous, with a slightly slower rate of spread following a spill. Hydraulic fluids are used in various equipment, such as A-frames, cranes, winches and ROVs. Failure of Hydraulic fluids are used in various equ		<ul> <li>miscellaneous chemicals (e.g., cleaning fluids)</li> </ul>
<ul> <li>mechanical failure of equipment, such as tank or pipework failure</li> <li>inadequate handling and storage</li> <li>insufficient fastening or inadequate bunding</li> <li>firefighting foam released during an unplanned incident</li> <li>bunkering</li> <li>A release of hydrocarbon liquids or chemicals may result in impacts to water quality and hence sensitive environmental receptors.</li> </ul> Extent Small spills may occur when the chemicals or hydrocarbons are in use or from leaks within the storage area and can potentially be released into the marine environment. The maximum volume of chemicals that could be released during routine operations is likely to be incidental and minor, with bunding in place to retain substances in the event of a leak. Operational experience indicates typical minor spill volumes are <10 L. Dilution from discharges in open waters is rapid, with 1 in 1,000 dilution usually occurring within 30 minutes (Costello and Read, 1994). If the spill is not contained on deck, a release to the marine environment would likely disperse rapidly within the OA. Hydraulic fluids and lubricating fluids behave similarly to marine diesel oil (MDO) when spilt in the marine environment (see Section 7.6 for information on unplanned release of MDO). Hydraulic fluids are medium oils of light to moderate viscosity with a relatively rapid spreading rate, and will dissipate quickly, particularly in high sea states. Lubricating oils are more viscous, with a slightly slower rate of spread following a spill. Hydraulic fluids are used in various equipment, such as A-frames, cranes, winches and ROVs. Failure of hydraulic line may result in the loss of hydraulic fluid to the environment. Operational experience indicates typical volumes released due to hydraulic fluid to the onvironment. Operational experience indicates typical volumes released into the marine environment. Chemical bunkering is a monitored event, allowing for almost immediate shutdown of the		An accidental release of minor volumes of chemicals and hydrocarbon liquids into the marine environment can potentially occur from:
<ul> <li>inadequate handling and storage</li> <li>insufficient fastening or inadequate bunding</li> <li>firefighting foam released during an unplanned incident</li> <li>bunkering</li> <li>A release of hydrocarbon liquids or chemicals may result in impacts to water quality and hence sensitive environmental receptors.</li> <li>Extent</li> <li>Small spills may occur when the chemicals or hydrocarbons are in use or from leaks within the storage area and can potentially be released into the marine environment.</li> <li>The maximum volume of chemicals that could be released during routine operations is likely to be incidental and minor, with bunding in place to retain substances in the event of a leak. Operational experience indicates typical minor spill volumes are &lt;10 L.</li> <li>Dilution from discharges in open waters is rapid, with 1 in 1,000 dilution usually occurring within 30 minutes (Costello and Read, 1994). If the spill is not contained on deck, a release to the marine environment (see Section 7.6 for information on unplanned release of MDO). Hydraulic fluids are medium oils of light to moderate viscosity with a relatively rapid spreading rate, and will dissipate quickly, particularly in high sea states. Lubricating oils are more viscous, with a slightly slower rate of spread following a spill. Hydraulic fluids are used in various equipment, such as the hose or fittings, combined with a failure in procedure to shut off fuel pumps for a period of up to three minutes, may result in approximately 10 m<sup>3</sup> of chemicals released into the marine environment. Chemical bunkering is a montived event, allowing for almost immediate shutdown of the bunkering activity.</li> <li>In the event of a fire emergency, firefighting foam will be used, which would then be discharged directly overboard or through deck drainage systems.</li> <li>The environment that may be affected for hydrocarbon liquids or chemical release d chemical or hydrocarbon induces in a decrease in water quality</li></ul>		<ul> <li>mechanical failure of equipment, such as tank or pipework failure</li> </ul>
<ul> <li>insufficient fastening or inadequate bunding</li> <li>firefighting foam released during an unplanned incident</li> <li>bunkering</li> <li>A release of hydrocarbon liquids or chemicals may result in impacts to water quality and hence sensitive environmental receptors.</li> <li>Extent</li> <li>Small spills may occur when the chemicals or hydrocarbons are in use or from leaks within the storage area and can potentially be released into the marine environment.</li> <li>The maximum volume of chemicals that could be released during routine operations is likely to be incidental and minor, with bunding in place to retain substances in the event of a leak. Operational experience indicates typical minor spill volumes are &lt;10 L.</li> <li>Dilution from discharges in open waters is rapid, with 1 in 1,000 dilution usually occurring within 30 minutes (Costello and Read, 1994). If the spill is not contained on deck, a release to the marine environment (see Section 7.6 for information on unplanned release of MDO). Hydraulic fluids are medium oils of light to moderate viscosity with a relatively rapid spreading rate, and will dissipate quickly, particularly in high sea states. Lubricating oils are more viscous, with a slightly slower rate of spread following a spill. Hydraulic fluids are used to hydraulic fluids are ranes, winches and ROVs. Failure of hydraulic lines may result in the loss of hydraulic fluid to the environment. Operational experience indicates typical volumes released due to hydraulic fluid to the environment. Operational experience indicates typical volumes released due to hydraulic fluid to the environment. Operational experience and allowing for almost immediate shutdown of the bunkering activity.</li> <li>In the event of a fire emergency, firefighting foam will be used, which would then be discharged directly overboard or through deck drainage systems.</li> <li>The environment that may be affected for hydrocarbon liquids or chemical release resulting in a decr</li></ul>		inadequate handling and storage
<ul> <li>firefighting foam released during an unplanned incident</li> <li>bunkering</li> <li>A release of hydrocarbon liquids or chemicals may result in impacts to water quality and hence sensitive environmental receptors.</li> <li>Extent</li> <li>Small spills may occur when the chemicals or hydrocarbons are in use or from leaks within the storage area and can potentially be released into the marine environment.</li> <li>The maximum volume of chemicals that could be released during routine operations is likely to be incidental and minor, with bunding in place to retain substances in the event of a leak. Operational experience indicates typical minor spill volumes are &lt;10 L.</li> <li>Dilution from discharges in open waters is rapid, with 1 in 1,000 dilution usually occurring within 30 minutes (Costello and Read, 1994). If the spill is not contained on deck, a release to the marine environment would likely disperse rapidly within the OA.</li> <li>Hydraulic fluids and lubricating fluids behave similarly to marine diesel oil (MDO) when spilt in the marine environment (see Section 7.6 for information on unplanned release of MDO). Hydraulic fluids are medium oils of light to moderate viscosity with a relatively rapid spreading rate, and will dissipate quickly, particularly in high sea states. Lubricating oils are more viscous, with a slightly slower rate of spread following a spill. Hydraulic fluids are used in various equipment, such as A-frames, cranes, winches and ROVs. Failure of hydraulic lines may result in the loss of hydraulic fluid to the environment. Operational experience indicates typical volumes released due to hydraulic line failure are &lt;20 L.</li> <li>During bunkering activities, a total rupture or failure of bunker transfer equipment, such as the hose or fittings, combined with a failure in procedure to shut off fuel pumps for a period of up to three minutes, may result in approximately 10 m<sup>3</sup> of chemicals released into the marine environment. Chemical bunkering is</li></ul>		insufficient fastening or inadequate bunding
<ul> <li>bunkering         A release of hydrocarbon liquids or chemicals may result in impacts to water quality and hence sensitive environmental receptors.     </li> <li>Extent         Small spills may occur when the chemicals or hydrocarbons are in use or from leaks within the storage area and can potentially be released into the marine environment.         The maximum volume of chemicals that could be released during routine operations is likely to be incidental and minor, with bunding in place to retain substances in the event of a leak. Operational experience indicates typical minor spill volumes are &lt;10 L.         <p>Dilution from discharges in open waters is rapid, with 1 in 1,000 dilution usually occurring within 30 minutes (Costello and Read, 1994). If the spill is not contained on deck, a release to the marine environment would likely disperse rapidly within the OA.         Hydraulic fluids and lubricating fluids behave similarly to marine diesel oil (MDO) when spilt in the marine environment (see Section 7.6 for information on unplanned release of MDO). Hydraulic fluids are medium oils of light to moderate viscosity with a relatively rapid spreading rate, and will dissipate quickly, particularly in high sea states. Lubricating oils are more viscous, with a slightly slower rate of spread following a spill. Hydraulic fluids are used in various equipment, such as A-frames, cranes, winches and ROVs. Failure of hydraulic lines may result in the loss of hydraulic fluid to the environment. Operational experience indicates typical volumes released due to hydraulic fluid to the environment. Deprational experience indicates typical volumes released into the marine environment. Chemical bunkering is a monitored event, allowing for almost immediate shutdown of the bunkering activity.         In the event of a fire emergency, frefighting foam will be used, which would then be discharged directly overboard or through deck drainage systems.        The environ</p></li></ul>		firefighting foam released during an unplanned incident
A release of hydrocarbon liquids or chemicals may result in impacts to water quality and hence sensitive environmental receptors.         Extent       Small spills may occur when the chemicals or hydrocarbons are in use or from leaks within the storage area and can potentially be released into the marine environment.         The maximum volume of chemicals that could be released during routine operations is likely to be incidental and minor, with bunding in place to retain substances in the event of a leak. Operational experience indicates typical minor spill volumes are <10 L.         Dilution from discharges in open waters is rapid, with 1 in 1,000 dilution usually occurring within 30 minutes (Costello and Read, 1994). If the spill is not contained on deck, a release to the marine environment would likely disperse rapidly within the OA.         Hydraulic fluids and lubricating fluids behave similarly to marine diesel oil (MDO) when spilt in the marine environment (see Section 7.6 for information on unplanned release of MDO). Hydraulic fluids are medium oils of light to moderate viscosity with a relatively rapid spreading rate, and will dissipate quickly, particularly in high sea states. Lubricating oils are more viscous, with a slightly slower rate of spread following a spill. Hydraulic fluids are used in various equipment, such as A-frames, cranes, winches and ROVs. Failure of hydraulic lines may result in the loss of hydraulic fluid to the environment. Operational experience indicates typical volumes released due to hydraulic line failure are <20 L.         During bunkering activities, a total rupture or failure of bunker transfer equipment, such as the hose or fittings, combined with a failure in procedure to shut off fuel pumps for a period of up to three minutes, may result in approximately 10 m <sup>3</sup> of chemicals released into the marine e		bunkering
ExtentSmall spills may occur when the chemicals or hydrocarbons are in use or from leaks within the storage area and can potentially be released into the marine environment.The maximum volume of chemicals that could be released during routine operations is likely to be incidental and minor, with bunding in place to retain substances in the event of a leak. Operational experience indicates typical minor spill volumes are <10 L.Dilution from discharges in open waters is rapid, with 1 in 1,000 dilution usually occurring within 30 minutes (Costello and Read, 1994). If the spill is not contained on deck, a release to the marine environment would likely disperse rapidly within the OA. Hydraulic fluids and lubricating fluids behave similarly to marine diesel oil (MDO) when spilt in the marine environment (see Section 7.6 for information on unplanned release of MDO). Hydraulic fluids are medium oils of light to moderate viscosity with a relatively rapid spreading rate, and will dissipate quickly, particularly in high sea states. Lubricating oils are more viscous, with a slightly slower rate of spread following a spill. Hydraulic fluids are used in various equipment, such as A-frames, cranes, winches and ROVs. Failure of hydraulic lines may result in the loss of hydraulic fluid to the environment. Operational experience indicates typical volumes released due to hydraulic fluid to the environment. Operational experience indicates typical volumes released into the marine environment. Chemical bunkering is a monitored event, allowing for almost immediate shutdown of the bunkering activity. In the event of a fire emergency, firefighting foam will be used, which would then be discharged directly overboard or through deck drainage systems. The environment that may be affected for hydrocarbon liquids or chemical release resulting in a decrease in water quality is likely to be restricted to the ar		A release of hydrocarbon liquids or chemicals may result in impacts to water quality and hence sensitive environmental receptors.
The maximum volume of chemicals that could be released during routine operations is likely to be incidental and minor, with bunding in place to retain substances in the event of a leak. Operational experience indicates typical minor spill volumes are <10 L.	Extent	Small spills may occur when the chemicals or hydrocarbons are in use or from leaks within the storage area and can potentially be released into the marine environment.
<ul> <li>Dilution from discharges in open waters is rapid, with 1 in 1,000 dilution usually occurring within 30 minutes (Costello and Read, 1994). If the spill is not contained on deck, a release to the marine environment would likely disperse rapidly within the OA.</li> <li>Hydraulic fluids and lubricating fluids behave similarly to marine diesel oil (MDO) when spilt in the marine environment (see Section 7.6 for information on unplanned release of MDO). Hydraulic fluids are medium oils of light to moderate viscosity with a relatively rapid spreading rate, and will dissipate quickly, particularly in high sea states. Lubricating oils are more viscous, with a slightly slower rate of spread following a spill. Hydraulic fluids are used in various equipment, such as A-frames, cranes, winches and ROVs. Failure of hydraulic lines may result in the loss of hydraulic fluid to the environment. Operational experience indicates typical volumes released due to hydraulic line failure are &lt;20 L.</li> <li>During bunkering activities, a total rupture or failure of bunker transfer equipment, such as the hose or fittings, combined with a failure in procedure to shut off fuel pumps for a period of up to three minutes, may result in approximately 10 m<sup>3</sup> of chemicals released into the marine environment. Chemical bunkering is a monitored event, allowing for almost immediate shutdown of the bunkering activity. In the event of a fire emergency, firefighting foam will be used, which would then be discharged directly overboard or through deck drainage systems.</li> <li>The environment that may be affected for hydrocarbon liquids or chemical release resulting in a decrease in water quality is likely to be restricted to the area surrounding Activity vessels and contained within the OA.</li> </ul>		The maximum volume of chemicals that could be released during routine operations is likely to be incidental and minor, with bunding in place to retain substances in the event of a leak. Operational experience indicates typical minor spill volumes are <10 L.
Hydraulic fluids and lubricating fluids behave similarly to marine diesel oil (MDO) when spilt in the marine environment (see Section 7.6 for information on unplanned release of MDO). Hydraulic fluids are medium oils of light to moderate viscosity with a relatively rapid spreading rate, and will dissipate quickly, particularly in high sea states. Lubricating oils are more viscous, with a slightly slower rate of spread following a spill. Hydraulic fluids are used in various equipment, such as A-frames, cranes, winches and ROVs. Failure of hydraulic lines may result in the loss of hydraulic fluid to the environment. Operational experience indicates typical volumes released due to hydraulic line failure are <20 L. During bunkering activities, a total rupture or failure of bunker transfer equipment, such as the hose or fittings, combined with a failure in procedure to shut off fuel pumps for a period of up to three minutes, may result in approximately 10 m³ of chemicals released into the marine environment. Chemical bunkering is a monitored event, allowing for almost immediate shutdown of the bunkering activity. In the event of a fire emergency, firefighting foam will be used, which would then be discharged directly overboard or through deck drainage systems. The environment that may be affected for hydrocarbon liquids or chemical release resulting in a decrease in water quality is likely to be restricted to the area surrounding Activity vessels and contained within the OA.DurationThe duration of the impact is limited (minutes to hours) to the time the released chemical or hydrocarbon takes to disperse to below harmful concentrations. IMMR vessels will twically be present in the OA for 7		Dilution from discharges in open waters is rapid, with 1 in 1,000 dilution usually occurring within 30 minutes (Costello and Read, 1994). If the spill is not contained on deck, a release to the marine environment would likely disperse rapidly within the OA.
During bunkering activities, a total rupture or failure of bunker transfer equipment, such as the hose or fittings, combined with a failure in procedure to shut off fuel pumps for a period of up to three minutes, may result in approximately 10 m³ of chemicals released into the marine environment. Chemical bunkering is a monitored event, allowing for almost immediate shutdown of the bunkering activity. In the event of a fire emergency, firefighting foam will be used, which would then be discharged directly overboard or through deck drainage systems. The environment that may be affected for hydrocarbon liquids or chemical release resulting in a decrease in water quality is likely to be restricted to the area surrounding Activity vessels and contained within the OA.DurationThe duration of the impact is limited (minutes to hours) to the time the released chemical or hydrocarbon takes to disperse to below harmful concentrations. IMMR vessels will twiceally be present in the OA for 7		Hydraulic fluids and lubricating fluids behave similarly to marine diesel oil (MDO) when spilt in the marine environment (see Section 7.6 for information on unplanned release of MDO). Hydraulic fluids are medium oils of light to moderate viscosity with a relatively rapid spreading rate, and will dissipate quickly, particularly in high sea states. Lubricating oils are more viscous, with a slightly slower rate of spread following a spill. Hydraulic fluids are used in various equipment, such as A-frames, cranes, winches and ROVs. Failure of hydraulic lines may result in the loss of hydraulic fluid to the environment. Operational experience indicates typical volumes released due to hydraulic line failure are <20 L.
In the event of a fire emergency, firefighting foam will be used, which would then be discharged directly overboard or through deck drainage systems.         The environment that may be affected for hydrocarbon liquids or chemical release resulting in a decrease in water quality is likely to be restricted to the area surrounding Activity vessels and contained within the OA.         Duration       The duration of the impact is limited (minutes to hours) to the time the released chemical or hydrocarbon takes to disperse to below harmful concentrations. IMMR vessels will twically be present in the OA for 7		During bunkering activities, a total rupture or failure of bunker transfer equipment, such as the hose or fittings, combined with a failure in procedure to shut off fuel pumps for a period of up to three minutes, may result in approximately 10 m <sup>3</sup> of chemicals released into the marine environment. Chemical bunkering is a monitored event, allowing for almost immediate shutdown of the bunkering activity.
The environment that may be affected for hydrocarbon liquids or chemical release resulting in a decrease in water quality is likely to be restricted to the area surrounding Activity vessels and contained within the OA.         Duration       The duration of the impact is limited (minutes to hours) to the time the released chemical or hydrocarbon takes to disperse to below harmful concentrations. IMMR vessels will twicely be present in the OA for 7		In the event of a fire emergency, firefighting foam will be used, which would then be discharged directly overboard or through deck drainage systems.
Duration The duration of the impact is limited (minutes to hours) to the time the released chemical or hydrocarbon takes to disperse to below harmful concentrations. IMMR vessels will typically be present in the OA for 7		The environment that may be affected for hydrocarbon liquids or chemical release resulting in a decrease in water quality is likely to be restricted to the area surrounding Activity vessels and contained within the OA.
to 30 days in duration, every three to five years, or as needed.	Duration	The duration of the impact is limited (minutes to hours) to the time the released chemical or hydrocarbon takes to disperse to below harmful concentrations. IMMR vessels will typically be present in the OA for 7 to 30 days in duration, every three to five years, or as needed.

## 7.4.2 Nature and scale of environmental impacts

**Potential receptors:** physical environment (water quality); threatened, migratory or local fauna (marine mammals, marine reptiles, seabirds, sharks, rays and other fish); socioeconomic; and cultural features.

### 7.4.2.1 Physical environment

Minor volumes of hydrocarbons or chemicals released to the marine environment may result in a reduction in water quality. Releases will likely be temporary (minutes to hours), localised and limited to surface waters surrounding the Activity vessels.

Minor hydrocarbon and chemical releases in the OA are expected to disperse and dilute rapidly given the water depths (47 - 50 m), strong currents and wave action. The small volumes of worst-case releases are such that the potential for impacts on water quality will decline rapidly with time and distance at the sea surface. Impacts to benthic habitats are not expected given the location and water depths in the OA (47 - 50 m).



### 7.4.2.2 Threatened, migratory or local fauna

Changes to water quality could potentially lead to short-term impacts on transiting marine fauna (e.g., pelagic fish, sharks, marine mammals, marine reptiles and seabirds), some of which may have cultural significance as totems or cultural food sources. Marine pollution and contamination are identified as threats in the recovery plans and conservation advice for several protected marine species, such as the Recovery Plan for Marine Turtles in Australia 2017 – 2027 (CoA, 2017b).

Relatively low numbers of marine fauna are expected around Activity vessels due to the location and water depths of the OA (47 – 50 m). Minor hydrocarbon and chemical releases in the OA are expected to disperse and dilute rapidly, such that the potential for impacts on marine fauna will decline with time and distance at the sea surface. Sublethal or lethal impacts to marine fauna, including physical coating (e.g., seabirds at the surface) from entrained or surface hydrocarbons or hazardous chemicals are considered unlikely, given the nature and limited volumes of potential releases from Activity vessels.

### 7.4.2.3 Cultural features

No First Nations people feedback was provided about the potential impacts from unplanned minor releases of hydrocarbons and chemicals on geographically specific cultural features (excluding marine fauna species) during consultations for this GEP Coastal Waters OEMP or other Barossa Gas Project EPs (refer to Section 4.8).

In accordance with First Nations people cultural beliefs, if totemic species (e.g. turtles, fish, marine mammals and birds) are impacted by the Activity some believe this in turn can impact First Nations people and make them sick. The potential impacts to culturally significant marine fauna species (such as dreaming and totem species including marine mammals, turtles, fish and birds) are assessed in Section 7.4.2.2.

### 7.4.3 Environmental performance outcomes and control measures

The EPOs relating to this event is:

- No injury, or mortality to, EPBC Act 1999 listed marine fauna or marine fauna listed as threatened species under the *Territory Parks and Wildlife Conservation Act 1976* (NT) as a result of the Activity. (EPO-08)
- Zero unplanned discharge of hazardous and non-hazardous wastes into the marine environment from the Activity. (EPO-16)
- Zero unplanned discharge of hydrocarbons or chemicals to the marine environment from the Activity. (EPO-18)
- No significant impact to cultural features from the Activity (EPO-19).

An assessment of the environmental benefits and the potential costs or issues associated with control measures for this activity are described in Table 7-5 to demonstrate that potential risks are ALARP. Control measures that are adopted have associated EPSs and measurement criteria and are presented in Table 8-2. Rejected control measures have an ALARP evaluation provided to justify their rejection.

Table 7-5: Contro	ol measures evaluation	for minor release of	hydrocarbons and chemicals
-------------------	------------------------	----------------------	----------------------------

CM reference	Control measure	Environmental benefit	Potential cost/issues	Evaluation
Standard control m	easures			
BAO-CM-6.1.2	Vessels equipped and crewed in accordance with Australian maritime requirements, including Marine Order 30 (Prevention of Collisions) and Marine Order 21 (Safety and Emergency Arrangements) (administrative control)	Ensures contracted vessels are operated, maintained and crewed in accordance with industry standards and regulatory requirements. Ensures vessels meet Marine Assurance Standards to reduce the likelihood of vessel collision (such as minimum and working lighting for maritime safety).	Costs are expected as part of standard procedure.	Adopted – benefit of assuring vessels outweighs procedure compliance costs.
BAO-CM-6.4.5	HSE inductions will include applicable environmental requirements	Ensures that crew are aware of the stringent Coastal Waters OEMP, Santos and legislative requirements.	Administrative costs to update existing Santos procedure and induction materials, and train personnel.	Adopted

CM reference	Control measure	Environmental benefit	Potential cost/issues	Evaluation
	(administrative control)			
BAO-CM-6.4.10	Barossa Facilities and vessel planned maintenance to confirm equipment integrity is maintained in accordance with manufacturers guidelines (administrative control)	Ensures vessel and equipment is maintained through routine checks, reducing the likelihood of leaks or loss of integrity. Maintenance and pre- deployment inspection on ROV completed as scheduled to reduce the risk of hydraulic fluid releases to the marine environment. Ensures bunkering equipment is maintained through routine checks via: • visual inspections • string hydrotest. Maintained bunkering equipment will reduce likelihood of loss of integrity events during transfers.	Cost of implementing procedures.	Adopted – environmental benefits of maintaining offtake equipment integrity outweigh the costs.
BAO-CM-6.7.5	Apply the Santos chemical selection process for all chemicals planned to be discharged (Section 2.7) (administrative control)	Under the procedure, CHARM-rated gold/silver and non-CHARM Group E/D chemicals managed under the OCNS, or OSPAR PLONOR list, or chemicals risk assessed by Santos and deemed environmentally acceptable, will be selected (Section 2.7). Reduces the potential impacts if accidental release occurs. Therefore, the production and other chemicals will pose little or no risk to the environment. Reduces the potential impacts to culturally significant marine species, such as marine turtles and marine mammals.	Cost of implementing procedures. Range of chemicals reduced with potentially higher costs for alternative products.	Adopted – environmental benefits of using environmentally acceptable chemicals outweigh procedural implementation and operational costs.
BAO-CM-7.1.1	Implement standards and procedures for lifting equipment to reduce risk of dropped objects during lifting (administrative control)	Impacts to the environment are reduced by preventing dropped objects and dragged objects during lifting operations.	Administrative costs to implement procedures, update induction materials and train personnel.	Adopted – environmental benefits of ensuring the procedure are followed and measures implemented outweigh the costs.
BAO-CM-7.1.2	Dropped objects (incident) management (administrative control)	Impacts to the environment are reduced by preventing dropped objects and by retrieving dropped objects unless the environmental consequences of the dropped object are negligible or there are risks to safety.	Cost of implementing procedures.	Adopted – environmental benefits outweigh costs.

CM reference	Control measure	Environmental benefit	Potential cost/issues	Evaluation
BAO-CM-7.1.3	International Maritime Dangerous Goods Code (administrative control)	Regulatory requirement that reduces the risk of an environmental incident, such as an accidental container release to sea or unintended chemical reaction.	Cost of implementing procedures.	Adopted – environmental benefits of ensuring procedures are followed outweigh the costs; plus, the control is a legislated requirement.
BAO-CM-7.1.4	Chemicals and hydrocarbons will be managed in accordance SDS to reduce risk of release to the marine environment (administrative control)	Reduces the risk of accidental discharge to sea by controlling the storage, handling, and clean-up of chemicals.	Cost of implementing procedures. Regulatory requirement to manage hazardous chemicals.	Adopted – environmental benefits of ensuring procedures are followed outweigh the costs, plus the control is a legislated requirement.
BAO-CM-7.4.1	ROV operations undertaken in accordance with good industry practice (administrative control)	Maintenance (as per manufacturer specifications) and pre- deployment inspection on ROV completed as scheduled to reduce the risk of hydraulic fluid releases to the marine environment.	Cost of implementing procedures.	Adopted – environmental benefits outweigh the costs.
BAO-CM-7.4.2	Bulk liquid transfer procedure (administrative control)	The procedure provides details about the chemical bunkering process to be undertaken. Implementing the procedure reduces the potential for release during bunkering. Requires use of dry-break coupling (on bunkering hose) and breakaway coupling, which limit the chemical losses in an emergency.	Costs associated with ensuring the procedure is in place, up to date and implemented.	Adopted – environmental benefits of ensuring the procedure is followed and measures implemented outweigh the costs.
BAO-CM-7.4.3	Vessel spill response plans (SOPEP/SMPEP) (administrative control)	Implements onboard response plans to deal with unplanned hydrocarbon releases and spills quickly and efficiently to reduce impacts to the marine environment.	Administrative costs of demonstrating vessel contractor compliance. Generally undertaken by vessel contractor so time for Santos personnel to confirm that a Shipboard Oil Pollution Emergency Plan (SOPEP) /Shipboard marine pollution emergency plan (SMPEP) is in place.	Adopted – regulatory requirement, must be adopted.
BAO-CM-7.4.4	Spill clean-up kits available in high-risk areas (protective control)	Reduces the risk of spills and leaks to sea by controlling the clean-up of minor spills.	Cost of implementing procedures.	Adopted

CM reference	Control measure	Environmental benefit	Potential cost/issues	Evaluation
Additional control measures				
N/A	Eliminate ROV activities (elimination control)	Eliminates accidental hydrocarbon releases to the marine environment due to equipment failure.	ROVs contain minimal hydrocarbons (<5 L of hydraulic fluid) and as they are inspected and maintained, the risk of failure is very low. Using ROVs for IMMR activities reduces seabed disturbance, length of time in field, safety and environmental risks.	Not adopted – not technically or environmentally feasible to eliminate ROV activities. Hydrocarbon releases due to ROV failure has a very low risk and is considered sufficiently managed under ROV inspection and maintenance procedures (BAO- CM-7.4.1).
N/A	ROVs to use biodegradable hydraulic fluids only (substitution control)	Using a biodegradable hydraulic fluid reduces potential spill impacts as the oil is less persistent in the marine environment.	ROVs contain minimal hydrocarbons (<5 L of hydraulic fluid) that is likely to be a synthetic blend base oil (inherently biodegradable). ROVs are inspected and maintained, and the risk of failure is considered very low.	Not adopted – based on the cost to replace or modify the ROVs. The synthetic blend base oil that may be released during ROV failure has a very low risk and is considered sufficiently managed under the ROV inspection and maintenance procedures (refer to BAO-CM-7.4.1).
N/A	Zero chemical bunkering via hose (elimination control)	Removes spill risk from hose operations.	Cost associated with large transfers of chemicals via drums or containers. Additional environmental risks (such as dropped objects) associated with transferring chemicals via drums or containers. Additional trips to port increases health and safety risks.	Not adopted – storage of chemicals would result in unacceptable transfer of environmental risks to health and safety and operational risks. Would not eliminate the risk of accidental releases.

## 7.4.4 Environmental impact and risk assessment

Receptors	<ul> <li>Physical environment (water quality)</li> <li>Threatened, migratory or local fauna (fish [including sharks and rays], marine mammals, marine reptiles and seabirds)</li> <li>Socioeconomic (commercial fishing, tourism and recreation)</li> <li>Cultural features</li> </ul>
Consequence	I – Negligible
If a abamical is smilt	

If a chemical is spilt, the largest spill would likely be less than 10 L. Impacts to water quality would be expected to be very short-term and localised given the limited volumes that could be spilled. Due to the dispersive nature of the ocean environment and water depths within the OA, impacts to benthic habitats, including Shepparton Shoal are not predicted.

The water foaming agents in aqueous film forming foam (AFFF) may be harmful to marine organisms. Most of these foams have high oxygen demand and the toxicity of the detergents, solvents and other components in the foams may result in adverse effects to marine organisms. However, these effects are greatly diminished in the offshore marine environment due to the natural dilution from wind, wave and currents. The release of these foams is restricted to an emergency event.

If a minor hydrocarbon spill occurs, the quantities would likely be limited to 20 L. The small volumes, dilution and dispersion from natural weathering processes such as ocean currents and evaporation are such that spills will be limited in area and duration. The susceptibility of marine fauna to hydrocarbons depends on hydrocarbon type and exposure duration; however, given that exposures would be limited in extent and duration, exposure to marine fauna from this potential hazard is considered very low. The small volumes of worst-case discharges are such that the potential for impacts to receptors will decline rapidly with time and distance at the sea surface.

Harmful effects are not expected to the benthic community due to the water depths.

Near the sea surface, fish can detect and avoid contact with surface slicks and, as a result, fish mortalities rarely occur in open waters from surface spills (Kennish, 1997; Scholz et al., 1992). Therefore, pelagic fish species (e.g. sharks) are generally not highly susceptible to impacts from hydrocarbon spills. In offshore waters near the release point, pelagic fish are at risk of exposure to the more toxic aromatic components of the hydrocarbons. However, pelagic fish in offshore waters are highly mobile; therefore, it is unlikely they would be exposed to toxic components for long periods in this spill scenario. Components with higher toxicity would also rapidly evaporate and concentrations would significantly diminish with distance from the spill site, limiting the potential area of impact.

Marine pollution is identified as potential threats to several marine fauna species (that may be present in the OA) in relevant recovery plans and conservation advice (Table 3-13) and to MNES (DCCEEW, 2022b).

Given the negligible consequence on the physical environment or species, subsequent impacts to socioeconomic receptors (including commercial fishing, tourism and recreation) and cultural features are not anticipated.

A very small (less than 20 L) chemical or hydrocarbon spill would not result in a decreased population size at a local or regional scale or long-term reduction to water and sediment quality, but may be detectable, it is expected that a spill of this nature would result in I - Negligible consequence.

Likelihood	C – Possible
LINGIIIIOOU	

The likelihood of releasing minor volumes of chemicals (<10 L) or hydrocarbons (<20 L) to the environment during routine operations is considered D – Occasional.

**Residual Risk** The residual risk is considered Very Low.

### 7.4.5 Demonstration of as low as reasonably practicable

The use and storage of chemicals and hydrocarbon liquids is a requirement to undertake the Activity, and there are no reasonably practicable alternatives that meet operational requirements. All reasonably practicable control measures were reviewed and those adopted are considered appropriate to manage the residual risk to a Very Low level. The proposed management controls are in accordance with Santos' risk management criteria and are considered appropriate to reduce the risk to ALARP.

## 7.4.6 Acceptability evaluation

Is the risk ranked between Very Low and Medium?	Yes – residual risk is ranked as Very Low.
Is further information required to validate the consequence assessment?	No – potential impacts and risks are well understood through the information available.
Are the risks and impacts consistent with the principles of ecologically sustainable development (ESD)?	Yes – activity evaluated in accordance with Santos' Offshore Division Environmental Hazard Identification and Assessment Guideline, which considers principles of ESD.
Have the acceptable levels of impact and risks been informed by relevant species recovery plans, threat abatement plans and conservation advice and Australian marine park zoning objectives?	Yes – while several plans identify pollution as a threat to marine fauna, negligible impacts are predicted for this Activity. Control measures implemented will reduce the risk of minor releases to species identified in the following relevant species conservation advice and recovery plans, as also set out in Table 3-13.
	Conservation advice:
	<ul> <li>Approved Conservation Advice for <i>Pristis clavata</i> (Dwarf Sawfish) (DEWHA, 2009)</li> </ul>
	Approved Conservation Advice for Green Sawfish (DEWHA, 2008c)
	<ul> <li>Approved Conservation Advice for <i>Pristis pristis</i> (largetooth sawfish) (TSSC, 2014b)</li> </ul>
	<ul> <li>Approved Conservation Advice for <i>Glyphis garricki</i> (northern river shark) (TSSC, 2014a)</li> </ul>
	<ul> <li>Approved Conservation Advice for <i>Rhincodon typus</i> (whale shark) (TSSC, 2015g)</li> </ul>
	<ul> <li>Approved Conservation Advice for Balaenoptera physalus (fin whale) (TSSC, 2015c)</li> </ul>



	<ul> <li>Approved Conservation Advice for Balaenoptera borealis (sei whale) (TSSC, 2015b)</li> </ul>
	Recovery plans:
	<ul> <li>Recovery Plan for the White Shark (<i>Carcharodon carcharias</i>) (Department of Sustainability, Environment, Water, Population and Communities ((DSEWPaC, 2013)</li> </ul>
	<ul> <li>Conservation Management Plan for the Blue Whale - A Recovery Plan under the Environment Protection and Biodiversity Conservation Act 1999 2015–2025 (CoA, 2015a)</li> </ul>
	Recovery Plan for Marine Turtles in Australia 2017–2027 (CoA, 2017b)
	Wildlife Conservation Plan for Seabirds (CoA, 2020)
	Habitat degradation or modification is identified in many conservation advice, however the nature of these discharges does not result in habitat degradation.
	The OA does not intersect any AMP or protected area.
	For the identified plans, the objectives of those plans are achieved through the adoption of performance outcomes and the control measures outlined in Section 7.4.3. Santos considers that the level of risk of minor releases (surface and subsea) is not inconsistent with these plans.
Are performance outcomes, control measures and associated performance standards consistent with legal and regulatory requirements?	Yes – management measures are consistent with <i>Protection of the Sea</i> ( <i>Prevention of Pollution from Ships</i> ) <i>Act 1983</i> (Cth), <i>Navigation Act 2012</i> ( <i>Cth)</i> , MARPOL Annex V and Marine Order 91 (Marine pollution prevention – oil).
	Through acceptance of this GEP Coastal Waters OEMP, legislative and regulatory requirements will be met as per Appendix B.
Are performance outcomes, control measures and associated performance standards consistent with Santos' Environment, Health and Safety Policy?	Yes – aligns with Santos' Environment, Health and Safety Policy (Appendix A).
Are performance outcomes, control measures and associated performance standards consistent with industry standards?	Yes – the most recent and comparable EPs accepted by NOPSEMA were reviewed for consistency with the performance outcomes, control measures and associated performance standards proposed in this GEP Coastal Waters OEMP.
Have performance outcomes, control measures and associated performance standards taken into consideration Relevant Person feedback?	Yes – no objections or claims were raised regarding a potential minor release.
Are performance standards such that the impact or risk is considered to be ALARP?	Yes – ALARP assessment conducted, with additional control measures adopted.

No Relevant Person concerns have been raised regarding the accidental release of minor volumes of chemicals and hydrocarbons, and the proposed controls will reduce the residual risk to Very Low and ALARP. Based on an assessment of Santos' acceptability criteria and with the control measures in place, residual risks are considered acceptable.



## 7.5 Subsea release of dry natural gas

## 7.5.1 Description of event

Event	Subsea release of gaseous hydrocarbon from:
	<ul> <li>part of the Barossa GEP within of the OA impacted by IMMR activities or external event.</li> </ul>
	A subsea release of gaseous hydrocarbon within the OA during normal operations could be caused by one or a combination of the following:
	External impact from the following sources:
	<ul> <li>dropped objects from supply/support/third party vessel</li> </ul>
	<ul> <li>damage to pipeline during IMMR campaigns (e.g., repairs not undertaken to meet design specifications)</li> </ul>
	<ul> <li>future construction in the area from external parties</li> </ul>
	GEP integrity failure (e.g., fatigue/stress, internal and external corrosion, erosion)
	operating outside design envelope (e.g., overpressure, exceeding design temperature)
	<ul> <li>natural hazards (e.g., cyclone or seismic activity)</li> </ul>
	Discharge of a maximum volume up to 1,080 MMscf (24,030 tonnes) of dry gas (ref Pipeline Consequence Modelling Report) forming a large plume in the water column and dispersing into the atmosphere.
	A gas plume would be released from the Barossa GEP in the event of a rupture. The plume would move towards the surface, with some of the gas becoming dissolved in seawater as the plume rises. A worst-case rupture would lead to the formation of a large gas cloud, which would rapidly disperse in the atmosphere. Methane (the main component of the dry gas) is lighter than air and would rise into the atmosphere, away from the release location. A gas cloud could potentially form an explosive mix, which if ignited, results in injury/death and/or damage to property.
	The scale of a Barossa GEP release is dependent on the nature of the loss of containment. Small 'pinhole' leaks will result in a stream of bubbles which may dissolve and disperse before reaching the surface. A major rupture caused by a catastrophic failure has the potential to release a volume of up to 1080 MMscfd (24,030 t) of dry gas (Pipeline Consequence modelling report) forming a large plume in the water column and dispersing into the atmosphere over a period of approximately three days. A rupture is considered to be the worst-case credible release from the Barossa GEP.
	As the Barossa GEP transports dry natural gas with no liquid phase hydrocarbons, a loss of containment would not release any liquid phase hydrocarbons to the environment. Given that the contents of the pipeline consist entirely of dehydrated gas, condensation of gas phase components upon release is not expected due to the pressure and temperature differential between the pipeline contents and the receiving environment.
Extent	The dry natural gas within the Barossa GEP is contained at a relatively high pressure of up to 180 barG. The extent of a leak from the Barossa GEP would depend on the nature of the rupture but is expected to be limited to within hundreds of metres of the rupture location. Small 'pinhole' leaks may result in a stream of bubbles that could dissolve before reaching the surface.
Duration	Potentially harmful concentrations are limited to a very short period (days) immediately following the release.

#### Dry natural gas

The Barossa GEP consists of dry natural gas that is predominantly methane (approximately 88%), carbon dioxide (up to 6%), hydrogen sulphide (0.0015%) and approximately 6% other organic compounds (including ethane, propane, butane and isobutane, and nitrogen) (Santos, 2024). However, the gas composition can vary. Physical properties indicate that dry natural gas is highly flammable and will volatilise from the aquatic environment rapidly. It is noted that in practice, acute and chronic effects would not typically be observed (Shell, 2019).

### 7.5.2 Nature and scale of environmental impacts

**Potential receptors**: physical environment (water and air quality); threatened, migratory or local fauna (marine mammals, marine reptiles, sharks and rays, other fish, and birds); socioeconomic (other marine users); and cultural features.

### 7.5.2.1 Physical environment

A pipeline rupture and subsequent release of dry natural gas would result in a localised and short-term reduction in water and air quality. The plume would move towards the surface, as methane (the main component of dry natural gas) is lighter than air, with some of the gas becoming dissolved in seawater as the plume rises. Any dissolved gas in the water column is expected to disperse rapidly in the OA given the water depths, strong currents and wave action. A worst-case rupture would lead to the formation of a minor gas cloud, which would rapidly disperse in the



atmosphere. Potential changes to water and air quality are expected to be short-term (within days) and limited to within hundreds of meters of the rupture site.

The seabed in the OA adjacent to the GEP is characterised as featureless silty, shelly sand, with very sparse (<1%) epibiota (mainly soft corals and crinoids) (RPS, 2023). Any seabed disturbance impacts (e.g., scouring) are expected to be limited to the immediate vicinity of a pipeline rupture. Given the mobile nature of sediments and high current speeds in the OA, the seabed is expected to infill naturally with sediments and detrital matter, returning to near its original state over a short time (weeks to months). Benthic habitats would remain viable and are expected to recolonise through the recruitment of new colonists from planktonic larvae in adjacent undisturbed areas (Guerra-Garcia & Garcia-Gomez, 2006).

### 7.5.2.2 Threatened, migratory or local fauna

A hydrocarbon release will cause a decline in water quality that may have chemical (e.g., toxicity) impacts on marine fauna. Pipeline rupture may also cause changes in fauna behaviour as individuals avoid turbulent water around the release. A worst case rupture would lead to the formation of a minor gas cloud that could potentially impact air-breathing fauna, such as marine mammals, reptiles, and birds. Individuals in the immediate vicinity of the release may be at risk of asphyxiation, potentially resulting in death. However, this effect would be localised (within 500 m) and short-term as gas rapidly disperses into the atmosphere.

The recovery plans and conservation advice for several marine fauna species identify pollution and/or habitat degradation as a threat (e.g., marine turtles, marine mammals, sawfish); however, they are primarily focussed on agricultural, terrestrial industrial, domestic and oil pollution. Given the location and habitat within the OA, contact between marine fauna and an unplanned release of dry natural gas will likely be for a short duration and not sufficient to cause a toxic effect.

### 7.5.2.3 Socioeconomic

Pipeline rupture and subsequent release of dry natural gas could impact the health and safety of other marine users, such as fishers (commercial and traditional), tourism operators, and recreational users. A dry natural gas cloud could potentially form an explosive mix that, if ignited, results in injury/death and/or property damage. However, exclusion zones around Activity vessels would restrict other marine users within 500 m. An unplanned release would also activate an emergency response plan (refer to 8.2.6) that ensures other marine users are advised of the hazard.

### 7.5.2.4 Cultural features

No First Nations people feedback was provided about potential impacts from an unplanned release of dry natural gas release to any geographically specific cultural features (excluding marine fauna) during consultation for this GEP Coastal Waters OEMP (refer to Section 4.8). An important outcome of Dr Corrigan's research is that no sacred sites or dreamings are shown to be directly impacted by the DPD project footprint (where this OA and Activity are located), although this is not to say that some persons do not have fears that this could be the case in the event of an unplanned event (Corrigan, 2024).

In the 2022 Statement of Reasons Requests, the Tiwi clan members raised their concern regarding traditional hunting of marine species and totem species. The First Nations people maintain a continuing spiritual connection with sea country, including caring for sea country and access to cultural food sources. The potential impact to marine fauna is likely to be limited to transiting individuals and is unlikely to result in significant impacts to marine species at the individual or population level (refer to Section 7.5.2.2). As a result, unplanned release of dry natural gas is not anticipated to affect traditional hunting practices or resources.

In accordance with First Nations people cultural beliefs, if totemic species (e.g. turtles) are impacted by the Activity some believe this in turn can impact First Nations people and make them sick. The potential impacts to culturally significant marine fauna species (such as dreaming dreaming and totem species including marine mammals, reptiles and birds) are assessed in Section 7.5.2.2.

## 7.5.3 Environmental performance outcomes and control measures

The EPO relating to this event is:

- No injury, or mortality to, EPBC Act 1999 listed marine fauna or marine fauna listed as threatened species under the *Territory Parks and Wildlife Conservation Act 1976* (NT) as a result of the Activity (EPO-08)
- Atmospheric emissions associated with the Activity will meet all regulatory source emission standards (EPO-09)
- Zero unplanned discharge of hydrocarbons or chemicals to the marine environment from the Activity (EPO-18)



- No significant impact to cultural features from the Activity (EPO-19)
- No impacts to underwater cultural heritage from the Activity (EPO-20)

An assessment of the environmental benefits and the potential costs or issues associated with control measures for this Activity are described in Table 7-6 to demonstrate that potential risks are ALARP. Control measures that are adopted have associated EPSs and measurement criteria and are presented in Table 8-2. Rejected control measures have an ALARP evaluation provided to justify their rejection.

Table 7-6: Control measures	evaluation for subsea	release of dry natural gas
-----------------------------	-----------------------	----------------------------

CM reference	Control measure	Environmental benefit	Potential cost/issues	Evaluation
Standard control meas	sures			
BAO-CM-6.1.2	Vessels equipped and crewed in accordance with Australian maritime requirements, including Marine Order 30 (Prevention of Collisions) and Marine Order 21 (Safety and Emergency Arrangements) (administrative control)	Ensures vessels meet Marine Assurance Standards to reduce the likelihood of vessel collision (such as minimum and working lighting for maritime safety).	Cost associated with implementing procedures.	Adopted – benefit of assuring vessels outweighs procedure compliance costs.
BAO-CM-6.4.5	HSE inductions will include applicable environmental requirements (administrative control)	Ensures that crew are aware of the stringent Coastal Waters OEMP, Santos and legislative requirements.	Administrative costs to update existing Santos procedure and induction materials, and train personnel.	Adopted - benefits considered to outweigh costs.
BAO-CM-6.4.10	Barossa Facilities and vessels planned maintenance system to confirm equipment integrity is maintained in accordance with manufacturer guidelines (administrative control)	Ensures vessel and equipment integrity is maintained through routine checks.	High cost of maintaining equipment and managing the maintenance system.	Adopted – environmental benefits outweigh the costs.
BAO-CM-6.6.1	Notify AHS and AMSA MSI prior to relevant Activity (administrative control)	Maritime notifications ensure marine users are informed of the proposed activities, reducing the likelihood of unplanned interactions. Subsea infrastructure will be clearly marked on Australian nautical charts published by the Australian Hydrographic Office (AHO), alerting other marine users to the presence of activity vessels and exclusion zones and restrictions, reducing the likelihood of vessel collision and fishing gear snagging.	Cost and time perform notifications.	Adopted – benefits considered to outweigh costs.
BAO-CM-6.6.4	Activity undertaken in accordance with Santos HSE management and	Ensures contracted vessels are operated, maintained, and crewed in accordance	Costs associated with personnel time in checking vessels.	Adopted – benefit of assuring vessels outweighs procedure

CM reference	Control measure	Environmental benefit	Potential cost/issues	Evaluation
	marine vessel vetting processes (Santos' Offshore Marine Assurance Procedure) (administrative control)	with industry standards and regulatory requirements.		compliance costs.
BAO-CM-7.1.1	Implement standards and procedures for lifting equipment to reduce risk of dropped objects during lifting (administrative control)	Impacts to the environment are reduced by preventing dropped objects and dragged objects during lifting operations. Reduces the chance of a dropped suspended load. Load-bearing lifting equipment engineering standards and appropriate lifting procedures factor in technical and environmental variables to minimize the risk of losing control of a suspended load.	Cost of implementing procedures.	Adopted – environmental benefits of preventing dropped objects outweigh the procedural compliance costs.
BAO-CM-7.1.2	Dropped objects (incident) management (administrative control)	Impacts to the environment are reduced by preventing dropped objects and by retrieving dropped objects unless the environmental consequences of the dropped object are negligible or there are risks to safety.	Cost of implementing procedures.	Adopted – benefits of ensuring procedures are developed and followed outweigh the costs of personnel time.
BAO-CM-7.6.1	Emergency Response Plan (ERP) (administrative plan)	Mitigates the impact of a potential leak in the Barossa GEP. The ERP is based on the safety case for the pipeline.	Cost of implementing the procedure.	Adopted – benefits of ensuring procedures are developed and followed outweigh the costs of personnel time.
BAO-CM-7.6.2	Pipeline operating procedures (administrative control)	This control is effective in maintaining the integrity of the pipeline by providing the limitations within which the pipeline can be safely operated. This is done by relying on design specifications and standards, which are well-developed through extensive experience within Santos and the industry more broadly.	Personnel costs of ensuring appropriate procedures are in place and followed, including compliance inspections/reviews.	Adopted – benefits of ensuring procedures are developed and followed outweigh the costs of personnel time.
BAO-CM-7.6.3	Pipeline Integrity Management Plan (administrative control)	This control is effective in maintaining the integrity of the pipeline by ensuring preventative and	Personnel costs of ensuring appropriate procedures are in place and followed, including compliance inspections / surveys /	Adopted – benefits of ensuring procedures are developed and followed outweigh the costs of personnel time and expenses

CM reference	Control measure	Environmental benefit	Potential cost/issues	Evaluation
		reactive inspections and maintenance/repairs are performed using a risk-based approach.	reviews. Survey expenses.	
BAO-CM-7.6.5	Repairs to the Pipeline carried out to design specification (administrative control)	Ensures pipeline repairs are consistent with design specifications. Repairs conducted incorrectly may increase the likelihood of a failure with environmental and safety impacts.	Costs of repairs to be carried out in accordance with the Offshore Standard for Submarine Pipeline Systems.	Adopted – benefits outweigh the costs of undertaking appropriate repairs.
BAO-CM-7.6.9	Barossa Nearshore Gas Export Pipeline Operations Pipeline Management Plan	Details alarms and required emergency response in the event of a loss of containment.	Administrative costs of preparing procedures.	Adopted – benefits considered to outweigh costs.
Additional control mea	sures			
N/A	Eliminate lifting in the Operational Area (elimination control)	Reduces the risk of dropped objects.	Lifting is an essential activity for IMMR activities.	<b>Not adopted</b> – not feasible to eliminate lifting in the field.

## 7.5.4 Environmental impact and risk assessment

Receptors	<ul> <li>Physical environment (water quality, air quality)</li> <li>Threatened, migratory or local fauna (marine mammals, marine reptiles, sharks, rays, other fish, and birds)</li> </ul>
	<ul><li>Socioeconomic (commercial fishing, traditional fishing, tourism, recreation, shipping and defence)</li><li>Cultural features</li></ul>
Consequence	III – Moderate

#### Physical environment and habitat

Impacts to water and air quality would be expected, but due to the dispersive nature of the ocean environment and water depths, impacts are expected to be short-term and localised. Potential impacts to the physical environment and habitat (water and air quality) from a large or small release is considered in the OA is considered to be II - Minor.

#### Threatened, migratory or local fauna

A dry natural gas release is unlikely to have widespread ecological effects, given the nature of the product, short duration and the limited volume that could be released, and the transient nature of marine fauna in the OA. This unplanned event is not considered to have the potential for significant impacts to marine fauna species at the population level, including cultural significance. Potential impacts to marine fauna are considered to be II - Minor.

#### Socio-economic

The release of 24,030 tonnes of dry natural gas forming a gas cloud has the potential to result in injury/death of other marine users (such as traditional / commercial fishers) and or property damage of vessels that may be in the area. If the release is due to IMMR activities, a 500 m exclusion zone will be in force around the Activity vessels and subsequent impacts to socioeconomic receptors including commercial fishing and other marine users are not anticipated to be significant. If a rupture occurs when a 500 m exclusion zone is not in place (i.e. it is not related to IMMR activities) there is the potential to impact a number of marine users. The worst-case consequence to socio-economic receptors was assessed as III – Moderate if a release impacts a number of marine users.

#### **Cultural features**

For assessment of impacts to marine species of cultural significance, refer to the above impacts and risks to threatened, migratory or local fauna.

Likelihood	A – Remote
A pipeline rupture inc to be remote due to the	ident caused by IMMR activities or an external impact with the control measures in place is considered ne volume of 3rd party vessels that are not under Santos control that may be present.
Posidual Pisk	The residual risk is considered Very Low

## 7.5.5 Demonstration of as low as reasonably practicable

A thorough set of controls has been proposed to minimise the risk of damage to the existing Barossa GEP and any subsequent environmental consequences of a dry natural gas release.

All reasonably practicable control measures were reviewed and those adopted are considered appropriate to manage the residual risk to Very Low. The proposed management controls are in accordance with Santos' risk management criteria and are considered appropriate to reduce the risk to ALARP.

## 7.5.6 Acceptability evaluation

Is the risk ranked between Very Low and Medium?	Yes – residual risk is ranked Very Low.
Is further information required to validate the consequence assessment?	No – potential impacts and risks are well understood through the information available.
Are the risks and impacts consistent with the principles of ecologically sustainable development (ESD)?	Yes – activity evaluated in accordance with Santos' Offshore Division Environmental Hazard Identification and Assessment Guideline, which considers principles of ESD.
Have the acceptable levels of impact and risks been informed by relevant species recovery plans, threat abatement plans and conservation advice and Australian marine park zoning objectives?	Yes – while several plans identify pollution as a threat to marine fauna, significant impacts are not predicted for this Activity. Control measures implemented will reduce the risk of a subsea release of gaseous hydrocarbon to species identified in the following relevant species recovery plans, conservation advice, wildlife conservation plans and other management plans/guidelines, as also set out in Table 3-13.
	Conservation advice
	<ul> <li>Approved Conservation Advice for <i>Pristis clavata</i> (Dwarf Sawfish) (DEWHA, 2009)</li> </ul>
	Approved Conservation Advice for Green Sawfish (DEWHA, 2008c)
	<ul> <li>Approved Conservation Advice for <i>Pristis pristis</i> (largetooth sawfish) (TSSC, 2014b)</li> </ul>
	<ul> <li>Approved Conservation Advice for <i>Glyphis garricki</i> (northern river shark) (TSSC, 2014a)</li> </ul>
	<ul> <li>Approved Conservation Advice for <i>Glyphis glyphis</i> (speartooth shark) (DoE, 2014a)</li> </ul>
	<ul> <li>Approved Conservation Advice for <i>Rhincodon typus</i> (whale shark) (TSSC, 2015g)</li> </ul>
	<ul> <li>Approved Conservation Advice for Balaenoptera physalus (fin whale) (TSSC, 2015c)</li> </ul>
	<ul> <li>Approved Conservation Advice for Balaenoptera borealis (sei whale) (TSSC, 2015b)</li> </ul>
	<ul> <li>Approved Conservation Advice for <i>Dermochelys coriacea</i> (Leatherback Turtle) (DEWHA, 2008b)</li> </ul>
	Recovery plans
	<ul> <li>Recovery Plan for the White Shark (<i>Carcharodon carcharias</i>) (Department of Sustainability, Environment, Water, Population and Communities (DSEWPaC, 2013)</li> </ul>
	<ul> <li>Conservation Management Plan for the Blue Whale - A Recovery Plan under the Environment Protection and Biodiversity Conservation Act 1999 2015–2025 (CoA, 2015a)</li> </ul>
	Recovery Plan for Marine Turtles in Australia 2017–2027 (CoA, 2017)
	Wildlife Conservation Plan for Seabirds (CoA, 2020)
	The OA does not intersect any AMP or protected area.
	For the identified plans, the objectives of those plans are achieved through the adoption of performance outcomes and the control measures outlined in Section 7.5.3. Santos considers that the level of risk of a subsea release of gaseous hydrocarbon is not inconsistent with these plans.

Are performance outcomes, control measures and associated performance standards consistent with legal and regulatory requirements?	Yes – Relevant legislative requirements and standard industry practices have been applied to control the risk. Through acceptance of this GEP Coastal Waters OEMP, legislative and regulatory requirements will be met as per Appendix B.
Are performance outcomes, control measures and associated performance standards consistent with Santos' Environment, Health and Safety Policy?	Yes – aligns with Santos' Environment, Health and Safety Policy (Appendix A).
Are performance outcomes, control measures and associated performance standards consistent with industry standards?	Yes – the most recent and comparable EPs accepted by NOPSEMA were reviewed for consistency with the performance outcomes, control measures and associated performance standards proposed in this GEP Coastal Waters OEMP.
Have performance outcomes, control measures and associated performance standards taken into consideration Relevant Person feedback?	Yes – Relevant feedback relating to a potential unplanned dry natural gas release has been considered when evaluating performance outcomes, control measures and associated performance standards. Where relevant, control measured implemented based on Relevant Persons feedback for other Barossa EPs have been adopted in this GEP Coastal Waters OEMP. Additional performance outcomes (EPO-19, EPO-20) have been adopted
	based on Relevant Persons feedback on other Barossa EPs.
Are performance standards such that the impact or risk is considered to be ALARP?	Yes – ALARP assessment conducted, with no additional control measures adopted.

Relevant Persons concerns have been evaluated regarding this aspect and additional performance outcomes have been adopted. The proposed controls will reduce the residual risk to Very Low and ALARP. Therefore, Santos considers the residual risk associated with the unplanned release of dry natural gas to be reduced to an acceptable level.

The potential impacts from a dry gas release from a pipeline rupture are broadly acceptable based on the residual risk ranking and considerations outlined above.



## 7.6 Unplanned release of MDO

## 7.6.1 Description of event

Event	Worst-case credible MDO release
	Unplanned loss of containment of hydrocarbons into the marine environment is a credible risk during IMMR activities. A credible worst-case release scenario of MDO to the marine environment could be a collision between 2 activity vessels or an activity vessel and a third party. Such a collision could rupture a fuel tank at the sea surface resulting in the release of MDO to sea.
	The AMSA (2015) Technical Guidelines for Preparing Contingency Plans for Marine and Coastal Facilities recommend that the spill scenario for modelling and impact assessment should be based on 50% of the largest single fuel tank volume if protected by a double–hull or the largest single unprotected fuel tank volume. It was estimated that a typical IMMR vessel could have an MDO fuel tank of 600 m <sup>3</sup> with double–hull wing protection equivalent. Based on the class of vessel to be used for the activity, the largest credible loss of containment scenario identified is a surface release of MDO from a vessel as a result of an external impact (vessel collision), which ruptures an MDO tank was conservatively estimated to be 300 m <sup>3</sup> over one hour.
	Bunkering incident
	Also considered in this section is a much smaller volume refuelling incident (fuel hose failure or rupture, coupling failure or tank overfilling) where vessel fuel bunkering would need to be stopped manually. Fuel released before pumping stops and fuel remaining in the transfer line may be released to the environment. Spill volumes were determined from transfer hose inventory and spill prevention measures, including 'drybreak' or 'breakaway' couplings, rapid shutdown of fuel pumps and spill response preparedness, with 10 m <sup>3</sup> considered to be the maximum volume that could be released from the hose before shutdown.
	A collision scenario between a vessel and another vessel (third-party or other Santos vessel) could occur due to factors such as human error, poor navigation, vessel equipment failure or poor weather. A number of prerequisite conditions must exist for a vessel collision to result in the loss of fuel to the environment:
	The vessel must be involved in a collision: collisions involving offshore support vessels, comparable to those that will be used during the activity, are very uncommon. Statistics compiled by the Australian Transport Safety Bureau indicated offshore support vessels were involved in only one collision-related incident between 2011 and 2012, and no pollution-related incidents from offshore support vessels were recorded in the same time period.
	The collision must occur with sufficient force to rupture a fuel tank: fuel tanks are typically located at various positions around a vessel within the hull.
	The rupture must be of such a nature that the fuel can be released into the environment: a tank rupture must be above or near the fuel level within the tank to result in a loss of containment from the tank. Once lost from the tank, fuel may leak to the environment or drain into the vessel hull. Fuel from ruptured tanks may be transferred to other tanks onboard, reducing the volume in the ruptured tank.
	Note, it is not credible that the total storage volume of the vessel would be lost, as MDO is stored in more than one tank.
	All vessels used to undertake activities within the scope of this GEP Coastal Waters OEMP will be fuelled using MDO or lighter fuels (such as MGO). Heavier fuel types, such as intermediate fuel oil or HFO, will not be used.
Extent	Floating oil at, or above, 1 g/m <sup>2</sup> concentrations were predicted to extend up to 55 km from the release location. However, the distances reduced to 38 km and 21 km as the thresholds increase to 10 g/m2 and 50 g/m2, respectively.
	The Afghan Shoal and Shepparton Shoal were the only environmental value areas (EVAs) predicted to be exposed by floating oil at, or above, 1 g/m <sup>2</sup> at 0.33% and 2% probability of exposure, respectively. It took a minimum of 7 hours and 66 hours to reach the two receptors, respectively.
	Oil accumulation was forecasted for Tiwi Islands and Vernon Islands CR with probabilities of exposure of 0.33% and 1.67%, respectively, at, or above, 10 g/m <sup>2</sup> threshold. No accumulation of oil was predicted at, or above, 100 g/m <sup>2</sup> threshold. The minimum time before oil accumulation was 230 hours for the Tiwi Islands, whilst the maximum volume of oil ashore was <1 m <sup>3</sup> predicted for both EVAs.
	The maximum distance from the release location to the 10 ppb entrained hydrocarbon threshold, was 229 km. Shepparton Shoal was the EVA with the shortest predicted time to exposure at 8 hours and also had the highest probability of exposure at 29%, whilst the maximum concentration was 1,435 ppb.
	Dissolved hydrocarbon concentrations exceeding 10 ppb may potentially occur 49 km from the release location with the distance reducing to 23 km as the threshold increased to 50 ppb. Notably, no concentrations at, or above, the 400 ppb threshold were predicted. The highest probability of exposure was forecasted for Outer Oceanic Shoals AMP (1.33%) at, or above, the 10 ppb threshold and this receptor registered the shortest time before exposure at 30 hours. Additionally, Outer Oceanic Shoals AMP recorded the highest concentration at 35 ppb (RPS April 2024).



## Duration Constant

An unplanned release may occur during operational activities within the OA.

Release estimated to occur over one hour. Hydrocarbons would persist within the environment for a longer period of time, although MDO is expected to weather quickly through evaporation and dispersion.

### 7.6.2 Nature and scale of environmental impacts

Potential receptors: physical environment (water quality, shoals and banks, benthic habitats), threatened or migratory fauna (marine mammals, marine reptiles, fish [including sharks and rays] and birds), protected areas (AMPs and KEFs), socioeconomic receptors (fisheries, tourism, recreation and other third-party operators); and cultural features (including Native Title, ILUAs, sacred sites and sea country).

A hydrocarbon release will cause a decline in water quality and may cause chemical (e.g. toxicity) and physical (e.g. coating of emergent habitats, oiling of wildlife at sea surface) impacts to marine species. The severity of the impact of a hydrocarbon release depends on the magnitude of the release (i.e., extent, duration), prevailing weather conditions and sensitivity of the receptor. The nature and scale of a hydrocarbon release is described throughout this section for a vessel collision scenario, given smaller hydrocarbon releases (from refuelling, maximum of 10 m<sup>3</sup>) will impact a smaller area than a vessel collision (modelled spill of 300 m<sup>3</sup>).

## 7.6.3 Spill modelling overview

Hydrocarbon spill modelling (RPS, 2024b; Appendix F) was performed using an advanced three-dimensional trajectory and fates model: Spill Impact Model Application Package (SIMAP). The SIMAP model calculates the transport, spreading, entrainment and evaporation of spilled hydrocarbons over time, based on the prevailing wind and current conditions and the physical and chemical properties. The modelling does not take into consideration any of the spill prevention, mitigation and response capabilities that would be implemented in response to the spill.

The modelling study was performed in stages. Firstly, a ten-year wind and current dataset (2010 to 2019) that includes the combined influence of large-scale ocean and tidal currents was prepared. Secondly, the currents, local winds and detailed hydrocarbon characteristics were used as inputs in SIMAP to simulate the drift, spread, weathering and fate of the spilled hydrocarbon.

Modelling was conducted using a stochastic (or probabilistic) approach, which involved running 100 spill simulations per season (summer [October to the following March]; transitional periods [April and September] and winter [May to August]), with each simulation having the same spill information (spill volume, duration and composition of hydrocarbons) but randomly selected start time to ensure a range of wind and current conditions were assessed. Once all 300 simulations were run, the results were combined to determine the annualised potential exposure to the surrounding waters, shorelines and sensitive receptors based on the thresholds outlined in the NOPSEMA Oil Spill Modelling Bulletin (NOPSEMA, 2019).

Deterministic modelling is the predictive modelling of a single incident subject to a single sample of wind and weather conditions over time. Deterministic spill dispersion modelling is provided in Section 6 of the GEP NT waters OPEP, which includes all results relevant to spill response.

Deterministic modelling is often paired with stochastic modelling to place the large stochastic footprint into perspective. This deterministic analysis is generally a single run selected from the stochastic analysis and serves as the basis for developing the spill response or scientific monitoring plans. Deterministic modelling was also performed for the worst-case scenario to understand the potential area of influence that could be expected from the worst case MDO spill event. Table 7-7 summarises the model settings and assumptions.

Parameter	Scenario
Scenario description	Fuel tank rupture
Location	Commonwealth/NT waters boundary – KP 23 Easting - 618,128.5 Northing - 8,663,104.1
Number of randomly selected spill start times	300 total (100 per season)
Oil type	MDO
Spill volume	300m <sup>3</sup>
Release duration	1 hour

Table 7-7: Summary of model settings and assumptions for the vessel collision scenario

Simulation length

30 days



## 7.6.4 Hydrocarbon and weathering characteristics

MDO has a density of 890 kg/m<sup>3</sup> (API of 37.6) and a low pour point of -9°C. The low viscosity (14 cP at 25°C) indicates this hydrocarbon will spread quickly when released and will form a thin- to low-thickness film on the sea surface, increasing the rate of evaporation.

As presented in Table 7-8, about 4% of the MDO mass should evaporate within the first 12 hours (Boiling point (BP) <  $180^{\circ}$ C); a further 32% should evaporate within the first 24 hours ( $180^{\circ}$ C < BP <  $265^{\circ}$ C); and an additional 54% should evaporate over several days ( $265^{\circ}$ C < BP <  $380^{\circ}$ C). Approximately 10% (by mass) of MDO will not evaporate, though will decay slowly over time.

MDO is categorised as a Group II oil (light-persistent) according to ITOPF (2022) and United States Environmental Protection Agency/United States Coast Guard classifications. The classification is based on the specific gravity of hydrocarbons in combination with relevant boiling point ranges.

MGO is a Group II oil hydrocarbon with a 'light persistent' classification. While MDO and MGO are similar, MGO has a marginally higher density than MDO and is based on the lighter distillates, which results in a low viscosity. MGO is considered an ultra-low sulfur fuel and emissions from MGO contain significantly less particulate matter than other fuel types. Given the similarities in MGO and MDO properties, MDO is presented in Table 7-8 and has been used for the purpose of spill modelling.

#### Table 7-8: Properties of marine diesel oil

Parameter		Marine diesel oil	
Density (kg/m <sup>3</sup> )		890	
American Petroleum Institute		27.5	
Dynamic viscosity (cP)		14.0 (at 25°C)	
Pour point (°C)		-9.0	
Hydrocarbon property category		Group II	
Hydrocarbon property classification		Light persistent	
Boiling point (°C)			
Non-persistent	Volatiles <180	4	
	Semi-volatiles 180 to 265	32	
	Low volatiles 265 to 380	54	
Persistent	Residual >380	10	

Source: RPS (2024b)

A series of weathering tests were conducted to illustrate the potential behaviour following a 50 m<sup>3</sup> instantaneous surface release of MDO when exposed to:

- five-knot (2.6 m/s) constant wind speed, 27°C water temperature and currents
- variable wind speeds (1 to 12 m/s or 2 to 24 knots), 27°C water temperature and currents.

The first case is indicative of the potential weathering rates under calm conditions that would not generate entrainment, while the second case would be more representative of the moderate winds experienced over the region.

The mass balance forecast for the constant wind case (Figure 7-1) shows that around 41% of the MDO has evaporated within 24 hours. Evaporation will slow considerably and be subject to more gradual decay through biological and photochemical processes.

Under the variable wind speed case (Figure 7-2), after 24 hours 40% of the mass has evaporated, 31% has entrained and 29% remains on the water surface. Due to the higher wind speeds and breaking waves, entrainment of the MDO into the water column is shown to occur. While the MDO is entrained it will decay at a higher rate of 1% per day or 7.7% after seven days due to biological and photochemical degradation, compared to a rate of 0.14% per day and total of approximately 1% after seven days for the constant-wind case. Given the proportion of entrained MDO and the tendency for it to remain mixed in the water column, the remaining hydrocarbons will decay over timescales of several weeks.



Figure 7-1: Mass balance plot for an instantaneous 50 m<sup>3</sup> surface release of MDO subjected to a constant 5 knot (2.6 m/s) wind, currents and 27°C water temperature



Figure 7-2: Mass balance plot for an instantaneous 50 m<sup>3</sup> surface release of marine diesel oil subjected to variable wind speeds (1 to 12 m/s or 2 to 23 knots), currents and 27°C water temperature



## 7.6.5 Hydrocarbon exposure thresholds

To inform the environmental assessment, it is important to understand the profile of the concentrations of hydrocarbons after a spill. To do this, NOPSEMA recommends identifying hydrocarbon exposure values that broadly reflect the range of consequences that could occur at certain concentrations (NOPSEMA, 2019). The exposure values that have been applied to this GEP Coastal Waters OEMP are provided in Table 7-9.

To identify appropriate exposure values, Santos has followed the advice provided by NOPSEMA in Bulletin #1 Oil Spill Modelling (2019) and scientific literature. The selected hydrocarbon exposure values are discussed in Table 7-10 to Table 7-13. These tables explain how the exposure value is relevant to the risk evaluation and provides context on how that exposure value is used to inform response planning (which is addressed further in the GEP NT waters OPEP).

|--|

Hydrocarbon phase	Exposure value		
	Low	Moderate	High
Floating (g/m <sup>2</sup> )	1	10	50
Shoreline accumulation (g/m <sup>2</sup> )	10	100	1,000
Dissolved aromatics (ppb)	10	50	400
Entrained (ppb)	10	-	100

The low exposure values contours (Figure 7-3), which approximate a range of potential socioeconomic effects, are used as a predictive tool to set the outer boundaries of the EMBA, presented in Section 3. A 'best fit' line is drawn around the outermost limits of the low exposure value contours for all three phases of hydrocarbons (floating, dissolved and entrained) in all seasons. This results in a highly conservative and comprehensive basis to plan and prepare for spill response.

These low exposure values are not considered to be representative of a biological impact, but they are adequate for identifying the full range of environmental receptors that might be contacted by surface and sub-surface floating hydrocarbons (NOPSEMA, 2019) and a visible sheen may be apparent.

Determining exposure values that may be representative of biological impact is complex, since the degree of impact will depend on the sensitivity of the receptors contacted, the duration of the exposure, and the toxicity of the hydrocarbon type making the contact. The toxicity of a hydrocarbon will also change over time, due to weathering processes altering the composition of the hydrocarbon.

To inform the environmental assessment, exposure values that may be representative of biological impact have also been identified. These are called 'moderate exposure values' (defined by the moderate exposure value area, or MEVA) and 'high exposure values' (defined by the high exposure value area, or HEVA), and are shown in Figure 7-3. Moderate and high exposure values are modelled for each fate of hydrocarbon to identify what contact is predicted for surface (floating hydrocarbon), subsurface (entrained hydrocarbon and dissolved aromatic hydrocarbons) and shoreline accumulation of hydrocarbon at sensitivities.



Figure 7-3: Low exposure value contours used to define the EMBA and shoreline accumulation

	ARAFURA
GARIG GUNAK	and and
BARLU MARINE	ST BAR
	4 5
	har har h
~ with the til	5117
The state	
	E
5 53	J
for a	
	$ \sim \sim$
/	~
	$\leq$
	1
	/
	$\sim$ 1
	V
oastal waters Operational A	rea (OA)
a GEP - Commonwealth	
a GEP - NT Coastal Waters	
a GEP - NT Internal Waters	
Harbour region management	nt 🛛
ary	
mmonwealth waters bounda	ry
ial sea baseline	
ne accumulation @ 10g/m2	across all
IS	
re contours	
ved low exposure threshold	
ed low exposure threshold (	EMBA)
g low exposure threshold	1.00
onal Park	
Gunak Barlu	
narine parks (DCCEEW)	
al Park Zone (IUCN II) (Marin	ne
al Park Zone (IUCN II) in the	SE)
Protection Zone (IUCN IV)	
e Use Zone (IUCN VI)	
Purpose Zone (Trawl) (ILIC	N VD
Purpose Zone (IIICN VI)	
	Date: 18/12/2024
Scale: Map 1	:2,500,000 @ A4
AU213015 AU213015087 001 A5 GEP LowExpos	ureContours v2



Figure 7-4: Sensitive receptors within the MEVA



## Table 7-10: Floating hydrocarbons exposure values

Surface hydrocarbons concentration	Exposure value	Description
(g/m²)		
1	Low	Risk evaluation
		It is recognised a lower floating hydrocarbon concentration of $1 \text{ g/m}^2$ (equivalent to a thickness of 0.001 mm or 1 ml of hydrocarbon per m <sup>2</sup> ) is visible as a rainbow sheen on the sea surface. Although this is lower than the exposure value for ecological impacts, it may be relevant to socioeconomic receptors and has been used as the exposure value to define the spatial extent of the low exposure and EMBA from floating hydrocarbon.
		Response planning
		Contact at 1 g/m <sup>2</sup> (as predicted by hydrocarbon spill trajectory modelling) is used for operational and scientific monitoring planning, as described in the Northern Australia Operational and Scientific Monitoring Bridging Implementation Plan.
10	Moderate	Risk evaluation
		There is a paucity of data about floating hydrocarbon concentrations with respect to impacts to marine organisms. Hydrocarbon concentrations for registering biological impacts resulting from contact of surface slicks have been estimated by different researchers at about 10 to 25 g/m <sup>2</sup> (French et al., 1999; Koops et al., 2004; NOAA, 2002). The impact of floating hydrocarbon on birds is better understood than on other receptors. A conservative exposure value of 10 g/m <sup>2</sup> has been applied to impacts from floating hydrocarbon) in this GEP Coastal Waters OEMP. Although based on birds, this hydrocarbon exposure value is also considered appropriate for turtles, seasnakes and marine mammals (Natural Resource Damage Assessment Model for Coastal and Marine Environments, 1997). This value has been used to define the MEVA.
		Response planning
		Contact at 10 g/m <sup>2</sup> is not specifically used for spill response planning.
50	High	Risk evaluation
		At greater thicknesses, the potential for impact of floating hydrocarbon to wildlife increases. All other things being equal, contact to wildlife by floating hydrocarbon at 50 g/m <sup>2</sup> is expected to result in a greater impact. This value has been used to define the HEVA.
		Response planning
		Containment and recovery effectiveness drops significantly with reduced hydrocarbon thickness (McKinney et al., 2017; NOAA, 2014). McKinney et al. (2017) tested the effectiveness of various hydrocarbon skimmers at various hydrocarbon thicknesses. Their results showed the hydrocarbon recovery rate of skimmers dropped significantly when hydrocarbon thickness was less than 50 g/m <sup>2</sup> (less than Bonn Agreement Code 4). Hence, 50 g/m <sup>2</sup> has been set as a guide for planning effective containment and recovery operations.
		equivalent to hydrocarbon observed as discontinuous or continuous true colour) is considered to be a lower limit for effective dispersant operations and is therefore considered for planning.

## Table 7-11: Shoreline hydrocarbon accumulation exposure values

Shoreline Accumulation (g/m <sup>2</sup> )	Exposure Value	Description
10	Low	<ul> <li>Risk evaluation</li> <li>An accumulated concentration of hydrocarbon above 10 g/m<sup>2</sup> on shorelines is considered to represent a level of socioeconomic effect (NOPSEMA, 2019). For example, reduction in visual amenity of shorelines. This value has been used in previous studies to represent a low contact value for interpreting shoreline accumulation modelling results (French-McCay, 2005a, 2005b) and is used to define the low exposure and EMBA.</li> <li>Response planning</li> <li>Not specifically used for response planning because it is below the limit that can be effectively cleaned.</li> </ul>



Shoreline Accumulation (g/m <sup>2</sup> )	Exposure Value	Description
100	Moderate	<b>Risk evaluation</b> The impact exposure value for exposure to hydrocarbons stranded on shorelines is derived from levels likely to cause adverse impacts to marine or coastal fauna and habitats. These habitats and marine fauna known to use shorelines are most at risk of exposure to shoreline accumulations of hydrocarbon, due to smothering of intertidal habitats (such as mangroves and emergent coral reefs) and coating of marine fauna. Environmental risk assessment studies (French-McCay, 2009) report a hydrocarbon thickness of 0.1 mm (100 g/m <sup>2</sup> ) on shorelines is assumed as the lethal exposure value for invertebrates on hard substrates (rocky, artificial or human-made) and sediments (mud, silt, sand or gravel) in intertidal habitats. Therefore, a conservative exposure value for impacts of 100 g/m <sup>2</sup> has been applied to impacts from shoreline accumulation of hydrocarbons. This value has been used to define the MEVA.
		<b>Response planning</b> A shoreline concentration of 100 g/m <sup>2</sup> , or above, is likely to be representative of the minimum limit the hydrocarbon can be effectively cleaned (AMSA, 2015; NOPSEMA, 2019) and is therefore used as a guide for shoreline clean-up planning. This exposure value equates to approximately half a cup of hydrocarbon per square metre of shoreline contacted.
1000	High	<ul> <li>Risk evaluation</li> <li>At greater thicknesses, the potential for impact of accumulated hydrocarbon to shoreline receptors increases. All other things being equal, accumulation of hydrocarbon above 1,000 g/m<sup>2</sup> is expected to result in a greater impact. This value has been used to define the HEVA.</li> <li>Response planning</li> <li>As hydrocarbons increase in thickness the effectiveness of hydrocarbon recovery techniques increases. This value can therefore be used to prioritise hydrocarbon recovery efforts, assuming hydrocarbon recovery is deemed to have an environmental benefit.</li> </ul>

## Table 7-12: Dissolved aromatic hydrocarbon exposure values

Dissolved hydrocarbons (ppb)	Exposure value	Description
10	Low	Risk evaluation
		Dissolved aromatic hydrocarbons (DAH) include the monoaromatic hydrocarbons (compounds with a single benzene ring, such as benzene, toluene, ethylbenzene and xylene [BTEX]) and polycyclic aromatic hydrocarbons (PAHs) (compounds with multiple benzene rings, such as naphthalenes and phenanthrenes). These compounds have a greater bioavailability than hydrocarbons and are the main contributors to hydrocarbon toxicity. The toxicity of DAHs is a function of the concentration and duration of exposure by sensitive receptors, with greater concentration and exposure time causing more severe impacts. Typically tests of toxicity done under laboratory conditions measure toxicity as A proportion of test organisms affected (such as 50% mortality or LC50) at the end of a set time, often 48 or 96 hours.
		French-McCay (2002) found LC50 for dissolved PAHs with a 96-hour exposure range between 30 ppb for sensitive species (2.5th-percentile species) and 2260 ppb for insensitive species (97.5th-percentile species), with an average of about 250 ppb. The range of LC50s for PAHs obtained under turbulent conditions (this includes fine hydrocarbon droplets) was 6 ppb to 410 ppb, with an average of 50 ppb (French-McCay, 2002).
		More recently, French-McKay (2018) described in-water thresholds as 10 to 100 $\mu$ g/L (equivalent to ppb). For the effect of ultraviolet on PAH toxicity, French-McKay et al. (2018) use the findings of the Deepwater Horizon Oil Spill to adjust for this by reducing the water column exposure thresholds by ten times in the top 20 m of the water column.
		The dissolved hydrocarbon 10 ppb exposure value has been used to inform the low exposure and EMBA. An exposure value of 10 ppb is appropriate as it is a concentration that could have some potential negative effect.
		Response planning
		Can assist in establishing planning area for scientific monitoring based on potential for exceedance of water quality triggers (NOPSEMA, 2019).



Dissolved hydrocarbons (ppb)	Exposure value	Description
50	Moderate	Risk evaluation
		Approximates potential toxic effects, particularly sublethal effects to sensitive species (see the above text). Consistent with NOPSEMA (2019). This value has been used to define the MEVA.
		Ecotoxicology tests on a broad range of representative taxa of ecological relevance for mainly tropical Australia were conducted to inform the assessment of the potential for toxicity impacts from unweathered (as in, fresh) and weathered Barossa condensate to sensitive marine biota. The ecotoxicity testing focused on the DAH concentration of the water-accommodated fraction, as these hydrocarbons are more biologically available to organisms through absorption into their tissues when compared with entrained hydrocarbons (Jacobs, 2016b). Based on the ecotoxicology tests, the dissolved aromatic exposure values applied in this GEP Coastal Waters OEMP are considered highly conservative for the Barossa condensate. Specifically, the moderate exposure values of 50 ppb for 95% species protection for DAH is approximately 23 times more conservative than that for the Barossa condensate (1146 ppb for the 95% species protection threshold).
		Response planning
		Encompassed by response to 10 ppb. There is nothing different for higher exposure values.
400	High	Risk evaluation
		Approximates toxic effects, including lethal effects to sensitive species (NOPSEMA, 2019). This value has been used to define the HEVA.
		Response planning
		Encompassed by response to 10 ppb. There is nothing different for higher exposure values.

## Table 7-13: Entrained hydrocarbon exposure values

Entrained hydrocarbons (ppb)	Exposure value	Description
10	Low	Risk evaluation
		Entrained hydrocarbons, as opposed to DAHs, are hydrocarbon droplets suspended in the water column and insoluble. Entrained hydrocarbons are not as bioavailable to marine organisms compared with DAHs and on that basis are considered to be less toxic, especially over shorter exposure timeframes. Entrained hydrocarbons still have potential effects on marine organisms through direct contact with exposed tissues and ingestion (National Research Council, 2005). However, the level of exposure causing effects is considerably higher than for DAHs.
		Much of the published scientific literature does not provide sufficient information to determine if toxicity is caused by entrained hydrocarbons, but rather the toxicity of total hydrocarbons which includes both dissolved and entrained components. Variations in the methodology of the total water-accommodated fraction (entrained and dissolved) may account for much of the observed wide variation in reported exposure values, which also depend on the test organism types, duration of exposure, hydrocarbon type and the initial hydrocarbon concentration. Total hydrocarbon toxicity acute effects of total hydrocarbon as LC50 for molluscs range from 500 to 2000 ppb (Clark et al., 2001; Long & Holdway, 2002). A wider range of LC50 values have been reported for species of crustacea and fish from 100 to 258,000,000 ppb (Gulec et al., 1997; Gulec & Holdway, 2000; Clark et al., 2001) and 45 to 465,000,000 ppb (Gulec & Holdway, 2000; Barron et al., 2004), respectively.
		The 10 ppb exposure value has been used to inform the EMBA and represents the very lowest concentration and corresponds generally with the lowest trigger levels for chronic exposure for entrained hydrocarbons in the ANZECC & ARMCANZ (2000) water quality guidelines. This is consistent with NOPSEMA (2019) guidance.
		Response planning
		Can assist in establishing planning area for scientific monitoring based on potential for exceedance of water quality triggers (NOPSEMA, 2019).
100	Moderate	Risk evaluation



Entrained hydrocarbons (ppb)	Exposure value	Description
		The 100 ppb exposure value is considered to be more representative of sub-lethal impacts to most species and lethal impacts to sensitive species, based on toxicity testing as described above. This is considered conservative, as toxicity to marine organisms from hydrocarbon is likely to be driven by the more bioavailable dissolved aromatic fraction, which is typically not differentiated from entrained hydrocarbon in toxicity tests using water-accommodated fractions. Given entrained hydrocarbon is expected to have lower toxicity than dissolved aromatics, especially over time periods where these soluble fractions have dissolved from entrained hydrocarbon, the higher moderate exposure value for entrained hydrocarbon over DAH (100 versus 50 ppb) is considered appropriate. This value has been used to define the MEVA.
		Note NOPSEMA does not define a moderate exposure value for entrained hydrocarbon and 100 ppb is defined as the high exposure value. However, Santos has adopted 100 ppb as the moderate exposure level for impact assessment purposes in the absence of a NOPSEMA-defined moderate value and based on existing literature (Bridges et al., 2018; French-McCay, 2016; French-McCay, 2018). <b>Response planning</b>
		Encompassed by response to 10 ppb. There is nothing different for higher exposure values.

### 7.6.6 Spill risk assessment approach

A consistent risk assessment approach is applied to each unplanned hydrocarbon release scenario. The approach for hydrocarbon spills involves several steps outlined below:

- 1. Identify the spatial extent of the EMBA. The EMBA is used to describe the existing environment and the values and sensitivities within it (Section 3)
- 2. Identify the MEVA where there is the potential for impact to biological receptors at moderate exposure levels or above
- 3. Identify areas of high environmental value (HEV) within the MEVA
- 4. Identify areas of HEV within the EMBA (HEVs are described in Section 7.6.6.2)
- 5. Identify hotspots and evaluate the impacts and risks to them (as described in Section 4). Hot spots are a subset of HEVs and their determination is described in Section 7.6.6.3.
- 6. Identify priorities for response (for consideration in the GEP NT waters OPEP) and monitoring (for consideration in the Northern Australia Operational and Scientific Monitoring Bridging Implementation Plan).

### 7.6.6.1 Spill environment that may be affected

For activities where there is the potential for multiple spill scenarios, the spill scenario, or combination of spill scenarios (e.g., vessel collision), resulting in the greatest spatial extent for potential contact with hydrocarbons is used to define the overall EMBA for the Activity. In this case, the vessel collision spill has a far greater EMBA than the refuelling scenario. The MEVA is also defined as the area within the EMBA with greater concentrations of hydrocarbons which may result in impacts to receptors (Section 7.6.3).

### 7.6.6.2 Areas of high environmental value

Within the EMBA areas that are considered to have high environmental value (HEV), include the following:

- protected area status used as an indicator of the biodiversity values contained within that area, where a World Heritage Property, Ramsar wetland and Marine Protected Area will score higher than areas with no protection assigned
- BIAs and HC of listed Threatened species spatially defined areas where aggregations of individuals of a species are known to display biologically important behaviour, such as breeding, feeding, resting or migration. Each one of these within the predefined areas contributes to the score.
- sensitivity of habitats to impact from hydrocarbons in accordance with the guidance document Sensitivity Mapping for Oil Spill Response produced by International Petroleum Industry Environmental Conservation Association (IPIECA) (2022), the IMO and International Association of Oil and Gas Producers (IOGP)
- sensitivities of receptors with respect to hydrocarbon-impact pathways



- status of zones within protected areas (as in, IUCN [1a] and sanctuary zones compared to IUCN [VI] and multiple use zones)
- listed species status and predominant habitat (surface versus subsurface)
- social values; as in, socioeconomic and heritage features like commercial fishing, recreational fishing, defence and military exercises, tourism, amenities, aquaculture and cultural features.

Tallied scores for each predefined area are then ranked from 1 to 5, with an assignment of 1 representing areas of the highest ecological value and those with 5 representing the areas of the lowest ecological value.

### 7.6.6.3 Hot spots

While the modelled EMBA will be considered during risk assessment and spill response planning, it is best practice to concentrate greatest effort and level of detail on those parts of the EMBA that have:

- the greatest intrinsic ecological value as in, HEVs ranked 1 to 3
- the highest probability of contact by hydrocarbons (either floating or entrained)
- the greatest potential concentration or volume of hydrocarbon accumulating at the receptor.

These areas are termed 'Hot Spots'. Defining Hot Spots is typically the first step in undertaking detailed spill risk assessment and spill response planning. Hot Spots are a subset of HEVA areas that:

- Have the highest probability of contact (at least higher than 5%) above the impact assessment exposure value for surface hydrocarbons and shoreline accumulation based on modelling results
- Receive the greatest concentration or volume of oil, either floating or stranded oil, entrained oil or dissolved aromatic hydrocarbons above contact exposure values described in Section 7.6.5
- Additional areas may be selected as Hotspots for detailed risk assessment, for example if stakeholder consultation has identified areas of particular concern that are not already included in the risk assessment. Additional discretionary hotspots may also be included where they do not strictly meet all of the criteria of a hotspot e.g. a HEV ranked 1-3 with <5% probability, or a HEV ranked 4 or 5 with >5% probability, depending on the concentrations and volumes of hydrocarbons presented in the modelling report. When a discretionary hotspot is added it will be identified as 'discretionary' and the rationale for its inclusion as a hotspot will be described.

Two 'Hot Spots' have been identified under the spill scenarios in this GEP Coastal Waters OEMP (refer to Section 7.6.10.1).

#### 7.6.7 Potential hydrocarbon impact pathways and nature and scale of impact

To help inform the hydrocarbon spill risk assessment, receptors within the EMBA and potential impact pathways have been defined (Table 7-14). The potential impact pathways consider physical and chemical pathways. Physical pathways include contact from floating hydrocarbon, accumulated shoreline hydrocarbon, or entrained hydrocarbon droplets. Chemical pathways include ingestion, inhalation or contact from any hydrocarbon phase. These are summarised in Table 7-14 and the information is drawn upon within the hydrocarbon risk assessment for the spill scenario.

Table 7-15 further describes the nature and scale of the hydrocarbon spills for this Activity on marine fauna and socioeconomic receptors found within the MEVA.

Table 7-14: Physical and chemic	I pathways for h	ydrocarbon exposure and	potential impacts to receptors
---------------------------------	------------------	-------------------------	--------------------------------

Receptor	Physical pathway	Potential impacts	Chemical pathway	Potential impacts
Rocky shorelines	<ul> <li>Shoreline loading and attachment may result in thin and sporadic coating of hydrocarbon residues.</li> <li>Degree of hydrocarbon coating is dependent upon the energy of the shoreline area, the type of rock formation and continual biodegradation of the hydrocarbon</li> <li>Lighter hydrocarbons, such as MDO are less likely to smother the rocks.</li> </ul>	Impacts to flora (mangroves) and fauna further described below.	Chemical pathway to fauna and flora via adsorption through cellular membranes and soft tissue, ingestion, irritation or burning on contact and inhalation.	Impacts to flora (mangroves) and fauna further described below.
Sandy beaches	Shoreline loading and water movement may allow hydrocarbon residue to filter down into sediments, continue to biodegrade on the surface or remobilise into surf zone. Degree of loading is dependent upon the energy and tidal reach of the shoreline, the type of the sandy shore and continual weathering of the hydrocarbon.	<ul> <li>Direct impacts on birds and turtles from becoming exposed to the hydrocarbons at the beach (e.g. loss of food source, coating, inhalation, ingestion).</li> <li>Direct impacts to infauna from exposure to hydrocarbons.</li> </ul>	Chemical pathway to fauna and flora via adsorption through cellular membranes and soft tissue, ingestion, irritation or burning on contact and inhalation.	Indirect impacts to nesting and foraging habitats for birds and turtles. Direct impacts (mortality) to infauna through toxic effects and smothering.
Intertidal platforms	Shoreline loading and water movement may allow hydrocarbon residue to filter down into sediments (such as within wetlands) or continue to biodegrade on the surface or remobilise into surf zone. Degree of loading is dependent upon the energy and tidal reach of the shoreline, the type of the substrate and continual weathering of the hydrocarbon.	<ul> <li>Direct impacts on birds and turtles from becoming exposed to the hydrocarbons (e.g. loss of food source, coating, inhalation, ingestion)</li> <li>Direct impacts to infauna from exposure to hydrocarbons</li> <li>Impacts to shoreline habitats contacted within the MEVA are likely to be more prolonged from an HFO release due to its persistent nature.</li> </ul>	Chemical pathway to fauna and flora via adsorption through cellular membranes and soft tissue, ingestion, irritation or burning on contact and inhalation.	Indirect impacts to foraging habitats for birds. Direct impacts (mortality) to infauna through toxic effects and smothering.

Receptor	Physical pathway	Potential impacts	Chemical pathway	Potential impacts
		<ul> <li>Light hydrocarbons such as MDO reaching intertidal platforms are likely to be heavily weathered, reducing the toxic effects.</li> </ul>		
Shallow sub-tidal soft sediments	Hydrocarbon residue in the shallow waters adjacent to shorelines may settle to filter down into sediments. Degree of loading is dependent upon the energy and tidal reach of the shoreline, the type of the substrate and continual weathering of the hydrocarbon.	<ul> <li>Direct impacts on birds and turtles from becoming exposed to the hydrocarbons (e.g. loss of food source, coating, inhalation, ingestion)</li> <li>Direct impacts to infauna from exposure to hydrocarbons.</li> </ul>	Adsorption via cellular membranes and soft tissue, ingestion, irritation or burning on contact and inhalation.	Indirect impacts to foraging habitats for turtles and fish. Direct impacts (mortality) to infauna through toxic effects and smothering.
Mangroves	Coating of root system and pneumatophores, reducing air and salt exchange. Degree of coating is dependent upon the energy and tidal reach of the shoreline, the type of the substrate and continual weathering of the hydrocarbon.	<ul> <li>Yellowing of leaves</li> <li>Defoliation</li> <li>Increased sensitivity to stressors</li> <li>Tree death</li> <li>Reduced growth</li> <li>Reduced reproductive output</li> <li>Reduced seed viability.</li> </ul>	External contact by hydrocarbon and adsorption across cellular membranes.	<ul> <li>Yellowing of leaves</li> <li>Defoliation</li> <li>Increased sensitivity to stressors</li> <li>Tree death</li> <li>Reduced growth</li> <li>Reduced reproductive output</li> <li>Reduced seed viability</li> <li>Growth abnormalities.</li> </ul>
Seagrasses and macroalgae	Most seagrasses in the EMBA are sub-tidal, although there may be small areas of intertidal seagrasses. Sub- tidal seagrasses are unlikely to be exposed to floating hydrocarbons, but may be contacted by entrained or dissolved fractions, which can be absorbed into tissues. The potential for toxic effects of entrained hydrocarbons may be reduced by weathering processes that should lower	<ul> <li>Bleaching or blackening of leaves</li> <li>Defoliation</li> <li>Reduced growth</li> <li>Fouling.</li> </ul>	External contact by hydrocarbon and adsorption across cellular membranes.	<ul> <li>Mortality</li> <li>Bleaching or blackening of leaves</li> <li>Defoliation</li> <li>Disease</li> <li>Reduced growth</li> <li>Reduced reproductive output</li> <li>Reduced seed and propagule viability.</li> </ul>

Receptor	Physical pathway	Potential impacts	Chemical pathway	Potential impacts
	<ul> <li>the content of soluble aromatic components before contact occurs. Long-term impacts to seagrass are unlikely unless hydrocarbons are retained within the seagrass meadow for a sustained duration (Wilson and Ralph 2011).</li> <li>If contacted by floating hydrocarbons, intertidal seagrasses are vulnerable to smothering, which can lead to mortality if it coats their flowers, leaves and stems (Taylor &amp; Rasheed 2011).</li> </ul>			
Hard and soft corals (coral reefs)	Coating of polyps, shading resulting in reduction on light availability. Degree of coating is dependent upon the metocean conditions, dilution, if corals are emergent at all and continual weathering of the hydrocarbon.	<ul> <li>Bleaching</li> <li>Increased mucous production</li> <li>Reduced growth.</li> </ul>	External contact by hydrocarbon and adsorption across cellular membranes.	<ul> <li>Mortality</li> <li>Cell damage</li> <li>Reduced metabolic capacity</li> <li>Reduced immune response</li> <li>Disease</li> <li>Reduced growth</li> <li>Reduced reproductive output</li> <li>Reduced egg and larval success</li> <li>Growth abnormalities.</li> <li>(Loya &amp; Rinkevich, 1980; White et al. 2012; Fisher et al. 2014).</li> </ul>
Non-coral benthic invertebrates	<ul> <li>Coating of adults, eggs and larvae</li> <li>Degree of coating is dependent upon the energy and tidal reach of the shoreline, the type of the receptor and continual</li> </ul>	<ul><li>Mortality</li><li>Behavioural disruption</li><li>Impaired growth.</li></ul>	<ul> <li>Ingestion and inhalation</li> <li>External contact and adsorption across exposed skin and cellular membranes</li> <li>Uptake of DAH across cellular membranes.</li> </ul>	<ul> <li>Mortality</li> <li>Increases in bacterial abundance leading to opportunistic community structure</li> <li>Decrease in species richness, abundance and diversity</li> </ul>

Receptor	Physical pathway	Potential impacts	Chemical pathway	Potential impacts
Ohada area and alk as fish	weathering of the hydrocarbon.		Reduced mobility and capacity for oxygen exchange.	<ul> <li>Reduced growth</li> <li>Impaired growth</li> <li>Growth abnormalities</li> <li>Behavioural disruption.</li> <li>(Schwing et al. 2020; Montagna et al. 2013; Baguley et al. 2015).</li> </ul>
Sharks, rays and other fish	Coating of adults but primarily eggs and larvae – reduced mobility and capacity for oxygen exchange.	<ul> <li>Mortality</li> <li>Oxygen debt</li> <li>Starvation</li> <li>Dehydration</li> <li>Increased predation</li> <li>Behavioural disruption.</li> </ul>	<ul> <li>Ingestion</li> <li>External contact and adsorption across exposed skin and cellular membranes</li> <li>Uptake of DAH across cellular membranes (for example, gills)</li> <li>Due to the filter-feeding nature of whale sharks, they may be susceptible to ingesting floating and entrained hydrocarbons, particularly if foraging at or near the sea surface.</li> </ul>	<ul> <li>Mortality</li> <li>Decrease in biomass</li> <li>Cell damage</li> <li>Starvation</li> <li>Increased predation</li> <li>Delayed growth</li> <li>Reduced reproductive output</li> <li>Reduced egg and larval success</li> <li>Growth abnormalities</li> <li>Behavioural disruption</li> <li>Reduced immune response</li> <li>Change in community structure</li> <li>Decrease in species richness, abundance and diversity.</li> <li>(Lewis et al. 2020; Ainsworth et al. 2018; Fisher, 2016).</li> </ul>
Birds (seabirds and shorebirds)	Physical coating occurs upon contact of contaminated shorelines and/or exposure to floating oil during foraging at sea or resting at the sea surface.	<ul> <li>Feather and skin irritation and damage, with the potential to cause secondary impacts such as:</li> <li>physical restriction of flight and swimming movement</li> <li>reduced buoyancy</li> <li>more vulnerable to predation</li> </ul>	Ingestion (during feeding or preening). External contact and adsorption across exposed skin and membranes.	<ul> <li>Reduced metabolic capacity</li> <li>Reduced immune response</li> <li>Reduced growth</li> <li>Reduced hatchling success</li> <li>Reduced reproductive output</li> <li>Growth abnormalities</li> <li>Behavioural disruption</li> <li>Mortality</li> </ul>

Receptor	Physical pathway	Potential impacts	Chemical pathway	Potential impacts
		<ul> <li>potential for secondary infections</li> <li>mortality</li> <li>hypothermia or impairing of the waterproofing of feathers</li> <li>disruption to feeding or starvation</li> <li>disruption to breeding</li> <li>disruption to migration.</li> </ul>		<ul> <li>Potential for secondary infections.</li> <li>(Deepwater Horizon Oil Spill Natural Resource Trustee Council, 2016; Unlu et al. 2018).</li> </ul>
Marine reptiles	Physical coating occurs upon contact of contaminated shorelines and/or exposure to floating oil when at the sea surface. Eggs may also become contaminated during laying, either from the laying female or the contaminated sand.	<ul> <li>Irritation of eyes and mouth and potential illness, which may cause secondary impacts such as:</li> <li>mortality</li> <li>disruption to feeding or starvation</li> <li>physical restriction</li> <li>behavioural disruption.</li> </ul>	<ul> <li>Inhalation</li> <li>Ingestion</li> <li>External contact and adsorption across exposed skin and membranes</li> <li>Contamination of eggs</li> <li>Exposure of turtle habitats.</li> </ul>	<ul> <li>Reduced metabolic capacity</li> <li>Reduced immune response</li> <li>Reduced growth</li> <li>Reduced hatchling success</li> <li>Reduced reproductive output</li> <li>Growth abnormalities</li> <li>Behavioural disruption</li> <li>Mortality</li> <li>Potential for secondary infections.</li> <li>(Deepwater Horizon Oil Spill Natural Resource Trustee Council, 2016; Unlu et al. 2018).</li> </ul>
Marine mammals	<ul> <li>Coating of feeding apparatus in some species (baleen whales) from exposure to floating hydrocarbons</li> <li>Potential to coat the sensory hairs around the mouths of dugongs which can impact feeding.</li> </ul>	<ul> <li>Irritation of eyes and mouth, damage to fur and potential illness, which may cause secondary impacts such as:</li> <li>mortality</li> <li>disruption to feeding and starvation</li> <li>physical restriction</li> <li>behavioural disruption.</li> </ul>	<ul> <li>Inhalation</li> <li>Ingestion</li> <li>External contact and adsorption across exposed skin and membranes.</li> </ul>	<ul> <li>Mortality</li> <li>Cell damage, lesions</li> <li>Secondary infections</li> <li>Reduced metabolic capacity</li> <li>Reduced immune response</li> <li>Disease</li> <li>Reduced growth</li> <li>Reduced reproductive output</li> <li>Growth abnormalities</li> <li>Behavioural disruption</li> </ul>
Receptor	Physical pathway	Potential impacts	Chemical pathway	Potential impacts
------------------------------------	------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------	------------------------------------------------------------------------------------------------	--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------	---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------
				<ul> <li>Lung, respiratory and adrenal impairment.</li> <li>(Deepwater Horizon Oil Spill Natural Resource Trustee Council, 2016; Bejder &amp; Gartner, 2016).</li> </ul>
Plankton	<ul> <li>Coating of feeding apparatus</li> <li>Reduced mobility and capacity for oxygen exchange.</li> </ul>	<ul> <li>Mortality</li> <li>Behavioural disruption (for example, reduced mobility).</li> </ul>	<ul> <li>Inhalation</li> <li>Ingestion</li> <li>External contact.</li> </ul>	<ul> <li>Mortality</li> <li>Impairment of biological activities (for example, feeding, respiration)</li> <li>Reduced mobility</li> <li>Cell damage</li> <li>Reduced growth</li> <li>Reduced reproduction</li> <li>Increased opportunistic species impacting community structure</li> <li>Decrease in species density and richness</li> <li>Decrease in total primary production.</li> <li>(Ozhan, 2014).</li> </ul>
Water quality and sediment quality	<ul> <li>Presence of hydrocarbon residue in the water, which may filter down to sediments or continue to biodegrade on the surface</li> <li>Degree of loading in the water column is dependent upon the influence of wave energy and tidal range.</li> </ul>	Impacts to flora and fauna, as discussed in rows above.	<ul> <li>Adsorption via cellular membranes and soft tissue, ingestion, irritation or burning on contact and inhalation</li> <li>Impacts to flora and fauna, as discussed in rows above.</li> </ul>	Impacts to flora and fauna, with emphasis on the ecosystem impacts of: • trophic shifts • community structure shifts • reduced growth • impaired reproduction • adverse health effects. (Deepwater Horizon Oil Spill Natural Resource Trustee Council, 2016).

Receptor	Physical pathway	Potential impacts	Chemical pathway	Potential impacts
Protected areas	Coating of benthic habitats and marine fauna and flora within protected areas, as discussed in rows above.	<ul> <li>Mortality, injury or behavioural disruption to marine fauna</li> <li>Death or impairment of habitats within protected areas</li> <li>Reduction in the quality of the marine environment within protected areas</li> <li>Environmental value of protected areas is degraded.</li> </ul>	Impacts to flora and fauna, as discussed in rows above.	<ul> <li>Mortality, injury or behavioural disruption to marine fauna</li> <li>Death or impairment of habitats within protected areas</li> <li>Reduced growth of benthic habitats</li> <li>Reduction in the quality of the marine environment within protected areas</li> <li>Environmental value of protected areas is degraded.</li> </ul>
Socioeconomic environment (commercial and recreational fisheries, tourism, shipping, defence)	<ul> <li>Presence of hydrocarbon residue in the water, which may filter down to sediments or continue to biodegrade on the surface</li> <li>Presence of weathered hydrocarbon on the shoreline</li> </ul>	<ul> <li>Degradation of UCH sites</li> <li>Disruption to tourism, recreation, defence and military exercises or shipping activities</li> <li>Displacement of commercial or recreational fishing</li> <li>Reduction in natural resources.</li> </ul>	Impacts to water quality, sediment quality, flora and fauna, as discussed in rows above.	<ul> <li>Mortality, injury or behavioural disruption to marine fauna relevant to commercial, and recreational fisheries or to tourism</li> <li>Loss or degradation of habitats within protected areas</li> <li>Reduced growth of benthic habitats</li> <li>Reduction in the quality of the marine and shoreline environment within protected areas</li> <li>Socio-economic value of protected areas is degraded.</li> </ul>
Cultural features (native title, ILUAs, IPAs, sacred sites, marine parks, cultural fishing, hunting and gathering, marine fauna representing totemic species or species associated with dreaming and sea country)	<ul> <li>Presence of hydrocarbon residue in the water, which may filter down to sediments or continue to biodegrade on the surface</li> <li>Presence of weathered hydrocarbon on the shoreline.</li> </ul>	<ul> <li>Hydrocarbons may be present in areas with cultural features (e.g. ILUAs, IPAs, sacred sites, marine parks, cultural fishing, hunting and gathering and sea country)</li> <li>Displacement of traditional uses of environment.</li> </ul>	Impacts to water quality, sediment quality, flora and fauna, as discussed in rows above.	<ul> <li>Mortality, injury or behavioural disruption to marine fauna. In accordance with First Nations people cultural beliefs, if totemic species (e.g. turtles, fish, marine mammals and birds) are impacted by the Activity some believe this in turn can impact First Nations</li> </ul>

Receptor	Physical pathway	Potential impacts	Chemical pathway	Potential impacts
		<ul> <li>Reduction in natural resources with cultural significance, refer above rows for potential impacts on fauna.</li> </ul>		people and make them sick. The potential impacts to culturally significant marine fauna species (such as dreaming and totem species including marine mammals, marine reptiles, fish and birds) are assessed separately above.
				<ul> <li>Loss or degradation of habitats of cultural value</li> </ul>
				Reduction in the quality of the marine and shoreline environment, including environment with cultural significance
				Cultural value of cultural features is degraded.

#### Table 7-15: Nature and scale of hydrocarbon spills on environment and socioeconomic receptors within the moderate exposure value area (MEVA)

Receptor	Impacts of hydrocarbon spills	
	Entrained and dissolved aromatic hydrocarbons in the water column	Floating hydrocarbons
Threatened/Migrat	ory fauna	
Marine mammals	There is potential for sublethal or lethal impacts to marine mammals and impacts to reproduction and behaviour from an accidental release of hydrocarbons. A wide range of effects from hydrocarbons have been reported in cetaceans including poor body condition, calcium imbalance, inflammation, reproductive failure, lung and adrenal gland damage, altered hepatobiliary function, immune changes and increased susceptibility to infections, impaired stress response, and death (Godard-Codding and Collier, 2018).	Marine mammals are at risk of direct contact with floating hydrocarbons at the moderate threshold when surfacing within slick. Effects include irritation of eyes or mouth and potential illness. The direct physical coating of marine mammals with hydrocarbons is more likely to occur with more persistent hydrocarbons such as. Surface respiration could lead to accidental inhalation of hydrocarbons or result in the coating of sensitive epidermal surfaces. Accidental ingestion could also occur through the ingestion of hydrocarbon during feeding or the ingestion of contaminated prey. Inhalation of vapours or the ingestion of hydrocarbons can potentially have lethal effects due to damage to the whale's respiratory and nervous systems. However, cetaceans and dugongs are highly mobile, capable of long mirritiane and tunional to a tunional to a tunional to a sufficient of the MEV(A. Experimental to a sufficient of the MEV).

		and field observations indicate that whales and dolphins may be able to detect and actively avoid floating hydrocarbon slicks, but this may not always be possible and exposure to floating oil may still occur (Smith <i>et al.</i> 1983, Geraci and St. Aubin 1990).
	Marine mammal and the potential of them occurring within the MEVA are prese whale) and two as Vulnerable (fin whale and sei whale). There are no BIAs for p they are likely to be transient and in low numbers. Fin whales are unlikely to occ	nted in Section 3.2.13.1. Of these, one is listed as Endangered (pygmy blue bygmy blue whales or sei whales identified within the EMBA and, if present, cur in the EMBA.
	A vessel collision releasing large volumes of MDO in the OA has potential to en dolphin in the vicinity of Darwin Harbour. Impacts may include behavioural imparare circumstances, mortality.	compass a small portion of the breeding BIA for the Indo-Pacific humpback acts (such as avoidance of impacted areas), sub-lethal biological effects and, in
	Dugongs are known to occur in coastal waters, including those of the Tiwi Island Indonesian offshore islands, particularly in areas of seagrass. Direct impacts to hydrocarbon or through direct exposure to hydrocarbons. Dugongs could also b feeding areas.	d such as the seagrass sites on the north-west of Melville Island, and around dugongs could occur through foraging or ingesting seagrass coated with indirectly affected if hydrocarbons cause the dieback of seagrass, reducing
Marine reptiles	There is potential for sublethal or lethal impacts to marine reptiles from an accidental release of hydrocarbons. Exposure can alter biochemical and haematological parameters, weight, skin function, metabolism, immune responses, diving patterns, and respiration (Ruberg et al., 2021). Marine turtles are susceptible to the effects of hydrocarbon spills during all life stages and are not expected to exhibit avoidance behaviour if they encounter hydrocarbon spills.	Marine turtles are at risk of direct contact with floating hydrocarbons when surfacing within slick. Effects include irritation of eyes or mouth and potential illness as adults can suffer mucus membrane inflammation, increasing susceptibility to infection (ITOPF, 2011). Surface respiration could lead to accidental ingestion of hydrocarbons or result in the coating of sensitive epidermal surfaces. Breathing and inhalation of toxic vapours may occur from exposure to hydrocarbons in surface waters.
		Physical coating of marine turtles also occurs upon contact of contaminated shorelines. Eggs may also become contaminated during laying, either from the laying female or the contaminated sand.

	Marine reptiles and the potential of them occurring within the EMBA and MEVA were identified, including loggerhead, green, leatherback, hawksbill, flatback an the EMBA and MEVA.	are presented in Section 3.2.13. Seven species of threatened marine reptile d olive ridley turtles. The migratory saltwater crocodile was also identified within
	Various BIAs and habitat critical to the survival of marine turtles in proximity to t turtles.	he Tiwi Islands are within the MEVA. This includes BIAs seven species of
	A vessel releasing MDO in the OA may lead to a greater probability of impact to Potential impacts offshore would be greatest during the internesting season: be ridley turtles. Population level impacts are considered unlikely as the hydrocarb survival of these species.	o flatback and olive ridley turtles, given the proximity to the Tiwi Islands. tween June and September for flatback turtles and April to August for olive ons are not predicted to contact the entire BIAs or areas of habitat critical to the
	Hydrocarbons may accumulate on shorelines, including Tiwi Islands and Indone nesting beaches seasonally to reproduce, which makes them vulnerable to impa- females and emergent hatchlings (Lauritsen et al., 2017). Potential impacts wou as a result of vessel collision in the OA may result in small quantities of MDO ac interacting with the nesting beaches is likely to represent the persistent fraction the hydrocarbon weathers, the potential impact of egg viability is reduced. Fresl 2002). Adult and juvenile turtles during nesting seasons may become coated in hydrocarbons as they pass through the affected area. While turtle eggs are unli the high tide level, they may be directly exposed through the transfer of hydrocar Seasnakes may be found throughout the MEVA particularly at nearby shoals a	esian Islands where turtle nesting beaches are present. Marine turtles rely on acts from hydrocarbon accumulated on shorelines, through oiling of nesting uld be greatest during the peak nesting periods. A worst-case release of MDO occumulating on the Tiwi Islands (<1 m <sup>3</sup> ). Any accumulated hydrocarbons in the form of viscous liquid and as tar balls as the hydrocarbon weathers. As in hydrocarbons may have a significant impact on success rate (Milton et al., the hydrocarbon as they move to and from shore and may also ingest kely to be exposed to shoreline hydrocarbons, as most turtles nest well above arbons from the oiled female turtle (Shigenaka, 2003).
	from direct contact with surface hydrocarbons are likely to be similar to those ex mucous membranes of the eyes, nose and throat. Saltwater crocodiles may be	perienced by marine turtles; for example, potential skin damage and irritation of present in the inshore/coastal areas and could be contacted by hydrocarbons.
Birds (seabirds and migratory shorebirds)	There is potential for injury or mortality to seabirds and shorebirds and a change in their behaviour from an accidental release of hydrocarbons. Seabirds may encounter entrained hydrocarbons while diving and foraging. Seabirds and shorebirds encounter hydrocarbon contaminated materials when foraging at intertidal areas. Lethal or sub-lethal physical and toxic effects include those such as such as irritation of eyes or mouth and potential illness.	Seabirds are particularly vulnerable to floating hydrocarbons. As most fish survive beneath floating slicks, they will continue to attract foraging seabirds, which typically do not exhibit avoidance behaviour. Smothering can lead to reduced water-proofing of feathers and ingestion while preening. In addition, direct contact with hydrocarbons can erode feathers, causing chemical damage to the feather structure that subsequently affects ability to thermoregulate and maintain buoyancy on water.
	Hydrocarbons from worst-case releases may accumulate on shorelines includin	Physical coating may also occur on contact of contaminated shorelines.
	in the OA may also result in smaller quantities of MDO accumulating on the Tiw	i Islands ( $<1 \text{ m}^3$ ).
	It is possible seabird populations can recover from large-scale spills. For examp number of non-breeders in the population that may buffer the loss of reproductiv adult losses can be more rapidly replaced (Oates, 2016). Other long-term studie Prestige hydrocarbon spill in the North Atlantic, had not recovered to pre-spill le predicting population recovery times is difficult, as the effects of hydrocarbon population dynamics (Oates, 2016).	ble, species with long life spans and high survival rates contain a substantial ve adults, while other species have a higher reproductive potential such that es have indicated seabird populations affected by significant spills, such as the vels eight to ten years after the spill occurred. However, it is acknowledged ollution cannot always be differentiated from natural environmental variation and

Sharks, rays and fish	There is potential injury or mortality to sharks, rays and fish and a change in their behaviour from an accidental release of hydrocarbons. As fish dwell in the water column, impacts are most likely from exposure to entrained or dissolved hydrocarbons, through the pathways of ingestion or the coating of gill structures, resulting in reduced oxygen exchange and incidence of irritation and infection. Fish may also ingest hydrocarbon droplets or contaminated food, leading to reduced growth.	While fish, sharks and rays do not generally break the sea surface, individuals may feed at the surface. Prolonged exposure to floating hydrocarbons by fish, shark and ray species is unlikely. Due to the filter-feeding nature of whale sharks, they may be susceptible to ingesting floating and entrained hydrocarbons, particularly if foraging at or near the sea surface.
	There is potential for localised mortality of fish eggs and larva due to reduced water quality and toxicity. Based on the modelling results (Section 7.6.3), fish eggs and larva will be exposed to hydrocarbons in the top 25 m of the water column, with the highest concentrations in the upper 10 m of the water column and areas close to the spill source.	
	Demersal fish are highly unlikely to be impacted by the hydrocarbon releases, as they generally inhabit waters near the seabed (hydrocarbons will be concentrated in the upper 25 m of the water column). Environmental monitoring of pelagic and demersal fishes immediately after the Montara oil spill indicated fish were exposed to hydrocarbons, although no adverse effects were detected (Gagnon & Rawson, 2011). Further sampling and testing over time indicated fish captured in proximity to the Montara wellhead were comparable to those collected from reference sites (Gagnon & Rawson, 2012).	
	Seven threatened species of fish and sharks were identified by the PMST, inclu and green sawfish (NT-listed Vulnerable) and northern river shark. Site-attached to hydrocarbons at harmful levels for longer durations.	ding the white shark, whale shark, speartooth shark, freshwater sawfish, dwarf d fish associated with shallow shoals and banks in the MEVA may be exposed
	Whale sharks do not spend all their time in surface water; rather, routinely move can spend most of their time near the seafloor, reducing the likelihood of impact not expected to exceed depths greater than approximately 25 m.	e between surface and to depths of greater than 30 m, and in offshore regions t, given the modelling (Section 7.6.3) predicts hydrocarbon concentrations are
Plankton	·	

Plankton (including zooplankton, fish and coral larvae)	Direct exposure of plankton to hydrocarbons may result in lethal or sublethal impacts to plankton and impact mobility, feeding and respiration. Plankton could include the eggs and larvae of marine invertebrates and fish; therefore, entrained hydrocarbon could have secondary impacts on recruitment of invertebrate and fish species. Based on the modelling results (Section 7.6.3) plankton will be exposed to hydrocarbons in the top 25 m of the water column, with the highest concentrations in the upper 10 m of the water column and areas close to the spill source.	Plankton utilising the sea surface layer could be impacted by floating hydrocarbon.
	Some studies have shown no obvious influence of hydrocarbon spills on plankton community structure (Varela et al., 2006), which could be a result of rapid replacement of stocks from adjacent areas due to water circulation (Batten et al., 1998). Other studies, however, have found the concentrations of phytoplankton reduced in the short term, and in the medium term, as outbreaks of algal blooms occurring where the Chlorophyll-a concentration increased (Lee et al., 2009; Sheng et al., 2011), particularly under warmer weather conditions (Tang et al., 2019) and in low energy environments such as coastal coves (Zhou et al., 2014).	
	Once water quality returns to background levels, it is anticipated plankton communities can return to normal densities and community structures due to their ability to produce large numbers of eggs and juveniles, their wide distribution, and rapid water exchange.	
	The MEVA has the potential to overlap with spawning areas of fish species; how times for species. Some impacted spawn may be of commercial interest (refer s larvae that occurs in the lifecycle of most fish species provides a buffer for recrudetectable impact on adult fish populations (ITOPF, 2014).	vever, the extent of impacts to plankton contact will depend on the spawning socioeconomic receptors below). The typical mass over-production of eggs and uitment, which further reduces the likelihood that a spill would have a significant
Benthic communit	ties	
Benthic communities	Shallow banks and shoals within the top 20 m of the water column occur within hydrocarbons may contact Shepparton Shoal, Harries bank, and Van Cloon-De	the MEVA. Modelling results (Section 7.6.3) show entrained or dissolved ep shoals.
	Banks and shoals support a diverse and varied range of benthic communities, r and Blackwood shoals and Lynedoch Bank, which are adjacent to the EMBA, re fish and pelagic fish (Heyward et al., 2012, 1997b). It is expected other shoals in would be characterised by similar communities.	eef-building soft corals, hard corals and filter-feeders. Surveys of Tassie, Evans ecorded coral and algae species, filter-feeder communities, sponges, demersal n the EMBA – such as Margaret Harris Bank and Van Cloon-Deep shoals –
	Benthic communities on the banks and shoals are vulnerable to hydrocarbons. ecosystem, affecting species of fish communities and other marine invertebrates hydrocarbons while feeding over the area. This may cause mortality or sublethal activity and reduced growth rates.	The loss of habitat-forming benthic biota may impact an entire bank or shoal s. Filter feeders are particularly susceptible as they are likely to directly ingest I impacts such as alteration in respiration rates, decreases in filter-feeding



Shoreline habitats	
Shoreline habitats	Rocky and sandy shorelines occur within the MEVA throughout the coastlines of the NT. Based on the modelling (Section 7.6.3), there is the potential (albeit in low probabilities) for accumulation of hydrocarbons at the moderate threshold at multiple shorelines locations. A worst-case release of MDO as a result of vessel collision in the OA may also result in small quantities of MDO accumulating on the Tiwi Islands (<1 m <sup>3</sup> ).
	The severity of impact of hydrocarbon on rocky shorelines largely depends on the hydrocarbon type, the incline of the rocky shoreline and the energy environment. On steep or vertical rock faces on wave-exposed coasts, there is likely to be little impact from a spill event, as the hydrocarbon does not typically accumulate due to wave action. Lighter hydrocarbons, such as MDO and condensates, are less likely to smother the rocks.
	Sandy beach ecosystems are attributable to the benthic invertebrate fauna – such as polychaetes, molluscs, marine crustaceans, semi terrestrial crustaceans and insects – inhabiting the sediments. However, sandy beaches also provide important habitats for nesting turtles, breeding and foraging seabirds, and shorebirds (impacts discussed in prior section). The long-term persistence of the hydrocarbons on sandy beaches will depend on the wave exposure and concentrations within sediments.
	Shoreline contact at the low threshold is anticipated to result in a reduction in visual amenity of shorelines only.
Intertidal/subtidal	habitats
Seagrasses and macroalgae	Seagrasses and macroalgae occur within the MEVA along the coastlines of the NT. Based on the modelling (Section 7.6.3) there is the potential (albeit in low probabilities) for accumulation of hydrocarbons at the moderate threshold at multiple locations where seagrasses are present. A worst-case release of MDO as a result of vessel collision in the OA may also result in small quantities of MDO accumulating on the Tiwi Islands (<1 m <sup>3</sup> ).
	Most seagrasses are subtidal, although there may be relatively small areas of intertidal seagrasses. The potential for toxicity effects of entrained hydrocarbon may be reduced by weathering processes that should serve to lower the content of soluble aromatic components before contact occurs. Hydrocarbons are expected to be highly weathered before reaching shallow areas where seagrasses may occur. The highest impact on seagrasses have been observed when leaves of intertidal plants have been exposed to direct contact with hydrocarbons (Durako et al., 1993; Jackson et al., 1989). Smothering through algal blooms (Jacobs, 1980), shoot mortality (Peirano et al., 2005) and a reduction in seagrass tolerance to other stress factors (Zieman et al., 1984) have also been documented as a result of hydrocarbon spills. Long-term impacts to seagrass are unlikely unless hydrocarbon is retained within the seagrass meadow for a sustained duration (Wilson & Ralph, 2011).
Mangroves	Intertidal mangrove habitats occur within the MEVA, along the coastlines of the NT. Based on the modelling (Section 7.6.3), there is the potential (albeit in low probabilities) for accumulation of hydrocarbons at the moderate threshold at multiple locations where mangroves are present. A worst-case release of MDO as a result of vessel collision in the OA may also result in small quantities of MDO accumulating on the Tiwi Islands (<1 m <sup>3</sup> ).
	The severity of exposure for mangroves largely depends on the amount and type of hydrocarbon entering the intertidal zone (Duke, 2016). While heavy hydrocarbons are particularly proficient at coating and smothering small plants and aerial root systems, lighter hydrocarbons such as MDO with low specific gravity, are more toxic to mangroves (Hensel et al., 2014; Connolly et al., 2020). The potential for toxicity effects from hydrocarbons may be reduced overtime by weathering processes that should serve to lower the content of soluble aromatic components.
	Observations of offshore hydrocarbon spill events have shown large scale hydrocarbon spills can result in persistent or permanent loss of mangrove habitat, with some capacity to recover over time (Duke, 2016).
Intertidal platforms	Intertidal platforms and mudflats occur within the MEVA, along the coastlines of the NT and Tiwi islands. Based on the modelling (Section 7.6.3), there is the potential (albeit in low probabilities) for accumulation of hydrocarbons at the low threshold at multiple locations where intertidal sand and mudflats are present. A worst-case release of MDO as a result of vessel collision in the OA may also result in smaller quantities of MDO accumulating on the Tiwi Islands (<1 m <sup>3</sup> ).
	Intertidal platforms and mudflats are typically a low-energy environment heavily influenced by tidal cycle. They therefore have the potential to trap hydrocarbons, increasing their susceptibility to impacts. Sediment quality in mudflats will be reduced in the area of the mud or sand flat from hydrocarbon accumulation, with finer sediments being more susceptible as persistent hydrocarbons which can penetrate through animal burrows and root pores. Intertidal

	mudflats provide important resting and feeding areas for migratory bird species.		
	MDO reaching intertidal platforms are likely to be heavily weathered, reducing the toxic effects.		
Socioeconomic			
Commercial, recreational and traditional fisheries	Hydrocarbons in the water column can have toxic effects on fish (as outlined above) and lead to a reduction in catch rates. Fish may also be tainted by the hydrocarbons, rendering them unsafe for human consumption. Impacts on spawning fish can also result in impacts to commercial fisheries.	In addition to the effects of entrained and DAHs, exclusion zones surrounding a spill can directly impact fisheries by restricting access for fishers. Weathered slicks may form tar balls, which may result in oiling of nets and fishing infrastructure.	
	A number of commercial fisheries may operate within the MEVA, given the exte fishing activities caused by the physical presence of the slick, loss of catch, dec	nt. Impacts to these fisheries from a spill include a disruption or displacement of line in commercially important fish stocks and suspension of fishing operations.	
	Southern bluefin tuna are known to spawn within the MEVA; therefore, a hydrocarbon spill occurring during spawning or movement from spawning grounds to the southern coast could have effects on the commercial fishery stock. It is likely other commercial fish that are targeted in the region (refer to Section 3.2.14.1) could also be affected if spawning occurs during a hydrocarbon spill event.		
	Exposure to entrained and dissolved oils could result in the accumulation of hydro fish flesh. Connell and Miller (1981a, 1981b) compiled a summary of studies list hydrocarbons. The results contained in their review indicate tainting of fish occu 300,000 ppb) of hydrocarbons in the water, for durations of 24 hours or more, w entrained hydrocarbons are predicted to exceed the moderate exposure value a Although it is difficult to assess how long fish might be exposed for, small, less r detected to fisheries on a stock level, although it is more likely natural variation a hydrocarbon spill. This would most likely be the case for fisheries species that direct impacts to fish or to fish habitats (for example, seagrass, coral reef, mang expected to retain a taint for longer than a week after exposure to entrained or of Babcock, 2016)	Brocarbon in fish tissues to the extent that could result in hydrocarbon taint of sing the exposure value concentrations at which tainting occurred for ars when they are exposed to ambient concentrations of 4 to 300 ppm (4000 to with response to phenols and naphthenic acids being the strongest. Given at some locations in the MEVA, hydrocarbon taint is possible in fish flesh. mobile fishes would be more susceptible. It is possible impacts could be in fish abundance would be on a greater scale than any impacts attributable to the scale shallow waters around the banks and shoals and could occur through grove habitats which are present within the MEVA). In general, fish are not dissolved hydrocarbons (Gagnon & Holdway, 2000, cited in Westera &	
	The same negative impacts could also occur to important traditional Indonesian	and recreational fish target species.	
	Commercial, recreational and traditional fisheries may be impacted within the E	MBA due to wider implications of taint on fish species.	
Recreation and tourism	There is limited tourism and recreation in remote, offshore waters; however, sor displace recreation and tourism users from the EMBA, and impact upon natural Contact at the low exposure threshold has the potential to result in a reduction i long-term impacts to tourism and recreation activities.	ne shoals and banks may be frequented. A hydrocarbon spill may temporarily resources (such as fish) targeted and seascapes valued by these users. n visual amenity of shorelines. It is considered highly unlikely there will be	
Shipping	MDO in the water column will have no effect on shipping.	Exclusion zones surrounding a spill will reduce access for shipping vessels for the duration of the response undertaken for spill clean-up (if applicable. Ships may have to chart alternative routes, leading to potential delays and increased costs.	
Defence	The level of defence activities performed near the OAs is low, though the MEVA zone surrounding a spill has the potential to adversely affect defence activities. minimal.	A does overlap some of the NAXA for the military training exercise. An exclusion Interference with defence activities due to a hydrocarbon spill is expected to be	

Shipwrecks	Floating hydrocarbons will have no impact on shipwrecks as all shipwrecks within the MEVA are submerged and therefore will not extensively be contacted by floating hydrocarbons. The potential for in-water hydrocarbons to impact on shipwrecks is poorly documented. Based on the modelling results (Section 7.6.3 hydrocarbons are present in the top 25 m of the water column; therefore, extensive contact with submerged shipwrecks is not anticipated. Exposure to hydrocarbon may alter bacterial community composition (biofilms) inhabiting shipwrecks, possibly altering corrosion potential (Salerno et al., 2018). The biofilms promote the recruitment of macro-organisms and can form protective surfaces that may decrease access for abiotic corrosion and may assist with the preservation of historic metal shipwrecks (dependent on the environmental conditions). Further studies have provided evidence that exposure of shipwreck surfaces to residual spill contaminants has the potential to alter biofilm taxonomy and functional potential, which may place the biodiversity and the preservation of historic metal structures in the deep sea at risk (Mugge et al., 2019).
Cultural features	Marine resource use by Indigenous people is generally restricted to coastal waters. Fishing, hunting and the maintenance of maritime cultures and heritage through ritual, stories and traditional knowledge continue as important uses of the nearshore region and adjacent areas. While the MEVA is largely offshore, it may overlap with cultural features. Impacts to these features from a spill include, but are not limited to, a disruption/displacement of cultural activities caused by the physical presence of hydrocarbon, decline in traditional food sources and / or mortality of fauna with cultural significance e.g. totemic species.
Existing energy industry	A number of energy industry operators have existing infrastructure within, and would transit through, the MEVA (such as Santos Bayu-Undan and INPEX Ichthys gas export pipelines). An exclusion zone surrounding a spill has the potential to adversely affect such operators. Interference of existing energy industry activities due to a hydrocarbon spill is expected to be minimal.
Protected areas	
Marine parks and	A number of marine parks overlap the EMBA (listed in Table 3-8). The EMBA overlaps the:
Commonwealth	Oceanic Shoals Marine Park
nentage aleas	Joseph Bonaparte Gulf
	Garig Gunak Barlu
	The marine parks are many kilometres from the OA, therefore any contact will be with highly weathered hydrocarbons, reducing the potential impacts. Hydrocarbons contacting marine parks may impact the value of the marine parks for a period. These values include:
	natural
	cultural
	socioeconomic.
	Section 3.2.12.1 details the values of the individual marine parks.
	Natural values
	Extensive contact with deeper features such as KEFs associated with the marine parks is not predicted, given the modelling predicts hydrocarbon concentrations are not expected to exceed depths greater than approximately 25 m (Section 7.6.3).
	Marine parks support increased productivity or abundance of marine fauna that use the waters – including plankton, pelagic invertebrates and fish, marine mammals, marine reptiles and seabirds – which may be impacted by hydrocarbons, as previously described in this table.
	Socioeconomic values
	Marine parks may be used by a number of other users, including tourism and recreational fisheries, and may be impacted by hydrocarbons, as previously described in this table.
KEFs	KEFs are described in Section 3.2.12.3.



	Two KEFs overlap the EMBA:
	Carbonate bank and terrace system of the Van Diemen Rise
	Pinnacles of the Bonaparte Basin
	While some features associated with the KEFs are subtidal or submerged and would not be directly contacted by a surface slick, they all may support increased productivity or abundance of marine fauna that use surface waters above the features – including plankton, pelagic invertebrates and fish, marine mammals, marine reptiles, and seabirds – which may be impacted by hydrocarbons, as previously described in this table.
	KEFs are typically geomorphic features. The likelihood of extensive impact is reduced, given the modelling (Section 7.6.3) predicts hydrocarbon concentrations are not expected to exceed depths greater than approximately 25 m.
Ramsar wetlands	There are no Ramsar wetlands in the EMBA
Threatened ecological communities	There are no threatened ecological communities within the EMBA.



### 7.6.8 Stochastic spill dispersion modelling results

The spill modelling results at or above moderate exposure values (as used to define the MEVA) are summarised below for an MDO release from an IMMR vessel (Figure 7-3). More detailed results are provided in Appendix F.

#### 7.6.8.1 Accumulated shoreline hydrocarbon

Modelling results for accumulated shoreline hydrocarbon indicate:

- the highest probability of shoreline hydrocarbon accumulation at the 10 g/m<sup>2</sup> threshold is predicted for the Vernon Islands Conservation Reserve (less than 2%) and Tiwi Islands (less than 1%), which has a maximum volume of hydrocarbon ashore of <1 m<sup>3</sup>.
- the shortest time for shoreline hydrocarbon accumulation at the 10 g/m<sup>2</sup> threshold is predicted at Tiwi Islands after 230 hours (approximately six days) and 261 hours at the Vernon Islands Conservation Reserve after the commencement of the spill.

### 7.6.8.2 Floating hydrocarbon greater than 10 g/m<sup>2</sup>

Modelling results for floating hydrocarbon greater than 10  $g/m^2$  indicate floating hydrocarbon may extend up to 38 km from the release location.

### 7.6.8.3 Entrained hydrocarbon greater than 100 ppb

Modelling results for entrained hydrocarbon greater than 100 ppb indicate:

- entrained hydrocarbon may occur within 0 to 25 m water depth, with a maximum distance from the release location of 74 km
- the shortest time for entrained hydrocarbon exposure at any receptor is predicted for Shepparton Shoal (15 hours)
- the worst-case concentration of entrained hydrocarbons is predicted at Shepparton Shoal as 1,435 ppb.

#### 7.6.8.4 Dissolved hydrocarbon greater than 50 ppb

Modelling results for dissolved hydrocarbon greater than 50 ppb indicate that no sensitive receptors will be contacted.

#### 7.6.9 Environmental performance outcomes and control measures

The EPOs relating to this event are:

- No injury, or mortality to, EPBC Act 1999 listed marine or marine fauna listed as threatened species under the *Territory Parks and Wildlife Conservation Act 1976* (NT) as a result of the Activity (EPO-08)
- Zero unplanned discharge of hydrocarbons or chemicals to the marine environment from the Activity (EPO-18)
- No significant impact to cultural features from the Activity (EPO-19)
- No impacts to underwater cultural heritage from the Activity (EPO-20).

An assessment of the environmental benefits and the potential costs or issues associated with control measures for this Activity are described in Table 7-16 to demonstrate the potential risks are ALARP. Control measures that are adopted have associated EPSs and measurement criteria and are presented in Table 8-2. Rejected control measures have an ALARP evaluation provided to justify their rejection.

Selection of oil spill response strategies and associated performance outcomes, control measures and performance standards, including those required to maintain preparedness and for response, are detailed within the Barossa GEP NT waters OPEP. The GEP NT waters OPEP contains an evaluation of oil spill preparedness arrangements to demonstrate that oil spills will be mitigated to ALARP.

CM reference	Control measure	Environmental benefit	Potential cost/issues	Evaluation
Standard control m	easures			1
BAO-CM-6.1.2	Vessels equipped and crewed in accordance with Australian maritime requirements, including Marine Order 30 (Prevention of Collisions) and Marine Order 21 (Safety and Emergency Arrangements) (administrative control)	Ensures contracted vessels are operated, maintained and crewed in accordance with industry standards and regulatory requirements. Ensures vessels meet Marine Assurance Standards to reduce the likelihood of vessel collision (such as minimum and working lighting for maritime safety).	Cost associated with implementing procedures.	Adopted – benefit of assuring vessels outweighs procedure compliance costs.
BAO-CM-6.4.10	Barossa Facilities and vessels planned maintenance system to confirm equipment integrity is maintained in accordance with manufacturers guidelines (administrative control)	Requires that equipment is maintained and certified, reducing the probability of an unplanned MDO spill. Reduces risk of vessel collision and refuelling incidents because equipment is operating within planned maintenance requirements.	High cost of maintaining vessel equipment and managing the maintenance system.	Adopted – environmental benefits of ensuring vessels are maintained outweigh the costs.
BAO-CM-6.6.1	Notify AHS and AMSA MSI prior to relevant Activity (administrative control)	Ensures other marine users are aware of the presence of the vessels and the relatively slow speed and restricted manoeuvrability. Alerts other marine users to the presence of Activity vessels and 500 m exclusion zone around the installation vessels, thus reducing the likelihood of vessel collision and fishing gear snagging.	Negligible costs.	Adopted – it is a regulatory requirement.
BAO-CM-6.6.4	Activity undertaken in accordance with Santos HSE management and marine vessel vetting processes (Santos' Offshore Marine Assurance Procedure) (administrative control)	Santos marine vetting process ensures vessel lighting, radios and equipment are inspected and maintained so that other marine users are aware of the vessel's physical presence, thus reducing the	Costs associated with personnel time in checking vessels.	Adopted – benefit of assuring vessels outweighs procedure compliance costs.

### Table 7-16: Control measures evaluation for surface release of MDO from a vessel

CM reference	Control measure	Environmental benefit	Potential cost/issues	Evaluation
		potential for interaction and collision.		
BAO-CM-6.6.6	Vessel speed restrictions within 500 m of IMMR vessels and campaign vessels (substitution control)	Reduces the likelihood and consequence of collisions (causing harm) as fauna have longer to detect and avoid the vessel by restricting vessel speeds in the OA to 8 knots or less within 500m of IMMR vessels and campaign vessels. Reduces the potential impacts to culturally significant marine species, including totemic species, such as marine turtles and marine mammals.	Administrative costs to update existing Santos procedure and induction materials, and train personnel.	Adopted
BAO-CM-7.4.2	Bulk liquid transfer procedure (administrative control)	The procedure provides details about the fuel bunkering process to be undertaken. Implementing the procedure reduces the potential for release during bunkering. Requires use of dry-break coupling (bunkering hose) and breakaway coupling, which limit the fuel losses in an emergency.	Personnel costs associated with ensuring procedures are in place and implemented during refuelling.	Adopted – environmental benefits of ensuring the procedure is followed and measures implemented outweigh the costs.
BAO-CM-7.4.3	Vessel spill response plans (SOPEP/SMPEP) (administrative control)	Implements onboard response plans (SOPEP/SMPEP) to deal with unplanned hydrocarbon releases quickly and efficiently to reduce impacts to the marine environment.	Administrative costs of demonstrating vessel contractor compliance. Generally undertaken by vessel contractor so time for Santos personnel to confirm that a SOPEP/SMPEP is in place.	Adopted – it is a regulatory requirement.
BAO-CM-7.7.8.15	Accepted Barossa GEP NT waters OPEP (administrative control)	Implements response plans to deal with an unplanned hydrocarbon release quickly and efficiently to reduce impacts to the marine environment.	Personnel and administrative costs associated with preparing documents, ongoing management (spill response exercises) and implementation of OPEP.	Adopted – it is a regulatory requirement.
Additional control r	neasures			
N/A	Response equipment above	May allow for quicker response to a spill as	Vessel storage restrictions. Large	Not adopted – not feasible due to lack of room on

CM reference	Control measure	Environmental benefit	Potential cost/issues	Evaluation
	and beyond SOPEP/SMPEP requirements (such as booms) on vessels ready to respond to a loss of hydrocarbons (protective control)	resources will be within proximity.	costs associated with a dedicated resource on location.	vessels, the large cost associated with dedicated resources on location, and the proximity of response equipment in Darwin. Costs considered grossly disproportionate to the low risk of a vessel collision or large MDO release.
N/A	No fuel bunkering via hose (elimination control)	Removes spill risk from hose operations.	Cost associated with transfer of MDO via drums or containers and introduction of new risks related to dropped objects and vessel transfers. Not possible to modify vessel to allow additional fuel storage.	Not adopted – eliminating bunkering via hoses introduces new risks related to dropped objects and vessel transfers. The bunkering method is consistent with industry and maritime practices.
N/A	Require all support vessels to be double hulled (engineering control)	Reduces the likelihood of a loss of hydrocarbon inventory minimising potential environmental impact.	Vessels are subject to availability and must meet Santos' standards during activities. The requirement for a double hull on vessels would limit the number of vessels available. Also, there is a high cost associated with refitting vessels with double hulls if required.	Not adopted – large costs associated with vessel selection. Having the Activity schedule determined by vessel availability is considered grossly disproportionate compared to the low risk of a vessel collision or large MDO release.

### 7.6.10 Environmental impact and risk assessment

### 7.6.10.1 Identification of hotspots for consequence assessment

Hotspots that are predicted to be contacted by hydrocarbons within the LEVA and MEVA for an unplanned release of MDO are listed in Table 7-17. The values and sensitives associated with these areas are described in Section 3. These hot spots meet the criteria as described in Section 7.6.6.3.

Note the worst-case values were taken from the modelling scenarios to identify the hot spots and therefore is taken from any season and any hydrocarbon phase at any water depth.

Table 7-17: Identified high e	nvironmental value and	hotspot receptors
-------------------------------	------------------------	-------------------

Receptor		Hotspot		
	Low (LEVA)	Moderate (MEVA)	High (HEVA)	
Afghan Shoal	~	~		No
Beagle Gulf-Darwin Coast (including Darwin Harbour)	~			No
Cobourg Peninsula-Nhulunbuy	~			No
Djukbinj NP	~			No
Flat Top Bank	~			No
Hancox Shoal	~			No
Harris Reef	~			No

Receptor	Exposure values			Hotspot
	Low (LEVA)	Moderate (MEVA)	High (HEVA)	
Joseph Bonaparte Gulf East Coast	~			No
Joseph Bonaparte Gulf Marine Park	~			No
Lowry Shoal	$\checkmark$			No
Marsh Shoal	~			No
Moresby Shoals	~			No
Newby Shoal	~			No
Shepparton Shoal	~	~	$\checkmark$	Yes
Skottowe Shoal	~			No
The Boxers Area	~	~		No
Tiwi Islands	~	~		Yes
Van Diemen Gulf Coast	~			No
Van Diemen Gulf Shoal	~			No
Vernon Islands Conservation Reserve	~			No

### 7.6.10.2 Impact, likelihood and consequence ranking – surface release of MDO

Receptors	Physical environment and habitat
	Protected areas
	Threatened, migratory or local fauna
	Socio-economic
	Cultural features
Consequence	III – Moderate

The consequence assessment for each receptor category is summarised below. Potential impact pathways (physical and chemical) of hydrocarbon exposure for receptors are summarised in Table 7-14, and potential impacts to receptors that may be found within the MEVA are further described in Table 7-15.

#### Physical environment and habitat

It is likely that water quality will be reduced due to hydrocarbon contamination, both at the sea surface and in the upper water column as a result of entrained and dissolved hydrocarbons, at the location of the spill as well as within surrounding marine waters. Given the light nature of MDO, it undergoes rapid spreading and evaporation losses in warm waters and any floating hydrocarbons will be temporary. Water quality changes within the water column are also expected to be temporary, due to the rapid natural degradation and dispersion of MDO in the marine environment.

Afghan Shoal, Shepparton Shoal, and The Boxers Area are within the MEVA. Banks and shoals support a diverse and varied range of benthic communities, reef-building soft corals, hard corals and filter-feeders (Heyward et al., 2017, 2015b). Shoals and banks close to the release have the greatest potential to be contacted by entrained hydrocarbons; however, at relatively low probabilities (for example, up to 24% at Shepparton Shoal).

Shallower shoals (for example, the top of the shoal is within the top 25 m of the water column) within the MEVA are more likely to be contacted by entrained hydrocarbons. Lethal and sub-lethal effects to filter feeders from hydrocarbons include mortality and changes in population recruitment, growth and reproduction which may lead to changes in community composition and structure (Wei et al., 2012). Filter feeders are particularly susceptible as they are likely to directly ingest hydrocarbons while feeding. This may cause mortality or sub-lethal impacts such as alteration in respiration rates, decreases in filter-feeding activity, reduced growth rates, biochemical effects (Keesing & Edgar, 2016). However, as the hydrocarbon concentration decreases and weathers, the communities are expected to recover.

The Tiwi Islands shoreline may accumulate hydrocarbons in low volumes. This location includes areas of benthic coral reefs, seagrass and mangroves. Hydrocarbon coating of prop roots of mangroves can occur from surface hydrocarbons when they are deposited on the aerial roots. Hydrocarbons deposited on the aerial roots can block the pores used by the plants to breathe or interfere with the trees' salt balance resulting in sub-lethal and potentially lethal effects. Mangroves can also be impacted by entrained aromatic hydrocarbons that may adhere to sediment particles. In low-energy environments such as mangroves, deposited sediment-bound hydrocarbons are unlikely to be removed naturally by wave action and may be deposited in layers by successive tides (NOAA, 2014).

Tidal mudflats, like mangroves, are a low-energy environment and are, therefore, susceptible to potential impacts from persistent surface or stranded hydrocarbons. Hydrocarbons in contaminated sediments can persist for years and significantly

impact benthic infauna and their dependent migratory shorebird populations (Duke and Burns, 2003). Saenger (1994) noted that mudflats were the most severely affected habitat 2 years after the Gulf War spill, with no sign of living epibiota. However, the hydrocarbon type in the Gulf was crude oil with a larger fraction of persistent components, compared to MDO. Given the low persistent hydrocarbons in MDO, the persistence of hydrocarbons is expected to be short-term.

Seagrasses in the subtidal and intertidal zones have different degrees of exposure to hydrocarbon spills. Subtidal seagrass is generally considered much less vulnerable to surface hydrocarbon spills than intertidal seagrass, primarily because freshly spilled hydrocarbons float under most circumstances. Dean et al. (1998) found that hydrocarbons mainly affect flowering. Therefore, species that can spread through apical meristem growth (growth at the tips of the root) are not as affected (such as Zostera, Halodule and Halophila species).

MDO tends to entrain within the water column, which can lead to seagrass coming into contact with or absorbing the watersoluble fraction. Contact and absorption have the potential to reduce photosynthesis and tolerance to other stress factors (Runcie et al., 2010; Taylor and Rasheed, 2011). Seagrass in the intertidal zone, such as that of the Tiwi Islands, is particularly vulnerable as it may come into direct contact with surface hydrocarbons and entrained components, which can smother and kill seagrasses if it coats their leaves and stems (Taylor and Rasheed, 2011). This conclusion is supported by Howard et al. (1989), who noted that surface hydrocarbon spills that become stranded on the seagrass and smother it during the rise and fall of the tide could result in reduced growth rates, blackened leaves and mortality. Wilson and Ralph (2011) concluded that long-term impacts to seagrass are unlikely unless hydrocarbon is retained within the seagrass meadow for a sustained duration.

Contact by hydrocarbons may result in a localised decrease in ecological value of the shoreline due to the associated toxic components of hydrocarbons. Secondary impacts may occur to the fauna using the shoreline, as described in the next subsection.

Potential impacts to the physical environment and habitat are expected to be III-Moderate, due to the potential for a significant loss of area and/or function of the local physical environment and habitat. Though the evaporative and dispersive nature of MDO, which largely remains in the top 25 m of the water column, and the low volume of shoreline accumulation (for example, <1 m<sup>3</sup> at Tiwi Islands) does reduce the potential for long term effects.

Water soluble hydrocarbon fractions associated with surface slicks also cause high coral mortality (Shigenaka, 2001) via direct physical contact of hydrocarbon droplets with sensitive coral species (such as the branching coral species). Inter-tidal and shallow water corals may be impacted by surface and entrained hydrocarbons. Impacts may include increased mortality and sub-lethal effects such as changes in feeding, bleaching (loss of zooxanthellae), and increased mucous production, resulting in reduced growth rates and impaired reproduction (Negri and Heyward, 2000). The habitat around the Tiwi Islands is restricted to coastal reef areas and inter-tidal platforms. Given the patchy distribution of inter-tidal and shallow water corals and the non-persistent nature of the hydrocarbon, impacts to corals in the event of an MDO release are expected to be restricted to sub-lethal impacts.

#### Threatened or Migratory fauna

In the event of a surface release of MDO, a reduction in water quality (described above) has the potential to impact marine fauna within the MEVA, as described in Table 7-15. Impacts would be greatest within several kilometres of the release location, where the hydrocarbon is at its thickest on the sea surface and where the toxic aromatic components of the MDO will be at their highest concentration. Given the nature of the release (at surface), hydrocarbons are predicted to remain the top 25 m of the water column; therefore, extensive contact with marine fauna below this level is not anticipated. Upon release to the marine environment, the MDO will also rapidly lose toxicity with time and will spread thinner at the surface as evaporation continues or due to entrainment within the water column.

#### Plankton

Plankton communities may be impacted by a hydrocarbon release, particularly entrained fractions. Toxic effects from exposure to entrained hydrocarbons may cause impacts such as blocked filter feeding organs and impacts resulting from ingesting hydrocarbons. Modelling of the credible release scenario predicts that entrained hydrocarbons above impact thresholds are expected to be highly localised around the release location. Given the high productivity of planktonic communities and the nature and scale of the credible release, these impacts are expected to be temporary and highly localised to the release location.

#### Seabirds and migratory shorebirds

The Wildlife Conservation Plan for Seabirds (CoA, 2020) identified pollution as a threat to seabirds and their habitats. As outlined in the Wildlife Conservation Plan for Seabirds (CoA, 2020), one of the objectives is to enhance contingency plans to prevent and respond to environmental emergencies that impact seabirds and their habitats, which is adopted in the control measure OPS-7.5.2 (refer to Table 7-16).

#### Marine mammals

The MEVA does not overlap any marine mammal BIAs, however a number of marine mammal species may come into contact with hydrocarbons either on the sea surface or within the water column. Potential impacts are likely to be limited to individuals that may be transiting through the area, with potential for coating of baleen (in whales) and ingestion of oiled prey (plankton and fish), as described in Table 7-15. Impacts to overall population viability or ecosystems are not anticipated.

Dugongs are known to occur in coastal waters, including those of the Tiwi Islands such as the seagrass sites on the northwest of Melville Island. Direct impacts to dugongs could occur through foraging or ingesting seagrass coated with hydrocarbon. Dugongs could also be indirectly affected if the released hydrocarbons cause the dieback of seagrass, reducing dugong feeding area. Impacts at a population level are considered highly unlikely as the extent of the MDO release is not anticipated to result in the loss of entire seagrass meadow habitats.

Pelagic and demersal fish communities (including sharks and rays)

Fish mortalities are rarely observed to occur as a result of hydrocarbon releases (ITOPF, 2014). This has generally been attributed to the possibility that pelagic fish can detect and avoid surface waters underneath hydrocarbon releases by swimming into deeper water or away from the affected areas. Fish that have been exposed to dissolved aromatic hydrocarbons are capable of eliminating the toxicants once in clean water, thus individuals exposed to a release are likely to recover (King et al. 1996). Where fish mortalities have been recorded, the releases (resulting from the groundings of the Amoco Cadiz [1978] and Florida [1969] tankers, which were significantly bigger than the worst-case credible release scenario considered in this GEP Coastal Waters OEMP) occurred in sheltered bays, which limited the ability of fish to access clean water and eliminate toxicants. Given the nature and scale of the credible release scenario and the open-ocean environment of the credible release location, impacts to pelagic and demersal fish are expected to be highly localised and temporary.

#### Marine reptiles

The MEVA overlaps various marine turtle BIAs and internesting buffer HC in proximity to the Tiwi Islands. Marine turtle species may come into contact with hydrocarbons either on the sea surface or within the water column, but any potential impacts (as described in Table 7-15) are likely to be limited to individuals that may be transiting through the area or feeding at nearby submerged shoals and banks. Hydrocarbons are not predicted to contact the entire BIAs or areas of habitat critical to the survival of these species.

Approximately 260 km of sandy beaches surround the Tiwi Islands, many of which are documented to host turtle nesting. It is important to acknowledge that turtles have a strong affinity for specific nesting beaches and are unlikely to relocate to an alternative beach if their preferred nesting site is affected by hydrocarbons. Deterministic modelling predicts that the longest length of oiled shoreline at the moderate exposure threshold was 3 km with a low probability (0.33%) of occurring. At the end of this modelling simulation (30 days), less than 1% of the total MDO volume remained ashore. No high (>1,000 g/m<sup>2</sup>) shoreline exposure was predicted during the model simulation. Therefore, even considering the longest length of oiled shoreline predicted by the model, it will not have a significant impact on the nesting turtle population, and the duration of the impact will be limited.

Turtle nests are also typically located above the high water mark, typically the highest point along the shoreline that stranded oil will reach. Direct contact between turtle eggs and the stranded hydrocarbons is very unlikely. Nesting females and hatchlings emerging from nests may be exposed to stranded hydrocarbons when moving on nesting beaches, potentially resulting in contamination. Exposure may result in light oiling of nesting females and hatchlings, subsequently leading to sub-lethal effects such as skin irritation; no mortality is expected. Given the non-persistent nature of MDO and low levels of hydrocarbons potentially stranding on shorelines, the potential for impacts to nesting turtles, egg clutches and hatchlings on beaches is considered low.

Given the non-persistent nature of the MDO, along with the expected rapid evaporation and dispersion, the timeframe during which marine turtles may be exposed to hydrocarbons above impact thresholds is low. The spatial extent of the MEVA, along with the wide distribution of turtle species in the region, indicates impact to overall population viability or ecosystems is not anticipated. Potential impacts would be greatest during the internesting season for flatback and olive ridley turtles; between June and September for flatback turtles and April to August for olive ridley turtles.

#### Summary

The Tiwi Islands may accumulate a very small amount of hydrocarbons, which could impact marine fauna that use these areas such as shorebirds and turtles. Impacts to turtles could occur from hydrocarbons that accumulate on turtle nesting beaches, with the greatest impact being during nesting seasons. Turtle nests are typically made above the high water mark, which is typically the highest point along the shoreline that hydrocarbon will reach. As such, direct contact between turtle eggs and the hydrocarbons is very unlikely. Given the low volumes (<1 m<sup>3</sup>) and non-persistent nature of MDO on shorelines, the impact to nesting beaches (including nesting turtles, egg clutches and hatchlings) is anticipated to relate to a very temporary local disruption of individual turtles using the nesting beach, if the spill was to occur during nesting season.

The potential sensitive receptors in the surrounding areas of the hydrocarbon release include fish, marine mammals, marine reptiles, and seabirds. Potential impacts (as described in Table 7-15) to Threatened or Migratory fauna are expected to be III – Moderate and relate to a temporary disruption to local populations. Impact to overall population viability or ecosystems is not anticipated.

#### **Protected areas**

The MEVA does not overlap with any marine protected areas.

#### Socio-economic (fisheries, tourism, recreation, and other third-party operators)

There is potential for temporary disruption to fishing activities (traditional, recreational and commercial) and tourism and recreational activities if the surface, shoreline or entrained hydrocarbons moves through frequented areas. However, the high rate of evaporation means that little MDO will become entrained and few aromatic hydrocarbons to become dissolved. Given the volume of MDO that could potentially be released, it is unlikely that impacts could be detected to fisheries on a stock level although it is more likely that natural variation in fish abundance would be on a greater scale than any impacts attributable to a hydrocarbon spill. A hydrocarbon release may also temporarily displace activities such as fishing, tourism and recreation from within sections of the MEVA. This displacement would be localised and short-term (days). A hydrocarbon release may result in tainting of fished species. This could potentially result in commercial fishers being unable to sell their catch, which may result in a loss of income or other fishers unable to eat their catch. Spilt hydrocarbons may also contaminate fishing gear, which may require cleaning. Potential impacts to fishing activity are expected to relate to a short-term, but potentially significant, loss of value to the local industry due to local disruptions and displacement of fishing ground.

Other energy operations in the region may also be disrupted in the event of a hydrocarbon release (such as Santos' Bayu-Undan operations) and defence and military exercises and commercial shipping may be excluded or displaced temporarily.

Potential impacts (as described in Table 7-15) to socio-economic receptors are expected to be III-Moderate and relate to a temporary, local disruption or displacement in activities.

#### **Cultural features**

An important outcome of Dr Corrigan's research is that no sacred sites or dreamings are shown to be directly impacted by the DPD project footprint (where this Activity OA is located), although this is not to say that some persons do not have fears that this could be the case in the event of an unplanned event (Corrigan, 2024).

The EMBA overlaps cultural features (Section 3.2.14.7). Impacts to cultural features, including a disruption/displacement of cultural activities caused by the physical presence of the hydrocarbon, decline in traditional food sources and / or mortality of fauna with cultural significance and contact to sacred sites, may result in the event of a significant spill of hydrocarbons. In the 2022 Statement of Reasons Requests, the Tiwi clan members raised their concern regarding traditional hunting of marine species and totem species. The First Nations people maintain a continuing spiritual connection with sea country, including caring for sea country and access to cultural food sources. Potential impacts to cultural features from a hydrocarbon spill may also include a decline in traditional food sources or mortality of fauna with cultural significance. The potential impacts to culturally significant marine fauna species (such as dreaming and totem species including marine mammals, marine reptiles, fish and birds) are assessed separately above. The Tiwi Islands have been listed as a hotspot (Table 7-17) on the basis of First Nations cultural heritage significance.

Potential impacts (as described in Table 7-15) to cultural features are expected to be III-Moderate and relate to a temporary, local disruption or displacement in activities.

Likelihood B – Unlikely

The likelihood of a hydrocarbon release occurring due to a vessel collision is unlikely, given the set of mitigation and management controls in place. External impacts to vessels have not occurred within Santos and controls are in place that limit such events.

The OA is close to the Commonwealth/Territory waters boundary, is an area of high shipping traffic due to its proximity to Darwin and is therefore considered a greater risk of collision The likelihood of a collision event occurring in the OA is unlikely.

The likelihood of a vessel collision releasing hydrocarbons to the environment resulting in an III-Moderate consequence is considered to be B – Unlikely.

**Residual risk** The residual risk is considered Low.

### 7.6.11 Demonstration of as low as reasonably practicable

The use of vessels is integral to the Activity. Therefore, vessels and associated risks of unplanned hydrocarbon releases cannot be completely eliminated.

All reasonably practicable control measures have been reviewed and those adopted are considered appropriate to manage the residual risk to a Low level. The proposed management controls are in accordance with the Santos risk management criteria and are considered appropriate to manage the risk to ALARP.

In terms of spill response activities, Santos will implement hydrocarbon spill response as specified within the Barossa GEP NT waters OPEP. A detailed ALARP assessment on the adequacy of arrangements available to support spill response strategies and control measures is presented in the Barossa GEP NT waters OPEP.

### 7.6.12 Acceptability evaluation

Is further information required to validate the consequence assessment?No – potential impacts and risks are well understood through the information available.Are risks and impacts consistent with the principles of ESD?Yes – Activity evaluated in accordance with Santos' Offshore Division Environmental Hazard Identification and Assessment Guideline which considers principles of ESD:• The impacts from the spill scenario are inherently inconsistent with principles of ESD, given the nature and scale of impacts. Control measures are applied to ensure the impacts and risks from activities are managed to ALARP and an acceptable level.Have the acceptable levels of impact and risks been informed by relevant species recoveryYes – Control measures implemented will reduce the risk of an unplanned release of MDO to species identified in the following relevant	Is the risk ranked between Very Low to Medium?	Yes – residual risk is ranked as Low.
Are risks and impacts consistent with the principles of ESD?Yes – Activity evaluated in accordance with Santos' Offshore Division Environmental Hazard Identification and Assessment Guideline which considers principles of ESD:• The impacts from the spill scenario are inherently inconsistent with principles of ESD, given the nature and scale of impacts. Control measures are applied to ensure the impacts and risks from activities are managed to ALARP and an acceptable level.Have the acceptable levels of impact and risks been informed by relevant species recoveryYes – Control measures implemented will reduce the risk of an unplanned release of MDO to species identified in the following relevant	Is further information required to validate the consequence assessment?	No – potential impacts and risks are well understood through the information available.
<ul> <li>The impacts from the spill scenario are inherently inconsistent with principles of ESD, given the nature and scale of impacts. Control measures are applied to ensure the impacts and risks from activities are managed to ALARP and an acceptable level.</li> <li>Have the acceptable levels of impact and risks been informed by relevant species recovery</li> <li>Yes – Control measures implemented will reduce the risk of an unplanned release of MDO to species identified in the following relevant</li> </ul>	Are risks and impacts consistent with the principles of ESD?	Yes – Activity evaluated in accordance with Santos' Offshore Division Environmental Hazard Identification and Assessment Guideline which considers principles of ESD:
Have the acceptable levels of impact and risks been informed by relevant species recovery Yes – Control measures implemented will reduce the risk of an unplanned release of MDO to species identified in the following relevant		<ul> <li>The impacts from the spill scenario are inherently inconsistent with principles of ESD, given the nature and scale of impacts. Control measures are applied to ensure the impacts and risks from activities are managed to ALARP and an acceptable level.</li> </ul>
plans, threat abatement plans, conservation advice, wildlife conservation plans and Australian and other management plans/guidelines, as also set out in Table 3-13.	Have the acceptable levels of impact and risks been informed by relevant species recovery plans, threat abatement plans, conservation advice, wildlife conservation plans and Australian	Yes – Control measures implemented will reduce the risk of an unplanned release of MDO to species identified in the following relevant species recovery plans, conservation advice, wildlife conservation plans and other management plans/guidelines, as also set out in Table 3-13.
marine park zoning objectives)? Conservation advice	marine park zoning objectives)?	Conservation advice
Approved Conservation Advice for <i>Pristis clavata</i> (Dwarf Sawfish) (DEWHA, 2009)		<ul> <li>Approved Conservation Advice for <i>Pristis clavata</i> (Dwarf Sawfish) (DEWHA, 2009)</li> </ul>
Approved Conservation Advice for Green Sawfish (DEWHA, 2008c)		Approved Conservation Advice for Green Sawfish (DEWHA, 2008c)
Approved Conservation Advice for <i>Pristis pristis</i> (largetooth sawfish) (DoE, 2014b)		<ul> <li>Approved Conservation Advice for <i>Pristis pristis</i> (largetooth sawfish) (DoE, 2014b)</li> </ul>
Approved Conservation Advice for <i>Glyphis garricki</i> (northern river shark) (TSSC, 2014a)		<ul> <li>Approved Conservation Advice for <i>Glyphis garricki</i> (northern river shark) (TSSC, 2014a)</li> </ul>

•	Approved Conservation Advice for <i>Glyphis glyphis</i> (speartooth shark) (DoE, 2014a)
•	Approved Conservation Advice for <i>Rhincodon typus</i> (whale shark) (TSSC, 2015g)
•	Approved Conservation Advice for <i>Balaenoptera physalus</i> (fin whale) (TSSC, 2015c)
•	Approved Conservation Advice for <i>Balaenoptera borealis</i> (sei whale) (TSSC, 2015b)
•	Approved Conservation Advice for <i>Limnodromus</i> semipalmatus (Asian dowitcher) (DCCEEW, 2024f)
•	Approved Conservation Advice for <i>Limosa limosa</i> (black-tailed godwit) (DCCEEW, 2024e)
•	Approved Conservation Advice for <i>Calidris tenuirostris</i> (great knot) (DCCEEW, 2024d)
•	Approved Conservation Advice for <i>Charadrius leschenaultii</i> (greater sand plover) (DCCEEW, 2023e)
•	Approved Conservation Advice for <i>Pluvialis squatarola</i> (grey plover) (DCCEEW, 2024d)
•	Approved Conservation Advice for <i>Limosa lapponica baueri</i> (Alaskan bar-tailed godwit) (DCCEEW, 2024k)
•	Approved Conservation Advice for <i>Calidris canutus</i> (red knot) (DCCEEW, 2024c)
•	Approved Conservation Advice for <i>Arenaria interpres</i> (ruddy turnstone) (DCCEEW, 2024I)
•	Approved Conservation Advice for <i>Calidris acuminata</i> (sharp-tailed sandpiper) (DCCEEW, 2024b)
•	Approved Conservation Advice for <i>Xenus cinereus</i> (terek sandpiper) (DCCEEW, 2024i)
•	Approved Conservation Advice for <i>Rostratula australis</i> (DSEWPaC, 2013)
•	Conservation Advice for <i>Charadrius mongolus</i> (lesser sand plover) (DCCEW, 2024a)
•	Approved Conservation Advice for <i>Rostratula australis</i> (Australian painted snipe) (TSSC, 2013)
•	Approved Conservation Advice for <i>Dermochelys coriacea</i> (Leatherback Turtle) (DEWHA, 2008b)
•	Conservation Advice for <i>Tringa nebularia</i> (common greenshank) (DCCEEW, 2024h)
•	Approved Conservation Advice for <i>Calidris ferruginea</i> (Curlew Sandpiper) (TSSC, 2015e)
•	Approved Conservation Advice for <i>Numenius madagascariensis</i> (Eastern Curlew) (TSSC, 2015f)
Re	covery plans
•	Sawfish and River Sharks Multispecies Recovery Plan (CoA, 2015b)
•	Recovery Plan for the White Shark ( <i>Carcharodon carcharias</i> ) (Department of Sustainability, Environment, Water, Population and Communities (DSEWPaC, 2013)
•	Recovery Plan for the Grey Nurse Shark ( <i>Carcharias taurus</i> ) (CoA, 2014b)
•	Conservation Management Plan for the Blue Whale - A Recovery Plan under the <i>Environment Protection and Biodiversity Conservation</i> <i>Act 1999</i> 2015–2025 (CoA, 2015a)
•	Recovery Plan for Marine Turtles in Australia 2017–2027 (CoA, 2017b)
•	National Recovery Plan for the Australian Painted Snipe ( <i>Rostratula australis</i> ) (DCCEEW, 2022a)
•	Wildlife Conservation Plan for Seabirds (CoA, 2020)
•	Wildlife Conservation Plan for Migratory Shorebirds (CoA, 2015c)
Ot	her management plans/guidelines
•	Marine bioregional plans for the NMR (CoA, 2012).

	For the identified plans, the objectives of those plans are achieved through the adoption of the performance outcomes and control measures outlined in Section 7.6.9. Santos considers that the level of risk of an unplanned release of MDO is not inconsistent with these plans.
Are performance outcomes, control measures and associated performance standards consistent with legal and regulatory requirements?	Yes – management consistent with Safety Case, Marine Safety (Domestic Commercial Vessel) National Law Act 2012, Navigation Act 2012, Marine Order 30: Prevention of Collisions, Marine Order 21: Safety of Navigation and Emergency Procedures, Protection of the Sea (Prevention of Pollution from Ships) Act 1983, MARPOL Annex I (Prevention of Pollution by Oil), Marine Order 91: Marine Pollution Prevention – Oil and National Plan for Maritime Environmental Emergencies (AMSA, 2020).
	Through acceptance of this GEP Coastal Waters OEMP, legislative and regulatory requirements will be met as per Appendix B.
	The GEP Coastal Waters OEMP is also compliant with the conditions of EPBC approval EPBC 2022/09372.
Are performance outcomes, control measures and associated performance standards consistent with Santos' Environment, Health and Safety Policy?	Yes – aligns with Santos' Environment, Health and Safety Policy (Appendix A).
Are performance outcomes, control measures and associated performance standards consistent with industry standards?	Yes – the most recent and comparable Operations EPs accepted by NOPSEMA have been reviewed for consistency with the performance outcomes, control measures and associated performance standards proposed in this GEP Coastal Waters OEMP.
Have performance outcomes, control measures and associated performance standards taken into consideration Relevant Person feedback	Yes – Relevant Persons feedback received during consultation for this activity has been considered when evaluating performance outcomes, control measures and associated performance standards. Where relevant, control measures implemented based on Relevant Persons feedback for other Barossa EPs have been adopted in this GEP Coastal Waters OEMP.
	Santos will notify all Relevant Persons who have requested notification in the event of a spill. As a result, Table 8-6 and the OPEP have also been updated to reflect additional requests for notifications in the event of a spill.
Are performance standards such that the impact or risk is considered to be ALARP?	Yes – ALARP assessment conducted, with no additional control measures adopted.



### 7.7 Contingency spill response operations

### 7.7.1 Spill response strategies

The spill response strategies that may be adopted in the event of a hydrocarbon spill from this Activity have been identified in the Barossa GEP NT waters OPEP. These are generally strategies that have been implemented in the past or are considered good industry practice. The Barossa GEP NT waters OPEP contains an evaluation of hydrocarbon spill preparedness arrangements to demonstrate hydrocarbon spills will be mitigated to ALARP. An environmental assessment of these spill response strategies has been conducted as presented below.

An overview of the hydrocarbon spill scenario considered for this Activity and relevant to spill response operations is provided in Section 7.6.

### 7.7.2 Description of event

In the event of a hydrocarbon spill, response strategies will be implemented to reduce environmental impacts to ALARP. Strategies will be selected through a net environmental benefit analysis (NEBA). Spill response will be under the direction of the relevant control agency, as defined in the Barossa GEP NT waters OPEP, which may be Santos, another agency or both. In all instances, Santos will undertake a 'first-strike' spill response and will act as the Control Agency until the designated Control Agency assumes control. The response strategies considered to be appropriate for the worst-case hydrocarbon spill scenarios identified for the Activity are provided in the Barossa GEP NT waters OPEP and comprise: • monitor and evaluate
containment and recovery
mechanical dispersion
shoreline protection and deflection
shoreline clean-up
oiled wildlife response
operational and scientific monitoring
waste management.
While response strategies are intended to reduce the environmental consequences of a hydrocarbon spill, poorly planned and coordinated response activities can result in a lack of or inadequate information being available upon which poor decisions can be made, exacerbating or causing further environmental harm. An inadequate level of training and guidance when implementing spill response strategies can also result in environmental harm over and above that already caused by the spill.
Spill response could occur anywhere within the EMBA for the worst-case spill scenarios.
The spill response effort as a whole will exceed the duration of the worst-case spill, due to persistence of the hydrocarbon in the environment and the requirement to remove hydrocarbons and monitor impacts and recovery to sensitive receptors. The Barossa GEP NT waters OPEP provides further detail about the likely duration of specific response strategies.

### 7.7.3 Nature and scale of environmental impacts

**Potential receptors**: Physical environment and habitat, protected areas, threatened, migratory, or local fauna, socio-economic and cultural features.

Light emissions		
Spill response activities Vessels may operate n	s will involve the use of vessels, which are required, at a minimum, to display navigational lighting. ear shoreline areas during spill response activities.	
Spill response activities require lighting.	s will also involve onshore operations, including the use of vehicles and temporary camps, which may	
Potential receptors	Protected areas	
	Threatened, migratory or local fauna	
	Cultural Features	
Lighting may cause bel consequence during ke and Migratory fauna (S further detail about the	navioural changes to fish, mammals, birds and marine turtles that can have a heightened by lifecycle activities, such as turtle nesting and hatching. Turtles and birds, which include Threatened ection 3.2.13), have been identified as key fauna susceptible to lighting impacts. Section 6.2 provides nature and scale of light emission impacts.	
Spill response activities that require lighting may occur anywhere within the MEVA (refer to Section 7.7.1), including in protected areas and close to shoals. This could result in indirect impacts on the values of the protected areas.		



During nesting and hatching season (primarily over summer months), lighting may cause behavioural impacts to turtles, including aborted nesting attempts and disorientation of newly hatched turtles, which may increase the hatchling mortality rate.

Spill response activities may also occur on shorelines used by nesting and feeding birds, including seabirds and shorebirds. Lighting can cause disorientation in flying birds, disrupt nesting and breeding behaviours, and impact on the ability of birds to forage. Disturbance to feeding migratory shorebirds may reduce their ability to replenish energy reserves and alter the timing and success of migratory flights.

Lighting impacts to fauna are not considered to have the potential to impact supported industries such as tourism.

Lighting from response activities may impact marine fauna of cultural significance.

#### Noise emissions

Spill response activities will involve the use of aircraft and vessels, which will generate noise both offshore and in nearshore locations within the EMBA.

Spill response activities will also involve the use of equipment on coastal areas during clean-up of shorelines, such as pumps and vehicles, for accessing shoreline areas; and for supporting temporary camps, such as diesel generators.

Potential receptors	Threatened, migratory or local fauna
	Protected areas
	Socio-economic receptors

Cultural Features

Underwater noise from the use of vessels may impact marine fauna, such as fish (including commercial species), marine reptiles and marine mammals, in the worst instance causing physical injury to hearing organs but more likely causing short-term behavioural changes; for example, temporary avoidance of the area, which may impact key lifecycle processes such as spawning, breeding and calving. Underwater noise can also mask communication or echolocation used by cetaceans. Section 6.1 provides details about potential noise emission impacts.

Vessels may also need to enter marine parks and other areas used for tourism, commercial and recreational fishing, and traditional purposes.

Noise and vibration from terrestrial activities on shorelines has the potential to cause behavioural disturbance to coastal fauna, including protected seabirds and turtles. Shoreline activities involving the use of noise-generating equipment may occur in important nesting areas for turtles and roosting and feeding areas for shorebirds.

As a consequence of impacts to fauna – including shorebirds, marine mammals, fish – noise has the potential to impact supported industries such as tourism and commercial fishing and recreational values of marine parks.

Noise from response activities may impact marine fauna of cultural significance.

#### **Atmospheric emissions**

The use of fuels to power vessel engines, generators and mobile equipment used during spill response activities will result in emissions of GHG, such as CO<sub>2</sub>, CH<sub>4</sub> and N<sub>2</sub>O, along with non-GHGs such as SO<sub>X</sub> and NO<sub>X</sub>. Emissions will result in a localised decrease in air quality.

Potential receptors	Physical environment and habitat
	Threatened, migratory or local fauna
	Socio-economic

Atmospheric emissions from spill response equipment will be localised, and the use of mobile equipment, vessels and vehicles is not considered to create emissions on a scale where noticeable impacts would be predicted. Emissions may occur in protected areas and areas where tourism is important; however, the scale of the impact relative to potential hydrocarbon spill impacts is not considered great. Section 6.3 provides further details about the nature and scale of air emission impacts.

#### **Operational discharges and waste**

Operational discharges include those routine discharges from vessels used during spill response, which may include:

- deck drainage
- putrescible waste and sewage
- · cooling water from operation of engines
- bilge water
- ballast water
- brine discharge.

In addition, there are specific spill response discharges and waste creation that may occur, including:

- cleaning of oily equipment, vessels and vehicles
- decanting of water back into the marine environment from containment and recovery operations
- flushing water for the cleaning of shoreline habitats
- sewage and putrescible and municipal waste at offshore staging sites
- creation, storage, transport and disposal of oily waste and contaminated organics.

Potential receptors	Physical environment and habitat
	Protected areas
	Threatened, migratory or local fauna
	Socio-economic
	Cultural Features
Operational discharges	from vessels may expete a leasting d and temperature duction in marine water quality. Effects include

Operational discharges from vessels may create a localised and temporary reduction in marine water quality. Effects include nutrient enrichment, toxicity, turbidity, and temperature and salinity increases, as detailed in Section 6.4. Discharge could potentially occur adjacent to marine habitats, such as corals, seagrass and macroalgae, and in protected areas, which support a more diverse faunal community; however, discharges are still expected to be localised and temporary.

Cleaning of hydrocarbon-contaminated equipment, vehicles and vessels has the potential to spread hydrocarbon from contaminated areas to areas not impacted by a spill, potentially spreading the impact area and moving hydrocarbon into a more sensitive environment.

The decanting of oily water back into the marine environment during containment and recovery activities has the potential to impact marine organisms from the toxic effects from hydrocarbons, however, given the marine environment would already be contaminated with hydrocarbons there is limited potential for an increase in impact, unless the discharge spreads the contamination to a previously uncontaminated area.

Flushing of hydrocarbon from shoreline habitats is a clean-up technique designed to remove hydrocarbon from the receptor that has been oiled and remobilise it back into the marine environment. It results in further dispersion of the hydrocarbon. The process of flushing has the potential to physically damage shoreline receptors such as mangroves and rocky shoreline communities, increase levels of erosion, and create an additional and potentially higher level of impact than if the habitat was left to bioremediate.

Sewage and putrescible and municipal waste will be generated from offshore activities at temporary staging and mooring areas, and onshore activities at temporary camps, which may include toilet and washing facilities. These wastes have the potential to impact water quality, attract fauna, impact habitats, flora and fauna, and reduce the aesthetic value of the environment, which may be within protected areas. Disturbance may also impact cultural values of an area. The creation, storage, transport and disposal of oily waste and contaminated organics has the potential to spread impacts of hydrocarbon to areas, habitats and fauna not previously contaminated. Sewage and putrescible and municipal waste generated onshore will be stored and disposed of at approved locations.

Operational discharges from response operations may impact marine fauna of cultural significance.

#### Seabed and habitat disturbance, marine fauna interaction

The movement and operation of vessels, vehicles, personnel and equipment, the undertaking of clean-up activities, and the setup of temporary camp areas during spill response activities have the potential to disturb the physical environment and marine and coastal habitats and fauna, which may occur within protected areas. Disturbance may also impact socio-economic values of an area.

Vessel movement and transportation could potentially introduce to nearshore areas invasive marine species attached as biofouling, while vehicle and equipment movement could spread non-indigenous flora and fauna. Spill response operations can impact on wildlife via vessel strikes and behavioural changes due to physical presence of personnel and equipment. Oiled wildlife response activities may also involve deliberate disturbance (hazing), capture, handling, cleaning, rehabilitation, transportation, and release of wildlife, which could lead to additional impacts to wildlife.

**Potential receptors** 

- Physical environment and habitat
- Protected areas
- Threatened, migratory and local fauna
- Socio-economic
- Cultural Features

The use of vessels may disturb benthic habitats in coastal waters, including corals, seagrass, mangroves and macroalgae. Impacts to habitats from vessels include damage through the deployment of anchors, nearshore booms, mooring lines and from grounding.

Vessel use in shallow coastal waters also increases the chance of contact with or physical disturbance of marine megafauna such as turtles and dugongs. Booms create a physical barrier on the surface waters that has the potential to injure or entangle passing marine fauna that are either surface-breathing or -feeding.

Vehicles, equipment, personnel and cleaning activities during shoreline response activities have the potential to damage coastal habitats, such as dune vegetation, mangroves and habitats important to threatened and migratory fauna, including nests of turtles and birds and bird roosting and feeding areas. Shoreline clean-up may involve the physical removal of substrates that could cause impact to habitats and coastal hydrodynamics and alter erosion or accretion rates.

The presence of camp areas, although relatively short-term, may disrupt normal behaviour of coastal species, such as shorebirds and turtles, and could potentially interfere with nesting and feeding behaviours.

Oiled wildlife response may include the hazing, capture, handling, cleaning, rehabilitation, transportation, cleaning and release of wildlife susceptible to oiling, such as birds and marine turtles. While oiled wildlife response is aimed at having a net benefit, poor responses can potentially create additional stress and exacerbate impacts from oiling, interfere with lifecycle processes, hamper recovery and, in the worst instance, increase levels of mortality.

Impacts and risks from invasive marine species are described in Section 7.2 and are not described further in this section. Impacts from invasive terrestrial species are similar in that the invasive species, such as weeds, can outcompete local

species and interfere with ecosystem processes. Non-native species may be transported attached to equipment, vehicles and clothing. Such an introduction would be especially detrimental to wilderness areas or protected terrestrial reserves, which may have a relatively undisturbed flora and fauna community.

The disturbance to marine and coastal natural habitat, as well as the potential for disruption to culturally sensitive areas, may occur in specially protected areas such as marine parks, and may have flow-on impacts to socio-economic values and industry, such as tourism and fisheries.

#### Interactions with other marine users

Spill response activities may involve the use of vessels and equipment in areas used by the general public or industry in Australia and potentially Indonesia. The mobilisation of spill response personnel into Forward Operating Bases may also place increased demands on local accommodation and other businesses.

Potential receptors Socio-economic

The use of vessels in the offshore environment and the undertaking of spill response activities may exclude the general public and industry use of the affected environment. As well as impacting recreational activities (such as recreational fishing) of the general public, this may impact on revenue with respect to industries such as commercial fishing and interrupt military exercises. The mobilisation of personnel to regional communities has the potential to affect the local community through demands on local accommodation and business, reducing the availability of services to members of the public.

### 7.7.4 Environmental performance outcomes and control measures

An assessment of the environmental benefits and the potential costs or issues associated with control measures relevant to response vessels for this Activity are described in Table 7-18 to demonstrate that the potential impacts from this aspect are ALARP. Control measures that are adopted have associated EPSs and measurement criteria and are presented in Table 8-2.

Control measures that are more specific to spill response are presented in the Barossa GEP NT waters OPEP. Control measures that are adopted have associated EPSs and measurement criteria, which are presented in the relevant strategy sections of the Barossa GEP NT waters OPEP.

CM reference	Control measure	Environmental benefit	Potential cost/issues	Evaluation
Standard control meas	sures			
BAO-CM-6.1.1	Apply Santos' Protected Marine Fauna Interaction Procedure to vessel and helicopter activities when in vicinity of cetaceans and turtles (isolation control)	Refer to Table 7-4	Refer to Table 7-4	Adopted – refer to Table 7-4
BAO-CM-6.2.1	Lighting limited to that required for safe work conditions and navigational purposes. (isolation control)	Refer to Table 6-13	Refer to Table 6-13	Adopted – refer to Table 6-13
BAO-CM-6.3.17	Pursuant to Marine Order 97 (vessels), relevant vessels will have a current International Air Pollution Prevention (IAPP) Certificate or equivalent and Ship Energy Efficiency Management Plan (SEEMP) (administrative control)	Refer to Table 6-18	Refer to Table 6-18	Adopted – refer to Table 6-18
BAO-CM-6.4.5	HSE inductions will include applicable	Ensures that crew are aware of the stringent Coastal Waters OEMP, Santos and	Administrative costs to update existing Santos procedure and induction materials,	Adopted

#### Table 7-18: Control measures evaluation for contingency spill response operations



CM reference	Control measure	Environmental benefit	Potential cost/issues	Evaluation
	environmental requirements (administrative control)	legislative requirements.	and train personnel.	
BAO-CM-6.7.2	Routine discharges of treated sewage and grey water, in accordance with Marine Order 96 (Marine Pollution Prevention – Sewage), (administrative control)	Refer to Table 6-22	Refer to Table 6-22	Adopted – refer to Table 6-22
BAO-CM-6.7.5	Apply the Santos chemical selection process for chemicals planned to be discharged (Section 2.7) (administrative control)	Refer to Table 6-22	Refer to Table 6-22	Adopted – refer to Table 6-22
BAO-CM-6.7.6	Routine discharges of treated bilge and deck water from vessels will comply with Marine Order 91 and <i>Marine</i> <i>Pollution Act 1999</i> (NT), as applicable (administrative control)	Refer to Table 6-22	Refer to Table 6-22	Adopted – refer to Table 6-22
BAO-CM-7.7.8.15	Accepted Barossa GEP NT waters OPEP (administrative control)	Refer to Table 7-16 and the Barossa GEP NT waters OPEP	Refer to Table 7-16 and the Barossa GEP NT waters OPEP	Adopted – refer to Table 7-16 and the Barossa GEP NT waters OPEP

### 7.7.5 Environmental impact assessment

Key receptors	Consequence level	
Spill response operations – light emissions		
Threatened, migratory or local fauna	The receptors considered most sensitive to lighting from vessel operations are seabirds, migratory shorebirds and marine turtles. After restricting night-time operations of spill response vessels, which will demobilise to mooring areas offshore with safety lighting only (as specified by controls in the Barossa GEP NT waters OPEP), impacts from vessels are considered to be I – Negligible.	
Physical environment and habitat		
Threatened ecological communities		
Protected areas		
Socio-economic		
Cultural Features		
Overall worst-case consequence	I – Negligible	
Spill response operations – noise emissions		
Threatened, migratory or local fauna	The receptors considered most sensitive to vessel noise are cetaceans. However, after adopting control measures to limit close interaction with protected fauna (as in, Protected adoption of the protected adoption of the protected adoption of the protected fauna (as in, Protected adoption of the pr	
Physical environment and habitat	Marine Fauna Interaction and Sighting Procedure), a temporary behavioural disturbance is expected only with a consequence of I – Negligible.	
Threatened ecological communities		
Protected areas		
Socio-economic		

Key receptors	Consequence level			
Cultural Features				
Overall worst-case consequence	I – Negligible			
Spill response operations -	Spill response operations – atmospheric emissions			
Threatened, migratory or local fauna	Atmospheric emissions from spill response equipment will be localised, and impacts to even the most sensitive fauna, such as birds, are expected to be I – Negligible.			
Physical environment and habitat				
Threatened ecological communities				
Protected areas				
Socio-economic				
Overall worst-case consequence	I – Negligible			
Spill response operations -	- operational discharges and waste			
Threatened, migratory or local fauna	Operational discharges from vessels may create a localised and temporary reduction in marine water quality, which has the potential to impact shallow marine habitats in particular. However,			
Physical environment and habitat	to shorelines, discharges will have a negligible impact to habitats, fauna or protected area values.			
Threatened ecological communities	Decanting from containment and recovery operations would only occur if approval was provided by the relevant Jurisdictional Authority (as specified by controls in the Barossa GEP			
Protected areas	NI waters OPEP), otherwise all collected oil and water will remain in the collection tanks, and all will be treated as collected waste.			
Socio-economic	Washing of vessels and equipment will occur only in defined offshore hot zones, preventing			
Cultural Features	Impacts to shallow habitats. Sewage, putrescible waste and municipal waste generated onshore will be stored and			
	disposed of at approved locations.			
	The storage, transport and disposal of hydrocarbon-contaminated waste arising from spill response operation actions will be managed by Santos' appointed waste management contractor, and dedicated waste containment areas will prevent the spreading or leaching of hydrocarbon contamination.			
	Operational discharges from spill response operations are expected to be II – Minor.			
Overall worst-case consequence	II – Minor			
Spill response operations -	- seabed and benthic habitat disturbance, marine fauna interaction			
Threatened, migratory or local fauna	The use of vessels has the potential to disturb benthic habitats, including sensitive shoal habitats such as corals and macroalgae. A review of shallow water habitats and of bathymetry			
Physical environment and habitat	and the establishment of demarcated areas for access and anchoring will reduce the level of impact to I – Negligible.			
Threatened ecological communities	the impact to the protected areas from physical disturbance is therefore also considered II – Minor.			
Protected areas	In the event of shoreline clean-up operations there is the potential for ground disturbance from			
Socio-economic	The main direct disturbance to fauna would be the hazing, capture, handling, transportation,			
Cultural Features	cleaning and release of wildlife susceptible to oiling impacts, such as birds and marine turtles. This would only be done if this intervention were to deliver a net benefit to the species, but it may result in a II – Minor consequence after complying with the Santos Oiled Wildlife Response Framework and Northern Territory Oiled Wildlife Response Plan.			
Overall worst-case consequence	II – Minor			



Key receptors	Consequence level		
Spill response operations -	Spill response operations – disruption to other users of marine and coastal areas and townships		
Socio-economic	The use of vessels in the offshore environment and spill response activities may exclude general public and commercial industries (such as fishing). Note this is distinct from the socio- economic impact of a spill itself, as described in Section 7.6. After applying control measures, it is considered the additional impact of spill response activities on affected industries would be II – Minor.		
Overall worst-case consequence	II – Minor		

### 7.7.6 Demonstration of as low as reasonably practicable

A NEBA is the primary tool used during spill response to evaluate response strategies and has the goal of selecting strategies that result in the least net impact to key environmental sensitivities. The NEBA process will identify and compare net environmental benefits of alternative spill response options, and will consider stakeholder input where relevant to inform the evaluation of impacts to socioeconomic sensitivities. The NEBA will effectively determine whether an environmental benefit will be achieved through implementing a response strategy or by undertaking no response. The NEBA will be undertaken by the relevant Controlling Agency for the Activity. For those activities under the control of Santos, the Incident Management Team (IMT) Environment Unit Leader will be responsible for reviewing the priority receptors and selected response strategies identified in this GEP Coastal Waters OEMP and coordinating the NEBA for each operational period. This will demonstrate that, at the strategy level, the response operations reduce additional environmental impacts to ALARP.

Spill response activities will be conducted in offshore waters using vessels and aircraft. The greatest potential for additional impacts from implementing spill response is considered to be on wildlife in offshore waters from oiled wildlife response activities.

Santos, together with the Controlling Agency for spill response, will apply appropriate processes and standards to ensure spill response impacts are reduced to a level that is ALARP.

All reasonably practicable control measures have been reviewed and those adopted are considered appropriate to manage the impacts such that the residual consequence is assessed to be II – Minor. The proposed control measures are in accordance with the Santos risk management criteria and are considered appropriate to manage impacts to ALARP.

### 7.7.7 Acceptability evaluation

Is the consequence ranked as I or II?	Yes – maximum consequence is II – Minor from contingency spill response operations.
Is further information required to validate the consequence assessment?	No – potential impacts and risks are well understood through the information available.
Are the risks and impacts consistent with the principles of ESD?	Yes – Activity evaluated in accordance with Santos' Offshore Division Environmental Hazard Identification and Assessment Guideline which considers principles of ecologically sustainable development.
Have the acceptable levels of impact and risks been informed by relevant species recovery plans, threat abatement plans, conservation advice, wildlife conservation	Yes – Control measures implemented will reduce the impact of contingency spill response operations to species identified in the following relevant species recovery plans, conservation advice, wildlife conservation plans and other management plans/guidelines, as also set out in Table 3-13.
plans and Australian marine park zoning	Conservation advice
objectives) ?	<ul> <li>Approved Conservation Advice for <i>Pristis clavata</i> (Dwarf Sawfish) (DEWHA, 2009)</li> </ul>
	Approved Conservation Advice for Green Sawfish (DEWHA, 2008c)
	<ul> <li>Approved Conservation Advice for <i>Pristis pristis</i> (largetooth sawfish) (DoE, 2014b)</li> </ul>
	<ul> <li>Approved Conservation Advice for <i>Glyphis garricki</i> (northern river shark) (TSSC, 2014a)</li> </ul>
	<ul> <li>Approved Conservation Advice for <i>Glyphis glyphis</i> (speartooth shark) (DoE, 2014a)</li> </ul>
	<ul> <li>Approved Conservation Advice for <i>Rhincodon typus</i> (whale shark) (TSSC, 2015g)</li> </ul>
	<ul> <li>Approved Conservation Advice for <i>Balaenoptera physalus</i> (fin whale) (TSSC, 2015c)</li> </ul>

	Approved Conservation Advice for <i>Balaenoptera borealis</i> (sei whale)     (TSSC, 2015b)
	Approved Conservation Advice for <i>Limnodromus semipalmatus</i> (Asian
	dowitcher) (DCCEEW, 2024f)
	<ul> <li>Approved Conservation Advice for Limosa limosa (black-tailed godwit) (DCCEEW, 2024e)</li> </ul>
	<ul> <li>Approved Conservation Advice for Calidris tenuirostris (great knot) (DCCEEW, 2024d)</li> </ul>
	Approved Conservation Advice for <i>Charadrius leschenaultii</i> (greater sand plover) (DCCEEW, 2023e)
	<ul> <li>Approved Conservation Advice for <i>Pluvialis squatarola</i> (grey plover) (DCCEEW, 2024d)</li> </ul>
	<ul> <li>Approved Conservation Advice for Limosa lapponica baueri (Alaskan bar- tailed godwit) (DCCEEW, 2024k)</li> </ul>
	<ul> <li>Approved Conservation Advice for Calidris canutus (red knot) (DCCEEW, 2024c)</li> </ul>
	<ul> <li>Approved Conservation Advice for Arenaria interpres (ruddy turnstone) (DCCEEW, 2024I)</li> </ul>
	<ul> <li>Approved Conservation Advice for Calidris acuminata (sharp-tailed sandpiper) (DCCEEW, 2024b)</li> </ul>
	<ul> <li>Approved Conservation Advice for Xenus cinereus (terek sandpiper) (DCCEEW, 2024i)</li> </ul>
	• Approved Conservation Advice for <i>Rostratula australis</i> (DSEWPaC, 2013)
	<ul> <li>Conservation Advice for Charadrius mongolus (lesser sand plover) (DCCEW, 2024a)</li> </ul>
	• Approved Conservation Advice for <i>Rostratula australis</i> (Australian painted snipe) (TSSC, 2013).
	Recovery plans
	Sawfish and River Sharks Multispecies Recovery Plan (CoA, 2015b)
	<ul> <li>Recovery Plan for the White Shark (<i>Carcharodon carcharias</i>) (Department of Sustainability, Environment, Water, Population and Communities (DSEWPaC, 2013)</li> </ul>
	Recovery Plan for the Grey Nurse Shark (Carcharias taurus) (CoA, 2014b)
	<ul> <li>Conservation Management Plan for the Blue Whale - A Recovery Plan under the Environment Protection and Biodiversity Conservation Act 1999 2015–2025 (CoA, 2015a)</li> </ul>
	Recovery Plan for Marine Turtles in Australia 2017–2027 (CoA, 2017b)
	<ul> <li>National Recovery Plan for the Australian Painted Snipe (Rostratula australis) (DCCEEW, 2022a)</li> </ul>
	Wildlife Conservation Plan for Seabirds (CoA, 2020)
	Wildlife Conservation Plan for Migratory Shorebirds (CoA, 2015c).
	Other management plans/guidelines
	<ul> <li>National Light Pollution Guidelines for Wildlife (DCCEEW, 2023c)</li> </ul>
	Marine bioregional plans for the NMR (CoA, 2012).
	For the identified plans, the objectives of those plans are achieved through the adoption of performance outcomes and the control measures outlined in Section 7.7.4. Santos considers that the level of potential impact from contingency spill response operations is not inconsistent with these plans.
	Management is also consistent with the zoning of the Australian marine parks in that risks have been reduced to ALARP; for example, implementation of spill response activities will limit impacts, thereby conserving the marine park values as required by the North Marine Parks Network Management Plan (DNP, 2018a) and North-West Marine Parks Network Management Plan (DNP, 2018b).
Are performance outcomes, control measures and associated performance standards consistent with legal and	Yes – management consistent with and National Plan for Maritime Environmental Emergencies (AMSA, 2020), among other legislation identified in Appendix B.
regulatory requirements?	Through acceptance of this GEP Coastal Waters OEMP, legislative and regulatory requirements will be met as per Appendix B.



Are performance outcomes, control measures and associated performance standards consistent with Santos' Environment, Health and Safety Policy?	Yes – aligns with Santos' Environment, Health and Safety Policy (Appendix A).
Are performance outcomes, control measures and associated performance standards consistent with industry standards?	Yes – the most recent and comparable Operations EPs accepted by NOPSEMA have been reviewed for consistency with the performance outcomes, control measures and associated performance standards proposed in this GEP Coastal Waters OEMP.
Have performance outcomes, control measures and associated performance standards taken into consideration Relevant Person feedback	Yes – Relevant Persons feedback received during consultation for this activity has been considered when evaluating performance outcomes, control measures and associated performance standards. Where relevant, control measures implemented based on Relevant Persons feedback for other Barossa EPs have been adopted in this GEP Coastal Waters OEMP. Additional EPOs adopted. No additional CMs adopted.
Are performance standards such that the impact or risk is considered to be ALARP?	Yes – ALARP assessment conducted, with no additional control measures adopted.

The consequence of spill response operations on receptors is assessed as II – Minor. Based on an assessment of Santos' acceptability criteria and with the control measures in place, potential impacts are considered acceptable.

### 8. Implementation strategy

#### Offshore Petroleum and Greenhouse Gas Storage (Environment) Regulations 2023 (OPGGS(E)R 2023) requirements

Section 22 Implementation strategy for environment plan

22(1) The environment plan must contain an implementation strategy for the activity in accordance with this section.

#### Consultation and compliance

22(16) The implementation strategy must comply with the Act, this instrument, any other regulations made under the Act, and any other environmental legislation applying to the activity.

This section describes the implementation strategy for this GEP Coastal Waters OEMP which follows a PLAN-DO-CHECK-ACT cycle and is structured accordingly. The implementation strategy is based on the assessment of impacts and risks and describes how the control measures (Table 8-2) will be implemented to achieve the environmental performance outcomes (Table 8-1) and performance standards (Table 8-2). It describes the systems, practices and procedures in place to plan, implement, monitor and manage the activities so environmental risks and impacts are continually being reduced to ALARP and are acceptable.



## 8.1 Environmental performance outcomes, control measures and performance standards

### 8.1.1 Environmental performance outcomes and standards

# OPGGS(E)R 2023 Requirements Section 21 Environmental assessment Environmental performance outcomes and standards 21(7) The environment plan must:

- a) set environmental performance standards for the control measures identified under paragraph (5)(c); and
- b) set out the environmental performance outcomes for the activity against which the performance of the titleholder in protecting the environment is to be measured; include measurement criteria that the titleholder will use to determine whether each environmental performance outcome and environmental performance standard is being met.

To ensure environmental risks and impacts will be reduced to ALARP and be of an acceptable level, environmental performance outcomes have been defined and are listed in Table 8-1, except those relating to hydrocarbon spill response, which are listed in the Barossa GEP NT Waters OPEP. These outcomes will be achieved by



implementing the identified control measures to the defined environmental performance standards (Table 8-2), noting some control measures are applicable to multiple environmental performance outcomes.

#### Table 8-1: Environmental performance outcomes

Reference	Environmental performance outcomes
EPO-01	No vessel collisions or adverse interactions with other marine users.
EPO-02	Vessel speeds in operational areas will not exceed applicable restrictions, to reduce the risk of physical interactions between cetaceans / marine reptiles and vessels.
EPO-03	Zero incidents of injury/mortality of cetaceans/marine reptiles from collision with vessels.
EPO-04	Seabed disturbance to be limited to planned activities and impacts described as part of the Activity and will not occur outside the Operational Area.
EPO-05	No anchoring or mooring of vessels on shoals/banks.
EPO-06	No loss of equipment or cargo overboard from vessels.
EPO-07	No introduction, establishment or spread of IMS in the natural environment as a result of the Activity.
EPO-08	No injury of, or mortality to, EPBC Act 1999 listed marine fauna or marine fauna listed as threatened species under the <i>Territory Parks and Wildlife Conservation Act 1976</i> (NT) as a result of the Activity.
EPO-09	Atmospheric emissions associated with the Activity will meet all regulatory source emission standards.
EPO-10	Manage indirect GHG emissions associated with the Activity consistent with the temperature objectives of the Paris Agreement, including by implementing company-wide targets and strategies for Scope 3 emissions reduction at the Barossa Gas Project as appropriate (having regard to joint venture arrangements and Barossa operations) and supporting customers and suppliers to reduce their GHG emissions.
EPO-11	Undertake the Activity in a manner that is compliant with the requirements of the Safeguard Mechanism.
EPO-12	No light emissions from the Activity except as required for safe operations and working requirements.
EPO-14	Planned discharges will meet relevant maritime obligations and Santos chemical assessment and approval process.
EPO-15	No displacement of marine turtles from habitat critical during nesting/breeding (including internesting periods for turtles) and ensure biologically important behaviour can continue in biologically important areas.
EPO-16	Zero unplanned discharge of hazardous and non-hazardous wastes into the marine environment from the Activity.
EPO-18	Zero unplanned discharge of hydrocarbons or chemicals to the marine environment from the Activity.
EPO-19	No significant <sup>1</sup> impact to cultural features from the Activity.
EPO-20	No impacts to underwater cultural heritage from the Activity.

Note 1: Significant is defined in the Santos Environment Consequence Descriptors (Appendix G).

### 8.1.2 Control measures and performance standards

Hazards and associated environmental risks and impacts for the proposed activities have been systematically identified and assessed in this GEP Coastal Waters OEMP in accordance with Santos' Offshore Division Environmental Hazard Identification and Assessment Guideline. The control measures that will be used to manage identified environmental impacts and risks and the associated statements of performance required of the control measure (Environmental Performance Standards) are listed in Table 8-2. Measurement criteria outlining how compliance with the control measure and the expected environmental performance could be evidenced are also listed.

All control measures and EPSs and associated measurement criteria relating to hydrocarbon spill preparedness and response operations are contained within the Barossa GEP NT Waters OPEP.

#### Table 8-2: Environmental performance standards and measurement criteria

EPO no	Control measure	Environmental performance standard	Measurement criteria
EPO-03	<ul> <li>BAO-CM-6.1.1</li> <li>Apply Santos' Protected Marine Fauna Interaction and Sighting Procedure to vessel and helicopter activities when in the vicinity of cetaceans and turtles</li> </ul>	Vessel/s comply with Santos' Protected Marine Fauna Interaction and Sighting Procedure, which ensures compliance with EPBC Regulations 2000- Part 8 which includes controls for minimising the risk of collision with marine fauna.	Conformance checked on recei
EPO-08			Completed vessel statement of
EPO-16		Any vessel strikes with cetaceans will be reported in the National Ship Strike Database.	Conformance checked on Santo
		Helicopter contractor procedures comply with Santos' Protected Marine Fauna Interaction and Sighting Procedure, which ensures compliance with Part 8 of the Environment Protection and Biodiversity Conservation Regulations 2000, which includes controls for minimising interaction with marine fauna.	Helicopter contractor procedure Interaction and Sighting Proced
	The vessel master or crew will act as a wildlife observer and record sightings of cetaceans and turtles.	Recorded marine fauna observa Regulations – Part 8 Division 8. marine turtles), including initiation was operated within a caution z	
EPO-01 EPO-08	BAO-CM-6.1.2 Vessels equipped and crewed in accordance with	Vessels will be equipped and crewed in accordance with the Navigation Act 2012 (Cth) (as applicable for vessel size, type, and class), including implementing:	A Minimum Safe Manning Certi qualifications to meet the STCW and class).
EPO-09	Marine Order 30 (Prevention of Collisions) and	measures such as manning and watchkeeping.	Records of Santos marine vess
EPO-12 EPO-15	Arrangements)	<ul> <li>Marine Order 27 (Safety of navigation and radio equipment), including:</li> <li>radio equipment and communications</li> </ul>	type and class) to demonstrate
EPO-16		<ul> <li>navigation safety measures and equipment</li> <li>denger, urganey, and distance signals and measures</li> </ul>	<ul> <li>Global Maritime Distres maintained</li> </ul>
EPO-19		<ul> <li>Marine Order 30 (Prevention of Collisions), including: lights and signals as</li> </ul>	radio equipment availat
EPO-20		<ul> <li>applicable to vessel class per COLREGS requirements.</li> <li>Marine Order 70 (Vessel marine crew are trained and competent to navigate vessels</li> <li>Marine Order 71 (master's and Deck Officers), including: all master, mate and watchkeeper officer duties undertaken by crew certified as applicable to vessel class per International Convention on Standards of Training, Certification and Watchkeeping for Seafarers 1978 (STCW) requirements.</li> </ul>	electronic and/or paper
			A Vessel Cargo Ship Safety Eq lights, shapes, and means of ma accordance with COLREGS rec and class).
			Records of vessel crew STCW Manning Certificate (as applicat
			Non-compliance with relevant M action undertaken documented
EPO-19	BAO-CM-6.1.4 Inductions for all site-based workforce will include information on cultural heritage to raise awareness about the cultural and spiritual belief of First Nations people	Activity inductions contain information on cultural heritage and are completed by all	Records demonstrate cultural h
EPO-20		site-based workforce prior to commencement of Activity.	based workforce
EPO-12	BAO-CM-6.2.1	Vessel navigation lighting and equipment is compliant with COLREGS/Marine Orders	Vessel certification confirms con
EPO-19 Lighting limited to that required for safe work conditions and navigational purposes	30: Prevention of Collisions, Marine Orders 21: Safety of Navigation and Emergency Procedures, Navigation Act 2012 (Cth). Work lighting will be the minimum required to maintain safe working conditions for all areas where the crew are operating on the	Vessel crew induction outlines r maintain safe working condition	
		deck.	Inspection verifies no excessive work/navigation.
	HSE induction to crew includes minimising light emissions from vessel during night hours where possible.	Records demonstrate all projec Induction that includes minimisi	



ipt of marine fauna sighting datasheets.

f conformance.

tos' receipt of incident report.

es align with Santos' Protected Marine Fauna dure.

vations demonstrate adherence to EPBC 8.1 Interacting with cetaceans (and applied for ion of management measures for when the vessel zone.

ificate is in place and identifies minimum crew *N* requirements (as applicable for vessel size, type

sel vetting process (as applicable for vessel size, the following:

ss and Safety System (GMDSS) radio logbook

ble, working and tested at regular intervals r-based charts are available.

quipment Certificate demonstrates the vessel has naking sound signals and distress signals in quirements (as applicable for vessel size, type,

qualifications align with the Minimum Safe able for vessel size, type, and class)

Marine Orders 21, 27, 30 70 and 71 and corrective (as applicable for vessel size, type and class).

neritage awareness inductions completed by site-

ompliance with applicable regulations.

requirement to keep work lighting to a minimum to

e light being used beyond that required for safe

ct personnel have attended the Activity HSE ing light emissions.

EPO no	Control measure	Environmental performance standard	Measurement criteria
EPO-12 EPO-19	BAO-CM-6.2.2 Additional lighting management (as recommended in the National Light Pollution Guidelines for Wildlife (DCCEEW, 2023h) implemented in the OA when undertaking activities within 3.3 km of turtle BIA or HC, where it does not impact the ability of light to safely illuminate the work area	<ul> <li>When undertaking activity within 3.3. km of known turtle BIA or habitat critical, additional measures implemented to minimise direct light spill on the ocean surface will include:</li> <li>turning off lights not in use</li> <li>closing curtains</li> <li>adjusting orientation of lights</li> <li>installing shielding where it does not impact the ability of light to safely illuminate the work area.</li> </ul>	Completed vessel statement of
EPO-12	BAO-CM-6.2.3	Vessel searchlights shall only be operated when retrieving AUVs at night or in the	Training and induction records f
EPO-19	Vessel searchlights will only be operated when retrieving AUVs at night or in the event of an emergency	event of an emergency to minimise light emissions.	be operated only in an emerger
EPO-11	BAO-CM-6.3.1	Daily vessel fuel use monitoring.	Fuel use monitoring recorded in
	Monitoring of vessel fuel consumption and vessel speed management to reduce fuel use	Vessels instructed, prior to vessel sailing, to sail at 'economic' speed specific to each vessel to reduce fuel use, subject to operational requirements.	Vessel speed instructions issue of compliance recorded in daily
EPO-11	BAO-CM-6.3.9 Reporting of GHG emissions as per the NGER Scheme	NGERS reporting is compliant with requirements set by Clean Energy Regulator and NGER report is lodged annually.	Records show that National Gre Mechanism) Rule 2015 has bee relevant Barossa facility emission set by the Clean Energy Regula
EPO-11	BAO-CM-6.3.11 The purchase and/or surrender of Australian Carbon Credit Units (ACCUs) or Safeguard Mechanism Credits (SMCs) required under the NGER (Safeguard Mechanism) Rule 2015 for any non-reservoir emissions from the Barossa facility above the annual baseline as determined by the Clean Energy Regulator.	If there are non-reservoir emissions from the Barossa facility above the annual baseline Australian Carbon Credit Units (ACCUs) or Safeguard Mechanism Credits (SMCs) will be purchased or surrendered as required under the NGER (Safeguard Mechanism) Rule 2015.	Records demonstrate net GHG baseline under the Safeguard M
EPO-11	<ul> <li>BAO-CM-6.3.12</li> <li>Implement an Operations GHG Emissions Management Plan (GHGEMP) as described in Section 8.3.2.12 to manage facility direct GHG emissions to ALARP over the life of the Activity, inclusive of: <ul> <li>Emissions Performance target setting (Section 8.2.4)</li> <li>Critical Equipment Maintenance (Section 8.3.2.3.1)</li> <li>Methane emissions management (Section 8.3.2.11)</li> <li>Decarbonisation opportunity management (Section 8.5.6)</li> </ul> </li> </ul>	The GHGEMP will be implemented in accordance with Section 8.3.2.12.	Records demonstrate impleme
		The GHGEMP will be reviewed annually.	standards listed for this control Records demonstrate annual re
		Emissions performance targets are set and tracked as described in Section 8.2.4	Records demonstrate both ongo performance targets and annua
		Emissions monitoring or emissions control equipment maintenance as described in Section 8.3.2.3.1.	Records demonstrate maintena control equipment according to
		Implementation of the Decarbonisation Opportunity Management process for the Barossa facility as described in Section 8.5.6	Records demonstrate Decarbor and reviewed annually in accord 8.5.6. Records demonstrate pro Opportunity Management plan.
EPO-09	BAO-CM-6.3.16 MARPOL-compliant (Marine Order 97) fuel oil will be used by vessels to reduce atmospheric emissions.	Vessels contracted whose practices comply with Marine Order 97 (including use of fuel oil) as applicable to vessel size, type, and class to reduce atmospheric emissions.	Fuel supply specifications show
EPO-09	BAO-CM-6.3.17 Pursuant to Marine Order 97, relevant vessels will have a current International Air Pollution	Vessels contracted will maintain a current International Air Pollution Prevention (IAPP) Certificate and/or Engine IAPP Certificate and/or International Energy Efficiency Certificate (IEEC) (or equivalent)', which certifies that measures to prevent	Current IAPP Certificate or equi



conformance.

for Vessel Masters detail that search lights are to ncy or when retrieving AUVs at night.

daily vessel performance reports.

ed to vessel prior to vessel sailing and statement vessel performance reports.

eenhouse and Energy Reporting (Safeguard en used to measure, report, and manage the ons and they are compliant with the requirements ator.

emissions managed within the applicable Mechanism.

ntation of the GHGEMP against the performance measure.

evision of the GHGEMP.

oing monthly reviews against emissions al review against emissions estimates.

ance for emissions monitoring and emissions system/equipment criticality requirements.

nisation Opportunity Management plan developed dance with the process described in Section ogress in implementing Decarbonisation

fuel is MARPOL-compliant on vessels.

ivalent in place for relevant vessels

EPO no	Control measure	Environmental performance standard	Measurement criteria
	Prevention (IAPP) Certificate or equivalent Ship Energy Efficiency Management Plan (SEEMP)	ozone-depleting substance emissions and reduce NOx, SOx and incineration emissions during the activity are in place.	
EPO-09	BAO-CM-6.3.18 Santos' vessel vetting process to include evaluation of vessel emissions and the potential for use of alternative fuels.	Vessel vetting includes evaluation of vessel emissions and the potential for use of alternative fuels to reduce scope 3 emissions.	Completed documentation of ev
EPO-10	<ul> <li>BAO-CM-6.3.21</li> <li>GHG emissions reduction initiatives of suppliers for the Barossa Gas Project will be evaluated in the tender evaluation process via development and implementation of a framework for identifying, assessing and implementing emissions reduction opportunities for all Barossa Gas Project supplier contracts of \$30m+ value.</li> <li>Through the data collection and tender evaluation process, opportunities to collaborate on emissions reduction initiatives and low carbon alternatives will be sought, including the potential to support suppliers in respect of: <ul> <li>investments in innovations in technology;</li> <li>research programs;</li> <li>education and training relating to the adoption of emissions reduction policies and processes; and/or</li> <li>monitoring programs</li> </ul> </li> <li>The tender evaluation framework will be reviewed and refined to ensure it is adaptive to advancements in technology, data collected and other opportunities to encourage reductions in GHG emissions.</li> </ul>	Data will be collected and recorded via the tender process. GHG emissions reduction initiatives will be evaluated during the tender process via a tender evaluation framework. Collaborating with Barossa Gas Project suppliers on initiatives may be adopted subject to a feasibility analysis, the willingness of suppliers to collaborate, and value to the environment. Annual review of the threshold spend will be conducted to ensure the most emissions intensive activities are captured. The tender evaluation framework will be reviewed and refined annually to ensure it is adaptive to advancements in technology, data collected and other opportunities to encourage reductions in GHG emissions.	Records demonstrate developm framework, tender scope of wor Records will evidence collabora investment spend and the imple explanation where this has not of review of threshold spend has b Records demonstrate annual re ensure it is adaptive to advance opportunities to encourage redu
EPO-10	<ul> <li>BAO-CM-6.3.22</li> <li>Annual engagement with suppliers with Barossa Gas Project supplier contracts of \$30m+ value to request GHG emissions data for Barossa activities. Data sought would include:</li> <li>quantitative and qualitative climate-related targets (including for Scopes 1, 2 and 3 emissions);</li> <li>information about the supplier's approach to setting, reviewing and monitoring progress against each target;</li> <li>information about the supplier's performance against each climate-related target, including GHG emissions data and measurement approach, inputs and assumptions, over the past year;</li> <li>the supplier's use over the past year, and planned use, of carbon credits to offset GHG emissions;</li> <li>information regarding the supplier's climate- related risks and opportunities, including</li> </ul>	Data will be collected annually and evaluated on an annual basis against Santos' scope 3 emissions estimations, Santos' Scope 3 equivalent climate targets and Climate Transition Action Plan. Review to be undertaken annually of options for improvements in the management of or reduction of Scope 3 emissions arising from this engagement.	Records will evidence annual su evaluation of responses. Records will evidence annual ve associated with Barossa Gas Pr equivalent climate targets and C practicable through the above e Records will demonstrate review management of or reduction of

aluation/

ment and utilisation of tender evaluation ork and evaluation forms.

ation with Barossa Gas Project suppliers including ementation of initiatives and programs, including occurred. Records demonstrate that annual been conducted.

evisions of the tender evaluation framework to ements in technology, data collected and other uctions in GHG emissions.

supplier data requests, responses received and

verification of GHG emissions estimates Project suppliers and evaluation against Scope 3 Climate Transition Action Plan to the extent engagement with suppliers.

w of any options for improvements in the Scope 3 emissions arising from this engagement.

EPO no	Control measure	Environmental performance standard	Measurement criteria
	information regarding the supplier's emissions reduction initiatives (if any).		
	Data will be used to verify GHG emissions estimates associated with our suppliers and track performance against Santos' Scope 3 equivalent targets and Climate Transition Action Plan.		
EPO-10	<ul> <li>BAO-CM-6.3.24</li> <li>Supporting Barossa Gas Project suppliers to reduce GHG emissions by:</li> <li>promoting global measurement and reporting standards by participating in relevant industry associations and collaboration initiatives; and</li> <li>advocating for policy frameworks that enable a consistent approach to carbon emissions management.</li> </ul>	Annual review of current or new industry forums, associations and initiatives warranting participation/involvement will be undertaken. Participation will take the form of attendance at events and actions arising, which may include ongoing collaboration and/or engagement with policy makers and other stakeholders, as relevant. Review to be undertaken annually of any options for improvement in GHG emissions management or GHG emissions reductions arising from such engagement.	Records will demonstrate annu associations and initiatives war Records will demonstrate partic collaboration initiatives, and en stakeholders, as relevant. Records will demonstrate cons GHG emissions management of engagement.
EPO-09	BAO-CM-6.4.1 Ozone depleting substance (ODS) and lower global warming potential (GWP) refrigerants use and handling procedures	ODS on vessels is managed in accordance with Marine Order 97 (vessels) and MARPOL Annex VI to reduce the risk of an accidental release of ODS to air.	Completed ODS Record Book with Marine Order 97 (vessels)
EPO-09	BAO-CM-6.4.3 Vessel waste incineration management	Waste incineration on vessels is managed in accordance with Marine Order 97/MARPOL Annex V to minimise atmospheric emissions.	Completed vessel waste record Marine Order 97
EPO-09	BAO-CM-6.4.4 National Pollutant Inventory (NPI) Reporting	NPI reporting is lodged as per the NPI submission requirements.	Records show that NPI reports
All	BAO-CM-6.4.5 HSE inductions will include applicable environmental requirements	All project personnel will attend HSE inductions which will include environmental requirements as required by this GEP Coastal Waters OEMP to reduce environmental impacts and risks to ALARP and acceptable levels.	Records demonstrate all project Induction and that HSE induction required by this GEP Coastal V ALARP and acceptable levels.
EPO-08 EPO-09 EPO-13	BAO-CM-6.4.10 Barossa Facilities and vessels planned maintenance system to confirm equipment integrity is maintained in accordance with manufacturers guidelines.	Documented maintenance program is in place for lifting equipment on vessels that provides a status on the maintenance of equipment in accordance with manufacturers guidelines.	Records from Santos vessel ve cranes and lifting equipment or manufacturer guidelines.
EPO-14 EPO-15 EPO-16 EPO-17		<ul> <li>Ensure bunkering equipment is maintained to reduce the likelihood of loss of integrity events during transfers:</li> <li>visual inspection of the integrity of the prior to bunkering</li> <li>test date/certification of the hose is checked prior to bunkering</li> </ul>	Completed pre-bunkering chec the test date/certification is valid
EPO-19		Documented maintenance program is in place for discharge monitoring equipment to ensure they are operating within their design parameters and analysers are calibrated in accordance with manufacturer guidelines.	Records show maintenance of with manufacturer guidelines.
EPO-04	BAO-CM-6.5.2 Maintain a subsea infrastructure inventory	Maintain a comprehensive and accurate inventory of subsea infrastructure and locations, including tracking of subsea infrastructure brought into the OA. An accurate inventory will reduce permanent seabed disturbance as it will enable Santos to fulfill future decommissioning/removal responsibilities.	Subsea infrastructure inventory the project.
EPO-04	BAO-CM-6.5.3 Span correction procedures to be developed, if required	If required, a span-correction procedure will be developed to provide clear direction on how spans shall be rectified and surveyed to minimise seabed disturbance.	A copy of a span rectification p requirements.

I review of current or new industry forums,
anting participation/involvement.

cipation in relevant industry association and gagement with policy makers and other

ideration of any potential for improvements in or emissions reductions arising from this

or recording system is on vessel in accordance and MARPOL VI.

book or recording system in accordance with

have been lodged as per NPI submission

ct personnel have attended the Activity HSE ons include environmental requirements as Vaters OEMP to reduce environmental impacts to

etting process confirm planned maintenance of n vessels undertaken in accordance with

klist show bunkering equipment is checked and

discharge monitoring equipment in accordance

records completed and maintained throughout

rocedure (or similar) demonstrating alignment to
EPO no	Control measure	Environmental performance standard	Measurement criteria
EPO-04 EPO-05 EPO-20 EPO-21	BAO-CM-6.5.4 Vessels will not anchor under routine operations within the OA however may anchor during emergency conditions.	Vessels will not anchor under routine operations within the OA however may anchor during emergency conditions	No reported or recorded inciden the event of emergency conditio
EPO-01 EPO-04 EPO-19 EPO-20 EPO-21	BAO-CM-6.6.1 Notify AHS and AMSA MSI prior to relevant Activity	AHS Notice to Mariners and AMSA MSI will be notified prior to relevant Activity to reduce the likelihood of unplanned interactions with other vessels.	Consultation records demonstra information to generate Notice to
EPO-01 EPO-07 EPO-09 EPO-17 EPO-18 EPO-19 EPO-20	BAO-CM-6.6.4 Activity undertaken in accordance with Santos HSE management and marine vessel vetting processes (Santos' Offshore Marine Assurance Procedure)	Vessels selected and onboarded in accordance with Santos' Offshore Marine Assurance Procedure to ensure contracted vessels equipment is maintained in accordance with Santos and industry standards, and regulatory requirements to ensure other marine users are aware of physical presence thus reducing the potential for interaction or collision.	Completed documentation in ac Assurance Procedure.
EPO-02 EPO-19 EPO-20 EPO-21	BAO-CM-6.6.6 Vessel speed restrictions within 500m around the IMMR vessels and campaign vessels	Restrict vessel operating speeds to 8 knots or less within 500m safety zone around IMMR vessels and campaign vessels to reduce the likelihood of unplanned interactions with other vessels with the exception of other fast rescue craft undertaking drills and training or responding to an emergency situation" Project induction material includes an environmental requirements section that details speed limit requirements to reduce the likelihood of unplanned interactions with other vessels.	Vessel speeds in exceedance o documentation and corrective ac of other fast rescue craft underta emergency situation". Induction records confirm all pro induction, and that induction inc
EPO-01	BAO-CM-6.6.7 Communications plan will be implemented for engagement prior to and during the Activity that may impact marine users, to raise awareness of the activity.	A communications plan will be developed by end of the first quarter of each year that identifies key Barossa activities that may impact other marine users and set out how we will communicate the location, timing, and nature of the identified Barossa activities to marine users that may be impacted. Communications will occur in accordance with Table 8-13	Consultation records demonstra communications plan, and any c
EPO-01	BAO-CM-6.6.8 Charting of infrastructure on nautical charts	Activity infrastructure will be clearly marked on Australian nautical charts published by the AHO to reduce the likelihood of unplanned interactions with other vessels.	Evidence of confirmation that Au show infrastructure installed as p Coastal Waters OEMP.
EPO-14 EPO-17 EPO-20	BAO-CM-6.7.2 Routine discharges of treated sewage and grey water, in accordance with Marine Order 96 (Marine Pollution Prevention – Sewage)	<ul> <li>Valid International Sewage Pollution Prevention (ISPP) Certificate (as relevant to vessel class and type) that details the vessel has a:</li> <li>MARPOL approved sewage treatment plant</li> <li>sewage comminuting and disinfecting system</li> <li>sewage holding tank sized appropriately to contain all generated waste (black</li> </ul>	A copy of valid International Sev demonstrating the vessel has a relevant to relevant to vessel cla Where the vessel does not have records of sewage treated using system are maintained in an Off
			Where the vessel does not have records of sewage not comminu Log Book (or similar) that record that discharge occurred at a dist

ts of anchoring	occurring within	n the OA,	unless in
ns.			

ate AHS and AMSA MSI provided sufficient to Mariners prior to relevant activities.

ccordance with Santos' Offshore Marine

of 8 knots are contained in incident reports action undertaken documented. With the exception taking drills and training or responding to an

oject personnel have completed the project cludes information specified in the EPS.

ate implementation and annual review of a consultation feedback received.

Australian nautical charts published by the AHO s part of the activities described in the GEP

ewage Pollution Prevention (ISPP) Certificate a MARPOL approved sewage treatment plant (as lass and type).

ve a MARPOL approved sewage treatment plant, ng an approved comminuted and disinfecting fficial Log Book (or similar) that records discharge ifies that discharge occurred at a distance of more nd.

ve a MARPOL approved sewage treatment plant, uted or disinfected are maintained in an Official rds discharge locations and volumes and verifies stance of more than 12 NM from the nearest land.

EPO no	Control measure	Environmental performance standard	Measurement criteria
EPO-14 EPO-19	EPO-14BAO-CM-6.7.3EPO-19Deck cleaning product selection according to MARPOL Annex V (and Marine Order 93: Noxious liquid)	Deck cleaning products planned to be released to sea from the vessels meet the criteria for not being harmful to the marine environment according to MARPOL Annex	Safety Data Sheet and product deck cleaning products comply
EPO-20		v.	Records demonstrate the chem all deck cleaning chemicals.
			Completed vessel inspection ch
EPO-13 EPO-14 EPO-19 EPO-20	BAO-CM-6.7.5 Apply the Santos chemical selection process for chemicals planned to be discharged (Section 2.7).	Chemicals planned to be discharged to sea are Gold/Silver/D or E rated through OCNS, or PLONOR substances listed by OSPAR, or have a complete risk assessment as per Santos Offshore Division Operations Chemical Approval Procedure so that only environmentally acceptable products are discharged.	Records demonstrate the chem been implemented for all chem
EPO-14 EPO-19 EPO-20	BAO-CM-6.7.6 Routine discharges of treated bilge and deck water from vessels will comply with Marine Order	Pursuant to Marine Order 91, support vessels larger than 400 t will have an International Oil Pollution Prevention Certificate, which certifies that required measures to reduce impacts of planned oil discharges are in place	A copy of a current Internationa
	91 and <i>Marine Pollution Act 1999</i> (NT), as applicable.	Machinery space bilge/oily water for support vessels larger than 400 t shall have IMO approved oil filtering equipment (oil/water separator) with an on-line monitoring device to measure Oil in Water (OIW) content to be less than 15 ppm prior to discharge.	Supplement to the IOPP Certifi approved oil / water separator less than 15 ppm OIW (as releven
		Oily mixtures (bilge water) only discharged to sea in accordance with Marine Order 91 and <i>Marine Pollution Act 1999</i> (NT), as applicable.	Evidence of a current and main
EPO-22	BAO-CM-6.7.11 Contractor contingency pipeline preservation	Contractor contingency pipeline preservation procedure and specification will be implemented and will include:	A copy of the contractor conting specification are aligned with re
procedure and specification	<ul> <li>treatment chemicals selected will be Gold (OCNS) or pseudo-CHARM rated Gold</li> </ul>	Records demonstrate the chemrelevant chemicals.	
			Records demonstrate that the of (OCNS) or pseudo-CHARM rat
EPO-20 EPO-22	BAO-CM-6.7.12 Contractor contingency pipeline major repair procedure to be developed in the event a major repair is required.	The contractor will develop a major repair procedure, in the case that a major repair is required, and will include management of treated seawater to reduce the likelihood of an unplanned treated sea water release.	Records confirm a pipeline maj event of a major repair is requir seawater.
EPO-06BAO-CM-7.1.1EPO-17Implement stan equipment to re during lifting.EPO-20EPO-21	BAO-CM-7.1.1 Implement standards and procedures for lifting equipment to reduced risk of dropped objects during lifting.	<ul> <li>Lifting operations comply with lifting IMMR procedures that include provision for:</li> <li>lift plans as required</li> <li>ROV monitored lifts, where required</li> <li>establishing and maintaining communications between relevant parties (e.g. ROV / crane operator) in line with roles and responsibilities</li> <li>Weather / sea state restrictions</li> <li>operational procedural guidelines lifting equipment (inspection and certification)</li> <li>training and competency.</li> </ul>	Records and Lifting Equipment
		When safe and practicable objects dropped overboard are recovered per BAO-CM- 7.1.2.	Fate of dropped objects detaile

t supplier supplementary data as required show with MARPOL Annex V.

nical selection process has been implemented for

hecklists.

nical selection process (Section 8.3.2.10) has icals planned to be discharged.

al Oil Pollution Prevention (IOPP) certificate

icate that indicates that the vessel has an with online monitoring calibrated to discharge at vant to relevant to vessel class and type).

ntained Oil Record Book.

gency pipeline preservation procedure and equirements listed in the EPS of BAO-CM-6.7.11

nical selection procedure was implemented for all

chemical treatment product selected is a Gold ted Gold.

jor repair procedure (or similar) is in place in the red and includes management of treated

Register shows lifting equipment is certified.

udes management of lifts and avoidance of

ed in incident documents.

EPO no	Control measure	Environmental performance standard	Measurement criteria
EPO-06 EPO-17 EPO-19 EPO-20 EPO-21	BAO-CM-7.1.2 Dropped objects (incident) management.	<ul> <li>For all dropped objects, dropped object (incident) management includes the following: <ul> <li>assessment of environmental risk</li> <li>assessment of feasibility of object recovery, where safe and practicable to do so</li> <li>implementing outcomes of the assessment.</li> </ul> </li> </ul>	Incident documentation details considerations and outcomes.
EPO-06 EPO-17 EPO-18 EPO-19 EPO-20 EPO-21	BAO-CM-7.1.3 International Maritime Dangerous Goods Code.	Dangerous goods managed in accordance with International Maritime Dangerous Goods Code to reduce the risk of an environmental incident, such as an accidental release to sea or unintended chemical reaction.	Records demonstrate that dang Vessels (RAVs) and foreign ves Order 41 (Division 4, Regulation completed multimodal dangeror
EPO 18 EPO 19	BAO-CM-7.1.4 Chemicals and hydrocarbons will be managed in accordance with SDS to reduce risk of release to the marine environment.	Chemicals and hydrocarbons managed in accordance with SDS in relation to safe handling and storage, spill response and emergency procedures, and disposal considerations.	Records of contractor vessel au with chemical and hydrocarbon
EPO-07 EPO-18 EPO-19	BAO-CM-7.2.1 Develop and implement a biosecurity management plan in consultation with and approved by the Department of Agriculture Forestry and Fisheries (DAFF).	<ul> <li>Vessels on contract to Santos are managed to low risk in accordance with the biosecurity management plan and Santos Offshore Division Invasive Marine Species Management Plan before movement or transit into or within the invasive marine species management zone, which requires: <ul> <li>compliance with the Biosecurity Act 2015</li> <li>assessment of applicable vessels using the IMS Management Plan risk assessment</li> <li>the management of immersible equipment to low risk</li> <li>accurately reporting information in accordance with Section 193 of the Biosecurity Act 2015</li> <li>compliance with the Biosecurity Status Document conditions</li> <li>maintain a Biofouling Management Plan and Ballast Water Management Plan as required</li> <li>read and understand the Biosecurity Status Document directions and conditions and keep a copy of the current version on board the vessel, for the duration of the voyage in Australia.</li> </ul> </li> </ul>	Record of Biosecurity Status Do Completed Pre-Arrival forms.
		Vessels mobilising from international locations will complete an IMS risk assessment, before first mobilisation to the OA, as described in Santos Offshore Division Invasive Marine Species Management Plan. The IMS risk assessment assigns a final risk category of low, acceptable, uncertain or high to vessels based on a range of information including last port of call, age of antifouling coating, internal sea water systems and niche management. If a risk category of uncertain or high is assigned, management responses will include inspections, cleaning or treatment of internal seawater systems (or a combination of these actions).	Records of IMS risk assessmer
		Vessels receive entry clearance from DAFF (Seaports) as necessary (or as applicable to their location and movements).	Records show a completed Que Biosecurity Control Determinati advance where practicable. Letter received from DAFF india low IMS risk status and entry cl
			Records show ballast water ma

the dropped objects management assessment,

gerous goods carried on Registered Australian essels are shipped in accordance with Marine on 16), and appropriate records including a bus goods form are kept.

#### udits and/or inspections demonstrate compliance o storage and handling requirements.

ocument for applicable vessels.

nt.

estionnaire for Biosecurity Exemptions for ion issued to Seaports at least one month in

cating that the vessels (as necessary) achieve a learance is granted.

anagement is implemented.

EPO no	Control measure	Environmental performance standard	Measurement criteria
		Pursuant to the Biosecurity Act 2015 and Australian Ballast Water Management Requirements 2017, vessels carrying ballast water and engaged in international voyages shall manage ballast water so marine pest species are not introduced.	Completed ballast water record
EPO-07 EPO-18	BAO-CM-7.2.2 Vessels undertake ballast water management or	Ballast water discharges will comply with the Australian Ballast Water Management Requirements (DAWE, 2020a), which implements the requirements of the Biosecurity	Records demonstrating a ballas is maintained.
EPO-19	treatment to achieve low risk ballast water.	Ships' Ballast Water and Sediments (as appropriate for vessel class)	If the vessel cannot demonstrat discharge logs confirm no disch including any ports.
			An International Ballast Water M demonstrates the principal balla with D-2 standards.
			A Biosecurity Status Document arriving from international locati ballast water risk assessment (f
EPO-07 EPO-18 EPO-19	BAO-CM-7.2.3 Vessels equipped with effective anti-fouling coatings.	Vessels will have a suitable anti-fouling coating in accordance with the Protection of the Sea (Harmful Anti-fouling Systems) Act 2006 (Cth) (as applicable for vessel class and type), including:	A copy of an approved internati each vessel class and type.
		<ul> <li>Marine Order 98 (Marine Pollution – Anti-fouling Systems) including (as required by vessel class):</li> <li>a valid international anti-fouling system certificate.</li> <li>Vessel anti-foulant system maintained in compliance with International Convention on the Control of Harmful Anti Fouling Systems on Ships where applicable.</li> </ul>	
EPO-07 EPO-18 EPO-19	BAO-CM-7.2.4 Vessels equipped with Marine Growth Prevention System (MGPS).	Vessels will have a marine growth prevention system or appropriate manual treatment systems.	Biosecurity management record prevention system or appropria
EPO 18 EPO 19	BAO-CM-7.4.1 ROV operations undertaken in accordance with	Scheduled preventive maintenance on ROV completed as per manufacturer specifications to reduce the risk of hydraulic fluid releases to sea.	Vessel contractor written verific specifications.
	good industry practice.	ROV pre-mobilisation audit completed to reduce the risk of hydraulic fluid releases to sea.	Records show a pre-mobilisation
EPO-17 EPO-18 EPO-19	BAO-CM-7.4.2 Bulk liquid transfer procedure	<ul> <li>Bunkering operation procedure is in place and includes key requirements to prevent spills to the environment such as:</li> <li>when bunkering activities can occur (hose connection restricted to daylight hours)</li> <li>roles and responsibilities for bunkering operations</li> <li>hoses have dry break couplings</li> <li>bunkering activity communication requirements</li> <li>hose integrity inspection</li> <li>requirement to be DP Class 2 vessels.</li> </ul>	Records show a completed bur undertaken.

book or log is maintained.

st water record system (electronic or in hard copy)

te it meets D-2 standards, records of ballast water harge within 12 nautical miles of coastlines

Management Certificate is in place for vessels and ast water management method is in accordance

t showing an approved ballast status (for vessels tions) or a low-risk exemption through a domestic (for domestic vessels).

ional anti-fouling system certificate, as relevant to

ds demonstrate vessels have a marine growth ate manual treatment systems.

cation demonstrates compliance with manufacturer

on audit completed for all ROV operations.

nkering checklist prior to any bunkering is

EPO no	Control measure	Environmental performance standard	Measurement criteria
EPO-19 EPO-20	BAO-CM-7.4.3 Vessel spill response plans (SOPEP/SMPEP)	Vessels have and implement checklists and a SOPEP or SMPEP pursuant to MARPOL Annex I. These measures reduce the likelihood of a spill entering the marine environment is reduced.	Approved SOPEP or SMPEP in
			Spill details contained in incider
		Spill response exercises conducted in accordance with SOPEP/SMPEP to ensure personnel are prepared.	Spill exercise records or eviden SOPEP/SMPEP requirements.
EPO-17 EPO-18 EPO-19	BAO-CM-7.4.4 Spill clean-up kits available in high-risk areas	Selection of vessel contractor is subject to Santos marine vessel vetting processes, specifically spill kits stocked and ready for use by trained personnel in the event of a spill to the marine environment.	Vessel audit process confirm sp with the manufacturer's guidanc
EPO-09 EPO-18 EPO-19 EPO-20	BAO-CM-7.6.1 Emergency response plan (ERP)	ERP details the requirements for preparedness and response to emergencies and crises to protect people and the environment. ERP is initiated to activate isolation of the flowline, pipeline and wells in the event the integrity of a pipeline and valve is compromised or there is an unplanned hydrocarbon release.	Completed incident documentat triggered by a release of hydroc
EPO-09 EPO-18 EPO-19 EPO-20	BAO-CM-7.6.2 Pipeline operating procedures	The Pipeline is operated within design envelope and maintained consistent with the Pipeline operating procedures reducing the potential of a pipeline rupture/subsea leak and hydrocarbon release.	Record logs show pipeline oper procedure. Inspection, review and mainten
EPO-09 EPO-18 EPO-19 EPO-20	BAO-CM-7.6.3 Pipeline Integrity Management Plan	The integrity of the pipeline is maintained consistent with the Pipeline Integrity Management Plan reducing the potential of a pipeline rupture/subsea leak and hydrocarbon release.	Pipeline maintenance and inspe
EPO-09 EPO-18 EPO-19 EPO-20	BAO-CM-7.6.5 Repairs to the Pipeline carried out to design specification	Pipeline repairs are carried out consistent with design specifications, including Pipeline Integrity Management Plan and Offshore Standard for Submarine Pipeline Systems, ensuring that infrastructure integrity remains in line with standards and specifications. The above reduced the possibility of a release of hydrocarbons resulting from a loss in infrastructure integrity.	Records demonstrate repairs to Pipeline Integrity Management I Pipeline Systems.
EPO-09 EPO-18 EPO-19 EPO-20	BAO-CM-7.6.9 Barossa Nearshore Gas Export Pipeline Operations Pipeline Management Plan	Barossa Nearshore Gas Export Pipeline Operations Pipeline Management Plan will be in place to identify hazards that have the potential to cause a loss of containment, detail the risks and identify physical barriers and safety management systems required to reduces risks to ALARP. If a loss of containment from the Pipeline was to occur, it will be managed in accordance with the Barossa Nearshore Gas Export Pipeline Operations Pipeline Management Plan, which details alarms and required emergency response in the event of a loss of containment.	Records demonstrate alarms ar in accordance with the Barossa Pipeline Management Plan.

place.

nt documentation.

nce of a spill exercise aligned with the vessel

pill kits stocked and ready for use in accordance ce.

ation shows ERP implemented as applicable if carbons.

ration has been in accordance with the relevant

ance records.

ection records.

o the Pipeline carried out in accordance with the Plan and DNV Offshore Standard for Submarine

re maintained and emergency response enacted a Nearshore Gas Export Pipeline Operations

EPO no	Control measure	Environmental performance standard	Measurement criteria
EPO-18 EPO-19	BAO-CM-7.7.8.6 Inspection of hydrocarbon containing equipment	The Barossa Project Integrity Management Plan – Subsea is implemented to ensure subsea infrastructure integrity is maintained, reducing likelihood of release to the marine environment. The plan includes:	Campaign-specific inspection re maintenance with the Barossa F
EPO-20		<ul> <li>inspection frequencies aligned with the RBI plan, including 3-yearly GVI of the GEP</li> <li>inspections methodologies (including GVI and CP)</li> <li>post-cyclone survey requirements, after a significant cyclonic event.</li> </ul>	GEP integrity records detailing ( inspection schedules, along with Any finding, maintenance or futu presented within the Maintenan
EP-19 EP-20	BAO-CM-7.7.8.15 Accepted Barossa GEP NT waters OPEP	In the event of an oil spill to the sea, the Barossa GEP NT Waters OPEP requirements are implemented to mitigate the environmental impact.	Completed incident documentat implemented as applicable.



records demonstrate ongoing inspection and Project Integrity Management Plan – Subsea.

CP surveys and inspection campaigns including the current status of the GEP.

ture monitoring will also be recorded and nce records.

ation shows the Barossa GEP NT Waters OPEP



## 8.2 Plan

The activities covered under 'Plan' describe an established and defined environmental management system, leadership commitments, and specific measures in the EMS that demonstrate how the activity will be managed and monitored to ensure that the EPOs and EPSs are met, within an ALARP performance target setting framework.

## 8.2.1 Environmental Management System

**OPGGS(E)R 2023 Requirements** 

Section 22 Implementation strategy for environment plan

Environmental management system

22(2) The implementation strategy must contain a description of the environmental management system for the activity, including specific measures to be used to ensure that, for the duration of the activity:

- (a) the environmental impacts and risks of the activity continue to be identified and reduced to a level that is as low as reasonably practicable; and
- (b) control measures detailed in the environment plan are effective in reducing the environmental impacts and risks of the activity to as low as reasonably practicable and an acceptable level; and
- (c) environmental performance outcomes and environmental performance standards in the environment plan are being met.

Santos as a titleholder and Operator of the Barossa facilities, including the GEP, is accountable for implementation and compliance with this GEP Coastal Waters OEMP. BW Offshore (BWO) is the owner of the BW Opal FPSO and provides crewing for its day-to-day operation and maintenance under a service contract with Santos.

Santos operates the Barossa facilities in accordance with an asset-specific management system: the Barossa Management System (BMS). The BMS is the environmental management system for the Activity. The BMS provides the framework and sets the mandatory requirements to manage and operate the Barossa facilities, including complying with the requirements of this GEP Coastal Waters OEMP.

As the transfer of dry gas in the GEP is reliant on the operation of the FPSO and any GEP IMMR activities are managed under the FPSO permit to work, the Barossa Management System is presented in its entirety as shown in the Barossa Production Operations EP. However, FPSO specific personnel are not directly responsible for IMMR activities in coastal waters and therefore no FPSO-specific roles are reflected in the roles and responsibility section of this GEP Coastal Waters OEMP.

## 8.2.1.1 Barossa Management System

The Barossa Management System integrates the relevant parts of the Santos and BWO management systems into a cohesive framework through the alignment of policies, standards and procedures.

Given the FPSO is provided under a service contract from BWO, the Barossa Management System was developed to leverage the BWO management system for the FPSO. The Barossa Management System comprises:

- BWO documents, procedures, and applications from the BWO management system that apply to Barossa FPSO work activities; and
- Santos documents from the Santos Management System that apply to operation of the Barossa subsea infrastructure, including the GEP, and supporting activities.

The BMS supports implementation of this GEP Coastal Waters OEMP requirements. The BMS also meets the minimum requirements of the Santos Policies and Operating Standards related to environment, health, and safety (EHS), and security. Santos as the operator of the Barossa facilities is accountable for the health and safety of all personnel, and the environmental performance of the Barossa facilities.

To ensure that all users and stakeholders of the BMS can identify and access applicable management system documentation for respective Barossa infrastructure and activities, the Barossa Management System Interface Plan serves to clearly explain the system hierarchy, scope boundaries and content of the BMS. The BMS is jointly managed by BWO and Santos and is hosted in an online location accessible to all persons engaged in Barossa related activities.

Future changes to BMS documentation and tools are managed via a change management process (Section 8.5.5). If the change under consideration results in an OEMP requirement no longer being met, or if a conflicting requirement is proposed for implementation, such changes are required to be assessed via the Environment Plan MoC process (Section 8.5.2) to determine if an OEMP revision or resubmission is required.



All field work, including subsea operations, is controlled by the FPSO Permit Process (Section 8.3.2.4), and all environmental incidents are recorded in Santos' incident management database.

The Santos Risk Matrix is the foundation for all risk assessments and management of change activities. BWO risk assessment and management of change applications are used for the FPSO, with Santos' systems applied to subsea infrastructure, including the GEP. Where a risk assessment or management of change is required to span both topsides and subsea, including the GEP, the Santos application governs the overall process. BWO risk assessment and management of change processes align with the Santos Risk Matrix. Risk management processes are further described in Section 8.3.2.1.

The BMS implements the requirements of the Barossa Production Operations EP and this GEP Coastal Waters OEMP at the Barossa facilities through the relevant, standards, procedures, processes, and tools. This includes:

- The standards, procedures, and tools for FPSO operations (out of scope of this OEMP). These standards, procedures and tools satisfy the minimum EHS requirements of Santos' Policies and Operating Standards and key procedures, and the requirements of this EP.
- The Santos standards, procedures, and tools for operating the subsea infrastructure, including the GEP.
- The Santos standards, procedures, and tools for business and supporting activities, including logistics / helicopter operations, drilling and completions activities, project management, etc.

Specific standards, procedures, processes and tools relevant to implementation of this GEP Coastal Waters OEMP , are described in the subsequent sections of the implementation strategy.

The BMS is continually improved and updated over the life of the Activity, responding to learnings from internal or industry wide incidents, changes in technology, regulations, processes, plant, and systems.

The Santos General Manager - Darwin is accountable for implementation of the BMS.

Figure 8-1 presents the key aspects of the BMS.



Figure 8-1: Key elements of the Barossa Management System

## 8.2.1.2 Environment, Health and Safety Policy

Santos' Environment, Health and Safety Policy (Appendix A) clearly sets out Santos' strategic environmental objectives and the commitment of the management team to continuously improving environmental performance. This GEP Coastal Waters OEMP has been prepared in accordance with this policy.



## 8.2.1.3 Standards, procedures and tools

The standards, processes, procedures, and tools contained within the BMS, support the implementation of control measures to achieve environmental performance outcomes and standards identified in this GEP Coastal Waters OEMP. The OEMP requirements have been mapped to relevant standards, processes, procedures and tools within the BMS, and those relevant to OEMP implementation are described in the subsequent sections of the implementation strategy. For example, processes for equipment maintenance to manage related impacts/risks of the Activity.

## 8.2.2 Leadership, accountability and responsibility

#### **OPGGS(E)R 2023 Requirements**

Section 22 Implementation strategy for environment plan

#### Responsibilities of employees and contractors

22(3) The implementation strategy must establish a clear chain of command, setting out the roles and responsibilities of employees and contractors in relation to the implementation, management and review of the environment plan, including during emergencies or potential emergencies.

## 8.2.2.1 Organisation

Figure 8-2 presents the indicative Barossa offshore organisation structure in place throughout all activities conducted under the Production Operations EP and IMMR activities relevant to this GEP Coastal Waters OEMP.



#### Figure 8-2: Barossa Offshore Organisation Structure<sup>47</sup>

The primary interface between the Santos and Campaign/ IMMR vessel is between the Santos IMMR Project Manager and the Santos Company Site Representative.

Figure 8-3 presents the indicative Barossa onshore organisation structures for the operations phase.

<sup>&</sup>lt;sup>47</sup> All organisation charts used in this OEMP are accurate at the time of submission.



Figure 8-3: Santos Onshore Organisation Structure

## 8.2.2.2 Roles and Responsibilities

The Santos General Manager - Darwin is the owner of the BMS and has accountability for implementation.

The Santos' Barossa Production Manager reporting to the Santos General Manager - Darwin, is responsible for all aspects of the Barossa facilities' performance and is accountable for ensuring compliance with all internal and external regulatory requirements, including the implementation, management and review of this GEP Coastal Waters OEMP.

The Santos' Offshore Installation Manager (OIM) is the individual (the 'Operator's representative' at the facility) who has day-to-day management and control on board of the facility, and absolute authority for the safety of the facility and all personnel on board, following connection of the FPSO to the STP Buoy. The OIM is responsible for operating the Barossa facilities in compliance with the GEP Coastal Waters OEMP, regulations and procedures.

The effective implementation of this GEP Coastal Waters OEMP requires collaboration and cooperation among Santos and its contractors. The chain of command and accountabilities of personnel in relation to implementation, management and review of this GEP Coastal Waters OEMP is outlined in Table 8-3. It is also outlined in the Barossa GEP NT Waters OPEP for spill response. All commitments in this GEP Coastal Waters OEMP will be tracked and monitored against the performance standards and the responsible role.

Table 8-3: Chain of comman	nd, key leadership	o roles and res	ponsibilities
----------------------------	--------------------	-----------------	---------------

Role	Accountabilities/Responsibilities
All Personnel	
Office-based roles and offshore roles	<ul> <li>Understand the relevant standards and procedures that apply to their area of work.</li> <li>Understand the environmental risks and control measures that apply to their area of work.</li> <li>Carry out assigned activities in accordance with approved procedures and the GEP Coastal Waters OEMP.</li> <li>Follow instructions from relevant supervisor with respect to environmental protection.</li> <li>Cease operations which are deemed to present an unacceptable risk to the environment.</li> <li>Participate in environmental assurance activities and inspections as required.</li> <li>Prompt reporting of environmental hazards/incidents to their supervisor and assist in event investigation.</li> </ul>

Role	Accountabilities/Responsibilities			
Office-based roles	Office-based roles			
Executive Vice President Western Australia (WA), Northern Australia (NA) & Timor Leste (TL)	<ul> <li>Accountable for HSE at Santos operated WA, NA, and TL facilities.</li> <li>Approval of any changes with a risk level of 'High', if acceptable and ALARP.</li> </ul>			
General Manager – Darwin	<ul> <li>Accountable for implementation of the Barossa Management System.</li> <li>Approval of any changes with a risk level of 'High'.</li> <li>Enforces compliance with local laws and ensure that regulatory requirements are maintained.</li> <li>Drives development of strategy to deliver continuous improvement in all aspects of Santos Operations.</li> <li>Ensures that positive relationships are developed and maintained with internal and</li> </ul>			
	external stakeholders.			
Production Manager – Barossa (Santos)	<ul> <li>Accountable for compliance with all internal and external regulatory requirements, including the overall management and implementation of the Barossa Management System.</li> </ul>			
	<ul> <li>Accountable for subsea processes and procedures that support this GEP Coastal Waters OEMP implementation</li> </ul>			
	<ul> <li>Responsible for leading annual GEP Coastal Waters OEMP performance reviews to review effectiveness of control measures in reducing environmental impacts and risks of the activity to ALARP and acceptable levels and identifying opportunities for continuous improvement.</li> </ul>			
	<ul> <li>Responsible for communication of any changes to the Activity that may affect the risk and impact assessment, EPOs, control measures, EPSs and MC detailed in this OEMP to the Santos HSE team.</li> </ul>			
	Responsible for all aspects of the asset's performance.			
	<ul> <li>Responsible for driving continuous improvement. Develops, monitors, and improves strategies to ensure KPIs are met or exceeded.</li> </ul>			
	<ul> <li>Accountable for managing marine vessel vetting as per the Marine Assurance Standard for field vessels as per Section 8.3.2.5 and 8.4.6.</li> </ul>			
	<ul> <li>Provides visible leadership and demonstrable commitment to the development and sustainability of Santos HSE culture.</li> </ul>			
	Accountable for implementation of the Operations management processes and plans.			
	Provides resources for HSE management.			
	Accountable for training and competency program for Santos Barossa personnel.			
	• Responsible for the implementation of risk management as per Section 8.3.2.1.			
Manager – Environment	• Provides leadership and guidance in all matters relating to environmental performance.			
(Santos)	<ul> <li>Drives the application of consistent environment culture and behaviours to ensure alignment with the overall Santos values and objectives.</li> </ul>			
	Responsible for obtaining environment-related regulatory approvals.			
	<ul> <li>Accountable for this GEP Coastal Waters OEMP compliance assurance program and activities.</li> </ul>			
	Accountable for environmental performance monitoring activities			
	<ul> <li>Responsible for analysis of environmental performance data and communication of findings to the Santos production manager</li> </ul>			
	<ul> <li>Provides implementation oversight of the environment components of the Barossa Management System.</li> </ul>			
	<ul> <li>Provides oversight and leadership in security and emergency management to ensure that adequate capability and structure is in place to respond to an oil pollution emergency as per the Barossa GEP NT Waters OPEP.</li> </ul>			
	Leads environmental incident investigations as per Section 8.4.5.			
Environmental Advisor	Responsible for the management and review of this GEP Coastal Waters OEMP.			
(Santos)	<ul> <li>Responsible for this GEP Coastal Waters OEMP compliance assurance program and activities.</li> </ul>			
	• Prepares, maintains and distributes the GEP Coastal Waters OEMP assurance register.			

Role	Accountabilities/Responsibilities
	Completes GEP Coastal Waters OEMP inspections and audits.
	Completes GEP Coastal Waters OEMP inductions and promotes general awareness.
	<ul> <li>Responsible for this GEP Coastal Waters OEMP performance monitoring, collation of associated data and records for analysis.</li> </ul>
	Contributes to environmental incident management and investigations.
	Provides operational HSE oversight and advice.
	Facilitates the development and implementation of MoC documents.
	<ul> <li>Provides incident reports, compliance reports and notifications to DME.</li> </ul>
	<ul> <li>Responsible for fulfilment of Relevant Person consultation and communication requirements.</li> </ul>
	<ul> <li>Responsible for communicating this GEP Coastal Waters OEMP requirements to subcontractors.</li> </ul>
Stakeholder Coordinator (Santos)	• Responsible for implementation of the steps described in Section 8.4.9 relating to post acceptance consultation throughout the duration of the Activity
	Maintains a Relevant Persons contact and information database
	Maintains a Relevant Persons Notification Log specific to the GEP Coastal Waters OEMP
	<ul> <li>Maintains records of all Relevant Persons correspondence specific to the GEP Coastal Waters OEMP</li> </ul>
	<ul> <li>Before the Activity begins and on advice of the Santos Barossa Environmental Adviser, notifies the Relevant Persons listed, or as revised, in Table 8-13</li> </ul>
	<ul> <li>Is available before, during and after the Activity to ensure opportunities are available for Relevant Persons to provide feedback</li> </ul>
	Prepares quarterly updates
Emergency Response	Providing overarching incident and crisis management responsibility
Advisor (Santos)	<ul> <li>Manages the Crisis Management Team and IMT personnel training program</li> </ul>
	<ul> <li>Reviews and assesses competencies for Crisis Management Team, IMT, and field based Incident Response Team members</li> </ul>
	Manages the duty roster system for Crisis Management Team and IMT personnel
	Manages the maintenance and readiness of incident response resources and equipment
Oil Spill Response Advisor (Santos)	<ul> <li>Provides ongoing guidance, framework and direction on preparing the Barossa GEP NT Waters OPEP relevant to this Activity</li> </ul>
	<ul> <li>Develops and maintains arrangements and contracts for incident response support from third parties</li> </ul>
	<ul> <li>Develops and defines objectives, strategies and tactical plans for response preparedness defined in the Barossa GEP NT Waters OPEP and IRP</li> </ul>
	<ul> <li>Undertakes assurance activities on arrangements outlined within the Barossa GEP NT Waters OPEP</li> </ul>
Manager – Engineering	Responsible for Santos Engineering Change Management Process as per Section 8.5.5
WA (Santos)	<ul> <li>Responsible for the implementation of the subsea maintenance and integrity management plan</li> </ul>
	<ul> <li>Provides engineering support and technical assurance resources to subsea, through MoC processes and risk assessments as required.</li> </ul>
Santos Construction and Fixed Assets	<ul> <li>Responsible for assigning a suitable IMMR Project Manager to complete the activities in accordance with this GEP Coastal Waters OEMP</li> </ul>
Programme Lead – WA, NA & TL	<ul> <li>Accountable for ensuring the IMMR Activity has sufficient resources to execute the activities in accordance with this GEP Coastal Waters OEMP</li> </ul>
	Responsible for coordination of Subsea IMMR Project Managers across Santos Assets
	<ul> <li>Responsible ensuring compliance with the systems and processes used by Santos to execute the activities in accordance with this GEP Coastal Waters OEMP</li> </ul>
	<ul> <li>Responsible for collating and reviewing the reporting and KPIs associated with the execution of IMMR activities across Santos Assets (including environmental incidents)</li> </ul>
	<ul> <li>Responsible for collating the IMMR budget and delivering the scope across the Santos subsea assets in conjunction with the subsea integrity engineering team</li> </ul>

Role	Accountabilities/Responsibilities	
Santos Manager Maintenance – WA, NA	<ul> <li>Accountable for ensuring that there are sufficient IMMR Project Manager resources to execute subsea IMMR activities across Santos Assets</li> </ul>	
& TL	<ul> <li>Accountable to Santos Production Manager for the reporting and KPIs associated with the execution of IMMR activities across Santos Assets (including environmental incidents)</li> </ul>	
	<ul> <li>Accountable for delivering the IMMR budget across the Santos subsea assets to each Asset Production Manager (budget owner) in conjunction with the subsea integrity engineering team lead.</li> </ul>	
Santos IMMR Project Manager	<ul> <li>Accountable for managing marine vessel vetting for vessel required for GEP IMMR as per hierarchy of procedures in Section 8.3.2.5</li> </ul>	
	<ul> <li>Accountable for inclusion of the HSE exhibit in contract tender documents, and evaluation of tenders against this GEP Coastal Waters OEMP requirements</li> </ul>	
	<ul> <li>Accountable for implementation of IMMR activities in accordance with this GEP Coastal Waters OEMP</li> </ul>	
	<ul> <li>Accountable for communicating any changes to the Activity that may affect the risk and impacts assessment, EPOs, EPSs and MC detailed in this GEP Coastal Waters OEMP to the Santos HSE team</li> </ul>	
	<ul> <li>Responsible for providing the resources required to enable the commitments in this GEP Coastal Waters OEMP to be maintained</li> </ul>	
	<ul> <li>Accountable for confirming the reporting of environmental incidents meets both external and Santos incident reporting requirements</li> </ul>	
	<ul> <li>Responsible for liaising with Santos Environmental Advisor on environmental incidents and what constitutes a reportable incident</li> </ul>	
	<ul> <li>Accountable for tracking and closing out of any corrective actions raised from environmental audits as required by this GEP Coastal Waters OEMP</li> </ul>	
	<ul> <li>Accountable for the Control of Work process and procedures onboard IMMR vessels as per Section 8.3.2.4</li> </ul>	
Santos IMMR Contractor Project	<ul> <li>Responsible for implementing the IMMR activity in accordance with this GEP Coastal Waters OEMP</li> </ul>	
Manager	<ul> <li>Responsible for providing the resources required to enable the IMMR commitments as relevant to this GEP Coastal Waters OEMP to be implemented</li> </ul>	
	• Responsible for biosecurity assurance for all activity vessels mobilised to the OAs (Section 8.3.2.9).	
	Responsible for all crew attending HSE inductions and saving attendance records	
	Responsible for reporting and investigating incidents, as required	
Santos Aviation Specialist	<ul> <li>Responsible for approving aircraft operators and aircraft types in accordance with Aviation Procedure as per Section 8.3.2.6</li> </ul>	
Offshore roles		
Santos Barossa Offshore Installation	<ul> <li>Person in charge on board the FPSO (noting the FPSO is out of scope of this GEP Coastal Waters OEMP).</li> </ul>	
Manager (OIM)	<ul> <li>Person in charge for any matter which may affect the safety of people, cause damage to the Barossa facilities or the environment.</li> </ul>	
	<ul> <li>Responsible for compliance with control measures identified in this GEP Coastal Waters OEMP to achieve EPOs and EPSs.</li> </ul>	
	<ul> <li>Responsible for implementation of supporting processes described in the GEP Coastal Waters OEMP implementation strategy e.g. Santos chemical management</li> </ul>	
	Acts as Emergency Commander during emergency response situations.	
	<ul> <li>Responsible for all works meeting the requirements of the Permit To Work (PTW) system (Section 8.3.2.4).</li> </ul>	
	<ul> <li>Responsible for ensuring the Barossa facilities are maintained as per the Maintenance Management System.</li> </ul>	
	<ul> <li>Responsible for operating the Barossa facilities in compliance with the GEP Coastal Waters OEMP, regulations and procedures.</li> </ul>	
	<ul> <li>Responsible for ensuring personnel on the facility are suitably trained and competent for their roles and duties.</li> </ul>	
	<ul> <li>Promotes an HSE culture amongst offshore crew.</li> </ul>	
	Responsible for investigating and reporting all incidents in a timely manner.	

Role	Accountabilities/Responsibilities				
Santos Company Site representative (IMMR	<ul> <li>Confirms contractors undertake the Activity in a manner consistent with this GEP Coastal Waters OEMP</li> </ul>				
and Vessel activities)	<ul> <li>Confirms the management measures detailed in this GEP Coastal Waters OEMP are implemented</li> </ul>				
	<ul> <li>Confirms that the Vessel Master and all crew adhere to the requirements of this GEP Coastal Waters OEMP</li> </ul>				
	<ul> <li>Advises the Santos IMMR Project Manager of any activity changes that may lead to a non-conformance with the requirements with this GEP Coastal Waters OEMP</li> </ul>				
	Reports environmental incidents to the IMMR Project Manager				
	<ul> <li>Management of offshore greenhouse gas emission sources.</li> </ul>				
Contractor Vessel Masters (IMMR and	Responsible for compliance with all HSE laws, conventions and approvals (such as safety case)				
Vessels)	<ul> <li>Responsible for conformance with the GEP Coastal Waters OEMP as relevant to vessel activities</li> </ul>				
	Reports any new, or increase in, HSE risk or impact				
	Responsible for compliance with MoC procedures				
	Empowers personnel to 'stop the job' due to HSE concerns				
	<ul> <li>Responsible for compliance with reporting requirements for all HSE incidents, hazards and non-conformances</li> </ul>				
	<ul> <li>Facilitates HSE investigations and is responsible for the implementation of corrective actions</li> </ul>				
	<ul> <li>Responsible for compliance with requirements for crew to be competent and prepared to respond to HSE incidents</li> </ul>				
HSE Advisers (Santos and/or contractor)	<ul> <li>Responsible for supporting the Santos OIM and/or Senior Client Site Representative to implement the requirements within this GEP Coastal Waters OEMP and assisting to collect and record environmental assurance evidence</li> </ul>				
	<ul> <li>Responsible for supporting the Santos OIM and/or Senior Client Site Representative to report environmental incidents or breaches of outcomes or standards outlined in this GEP Coastal Waters OEMP and develop, track and close out corrective actions for incidents and breaches are in a timely manner</li> </ul>				
	<ul> <li>Responsible for completing periodic environmental inspections/reviews, including waste audits, and developing, tracking and closing out corrective actions from inspections are in a timely manner</li> </ul>				
	<ul> <li>Responsible for reviewing contractors' procedures and providing input into toolbox talks and job safety analyses</li> </ul>				
	<ul> <li>Responsible for providing day-to-day environmental support for activities in consultation with the Santos Barossa Environmental Adviser</li> </ul>				

## 8.2.3 Adverse weather preparedness

Adverse weather preparedness is described in Section 8.2.3 of the Barossa Production Operations EP and is out of scope of this GEP Coastal Waters OEMP. The FPSO Adverse Weather Plan outlines operational precautions leading up to and during a cyclone, which may impact the transport of dry natural gas from the Barossa field through the GEP in Coastal Waters. Requirements for resuming operations post adverse weather events are also described in the FPSO Adverse Weather Plan.

The Barossa Nearshore Gas Export Pipeline Operations Pipeline Management Plan (of relevance to this OEMP) outlines requirements for inspecting the GEP post cyclone events.

IMMR vessels working in the cyclone season would have their own adverse weather plan.

## 8.2.4 GHG Emissions Management

The GHG emissions management for the Barossa Gas Project are considered on the basis of all the facilities including the GEP. Estimates for GHG and atmospheric emissions included in the Barossa Production Operations EP and this OEMP are the basis for evaluation of impacts and risks from the Activity and represent a threshold that if exceeded would trigger an EP MoC assessment (Section 8.5.2).

In demonstrating that risks and impacts relating to Barossa Gas Project's GHG and atmospheric emissions are reduced to ALARP over the life of the Activity, targets are set annually and progress tracked monthly and annually, before new annual targets are set. In support of reducing emissions to ALARP, emissions performance targets are

also designed to drive continuous improvement in emissions performance. Figure 8-4 shows the emissions target setting cycle for the Barossa Gas Project during steady state operations.



Figure 8-4: Emissions Target Setting Cycle

Inputs to setting the performance targets consider:

- EP emissions estimate inputs and assumptions
- Operating experience
- Forecast activities planned and contingency

Annual performance targets require approval of both the relevant Santos Environment Manager for the Barossa asset, and the Barossa Production Manager.

Performance targets will be tracked on a monthly frequency as part of Operations Governance Forums (Section 8.5.3.1), which include separate forums with representatives from both Asset leadership and Regional Business Unit leadership. Operations Governance Forums will track both leading and lagging indicators of relevance to performance targets e.g. emissions control equipment performance and reliability and will consider both year to date performance and forecast performance against targets. In reviewing monthly performance, emissions improvement initiatives are considered. Where improvement opportunities are identified and endorsed for further evaluation, they are managed per the decarbonisation opportunity management process (Section 8.5.6). Operations governance KPI dashboards are shared with the workforce for awareness, and to solicit feedback about improvement opportunities (Section 8.3.1.1.2).

If deviations against annual performance targets are forecast to occur, an internal performance target deviation is developed, which requires an ALARP justification and approval from both the relevant Environment Manager and Barossa Production Manager. If the emissions estimates are likely to be exceeded, a management of change assessment (Section 8.5.2) is undertaken to determine if a revision and resubmission is required.

A baseline facility "energy efficiency" target will be established after the first year of steady state operations to inform monthly tracking of energy efficiency performance, and potential improvement opportunities. Efficiency improvement opportunities will be managed via the Barossa decarbonisation opportunity management process (Section 8.5.6).

At the end of the 12-month cycle, performance against targets and emissions estimates is reviewed, continuous improvement opportunities and adaptive management measures considered, and assumptions and targets are reviewed and re-set for the next 12-month cycle.

The majority of Barossa Gas Project direct (Scope 1) emissions are associated with FPSO processing activities and control measures to address associated impacts and risks can be found in the Production Operations Environment Plan in Section 8.2.4. Methane emissions management for Scope1 emissions over the life of the



Barossa Gas Project is addressed in Section 8.3.2.11. Control measures to address Activity GHG emissions for this OEMP are addressed in Section 6.3 of the OEMP.

## 8.2.5 Santos Decommissioning Strategy

Decommissioning lifecycle planning covers:

- Decommissioning considerations for new facilities or modifications to existing facilities (Appraise, Select and Define).
- Decommissioning and maintenance considerations during facility Operate phase facility management.
- Managing decommissioning opportunities when preparing for End of Field Life (EOFL) and surrender of production license/lease.
- Development and maintenance of restoration cost estimates for use in financial provisioning.

The Barossa GEP infrastructure in the GEP Coastal Waters OEMP is planned to be fully utilised over the lifecycle of the Barossa Gas Project, which is expected to be approximately 25 years. While no Activity infrastructure is planned to be decommissioned as part of this GEP Coastal Waters OEMP, all infrastructure has been selected and designed to allow for removal when no longer used or to be used.

As part of Santos' assets life cycle management requirements, decommissioning execution strategies will be matured throughout the life of the project. Santos' decommissioning strategy is to manage all equipment over the life cycle of the activity to facilitate removal at the time of decommissioning through appropriate design, inspection and maintenance practices.

Decommissioning and removal of property with no further use prior to end of field life, will be addressed under the future decommissioning plan. The future decommissioning plan for infrastructure within the OA will meet all applicable regulatory requirements in force at the relevant time. Santos decommissioning strategy is described below.

#### 8.2.5.1 Decommissioning in operations

Decommissioning planning forms part of the strategic planning for how an operational asset is effectively and efficiently managed throughout the Operate phase in accordance with the Asset Long Term Plan (LTP) and Asset Reference Plan (ARP) for each Santos operated asset.

### 8.2.5.2 Facility decommissioning planning

Decommissioning planning and costing start early during field development to ensure development decisions account for decommissioning strategy and costs. In accordance with the Santos Opportunity Development Process, decommissioning philosophies must be identified, and for high risk/complexity and high opportunity cost assets, considered during selection of a preferred concept.

During Field Development Planning for any opportunity, consideration must be given to the means of decommissioning and how this may impact on the design, key decisions, and economics. The regulatory environment may change during the life of an asset; therefore, all equipment must be designed for the base case of full removal and (in the case for offshore developments) onshore disposal. Further guidance on typical expectations for offshore developments can be found in the <u>Australian Government's offshore petroleum</u> decommissioning guideline and NOPSEMA's information paper on <u>Planning for proactive decommissioning</u>.

Santos Decommissioning Project Delivery Process can be seen presented within Figure 8-5.



#### Figure 8-5: Santos Project Delivery Process for Decommissioning

### 8.2.5.3 Barossa decommissioning planning

Barossa will have a documented decommissioning strategy or plan that considers all obligations, including regulatory requirements, Joint Venture agreements, timing constraints and credible cost estimates, at all stages of it's lifecycle:

- Decommissioning planning will be aligned with Project Delivery Process requirements (Figure 8-5)
- Financial provisioning for decommissioning will be underpinned by reasonable cost estimates and comply with the relevant Accounting Practice. The basis of the cost estimate will be in compliance with the regulatory requirements of the relevant jurisdiction governing the project. Timing assumptions for the realisation of decommissioning liabilities shall be in accordance with the governing regulatory requirements
- Regulator engagement and notification will be maintained at a level of detail and frequency suitable to the life cycle stage and regulations applying to the asset. Santos holds regular decommissioning update meetings with relevant regulators to update them on planning progress across all assets.

The asset systems identified as key for decommissioning are designed to facilitate infrastructure flushing, cleaning, and decommissioning processes, and will be maintained as required. These systems will continue to operate in compliance with standard IMMR protocols until the conclusion of the operational lifespan. To ensure that the systems used in conjunction with operations are adequately maintained throughout their operational life, these requirements will be incorporated into the relevant operational documentation.

#### 8.2.5.3.1 Subsea infrastructure decommissioning

To satisfy future decommissioning obligations, all subsea infrastructure has been designed to be feasible to remove. Detailed decommissioning documents, plans and procedures will be produced as detailed within **Figure 8-5**., while prioritising optimal environmental outcomes and the latest technological advancements available at that time.

#### 8.2.5.3.2 Gas export pipeline decommissioning

The Barossa infrastructure has been designed and will be installed and operated so that it is feasible to remove, this includes the GEP. Design features and maintenance plans for the GEP, which allow removal to occur at the end of field life, are detailed in Table 8-4.



## Table 8-4: Design features and maintenance plans to enable removal of the gas export pipeline and ancillary infrastructure at decommissioning

Infrastructure	Key elements to facilitate maintenance & removal
Gas Export Pipeline • 26" x 8.265km from KP 23 (NT/Commonwealth boundary) to KP 31.265 (Territorial sea Baseline)	<b>Design:</b> Cathodic protection system provides design life (25yr) protection. <b>Maintenance</b> : Risk-based inspection regime to monitor system integrity. <b>Removal:</b> Two options considered: reverse S-lay and pipeline cutting and lifting for retrievals
Grout bags	<b>Design:</b> Designed to maintain integrity for design life (25 year). <b>Maintenance</b> : Risk-based inspection regime to monitor system integrity. <b>Removal:</b> Retrieval by ROV and ROV basket.

## 8.2.6 Emergency preparedness and response

#### **OPGGS(E)R 2023 Requirements**

#### Section 22 Implementation strategy for environment plan

#### Oil pollution emergency response

22(8) The implementation strategy must contain an oil pollution emergency plan and provide for the updating of the plan. 22(9) The oil pollution emergency plan must include adequate arrangements for responding to and monitoring oil pollution,

- 22(9) The oil pollution emergency plan must include adequate arrangements for responding to and monitoring oil pollution, including the following:
  - the control measures necessary for timely response to an emergency that results or may result in oil pollution;
  - the arrangements and capability that will be in place, for the duration of the activity, to ensure timely implementation of the control measures, including arrangements for ongoing maintenance of response capability;
  - the arrangements and capability that will be in place for monitoring the effectiveness of the control measures and ensuring that the environmental performance standards for the control measures are met;
  - the arrangements and capability in place for monitoring oil pollution to inform response activities.

22(10) The implementation strategy must provide for monitoring of impacts to the environment from oil pollution and response activities that:

- is appropriate to the nature and scale of the risk of environmental impacts for the activity; and
- is sufficient to inform any remediation activities.

22(11) The implementation strategy must include information demonstrating that the response arrangements in the oil pollution emergency plan are consistent with the national system for oil pollution preparedness and response.

#### Testing oil pollution emergency response arrangements

22(12) The implementation strategy must include arrangements for testing the response arrangements in the oil pollution emergency plan. The testing arrangements must be appropriate to the response arrangements and to the nature and scale of the risk of oil pollution for the activity.

22(13) The testing arrangements must include:

- a statement of the objectives of testing; and
- a proposed schedule of tests; and
- mechanisms to examine the effectiveness of response arrangements against the objectives of testing; and
- mechanisms to address any recommendations arising from tests.
- (14) For the purposes of paragraph (13)(b), the proposed schedule of tests must provide for the following:
  - testing the response arrangements when they are introduced;
  - testing the response arrangements when they are significantly amended;
  - testing the response arrangements not later than 12 months after the most recent test;
  - if a new location for the activity is added to the environment plan after the response arrangements have been tested, and before the next test is conducted-testing the response arrangements in relation to the new location as soon as practicable after it is added to the plan;
  - if a facility becomes operational after the response arrangements have been tested and before the next test is scheduled to be conducted-testing the response arrangements in relation to the facility when it becomes operational.

Vessels are required to have and implement incident response plans, such as an emergency response plan and SMPEP or SOPEP. Regular incident response drills and exercises – for example, as defined in an emergency



response plan, SMPEP or SOPEP – are performed to refresh the crew in using equipment and implementing incident response procedures.

Santos will implement the Barossa GEP NT Waters OPEP in the event of a hydrocarbon spill. The Barossa GEP NT Waters OPEP details how Santos will prepare and respond to a spill event and meet the requirements of the OPGGS(E)R 2023.

### 8.2.6.1 Emergency response overview

The Santos Crisis, Incident and Emergency investigation, defines the requirements for emergency response preparedness for all facilities, including:

- Emergency Response Plans (ERPs) are in place for credible operational risks and scenarios with trained Emergency Response Team (ERT) personnel and equipment / resources in place to execute the plans.
- An Incident Management Plan (IMP) and trained Incident Management Team (IMT) are in place to provide operational support for escalating emergencies.
- A Crisis Management Plan (CMP) and trained CMT are in place to manage incidents or issues with significant, strategic implications for the organisation.
- An annual schedule of exercises is developed and executed to test effectiveness of the plans and to build and maintain response capability.

Emergency management for the Barossa facilities follows a three level response framework as presented in Table 8-5.

An ERP, IMP and CMP are in place which define the requirements for emergency response preparedness. These plans apply for the GEP.

A trained ERT, IMT and CMT are in place to execute these response plans and provide support during emergencies.

Response Level	Description
Level 1 Emergency Response	Incidents that can be controlled by using resources normally available at the facility concerned, without the need to mobilise the IMT or other external assistance. This level would include sending personnel ashore for medical examination of a non-emergency nature. Generally, a Level 1 event is managed by the ERT.
	A Level 1 event is escalated to Level 2 at any time support is required from a Business Division IMT or external response agencies.
Level 2 Incident Management	Incidents that cannot be controlled by the facilities resources alone and require external support and resources to address the situation. PERTH IMT and/or Santos Support Teams / Oil Spill Response Team will be activated to provide operational and technical resources management and liaise with authorities and mutual aid organisations. Generally, a Level 2 event is managed by the IMT.
Level 3 Crisis Management	Incidents which have a wide-ranging impact on the company's strategic and enterprise interests and/or crisis status, warranting establishment of the Santos CMT. A large scale, sustained response requires the engagement of significant external resources and response vendors. A level 3 event may not have resulted from, or result in, ERT or IMT activation and can be a standalone event. Generally, a Level 3 event is managed by the CMT.

#### Table 8-5: Emergency response levels

#### 8.2.6.2 Emergency response

The ERP addresses the credible operational emergency risks and scenarios for the GEP. The ERP has been developed in accordance with NOPSEMA's Emergency Planning Guidance Note, and Safe Work Australia's Guidelines for Emergency Plans. The plan includes incident management guides for the identified emergency scenarios, including emergency scenarios with potential significant environmental consequences (such as a potential vessel collision). These guides assist the facility emergency response team by describing the actions to be undertaken during emergency scenarios, including initial response to an incident, and follow on actions.

A trained ERT are the designated FPSO based emergency responders who respond to tactically manage an emergency event / incident. The ERT are directed by the Emergency Commander and coordinated locally by the ERT Leader (On-Scene Commander). The Emergency Commander is supported by the Emergency Command Team who operate at the Emergency Command Centre / CCR. In the event of an emergency the OIM assumes the role of the Emergency Commander and is responsible for the implementation and supervision of procedures in the event of an incident.



The ERT roles and responsibilities are outlined in the ERP. Incumbent and back up personnel are also identified for each ERT role. Facility crewing is commensurate with the ERP / ERT requirements.

### 8.2.6.3 Incident management

Level 2 incident management is enacted for incidents / emergencies that cannot be controlled by the facilities resources alone and require external support and resources to address the situation (i.e. Level 2 incidents / emergencies). A trained IMT are in place to provide operational support for escalating incidents / emergencies with a focus on:

- Managing / supporting the incident / emergency overall.
- People, environment, assets and recovery.
- Operational continuity and recovery.
- Divisional stakeholder management.

The WANATL IMP establishes Santos' incident management arrangements to:

- Guide the IMT in emergency preparedness, emergency response and operational recovery.
- Support site / facility ERTs during emergencies.
- Undertake incident action planning to manage the consequences of an emergency event.
- Ensure WANATL incident management preparedness.

The IMP interfaces with the ERP and the Barossa GEP NT Waters OPEP.

The IMP also provides incident scenario guides which provide guidance on potential IMT tasks associated with various potential emergency scenarios, including loss of containment, fire/explosion, vessel incidents and severe weather / natural disasters.

The IMT manages the operational consequences and impacts of an emergency event. The IMT are led by the IMT Lead, who provides leadership to:

- Ensure support is provided to the affected site / facility / field operations during actual / potential emergencies.
- Ensure the safety and protection of the company's people, environment, assets and reputation.
- Minimise operation production and business interruption.
- Determine and mitigate the operational risks arising from the emergency.
- Lead operational and business recovery.

The Incident Management Manual incudes duty cards which define each IMT role including their role, responsibility and specific actions to be performed during an incident.

### 8.2.6.4 Crisis management

The CMP and trained CMT are in place to manage incidents or issues with significant, strategic implications for the organisation. The CMP defines:

- Criteria and procedure for activating the plan (both incident and issue triggers), including CEO consultation.
- Roles and responsibilities of the CMT.
- CMT selection and training requirements.
- Crisis Management Process and the CMT response objectives, actions, processes, and tools.
- Internal and external stakeholder communication plans and protocols.
- Strategies for managing a long term crisis.
- Criteria and procedure for calling an end to the crisis and standing down the CMT.

The CMT is convened for the purpose of managing crisis events (Level 3 emergencies / incidents / events) The CMT is comprised of executive leadership, governed by the Crisis Management Chair (CEO or delegate). The Crisis Management Lead (appointed by the Crisis Management Chair) leads the CMT. The CMT:

• conducts crisis management operations, including assessing the crisis event and setting clear objectives to manage and recover from the crisis;



- directs the enterprise level strategic response; and
- addresses strategic implications and outcomes.

Support Teams support the CMT, when warranted, as business function subject matter experts. The Support Team functions will focus on strategies and actions set by the CMT.

#### 8.2.6.5 IMMR campaign emergency response

Contracted IMMR vessels shall have a vessel specific ERP covering vessel emergencies. This typically includes major emergencies such as vessel collision, loss of vessel stability, etc. The plan shall comply with the relevant regulatory bodies requirements (e.g. AMSA, Flag State and DME).

Bridging documentation shall also be prepared providing campaign-specific guidance, including explaining the emergency response interfaces between Santos and the IMMR vessel contractor, and the process for providing a coordinated response for emergencies at the Barossa facilities.

## 8.3 Do

The activities under 'Do' describe competency and training processes relevant to implementation of the GEP Coastal Waters OEMP, and specific measures relevant to implementation of the requirements of the environmental plan.

## 8.3.1 Workforce EP Awareness and Training

OPGGS(E)R 2023 Requirements

Section 22 Implementation strategy for environment plan

Responsibilities of employees and contractors

22(4) The implementation strategy must include measures to ensure that each employee or contractor working on, or in connection with, the activity is aware of the employee's or contractor's responsibilities in relation to the environment plan, including during emergencies or potential emergencies, and has the appropriate competencies and training.

This section describes the mechanisms that will be in place, so each employee and contractor is aware of his or her responsibilities in relation to the GEP Coastal Waters OEMP and has appropriate training and competencies.

### 8.3.1.1 Workforce EP Awareness

#### 8.3.1.1.1 Inductions

The induction process is in place to ensure that all personnel associated with, working on or visiting the Barossa facilities are provided the necessary awareness, knowledge and competence appropriate for their role, to ensure that personnel are aware of the environmental impacts and risks associated with the facilities and the environmental management requirements arrangements described in this GEP Coastal Waters OEMP.

All new personnel to the Barossa facilities must complete several inductions prior to commencement of work duties, including company, facility and activity specific awareness induction packages (depending on role). The induction will include an OEMP awareness component, which covers environmental management commitments and requirements to be implemented as described in this GEP Coastal Waters OEMP, tailored to the environmental impacts/risks of each activity phase.

The GEP Coastal Waters OEMP awareness training for vessel-based activities includes training for:

- Overview of Santos' Environment, Health and Safety Policy (Appendix A) and the Barossa Management System
- the applicable environment legislative requirements
- environmental values and sensitivities relevant to the Barossa facilities and Operational Areas (such as protected marine areas, sensitive environmental periods)
- maritime underwater cultural heritage and First Nations cultural heritage awareness
- key environmental hazards and control measures as relevant (including but not limited to GHG emissions, atmospheric emissions, waste management, spill prevention, chemical storage, fauna interactions and vessel speeds and interactions with other marine users)
- hazard and incident reporting and notifications
- regulatory compliance reporting (including fauna interactions)



- oil pollution emergency response requirements (such as the Barossa GEP NT Waters OPEP).
- where to locate copies (paper and digital) of the GEP Coastal Waters OEMP and GEP NT Waters OPEP
- records of induction attendance will be maintained.

For key leadership roles associated with the Activity, a focused environmental awareness package will be provided describing key responsibilities under this GEP Coastal Waters OEMP.

#### 8.3.1.1.2 Workforce involvement and communication

Daily operational meetings will include environmental management as a standing agenda item. It is a requirement that supervisors attend daily operational meetings and all personnel attend daily toolbox or pre-shift meetings. Toolbox or pre-shift meetings will be held to plan jobs and discuss work tasks, including environmental risks and their control measures.

Environmental performance will be monitored and reported during the Activity by the Environment Manager, and performance metrics (such as the number of environmental incidents) will be regularly communicated to the workforce via the daily operational meetings. The Operations Governance Forum (Section 8.5.3.1) KPI dashboard (inclusive of environmental performance KPIs) will be shared with the workforce during the Activity where applicable via HSE Information Feedback meetings , with members of the workforce encouraged to contribute opportunities for improvement to be considered/evaluated at monthly Operations Governance Forums (Section 8.3.1.1.2).

Workforce involvement and environmental awareness will also be promoted by reminding offshore personnel about obligations communicated during environmental awareness inductions (Section 8.3.1.1.2) to report marine fauna sightings and marine pollution; for example, oil sheens on water, dropped objects.

#### 8.3.1.1.3 Operations Environmental Management Plan Compliance Registers

The key mechanism for ensuring all personnel involved in the operation of the Barossa facilities across all petroleum activity phases in this Activity are aware of the environmental control measures and performance standards adopted in this GEP Coastal Waters OEMP is via the provision of OEMP compliance registers. Each register is tailored to the requirements of each activity phase. Responsibilities are assigned for each OEMP compliance requirement, consistent with the roles and responsibilities in Section 8.2.2.2, as relevant to each activity phase. The Environment Advisor is responsible for preparing, maintaining and distributing the OEMP Compliance registers. The registers will be updated as required following OEMP revisions.

Santos is responsible for ensuring all personnel are aware of the environmental control measures and performance standards as adopted in the GEP Coastal Waters OEMP. The OEMP assurance compliance register will be incorporated into Santos systems.

For vessel-based activities, the GEP Coastal Waters OEMP compliance register as relevant to vessel based activities will be used by the Vessel Master for ensuring the environmental control measures and performance standards relevant to the activity are implemented.

#### 8.3.1.1.4 Contractor health, safety and environment requirements

The minimum requirements and expectations for HSE management of contractors and subcontractors is described in the HSE Contractor Management Procedure, including a specific contractual HSE exhibit for project specific scopes of work. The HSE exhibit has a detailed environmental requirements section for:

- review of GEP Coastal Waters OEMP requirements applicable to the contract scope
- the contractor to comply with the control measures specific to the contract scope, and identify any new environmental risks and propose additional control measures to reduce the risk to ALARP. The Santos Environment MoC assessment (refer Section 8.5.2) is triggered if there are changes to the Activity and associated environment risk, as assessed in the accepted EP.
- understanding and compliance with applicable environmental legislation
- · key activities to support continuous environmental improvement
- chemical selection and approvals
- prohibited materials and chemicals (if/as applicable)
- vessel requirements (if relevant to the scope) to comply with the GEP Coastal Waters OEMP.

#### 8.3.1.2 Workforce Training and Competency

All members of the Barossa facilities workforce will complete relevant training and hold qualifications and certificates for their role. Santos and its contractors are individually responsible for ensuring their personnel are



qualified and trained. The systems, procedures and responsible persons will vary and will be managed through the use of online databases, staff onboarding process, training departments and other means.

Personnel qualification and training records will be sampled before and during an activity. Such checks will be performed during the procurement process, facility acceptance testing, inductions, crew change, and operational inspections and audits. Ongoing checks will be undertaken to check and verify training needs are being identified and training undertaken to maintain a competent workforce.

Table 8-6 describes the standard, procedures and plans that manage the training and competence of personnel at the Barossa facilities to ensure they can effectively and safely perform their work in order to ensure compliance and regulatory obligations are met.

The training may be provided through learning modules, with vendor training programs and accredited training provided by external training providers as required, as well as on the job training / job shadowing and coaching.

The Santos Manager of Training and Competency Assurance (Operations and Maintenance) is responsible for the training and competency of the Santos workforce undertaking IMMR activities associated with the GEP

As per the Santos Competency and Training Technical Standard, records of competency requirements and the employee's assessment in relation to those requirements will be retained in the Learning Management System (LMS) enabling tracking, monitoring and reporting.

#### Table 8-6: Training Standard, Procedures and Plans

Title	Description
Competency and Training Technical Standard	Ensures appropriate and relevant training is provided to personnel so they are able to effectively perform their work; and compliance and regulatory obligations are met. The Technical Standard requires the assessment of organisational and individual capability via competency frameworks, training needs analysis and training delivery (including frequency of training/ competency checks) through the core curriculum programme or individual / group training.
Competency Framework Procedure	The framework delivers standardised competency-based training and an assessment curriculum that provides fit-for-purpose training and assessment to confirm personnel are competent to perform work against established and pre-defined performance. The process includes competency-based assessment undertaken by assessors with nationally accredited qualification and includes assessment by a combination of: <ul> <li>Attendance at a formal training course.</li> <li>Observation during work activities.</li> <li>Questioning (oral, written or online).</li> </ul>
	Collection of historical evidence.
Competency Assurance Plan	This plan ensures that all contract personnel are trained and assessed as competent to carry out their assigned tasks in line with environmental performance requirements in this OEMP, including the required training and certification standards which apply for core crew on the Barossa facilities, to ensure personnel have the necessary level of competence to conduct the work in the operation, design, maintenance, and emergency response for the Barossa facilities. Training records for all crew are maintained and available to the OIM and key Santos operations personnel onshore. To meet the requirements of Australian legislation / regulations, the Santos Competency Framework Procedure, a matrix has been developed that defines the training and competency requirements considered either mandatory or beneficial for each role. For example roles that are responsible for emissions or discharges critical equipment will identify the required competencies in operating, inspecting, maintaining or using equipment that is critical to achieve environmental performance outcomes and standards as defined in this OEMP. Personnel mobilised on vessels involved vessel based activities will be subject to competency testing criteria, including site-specific competency assessment, verification of certificates and work experience (as required), as per the contractor management and marine assurance processes.
Competency Verification	The process for verifying competence. Competence is assessed by questioning and observation during work activities, collecting historical evidence of competence (including evidence of accredited / recognised prior learning, applicable licences, records of performance, etc), and attendance at formal training courses.

For IMMR campaigns, the training and competency requirements are typically documented in the HSE Management Plan developed for each specific IMMR campaign. All personnel mobilised on an IMMR vessel are subject to competency testing criteria, including site-specific competency assessment, verification of certificates and work experience.

Emergency preparedness and response (including for oil spill response) training requirements are addressed in Section 8.3.1.3.



Members of the workforce that have been allocated roles within the PTW system are required to complete the training module relevant to their role. The PTW system training requirements for the various roles on the Barossa facilities are defined in the training and competency matrix, and includes delivering training on the PTW system structure, permit planning and preparation, permit coordination, authorisation and arrangement, permit execution, revalidation and closing, risk assessments, mechanical (process) isolations.

### 8.3.1.3 Emergency response training, drills and exercises

All personnel appointed to emergency response roles undergo suitable training and participate in preparedness activities and exercises to enable them to effectively perform their roles. All positions that are required to have specific emergency response training are identified. The minimum training requirements for key roles are presented in Table 8-7.

Other members of the Emergency Management Team shall undertake PMAOMIR322 'Manage Incident Response Information' (or an equivalent training course), with members of the ERT also trained as appropriate for their role.

The Drills and Exercises Procedure, describes the facility emergency response drills and exercises. The drills and exercises are based on the identified emergency scenarios, and include liaison with, and involvement of, external response organisations and other operators where appropriate.

The minimum emergency exercise requirements are:

- Annual IMT exercise with facility participation.
- Annual Level 3 oil spill exercise involving external parties—the Santos facility involved in this exercise is dependent on the worst case scenario defined in the Santos operated facilities OPEPs.

Reviews are conducted after each exercise or activation to identify any opportunities for improvement or sharing of learnings (including good practices).

Additional training and competency requirements for relevant personnel specific to spill response are provided in the Barossa GEP NT Waters OPEP.

- An overview of Santos hydrocarbon spill response training and competency requirements are provided in dashboards for key responder roles. The roles are consistent with Santos' crisis and emergency management incident control structure.
- Santos' Oil Spill Response Advisor(s) are responsible for maintaining hydrocarbon spill preparedness competency. This includes the identification and development of approved competency and noncompetency-based courses, identification of relevant personnel required to undertake training, and ensuring training records are maintained. Minimum Santos capabilities will continue to be identified and documented.

Table 8-7: Emergency re	esponse training
-------------------------	------------------

Role	Minimum training requirements		
Emergency Commander (EC)	<ul> <li>PMAWHS511 Manage Emergency Incidents</li> <li>PMAOMIR322 Manage incident response information</li> <li>Oil Spill Familiarisation (AMOSC)</li> </ul>		
IMT Leader	PMAOMIR418 'Coordinate Incident Response'		

### 8.3.2 Supporting Management Processes and Procedures

This section describes the 'specific measures' of the Barossa Management System that define how the activity will be managed and monitored to ensure that the EPOs and EPSs are met.

### 8.3.2.1 Risk Management

Santos operates under an overarching Risk Management Policy that applies to this Activity. The company Risk Management, Investigation and Assurance operating standard underpins the Risk Management Policy and is consistent with the requirements of AS/NZS ISO 31000:2018, Risk Management – Guidelines (ISO, 2018). Risk management processes are implemented on an ongoing basis during the Activity to manage risks to personnel, the Barossa facilities and the environment. The key steps to risk management are illustrated in Figure 5-1.

The forum used to undertake environment risk assessment is the environmental hazard workshop, referred to as an ENVID, which is described in Section 4 of Santos' Offshore Division Environmental Hazard Identification and Assessment Guideline, and summarised in Section 5.2.



## 8.3.2.2 Asset Integrity Management

The Subsea Integrity Management Procedure sets out the requirements for managing the integrity of the Barossa subsea equipment. The procedure requires that Integrity Management Plans are developed which review and verify credible damage and failure mechanisms identified in the design phase and as part of ongoing operational and integrity reviews.

RBI programs are developed to ensure that maintenance of the subsea equipment is appropriately prioritised, ensuring risks are managed proactively. The RBI programs are based on the findings of several workshops held to risk assess the identified threats to the Barossa subsea infrastructure and develop inspection requirements and associated cycles to maintain the integrity of each component.

The Corrosion Monitoring Procedure defines the philosophy, procedure, and reporting requirements for internal corrosion monitoring of pipelines / risers and subsea pressure piping.

The Subsea Inspection Procedure and Underwater Inspection Manual, describe the inspection philosophy, inspection types and reporting requirements for subsea assets, including marine growth and post cyclone surveys, offshore cathodic protection, and includes offshore structures and pipelines.

### 8.3.2.3 Maintenance Management System

Systems and equipment are maintained to achieve a required level of functionality. A risk-based approach is used to develop the inspection, maintenance and testing requirements. The likelihood of equipment degradation leading to failure is considered when setting up planned maintenance tasks and intervals. Maintenance activities shall be planned, as far as possible with consideration for frequency, criticality, resource optimisation and shutdown alignment. The maintenance program is prepared by systematic analysis of function criticality and application of maintenance strategies and is required to be adaptive to change in criticality or emerging maintenance technologies. The main maintenance strategy is condition based, non-intrusive maintenance which includes watch keeping, vibration analyses, lube oil sampling and other condition-based techniques where potential failure conditions can be monitored.

The gas export pipeline is designed to avoid the requirement for preventative maintenance (i.e. regular maintenance activities necessary to maintain integrity/operability. The subsea equipment condition will be regularly assessed to monitor its condition and detect if corrective maintenance is required.

Inspection findings and any resultant maintenance requirements are manages through the Santos MMS.

### 8.3.2.3.1 Critical Equipment Maintenance

Each maintainable item is assigned a criticality level in the Santos MMS based on the safety or environmental consequence of failure. Production downtime is also considered in this determination. The criticality then determines the priority that preventative or corrective maintenance work orders are assigned in the scheduling of maintenance work, to achieve desired performance functionality/reliability/availability of systems/equipment.

Equipment that has a function that is necessary to achieve or verify environmental performance standards and that may impact compliance with an environmental performance outcome (refer Section 8.1), are assigned a criticality rating reflective of the importance of the performance of this equipment. Breakdown of systems or equipment with a criticality level of 1 receive the highest priority for corrective maintenance and the frequency of planned maintenance is appropriate to meet the required availability of the equipment. Changes to the maintenance regime for criticality 1 equipment is subject to a risk assessment and approval process. The Santos management requirements of the maintenance does not go overdue without a deferral risk assessment, and that corrective maintenance is completed in the required timeframe.

A deferral process is in place to initiate a risk assessment well before any critical equipment maintenance becomes overdue. In raising deferrals, consideration is given to the likelihood of failure of that element during the deferral period. The deferral risk assessment process shall establish the potential for increased environmental impact or risk, or non-compliance with an associated EPS. In such cases the Santos Environment Manager will be included in the risk assessment process. The deferral for equipment that has a critical function in reducing environmental risks and impacts to ALARP and acceptable levels, or maintaining compliance with the OEMP requires approval from the Santos Production Manager, who is accountable for compliance with all internal and external regulatory requirements, including compliance with the OEMP.

## 8.3.2.4 Control of Work

The Control of Work process and procedures are for vessel-based activities in along the GEP (IMMR) so that work activities are executed in a safe manner. The Control of Work process and procedures include:

- Stop Work Authority. All personnel have the permission and responsibility to stop a work / task or to decline
  to perform a task where they believe there is a threat to the health or safety of people or the environment.
  Individuals calling to stop work should inform their supervisor / person in charge of the work immediately.
  When a safety concern is raised and work is stopped, the work shall also be reassessed. The Stop Work
  Authority and the workforce's responsibility to stop unsafe work, forms part of the induction program for all
  personnel arriving on board and forms part of facility toolbox talks.
- Safety Observation System. A safety observation system is in place to identify, record, report and track safety, environment or integrity hazards observed on the facility. Submitted observations are reviewed by the facility management team. Corrective actions which result from such identified hazards shall be documented, assigned to a person for close-out and tracked. Regular feedback back is provided to facility personnel on close-out of corrective actions. The facility induction covers use of the safety observation system.
- Performing task planning, including for PTW, daily reporting, short, medium, and long term planning, and managing unplanned tasks.
- Task Risk Assessments for hazard identification and risk control for high-risk, complex, and multitask
  offshore operations and maintenance activities, and special operations. Task Risk Assessments are an
  integrated part of the PTW process.
- The PTW Procedure describes the management of risks to health and safety arising from the interaction between work activities and process hazards at the Barossa facilities. The PTW Procedure is implemented for work activities carried out by all personnel on the Barossa facilities and whose work activities are under the influence and control of Barossa operations. The procedure documents hazard controls so that work can be carried out in an environment where risks are eliminated or minimised to ALARP. Members of the workforce that have been allocated roles within the PTW system are required to complete the training module relevant to their role (refer to Section 8.3.1.2). Vessels requesting entry to the 500m safety zone are required to complete a pre-entry checklist. A PTW and/or risk assessments may also be required depending on the types of activities a vessel is undertaking within the 500m safety zone. IMMR vessels shall operate under their own PTW system (refer to Section 8.3.2.4).
- Process safety, including override management and inhibit management.
- Defining the requirement for suitable bridging arrangements to be developed between the Barossa Control of Work and contractor control of work procedures and processes.
- Conducting Toolbox Talks to discuss the hazards, risks and associated controls and procedures for work activities.
- Managing hazardous operations, including vessel operations, lifting operations. Management of these hazardous operations is described in Sections 8.3.2.5 and 8.3.2.7.

## 8.3.2.5 Vessel operations

Santos manages marine vessel operations using a hierarchy of procedures, outlined below. These requirements for vessel acceptance criteria include technical, personnel (e.g. crew competencies) and operational requirements for marine vessels engaged by Santos.

Vessel Type	Vessel Acceptance Criteria
IMMR Vessels	The specific requirements for managing vessels performing Inspection Monitoring Maintenance and Repair (IMMR) or installation support vessels activities in the field are defined in campaign specific interface management plans. Such plans will typically define the vessels planned activities, communications protocols, PTW requirements, roles and responsibilities, interface management procedures and controls, emergency response protocols and incident reporting and notification requirements. A Santos CSR is present on IMMR and installation vessels in the field to provide oversight of the day to day activities on board the vessel.

#### 8.3.2.5.1 Marine operations manual

The Marine Operations Manual describes the standards of marine operations to achieve safe and efficient outcomes. The Marine Operations Manual details:

- standard operating procedures for all vessels under contract with Santos
- compliance requirements for relevant maritime legislation and relevant guidelines, standards and codes



- compliance requirements for international conventions and agreements, including:
  - International Convention for the Control and Management of Ships' Ballast Water and Sediments 2004
  - SOLAS 1974 and its Protocol of 1988
  - o International Convention for the Prevention of Pollution from Ships 1973/1978 (MARPOL 73/78)
  - o Convention on the International Regulations for Preventing Collisions at Sea 1972 (COLREGS)
  - International Convention on Standards of Training, Certification and Watchkeeping (STCW) for Seafarers, 1978.
- compliance requirements for industry standards as set up by:
  - OCIMF
  - o IMCA
  - Guidelines for Offshore Marine Operations
  - Nautical Institute.
- Marine assurance
  - vessel vetting criteria
  - marine personnel competency
- Santos and contractor standards, procedures and best practice management to comply with the control
  measures and performance standards (as set out in Table 8-2) in this GEP Coastal Waters OEMP as
  relevant to vessels, including:
  - vessels' safety of navigation
  - vessels using DP systems
  - vessels' bunkering procedures
  - o crew competency and training records
  - biosecurity management
  - o chemical storage and handling procedures
  - o discharge management procedures
  - waste management procedures
  - anchoring procedures
  - vessel and equipment maintenance procedures as per the vessel-specific safety management system.

### 8.3.2.6 Aviation operations

The Aviation Procedure which sets out the requirements for the safe, effective, and efficient management of all aviation operations and activities.

Operations must meet CASA's regular public transport (RPT) standards and published aircraft performance criteria, and independent evaluations or audits will take place at no less than 12 month intervals.

## 8.3.2.7 Lifting operations

Lifting operations on an IMMR vessel are managed under the IMMR vessel contractor's procedures. Contracted IMMR vessels are vetted under the Santos Marine Assurance process described in Section 8.4.6.

## 8.3.2.8 Waste management

Waste management will be undertaken in a manner consistent with Santos' waste management processes, including application of the waste management hierarchy, classification and segregation of waste streams, appropriate storage, transportation requirements, record management (e.g. waste inventories and tracking), use of licenced contractors/facilities and auditing.

Wastes from vessels are sent to shore and are maintained as per vessel class in accordance with relevant Marine Orders. This includes recording waste volumes in a Garbage Record Book.



## 8.3.2.9 Biosecurity management

### 8.3.2.9.1 Ballast water management

#### 8.3.2.9.1.1 Summary of requirements

The Australian ballast water management requirements set out the obligations on vessel operators regarding managing ballast water and ballast tank sediment when operating within Australian seas. These requirements include legislative obligations under the Biosecurity Act 2015 (Cth) and the International Convention for the Control and Management of Ships' Ballast Water and Sediments. The requirements provide guidance for vessel operators on best practice policies and apply to all vessels operating internationally and domestically in Australia. All vessels designed to carry ballast water (as applicable to vessel class) are required to carry the following:

- a valid ballast water management plan
- a valid international ballast water management certificate
- a type approval certificate specific to the type of ballast water management system installed (if installed)
- maintenance of a complete and accurate record of all ballast water movements, including those conducted in Australian waters.

Ballast water exchange should be conducted in areas at least 12 Nm from the nearest land and in water at least 50 m deep (having regard to the D-2 standard exemptions in the Australian Ballast Water Management Requirements [DAWE, 2020a]). Volumetric exchange must be at least 95% of the relevant tank.

Records on ballast water exchange must include the start and finish times and geographic coordinates of the operation.

All ballast water management equipment, such as pumps, will be maintained per the vessel's preventive maintenance system and regularly tested to ascertain accurate calculations for ballast water exchange operations.

#### 8.3.2.9.1.2 Australian pre-arrival report

All international vessels (intending to ballast) must submit a pre-arrival report (through the Maritime Arrival Reporting System [MARS]) at least 12 hours prior to arrival. The Ballast Water Report will be assessed by the DAFF through MARS, and a response will be issued through the Biosecurity Status Document. Domestic vessels can request a low-risk exemption through a domestic risk assessment through MARS.

MARS is the online portal used by commercial vessel masters and shipping agents to submit the reports required of all international vessels seeking Australian biosecurity clearance and to request services such as coastal strip, waste removal, ship sanitation certification and crew change.

DAFF will request evidence from vessels with a ballast water management system of:

- a valid ballast water management plan specific to the vessel (consistent with the Ballast Water Management Convention)
- a valid ballast water management certificate, or certificate of compliance, that is approved by a port state administration, or a recognised survey authority (consistent with the Convention)
- ballast water management records clearly demonstrate the ballast water management system has been operated consistently with the ballast water management plan.

A DAFF biosecurity officer may board the vessel to verify the pre-arrival report and personnel proficiency in the operation and maintenance of the ballast water management system.

#### 8.3.2.9.2 Biofouling management

IMS may be present as biofouling on the vessel hull or within piping, sea chests, etc. Biofouling, which may be found on and in a vessel, reflects the vessel's design, construction, maintenance and operations. Each of these aspects introduces particular biofouling vulnerabilities but also offers opportunities to limit the extent and development of biofouling, with commensurate reduction in biosecurity risks.

#### 8.3.2.9.2.1 Summary of requirements

Biofouling management for international vessels will comply the Australian biofouling management requirements (DAFF, 2023), which implements the requirements of the *Biosecurity Act 2015* (Cth) and the IMO 2023 Guidelines for the Control and Management of Ships' biofouling to Minimize the Transfer of Invasive Aquatic Species.

Under the *Biosecurity Regulation 2016* (Cth), all operators of vessels intending to enter Australian territorial waters must provide information relating to biofouling management through the mandatory pre-arrival report 12–96 hours prior to arrival. In addition, the vessel operator must demonstrate proactive management of biofouling by implementing one of the 3 accepted proactive biofouling management options:



- implementation of an effective biofouling management plan and record book
- cleaned all biofouling within 30 days prior to arriving in Australian territory
- implementation of an alternative biofouling management method pre-approved by the department.

Vessels mobilised to the OAs from international or domestic waters must also comply with the National biofouling management guidelines for the petroleum production and exploration industry (Marine Pest Sectoral Committee, 2009).

#### 8.3.2.9.2.2 Vessel risk assessment

This includes:

- completing a biofouling risk assessment
- implementing mitigation measures commensurate with the level of risk.

Figure 8-6 illustrates the biofouling risk assessment process. Factors that will inform risk include:

- timing of marine pest risk assessment relative to the Activity vessel mobilisation to provide sufficient time to implement control measures in cases where management is warranted
- Activity vessel location history since last dry dock and clean to inform whether the Activity vessel may have been exposed to high-risk ports/locations
- level of biofouling and the presence of species of concern (particularly the presence of marine pests) within biofouling communities on the vessels associated with the Activity (often informed by biofouling record books and/or maintenance/cleaning or inspection programs)
- operational profile relevant to biosecurity risk such as operating speed, time alongside a facility and the need for ballast exchanges within the title area
- receiving environment including the presence of shallow-water sensitivities near the Activity and the presence and area of non-biocidal surfaces on facilities that could harbour marine pests
- presence and effectiveness of external and internal marine growth prevention systems including effectiveness and integrity of anti-fouling coatings and functionality of internal treatment systems
- qualifications and competency of those conducting and reviewing the risk assessment and making management decisions.

#### 8.3.2.9.2.3 Vessel risk status

Vessels must achieve a 'low' risk status to demonstrate to the government that Santos has taken all reasonable measures to minimise the risk of IMS. The risk assessment categorises the vessel's risk status as:

- low low risk of introducing IMS; no additional management measures required
- uncertain risk of introducing IMS is not apparent; precautionary approach adopted, additional management measures required to achieve low status
- high high risk of introducing IMS; additional management measures will be required.

#### 8.3.2.9.2.4 Potential management measures to achieve low risk status

The outcome of the risk assessment will determine the management measures required. If the vessel is deemed as 'low' risk status, no other measures are required (providing the vessel does not exceed the 7 day threshold at stationary or slow speed, in waters outside Australia).

For vessels that are assessed as having an 'uncertain' or 'high' risk, contractors will engage a qualified IMS inspector to conduct inspections and/or provide advice on obtaining 'low' status. Table 8-8 lists mitigation measures that can be applied to achieve 'low' risk status.

#### Table 8-8: Biofouling mitigation measures

Mitigation measure	Overview
IMS inspection	Visual inspection of submerged surfaces and niche areas by a qualified biosecurity inspector to better understand the actual biosecurity risk.
In-water cleaning	The appropriateness of in-water cleaning operations must be a decision made closely with an IMS inspector on a case-by-case basis. Many factors will be considered, including:



Mitigation measure	Overview
	<ul> <li>degree and type of biofouling</li> <li>location of biofouling on the vessel.</li> <li>Before undertaking in-water cleaning within Australia, approval from the relevant state/territory authority must be granted and conditions may be imposed. Application must be made to the administering authority (harbour master, local government or state/territory environmental protection agency) at least 5 working days before the proposed start of work.</li> </ul>
Dry docking cleaning	Dry docking and removing/cleaning biofouling will include hull surfaces, niche areas such as sea chests, all retractable equipment such as thrusters, intakes and outlets, anodes and voids.
Temporal or spatial controls	Temporal or spatial controls to limit vessel exposure to sources of risk.
Applying anti fouling coating	Depending on its age, the vessel may require a new anti-fouling coating to be applied by professional operators. The anti-fouling coating type will be based on technical advice. All vessels more than 400 gross tonnage require a valid anti-fouling system certificate.
Treating internal seawater systems	In the absence of a marine growth prevention system, internal seawater systems may need to be cleaned. Cleaning actions may include: dehydration heat physical removal chemical treatment. Ideally, treating internal seawater systems will be undertaken before the vessel is mobilised to Australia. If chemical treatments are to be undertaken within Australian waters, advice must be sought from the Australian Pesticides and Veterinary Medical Authority (https://apvma.gov.au/) in relation to permit and reporting requirements—it is prohibited to clean internal systems in Australian waters without a permit.



Source: Marine Pest Sectoral Committee (2009)

#### Figure 8-6: Generic biofouling risk assessment process

### 8.3.2.10 Chemical management

Chemicals will be selected in accordance with the Santos chemical selection process (refer Section 2.7) and managed in accordance with the contractor management system.

A Safety Data Sheet (SDS) is provided for each for each hazardous substance. The SDS describes its physical and chemical properties, health and physical hazards, safe handling and storage, emergency procedures and disposal considerations.

### 8.3.2.11 Methane Emissions Management

Santos' approach to methane, as outlined in the 2024 Annual Report, focuses on three key areas:

• **Detect, measure and validate**: Activities that detect and accurately measure methane emissions using a combination of evidence-based theoretical techniques and real-time technologies. The utilisation of various methods and technologies permits validation of results and comparison against reported emissions. Our most material emissions are assessed and prioritised accordingly.



- **Monitor and mitigate**: Asset-led programs that incorporate surveillance of emissions using different techniques and technologies. These programs permit prioritisation of our most material emissions, associated reparation feasibility assessment and value impact to the business.
- Engagement and leadership: Interaction with stakeholders across the methane value chain to collaborate on solutions. This includes engagement and collaboration with our peers on approaches for methane measurement and reduction.

Actions implemented under each of these pillars may change year-on-year as our approach evolves, but in 2024 included, at the corporate level:

- Becoming a signatory of the OGCI Aiming for Zero Initiative, committing to 'near zero' methane emissions from operated assets by 2030<sup>48</sup>
- Became a signatory to the World Bank's Zero Routine Flaring Initiative, committing to avoid routine flaring in new oil field developments and end routine flaring at existing oil facilities by 2030, where economically viable.
- Participation in a satellite monitoring campaign to validate knowledge of methane emissions within the areas we operate
- Completed a methane emissions materiality assessment across our operated assets
- Participated in the Australian Energy Producers methane working group and Climate Leaders Coalition methane workstream

Methane emissions from Santos portfolio operations, equated to approximately 10 per cent of total gross operated Scope 1 emissions in 2024. Santos also achieved a methane emissions intensity of 0.16%, which is below the Oil and Gas Climate Initiative's (OGCI) 2025 target of 0.20%.

Whilst methane emissions management is shaped by corporate targets and RBU management objectives, it is operationalised at the asset level taking into consideration factors including NGERs legislative reporting requirements, integrity and safety risk, materiality of emission sources, applicability of methane detection, measurement and reduction technologies and stage of asset-life.

As the operator of the Barossa facilities, Santos will operationalise the three methane emissions reduction pillars to Barossa operations having regard to its joint venture arrangements. Operationalisation of these pillars will be in the form of specific control measures and environmental performance standards (refer to BAO-CM-6.3.12) which aims to ensure that methane, along with other GHG emissions, are reduced to ALARP throughout the life of the Activity.

Methane emissions reduction opportunities are also considered as part of Santos' ongoing decarbonisation opportunity management process as described in Section 8.5.6.

## 8.3.2.12 Greenhouse Gas Emissions Management Plan

Management of facility Scope 1 GHG emissions over the life of the Barossa Gas Project will be governed under the Operations Greenhouse Gas Emissions Management Plan (GHGEMP) to ensure a coordinated approach across all aspects of GHG emissions management described in the implementation strategy. For the purpose of the Barossa Gas Project, the facilities include the FPSO and the GEP. The purpose of the Operations GHGEMP is to reduce Scope 1 emissions from facility operations to ALARP over the life of the Barossa Gas Project. The Barossa Production Manager will be accountable for the Operations GHGEMP. The GHGEMP will take effect from the commencement of steady-state operations.

The scope of the Operations GHGEMP includes all FPSO and subsea emissions sources (including the GEP – of which this Activity is a component), upstream of the DLNG facility custody transfer point at the DLNG facility beach valve. That is, the scope of the Operations GHGEMP covers emissions from Operations under the Barossa Production Operations EP but also emissions from operation of the GEP in NT waters including this GEP Coastal Waters OEMP (noting that GHG emissions from operation of the GEP will be limited to fugitive emissions from conveyance of gas through the pipeline and vessel emissions from infrequent IMMR vessel activity).

To ensure ongoing GHG emissions management in line with evolving best practice standards, and because the GHGEMP captures activities outside the scope of this GEP Coastal Waters OEMP, the GHGEMP will be a living document within the Barossa Management System (Section 8.2.1.1) and not annexed to this GEP Coastal Waters OEMP.

The following protocols, procedures, systems and measures as detailed further within the implementation strategy at the sections cross-referenced below, will be incorporated in, and form part of, the GHGEMP:

<sup>&</sup>lt;sup>48</sup> Page 90 of Santos 2024 Annual Report <u>https://www.santos.com/wp-content/uploads/2025/02/FINAL-Appendix-4E-and-2024-Annual-Report.pdf</u>



- Emissions Performance Target Setting (Section 8.2.4)
- Critical Equipment Maintenance (Section 8.3.2.3.1)
- Methane Emissions Management (Section 8.3.2.11)
- Decarbonisation Opportunity Management (Section 8.5.6)

Adaptive management, to address areas of uncertainty, is integral to emissions performance target setting (Section 8.2.4), methane emissions management (Section 8.3.2.11) and decarbonisation opportunity management (Section 8.5.6).

Due to its importance as a governing document, the GHGEMP has been adopted as a specific control measure (BAO-CM-6.3.12) with corresponding environmental performance standards (Table 8-2).

In addition to any updates to the GHGEMP triggered as a result of annual performance reviews (Section 8.5.3.2) and the annual emissions performance target setting cycle (Section 8.2.4), the GHGEMP will be reviewed and updated every five years at a minimum.

## 8.3.3 Other Measures

During the preparation of this Coastal Waters OEMP, including as a result of consultation with Relevant Persons, Santos has identified additional measures which it considers are appropriate to implement. These measures are not control measures, as defined in the OPGGS(E)R, because they are not intended to be used by Santos as a basis for managing environmental impacts and risks. Some measures are not properly characterized as 'control measures' in respect of the Activity because they relate to operations outside of the operational area, which are not regulated under this GEP Coastal Waters OEMP. Notwithstanding this, Santos considers it appropriate to adopt the following measures as part of its implementation strategy:

- To build community confidence and capacity, Santos will offer training to Tiwi Islands Ranger Groups, other Tiwi people nominated by Tiwi Land Council Trustees and Croker Island Ranger Groups for rapid assessment for hydrocarbon spill incidents. Training will be provided subject to the interest, availability and the participation of the Ranger Groups.
- Santos to advise relevant Tiwi clan member if any bones are identified during the term of this GEP Coastal Waters OEMP.

## 8.4 Check

The 'Check' section describes surveillance and assurance activities to check effectiveness of the implementation strategy and associated control measures to achieve environmental performance outcomes (Table 8-1) and performance standards (Table 8-2) in this GEP Coastal Waters OEMP.

#### **OPGGS(E)R 2023 Requirements**

#### Section 22 Implementation strategy for environment plan

#### Monitoring and reporting

22(5) The implementation strategy must provide for sufficient monitoring, recording, audit, management of non-conformance and review of the titleholder's environmental performance and the implementation strategy to ensure that the environmental performance standards in the environment plan are being met.

## 8.4.1 Assurance framework

Santos Assurance Procedure defines the framework and requirements for the planning and application of assurance activities for Santos Operations.

Level 1 assurance (such as the inspections described in Section 8.4.3) is the day-to-day assessment and management of risk undertaken by frontline leadership (a focus on hazards), which includes:

- Ensuring that controls are in place and operating effectively for the management of risk.
- Monitoring and maintaining compliance with relevant controls.
- Escalating identified material or systemic control failures or weaknesses and ensuring that any ongoing risk
  exposure is managed in line with the relevant risk management procedure.

Level 2 assurance (such as the audits described in Section 8.4.4) is management oversight through an annual assurance plan, based on key risks and controls (typically as described in relevant approvals such as the OEMP),

and the underpinning systems. Typically, this level of assurance is conducted by management and senior management. Findings of Level 2 assurance activities resulting in necessary control improvement actions to effectively manage risk exposure are recorded, monitored and when completed, closed. Emerging trends and material or systemic control failures or weaknesses identified through these assurance activities must be promptly escalated to the relevant managers and any ongoing risk exposure must be managed in line with the relevant risk management procedure.

Level 3 assurance is independent auditing, undertaken by independent internal or external auditors and is independent of Operations. The integrated assurance plan is governed by the Executive Functional Governance Committee of Santos, with a focus on audit of Technical and Operating Standards and "selection of risk-based audits". Findings from this assurance activity resulting in improvement actions are recorded, monitored and closed out in an approved system within agreed timeframes and the internal audit team notified upon completion. Key findings from these audits are required to be shared with the Santos leadership team.

Inspections and audits are undertaken to confirm compliance with the performance outcomes, controls measures and environment performance standards in the OEMP and confirm that the control measures are effective at reducing the environment impacts and risks. They will also identify potential new or changes to existing environmental impacts and risk, and methods for reducing these to ALARP.

Santos will retain accountability for assurance of all aspects of the Barossa Management System for all stages of the Activity including undertaking assurance activities in a manner consistent with the Barossa Project Environmental Compliance and Assurance Plan (ECAP). The ECAP outlines a process that enables the planning, collection and verification of environmental assurance evidence across the life of the Activity to measure compliance against the EPOs, EPSs, and measurement criteria for this GEP Coastal Waters OEMP. The ECAP is centred around the GEP Coastal Waters OEMP Compliance Register, which is the primary compliance tool that has a touchpoint at all stages of the environmental assurance process and directly shapes environmental assurance requirements, tools, schedules, verification and reporting. The GEP Coastal Waters OEMP Compliance Register identifies all applicable environmental requirements within the OEMP (EPOs, EPSs, measurement criteria, notifications), assigns responsibilities and applies verification controls, timing and tasks to each environmental compliance requirement. It will be first developed following acceptance of this GEP Coastal Waters OEMP and prior to commencement of the activities. It will be reviewed and revalidated following 12 months of steady state operations, and then again follow the 5-year revision of the GEP Coastal Waters OEMP. Planned inspections and audits against the GEP Coastal Waters OEMP Compliance Register are described in Section 8.4.3 and 8.4.4.

## 8.4.2 Monitoring, recording of emissions and discharges

#### **OPGGS(E)R 2023 Requirements**

Section 34 Criteria for acceptance of an environmental plan

For the purposes of section 33, the criteria for acceptance of an environment plan (the *environment plan acceptance criteria*) for an activity are that the plan:

includes an appropriate implementation strategy and monitoring, recording and reporting arrangements.

Section 22 Implementation strategy for environment plan

Monitoring and reporting

22(6) The implementation strategy must provide for sufficient monitoring of, and maintaining a quantitative record of, emissions and discharges (whether occurring during normal operations or otherwise), such that the record can be used to assess whether the environmental performance outcomes and environmental performance standards in the environment plan are being met.

The following sections describe monitoring and recording of emissions and discharges for each stage of the Activity.

The purpose of emissions and discharges monitoring is to assess compliance with the GEP Coastal Waters OEMP requirements and to reduce environmental impacts and risks to ALARP. Monitoring data may also inform identification of improvement opportunities as part of continuous improvement over the life of the Activity.

Monitoring and recording of emissions and discharges for each stage of the Activity, as described in the sections that follow, will also apply to contingency activities depending on which stage of the Activity they occur.

## 8.4.2.1 Initial Start-up

### 8.4.2.1.1 Vessel emission and discharge monitoring

Discharges to the marine environment associated with this Activity will be recorded and controlled in accordance with requirements under relevant Marine Orders and MARPOL requirements.



Santos and vessel contractors will maintain records so emissions and discharges can be determined or estimated. Such records will be maintained for a period of five years. Contractors are required to make these records available upon request.

For vessel activities Santos will maintain records of discharges or emissions, to the environment as described in Table 8-9.

	Table	8-9:	Monitorir	g of	vessel	emissions	and	discharge
--	-------	------	-----------	------	--------	-----------	-----	-----------

Vessel Discharge/emission	Parameter	Quantitative record	Frequency	
Atmospheric and GHG emissions	Fuel volume	Calculations based on measured fuel use in accordance with NPI/NGER reporting scheme requirements	Fuel use tracked daily, emissions calculated at end of campaign	
Oily water / bilge	Volume and location	Oil Record Book* or equivalent report	As required	
Ballast water	Volume and location	Ballast water log**	As required	
Treated sewage and greywater	Volume and location	Estimated based on POB and days on location	Once per campaign	
Unplanned discharge of: <ul> <li>solid objects</li> <li>hazardous liquids</li> </ul>	Volume	DME recordable or reportable incident reports	As required	
Unplanned hydrocarbon release	Volume	DME recordable or reportable incident reports	As required	

\* Maintained as per vessel class in accordance with relevant regulations

\*\* Maintained as per Australian Ballast Water Management Requirements 2020

### 8.4.2.2 Steady state operations

#### 8.4.2.2.1 Vessel emission and discharge monitoring

Discharges to the marine environment associated with this Activity will be recorded and controlled in accordance with requirements under relevant Marine Orders and MARPOL requirements.

Santos and vessel contractors will maintain records so emissions and discharges can be determined or estimated. Such records will be maintained for a period of five years. Contractors are required to make these records available upon request.

For vessel activities Santos will maintain records of discharges or emissions, to the environment as described in Table 8-10.

#### Table 8-10: Monitoring of vessel emissions and discharges

Vessel Discharge/emission	Parameter	Quantitative record	Frequency
Atmospheric and GHG emissions	Fuel volume	Calculations based on measured fuel use in accordance with NPI/NGER reporting scheme requirements	Fuel use tracked daily, emissions calculated at end of campaign
Oily water / bilge	Volume and location	Oil Record Book* or equivalent report	As required
Ballast water	Volume and location	Ballast water log**	As required
Treated sewage and greywater	Volume and location	Estimated based on POB and days on location	Once per campaign
Unplanned discharge of: • solid objects • hazardous liquids	Volume	DME recordable or reportable incident reports	As required

Vessel Discharge/emission	Parameter	Quantitative record	Frequency
Unplanned hydrocarbon release	Volume	DME recordable or reportable incident reports	As required
* Maintained as per vessel clas			
** Maintained as per Australiar			

Discharges from vessels are also monitored as per vessel class in accordance with relevant Marine Orders and MARPOL requirements.

#### 8.4.2.2.2 GEP emissions monitoring

Fugitive emissions are not actively metered; they will be estimated annually based on default NGER factors. If a major repair is required as an unplanned activity, then fugitive emissions would be estimated based on default NGER factors based on design parameters of pipeline length. Measurement data used for emissions calculations is recorded and stored in a production historian database.

## 8.4.3 Inspections

Inspections are the primary tool to check compliance with the relevant commitments (including CMs and EPSs) in the GEP Coastal Waters OEMP. Inspections can be desktop or field based. Inspections will be undertaken throughout all phases of the Activity and will utilise the GEP Coastal Waters OEMP Compliance Register (refer Section 8.4.1) as the basis to inform their scope.

#### Vessels

An environmental inspection program will be implemented for all vessels carrying out activities under this GEP Coastal Waters OEMP. Inspections are used to assess the vessels' compliance against requirements outlined in this GEP Coastal Waters OEMP, and at the end of all vessel campaigns, records will be collected from the vessel contractor to demonstrate compliance with the GEP Coastal Waters OEMP for performance reporting. Table 8-11 outlines the inspection schedule and scope for vessels. Inspection reports will be distributed for review to Santos relevant personnel (e.g. Operations Superintendent, Santos Company Site representatives), and HSE Department representatives.

#### Table 8-11: OEMP Inspection Schedule and Scope for Vessels

Activity Phase	Inspection	Frequency	Scope
Pre-mobilisation	Pre-mobilisation inspection	Once prior to entering the Operational Area	Environmental pre-mobilisation inspection will examine the level of mobilisation readiness of the vessel prior to entering the Operational Area, undertaken by a suitably experienced Santos or Contractor HSE Advisor. The inspections include a review of readiness against GEP Coastal Waters OEMP commitments.
Activity	Execution Inspections	Monthly - the first inspection will be undertaken within 2 weeks of entering the field. If Activity duration is <1 month then an inspection will be undertaken during the Activity	General ongoing environment inspections against all GEP Coastal Waters OEMP commitments and as relevant plans/ procedures/ other requirement using a risk based approach.

Any opportunities for improvement that cannot be addressed as part of inspection close out, will be considered as part of annual performance reviews and identification of continuous improvement opportunities (Section 8.5.3).

### 8.4.4 Audits

Santos maintains Activity audit plans and schedules that are frequently reviewed and updated. Audits will typically include for assessing compliance with relevant GEP Coastal Waters OEMP commitments as well as the supporting management processes and procedures described in the implementation section of the GEP Coastal Waters OEMP. For Contractors the audit will review their processes and procedures that support implementation of and compliance with the commitments in the GEP Coastal Waters OEMP.

Table 8-12 outlines the GEP Coastal Waters OEMP audit schedule. Further information on assurance activities for vessels is described in Section 8.4.6.

Audits will be undertaken in a manner consistent with the Santos' Assurance Procedure and the Barossa Project Environmental Compliance Assurance Plan (ECAP). Santos will retain accountability for assurance of all aspects of the Barossa Management System, for all stages of the Activity.


### Table 8-12: OEMP Audit Schedule

Activity Phase	Audit Type	Frequency	Scope
Pre mobilisation inspection	Field Audit of GEP Coastal Waters OEMP	Once (minimum), in conjunction with the pre- mobilisation inspection (refer Table 8-11)	Aspects of the OEMP implementation strategy relevant to the activity phase, with a focus on operational readiness and compliance with the OEMP ahead of entering the Operational Area.
Vessel campaigns (for vessels contracted for >3 mths duration)	Desktop or Field Audit of GEP Coastal Waters OEMP	Once per campaign	Aspects of the OEMP implementation strategy as relevant to the Activity.
All phases	Desktop or Field Audit of GEP Coastal Waters OEMP	As required – in the event that assurance data (inspection reports, monitoring data) is trending toward non-compliance a proactive audit will be undertaken.	Risk specific, with a focus on implementation aspects.

Santos' audit plans and schedules are reviewed and updated at the beginning of each activity phase or calendar year (following commencement of steady state operations) and cover all Santos' facilities and activities.

During steady state operations, selected risk areas/activities are identified to review environmental performance against the EPOs and EPSs and verify that control measures are effective in reducing the environmental risks and impacts of the activity to an ALARP and acceptable level. The audit also includes review of conformance with selected aspects of the GEP Coastal Waters OEMP implementation strategy

Vessel HSE audits by the Santos HSE department are performed in accordance with the applicable requirements of any Environment Plan, Pipeline Management Plan or campaign specific HSE Management Plans. The GEP Coastal Waters OEMP compliance register (refer Section 8.4.1) will capture all OEMP requirements related to marine vessels (EPOs, EPSs, measurement criteria, notifications).

OPEP assurance activities, including testing arrangements and audits, are described in the GEP NT Waters OPEP (Section 5.6 of the OPEP).

Audit findings may include opportunities for improvement and non-conformances. Audit nonconformances are managed as described in Section 8.4.1. Any opportunities for improvement that can't be closed out as part of audit close out, will be considered as part of annual performance reviews and identification of continuous improvement opportunities (Section 8.4.2).

Audit findings and conformance with the Audit Schedule in Table 8-12 will be provided to the Vessel Master at the completion of the audit, with recommendations and actions to be tracked to close out. The audit findings will also be reported in the Annual Performance Report.

## 8.4.5 Incident Investigations

The Event Reporting & Investigation Technical Standard describes the requirements for reporting and investigating events that can cause harm to people, the environment and/or assets. These requirements ensure:

- Learning from events.
- Preventative and corrective actions are identified.
- Actions are tracked to closure.

The scope of the standard includes hazards, near misses, incidents, occupational injuries and illness, nonconformances and events requiring technical investigations.

Incident investigations will demonstrate consideration of the following:

- the availability and feasibility of adopting new or improved technology that would minimise the risk of incident recurrence
- the incorporation, where relevant, of lessons learned from the incident into the GEP Coastal Waters OEMP, as part of the continuous improvement cycle
- assessment of existing EPOs, EPSs and CMs to determine if they can continue to be achieved

Where changes are required to the GEP Coastal Waters OEMP, based on the outcomes of the above considerations, the Environment Plan MoC process outlined in Section 8.5.2 will be followed.



## 8.4.6 Marine Assurance

The Marine Assurance Standard requires that all vessels engaged for the Activity are to be vetted and applies to activities described in the GEP Coastal Waters OEMP. The vetting process is based on industry standards and best practices, along with considerations of guidelines and recommendations from recognised industry organisations such as Oil Companies International Marine Forum (OCIMF) and International Maritime Contractors Association (IMCA), and international regulatory agencies like the IMO and vessel classification societies.

The Marine Assurance Standard provides the framework to define the minimum controls, expectations and guidance for the selection and engagement of contractors and their vessels to minimise the risk of harm to people, environment, or assets. This includes requirements relating to:

- contractor assurance;
- vessel assurance and suitability;
- operational capability;
- ongoing vessel assurance; and
- change management.

All contracted vessel operators are audited against their respective Management Systems to ensure they have policies and procedures that define the safe operation of their vessels, verify their compliance with these and their continual improvement process.

Vessels are audited annually by an OCIMF or IMCA accredited inspectors against the respective Offshore Vessel Inspection Database (OVID) report or a Common Marine Inspection Document (CMID). For vessels where the OVID and/or CMID are not valid or available, a Santos approved inspection report is required.

The Marine Assurance Standard includes the following requirements relating to the ongoing assurance of vessels:

- Vessel visits are performed as required by the Santos Marine Superintendent or other Santos
  representative (or nominated Contractor representatives) throughout the charter period. The purpose of
  such visits is to ensure Santos' expectations are being upheld, there is compliance and conformity with
  operational procedures and a safety culture is being fostered.
- Reviews of the vessel Contactor's Management System are performed every three years, or whenever there is considered a requirement, such as following a major incident, repeated incidents, incidents which may impact the vessel State, Class or Flag status or if a vessel operator has been detained by a regulatory authority.
- Reviews of crew experience are undertaken whenever key personnel crew members change.
- Control measures relevant to crewed vessels (such as BAO-CM-6.1.1 and BAO-CM-6.1.2) apply equally to
  crewed support vessels, noting that the Santos marine assurance process will be tailored to measures
  specific to USVs to achieve an equivalent level of environmental performance. For example, a 360 degree
  camera will be required to comply with Part 8 of the compliance with Part 8 of Environment Protection and
  Biodiversity Conservation Regulations 2000 (BAO-CM-6.1.1) and the Vessel Master (person responsible
  for remotely operating the USV) will be subject to equivalent vessel crew Australian maritime requirements.

## 8.4.7 Reporting and Notifications

### **OPGGS(E)R 2023 Requirements**

Section 22 Implementation strategy for environment plan

22(7)

The implementation strategy must state when the titleholder will report to NOPSEMA in relation to the titleholder's environmental performance for the activity. The interval between reports must not be more than 12 months.

Note: Section 51 requires a titleholder to report on environmental performance at the times or intervals set out in the environment plan.

Where section 22 of the OPGGS(E)R (above) notes reporting requirements to NOPSEMA, Santos will report to DME, as the relevant regulatory authority for administering the OPGGS(E)R in NT Coastal Waters.

Santos will also comply with any reporting requirements as per the conditions of EPBC Act approval EPBC 2022/09372 and EP Act approval EP 2022/022-001.

Regulatory and other notification and compliance reporting requirements are summarised in Table 8-13 and Table 8-14.

### Table 8-13: Activity notification and reporting requirements

Initiation	Required Information	Timing	Туре	Recipient	
Before the Activity					
Australian Hydrographic Office (AHO)	Notification of proposed start and end dates and any other relevant information for the Notice to Mariners to be issued.	No less than four working weeks before operations.	Written	AHO datacentre@hydro.gov.au	
Australian Maritime Safety Authority (AMSA)	<ul> <li>AMSA's Joint Rescue Coordination Centre (JRCC) requires the:</li> <li>vessel details (including name, callsign and Maritime Mobile Service Identity)</li> <li>satellite communications details (including International Maritime Satellite C (INMARSAT-C) and satellite telephone numbers)</li> <li>area of operation</li> <li>requested clearance from other vessels</li> <li>any other information that may contribute to safety at sea</li> <li>when operations start and and</li> </ul>	At least 48 hours before operations begin.	Written	AMSA's JRCC rccaus@amsa.gov.au	
DLI (NT), Regional Harbourmaster	Notification of proposed start and end dates of planned IMMR activities and any other relevant information for the Notice to Mariners to be issued	No less than 4 weeks before vessel operations begin	Written	Regional Harbourmaster	
Marine user notifications to Relevant Persons identified in Table 8-14 (as may be updated from time to time).	Prior notification to OA marine users of planned Activity commencement.	At least 48 hours before IMMR activities begin	Written	As indicated in Table 8-14 by email.	
<b>DAFF</b> – Biosecurity (international vessels, aircraft and personnel) (refer Section 4)	<ul> <li>In accordance with control measure BAO-CM-7.2.1, Santos will:</li> <li>pursuant to the <i>Biosecurity Act 2015</i> and the Biosecurity (<i>Exposed Conveyances – Exceptions from Biosecurity Control</i>) Determination 2016, undertake a vessel biosecurity risk and be assessed as 'low' by the Commonwealth Department of Agriculture, Fisheries and Forestry before interacting with domestic vessels and aircraft</li> <li>undertake pre-arrival approval for the vessels (where applicable) using the Maritime Arrivals Reporting System (MARS) to meet the DAFF biosecurity reporting obligations.</li> </ul>	At least one month before Activity begins. Maritime Arrivals Reporting System reporting at least 12 hours before arrival.	Written	DAFF Biosecurity (vessels, aircraft and personnel) <u>https://www.agriculture.gov.au</u> /biosecurity-trade/aircraft- vessels-military/vessels/mars	
DAFF (Fisheries)	Prior notification of planned Activity commencement for the purpose of awareness of potential impacts to Commonwealth fishery licence holders.	No less than 4 weeks prior to the start of activities.	Written	DAFF	

Initiation	Required Information	Timing	Туре	Recipient
Department of Defence (DoD)	<ul> <li>Prior notification of planned Activity commencement, for the purposes of:</li> <li>consideration of Defence activities</li> </ul>	No less than 5 weeks prior to the start of activities.	Written	DoD
	consideration of restricted airspace.			
NT Department of Agriculture and Fisheries (DAF)	Prior notification of planned Activity commencement for the purpose of awareness of potential impacts to NT State fishery licence holders.	No less than 4 weeks prior to the start of activities.	Written	NT DAF
Northern Territory Seafood Council (NTSC)	Prior notification of planned Activity commencement for the purpose of awareness of potential impacts to NT State fishery licence holders.	No less than 4 weeks prior to the start of activities.	Written	NTSC
Quarterly updates	The Activity will be included in the Quarterly Update until the Activity has ended. In the event that distribution of this update does not correlate with the Activity schedule, notifications will be provided to identified relevant commercial fishers within the OA before and after the Activity.	Quarterly	Online on Santos' website and automated notifications to registered/ subscribed interested parties	Relevant Persons and any other interested party who has registered or subscribed for quarterly updates.
OPGGS(E)R 54 – Notifications DME– Energy Division must be given written notice that the Activity is to begin	Prior notification of planned Activity commencement.	At least 10 days before the Activity begins.	Written	DME (NT) – Energy Division
During the Activity				
OPGGS(E)R 50 – Recordable Incidents DME – Energy Division must be notified of a breach of an EPO or EPS in the GEP Coastal Waters OEMP that applies to the Activity that is not a reportable incident.	<ul> <li>The record must include:</li> <li>a record of all recordable incidents that occurred during the calendar month; and</li> <li>all material facts and circumstances concerning the recordable incidents that the titleholder knows or is able, by reasonable search or enquiry, to find out; and</li> <li>any action taken to avoid or mitigate any adverse environmental impacts of the recordable incidents; and</li> <li>the corrective action that has been taken, or is proposed to be taken, to stop, control or remedy the recordable incident; and the action that has been taken, or is proposed to be taken, to prevent a similar incident occurring in the future.</li> </ul>	As soon as practicable after the end of the calendar month, and in any case, not later than 15 days after the end of the calendar month.	Written	DME (NT) – Energy Division

Initiation	Required Information	Timing	Туре	Recipient
OPGGS(E)R 24(c), 47 & 48 – Reportable Incident A reportable incident is defined as per Section 8.4. DME must be notified of any reportable incidents. Following an initial oral	<ul> <li>The oral notification must contain:</li> <li>all material facts and circumstances concerning the reportable incident known or that could be found out by reasonable search or enquiry</li> <li>any action taken to avoid or mitigate any adverse environmental impacts of the reportable incident</li> <li>the corrective action that has been taken, or is proposed to be taken, to stop, control or remedy the reportable incident.</li> </ul>	As soon as practicable, and in any case not later than 2 hours after the first occurrence of a reportable incident, or if the incident was not detected at the time of the first occurrence, at the time of becoming aware of the reportable incident.	Oral	DME (NT) – Energy Division
notification to DME, a written record of the notification must be given to other government agencies in accordance with s 47(3) of the OPGGS(E)R.	A written record of the oral notification must be submitted. The written record is not required to include anything that was not included in the oral notification.	As soon as practicable after the oral notification.	Written	DME (NT) – Energy Division (in its capacity as the relevant regulatory authority for administering the OPGGS(E)R in NT Coastal Waters and for administering petroleum titles in NT Coastal Waters). Department of the responsible State or NT Minister (noting this is also DME – Energy Division
	<ul> <li>A written report must contain:</li> <li>all material facts and circumstances concerning the reportable incident known or that could be found out by reasonable search or enquiry</li> <li>any action taken to avoid or mitigate any adverse environmental impacts of the reportable incident</li> <li>the corrective action that has been taken, or is proposed to be taken, to stop, control or remedy the reportable incident</li> <li>the action that has been taken, or is proposed to be taken, to prevent a similar incident occurring in the future</li> </ul>	Must be submitted as soon as practicable, and in any case not later than 3 days after the first occurrence of the reportable incident unless DME (NT) – Energy Division specifies otherwise. Same report to be submitted to DME and the Department of the responsible State or NT Minister (noting this is also DME – Energy Division) within 7 days after giving the written report to DME.	Written	NOPSEMA https://securefile.nopsema.go v.au/filedrop/submissions NOPTA reporting@nopta.gov.au DME – Energy Division

Initiation	Required Information	Timing	Туре	Recipient
OPGGS(E)R 22(7) & 51 – Environmental Performance DME must be notified of the environmental performance at the intervals provided for in the GEP Coastal Waters OEMP	Report must contain sufficient information to determine whether or not EPO and EPS in the GEP Coastal Waters OEMP have been met. Report will also address progress of Santos' identification and/or implementation of sea country initiatives.	An environmental performance report will be submitted to DME (NT) – Energy Division annually from the date of acceptance of this GEP Coastal Waters OEMP.	Written	DME (NT) – Energy Division
Notifiable incident (incident causing or threatening material or significant	<ul><li>The information must consist of the following:</li><li>time, date, nature, duration and location of the incident</li></ul>	A written report as soon as practicable (and in any case within 24 hours) after the person observes or becomes aware of the incident; or if in an emergency:	Oral (if in an emergency)	CEO of DEPWS NT EPA pollution hotline: 1800 064 567

Initiation	Required Information	Timing	Туре	Recipient
<ul> <li>environmental harm) as per:</li> <li>EP Act (NT) Pt 9, Div 8</li> <li>Environment Protection Regulations 2020 (NT) Pt 10</li> <li>There is also a requirement to report environmental incidents that cause, or threaten or may threaten to cause pollution resulting in material or significant environmental harm under Part 3 of the Waste Management and Pollution Control Act 1998 (NT) (WMPC Act). Notifications under the WMPC Act will be made a the NT EDA</li> </ul>	<ul> <li>the location of the place where the environmental harm is likely to occur</li> <li>the nature, the estimated quantity or volume and the concentration of any pollution involved</li> <li>the circumstances in which the incident occurred (including the cause of the incident, if known)</li> <li>the action taken or proposed to be taken to deal with the incident and any resulting environmental harm, if known</li> </ul>	<ul> <li>notifying the CEO orally of the required information; and</li> <li>giving the CEO a written notice containing the required information within 24 hours after the oral notice is given</li> </ul>	Written	CEO of DEPWS environmentalregulation@nt.g ov.au
AMSA Reporting	Titleholder agrees to notify AMSA of any marine pollution incident <sup>49</sup> .	Notification within two hours of incident.	Verbal	AMSA JRCC
	Harmful Substances Report and Situation Report available online (refer to GEP NT Waters OPEP).	Harmful Substances Report as requested by AMSA after verbal notification.	Written	AMSA JRCC

<sup>&</sup>lt;sup>49</sup> For clarity and consistency across Santos' regulatory reporting requirements, Santos will meet the requirement of reporting marine oil pollution by reporting oil spills assessed to have an environmental consequence of Moderate or higher in accordance with Santos' environmental impact and risk assessment process outlined in Section 5.

Initiation	Required Information	Timing	Туре	Recipient
Director of National Parks Reporting Notification of the event of oil pollution within a marine park or where an oil spill response action must be taken within a marine park (requested through consultation)	<ul> <li>The DNP should be made aware of oil and gas pollution incidences that occur within a marine park or are likely to impact on a marine park as soon as possible. Notification should be provided to the 24hour Marine Compliance Duty Officer on 0419 293 465. The notification should include:</li> <li>titleholder details</li> <li>time and location of the incident, including name of marine park likely to be affected</li> <li>proposed response arrangements as per the GEP NT Waters OPEP, such as dispersant, containment</li> <li>confirmation of providing access to relevant monitoring and evaluation reports when available</li> <li>contact details for the response coordinator.</li> </ul> <i>Note:</i> the DNP may request daily or weekly Situation Reports, depending on the scale and severity of the pollution incident.	Verbal notification as soon as reasonably practicable.	Verbal	DNP
DCCEEW Reporting	Notification of any harm or mortality to an EPBC-listed species	Within seven days to	Written	DCCEEW
Any harm or mortality to	of marine fauna, whether attributable to the Activity of hot.	EPBC.permits@environment.gov.au.		
Threatened marine fauna	If matters of national environmental significance are considered at risk from a spill or response strategy, or where there is death or injury to a protected species.	Email notification as soon as practicable.	Written	DCCEEW (Director of monitoring and audit section)
Marine fauna sighting data	Marine fauna sighting data recorded in the marine fauna sighting database.	As soon as practicable.	Written	DCCEEW
	Underwater cultural heritage details recorded in online database if discovered during Activity.	As soon as practicable; in any case, no later than three months after the end of the Activity.	Written	DCCEEW
	Reports to be provided to DCCEEW outlining findings from IMMR activities conducted in the Oceanic Shoals Marine Park	Following IMMR survey Activity (every 3 years)	Written	DCCEEW
Australian Marine Mammal Centre Reporting	Ship strike report provided to the Australian Marine Mammal Centre: <u>https://data.marinemammals.gov.au/report/shipstrike</u> .	As soon as practicable.	Written	DCCEEW
Any ship strike incident with cetaceans will also be reported to the National Ship Strike Database				

Initiation	Required Information	Timing	Туре	Recipient
NT Department of Environment, Parks and Water Security All actual or impending	Verbal reporting will consist of transfer of information to conduct a coordinated emergency response. All reporting will be performed by the vessel master as per the vessel specific SOPEP/SMPEP.	As soon as practicable.	Verbal	NT DEPWS. NT EPA (Pollution Response Hotline; Environmental Operations)
spills in NT waters	Written reports will contain all material facts and circumstances concerning the reportable incident, actions taken to avoid or mitigate any adverse impacts, and corrective action taken.	Written report as soon as practicable.	Written	NT DEPWS, NT EPA (Pollution Response Hotline; Environmental Operations)
NT Department of Environment, Parks and Water Security (DEPWS) Marine wildlife incident or injury	Notification of occurrences of stranded, sick, injured or entangled marine wildlife	As soon as practicable.	Oral	Marine Wild Watch (1800 453 941)
AFMA	Verbal notification if any spill may affect Commonwealth fisheries within the EMBA.	Verbal notification within eight hours.	Verbal	AFMA
DLPE – Heritage Branch (NT) Heritage Act 2011 (NT) section 114 reporting	<ul> <li>Notification of discovery of place or object known to be an Aboriginal or Macassan archaeological place or object.</li> <li>The report to include: <ul> <li>A description of the place or object</li> <li>Its location</li> </ul> </li> <li>The person's name and address</li> <li>If known, the name and address of the owner or occupier of the place where the object is located</li> </ul>	As soon as reasonably practicable.	Written	CEO of DLPE – Heritage Branch (NT)
Wickam Point Deed liaison committee First Nations UCH discovery	Notification of discovery of any First Nations UCH.	As soon as reasonably practicable.	Written	Wickham Point Deed liaison committee
AHO, AMSA	Notification of updates to both AHO and AMSA (JRCC) on progress and, importantly, any changes to the intended operations.	As soon as possible.	Written	AMSA's JRCC AHO
<b>Tiwi Resources</b> (Ranger Coordinator),	Notification of all spills heading towards the Tiwi Islands.	Within 8 hours of incident being identified	Oral – by phone call	Tiwi Resources (Ranger Coordinator), Tiwi Land

Initiation	Required Information	Timing	Туре	Recipient
Tiwi Land Council and Munupi Clan members.	Follow up email notification outlining details of incident.	After oral notification.	Written	Council and nominated Munupi Clan members (per Table 7-1 from GEP NT Waters OPEP), subject to obtaining relevant email addresses.
Greenhouse gas emissions	Reporting of greenhouse gas emissions, energy production and consumption under the NGER Reporting Scheme.	Annually.	Written	Clean Energy Regulator
Pollutant emissions	National Pollution Inventory reporting is lodged as per the National Pollution Inventory submission requirements.	Annually.	Written	National Pollution Inventory
Tiwi Resources (Ranger Coordinator), Tiwi Land Council and the delegated Clan	Notification of all spills heading towards the Tiwi Islands.	Within eight hours of incident being identified.	Verbal phone call notification	Tiwi Resources (Ranger Coordinator), Tiwi Land Council and the delegated Clan Trustees
Trustees	Follow up with email outlining details of incident.	After oral notification.	Written	Tiwi Resources (Ranger Coordinator), Tiwi Land Council and the delegated Clan Trustees (per the GEP NT Waters OPEP), subject to obtaining relevant email addresses
First Nation Consultative Committees	Notification of all spills heading towards the relevant parties interests	Within eight hours of incident being identified.	Verbal phone call notification	First Nation Consultative Committees via Committee Chairs
	Follow up with email outlining details of incident.	After oral notification.	Written	First Nation Consultative Committees, via Committee Chairs (per the GEP NT Waters OPEP), subject to obtaining relevant email addresses
Other First Nations groups as agreed through the post acceptance consultation	Notification of all spills heading towards the relevant parties interests.	Within eight hours of incident being identified.	Verbal phone call notification	Other First Nation Groups, as agreed through the post acceptance consultation implementation process and through the NLC.
and through the NLC	Follow up with email outlining details of incident.	After oral notification.	Written	Other First Nation Groups, as agreed through the post acceptance consultation implementation process and through the NLC.

Initiation	Required Information	Timing	Туре	Recipient
Vocus	Notification in the event of a spill requiring clean-up activities that may impact its infrastructure	As soon as possible	Written	Vocus
Environmental Approval (EPBC 2022/09372) for Darwin Pipeline Duplication Project Compliance Records and annual data reporting	Condition 14 The approval holder must maintain accurate and complete compliance records. Condition 16 The approval holder must ensure that any monitoring data (including sensitive ecological data), surveys, maps, and other spatial and metadata required under the conditions of this approval are prepared in accordance with the Guidelines for biological survey and mapped data, Commonwealth of Australia 2018, or as otherwise specified by the Minister in writing. Condition 17 The approval holder must ensure that any monitoring data (including sensitive ecological data), surveys, maps, and other spatial and metadata required under the conditions of this approval are prepared in accordance with the Guide to providing maps and boundary data for EPBC Act projects, Commonwealth of Australia 2021, or as otherwise specified by the Minister in writing.	Condition 18 The approval holder must submit all monitoring data (including sensitive ecological data), surveys, maps, other spatial and metadata and all species occurrence record data (sightings and evidence of presence) electronically to the department within 20 business days of each anniversary of the date of this approval decision. Condition 15 If the department makes a request in writing, the approval holder must provide electronic copies of compliance records to the department within the timeframe specified in the request.	Written	DCCEW

Initiation	Required Information	Timing	Туре	Recipient
Environmental	Condition 19	Condition 22	Written	DCCEEW
Approval (EPBC 2022/09372) for Darwin Pipeline	The approval holder must prepare a compliance report for each 12-month period following the date of this approval decision (or	The approval holder must:		
Duplication Project	as otherwise agreed to in writing by the Minister).	Condition 22, a)		
Annual Compliance Reporting	Condition 20 Each compliance report must be consistent with the Annual Compliance Report Guidelines, Commonwealth of Australia 2023.	Publish each compliance report on the website within 60 business days following the end of the 12-month period for which that compliance report is required.		
		Condition 22, b)		
	Condition 21 Each compliance report must include:	Notify the department electronically, within 5 business days of the date of publication that a compliance report has been		
	Condition 21, b)	published on the website.		
	Accurate and complete details of compliance and any non- compliance with the conditions and the plans, and any incidents.	Condition 22, c) Provide the weblink for the compliance		
	Condition 21, c)	report in the notification to the department.		
	One or more shapefile showing all clearing of protected matters, and/or their habitat, undertaken within the 12-month period at the end of which that compliance report is prepared.	Condition 22, d) Keep all published compliance reports required by these conditions on the website		
	Condition 21, d)	until the expiry date of this approval.		
	A schedule of all plans in existence in relation to these conditions and accurate and complete details of how each plan is being implemented.	Condition 22, e) Exclude or redact sensitive ecological data from compliance reports published on the website or otherwise provided to a member of the public.		
		If sensitive ecological data is excluded or redacted from the published version, submit the full compliance report to the department within 5 business days of its publication on the website and notify the department in writing what exclusions and redactions have been made in the version published on the website.		

Initiation	Required Information	Timing	Туре	Recipient
Environmental Approval (EPBC 2022/09372) for Darwin Pipeline Duplication Project Independent Audit	Condition 27 For each independent audit, the approval holder must: Condition 27, a) Provide the name and qualifications of the nominated independent auditor, the draft audit criteria, and proposed timeframe for submitting the audit report to the department prior to commencing the independent audit. Condition 27, b) Only commence the independent audit once the nominated independent auditor, audit criteria and timeframe for submitting the audit report have been approved in writing by the department. Condition 27, c) Submit the audit report to the department for approval within the timeframe specified and approved in writing by the department. Condition 28 Each audit report must report for the period preceding that audit report. Condition 29 Each audit report must be completed to the satisfaction of the Minister and be consistent with the 'Environment Protection and Biodiversity Conservation Act 1999 Independent Audit and Audit Report Guidelines, Commonwealth of Australia 2019'.	Condition 26 The approval holder must ensure that an independent audit of compliance with the conditions is conducted at three (3) years after the commencement of the Action, and at any time upon the direction of the Minister. Condition 27 For each independent audit, the approval holder must: Condition 27, d) Publish the audit report on the website within 15 business days of the date of the department's approval of the audit report. Condition 27, e) Keep the audit report published on the website until this approval expires.	Written	DCCEEW

Initiation	Required Information	Timing	Туре	Recipient
Environmental Approval (EPBC 2022/09372) for Darwin Pipeline Duplication Project Reporting Non- Compliance	Condition 24 The approval holder must specify in the notification: Condition 24, a) Any condition or commitment made in a plan which has been or may have been breached. Condition 24, b) A short description of the incident and/or potential non- compliance and/or actual non-compliance. Condition 24, c) The location (including co-ordinates), date and time of the incident and/or potential non-compliance and/or actual non- compliance. Condition 25 The approval holder must provide to the department in writing, within 12 business days of becoming aware of any incident and/or potential non-compliance and/or actual non-compliance, the details of that incident and/or potential non-compliance and/or actual non-compliance with the conditions or commitments made in a plan. The approval holder must specify: Condition 25, d) Any corrective action or investigation which the approval holder has already taken. Condition 25, f) The method and timing of any corrective action that will be undertaken by the approval holder.	Condition 23 The approval holder must notify the department electronically, within 2 business days of becoming aware of any incident and/or potential non-compliance and/or actual non-compliance with the conditions or commitments made in a plan. Condition 25 The approval holder must provide to the department in writing, within 12 business days of becoming aware of any incident and/or potential non-compliance and/or actual non-compliance, the details of that incident and/or potential non-compliance and/or actual non-compliance with the conditions or commitments made in a plan.	Written	DCCEEW
End of the Activity				
OPGGS(E)R Section 54 – Notifications DME must be notified that the activity has ended	Complete NOPSEMA's Section 54 Start or End of Activity Notification Form.	Within ten days after the Activity ends.	Written	NOPSEMA https://securefile.nopsema.go v.au/filedrop/submissions

Initiation	Required Information	Timing	Туре	Recipient
OPGGS(E)R Section 46 OEMP ends when titleholder notifies completion and the regulator accepts the notification DME must be notified that the activity has ended and all OEMP obligations have been completed	Complete NOPSEMA's Section 46 – End of Operation of Environment Plan form <sup>50</sup>	At the completion of the Activity and all EP obligations.	Written	NOPSEMA https://securefile.nopsema.go v.au/filedrop/submissions
AMSA (JRCC) Consultation	Notification that the IMMR activity has completed.	Within ten days of completion.	Written	JRCC
АНО	Notification that the IMMR activity has completed.	Within ten days of completion.	Written	АНО
DAFF	Notification that the IMMR activity has completed.	Within 10 days of completion.	Written	DAFF
DoD	Notification that the IMMR activity has completed.	Within 10 days of completion.	Written	DoD
NT DAF	Notification that the IMMR activity has completed.	Within 10 days of completion.	Written	NT DAF
NTSC	Notification that the IMMR activity has completed.	Within 10 days of completion.	Written	NTSC
Other Marine Users identified in Table 8-14 (as may be updated from time to time).	Notification that IMMR activities have been completed.	Within ten days of completion.	Written	Other Marine Users active in the Operational Area.
Environmental Approval (EPBC 2022/09372) for Darwin Pipeline Duplication Project Completion of the Action	Condition 30 The approval holder must notify the department electronically 60 business days prior to the expiry date of this approval, that the approval is due to expire.	60 business days prior to the expiry date of this approval	Written	DCCEEW

<sup>&</sup>lt;sup>50</sup> https://www.nopsema.gov.au/sites/default/files/documents/Regulation%2046%20-%20End%20of%20Operation%20ef%20Environment%20Plan%20%28A346625%29%20form.docx

Initiation	Required Information	Timing	Туре	Recipient
Environmental Approval (EPBC 2022/09372) for Darwin Pipeline Duplication Project Completion of the action	Condition 31 Within 20 business days after the completion of the Action, and, in any event, before this approval expires, the approval holder must notify the department electronically of the date of completion of the Action and provide completion data. The approval holder must submit any spatial data that comprises completion data as a shapefile.	Within 20 business days after the completion of the Action	Written	DCCEEW

Other Marine Users Communications plan will be implemented by the end of the first quarter of each year, for engagement prior to and during the Activity that may impact marine users, to raise awareness of the activity. The plan will:

- Identify key Barossa activities that may impact other marine users.
- Leverage the regional engagement model and the database of relevant authorities' person and organisations (refer section 8.4.9), to identify other marine users whose activities may be impacted.
- Set out the method/s and frequency/timing of communications (inclusive of activity notifications contained in Table 8-13).
- Communicate the location, timing, and nature of the identified Barossa activities
- Set out person/s accountability to undertake the communications activities set out in the plan.
- Contain mechanism for stakeholders to contact Santos with any queries or complaints.
- Require that records of engagement and complaints be maintained.
- Where practical communications will occur at least two weeks before the activities take place

### Table 8-14: Marine user notification recipients

Person to be issued marine user notifications	Notification recipient
Aquarium Fishery licence-holders (NT)	NTSC and NT DAF
Australian Border Force (ABF)	ABF
Australian Fisheries Management Authority (AFMA)	AFMA
Australian Institute of Marine Science (AIMS)	AIMS
Australian Southern Bluefin Tuna Industry Association (ASBTIA)	ASBTIA
Demersal Fishery licence-holders (NT)	NTSC and NT DAF
Department of Defense – Navy (DoD – Navy)	DoD – Navy
Eni Australia Ltd	Eni Australia Ltd
INPEX Ichthys Pty Ltd	INPEX Ichthys Pty Ltd
Northern Prawn Fishery commercial licence-holders	NPFI
Northern Prawn Fishing Industry Pty Ltd (NPFI)	NPFI
NT Department Mining and Energy	NT DME
NT Department of Agriculture and Fisheries	NT DAF
NT Guided Fishing Industry Association	NT Guided Fishing Industry Association
Northern Territory Seafood Council (NTSC)	NTSC
Offshore Net and Line Fishery licence-holders	NTSC and NT DAF

Person to be issued marine user notifications	Notification recipient
Pearl Oyster Fishery licence-holders	NTSC and NT DAF
Small Pelagic (Development) Fishery licence-holders	NTSC and NT DAF
Southern Bluefin Tuna Fishery licence-holders	ASBTIA and AFMA
Spanish Mackerel Fishery licence-holders	NTSC and NT DAF
Timor Reef Fishery commercial licence holders	NTSC and NT DAF
Top End Tourism	Top End Tourism
Tourism NT	Tourism NT
Vocus	Vocus
Western Skipjack Tuna Fishery licence-holders	ASBTIA and AFMA
Western Tuna and Billfish Fishery licence-holders	ASBTIA and AFMA
Woodside Energy Ltd	Woodside Energy Ltd



### 8.4.7.1 Incident reporting

OPGGS(E)R 2023 Requirements			
Section 24 Other information in the environment plan			
24(c) details of all reportable incidents in relation to the proposed activity.			
Section 47 Notifying reportable incidents			
47(1) A titleholder commits an offence of strict liability if:			
a. the titleholder undertakes an activity under the title; and			
b. there is a reportable incident for the activity; and			
<li>c. the titleholder does not notify NOPSEMA of the reportable incident in accordance with subsection (2).</li>			
Penalty: 40 penalty units.			
47(2) For the purposes of paragraph (1)(c), the notification:			
d. must be given as soon as practicable, and in any case not later than 2 hours, after:			
1. the first occurrence of the reportable incident; or			
<ol><li>if the reportable incident was not detected by the titleholder at the time of the first occurrence—the time the titleholder becomes aware of the reportable incident; and</li></ol>			
e. must be oral; and			
f. must include:			
<ol> <li>all material facts and circumstances concerning the reportable incident that the titleholder knows or is able, by reasonable search or enquiry, to find out; and</li> </ol>			
4. any action taken to avoid or mitigate any adverse environmental impacts of the reportable incident; and			
5. the corrective action that has been taken, or is proposed to be taken, to stop, control or remedy the reportable incident.			
47(3) As soon as practicable after the titleholder notifies a reportable incident, the titleholder must give a written record of the notification to:			
g. NOPSEMA; and			
h. the Titles Administrator; and			
<ul> <li>if the incident occurred in the offshore area of a State—the Department of the responsible State Minister; and</li> </ul>			
<li>j. if the incident occurred in the Principal Northern Territory offshore area—the Department of the responsible Northern Territory Minister.</li>			
47(4) The titleholder is not required to include in the record anything that was not included in the notification.			
Where section 47 of the OPGGS(E)R (above) notes reporting requirements to NOPSEMA, Santos will report to DME, as the relevant regulatory authority for administering the OPGGS(E)R in NT Coastal Waters.			
Santos will also comply with any reporting requirements as per the conditions of EPBC Act approval EPBC 2022/09372 and EP Act approval EP 2022/022-001.			

The requirements for reporting, investigating, and learning from unplanned or uncontrolled events that have or could have resulted in harm to people, the environment or company assets are defined in the Incident Reporting, Procedure.

All personnel will be informed, through inductions and daily operational meetings, of their duty to report HSE incidents and hazards. Reported HSE incidents and hazards will be shared during daily operational meetings and will be documented in Santos' incident management system. HSE incidents will be investigated using root cause analysis.

Environmental recordable and reportable incidents will be reported to DME as required, in accordance with Table 8-13. The incident reporting requirements will be provided to all crew on board the facilities and vessels, with special attention to the reporting timeframes to provide for accurate and timely reporting.

For the purposes of this Activity, in accordance with Section 5 of the OPGGS(E)R:

- A recordable incident for an activity means a breach of an EPO or EPS, in the GEP Coastal Waters OEMP that applies to the Activity, that is not a reportable incident.
- A reportable incident for an activity means an incident relating to the Activity that has caused, or has the potential to cause, moderate to significant environmental damage.



For the purposes of this GEP Coastal Waters OEMP, a reportable incident is one that is assessed to have an environmental consequence of moderate or higher in accordance with the Santos environmental impact and risk assessment process outlined in Section 5. Of the planned and unplanned events assessed within this Coastal Waters OEMP, the items identified to have a potential consequence level of moderate or higher if the event were to occur and would therefore be a reportable incident were:

- introduction of invasive marine species (IV Major in OA)
- surface release of marine diesel oil from a vessel (III Moderate)
- subsea release of gaseous hydrocarbons (III Moderate)

### 8.4.8 Document Management

#### OPGGS(E)R 2023 Requirements

#### Section 52 Storage of records

#### Environment plan

52(1) A titleholder must store an environment plan for an activity under the title, in a way that makes retrieval of the environment plan reasonably practicable, during the following periods:

- when the environment plan is in force for the activity;
- for 5 years beginning on the day that the environment plan ceases to be in force for the activity.

52(2) A titleholder commits an offence of strict liability if the titleholder does not comply with subsection (1). Penalty: 30 penalty units.

#### Records and reports required under provisions of this instrument

52(3) A titleholder must store the following documents, in a way that makes retrieval of the document reasonably practicable, for a period of 5 years beginning on the day the document is given or submitted to NOPSEMA:

- (a) a written record of a notification by the titleholder under section 47;
- (b) a written report given or submitted by the titleholder under section 48, 49, 50 or 51.

52(4) A titleholder commits an offence of strict liability if the titleholder does not comply with subsection (3). Penalty: 30 penalty units.

#### Other records and reports

52(5) A titleholder commits an offence of strict liability if the titleholder:

- (a) creates a record or report mentioned in subsection (7); and
- (b) either:
  - (i) does not store the record or report; or

(ii) stores the record or report in a way that does not make retrieval of the record or report reasonably practicable. Penalty: 30 penalty units.

52(6) Subsection (5) does not apply if the failure to store the record or report, or failure to store it in a way that makes retrieval reasonably practicable, occurs more than 5 years after the day that the record or report was created.

Note: A defendant bears an evidential burden in relation to the matter in subsection (6) (see subsection 13.3(3) of the Criminal Code).

52(7) For the purposes of paragraph (5)(a), the records and reports are the following:

- (a) records relating to environmental performance, or the implementation strategy, under the environment plan in force for an activity under the title;
- (b) records of emissions and discharges into the environment made in accordance with the environment plan in force for an activity under the title;
- (c) records of calibration and maintenance of monitoring devices used in accordance with the environment plan in force for an activity under the title;
- (d) written reports (including monitoring, audit and review reports) about environmental performance, or about the implementation strategy, under the environment plan in force for an activity under the title.



This GEP Coastal Waters OEMP and GEP NT Waters OPEP, as well as approved MoC documents, are controlled documents; current versions will be available on Santos' intranet. Santos' contractors are also required to maintain current versions of these documents.

Environmental performance outcomes and standards will be measured based on the measurement criteria listed in Table 8-2. Such records will be maintained for a period of five years. Contractors are required to make these records available upon request.

### 8.4.8.1 Information management and document control

The Barossa Information Management System is comprised of a suite of applications, configured to support operations and accessible by relevant Santos personnel. Key aspects of relevance to the GEP Coastal Waters OEMP include:

- Engineering Information Management Systems (EIMS)
- Operating manuals and procedures, Operating Procedure Guides and Work Instructions (standards, processes,
- Process surveillance and analysis.
- Santos IRR for operational risk assessment of the subsea infrastructure.
- MoC management systems.
- Incident reporting systems (including HSSE and technical).
- Productions and emissions reporting.

A library of controlled documents specific to the Barossa facility operations, including this GEP Coastal Waters OEMP and the GEP NT Waters OPEP, is accessible by personnel on the FPSO via a dedicated homepage. Each document has an 'Approver' and an 'Owner' who are responsible for the effectiveness of the requirements in that document. All documents will be updated as necessary in response to:

- Requirements for revision in accordance with the relevant Regulations.
- Changes in legislation, standards and/or codes.
- Findings / lessons learned from hazards reported and incident investigations.
- Audit actions.
- Changes to plant, equipment, and operating parameters.
- Changes in operating practices.
- Significant increase in the level of risk associated with a work activity or operation.
- Introduction of new technology, materials, or services.

The current approved version of controlled documents can always be readily located and accessed by the workforce through the information management system. Santos' contractors are also required to maintain current versions of these documents.

### 8.4.8.2 Operations Environmental Management Plan Review

Outside of any GEP Coastal Waters OEMP reviews and revisions triggered by the Environment Plan MoC process (Section 8.5.2), the GEP Coastal Waters OEMP will be reviewed as part of the 5-year major revision and resubmission in line with Regulation 41.

## 8.4.9 Post acceptance consultation implementation strategy

OPGGS(E)R 2023 Requirements	
Section 22 Implementation strategy for environment plan	
Consultation and compliance	
22(15) The implementation strategy must provide for appropriate consultation with:	
a) relevant authorities of the Commonwealth, a State or a Territory; and	
b) other relevant interested persons or organisations.	



Post-acceptance consultation activities for this GEP Costal Waters OEMP will be principally supported by Santos' existing relationships with those relevant interested persons and organisations whose functions, interests and activities may be affected by the Activity.

Santos recognises and respects the preference of relevant government authorities and other relevant interested persons and organisations to determine the frequency and method of updates, in addition to the written quarterly updates outlined in this strategy below.

### 8.4.9.1 First Nations people and groups

Santos will undertake post acceptance consultation over the life of the Activity with First Nations representative organisations.

Santos will provide quarterly written Activity updates via land councils and Aboriginal Corporations, specifically to:

- GDA
- KLC
- LDC
- LNAC
- NLC
- TLC
- Wickham Point Deed liaison committee
- Quarterly written Activity updates will also be provided to:
  - o Tiwi Clan Trustees for each Clan via TLC
  - First Nations Consultative Committees via Committee Chairs.

Having regard to Santos' experience consulting with First Nations groups, and feedback from First Nations relevant persons, Santos considers that consultation through representative bodies provides an appropriate mechanism for ongoing consultation with First Nations relevant interested persons.

Representative bodies provide for regular, culturally appropriate engagement, including processes for dissemination of information to First Nations Elders, cultural leaders and communities in a manner that is readily accessible and culturally appropriate.

### 8.4.9.2 Local governments, communities and industry

As part of Santos' community engagement efforts, Santos will provide quarterly written Activity updates to regional local government and associated communities.

Santos will also provide quarterly written Activity updates to the commercial fishing industry, which is the industry most likely to be affected by proposed offshore activities. Santos will provide quarterly written Activity updates to those representative organisations whose membership are most likely to be affected, specifically to NPFI and NTSC.

### 8.4.9.3 Post-acceptance consultation implementation strategy – approach

Santos will provide to those organisations identified above quarterly written Activity updates. The updates will also be posted on Santos' website, with notifications to registered/subscribed interested parties.

Activity notifications and reports will be made in accordance with Table 8-13 and Table 8-14. The notifications and reports are based on legislative requirements, standing arrangements with particular Relevant Persons, Relevant Persons' requests for notification made during Section 25 of the OPGGS(E)R 2023 consultation or as otherwise deemed appropriate by Santos.

Santos will apply the regional engagement model to consider the preferences of relevant government authorities and other relevant interested persons and organisations when determining the frequency and method of additional updates.

A community lead for each region (e.g. NT Community Affairs Manager) oversees the development and implementation of engagement related plans, such as community investment plan and provision of information updates on Santos' activities. A core aim is to build long term relationships with key local stakeholders through regular engagement.

The regional engagement model is bespoke for each area so it can incorporate the preferences of local stakeholders and updated from time to time to reflect those preferences. For example, the NT model currently



includes the use of a Darwin shopfront which is open to the public and a NT based First Nations Engagement Adviser. These plans also consider the community commitments (e.g. post GEP Coastal Waters OEMP engagement) for each region. For example, the NT model currently includes quarterly meetings with Larrakia people through the Wickham Point Deed liaison committee.

Santos will continue to accept, assess and respond to post acceptance consultation feedback during the life of the Activity under this GEP Coastal Waters OEMP. Records of any post acceptance consultation will be maintained in an appropriate Santos consultation database.

During the GEP Coastal Waters OEMP validity period, Santos will also:

- Review information sources that may give rise to additional or new relevant interested persons or organisations, as part of planned consultation activities to support future approvals.
- Request recipients of Santos' Quarterly Update to advise Santos of other organisations who may be relevant interested persons or organisations or who may be relevant Commonwealth, State or Territory authorities with respect to particular regional activities. The Quarterly Update is sent to a diverse range of organisations and provides information about Santos' proposed, existing and completed activities.

Additional new potentially relevant interested persons or organisations will be engaged and provided information about the accepted activity, as well as information about the post-acceptance consultation process and opportunities to provide input or receive activity updates.

Additional new relevant interested persons or organisations will also be added to the distribution list for its Quarterly Update, unless they request that they not be added.

If, during the course of post acceptance consultation, Santos receives information demonstrating a new or increased environmental impact or risk that is not provided for in this GEP Coastal Waters OEMP, as in force at the time, Santos will apply its MoC process outlined in Section 8.5.2.

Santos will maintain a database of relevant authorities, and other relevant interested persons and organisations for the Activity under this GEP Coastal Waters OEMP. This includes updating its database in light of post acceptance consultation, including identification of new relevant interested persons or organisations.

## 8.5 Act

Within 'Act' are described the processes for ongoing environmental impacts and risks to ensure they remain at ALARP and acceptable levels for the life of the Activity.

## 8.5.1 Non-conformance Management

OEMP non-conformances will be addressed and resolved by a systematic corrective action process, as outlined in Santos' Assurance Operating Standard and the Assurance Procedure. Non-conformances arising from audits and inspections will be entered into Santos' incident and action tracking management system (as in, HSE Toolbox). Once entered, corrective actions, timeframes and responsible persons (including action owners and event validators) will be assigned. Corrective action 'close out' will be monitored using a management escalation process. Any non-conformances that trigger recordable or reportable incident reporting to DME will be managed per Section 8.4.5.

## 8.5.2 Environment Plan Management of Change

The Environment Plan (EP) MoC process is applicable to the GEP Coastal Waters OEMP. It provides a systematic approach to initiate, assess, approve, implement and close out actions associated with a change in the Activity. Implementation of the EP MoC process is designed so that all activities undertaken by Santos is in full compliance with regulatory approvals and conditions and that changes have been properly considered, risk assessed, approved and communicated to all appropriate stakeholders accompanied by a detailed record of the change in Activity.

For any change with identified environmental impacts or risks, an impact / risk assessment will be undertaken to ensure that impacts and risks from the change can be managed to meet relevant EPOs, and be ALARP and acceptable. For a change to proceed, the associated environmental impacts and risks must be demonstrated to be acceptable and ALARP.

The MoC process considers sections 18, 19, 38 and 39 of the OPGGS(E)R 2023 and determines if:

- in respect of existing activities for the purposes of ss 18, 19 and 39, a proposed change can proceed and the manner in which it can proceed; and
- whether a proposed change or activity constitutes a 'new activity' for the purposes of s 38.



The MoC process applies to:

- new activities, assets, equipment, processes or procedures proposed to be undertaken or implemented that have the potential to impact on the environment and have not been assessed or authorised previously
- changes arising from any aspect of the approved activity including proposed changes to activities, assets, equipment, processes or procedures that have the potential to impact on the environment
- changes to the existing environment as relevant to the Activity
- changes to performance standards
- receipt of information form stakeholders about the impacts or risks of the Activity
- new information that becomes available after the GEP Coastal Waters OEMP acceptance
- changes, updates or environmental performance improvements identified from incident investigations, emergency response activities or annual audits
- an assurance check for a specific activity (e.g. annual audits)
- for GHG emissions, changes or updates following changes to Santos' corporate targets and policies regarding scope 3 equivalent emissions, or following the collection of data from suppliers, the DLNG onshore processing facility and customer and evaluation of opportunities to collaborate on emissions reductions initiatives (see BAO-CM-6.3.21 and 6.3.22,)

The MoC procedure also allows for the assessment of information that may become available after acceptance of the GEP Coastal Waters OEMP.

Some of the above circumstances where the MoC process applies are expanded on below as follows:

- When further feedback is received from external stakeholders after OEMP acceptance, consideration will be given as to whether it includes information concerning the environmental impacts or risks of Santos' activities, and if so, whether these impacts or risks are provided for in the relevant approval documentation (e.g. in this GEP Coastal Waters OEMP).
- Santos will also have regard to the guidance in *Munkara* that a 'new' significant environmental impact or
  risk means objective facts and circumstances arising after the approval of the OEMP.<sup>51</sup> If the impact or risk
  is not provided for in the OEMP and is new, the MoC process will be initiated in a timely manner in order for
  the significance of the new information, and any new or increased impacts or risks to be assessed.

The MoC procedure also includes an assurance check process.

• Where there is an identified change from the accepted OEMP content, a check is done to test the 'significance' of the change, to determine whether it can be accommodated which may then result in an MoC as described above.

Once the MoC process is complete:

- Accepted changes become part of the in-force GEP Coastal Waters OEMP or the GEP NT Waters OPEP, are tracked on a register and are made available on Santos' intranet. Where appropriate, the OEMP compliance register will be updated so that CM or EPS changes are communicated to the workforce and implemented. Any MoC will be distributed to the relevant roles identified in Table 8-3, and the most relevant management position is responsible for communication and implementation of the MoC. This may include crew meetings, briefings or communications as appropriate for the change.
- The MoC procedure will determine whether a revision of the OEMP is required and whether that revision must be submitted to DME and assessed by NOPSEMA. Additional consultation with Relevant Persons may be appropriate in order to complete the MoC process, depending on the nature and scale of the change.
- If re-submission of the OEMP to the Regulator is triggered as a result of the MoC assessment, standard
  practice of document revision will apply. If re-submission of the OEMP to the Regulator is not triggered, but
  an update to the latest version of the OEMP is required to reflect the changes from the MoC assessment,
  then this type of revision is called a "minor" revision.

<sup>&</sup>lt;sup>51</sup> Munkara at [232].

Santos Ltd | Barossa Gas Project



### 8.5.3 **Performance Review and Continuous improvement**

Review of environmental performance and the implementation strategy will occur at regular intervals over the life of the Activity as part of existing business processes, or stand-alone reviews specific to the GEP Coastal Waters OEMP. These processes are described below for steady state operations.

### 8.5.3.1 Monthly Operations Governance Forums

Santos Operations Governance requires that the governance process:

- Assess and evaluate adequacy of existing performance indicators and supplement with new performance indicators (as required).
- Track and monitor health of performance indicators against targets.
- Identify gaps and focus areas for improvement, with performance below threshold requiring a plan to improve, endorsed by Leaders in the Business Unit and Function.
- Report and communicate performance in governance forums to facilitate and enable discussion for improvement initiatives.

Operations Governance Forums are held monthly to review operations and maintenance performance and KPIs, and will include KPIs relevant to evaluating the effectiveness of this GEP Coastal Waters OEMP implementation strategy.

### 8.5.3.2 Annual Environmental Performance Reviews

Review of environmental performance with Operations leadership against the requirements of the GEP Coastal Waters OEMP will occur annually in the lead up to the preparation of the annual performance report. The review will be informed primarily by the results of OEMP monitoring and assurance activities, including inspections and audits. These are described in Section 8.4. The purpose of the annual performance review is to review effectiveness of the implementation strategy, test the adequacy of resources to implement the requirements of the OEMP, assess effectiveness of the Environment Management System (EMS) (Section 8.2.1) and to assess if there are any necessary changes to the management of environmental impacts and risks for the activity.

### 8.5.3.3 Continuous Improvement

The scope of the annual performance review process is to also identify opportunities to improve environmental performance over the life of the Activity. Inputs to improvement opportunity identification will include the following:

- improvements identified from quarterly review of HSE key performance indicators (leading and lagging), where KPIs are relevant to management of Activity impacts and risks in this OEMP.
- opportunities for improvement identified from OEMP assurance processes including emissions/discharges monitoring, audits, inspections, environmental incident investigations and after action reviews
- opportunities for improvement identified during Environment Plan MoC reviews (Section 8.5.2)
- emergence of new knowledge or technologies relevant to management of environmental impacts and risks of the Activity
- any changes to Commonwealth or Northern Territory legislation, regulation or policy of relevance to management of impacts and risks from the Activity.
- issues raised during the ongoing consultation process (Section 8.4.9).

To ensure Santos maintains up-to-date knowledge of external sources of information that may be relevant to identification of continuous improvement opportunities, the following is undertaken:

- 1. maintain membership of Australian Energy Producers (formerly Australian Petroleum Production and Exploration Association), which provides a mechanism for communicating potential changes in legislation, industry practice and other issues that may affect OEMP implementation;
- 2. undertake annual spill response exercises, involving relevant external response organisations, to check spill response arrangements and capability are adequate;
- 3. subscribe to various regulator updates; and
- 4. have regular liaison meetings with regulators.



## 8.5.4 Environmental Impact/Risk Review

A review of environmental impacts and risks via an ENVID workshop will be conducted 12 months after commencement of steady state operations, and then once every five years to inform the 5-yearly revision of the GEP Coastal Waters OEMP in line with Regulation 41.

### 8.5.5 Engineering Change Management

Engineering change management processes will be used to scope and guide engineering and operations changes and determine reviews required to appropriately manage risks. Change management is critical to ensure all changes, no matter how minor or small, are effectively captured, assessed, approved, and managed with due rigour by competent people to ensure that any change will not compromise integrity, safe operations, the safety of personnel or protection of the environment. The change management process applies to all technical substitutions, modifications, additions, or deletions of plant or equipment; systems and procedures; critical positions or personnel.

Each change has an owner responsible for managing the change through its lifecycle, from inception through to ultimate close-out. Change requests evaluations shall include operations and technical disciplines as appropriate to the requested change. Appropriate communication and consultation with the parties affected by the change will also be conducted. Close-out will confirm that any lessons learned from implementation are captured and communicated appropriately.

There will be two complementary change management of processes in place:

- Santos Engineering Change Management Process
  - Utilised for STO change impacting field wide operations (including regulatory etc)
  - Covers multiple scopes (Subsea, GEP etc), ensuring integration with FPSO operations where required to maintain field-wide integrity and regulatory compliance
  - References supporting BWO process where required
- BWO Engineering Change Management Process
  - The BWO Engineering Change Management Process applies to FPSO-specific modifications. Where an FPSO change has potential implications for the GEP pipeline or subsea infrastructure, integration with the Santos Engineering Change Management Process will be required
  - Focused on FPSO-specific modifications, with Santos oversight where broader field-wide impacts exist.
  - Links to the Santos engineering change management process when FPSO-initiated changes impact the GEP, subsea infrastructure, or other field-wide assets

Any change that spans both the FPSO and subsea (including GEP), or that is relevant to requirements of the GEP Coastal Waters OEMP, will be governed under the Santos engineering change management process ensuring regulatory compliance and alignment with field-wide integrity management.

Both change management processes include an HSE review at initiation, which will determine whether potential changes impact achievement or verification of environmental performance standards or compliance with an environmental performance outcome. Any change that results in an outcome that may affect management of impacts and risks of the Activity will trigger the Environment Plan MoC process (Section 8.5.2).

Each change has an owner responsible for managing the change through its lifecycle, from inception through to ultimate close-out. Change requests evaluations shall include operations and technical disciplines as appropriate to the requested change. Appropriate communication and consultation with the parties affected by the change will also be conducted.

### 8.5.6 Decarbonisation Opportunity Management

Each Santos Regional Business Unit (RBU) is required to submit a decarbonisation plan for each regional asset as part of the annual corporate Long Term Planning Process. These plans contain an articulation of asset specific carbon emissions forecasts, material carbon sources and decarbonisation strategy; and outline opportunities to meet regulatory requirements and corporate targets. The plans outline identified projects to achieve regulatory compliance, such as Australia's Safeguard Mechanism, and Santos' path to net-zero Scope 1 and 2 emissions (avoid & reduce emissions). Any remaining gap to net-zero Scope 1 and 2 emissions is evaluated for offset solutions.

Each RBU's decarbonisation opportunities will vary depending upon their business. RBUs will first look to avoid emissions and then to reduce by implementing energy efficiency and lower emission technologies such as CCS (carbon capture and storage). Operating efficiency projects can include reducing fuel, flare and vent (out of scope of this OEMP), electrification (out of scope of this OEMP), improving fuel efficiency and other projects which

improve energy efficiency in operations. Opportunities are added to an asset-specific decarbonisation plan (given that this section of the GEP is part of the Barossa facilities), where they are evaluated and prioritised within the constraints of technical feasibility, cost, resource availability and other factors. Asset managers are responsible for identifying asset specific decarbonisation opportunities for all assets, including Barossa, for inclusion in RBU decarbonisation plans. RBU long-term plans, including the decarbonisation opportunities, are consolidated by the Corporate planning group and then reviewed and discussed with executive leadership to determine the best way to achieve required and targeted emissions reductions.

Based upon this discussion, guidance is provided to the RBUs on projects to include in their budget noting that only sanctioned major projects are considered committed (sanctioned major projects have progressed through the corporate project development process and have an approved final investment decision). The corporate group also considers new technology and other initiatives that may be outside the business unit remit to decarbonise. Sanctioned projects must meet minimum investment criteria set by the corporation based upon several metrics including net present value, internal rate of return, pay back, capital efficiency and marginal abatement cost (for decarbonisation projects). The marginal abatement cost is used to create a cost curve comparing the relative costs of individual projects to understand cost-effectiveness of options.

Projects that do not meet the capital threshold can be approved through the regional business unit's annual budget.

Project opportunity lists are maintained by each RBU and a compilation of current/ongoing opportunities are included in the annual decarbonisation plan. Following delivery, opportunities are validated prior to closeout.

As an individual asset, Barossa will form part of the RBU decarbonisation plan. For Barossa, opportunities identified during the Design/Execute phases, which remain applicable during the Operations phase, will be captured in the applicable RBU decarbonisation plan. Additionally, and as a new facility, a post-start-up workshop will be held within the first 18 months of Barossa steady-state operations for opportunity identification, once the facility is well understood and areas for long-term emissions intensity improvement become evident.

During steady-state operations, opportunities will be added to a formal tracking register where they are evaluated and prioritised, having regard to technical feasibility, cost, resource availability and other factors such as synergistic benefits e.g. opportunities that provide a GHG emissions and atmospheric emissions benefit. Improvement opportunities identified from the performance target setting process (Section 8.2.4) will assist with identification of decarbonisation opportunities over the life of the Activity.

Santos reviews and revises its internal planning processes and incorporates changes in requirements and timing as needed to meet changing environmental and internal requirements and to continuously improve internal processes. RBUs are accountable for their asset decarbonisation plans and corporate teams are accountable for validation of these plans and estimated emission reductions. These processes are subject to external audit.

Santos reports annually on its Climate initiatives. The most recent Santos Climate Report can be found on page 68 of the 2024 Annual report: <u>https://www.santos.com/wp-content/uploads/2025/02/FINAL-Appendix-4E-and-2024-Annual-Report.pdf</u>.



# 9. References

- AAPA (Aboriginal Areas Protection Authority) (2022). What is a sacred site. Available at: https://www.aapant.org.au/sacredsites.
- AAPA (Aboriginal Areas Protection Authority) (2023). Abstract of records Tiwi islands 202308046.
- Air Services Australia, 2020. Short-Term Noise Monitoring Program, Strathmore Heights, Essendon Airport, Melbourne. Version 1.0. Available online: <u>https://www.airservicesaustralia.com/wp-content/uploads/Short-Term-Monitoring-Program-</u> Strathmore-Heights-VIC-September-2020.pdf.
- AMSA (Australian Maritime Safety Authority) (2024). Vessel Traffic Data May 2024. Available at: https://www.operations.amsa.gov.au/spatial/DataServices/DigitalData
- Anderson PK (1982) 'Studies of dugongs at Shark Bay, Western Australia I. Analysis of population size, composition, dispersion and habitat use on the basis of aerial survey', *Australian Wildlife Research*, 9(1): 69–84, doi: https://doi.org/10.1071/WR9820069
- Anderson PK (1997) 'Shark Bay dugongs in summer. I: Lek mating', *Behaviour*, 134(5): 433–462, doi: https://doi.org/10.1163/156853997X00629
- Anderson, TJ, Nichol, S, Radke, L, Heap, AD, Battershill, C, Hughes, M, Siwabessy, PJ, Barrie, V, Alvarez de Glasby, B, Tran, M, Daniell, J & Shipboard Party, (2011) Seabed environments of the eastern Joseph Bonaparte Gulf, northern Australia: GA0325/Sol5117-post survey report., record 2011/08, Geoscience Australia, Canberra.
- Ansell DV, Dicks B, Guenette CC, Moller TH, Santner RS and White IC (2001) 'A Review of the Problems Posed by Spills of Heavy Fuel Oils', International Tanker Owners Pollution Federation (ITOPF), Paper presented at 2001 International Oil Spill Conference, March 26–29 2001, Tampa Florida, available from https://www.itopf.org/fileadmin/data/Documents/Papers/iosc2001.pdf
- Armstrong, A.J., Armstrong, A.O., McGregor, F., Richardson, A.J., Bennett, M.B., Townsend, K.A., Hays, G.C., van Keulen, M., Smith, J. & Dudgeon C.L. (2020). Satellite Tagging and Photographic Identification Reveal Connectivity Between Two UNESCO World Heritage Areas for Reef Manta Rays. Front. Mar. Sci. 7:725.
- Aubé, M., Franchomme-Fosse, L., Robert-Staehler, P. and Houle, V. (2005). Light pollution modelling and detection in a heterogeneous environment: Toward a night-time aerosol optical depth retrieval method. Proc. SPIE 2005, 5890, 248– 256, https://doi.org/10.1117/12.615405.
- Aulich, M.G., McCauley, R.D., Saunders, B.J. & Parsons, M.J. (2019). Fin whale (Balaenoptera physalus) migration in Australian waters using passive acoustic monitoring. Sci Rep 9, 8840.
- Aulich, M.G., McCauley, R.D., Miller, B. S., Samaran, F., Giacomo, G., Saunders, B. J., & Erbe, C. (2022). Seasonal Distribution of the Fin Whale (Balaenoptera physalus) in Antarctic and Australian Waters Based on Passive Acoustics. Front. Mar. Sci., 9.
- Aumann, T. and Baker-Gabb, D.J. (1991) The ecology and status of the Red Goshawk in northern Australia. Royal Australasian Ornithologists Union: Melbourne.
- Australian and New Zealand Guidelines (ANZG) (2018) Australian and New Zealand Guidelines for Fresh and Marine Water Quality. https://www.waterquality.gov.au/anz-guidelines
- Baker C, Potter A, Tran M, & Heap AD. (2008). Geomorphology and sedimentology of the North-west Marine Region of Australia. Record 2008/07, Geoscience Australia, Canberra, A
- Baldwin R., Hughes, G.R. & Prince, R.I.T. (2003). Loggerhead turtles in the Indian Ocean. Bolten, I.A.B & Witherington, B.E (eds) Loggerhead Sea Turtles, Smithsonian Books, Washington.
- Bamford, M., Watkins, D., Bancroft, W., Tischler, G. & Wahl, J. (2008). Migratory Shorebirds of the East Asian-Australasian Flyway Migratory Shorebirds of the East Asian – Australasian Flyway; Population Estimates and Internationally Important Sites. Available from: www.wetlands.org/publications/migratory-shorebirds-of-the-east-asian-australasian-flywaypopulation -estimates-and-internationally-important-sites (Accessed: 10/07/23).
- Bannister, J.L., Kemper, C.M., & Warneke, R.M. (1996). The Action Plan for Australian Cetaceans. Australian Nature Conservation Agency, Canberra, Australian Capital Territory.
- Bartes S & Braccini M (2021) Potential expansion in the spatial distribution of subtropical and temperate west Australian sharks. Journal of Fish Biology 2021, 1–4.
- Bell Helicopter, 2012. Bell 412EP Product Specifications. Available online: https://web.archive.org/web/20120911170639/http://www.bellhelicopter.com/MungoBlobs/38/570/412E P\_ProductSpecs02-2012.pdf.
- Bennelongia (2011). Port Hedland Migratory Shorebird Survey Report and Impact Assessment. Prepared for BHP Billiton Iron Ore. Available from: www.bhp.com/-/media/bhp/regulatory-information-media/iron-ore/western-australia-ironore/0000/supporting-documentation-for-response-to-submissions-technical-appendices-1-16/appendix-5-port-hedlandshorebird-assessment.pdf, accessed March 2024
- Birdlife Australia (2020). National Directory of Important Migratory Shorebird Habitat. Prepared for Department of Agriculture Water and the Environment, August 2020.



- Birkmanis, C.A., Partridge, J.C., Simmons, L.W., Heupel, M.R. & Sequeira, A. (2020). Shark conservation hindered by lack of habitat protection. Global Ecology and Conservation, 21, e00862.
- Blake, S., Geary, N. and Wilczynska, M., 2022. Standard Operating Procedure for Offshore Chemical Hazard Assessment. Part 1: Core Elements. Centre for Environment, Fisheries & Aquaculture Science (CEFAS). Netherlands (NL) Protocol OCNS 011.
- Bond, T., Partridge, J.C., Taylor, M.D., Langlois, T.J., Malseed, B.E., Smith, L.D., McLean, D.L. (2018). Fish associated with a subsea pipeline and adjacent sea floor of the North West Shelf of Western Australia. Marine Environmental Research. Available from: https://doi.org/10.1016/j.marenvres.2018.08.003.
- Bouchet, P., Thiele, D., Marley, S., Waples, K., Weisenberger, F., Balanggarra Rangers, Bardi Jawi Rangers, Dambimangari Rangers, Nyamba Baru Rangers, Nyul Nyul Rangers, Uunguu Rangers, & Raudino, H. (2021). Regional Assessment of the Conservation Status of Snubfin Dolphins (Orcaella heinsohni) in the Kimberley Region, Western Australia. Front. Mar. Sci. 7.
- Bureau of Meteorology (BoM) (2025) Summary statistics Pirlangimpi Airport (Melville Island). Australian Government Bureau of Meteorology, Canberra. Available from: www.bom.gov.au (Accessed: 10/02/25).
- Burgess, K.B., Couturier, L.I.E., Marshall, A.D., Richardson, A.J., Weeks, S.J., Bennett, M.B., (2016). Manta birostris, predator of the deep? Insight into the diet of the giant manta ray through stable isotope analysis. Royal Society Open Science. doi:10.1098/rsos.160717.
- Burton, C., Bouchet, P. J., Gill, P., & Marley, S. A. 2023. Evidence of likely foraging by pygmy blue whales in the Timor Trough during the late austral winter and early austral spring. Marine Ecology Progress Series, 718, 99-117
- Butler, I, Patterson, H, Bromhead, D, Galeano, D, Timmiss, T, Woodhams, J and Curtotti, R (2023). Fishery status reports 2023, Australian Bureau of Agricultural and Resource Economics and Sciences, Canberra. CC BY 4.0. https://doi.org/10.25814/vgp4-xr81.
- Cáceres-Farias, L., Reséndiz, E. & Espinoza, J. (2022). Threats and Vulnerabilities for the Globally Distributed Olive Ridley (*Lepidochelys olivacea*) Sea Turtle: A Historical and Current Status Evaluation. Animals. 12(14):1837.Calnan, T. (2006). An assessment of the conservation values of the Gunn Peninsula/Vernon Islands area and the impacts of the proposed Glyde Point heavy industry and residential estate. Prepared for Environment Centre NT and Australian Marine Conservation Society, Darwin.
- Calnan, T (2006) An assessment of the conservation values of the Gunn Peninsula/Vernon Islands area and the impacts of the proposed Glyde Point heavy industry and residential estate. Prepared for Environment Centre NT (ECNT) and Australian Marine Conservation Society (AMCS), Darwin.
- Cannell, B., Hamilton, S. and Driessen, J. (2019). Wedge-tailed Shearwater Foraging Behaviour in the Exmouth Region. Report for Woodside Energy Ltd. University of Western Australia and Birdlife Australia, 36pp <u>https://direct.birdlife.org.au/documents/wedge-tailed%20shearwater%20foraging%20behaviour.pdf</u>
- Cardno (2015). Ichthys Nearshore Environmental Monitoring Program: Turtle and dugong post-dredging report. Report prepared by Cardno (NSW/ACT) Pty Ltd, Sydney, for INPEX Operations Australia Pty. Ltd., Perth
- Ceccarelli, D., McCrea, I., Collis, M. & Nicoll, R. (2011). Australia's last great whale haven Cetacean distribution and conservation needs in the north-west marine region. International Fund for Animal Welfare, 1–68.
- CER (Clean Energy Regulator). (2024). GHG Scope definitions. <u>https://cer.gov.au/schemes/national-greenhouse-and-energy-</u> reporting-scheme/about-emissions-and-energy-data/emissions-and-energy-types#types-of-emissions, accessed April 2024.
- Chapman, C.C., Lea, M.A., Meyer, A., Sallée, J.B. and Hindell, M., 2020. Defining Southern Ocean fronts and their influence on biological and physical processes in a changing climate. Nature Climate Change, 10(3).
- Chatto, R. (2001). The distribution and status of colonial breeding seabirds in the Northern Territory. Technical Report 70, Parks and Wildlife Commission of the Northern Territory, Darwin.
- Chatto, R. (2003). The distribution and status of shorebirds around the coast and coastal wetlands of the Northern Territory. (Technical Report 73, Parks and Wildlife Commission of the Northern Territory, Darwin).
- Chatto, R. & Baker, B. (2008). The distribution and status of marine turtle nesting in the Northern Territory (Technical Report No. 77). Department of Natural Resources, Environment, the Arts and Sport. Darwin.
- Chevron (2011). Final Environmental Impact Statement/Response to Submissions on the Environmental Review and Management Programme for the Proposed Wheatstone Project. Chevron.
- Church, J.A., Hunter, J.R., Mcinnes, K. and White, N.J., 2006. Sea-level rise around the Australian coastline and the changing frequency of extreme events. Australian Meteorological Magazine 55, 253–260. <u>https://doi.org/10.1016/j.gloplachs.2006.04.001</u>.
- Clarke, R. (2011). Complete list of birds recorded near or at Ashmore Reef. Available from: www.dcceew.gov.au/sites/files/env/pages/files/ashmore-reef-bird-list.pdf. (Accessed: 10/07/23).
- CoA (Commonwealth of Australia) The Kenbi (Cox Peninsula) Land Claim No. 37. Report and recommendations of the former Aboriginal Land Commissioner, Justice Gray, to the Minister for Aboriginal and Torres Strait Islander Affairs and to the Administrator of the Northern Territory. December 2000. https://nla.gov.au/nla.cat-vn1185816



- CoA (Commonwealth of Australia) (2005). PB26- Northwest IMCRA Transition. Department of the Environment and Heritage, Canberra, Australia.
- CoA (Commonwealth of Australia) (2006). A Guide to the Integrated Marine and Coastal Regionalisation of Australia Version 4.0. Department of the Environment and Heritage, Canberra, Australia.
- CoA (Commonwealth of Australia) (2012). Marine bioregional plan for the North Marine Region. Department of the Environment, Water, Heritage and the Arts, Canberra, Australian Capital Territory.
- CoA (Commonwealth of Australia) (2015a). Conservation Management Plan for the Blue Whale. Commonwealth of Australia.
- CoA (Commonwealth of Australia) (2015b). Sawfish and River Sharks Multispecies Recovery Plan. Department of the Environment, Canberra.
- CoA (Commonwealth of Australia) (2015c). Wildlife Conservation Plan for Migratory Shorebirds. Department of the Environment, Canberra.
- CoA (Commonwealth of Australia) (2017a). National Strategy for Reducing Vessel Strike on Cetaceans and other Marine Megafauna 2017. Commonwealth of Australia.
- CoA, (Commonwealth of Australia). (2017b). Recovery Plan for Marine Turtles in Australia 2017–2027. Department of Climate Change, Energy, the Environment and Water, Canberra.
- CoA (Commonwealth of Australia) (2020). Wildlife Conservation Plan for Seabirds. Commonwealth of Australia, Canberra.
- CoA (Commonwealth of Australia) (2021). National Indigenous Australians Agency, Woolner/ Mary River Region Land Claim No. 192. Report of the Aboriginal Land Commissioner, the Hon John Mansfield AM QC, to the Minister for Indigenous Australians and to the Administrator of the Northern Territory. December 2021. https://www.niaa.gov.au/sites/default/files/publications/alc-woolner-mary-river-report.pdf
- CoA (Commonwealth of Australia) (2023) Wulna (Northern Territory Portion 2001) Land Claim No. 155. Report and recommendations of the Hon John Mansfield AM QC, to the Minister for Indigenous Australians and to the Administrator of the Northern Territory. December 2023. https://www.niaa.gov.au/sites/default/files/publications/wulna-land-claim-155.pdf
- CoA (Commonwealth of Australia) (2023). Aboriginal Land Commissioner Report for the year ended 30 June 2023. Available from: www.niaa.gov.au/sites/default/files/documents/publications/4387%20-%20ALC%20Annual%20Report%202022-23\_ACCv4.pdf, accessed July 2024.
- Coastal Frontiers, 2017. Request for approval, Geophysical Survey Activities near Encinitas, CA. http://www.slc.ca.gov/Programs/OGPP/Survey/2017/CoastalFrontiers\_PreSurveyNotification\_DecJan1718.pdf.
- ConocoPhillips (2018) Barossa Area Development Offshore Project Proposal. ConocoPhillips, Perth. https://info.nopsema.gov.au/offshore\_projects/26/show\_public
- Commonwealth of Australia (CSIRO and BOM), 2024. State of the Climate 2024. Published by the Commonwealth of Australia
- ConocoPhillips (2019) Tiwi Islands Sensitivity Mapping. Unpublished report prepared by Jacobs for ConocoPhillips. Document Number BAA-200 0255.
- Corrigan (2023) General Direction # 1898. Assessment to identify any underwater cultural heritage places along the Barossa pipeline route to the west and northwest of the Tiwi Islands, Northern Australia. 15 September 2023. https://www.santos.com/wp-content/uploads/2023/10/Dr-Corrigan-Report-and-Addendum-Redacted.pdf
- Corrigan (2024). First Nations spiritual and cultural values in relation to the Darwin Pipeline Duplication Project. Report prepared for Santos Ltd., Perth, Western Australia.
- Cosmos Archaeology (2022). Santos (Barossa) Gas Export Pipeline Additional and Nearshore Barossa GEP Stage (Beagle Gulf and Darwin Harbour) Maritime Heritage Assessment. Report prepared for Santos Ltd., Perth, Western Australia.
- Cosmos Archaeology (2023). Santos (Barossa) Offshore Development Infield Infrastructure Maritime Heritage Assessment.
- Costello, M.J. and Read, P (1994). Toxicity of sewage sludge to marine organisms: A review, Marine Environmental Research, 37(1), 23–46. Available from: https://doi.org/10.1016/0141-1136(94)90061-2.
- Couturier, L., Jaine, F. & Kashiwagi, T. (2015). First photographic records of the giant manta ray Manta birostris off eastern Australia. PeerJ 3:e742.
- CSIRO (Commonwealth Scientific and Industrial Research Organisation) Bureau of Meteorology (2017). Climate change in Australia information for Australia's natural resource management regions. CSIRO, Canberra.
- DAFF (Department of Agriculture, Fisheries and Forestry) (2023). Australian biofouling management requirements (Version 2). Department of Agriculture, Fisheries and Forestry, Canberra. Available at: www.agriculture.gov.au/sites/default/files/documents/Australian-biofouling-management-requirements.pdf
- DAWE (Department of Agriculture, Water and the Environment) (2020). Australian Ballast Water Management Requirements. Version 8. Department of Agriculture, Water and the Environment, Canberra.
- DAWE (Department of Agriculture, Water and the Environment) (2018). Marine Pest Plan 2018–2023: the National Strategic Plan for Marine Pest Biosecurity, Department of Agriculture and Water Resources, Canberra, May. CC BY 4.0.



- DCCEEW (Department of Climate Change, Energy, Environment and Water) (2021). State of the Environment. Commonwealth of Australia. <u>https://soe.dcceew.gov.au/</u>.
- DCCEEW (Department of Climate Change, Energy, the Environment and Water) (2022). Quarterly Update of Australia's National Greenhouse Gas Inventory: December 2022, DCCEEW. Available from: https://www.dcceew.gov.au/sites/default/files/documents/nggi-guarterly-update-dec-2022.pdf, accessed July 2023.
- DCCEEW (Department of Climate Change, Energy, the Environment and Water) (2022a). What's protected under the EPBC Act. Australian Government, Canberra. https://www.dcceew.gov.au/environment/epbc/what-is-protected
- DCCEEW (Department of Climate Change, Energy, the Environment and Water) (2022a). National Recovery Plan for the Australian Painted Snipe (*Rostratula australis*). DCCEEW. Available from: www.dcceew.gov.au/sites/default/files/documents/national-recovery-plan-australian-painted-snipe-2022.pdf, accessed 2024
- DCCEEW (Department of Climate Change, Energy, the Environment and Water) (2022b) Species Profile and Threats Database. Available from: https://www.environment.gov.au/sprat. Accessed January to February 2023.
- DCCEEW (Department of Climate Change, Energy, the Environment and Water) (2023a) Guidelines for working in the near and offshore environment to protect Underwater Cultural Heritage, Department of Climate Change, Energy, the Environment and Water, Canberra, October CC BY 4.0. Available at: dcceew.gov.au/publications.
- DCCEEW (Department of Climate Change, Energy, the Environment and Water) (2023b) Directory of Important Wetlands in Australia. Available from: https://www.dcceew.gov.au/water/wetlands/australian-wetlands-database/directory-important-wetlands. Accessed: 20/09/23
- DCCEEW (Department of Climate Change, Energy, the Environment and Water) (2023c) National Light Pollution Guidelines for Wildlife. Available from: https://www.dcceew.gov.au/sites/default/files/documents/national-light-pollution-guidelineswildlife.pdf
- DCCEEW (Department of Climate Change, Energy, the Environment and Water) (2023d) Australia's Emissions Projections 2023. DCCEEW, Canberra. Available from: <u>www.dcceew.gov.au/sites/default/files/documents/australias-emissions-</u> <u>projections-2023.pdf</u>
- DCCEEW (Department of Climate Change, Energy, the Environment and Water) (2023e). Conservation Advice for *Charadrius leschenaultii* (greater sand plover). Canberra: Department of Climate Change, Energy, the Environment and Water. Available from: <u>http://www.environment.gov.au/biodiversity/threatened/species/pubs/877-conservation-advice-18122023.pdf</u>.
- DCCEEW (Department of Climate Change, Energy, the Environment and Water) (2023f). Conservation Advice for *Erythrotriorchis radiatus* (red goshawk). Canberra: Department of Climate Change, Energy, the Environment and Water. Available from: https://www.environment.gov.au/biodiversity/threatened/species/pubs/942-conservation-advice-31032023.pdf
- DCCEEW (Department of Climate Change, Energy, the Environment and Water) (2024a). Conservation Advice for *Charadrius* mongolus (lesser sand plover). DCCEEW, Canberra. Available from: www.environment.gov.au/biodiversity/threatened/species/pubs/879-conservation-advice-05052016.pdf
- DCCEEW (Department of Climate Change, Energy, the Environment and Water) (2024b). Conservation Advice for *Calidris acuminata* (sharp-tailed sandpiper). DCCEEW, Canberra. Available from: www.environment.gov.au/biodiversity/threatened/species/pubs/874-conservation-advice-05012024.pdf
- DCCEEW (Department of Climate Change, Energy, the Environment and Water) (2024c). Conservation Advice for *Calidris canutus* (red knot). DCCEEW, Canberra. Available from: www.environment.gov.au/biodiversity/threatened/species/pubs/855-conservation-advice-05012024.pdf
- DCCEEW (Department of Climate Change, Energy, the Environment and Water) (2024d). Conservation Advice for *Calidris tenuirostris* (great knot). DCCEEW, Canberra. Available from: www.environment.gov.au/biodiversity/threatened/species/pubs/862-conservation-advice-05012024.pdf
- DCCEEW (Department of Climate Change, Energy, the Environment and Water) (2024e). Conservation Advice for *Limosa limosa* (black-tailed godwit). DCCEEW, Canberra. Available from: www.environment.gov.au/biodiversity/threatened/species/pubs/845-conservation-advice-05012024.pdf
- DCCEEW (Department of Climate Change, Energy, the Environment and Water) (2024f). Conservation Advice for *Limnodromus semipalmatus* (Asian dowitcher). DCCEEW, Canberra. Available from: www.environment.gov.au/biodiversity/threatened/species/pubs/843-conservation-advice-05012024.pdf
- DCCEEW (Department of Climate Change, Energy, the Environment and Water) (2024g). Conservation Advice for *Pluvialis* squatarola (grey plover). DCCEEW, Canberra. Available from: www.environment.gov.au/biodiversity/threatened/species/pubs/865-conservation-advice-05012024.pdf
- DCCEEW (Department of Climate Change, Energy, the Environment and Water) (2024h). Conservation Advice for *Tringa nebularia* (common greenshank). DCCEEW, Canberra. Available from: www.environment.gov.au/biodiversity/threatened/species/pubs/832-conservation-advice-05012024.pdf
- DCCEEW (Department of Climate Change, Energy, the Environment and Water) (2024i). Conservation Advice for *Xenus cinereus* (terek sandpiper). DCCEEW, Canberra. Available from: www.environment.gov.au/biodiversity/threatened/species/pubs/59300-conservation-advice-05012024.pdf



- DCCEEW (Department of Climate Change, Energy, the Environment and Water) (2024j). Species Profile and Threats Database, Department of the Environment, Canberra. Available from: https://www.environment.gov.au/cgi-bin/sprat/public/sprat.pl. Accessed: 04/07/23.
- DCCEEW (Department of Climate Change, Energy, the Environment and Water) (2024k). Conservation Advice for *Limosa lapponica baueri* (Alaska bar-tailed godwit). DCCEEW, Canberra. Available from: http://www.environment.gov.au/biodiversity/threatened/species/pubs/86380-conservation-advice-05012024.pdf
- DCCEEW (Department of Climate Change, Energy, the Environment and Water) (2024I). Conservation Advice for Arenaria interpres (ruddy turnstone). Canberra: Department of Climate Change, Energy, the Environment and Water. Available from: http://www.environment.gov.au/biodiversity/threatened/species/pubs/872-conservation-advice-05012024.pdf. In effect under the EPBC Act from 05-Jan-2024.
- Debus, S.J. and Czechura, G.V. (1988). The red goshawk Erythrotriorchis radiatus: a review. Australian Bird Watcher 12
- Deutsch, C. J., Castelblanco-Martínez, D. N., Groom, R., & Cleguer, C. (2022). Movement behavior of manatees and dugongs:
   I. Environmental challenges drive diversity in migratory patterns and other large-scale movements. In Ethology and behavioral ecology of sirenia (pp. 155-231). Cham: Springer International Publishing.
- Department of Natural Resources, Environment, The Arts and Sport (NRETAS). (2009a). Sites of Conservation Significance Tiwi Islands. Department of Natural Resources, Environment, The Arts and Sport, Darwin, Northern Territory. Available at: http://www.lrm.nt.gov.au/\_\_data/assets/pdf\_file/0017/13931/09\_tiwi.pdf.
- DEPWS (Department of Environment, Parks and Water Security) (2021a). Threatened species of the Northern Territory. Australian painted-snipe *Rostratula australis*. Northern Territory Government, Darwin.
- DEPWS (Department of Environment, Parks and Water Security) (2021b). Threatened species of the Northern Territory. Bartailed godwit (western Alaskan) Limosa Iapponica baueri. Northern Territory Government, Darwin.
- DEPWS (Department of Environment, Parks and Water Security) (2021b). Threatened species of the Northern Territory. Hooded robin (Tiwi) Melanodryas cucullata melvillensis. Northern Territory Government, Darwin.
- DEPWS (Department of Environment, Parks and Water Security) (2021c). Threatened species of the Northern Territory. Great knot *Calidris tenuirostris*. Northern Territory Government, Darwin.
- DEPWS (Department of Environment, Parks and Water Security) (2021d). Threatened species of the Northern Territory. Lesser sand plover *Charadrius mongolus*. Northern Territory Government, Darwin.
- DEPWS (Department of Environment, Parks and Water Security) (2021e). Threatened species of the Northern Territory. Masked owl (northern mainland) *Tyto novaehollandiae Kimberli*. Northern Territory Government, Darwin.
- DEPWS (Department of Environment, Parks and Water Security) (2021f). Threatened species of the Northern Territory. Masked owl (Tiwi) *Tyto novaehollandiae melvillensis.* Northern Territory Government, Darwin.
- DEPWS (Department of Environment, Parks and Water Security) (2021g). Threatened species of the Northern Territory. Red goshawk *Erythrotriorchis radiates*. Northern Territory Government, Darwin.
- DEPWS (Department of Environment, Parks and Water Security) (2021f). Threatened species of the Northern Territory. Masked owl (Tiwi) Tyto novaehollandiae melvillensis. Northern Territory Government, Darwin.
- DEPWS (Department of Environment, Parks and Water Security) (2021g). Threatened species of the Northern Territory. Red goshawk *Erythrotriorchis radiates*. Northern Territory Government, Darwin.
- DEPWS (Department of Environment, Parks and Water Security) (2021i). Threatened species of the Northern Territory. Grey falcon Falco hypoleucos. Northern Territory Government, Darwin.
- DEWHA (Department of the Environment, Water, Heritage, and the Arts) (2008a) North-West Marine Bioregional Plan: Bioregional Profile: A Description of the Ecosystems, Conservation Values and Uses of the North-West Marine Region. Department of Environment Water Heritage and the Arts, Canberra, Australian Capital Territory.
- DEWHA (Department of the Environment, Water, Heritage and the Arts) (2008b). Approved Conservation Advice for *Dermochelys coriacea* (Leatherback Turtle). Canberra: Department of the Environment, Water, Heritage and the Arts. Available from: http://www.environment.gov.au/biodiversity/threatened/species/pubs/1768-conservation-advice.pdf. In effect under the EPBC Act from 08-Jan-2009.
- DEWHA (Department of the Environment, Water, Heritage and the Arts) (2008c). Approved Conservation Advice for Green Sawfish. Department of the Environment, Water, Heritage and the Arts, Canberra.
- DEWHA (Department of the Environment, Water, Heritage and the Arts) (2009). Approved Conservation Advice for *Pristis clavata* (Dwarf Sawfish). Department of the Environment, Water, Heritage and the Arts, Canberra.
- DISER (Department of Industry, Science, Energy and Resources), (2022). Guideline: Offshore petroleum decommissioning. In relation to the Offshore Petroleum and Greenhouse Gas Storage Act 2006. Effective 2 March 2022.
- DIT (Department for Infrastructure and Transport). (2021). Aboriginal Sites, Objects and Ancestral Remains Discovery Procedure. Environment and Heritage Technical Manual
- DPIF (Department of Primary Industries and Fisheries) (2014) Fishery Status Reports 2012, Fishery Report No 113. Department of Primary Industries and Fisheries, Darwin, NT.



- DNP (Director of National Parks) (2018a). North-west Marine Parks Network Management Plan 2018, Director of National Parks, Canberra.
- DNP (Director of National Parks) (2018b). North Marine Parks Network Management Plan 2018, Director of National Parks, Canberra.
- DoE (Department of the Environment) (2013). Matters of National Environmental Significance. Significant impact guidelines 1.1 Environment Protection and Biodiversity Conservation Act 1999. Australian Government, Canberra.
- DoE (Department of the Environment) (2014a). Approved Conservation Advice for *Glyphis glyphis* (speartooth shark). Department of the Environment, Canberra.
- DoE (Department of the Environment) (2014b). Recovery plan for the grey nurse shark (Carcharias taurus). Canberra.
- DoEE (Department of the Environment and Energy) (2017). Australian National Guidelines for Whale and Dolphin Watching 2017. Australian Government, Canberra.
- DoEE (Department of the Environment and Energy) (2018). Threat abatement plan for the impacts of marine debris on the vertebrate wildlife of Australia's coasts and oceans.
- DoEE (Department of the Environment and Energy) (2019). Loss of terrestrial climatic habitat caused by anthropogenic emissions of greenhouse gases. <u>https://www.environment.gov.au/climate-change/climate-solutions-package</u>.
- DPIF (Department of Primary Industries and Fisheries) (2014). Fishery Status Reports 2012, Fishery Report No 113. Northern Territory Government, Darwin.
- DSEWPaC (Department of Sustainability, Environment, Water, Population and Communities) (2010). Wetlands and the Ramsar Convention. Canberra.
- DSEWPaC (Department of Sustainability, Environment, Water, Population and Communities) (2012a). Marine Bioregional Plan for the North Marine Region. Canberra, Australian Capital Territory.DSEWPaC (Department of Sustainability, Environment, Water, Population and Communities) (2012b). Species group report card – marine reptiles. Supporting the marine bioregional plan for the North Marine Region. Canberra, Australian Capital Territory.
- DSEWPaC (Department of Sustainability, Environment, Water, Population and Communities) (2012c). Species group report card sharks and sawfishes. Supporting the marine bioregional plan for the North Marine Region. Canberra, Australian Capital Territory.
- DSEWPaC (Department of Sustainability, Environment, Water, Population and Communities) (2012d). Species group report card – seabirds. Supporting the marine bioregional plan for the North Marine Region. Canberra, Australian Capital Territory.
- DSEWPaC (Department of Sustainability, Environment, Water, Population and Communities) (2012e). Species group report card – seabirds. Supporting the marine bioregional plan for the North-west Marine Region. Commonwealth of Australia.
- DSEWPaC (Department of Sustainability, Environment, Water, Population and Communities). (2012f) Marine bioregional plan for the North-west Marine Region. Department of Sustainability, Environment, Water, Population and Communities, Canberra, ACT. 269 pp.
- DSEWPaC (Department of Sustainability, Environment, Water, Population, and Communities) (2013). Recovery plan for the white shark (Carcharodon carcharias). Canberra.
- Dunlop, M., Hilbert, D., Ferrier, S., House, A., Liedloff, A., Prober, S., Smyth, A., Martin, T., Harwood, T., Williams, K., Fletcher, C. and Murphy, H. (2012). The Implications of Climate Change for Biodiversity, Conservation and the National Reserve System: Final Synthesis. CSIRO Climate Adaptation Flagship. https://doi.org/10.4225/08/5850384d796c6.
- Dwyer, R. G., Rathbone, M., Foote, D. L., Bennett, M., Butcher, P. A., Otway, N. M., Louden, B. M., Jaine, F. R., Franklin, C. E., & Kilpatrick, C. (2023). Marine reserve use by a migratory coastal shark, Carcharias taurus. Biological Conservation, 283, 110099.
- Farcas, A., Thompson, P.M. and Merchant, N.D. (2016). Underwater noise modelling for environmental impact assessment, Environmental Impact Assessment Review, Volume 57, 2016, Pages 114–122, ISSN 0195-9255,https://doi.org/10.1016/j.eiar.2015.11.012.
- Ferreira, L. C., Thums, M., Fossette, S., Wilson, P., Shimada, T., Tucker, A. D., Pendoley, K., Waayers, D., Guinea, M. L., Loewenthal, G., King, J., Speirs, M., Rob, D., & Whiting, S.D. (2021). Multiple satellite tracking datasets inform green turtle conservation at a regional scale. Divers. Distrib. 27: 249–266.
- Ferreira, L.C., Jenner, C., Jenner, M., Udyawer, V., Radford, B., Davenport, A., Moller, L., Andrews-Goff, V., Double, M. and Thums, M. (2024). Predicting suitable habitats for foraging and migration in Eastern Indian Ocean pygmy blue whales from satellite tracking data. Movement Ecology, 12(1), p.42.
- Finneran, J.J., Henderson, E., Houser, D.S., Jenkins, K., Kotecki, S. and Mulsow, J. (2017). Criteria and Thresholds for U.S. Navy Acoustic and Explosive Effects Analysis (Phase III). Technical report by Space and Naval Warfare Systems Center Pacific (SSC Pacific).
- FITRI, L., & FORD, H. A. FORAGING BEHAVIOUR OF HOODED ROS: INS.
- FRDC (Fisheries Research and Development Corporation) (2019). Northern Territory Fisheries and Aquaculture Industry 2017/18: Economic Contributions Summary. FRDC project 2017-210.



- Frawley, G. (2003). The International Directory of Civil Aircraft, 2003–2004, p. 44. Aerospace Publications Pty Ltd. ISBN 1-875671-58-7.
- Fugro (2015) Barossa Field Meteorological, Current Profile, Wave and CTD Measurements Final Report. Reporting Period: 8 July 2014 to 16 July 2015. Report prepared for ConocoPhillips Australia Pty Ltd., Perth, Western Australia.
- Garnet, S.T., Szabo, J.K. & Dutson, G. (2011). The Action Plan for Australian Birds 2010. CSIRO Publishing, Melbourne.
- Gaughan, D. J., & Santoro, K. (2021) Status reports of the fisheries and aquatic resources of Western Australia 2019/20. State of the fisheries.
- Genesis Oil and Gas Consultants. (2011). Review and Assessment of Underwater Sound Produced from Oil and Gas Sound Activities and Potential Reporting Requirements under the Marine Strategy Framework Directive. Department of Energy and Climate Change, Aberdeen, Scotland.
- Gotz, T., G. Hastie, L.T. Hatch, O. Raustein, B.L. Southall, M. Tasker, and F. Thomsen. (2009). Overview of the Impacts of Anthropogenic Underwater Sound in the Marine Environment, London: OSPAR Commission, <u>http://mhk.pnnl.gov/wiki/index.php/Overview Of The Impacts Of Anthropogenic Underwater Sound In The Marine Environment</u>.
- Gilmour, J., Speed, C.W., Babcock, R., 2016. Coral reproduction in Western Australia. PeerJ 4: e2010. doi:10.7717/peerj.2010
- Gleiss, A., Wright, S., Liebsch, N. and Wilson, R. (2013). Contrasting diel patterns in vertical movement and locomotor activity of whale sharks at Ningaloo Reef. Marine Biology 160(11): 2981–2992.
- Green ME, Simpfendorfer CA & Devloo-Delva F (2022) Population Structure and Connectivity. In: JC Carrier, CA Simpfendorfer, MR Heithaus & KE Yopak (eds), Biology of Sharks and Their Relatives, Third Edition. CRC Press.
- Groeneveld, J.C., Cliff, G., Dudley, S.F.J., Foulis, A.J., Santos, J. & Wintner, S.P. (2014). Population structure and biology of shortfin mako, Isurus oxyrinchus, in the south-west Indian Ocean. Marine and Freshwater Research 65(12): 1045–1058. Hanf, D., Hodgson, A.J., Kobryn, H., Bejder, L. & Smith, J.N. (2022). Dolphin Distribution and Habitat Suitability in Northwestern Australia: Applications and Implications of a Broad-Scale, Non-targeted Dataset. Front. Mar. Sci. 8:733841.
- Guerra-Garcia, J.M. and Garcia-Gomez, J.C. (2006). Recolonization of defaunated sediments: Fine versus gross sand and dredging versus experimental trays. Estuarine, Coastal and Shelf Science. Vol. 68 (1-2), pp. 328-342.
- Guinea, M.L. (2006). Sea turtles, sea snakes and dugongs of Scott Reef, Seringapatam Reef and Browse Island with notes on West Lacepede Island. Report to URS. Charles Darwin University, Darwin, NT.
- Guinea, M. (2013). Surveys of the Sea Snakes and Sea Turtles on Reefs of the Sahul Shelf: Monitoring Program for the Montara Well Release Timor Sea Monitoring Study S6 Sea Snakes/Turtles. Charles Darwin University. Darwin, Northern Territory.
- Guzman, H.M., Gomez, C.G., Hearn, A. & Eckert, S.A. (2018). Longest recorded trans-Pacific migration of a whale shark (Rhincodon typus). Mar Biodivers Rec 11, 8.
- Hamel, M.A., McMahon, C.R. and Bradshaw, C.J.A., 2008. Flexible inter-nesting behaviour of generalist olive ridley turtles in Australia. Journal of Experimental Marine Biology and Ecology 359: 47–54. doi:10.1016/j.jembe.2008.02.019.
- Hanf, D., Hodgson, A. J., Kobryn, H., Bejder, L., & Smith, J. N. (2022). Dolphin distribution and habitat suitability in north western Australia: applications and implications of a broad-scale, non-targeted dataset. Frontiers in Marine Science, 8, 733841.
- Harewood, A. and Horrocks, J.A. (2008). Impacts of coastal development on hawksbill hatchling survival and herpetological impacts of coastal development on hawksbill hatchling survival and swimming success during the initial offshore migration. Biological Conservation, 141, 394–401.
- Hazel, J., Lawler, I.R. and Hamann, M. (2009). Diving at the shallow end: Green turtle behaviour in near-shore foraging habitat. Journal of Experimental Marine Biology and Ecology 371, 84–92. https://doi.org/10.1016/j.jembe.2009.01.007.
- Heatwole, H. (1975). Voluntary submergence times of marine snakes. Marine Biology 32, 205–213.
- Heyward, A.J., Pincerato, E.J., Smith, L. (eds.) (1997) Big Bank Shoals of the Timor Sea: And Environmental Resource Atlas. BHP Petroleum, Melbourne, Victoria.
- Heyward, A., Radford, B., Cappo, M., Case, M., Stowar, M., Colquhoun, J. & Cook, K (2017) Barossa Environmental Baseline Study, Regional Shoals and Shelf Assessment 2015 Final Report. Report prepared for ConocoPhillips Australia Pty Ltd. Perth, Western Australia.
- Higgins, P.J. & J.M. Peter (2002). Handbook of Australian, New Zealand and Antarctic Birds. In: Pardalotes to Shrike-thrushes. Volume 6. Melbourne: Oxford University Press.
- Hobbs, J. and Willshaw, K. (2015). Unusual behavior and habitat use of a solitary male dugong inhabiting coral reefs at the Cocos (Keeling) Islands. Marine Biodiversity. 46 (1): pp. 31-32.
- Hoegh-Guldberg, O., Jacob, D., Taylor, M., Bindi, M., Brown, S., Camilloni, I., Diedhiou, A., Djalante, R., Ebi, K.L., Engelbrecht, F., Guiot, J., Hijioka, Y., Mehrotra, S., Payne, A., Seneviratne, S.I., Thomas, A., Warren, R. and Zhou, G. (2018). Impacts of 1.5°C Global Warming on Natural and Human Systems, in: Global Warming of 1.5°C. An IPCC Special Report on the Impacts of Global Warming of 1.5°C above Pre-Industrial Levels and Related Global Greenhouse Gas Emission Pathways, in the Context of Strengthening the Global Response to the Threat of Climate Change, Sustainable Development, and Efforts to Eradicate Poverty.



- Hoschke, A.M., Whisson, G.J. & Haulsee, D. (2023). Population distribution, aggregation sites and seasonal occurrence of Australia's western population of the grey nurse shark Carcharias taurus. Endang Species Res 50:107-123.
- Hughes, T.P., Kerry, J.T., Álvarez-Noriega, M., Álvarez-Romero, J.G., Anderson, K.D., Baird, A.H., Babcock, R.C., Beger, M., Bellwood, D.R., Berkelmans, R. (2017). Global warming and recurrent mass bleaching of corals. Nature 543: 373–377
- INPEX Browse (2010). Ichthys Gas Field Development Project: draft environmental impact statement. INPEX Browse, Perth.
- Intecsea (2018). Barossa FEED Subsea Facilities and Export Pipeline Shipping and Fishing Impact Study. Report prepared for ConocoPhillips Australia Pty Ltd.
- Intecsea (2022). BAS-201 1041: Nearshore Pipeline Route Selection and Span Rectification Report. Prepared by Intecsea for Santos (unpublished).
- Interim Marine and Coastal Regionalisation for Australia Technical Group (IMCRATG) (1998) Interim Marine and Coastal Regionalisation for Australia: an ecosystem-based classification for marine and coastal environments. Version 3.3. Environment Australia, Commonwealth Department of the Environment. Canberra.
- International Maritime Organisation (IMO) (2023). 2023 Guidelines for the control and management of ships' biofouling to minimize the transfer of invasive aquatic species (Biofouling Guidelines). Resolution MEPC.378(80) (adopted on 7 July 2023). Available at:

https://www.cdn.imo.org/localresources/en/KnowledgeCentre/IndexofIMOResolutions/MEPCDocuments/MEPC.378%288 0%29.pdf

- International Organization for Standardization (ISO), (2018). AS ISO 31000:2018 Risk management Guidelines. https://www.iso.org/standard/65694.html
- Intergovernmental Panel on Climate Change (IPCC), 2018. Global Warming of 1.5°C.An IPCC Special Report on the impacts of global warming of 1.5°C above pre-industrial levels and related global greenhouse gas emission pathways, in the context of strengthening the global response to the threat of climate change, sustainable development, and efforts to eradicate poverty [Masson-Delmotte, V., P. Zhai, H.-O. Pörtner, D. Roberts, J. Skea, P.R. Shukla, A. Pirani, W. Moufouma-Okia, C. Péan, R. Pidcock, S. Connors, J.B.R. Matthews, Y. Chen, X. Zhou, M.I. Gomis, E. Lonnoy, T. Maycock, M. Tignor, and T. Waterfield (eds.)]. Cambridge University Press, Cambridge, UK and New York, NY, USA, 616 pp. https://doi.org/ 10.1017/9781009157940.
- IPCC (Intergovernmental Panel on Climate Change). (2021). Climate Change 2021: The Physical Science Basis. International Panel on Climate Change. <u>https://www.ipcc.ch/report/ar6/wg1</u>
- IPCC (Intergovernmental Panel on Climate Change). (2023). Climate Change 2023. https://www.ipcc.ch/report/ar6/syr/
- IUCN-MMPATF (International Union for Conservation of Nature Marine Mammal Protected Areas Task Force) (2023a). Available from: https://www.marinemammalhabitat. Org/portfolio-item/eastern-indian-ocean-blue-whale-migratory-route, accessed: 04/07/23.
- IUCN-MMPATF (International Union for Conservation of Nature Marine Mammal Protected Areas Task Force) (2023b). Available at: https://www.marinemammalhabitat. Org/portfolio-item/western-australian-humpback-whale-migration-route, accessed: 04/07/23.
- Jacobs (2016) Barossa Environmental Studies Benthic Habitat Report. Report prepared for ConocoPhillips. Perth, Western Australia
- Jacobs. (2019). Tiwi Islands Sensitivity Mapping. Unpublished report prepared by Jacobs for ConocoPhillips. Document Number IW193500.AA.RPT-0001 | 0.
- Jaensch (1993) Directory of important wetlands in Australia. Compiled for the Wildlife Division, Conservation Commission of the Northern Territory, January-February 1993. Updated by P. Whitehead and R. Chatto November 1995.
- Jakobs, S. & Braccini, M. (2019). Acoustic and conventional tagging support the growth patterns of grey nurse sharks and reveal their large-scale displacements in the west coast of Australia. Mar Biol 166:150.
- JASCO. (2020). Underwater Noise Impacts on Marine Fauna: Technical Appendix. Document 02028, Version 1.1. Technical Appendix by JASCO Applied Sciences for Santos WA Energy Ltd. (unpublished).
- Jefferson, T.A., & H.C. Rosenbaum (2014). Taxonomic revision of the humpback dolphins (Sousa spp.), and description of a new species from Australia. Marine Mammal Science. 30(4):1494-1541.
- Jenner, K., Jenner, M., McCabe, K. (2001). Geographical and temporal movements of humpback whales in Western Australian waters. Australian Petroleum Production & Exploration Association Journal 41, 692–707.
- Jensen, A. and Silber, G. (2004). Large whale ship strike database (NOAA Technical Memorandum). National Marine Fisheries Service, Silver Spring, Maryland.
- Jimenez-Arranz, G., Banda, N., Cook, S. and Wyatt, R. (2020). Review on Existing Data on Underwater Sounds Produced by the Oil and Gas Industry. Prepared by Seiche Ltd for E&P Sound and Marine Life Joint Industry Programme. Project Reference P783.
- Kato, H. (2002). Bryde's Whales Balaenoptera edeni and B. brydei. In: Perrin W.F., B. Wrsig & H.G.M. Thewissen, eds. Encyclopedia of Marine Mammals. Page(s) 171-177. Academic Press.
- Kelly, T. & Przeslawski, R (2012) The ecology and morphology of sponges and octocorals in the north eastern Joseph Bonaparte Gulf. Geoscience Australia, Record 2012/67, Canberra: 81 pp.



- Knuckey, C.G., Trainor, C.R., Firth, R.S.C. & Sansom, J.L. (2013). A record of the Endangered Australian Painted Snipe Rostratula australis (Gould, 1838) in the Fortescue valley, Pilbara region. Wader Study Group Bull. 120(1).
- Kyne, P.M., Heupel, M.R., White, W.T. & Simpfendorfer, C.A. (2021a). The Action Plan for Australian Sharks and Rays 2021. National Environmental Science Program, Marine Biodiversity Hub, Hobart.
- Kyne P.M., Davies C-L, Devloo-Delva, F., Johnson, G., Amepou, Y., Grant, M.I., Green, A., Gunasekara, R.M., Harry, A.V., Lemon, T., Lindsay, R., Maloney, T., Marthick, J., Pillans, R.D., Saunders, T., Shields, A., Shields, M. & Feutry, P. (2021b) Molecular analysis of newly-discovered geographic range of the threatened river shark Glyphis glyphis reveals distinct populations. Report to the National Environmental Science Program, Marine Biodiversity Hub. Charles Darwin University and CSIRO.
- Laist, D.W., Knowlton, A.R., Mead, J.G., Collet, A.S. and Podesta, M. (2001). Collisions between ships and whales. Marine Mammal Science 17, 35–75.
- Landbridge Darwin Port (2024). Vessel Visits, Landbridge Darwin Port. Available from: https://www.darwinport.com.au/trade/vessel-visits. Accessed 19 January 2024.
- Lear, K.O., Fazeldean, T., Bateman, R.L., Inglebrecht, J. & Morgan, D.L. (2023). Growth and morphology of Critically Endangered green sawfish Pristis zijsron in globally important nursery habitats. Mar Biol 170:70.Li F, Qiao Z, Duan Q and Nevo E. (2021). Adaptation of mammals to hypoxia. Animal Model Exp Med. 2021 Nov 29;4(4):311-318. doi: 10.1002/ame2.12189. PMID: 34977482; PMCID: PMC8690989.
- Lilleyman, A., Garnett, S.T., Rogers, D.I & Lawes, M.J. (2016). Trends in relative abundance of the Eastern Curlew in Darwin. Stilt, 68: 25–30.
- Limpus, C.J. (1971). Sea turtle ocean finding behaviour. Search, 2, 385–387.
- Limpus, C.J. & McLachlin, N. (1994). The conservation status of the Leatherback Turtle, Dermochelys coriacea, in Australia. In: James R (ed.) Proceedings of the Australian Marine Turtle Conservation Workshop, Gold Coast 14-17 November 1990. Pp. 63-67. Queensland Department of Environment and Heritage. Canberra: ANCA.
- Limpus, C.J. (2007). A biological review of Australian marine turtles. 5. Flatback turtle, *Natator depressus* (Garman). The State of Queensland. Environmental Protection Agency.
- Limpus, C.J. (2008). A biological review of Australian marine turtles. 1. Loggerhead turtle, Caretta (Linneaus). The State of Queensland. Environmental Protection Agency.
- Limpus, C.J. (2009). A biological review of Australian marine turtles. 6. Leatherback turtle, Dermochelys coriacea (Vandelli). The State of Queensland. Environmental Protection Agency.
- Limpus, C.J. and Kamrowski, R.L. (2013). Ocean-finding in marine turtles: the importance of low horizon elevation as an orientation cue. Behaviour, 150, 863–893.
- Lindquist, D.C., Shaw, R.F. and Hernandez Jr, F.J. (2005). Distribution patterns of larval and juvenile fishes at offshore petroleum platforms in the north central Gulf of Mexico. Estuarine, Coastal and Shelf Science 62:655–665.
- Lohmann, K. and Fittinghoff-Lohmann, C. (1992). Orientation to oceanic waves by green turtle hatchlings. The Journal of Experimental Biology. Pg 1 13. https://doi.org/10.17615/ny34-kb57.
- Lohmann, K.J., Witherington, B.E., Lohmann, C.M.F. and Salmon, M. (1997). Orientation, navigation, and natal beach homing in sea turtles. In 'The Biology of Sea Turtles'. (Eds. P. Lutz and J. Musick.) pp. 107–136. (CRC Press: Boca Raton, FL.)
- Longcore, T. and Rich, C. (2016). Artificial night lighting and protected lands: Ecological effects and management approaches. Natural Resource Report NPS/NRSS/NSNS/NRR—2016/1213. National Park Service, Fort Collins, Colorado.
- Lorne, J.K. and Salmon, M., 2007. Effects of exposure to artificial lighting on orientation of hatchling sea turtles on the beach and in the ocean. Endangered Species Research 3, 23–30.
- Marine Pest Sectoral Committee, 2009. National biofouling management guidelines for the petroleum production and exploration industry, Department of Agriculture and Water Resources, Canberra.
- Marsh, H. (n.d.) Fauna of Australia. 57. Dugongidae. Available from: www.dcceew.gov.au/sites/default/files/env/pages/a117ced5-9a94-4586-afdb-1f333618e1e3/files/57-ind.pdf, accessed March 2024.
- Marshall, A., Barreto, R., Carlson, J., Fernando, D., Fordham, S., Francis, M.P., Derrick, D., Herman, K., Jabado, R.W., Liu, K.M., Rigby, C.L. & Romanov, E. (2022a). Mobula birostris (amended version of 2020 assessment). The IUCN Red List of Threatened Species 2022: e.T198921A214397182.
- Marshall, A., Barreto, R., Carlson, J., Fernando, D., Fordham, S., Francis, M.P., Herman, K., Jabado, R.W., Liu, K.M., Pacoureau, N., Rigby, C.L., Romanov, E. & Sherley, R.B. (2022b). Mobula alfredi (amended version of 2019 assessment). The IUCN Red List of Threatened Species 2022: e.T195459A214395983.
- McAuley, R., Bruce, B., Keay, I., Mountford, S. & Pinnell, T. (2016). Evaluation of passive acoustic telemetry approaches for monitoring and mitigating shark hazards off the coast of Western Australia. Fisheries Research Report No. 273, Department of Fisheries, Western Australia. 84pp.
- McCauley, R. (1998). Radiated underwater noise measured from the drilling rig Ocean General, rig tenders Pacific Ariki and Pacific Frontier, fishing vessel Reef Venture and natural sources in the Timor Sea, Northern Australia. (Report No. C98-20). Centre for Marine Science and Technology, Curtin University of Technology, Perth.


- McCauley, R.D. and Duncan, A.J. (2003). Underwater Acoustic Environment, Otway Basin, Victoria. Report prepared for Woodside Energy and Curtin University Centre for Marine Science and Technology, Perth, Western Australia.
- McCauley, R.D. (2012). Fish choruses from the Kimberley, seasonal and lunar links as determined by long term sea noise monitoring. Australian Acoustics Society, Centre for Marine Science and Technology, Curtin University, Perth.
- McLean, D.L., Partridge, J.C., Bond, T., Birt, M.J., Bornt, K.R., Langlois, T.J. (2017). Using industry ROV videos to assess fish associations with subsea pipelines. Continental Shelf Research 141, 76–97. Available from: https://doi.org/10.1016/j.csr.2017.05.006, accessed April 2023.
- McPherson, C., Kowarski, K., Delarue, J., Whitt, C., MacDonnell, J. & Martin, B. (2016). Passive acoustic monitoring of ambient noise and marine mammals – Barossa Field (JASCO Document No. 00997). JASCO Applied Sciences, Capalaba.
- Meekan, M., Wilson, S.G., Halford, A. and Retzel, A. (2001). A comparison of catches of fishes and invertebrates by two light trap designs, in tropical NW Australia. Marine Biology. 139. 373–381. 10.1007/s002270100577.
- Meekan, M. & Radford, B. (2010). Migration patterns of whale sharks: A summary of 15 satellite tag tracks from 2005 to 2008. Australian Institute of Marine Science, Perth, Western Australia.
- Menkhorst, P., Rogers, D., Clarke, R., Davies, J., Marsack, P. & Franklin, K. (2017). The Australian Bird Guide. CSIRO Publishing.
- Milicich, M.J., Meekan, M.G. and Doherty, P.J. (1992). Larval supply: a good predictor of recruitment in three species of reef fish (Pomacentridae). Mar Ecol Prog Ser. 86: 153–166.
- Milne, C., 2019. CHC Helicopters: From Rigs to Rescue. Written for Australian Aviation. https://australianaviation.com.au/2019/07/chc-helicopters-from-rigs-to-rescue/.
- Minchin, D.; Gollasch, S. (2002). Vectors—how exotics get around. In: Leppakoski, E.; Gollasch, S.; Olenin, S. ed. Invasive aquatic species of Europe: distribution, impact and management. London, Kluwer. Pp. 183–192.
- Momigliano, P. & Jaiteh, V.F. (2015). First records of the grey nurse shark Carcharias taurus (Lamniformes: Odontaspididae) from oceanic coral reefs in the Timor Sea. Marine Biodiversity Records, 8.
- NT Government (n.d). Threatened Species of the Northern Territory. Green Turtle Chelonia mydas. The Northern Territory Government, Darwin.
- National Offshore Petroleum Safety and Environmental Management Authority (NOPSEMA), 2019. *Guidance Note: Environment Plan Content Requirements (N04750-GN1344)*, Revision No. 4, National Offshore Petroleum Safety and Environmental Management Authority, Perth, Western Australia, available from https://www.nopsema.gov.au/assets/Guidance-notes/A339814.pdf [accessed 17 Apr 2020].
- National Offshore Petroleum Safety and Environmental Management Authority (NOPSEMA), 2019. *Guideline: Environment plan decision making N-04750-GL1721*, Rev. 6, National Offshore Petroleum Safety and Environmental Management Authority, Perth, Western Australia, available from https://www.nopsema.gov.au/assets/Guidelines/A524696.pdf [accessed 25 Mar 2020].
- National Offshore Petroleum Safety and Environmental Management Authority (NOPSEMA), 2020. Section 572 Maintenance and removal of property (Policy N-00500-PL1903 A720369). National Offshore Petroleum Safety and Environmental Management Authority, Perth.
- National Offshore Petroleum Safety and Environmental Management Authority (NOPSEMA), 2021. Oil Pollution Risk Management. (Document number: N-04750-GN1488 A382148). NOPSEMA, Perth. Available from: www.nopsema.gov.au/sites/default/files/documents/2021-07/A382148.pdf, accessed March 2024.
- National Offshore Petroleum Safety and Environmental Management Authority (NOPSEMA), 2023. Consultation in the course of preparing an environment plan. (Document number: N-04750-GL2086 A900179). NOPSEMA, Perth.
- National Offshore Petroleum Safety and Environmental Management Authority (NOPSEMA), 2024a. Consultation in the course of preparing an environment plan. (Document number: N-04750-GL2086 A900179). NOPSEMA, Perth. Available from: www.nopsema.gov.au/sites/default/files/documents/Consultation%20in%20the%20course%20of%20preparing%20an%2 0Environment%20Plan.pdf, accessed May 2024.
- National Offshore Petroleum Safety and Environmental Management Authority (NOPSEMA), 2024b. Consultation with Commonwealth agencies with responsibilities in the marine area. (Document number: N-04750-GL1887 A705589). NOPSEMA, Perth. Available from: www.nopsema.gov.au/sites/default/files/documents/Consultation%20with%20agencies%20with%20responsibilities%20in %20the%20Commonwealth%20marine%20area.pdf, accessed February 2023.
- National Offshore Petroleum Safety and Environmental Management Authority (NOPSEMA), 2024c. Environment plan decision making. (Document number: N-04750-GL1721 A524696). NOPSEMA, Perth. Available from: www.nopsema.gov.au/sites/default/files/documents/Environment%20plan%20decision%20making%20guideline.pdf, accessed March 2024.
- National Offshore Petroleum Safety and Environmental Management Authority (NOPSEMA), 2024d. Environment plan content requirement. (N-04750-GN1344 A339814). NOPSEMA, Perth. Available from: www.nopsema.gov.au/sites/default/files/documents/Environment%20Plan%20Content%20Requirements%20Guidance% 20Note.pdf, accessed March 2024.



- National Offshore Petroleum Safety and Environmental Management Authority (NOPSEMA), 2024e. Oil Pollution Riks Management. (Document number: N-04750-GN1488 A382148). NOPSEMA, Perth. Available from: <u>https://www.nopsema.gov.au/sites/default/files/documents/Oil%20Pollution%20Risk%20Management.pdf</u>, accessed October 2024.
- Newman, S.J., Wise, B.S., Santoro, K.G. and Gaughan, D.J. (Eds.), (2021) Status Reports of the Fisheries and Aquatic Resources of Western Australia 2020/21: The State of the Fisheries. Department of Primary Industries and Regional Development, Western Australia.
- Nguyen, K.Q. and Winger, P.D. (2019). Artificial light in commercial industrialized fishing applications: a review. Reviews in Fisheries Science & Aquaculture 27: 106–126.
- NT Government (2023) Aquaculture Species. Available from: https://nt.gov.au/marine/aquaculture/commercial/aquaculture-species. (Accessed: 4/10/23)
- NT Government (2023b). Commercial fishing. Available from: https:// https://nt.gov.au/marine/commercial-fishing. Accessed April 2023.
- NT Tourism (2023a) Tourism Sector Development, Cruise. Available from: <u>https://www.tourismnt.com.au/research-strategies/sector-development/cruise</u>. (Accessed: 4/10/23)
- NT Tourism (2023b). Combined (Domestic and International) Snapshot Year Ending (YE) March 2024. Available from: https://tourismnt.com.au/system/files/uploads/files/2024/Combined\_snapshot%20YE%20Mar%2024.pdf (Accessed: 14/11/24).
- O'Malley, C. (2006). National Recovery Plan for the Gouldian Finch (Erythrura gouldiae). WWF-Australia, Sydney and Parks and Wildlife NT, Department of Natural Resources, Environment and the Arts, NT Government, Palmerston.
- OSPAR, 2009. Assessment of the environmental impact of underwater noise. Available online: https://qsr2010.ospar.org/media/assessments/p00436\_JAMP\_Assessment\_Noise.pdf
- OzArk Environment & Heritage (2024). Darwin Pipeline Duplication Project Desktop First Nations Archaeological Assessment Report.
- Padovan, A (1997) The water quality of Darwin Harbour October 1990 November 1991. Report 34/1997D. Prepared for Department of Lands, Planning and Environment. <u>https://hdl.handle.net/10070/674282</u>
- Palmer, C., Brooks, L., Parra, G. J., Rogers, T., Glasgow, D., & Woinarski, J. C. (2014). Estimates of abundance and apparent survival of coastal dolphins in Port Essington harbour, Northern Territory, Australia. *Wildlife Research*, *41*(1), 35-45.
- Parnell, E. (2003). The effects of sewage discharge on water quality and phytoplankton of Hawai'ian coastal waters. Marine Environmental Research, 55(4), 293–311.
- Parra, G. J., & Cagnazzi, D. (2016). Conservation status of the Australian humpback dolphin (Sousa sahulensis) using the IUCN Red List Criteria. In *Advances in marine biology* (Vol. 73, pp. 157-192). Academic Press.
- Parry, D.L. and Munksgaard, N.C (1995) Physiochemical Baseline Data for Darwin Harbour-East Arm Port Development. Northern Territory University, Darwin NT
- Patterson, H., Bromhead, D., Galeano, D., Larcombe, J., Timmiss, T., Woodhams, J. and Curtotti, R., (2022) Fishery status reports 2022, Australian Bureau of Agricultural and Resource Economics and Sciences, Canberra. CC BY 4.0. https://doi.org/10.25814/gx9r-3n90.
- Peel, D., Smith, J., & Childerhouse, S. (2018). Vessel strike of whales in Australia: The challenges of analysis of historical incident data. Frontiers in Marine Science, 5, 69.
- Pendoley, K. and Kamrowski, R.L. (2016). Sea-finding in marine turtle hatchlings: What is an appropriate exclusion zone to limit disruptive impacts of industrial light at night? Journal for Nature Conservation, 30 (2016) 1–11.
- Pendoley (2019). ConocoPhillips Barossa Project potential impacts of pipeline installation activities on marine turtles literature update (No. J54001 Rev 2). 5 July 2019. Unpublished report prepared by Pendoley Environmental Pty Ltd for Jacobs.
- Pendoley Environmental (2022). Barossa Pipelay Light Modelling 2022. Report prepared for Santos Pty Ltd, 15 August 2022.
- Pendoley (2023) Desktop Study Tiwi Turtle Programs. Pendoley Environmental for Santos, June 2023.
- Perez, M.A., Limpus, C.J., Hofmeister, K., Shimada, T., Strydom, A., Webster, E. & Hamann, M. (2022). Satellite tagging and flipper tag recoveries reveal migration patterns and foraging distribution of loggerhead sea turtles (Caretta caretta) from eastern Australia. Mar Biol 169: 80.
- Pogonoski, J.J., Pollard, D.A. & Paxton J.R. (2002). Conservation Overview and Action Plan for Australian Threatened and Potentially Threatened Marine and Estuarine Fishes. Environment Australia.
- Popper, A.N., Hawkins, A.D., Fay, R.R., Mann, D.A., Bartol, S.M., Carlson, T.J., Coombs, S., Ellison, W.T., Gentry, R.L., Halvorsen, M.B., Løkkeborg, S., Rogers, P., Southall, B.L., Zeddies, D.G. and Tavolga, W.N. (2014). ASA S3/SC1.4 TR-2014 sound exposure guidelines for fishes and sea turtles: a technical report prepared by American National Standards Institute Accredited Standards Committee S3/SC1 and registered with American National Standards Institute. Springer, New York.

Posamentier, H. W (2023) Barossa Field Seafloor Late Pleistocene/Holocene Depositional Environment." Unpublished report.



- Przeslawski, R., Daniell, J., Anderson, T., Barrie, J.V., Battershill, C., Heap, A., Hughes, M., Li, J., Potter, A., Radke, R., Siwabessy, J., Tran, M, Whiteway, T. & Nichol, S (2011) Seabed Habitats and Hazards of the Joseph Bonaparte Gulf and Timor Sea, Northern Australia. Geoscience Australia, record 2011/40. Geoscience Australia, Canberra, Australian Capital Territory.
- PWSNT (Parks and Wildlife Service Northern Territory). (2003). Draft Management Program for the Dugong (Dugong dugong) in the Northern Territory of Australia 2003-2008. Department of Planning, Infrastructure and Environment, Darwin, Northern Territory.
- Radford, B., Heyward, A., Birt, M.J., Case, M., Colquhoun, J., Currey-Randall, L.M., Stowar, M.J., Vaughan, B.I. & Wyatt, M. (2019) Oceanic Shoals Commonwealth Marine Reserve (CMR) final quantitative report on benthic habitats (Report No. AIMS/COP/RPT/002/2019). Australian Institute of Marine Science, Perth.
- Raudino HC, Douglas CR, Waples KA (2018). How many dolphins live near a coastal development? Regional Studies in Marine Science. 2018;19:25–32.
- Reputex, (2024). Market Outlook Australian Carbon Price, Supply & Demand Outlook September 2024
- Richardson, W.J., Greene Jr, C.R., Malme, C.I. and Thomson, D.H. (1995). Marine Mammals and Noise. Academic Press, San Diego.
- Rigby, C.L., Barreto, R., Carlson, J., Fernando, D., Fordham, S., Francis, M.P., Jabado, R.W., Liu, K.M., Marshall, A., Pacoureau, N., Romanov, E., Sherley, R.B. & Winker, H. (2019). Isurus paucus. The IUCN Red List of Threatened Species 2019: e.T60225A3095898. Accessed: 28/06/23.Roelof, A., Coles, R and Smit N. (2005). A survey of intertidal seagrass from Van Diemen Gulf to Castlereagh Bay, Northern Territory, and from Gove to Horn Island, Queensland. Report to the National Oceans Office. March 2005.
- RPS (2023). Barossa DPD Pipeline Benthic Survey Report (Ref: BAS-210-0014). Rev1, 13 January 2023. RPS, Perth.
- RPS (2010). Marine megafauna report, Browse marine megafauna study. RPS Environment and Planning Pty Ltd. Perth.
- RPS (2024a). Santos DPD treated seawater modelling. (Document Number GOC367756). RPS, Brisbane.
- RPS (2024b). Santos Barossa Gas Export Pipeline Northern Territory Coastal Waters and Internal Waters) Hydrocarbon Spill Modelling report. Rev2. GOC347879.
- Ruiz, G. M.; Carlton, J. T; Grosholz, E. D; Hines, A. H. (1997). Global invasions of marine and estuarine environments by nonindigenous species: mechanisms, extent and consequences. American Zoology 37:621–632.
- Sahri, A., C. Jak, M. I. H. Putra , A. J. Murk, V. Andrews-Goff, M. C. Double, R. van Lammeren. (2022). Telemetry-based home range and habitat modelling reveals that the majority of areas important for pygmy blue whales are currently unprotected. Biological Conservation 272:109594.
- Salmon, M., Wyneken, J., Fritz, E. and Lucas, M. (1992). Sea finding by hatchling sea turtles: role of brightness, silhouette and beach slope orientation cues. Behaviour, 122.
- Santos (2020). Barossa Project Design Basis, BAA-100 0047, Rev 7
- Santos (2022). Bayu-Undan to Darwin Gas Export Pipeline Environment Plan (Document # 7710-057-EIS-0001 Rev 3). Santos.
- Santos (2024). Barossa Gas Export Pipeline Safety Case. BAS-202-0005 Revision 6
- Shell (2019). Safety Data Sheet Natural gas, dried. Available from: https://www.shell.com/business-customers/chemicals/safeproduct-handling-and-transportation/chemicals-safety-data-sheets.html.
- Simmonds, M.P., Dolman, S.J. and Weilgart, L. (eds.) (2004). Oceans of Noise. Whale and Dolphin Conservation Society. Available from: https://uk.whales.org/wp-content/uploads/sites/6/2018/08/Oceans-of-Noise.pdf, accessed July 2023.
- Solomon, S., Manning, M., Marquis, M., Qin, D. (2007). Climate change 2007-the physical science basis: Working group I contribution to the fourth assessment report of the IPCC. Cambridge
- Spalding, M.D., Brown, B.E. (2015). Warm-water coral reefs and climate change. Science 350: 769–771.
- Stapput, K. and Wiltschko, W. (2005). The sea-finding behavior of hatchling olive ridley sea turtles, Lepidochelys olivacea, at the beach of San Miguel (Costa Rica). Naturwissenschaften, 92(5), pp. 250–253.
- Steffen, W., Burbidge, A.A., Hughes, L., Kitching, R., Lindenmayer, D., Musgrave, W., Staford Smith, M. and Werner, P. (2009). Australia's biodiversity and climate change: A strategic assessment of vulnerability of Australia's biodiversity to climate change. A report to the Natural resource Management Ministerial Council commissioned by Australian Government. CSIRO Publishing.
- Stevens JD, Pillans RD and Salini JP (2005). Conservation assessment of *Glyphis glyphis* (speartooth shark), *Glyphis garricki* (northern river shark), Pristis microdon (freshwater sawfish) and Pristis zijsron (green sawfish). Report to Department of Environment and Heritage, Canberra, Australia.
- Surman, C. (2002). Survey of the marine avifauna at the Laverda-2 appraisal well (WA-271-P) Enfield Area Development and surrounding waters. Report prepared for Woodside Energy Ltd., Perth.
- Thomson, P.G., Pillans, R., Jaine, F.R., Harcourt, R.G., Taylor, M.D., Pattiaratchi, C.B. & McLean, D.L. (2021). Acoustic Telemetry Around Western Australia's Oil and Gas Infrastructure Helps Detect the Presence of an Elusive and Endangered Migratory Giant. Frontiers in Marine Science. 8.



- Thorburn, D.C., Morgan, D.L., Rowland A.J. & Gill, H.S. (2007). Freshwater sawfish *Pristis microdon* Latham, 1794 (Chondrichthyes: Pristidae) in the Kimberley region of Western Australia. Zootaxa. 1471:27-41.
- Thorburn, D.C., Morgan, D.L., Rowland, A.J., Gill, H.S. & Paling, E. (2008). Life history notes of the critically endangered dwarf sawfish, Pristis lavate, Garman 1906 from the Kimberley region of Western Australia. Environmental Biology of Fishes. Vol. 83:139–145. Thums, M., Australian Institute of Marine Science, Whiting., S., Department of Environment and Conservation, Reisser, J., ULUU., Pendoley, K., Pendoley Environmental Pty Ltd. (2013). Tracking sea turtle hatchlings A pilot study using acoustic telemetry Tracking sea turtle hatchlings A pilot study using acoustic telemetry. February 2013 Journal of Experimental Marine Biology and Ecology
- Thums, M., Whiting, S.D, Reisser, J.W., Pendoley, K.L., Pattiaratchi C.B., Harcourt, R.G., Mcmahon, C.R. and Meekan, M.G. (2013). Tracking sea turtle hatchlings—A pilot study using acoustic telemetry. Journal of Experimental Marine Biology and Ecology, 440, 156–163.
- Thums, M., Whiting, S.D, Reisser, J.W., Pendoley, K.L., Pattiaratchi C.B., Proietti, M., Hetzel, Y., Fisher, R. and Meekan, M.G. (2016). Artificial light on water attracts turtle hatchlings during their near shore transit. Royal Society Open Science 3, DOI: 10.1098/rsos.160142.
- Thums, M., Rossendell, J., Fisher, R. & Guinea, M.L. (2020). Nesting ecology of flatback sea turtles Natator depressus from Delambre Island, Western Australia. Marine and Freshwater Research. 71: 443-451
- Thums, M., Ferreira, L. C., Jenner, C., Jenner M., Harris, D., Davenport, A., Andrews-Goff, V., Double, M., Möller, L., Attard, C., Bilgmann, K., Thomson, P. & McCauley, R. (2021). Understanding pygmy blue whale movement and distribution off northwestern Australia. The APPEA Journal. 61, 505-511.
- Tiwi Land Council (2021). Tiwi Islands Sea Country: An information booklet supporting Tiwi people's aspirations for a Marine Indigenous Protected Area. Tiwi Land Council.
- Tiwi Land Council (2022). Sea Turtles. Available at: https://www.tiwilandcouncil.com/index.cfm?fuseaction=page&p=297&id=64&smid=135&ssmid=69#:~:text=Tiwi%20peopl e%20have%20strong%20cultural%20and%20subsistence%20links,most%20of%20the%20islands%20free%20from%20 harvest%20pressure. Accessed 2 September 2022.
- Trainor, C.R., Knuckey, C. & Firth, R.S.C. (2017). New bird records from the Fortescue Marsh and nearby claypans, Pilbara bioregion, Western Australia. Aus. Field. Ornithology. 33:61–81.
- Tucker, A.D., Pendoley, K.L., Murray, K. et al. (2021). Regional Ranking of Marine Turtle Nesting in Remote Western Australia by Integrating Traditional Ecological Knowledge and Remote Sensing. Remote Sens. 13: 4696. Whiting, S.D., Hadden, K., Long, J.L., Lauder, A.D.K., Kleidon, A. & Cook, K. (2007). Sea Turtle Conservation and Education on the Tiwi Islands. Final Natural Heritage Report. Canberra. Australian Government.
- TSSC (Threatened Species Scientific Committee) (2014a). Approved Conservation Advice for *Glyphis garricki* (northern river shark). Department of Environment, Canberra.
- TSSC (Threatened Species Scientific Committee) (2014b). Approved Conservation Advice *for Pristis pristis* (largetooth sawfish). Department of Environment, Canberra.
- TSSC (Threatened Species Scientific Committee) (2015c). Conservation Advice *Balaenoptera borealis* sei whale. Department of Environment, Canberra.
- TSSC (Threatened Species Scientific Committee) (2015d). Conservation Advice *Balaenoptera physalus* fin whale. Department of Environment, Canberra.
- TSSC (Threatened Species Scientific Committee) (2015e). Conservation Advice *Calidris ferruginea* curlew sandpiper. Department of Environment, Canberra.
- TSSC (Threatened Species Scientific Committee) (2015f). Conservation Advice *Numenius madagascariensis* eastern curlew. Department of Environment, Canberra.
- TSSC (Threatened Species Scientific Committee) (2015g). Conservation Advice *Rhincodon typus* whale shark. Department of the Environment, Canberra.
- TSSC (Threatened Species Scientific Committee) (2018). Listing Advice *Sphyrna lewini* scalloped hammerhead. Department of the Environment and Energy, Canberra.
- TSSC (Threatened Species Scientific Committee) (2020). Conservation Advice Falco hypoleucos (Grey Falcon). Threatened Species Scientific Committee. Department of the Environment and Energy, Canberra. Available from: www.environment.gov.au/biodiversity/threatened/species/pubs/929-conservation-advice-09072020.pdf, accessed 15 March 2023.
- TSSC (Threatened Species Scientific Committee) (2022). Listing Advice Megaptera novaeangliae (Humpback Whale). Threatened Species Scientific Committee. Department of the Environment and Energy, Canberra. Available from: www.environment.gov.au/biodiversity/threatened/species/pubs/38-listing-advice-26022022.pdf, accessed 15 March 2023.
- Udyawer, V., Thums, M., Ferreira, L. C., Tulloch, V., & Kyne, P. M. (2021). Distribution and Habitat Suitability of Threatened and Migratory Marine Species in Northern Australia. National Environmental Research Program Marine Biodiversity Hub.
- United Nations, 2023. Globally harmonised system of classification and labelling of chemicals. Available online at: <u>GHS/Rev.10e</u> (<u>unece.org</u>). Accessed 16/09/2024.



- United Nations Framework Convention on Climate Change, 2018. Katowice climate change package. <u>https://unfccc.int/process-and-meetings/the-paris-agreement/the-katowice-climate-package/katowice-climate-package</u>
- UNFCCC (United Nations Framework Convention on Climate Change). (2021). Glasgow Climate Pact. https://unfccc.int/sites/default/files/resource/cop26\_auv\_2f\_cover\_decision.pdf.
- United Nations Framework Convention on Climate Change, 2022. NDC Registry Australia. <u>https://unfccc.int/documents/510664</u>.
- UNFCCC (United Nations Framework Convention on Climate Change). (2023). Outcome of the first global stocktake. Draft decision, distributed 13 December 2023. FCCC/PA/CMA/2023L.17. https://unfccc.int/documents/636608.
- Vignaud, T.M., Maynard, J.A., Leblois, R., Meekan, M.G., Vázquez-Juárez, R., Ramírez-Macías, D., Pierce, S.J., Rowat, D., Berumen, M.L., Beeravolu, C., Baksay, S. & Planes, S. (2014). Genetic structure of populations of whale sharks among ocean basins and evidence for their historic rise and recent decline. Mol Ecol. 23(10):2590-601.
- Walker, D.I. and McComb, A.J. (1990). Salinity response of the seagrass *Amphibolus antarctica*: an experimental validation of field results. Aquatic Botany, 36, 359–366.
- Ward, S. (2010). Biodiversity North. Department of Natural Resources and Environment Tasmania (NRETAS) Survey protocol for masked owls in the NT Tyto novaehollandiae (north Australian mainland subspecies T. n. kimberli and Tiwi subspecies T. n. melvillensis).
- Watkins, D. (1993). A National Plan for Shorebird Conservation in Australia. Australasian Wader Studies Group. RAOU Report No. 90.
- Wessex Archaeology (2023) Barossa Gas Export Pipeline Submerged Palaeolandscapes Archaeological Assessment Technical Report. Wessex Archaeology.
- West, L.D., Lyle, J.M., Matthews, S.R., Stark, K.E. & Steffe, A.S (2022) Survey of recreational fishing in the Northern Territory: 2018 to 2019. Available from: www.industry.nt.gov.au/survey-recreational-fishing-nt-2018-2019-report.pdf. (Accessed: 14/03/23).
- Whiting, S., Long, J., Hadden, K. & Council, T.L. (2005). Identifying the links between nesting and foraging grounds for the Olive Ridley (Lepidochelys olivacea) sea turtles in northern Australia. Report to the Department of the Environment and Water Resources.
- Whiting, S.D., Hadden, K., Long, J.L., Lauder, A.D.K., Kleidon, A. & Cook, K. (2007). Sea Turtle Conservation and Education on the Tiwi Islands. Final Natural Heritage Report. Canberra. Australian Government.
- Whiting, S.D. (2008). Movements and distribution of dugongs (Dugong dugon) in a macro-tidal environment in northern Australia. Australian Journal of Zoology. 56, 215–222.
- Whittock, P.A., Pendoley, K.L. & Hamann, M. (2016). Using habitat suitability models in an industrial setting: the case for internesting flatback turtles. Ecosphere, 7, e01551.
- Wilson, S., Polovina, J., Stewart, B. & Meekan, M. (2006). Movements of whale sharks (*Rhincodon typus*) tagged at Ningaloo Reef. Marine Biology, Vol. 147, 1157–1166.
- Wilson, P., Thums, M., Pattiaratchi, C., Meekan, M., Pendoley, K., Fisher, R. and Whiting, S., 2018. Artificial light disrupts the nearshore dispersal of neonate flatback turtles Natator depressus. Marine Ecology Progress Series 600: 179–192.
- Witherington, B.E. and Martin, R.E. (2003). Understanding, assessing and resolving light-pollution problems on sea turtle nesting beaches. Florida Marine Research Institute Technical Report TR-2 3rd Edition Revised, Florida Department of Environmental Protection, Tequesta, Florida, United States of America.
- Woinarski, J., Hadden, K., Hicks, J., and McLeod, D. (2003). Biodiversity Conservation on the Tiwi Islands, Northern Territory: Part 3. Management and Planning for biodiversity conservation. Department of Infrastructure, Planning and Environment, Darwin.
- Wolanski, E., 1994. Physical oceanographic processes of the Great Barrier Reef.CRC Press. Boca Raton.
- Woodside Energy Limited (2014). Browse FLNG Metocean Design Basis. Prepared for Woodside, Report reference JJ0013RT000002 Rev 5.
- World Resources Institute. (2021). State of Climate Action 2021: Systems Transformations Required to Limit Global Warming to 1.5°C, https://www.wri.org/research/state-climate-action-2021

# Appendix A

Santos Environment, Health and Safety policy

# Environment, Health & Safety



Policy

#### **Our Commitment**

Santos is committed to being the safest gas company wherever we have a presence and preventing harm to people and the environment

#### **Our Actions**

We will:

- 1. Integrate environment, health and safety management requirements into the way we work
- 2. Comply with all relevant environmental, health and safety laws and continuously improve our management systems
- 3. Include environmental, health and safety considerations in business planning, decision making and asset management processes
- 4. Identify, control and monitor risks that have the potential for harm to people and the environment, so far as is reasonably practicable
- 5. Report, investigate and learn from our incidents
- 6. Consult and communicate with, and promote the participation of all workers to maintain a strong environment, health and safety culture
- 7. Empower our people, regardless of position, to "Stop the Job" when they feel it necessary to prevent harm to themselves, others or the environment
- 8. Work proactively and collaboratively with our stakeholders and the communities in which we operate
- 9. Set, measure, review and monitor objectives and targets to demonstrate proactive processes are in place to reduce the risk of harm to people and the environment
- 10. Report publicly on our environmental, health and safety performance

#### Governance

The Environment Health Safety and Sustainability Committee is responsible for reviewing the effectiveness of this policy.

This policy will be reviewed at appropriate intervals and revised when necessary to keep it current.

#### Kevin Gallagher

Managing Director & CEO

Status: APPROVED

Document Owner:	David Banks, Chief Operating Officer			
Approved by:	The Board	Version:	3	
-	· · · · · · · · · · · · · · · · · · ·	-		

15 August 2022

# **Appendix B**

# Requirements applicable to the Activity

- Table B-1: Summary of Relevant Commonwealth Legislation
- Table B-2: Summary of Relevant Northern Territory Legislation
- Table B-3: Summary of Relevant International Agreements and Conventions
- Table B-4: EPBC Act Approval (2022/09372) Compliance Table
- Table B-5: EP Act Approval (2022/022-001) Compliance Table

#### Table B-1: Summary of Relevant Commonwealth Legislation

Requirement for Environmental Management	Administering Authority	Summary of requirement and how it will be met	OEMP section
Aboriginal and Torres Strait Islander Heritage Protection Act 1984 (Cth) (ATSIHP Act)	Commonwealth – Attorney-General's Department DCCEEW	The ATSHIP Act provides for the preservation and protection from injury or desecration areas and objects in Australia and Australian waters that are of significance to Aboriginal people in accordance with Aboriginal tradition. The Minister for the Environment and Water may make a declaration to protect such areas and objects. The Act also requires the discovery of Aboriginal remains to be reported to the Minister. The ATSIHP Act is not directly relevant to the Activity as there are no areas or objects within the OA or the EMBA the subject of a 'significant Aboriginal areas' or a 'significant Aboriginal object' declaration under the ATSIHP Act. Further, there are no requirements arising under the ATSIHP Act that apply to the environmental management of the Activity. However, in the event that such areas or objects are declared in the future, this Act could potentially become relevant to the activities. Accordingly, this Act has been identified for completeness. Santos notes that on 23 October 2023 it was informed by the DCCEEW that applications had been received under the ATSIHP Act in relation to certain areas of the sea. Santos understands that these areas are at least 31.5 km from the OA but overlap a small portion of the outer limits of the EMBA. Santos understands that no decisions have been made by the Minister in relation to the applications at the time of writing.	There are no requirements arising under the ATSIHP Act that apply to the environmental management of the Activity. Refer to Section 3.2.14.7 and Section 3.2.15 – in relation to relevant heritage values and cultural features more broadly.
Aboriginal Land Rights (Northern Territory) Act 1976 (Cth) (ALR Act)	Commonwealth – Attorney-General's Department Department of the Prime Minister and Cabinet	The main purpose of this Act is to provide for the granting of traditional Aboriginal land in fee simple to be held by Aboriginal Land Trusts for the benefit of Aboriginal people entitled by Aboriginal tradition to the use or occupy the land. The ALR Act is not directly relevant to the environmental management of the Activity. There is no Aboriginal land either claimed or granted under the ALR Act, or sea closures put into effect in accordance with that Act, that overlap with the OA. There are no predicted impacts to land or nearshore locations (including the Tiwi Islands) declared under the ALR Act associated with the Activity. However, the EMBA for the Activity, associated with an unplanned MDO spill, does overlap with Aboriginal land declared under the ALR Act. As such, this Act has been included to give context to Santos' consultation with relevant Land Councils established under the Act.	Section 3.2.15 – Cultural features Section 4– Consultation
Australian Maritime Safety Authority Act 1990 (Cth) (AMSA Act)	AMSA	This Act establishes the Australian Maritime Safety Authority which manages the National Plan for Maritime Environmental Emergencies in coordination with industry. AMSA is also responsible for administering Marine Orders in Commonwealth waters. The Act aims to promote maritime safety, protect the marine environment from pollution and environmental damage from ships, provide for a national search and rescue service and promote the efficient provision of service by AMSA. AMSA is the lead agency for responding to oil spills in the marine environment and is responsible for the Australian National Plan for Maritime Environmental Emergencies. While the Act does not contain any explicit requirements relevant to the Activity, it establishes and sets out the functions of AMSA, which relate to environmental management including in respect of response to spill events and	Table 4-9 – Summary of Relevant Persons Table 4-11 – Consultation Summary Table – Commonwealth Government Agency or Authority Sections 7.4 to 7.5 – Hydrocarbon spills

Requirement for Environmental Management	Administering Authority	Summary of requirement and how it will be met	OEMP section
		administration of Marine Orders. The Act applies to the use of any vessel associated with operations and is relevant to the activity in respect of any unplanned pollution from ships. AMSA has also been consulted as a Relevant Person and will be notified throughout activities in accordance with Table 8-13. AMSA's relevant functions are described in Table 4-9.	Section 7.7 – Contingency spill response operations Table 8-13 – Activity notification and reporting requirements
Biosecurity Act 2015 (Cth) Biosecurity Regulations 2016 (Cth) Biosecurity Amendment (Biofouling Management) Regulations 2021 (Cth) Australian Ballast Water Management Requirements, Version 8 Australian Biofouling Management Requirements (DAWE 2022)	Commonwealth – Department of Agriculture, Fisheries and Forestry	This Act and its supporting regulations are the primary legislative means for managing diseases and pests that may cause harm to human, animal or plant health, or the environment. This Act includes provisions for ballast water management plans and certificates, record keeping obligations and powers to ensure compliance. This Act includes mandatory controls on the use of seawater as ballast in ships and the declaration of sea vessels voyaging out of and into Commonwealth waters. The Regulations stipulate that all information regarding the voyage of the vessel and the ballast water is declared correctly to the quarantine officers. The Australian Ballast Water Management Requirements outline the mandatory ballast water management requirements to reduce the risk of introducing harmful aquatic organisms into Australia's marine environment through ballast water from international vessels. These requirements are enforceable under the <i>Biosecurity Act 2015</i> (Cth) and include obligations under the International Convention for the Control and Management of Ships' Ballast Water and Sediments. This Act and Regulations apply to all foreign vessels operating in Australian waters and these vessels are required to comply with the requirements of this Act, the Regulations, the Australian Ballast Water Management Requirements, and the Australian Biofouling Management Requirements.	Section 7.2 – Introduction of invasive marine species Section 8.3.2.9 – Biosecurity management
<i>Climate Change Act</i> 2022 (Cth) (Climate Act)	Commonwealth – Climate Change Authority	The Climate Act commenced in September 2022. The Climate Act sets out Australia's net-zero commitments and codifies Australia's net 2030 and 2050 GHG emissions reductions targets under the Paris Agreement. While the oil and gas sector is not subject to direct obligations under this Acts, it legislates Australia's emissions net zero targets by 2050. The Santos Climate Change Policy and target to become a net-zero scope 1 and 2 GHG emissions by 2040 are aligned with these Acts. The activities covered under this GEP Coastal Waters OEMP are consistent with the principles of ESD and ALARP to mitigate GHG emissions.	Section 6.3 – Atmospheric and greenhouse gas emissions
Environment Protection and Biodiversity Conservation Act	Commonwealth – DCCEEW NOPSEMA	While the OPGGS (E) Regulations under the OPGGS Act (see below) regulate day to day petroleum activities and apply to any activity that may have an impact on the environment, the EPBC Act regulates the assessment and approval of proposed actions that are likely to have a significant impact on a matter of National Environmental Significance (MNES). Actions that are	Section 3 – Description of the environment

Requirement for Environmental Management	Administering Authority	Summary of requirement and how it will be met	OEMP section
1999 (Cth) (EPBC Act) Environment Protection and Biodiversity Conservation Regulations 2000 (Cth) (EPBC Regulations)	DNP	<ul> <li>likely to have a significant impact on a MNES typically require referral under the EPBC Act, and the assessment process is administered by the DCCEEW. To protect, maintain and enhance recovery of certain threatened species and ecological communities listed under the EPBC Act, DCCEEW may prepare conservation management plans in the form of conservation advice or recovery plans. Australian Marine Parks (AMP) are established under the EPBC Act, and each AMP zone is based on the principles of the Australian International Union for Conservation of Nature (IUCN). Each of the AMPs have a management plan to give effect to management principles and objectives. Schedule 8 of the EPBC Regulations outlines the IUCN Reserve Management Principles. In addition, the EPBC Regulations provide for the protection and conservation of cetaceans, and create various offences for actions that may endanger them.</li> <li>This Act is relevant to the environmental management of the Activity and applies to all aspects of the Activity that have the potential to impact MNES.</li> <li>The DPD Project, inclusive of the NT Coastal Waters OA, was referred under the EPBC Act (EPBC 2022/09372) and assessed as a controlled action. This approval constitutes the Commonwealth Government's primary approval for the DPD Project and authorises the installation, precommissioning, operation and decommissioning of part of the Barossa GEP addressed in this GEP Coastal Waters OEMP. The DPD Project was deemed a 'controlled action' on 6 December 2022 and following assessment, was approved on 15 March 2024.</li> <li>Relevant conditions of EPBC approval 2022/0937 are included in Table B-4. Santos will comply with the requirements of Environmental Approval (EPBC 2022/09372) to the extent applicable in NT Coastal Waters (refer to Table B-4).</li> </ul>	Section 6 – Planned activities impact assessment Section 7 – Unplanned events risk assessment
Fisheries Management Act 1991 (Cth) (FM Act)	Commonwealth - Department of Agriculture, Fisheries and Forestry Commonwealth – Australian Fisheries Management Authority	Management plans for fisheries are established under the FM Act, and the FM Act also sets out the legislative basis for Statutory Fishing Rights (SFRs), licences and permits. The Act defines the Australian Fishing Zone (AFZ) and provides for the majority of Commonwealth fisheries offences. The Act also establishes the functions of the Australian Fisheries Management Authority (AMFA), including in relation to the pursuit of ecologically sustainable development. The FM Act is not directly relevant to the environmental management of the Activity. However, in the event of a spill, the Act provides the regulatory framework for any necessary fisheries management decisions in Commonwealth waters. Further, the AFMA is responsible for managing Commonwealth fisheries and is a relevant agency where the Activity has the potential to impact on fisheries resources in AFMA managed fisheries. The OA and EMBA overlap four Commonwealth commercial fisheries managed by the AFMA. Accordingly, this Act has been identified for completeness (and to provide context for the consultation undertaken by Santos with the AFMA in the course of preparing this environment plan).	Section 3.2.14.1 – Commercial Fisheries Section 4.8 – Santos' consultation methodology Table 4-11– Consultation Summary Table – Commonwealth Government Agency or Authority Section 6 - Planned activities impact assessment Section 7 - Unplanned events risk assessment

Requirement for Environmental Management	Administering Authority	Summary of requirement and how it will be met	OEMP section
Hazardous Waste (Regulation of Exports and Imports) Act 1989 (Cth)	Commonwealth – DCCEEW	This Act regulates the import, export and transport of hazardous waste. The Act aims to ensure that exported, imported or transited waste is managed in an environmentally sound manner so that human beings and the environment, both within and outside Australia, are protected from the harmful effects of the waste. The Act gives effect to the Basel Convention on the Control of Transboundary Movements of Hazardous Wastes and their Disposal, 1972 (commonly referred to as the Basel Convention). This Act applies to the import, export and transport of hazardous wastes required for the Activity, which will comply with the requirements of the Act.	Section 6.4 – Operational discharges Section 8.4.2 – Monitoring and recording emissions and discharges
Marine Orders	Commonwealth - AMSA	<ul> <li>Marine Orders are subordinate rules made pursuant to the Navigation Act 2012 (Cth), the Protection of the Sea (Prevention of Pollution from Ships) Act 1983 (Cth), Protection of the Sea (Harmful Anti-fouling Systems) Act 2006 (Cth) and the Marine Safety (Domestic Commercial Vessel) National Law Act 2012 (Cth) affecting the maritime industry. They are a means of implementing Australia's international maritime obligations by giving effect to international conventions in Australian law.</li> <li>There are two series of marine orders, being those made to give effect to international obligations and standards and apply to regulated Australian vessels, foreign vessels and some domestic commercial vessels (Marine Orders 1-98). In addition, Marine Orders 500-507 are made under the Marine Safety (Domestic Commercial Vessel) National Law Act 2012 (Cth).</li> <li>Various Marine Orders apply to the Activity, including in relation to discharges and emissions. The Marine Orders relevant to this GEP Coastal Waters OEMP include:</li> <li>Marine Order 21: Safety and emergency arrangements</li> <li>Marine Order 30: Prevention of collisions</li> <li>Marine Order 41: Carriage of dangerous goods</li> <li>Marine Order 71: Masters and deck officers</li> <li>Marine Order 91: Marine pollution prevention – oil.</li> <li>Marine Order 91: Marine pollution prevention – packaged harmful substances</li> <li>Marine Order 95: Marine pollution prevention – packaged harmful substances</li> <li>Marine Order 95: Marine pollution prevention – sewage</li> <li>Marine Order 95: Marine pollution prevention – sewage</li> <li>Marine Order 97: Marine pollution prevention – sir pollution</li> <li>Marine Order 96: Marine pollution prevention – air pollution</li> </ul>	Section 2.5 – Vessel and helicopter operations Section 6 – Planned activities impact assessment Section 7 – Unplanned events risk assessment Section 8.4.2 – Monitoring and recording emissions and discharges

Requirement for Environmental Management	Administering Authority	Summary of requirement and how it will be met	OEMP section
		Discharges to the marine environment caused by the Activity will be recorded and controlled in accordance with relevant Marine Orders. Santos has also implemented control measures directed to ensuring compliance with Marine Orders.	
Marine Safety (Domestic Commercial Vessel) National Law Act 2012 (Cth) Marine Safety (Domestic Commercial Vessel) National Law Regulation 2013 (Cth)	Commonwealth – AMSA	This Act is a single regulatory framework for the certification, construction, equipment, design and operation of domestic commercial vessels inside Australia's exclusive economic zone. The Act names AMSA as the National Marine Safety Regulator and confers functions on AMSA in relation to marine safety, including that AMSA may make and maintain Marine Orders. The Regulations under the Act set out the definition of a vessel and details and requirements of the accredited marine surveyor scheme. The Act also sets requirements in relation to the survey of marine vessels which any Australian Activity vessels must comply with. All vessel movements associated with the Activity will be governed by AMSA marine safety regulations provided for under this Act, in addition to NT requirements, to the extent relevant. This Act also imposes duties on owners, masters and crew of domestic commercial vessels in relation to the safety of the vessel, relevant to owners, masters and crew of any Australian Activity vessels under this GEP Coastal Waters OEMP. Santos, when engaging vessel contractors, shall assure the vessel contractors compliance with applicable maritime law and regulations.	Section 6.5 – Interaction with other marine users Section 7.6 – Surface release of MDO from a vessel Section 7.7 – Contingency spill response operations
National Biofouling Management Guidance for the Petroleum Production and Exploration Industry 2009	Commonwealth – Department of Agriculture, Fisheries and Forestry	The guidance document provides recommendations for the management of biofouling hazards by the petroleum industry. The recommendations and biofouling controls set out within this document will be applied to the Activity in order to reduce the risk of the introduction of an IMS.	Section 7.2 – Introduction of invasive marine species
National Environmental Protection Council Act 1994 (Cth)	Commonwealth - DCCEEW	This Act establishes the National Environmental Protection Council (NEPC) that sets National Environmental Protection Measures (NEPMs). NEPMs are a set of national objectives designed to assist in protecting or managing particular aspects of the environment, to ensure that Australians have equivalent protection from air, water, soil and noise pollution. This Act is mirrored in all States and Territories. The Activity will be undertaken in line with the principles of ecologically sustainable development, and impacts and risks resulting from these activities relevant to National Environment Protection Measures national objectives will be demonstrated to be ALARP and acceptable.	Section 6.3 – Atmospheric and greenhouse gas emissions Section 8.4.2– Monitoring and recording emissions and discharges
National Greenhouse and Energy Reporting Act 2007 (Cth) (NGER Act)	Commonwealth – DCCEEW Clean Energy Regulator	The NGER Act provides for a single national reporting framework for the reporting and dissemination of information about greenhouse gas emissions, greenhouse gas projects and energy use and production of corporations. The NGER Act applies to the atmospheric emissions generated by fugitive emissions from pipeline transmission of dry gas (scope 1 emissions).	Section 6.3 – Atmospheric and greenhouse gas emissions

Requirement for Environmental Management	Administering Authority	Summary of requirement and how it will be met	OEMP section
National Greenhouse and Energy Reporting (Safeguard Mechanism) Rule 2015 (Cth)	Climate Change Authority	The Safeguard Mechanism is also administered under the NGER Act. The Safeguard Mechanism applies to the Barossa Gas Project gas production activities, specifically, the Barossa Gas Project gas field will be a designated large facility under the NGER Act, and, as such will be subject to the Safeguard Mechanism. This means that Santos, among other things, will have an obligation to ensure that the net covered emissions of GHGs from the production of gas at the Barossa gas field do not exceed the applicable baseline.	
Native Title Act 1993 (Cth) (Native Title Act)	Commonwealth – Attorney-General's Department Commonwealth – Department of the Prime Minister and Cabinet National Native Title Tribunal Federal Court of Australia	The NT Act recognises the rights and interests of Aboriginal and Torres Strait Islander people in land and waters according to their traditional laws and customs, and creates processes through which native title can be recognised and protected. Under s 280(2) of the OPGGS Act, petroleum activities must be carried out in a manner that does not interfere with the enjoyment of native title rights and interests under the NT Act to a greater extent than necessary. In addition, under s 124 of the PSL Act, activities carried out under a pipeline licence must be carried out in a manner that does not interfere with registered native title rights and interests Act to a greater extent than necessary. The NT Act is not directly relevant to environmental management of the Activity. There are no native title claims or determinations within the OA or EMBA. However, the NLC is a Representative Aboriginal/Torres Strait Islander Body under the NT Act for parts of the OA and EMBA. Accordingly, this Act has been identified for completeness (and to provide context for the consultation undertaken by Santos with the NLC in the course of preparing this GEP Coastal Waters OEMP).	Section 3.2.15 – Cultural features Section 4.8 Consultation Summary Table – First Nations People and Groups (NLC)
Navigation Act 2012 (Cth)	AMSA (operational) Commonwealth - Department of Infrastructure, Transport, Regional Development, Communication and the Arts	This Act aims to promote the International Convention for the Safety of Life at Sea 1974 (SOLAS) and safe navigation, prevent pollution of the marine environment and ensure that AMSA has the power to carry out inspections of vessels, and enforce national and international standards. Specifically, this Act empowers AMSA to make Marine Orders, which are legislative instruments, with respect to any matter for which provision must or may be made by the Regulations. AMSA has the authority and responsibility for the operational activities under the Act, including vessel certification, seafarers qualifications, marine pollution prevention, monitoring and enforcement activities. All vessel movements associated with the Activity will comply with the Act and Marine Orders made under the Act, as well as NT requirements (see <i>Marine Act 1981</i> (NT)). See Marine Orders, above.	Section 2.5 – Vessel and helicopter operations Section 6.5 – Interactions with other marine users Section 6.2 – Light emissions Section 6.3 – Atmospheric and greenhouse gas emissions Section 6.6.3 – Environmental performance outcomes and control measures Section 7.1 – Release of solid objects Section 7.2– Introduction of invasive marine species

Administering Authority	Summary of requirement and how it will be met	OEMP section
Commonwealth	Detroloum our longtion and doubloom on the stimitics in Australia's offehere errors are subject to the	Section 7.3 – Marine fauna interaction Section 7.4 – Minor Release of hydrocarbons and chemicals Section 7.6 – Surface release of MDO from a vessel Section 7.7 – Contingency spill response operations
Commonwealth – NOPSEMA Commonwealth – Department of Industry, Science and Resources Department of Mining and Energy	Petroleum exploration and development activities in Australia's offshore areas are subject to the environmental requirements specified in the OPGGS Act and associated Regulations. The OPGGS Act contains a broad requirement for titleholders to operate in accordance with 'good oil-field practice'. Specific environmental provisions relating to work practices essentially require operators to control and prevent the escape of wastes and petroleum. The Act also requires that activities are performed in a manner that does not unduly interfere with other rights or interests, including the conservation of the resources of the sea and sea-bed, such as fishing or shipping. In some cases, where there are environmental sensitivities or multiple use issues, it may be necessary to apply special conditions to an exploration permit area. The holder of a petroleum title must maintain adequate insurance against expenses or liabilities arising from activities in the title, including expenses relating to clean-up or other remedying of the effects of the escape of petroleum.	Requirements under the OPGGS Act and associated Regulations are addressed throughout this GEP Coastal Waters OEMP.
	The OPGGS(E) Regulations provide an objective-based regime for managing environmental performance for Australian offshore petroleum exploration and production activities in areas of Commonwealth jurisdiction. The OPGGS(E) Regulations are applied to petroleum operations within NT Coastal Waters by the PSL Act and the <i>NT Petroleum (Submerged Lands) (Application of Commonwealth Laws) Regulations 2004</i> (NT). Key objectives of the OPGGS(E) Regulations include to ensure that a petroleum activity carried out in an offshore area is:	
	<ul> <li>carried out in a way that is consistent with the principles of ecologically sustainable development as set out in section 3A of the EPBC Act; and</li> <li>carried out in a manner by which the environmental impacts and risks of the activity will be reduced to ALARP and be of an acceptable level.</li> <li>This Coastal Waters OEMP demonstrates that the Activity will be undertaken in line with the principles of ecologically sustainable development, and that the environmental impacts and risks</li> </ul>	
	Administering Authority	Administering Authority         Summary of requirement and how it will be met           Commonwealth – NOPSEMA Commonwealth – NOPSEMA Commonwealth – NOPSEMA Commonwealth – NOPSEMA Commonwealth – Department of industry, Science and Resources Department of Mining and Energy         Petroleum exploration and development activities in Australia's offshore areas are subject to the environmental requirements specified in the OPGGS Act and associated Regulations. The OPGGS Act contains a broad requirement for titleholders to operate in accordance with 'good oil-field Department of industry, Science and Resources           Department of Wining and Energy         The Act also requires that activities are performed in a manner that does not unduly interfere with other rights or interests, including the conservation of the resources of the sea and sea-bed, such as fishing or shipping. In some cases, where there are environmental sensitivities or multiple use issues, it may be necessary to apply special conditions to an exploration permit area. The holder of a petroleum title must maintain adequate insurance against expenses or liabilities arising from activities in the title, including expenses relating to clean-up or other remedying of the effects of the escape of petroleum. The OPGGS(E) Regulations provide an objective-based regime for managing environmental performance for Australian offshore petroleum exploration and production activities in areas of Commonwealth Laws) Regulations 2004 (NT). Key objectives of the OPGGS(E) Regulations include to ensure that a petroleum activity carried out in an offshore area is: • carried out in a way that is consistent with the principles of ecologically sustainable development as set out in section 3A of the EPBC Act; and • carried out in a way that is consistent with the principles of ecologically sustainable development as set out in section 3A of the EPBC Act; a

Requirement for Environmental Management	Administering Authority	Summary of requirement and how it will be met	OEMP section
Ozone Protection and Synthetic Greenhouse Gas Management Act 1989 (Cth) (and associated Regulations)	Commonwealth – DCCEEW	This Act regulates the manufacture, importation and use of ozone depleting substances (ODSs) (typically used in fire-fighting equipment and refrigerants) and synthetic greenhouse gases, and is applicable to the handling of any ODS. The Act provides a licensing system for the import, export and manufacture of ODSs and equipment containing ODSs, while the Regulations control the enduse of ODSs, which are licenced by DCCEEW. While the Activity does not include import, export or manufacture activities of ODS, this Act applies where ODS is found on Activity vessel refrigeration systems (which is a rare occurrence). The Activity vessels may use ODSs which would be regulated under this Act. Santos, when engaging vessel contractors, shall assure the vessel contractors compliance with applicable maritime law and regulations. Relevant Activity vessels will follow ODS handling procedures.	Section 6.3 – Atmospheric and greenhouse gas emissions
Protection of the Sea (Powers of Intervention) Act 1981 Protection of the Sea (Powers of Intervention) Regulations 1983 (Cth)	AMSA Commonwealth – Department of Infrastructure, Transport, Regional Development, Communication and the Arts	The Act authorises the Commonwealth (through AMSA) to take measures for the purpose of protecting the sea from pollution by oil and other noxious substances discharged from ships and provides legal immunity for persons acting under an AMSA direction, including measures provided for under both of the International Convention Relating to Intervention on the High Seas in Cases of Oil Pollution Casualties and the Protocol Relating to Intervention on the High Seas in Cases of Pollution by Substances other than Oil, 1973. The Regulations set out requirements to notify AMSA in respect of changes in the ownership or master of a vessel. This Act applies to vessel discharges and movements associated with the Activity, and Santos is required to comply with the Act in the event of a spill of oil or noxious subjects from a ship. Further, the Act confers powers on AMSA to take action in the event of a spill of oil or noxious subjects from a ship, which functions are relevant in the event of an MDO spill arising from activities under this GEP Coastal Waters OEMP.	Section 6.5 – Interaction with other marine users Section 6.4 – Operational discharges Section 7.4 – Minor release of hydrocarbons and chemicals Section 7.6 – Surface release of MDO from a vessel Section 7.7 – Contingency spill response operations
Protection of the Sea (Prevention of Pollution from Ships) Act 1983 (Cth) Protection of the Sea (Prevention of Pollution from Ships) (Orders) Regulations 1994 (Cth)	AMSA Commonwealth – Department of Infrastructure, Transport, Regional Development, Communication and the Arts	This Act relates to the protection of the sea from pollution by oil and other harmful substances discharged from ships and implements into domestic law Australia's obligations under the MARPOL convention, which sets out the legislative obligations relating to the prevention of accidental and operational marine environment pollution from shipping. This Act disallows any harmful discharge of sewage, oil and noxious substances into the sea and sets the requirements for a shipboard waste management plan. This Act also provides for the making of marine orders relating to marine pollution prevention, which give effect to relevant regulations of Annexes I, II, III, IV, V and VI of MARPOL 73/78. This Act applies to vessel discharges and movements associated with the Activity. Santos and its contractors must comply with relevant requirements under this Act and Regulations in respect of Activity vessels, including requirements to have a shipboard oil pollution emergency plan and a marine pollution emergency plan. Santos notes that the requirement to maintain a ship energy	Section 6.3 – Atmospheric and greenhouse gas emissions Section 6.4 – Operational discharges Section 7.4 – Minor release of hydrocarbons and chemicals Section 7.6 – Surface release of MDO from a vessel Section 7.7 – Contingency spill response operations

Requirement for Environmental Management	Administering Authority	Summary of requirement and how it will be met	OEMP section
		efficiency management plan is not applicable to Activity vessels as the vessels will not be engaged on an overseas voyage when undertaking activities under this GEP Coastal Waters OEMP.	
Protection of the Sea (Civil Liability of Bunker Oil Pollution Damage) Act 2008	AMSA	This Act implements the requirements for the International Convention on Civil Liability for Bunker Oil Pollution Damage imposing insurance certification requirements in respect of regulated Australian vessels carrying more than 2,000 tonnes of oil in bulk as cargo. Activities under this GEP Coastal Waters OEMP do not involve the use of any vessels carrying over 2,000 tonnes of oil, as regulated under the Act.	Section 7.6 – Surface release of MDO from a vessel
Protection of the Sea (Harmful Antifouling Systems) Act 2006	Commonwealth, Department of Infrastructure, Transport, Regional Development, Communications and the Arts AMSA	This Act relates to protecting the sea from the effects of harmful anti-fouling systems. It prohibits the use of harmful organotins in anti-fouling paints used on ships. This is also implemented through Marine Order 98 (Marine pollution – anti-fouling systems). This Act applies to vessel movements in Australian waters associated with the Activity, which are required to have biofouling systems in place to prevent introduction of IMS and harmful impacts on Australian biodiversity.	Section 7.2 – Introduction of invasive marine species
Underwater Cultural Heritage Act 2018 (Cth) (UCH Act) Underwater Cultural Heritage (Consequential and Transitional Provisions) Act 2018 (Cth)	Commonwealth – DCEEW The NT Heritage Branch has jurisdiction over Commonwealth waters North of the NT	<ul> <li>The UCH Act replaces the <i>Historic Shipwrecks Act</i> 1976 (Cth) and extends protection to other wrecks such as submerged aircrafts, human remains and other types of underwater cultural heritage including Aboriginal and Torres Strait Islander Underwater Cultural Heritage.</li> <li>Under the UCH Act, heritage that has been in Commonwealth / Australian waters for at least 75 years is automatically protected, while other heritage can be declared to be protected by the Minister. It is an offence to interfere with heritage covered by this Act.</li> <li>Key obligations imposed under the UCH Act include: <ul> <li>not disturbing protected underwater heritage during the course of a proposed action without a permit;</li> <li>observing the requirements of protected zones and obtaining a permit to enter one if required; and</li> <li>providing notification in respect of the discovery of any suspected underwater heritage identified during the course of proposed action within 21 days of discovery.</li> </ul> </li> <li>There is one UCH site within the OA protected under the UCH Act, the wreck of Japanese submarine 1-124 with an 800m radial exclusion zone (refer Section 3.2.14.7. In addition, several known historic shipwrecks occur within the EMBA. Some unlocated wrecks could fall within the boundaries of the OA or EMBA. Despite this, there is no predicted impact to cultural heritage values in relation to these shipwrecks resulting from activities under the GEP Coastal Waters OEMP, including from unplanned risks. Although there are no presently predicted impacts, the UCH Act improved which is provide in particle of the order are no presently predicted impacts, the UCH Act improved which is provide on particle of the order are no presently predicted impacts, the UCH Act improved which is provide on the particle of the proved heritage her</li></ul>	Section 3.2.14.7 – Underwater cultural heritage Section 3.2.15 – Cultural features Section 6.4 – Seabed and benthic habitat disturbance.



Requirement for Environmental Management	Administering Authority	Summary of requirement and how it will be met	OEMP section
		waters and the OA. The UCH Act requires that that anyone who finds an article of underwater cultural heritage which appears to be of an archaeological character needs to notify the relevant authorities, via online form.	

#### **Table B-2:** Summary of Relevant Northern Territory Legislation

State Legislation	Administering Authority	Summary of relevant requirement and how it will be met	EP Section
Aboriginal Land Act 1978 (NT)	Department of the Chief Minister and Cabinet (DCMC)	This Act provides for the access to Aboriginal land, certain roads bordered by Aboriginal land and the seas adjacent to Aboriginal land. While there are no planned activities associated with this GEP Coastal Waters OEMP that require access to Aboriginal land under the Act, there may be contingency spill response activities undertaken, in response to an unplanned spill event, that require access to Aboriginal land and adjacent seas, under the direction of the relevant Control Agency.	Section 7 – Unplanned events risk assessment
Dangerous Goods Act 1998 (NT) and Dangerous Goods Regulations 1985 (NT)	NT - Department of the Attorney-General and Justice	This Act relates to the handling of certain dangerous goods within the NT. The Regulations provide requirements for the safe handling, storage and transportation of dangerous goods, including the provision of adequate training for personnel, suitable labelling, storage facilities and on-site emergency response capability. This Act applies in relation to the handling of dangerous goods in NT waters.	Section 6.4 – Operational Discharges
Environmental Offences and Penalties Act 1996 (NT)	Department of Environment Parks and Water Security (DEPWS)	This Act, defines levels and penalties for environmental offences under various Acts, including the NT EP Act, <i>Waste Management and Pollution Control Act 1998</i> and the <i>Water Act 1992</i> . Santos will ensure the activity complies with the requirements of these Acts. The Act is not directly relevant to the environmental management of the activity. However, the Act may also apply to the extent that a credible spill scenario or other discharge may result in impact to NT waters.	Section 6 – Planned activities impact assessment Section 7 – Unplanned events risk assessment

State Legislation	Administering Authority	Summary of relevant requirement and how it will be met	EP Section
Environment Protection Act 2019 (NT)	NT - Department of Environment, Parks and Water Security NT EPA	This Act is the principle environment legislation in the NT. The Act establishes the NT EPA whose key objectives are to promote ecologically sustainable development, protect the environment, and promote key effective waste management and minimisation strategies. The Act also provides for environmental impact assessment and approval for specific actions that may have a significant impact on the environment or that meet a referral trigger. In December 2023, the DPD Project (in NT waters) was approved (EP2022/022-001) under the Act following assessment by the NT EPA, subject to certain conditions and associated management plans relating to environmental management for activities in NT jurisdiction, including the Activity in the OA (refer to Table B-5). The Act may also apply to the extent that a credible spill scenario may result in impact to NT waters. Santos will comply with requirements of Environmental Approval (EP2022/022-001) applicable to the Activity in NT Coastal Waters.	Section 7.4 – Minor release of hydrocarbons and chemicals Section 7.6 – Surface release of MDO from a vessel Section 7.7 – Contingency spill response operations
Environment Protection (National Pollutant Inventory) Objective 2004 (NT)	NT EPA	The National Pollutant Inventory (NPI) provides information on the types and amounts of certain substances being emitted to the air, land and water or transported in waste. This is an objective under the <i>Waste Management and Pollution Control Act</i> that provides for compulsory reporting of air emissions by certain facilities, in accordance with the Commonwealth National Environment Protection (National Pollutant Inventory) Measure. Reporting for GHG emissions associated with the Activity will comply with these requirements.	Section 6.3 – Atmospheric emissions
Fisheries Act 1988 (NT) Fisheries Regulations 1992 (NT)	NT - Department of Industry, Tourism and Trade – Fisheries Division	The <i>Fisheries Act 1988</i> (NT) provides for the regulation, conservation and management of fisheries and fishery resources so as to maintain their sustainable utilisation, to regulate the sale and processing of fish and aquatic life, and for related purposes. There are no requirements directly relevant to the environmental management of the Activity. However, in the event of an emergency, the Act provides the regulatory framework for the Joint Authority Fishery (such as the Timor Reef Fishery) to make any necessary fisheries management decisions. The OA overlaps six NT managed fisheries, and the EMBA overlaps 12 NT managed fisheries regulated under this Act. Accordingly, this Act has been identified for completeness (and to provide context for the consultation undertaken by Santos with the NT Department of Industry, Tourism and Trade in the course of preparing this GEP Coastal Waters OEMP).	Section 3.2.14.1 – Commercial fisheries Table 4-12 – Consultation Summary Table – NT Government Agency or Authority Section 7 – Unplanned events risk assessment

State Legislation	Administering Authority	Summary of relevant requirement and how it will be met	EP Section
Heritage Act 2011 (NT)	NT - Department of Territory Families, Housing and Communities	This Act establishes the NT Heritage Council and governs protection of both natural and cultural heritage places within the NT jurisdiction by establishing heritage offences and regulating activities that may impact heritage places and objects, including through a process for obtaining works approvals. While the Activity is not likely to impact natural and cultural heritage places or objects in the NT, the Act is relevant in the event that unplanned events may impact natural and cultural places, or objects in the NT, constituting an offence under the Act. Santos has conducted a maritime archaeological heritage assessment of the DPD Project, inclusive of the pipeline route within the GEP Coastal Waters OEMP OA, as per an Archaeological Scope of Work provided by the Heritage Branch of the DTFHC (refer Section 3.2.14.7). The EPOs, Control Measures, EPSs, supporting management processes/procedures and reporting requirements within this GEP Coastal Waters OEMP have been developed to ensure that underwater cultural heritage is protected during the Activity and requirements of the <i>Heritage Act 2011</i> (NT) are met.	Section 3.2.14.7 – Underwater cultural heritage Section 7.6 – Surface release of MDO from a vessel Section 7.7 – Contingency spill response operations
Marine Pollution Act 1999 (NT) Marine Pollution Regulations 2003 (NT)	NT – Department of Environment, Parks and Water Security	<ul> <li>This Act protects the NT marine and coastal environment from ship-sourced pollution, including litter and rubbish, hydrocarbons and substances that may be hazardous to the marine environment (including substances that may be in ballast and grey water). This Act also gives effect to the following annexes of MARPOL in NT waters: <ul> <li>Annex I (which deals with pollution by oil)</li> <li>Annex II (which deals with pollution by noxious liquid substances in bulk)</li> <li>Annex III (which deals with pollution by harmful substances in packaged form)</li> <li>Annex V (which deals with pollution by garbage).</li> </ul> </li> <li>This Act is applicable to the extent that unplanned events may impact NT waters and regulates ship-sourced pollution in NT waters, inclusive of the OA, prohibiting certain discharges in line with MARPOL (annexures I, II, III and V).</li> </ul>	Section 6.4 – Operational discharges Section 7.1 – Release of solid objects Section 7.4 – Minor release of hydrocarbons and chemicals Section 7.6 – Surface release of MDO from a vessel Section 7.7 – Contingency spill response operations

State Legislation	Administering Authority	Summary of relevant requirement and how it will be met	EP Section
Northern Territory Aboriginal Sacred Sites Act 1989 (NT) Northern Territory Aboriginal Sacred Sites Regulations 2004 (NT)	NT – Aboriginal Areas Protection Authority (AAPA)	This Act establishes procedures for the protection and registration of sacred sites and the avoidance of sacred sites in the development and use of land. The Act also provides for entry onto sacred sites and specifies the conditions that apply to such entry, and establishes the Aboriginal Areas Protection Authority, who is responsible for, among other things, the enforcement of the Act. The AAPA has issued Authority Certificates (C2022-098 which covers potential seabed disturbance along the pipeline route in NT waters and a nominal ~1,000m buffer each side of the pipeline route, including the DPD pipeline route in the OA. Based on AAPA's research findings, there were no registered or recorded sacred sites, protected under the <i>Northerm Territory Aboriginal Sacred Sites Act 1989</i> (NT) (NTASS Act), identified within or adjacent to the area represented by a 1,000m buffer each side of the pipeline route within the OA, nor any specific certificate conditions related to activities within the buffer area. In addition, there are many NT coastal sites along the mainland and island coastlines and potentially the surrounding waters that overlap the EMBA that are protected under the Act (whether registered, recorded, or not). There are no registered sacred sites in the OA. As a result, no credible impacts to known sites are expected from planned activities. However, the Act will be applicable in the unlikely event	Section 3.2.15 – Cultural features Section 7.6 – Surface release of MDO from a vessel Section 7.7 – Contingency spill response operations
		that an unplanned event may impact sacred sites protected under this Act, constituting an offence.	
NT Petroleum (Submerged Lands) (Application of Commonwealth Laws) Regulations 2004	NT DME	Associated regulations of the PSL Act. These regulations apply certain Commonwealth laws to the submerged lands in the NT. The Activity falls under the jurisdiction of the PSL Act and the associated <i>NT Petroleum</i> ( <i>Submerged Lands</i> ) (Application of Commonwealth Laws) Regulations 2004. Planned and unplanned events may impact on coastal waters.	Requirements under the NT Petroleum (Submerged Lands) (Application of Commonwealth Laws) Regulations 2004 are addressed throughout this GEP Coastal Waters OEMP.
Petroleum (Submerged Lands) Act 1981 (PSL Act)	NT DME	The PSL Act allows for the creation of provisions with respect to the exploration for and the exploitation of the petroleum resources, and certain other resources, of certain submerged lands adjacent to the coasts of the NT and for other purposes. The PSL Act applies to NT Coastal Waters, that is, waters between the NT/Commonwealth waters boundary and the Territorial Sea Baseline. The Activity falls under the jurisdiction of the PSL Act. Planned and unplanned events may impact on coastal waters.	Requirements under the PSL Act and associated Regulations are addressed throughout this GEP Coastal Waters OEMP.

State Legislation	Administering Authority	Summary of relevant requirement and how it will be met	EP Section
Validation (Native Title) Act 1994 (NT)	Department of the Attorney-General	This Act, administered by the Department of the Attorney-General, validates certain acts attributable to the Territory, to make provision for the effect of certain acts attributable to the Territory on native title and for related purposes.	Section 3.2.15.3
		There are no planned activities associated with this GEP Coastal Waters OEMP that will affect determined areas under this Act. The EMBA associated with a worst-case spill event does not intersect the Larrakia Native Title determination. Refer to entry for the <i>Native Title Act 1993</i> (Cth) (NT Act).	
Waste Management and Pollution Control Act 1998 (NT) (WMPC Act)	NT EPA NT – Department of Environment, Parks and Water Security	This Act provides for the protection of the NT environment though encouraging effective waste management and pollution prevention and control practices. Under the WMPC Act, environmental protection approvals and licences are required to authorise specific activities and operations.	Section 6.4 – Operational discharges
		Santos intends to conduct the Activity consistent with the general environmental duty (Section 12 of the Act), whereby a person must not carry out any activity that results in pollution that causes or is likely to cause environmental harm, unless measures to prevent or minimise the harm have been taken, and a duty to notify the administering authority where an incident has caused or threatens to cause serious or material environmental harm.	
		This Act is applicable to the extent that an unplanned event may impact NT lands and waters, in which case, spill response operations will be undertaken in accordance with plans produced under this Act (e.g., the NT Oil Spill Contingency Plan), in consultation with relevant NT response agencies	

#### **Table B-3:** Summary of Relevant International Agreements and Conventions

International agreements and conventions	Summary of relevant requirement and how it will be met	EP section
1997 Treaty between Australia and Indonesia establishing the EEZ Boundary and Certain Seabed Boundaries (Perth Treaty)	This treaty has been signed but not yet ratified. When ratified, the treaty will finalise the EEZ boundary between Australia and Indonesia. Under the Perth Treaty, there are areas of overlapping jurisdiction where Australia exercises seabed jurisdiction including exploration for petroleum, and Indonesia exercises water column jurisdiction including fishing rights. There is no overlap with the EMBA of the GEP Coastal Waters OEMP.	Section 3.2.14.1 – Commercial fisheries

International agreements and conventions	Summary of relevant requirement and how it will be met	EP section
Agreement Between the Government of Australia and the Government of Japan for the Protection of Migratory Birds in Danger of Extinction and Their Environment 1974 (commonly referred to as the Japan Australia Migratory Bird Agreement [JAMBA])	This agreement recognises the special international concern for protecting Migratory birds and birds in danger of extinction that migrate between Australia and Japan. The agreement is implemented in the EPBC Act. Birds listed on the annex to this agreement must be placed on the migratory species list under the EPBC Act. Only relevant insofar that a credible spill scenario may result in impacts to migratory seabirds foraging in the OA or EMBA.	Section 3.2.13 – Threatened and migratory fauna Section 7.6 – Surface release of MDO from a vessel Section 7.7 – Contingency spill response operations
Agreement Between the Government of Australia and the Government of the People's Republic of China for the Protection of Migratory Birds and Their Environment 1986 (commonly referred to as the China Australia Migratory Bird Agreement [CAMBA])	This agreement recognises the special international concern for protecting Migratory birds and birds in danger of extinction that migrate between Australia and China. The agreement is implemented in the EPBC Act. Birds listed on the annex to this agreement must be placed on the migratory species list under the EPBC Act. Only relevant insofar that a credible spill scenario may result in impacts to migratory seabirds foraging in the OA or EMBA.	Section 3.2.13 – Threatened and migratory fauna Section 7.6 – Surface release of MDO from a vessel Section 7.7 - Contingency spill response operations
Agreement Between the Government of Australia and the Government of the Republic of Korea for the Protection of Migratory Birds and Their Environment 1986 (commonly referred to as the Republic of Korea Australia Migratory Bird Agreement [ROKAMBA])	This agreement recognises the special international concern for the protection of migratory birds and birds in danger of extinction that migrate between Australia and Korea. The agreement is implemented in the EPBC Act. Birds listed on the annex to this agreement must be placed on the migratory species list under the EPBC Act. Only relevant insofar that a credible spill scenario may result in impacts to migratory seabirds foraging in the OA or EMBA.	Section 3.2.13 – Threatened and migratory fauna Section 7.6 – Surface release of MDO from a vessel Section 7.7 – Contingency spill response operations

International agreements and conventions	Summary of relevant requirement and how it will be met	EP section
Convention for the Control of Transboundary Movements of Hazardous Wastes and Their Disposal	This convention deals with the transboundary movement of hazardous wastes, particularly by sea. The Hazardous Waste (Regulation of Exports and Imports) Act 1989 (Cth) gives effect to the Basel Convention in Australian law. The overarching objective of the Basel Convention is to protect human health and the environment against the adverse effects of hazardous wastes.	Section 8.4.2 – Monitoring and recording emissions and discharges
	requirements of the Basel Convention in respect of any import, export and transport of hazardous waste for the Activity).	
Convention on Wetlands of International Importance (Ramsar Convention)	The Ramsar Convention provides a framework for the conservation and wise use of wetlands and their resources. The EPBC Act gives effect to the Ramsar Convention by providing specific protection for wetlands recognised by the Ramsar Convention under Parts 2, 3 and 4 of the EPBC Act.	Section 3.2.12.2 – Wetlands of National and international importance
	The EMBA intersects with one Ramsar wetland – Cobourg Peninsula While no impacts are expected to this Ramsar wetland, this convention is applicable insofar as the credible spill scenario may result in impacts to Ramsar wetlands.	
Convention on the International Regulations	The COLREGS are a set of rules that apply to prevent the collision of vessels at sea and apply to vessels navigating waters outside of the COLREGS demarcation lines. The COLREGS rules, include, among other things, requirements to travel at safe speeds and keep watch keepers on deck of vessels, to reduce the likelihood of collisions. Activity vessels will comply with the COLREGS, including in particular, through the use of appropriate lights and	Section 6.5 – Interactions with other marine users
for Preventing Collisions at		Section 6.2 – Light emissions
		MDO from a vessel
	Act 2012 (Cth) and Marine Orders (see above).	Section 7.7 – Contingency spill response operations
Convention on Oil Pollution Preparedness, Response	This convention comprises national arrangements for responding to oil pollution incidents from ships, offshore oil facilities, sea ports and oil handling. The convention recognises that in the event of a pollution incident, prompt and	Section 7.6 – Surface release of MDO from a vessel
and Co-operation 1990 (OPRC 90)	effective action is essential. Parts of this convention are implemented by the Protection of the Sea (Prevention of Pollution from Ships) Act 1983 (Cth).	Section 7.7 – Contingency spill response operations
	The convention is applicable to the Activity in the event of a worst-case credible spill scenario, which may enact a national arrangement for response. Refer to the <i>Protection of the Sea (Prevention of Pollution from Ships) Act 1983</i> (Cth).	Section 7.5 – Subsea release of dry natural gas
Convention on the Conservation of Migratory	The Bonn Convention aims to improve the status of all threatened migratory species through national action and international agreements between range states of particular groups of species.	Section 3 – Description of the environment
Species of Wild Animals 1979 (Bonn Convention)	This convention is only relevant insofar as a credible spill scenario may result in impact to MNES protected migratory species.	Section 7.6 – Surface release of MDO from a vessel
		Section 7.7 – Contingency spill response operations
		Section 7.5 – Subsea release of dry natural gas

International agreements and conventions	Summary of relevant requirement and how it will be met	EP section
The Paris Agreement on Climate Change 2015	The Paris Agreement aims to tackle climate change and its negative impacts. It sets the long term goal of substantially reducing global GHG emissions to limit global temperature rise this century to well below 2°C above preindustrial levels while pursuing efforts to limit the temperature increase even further to 1.5°C to prevent dangerous human-induced interference with the climate system. Under the Paris Agreement, Australia must submit emissions reduction commitments known as nationally determined contributions. While this Agreement is not directly relevant to the environmental management of the Activity, it provides the international framework and context around Australia's nationally determined contributions and greenhouse gas emissions, which are reflected in domestic law in the Climate Change Act (see above). This helps establish the	Section 6.3 – Atmospheric emissions
	See also the <i>Climate Change Act 2022</i> (Cth) and the Safeguard Mechanism above.	
International Convention for the Control and Management of Ships' Ballast Water and Sediments 2004 (Ballast Water Convention)	The Ballast Water Convention was adopted by the IMO and entered into force globally in 2017. It aims to prevent the spread of harmful aquatic organisms from one region to another, by establishing standards and procedures for managing and controlling ships' ballast water and sediments. Thus, ballast water management systems must be approved in accordance with this convention. From 8 September 2017, all vessels that use ballast water are required to meet the Regulation D-2 discharge standard of this Convention at their next renewal survey. In Australia, Implementation of the convention is also provided for under the <i>Navigation Act 2012</i> (Cth) and the <i>Biosecurity Act 2015</i> (Cth). This convention applies to all foreign vessels operating in Australian waters that have the potential to introduce IMS and/or utilise or conduct ballast water exchange. Refer to Australian Ballast Water Management Requirements.	Section 7.2 – Introduction of invasive marine species Section 8.3.2.9 – Biosecurity management
International Convention on Civil Liability for Bunker Oil Pollution Damage 1969	This convention provides a mechanism for ensuring the payment of compensation for oil pollution damage. In Australia, the convention is enacted under the <i>Protection of the Sea (Civil Liability of Bunker Oil Pollution Damage) Act 2008</i> (Cth). This convention applies in the event of a large-scale spill scenario associated with the Activity. Refer to <i>Protection of the Sea (Civil Liability of Bunker Oil Pollution Damage) Act 2008</i> (Cth).	Section 7.6 – Surface release of MDO from a vessel Section 7.7 – Contingency spill response operations
International Convention on the Control of Harmful Anti-fouling Systems on Ships	This convention prohibits the use of harmful organotin compounds in anti-fouling paints used on ships and establishes a mechanism to prevent the potential future use of other harmful substances in anti-fouling systems. Santos will ensure that the vessels utilised for the Activity maintain anti-fouling systems in accordance with the convention. In addition, Santos will ensure it obtains an approved International Anti-Fouling Systems Certificate. See also <i>Protection of the Sea (Harmful Anti-fouling Systems) Act 2006</i> (Cth).	Section 8.1 Control measures and environmental performance standards

International agreements and conventions	Summary of relevant requirement and how it will be met	EP section
International Convention for the Prevention of Pollution from Ships 1973/1978 (MARPOL 73/78)	This convention and protocol (together known as MARPOL 73/78) build on earlier conventions in the same area. MARPOL is concerned with operational discharges of pollutants from ships. It contains six Annexes, dealing respectively with oil, noxious liquid substances, harmful packaged substances, sewage, garbage and air pollution. Detailed rules are laid out as to the extent to which (if at all) such substances can be released in different sea areas. The legislation giving effect to MARPOL in Australia is the <i>Marine Pollution Act 1999</i> (NT), <i>Protection of the</i> <i>Sea (Prevention of Pollution from Ships) Act 1983</i> (Cth), the <i>Navigation Act 2012</i> (Cth) and several Parts of Marine Orders made under this legislation. This convention applies to vessel discharges and movements associated with the Activity. Santos will ensure that all required audits and inspections of relevant contracted vessels assess compliance with the laws of the international shipping industry, including MARPOL. Refer also to the <i>Marine Pollution Act 1999</i> (NT), the <i>Protection of the Sea (Prevention of Pollution from Ships) Act</i>	Section 2.5 – Vessel and helicopter operations Section 6 – Planned activities impact assessment Section 7 – Unplanned events risk and impact assessment
	1983 (Cth), Navigation Act 2012 (Cth) and Marine Orders.	
International Convention for the Safety of Life at Sea 1974 (SOLAS) and its Protocol of 1988	This convention is generally regarded as the most important of all international treaties concerning the safety of merchant ships. In Australia, the convention is implemented by the <i>Navigation Act 2012</i> (Cth) and Marine Orders under that Act. The convention has been considered in relation to certain safety aspects of the Activity (such as in relation to the management of impacts associated with light emissions). Refer to the <i>Navigation Act 2012</i> (Cth) and Marine Orders.	Section 6 – Planned activities impact assessment Section 8.3.2 – Supporting management processes and procedures
International Convention on Standards of Training, Certification and Watchkeeping (STCW) for Seafarers 1978	This convention prescribes internationally agreed minimum standards relating to training, certification and watchkeeping for seafarers. This convention is given effect in Australia by Marine Order 71 (Masters and Deck Officers). Santos has implemented control measures directed to ensuring compliance with this Convention and with Marine Orders.	Section 8.1.1 – Environmental performance outcomes Section 8.4.6– Marine Assurance
International Convention Relating to Intervention on the High Seas in Cases of Oil Pollution Casualties 1969	Under this convention a coastal state may take action to prevent, mitigate or eliminate danger to its coastline or related interests from pollution by oil or the threat thereof, following upon a maritime casualty. In Australia, this convention is enacted under the <i>Protection of the Sea (Powers of Intervention) Act 1981</i> (Cth) This convention is relevant in the unlikely event of a large-scale spill scenario associated with the Activity where that spill is likely to affect the shoreline of a coastal state. Refer to <i>Protection of the Sea (Powers of Intervention) Act 1981</i> (Cth).	Section 6.4 – Operational Discharges Section 7.6 – Surface release of MDO from a vessel Section 7.7 – Contingency spill response operations
International Maritime Dangerous Goods (IMDG) Code 1994	The IMDG Code was developed as a uniform international code for the transport of dangerous goods by sea covering such matters as packing, marking, labelling and stowage of dangerous goods. Dangerous marine goods that are shipped for the Activity will be stored, handled and transported in line with this code to reduce the risk of an environmental incident and Santos has implemented control measures directed to ensuring compliance with this Code.	Section 7.1 – Release of solid objects

International agreements and conventions	Summary of relevant requirement and how it will be met	EP section
United Nations Convention on Biological Diversity 1992	This convention is the international legal instrument for the conservation of biological diversity, the sustainable use of its components and the fair and equitable sharing of the benefits arising out of the utilization of genetic resources. The overall objectives of the convention are to encourage actions leading to a sustainable future. Australia ratified this convention in June 1993 and the convention came into force in December 1993.Implementation of the measures provided for in the convention is achieved under the EPBC Act (in addition to other plans, policies and programmes at the Commonwealth, State, Territory and Local Government level). The convention is relevant only insofar as the Activity may interact with MNES (threatened and migratory species) that are protected under the EPBC Act.	Section 3.2 – Existing environment Section 6 – Planned activities impact assessment Section 7 – Unplanned event risk assessment.
United Nations Convention on the Law of the Sea (UNCLOS) 1982	<ul> <li>Part XII of the convention sets up a general legal framework for protecting the marine environment. The convention imposes obligations on State Parties to prevent, reduce and control marine pollution from the various major sources, including pollution from land, from the atmosphere, from vessels and from dumping (Articles 207 to 212). Subsequent articles provide a regime for enforcing national marine pollution laws in the many different situations that can arise. UNCLOS also defines maritime zones, including the territorial sea, contiguous zone, exclusive economic zone and the continental shelf.</li> <li>Australia signed the agreement relating to the implementation of Part XI of the Convention in 1982, and UNCLOS in 1994.</li> <li>The convention is relevant to the extent that Santos will comply with MARPOL through the following relevant Marine Orders relating to marine pollution prevention that have been put in place to give effect to relevant regulations provided in Annexes I, II, III, IV, V and VI of MARPOL 73/78:</li> <li>Marine Order 91: Marine pollution prevention – noxious liquid substances</li> <li>Marine Order 95: Marine pollution prevention – packaged harmful substances</li> <li>Marine Order 95: Marine pollution prevention – garbage</li> <li>Marine Order 97: Marine pollution prevention – air pollution.</li> </ul>	Section 6.5 – Interaction with other marine users Section 6.4 – Operational discharges Section 7.6 – Surface release of MDO from a vessel Section 7.7 – Contingency spill response operations
United Nations Educational, Scientific and Cultural Organization Convention on the Protection of Underwater Cultural Heritage 2001	This Convention provides a framework preservation and protection of underwater cultural heritage. This includes traces of human existence of cultural, historical, or archaeological nature that have been submerged for at least 100 years. This Convention is aligned with the sustainable development objectives of the United Nations Agenda 2030. This convention provides the framework to protect and reduce the impact of Barossa production operations on underwater cultural heritage.	Section 3.2.14.7 – Underwater cultural heritage

International agreements and conventions	Summary of relevant requirement and how it will be met	EP section
United Nations Framework Convention on Climate Change 1992	The objective of the convention is to stabilise greenhouse gas concentrations in the atmosphere at a level that would prevent dangerous interference with the climate system. Australia ratified the convention in December 1992 and it came into force on 21 March 1994. The Paris Agreement was agreed under the convention.	Section 6.3 – Atmospheric emissions.
	The convention is relevant to the extent that to reduce impact of GHG emissions associated with the Activity, Santos will comply with MARPOL (Marine Order 97: Marine pollution prevention – air pollution) and will require the use of low sulphur fuel.	



ID	EPBC 2022/09372 Conditions	Conditions applicable to the Activity		
Part A –	Part A – Avoidance, mitigation and compensation conditions			
1	To avoid and mitigate harm to protected matters, the approval holder must not undertake the Action outside the project area.	Authorised activities under this GEP Coastal Waters OEMP are limited to the OA defined in Section 2.2. The OA is within the approved Project Area described in EPBC 2022/09372 thereby limiting activities under this GEP Coastal Waters OEMP to the EPBC approved project area.		
2	To avoid and mitigate impacts on the environment of Commonwealth marine areas and avoid and mitigate harm to protected matters within the project area, the approval holder must:	Refer to Conditions 2a to 2f below.		
2, a)	Ensure that no significant impact to protected matters occurs from potentially harmful substances released into the marine environment during any pre-construction and/or construction activities.	Not applicable to the Activity. DPD pre-construction and construction activities are covered by other Environment Plans, and are outside the scope of the activities authorised under this GEP Coastal Waters OEMP.		
2, b)	Ensures that a Marine Fauna Observer is present at all times during daylight hours during pre-construction and construction operations and continuously monitors and records marine fauna present in the observation zone <sup>52</sup> and is adequately equipped to do so.	Not applicable to the Activity. DPD pre-construction and construction activities are covered by other Environment Plans/ Environmental Management Plans, and are outside the scope of the activities authorised under this GEP Coastal Waters OEMP.		
2, c)	Cease any hydraulic hammering, or use of an Xcentric Ripper tool, or operation of trenching equipment at the direction of the Marine Fauna Observer if marine fauna are sighted within the exclusion zone.	This is not applicable to the Activity. This Environment Plan does not address any hydraulic hammering, or use of an Xcentric Ripper tool, or operation of trenching equipment, as these activities are not undertaken as part of the Activity covered by this GEP Coastal Waters OEMP.		
2, d)	Ensure that, if operations have ceased in accordance with condition 2.c), that use of an Xcentric Ripper tool and/or operation of trenching equipment does not recommence until marine fauna have moved away from the	Not applicable to the Activity. Use of an Xcentric Ripper tool and/or operation of trenching equipment as these activities are not undertaken as part of the Activity covered by this GEP Coastal WatersOEMP.		

#### Table B-4: EPBC Act Approval (EPBC 2022/09372) conditions applicable to the Activity

<sup>&</sup>lt;sup>52</sup> The observation zone defined in the Conditions is in relation to the Marine Megafauna Noise Management Plan which is 150 m. However, this OEMP refers to Part 8 of the EPBC Regulations 2000 which requires cautionary zones of 150 m for dolphins and 300 m for whales and therefore in meeting the Part 8 of the EPBC Regulations 2000 the Conditions will also be met.

ID	EPBC 2022/09372 Conditions	Conditions applicable to the Activity	
	exclusion zone and have not been observed for a minimum of 10 minutes.		
2, e)	Ensure that, if operations have ceased in accordance with condition 2.c), that hydraulic hammering does not recommence until marine fauna have moved away from the exclusion zone and have not been observed for a minimum of 30 minutes.	Not applicable to the Activity. Hydraulic hammering is not undertaken as part of the Activity covered by this GEP Coastal Waters OEMP.	
2, f)	"Initiate a soft start procedure during any initial or subsequent startup activities involving hydraulic hammering, and/or use of an Xcentric Ripper tool, and/or operation of trenching equipment	Not applicable to the Activity. Hydraulic hammering, and/or use of an Xcentric Ripper tool, and/or operation of trenching equipment activities are not undertaken as part of the Activity covered by this GEP Coastal Waters OEMP.	
Action N	Action Management Plans		
3	To avoid and mitigate impacts on any underwater cultural heritage in the environment of Commonwealth marine areas, the approval holder must:	Refer to Conditions 3a to 3c.	
3, a)	Submit a Protocol for Protecting Underwater Cultural Heritage (PPUCH) within the Commonwealth marine area to the department for the Minister's approval which must include:	This condition was met by submitting the PPUCH in May 2024, this protocol was approved by the Minister in writing on 14 June 2024. Additionally, the PPUCH is not relevant to the activities the subject of this GEP Coastal Waters OEMP, given it relates to the management of UCH in the Commonwealth Marine Area.	
3, a), i)	Details of how any underwater cultural heritage present within the Commonwealth marine area will be avoided.	This condition is met by details described in Section 2 of the PPUCH. Additionally, the PPUCH is not relevant to the activities the subject of this GEP Coastal Waters OEMP, given it relates to the management of UCH in the Commonwealth Marine Area.	
3, a), ii)	Detailed impact control and management measures (if required) to ensure no harm to any underwater cultural heritage present within the Commonwealth marine area.	This condition is met by details in Section 3 of the PPUCH. Additionally, the PPUCH is not relevant to the activities the subject of this GEP Coastal Waters OEMP, given it relates to the management of UCH in the Commonwealth Marine Area.	
3, a), iii)	A commitment to engage a suitably qualified underwater archaeologist to advise on any items of potential underwater cultural heritage identified during construction and any related activities impacting the sea floor (if required).	This condition is met by commitments made in Section 4 of the PPUCH. Additionally, the PPUCH is not relevant to the activities the subject of this GEP Coastal Waters OEMP, given it relates to the management of UCH in the Commonwealth Marine Area	
3, a), iv)	Detailed procedures and reporting to be implemented if underwater cultural heritage is	This condition is met by details in Section 5 of the PPUCH. Additionally, the PPUCH is not relevant to the activities the subject of this GEP Coastal Waters OEMP, given it relates to the management of UCH in the Commonwealth Marine Area.	



ID	EPBC 2022/09372 Conditions	Conditions applicable to the Activity
	discovered, including a commitment to notify the department within 21 calendar days of identifying any underwater cultural heritage of clear archaeological character identified by a suitably qualified underwater archaeologist.	
3, a), v)	Details of the process to be followed where any variations are required to be made to the PPUCH, including a requirement for any revised PPUCH to be submitted to the department for the Minister's approval, unless taking the action in accordance with the revised PPUCH would not be likely to have a new or increased impact.	This condition is met by details in Section 6 of the PPUCH. Additionally, the PPUCH is not relevant to the activities the subject of this GEP Coastal Waters OEMP, given it relates to the management of UCH in the Commonwealth Marine Area.
3, b)	Not commence the action unless the Minister has approved the PPUCH in writing.	The Minister approved the PPUCH in writing on 14 June 2024. Additionally, the PPUCH is not relevant to the activities the subject of this OEMP, given it relates to the management of UCH in the Commonwealth Marine Area.
3, c)	Implement the PPUCH prior to the commencement of any activities involving impact to the sea floor.	Not applicable. The PPUCH is not relevant to the activities the subject of this GEP Coastal Waters OEMP, given it relates to the management of UCH in the Commonwealth Marine Area.
4	The approval holder must implement the following Action Management Plans to avoid and mitigate harm as a result of the Action on protected matters. The approval holder must commence implementing each management plan from the commencement of the Action and continue to implement them at least until the completion of the Action.	Refer to Conditions 4a to 4e.
4, a)	Acid Sulfate Soils and Dewatering Management Plan	This condition is not applicable. The Acid Sulfate Soils and Dewatering Management Plan is required for onshore construction activities, which are not undertaken as part of the Activity covered by this GEP Coastal Waters OEMP.
4, b)	Marine Megafauna Noise Management Plan	Not applicable. The Marine Megafauna Noise Management Plan is not applicable to the Activity it is required for noise generated by construction activities within the portion of the EPBC approved Project Area (per EPBC 2022/09372) within the OA, which are not undertaken as part of the Activity covered by this GEP Coastal Waters OEMP.
4, c)	Trenching and Spoil Disposal Management and Monitoring Plan	Not applicable. The Trenching and Spoil Disposal Management and Monitoring Plan is required for Trenching and Spoil Disposal activities, which are not undertaken as part of the Activity covered by this GEP Coastal Waters OEMP.
4, d)	Onshore Construction Environmental Management Plan	Not applicable. The Onshore Construction Environmental Management Plan is required for onshore construction activities, which are not undertaken as part of the Activity covered by this GEP Coastal Waters OEMP.



ID	EPBC 2022/09372 Conditions	Conditions applicable to the Activity	
4, e)	Offshore Construction Environmental Management Plan	Not applicable. The Offshore Construction Environmental Management Plan is required for offshore construction activities, which are not undertaken as part of the Activity covered by this GEP Coastal Waters OEMP.	
5	The approval holder must achieve the following environmental outcomes in implementing the plans required under condition 4):	Refer to Conditions 5a to 5e.	
5, a)	No significant impact to protected matters from intertidal or onshore earthworks relating to the Acid Sulfate Soils and Dewatering Management Plan.	This condition is not applicable. The Acid Sulfate Soils and Dewatering Management Plan is required for onshore construction activities, which are not undertaken as part of the Activity covered by this GEP Coastal Waters OEMP.	
5, b)	The environmental performance objective of no significant impacts to protected marine fauna from noise generated during the DPD construction activities, and performance criteria detailed in table 8-2 of the Marine Megafauna Noise Management Plan.	This is not applicable to the Activity. This Environment Plan does not address pre-construction or construction activities, as these activities are not undertaken as part of the Activity covered by this GEP Coastal Waters OEMP.	
5, c)	All environmental performance objectives and performance criteria detailed in table 8-2; 8-9; 8-13; 8-16; 8-19; 8-21; 8-23; 8-26; 8-29; 8-31; and 8-34 of the Trenching and Spoil Disposal Management and Monitoring Plan.	Not applicable. The Trenching and Spoil Disposal Management and Monitoring Plan is not applicable to the Activity covered by this GEP Coastal Waters OEMP.	
5, d)	All environmental performance objectives and performance criteria detailed in table 7-2 to table 7-18 inclusive, of the Onshore Construction Environmental Management Plan.	Not applicable. The Onshore Construction Environmental Management Plan is not applicable to the Activity covered by this GEP Coastal Waters OEMP.	
5, e)	All environmental performance objectives and performance criteria detailed in table 7-5 to table 7-41 inclusive, of the Offshore Construction Environmental Management Plan.	Not applicable. The Offshore Construction Environmental Management Plan is not applicable to the Activity covered by this GEP Coastal Waters OEMP.	
Part B -	Part B – Administrative Conditions		
Revision	Revision of Action Management Plans		
6	The approval holder may, at any time, apply to the Minister for a variation to an action management plan approved by the Minister, by submitting an application in accordance with the requirements of section 143A of the EPBC Act. If the Minister approves a revised action	Not applicable because the Action management plans do not apply to the Activity covered by this GEP Coastal Waters OEMP.	

ID	EPBC 2022/09372 Conditions	Conditions applicable to the Activity
	management plan (RAMP) then, from the date specified, the approval holder must implement the RAMP in place of the previous action management plan.	
7	The approval holder may choose to revise an action management plan approved by the Minister under conditions 4 and 5, or as subsequently revised in accordance with these conditions, without submitting it for approval under section 143A of the EPBC Act, if the taking of the action in accordance with the RAMP would not be likely to have a new or increased impact.	Not applicable because the Action management plans do not apply to the Activity covered by this GEP Coastal Waters OEMP.
8	If the approval holder makes the choice under condition 7 to revise an action management plan (RAMP) without submitting it for approval, the approval holder must:	Not applicable because the Action management plans do not apply to the Activity covered by this GEP Coastal Waters OEMP.
8, a)	Notify the department electronically that the approved action management plan has been revised and provide the department with:	Not applicable because the Action management plans do not apply to the Activity covered by this GEP Coastal Waters OEMP.
8, a), i)	An electronic copy of the RAMP.	Not applicable because the Action management plans do not apply to the Activity covered by this GEP Coastal Waters OEMP.
8, a), ii)	An electronic copy of the RAMP marked up with track changes to show the differences between the approved action management plan and the RAMP.	Not applicable because the Action management plans do not apply to the Activity covered by this GEP Coastal Waters OEMP.
8, a), iii)	An explanation of the differences between the approved action management plan and the RAMP.	Not applicable because the Action management plans do not apply to the Activity covered by this GEP Coastal Waters OEMP.
8, a), iv)	The reasons the approval holder considers that taking the Action in accordance with the RAMP would not be likely to have a new or increased impact.	Not applicable because the Action management plans do not apply to the Activity covered by this GEP Coastal Waters OEMP.
8, a), v)	Written notice of the date on which the approval holder will implement the RAMP (RAMP implementation date), being at least 20 business days after the date of providing notice	Not applicable because the Action management plans do not apply to the Activity covered by this GEP Coastal Waters OEMP.

ID	EPBC 2022/09372 Conditions	Conditions applicable to the Activity	
	of the revision of the action management plan, or a date agreed to in writing with the department.		
8, b)	Subject to condition 10, implement the RAMP from the RAMP implementation date.	Not applicable because the Action management plans do not apply to the Activity covered by this GEP Coastal Waters OEMP.	
9	The approval holder may revoke its choice to implement a RAMP under condition 7 at any time by giving written notice to the department. If the approval holder revokes the choice under condition 7, the approval holder must implement the action management plan in force immediately prior to the revision undertaken under condition 7.	Not applicable because the Action management plans do not apply to the Activity covered by this GEP Coastal Waters OEMP.	
10	If the Minister notifies the approval holder that the Minister is satisfied that the taking of the Action in accordance with the RAMP would be likely to have a new or increased impact, then:	Not applicable because the Action management plans do not apply to the Activity covered by this GEP Coastal Waters OEMP.	
10, a)	Condition 7 does not apply, or ceases to apply, in relation to the RAMP.	Not applicable because the Action management plans do not apply to the Activity covered by this GEP Coastal Waters OEMP.	
10, b)	The approval holder must implement the action management plan specified by the Minister in the notice.	Not applicable because the Action management plans do not apply to the Activity covered by this GEP Coastal Waters OEMP.	
11	At the time of giving the notice under condition 10, the Minister may also notify that for a specified period of time, condition 7 does not apply for one or more specified action management plans.	Not applicable because the Action management plans do not apply to the Activity covered by this GEP Coastal Waters OEMP.	
Notificat	Notification of Date of Commencement of the Action		
12	The approval holder must notify the department electronically of the date of commencement of the Action, within 5 business days following commencement of the Action.	Santos notified the department via email within 5 business days following commencement of the Action.	
13	The approval holder must not commence the Action later than five (5) years after the date of this approval decision.	As the Action was commenced in 2024 (i.e. within 5 years of the date of the approval decision), this condition has been satisfied.	



ID	EPBC 2022/09372 Conditions	Conditions applicable to the Activity	
Complia	Compliance Records		
14	The approval holder must maintain accurate and complete compliance records.	This condition is demonstrated in: Table 8-13: Activity notification and reporting requirements	
15	If the department makes a request in writing, the approval holder must provide electronic copies of compliance records to the department within the timeframe specified in the request.	This condition is demonstrated in: Table 8-13: Activity notification and reporting requirements	
16	The approval holder must ensure that any monitoring data (including sensitive ecological data), surveys, maps, and other spatial and metadata required under the conditions of this approval are prepared in accordance with the Guidelines for biological survey and mapped data, Commonwealth of Australia 2018, or as otherwise specified by the Minister in writing.	This condition is demonstrated in: Table 8-13: Activity notification and reporting requirements	
17	The approval holder must ensure that any monitoring data (including sensitive ecological data), surveys, maps, and other spatial and metadata required under the conditions of this approval are prepared in accordance with the Guide to providing maps and boundary data for EPBC Act projects, Commonwealth of Australia 2021, or as otherwise specified by the Minister in writing.	This condition is demonstrated in: Table 8-13: Activity notification and reporting requirements	
18	The approval holder must submit all monitoring data (including sensitive ecological data), surveys, maps, other spatial and metadata and all species occurrence record data (sightings and evidence of presence) electronically to the department within 20 business days of each anniversary of the date of this approval decision.	This condition is demonstrated in: Table 8-13: Activity notification and reporting requirements	
Annual C	Annual Compliance Reporting		
19	The approval holder must prepare a compliance report for each 12-month period following the	This condition is demonstrated in: Table 8-13: Activity notification and reporting requirements	



ID	EPBC 2022/09372 Conditions	Conditions applicable to the Activity
	date of this approval decision (or as otherwise agreed to in writing by the Minister).	
20	Each compliance report must be consistent with the Annual Compliance Report Guidelines, Commonwealth of Australia 2023.	This condition is demonstrated in: Table 8-13: Activity notification and reporting requirements
21	Each compliance report must include:	Refer to Conditions 21b to d.
21, b <sup>53</sup> )	Accurate and complete details of compliance and any non-compliance with the conditions and the plans, and any incidents.	This condition is demonstrated in: Table 8-13: Activity notification and reporting requirements
21, c)	One or more shapefile showing all clearing of protected matters, and/or their habitat, undertaken within the 12-month period at the end of which that compliance report is prepared.	This condition is demonstrated in: Table 8-13: Activity notification and reporting requirementsTable 8-13: Activity notification and reporting requirements
21, d)	A schedule of all plans in existence in relation to these conditions and accurate and complete details of how each plan is being implemented.	This condition is demonstrated in: Table 8-13: Activity notification and reporting requirements
22	The approval holder must:	Refer to Conditions 22a to 22f.
22, a)	Publish each compliance report on the website within 60 business days following the end of the 12-month period for which that compliance report is required.	This condition is demonstrated in: Table 8-13: Activity notification and reporting requirements
22, b)	Notify the department electronically, within 5 business days of the date of publication that a compliance report has been published on the website.	This condition is demonstrated in: Table 8-13: Activity notification and reporting requirements
22, c)	Provide the weblink for the compliance report in the notification to the department.	This condition is demonstrated in: Table 8-13: Activity notification and reporting requirements
22, d)	Keep all published compliance reports required by these conditions on the website until the expiry date of this approval.	This condition is demonstrated in: Table 8-13: Activity notification and reporting requirements

<sup>&</sup>lt;sup>53</sup> Note: The numbering convention follows the original sequence as provided in EPBC 2022/09372.



ID	EPBC 2022/09372 Conditions	Conditions applicable to the Activity	
22, e)	Exclude or redact sensitive ecological data from compliance reports published on the website or otherwise provided to a member of the public.	This condition is demonstrated in: Table 8-13: Activity notification and reporting requirements	
22, f)	If sensitive ecological data is excluded or redacted from the published version, submit the full compliance report to the department within 5 business days of its publication on the website and notify the department in writing what exclusions and redactions have been made in the version published on the website.	This condition is demonstrated in: Table 8-13: Activity notification and reporting requirements	
Reportin	Reporting Non-Compliance		
23	The approval holder must notify the department electronically, within 2 business days of becoming aware of any incident and/or potential non-compliance and/or actual non-compliance with the conditions or commitments made in a plan.	This condition is demonstrated in: Table 8-13: Activity notification and reporting requirements	
24	The approval holder must specify in the notification:	Refer to Conditions 24a to 24c.	
24, a)	Any condition or commitment made in a plan which has been or may have been breached.	This condition is demonstrated in: Table 8-13: Activity notification and reporting requirements	
24, b)	A short description of the incident and/or potential non-compliance and/or actual non-compliance.	This condition is demonstrated in: Table 8-13: Activity notification and reporting requirements	
24, c)	The location (including co-ordinates), date and time of the incident and/or potential non-compliance and/or actual non-compliance.	This condition is demonstrated in: Table 8-13: Activity notification and reporting requirements	
25	The approval holder must provide to the department in writing, within 12 business days of becoming aware of any incident and/or potential non-compliance and/or actual non-compliance, the details of that incident and/or potential non-compliance and/or actual non-compliance with the conditions or commitments made in a plan. The approval holder must specify:	This condition is demonstrated in: Table 8-13: Activity notification and reporting requirements Refer to Conditions 25d to 24f.	


ID	EPBC 2022/09372 Conditions	Conditions applicable to the Activity
25, d <sup>54</sup> )	Any corrective action or investigation which the approval holder has already taken.	This condition is demonstrated in: Table 8-13: Activity notification and reporting requirements
25, e)	The potential impacts of the incident and/or non-compliance.	This condition is demonstrated in: Table 8-13: Activity notification and reporting requirements
25, f)	The method and timing of any corrective action that will be undertaken by the approval holder.	This condition is demonstrated in: Table 8-13: Activity notification and reporting requirements
Independ	dent Audit	
26	The approval holder must ensure that an independent audit of compliance with the conditions is conducted at three (3) years after the commencement of the Action, and at any time upon the direction of the Minister.	This condition is demonstrated in: Table 8-13: Activity notification and reporting requirements
27	For each independent audit, the approval holder must:	Refer to Conditions 27a to 27e.
27, a)	Provide the name and qualifications of the nominated independent auditor, the draft audit criteria, and proposed timeframe for submitting the audit report to the department prior to commencing the independent audit.	This condition is demonstrated in: Table 8-13: Activity notification and reporting requirements
27, b)	Only commence the independent audit once the nominated independent auditor, audit criteria and timeframe for submitting the audit report have been approved in writing by the department.	This condition is demonstrated in: Table 8-13: Activity notification and reporting requirements
27, c)	Submit the audit report to the department for approval within the timeframe specified and approved in writing by the department.	This condition is demonstrated in: Table 8-13: Activity notification and reporting requirements
27, d)	Publish the audit report on the website within 15 business days of the date of the department's approval of the audit report.	This condition is demonstrated in: Table 8-13: Activity notification and reporting requirements

<sup>&</sup>lt;sup>54</sup> Note: The numbering convention follows the original sequence as provided in EPBC 2022/09372.



ID	EPBC 2022/09372 Conditions	Conditions applicable to the Activity
27, e)	Keep the audit report published on the website until this approval expires.	This condition is demonstrated in: Table 8-13: Activity notification and reporting requirements
28	Each audit report must report for the period preceding that audit report.	This condition is demonstrated in: Table 8-13: Activity notification and reporting requirements
29	Each audit report must be completed to the satisfaction of the Minister and be consistent with the 'Environment Protection and Biodiversity Conservation Act 1999 Independent Audit and Audit Report Guidelines, Commonwealth of Australia 2019'.	This condition is demonstrated in: Table 8-13: Activity notification and reporting requirements
Complet	ion of the Action	
30	The approval holder must notify the department electronically 60 business days prior to the expiry date of this approval, that the approval is due to expire.	This condition is demonstrated in: Table 8-13: Activity notification and reporting requirements
31	Within 20 business days after the completion of the Action, and, in any event, before this approval expires, the approval holder must notify the department electronically of the date of completion of the Action and provide completion data. The approval holder must submit any spatial data that comprises completion data as a shapefile.	This condition is demonstrated in: Table 8-13: Activity notification and reporting requirements

#### Table B-5: EP Act Approval (EP 2022/022-001) conditions applicable to the Activity

ID	EP 2022/022-001 Conditions	Conditions applicable to the Activity			
1 Lin	1 Limitations and extent of action				
1.1	All activities must be carried out in the approved extent.	Authorised activities under this GEP Coastal Waters OEMP are limited to the OA defined in Section 2.2. The OA is within the approved Project Area described in EP 2022/022-001 thereby limiting activities under this GEP Coastal Waters OEMP to the approved extent.			
1.2	Activities must not exceed the limitations in Table 1. <u>Table 1 Limitations</u> Dredging: No more than 500,000 m <sup>3</sup> of material to be dredged for subsea pipeline trenches in the approved extent. Spoil disposal: Spoil disposal may only occur within the 625 ha spoil area in the approved extent.	Not applicable to the Activity. DPD pre-construction and construction activities are covered by other Management Plans and are outside the scope of the activities to be undertaken under this GEP Coastal Waters OEMP.			
2 Ma	nagement Plans				
2.1	<ul> <li>The approval holder must implement and comply with the following plans:</li> <li>(1) Onshore construction environment management plan Document number BAS-210 0025)</li> <li>(2) Offshore construction environment management plan Document number BAS-210 0024)</li> <li>(3) Marine megafauna noise management plan Document number BAS-210 0045</li> <li>(4) Cultural heritage management plan (CHMP), as required by condition 4; and</li> <li>(5) Trenching management plan (TMP), updated as required by condition 6.</li> </ul>	The only management plan applicable to the Activity covered by this GEP Coastal Waters OEMP is the Cultural heritage management plan (CHMP). A Cultural Heritage Management Plan (CHMP) (BAS-210 0208) has been developed and is being implemented. The other specified management plans are not applicable because they all relate to construction activities and do not apply to the Activity covered by this GEP Coastal Waters OEMP.			
Culture and heritage					
3 En	3 Environmental outcomes				



ID	EP 2022/022-001 Conditions	Conditions applicable to the Activity
3.1	The approval holder must ensure the action achieves the following environmental outcomes:	Specific EPOs for both Planned and Unplanned events are presented in Section 6 and Section 7 respectively.
	<ul><li>(1) protect Aboriginal cultural values; and</li><li>(2) protect maritime heritage, including shipwrecks.</li></ul>	These outcomes will be achieved by implementing the identified control measures to the defined EPSs, and in doing so, will result in compliance with this condition.
4 Cul	tural heritage management plan	
4.1	To support achieving the environmental outcomes required by condition 3-1 a CHMP must be prepared by a suitably qualified and experienced person.	A Cultural Heritage Management Plan (CHMP) (BAS-210 0208) has been developed by a suitably qualified and experienced person on behalf of Santos to protect First Nations and maritime cultural heritage.
4.2	The CHMP must be prepared in consultation with the Northern and Tiwi Land Councils.	The CHMP was developed in consultation with the Northern and Tiwi Land Councils.
4.3	The CHMP must be submitted to the Minister at least 10 days prior to commencement of trenching activity.	Not applicable to the Activity. The use of/or operation of trenching equipment activities is not being undertaking as part of the Activity covered by this GEP Coastal Waters OEMP. The CHMP was submitted to the NT Minister for Environment, Climate Change and Water Security as required.
4.4	<ul> <li>The CHMP must include measures to provide for:</li> <li>(1) cultural heritage induction procedures for site personnel;</li> <li>(2) an internal heritage clearance process prior to trenching and pipe laying activities;</li> <li>(3) further archaeological survey and assessment if warranted by alignment changes;</li> <li>(4) procedures for anchoring and the establishment of anchor exclusion zones;</li> <li>(5) procedures to mitigate risks to unexpected maritime heritage objects, including a stop work protocol, developed in consultation with, and to the satisfaction of, the Heritage Branch of DTFHC;</li> <li>(6) measures for ongoing consultation and engagement on cultural heritage values with stakeholders;</li> <li>(7) the requirement to update the CHMP if stakeholder engagement identifies additional information about cultural heritage values that warrants additional measures to be implemented to achieve the environmental outcomes required by condition 3-1; and</li> <li>(8) detail of how compliance would be monitored and reported and how the outcomes of investigative and/or adaptive management actions would be notified to the relevant government authorities.</li> </ul>	The CHMP (BAS-210 0208) has been submitted to NT EP and published on their website. The CHMP includes the measures required under this condition. Available at: https://environment.nt.gov.au/environment/environmental-assessment-and-approvals/environmental-approvals/darwin-pipeline-duplication

# **Santos**

ID	EP 2022/022-001 Conditions	Conditions applicable to the Activity
5 En	vironmental outcomes	
5.1	5-1 The approval holder must ensure the action achieves the following environmental outcomes:	Specific EPOs for both Planned and Unplanned events are presented in Section 6 and Section 7 respectively.
	<ol> <li>no material environmental harm to the environmental values of water, or declared beneficial uses of Darwin Harbour beyond the zone of influence;</li> <li>no material environmental harm to benthic habitats and communities</li> </ol>	These outcomes will be achieved by implementing the identified control measures to the defined EPSs, and in doing so, will result in compliance with this condition.
	beyond the zone of impact; and	
	(3) risks of physical injury, mortality, behavioural changes and health impacts on marine megafauna are minimised	
6 Tre	enching management	



ID	EP 2022/022-001 Conditions	Conditions applicable to the Activity	
6.1	To support achieving the environmental outcomes required by conditions 5-1(1) and 5-1(2) the TMP must be updated, and submitted to the Minister at least 10 days prior to commencement of trenching activity.	Not applicable to the Activity. The use of/or operation of trenching equipment activities is not being undertaking as part of the Activity covered by this GEP Coastal Waters OEMP.	
6.2	To support achieving the environmental outcomes required by conditions 3-1(1) and 3-1(2), the TMP must be updated to reflect any additional management measures related to trenching that arise as a result of the stakeholder engagement under condition 4-4(7).	Not applicable to the Activity. The use of/or operation of trenching equipment activities is not being undertaking as part of the Activity covered by this GEP Coastal Waters OEMP.	
6.3	6-3 The TMP must include:	Not applicable to the Activity. The use of/or operation of trenching equipment activities is not being	
	(1) a detailed monitoring program that includes:	undertaking as part of the Activity covered by this GEP Coastal Waters OEMP.	
	<ul> <li>(a) parameters to be monitored to detect impacts of trenching activity and spoil disposal, including turbidity (NTU);</li> </ul>		
	<ul><li>(b) location, method and frequency of monitoring, including establishing baseline values of water quality;</li></ul>		
	<ul> <li>(c) quantitative trigger values to indicate when management actions are required;</li> </ul>		
	<ul> <li>(d) monitoring and management actions to be implemented if trigger values are exceeded;</li> </ul>		
	(e) reporting action to be undertaken in the event that trigger values are exceeded.		
6.4	A written review and endorsement from an independent qualified person, stating that the TMP appropriately identifies and appropriately mitigates any environmental risk and complies with the conditions of the approval, must be provided to the Minister with the TMP.	Not applicable to the Activity. The use of/or operation of trenching equipment activities is not being undertaking as part of the Activity covered by this GEP Coastal Waters OEMP.	
7 Hyc	Irotest fluid		
7.1	The approval holder must ensure that there is no planned discharge of hydrotest fluid within NT waters.	Not applicable to the Activity. Hydrotesting is not part of the planned Activity covered by this GEP Coastal Waters OEMP.	
7.2	The approval holder must ensure that any contingency discharge of hydrotest fluid is undertaken in a manner and at a rate such that marine water quality, within a 40 m radius of the discharge location, returns to ambient levels within 12 hours of cessation of discharge.	Hydrotesting is not part of the planned Activity covered by this GEP Coastal Waters OEMP. A contingency discharge of hydrotest water is presented in Section 7.5 as a result of an unplanned rupture of the pipeline. Santos will comply with Condition 7-2.	
8 Mai	3 Maintenance and decommissioning		



ID	EP 2022/022-001 Conditions	Conditions applicable to the Activity	
8.1	Any part of the pipeline or associated infrastructure that will not, or will no longer, be required for use must be decommissioned by the approval holder as soon as practicable.	Section 8.2.5outlines the Santos decommissioning strategy for compliance with this condition.	
8.2	The pipeline and associated infrastructure that is not in operation must be maintained in a condition appropriate for use or decommissioning.	Section 2.6 outlines Inspection, Maintenance, Monitoring and Repair activities during the life of the pipeline and Section 8.2.5outlines the Santos decommissioning strategy for compliance with this condition.	
8.3	The approval holder must ensure that decommissioning of the pipeline and associated infrastructure achieves the environmental outcomes	Not applicable to the Activity. Decommissioning is not part of the planned Activity covered by this GEP Coastal Waters OEMP.	
	identified in condition 5-1.	Section 8.2.5outlines the Santos decommissioning strategy.	
General	conditions		
9 Rev	vision of plans		
9.1	The approval holder may review and revise any management plan required by Condition 2 and must provide the following to the Minister within 10 business days prior to any amendment(s) being implemented; (1) the revised plan(s)	Section 8.5.2 Document management outlines the process that will be followed in the event a management plan must be revised. This condition will be complied with as required for the Activity covered by this GEP Coastal Waters	
	<ul> <li>(1) the revised plan(s)</li> <li>(2) a tabulated summary of the amendment(s) with document references;</li> <li>(3) reasons for the amendment(s);</li> </ul>	Condition 9.1 (5) is not applicable to the activity as the use of/or operation of trenching equipment activities is not being undertaking as part of the Activity covered by this GEP Coastal Waters OEMP, so a TMP is not required.	
	<ul><li>(4) an assessment of environmental risks and potential impacts associated with the amendment(s); and</li></ul>		
	(5) if the TMP is updated, a written review and endorsement from an independent qualified person that the TMP addresses conditions 3- 1 and 5.		
9.2	The approval holder must implement the action to comply with the latest revision of management plans required by condition 2.	This condition will be complied with as required for the Activity covered by this GEP Coastal Waters OEMP.	
10 Cor	10 Commencement of action		



ID	EP 2022/022-001 Conditions	Conditions applicable to the Activity	
10.1	This approval expires five years after the date on which it is granted, unless	Not applicable as the trenching activity referred to in this condition commenced before that date.	
	trenching activity has commenced on or before that date.		
10.2	The approval holder must provide notification in writing to the Minister, at least 5 business days prior to the commencement of trenching activity.	Not applicable to the activity as the use of/or operation of trenching equipment activities is not being undertaking as part of the Activity covered by this GEP Coastal Waters OEMP. Notification of the commencement of trenching was given to the Minister as required.	
11 Change of contact details			
11.1	The approval holder must notify the Minister in writing of any change of its name, physical address or postal address for the serving of notices or other correspondence within 10 business days of such change.	This condition will be complied with as required for the Activity covered by this GEP Coastal Waters OEMP.	
12 Env	12 Environmental performance reporting		



ID	EP 2022/022-001 Conditions	Conditions applicable to the Activity
12.1	<ul> <li>The approval holder must:</li> <li>(1) within six months of the completion of commissioning carried out under this approval, obtain from an independent qualified person, a report on the environmental performance of the action and compliance with the conditions of this environmental approval; and</li> <li>(2) submit the report to the Minister within 30 days of its completion.</li> </ul>	Not applicable to the activity as commissioning is not being undertaking as part of the Activity covered by this GEP Coastal Waters OEMP.
12.2	The report required by condition 12-1(1) must:	Not applicable to the activity as commissioning is not being undertaking as part of the Activity covered
	<ol> <li>provide all monitoring data and reportable incidents required by the conditions of this approval;</li> </ol>	by this GEP Coastal Waters OEMP.
	(2) provide an analysis and interpretation of monitoring data to demonstrate whether compliance with the requirements of conditions has been achieved;	
	(3) include an assessment of the effectiveness of monitoring, management and contingency measures implemented to comply with the requirements of condition 5-1(1) and 5-1(2);	
	(4) include a comparison of the predicted impacts of the action, including trenching activity and spoil disposal, and the actual impacts of the action as verified by environmental monitoring data compared with baseline survey data;	
	(5) be endorsed by the approval holder or a person delegated to sign on the approval holder's behalf;	
	(6) include a statement as to whether the approval holder has complied with the conditions of this approval; and	
	<ul><li>(7) identify all non-compliances and describe corrective and preventative actions taken.</li></ul>	
13 Pro	vision of environmental data	
13.1	All environmental monitoring data required to be collected or obtained under this environmental approval must be retained by the approval holder for a period of not less than 10 years commencing from the date that the data is collected or obtained.	This condition will be complied with as required for the Activity covered by this GEP Coastal Waters OEMP.
13.2	The approval holder must, as and when directed by the Minister, provide any environmental data (including sampling design, sampling methodologies, empirical data and derived information products such as maps) relevant to the assessment of the action and implementation of this environmental approval, to the Minister in the form and manner and at the intervals specified in the direction.	This condition will be complied with as required for the Activity covered by this GEP Coastal Waters OEMP.

# Santos

# **Appendix C**

# **EPBC Act Protected Matters Searches**



Australian Government

**Department of Climate Change, Energy, the Environment and Water** 

# **EPBC** Act Protected Matters Report

This report provides general guidance on matters of national environmental significance and other matters protected by the EPBC Act in the area you have selected. Please see the caveat for interpretation of information provided here.

Report created: 19-Mar-2025

Summary Details Matters of NES Other Matters Protected by the EPBC Act Extra Information Caveat Acknowledgements

# Summary

# Matters of National Environment Significance

This part of the report summarises the matters of national environmental significance that may occur in, or may relate to, the area you nominated. Further information is available in the detail part of the report, which can be accessed by scrolling or following the links below. If you are proposing to undertake an activity that may have a significant impact on one or more matters of national environmental significance then you should consider the Administrative Guidelines on Significance.

World Heritage Properties:	None
National Heritage Places:	None
Wetlands of International Importance (Ramsar	1
Great Barrier Reef Marine Park:	None
Commonwealth Marine Area:	2
Listed Thus stops and Easternian Company with su	NIARA
Listed Inreatened Ecological Communities:	None
Listed Threatened Ecological Communities: Listed Threatened Species:	None 69

# Other Matters Protected by the EPBC Act

This part of the report summarises other matters protected under the Act that may relate to the area you nominated. Approval may be required for a proposed activity that significantly affects the environment on Commonwealth land, when the action is outside the Commonwealth land, or the environment anywhere when the action is taken on Commonwealth land. Approval may also be required for the Commonwealth or Commonwealth agencies proposing to take an action that is likely to have a significant impact on the environment anywhere.

The EPBC Act protects the environment on Commonwealth land, the environment from the actions taken on Commonwealth land, and the environment from actions taken by Commonwealth agencies. As heritage values of a place are part of the 'environment', these aspects of the EPBC Act protect the Commonwealth Heritage values of a Commonwealth Heritage place. Information on the new heritage laws can be found at <a href="https://www.dcceew.gov.au/parks-heritage/heritage">https://www.dcceew.gov.au/parks-heritage/heritage</a>

A <u>permit</u> may be required for activities in or on a Commonwealth area that may affect a member of a listed threatened species or ecological community, a member of a listed migratory species, whales and other cetaceans, or a member of a listed marine species.

Commonwealth Lands:	2
Commonwealth Heritage Places:	None
Listed Marine Species:	118
Whales and Other Cetaceans:	15
Critical Habitats:	None
Commonwealth Reserves Terrestrial:	None
Australian Marine Parks:	3
Habitat Critical to the Survival of Marine Turtles:	3

# Extra Information

This part of the report provides information that may also be relevant to the area you have

State and Territory Reserves:	3
Regional Forest Agreements:	None
Nationally Important Wetlands:	1
EPBC Act Referrals:	36
Key Ecological Features (Marine):	1
Biologically Important Areas:	6
Bioregional Assessments:	None
Geological and Bioregional Assessments:	None

# **Details**

# Matters of National Environmental Significance

Wetlands of International Importance (Ramsar Wetlands)	[Resource Information]
Ramsar Site Name	Proximity
Cobourg peninsula	Within Ramsar site

# **Commonwealth Marine Area**

Approval is required for a proposed activity that is located within the Commonwealth Marine Area which has, will have, or is likely to have a significant impact on the environment. Approval may be required for a proposed action taken outside a Commonwealth Marine Area but which has, may have or is likely to have a significant impact on the environment in the Commonwealth Marine Area.

[Resource Information]

**Feature Name** 

Commonwealth Marine Areas (EPBC Act)

Commonwealth Marine Areas (EPBC Act)

Listed Threatened Species		[Resource Information]	
Status of Conservation Dependent and Extinct are not MNES under the EPBC Act. Number is the current name ID.			
Scientific Name	Threatened Category	Presence Text	
BIRD			
Arenaria interpres			
Ruddy Turnstone [872]	Vulnerable	Foraging, feeding or related behaviour known to occur within area	
Calidris acuminata			
Sharp-tailed Sandpiper [874]	Vulnerable	Foraging, feeding or related behaviour known to occur within area	
Calidris canutus			
Red Knot, Knot [855]	Vulnerable	Species or species habitat known to occur within area	

Calidris ferruginea

Curlew Sandpiper [856]

**Critically Endangered** Species or species habitat known to occur within area

Scientific Name	Threatened Category	Presence Text
Calidris tenuirostris		
Great Knot [862]	Vulnerable	Foraging, feeding or related behaviour known to occur within area
Charadrius leschenaultii Greater Sand Plover, Large Sand Plover [877]	Vulnerable	Species or species habitat known to occur within area
<u>Charadrius mongolus</u> Lesser Sand Plover, Mongolian Plover [879]	Endangered	Foraging, feeding or related behaviour known to occur within area
Chloebia gouldiae listed as Erythrura gou	Ildiae	
Gouldian Finch [90091]	Endangered	Species or species habitat likely to occur within area
Epthianura crocea tunneyi Alligator Rivers Yellow Chat, Yellow Chat (Alligator Rivers) [67089]	Endangered	Species or species habitat may occur within area
Erythrotriorchis radiatus Red Goshawk [942]	Endangered	Species or species habitat known to occur within area
<u>Falco hypoleucos</u> Grey Falcon [929]	Vulnerable	Species or species habitat likely to occur within area
<u>Geophaps smithii smithii</u> Partridge Pigeon (eastern) [64441]	Vulnerable	Species or species habitat known to occur within area
Limnodromus semipalmatus		
Asian Dowitcher [843]	Vulnerable	Species or species

within area

# Limosa lapponica baueri

## Nunivak Bar-tailed Godwit, Western Alaskan Bar-tailed Godwit [86380]

Endangered

Species or species habitat known to occur within area

### Limosa limosa

Black-tailed Godwit [845]

Endangered

Scientific Name	Threatened Category	Presence Text
<u>Melanodryas cucullata melvillensis</u> Tiwi Islands Hooded Robin, Hooded Robin (Tiwi Islands) [67092]	Critically Endangered	Species or species habitat likely to occur within area
<u>Numenius madagascariensis</u> Eastern Curlew, Far Eastern Curlew [847]	Critically Endangered	Species or species habitat known to occur within area
Phaethon rubricauda westralis Red-tailed Tropicbird (Indian Ocean), Indian Ocean Red-tailed Tropicbird [91824]	Endangered	Species or species habitat may occur within area
<u>Pluvialis squatarola</u> Grey Plover [865]	Vulnerable	Foraging, feeding or related behaviour known to occur within area
Rostratula australis Australian Painted Snipe [77037]	Endangered	Species or species habitat may occur within area
Sternula albifrons Little Tern [82849]	Vulnerable	Breeding known to occur within area
<u>Tringa nebularia</u> Common Greenshank, Greenshank [832]	Endangered	Species or species habitat known to occur within area
Tyto novaehollandiae kimberli Masked Owl (northern) [26048]	Vulnerable	Species or species habitat known to occur within area
<u>Tyto novaehollandiae melvillensis</u> Tiwi Masked Owl, Tiwi Islands Masked Owl [26049]	Endangered	Species or species habitat known to occur within area

Xenus cinereus

Terek Sandpiper [59300]

## Vulnerable

Foraging, feeding or related behaviour known to occur within area

## FROG

Uperoleia daviesae

Howard River Toadlet, Davies's Toadlet Vulnerable [85375]

Species or species habitat known to occur within area



Scientific Name	Threatened Category	Presence Text
Antechinus bellus Fawn Antechinus [344]	Vulnerable	Species or species habitat likely to occur within area
Balaenoptera borealis Sei Whale [34]	Vulnerable	Species or species habitat may occur within area
Balaenoptera musculus Blue Whale [36]	Endangered	Species or species habitat likely to occur within area
Balaenoptera physalus Fin Whale [37]	Vulnerable	Species or species habitat may occur within area
Conilurus penicillatus Brush-tailed Rabbit-rat, Brush-tailed Tree-rat, Pakooma [132]	Vulnerable	Species or species habitat known to occur within area
Dasyurus hallucatus Northern Quoll, Digul [Gogo-Yimidir], Wijingadda [Dambimangari], Wiminji [Martu] [331]	Endangered	Species or species habitat likely to occur within area
Macroderma gigas Ghost Bat [174]	Vulnerable	Species or species habitat likely to occur within area
Mesembriomys gouldii gouldii Black-footed Tree-rat (Kimberley and mainland Northern Territory), Djintamoonga, Manbul [87618]	Endangered	Species or species habitat likely to occur within area
Mesembriomys gouldii melvillensis Black-footed Tree-rat (Melville Island) [87619]	Vulnerable	Species or species habitat known to occur within area

Orcaella heinsohni

Australian Snubfin Dolphin [81322]

Vulnerable

Species or species habitat known to occur within area

Petrogale concinna canescens Nabarlek (Top End) [87606]

Endangered

Species or species habitat may occur within area

Scientific Name	Threatened Category	Presence Text
Phascogale pirata	3,	
Northern Brush-tailed Phascogale [82954]	Vulnerable	Species or species habitat likely to occur within area
Saccolaimus saccolaimus nudicluniatus		
Bare-rumped Sheath-tailed Bat, Bare- rumped Sheathtail Bat [66889]	Vulnerable	Species or species habitat likely to occur within area
Sminthopsis butleri		
Butler's Dunnart [302]	Vulnerable	Species or species habitat known to occur within area
Sousa sahulensis		
Australian Humpback Dolphin [87942]	Vulnerable	Breeding known to occur within area
Trichosurus vulpecula arnhemensis		
Northern Brushtail Possum [83091]	Vulnerable	Species or species habitat known to occur within area
Xeromys myoides		
Water Mouse, False Water Rat, Yirrkoo [66]	Vulnerable	Species or species habitat likely to occur within area
PLANT		
Burmannia championii listed as Burmann	ia sp. Bathurst Island (R.F	Fensham 1021)
[93461]	Endangered (listed as Burmannia sp. Bathurst Island	Species or species habitat likely to occur within area
Elaeocarpus miegei		
[65147]	Endangered	Species or species habitat may occur within area
Hoya australis subsp. oramicola		
a vine [55436]	Vulnerable	Species or species habitat known to occur within area

Stylidium ensatum a triggerplant [86366]

Endangered

Species or species habitat known to occur within area

# Tarennoidea wallichii [65173]

Endangered

Scientific Name	Threatened Category	Presence Text
<u>Typhonium jonesii</u>		
a herb [62412]	Endangered	Species or species habitat likely to occur within area
Typhonium mirabile		
a herb [79227]	Endangered	Species or species habitat likely to occur within area
Xylopia monosperma		
a shrub [82030]	Endangered	Species or species habitat likely to occur within area
REPTILE		
Acanthophis hawkei		
Plains Death Adder [83821]	Vulnerable	Species or species habitat likely to occur within area
Caretta caretta		
Loggerhead Turtle [1763]	Endangered	Foraging, feeding or related behaviour known to occur within area
Chelonia mydas		
Green Turtle [1765]	Vulnerable	Breeding known to occur within area
Dermochelvs coriacea		
Leatherback Turtle, Leathery Turtle, Luth [1768]	Endangered	Species or species habitat known to occur within area
Eretmochelvs imbricata		
Hawksbill Turtle [1766]	Vulnerable	Foraging, feeding or related behaviour known to occur within area
Lepidochelys olivacea		
Olive Ridley Turtle, Pacific Ridley Turtle	Endangered	Breeding known to

# Natator depressus

Flatback Turtle [59257]

Vulnerable

Breeding known to occur within area

Tiliqua scincoides intermedia

Northern Blue-tongued Skink [89838]

Critically Endangered Species or species habitat likely to occur within area

Scientific Name	Threatened Category	Presence Text
Varanus mertensi		
Mertens' Water Monitor, Mertens's Water Monitor [1568]	Endangered	Species or species habitat known to occur within area
Varanus mitchelli		
Mitchell's Water Monitor [1569]	Critically Endangered	Species or species habitat likely to occur within area
SHARK		
Carcharodon carcharias		
White Shark, Great White Shark [64470]	Vulnerable	Species or species habitat may occur within area
Glyphis garricki		
Northern River Shark, New Guinea River Shark [82454]	Endangered	Breeding known to occur within area
Glyphis glyphis		
Speartooth Shark [82453]	Critically Endangered	Species or species habitat known to occur within area
Pristis clavata		
Dwarf Sawfish, Queensland Sawfish [68447]	Vulnerable	Species or species habitat known to occur within area
Pristis pristis		
Freshwater Sawfish, Largetooth Sawfish, River Sawfish, Leichhardt's Sawfish, Northern Sawfish [60756]	Vulnerable	Species or species habitat likely to occur within area
Pristis zijsron		
Green Sawfish, Dindagubba, Narrowsnout Sawfish [68442]	Vulnerable	Species or species habitat known to occur within area
Rhincodon typus		
Whale Shark [66680]	Vulnerable	Species or species habitat may occur within area

<u>Sphyrna lewini</u> Scalloped Hammerhead [85267]

Conservation Dependent

Listed Migratory Species			[Resource Information]
Scientific Name	Threatened Category	Presence Text	
Migratory Marine Birds			

Scientific Name <u>Anous stolidus</u> Common Noddy [825]

Apus pacificus Fork-tailed Swift [678]

Calonectris leucomelas Streaked Shearwater [1077]

<u>Fregata ariel</u> Lesser Frigatebird, Least Frigatebird [1012]

Fregata minor Great Frigatebird, Greater Frigatebird [1013]

Phaethon lepturus White-tailed Tropicbird [1014]

Sternula albifrons Little Tern [82849]

Vulnerable

Threatened Category Presence Text

Species or species habitat likely to occur within area

Species or species habitat likely to occur within area

Species or species habitat known to occur within area

Species or species habitat known to occur within area

Species or species habitat known to occur within area

Species or species habitat may occur within area

Breeding known to occur within area

Species or species habitat known to

occur within area

Migratory Marine Species Anoxypristis cuspidata Narrow Sawfish, Knifetooth Sawfish [68448]

Balaenoptera borealis Sei Whale [34]

Vulnerable

Species or species habitat may occur within area

Balaenoptera edeni Bryde's Whale [35]

Species or species habitat may occur within area

Balaenoptera musculus

Blue Whale [36]

Endangered

Scientific Name	Threatened Category	Presence Text
Balaenoptera physalus		
Fin Whale [37]	Vulnerable	Species or species habitat may occur within area
Carcharhinus longimanus		
Oceanic Whitetip Shark [84108]		Species or species habitat may occur within area
Carcharias taurus		
Grey Nurse Shark [64469]		Species or species habitat may occur within area
Carcharodon carcharias		
White Shark, Great White Shark [64470]	Vulnerable	Species or species habitat may occur within area
Caretta caretta		
Loggerhead Turtle [1763]	Endangered	Foraging, feeding or related behaviour known to occur within area
Chelonia mydas		
Green Turtle [1765]	Vulnerable	Breeding known to occur within area
Crocodylus porosus		
Salt-water Crocodile, Estuarine Crocodile [1774]		Species or species habitat likely to occur within area
Dermochelys coriacea		
Leatherback Turtle, Leathery Turtle, Luth [1768]	Endangered	Species or species habitat known to occur within area
Duaona duaon		
Dugong [28]		Species or species habitat known to occur within area

Eretmochelys imbricata Hawksbill Turtle [1766]

Vulnerable

Foraging, feeding or related behaviour known to occur within area

Isurus oxyrinchus Shortfin Mako, Mako Shark [79073]

Isurus paucus Longfin Mako [82947]

Lepidochelys olivacea Olive Ridley Turtle, Pacific Ridley Turtle Endangered [1767]

Megaptera novaeangliae Humpback Whale [38]

# Mobula alfredi as Manta alfredi

Reef Manta Ray, Coastal Manta Ray [90033]

# Mobula birostris as Manta birostris

Giant Manta Ray [90034]

Natator depressus Flatback Turtle [59257]

Vulnerable

Vulnerable

Orcaella heinsohni

Australian Snubfin Dolphin [81322]

Orcinus orca Killer Whale, Orca [46]

Pristis clavata

Dwarf Sawfish, Queensland Sawfish [68447]

Pristis pristis

Freshwater Sawfish, Largetooth Sawfish, River Sawfish, Leichhardt's Sawfish, Northern Sawfish [60756]

Vulnerable

Vulnerable

Threatened Category **Presence Text** 

> Species or species habitat likely to occur within area

Breeding known to occur within area

Species or species habitat likely to occur within area

Species or species habitat likely to occur within area

Species or species habitat likely to occur within area

Breeding known to occur within area

Species or species habitat known to occur within area

Species or species habitat may occur within area

Species or species habitat known to occur within area

Species or species habitat likely to occur within area

### Pristis zijsron

Green Sawfish, Dindagubba, Narrowsnout Sawfish [68442]

Species or species habitat known to occur within area

Rhincodon typus Whale Shark [66680]

Vulnerable

Vulnerable

Species or species habitat may occur within area

Scientific Name	Threatened Category	Presence Text
<u>Sousa sahulensis as Sousa chinensis</u>		
Australian Humpback Dolphin [87942]	Vulnerable	Breeding known to occur within area
Tursiops aduncus (Arafura/Timor Sea po	pulations)	
Spotted Bottlenose Dolphin (Arafura/Timor Sea populations) [78900]		Species or species habitat known to occur within area
Migratory Terrestrial Species		
Cecropis daurica		
Red-rumped Swallow [80610]		Species or species habitat may occur within area
Cuculus optatus		
Oriental Cuckoo, Horsfield's Cuckoo [86651]		Species or species habitat known to occur within area
Hirundo rustica		
Barn Swallow [662]		Species or species habitat likely to occur within area
Motacilla cinerea		
Grey Wagtail [642]		Species or species habitat may occur within area

Motacilla flava Yellow Wagtail [644]

Species or species habitat likely to occur within area

Migratory Wetlands Species

Acrocephalus orientalis Oriental Reed-Warbler [59570]

Actitis hypoleucos Common Sandpiper [59309]

Species or species habitat known to occur within area

Species or species habitat may occur

within area

Arenaria interpres Ruddy Turnstone [872]

Vulnerable

Foraging, feeding or related behaviour known to occur within area

Calidris acuminata Sharp-tailed Sandpiper [874]

Vulnerable

Scientific Name	Threatened Category	Presence Text
Calidris alba		
Sanderling [875]		Foraging, feeding or related behaviour known to occur within area
Calidris canutus		
Red Knot, Knot [855]	Vulnerable	Species or species habitat known to occur within area
Calidris ferruginea		
Curlew Sandpiper [856]	Critically Endangered	Species or species habitat known to occur within area
Calidris melanotos		
Pectoral Sandpiper [858]		Species or species habitat may occur within area
Calidris ruficollis		
Red-necked Stint [860]		Foraging, feeding or related behaviour known to occur within area
Calidris subminuta		
Long-toed Stint [861]		Foraging, feeding or related behaviour known to occur within area
Calidris tenuirostris		
Great Knot [862]	Vulnerable	Foraging, feeding or related behaviour known to occur within area
Charadrius dubius		
Little Ringed Plover [896]		Foraging, feeding or related behaviour known to occur within

# Charadrius leschenaultii

# Greater Sand Plover, Large Sand Plover Vulnerable [877]

Species or species habitat known to occur within area

area

Charadrius mongolus

# Lesser Sand Plover, Mongolian Plover Endangered [879]

Charadrius veredus

Oriental Plover, Oriental Dotterel [882]

Gallinago megala Swinhoe's Snipe [864]

Gallinago stenura Pin-tailed Snipe [841]

<u>Glareola maldivarum</u> Oriental Pratincole [840]

Limicola falcinellus Broad-billed Sandpiper [842]

Limnodromus semipalmatus Asian Dowitcher [843]

Vulnerable

Limosa lapponica Bar-tailed Godwit [844]

Limosa limosa Black-tailed Godwit [845]

Endangered

Presence Text

Threatened Category

Foraging, feeding or related behaviour known to occur within area

Foraging, feeding or related behaviour known to occur within area

Foraging, feeding or related behaviour likely to occur within area

Foraging, feeding or related behaviour known to occur within area

Foraging, feeding or related behaviour known to occur within area

Species or species habitat likely to occur within area

Species or species habitat known to occur within area

Foraging, feeding or related behaviour known to occur within area

Numenius madagascariensis

Eastern Curlew, Far Eastern Curlew [847]

Critically Endangered

Species or species habitat known to occur within area

Numenius minutus

Little Curlew, Little Whimbrel [848]

Scientific Name	Threatened Category	Presence Text
Numenius phaeopus		
Whimbrel [849]		Foraging, feeding or related behaviour known to occur within area
Pandion haliaetus		
Osprey [952]		Species or species habitat known to occur within area
<u>Pluvialis fulva</u>		
Pacific Golden Plover [25545]		Foraging, feeding or related behaviour known to occur within area
Pluvialis squatarola		
Grey Plover [865]	Vulnerable	Foraging, feeding or related behaviour known to occur within area
Tringa brevines		
Grey-tailed Tattler [851]		Foraging, feeding or related behaviour known to occur within area
Tringa glareola		
Wood Sandpiper [829]		Foraging, feeding or related behaviour known to occur within area
Tringa incana		
Wandering Tattler [831]		Foraging, feeding or related behaviour known to occur within area
Tringa nebularia		
Common Greenshank, Greenshank [832]	Endangered	Species or species habitat known to occur within area

Tringa stagnatilis

# Marsh Sandpiper, Little Greenshank [833]

Foraging, feeding or related behaviour known to occur within area

Xenus cinereus

Terek Sandpiper [59300]

Vulnerable

# Other Matters Protected by the EPBC Act

Commonwealth Lands		[Resource ]	nformation
The Commonwealth area listed below the unreliability of the data source, all Commonwealth area, before making a department for further information.	may indicate the presence of proposals should be checked definitive decision. Contact	of Commonwealth land in this vided as to whether it impacts on a the State or Territory governme	inity. Due to
Commonwealth Land Name		State	
Defence			
Defence - QUAIL ISLAND BOMBING	RANGE [70003]	NT	
Unknown			
Commonwealth Land - [70995]		NT	
Listed Marine Species		[Resource ]	nformation ]
Scientific Name	Threatened Category	Presence Text	
Bird			
Acrocephalus orientalis			
Oriental Reed-Warbler [59570]		Species or species habitat may occur within area overfly marine area	
<u>Actitis hypoleucos</u>			
Common Sandpiper [59309]		Species or species habitat known to occur within area	
<u>Anous stolidus</u>			
Common Noddy [825]		Species or species habitat likely to occur within area	
Anseranas semipalmata			
Magpie Goose [978]		Species or species habitat may occur within area overfly marine area	
<u>Apus pacificus</u>			
Fork-tailed Swift [678]		Species or species habitat likely to occur	

marine area

within area overfly

<u>Arenaria interpres</u> Ruddy Turnstone [872]

Vulnerable

Scientific Name	Threatened Category	Presence Text
Bubulcus ibis as Ardea ibis		
Cattle Egret [66521]		Species or species habitat may occur within area overfly marine area
Calidris acuminata		
Sharp-tailed Sandpiper [874]	Vulnerable	Foraging, feeding or related behaviour known to occur within area
Calidris alba		
Sanderling [875]		Foraging, feeding or related behaviour known to occur within area
Calidris canutus		
Red Knot, Knot [855]	Vulnerable	Species or species habitat known to occur within area overfly marine area
Calidris ferruginea		
Curlew Sandpiper [856]	Critically Endangered	Species or species habitat known to occur within area overfly marine area
Calidris melanotos		
Pectoral Sandpiper [858]		Species or species habitat may occur within area overfly marine area
Calidris ruficollis		
Red-necked Stint [860]		Foraging, feeding or related behaviour known to occur within area overfly marine area
Calidris subminuta		

Long-toed Stint [861]

Calidris tenuirostris Great Knot [862]

Vulnerable

Foraging, feeding or related behaviour known to occur within area overfly marine area

Foraging, feeding or related behaviour known to occur within area overfly marine area

Calonectris leucomelas Streaked Shearwater [1077]

<u>Cecropis daurica as Hirundo daurica</u> Red-rumped Swallow [80610]

<u>Chalcites osculans as Chrysococcyx osculans</u> Black-eared Cuckoo [83425]

<u>Charadrius dubius</u> Little Ringed Plover [896]

<u>Charadrius leschenaultii</u>

Greater Sand Plover, Large Sand Plover Vulnerable [877]

Charadrius mongolus Lesser Sand Plover, Mongolian Plover Endangered [879]

Charadrius ruficapillus Red-capped Plover [881]

### <u>Charadrius veredus</u> Oriental Plover, Oriental Dotterel [882]

Threatened Category F

Presence Text

Species or species habitat known to occur within area

Species or species habitat may occur within area overfly marine area

Species or species habitat likely to occur within area overfly marine area

Foraging, feeding or related behaviour known to occur within area overfly marine area

Species or species habitat known to occur within area

Foraging, feeding or related behaviour known to occur within area

Foraging, feeding or related behaviour known to occur within area overfly marine area

Foraging, feeding or related behaviour

Fregata ariel Lesser Frigatebird, Least Frigatebird [1012] known to occur within area overfly marine area

Species or species habitat known to occur within area

Fregata minor Great Frigatebird, Greater Frigatebird [1013]

Gallinago megala Swinhoe's Snipe [864]

Gallinago stenura Pin-tailed Snipe [841]

<u>Glareola maldivarum</u> Oriental Pratincole [840]

Haliaeetus leucogaster White-bellied Sea-Eagle [943]

<u>Himantopus himantopus</u> Pied Stilt, Black-winged Stilt [870]

<u>Hirundo rustica</u> Barn Swallow [662] Threatened Category Pres

**Presence Text** 

Species or species habitat known to occur within area

Foraging, feeding or related behaviour known to occur within area overfly marine area

Foraging, feeding or related behaviour likely to occur within area overfly marine area

Foraging, feeding or related behaviour known to occur within area overfly marine area

Species or species habitat known to occur within area

Foraging, feeding or related behaviour known to occur within area overfly marine area

Species or species habitat likely to occur within area overfly marine area

Limicola falcinellus

### Broad-billed Sandpiper [842]

Foraging, feeding or related behaviour known to occur within area overfly marine area

# Limnodromus semipalmatus Asian Dowitcher [843]

Vulnerable

Species or species habitat likely to occur within area overfly marine area

Scientific Name	Threatened Category	Presence Text
Limosa lapponica		
Bar-tailed Godwit [844]		Species or species habitat known to occur within area
Limosa limosa		
Black-tailed Godwit [845]	Endangered	Foraging, feeding or related behaviour known to occur within area overfly marine area
Merops ornatus		
Rainbow Bee-eater [670]		Species or species habitat may occur within area overfly marine area
Motacilla cinerea		
Grey Wagtail [642]		Species or species habitat may occur within area overfly marine area
Motacilla flava		
Yellow Wagtail [644]		Species or species habitat likely to occur within area overfly marine area
Numenius madagascariensis		
Eastern Curlew, Far Eastern Curlew [847]	Critically Endangered	Species or species habitat known to occur within area
Numenius minutus		
Little Curlew, Little Whimbrel [848]		Foraging, feeding or related behaviour known to occur within area overfly marine area
Numenius phaeopus		
Whimbrel [849]		Foraging, feeding or

Pandion haliaetus

Osprey [952]

Phaethon lepturus White-tailed Tropicbird [1014] known to occur within area

related behaviour

Species or species habitat known to occur within area

Species or species habitat may occur within area

Scientific Name	Threatened Category	Presence Text
Pluvialis fulva		
Pacific Golden Plover [25545]		Foraging, feeding or related behaviour known to occur within area
Pluvialis squatarola		
Grey Plover [865]	Vulnerable	Foraging, feeding or related behaviour known to occur within area overfly marine area
Rhipidura rufifrons		
Rufous Fantail [592]		Species or species habitat known to occur within area overfly marine area
Rostratula australis as Rostratula bendha	alensis (sensu lato)	
Australian Painted Snipe [77037]	Endangered	Species or species habitat may occur within area overfly marine area
Sternula albifrons as Sterna albifrons		
Little Tern [82849]	Vulnerable	Breeding known to occur within area
Stiltia isabella		
Australian Pratincole [818]		Foraging, feeding or related behaviour known to occur within area overfly marine area
Tringa brevipes as Heteroscelus brevipe	S	
Grey-tailed Tattler [851]		Foraging, feeding or related behaviour known to occur within area
Tringa glareola		
Wood Sandpiper [829]		Foraging, feeding or related behaviour

Tringa incana as Heteroscelus incanus Wandering Tattler [831] known to occur within area overfly marine area

Scientific Name	Threatened Category	Presence Text
<u>Tringa nebularia</u> Common Greenshank, Greenshank [832]	Endangered	Species or species habitat known to occur within area overfly marine area
<u>Tringa stagnatilis</u> Marsh Sandpiper, Little Greenshank [833]		Foraging, feeding or related behaviour known to occur within area overfly marine area
Xenus cinereus		
Terek Sandpiper [59300]	Vulnerable	Foraging, feeding or related behaviour known to occur within area overfly marine area
Fish		
Bhanotia fasciolata		
Bhanotia fasciolata Corrugated Pipefish, Barbed Pipefish [66188]		Species or species habitat may occur within area
Bhanotia fasciolata Corrugated Pipefish, Barbed Pipefish [66188]		Species or species habitat may occur within area
<ul> <li><u>Bhanotia fasciolata</u></li> <li>Corrugated Pipefish, Barbed Pipefish [66188]</li> <li><u>Campichthys tricarinatus</u></li> <li>Three-keel Pipefish [66192]</li> </ul>		Species or species habitat may occur within area Species or species habitat may occur within area
<ul> <li>Bhanotia fasciolata</li> <li>Corrugated Pipefish, Barbed Pipefish [66188]</li> <li>Campichthys tricarinatus</li> <li>Three-keel Pipefish [66192]</li> <li>Choeroichthys brachysoma</li> <li>Pacific Short-bodied Pipefish, Short-bodied Pipefish [66194]</li> </ul>		Species or species habitat may occur within area
<ul> <li>Bhanotia fasciolata</li> <li>Corrugated Pipefish, Barbed Pipefish [66188]</li> <li>Campichthys tricarinatus</li> <li>Three-keel Pipefish [66192]</li> <li>Choeroichthys brachysoma</li> <li>Pacific Short-bodied Pipefish, Short-bodied Pipefish [66194]</li> <li>Choeroichthys suillus</li> </ul>		<text></text>
<ul> <li>Bhanotia fasciolata</li> <li>Corrugated Pipefish, Barbed Pipefish [6188]</li> <li>Campichthys tricarinatus</li> <li>Three-keel Pipefish [66192]</li> <li>Choeroichthys brachysoma</li> <li>Pacific Short-bodied Pipefish, Short-bodied Pipefish [66194]</li> <li>Choeroichthys suillus</li> <li>Pig-snouted Pipefish [66198]</li> </ul>		<ul> <li>Species or species habitat may occur within area</li> </ul>

Fijian Banded Pipefish, Brown-banded Pipefish [66199]

Species or species habitat may occur within area

Corythoichthys flavofasciatus

Reticulate Pipefish, Yellow-banded Pipefish, Network Pipefish [66200]

Corythoichthys haematopterus Reef-top Pipefish [66201] Species or species habitat may occur within area

Species or species habitat may occur within area

<u>Corythoichthys intestinalis</u> Australian Messmate Pipefish, Banded Pipefish [66202]

<u>Corythoichthys schultzi</u> Schultz's Pipefish [66205]

Cosmocampus banneri Roughridge Pipefish [66206]

### Doryrhamphus dactyliophorus

Banded Pipefish, Ringed Pipefish [66210]

### Doryrhamphus excisus

Bluestripe Pipefish, Indian Blue-stripe Pipefish, Pacific Blue-stripe Pipefish [66211]

Doryrhamphus janssi Cleaner Pipefish, Janss' Pipefish [66212]

Festucalex cinctus Girdled Pipefish [66214]

<u>Filicampus tigris</u> Tiger Pipefish [66217]

Halicampus brocki Brock's Pipefish [66219] Threatened Category

Presence Text

Species or species habitat may occur within area

Halicampus dunckeri

Red-hair Pipefish, Duncker's Pipefish [66220]

<u>Halicampus grayi</u> Mud Pipefish, Gray's Pipefish [66221] Species or species habitat may occur within area

Species or species habitat may occur within area

Halicampus spinirostris Spiny-snout Pipefish [66225]

### Haliichthys taeniophorus

Ribboned Pipehorse, Ribboned Seadragon [66226]

## Hippichthys cyanospilos

Blue-speckled Pipefish, Blue-spotted Pipefish [66228]

## Hippichthys parvicarinatus

Short-keel Pipefish, Short-keeled Pipefish [66230]

<u>Hippichthys penicillus</u> Beady Pipefish, Steep-nosed Pipefish [66231]

<u>Hippocampus histrix</u> Spiny Seahorse, Thorny Seahorse [66236]

<u>Hippocampus kuda</u> Spotted Seahorse, Yellow Seahorse [66237]

<u>Hippocampus planifrons</u> Flat-face Seahorse [66238]

<u>Hippocampus spinosissimus</u> Hedgehog Seahorse [66239] Threatened Category

Presence Text

Species or species habitat may occur within area

Micrognathus micronotopterus Tidepool Pipefish [66255]

Solegnathus hardwickii

# Pallid Pipehorse, Hardwick's Pipehorse [66272]

Species or species habitat may occur within area

Species or species habitat may occur within area

Solegnathus lettiensis Gunther's Pipehorse, Indonesian Pipefish [66273]

#### Solenostomus cyanopterus

Robust Ghostpipefish, Blue-finned Ghost Pipefish, [66183]

#### Syngnathoides biaculeatus

Double-end Pipehorse, Double-ended Pipehorse, Alligator Pipefish [66279]

### Trachyrhamphus bicoarctatus

Bentstick Pipefish, Bend Stick Pipefish, Short-tailed Pipefish [66280]

## Trachyrhamphus longirostris

Straightstick Pipefish, Long-nosed Pipefish, Straight Stick Pipefish [66281]

#### Mammal

Dugong dugon Dugong [28]

### Reptile

### Aipysurus duboisii

Dubois' Sea Snake, Dubois' Seasnake, Reef Shallows Sea Snake [1116]

### <u>Aipysurus laevis</u>

Olive Sea Snake, Olive-brown Sea Snake [1120]

# <u>Aipysurus mosaicus as Aipysurus eydouxii</u> Mosaic Sea Snake [87261]

Threatened Category

Presence Text

Species or species habitat may occur within area

Species or species habitat known to occur within area

Species or species habitat may occur within area

Species or species habitat may occur within area

Species or species habitat may occur within area

## Caretta caretta Loggerhead Turtle [1763]

Endangered

Foraging, feeding or related behaviour known to occur within area

Chelonia mydas Green Turtle [1765]

Vulnerable

Breeding known to occur within area
Scientific Name

<u>Crocodylus johnstoni</u> Freshwater Crocodile, Johnston's Crocodile, Johnstone's Crocodile [1773]

<u>Crocodylus porosus</u> Salt-water Crocodile, Estuarine Crocodile [1774]

# **Dermochelys coriacea**

Leatherback Turtle, Leathery Turtle, Luth Endangered [1768]

# Emydocephalus annulatus

Eastern Turtle-headed Sea Snake [1125]

Eretmochelys imbricata Hawksbill Turtle [1766]

Vulnerable

<u>Hydrelaps darwiniensis</u> Port Darwin Sea Snake, Black-ringed Mangrove Sea Snake [1100]

<u>Hydrophis atriceps</u> Black-headed Sea Snake, Banded Sea Snake [1101]

<u>Hydrophis coggeri</u> Cogger's Sea Snake, Black-headed Sea Snake [25925]

<u>Hydrophis elegans</u> Elegant Sea Snake, Bar-bellied Sea Snake [1104] Threatened Category Presence Text

Species or species habitat may occur within area

Species or species habitat likely to occur within area

Species or species habitat known to occur within area

Species or species habitat may occur within area

Foraging, feeding or related behaviour known to occur within area

Species or species habitat may occur within area

<u>Hydrophis hardwickii as Lapemis hardwickii</u> Spine-bellied Sea Snake [93516]

<u>Hydrophis inornatus</u> Plain Sea Snake [1107] Species or species habitat may occur within area

Species or species habitat may occur within area

# Scientific Name

e Ir iii as Disteira kingii

Threatened Category

<u>Hydrophis kingii as Disteira kingii</u> Spectacled Sea Snake [93511]

# Hydrophis macdowelli as Hydrophis mcdowelli

MacDowell's Sea Snake, Small-headed Sea Snake, [75601]

# Hydrophis major as Disteira major

Olive-headed Sea Snake [93512]

## Hydrophis ornatus Spotted Sea Snake, Ornate

Spotted Sea Snake, Ornate Reef Sea Snake [1111]

### Hydrophis pacificus Pacific Sea Snake Large-

Pacific Sea Snake, Large-headed Sea Snake [1112]

# <u>Hydrophis peronii as Acalyptophis peronii</u> Horned Sea Snake [93509]

Hydrophis platura as Pelamis platurus Yellow-bellied Sea Snake [93746]

# <u>Hydrophis stokesii as Astrotia stokesii</u> Stokes' Sea Snake [93510]

<u>Hydrophis zweifeli as Enhydrina schistosa</u> Australian Beaked Sea Snake [93514] Species or species habitat may occur within area

Lepidochelys olivacea

Olive Ridley Turtle, Pacific Ridley Turtle Endangered [1767]

Breeding known to occur within area

Natator depressus Flatback Turtle [59257]

Vulnerable

Breeding known to occur within area

Parahydrophis mertoni

Arafura Smooth Sea Snake, Northern Mangrove Sea Snake [1090] Species or species habitat may occur within area

Whales and Other Cetaceans		[Resource Information
Current Scientific Name	Status	Type of Presence
Mammal		
Balaenoptera borealis		
Sei Whale [34]	Vulnerable	Species or species
		habitat may occur
		within area
Balaenontera edeni		
Bryde's Whale [35]		Species or species
		habitat may occur
		within area
Balaenoptera musculus		
Blue whale [36]	Endangered	Species or species babitat likely to occur
		within area
Balaenoptera physalus		
Fin Whale [37]	Vulnerable	Species or species
		habitat may occur
		within area
Delphinus delphis		
Common Dolphin, Short-beaked		Species or species
Common Dolphin [60]		habitat may occur
		within area
Crompus grissus		
<u>Grampus griseus</u> Risso's Dolphin, Grampus [64]		Species or species
Nisso's Dolphin, Orampus [04]		habitat may occur
		within area
Megaptera novaeangliae		
Нитроаск vvnale [38]		Species or species babitat likely to occur
		within area
Orcaella heinsohni		
Australian Snubfin Dolphin [81322]	Vulnerable	Species or species
		habitat known to
		occur within area

<u>Orcinus orca</u> Killer Whale, Orca [46]

Species or species

habitat may occur within area

Pseudorca crassidens False Killer Whale [48]

Species or species habitat likely to occur within area

Sousa sahulensis

Australian Humpback Dolphin [87942]

Vulnerable

Breeding known to occur within area

Current Scientific Name

Stenella attenuata Spotted Dolphin, Pantropical Spotted Dolphin [51]

<u>Tursiops aduncus</u> Indian Ocean Bottlenose Dolphin, Spotted Bottlenose Dolphin [68418]

Tursiops aduncus (Arafura/Timor Sea populations) Spotted Bottlenose Dolphin (Arafura/Timor Sea populations) [78900]

<u>Tursiops truncatus s. str.</u> Bottlenose Dolphin [68417] Status

Type of Presence

Species or species habitat may occur within area

Species or species habitat likely to occur within area

Species or species habitat known to occur within area

Species or species habitat may occur within area

Australian Marine Parks	[Resource Information]
Park Name	Zone & IUCN Categories
Oceanic Shoals	Habitat Protection Zone (IUCN IV)
Joseph Bonaparte Gulf	Special Purpose Zone (IUCN VI)
Oceanic Shoals	Special Purpose Zone (Trawl) (IUCN VI)

Habitat Critical to the Survival of Marine Turtles		[Resource Information]
Scientific Name	Behaviour	Presence
All year (Jun - Aug)		
Natator depressus		
Flatback Turtle [59257]	Nesting	Known to occur
Dec - Jan		
Dermochelys coriacea		
Leatherback Turtle [1768]	Nesting	Known to occur



Lepidochelys olivacea Olive Ridley Turtle [1767]

# Nesting Known to occur

# Extra Information

State and Territory Reserves			[Resource Information]
Protected Area Name	Reserve Type	State	
Djukbinj	National Park	NT	
Garig Gunak Barlu	National Park	NT	
Garig Gunak Barlu	Marine Park	NT	

Nationally Important Wetlands	]	Resource Information ]
Wetland Name	State	
Cobourg Peninsula System	NT	

EPBC Act Referrals			[Resource Information]
Title of referral	Reference	Referral Outcome	Assessment Status
<u>Clarence Strait Offshore Tidal Energy</u> <u>Project</u>	2008/4660		Assessment
Darwin Pipeline Duplication (DPD) Project	2022/09372		Post-Approval
Darwin Pipeline Duplication DPD Project	2022/9166		Completed
Marine Route Survey for Subsea Fibre Optic Data Cable System - Australia West	2024/09826		Completed
Controlled action			
Bonaparte Liquified Natural Gas Project	2011/6141	Controlled Action	Post-Approval
Glyde Point Industrial Estate	2001/336	Controlled Action	Completed
Glyde Point Industrial Estate and Associated Infrastructure	2004/1506	Controlled Action	Completed
Hardwood Plantation	2001/229	Controlled Action	Post-Approval

Ichthys Gas Field, Offshore and onshore processing facilities and subsea pipeline

Kilimiraka Mineral Sands and Associated Infrastructure (Bathurst Island), NT 2008/4208 Controlled Action Post-Approval

2012/6587 Controlled Action Assessment Approach

Shipping Channel Enhancement

2010/5431 Controlled Action Completed

Title of referral	Reference	Referral Outcome	Assessment Status
Controlled action			
Tropical Tidal Testing Centre, Clarence Strait, 50km NE Darwin	2014/7299	Controlled Action	Guidelines Issued
Not controlled action			
2D Seismic Survey in Permit Areas WA-318-P & WA-319-P, near Cape Londonderry	2004/1687	Not Controlled Action	Completed
Construction and operation of Radar Infrastructure	2004/1406	Not Controlled Action	Completed
Marine Survey for the Australia- ASEAN Power Link AAPL	2020/8714	Not Controlled Action	Completed
Nexus Drilling Program NT-P66	2007/3745	Not Controlled Action	Completed
Project Sea Dragon Stage 1 Hatchery - Gunn Point, NT	2017/8092	Not Controlled Action	Completed
Not controlled action (particular manne	r)		
2D and 3D Seismic Survey	2011/6197	Not Controlled Action (Particular Manner)	Post-Approval
2D Marine Seismic Survey	2009/4728	Not Controlled Action (Particular Manner)	Post-Approval
<u>2D marine seismic survey of</u> Braveheart,Kurrajong,Sunshine and Crocodile	2006/2917	Not Controlled Action (Particular Manner)	Post-Approval
2D Seismic survey	2009/5076	Not Controlled Action (Particular Manner)	Post-Approval
3D Marine Seismic Survey	2009/4681	Not Controlled Action (Particular Manner)	Post-Approval

Bonaparte 2D & 3D marine seismic survey

2011/5962 Not Controlled Post-Approval Action (Particular Manner)

Bonaparte Basin Seabed Mapping Survey 2009/4951 Not Controlled Post-Approval Action (Particular Manner)

Bonaparte Seismic and Bathymetric Survey 2012/6295 Not Controlled Post-Approval Action

Title of referral	Reference	Referral Outcome	Assessment Status
Not controlled action (particular manne	er)		
		(Particular Manner)	
Joseph Bonaparte Gulf Seabed mapping survey	2010/5517	Not Controlled Action (Particular Manner)	Post-Approval
Kingtree & Ironstone-1 Exploration Wells	2011/5935	Not Controlled Action (Particular Manner)	Post-Approval
Marine Environmental Survey 2012	2012/6310	Not Controlled Action (Particular Manner)	Post-Approval
NT/P77 3D Marine Seismic Survey	2009/4683	Not Controlled Action (Particular Manner)	Post-Approval
Offshore Fibre Optic Cable Network Construction & Operation, Port Hedland WA to Darwin NT	2014/7223	Not Controlled Action (Particular Manner)	Post-Approval
Petrel MC2D Marine Seismic Survey	2010/5368	Not Controlled Action (Particular Manner)	Post-Approval
Removal of Potential Unexploded Ordnance within NAXA	2012/6503	Not Controlled Action (Particular Manner)	Post-Approval
Santos Petrel-7 Offshore Appraisal Drilling Programme (Bonaparte Basin)	2011/5934	Not Controlled Action (Particular Manner)	Post-Approval
Sonar and Acoustic Trials	2001/345	Not Controlled Action (Particular	Post-Approval

Manner)

# Westralia SPAN Marine Seismic Survey, WA & NT

# 2012/6463 Not Controlled Post-Approval Action (Particular Manner)

Referral decision2D Marine Seismic Survey2008/4623Referral DecisionCompleted

# Key Ecological Features

Key Ecological Features are the parts of the marine ecosystem that are considered to be important for the biodiversity or ecosystem functioning and integrity of the Commonwealth Marine Area.

 Name
 Region

 Carbonate bank and terrace system of the Van Diemen
 North

 Rise
 Rise

Biologically Important Areas		[Resource Information]
Scientific Name	Behaviour	Presence
Dolphins		
Sousa chinensis		
Indo-Pacific Humpback Dolphin [50]	Breeding	Known to occur
Marine Turtles		
Chelonia mydas		
Green Turtle [1765]	Foraging	Known to occur
Eretmochelys imbricata		
Hawksbill Turtle [1766]	Internesting	Likely to occur
Lepidochelys olivacea		
Olive Ridley Turtle [1767]	Foraging	Known to occur
Lepidochelys olivacea		
Olive Ridley Turtle [1767]	Internesting	Likely to occur
Natator depressus		
Flatback Turtle [59257]	Internesting	Likely to occur

# Caveat

# 1 PURPOSE

This report is designed to assist in identifying the location of matters of national environmental significance (MNES) and other matters protected by the Environment Protection and Biodiversity Conservation Act 1999 (Cth) (EPBC Act) which may be relevant in determining obligations and requirements under the EPBC Act.

The report contains the mapped locations of:

- World and National Heritage properties;
- Wetlands of International and National Importance;
- Commonwealth and State/Territory reserves;
- distribution of listed threatened, migratory and marine species;
- listed threatened ecological communities; and
- other information that may be useful as an indicator of potential habitat value.

## 2 DISCLAIMER

This report is not intended to be exhaustive and should only be relied upon as a general guide as mapped data is not available for all species or ecological communities listed under the EPBC Act (see below). Persons seeking to use the information contained in this report to inform the referral of a proposed action under the EPBC Act should consider the limitations noted below and whether additional information is required to determine the existence and location of MNES and other protected matters.

Where data is available to inform the mapping of protected species, the presence type (e.g. known, likely or may occur) that can be determined from the data is indicated in general terms. It is the responsibility of any person using or relying on the information in this report to ensure that it is suitable for the circumstances of any proposed use. The Commonwealth cannot accept responsibility for the consequences of any use of the report or any part thereof. To the maximum extent allowed under governing law, the Commonwealth will not be liable for any loss or damage that may be occasioned directly or indirectly through the use of, or reliance on the contents of this report.

# 3 DATA SOURCES

### Threatened ecological communities

For threatened ecological communities where the distribution is well known, maps are generated based on information contained in recovery plans, State vegetation maps and remote sensing imagery and other sources. Where threatened ecological community distributions are less well known, existing vegetation maps and point location data are used to produce indicative distribution maps.

### Threatened, migratory and marine species

Threatened, migratory and marine species distributions have been discerned through a variety of methods. Where distributions are well known and if time permits, distributions are inferred from either thematic spatial data (i.e. vegetation, soils, geology, elevation, aspect, terrain, etc.) together with point locations and described habitat; or modelled (MAXENT or BIOCLIM habitat modelling) using point locations and environmental data layers.

Where little information is available for a species or large number of maps are required in a short time-frame, maps are derived either from 0.04 or 0.02 decimal degree cells; by an automated process using polygon capture techniques (static two kilometre grid cells, alpha-hull and convex hull); or captured manually or by using topographic features (national park boundaries, islands, etc.).

In the early stages of the distribution mapping process (1999-early 2000s) distributions were defined by degree blocks, 100K or 250K map sheets to rapidly create distribution maps. More detailed distribution mapping methods are used to update these distributions when time permits.

## 4 LIMITATIONS

The following species and ecological communities have not been mapped and do not appear in this report:

- threatened species listed as extinct or considered vagrants;
- some recently listed species and ecological communities;
- some listed migratory and listed marine species, which are not listed as threatened species; and
- migratory species that are very widespread, vagrant, or only occur in Australia in small numbers.

The following groups have been mapped, but may not cover the complete distribution of the species:

• listed migratory and/or listed marine seabirds, which are not listed as threatened,

have only been mapped for recorded breeding sites; and

• seals which have only been mapped for breeding sites near the Australian continent

The breeding sites may be important for the protection of the Commonwealth Marine environment.

Refer to the metadata for the feature group (using the Resource Information link) for the currency of the information.

# Acknowledgements

This database has been compiled from a range of data sources. The department acknowledges the following custodians who have contributed valuable data and advice:

-Office of Environment and Heritage, New South Wales -Department of Environment and Primary Industries, Victoria -Department of Primary Industries, Parks, Water and Environment, Tasmania -Department of Environment, Water and Natural Resources, South Australia -Department of Land and Resource Management, Northern Territory -Department of Environmental and Heritage Protection, Queensland -Department of Parks and Wildlife, Western Australia -Environment and Planning Directorate, ACT -Birdlife Australia -Australian Bird and Bat Banding Scheme -Australian National Wildlife Collection -Natural history museums of Australia -Museum Victoria -Australian Museum -South Australian Museum -Queensland Museum -Online Zoological Collections of Australian Museums -Queensland Herbarium -National Herbarium of NSW -Royal Botanic Gardens and National Herbarium of Victoria -Tasmanian Herbarium -State Herbarium of South Australia -Northern Territory Herbarium -Western Australian Herbarium -Australian National Herbarium, Canberra -University of New England -Ocean Biogeographic Information System -Australian Government, Department of Defence Forestry Corporation, NSW -Geoscience Australia -CSIRO -Australian Tropical Herbarium, Cairns -eBird Australia -Australian Government – Australian Antarctic Data Centre -Museum and Art Gallery of the Northern Territory -Australian Government National Environmental Science Program

-Australian Institute of Marine Science

-Reef Life Survey Australia

-American Museum of Natural History

-Queen Victoria Museum and Art Gallery, Inveresk, Tasmania

-Tasmanian Museum and Art Gallery, Hobart, Tasmania

-Other groups and individuals

The Department is extremely grateful to the many organisations and individuals who provided expert advice and information on numerous draft distributions.

Please feel free to provide feedback via the Contact us page.

© Commonwealth of Australia

Department of Climate Change, Energy, the Environment and Water GPO Box 3090 Canberra ACT 2601 Australia +61 2 6274 1111

# Appendix D Relevant Persons Consultation and Advertising Materials

# **Appendix D: Relevant Persons Consultation Material**

# **Preliminary consultation email**

From: Consultation, Santos
Sent: Friday, February 9, 2024 1:51 PM
Subject: Santos Barossa Productions Operations - - Environment Plan and Environmental Management Plan - Preliminary Consultation

Consultation on Barossa Production Operations Activity covered by:

- the Barossa Production Operations Environment Plan (Commonwealth waters)
- the Barossa Operations Environmental Management Plan (Northern Territory waters)

### <u>Overview</u>

The Santos-operated Barossa Gas Project is an offshore gas and condensate project with the purpose of providing a new source of gas for the existing Darwin liquified natural gas (DLNG) facility at Wickham Point in the Northern Territory (NT).

Santos is contacting you as we are proposing to undertake Barossa Production Operations Activity in Commonwealth waters and NT waters. As part of obtaining authorisation for this activity, Santos is undertaking consultation for the following regulatory approvals:

- The Production Operations Environment Plan (EP) relating to the arrival and operations of the FPSO, operation of a subsea production system and supporting subsea infrastructure, and operation of a 285 km section of the GEP located in Commonwealth waters where offshore petroleum activities are regulated under the Offshore Petroleum and Greenhouse Gas Storage Act 2006 (Cth) (OPGGS Act).
- The Operations Environmental Management Plan (OEMP) which includes the:
  - 8.26 km section of the GEP in NT coastal waters covered by the *Petroleum* (Submerged Lands Act) 1981 (NT) (PSL Act); and
  - ~92 km section of the GEP inshore of NT waters covered by the *Energy Pipelines Act 1981* (NT) (Energy Pipelines Act).

Under section 25 of the *Offshore Petroleum and Greenhouse Gas Storage (Environment) Regulations* (Cth) 2023 (OPGGS Environment Regulations), in preparing the Environment Plan for Barossa Production Operations activities in Commonwealth waters, Santos is required to consult with relevant persons, which includes:

- Commonwealth Departments or agencies to which our proposed activities may be relevant;
- State/Territory Departments or agencies to which our proposed activities may be relevant;
- the Department of the responsible Northern Territory Minister; and
- persons or organisations whose functions, interests and activities may be affected by our proposed activities.

In preparing an OEMP for activities in Northern Territory coastal waters under the Northern Territory PSL Act and applied Commonwealth environmental regulations, Santos is required to consult with relevant persons.

You or your department, agency or organisation may be a relevant person for the purposes of the Barossa Production Operations EP or OEMP.



<u>https://www.santos.com/wp-content/uploads/2024/02/Production-</u> <u>Operations-Information-Booklet.pdf</u>A booklet containing information about these activities in Commonwealth and NT waters can be found online at Barossa Production Operations Activity Information Booklet

The booklet includes information on the proposed activities, potential impacts, risks and management measures and the presence, based on a review of publicly available information, of environmental, social, economic, and cultural features and/or values within the environment that may be affected. The booklet and further information are located

on the Production Operations section of the Santos website.

### Seeking information and what's next

At this stage, Santos is seeking information to better understand:

- if you are from a Department or agency, or a person (or organisation) whose functions, interests or activities may be affected by the activity proposed to be carried out under the EP or OEMP; and
- what the functions, interests or activities you or your organisation have that may be affected by the proposed Production Operations activities.

Please contact us at the earliest opportunity if you consider you may be a relevant person to allow time to initiate consultation with you. Please also let us know if you know of other Departments, agencies, persons, or organisations which you believe we should consult. You can do this online via the relevant person nomination form located at <u>Production Operations</u>, via return email at <u>offshore.consultation@santos.com</u> or by calling us toll free on **1800 267 600**.

### **Consultation**

Consultation for Production Operations Activity under Commonwealth environmental regulations will formally commence on **Monday 11 March 2024** with the consultation period closing on **Tuesday 9 April 2024**.

If you would like to provide information, please note that the information you provide will be included in documentation submitted to NOPSEMA and DITT (defined below) for assessment. This will include our assessment of the information you provide so that Santos can better understand the environmental risks and impacts from the activities and our response to you.

The information you provide during consultation will be used for the development of the following documents:

- an Environment Plan for the activity in Commonwealth waters, which will be assessed by the National Offshore Petroleum Safety and Environmental Management Authority (NOPSEMA); and
- an Operations Environmental Management Plan for the activity in Northern Territory coastal waters, which will be assessed by the Energy Division within the Northern Territory Department of Industry, Tourism and Trade (DITT).

Please let us know if you would like any particular information you provide to not be published. If requested, Santos will include your information in a separate report which will not be published on NOPSEMA's website. Santos will handle your information in accordance with our <u>Offshore Western Australia and Northern Territory Consultation</u> <u>Privacy Policy</u>. Importantly, we recognise that First Nations people and groups may have concerns about sharing culturally sensitive information so we will follow your guidance when undertaking consultation activities.



https://www.nopsema.gov.au/sites/default/files/documents/Consultation on offshore petroleum environment plans brochure.pdf

### Additional resources

NOPSEMA has published information that sets out titleholders' responsibilities for consultation, as well as opportunities for relevant persons to provide guidance for consultation expectations. Click the image to read the information in full.

We look forward to hearing from you soon. Regards

### **Barossa Consultation Coordinator**

Email: offshore.consultation@santos.com Phone: 1800 267 600

# **Consultation email**

From: Consultation, Santos
Sent: Monday, March 11, 2024 5:02 PM
Subject: Santos Barossa Productions Operations - - Environment Plan and Environmental Management Plan - Consultation

### Consultation on Barossa Production Operations Activity covered by:

- the Barossa Production Operations Environment Plan (Commonwealth waters)
- the Barossa GEP Operations Environmental Management Plan (Northern Territory waters)

Santos is contacting you as we are proposing to undertake Barossa Production Operations activities in Commonwealth waters and Northern Territory (NT) waters. As part of obtaining authorisation for this activity, Santos is undertaking consultation for the following regulatory approvals:

- The Production Operations Environment Plan (EP) relating to the arrival and operations of the FPSO, operation of a subsea production system and supporting subsea infrastructure, and operation of a 285 km section of the GEP located in Commonwealth waters where offshore petroleum activities are regulated under the *Offshore Petroleum and Greenhouse Gas Storage Act 2006* (Cth) (OPGGS Act).
- The GEP Operations Environmental Management Plan (OEMP) which includes the:
  - 8.26 km section of the GEP in NT coastal waters covered by the *Petroleum* (*Submerged Lands Act*) 1981 (NT) (PSL Act); and
  - ~92 km section of the GEP inshore of NT waters covered by the *Energy Pipelines Act 1981* (NT) (Energy Pipelines Act).

Under section 25 of the Offshore Petroleum and Greenhouse Gas Storage (Environment) Regulations (Cth) 2023 (OPGGS Environment Regulations), in preparing the EP for the activities in Commonwealth waters, Santos is required to consult with relevant persons, which includes:

Commonwealth Departments or agencies to which our proposed activities may be relevant;

State/Territory Departments or agencies to which our proposed activities may be relevant; the Department of the responsible Northern Territory Minister; and

persons or organisations whose functions, interests and activities may be affected by our proposed activities.

In preparing an OEMP for activities in Northern Territory coastal waters under the Northern Territory PSL Act and applied Commonwealth environmental regulations, Santos is required to consult with relevant persons.

On 9 February 2024, Santos contacted you to advise that consultation for Barossa Production Operations activities under Commonwealth environmental regulations would commence on 11 March 2024 and to seek information as to whether your department, agency or organisation may be a relevant person for the purposes of the EP or OEMP.

### Consultation

<u>As advised in the email of 9 February, consultation for Barossa Production Operations</u> activities under Commonwealth environmental regulations has now commenced, with the consultation period closing on **Tuesday, 9 April 2024**. During the consultation period we are seeking information on the environmental values in the operational area and the environment that may be affected by the proposed activities, and the environmental impacts and risks associated with the proposed activities.

### You can provide information via return email or call us toll free on 1800 267 600.

The information provided by you during consultation will be used for the development of the following documents:

- The EP for the activity in Commonwealth waters, which will be assessed by the National Offshore Petroleum Safety and Environmental Management Authority (NOPSEMA); and
- The OEMP for the activity in NT coastal waters, which will be assessed by the Energy Division within the NT Department of Industry, Tourism and Trade (DITT).



<u>https://www.santos.com/wp-content/uploads/2024/02/Production-</u> <u>Operations-Information-Booklet.pdf</u>A booklet containing information about these activities in Commonwealth and NT coastal waters has been prepared by Santos.

It includes information on the proposed activities, potential impacts, risks and management measures and the presence, based on a review of publicly available information, of environmental, social, economic, and cultural features and/or values within the environment that may be affected.

The booklet can be found online at <u>Barossa Production Operations</u>

<u>Activity Information Booklet</u> or by clicking on the image opposite. The booklet and further information are located on the <u>Production Operations</u> section of the Santos website.

Please note that the information you provide will be included in the documentation submitted to NOPSEMA and DITT for assessment. This will include our assessment of the information you provide so that Santos can better understand the environmental risks and impacts from the activities and our response to you.

Please let us know if you would like any particular information you provide to not be published. If requested, Santos will include your information in a separate report which will not be published on NOPSEMA's website. Santos will handle your information in accordance with our <u>Offshore Western Australia and Northern Territory Consultation</u> <u>Privacy Policy</u>.

Importantly, we recognise that Indigenous people and groups may have concerns about sharing culturally sensitive information so we will follow your guidance when undertaking consultation activities.

Relevant persons being consulted under the OPGGS Environment Regulations should note that they:

- are entitled to be given sufficient information to allow them to make an informed assessment of the possible consequences of the activity on their functions, interests or activities; and
- are entitled to be allowed a reasonable period for the consultation.

NOPSEMA has published information that sets out titleholders' responsibilities for consultation, as well as opportunities for relevant persons to provide guidance for consultation expectations. Click the image to read in full.



Santos has previously sought information to better understand:

- if you are from a government Department or agency, how the proposed Production Operations activities may be relevant to your Department or agency;
- if you know of other government Departments, agencies, persons or organisations which you believe we should consult; and
- what (if any) functions, interests or activities you or your organisation have that may be affected by the proposed activities.

You can still contact us with this information during consultation.

We look forward to hearing from you soon.

Regards

Barossa Consultation Coordinator Email: <u>offshore.consultation@santos.com</u> Phone: 1800 267 600 Production operations information booklet – original version

# Santos

INFORMATION BOOKLET BAROSSA PRODUCTION OPERATIONS ACTIVITY





## INTRODUCTION

The activities described in this booklet relate to the extraction, processing, and distribution of gas and condensate from the Barossa Field.

The Barossa Development facilities used in these activities consist of a Floating Production Storage and Offloading (FPSO) facility, subsea production wells, supporting in-field subsea infrastructure (Figure 1), a 285 km Gas Export Pipeline (GEP) in Commonwealth waters, and an 8.26 km section of the GEP in Northern Territory (NT) coastal waters; collectively termed in this booklet as "Production Operations Activity".

As part of obtaining authorisation for this activity, Santos is undertaking consultation for the following regulatory approvals:

- The Production Operations Environment Plan (EP) relating to the arrival and operations of the FPSO, operation of a subsea production system and supporting subsea infrastructure, and operation of a 285km section of the GEP located in Commonwealth waters where offshore petroleum activities are regulated under the Offshore Petroleum and Greenhouse Gas Storage Act 2006 (Cth) (OPGGS Act).
- The Operations Environmental Management Plan (OEMP) which includes the:
   8.26 km section of the GEP in NT coastal waters covered by the Petroleum (Submerged Lands Act) 1981 (NT)
   (PSL Act): and
  - -92km section of the GEP Inshore of NT waters covered by the Energy Pipelines Act 1981 (NT) (Energy Pipelines Act).

The term 'GEP' refers to the Gas Export Pipeline through which Barossa gas will be transported from the Barossa field to Darwin LNG. However, the scope of the GEP covered in this bookiet, is limited to the 8.26 km section of the GEP located in NT coastal waters.

The activities, environmental impacts, and risks for the GEP in NT waters (-92km) are broadly similar to those for the GEP in Commonwealth waters (described in this booklet). The activities, environmental impacts and risks specific to the GEP in NT waters, not covered in this booklet, will be provided in a separate factsheet.

The estimated life of the Barossa Development is 25 years, and the Production Operations EP and the OEMP will be reviewed every five years following initial regulator authorisation. This booklet provides a summary of the credible environmental impacts and risks associated with the first five years (also known as Barossa Phase 1) of the Production Operations Activity.

			PSO
	a state of the sta	A. constitution of	
Pipel Term	ine End ination (PLET)	Riser	Flowline End Terminations (FLET)
Umbilical Termination	Mooring Line	Flowline	Manifold
Assembly UTA)	Gas Expor	t Pipeline Chris	tmas Tree Umbilical

Figure 1: Schematic of the Barossa Field Subsea production system and infrastructure, and FPSO



# **ACTIVITY LOCATION**

The planned Production Operation Activity is confined to two key operational areas. These areas are simply termed Operational Area 1 (OA1) and Operational Area 2 (OA2) (Figure 2).

OA1: The Barossa field. This is the area in which the FPSO, subsea production system, and supporting subsea infrastructure will be used to process gas and condensate extracted from the Barossa wells. The area is confined to Commonwealth waters, approximately 285 km north-north-west of Darwin (the closest major populated centre), approximately 210 km north-west of the mainland NT coastline, and approximately 130 km north of the Tiwl Islands at the closest point (Seaguii Island).

OA2: The 285 km section of the GEP from OA1 to the Commonwealth waters/ NT waters boundary; and the 8.26 km section of the GEP situated in NT coastal waters between the Commonwealth waters/NT coastal waters boundary and the Territorial Sea Baseline (TSB). Not included in this section is the remaining GEP in NT waters (-92 kms). Information on this is covered elsewhere.



Figure 2: Location of the Barossa Operational Area 1 (OA1) in Commonwealth waters (OPPGS Act) and Operational Area 2 (OA2) in Commonwealth waters and NT coastal waters (PSL Act).

## SUMMARY OF ACTIVITIES

The Barossa FPSO, which will be known as BW Opal, is a permanently moored vessel which is able to freely rotate with the strongest wind direction and remain connected to subsea facilities. The operational design life for all Barossa facilities is 25 years. **Figures 3 and 4** depict key features of the FPSO and utilities/marine systems.

Gas and condensate extracted from the Barossa field will be processed on the FPSO to separate the natural gas and condensate (a light liquid hydrocarbon, straw coloured and flammable). The dry gas will be exported to the DLNG facility via the GEP, and the condensate will be transferred from the FPSO to offtake tankers for export approximately four to five times per year. The key activities proposed under the Production Operations EP are detailed below.

Most of the below activities are only applicable to OA1 as this is where the FPSO and subsea infrastructure is located, and the condensate offtakes will occur. The main activity in OA2 will be inspection, maintenance, monitoring, and repair (IMMR) of the GEP.

- FPSO arrival in the field: connection of the FPSO to the mooring buoy; equipment and systems testing (also termed commissioning), start-up operations.
- FPSO operations: process gas and condensate from the Barossa field. The gas and condensate separation and treatment
  systems have a gas export capacity of approximately 635 million standard cubic feet per day (a little over 7000 Olympicsize swimming pools) with a condensate processing capacity of approximately 10,000 barrels per day (a little more than hair
  an Olympic-size swimming pool) and a produced water processing maximum capacity of 20,000 barrels per day (over one
  Olympic-size swimming pool). Under normal operating conditions produced water discharge rates will be approximately a
  quarter (-5,000 barrels per day) of the maximum discharge capacity (-1/4 of an Olympic-size swimming pool). The FPSO
  generates its own electricity using Barossa production gas, and potable water supply. Living quarters are provided for the
  operations workforce.
- Gas export to DLNG: dry natural gas will be transported through the GEP for onshore processing at the DLNG facility.
   Storage and Offtake operations: storage of condensate onboard the FPSO and offloading of condensate to offtake tankers. Approximately 65,0000 barrels (- 40 Olympic-size swimming pools) will be offloaded approximately once every two to three months (four to five times per year).
- 4 | BAROSSA PRODUCTION OPERATIONS ACTIVITY INFORMATION BOOKLET



- Support operations: offshore support vessels periodically visit the FPSO to resupply materials (such as stores, consumables, chemicals and fuel) and return surplus goods and wastes to the Australian mainland for disposal or recycling. Helicopters will be used to transport the operations workforce to and from operations facilities.
- Subsea system and GEP Inspection, maintenance, monitoring and repair (IMMR): visual inspection of subsea infrastructure and/or the GEP using Remotely Operated Vehicles (a submersible craft used to perform underwater visual inspections operated from a vessel). This activity will be performed according to a planned inspection and maintenance schedule, or at other intervals if unplanned inspections or repairs are required.



Figure 3: FPSO key features



- O Condensate Offloading System Aft Service Crane
- O Helideck Parking Area Forward Pedestal Crane
- O Power Generation Exhaust
- O Communications & Radar O Aft Pedestal Crane O Thermal Oxidiser
- O Helideck Landing Area
- 0 Midship Pedestal Crane

Freefall Lifeboats

Engine Room Exhaust

Figure 4: FPSO utilities and marine systems

## **REGIONAL EXISTING ENVIRONMENT** SUMMARY

#### Environment that may be affected (EMBA)

Santos recognises the region's various environmental values and sensitivities. In an EP, although planned activity occurs in OA1 and OA2, it is common to present a geographically defined area of the environment that may be affected (EMBA) by an offshore activity e.g. an unplanned hydrocarbon spill.

In the case of the Production Operations Activity, the broadest extent of the EMBA, is determined by a potential loss of heavy fuel oil from a condensate offtake tanker due to impact from another vessel. Potential loss of heavy fuel oil is a risk associated with any large marine vessel and is managed through established maritime laws and safeguards. Barossa condensate offtake operations are a low frequency activity (four to five times a year) which further reduces the likelihood of such an event, which is already a very low probability of occurring.

Figure 5 depicts operational areas OA1 and OA2 and the EMBA (blue line). The EMBA is generated by modelling and represents Figure a depicts operation areas of a ratio data and one affected by 300 individual inter. The branch is generated by individual and interaction of the greatest geographical extent that could be affected by 300 individual hydrocarbon spill scenarios occurring simultaneously across the full range of seasonal conditions. It should be noted that an actual spill event is more accurately represented by only one of the 300 simulations from the modelling, meaning a much smaller geographical area would be affected in the event of an actual spill, and the EMBA does not take account of spill response mitigations which would reduce the extent of an unplanned spill. The primary purpose of the EMBA is to assist with spill response planning and preparedness in the unlikely event of a hydrocarbon spill. The EMBA also provides the basis for assessing the range of potential socio-economic impacts and establishes a planning area for scientific monitoring during an unplanned spill event.

The Moderate Exposure Value (MEVA) (pink line) represents the predicted extent of ecological impacts and is used to inform the environmental impact assessment and spill response plans. Beyond the MEVA, impacts to ecological receptors are not expected.

To learn more about spill modelling, exposure values and spill response, see NOPSEMA Spill Modelling Video.

#### **Regional protected and significant areas**

Figures 6, 7, and 8 lilustrate the boundaries and zonings of regional marine parks and reserves, key ecological features, wetlands, EMBA, the MEVA and the OAs. Figure 9 lilustrates the shoals and banks in relation to the OAs.



Figure 5: Production Operations Activities EMBA and MEVA in Commonwealth waters (OPGGS Act) and NT coastal waters (PSL Act)



Figure 6: Marine Parks and Other Protected Areas in relation to OA1, OA2, EMBA and MEVA



Figure 7: Key Ecological Features in relation to OA1, OA2, EMBA and MEVA



Figure 8: RAMSAR wetlands in relation to OA1, OA2, EMBA and MEVA



Figure 9: Shoals and Banks in the vicinity of OA1 and OA2

#### Marine fauna and biologically important areas

Biologically important areas (BIAs) are areas used by protected marine species for carrying out critical life functions, such as reproduction, feeding, migration or resting. BIAs are areas that contain habitat crucial to the survival of protected species and are defined by the Australian Government under the Environment Protection and Biodiversity Conservation Act 2009 (EPBC Act). As shown in Figure 10, some BIAs occur within the EMBA. These areas are known to include protected marine species such as whale sharks, pygmy blue whales, dugongs, olive ridley turtles, loggerhead turtles, green turtles, hawkslill turtles, flatback turtles, and 12 types of seabirds and shorebirds. In addition, the EMBA overlaps the spawning grounds for southern bluefin tuna, a listed species under the EPBC Act, between northern Western Australia and Java.

The BIAs for flatback turtles overlap with OA2. There are no BIAs within OA1. Two turtle species, the flatback turtle and olive ridley turtle, have critical habitat that overlap with OA2.

Activities in OA1 will be conducted in water depths ranging from approximately 220 – 280 m. There are a variety of highly mobile marine fauna that may transit OA1 in iow numbers, such as:

- Bryde's, blue, fin, humpback, sperm and sel whales
- orcas, Australian snubfin dolphin and spotted bottlenose dolphin
- dugongs (mostly in shallow waters)
   olive ridley, green, loggerhead, hawksbill, leatherback and
- olive ridley, green, loggerhead, hawksbill, leatherback an flatback turtles
- sea snakes
- whale sharks
- migratory seabirds and shorebirds
  fishes, sharks, rays and sawfish.

the Barossa Marine Studies Program.

An additional three species – the grey nurse shark, Omura's whale and the turtie-headed sea snake have been included as they were observed within or near OA1 and OA2 during

Santos recognises the region's various environmental values and sensitivities and has considered government guidance, including protected species management plans, recovery plans, conservation advice and threat abatement plans in the development of the Production Operations and OEMP, and has developed control measures to reduce impacts and risks to marine fauna and biologically important areas to as low as reasonably practicable and acceptable levels.



Figure 10: Biologically important areas in relation to OA1, OA2, EMBA and MEVA.



10 | BAROSSA PRODUCTION OPERATIONS ACTIVITY INFORMATION BOOKLET

### REGIONAL SOCIO-ECONOMIC SUMMARY

Socioeconomic activities that may occur within the OAs or EMBA might include commercial, recreational and traditional (subsistence) fishing, aquaculture, tourism, petroleum industry activities, defence activities, shipping and to a lesser extent in the deeper offshore waters, recreational fishing and tourism.

Underwater cultural heritage and cultural values may also exist across the region. Darwin will be the logistics hub and supply base for the capitals when stating Production Operations Activity bringing employment and economic benefits to the local community.

#### Nearest population centres

OA2, at its closest point, is located approximately seven kilometres from Bathurst Island, which is part of the Tiwi Islands. Darwin is the closest regional city, which approximately 285 km north-north-west of OA1.

#### Summary of other uses within the EMBA

Santos' existing understanding of the uses and values of the area and its strategies to reduce impacts or risks to these uses and values, will be supplemented with any new information obtained during consultation. Santos has set out in the list below a summary of the uses and values of the area based on existing information or previous consultation. Santos welcomes further information that may be provided during consultation to inform the Production Operations EP and OEMP.

# S

#### Commercial fishing

Santos recognises the presence and rights of commercial fishers within the operational area and EMBA. Within the OAs, interaction with some commercial fishing is possible. These fishertes include Northern Prawn, Spanish Mackerel, Pearl Oyster, Offshore Net and Line, and Demersal. Santos has been consulting with the relevant fisheries representative associations, licence-holders and government over many years.



#### Petroleum industry

Several oil and gas companies hold petroleum permits near the OAs; however, no established oil and gas operations are located within or in the immediate surrounds. The closest operational offshore production facilities and in-field subsea infrastructure are associated with the Santos-operated Bayu-Undan platform, located approximately 400km to the southwest of OAI and west of OA2.

# Tourism, recreational fishing and traditional fishing

The OA1 is located in offshore waters that are not likely to be accessed for tourism activities (e.g. charter boat operations) or recreational fishing, as these tend to be centred around nearshore waters, islands and coastal areas. However, previous consultation on a different Barossa Gas Project EP has identified one fishing charter operator who may on occasions conduct tours near Evans Shoal, approximately 62km west of OA1. Tourism activities may occur within OA2, but they are likely to be limited to vessels transiting the area to access other destinations within the region e.g. Islands, shoals, and shipwrecks. Indonesian and Timorese traditional fishers, as well as Australian recreational fishers, are expected to transit and fish in the EMBA. Santos continues to consult regarding recreational and traditional fishing and hunting within the EMBA.



#### Defence Activities

Several oil and gas companies hold petroleum permits near the OAs; however, no established oil and gas operations are located within or in the immediate surrounds. The closest operational offshore production facilities and in-field subsea infrastructure are associated with the Santos-operated Bayu-Undan platform, located approximately 400km to the southwest or OA1 and west of OA2.

# Telecommunications cables

#### relecommunications caples

The North-West Cable System (NWCS) is a submarine telecommunication fibre cable system located within the EMBA and crosses the GEP in the southern portion of OA1. It is located approximately 230 km and 30 km south of OA1 and OA2. Extending 2100 km from Darwin to Port Hediand, the NWCS connects Australia's remote northern and western regions, including offshore oil and gas facilities, with onshore locations.

## (°,°,°)

#### Listed Heritage

There are no world heritage properties, national heritage places or Commonwealth heritage places within the OAs; however, the EMBA (including the MEVA) overlaps the Ashmore Reef Marine Park, a Commonwealth heritage place. The closest World Heritage Property is the Kakadu National Park, located onshore in the NT. A small portion of the coastal edge overlaps the EMBA. (Figure 8)

There are no recorded Aboriginal heritage sites within the OAs. The Tiwi Islands are a declared Aboriginal reserve and a number of protected sacred sites under the Aboriginal Sacred Sites Act 1989 (NT) have been recorded on the Islands.

Under the Commonwealth Underwater Cultural Heritage Act 2018, Australia's underwater cultural heritage is protected in Commonwealth waters, such as shipwrecks, sunken aircrart and other types of underwater cultural heritage including Australia's Aboriginal and Torres Strait Islander underwater cultural heritage. No known shipwrecks are located within the OAs. Multiple known shipwrecks, sunken aircrafts, historic aircrafts and shipwrecks (greater than 75 years old) and other sites occur within the EMBA. Some unlocated wrecks could fail within the boundaries of the OAs or EMBA.

In the course of preparing the Barossa Drilling and Completions EP, SURF EP and commencing works under the GEP EP, Santos engaged independent consultants to investigate potential for underwater cultural heritage within OAL First Nations underwater cultural heritage is not relevant to OA1 due to its location in water depths beyond the extent of the ancient coastline at the 125 m water depth contour. A 262km section of the GEP within OA2 has been surveyed for

12 | BAROSSA PRODUCTION OPERATIONS ACTIVITY INFORMATION BOOKLET

both First Nations and other underwater cultural heritage

The results of those surveys concluded there are no specific underwater cultural heritage places along the Barossa GEP to which people, in accordance with Indigenous tradition, may have spiritual and cultural connections that may be affected by the activities covered by the GEP Environment Plan. Further similar surveys are planned for the remainder of OA2 in 2024.

L.

#### Shipping

The closest port to the OAs is Darwin Port, which is approximately 290km away from OA1 and 116km away from OA2. No designated shipping fairways overlap the OAs, however the southern end of OA2 is an area of high shipping activity.



#### Cultural Values

Traditional hunting and fishing continue to occur on the Tiwi Islands, although typically these occur within 3 nm of the shoreline.

Mapping exercises and workshops conducted on the Tiwi Islands have identified Aboriginal heritage sites along the northern, western and southern coastlines or the Tiwi Islands, including areas used for food collection, sacred sites camping sites and a dreaming site. These coastlines are within the EMBA but outside the OAs.

Santos has identified that the Croker Island native title determination (DCD1998/001) partiality overlaps the EMBA. The native title holders within the Croker Island native title determination are the Yuwurrumu members of the Mandilarri-lidugij, the Mangalara, the Murran, the GaduraMinaga and the Ngaynjaharr clans. The Larrakla native title determination (DCD2006/001) also partiality overlaps the EMBA. This determination found that native title does not exist within the claim area.

Mapping exercises and workshops conducted on the Tiwi Islands have identified Aboriginal heritage sites along the northern, western and southern coastlines or the Tiwi Islands, including areas used for food collection, sacred sites, camping sites and a dreaming site. These coastlines are within the EMBA but outside the OAs.

Santos acknowledges coastal First Nations peoples' connection with culture through Sea Country and is seeking to improve knowledge and understanding of cultural features within the EMBA, including through consultation with First Nations people and their relevant representative bodies.

### SUMMARY OF ENVIRONMENTAL IMPACTS AND RISKS

Environmental impact and risk assessment is the process by which proposed activities are assessed for their impacts (consequences) on the environment (physical, biological, socio-economic and cultural). For the purposes of assessing impacts and risks, proposed activities are divided into planned activities and unplanned events.

Planned activities occur within OA 1 and OA 2 can have unavoidable impacts, such as light, noise and atmospheric emissions, seabed disturbance, discharges to the marine environment, and interactions with other marine users. Unplanned events are not expected to occur but are considered so that contingency measures are in place should they ever eventuate. Unplanned events include dropped objects, introduction of invasive marine species, interactions with marine fauna, accidental discharges, or spills.

Planned activities are assessed based on consequence of impact. Unplanned events are assessed based on their potential impact (consequence) and likelihood of occurrence, which informs the associated risk level.

Santos has conducted an environmental assessment in order to consider the potential environmental impacts and risks associated with activities under this EP. The identification of potential impacts and risks, and the measures proposed to reduce these impacts and risks, may be revised and amended as a result of the consultation process. This includes information obtained during consultation to improve Santos' understanding of potential impacts and risks in regards to cultural values within the EMBA and adoption of any appropriate measures.

# PLANNED ACTIVITIES

Santos proposes to adopt a suite of control measures to reduce impacts and risks associated with planned activities to a level that results in a minor or negligible environmental consequence. These consequence levels are considered by Santos to be acceptable and to have been reduced to as low as reasonably practicable (ALARP). Figure 11 shows several emissions and discharges from planned activities associated with the operation of the FPSO.



Figure 11: Emissions and discharges from planned activities associated with the operation of the Floating Storage and Offloading (FPSO) facility.



# **GHG EMISSIONS**

The Production Operations EP will consider the contribution of emissions from the Barossa Development to national and global emissions and the potential indirect impacts of climate change on the Australian environment, noting that as a result of the complex nature of the global emissions system, climate change impacts cannot be meaningfully linked to any one activity or emissions source.

- GHG emissions can be categorised into Scope 1, Scope 2 and Scope 3
- Scope 1 direct emissions from sources that Santos owns or controls, due to fuel combustion, flaring, venting, CO2 removal and fugitive emissions.
- Scope 2 Indirect emissions from the generation of energy that Santos purchases for its operations including electricity purchased for ancillary activities such as office buildings.
- Scope 3 includes all indirect emissions not included in Scope 2. The vast majority of Scope 3 emissions from Santos' activities are emissions from the use of sold products.

The Production Operations Activity will not produce scope 2 emissions as It does not consume externally generated electricity or other forms of externally generated energy.

Total annual Barossa Scope 1 emissions are estimated to be 2.5 Mt CO2e (carbon dioxide equivalent), and total annual Scope 3 emissions are estimated to be 12.7 Mt CO2e.

#### What Impacts are expected?

Barossa GHG emissions (Scope 1 and 3) estimates account for approximately 0.86% of annual Australian GHG emissions (Department of Climate Change, Energy, the Environment and Water, 2022).

The GHG emissions attributable to the Barossa Development are not expected to be significant relative to national and international GHG emissions and are considered to be low risk.

#### How will Santos manage impacts?

Scope 1 emissions from Barossa are managed under Australian regulations and scope 3 emissions are managed using control measures consistent with the UN Paris Agreement, to which the Australian Government is a signatory. Key proposed control measures include:

#### Scope 1 emissions:

- Barossa will comply with Safeguard Mechanism obligations, including surrendering carbon credit units for any emissions above the Safeguard baseline for the assessment year.
- Barossa will implement a GHG management plan that minimises GHG emissions to ALARP and acceptable levels over the life of the field operations.
- Barossa facilities design has been optimised to reduce fuel, flare and vent (FFV) emissions, and to enable the possibility of future export of reservoir CO2 to a Carbon Capture and Storage (CCS) project.

#### Scope 3 emissions:

 Products generated from the Barossa Development will only be sold to customers from countries that are signatories to the Paris Agreement (or that have policies for reducing greenhouse gas emissions that are equivalent to policies required by the Paris Agreement), as at the date of the relevant contract of sale.



## ATMOSPHERIC EMISSIONS

Fuel consumption, flaring and venting excess gas is required to process gas and condensate which results in the release of air pollutants, such as sulphur oxides (SOX), nitrogen oxides (NOX) and volatile organic compounds (VOCs) to the atmosphere. This mix of continuous and infrequent (e.g. flaring of excess gas) sources of atmospheric emissions associated with operating the Barossa facilities may result in a temporary, localised reduction in air quality.

Intermittent flaring is expected to be of short duration during initial start-up operations and unplanned process trips/upsets during steady-state operations

Atmospheric emissions will also be generated from support vessel and helicopter operations.

All activities described above that may result in air emissions could be expected within OA1, however only emissions associated with vessel activities would be expected in OA2. In the offshore environment, air emissions quickly dissipate into the surrounding airshed.

#### What Impacts are expected?

Impacts are considered very localised and not significant. Seabirds and migratory shorebirds are unlikely to be impacted by the localised and temporary reduction in air quality.

The potential impact from the release of air emissions includes the decrease in air quality of the local airshed.

Behavioural impacts, such as avoidance, could be expected if seabirds fly in the vicinity of OA1. Impacts to threatened, migratory or local fauna (seabirds) are considered to be minor.

As Barossa's operational activities occur in remote offshore waters, Production Operations Activity emissions will not impact air quality in coastal towns. Atmospheric emissions will quickly dissipate into the surrounding atmosphere and are not considered to be a potential source of impact for protected areas or threatened ecological communities.

#### How will Santos manage impacts?

The FPSO power generation system reduces emissions to the atmosphere by primarily using production gas as fuel. Combined-cycle gas turbines (CCGT) will also be used to improve fuel use efficiency. CCGT are highly efficient and best practice resulting in reduced pollution. The steam turbines use low NOX burners, which significantly reduces NOX emissions.

Further FPSO facility design measures to reduce atmospheric emissions include:

- surplus waste gas burned to completely remove methane (minimal flaring)
- vapor recovery instead of releasing to atmosphere
   ozone Depieting Substances (ODS) are not used as refrigerants.

Santos proposes to adopt numerous control measures to manage vessel emissions, including requiring contractor vessels' compilance with MARPOL requirements for low-sulphur fuel and air pollution prevention certifications. ("MARPOL' is a reference to the international Convention for the Prevention of Pollution from Ships).

The control measures to be adopted are designed to be consistent with maritime regulations and petroleum industry standards.



## NOISE SOURCES

In OA1 the main FPSO noise sources are intermittent and short-term flaring during initial start-up operations and unplanned process trips/upsets during steady-state operations. FPSO power and processing equipment will also generate a continuous source of noise. Other noise sources include:

- support vessels; helicopters:
- ROVs, acoustic positioning systems and survey equipment · operation of subsea infrastructure such as wellheads, flowlines and valves.

In OA2 the types of noise generated by these activities can be categorised as either. impulsive (brief, high intensity) e.g. from operation of survey equipment or, non-impulsive noises (ongoing or continuous) e.g., from vessel engines. Noise emitted from activities in OA1 and OA2 are expected to be at low levels, similar to ambient noise levels in the region.

#### What Impacts are expected?

Santos has engaged subject matter experts to conduct the underwater noise assessments for the activities.

Noise emissions from the FPSO, helicopters, survey equipment and vessels may result in marine mammals (e.g. whales) changing their behaviour (e.g. avoidance or diving to avoid noise). This change in behaviour is expected to be localised (within the area of the noise source in OA1 or OA2) and short term (e.g. periods of minimal flaring in OA1). Noise emissions are not expected to cause long term population impacts (e.g. distribution & abundance).

Low level noise can occur from the operation of the pipeline which will dissipate to background levels within 100m of the pipeline. The GEP route (OA2) crosses two small areas of important turtle habitat and impacts to marine turtles in these areas are expected to be limited to behavioural (e.g. avoidance). Impacts to marine turties from any infrequent survey equipment use is also expected to result only in temporary and localised behavioural changes, given the low level of noise. Vessels will be moving when undertaking surveys and it is highly unlikely any individual would remain near the noise source for any length of time.

Other protected species of marine reptiles (such as sea snakes), seabirds and fish (such as sharks and sawfish) are not expected to be affected at the population level, given their wide distribution (in the case of sea snakes and sharks), distances to seabird breeding colonies, and preference for shallow coastal habitats (sawfish).

Noise emissions could result in behavioural changes in marine fauna within the Oceanic Shoais Marine Park, as OA2 is located in this area (Figure 4).

Noise is not expected to impact socio-economic receptors, including commercial fisheries, due to jow noise levels and jow socio-economic activity levels within and near the Operation Areas. Behavioural impacts to fish of potential commercial value would be restricted to within hundreds of metres of the noise source, a very small portion of the total available fishing area.

#### How will Santos manage impacts?

A source of significant underwater noise has been eliminated from the FPSO facility by designing the facility to be permanently moored without the use of a propulsion system.

Vapour recovery on the flare system reduces the frequency at which flaring occurs during operations and therefore reduces the amount of noise emitted during routine operations.

Activity vessels are required to comply with Santos's Protected Marine Fauna Interaction and Sighting Procedure to comply with regulatory requirements for managing fauna noise impacts. Marine assurance standards and planned vessel maintenance will minimise noise generated from vessels by ensuring contracted vessels are operated, maintained and crewed in accordance with industry standards and regulatory requirements.



## LIGHT SOURCES

Artificial lighting is required for operational and navigational safety during the activity. Light sources include safety and navigational lighting on vessels, campaign-specific lighting when needed, such as deploying or retrieving equipment or when ROVs are working underwater, and intermittent flaring from the FPSO.

#### What Impacts are expected?

Permanent safety and navigational lighting on the FPSO and intermittent flaring will result in light emissions in OA1. Light emissions are not expected to have an effect on adult turtles or hatchlings, given the offshore location and distance from the nearest turtles nesting beaches.

OA2 overlaps BIAs for marine turtles surrounding the Tiwi Islands. The closest turtle nesting beach is Cape Fourcroy, on the Tiwi Islands. Lighting emissions in OA2 will only occur from infrequent vessel inspection and maintenance activities in this location, which are of short durations. Vessel activities in OA2 are expected to produce similar light levels to other marine vessel activities in the region.

Impact to nesting females or hatchlings is not expected to occur. There is potential for hatchlings at sea to be attracted to light emissions if they are carried by currents to within approximately 3.3 km of an IMMR vessel. In the unlikely event hatchlings are attracted to vessel light, the proportion impacted is considered negligible when compared to the total number of hatchlings emerging from Bathurst Island beaches across the year. It will also be a temporary phenomenon, occurring during hours of darkness only. After sunrise, hatchling dispersal behaviour will resume. Displacement of individuals from critical habitat areas is therefore not a credible outcome.

Fish, sharks and birds have been shown to be attracted to artificial light sources, leading to a short-term localised increase in fauna activity, however large-scale changes in species abundance or distribution are unlikely.

#### How will Santos manage impacts?

The FPSO facility is equipped with a centralised battery system providing an uninterrupted power supply to the FPSO LED lighting which also allows for dimming and controlling in individual areas. Lighting is to be limited to that required for safe operations and navigation and will be compliant with maritime regulations (similar to other commercial vessels operating in the region).


## SEABED DISTURBANCE

Seabed disturbance will occur because of:

- · physical presence of installed subsea infrastructure and GEP on the seabed
- temporary placement and set down of equipment and subsea infrastructure on the seabed used during IMMR activities

### What Impacts are expected?

Seabed disturbance resulting from IMMR activities will be confined to the OAs and might result in localised disturbance under the subsea infrastructure and GEP. Seabed disturbance resulting from a subsea repair or replacement will be localised with a potential footprint of approximately 50 m<sup>2</sup> up to 1,600 m<sup>2</sup>.

Given the nature and relatively small scale of seabed disturbance, it is not expected to cause a decrease in local population size, area of occupancy of species, loss or disruption or critical habitat, or disruption to the breeding cycle or any protected marine fauna.

Given localised disturbance is restricted to the OAs, which is mostly bare sediment and does not contain any significant habitat features, the consequence level for the physical environment or habitat is negligible. Impacts to the seabed within the Oceanic Shoais Marine Park or overlapping key ecological features (KEFs) (Carbonate bank and terrace system of the Van Diemen Rise KEF and the Shelf break and slope of the Aratura Shelf KEF) are considered to be minor.

While OA1 does not overlap any marine turtle BIAs, the southern end of OA2 traverses internesting buffer habitat critical to survival for flatback and olive ridley turtles, overlaps a portion of the the internesting BIA for flatback turtles, and is adjacent to the internesting BIA for olive ridley turtles. Considering the water depth along the pipeline route in OA2 is greater than the maximum turtle interesting depth of 30 m , it is unlikely the species will be present in significant numbers or for significant periods. Any impact to marine turtles from seabed disturbance or resultant turbidity in both OA1 and OA2 would likely be temporary and negligible, based on the nature and scale of impact.

Seabed disturbance is not expected to impact commercial fisheries, based on the small size of disturbance compared with the total available fishing area.

### How will Santos manage Impacts?

During IMMR activities Santos' vessels will undertake safe and accurate placement of infrastructure using dynamic positioning to minimise seabed disturbance during placement. Santos will also maintain a comprehensive inventory of all installed equipment to enable recovery of all equipment during decommissioning to limit impacts to the seabed.



## INTERACTIONS WITH OTHER MARINE USERS

Other marine users will be displaced from part of OA1 over the life of Barossa operations, and temporarily restricted within parts of OA2 during IMMR. In OA2, the GEP may present a hazard to marine users due to the potential for snagging.

### What Impacts are expected?

Other marine users will have restricted access within petroleum safety zones (PSZ). A permanent 500m exclusion PSZ will extend around the outer edge of the Barossa Production Operations wells, the subsea infrastructure and mooring system in OA1. During IMMR activities along the GEP IN OA2, a temporary 500m PSZ will be maintained around vessel operations.

Commercial fishing, shipping, military exercises and other incidental marine traffic in the OAs are expected to be low frequency. The area marine users will be excluded from is small when compared to the large area available for their use.

### How will Santos manage impacts?

Santos will notify and communicate with other marine users using standard maritime notifications (e.g. Notice to Mariners) before, during and at the end of IMMR activities. Infrastructure locations will be marked on nautical charts. These proposed control measures are consistent with maritime regulations and industry practices.



## PRODUCED WATER DISCHARGES

Produced water is naturally occurring water that is extracted from the seabed along with hydrocarbons (condensate and gas in the case of the Barossa field). It is separated from the hydrocarbon components during processing and treated before being discharged to the marine environment from a pipe at least 10 m below the sea surface on the FPSO. This produced water consists of naturally occurring formation water (from the body of rock below the hydrocarbon formation), condensed water (water vapour present within the produced hydrocarbons which condenses when brought to the surface) as well as introduced water-soluble chemicals and other contaminants. While produced water treatment is performed before discharge, the effluent may contain residual inorganic (such as chemicals used for production) and organic (such as oil) contaminants.

The produced water treatment system is divided into two stages - removal of hydrocarbons through a filtration system and designed to handle 20,000 barreis per day (over one Olympic-size swimming pool). During operations the produced water discharge will vary from 3,500 to 5,000 barreis per day (bbl/day), or a quarter to a third of an Olympic-size swimming pool, with a peak rate after 11 years estimated up to 16,500 bbl/day (one Olympic-size swimming pool). Best available technology has been selected to remove oil-in-water concentrations to as low as reasonably practicable (ALARP) and the treatment system will operate well below its design capacity over the majority of the field life.

### What Impacts are expected?

Water quality may be impacted at the discharge point while the produced water is discharged (Figure 12). Discharge modelling has been undertaken for a conservative maximum discharge rate of 20,000 bbl/day (over one Olympic-size swimming pool). Under normai operating conditions produced water discharge rates will be approximately a quarter (-5,000 bbl/day) of the modelled maximum discharge rate. Modelling results indicated that species protection thresholds for waterborne contaminants is achieved at approximately 6 km from the FPSO. As a result, predicted impacts will be localised and considered minor.

Marine turtles may occur within the produced water mixing zone. It is possible individual turtles may traverse the mixing zone; however significant impacts are not expected to occur, and large numbers of animats are not expected to be exposed. That is because the discharge water depth and discharge location are not within the proximity of internesting turtle habitat, and there is minimal reef habitat in the mixing zone. Given marine turtles are transient through the produced water mixing zone, they will not be exposed to the produced water for enough time for contaminants to accumulate within their body. Behavioural impacts (such as avoidance) may occur to a small proportion (individuals) of a local population close to the produced water discharge.

Like turties, produced water exposure to plankton, fish, invertebrates and sharks is expected to be brief due to the transient nature of these animals.

Potential impacts to fishery resources are unlikely to result in changes in distribution and abundance of fish species outside the produced water mixing zone.



Figure 12: Predicted extent of the produced water dilution at a maximum discharge rate of 20,000 bbi/day.

## How will Santos manage impacts?

Activity discharges are to be managed through the application of Santos' Chemical Selection Process, designed so that environmentally acceptable process chemicals (which are likely to be mixed with produced water discharge) are used. Additives have been selected and optimised for biodegradability as well as low aquatic toxicity and bioaccumulation potential.

The produced water discharge will be continuously monitored for oil-inwater content and any reading over the limit of 30 mg/l over any 24hour period, will be diverted to a dedicated storage tank and returned to the produced water treatment system until suitable for discharge. Strict protocols will be in place for taking regular water samples and perform laboratory testing to ensure the produced water is within acceptable levels before disposing into the marine environment. The Production Operations Environment Plan will include a produced water daptive management plan that prescribes a water quality monitoring regime which enables the detection of potential impacts of produced water discharge on the marine environment and if remedial actions are necessary to retain the discharge within acceptable limits.



## OPERATIONAL DISCHARGES

Operational discharges associated with the activities may cause localised impacts to water quality in the direction of the prevailing current. The environment that may be affected by operational discharges will likely be contained within the OAs. Water quality conditions will return to normal within minutes to hours once discharging stops.

### FPSO facility and subsea system discharges

Operational discharges from the FPSO in OA1 will occur each day resulting in localised changes to water quality. Discharges of warm cooling water will include low concentrations of chiorine which break down quickly in the environment and is non-toxic at low concentrations. Minor discharges of water based hydraulic fluid used in the subsea system are classified by the offshore chemical notification scheme as being environmentally acceptable.

### Vessel discharges

The types of anticipated discharges in OA1 and OA2 are typical of most offshore commercial vessels and include deck runoff, treated sewage, grey water, machinery cooling water, blige water (treated via the oily water system), ballast water, macerated food scraps and brine (from water making). These discharges will be small in volume and released into surface waters.

### What Impacts are expected?

Sensitive receptors that may be impacted include plankton, fish, seabirds, marine turtles and mammals. Impacts to water quality will be localised and temporary occurring only during discharge.

Some fish and oceanic seabirds may be attracted to the FPSO by the discharge of food scraps. However, given the small quantities, intermittent nature of disposal and switt currents, any attraction is likely to be minor and is not anticipated to result in adverse impacts at an ecosystem or population level. Given the controls in place to manage the FPSO discharges in OAI in accordance with regulatory requirements, impacts to commercial fish species are not predicted.

Operational discharges in OA1 are predicted to quickly dilute and disperse in the offshore environment. Water quality changes will be localised and will occur only when the discharges occur. Given the temporary nature of activities within OA2 (limited to vessel based IMMR) and the relatively deep offshore environment with significant current and tidal action, impacts to water quality will be localised and will occur only for the duration of the discharge.

### How will Santos manage impacts?

Vessel discharges are to be managed to acceptable levels as regulated by maritime laws and conventions (e.g. management of sewage treatment systems and oily water systems), such as MARPOL and relevant Marine Orders. Santos also intends to implement management measures including waste management procedures and chemical management and selection procedures.

Santos procedures require that all operational chemicals used on the FPSO and chemicals potentially discharged to sea are risk assessed. Santos also implements general chemical management procedures to reduce the risk of accidental discharges.







## DROPPED OBJECTS

## How could dropped objects occur?

There is the potential for objects to be accidentally released to the marine environment from support vessels during steady state operations in OA1 or during IMMR activities in OA2. Dropped objects may be non-hazardous soild waste (e.g. paper, packaging materials), hazardous waste (e.g. batteries, aerosol cans etc.). A dropped object event could result from overfilling waste containers, unsecured objects during lifting operations or failed sea fastening.

### What environmental impacts could occur?

Objects that float could potentially move beyond the OAs. All non-buoyant objects are expected to sink to the seabed and remain within the OAs. This could cause localised and short-term damage to the seabed.

Marine debris (including plastics and microplastics) is listed as a potential threat to several marine fauna species. Depending on debris size of the dropped object, there is potential for entanglement or ingestion by marine fauna, including turtles and vertebrate wildlife, which could result in injury or death. However, given the limited quantities that might be dropped, impacts to fauna would be limited.

Considering the low frequency of such an unplanned event, even in a worst-case release of a solid object, impacts to fauna would be very localised and limited to individuals, and are not expected to result in impacts to the local population.

## How will Santos manage the risk?

Santos has numerous control measures to prevent dropped objects, and to mitigate consequence of impacts of an event does occur. These measures include:

- safety standards and procedures to reduce the likelihood of tools and other equipment being dropped during lifting operations
- waste management procedures to reduce the likelihood of windblown waste entering the marine environment
- Implementation of chemical selection processes and the international Maritime Dangerous Goods Code to limit the environmental impact of chemicals If lost overboard
- dropped objects, regardless of size, must be reported and attempts made to recover the object according to safety and environment criteria.

These control measures are designed to comply with maritime legislation. In addition, these control measures are consistent with applicable actions described in the relevant fauna recovery plans and conservation advice, reducing the residual risk to tow.



## INVASIVE MARINE SPECIES

### What are IMS?

Invasive marine species (IMS) are marine flora and fauna that have been introduced into a region that is beyond their natural range but have the ability to survive, and possibly thrive. The majority of climatically compatible IMS to northern Australia are found in south-east Asian countries.

### How might IMS be introduced?

Some IMS pose a significant risk to environmental values, biodiversity, ecosystem health, human health, fisheries, aquaculture, shipping, ports and tourism. The risk of introducing IMS is common for all maritime activities. The introduction of IMS may occur due to the following:

- biofouling on FPSO and vessels, external/Internal niches (such as sea chests and sea water systems) and routinely submerged equipment
- discharge of FPSO bailast water when the FPSO first transits from the
- International shipyard to the Barossa field

  discharge of high-risk ballast water where vessels have transited from
  international destinations.
- What environmental impacts could occur?

If successfully established, IMS can:

- outcompete native species for food or space
- prey on native species
- Impact fisheries or aquaculture
   Impact on human health through released toxins
- reduce coastal aesthetics
- cause damage to marine and industrial equipment and infrastructure.

The above impacts can result in flow-on detrimental effects to marine parks, tourism and recreation.

The ability of Invasive marine species to colonise a habitat depends on several environmental conditions. For example, highly disturbed environments (such as marinas) or shallower areas are more susceptible to colonisation than open-water environments (OA2 is 33 metres deep at its shallowest point and not considered sufficiently shallow to be conducive for IMS colonisation). OA1 provides an unfavourable habitat for IMS due to water depth (over 200 metres) and the long distance to the coast. These conditions limit light availability and have low habitat blodiversity with sparse epiblota, therefore, it is highly unlikely that IMS would be able to survive or colonise in OA1.

### How will Santos manage the risk?

The pathways and vessel mitigation measures for IMS Introduction are well established. The offfake tankers used for condensate export and specialised IMMR vessel(s) (If required) are sourced internationally, whilst the regular support vessels to and from the FPSO are sourced domestically. Vessels contracted to Santos, and vessel ballast, are to be managed according to control measures that comply with maritime regulations, industry practices, and the Blosecurity Act 2015. The FPSO and support vessels will also have ballast water management, vessel biofouling management and anti-fouling systems in place. With these control measures adopted, the residual risk of Introducing IMS is assessed as low and reduced to as low as reasonably practicable.

The initial mobilisation of the FPSO out of Singapore to the Barossa gas field will be managed under a quarantine management plan including arrangements for invasive species, biofouling and ballast water exchange



## INTERACTION WITH MARINE FAUNA

### How could interactions with marine fauna occur?

During the Production Operations Activity, approximately two vessels per week will travel between the Barossa field and Darwin servicing the FPSO, which is a minor increase relative to the existing levels of regional marine vessel traffic.

The highest potential for interactions with marine fauna, including potential accidental strike or collision resulting in injury or mortality, will be during IMMR vessel operations in OA2 where there is higher likelihood of marine fauna presence. In OA1, where marine fauna presence is of a lower likelihood, the FPSO will remain stationary once on location in position and support vessel movements within the operational area are limited and slow-moving, hence marine fauna interactions are not anticipated and are expected to be minimal.

Marine fauna such as marine mammals (such as whales and dolphins), marine turtles and whale sharks that swim at or near the water surface are most at risk from vessel collisions. Some of these species are threatened, and some marine fauna may have cultural significance.

Marine mammals (such as whales and dolphins) and whale sharks may transit through the OAs but are expected to be in low numbers in OA1 (Figure 10). Considering the relatively slow vessel speeds, short duration of activities, and the mobility of these species, it is unlikely that activity vessels will adversely interact with any individuals.

### How will Santos manage the risk?

The likelihood of marine fauna interaction resulting in injury or death is considered unlikely given the proposed implementation of the following control measures:

- Santos' Protected Marine Fauna Interaction and Sighting Procedure, which aligns with the Environment Protection and Biodiversity Conservation Regulations 2000. This procedure limits marine fauna approach distances and speed, allowing marine fauna to be avoided or to move away.
- Operational area vessel speed restrictions

The control measures are designed to align with management actions outlined in government-published fauna recovery plans and conservation advice. The risk of interactions with marine fauna is assessed as very low and reduced to as low as reasonably practicable and acceptable levels. The risk is no higher than for any other regional maritime activity.



## NON-HYDROCARBON LIQUID RELEASE

## How could non-hydrocarbon liquids be released?

Non-hydrocarbon liquids including miscellaneous chemicals for use during the Production Operations Activity and waste by-products are transferred to and from supply vessels to the FPSO in OA1. Examples of non-hydrocarbon liquids include chemicals used in the production process, domestic products used in the living quarters for cleaning and general maintenance products such as greases and paints.

An accidental release of non-hydrocarbon liquids into the marine environment has the potential to occur from:

- transferring, storing or using bulk products (e.g. production chemicals)
- mechanical failure of equipment, such as a tank or pipework failure
  handling and storage spills and leaks due to insufficient fastening or inadequate bunding
- floating hose failure or rupture, coupling failure or tank overfilling
- lifting and incorrect handling (e.g. dropped objects damaging storage containers)
- firefighting foam during an emergency response incident.

### What environmental impacts could occur?

A release of non-hydrocarbon ilquids may result in impacts to water quality and any sensitive environmental receptors.

The maximum volume of non-hydrocarbon liquids that could be released during routine operations is likely to be small and limited to the volume of individual storage containers. Individual containers stored on the FPSO include process chemicals and lube oil storage tanks (approximately 4.5 m<sup>6</sup>).

If the spill is not contained on deck, a release to the marine environment would likely disperse rapidly, with one in 1,000 dilution usually occurring within 30 minutes.

The environment that may be affected for non-hydrocarbon liquids releases resulting in a decrease in water quality is likely to be restricted to the immediate vicinity of the FPSO or support vessel and contained within the OAs.

Potential receptors include the physical environment (e.g. water and sediment quality, benthic habitats), threatened, migratory or local fauna (e.g. marine mammals, marine reptiles, sharks and rays, other flsh, and birds) and socioeconomic features of the environment (including cultural features).

### How will Santos manage the risk?

Santos has a suite of procedures to manage the selection, storage, handling and clean-up of non-hydrocarbon liquids releases. Vessels also have spill response plans. All chemicals are reviewed and accepted for use, and any chemical that might be discharged to the environment is assessed under the Santos chemical selection procedure to ensure environmental acceptability. These procedures will assist to minimise the likelihood of non-hydrocarbon liquid spills, and subsequent environmental consequences should they occur.

The control measures proposed to be adopted are designed to be consistent with maritime and petroleum industry standards and appropriate to manage the residual risks to as low as reasonably practicable and acceptable levels.



## MINOR LIQUID HYRDOCARBON RELEASES

## How could a minor liquid hydrocarbon release occur?

Minor releases refer to relatively small volumes of hydrocarbons from storage containers, transfer equipment and pipework on the FPSO or support vessels, that enters the marine environment. Typically, such spills occur as a result of human error during tank filling or storage container transfers. Most of these types of release occur within bunded deck areas, and are less than 1m3, however it remains possible for such spills to enter the marine environment.

### What environmental impacts could occur?

A localised decrease in water quality may occur, however due to the relatively small volumes impacts are expected to be short term as the hydrocarbon would rapidly dilute and dissolve into the ocean. Marine fauna may transit through the OAs and come into contact with the release. However it is expected impacts to fauna would be short term and result in behavioural changes, as they move away from the area where the spill occurred.

### How will Santos manage the risk?

A sulte of procedures will be in place to manage the handling and transfer of hydrocarbons on both support vessels and the FPSO. Response procedures such as stopping the source of the of the release and cleaning it up on deck to prevent it entering the ocean will be in place to manage minor releases should they occur.

The control measures proposed to be adopted are designed to be consistent with maritime and petroleum industry standards and appropriate to manage the residual risks to as low as reasonably practicable and acceptable levels.



## LARGER HYDROCARBON RELEASES

Larger volumes of hydrocarbons may accidentally be released during production operations. These include accidental splits from support vessels or the FPSO, as well as from subsea equipment (e.g. wells and flowlines). A range of different types of hydrocarbons that may be accidentally released are discussed below.

### MARINE DIESEL OIL OR MARINE GAS OIL

## How could a marine diesel or gas release occur?

Marine vessel fuels (marine diesel oil or marine gas oil) could be released to the environment if there is a collision event between two vessels. An accidental collision could occur due to factors such as human error, poor navigation, vessel equipment failure or poor weather. If a marine vessel collided with the FPSO, the vessel or FPSO huil may rupture and release fuel to the marine environment.

If a vessel fuel tank is ruptured a fuel called marine diesel could be released. The FPSO uses a lighter fuel for some of its power requirements called marine gas oil which could be released if a fuel tank is ruptured.

Although the risk is higher in OA1 than OA2, it should be noted that it is considered unlikely that a vessel collision would occur that would result in releasing fuel to the environment. A sequence of events would need to occur for a vessel collision to escalate to a large volume of fuel released to the environment, including:

- 1. the vessel must be involved in a collision
- 2. collision must occur with enough force to rupture a fuel tank
- rupture must be of such a nature that the fuel can be released into the environment.

## What environmental impacts could occur?

Marine diesel oil and marine gas oil fuels are typically characterised by a high percentage of volatile components (typically >95%), which will evaporate when on the sea surface over several days. A small fraction (typically <5%) or persistent hydrocarbons remains, which will not evaporate, and will decay over time. The heavier components of the fuels tend to become entrained in the upper water column as droplets in the presence of waves but can refloat to the surface if wave energies abate. Both marine diesel oil and marine gas oil fuels are expected to weather quickly through evaporation and dispersion and are unlikely to persist in the environment for a significant period.

Such releases will cause a decline in water quality and may cause chemical (e.g. toxicity) and physical impacts to marine species (e.g. Ingestion of hydrocarbons). The severity of the impact depends on the magnitude of the release (i.e. extent, duration) and sensitivity of the receptor, however, may include impacts to the physical environment, threatened or migratory marine fauna, protected and significant areas and socioeconomic receptors (fisheries, tourism, recreation, cultural features and other oil and gas operators).

### How will Santos manage the risk?

The FPSO in OA1 is fitted with a collision avoidance radar so it appears on the display of the triggering radars, providing range, bearing and identification information, alerting vessels to its presence. Santos has also designed the FPSO hull to be double-sided and double-bottomed, which provides two physical barriers between the fuel tanks and the marine environment for side impact, reducing the likelihood of fuel release in the event of a collision.

A petroleum safety zone (PSZ) will be established alerting other marine users to the presence of the FPSO in OA1 which includes precautions for marine activities (e.g. reduced speed limits, communication protocols and automatic identification systems to aid in their detection at sea). Third party vessels are not permitted to enter a PSZ, thereby reducing the likelihood of other interactions with the FPSO and support vessels. In OA2 during IMMR activities a similar exclusion zone will also be established restricting access to other marine users.

The Production Operations Activity facilities in OA1 & OA2 will be included on navigational charts making other vessels aware of the presence of Barossa facilities. Santos will also provide maritime notifications to relevant departments to ensure marine users are informed of vessel movements.

Santos has also developed response plans which will detail the actions to take to control the release and manage the cleanup activities in the event of a release.







## CONDENSATE

### How could a condensate release occur?

Barossa condensate has the potential to be released to the marine environment under several scenarios. Of those scenarios, three worst-case events are summarised below.

- In the event of a vessel collision (e.g. those described above for marine diesel oil or marine gas oil) which ruptures the FPSO condensate storage tank.
- in the event of an impact to, or failure of the subsea hydrocarbon containing equipment.

 In the event of an impact to, or failure of multiple production well barriers. Other scenarios exist that may result in other smaller condensate releases to the marine environment. All scenarios are very low probability of occurring.

### What environmental Impacts could occur?

Condensate, being a lighter hydrocarbon behaves in a similar fashion to marine diesel when released to the marine environment. The fate of the condensate will depend greatly on the proportion on the surface, which will be transported by prevailing currents and wind and can evaporate readily. Condensate that entrains or dissolves in the water column will be transported by prevailing current and, hence, will follow a different path.

As with the marine diesel oil, the heavier components contained in the condensate will have a strong tendency to physically entrain into the upper water column but can re-float to the surface if these energies abate.

Such releases will impact the marine environment much in the same way as marine diesel oil and marine gas oils, described above. However, in the event of a subsea condensate release, more entrainment of hydrocarbons in the water column could occur, rather than being present on the sea-surface.

## How will Santos manage the risk?

In the unlikely event a vessel collision occurs which ruptures an FPSO condensate tank, Santos will manage the risk in accordance with the accepted Production Operations OII Pollution Emergency Plan.

In the unlikely event of a subsea release in the OA1 area from subsea flowlines or wells, Santos has a range of operating procedures and plans to ensure that the integrity of the subsea infrastructure is maintained and well barriers are in place. Santos will submit a well operations management plan (WOMP) to NOPSEMA that will contain the full details of systems in place to ensure well design and integrity is managed for the well ilfecycle. All production wells must be in compliance with the NOPSEMA accepted WOMP at all times. Hydrocarbon containing subsea infrastructure is also within a petroleum safety zone (PSZ), which third party vessels are not permitted to enter, subsequently reducing any interaction with this infrastructure.

Santos is developing response plans which will detail the actions to take to control a release and manage cleanup activities in the unlikely event of a release.



## HEAVY FUEL OIL

## How could a heavy fuel oll release occur?

Heavy fuel oil is only used as fuel for offtake tankers who enter the Barossa field periodically (approximately once every three months) to load condensate from the FPSO. The only scenario that could lead to a release of heavy fuel oil is in the unlikely event of a vessel collision (described above for marine diesel oil or marine gas oil), where the offtake tanker huil and heavy fuel oil tak is ruptured.

### What environmental impacts could occur?

Heavy fuel oil is heavier and more persistent than marine diesel oils, marine gas oils and condensates. The fuel is often characterised by a very high density and a high dynamic viscosity, which does not evaporate as quickly as other lighter fuels. As the fuel has a high residual component, a portion is expected to become semisolid and can persist in the marine environment for extended periods.

Such releases will cause a decline in water quality and may cause chemical (e.g. toxicity) and physical impacts to marine species (e.g. ingestion of hydrocarbons, physical coating). The severity of the impact depends on the magnitude of the release (i.e. extent, duration) and sensitivity of the receptor, however, may include those to the physical environment, threatened or migratory marine fauna, protected and significant areas and socioeconomic receptors (fisheries, tourism, recreation, cultural features and other oil and gas operators). Given the persistent and sticky nature of heavy fuel oil, there is a higher risk of coating of the physical environment (e.g. shorelines) and marine fauna compared to the lighter fuels such as marine diesel oil and marine gas oil.

### How will Santos manage the risk?

Offtake tankers are third-party operated vessels. They are vetted following Santos' marine assurance procedure and international guidelines before acceptance for condensate offtake operations at the Barossa field. The use of tankers with double hulls and fully segregated ballast tanks is not only a requirement of the vetting process; it is a MARPOL requirement that is monitored by way of regular statutory inspections.

All offtake loading events are planned in advance, occur within a petroleum safety zone (PSZ), and are performed under strict operational procedures.

Offtake tankers are third-party operated vessels. They are vetted following Santos' marine assurance procedure and international guidelines before acceptance for condensate offtake operations at the Barossa field. The use of tankers with double hulls and fully segregated ballast tanks is not only a requirement of the vetting process; it is a MARPOL requirement that is monitored by way of regular statutory inspections.

All offtake loading events are planned in advance, occur within a petroleum safety zone (PSZ), and are performed under strict operational procedures.



## CONTINGENCY SPILL RESPONSE OPERATIONS

in the event of a hydrocarbon spill, response strategies will be imple mented to reduce environmental impacts to as low as reasonably practicable. The selection of strategies will be undertaken using the Net Environmental Benefits Assessment (NEBA) process. Spill response will be under the direction of the relevant control agency, as defined in the Production Operations Oil Pollution Emergency Plan (OPEP), which may be Santos, another agency or both. In all Instances, Santos will undertake a 'first-strike' spill response and will act as the control agency until the designated control agency assumes control. The response strategies considered to be appropriate for the worst-case splil scenarios identified for the activity are detailed in the OPEP and comprise

- source control (blowout preventer, relief well)
- monitor and evaluate
- mechanical dispersion
- shoreline protection and clean up · olied wildlife response
- scientific monitoring
- waste management.

Response strategies are intended to reduce the environmental consequences of a hydrocarbon spill, but poorly planned and coordinated response activities can result in a lack of, or inadequate, information being available, upon which poor decisions can be made, exacerbating or causing further environmental harm.

### What Impacts are expected?

Splil response operations may be required at any location within the EMBA. Potential environmental impacts include:

- Noise and light emissions generated by response vessels and equipment which may impact marine fauna, such as fish (including commercial species). marine reptiles and marine mammals.
- · Atmospheric emissions generated from response equipment and vessels are expected to be localised and are not considered to create emissions on a scale where noticeable impacts would be predicted.
- · Operational discharges and waste generated from response equipment and vessels are expected to be consistent with those of normal commercial ve operations and may create a localised and temporary reduction in marine rcial vessel water quality. Cleaning of hydrocarbon-contaminated equipment, vehicles and vessels has the potential to spread hydrocarbons from contaminated areas to areas not impacted by a spill. Sewage and other waste will be generated from offshore activities at temporary staging/mooring areas, which may include toilet and washing facilities. These wastes have the potential to impact water quality, impact habitats, and reduce the aesthetic value of the environment, which may be within protected areas.
- Physical presence and disturbance operating vessels during spill response operations has the potential to disturb the physical environment and marine habitats and fauna (e.g. vessel strike, behavioural changes) or cause disruption to other marine users, coastal areas, townships and commercial fishing.



### How will Santos manage the risk?

Santos will rely primarily on the Implementation of the Production Operations OPEP to manage the potential impacts associated with a spill response event. Other control measures that would be implemented include:

- procedure for interacting with marine fauna
   chemical selection process
- minimum lighting to meet maritime safety and navigation requirements
- air pollution prevention certification
   sewage and oily water treatment systems on vessels
- notify agreed stakeholders.

The implementation of spill response activities to reduce the potential impacts from a spill are required by legislation. The spill response options selected will be demonstrated to show a net environmental benefit, are standard industry practice and are consistent with relevant standards and guidelines, including the National Plan for Maritime Environmental Emergencies. The controls proposed are intended to reduce the consequences of the potential impacts to minor and as low as reasonably practicable and an acceptable level.



## SUMMARY OF SANTOS' RISK MANAGEMENT STRATEGY

## Santos has a management system that includes specific measures, to be used for the duration of the Production Operations Activity, which seek to confirm that:

- environmental impacts and risks continue to be identified for the duration of the activity and are reduced to as low as
  reasonably practicable and acceptable levels
- control measures are effective in reducing environmental impacts and risks to as low as reasonably practicable and acceptable levels
- environmental performance outcomes and standards set out in the EP and OEMP are being met
- there will be ongoing appropriate consultation with relevant authorities and other relevant interested persons or organisations
- the roles, accountabilities and responsibilities are defined and understood
- workforce training is completed and competencies assured
- emergency preparedness and response arrangements are in place
- incident reporting, investigation and follow-up is monitored
- · audits, inspections, reporting and notifications and document management are appropriately undertaken

## **APPROVALS PROCESS**

Production Operations Activities detailed in this booklet require a number of regulatory approvals. Primary environmental approvals required for Production Operations Activities are outlined below:

- An Offshore Project Proposal (the Barossa Offshore Project Proposal (OPP)) was developed for the Commonwealth waters component the Barossa Project and was accepted by NOPSEMA in March 2018. The Barossa OPP, at the time of submission, excluded approximately 23 km of GEP in Commonwealth waters which is subject to a separate EPBC Act approval process (refer balow).
- A referral under the EPBC Act, covering the Installation, operation and decommissioning of the remaining approximately 25 km of GEP in Commonwealth waters and the 100 km section of GEP in NT waters (inclusive of the 8.26 km in NT coastal waters) was submitted to the Department of Climate Change, Energy, the Environment and Water (DCCEEW) for assessment. The activity (referred to as the Darwin Pipeline Duplication Project) was determined to be a 'controlled action' under the EPBC Act and is currently being assessed on preliminary documentation.
- A referral under the NT Environment Protection Act 2019 (EP Act) for the construction, operation and decommissioning of the 100 km section of GEP in NT waters (part of the Darwin Pipeline Duplication Project) was submitted to the NT Environment Protection Authority (EPA) and was subsequently assessed by way of Supplementary Environmental Report. On 22 December 2023, the NT Minister for Environment, Climate Change and Water Security approved the action the subject of the referral, on the recommendation of the NT EPA.

In addition to the primary environmental approvals outlined above, activity-specific Environmental Plans (EPs) meeting the requirements of the Offshore Petroleum and Greenhouse Gas Storage (Environment) Regulations 2023 (Cth) (OPGGS Environment Regulations) are required. For Production Operations activities, the OPGGS Environment Regulations apply to the activities within OA1 in Commonwealth waters and OA2, spanning both Commonwealth waters (285 km) and coastal waters of the NT (8.26 km). The OPGGS Environment Regulations set out that an EP must (among other things):

- · comprehensively describe the activity to be carried out under the EP
- describe the environment that may be affected by the activity, including the values and sensitivities of that environment
- detail and evaluate the environmental impacts and risks for the relevant activity
- demonstrate that the impacts and risks of the activity will be reduced to as low as reasonably practicable and an acceptable level (and detail the control measures to be used to achieve this)
- demonstrate that Santos has consulted, in accordance with regulatory requirements, with each relevant person, including those whose functions, interests or activities may be affected by the activities to be carried out under the EP
- demonstrate that the measures ()f any) that Santos has adopted, or proposes to adopt, because of the consultations are appropriate

Santos is currently preparing the Production Operations EP for submission to NOPSEMA, covering Commonwealth waters Production Operations activities in OA1 and OA2.

Santos is also preparing an Operations Environmental Management Plan (OEMP) to cover the operation of the GEP in NT waters for submission to DIT. The OEMP will also cover the operation of the 8.26km GEP in NT coastal waters, under the PSL Act and OPGGS Environment Regulations and operation of the remaining -92km GEP covered under the Energy Pipelines Act. In order to meet its proposed schedule for the Barossa Gas Project, Santos is aiming to submit the Productions Operation EP to NOPSEMA and the OEMP to DITT in 2024 and, subject to regulatory acceptance, commence activities in 2025. The timeline for consultation has been developed by Santos to meet this objective, while still providing a reasonable period for meaningful consultation with relevant persons, having regard to Santos's regulatory obligations and to feedback from relevant persons.

## SEEKING INFORMATION AND WHAT'S NEXT

Santos is continuing its Barossa Gas Project consuitation efforts to further learn, understand and assess values and sensitivities of the environment that may be affected by our proposed activities, and potential environmental impacts and risks. There may be information Santos is not yet aware of but needs to properly understand to assess potential activity impacts and risks. Consultation may inform this. It may also inform what control measures are to be proposed to reduce environmental impacts and risks to as iow as reasonably practicable and to an acceptable level.

Santos is consulting on both the Production Operations EP (Commonwealth waters) and OEMP (NT waters) at the same time.

Scan this QR Code for more information on Barossa Production Operations Activity:



## YOUR INPUT IS IMPORTANT TO SANTOS:

In preparing an EP for submission to NOPSEMA, a titleholder must consult with each 'relevant person', including relevant Commonwealth, State and Northern Territory Departments or agencies and persons (or organisations) whose functions, interests or activities may be affected by the activity proposed to be carried out under an EP.

Relevant persons being consulted on EPs under the Offshore Petroleum and Greenhouse Gas Storage (Environment) Regulations 2023 (Cth) (OPGGS Environment Regulations) should note that they:

- are entitled to be given sufficient information to allow them to make an informed assessment of the possible consequences of the activity on their functions, interests or activities;
- are entitled to be allowed a reasonable period for the consultation; and
- may request particular information provided in consultation not be published.

If you request particular information not to be published, Santos will respect and abide by your request. Any information not to be published will be provided to NOPSEMA in a confidential report, separate from the published EP.

Your Input Is Important to Santos:

- so that we can understand the environmental values in the OAs and the environment that may be affected, the environmental impacts and risks associated with the activity, to inform development of the Productions Operations EP (Commonwealth waters) and OEMP (NT waters);
- to inform how consultation processes may need to be adapted for different relevant persons; and
- to ensure that we provide information to relevant persons in an appropriate and accessible manner.

If you think you may be a relevant person for the purposes of one of Santos' proposed activities, please contact Santos on: 1800 267 600 or email offshore.consultation@santos.com to seek to be included in consultations and to provide feedback on how you would like to be consulted (If a relevant person).

This can also be done using the form available by scanning the QR Code below:



Visit www.santos.com/barossa for more information on the Barossa Gas Project.

## NORTHERN TERRITORY WATERS

# GAS EXPORT PIPELINE OPERATIONS OVERVIEW FACTSHEET

## INTRODUCTION

Santos' NT Waters Gas Export Pipeline (GEP) Operations Environmental Management Plan (OEMP) covers operation of the Barossa GEP in NT waters, including the:

- 8.26km section of the GEP in NT coastal waters - covered by the Petroleum (Submerged Lands Act) 1981 (NT) (PSL Act); and
- ~92km section of the GEP in NT waters covered by the Energy Pipelines Act 1981 (NT) (Energy Pipelines Act)

The purpose of the Barossa GEP is to transport dry natural gas (not oil or condensate) from the Barossa gas field to the existing Darwin Liquified Natural Gas (DLNG) facility.

In December 2023, the NT Government approved the construction of the GEP in NT waters under the NT Environment Protection Act 2019 (NT EP Act) following assessment by the NT Environment Protection Authority (EPA). To obtain authorisation for operation of the GEP in NT waters, Santos will now submit the OEMP to the NT Department of Industry, Tourism and Trade (DITT) for assessment and approval under the PSL Act and the Energy Pipelines Act.

The estimated duration of Barossa Production Operations is 25 years. This factsheet provides a summary of the activity, environmental values and sensitivities relevant to operation of the GEP in NT waters and the impacts, risks and control measures associated with the activity.

The operation of the GEP in Commonwealth waters will be covered by an Environment Plan (EP) requiring acceptance by the National Offshore Petroleum Safety and Environmental Management Authority (NOPSEMA). Information on the Commonwealth waters Barossa Production Operations EP can be found in a separate information booklet on the Santos website (Production-Operations-Information-Booklet.pdf (santos.com)



## **KEY ACTIVITY** Inspection, Monitoring, Maintenance & Repair (IMMR)

Inspection, maintenance, monitoring and repair (IMMR) activities are the key activities related to operation of the Barossa GEP.

IMMR activities are performed to assure the ongoing integrity of the GEP and safe and reliable operations. The activities will occur within an Operational Area approximately 500m either side of the pipeline. The Operational Area and location of the GEP in NT waters are shown in Figure 1.

These activities will be infrequent and of relatively short duration (approximately three weeks), occurring approximately every three years, consistent with industry standards. Inspections during the first year of operations will confirm post-construction integrity of the GEP and inform the frequency of subsequent planned inspections.

Additional inspections may be performed following extreme weather events, seismic activity, or unplanned third-party interactions. Maintenance and repair activities may be undertaken if required.

Inspection activities typically include general visual inspection and close visual inspection. This is conducted using remotely operated vehicles (ROVs) from one or more vessels that have dynamic positioning capabilities. Sidescan sonar or multibeam echo sounder may be used as a screening tool to inform further targeted inspections by ROV.

Maintenance activities may include correction of free spans by targeted placement of sand or grout bags. Repair activities may include repair of damaged sections of pipe.



**EXISTING ENVIRONMENT** 

Darwin Harbour is a large, drowned river system approximately 500km2 in extent. It is comprised of three arms (East Arm, West Arm, and Middle Arm), which converge into a single unit before opening to the ocean and into Beagle Gulf in the north. Freshwater inflow from the Elizabeth River into the East Arm and the Blackmore and Darwin Rivers into the Middle Arm generally occurs between January and April creating more estuarine conditions.

The Darwin region supports several benthic habitats including mangroves, coral, seagrass and macroalgae. The bathymetry of the Operational Area in NT waters has been thoroughly investigated and is well understood. Recent surveys have shown that the seabed along the GEP route in NT waters is generally flat and featureless and typically less than 30m in depth.

Port Darwin's main channel is approximately 1,525m wide and 15 - 25m deep, with a maximum recorded depth of 36m (Lowest Astronomical Tide). The channel is generally deeper on the eastern side of the Harbour, while the western side is broader with shallower areas with intertidal flats and shoal being more extensive.

The benthic habitats along the pipeline route were found to be silty shelly sand habitat, with burrows and polychaete worm tubes. Biota commonly associated with this habitat type is sparse, including hydroids, soft corals (gorgonians, Junceella and Alcyoniidae), sea stars and sponges.

Dolphin species are the most recorded marine mammal in Darwin Harbour, with the Australian snubfin (Orcaella

heinsohni), Indo-Pacific humpback (Sousa chinensis) and Indo-Pacific spotted bottlenose (*Tursiops aduncus*) known to occur. Dugongs are also present, as monitoring estimates that approximately 180 to 300 individuals inhabit the Darwin Region.

There are six protected species of marine turtle known to occur in NT waters; of these only green (Chelonia mydas), hawksbill (Eretmochelys imbricata) and flatback (Natator depressus) turtles are known to occur in Darwin Harbour regularly. Olive ridley (Lepidochelys olivacea) and loggerhead (Caretta caretta) turtles are known to occasionally occur in Darwin Harbour, and leatherback turtles (Dermochelys coriacea) are found offshore as they are an oceanic species. The closest nesting sites are located at Cox Peninsula and Casuarina Beach, although these are not considered significant nesting areas.

Darwin Harbour supports an abundance of fish species across an array of habitats. There is a diverse range from small site-specific species such as gobies, cardinals, and pipefish to larger species of recreational and commercial importance such as mackerel, trevallies and barramundi. Barramundi account for 26% of all recreational fishing in the Northern Territory, making it the most targeted species.

Three protected sawfish species have been recorded within the Darwin Harbour region—the dwarf sawfish (Pristis clavata), freshwater sawfish (Pristis pristis or Prisitis microdon) and green sawfish (Pristis zijsron). However, they are unlikely to be encountered in the Operational Area.

## **Threatened Fauna**

Fauna that are listed as threatened or migratory species under the Environment Protection and Biodiversity Conservation Act 1999 (Cth) and which have been identified as potentially relevant to the Activity are listed in the table below.

Marine Reptiles	Marine Mammals	Birds
Flatback turtle	Australian snubfin dolphin	Asian dowitcher
Olive ridley turtle	Dugong	Common sandpiper
Green turtle	Indo-Pacific humpback dolphin	Grey plover
Hawksbill turtle	Spotted bottlenose dolphin	Oriental plover
Leatherback turtle		Osprey
Loggerhead turtle		
Salt-water crocodile		

## Socio-Economic

Socio-economic values that are potentially relevant to the Activity are summarised in the table below.

Activity/Value	Description	
NT commercial fisheries	Offshore Net and Line Fishery, the Spanish and Aquarium Fishery	
Recreational fishing	The Darwin Harbour/Surrounds fishing zo	
Traditional fishing	Traditional Australian Indigenous fishing in	
Shipping	Darwin Port - cruises, naval, livestock, dry	
Tourism	Fishing, boating, scuba-diving, sailing, wat	
Defence	Darwin air weapons range, North Australia	
Other industries	Telecommunications (e.g. Telstra cables)	
Maritime heritage	Numerous shipwrecks and sunken aircraft either the bombing of Darwin in 1942 or C	
First Nations heritage	land and sea in the region. Cultural, spiritu	





h Mackerel Fishery, the Coastal Line Fishery, and the Demersal Fishery

- one supports 63% of total Darwin fishing
- in NT waters occurs within inshore tidal waters
- bulk ore, oil and gas, general cargo/containers
- ter-skiing, and beach use
- an exercise area, Defence training area
- are located in Darwin Harbour. Most wrecks are associated with Cvclone Tracv in 1974.
- pied by the Larrakia people, they maintain an innate connection to the ual and heritage sites of significance are located throughout the region where traditional fishing and hunting continue to be practiced.



The key environmental impacts and risks from planned activities and unplanned events associated with GEP Operations are detailed below. As the key activities associated with GEP Operations, IMMR vessel activities are expected to occur approximately once every three years for approximately three weeks.

## **PLANNED ACTIVITIES**

Planned Activity	Impact	How will Santos manage impacts	Unplanned Event
LIGHT EMISSIONS	Behavioural impact to marine life (e.g. attraction and behavioural changes)	Lighting is to be limited on the activity vessels to what is required for safe operations and navigation. Lighting will be compliant with maritime regulations (similar to other commercial vessels operating in the region).	DROPPED OBJECT
NOISE EMISSIONS	Behavioural impact to marine life (e.g. avoidance)	Activity vessels are required to comply with Santos' Protected Marine Fauna Interaction and Sighting Procedure to comply with regulatory requirements for managing fauna noise impacts. Marine assurance standards and planned vessel maintenance will minimise noise generated from vessels by ensuring contracted vessels are operated, maintained, and crewed in accordance with industry standards and regulatory requirements.	MARINE FAUNA INTERACTION
SEABED DISTURBANCE	Behavioural impact to marine life (e.g. avoidance)	Activity vessels are required to comply with Santos' Protected Marine Fauna Interaction and Sighting Procedure to comply with regulatory requirements for managing fauna noise impacts. Marine assurance standards and planned vessel maintenance will minimise noise generated from vessels by ensuring contracted vessels are operated, maintained, and crewed in accordance with industry standards and regulatory requirements.	INVASIVE MARINE SPECIES (IMS)
VESSEL DISCHARGES	Sensitive receptors that may be impacted include plankton, fish, seabirds, marine turtles, and mammals. Impacts to water quality will be localised and temporary occurring only during discharge.	Vessel discharges are to be managed to acceptable levels, as regulated by applicable laws and conventions. Santos selects chemicals that are environmentally acceptable and limits their use to only what is needed. Vessels will have routine discharges such as small volumes of treated sewage, treated bilge water and macerated food scraps. Vessel discharges will be compliant with MARPOL requirements. ('MARPOL' is a reference to the International Convention for the Prevention of Pollution from Ships.)	NON-HYDROCARE LIQUID RELEASE
AIR EMISSIONS	Impact to air quality, contribution to national greenhouse gas levels, from IMMR vessel emissions	Santos proposes to adopt various control measures to manage vessel emissions, including vessel compliance with MARPOL requirements for low-sulphur fuel and air pollution prevention certifications.	MINOR LIQUID HYDROCARBON RELEASE

## **OTHER MARINE USERS**



Other marine users will be temporarily restricted from the area around vessels when performing IMMR activities. The GEP may also present a hazard to marine users due to the potential for snagging.

Santos will notify and communicate with other marine users using standard maritime notifications (e.g. Notice to Mariners) before, during and at the end of IMMR activities. Infrastructure locations will be marked on nautical charts. These proposed control measures are consistent with maritime regulations and industry practices.

# **UNPLANNED EVENTS**

# nplanned Event Impact **DROPPED OBJECTS**

Impacts to water quality, disturbance to seabed and marine life

## MARINE FAUNA INTERACTION



Disturbance to marine animals (e.g. fauna strike and behavioural changes such as avoidance)

## If established. IMS can:

- Outcompete native species for food or space • Prey on native species
- Impact fisheries
- Impact on human health through released toxins
- Cause damage to maritime equipment and infrastructure.

## **NON-HYDROCARBON** LIQUID RELEASE



Santos has a suite of procedures to manage the selection. An accidental release of nonstorage, handling, and clean-up of non-hydrocarbon liquids hydrocarbon liquids such as chemicals, releases. Vessels also have spill response plans. All chemicals may result in impacts to water quality are reviewed and accepted for use, and any chemical that and any sensitive environmental might be discharged to the environment is assessed under the receptors (such as fauna and habitat). Santos chemical selection procedure to ensure environmental acceptability.

### MINOR LIQUID **HYDROCARBON** RELEASE



A localised decrease in water quality may occur from minor hydrocarbon releases, such as, hydraulic fluids from vessel equipment. However due to the relatively small volumes, impacts are expected to be short term as the hydrocarbon would rapidly dilute and dissolve into the ocean.





## How will Santos manage impacts

С		

Procedures are in place to reduce the likelihood of tools and other equipment being dropped during lifting operations. Waste management procedures reduce the likelihood of windblown waste entering the marine environment. Dropped objects. regardless of size, will be reported and attempts to recover the object will occur, according to safety and environment criteria.

Santos' Protected Marine Fauna Interaction and Sighting Procedure limits marine fauna approach distances and speed, allowing marine fauna to be avoided or to move away. Vessel speed restrictions will also be in place in the Operational Area to reduce the likelihood of an unplanned interaction. It is unlikely that IMMR vessels will adversely interact with any individuals due to the slow vessel speeds, short duration/low frequency of activities, existing maritime traffic, and fauna mobility.

Vessels contracted to Santos are managed according to control measures that comply with maritime regulations, industry practices, and biosecurity legislation. Vessels will also have ballast water management, vessel biofouling management and anti-fouling systems in place.

A suite of procedures will be in place to manage the handling and storage of hydrocarbons on vessels. Response procedures such as stopping the source of the of the release and cleaning it up on deck to prevent it entering the ocean will be in place to manage minor releases should they occur.

PIPELINE

**GAS RELEASE** 

## Unplanned Event

## Impact

impacted.

### How will Santos manage impacts

Santos has proposed preventative and mitigation measures to manage the impacts and risks of an unplanned gas release from the GEP, such as

- Pipeline integrity management plan, monitoring procedures and emergency response procedures
- Procedures and standards for lifting equipment, IMMR procedures and contractor management standards
- Maritime notifications to ensure marine users are informed of a gas release event
- The GEP will be marked on nautical charts
- Emergency response procedures, pipeline depressurisation procedures (stop gas from flowing into the pipeline) and emergency pipeline repair plans will be implemented to minimise impacts in the event of a loss of containment from the Barossa GEP.

## MARINE DIESEL SPILL



Although highly unlikely, a spill from a collision between two vessels could rupture a fuel tank resulting in the release of vessel fuel to the sea. This would impact water quality and may cause chemical/physical impacts to marine species.

In the unlikely event of a gas release

from the gas export pipeline (GEP), gas

surface resulting in a visible bubble zone

at the sea surface and an associated gas

cloud before rapidly dispersing into the

atmosphere. Any marine fauna or marine

and future repairs of the pipeline may be

users in the vicinity of the gas release

would move vertically toward the sea

The risk of collision is reduced by managing interactions with marine users before and during the activity, with maritime notifications, automatic identification systems, navigational lighting, and exclusion zones in place. Operational procedures are designed to minimise refuelling incidents and spill response plans will be in place.

## SPILL RESPONSE OPERATIONS



If a spill occurs, response operations may be required at any location surrounding the Operational Area. Potential environmental impacts include those listed in the Planned Activities table. Santos will rely on its Oil Pollution Emergency Plan (OPEP) to manage the impacts from a spill response event. Control measures would include:

- · Procedure for interacting with marine fauna
- Chemical selection process
- Minimum lighting
- Air pollution prevention certification
- Sewage and oily water treatment systems on vessels
- Notification to agreed stakeholders



## **CONTACT US**

For further information or queries on the Production Operations activity, please contact Santos.

Phone 1800 267 600 or email offshore.consultation@santos.com

For more information about PO activity please scan this QR Code:



## **Updated email – additional risk**

# This email, or the content it contains, was sent to relevant entities following the identification of an additional risk.

Since our email to you of 11 March 2024 regarding the consultation Santos is undertaking for the Barossa Gas Project in relation to the proposed Production Operations Environment Plan (EP) and the Gas Export Pipeline Operations Environment Management Plan (OEMP), we advise that the Production Operations information booklet and Gas Export Pipeline Operation factsheet have been updated and that we have extended the consultation period until **21 May 2024**.

In providing this extension of time, we draw your attention to the updated information in the Production Operations information booklet and Gas Export Pipeline Operation factsheet. The information booklet and factsheet have each been updated to account for an additional risk associated with the proposed activity, namely a gas release in the unlikely event of an unplanned pipeline loss of containment. Here is a link to the <u>booklet</u> and a link to the <u>factsheet</u>. This risk, and the measures we propose to manage it, are summarised on page 31 of the updated information booklet and page 6 of the updated factsheet.

If you wish to provide any further input in light of this update, please call 1800 267 600 or email <u>offshore.consultation@santos.com</u> by the revised consultation closure date of **21 May 2024**. If we do not receive your input by this date, we infer that this means that you do not want Santos to consult with you further on the Productions Operations EP and OEMP.

If and when you provide your input, please let us know if you request particular information you provide during consultation not be published. If you make this request, the information will not be published as part of the plan, in accordance with relevant legislation. Sensitive information we need to give to the regulator to assess our plan will be provided in a separate report, rather than in the published plan. Santos will handle your information in accordance with our *Barossa Gas Project Consultation Privacy Policy*.

Regards,

Santos Offshore Consultation Team

Production operations information booklet – updated version for additional risk

INFORMATION BOOKLET BAROSSA PRODUCTION OPERATIONS ACTIVITY



## INTRODUCTION

The activities described in this booklet relate to the extraction, processing, and distribution of gas and condensate from the Barossa Field.

The Barossa Development facilities used in these activities consist of a Floating Production Storage and Offloading (FPSO) facility, subsea production wells, supporting in-field subsea infrastructure (Figure 1), a 285 km Gas Export Pipeline (GEP) in Commonwealth waters, and an 8.26 km section of the GEP in Northern Territory (NT) coastal waters; collectively termed in this booklet as "Production Operations Activity".

As part of obtaining authorisation for this activity, Santos is undertaking consultation for the following regulatory approvals:

- The Production Operations Environment Plan (EP) relating to the arrival and operations of the FPSO, operation of a subsea production system and supporting subsea infrastructure, and operation of a 285km section of the GEP located in Commonwealth waters where offshore petroleum activities are regulated under the Offshore Petroleum and Greenhouse Gay Storage Act 2006 (Cth) (OPGGS Act).
- The Operations Environmental Management Plan (OEMP) which includes the:
  - 8.26 km section of the GEP in NT coastal waters covered by the Petroleum (Submerged Lands Act) 1981 (NT) (PSL Act); and
  - -92km section of the GEP inshore of NT waters covered by the Energy Pipelines Act 1981 (N7) (Energy Pipelines Act).

The term 'GEP' refers to the Gas Export Pipeline through which Barossa gas will be transported from the Barossa field to Darwin LNG. However, the scope of the GEP covered in this booklet, is limited to the 8.26 km section of the GEP located in NT coastal waters.

The activities, environmental impacts, and risks for the GEP in NT waters (-92km) are broadly similar to those for the GEP in Commonwealth waters (described in this booklet). The activities, environmental impacts and risks specific to the GEP in NT waters, not covered in this booklet, will be provided in a separate factsheet.

The estimated life of the Barossa Development is 25 years, and the Production Operations EP and the OEMP will be reviewed every five years following initial regulator authorisation. This booklet provides a summary of the credible environmental impacts and risks associated with the first five years (also known as Barossa Phase 1) of the Production Operations Activity.



Figure 1: Schematic of the Barossa Field Subsea production system and infrastructure, and FPSO

## BAROSSA GAS PROJECT OVERVIEW

The Samos-operated Barossa Gas Project is an offshore gas and condensate project with the purpose of providing a new source of gas to the existing Darwin injuried natural gas (DLNG) facility at Workham Point in the NT. It is intended that natural gas and condensate would be extincted from the Barossa field, located in Commonwealth waters approximately 205 storneties offshore north-northwest from Darwin, initial processing would occur at the PPSO to separate the natural gas, water and condensates. The dry natural gas would be transported through the GEP for orshore processing at the DLNG facility. The condensate would be transferred from the FPSO facility to purpose-built tankers for international export.

DLNG facility at Wickham Point where Barossa gas will be sent for onehore processing.

## **ACTIVITY LOCATION**

The planned Production Operation Activity is confined to two key operational areas. These areas are simply termed Operational Area 1 (DA1) and Operational Area 2 (DA2) (Figure 2).

OA1: The Barossa field. This is the area in which the FPSO, subsea production system, and supporting subsea infrastructure will be used to process gas and condensate extracted from the Barossa wells. The area is confined to Commonwealth waters, approximately 285 km north-north-west of Darwin (the closest major populated centre), approximately 210 km north-west of the mainland NT coastline, and approximately 130 km north of the Tiwi Islands at the closest point (Seeguil Island).

OA2: The 285 km section of the GEP from OA1 to the Commonwealth waters/NT waters boundary; and the 8.26 km section of the GEP situated in NT coastal waters between the Commonwealth waters/NT coastal waters boundary and the Territorial Sea Baseline (TSB). Not included in this section is the remaining GEP in NT waters (-92 kms). Information on this is covered elsewhere:



Figure 2: Location of the Barossa Operational Area 1 (OA1) in Commonwealth waters (OPPGS Act) and Operational Area 2 (OA2) in Commonwealth waters and NT coastal waters (PSL Act).

## SUMMARY OF ACTIVITIES

The Barossa FPSO, which will be known as BW Opal, is a permanently moored vessel which is able to freely rotate with the strongest wind direction and remain connected to subsea facilities. The operational design life for all Barossa facilities is 25 years. Figures 3 and 4 depict key features of the FPSO and utilities/marine systems.

Gas and condensate extracted from the Barossa field will be processed on the FPSO to separate the natural gas and condensate (a light liquid hydrocarbon, straw coloured and flammable). The dry gas will be exported to the DLNG facility via the GEP, and the condensate will be transferred from the FPSO to offtake tankers for export approximately four to five times per year. The key activities proposed under the Production Operations EP are detailed below.

Most of the below activities are only applicable to OA1 as this is where the FPSO and subsea infrastructure is located, and the condensate offtakes will occur. The main activity in OA2 will be inspection, maintenance, monitoring, and repair (IMMR) of the GEP.

- FPSO arrival in the field: connection of the FPSO to the mooring buoy; equipment and systems testing (also termed commissioning), start-up operations.
- FPSO operations: process gas and condensate from the Barossa field. The gas and condensate separation and treatment
  systems have a gas export capacity of approximately 635 million standard cubic feet per day (a little over 7000 Olympicsize swimming pool) with a condensate processing capacity of approximately 1000 barrels per day (a little more than half
  an Olympic-size swimming pool) and a produced water processing maximum capacity of 20,000 barrels per day (or one
  Olympic-size swimming pool). Under normal operating conditions produced water discharge rates will be approximately a
  quarter (-5,000 barrels per day) of the maximum discharge capacity (-1/4 of an Olympic-size swimming pool). The FPSO
  generates its own electricity using Barossa production gas, and potable water supply. Living quarters are provided for the
  operations workforce.
- Gas export to DLNG: dry natural gas will be transported through the CEP for onshore processing at the DLNG facility.
- Storage and Offtake operations: storage of condensate onboard the FPSO and offloading of condensate to offtake tankers. Approximately 650,000 barrels (- 40 Olympic-size swimming pools) will be offloaded approximately once every two to three months (four to five times per year).



- Support operations: offshore support vessels periodically visit the FPSO to resupply meterials (such as stores, consumables, chemicals and fuel) and neturn surplus goods and wastes to the Australian mainland for disposal or recycling. Helicopters will be used to transport the operations workforce to and from operations facilities.
- Subsea system and GEP inspection, maintenance, monitoring and repair (IMMR): visual inspection of subsea infrastructure and/or the GEP using Remotely Operated Vahicles (a submensible craft used to perform underwater visual inspections operated from a vessel). This activity will be performed according to a planned inspection and maintenance schedule, or at other intervals if unplanned inspections or repairs are required.



Figure 3: FPSO key features



- Condensate Offloading System
- Aft Service Crane
- O Communications & Radar
- O Aft Pedestal Crane
- O Thermal Oxidiser

Figure 4: FPSO utilities and marine systems

- Forward Pedestal Crane Power Generation Exhaust O Midship Pedestal Crane
- D Helideck Landing Area
- Daughter Craft Freefall Lifeboats
- Engine Room Exhaust
- REGIONAL EXISTING ENVIRONMENT SUMMARY

### Environment that may be affected (EMBA)

Santos recognises the region's various environmental values and sensitivities. In an EP, although planned activity occurs in OA1 and CA2, it is common to present a geographically defined area of the environment that may be affected (EMBA) by an offshore activity e.g. an unplanned hydrocarbon spill.

In the case of the Production Operations Activity, the broadest extent of the EMBA, is determined by a potential loss of heavy fuel oil from a condensate offtake tanker due to impact from another vessel. Potential loss of heavy fuel oil is a risk associated with any large marine vessel and is managed through established maritime laws and safeguards. Barossa condensate offtake operations are a low frequency activity (four to five times a year) which further reduces the institution of such an event, which is already a very low probability of occurring.

Figure 5 depicts operational areas OAI and OA2 and the EMBA (blue line). The EMBA is generated by modeling and represents. the greatest geographical extent that could be affected by 300 individual hydrocarbon spill scenarios occurring simultaneously across the full range of seasonal conditions. It should be noted that an actual spill event is more accurately represented by only one of the 300 simulations from the modelling, meaning a much smaller geographical area would be affected in the event of an actual spil; and the EMBA does not take account of spil response mitigations which would reduce the extent of an unplanned spill. The primary purpose of the EMBA is to assist with spill response planning and preparedness in the unikely event of a hydrocarbon spill. The EMBA also provides the basis for assessing the range of potential socio-economic impacts and establishes a planning area for scientific monitoring during an unplanned spill event.

The Moderate Exposure Value (MEVA) (pink line) represents the predicted extent of ecological impacts and is used to inform the environmental impact assessment and spill response plans. Beyond the MEVA, impacts to ecological receptors are not expected.

To learn more about spill modelling, exposure values and spill response, see NOPSEMA Spill Modelling Video.



## Regional protected and significant areas

Figures 6, 7, and 6 illustrate the boundaries and zonings of regional marine parks and reserves, key ecological features, wetlands, EMBA, the MEVA and the OAs. Figure 9 illustrates the shoals and banks in relation to the OAs.



Figure 5: Production Operations Activities EMBA and MEVA in Commonwealth waters (OPGGS Act) and NT coastal waters (PSL Act)



Figure 6: Marine Parks and Other Protected Areas in relation to OA1, OA2, EMBA and MEVA



Figure 7: Key Ecological Features in relation to OA1, OA2, EMBA and MEVA



Figure 8: RAMSAR wetlands in relation to OA1, OA2, EMBA and MEVA



Figure 9: Shoals and Banks in the vicinity of OA1 and OA2

## Marine fauna and biologically important areas

Biologically important areas (BIAs) are areas used by protected marine species for carrying out critical life functions, such as reproduction, feeding, migration or resting. BIAs are areas that contain habitat crucial to the survival of protected species and are defined by the Australian Government under the Environment Protection and Biodiversity Conservation Act 2009 (EPBC Act). As shown in Figure 10, some BIAs occur within the EMBA. These areas are known to include protected marine species such as whale sharks, pygmy blue whales, dugongs, olive ridley turtles, loggerhead turtles, green turtles, hawksbill turtles, flatback turtles, and 12 types of seabirds and shorebirds. In addition, the EMBA overlaps the spawning grounds for southern bluefin tuna, a listed species under the EPBC Act, between northern Western Australia and Java.

The BIAs for flatback turtles overlap with OA2. There are no BIAs within OA1. Two turtle species, the flatback turtle and olive ridley turtle, have critical habitat that overlap with OA2.

Activities in OA1 will be conducted in water depths ranging from approximately 220 - 280 m. There are a variety of highly mobile marine fauna that may transit OA1 in low numbers, such as:

- Bryde's, blue, fin, humpback, sperm and sei whales
- orcas, Australian snubfin dolphin and spotted bottlenose dolphin
- dugongs (mostly in shallow waters)
- olive ridley, green, loggerhead, hawksbill, leatherback and flatback turtles
- sea snakes
- whale sharks
- migratory seabirds and shorebirds
- fishes, sharks, rays and sawfish.

An additional three species - the grey nurse shark, Omura's whale and the turtle-headed sea snake have been included as they were observed within or near OA1 and OA2 during the Barossa Marine Studies Program.

Santos recognises the region's various environmental values and sensitivities and has considered government guidance, including protected species management plans, recovery plans, conservation advice and threat abatement plans in the development of the Production Operations and OEMP, and has developed control measures to reduce impacts and risks to marine fauna and biologically important areas to as low as reasonably practicable and acceptable levels.



Figure 10: Biologically important areas in relation to OA1, OA2, EMBA and MEVA.



10 | BAROSSA PRODUCTION OPERATIONS ACTIVITY INFORMATION BOOKLET

## REGIONAL SOCIO-ECONOMIC SUMMARY

Socioeconomic activities that may occur within the OAs or EMBA might include commercial, recreational and traditional (subsistence) fishing, aquaculture, tourism, petroleum industry activities, defence activities, shipping and to a lesser extent in the deeper offshore waters, recreational fishing and tourism.

Underwater cultural heritage and cultural values may also exist across the region. Darwin will be the logistics hub and supply base for the Production Operations Activity bringing employment and economic benefits to the local community.

## Nearest population centres

OA2, at its closest point, is located approximately seven kilometres from Bathurst Island, which is part of the Tiwi Islands. Darwin is the closest regional city, which approximately 285 km north-north-west of OA1.

## Summary of other uses within the EMBA

Santos' existing understanding of the uses and values of the area and its strategies to reduce impacts or risks to these uses and values, will be supplemented with any new information obtained during consultation. Santos has set out in the list below a summary of the uses and values of the area based on existing information or previous consultation. Santos welcomes further information that may be provided during consultation to inform the Production Operations EP and OEMP.

## S

### Commercial fishing

Santos recognises the presence and rights of commercial fishers within the operational area and EMBA. Within the OAs, interaction with some commercial fishing is possible. These fisheries include Northern Prawn, Spanish Mackerel, Pearl Oyster, Offshore Net and Line, and Demersal. Santos has been consulting with the relevant fisheries representative associations, licence-holders and government over many years.



### Petroleum industry

Several oil and gas companies hold petroleum permits near the OAs; however, no established oil and gas operations are located within or in the immediate surrounds. The closest operational offshore production facilities and in-field subsea infrastructure are associated with the Santos-operated Bayu-Undan platform, located approximately 400km to the southwest of OA1 and west of OA2.



## Tourism, recreational fishing and traditional fishing

The OA1 is located in offshore waters that are not likely to be accessed for tourism activities (e.g. charter boat operations) or recreational fishing, as these tend to be centred around nearshore waters, islands and coastal areas. However, previous consultation on a different Barossa Gas Project EP has identified one fishing charter operator who may on occasions conduct tours near Evans Shoal, approximately 62km west of OA1. Tourism activities may occur within OA2, but they are likely to be limited to vessels transiting the area to access other destinations within the region e.g. islands, shoals, and shipwrecks. Indonesian and Timorese traditional fishers, as well as Australian recreational fishers, are expected to transit and fish in the EMBA. Santos continues to consult regarding recreational and traditional fishing and hunting within the EMBA.



### Defence Activities

No designated military and defence exercise areas occur within OA1. OA2 partially intersects the Darwin Air Weapons Range central practice area, which comprises practice and training areas used for offshore naval exercises and onshore weapons-firing training by the Department of Defence. Australian Border Force vessels also undertake civil and maritime surveillance (and enforcement) in Australian offshore waters, therefore they may transit through the operational areas and EMBA.

# ×

#### Telecommunications cables

The North-West Cable System (NWCS) is a submarine telecommunication fibre cable system located within the EMBA and crosses the GEP in the southern portion of OA1. It is located approximately 230 km and 30 km south of OA1 and OA2. Extending 2100 km from Darwin to Port Hedland, the NWCS connects Australia's remote northern and western regions, including offshore oil and gas facilities, with onshore locations.

## () Listed Heritage

There are no world heritage properties, national heritage places or Commonwealth heritage places within the OAs; however, the EMBA (including the MEVA) overlaps the Ashmore Reef Marine Park, a Commonwealth heritage place. The closest World Heritage Property is the Kakadu National Park, located onshore in the NT. A small portion of the coastal edge overlaps the EMBA. (Figure 8)

There are no recorded Aboriginal heritage sites within the OAs. The Tiwi Islands are a declared Aboriginal reserve and a number of protected sacred sites under the Aboriginal Sacred Sites Act 1989 (NT) have been recorded on the Islands.

Under the Commonwealth Underwater Cultural Heritage Act 2018, Australia's underwater cultural heritage is protected in Commonwealth waters, such as shipwrecks, sunken aircraft and other types of underwater cultural heritage including Australia's Aboriginal and Torres Strait Islander underwater cultural heritage. No known shipwrecks are located within the OAs. Multiple known shipwrecks, sunken aircrafts, historic aircrafts and shipwrecks (greater than 75 years old) and other sites occur within the EMBA. Some unlocated wrecks could fall within the boundaries of the OAs or EMBA.

In the course of preparing the Barossa Drilling and Completions EP, SURF EP and commencing works under the GEP EP, Santos engaged independent consultants to investigate potential for underwater cultural heritage within OAL First Nations underwater cultural heritage is not relevant to OAI due to its location in water depths beyond the extent

12 | BAROSSA PRODUCTION OPERATIONS ACTIVITY INFORMATION BOOKLET

of the ancient coastline at the 125 m water depth contour. A 262km section of the GEP within OA2 has been surveyed for both First Nations and other underwater cultural heritage

The results of those surveys concluded there are no specific underwater cultural heritage places along the Barossa GEP to which people, in accordance with Indigenous tradition, may have spiritual and cultural connections that may be affected by the activities covered by the GEP Environment Plan. Further similar surveys are planned for the remainder of OA2 in 2024.



#### Shipping

The closest port to the OAs is Darwin Port, which is approximately 290km away from OA1 and 116km away from OA2. No designated shipping fairways overlap the OAs, however the southern end of OA2 is an area of high shipping activity.



### Cultural Values

Traditional hunting and fishing continue to occur on the Tiwi Islands, although typically these occur within 3 nm of the shoreline.

Mapping exercises and workshops conducted on the Tiwi Islands have identified Aboriginal heritage sites along the northern, western and southern coastlines of the Tiwi Islands, including areas used for food collection, sacred sites, camping sites and a dreaming site. These coastlines are within the EMBA but outside the OAs.

Santos has identified that the Croker Island native title determination (DCD1998/001) partially overlaps the EMBA. The native title holders within the Croker Island native title determination are the Yuwurrumu members of the Mandilarri-Ildugij, the Mangalara, the Murran, the GaduraMinaga and the Ngaynjaharr clans. The Larrakia native title determination (DCD2006/001) also partially overlaps the EMBA. This determination found that native title does not exist within the claim area.

Mapping exercises and workshops conducted on the Tiwi Islands have identified Aboriginal heritage sites along the northern, western and southern coastlines of the Tiwi Islands, including areas used for food collection, sacred sites, camping sites and a dreaming site. These coastlines are within the EMBA but outside the OAs.

Santos acknowledges coastal First Nations peoples' connection with culture through Sea Country and is seeking to improve knowledge and understanding of cultural features within the EMBA, including through consultation with First Nations people and their relevant representative bodies.
## SUMMARY OF ENVIRONMENTAL IMPACTS AND RISKS

Environmental impact and risk assessment is the process by which proposed activities are assessed for their impacts (consequences) on the environment (physical, biological, socio-economic and cultural). For the purposes of assessing impacts and risks, proposed activities are divided into planned activities and unplanned events.

Planned activities occur within OA 1 and OA 2 can have unavoidable impacts, such as light, noise and atmospheric emissions, seabed disturbance, discharges to the marine environment, and interactions with other manne users. Unplanned events are not expected to occur but are considered so that contingency measures are in place should they ever eventuate. Unplanned events include dropped objects, introduction of invasive marine species, interactions with marine fauna, accidental discharges, or spills.

Planned activities are assessed based on consequence of impact. Unplanned events are assessed based on their potential impact (consequence) and likelihood of occurrence, which informs the associated risk level.

Santos has conducted an environmental assessment in order to consider the potential environmental impacts and risks associated with activities under this EP. The identification of potential impacts and risks, and the measures proposed to reduce these impacts and risks, may be revised and amended as a result of the consultation process. This includes information obtained during consultation to improve Santos' understanding of potential impacts and risks in regards to cultural values within the EMBA and adoption of any appropriate measures.

## **PLANNED ACTIVITIES**

Santos proposes to adopt a suite of control measures to reduce impacts and risks associated with planned activities to a level that results in a minor or negligible environmental consequence. These consequence levels are considered by Santos to be acceptable and to have been reduced to as low as reasonably practicable (ALARP). Figure 11 shows several emissions and discharges from planned activities associated with the operation of the FPSO.



Figure 11: Emissions and discharges from planned activities associated with the operation of the Floating Storage and Offloading (FPSO) facility.



## **GHG EMISSIONS**

The Production Operations EP will consider the contribution of emissions from the Barossa Development to national and global emissions and the potential indirect impacts of climate change on the Australian environment, noting that as a result of the complex nature of the global emissions system, climate change impacts cannot be meaningfully linked to any one activity or emissions source.

- GHG emissions can be categorised into Scope 1, Scope 2 and Scope
- Scope 1 direct emissions from sources that Santos owns or controls, due to fuel combustion, flaring, venting, CO2 removal and fugitive emissions.
- Scope 2 Indirect emissions from the generation of energy that Santos purchases for its operations including electricity purchased for ancillary activities such as office buildings.
- Scope 3 includes all indirect emissions not included in Scope 2. The vast majority of Scope 3 emissions from Santos' activities are emissions from the use of sold products.

The Production Operations Activity will not produce scope 2 emissions as it does not consume externally generated electricity or other forms of externally generated energy.

Total annual Barossa Scope 1 emissions are estimated to be 2.5 Mt CO2e (carbon dioxide equivalent), and total annual Scope 3 emissions are estimated to be 12.7 Mt CO2e.

### What impacts are expected?

Barossa GHG emissions (Scope 1 and 3) estimates account for approximately 0.86% of annual Australian GHG emissions (Department of Climate Change, Energy, the Environment and Water, 2022).

### The GHG emissions attributable to the Barossa Development are not expected to be significant relative to national and international GHG emissions and are considered to be low risk.

### How will Santos manage impacts?

Scope 1 emissions from Barossa are managed under Australian regulations and scope 3 emissions are managed using control measures consistent with the UN Paris Agreement, to which the Australian Government is a signatory. Key proposed control measures include:

### Scope 1 emissions:

- Barossa will comply with Safeguard Mechanism obligations, including surrendering carbon credit units for any emissions above the Safeguard baseline for the assessment year.
- Barossa will implement a GHG management plan that minimises GHG emissions to ALARP and acceptable levels over the life of the field operations.
- Barossa facilities design has been optimised to reduce fuel, flare and vent (FFV) emissions, and to enable the possibility of future export of reservoir CO2 to a Carbon Capture and Storage (CCS) project.

#### Scope 3 emissions:

Products generated from the Barossa Development will only be sold to
customers from countries that are signatories to the Paris Agreement
(or that have policies for reducing greenhouse gas emissions that are
equivalent to policies required by the Paris Agreement), as at the date
of the relevant contract of sale.



## **ATMOSPHERIC EMISSIONS**

Fuel consumption, flaring and venting excess gas is required to process gas and condensate which results in the release of air pollutants, such as sulphur oxides (SOX), nitrogen oxides (NOX) and volatile organic compounds (VOCs) to the atmosphere. This mix of continuous and infrequent (e.g. flaring of excess gas) sources of atmospheric emissions associated with operating the Barossa facilities may result in a temporary, localised reduction in air quality.

Intermittent flaring is expected to be of short duration during initial start-up operations and unplanned process trips/upsets during steady-state operations.

Atmospheric emissions will also be generated from support vessel and helicopter operations.

All activities described above that may result in air emissions could be expected within OA1, however only emissions associated with vessel activities would be expected in OA2. In the offshore environment, air emissions quickly dissipate into the surrounding airshed.

### What impacts are expected?

Impacts are considered very localised and not significant. Seabirds and migratory shorebirds are unlikely to be impacted by the localised and temporary reduction in air quality.

The potential impact from the release of air emissions includes the decrease in air quality of the local airshed.

Behavioural impacts, such as avoidance, could be expected if seabirds fly in the vicinity of OA1. Impacts to threatened, migratory or local fauna (seabirds) are considered to be minor.

As Barossa's operational activities occur in remote offshore waters, Production Operations Activity emissions will not impact air quality in coastal towns. Atmospheric emissions will quickly dissipate into the surrounding atmosphere and are not considered to be a potential source of impact for protected areas or threatened ecological communities.

### How will Santos manage impacts?

The FPSO power generation system reduces emissions to the atmosphere by primarily using production gas as fuel. Combined-cycle gas turbines (CCGT) will also be used to improve fuel use efficiency. CCGT are highly efficient and best practice resulting in reduced pollution. The steam turbines use low NOx burners, which significantly reduces NOx emissions.

Further FPSO facility design measures to reduce atmospheric emissions include:

- surplus waste gas burned to completely remove methane (minimal flaring)
- vapor recovery instead of releasing to atmosphere
- · ozone Depleting Substances (ODS) are not used as refrigerants

Santos proposes to adopt numerous control measures to manage vessel emissions, including requiring contractor vessels' compliance with MARPOL requirements for low-sulphur fuel and air pollution prevention certifications. ("MARPOL' is a reference to the International Convention for the Prevention of Pollution from Ships).

The control measures to be adopted are designed to be consistent with maritime regulations and petroleum industry standards.



### **NOISE SOURCES**

In OA1 the main FPSO noise sources are intermittent and short-term flaring during initial start-up operations and unplanned process trips/upsets during steadystate operations. FPSO power and processing equipment will also generate a continuous source of noise. Other noise sources include:

- support vessels;
- helicopters;
- + ROVs, acoustic positioning systems and survey equipment
- + operation of subsea infrastructure such as wellheads, flowlines and valves.

In OA2 the types of noise generated by these activities can be categorised as either: impulsive (brief, high intensity) e.g. from operation of survey equipment or, non-impulsive noises (ongoing or continuous) e.g., from vessel engines. Noise emitted from activities in OA1 and OA2 are expected to be at low levels, similar to ambient noise levels in the region.

### What impacts are expected?

Santos has engaged subject matter experts to conduct the underwater noise assessments for the activities.

Noise emissions from the FPSO, helicopters, survey equipment and vessels may result in marine mammals (e.g. whales) changing their behaviour (e.g. avoidance or diving to avoid noise). This change in behaviour is expected to be localised (within the area of the noise source in OA1 or OA2) and short term (e.g. periods of minimal flaring in OA1). Noise emissions are not expected to cause long term population impacts (e.g. distribution & abundance).

Low level noise can occur from the operation of the pipeline which will dissipate to background levels within 100m of the pipeline. The GEP route (OA2) crosses two small areas of important turtle habitat and impacts to marine turtles in these areas are expected to be limited to behavioural (e.g. avoidance). Impacts to marine turtles from any infrequent survey equipment use is also expected to result only in temporary and localised behavioural changes, given the low level of noise. Vessels will be moving when undertaking surveys and it is highly unlikely any individual would remain near the noise source for any length of time.

Other protected species of marine reptiles (such as sea snakes), seabirds and fish (such as sharks and sawfish) are not expected to be affected at the population level, given their wide distribution (in the case of sea snakes and sharks), distances to seabird breeding colonies, and preference for shallow coastal habitats (sawfish).

Noise emissions could result in behavioural changes in marine fauna within the Oceanic Shoals Marine Park, as OA2 is located in this area (Figure 4).

Noise is not expected to impact socio-economic receptors, including commercial fisheries, due to low noise levels and low socio-economic activity levels within and near the Operation Areas. Behavioural impacts to fish of potential commercial value would be restricted to within hundreds of metres of the noise source, a very small portion of the total available fishing area.

### How will Santos manage impacts?

A source of significant underwater noise has been eliminated from the FPSO facility by designing the facility to be permanently moored without the use of a propulsion system.

Vapour recovery on the flare system reduces the frequency at which flaring occurs during operations and therefore reduces the amount of noise emitted during routine operations.

Activity vessels are required to comply with Santos's Protected Marine Fauna Interaction and Sighting Procedure to comply with regulatory requirements for managing fauna noise impacts. Marine assurance standards and planned vessel maintenance will minimise noise generated from vessels by ensuring contracted vessels are operated, maintained and crewed in accordance with industry standards and regulatory requirements.



## LIGHT SOURCES

Artificial lighting is required for operational and navigational safety during the activity. Light sources include safety and navigational lighting on vessels, campaign-specific lighting when needed, such as deploying or retrieving equipment or when ROVs are working underwater, and intermittent flaring from the FPSO.

### What impacts are expected?

Permanent safety and navigational lighting on the FPSO and intermittent flaring will result in light emissions in OA1. Light emissions are not expected to have an effect on adult turtles or hatchlings, given the offshore location and distance from the nearest turtles nesting beaches.

OA2 overlaps BIAs for marine turtles surrounding the Tiwi Islands. The closest turtle nesting beach is Cape Fourcroy, on the Tiwi Islands. Lighting emissions in OA2 will only occur from infrequent vessel inspection and maintenance activities in this location, which are of short durations. Vessel activities in OA2 are expected to produce similar light levels to other marine vessel activities.

Impact to nesting females or hatchlings is not expected to occur. There is potential for hatchlings at sea to be attracted to light emissions if they are carried by currents to within approximately 3.3 km of an IMMR vessel. In the unlikely event hatchlings are attracted to vessel light, the proportion impacted is considered negligible when compared to the total number of hatchlings emerging from Bathurst Island beaches across the year. It will also be a temporary phenomenon, occurring during hours of darkness only. After sunrise, hatchling dispersal behaviour will resume. Displacement of individuals from critical habitat areas is therefore not a credible outcome.

Fish, sharks and birds have been shown to be attracted to artificial light sources, leading to a short-term localised increase in fauna activity, however large-scale changes in species abundance or distribution are unlikely.

### How will Santos manage impacts?

The FPSO facility is equipped with a centralised battery system providing an uninterrupted power supply to the FPSO LED lighting which also allows for dimming and controlling in individual areas. Lighting is to be limited to that required for safe operations and navigation and will be compliant with maritime regulations (similar to other commercial vessels operating in the region).



## SEABED DISTURBANCE

Seabed disturbance will occur because of:

- · physical presence of installed subsea infrastructure and GEP on the seabed
- temporary placement and set down of equipment and subsea infrastructure on the seabed used during IMMR activities

### What impacts are expected?

Seabed disturbance resulting from IMMR activities will be confined to the OAs and might result in localised disturbance under the subsea infrastructure and GEP. Seabed disturbance resulting from a subsea repair or replacement will be localised with a potential footprint of approximately 50 m<sup>2</sup> up to 1,600 m<sup>2</sup>.

Given the nature and relatively small scale of seabed disturbance, it is not expected to cause a decrease in local population size, area of occupancy of species, loss or disruption of critical habitat, or disruption to the breeding cycle of any protected marine fauna.

Given localised disturbance is restricted to the OAs, which is mostly bare sediment and does not contain any significant habitat features, the consequence level for the physical environment or habitat is negligible. Impacts to the seabed within the Oceanic Shoals Marine Park or overlapping key ecological features (KEFs) (Carbonate bank and terrace system of the Van Diemen Rise KEF and the Shelf break and slope of the Arafura Shelf KEF) are considered to be minor.

While OA1 does not overlap any marine turtle BLAs, the southern end of OA2 traverses internesting buffer habitat critical to survival for flatback and olive ridley turtles, overlaps a portion of the the internesting BLA for flatback turtles, and is adjacent to the internesting BLA for olive ridley turtles. Considering the water depth along the pipeline route in OA2 is greater than the maximum turtle interesting depth of 30 m, it is unlikely the species will be present in significant numbers or for significant periods. Any impact to marine turtles from seabed disturbance or resultant turbidity in both OA1 and OA2 would likely be temporary and negligible, based on the nature and scale of impact.

Seabed disturbance is not expected to impact commercial fisheries, based on the small size of disturbance compared with the total available fishing area.

### How will Santos manage impacts?

During IMMR activities Santos' vessels will undertake safe and accurate placement of infrastructure using dynamic positioning to minimise seabed disturbance during placement. Santos will also maintain a comprehensive inventory of all installed equipment to enable recovery of all equipment during decommissioning to limit impacts to the seabed.



## INTERACTIONS WITH OTHER MARINE USERS

Other marine users will be displaced from part of OA1 over the life of Barossa operations, and temporarily restricted within parts of OA2 during IMMR. In OA2, the GEP may present a hazard to marine users due to the potential for snagging.

### What impacts are expected?

Other marine users will have restricted access within petroleum safety zones (PSZ). A permanent S0Om exclusion PSZ will extend around the outer edge of the Barossa Production Operations wells, the subsea infrastructure and mooring system in OA1. During IMMR activities along the GEP IN OA2, a temporary S0Om PSZ will be maintained around vessel operations.

Commercial fishing, shipping, military exercises and other incidental marine traffic in the OAs are expected to be low frequency. The area marine users will be excluded from is small when compared to the large area available for their use.

### How will Santos manage impacts?

Santos will notify and communicate with other marine users using standard maritime notifications (e.g. Notice to Mariners) before, during and at the end of IMMR activities. Infrastructure locations will be marked on nautical charts. These proposed control measures are consistent with maritime regulations and industry practices.



## PRODUCED WATER DISCHARGES

Produced water is naturally occurring water that is extracted from the seabed along with hydrocarbons (condensate and gas in the case of the Barossa field). It is separated from the hydrocarbon components during processing and treated before being discharged to the marine environment from a pipe at least 10 m below the sea surface on the FPSO. This produced water consists of naturally occurring formation water (from the body of rack below the hydrocarbon formation), condensed water (water vapour present within the produced hydrocarbons which condenses when brought to the surface) as well as introduced water-soluble chemicals and other contaminants. While produced water treatment is performed before discharge, the effluent may contain residual inorganic (such as chemicals used for production) and organic (such as oil) contaminants.

The produced water treatment system is divided into two stages - removal of hydrocarbons through a filtration system and designed to handle 20,000 barrels per day (over one Olympic-size swimming pool). During operations the produced water discharge will vary from 3,500 to 5,000 barrels per day (bb/day), or a quarter to a third of an Olympic-size swimming pool, with a peak rate after 11 years estimated up to 16,500 bb/day (one Olympic-size swimming pool). Best available technology has been selected to remove oil-in-water concentrations to as low as reasonably practicable (ALARP) and the treatment system will operate well below its design capacity over the majority of the field life.

### What impacts are expected?

Water quality may be impacted at the discharge point while the produced water is discharged (Figure 12). Discharge modelling has been undertaken for a conservative maximum discharge rate of 20,000 bbl/day (over one Olympic-size swimming pool). Under normal operating conditions produced water discharge rates will be approximately a quarter (-5,000 bbl/day) of the modelled maximum discharge rate. Modelling results indicated that species protection thresholds for waterborne contaminants is achieved at approximately 6 km from the FPSO. As a result, predicted impacts will be localised and considered minor.

Marine turtles may occur within the produced water mixing zone. It is possible individual turtles may traverse the mixing zone; however significant impacts are not expected to occur, and large numbers of animals are not expected to be exposed. That is because the discharge water depth and discharge location are not within the proximity of internesting turtle habitat, and there is minimal reef habitat in the mixing zone. Given marine turtles are transient through the produced water mixing zone, they will not be exposed to the produced water for enough time for contaminants to accumulate within their body. Behavioural impacts (such as avoidance) may occur to a small proportion (individuals) of a local population close to the produced water discharge.

Like turtles, produced water exposure to plankton, fish, invertebrates and sharks is expected to be brief due to the transient nature of these animals.

Potential impacts to fishery resources are unlikely to result in changes in distribution and abundance of fish species outside the produced water mixing zone.



Figure 12: Predicted extent of the produced water dilution at a maximum discharge rate of 20,000 bbl/day.

### How will Santos manage impacts?

Activity discharges are to be managed through the application of Santos' Chemical Selection Process, designed so that environmentally acceptable process chemicals (which are likely to be mixed with produced water discharge) are used. Additives have been selected and optimised for biodegradability as well as low aquatic toxicity and bioaccumulation potential.

The produced water discharge will be continuously monitored for oil-inwater content and any reading over the limit of 30 mg/l over any 24hour period, will be diverted to a dedicated storage tank and returned to the produced water treatment system until suitable for discharge. Strict protocols will be in place for taking regular water samples and perform laboratory testing to ensure the produced water is within acceptable levels before disposing into the marine environment. The Production Operations Environment Plan will include a produced water adaptive management plan that prescribes a water quality monitoring regime which enables the detection of potential impacts of produced water discharge on the marine environment and if remedial actions are necessary to retain the discharge within acceptable limits.



## OPERATIONAL DISCHARGES

Operational discharges associated with the activities may cause localised impacts to water quality in the direction of the prevailing current. The environment that may be affected by operational discharges will likely be contained within the OAs. Water quality conditions will return to normal within minutes to hours once discharging stops.

### FPSO facility and subsea system discharges

Operational discharges from the FPSO in OA1 will occur each day resulting in localised changes to water quality. Discharges of warm cooling water will include low concentrations of chlorine which break down quickly in the environment and is non-toxic at low concentrations. Minor discharges of water based hydraulic fluid used in the subsea system are classified by the offshore chemical notification scheme as being environmentally acceptable.

#### Vessel discharges

The types of anticipated discharges in OA1 and OA2 are typical of most offshore commercial vessels and include deck runoff, treated sewage, grey water, machinery cooling water, bilge water (treated via the oily water system), ballast water, macerated food scraps and brine (from water making). These discharges will be small in volume and released into surface waters.

#### What impacts are expected?

Sensitive receptors that may be impacted include plankton, fish, seabirds, marine turtles and mammals. Impacts to water quality will be localised and temporary occurring only during discharge.

Some fish and oceanic seabirds may be attracted to the FPSO by the discharge of food scraps. However, given the small quantities, intermittent nature of disposal and swift currents, any attraction is likely to be minor and is not anticipated to result in adverse impacts at an ecosystem or population level. Given the controls in place to manage the FPSO discharges in OA1 in accordance with regulatory requirements, impacts to commercial fish species are not predicted.

Operational discharges in OA1 are predicted to quickly dilute and disperse in the offshore environment. Water quality changes will be localised and will occur only when the discharges occur. Given the temporary nature of activities within OA2 (limited to vessel based IMMR) and the relatively deep offshore environment with significant current and tidal action, impacts to water quality will be localised and will occur only for the duration of the discharge.

### How will Santos manage impacts?

Vessel discharges are to be managed to acceptable levels as regulated by maritime laws and conventions (e.g. management of sewage treatment systems and oily water systems), such as MARPOL and relevant Marine Orders. Santos also intends to implement management measures including waste management procedures and chemical management and selection procedures.

Santos procedures require that all operational chemicals used on the FPSO and chemicals potentially discharged to sea are risk assessed. Santos also implements general chemical management procedures to reduce the risk of accidental discharges.

# UNPLANNED EVENTS

Unplanned events are not expected to occur but are considered so that contingency measures are in place should they ever eventuate.

The following unplanned events have been identified for the Production Operations Activity:

- · release of solid objects
- introduction of invasive marine species
  interaction with marine fauna





## **DROPPED OBJECTS**

### How could dropped objects occur?

There is the potential for objects to be accidentally released to the marine environment from support vessels during steady state operations in OA1 or during IMMR activities in OA2. Dropped objects may be non-hazardous solid waste (e.g. paper, packaging materials), hazardous waste (e.g. batteries, aerosol cans etc.). A dropped object event could result from overfilling waste containers, unsecured objects during lifting operations or failed sea fastening.

### What environmental impacts could occur?

Objects that float could potentially move beyond the OAs. All non-buoyant objects are expected to sink to the seabed and remain within the OAs. This could cause localised and short-term damage to the seabed.

Marine debris (including plastics and microplastics) is listed as a potential threat to several marine fauna species. Depending on debris size of the dropped object, there is potential for entanglement or ingestion by marine fauna, including turtles and vertebrate wildlife, which could result in injury or death. However, given the limited quantities that might be dropped, impacts to fauna would be limited.

Considering the low frequency of such an unplanned event, even in a worst-case release of a solid object, impacts to fauna would be very localised and limited to individuals, and are not expected to result in impacts to the local population.

### How will Santos manage the risk?

Santos has numerous control measures to prevent dropped objects, and to mitigate consequence of impacts of an event does occur. These measures include:

- safety standards and procedures to reduce the likelihood of tools and other equipment being dropped during lifting operations
- waste management procedures to reduce the likelihood of windblown waste entering the marine environment
- implementation of chemical selection processes and the International Maritime Dangerous Goods Code to limit the environmental impact of chemicals if lost overboard
- dropped objects, regardless of size, must be reported and attempts made to recover the object according to safety and environment criteria.

These control measures are designed to comply with maritime legislation. In addition, these control measures are consistent with applicable actions described in the relevant fauna recovery plans and conservation advice, reducing the residual risk to low.



## **INVASIVE MARINE SPECIES**

### What are IMS?

Invasive marine species (IMS) are marine flora and fauna that have been introduced into a region that is beyond their natural range but have the ability to survive, and possibly thrive. The majority of climatically compatible IMS to northern Australia are found in south-east Asian countries.

### How might IMS be introduced?

Some IMS pose a significant risk to environmental values, biodiversity, ecosystem health, human health, fisheries, aquaculture, shipping, ports and tourism. The risk of introducing IMS is common for all maritime activities. The introduction of IMS may occur due to the following:

- biofouling on FPSO and vessels, external/internal niches (such as sea chests and sea water systems) and routinely submerged equipment
- discharge of FPSO ballast water when the FPSO first transits from the international shipyard to the Barossa field
- discharge of high-risk ballast water where vessels have transited from international destinations.

### What environmental impacts could occur?

If successfully established, IMS can:

- · outcompete native species for food or space
- prey on native species
- impact fisheries or aquaculture
- impact on human health through released toxins
- reduce coastal aesthetics
- cause damage to marine and industrial equipment and infrastructure.

The above impacts can result in flow-on detrimental effects to marine parks, tourism and recreation.

The ability of invasive marine species to colonise a habitat depends on several environmental conditions. For example, highly disturbed environments (such as marinas) or shallower areas are more susceptible to colonisation than open-water environments (CA2 is 33 metres deep at its shallowest point and not considered sufficiently shallow to be conducive for IMS colonisation). OA1 provides an unfavourable habitat for IMS due to water depth (over 200 metres) and the long distance to the coast. These conditions limit light availability and have low habitat biodiversity with sparse epibiota, therefore, it is highly unlikely that IMS would be able to survive or colonise in OA1.

### How will Santos manage the risk?

The pathways and vessel mitigation measures for IMS introduction are well established. The offtake tankers used for condensate export and specialised IMMR vessel(s) (if required) are sourced internationally, whilst the regular support vessels to and from the FPSO are sourced domestically. Vessels contracted to Santos, and vessel ballast, are to be managed according to control measures that comply with maritime regulations, industry practices, and the Biosecurity Act 2015. The FPSO and support vessels will also have ballast water management, vessel biofouling management and anti-fouling systems in place. With these control measures adopted, the residual risk of introducing IMS is assessed as low and reduced to as low as reasonably practicable.

The initial mobilisation of the FPSO out of Singapore to the Barossa gas field will be managed under a quarantine management plan including arrangements for invasive species, biofouling and ballast water exchange.



## INTERACTION WITH MARINE FAUNA

## How could interactions with marine fauna occur?

During the Production Operations Activity, approximately two vessels per week will travel between the Barossa field and Darwin servicing the FPSO, which is a minor increase relative to the existing levels of regional marine vessel traffic.

The highest potential for interactions with marine fauna, including potential accidental strike or collision resulting in injury or mortality, will be during IMMR vessel operations in OA2 where there is higher likelihood of marine fauna presence. In OA1, where marine fauna presence is of a lower likelihood, the FPSO will remain stationary once on location in position and support vessel movements within the operational area are limited and slow-moving, hence marine fauna interactions are not anticipated and are expected to be minimal.

Marine fauna such as marine mammals (such as whales and dolphins), marine turtles and whale sharks that swim at or near the water surface are most at risk from vessel collisions. Some of these species are threatened, and some marine fauna may have cultural significance.

Marine mammals (such as whales and dolphins) and whale sharks may transit through the OAs but are expected to be in low numbers in OA1 (Figure 10). Considering the relatively slow vessel speeds, short duration of activities, and the mobility of these species, it is unlikely that activity vessels will adversely interact with any individuals.

### How will Santos manage the risk?

The likelihood of marine fauna interaction resulting in injury or death is considered unlikely given the proposed implementation of the following control measures:

- Santos' Protected Marine Fauna Interaction and Sighting Procedure, which aligns with the Environment Protection and Biodiversity Conservation Regulations 2000. This procedure limits marine fauna approach distances and speed, allowing marine fauna to be avoided or to move away.
- Operational area vessel speed restrictions

The control measures are designed to align with management actions outlined in government-published fauna recovery plans and conservation advice. The risk of interactions with marine fauna is assessed as very low and reduced to as low as reasonably practicable and acceptable levels. The risk is no higher than for any other regional maritime activity.



## NON-HYDROCARBON LIQUID RELEASE

## How could non-hydrocarbon liquids be released?

Non-hydrocarbon liquids including miscellaneous chemicals for use during the Production Operations Activity and waste by-products are transferred to and from supply vessels to the FPSO in OA1. Examples of non-hydrocarbon liquids include chemicals used in the production process, domestic products used in the living quarters for cleaning and general maintenance products such as greases and paints.

An accidental release of non-hydrocarbon liquids into the marine environment has the potential to occur from:

- + transferring, storing or using bulk products (e.g. production chemicals)
- mechanical failure of equipment, such as a tank or pipework failure
- handling and storage spills and leaks due to insufficient fastening or inadequate bunding
- floating hose failure or rupture, coupling failure or tank overfilling
- lifting and incorrect handling (e.g. dropped objects damaging storage containers)
- · firefighting foam during an emergency response incident.

### What environmental impacts could occur?

A release of non-hydrocarbon liquids may result in impacts to water quality and any sensitive environmental receptors.

The maximum volume of non-hydrocarbon liquids that could be released during routine operations is likely to be small and limited to the volume of individual storage containers. Individual containers stored on the FPSO include process chemicals and lube oil storage tanks (approximately 4.5 m<sup>3</sup>).

If the spill is not contained on deck, a release to the marine environment would likely disperse rapidly, with one in 1,000 dilution usually occurring within 30 minutes.

The environment that may be affected for non-hydrocarbon liquids releases resulting in a decrease in water quality is likely to be restricted to the immediate vicinity of the FPSO or support vessel and contained within the OAs.

Potential receptors include the physical environment (e.g. water and sediment quality, benthic habitats), threatened, migratory or local fauna (e.g. marine mammals, marine reptiles, sharks and rays, other fish, and birds) and socioeconomic features).

### How will Santos manage the risk?

Santos has a suite of procedures to manage the selection, storage, handling and clean-up of non-hydrocarbon liquids releases. Vessels also have spill response plans. All chemicals are reviewed and accepted for use, and any chemical that might be discharged to the environment is assessed under the Santos chemical selection procedure to ensure environmental acceptability. These procedures will assist to minimise the likelihood of non-hydrocarbon liquid spills, and subsequent environmental consequences should they occur.

The control measures proposed to be adopted are designed to be consistent with maritime and petroleum industry standards and appropriate to manage the residual risks to as low as reasonably practicable and acceptable levels.



## MINOR LIQUID HYRDOCARBON RELEASES

## How could a minor liquid hydrocarbon release occur?

Minor releases refer to relatively small volumes of hydrocarbons from storage containers, transfer equipment and pipework on the FPSO or support vessels, that enters the marine environment. Typically, such spills occur as a result of human error during tank filling or storage container transfers. Most of these types of release occur within bunded deck areas, and are less than 1m3, however it remains possible for such spills to enter the marine environment.

### What environmental impacts could occur?

A localised decrease in water quality may occur, however due to the relatively small volumes impacts are expected to be short term as the hydrocarbon would rapidly dilute and dissolve into the ocean. Marine fauna may transit through the OAs and come into contact with the release. However it is expected impacts to fauna would be short term and result in behavioural changes, as they move away from the area where the spill occurred.

### How will Santos manage the risk?

A suite of procedures will be in place to manage the handling and transfer of hydrocarbons on both support vessels and the FPSO. Response procedures such as stopping the source of the of the release and cleaning it up on deck to prevent it entering the ocean will be in place to manage minor releases should they occur.

The control measures proposed to be adopted are designed to be consistent with maritime and petroleum industry standards and appropriate to manage the residual risks to as low as reasonably practicable and acceptable levels.



## LARGER HYDROCARBON RELEASES

Larger volumes of hydrocarbons may accidentally be released during production operations. These include accidental spills from support vessels or the FPSO, as well as from subsea equipment (e.g. wells and flowlines). A range of different types of hydrocarbons that may be accidentally released are discussed below.

### MARINE DIESEL OIL OR MARINE GAS OIL

### How could a marine diesel or gas release occur?

Marine vessel fuels (marine diesel oil or marine gas oil) could be released to the environment if there is a collision event between two vessels. An accidental collision could occur due to factors such as human error, poor navigation, vessel equipment failure or poor weather. If a marine vessel collided with the FPSO, the vessel or FPSO hull may rupture and release fuel to the marine environment.

If a vessel fuel tank is ruptured a fuel called marine diesel could be released. The FPSO uses a lighter fuel for some of its power requirements called marine gas oil which could be released if a fuel tank is ruptured.

Although the risk is higher in OA1 than OA2, it should be noted that it is considered unlikely that a vessel collision would occur that would result in releasing fuel to the environment. A sequence of events would need to occur for a vessel collision to escalate to a large volume of fuel released to the environment, including:

- 1. the vessel must be involved in a collision
- 2. collision must occur with enough force to rupture a fuel tank
- rupture must be of such a nature that the fuel can be released into the environment.

### What environmental impacts could occur?

Marine diesel oil and marine gas oil fuels are typically characterised by a high percentage of volatile components (typically >95%), which will evaporate when on the sea surface over several days. A small fraction (typically <5%) of persistent hydrocarbons remains, which will not evaporate, and will decay over time. The heavier components of the fuels tend to become entrained in the upper water column as droplets in the presence of waves but can refleat to the surface if wave energies abate. Both marine diesel oil and marine gas oil fuels are expected to weather quickly through evaporation and dispersion and are unlikely to persist in the environment for a significant period.

Such releases will cause a decline in water quality and may cause chemical (e.g. taxicity) and physical impacts to marine species (e.g. ingestion of hydrocarbons). The severity of the impact depends on the magnitude of the release (i.e. extent, duration) and sensitivity of the receptor, however, may include impacts to the physical environment, threatened or migratory marine fauna, protected and significant areas and socioeconomic receptors (fisheries, tourism, recreation, cultural features and other oil and gas operators).

### How will Santos manage the risk?

The FPSO in OA1 is fitted with a collision avoidance radar so it appears on the display of the triggering radars, providing range, bearing and identification information, alerting vessels to its presence. Santos has also designed the FPSO hull to be double-sided and double-bottomed, which provides two physical barriers between the fuel tanks and the marine environment for side impact, reducing the likelihood of fuel release in the event of a collision.

A petroleum safety zone (PSZ) will be established alerting other marine users to the presence of the FPSO in OA1 which includes precautions for marine activities (e.g. reduced speed limits, communication protocols and automatic identification systems to aid in their detection at sea). Third party vessels are not permitted to enter a PSZ, thereby reducing the likelihood of other interactions with the FPSO and support vessels. In OA2 during IMMR activities a similar exclusion zone will also be established restricting access to other marine users.

The Production Operations Activity facilities in OA1 & OA2 will be included on navigational charts making other vessels aware of the presence of Barossa facilities. Santos will also provide maritime notifications to relevant departments to ensure marine users are informed of vessel movements.

Santos has also developed response plans which will detail the actions to take to control the release and manage the cleanup activities in the event of a release.





## **PIPELINE GAS RELEASE**

### How could a dry natural gas release occur?

Although highly unlikely, dropped objects from vessels (Santos or a third party) may pose a risk of damage to the Barossa GEP. Damage from dropped objects could result in a potential rupture to the GEP that would release dry natural gas into the environment.

### What environmental impacts could occur?

Gas would move vertically toward the sea surface resulting in a visible bubble zone at the sea surface and an associated gas cloud before rapidly dispersing into the atmosphere.

Potential receptors include the physical environment (water and air quality); threatened, migratory or local fauna (marine mammals, marine reptiles, sharks and rays, other fish, and birds); socioeconomic (other marine users) and cultural features. Potentially the gas cloud may impact air-breathing fauna, such as marine mammals, marine reptiles and birds. Impacts to marine mammals, turtles and birds are not expected to be significant due to the very localised disturbance area and the rapid dispersion of the gas into the atmosphere. Following the initial gas release, marine fauna are expected to avoid the gas release location limiting the potential for any further impacts.

The gas release could also result in a loss of methane, contributing to GHG emissions. The contribution of the gas release to GHG emissions would be temporary and limited to the duration of the release event.

### Socio-economic receptors

For the Barossa GEP release, a gas cloud could form an explosive mix that, that if ignited, could result in injury/death of other marine users and property damage. In the highly unlikely event of a Barossa GEP release, safety exclusion zones may be established for the duration of the gas release (weeks) and pipeline repair works (months) which may also temporarily restrict access for fishers (traditional and commercial), tourism and recreational users.

### How will Santos manage the risk?

Preventative measures include pipeline design, targeted pipeline protection structures, and pipeline integrity management. Maritime notifications ensure marine users are informed of a gas release event, reducing the likelihood of unplanned interactions. The location of the Barossa GEP will be marked on nautical charts. The risk of impact to subsea infrastructure from dropped objects will be reduced through implementing procedures and standards for lifting equipment, IMMR, and contractor management. Emergency response procedures, pipeline depressurisation procedures (stop gas from flowing into the pipeline) and emergency pipeline repair plans will be implemented to minimise impacts in the event of a loss of containment from the Barossa GEP.



## **CONDENSATE RELEASE**

### How could a condensate release occur?

Barossa condensate has the potential to be released to the marine environment under several scenarios. Of those scenarios, three worst-case events are summarised below.

- In the event of a vessel collision (e.g. those described above for marine diesel oil or marine gas oil) which ruptures the FPSO condensate storage tank.
- In the event of an impact to, or failure of the subsea hydrocarbon containing equipment.

3. In the event of an impact to, or failure of multiple production well barriers.

Other scenarios exist that may result in other smaller condensate releases to the marine environment. All scenarios are very low probability of occurring.

### What environmental impacts could occur?

Condensate, being a lighter hydrocarbon behaves in a similar fashion to marine diesel when released to the marine environment. The fate of the condensate will depend greatly on the proportion on the surface, which will be transported by prevailing currents and wind and can evaporate readily. Condensate that entrains or dissolves in the water column will be transported by prevailing current and, hence, will follow a different path.

As with the marine diesel oil, the heavier components contained in the condensate will have a strong tendency to physically entrain into the upper water column but can re-float to the surface if these energies abate.

Such releases will impact the marine environment much in the same way as marine diesel oil and marine gas oils, described above. However, in the event of a subsea condensate release, more entrainment of hydrocarbons in the water column could occur, rather than being present on the sea-surface.

### How will Santos manage the risk?

In the unlikely event a vessel collision occurs which ruptures an FPSO condensate tank, Santos will manage the risk in accordance with the accepted Production Operations Oil Pollution Emergency Plan.

In the unlikely event of a subsea release in the OA1 area from subsea flowlines or wells, Santos has a range of operating procedures and plans to ensure that the integrity of the subsea infrastructure is maintained and well barriers are in place. Santos will submit a well operations management plan (WOMP) to NOPSEMA that will contain the full details of systems in place to ensure well design and integrity is managed for the well lifecycle. All production wells must be in compliance with the NOPSEMA accepted WOMP at all times. Hydrocarbon containing subsea infrastructure is also within a petroleum safety zone (PSZ), which third party vessels are not permitted to enter, subsequently reducing any interaction with this infrastructure.

Santos is developing response plans which will detail the actions to take to control a release and manage cleanup activities in the unlikely event of a release.



## MARINE DIESEL SPILL

### What environmental impacts could occur?

Although highly unlikely, a spill from a collision between two vessels could rupture a fuel tank resulting in the release of vessel fuel to the sea. This would impact water quality and may cause chemical/physical impacts to marine species.

### How will Santos manage the risk?

The risk of collision is reduced by managing interactions with marine users before and during the activity, with maritime notifications, automatic identification systems, navigational lighting, and exclusion zones in place. Operational procedures are designed to minimise refuelling incidents and spill response plans will be in place.



## SPILL RESPONSE OPERATIONS

### What environmental impacts could occur?

If a spill occurs, response operations may be required at any location surrounding the Operational Area. Potential environmental impacts include those listed in the Planned Activities table.

### How will Santos manage the risk?

Santos will rely on its Oil Pollution Emergency Plan (OPEP) to manage the impacts from a spill response event. Control measures would include:

- + Procedure for interacting with marine fauna
- Chemical selection process
- Minimum lighting
- Air pollution prevention certification
   Sewage and oily water treatment systems on vessels
- Notification to agreed stakeholders
- Notification to agreed stakeholders



## **HEAVY FUEL OIL**

### How could a heavy fuel oil release occur?

Heavy fuel oil is only used as fuel for offtake tankers who enter the Barossa field periodically (approximately once every three months) to load condensate from the FPSO. The only scenario that could lead to a release of heavy fuel oil is in the unlikely event of a vessel collision (described above for marine disel oil or marine gas oil), where the offtake tanker hull and heavy fuel oil tank is ruptured.

### What environmental impacts could occur?

Heavy fuel oil is heavier and more persistent than marine diesel oils, marine gas oils and condensates. The fuel is often characterised by a very high density and a high dynamic viscosity, which does not evaporate as quickly as other lighter fuels. As the fuel has a high residual component, a portion is expected to become semisolid and can persist in the marine environment for extended periods.

Such releases will cause a decline in water quality and may cause chemical (e.g. toxicity) and physical impacts to marine species (e.g. ingestion of hydrocarbons, physical coating). The severity of the impact depends on the magnitude of the release (i.e. extent, duration) and sensitivity of the receptor, however, may include those to the physical environment, threatened or migratory marine fauna, protected and significant areas and socioeconomic receptors (fisheries, tourism, recreation, cultural features and other oil and gas operators). Given the persistent and sticky nature of heavy fuel oil, there is a higher risk of coating of the physical environment (e.g. shorelines) and marine fauna compared to the lighter fuels such as marine diesel oil and marine gas oil.

### How will Santos manage the risk?

Offtake tankers are third-party operated vessels. They are vetted following Santos' marine assurance procedure and international guidelines before acceptance for condensate offtake operations at the Barossa field. The use of tankers with double hulls and fully segregated ballast tanks is not only a requirement of the vetting process; it is a MARPOL requirement that is monitored by way of regular statutory inspections.

All offtake loading events are planned in advance, occur within a petroleum safety zone (PSZ), and are performed under strict operational procedures.

Offtake tankers are third-party operated vessels. They are vetted following Santos' marine assurance procedure and international guidelines before acceptance for condensate offtake operations at the Barossa field. The use of tankers with double hulls and fully segregated ballast tanks is not only a requirement of the vetting process; it is a MARPOL requirement that is monitored by way of regular statutory inspections.

All offtake loading events are planned in advance, occur within a petroleum safety zone (PSZ), and are performed under strict operational procedures.



## CONTINGENCY SPILL RESPONSE OPERATIONS

In the event of a hydrocarbon spill, response strategies will be implemented to reduce environmental impacts to as low as reasonably practicable. The selection of strategies will be undertaken using the Net Environmental Benefits Assessment (NEBA) process. Spill response will be under the direction of the relevant control agency, as defined in the Production Operations Oil Pollution Emergency Plan (OPEP), which may be Santos, another agency or both. In all instances, Santos will undertake a 'first-strike' spill response and will act as the control agency until the designated control agency assumes control. The response strategies considered to be appropriate for the worst-case spill scenarios identified for the activity are detailed in the OPEP and comprise:

- source control (blowout preventer, relief well)
- monitor and evaluate
- mechanical dispersion
- shoreline protection and clean up
- oiled wildlife response
  scientific monitoring
- waste management.

Response strategies are intended to reduce the environmental consequences of a hydrocarbon spill, but poorly planned and coordinated response activities can result in a lack of, or inadequate, information being available, upon which poor decisions can be made, exacerbating or causing further environmental harm.

#### What impacts are expected?

Spill response operations may be required at any location within the EMBA. Potential environmental impacts include:

- Noise and light emissions generated by response vessels and equipment which may impact marine fauna, such as fish (including commercial species), marine reptiles and marine mammals.
- Atmospheric emissions generated from response equipment and vessels are expected to be localised and are not considered to create emissions on a scale where noticeable impacts would be predicted.
- Operational discharges and waste generated from response equipment and vessels are expected to be consistent with those of normal commercial vessel operations and may create a localised and temporary reduction in marine water quality. Cleaning of hydrocarbon-contaminated equipment, vehicles and vessels has the potential to spread hydrocarbons from contaminated areas to areas not impacted by a spill. Sewage and other waste will be generated from offshore activities at temporary staging/mooring areas, which may include toilet and washing facilities. These wastes have the potential to impact water quality, impact habitats, and reduce the aesthetic value of the environment, which may be within protected areas.
- Physical presence and disturbance operating vessels during spill response operations has the potential to disturb the physical environment and marine habitats and fauna (e.g. vessel strike, behavioural changes) or cause disruption to other marine users, coastal areas, townships and commercial fishing.



### How will Santos manage the risk?

Santos will rely primarily on the implementation of the Production Operations OPEP to manage the potential impacts associated with a spill response event. Other control measures that would be implemented include:

- · procedure for interacting with marine fauna
- chemical selection process
- · minimum lighting to meet maritime safety and navigation requirements
- air pollution prevention certification
- · sewage and oily water treatment systems on vessels
- notify agreed stakeholders.

The implementation of spill response activities to reduce the potential impacts from a spill are required by legislation. The spill response options selected will be demonstrated to show a net environmental benefit, are standard industry practice and are consistent with relevant standards and guidelines, including the National Plan for Maritime Environmental Emergencies. The controls proposed are intended to reduce the consequences of the potential impacts to minor and as low as reasonably practicable and an acceptable level.

## SUMMARY OF SANTOS' RISK MANAGEMENT STRATEGY

Santos has a management system that includes specific measures, to be used for the duration of the Production Operations Activity, which seek to confirm that:

- environmental impacts and risks continue to be identified for the duration of the activity and are reduced to as low as reasonably practicable and acceptable levels
- control measures are effective in reducing environmental impacts and risks to as low as reasonably practicable and acceptable levels
- · environmental performance outcomes and standards set out in the EP and OEMP are being met
- there will be ongoing appropriate consultation with relevant authorities and other relevant interested persons or organisations
- · the roles, accountabilities and responsibilities are defined and understood
- workforce training is completed and competencies assured
- emergency preparedness and response arrangements are in place
- incident reporting, investigation and follow-up is monitored
- audits, inspections, reporting and notifications and document management are appropriately undertaken.

## **APPROVALS PROCESS**

Production Operations Activities detailed in this booklet require a number of regulatory approvals. Primary environmental approvals required for Production Operations Activities are outlined below:

- An Offshore Project Proposal (the Barossa Offshore Project Proposal (OPP)) was developed for the Commonwealth waters component the Barossa Project and was accepted by NOPSEMA in March 2018. The Barossa OPP, at the time of submission, excluded approximately 23 km of GEP in Commonwealth waters which is subject to a separate EPBC Act approval process (refer below).
- A referral under the EPBC Act, covering the installation, operation and decommissioning of the remaining approximately 23 km of GEP in Commonwealth waters and the 100 km section of GEP in NT waters (inclusive of the 8.26 km in NT coastal waters) was submitted to the Department of Climate Change, Energy, the Environment and Water (DCCEEW) for assessment. The activity (referred to as the Darwin Pipeline Duplication Project) was determined to be a 'controlled action' under the EPBC Act and is currently being assessed on preliminary documentation.
- A referral under the NT Environment Protection Act 2019 (EP Act) for the construction, operation and decommissioning of the 100 km section of GEP in NT waters (part of the Darwin Pipeline Duplication Project) was submitted to the NT Environment Protection Authority (EPA) and was subsequently assessed by way of Supplementary Environmental Report. On 22 December 2023, the NT Minister for Environment, Climate Change and Water Security approved the action the subject of the referral, on the recommendation of the NT EPA.

In addition to the primary environmental approvals outlined above, activity-specific Environmental Plans (EPs) meeting the requirements of the Offshore Patroleum and Greenhouse Gas Storage (Environment) Regulations 2023 (Cth) (OPGGS Environment Regulations) are required. For Production Operations activities, the OPGGS Environment Regulations apply to the activities within OA1 in Commonwealth waters and OA2, spanning both Commonwealth waters (285 km) and coastal waters of the NT (8.26 km). The OPGGS Environment Regulations set out that an EP must (among other things):

- · comprehensively describe the activity to be carried out under the EP
- describe the environment that may be affected by the activity, including the values and sensitivities of that environment
   detail and evaluate the environmental impacts and risks for the
- relevant activity • demonstrate that the impacts and risks of the activity will be reduced to as low as reasonably practicable and an acceptable level (and detail the control measures to be used to achieve this)
- demonstrate that Santos has consulted, in accordance with regulatory requirements, with each relevant person, including those whose functions, interests or activities may be affected by the activities to be carried out under the EP
- demonstrate that the measures (if any) that Santos has adopted, or proposes to adopt, because of the consultations are appropriate

Santos is currently preparing the Production Operations EP for submission to NOPSEMA, covering Commonwealth waters Production Operations activities in OA1 and OA2.

Santos is also preparing an Operations Environmental Management Plan (OEMP) to cover the operation of the GEP in NT waters for submission to DITT. The OEMP will also cover the operation of the 8.26km GEP in NT coastal waters, under the PSL Act and OPGGS Environment Regulations and operation of the remaining -92km GEP covered under the Energy Pipelines Act.

In order to meet its proposed schedule for the Barossa Gas Project, Santos is aiming to submit the Productions Operation EP to NOPSEMA and the OEMP to DITT in 2024 and, subject to regulatory acceptance, commence activities in 2025. The timeline for consultation has been developed by Santos to meet this objective, while still providing a reasonable period for meaningful consultation with relevant persons, having regard to Santos's regulatory obligations and to feedback from relevant persons.

## SEEKING INFORMATION AND WHAT'S NEXT

Santos is continuing its Barossa Gas Project consultation efforts to further learn, understand and assess values and sensitivities of the environment that may be affected by our proposed activities, and potential environmental impacts and risks. There may be information Santos is not yet aware of but needs to properly understand to assess potential activity impacts and risks. Consultation may inform this. It may also inform what control measures are to be proposed to reduce environmental impacts and risks to as low as reasonably practicable and to an acceptable level.

Santos is consulting on both the Production Operations EP (Commonwealth waters) and OEMP (NT waters) at the same time.

Scan this QR Code for more information on Barossa Production Operations Activity:



### YOUR INPUT IS IMPORTANT TO SANTOS:

In preparing an EP for submission to NOPSEMA, a titleholder must consult with each 'relevant person', including relevant Commonwealth, State and Northern Territory Departments or agencies and persons (or organisations) whose functions, interests or activities may be affected by the activity proposed to be carried out under an EP.

Relevant persons being consulted on EPs under the Offshore Petroleum and Greenhouse Gas Storage (Environment) Regulations 2023 (Cth) (OPGGS Environment Regulations) should note that they:

- are entitled to be given sufficient information to allow them to make an informed assessment of the possible consequences of the activity on their functions, interests or activities;
- are entitled to be allowed a reasonable period for the consultation; and
- may request particular information provided in consultation not be published.

If you request particular information not to be published, Santos will respect and abide by your request. Any information not to be published will be provided to NOPSEMA in a confidential report, separate from the published EP.

Your input is important to Santos:

- so that we can understand the environmental values in the OAs and the environment that may be affected, the environmental impacts and risks associated with the activity, to inform development of the Productions Operations EP (Commonwealth waters) and OEMP (NT waters);
- to inform how consultation processes may need to be adapted for different relevant persons; and
- to ensure that we provide information to relevant persons in an appropriate and accessible manner.

If you think you may be a relevant person for the purposes of one of Santos' proposed activities, please contact Santos on: 1800 267 600 or email offshore.consultation@santos.com to seek to be included in consultations and to provide

feedback on how you would like to be consulted (if a relevant person).

This can also be done using the form available by scanning the QR Code below:



Visit www.santos.com/barossa for more information on the Barossa Gas Project.

## **Closeout email**

This email or the content it contains was sent as appropriate to relevant entities.

Good afternoon,

We refer to our previous correspondence regarding consultation for environment plans for Santos' Barossa Project Operations activities in Commonwealth waters and Northern Territory waters.

Between February and May this year, Santos provided opportunities for your organisation to seek to participate in consultation and provide input regarding these activities, the environment that may be affected by the proposed activities, and the environmental impacts and risks associated with the proposed activities.

Santos would like to thank you for your response and any input provided to date. With the consultation period now complete we consider that consultation has now closed for the purpose of Santos finalising and submitting environment plans for these activities to government regulators for assessment.

Regards

Barossa Consultation Coordinator Email: offshore.consultation@santos.com

Phone: 1800 267 600



## **Video/Animation**



**Barossa Production Operations** 

Santos SantosLtd Subscribe



## Photos, posters and iPad-based server-documents

The consultation setup, visual tools (posters) and document tools (iPad server-based documents) made available to relevant persons attending consultation workshops are shown below.



Consultation Session Set Up: 21 May 2024, Pirlangimpi, Sports & Social Club



Consultation Session Set Up: 15 May 2024, Wurrumiyanga, Mantiyupwi Motel



Consultation Session Set Up: 15 May 2024, Wurrumiyanga, Mantiyupwi Motel



Consultation Session Set Up: 6 March 2024, Wurrumiyanga



Consultation Session Set Up: 7 March 2024, Wurrumiyanga



Consultation Session Set Up: 10 April 2024, Wurrumiyanga



Consultation Session Set Up: 10 April 2024, Wurrumiyanga



Consultation Session Set Up: 5 March 2024, Milikapiti, Sports & Recreation Centre



Consultation Session Set Up: 5 March 2024, Milikapiti, Sports & Recreation Centre



Consultation Session Set Up: 5 March 2024, Milikapiti, Sports & Recreation Centre



Wed 22 May: 10.30am – Wurankuwu Clan & Malawu Clan\*\* (came to one joint session), Wurrumiyanga, Mantiyupwi Motel



Consultation Visual Tool: Photo of the A0 sized Production Operations Laminated Poster utilised at each consultation session



Consultation Visual Tools: Photo of the A0 sized Barossa Overview Laminated Poster utilised at each consultation session & NOPSEMA Consultation Brochures



Consultation Visual Tool: Photo of the A0 sized Laminated Poster (Pipe Weld on pipelay vessel) utilised at each consultation session



Consultation Tool: Photo of the A0 sized Laminated Poster (Pipelay Control Room, Underwater Images, ROV) utilised at each consultation session


Consultation Tool: A0 sized FPSO Construction Laminated Poster utilised at each consultation session



Consultation Visual Tool: A4 sized Oil and Condensate Laminated Image

+ Create or upload	
Gabor Feher My files	
A Home	
Image: My files     Image: My files     Modified ∨     File size ∨	Sharing
Photos         BAU Pipeline         ··· 12         4/25/2023         510 KB	😂 Shared
83 Shared Consult Material 5/7/2024 242 MB	😂 Shared
Dec 2023 12/4/2023 1.03 GB	Private
People     Documents     3/14/2023     102 KB	8 Shared
March 2024 Less than a minute 1.63 GB	😤 Shared
Arr May 2024 5/12/2024 2.10 GB	😂 Shared
Personal Vault 3/18/2023	Private
Barossa 3D Depth structure map.png 6/14/2023 1.15 MB	Private
MG_2193JPG 6/14/2023 5.52 MB	Private
MG_2199JPG 6/14/2023 4.44 MB	Private
MG_2256JPG 6/14/2023 7.19 MB	Private
MG_2259JPG 6/14/2023 6.26 MB	Private
kit box.PNG 6/15/2023 325 KB	Private
Kit.PNG 6/15/2023 534 KB	Private

Consultation Tools: Server Based Documents Prepared for IPAD Uploads, 2023 & 2024

			100	10000000000
10:51am Thu 5 Dec				4514
G Files				+
Name 1	Date	Size	Sharing	00
BAU Pipeline	25 Apr 2023	510 KB	a Shared	
Consult Material	7 May	241.6 MB	Shared	
Dec 2023	4 Dec 2023	1.03 G8		
B Documents	14 Mar 2023	102 KB	© Shared	
March session videos	23 Apr 2023	1.63 GB	# Shared	
May 2024	12 May	2.1 GB	a Shared	
Personal Vault				
Barossa 3D Depth structure map	14 Jun 2023	1.1 MB		
IM6_2193	14 Jun 2023	5.5 MB		
IMG_2199	14 Jun 2023	4.4 MB		
MG_2256	14 Jun 2023	7.2 MB		
IMG_2259	14 Jun 2023	6.3 MB		
1 items r	remaining to download.			
n Hone Files	(@) Estare		Photos	
		161-	1:1	
	Y IL IL I			
				L
caps lock A S D F G	н ј к	L		return
shift Z X C V B	NM	< > .	5	shift
Control option cmd		H Z cmd op	tion	

Photo of IPAD with 2023 & 2024 Documents Uploaded



8 People

Consultation Tools: Server Based Documents Prepared for IPAD Uploads, March 2024

S & Marr	ch session videos				4%
Contrait C	Personal				Q
Name	↑ Date	Size	Sharing		00
1.0 Consultation video (Jeffrey)	23 Apr 2023	458.4 MB	≅ Shared		
1.2 Condensate vs oil video	23 Apr 2023	67.1 MB	# Shared		
1.7 Drilling EP animation [FINAL]	23 Apr 2023	371.6 MB	a Shared		
Animation_Closer	23 Apr 2023	3.3 MB	a Shared		
FPSO 10000 Y storm event	23 Apr 2023	12.1 MB	a Shared		
Dipe laying animation	23 Apr 2023	6.8 MB	# Shared		
Santos-Barossa-Stakeholder-Engagement-v05_20221216.2	23 Apr 2023	402.5 MB	# Shared		
Barossa Stakeholder EngagementSafety (extended) [2023-03-17] (1)	25 Apr 2023	36.5 MB	# Shared		
Barossa Production Ops - [2024-03-07]	5 Apr	308.8 MB	# Shared		
🕞 Home 🖬 Files	0	ee shared (	Distore		
				1	
Contraction and the second sec					

Photo of IPAD with March 2024 Documents Uploaded

::: OneDrive				
+ Create or upload	🖄 Share 🚽 Download	Create album from folde	er	
Gabor Feher	My files > Dec 2023			
G Home				
My files			No. 1	
Photos	Steel Cement			
es Shared	Gas	and the second		in.e
🗓 Recycle bin	1.1 Barossa Gas Project 12/4/2023	Marine Seismic Surveys 12/4/2023	Sea Surface Exposure Ty 12/4/2023	S0155 - Barossa Animati 12/4/2023
Browse files by				

Consultation Tools: Server Based Documents Prepared for IPAD Uploads, December 2023



Photo of IPAD with December 2023 Documents Uploaded

::: OneDrive					P Se
+ Create or upload	are 🛓 Download 📳 Create album from folder				
Gabor Feher My fi	les > May 2024 &				
My files	Ď Name ↑ ~		Modified ~	File size $$	Sharing
Photos	1.FAQ		5/12/2024	34.6 KB	8 Shared
83 Shared	2.FAQ		5/12/2024	46.7 KB	88 Shared
W Recycle bin	all 3.FAQ.		5/12/2024	54.5 KB	8s Shared
People	5.F May 2024 Tiwi consultation script_Manupi (with DPD)_FINAL CLEAN.docx		5/12/2024	220 KB	88 Shared
0	5.F Prod OPS 2024 Consultation Presentation Tiwi Manupi (with DPD) FINAL CLEAN.pptx	··· 🖻	5/17/2024	1.18 GB	😂 Shared
	6.F May 2024 Tiwi consultation script_Production Ops_FINAL CLEAN.docx		5/12/2024	216 KB	89 Shared
	6.F Prod OPS 2024 Consultation Presentation_Tiwi_FINAL CLEAN.pptx		5/12/2024	632 MB	8s Shared
	8. Prod OPS 2024 Consultation Presentation_Tiwi_Script.docx		5/21/2024	208 KB	89 Shared
	Barossa Production Ops - [2024-03-07].mp4		4/7/2024	309 MB	8s Shared
	HFO_450m3 Winter Run83_Surface_Shoreline_Animation (3fps).mp4		4/5/2024	2.13 MB	용 Shared

#### Consultation Tools: Server Based Documents Prepared for IPAD Uploads, May 2024



Photo of IPAD with May 2024 Documents Uploaded

OneDrive				
+ Create or upload	🖄 Share 🚽 Download 🔳 Create album from folder			
Gabor Feher	My files → Consult Material &			
My files	O D Name ↑ ~	Modified ~	File size $$	Sharing
A Photos	A598152 - OPP APPENDICES.pdf	5/7/2024	89.9 MB	응 Shared
Shared	A598153 - OPP.pdf	5/7/2024	21.7 MB	용 Shared
U Recycle bin	A715657 - GEP EP.pdf	5/7/2024	27.4 MB	응 Shared
Browse files by	A720308 - GEP EP APPENDICES.pdf	5/7/2024	16.3 MB	& Shared
O rope	A1033886 - Drilling EP.pdf	5/7/2024	48.5 MB	용 Shared
	A1051117 - SURF EP.pdf	5/7/2024	37.9 MB	සී Shared

Consultation Tools: Server Based Documents Prepared for IPAD Uploads, May 2024



Photos of IPAD with May 2024 Documents Uploaded

## **PowerPoint presentation**

The content of this presentation was presented to various relevant persons throughout the consultation period.



## **Acknowledgement of Country**

I would like to begin by acknowledging the Traditional Custodians of the land and water on which work, and where we meet today.

We pay our respects to Elders past, present and emerging, and extend that respect to all Aboriginal and Torres Strait Islander peoples here today.



Santos

## **Welcome & Introductions**

We are here today to share information about our company and operations, seek information from you and listen to your questions about Santos and upcoming projects.



Santos

# SANTOS OPERATIONS & DARWIN LNG

# SANTOS - South Australia, Northern Territory Oil Search



Santos is a global energy company committed to increasingly cleaner energy and fuels production, with operations across Australia, Papua New Guinea, Timor-Leste and North America (Alaska).

For 70 years, Santos has been working in partnership with local communities, providing local jobs and business opportunities, safely developing its natural gas resources, and powering industries and households.



At Santos, our commitment is to be a global leader in the transition to cleaner energy and clean fuels, by helping the world decarbonise to reach net-zero emissions in an affordable and sustainable way.



Santos is one of Australia's biggest domestic gas suppliers and a leading LNG supplier in the Asia Pacific region.



We are committed to supplying critical fuels in a more sustainable way through decarbonising projects.



Our business focus: Safe, reliable operations & Minimise our social and environmental impacts.

## **Darwin LNG Facility & Operations**

Santos

6

7

- Located in Darwin at Wickham Point, Darwin LNG (DLNG) is a gas liquefaction and storage facility that started production in 2006.
- Gas has been provided to the Darwin LNG from the Bayu-Undan gas field, located in between Australia and Timor Leste. Darwin LNG has shipped its 835th and final LNG cargo from the Bayu-Undan field.
- DLNG generates about \$100 million a year in supply and service opportunities for Territory businesses.
- 100% Darwin residential Santos employee workforce and approx 180 local Darwin people work at the LNG Facility.
- DLNG established the NT's first LNG Process Operator Traineeships in 2010. 10% of those who have completed identified as an Aboriginal or Torres Strait Islander.



## Santos

## BAROSSA GAS PROJECT OVERVIEW & UPDATE

#### **Barossa Project Update**

- The drill rig has finished drilling the second well to just above the reservoir and installed plugs.
- A Tubing Head Spool has been installed on the second well.
- The rig is moving back to the first well to install the Christmas
- Tree.
  The Christmas tree is series of big valves and it works like a tap to open and close the well.



Christmas Tree Valves

STP Buoy tow to site is complete Mooring line hook up has commenced. Flowline installation vessel has arrived on site and commenced flowline installation operations.

#### **Barossa Project Update**



WIN

Derrick structure on drill rig

#### Santos

 15x FPSO mooring anchor/chain installed & tensioning complete
 PLET installed at each

 PLET installed at eac end of pipeline



- Baseline environmental monitoring program completed around Darwin harbour.
- Water quality, habitat monitoring & fish surveys

Joint Program funded by Barossa with the expertise of RPS, Curtin University, focus subsea, Larrakia Rangers.

- Installation of the 262km Gas Export Pipeline is complete up to DPD PLET.
- Audacia (pipelay vessel) has departed Australia.
   Fortitude is conducting post-lay works on the pipeline and will soon commence pipeline testing.

## **Barossa Project Update**

#### Santos



 ROV survey of hull for marine biofouling complete and a report will be prepared with that data.

## CONSULTATION SESSION

Consultation – reg 25, Offshore Petroleum and Greenhouse Gas Storage (Environment) Regulations 2023 (Cth)

(previously known as reg 11A, *Offshore Petroleum and Greenhouse Gas* (Environment) Regulations 2009 (Cth))

## Santos Consultation Introduction

Santos

12



#### Santos

## **Privacy Statement**

Santos Ltd and its related bodies corporate (together, we, our, us or Santos) collect personal information about you, such as your name and sensitive information about your indigenous heritage. We use this information to record your attendance at any meeting or other discussion with us, to provide you with information about our projects, to receive and respond to any information that you provide, to answer any questions you might have and for other purposes that we tell you about during your meeting or other discussion with us. Santos will handle any information that you provide in accordance with our Code of Conduct and our Confidentiality, IP and Privacy Procedure. You can ask us for a copy of this Privacy Notice or these other documents.

If you do not provide your personal information, we may not be able to identify you as the person who provided particular <u>information</u> or we may be unable to discuss any information you have provided with you further or respond to your questions. We may disclose your information to other companies within the Santos group, to third parties that help us run our business and to relevant government agencies and government departments.

Due to the global nature of our operations and business, your personal information may be accessed by or disclosed to <u>Santos</u> personnel outside Australia. We may also use overseas third parties to collect, transfer, store and handle your personal information. Some of the overseas countries that your personal information may be accessed from, disclosed or transmitted to or stored in include but are not limited to, Papua New Guinea and the United States of America.

You have a right to request a copy of any personal information that we hold about you, as well as a right to request that we correct any information that we hold about you that is inaccurate, out-of-date, incomplete, irrelevant or misleading. You can also make a complaint about how we have handled your personal information. Our Consultation Privacy Policies explain in more detail how you can exercise these rights, including how we will respond to your access or correction request or to any privacy complaint that you make. The Barossa Gas Project Consultation-privacy-policy, and the Northern Territory Consultation Privacy Policy is available on our website at https://www.santos.com/offshore-wa-and-nt-consultation-privacy-policy/. You can also contact us to request copies be provided to you.

You can contact us by:

- posting a letter addressed to us at 60 Flinders Street, Adelaide, South Australia, 5000;
- telephoning us on +61 8 8116 5000; or
- sending us an email at offshore.consultation@santos.com and compliance@santos.com.

# Santos BAROSSA PRODUCTION OPERATIONS ACTIVITY OVERVIEW

## Production Operations Activity - Summary Santo

Key A	ctivities	GEP Commonwealth W (OPGGS Act)	aters (A)	
Arrival FPSO	arrival & connection to mooring buoy		Level 1	B NTA1
Hook-up & Commissioning FPSO systems & subsea equipment	Commissioning activities (testing) and prepare for start-up	ENC.		
Production Operations FPSO systems & subsea equipment	gas & condensate is processed on the FPSO	GEP NT Coastal Waters (PSL Act)		Oth Compose all a
Offtake operations FPSO & tankers	condensate stored on the FPSO is removed 4- 5 times a year by offtake tankers			(then
Gas Export Pipeline Operations	dry gas is sent to DLNG via the GEP			Twilslands
Support Operations Vessels	transport food & materials			( o
Support Operations Helicopters	transport people to/from FPSO			CEP NI autor
Support Operations- IMMR GEP & subsea	planned underwater visual Inspection, Maintenance, Monitoring and Repair activities			DARWIN
equipment	using remotely operated vehicles	Santos	PRODUCTION OPERATIONS ACTIVITIES	Opatial Reference GDA2020 :

## **Production Operations Activity - FPSO**



- 359m long or approx. 2.5 MCG football fields
- 64 m wide ,
- Houses 140 people

- Total of 11.3 km in length
- Weigh 5500 tonnes or
- approx. 2200 land cruisers
- Anchor piles
   sink into seabed & attach to mooring buoy
- 19m high x 7m round
- Weigh 140 tonne each

## **Production Operations Activity - FPSO**

Santos



#### Santos

## BAROSSA PRODUCTION OPERATIONS PLANNED ACTIVITIES IMPACTS & **RISKS**

Presented by Barossa Team

#### **Production Operation Activity Impacts**

• Froduction		Santos
Planned Activities	Potential Impact	How we Manage (the rules we follow)
Light emissions	Behavioural impact to marine life (e.g., attraction)	We only use lights where needed for safe operations and to comply with relevant safety rules. We turn off lights when not required.
Noise emissions	Behavioural impact to marine life (e.g., avoidance)	The FPSO has been designed to reduce noise emissions. We follow procedures and look out for marine life (e.g. whales, dolphins, turtles) and we slow down and move away from them where possible if they are too close. Boats and equipment will be maintained to minimise noise.
Seabed disturbance (from inspections, maintenance, monitoring and repair)	Temporary loss of habitat	We will accurately and safely place infrastructure using dynamic positioning to <u>minimise</u> seabed disturbance. Records of all installed equipment will be kept so it can be removed during decommissioning.
Discharges (produced water, discharges from the FPSO, subsea equipment and boats)	Impact to water quality	We will manage discharges to acceptable levels and follow the standard rules for what boats can discharge. We will select chemicals that are environmentally acceptable, and we will limit their use to only what is needed. We will use a water treatment system for produced water discharges and the water will regularly be tested before release.
Air emissions	Impact to air quality	The FPSO has been designed to reduce air emissions. We ensure engines and other equipment are looked after and low emission fuel will be used.
Greenhouse gas emissions	Insignificant contribution to national and international greenhouse gas levels from direct and indirect emissions	The FPSO has been designed to reduce greenhouse gas emissions and we will have a Greenhouse Gas Management Plan to minimise greenhouse gas emissions over the life of operations. We will comply with the Australian Government Safeguard Mechanism and also legal and regulatory requirements for emissions reporting.
Physical presence (disturbance to other boats)	Exclude other users from activity area	We will communicate to other boats where we are and what we are doing. We mark the location of equipment on charts.

## Santos BAROSSA PRODUCTION OPERATIONS UNPLANNED EVENTS IMPACTS & RISKS

Presented by Barossa Team

## **Production Operation Activity Impacts**

Planned Activities	Potential Impact	How we Manage (the rules we follow)
Light emissions	Behavioural impact to marine life (e.g., attraction)	We only use lights where needed for safe operations and to comply with relevant safety rules. We turn off lights when not required.
Noise emissions	Behavioural impact to marine life (e.g., avoidance)	The FPSO has been designed to reduce noise emissions. We follow procedures and look out for marine life (e.g. whales, dolphins, turtles) and we slow down and move away from them where possible if they are too close. Boats and equipment will be maintained to minimise noise.
Seabed disturbance (from inspections, maintenance, monitoring and repair)	Temporary loss of habitat	We will accurately and safely place infrastructure using dynamic positioning to minimise seabed disturbance. Records of all installed equipment will be kept so it can be removed during decommissioning.
Discharges (produced water, discharges from the FPSO, subsea equipment and boats)	Impact to water quality	We will manage discharges to acceptable levels and follow the standard rules for what boats can discharge. We will select chemicals that are environmentally acceptable, and we will limit their use to only what is needed. We will use a water treatment system for produced water discharges and the water will regularly be tested before release.
Air emissions	Impact to air quality	The FPSO has been designed to reduce air emissions. We ensure engines and other equipment are looked after and low emission fuel will be used.
Greenhouse gas emissions	Insignificant contribution to national and international greenhouse gas levels from direct and indirect emissions	The FPSO has been designed to reduce greenhouse gas emissions and we will have a Greenhouse Gas Management Plan to minimise greenhouse gas emissions over the life of operations. We will comply with the Australian Government Safeguard Mechanism and also legal and regulatory requirements for emissions reporting.
Physical presence (disturbance to other boats)	Exclude other users from activity area	We will communicate to other boats where we are and what we are doing. We mark the location of equipment on charts.

## Production Operations Spill Animation – typical dry season simulation



Santos

26

## CLOSE OF CONSULTATION

Santos

## **COMMUNITY UPDATE**

**Employment and Training** 

#### Follow Up Answers to Questions Community Update

Торіс	Question
	How many First Nations people are employed on the Barossa Gas Project?
Employment &	How many First Nations People are employed with Santos altogether?
Opportunities	How do we find out about any new employment opportunities?

How do we apply for jobs and traineeships?

## **Employment Outcomes**

#### Pipelay Cultural Observers & Monitors:

- 5 Tiwi People engaged as Cultural Observers & Monitors on Allseas Pipelay and Survey Vessels.
- Review ROV Footage during surveys and provide support if any finds on seabed while the Barossa pipeline is being laid.

#### **Drilling Rig Assistant Roles:**

 7 First Nations people from Larrakia and Tiwi are employed in entry level offshore roles on the Santos contracted drill rig for Barossa.

#### Santos Offshore Health, Safety & Environment Advisor Traineeships

- 2 x HSE Advisor Traineeships with Santos, in partnership with Programmed Skilled Workforce to obtain a Certificate IV in HSE (<u>12-18 month</u> program).
- Recruitment process in final stages, with successful candidates on track to commence before July 2024.



Santos

## BAROSSA ABORIGINAL FUTURE FUND

**Community Benefits** 

#### Employment & Training Program Outcomes

#### Santos-Kaefer (funded by Darwin LNG)

- Support Indigenous peoples by equipping participants with skills and knowledge to prepare them for long-term employment opportunities.
- Onsite training in Darwin at Darwin LNG and Kaefer workshop in Darwin.
- Training areas include scaffolding, HSE Traineeship, University Degree (HR or Mech Engineering), Accounting/ Bookkeeping Traineeship, Trade Apprenticeship
- Successful candidates are employed by Kaefer.
- Applications for 2024 opportunities have now closed.
- If you or a family member would like to register interest for future opportunities, please contact Kaefer via:
- FNrecruitment@kaefer.com.au
- k.thorn@kaefer.com.au
- Kylie Thorn: 0474 159 206



# Santos is committed to supporting the communities where we operate

## NT Aboriginal Coastal Communities (including homelands & outstations)

The Barossa Aboriginal Future Fund (BAFF) will invest in areas that provide an enduring legacy for NT Aboriginal Coastal Communities by:

- · Improving community infrastructure
- Improving services that improve health, education, housing, community resilience and economic outcomes
- · Enabling communities to maintain cultural practices and carry out cultural obligations
- · Enabling communities to care for their country
- Building capacity to establish pathways to skilled, well-paying, secure jobs and business development opportunities.

NT Aboriginal Coastal Communities the Barossa projects operates in includes Tiwi Islands, Darwin-Daly-Wagait, West Arnhem, East Arnhem and Victoria River District.

# Santos Foundation - Supporting local PNG priorities

Healthcare, Community Development, Family & Sexual Violence, Youth Opportunities



We want to see the children excel in early childhood learning so they can continue to elementary schools equipped with the fundamental tools they need. Pale Mbipe of Hela Province, PNG Unlocking barriers to healthcare



#### "

We undertook repairs to the water supply, renovated the maternity ward, and repaired other buildings at the hospital. We will be able to diagnose and treat people very quickly now with these new services available in Koroba. Dr James Klintwa, Co Hele Provincial Health Authority and community leadership

Bel isi PNG case management



We nost events and learning sessions throughout the year aimed at raising awareness and imparting tools to address family violence. Leon Buskens, Santor Country Chair PNG

"

32

## **BAROSSA ABORIGINAL FUTURE FUND**

Santos wants to listen to NT Aboriginal Coastal Communities about the needs for their communities.

Santos is working with our Barossa Joint Venture Partners to establish legal structures with sound governance

The Santos NT Community Team will continue to visit Tiwi and other NT coastal regions even after consultation is complete.

If you want more information or want to share your ideas, you can visit the NT Community Team at the Santos shop in the Darwin Mall or contact us through:

35

Email: <u>enquiriesNT@santos.com</u> Phone: +61 8919 1900



## **PowerPoint presentation**

These slides were shown as part of close out meetings with Tiwi clans.



Santos

2

## **Acknowledgement of Country**

I would like to begin by acknowledging the Traditional Custodians of the land and water on which work, and where we meet today.

We pay our respects to Elders past, present and emerging, and extend that respect to all Aboriginal and Torres Strait Islander peoples here today.

## WELCOME AND INTRODUCTIONS

Santos

### **Welcome & Introductions**

Santos

We are here today to share information about our company and operations, seek information from you and listen to your questions about Santos and upcoming projects.



## Santos

## BAROSSA GAS PROJECT OVERVIEW & UPDATE

B

#### **Barossa Project Update**

- The drill rig has finished drilling the second well to just above the reservoir and installed plugs.
- A Tubing Head Spool has been installed on the second well.
  The rig is moving back to the first well to install the Christmas
- Tree.
- The Christmas tree is series of big valves and it works like a tap to open and close the well.



Christmas Tree Valves



Santos

6

#### **Barossa Project Update**

#### Santos



- STP Buoy tow to site is complete
- Mooring line hook up has commenced.
- Flowline installation vessel has arrived on site and commenced flowline installation operations.
- Installation of the 262km Gas Export Pipeline is complete up to DPD PLET. Audacia (pipelay vessel) has departed Australia.
- Fortitude is conducting post-lay works on the pipeline and will soon commence pipeline testing.

15x FPSO mooring anchor/chain installed •

DARWIN

- & tensioning complete PLET installed at each end of pipeline
- Baseline environmental monitoring program completed around Darwin harbour.
- Water quality, habitat monitoring & fish surveys

Joint Program funded by Barossa with the expertise of RPS, Curtin University, focus subsea, Larrakia Rangers.

## **Barossa Project Update- Anchor Installation**

A



## **Barossa Project Update**

#### Santos



• ROV survey of hull for marine biofouling complete and a report will be prepared with that data.

#### Santos

Planned Activities	Status
Bayu-Undan Gas Export Pipeline Operations (preservation phase)	Consultation closed. EP under assessment by NOPSEMA.
Tern-2 Plug and Abandonment	Consultation closed. EP under assessment by NOPSEMA.
Eos 3D Marine Seismic Survey	Consultation closed. EP to be submitted to NOPSEMA in Q2 2024.
G-11-AP Carbon Capture and Storage Appraisal Well (Astraea-1)	Consultation closed. EP submission to NOPSEMA in Q3 2024.



Santo

**Other Santos Activity Update 2024** 

Santos

#### CONSULTATION SESSION

Consultation – reg 25, Offshore Petroleum and Greenhouse Gas Storage (Environment) Regulations 2023 (Cth)

(previously known as reg 11A, *Offshore Petroleum and Greenhouse Gas* (Environment) Regulations 2009 (Cth))

### **Privacy Statement**

Santos Ltd and its related bodies corporate (together, we, our, us or Santos) collect personal information about you, such as your name and sensitive information about your indigenous heritage. We use this information to record your attendance at any meeting or other discussion with us, to provide you with information about our projects, to receive and respond to any information that you provide, to answer any questions you might have and for other purposes that we tell you about during your meeting or other discussion with us. Santos will handle any information that you provide in accordance with our Code of Conduct and our Confidentiality, IP and Privacy Procedure. You can ask us for a copy of this Privacy Notice or these other documents.

If you do not provide your personal information, we may not be able to identify you as the person who provided particular information or we may be unable to discuss any information you have provided with you further or respond to your questions. We may disclose your information to other companies within the Santos group, to third parties that help us run our business and to relevant government agencies and government departments.

Due to the global nature of our operations and business, your personal information may be accessed by or disclosed to Santos personnel outside Australia. We may also use overseas third parties to collect, transfer, store and handle your personal information. Some of the overseas countries that your personal information may be accessed from, disclosed or transmitted to or stored in include but are not limited to, Papua New Guinea and the United States of America.

You have a right to request a copy of any personal information that we hold about you, as well as a right to request that we correct any information that we hold about you that is inaccurate, out-of-date, incomplete, irrelevant or misleading. You can also make a complaint about how we have handled your personal information. Our Consultation Privacy Policies explain in more detail how you can exercise these rights, including how we will respond to your access or correction request or to any privacy complaint that you make. The Barossa Gas Project Consultation Privacy Policy is available on our website at https://www.santos.com/barossa/barossa/barossa-gas-project\_consultation-privacy-policy, and the Northern Territory Consultation Privacy Policy is available on our website at https://www.santos.com/offshore-wa-and-nt-consultation-privacy-policy/. You can also contact us to request copies be provided to you.

You can contact us by:

Santos

- posting a letter addressed to us at 60 Flinders Street, Adelaide, South Australia, 5000;
- telephoning us on +61 8 8116 5000; or
- · sending us an email at offshore.consultation@santos.com and compliance@santos.com.

13

#### BAROSSA PRODUCTION OPERATIONS ACTIVITY OVERVIEW

## **Production Operations Activity - FPSO**

Santos



#### **Production Operation Activity Impacts**

Santos

Presented by Barossa Team

Planned Activities	Potential Impact	How we Manage (the rules we follow)
Light emissions	Behavioural impact to marine life (e.g., attraction)	We only use lights where needed for safe operations and to comply with relevant safety rules. We turn off lights when not required.
Noise emissions	Behavioural impact to marine life (e.g., avoidance)	The FPSO has been designed to reduce noise emissions. We follow procedures and look out for marine life (e.g. whales, dolphins, turtles) and we slow down and move away from them where possible if they are too close. Boats and equipment will be maintained to minimise noise.
Seabed disturbance (from inspections, maintenance, monitoring and repair)	Temporary loss of habitat	We will accurately and safely place infrastructure using dynamic positioning to minimise seabed disturbance. Records of all installed equipment will be kept so it can be removed during decommissioning.
Discharges (produced water, discharges from the FPSO, subsea equipment and boats)	Impact to water quality	We will manage discharges to acceptable levels and follow the standard rules for what boats can discharge. We will select chemicals that are environmentally acceptable, and we will limit their use to only what is needed. We will use a water treatment system for produced water discharges and the water will regularly be tested before release.
Air emissions	Impact to air quality	The FPSO has been designed to reduce air emissions. We ensure engines and other equipment are looked after and low emission fuel will be used.
Greenhouse gas emissions	Insignificant contribution to national and international greenhouse gas levels from direct and indirect emissions	The FPSO has been designed to reduce greenhouse gas emissions and we will have a Greenhouse Gas Management Plan to minimise greenhouse gas emissions over the life of operations. We will comply with the Australian Government Safeguard Mechanism and also legal and regulatory requirements for emissions reporting.
Physical presence (disturbance to other boats)	Exclude other users from activity area	We will communicate to other boats where we are and what we are doing. We mark the location of equipment on charts.

#### Santos

#### Follow Up Answers to Questions

#### **Planned Activities**

Торіс	Question
	Is there a limit on how many times Santos can use the flare on the FPSO?
Greenhouse Gas Emissions	What happens if Santos exceeds the total emissions for the year? Do they shut down?
Environmental Research in the Operational Areas	What baseline studies for the environmental impacts that have been completed?
	Can Santos provide the environmental baseline studies?
Helicopter Activity	Will helicopters fly over Tiwi

## **Helicopter Activity**

Santos

- Helicopters will fly over Tiwi islands on average 3 times per week to transport people to the FPSO
- To manage any potential noise impacts, Santos intends to:
  - Limit noise impact by fly the Helicopter 1.8 km to 2.4km above Tiwi islands, existing aircraft fly lower than 0.6km. Noise levels at this height are unlikely to be heard any more/differently than other background air traffic.
  - Planned helicopter flight paths are over the eastern end of Melville Island, at its closest point in 22km from Seagull Island
- When helicopters begin flying over Tiwi's, if you do have concerns, you can raise them directly with Santos.

#### **Follow Up Requests**

#### PLANNED ACTIVITIES

#### **Production Operation**

- 1. Request for a map with the vessel movements and shipping routes
- 2. Request for a picture of the shipping chart that shows the 500-metre exclusion zone around the FPSO
- 3. Where is the operational area in relation to biologically important areas (threatened species)?
- 4. Provide evidence that the operational area is not in a whale migration pathway.

20



#### **Production Operations Unplanned Risks**

Santos

Unplanned Events (Accidents)	Unplanned Impact	How we manage (the rules we follow)
Dropped objects	Impacts to water quality, disturbance to seabed and marine life	We follow strict procedures to stop objects dropping overboard and we pick up objects when it is safe to do so.
Disturbing marine animals	Disturbance (e.g. collisions) to marine animals	We look out for marine life (e.g., whales, dolphins, turtles) and we slow down and move away from them where possible if they are too close. We follow rules that outline how we need to interact with marine animals.
Invasive marine life (marine pests)	Impacts to other marine life and industry if pests establish	We ensure boats have low risk for carrying marine pests before they arrive. We have plans and equipment in place on boats to prevent invasive marine species. The Federal government has strict rules we need to follow. The FPSO will have a quarantine management plan.
Chemical spill	Water quality and marine life impacts	We select chemicals that are environmentally friendly where possible and store them carefully. We have procedures for using and cleaning up chemicals.
Dry gas release	Impacts to marine life, water/air quality, other marine users and cultural features	The pipeline is designed to withstand impacts from dropped objects. We follow strict rules for lifting equipment and have emergency response procedures. The infrastructure location is marked on nautical charts. We have operating procedures and a safety case to prevent a loss of gas.
Minor hydrocarbon releases	Impacts to marine life-and water quality	We have procedures in place to manage the handling and transfer of hydrocarbons.
Larger hydrocarbon releases (marine diesel oil, marine gas oil, condensate, heavy fuel oil, well fluids)	Impacts to water quality, habitats and marine life, other marine users and cultural features	We make sure all the boats are following the rules for preventing collisions. We let other boats know where we are and what we are doing. We follow strict procedures for refueling. We will have well operations management plans setting out the systems in place to ensure well safety. We have plans (OPEP) arrangements in place for responding to spills.

#### Follow Up Answers to Questions Unplanned Risks (Accidents)

Торіс	Question
	Who manages the spill response training register?
	Will Santos be doing more spill response training in the Community?
Spill Response Training	Is spill response training only for Tiwi people or will other Communities
	be trained too?
0	If an oil spill were to occur and people were impacted, would there be
Compensation	compensation?









### Website



The Santos-operated Barossa Gas Project is an offshore gas and condensate project with the purpose of providing a new source of gas for the existing Darwin liquified natural gas (DLNG) facility at Wickham Point in the NT. It is intended that natural gas and condensate would be extracted from the Barossa field, located in Commonwealth waters approximately 285 kilometres offshore north-northwest from Darwin. Initial processing would occur at the Floating Production Storage and Offloading (FPSO) facility, to separate the natural gas, water and condensate extracted from the Barossa field. The dry natural gas would be transported through the GEP for onshore processing at the DLNG facility. The condensate would be transferred from the FPSO facility to purpose-built tankers for international export.

As part of obtaining authorisation for this activity, Santos is undertaking consultation for the following regulatory approvals:

- The Production Operations Environment Plan (EP) relating to the arrival and operations of the FPSO, operation of a subsea production system and supporting subsea infrastructure, and operation of a 285 km section of the GEP located in Commonwealth waters where offshore petroleum activities are regulated under the Offshore Petroleum and Greenhouse Gas Storage Act 2006 (Cth) (OPGGS Act).
- The Operations Environmental Management Plan (OEMP) which includes the:
  - > 8.26 km section of the GEP in NT coastal waters covered by the Petroleum (Submerged Lands Act) 1981 (NT) (PSL Act); and
  - > ~92 km section of the GEP inshore of NT waters covered by the Energy Pipelines Act 1981 (NT) (Energy Pipelines Act).

The estimated life of the Barossa Development is 25 years, and the Production Operations EP and the OEMP will be reviewed every five years following initial regulator authorisation. Productions Operations Activity is planned to commence in 2025.



Figure 1: Graphical image of the FPSO

#### **Approvals process**

The Commonwealth Government's independent expert regulator for offshore oil and gas development, the National Offshore Petroleum Safety Management Authority (NOPSEMA), accepted the Barossa Offshore Project Proposal (OPP) in March 2018. The OPP is the government's projectlevel environmental approval for offshore projects, with construction and operations subject to acceptance of activity-level environment plans (EPs). An EP:

- must provide a detailed environmental impact and risk assessment of the proposed offshore activity and demonstrate how those impacts and risks will be reduced to a level that is as low as reasonably practicable and acceptable for the life of the activity; and
- must describe how Santos has identified and consulted with relevant persons whose interests, functions and activities may be affected by the activities to be carried out under the EP.

Santos will be consulting in accordance with section 25 of the Offshore Petroleum and Greenhouse Gas Storage (Environment) Regulations 2023 (Cth) (OPGGS Environment Regulations) as the Production Operations EP is developed.

For Production Operations activity in Northern Territory waters, the OEMP will be submitted to the NT Department of Industry Tourism and Trade (DITT).

## Seeking information and what's next

Santos is consulting on both the Production Operations EP (Commonwealth waters) and OEMP (NT waters) at the same time.

Santos is continuing its Barossa Gas Project consultation efforts to further learn, understand and assess values and sensitivities of the environment that may be affected by our proposed activities, and potential environmental impacts and risks. There may be information Santos is not yet aware of but needs to properly understand to assess potential activity impacts and risks. Consultation may inform this. It may also inform what control measures are to be proposed to reduce environmental impacts and risks to as low as reasonably practicable and to an acceptable level.

In preparing an EP for submission to NOPSEMA, a titleholder must consult with each 'relevant person', including relevant Commonwealth, State and Northern Territory Departments or agencies and persons (or organisations) whose functions, interests or activities may be affected by the activity proposed to be carried out under an EP.

Relevant persons being consulted on under the OPGGS Environment Regulations should note that they:

- are entitled to be given sufficient information to allow them to make an informed assessment of the possible consequences of the activity on their functions, interests or activities;
- are entitled to be allowed a reasonable period for the consultation; and

may request particular information provided in consultation not be published.
If you do ask this, Santos will respect that, and the information will not be published under the OPGGS Environment Regulations. Information we need to give to NOPSEMA to assess our plan will be provided in a separate report (rather than in the published EP).

Your input is important to Santos:

- so that we can understand the environmental values in the operational area and the EMBA, and the environmental impacts and risks associated with the activity;
- to inform how consultation processes may need to be adapted for different relevant persons;
- to ensure that we provide information to people in an appropriate and accessible manner; and
- to assist with Santos' preparation of the EP.

If you consider you may be a relevant person, please contact us as soon as possible via email at offshore.consultation@santos.com or call 1800 267 600 to allow Santos to initiate consultation with you in relation to the proposed activities and so you can tell us how you would like to be consulted. Consultation closes on **Tuesday 9 April 2024**.

Santos is committed to undertaking genuine and meaningful consultation. We want to provide information for people to make informed assessments of the possible consequences of the proposed activities on them. Your input is important to us, and input will be considered in the development of our EP.

Visit santos.com/barossa for more information on the Barossa Development.

### **Privacy Notice**

Santos Limited, Santos NA Barossa Pty Ltd and their related bodies corporate (collectively, we, our, us or Santos) collect personal information about you (which may include sensitive information about your Indigenous heritage or clan group) that you provide in this form and in any consultation with you. We use this information to engage with you, to receive and respond to any feedback that you provide, to include relevant information in any reports that we prepare, and for the purposes otherwise set out in our <u>Barossa Gas Project Consultation Privacy Policy</u>. Santos will handle your information in accordance with our Code of Conduct, our Confidentiality, IP and Privacy Procedure and Barossa Gas Project Consultation Privacy Policy. You can ask us for a copy of any of these documents by contacting us using the details set out below.

The laws that relate to consultation regarding the environment plans may require us to collect personal information about you if your feedback is to be included in any report we prepare (including so that a copy of that report can be provided to you if you request). If you do not provide your personal information, we may not be able to identify you as the person who provided particular feedback (including in any report relating to consultation) or discuss any feedback you have provided with you further.

The information you provide will be collected by or on behalf of us and may be disclosed to other companies in the Santos group, to third parties that help us run our business, or as required by law (including collection and disclosure of information to relevant government agencies and departments to which we are required to provide reports). Your feedback may also be reflected in our environment plans (subject to any non-publication requests, as described below). We may disclose your personal information to recipients that are located outside of Australia, including in Papua New Guinea and the United States of America.

Our Barossa Gas Project Consultation Privacy Policy and related documents described in this Privacy Notice provide further information about how we store and use, and how you may access and correct, your personal information, and how you can lodge a complaint regarding the handling of your personal information (including how we will respond to that complaint). If you would like to request a copy of any personal information that we hold about you or request that we correct any such information that is inaccurate or incomplete, you can contact us in the following ways:

- posting a letter addressed to us at 60 Flinders Street, Adelaide;
- telephoning us on (08) 8116 5000; or
- sending us an email at <u>offshore.consultation@santos.com</u>.

You may request that any information you provide in this form, or during any consultation, not be published (including as part of any environment plan or related reports). You can make this request by selecting the option in the form below. If you select this option, the information will not be published and Santos may contact you to discuss your wishes.

#### **Relevant person nomination form**

#### Production Operations Environment Plan

By completing this form I consent to Santos using the information provided in this form for the purpose of consultation on the Production Operations Environment Plan in accordance with the Privacy Notice on this page and the <u>Barossa Gas Project Privacy Policy</u>.



## **Quarterly Update October 2024**

## Santos

#### **Stakeholder Consultation**

# Barossa Gas Project

#### In this edition



- DPD works continue in Darwin Harbour.
- Subsea infrastructure installation continues.
- FPSO on-track for sail-away in Q1 2025.
- Development wells produce good results.

#### Work progress highlights

The Barossa Gas Project is now more than 80 per cent complete with first gas expected in the third quarter of 2025. Offshore works for the Darwin Pipeline Duplication (DPD) will continue within Darwin Harbour for the remainder of 2024.

In remote Commonwealth waters off the Northern Territory (NT), the second phase of subsea infrastructure installation will be undertaken during the next quarter, and we will continue to drill our production wells.

The Floating Production, Storage and Offloading (FPSO) vessel, currently in Singapore, is expected to head to Australia in the first quarter of 2025.

#### **FPSO** construction



Installation of all 16 topside modules on the Floating Production Storage and Offloading vessel (FPSO) has been completed in Singapore.

The vessel, one of the largest FPSOs ever built, is now into the commissioning phase during which detailed inspections and testing of systems are being undertaken to ensure they are operating safely and to design parameters.

Once onshore commissioning has been completed, the FPSO will be towed from Singapore to Australia using ocean-going tugs. Sail-away is on track for Q1 in 2025.

Following the FPSO's arrival in the Barossa Field, it will be connected to the mooring system. Equipment and systems will all be tested during this commissioning stage in readiness for start-up operations.

The FPSO, which will be permanently located in the Field, (subject to receipt of the necessary approvals) is 358m long and 64 meters wide and can accommodate up to 140 personnel.

Santos	Level 7, 100 St George's Terrace Perth WA 6000 Australia	GPO Box 5624 Perth WA 6831	T: +61 8 6218 7100 F: +61 8 6218 7200	santos.com	1	

#### **Project Overview**

The Barossa Gas Development, located approximately 285km offshore from Darwin in the Northern Territory, comprises a Floating Production Storage and Offloading (FPSO) facility, a subsea production system, supporting in-field subsea infrastructure and a gas export pipeline (GEP).

Once complete, the Barossa Gas Development will transport natural gas from the offshore Barossa field via the GEP to the existing Darwin liquefied natural gas (DLNG) facility.

The development is expected to commence delivering natural gas in 2025. Barossa gas is a critical energy source and will ensure affordable and reliable energy for Santos' customers for decades to come.



The Barossa Gas Development will create and sustain hundreds of jobs in the Northern Territory and will inject billions of dollars into the local economy, supporting the continued regional development of northern Australia.



#### Subsea Infrastructure Installation

The second campaign for the installation of subsea infrastructure in the Barossa Field, located 285km offshore from Darwin, is due to commence in mid-November and continue until mid-March 2025, subject to weather conditions and operational performance.

During the first campaign, installation of the infield flowlines that connect the drill centres to the FPSO location was completed along with the installation of manifolds, risers and umbilicals. This phase also involved the installation of the suction anchor piles and hook-up of the Submerged Turret Production buoy for the FPSO mooring system. Flowline precommissioning activities for two drill centres were also completed. Campaign 2 will involve the installation of production manifolds, spools, umbilicals, flying leads and jumpers, designed to support the connection of subsea production wells to the FPSO. Further pre-commissioning will be undertaken to test the various connections and systems.

The third campaign will take place in 2025 when the drilling phase is completed.

Once all installation and testing has been completed, a preservation phase will commence that will maintain the integrity of the infrastructure in readiness for future operations expected to start in 2025.

	Level 7, 100 St George's Terrace	GPO Box 5624	12 +61 8 6218 7100	
Santos	Perth WA 6000 Australia	Perth WA 6831	+61 8 6218 7200	santos.con



#### **DPD** Project

The DPD Project effectively involves 'duplicating' the existing Bayu-Undan to Darwin pipeline in NT internal waters, NT coastal waters and Commonwealth waters. Once in operation the new pipeline will only be used for transporting natural gas (not oil or condensate) from the Barossa gas field to the Darwin LNG (DLNG) facility.

All work has been completed to date in accordance with an accepted Environment Management Plan. Since the commencement of works in NT internal waters in May 2024, significant progress has been achieved.

Construction of the trench in Darwin Harbour, within which the pipeline will be laid, has been completed along with installation of concrete mattresses for pre-lay span supports, pipeline crossings and cable crossings.

Onshore at Wickham Point, where the existing DLNG facility is located, construction of the shore crossing for the pipeline is nearing completion.

During the next quarter, trench maintenance activities will be undertaken in locations where build-up of sedimentation has occurred.

Installation of the rock bedding layer and pull-in wire at the Wickham Point shore crossing is being undertaken in readiness for shallow water pipelay and associated survey activities to commence in October. Installation of concrete mattresses for pre-lay span supports within Darwin Harbour.

The construction activities occur within a Project area, encompassing the pipeline route with an approximate 2 km buffer either side of the pipeline and the spoil disposal ground.

Other Darwin Harbour users are asked to exercise caution at all times and maintain a distance of at least 500 metres from any vessels undertaking the construction activities.

The activities are managed to ensure that environmental impacts and risks are avoided or mitigated to meet Santos and regulatory requirements.

The DPD Project in NT internal waters has received all primary approvals. The NT Government approved the DPD activity in NT jurisdiction under the NT Environment Protection Act following assessment by the NT Environment Protection Authority.

The DPD activity in NT and Commonwealth jurisdictions has also been assessed and approved under the Commonwealth Environment Protection and Biodiversity Conservation Act (EPBC Act).

Santos	Level 7, 100 St George's Terrace Perth WA 6000 Australia	GPO Box 5624 Perth WA 6831	1: +61 8 6218 7100 F: +61 8 6218 7200	santos.com	3
--------	-------------------------------------------------------------	-------------------------------	------------------------------------------	------------	---



Valaris MS-1 MODU (view from support vessel).

#### **Drilling and Completions**

Drilling of the six development wells is continuing at the Barossa Field, 285km offshore from Darwin. The moored semisubmersible mobile offshore drilling unit (MODU) is currently positioned at the South 1(SI) drill centre location, and is working on the fourth well.

All work has been completed to date in accordance with the accepted Environment Plan and without any environmental incidents.

The MODU is planned to move from the S1 to the South-2 (S2) drill centre location in October/November 2024 to complete the remaining work at that location.

Drilling activities occur within a 2.5-km radius of each well. During drilling, a 500-metre exclusion zone (known as a petroleum safety zone) is in place around the MODU. Vessels are prohibited from entering or being present in a petroleum safety zone without authorisation.

Work at the North-1 (N1) drill centre location is planned for subsequent quarters. The entire



Valaris MS-1 MODU helicopter deck.

drilling campaign, including completions and installation of subsea vertical trees, is still expected to take approximately 24 months in total. All schedule estimates are subject to weather and operational performance.

The MODU continues to be supported by helicopters and up to four support vessels which transit between the development area and onshore supply base in Darwin as required.

#### **GEP** installation

With all testing and pre-commissioning activities completed, the installed 262-kilometre GEP is safely preserved on the seabed in readiness for future operations expected to start in 2025.

The GEP will connect the Barossa field to the DPD and together these pipelines will transport natural gas from the Barossa field to the Darwin LNG (DLNG) facility.



The 'Fortitude' construction support vessel performing precommissioning activities on the GEP.

Santos	Level 7, 100 St George's Terrace Perth WA 6000 Australia	GPO Box 5624 Perth WA 6831	71 +61 8 6218 7100 15 +61 8 6218 7200	santos.com	4
--------	-------------------------------------------------------------	-------------------------------	------------------------------------------	------------	---



#### Work Opportunities

We continue to have 11 First Nations people engaged in Santos funded roles/ training programs that work offshore in positions related to the Barossa Gas Project.

These comprise four Tiwi people engaged as Cultural Observers and Monitors on the GEP pipelay and support vessels and seven Larrakia and Tiwi people engaged in offshore roles on the Santos' contracted drilling rig.

Two Santos Offshore Health, Safety and Environment (HSE) Advisor Trainees have been recruited and will undertake a formal 12-18-month traineeship to obtain a Certificate IV in HSE.

One of the trainees, a Tiwi woman, has commenced work on the Santos contracted drill rig while the other trainee will start offshore in October 2024.

Another Tiwi man, one of six indigenous trainees in the MS-1 drill rig training program, has transitioned from offshore assistant to a core roustabout position.

First Nations people represent 55% of Santos' current trainees and apprentices in the NT.

#### **Barossa Aboriginal Future Fund**

Santos' Barossa Aboriginal Future Fund (BAFF) aims to invest in projects that will directly benefit NT Aboriginal Coastal Communities.

The Fund will invest in projects and programs that improve community infrastructure, health, education, housing, and economic outcomes and support employment pathways.

The Fund will also support First Nations people to maintain cultural practices, carry out cultural obligations and care for country.

The Barossa Joint Venture Partners initial contribution to the Fund is \$10 million, with additional payments to be made into the Fund once Barossa starts producing and selling gas.

Santos is working with our Barossa Joint Venture Partners to finalise the revenue stream for the Fund and establish legal structures with sound governance.

Santos wants to listen to NT Coastal Aboriginal people about the needs for their communities.

If you want more information or want to share your ideas, you can visit the NT Community Team at the Santos shop in the Darwin Mall or contact us via email at enquiriesNT@cantos.com or phone on +61 8919 1900.

	Level 7, 100 St George's Terrace	GPO Box 5624	T: +61 8 6218 7100		
Santos	Perth WA 6000 Australia	Perth WA 6831	F +61 8 6218 7200	santos.com	5
#### **Environmental Approvals** and Consultation

#### DPD Project - NT waters

DPD activities in NT internal waters are being undertaken in compliance with the primary environmental approvals received from the NT and Commonwealth governments.

The NT Government approved the DPD Project, following the completion of assessment by the NT Environment Protection Authority under the Environment Protection Act 2019 (NT).

The DPD Project was also assessed and approved by the Commonwealth Department of Climate Change, Energy, the Environment and Water (DCCEEW) under the Environment Protection and Biodiversity Conservation Act 1999 (Cth).

#### DPD Project - Commonwealth and NT coastal waters

Additional environmental approvals are under assessment by the Regulator for the DPD Project activity in each of Commonwealth and NT coastal waters. Santos has submitted the Environment Plan (EP) for DPD installation (-23km section of pipeline) in Commonwealth waters and a Construction Environmental Management Plan (CEMP) for DPD installation (-8km section) in NT coastal waters. Consultation for these plans has been completed.

The Commonwealth waters EP is currently under assessment by NOPSEMA and can be viewed here The CEMP has been submitted to the NT Department of Industry, Tourism and Trade (NT-DITT) for assessment and can be viewed at Published environment plans | NT.GOV.AU

#### **Barossa Productions Operations -**Commonwealth and NT waters

Santos has submitted the Barossa Production Operations EP in Commonwealth waters (-285km section) to NOPSEMA for assessment. The link to the EP is accessible on the NOPSEMA website.



The EP covers (among other things) the following activities:

- · Arrival, hook-up and commissioning of the FPSO (section 2.5 of the EP);
- Initial start-up of production to a steady state (section 2.6 of the EP):
- Operation of the FPSO and subsea infrastructure and associated activities (section 2.7 of the EP); and
- Inspection, monitoring, maintenance and repair (IMMR) activities for subsea infrastructure and the Gas Export Pipeline (section 2.9 of the EP);

The EP also provides information on (among other things):

- the potential risks and impacts of these proposed activities; and
- the consultation undertaken by Santos on the proposed activities (section 4 of the EP).

If you have any queries about this EP, please contact us by phone or email (see details below).

Operations Environmental Management Plans (OEMPs) for Operational activities along the Barossa pipeline in NT coastal waters (-8km section) and NT internal waters (-92km section) will be submitted to the NT- DITT for assessment and approval.

Santos		Louis 7 100 Ct Coornels Terrae	GPO Box 5624 Perth WA 6831	T: +61 8 6218 7100 F: +61 8 6218 7200			6
				Phone: Email: Website:	1800 267 600 offshore.consult santos.com/bar	ation@santos.com ossa	
Contact Us			We are committed to ongoing engagement about our activities. If you require further information or would like to speak to us, please do not hesitate to contact us.				

# **Quarterly Update January 2025**

### Stakeholder Consultation

# Barossa Gas Project

### In this edition



- NT pipeline installation completed in Darwin Harbour
- SURF Campaign 2 more than 50 percent completed
- FPSO Living Quarters set to be operational
- Drilling of wells continues in Barossa Field

### Work progress highlights

The Barossa Gas Project is now more than 88 per cent complete with first gas expected in the third quarter of 2025.

Offshore works for the Darwin Pipeline Duplication (DPD) are nearing completion in NT Waters, to be immediately followed by installation of the final sections of pipeline in Commonwealth and NT Coastal Waters.

In remote Commonwealth waters off the Northern Territory (NT), the second phase of subsea infrastructure installation is continuing along with drilling of the production wells.

The Floating Production, Storage and Offloading (FPSO) vessel, currently in Singapore, is being prepared in readiness for its journey to Australia expected in the first quarter of 2025.

### FPSO construction



Commissioning of the various FPSO systems has continued to plan, including the Main Topsides Generators which have all been successfully fired.

Preparations are also being made for the habitation of the Living Quarters by the FPSO crew starting in January.

Once onshore commissioning has been completed, the FPSO will be towed from Singapore to Australia using ocean-going tugs. Sail-away is on track for Q1 in 2025.

Following the FPSO's arrival in the Barossa Field it will be connected to the mooring system. Equipment and systems will all be tested during this commissioning stage in readiness for start-up operations.

The FPSO, which will be permanently located in the Field, (subject to receipt of the necessary approvals) is 358m long and 64 meters wide and can accommodate up to 140 personnel.

Santos Level 7, 1 Perth W

Level 7, 100 St George's Terrace Perth WA 6000 Australia GPO Box 5624 Perth WA 6831

T: +61 8 6218 7100 F: +61 8 6218 7200

santos.com

1

### **Project Overview**

The Barossa Gas Development, located approximately 285km offshore from Darwin in the Northern Territory, comprises a Floating Production Storage and Offloading (FPSO) facility, a subsea production system, supporting in-field subsea infrastructure and a gas export pipeline (GEP).

Once complete, the Barossa Gas Development will transport natural gas from the offshore Barossa field via the GEP to the existing Darwin liquefied natural gas (DLNG) facility.

The development is expected to commence delivering natural gas in 2025. Barossa gas is a critical energy source and will ensure affordable and reliable energy for Santos' customers for decades to come.



The Barossa Gas Development will create and sustain hundreds of jobs in the Northern Territory and will inject billions of dollars into the local economy, supporting the continued regional development of northern Australia.



### Subsea Infrastructure Installation

The second campaign for the installation of subsea infrastructure in the Barossa Field, located 285km offshore from Darwin, commenced mid-November 2024 and will continue until end-March 2025, subject to weather conditions and operational performance.

Campaign 2 is approximately 50 percent complete with the S1 manifold and umbilicals having been installed, along with the spools, flying leads and jumpers at the S2 Drill Centre.

Pre-commissioning of the riser system and spools is due to commence in the near future, which will test the various connections and systems prior to production. All work has been completed to date in accordance with an accepted Environment Plan.



The third campaign will take place in Q2 / Q3 2025when the drilling phase at the NI Drill Centre is completed.

Once all installation and testing has been completed, a preservation phase will commence that will maintain the integrity of the infrastructure in readiness for future operations expected to start in 2025.

Santos

Level 7, 100 St George's Terrace GPO Box 5624 Perth WA 6000 Australia Perth WA 6831 T: +61 8 6218 7100 P: +61 8 6218 7200

santos.com

2



"Sandpiper" shallow water barge performing shorepull at DLNG



Installation of rock protection over the pipeline at the shore crossing

### **Darwin Pipeline Duplication**

The DPD Project effectively involves 'duplicating' the existing Bayu-Undan to Darwin pipeline in NT internal waters, NT coastal waters and Commonwealth waters. Once in operation the new pipeline will only be used for transporting natural gas (not oil or condensate) from the Barossa gas field to the Darwin LNG (DLNG) facility.

All work has been completed to date in accordance with an accepted Environmental Management Plan. Since the commencement of works in NT internal waters in May 2024, significant progress has been achieved.

In December the shallow water barge 'Sandpiper' completed installing the first 30kms of the 92 kms section of pipeline in NT internal waters. The Sandpiper commenced with the shore crossing at DLNG and then installed the pipeline through Darwin harbour before handing over to the dynamically positioned pipelay vessel 'Audacia' to complete the section of the pipeline in deeper NT internal waters. Installation of rock protection over the pipeline at the shore crossing at Wickham Point has been completed and installation of rock protection in deeper water within Darwin harbour continues.

Pipelay will then proceed into NT coastal waters (8 kms) and Commonwealth waters (23 kms) and is expected to be completed in early 2025. Pipeline survey and pre-commissioning will commence on the completion of pipelay activities.

The construction activities occur within a Project area, encompassing the pipeline route with an approximate 2 km buffer either side of the pipeline and the spoil disposal ground.

Other users of the marine environment are asked to exercise caution at all times and maintain a distance of at least 500 metres from any vessels undertaking the construction activities.

The activities are managed to ensure that environmental impacts and risks are avoided or mitigated to meet Santos and regulatory requirements.



The Audacia is now installing the pipeline in the deeper waters outside Darwin Harbour



Welding of pipeline sections aboard the Audacia

Santos

Level 7, 100 St George's Terrace Perth WA 6000 Australia

GPO Box 5624 T Perth WA 6831 P

T: +61 8 6218 7100 F: +61 8 6218 7200

### Drilling and Completions

Drilling of the six development wells is continuing at the Barossa Field, 285km offshore from Darwin. The moored semisubmersible mobile offshore drilling unit (MODU) is currently positioned at the North 1 (N1) drill centre location.

All work has been completed to date in accordance with the accepted Environment Plan and without any environmental incidents.

Work at the N1 drill centre location is planned to continue during the first half of 2025. The entire drilling campaign, including completions and installation of subsea vertical trees, is still expected to take approximately 24 months in total. All schedule estimates are subject to weather and operational performance.

Drilling activities occur within a 2.5-km radius of each well. During drilling, a 500-metre exclusion zone (known as a petroleum safety zone) is in place around the MODU. Vessels are prohibited from entering or being present in a petroleum safety zone without authorisation.



The Valaris MS-1 mobile offshore drilling unit

The MODU continues to be supported by helicopters and up to four support vessels which transit between the development area and onshore supply base in Darwin as required.

### Work Opportunities

Several First Nations people continue to be engaged in Santos' funded roles/ training programs that work offshore in positions related to the Barossa Gas Project.

These comprise five Larrakia and Tiwi people engaged in offshore assistant roles on the Santos' contracted drilling rig, one Tiwi man engaged as a Cultural Monitor on SURF activities and a Larrakia man engaged as a Cultural Monitor on DPD activities.

In addition, two Santos Offshore Health, Safety and Environment (HSE) Advisor Trainees have been recruited and are now undertaking a formal 18-24 month traineeship to obtain a Certificate IV in HSE. Both are currently working offshore on the Santos' contracted drill rig under the leadership of Santos HSE team members.

Another Tiwi man, one of the trainees in the offshore assistant training program on the Santos contracted drill rig, has transitioned from offshore assistant to a core roustabout position, where he continues in his role.

First Nations people represent more than 55% of Santos' current trainees and apprentices in the NT.

### **Barossa Aboriginal Future Fund**

Santos' Barossa Aboriginal Future Fund (BAFF) aims to invest in projects that will directly benefit NT Aboriginal Coastal Communities.

The Fund will invest in projects and programs that improve community infrastructure, health, education, housing, and economic outcomes and support employment pathways.

The Fund will also support First Nations people to maintain cultural practices, carry out cultural obligations and care for country.

The Barossa Joint Venture Partners initial contribution to the Fund is \$10 million, with additional payments to be made into the Fund once Barossa starts producing and selling gas.

Santos is working with our Barossa Joint Venture Partners to finalise the revenue stream for the Fund and establish legal structures with sound governance.

Santos wants to listen to NT Coastal Aboriginal people about the needs for their communities.

If you want more information or want to share your ideas, you can visit the NT Community Team at the Santos shop in the Darwin Mall or contact us via email at enquiriesNT@santos.com or phone on +61 8919 1900.

Santos Level

Level 7, 100 St George's Terrace GPO Box 5624 Perth WA 6000 Australia Perth WA 6831 T: +61 8 6218 7100 F: +61 8 6218 7200

### Environmental Approvals and Consultation

#### DPD Project - NT waters

DPD activities in NT internal waters are being undertaken in compliance with the primary environmental approvals received from the NT and Commonwealth governments.

The NT Government approved the DPD Project, following the completion of assessment by the NT Environment Protection Authority under the Environment Protection Act 2019 (NT).

The DPD Project was also assessed and approved by the Commonwealth Department of Climate Change, Energy, the Environment and Water (DCCEEW) under the Environment Protection and Biodiversity Conservation Act 1999 (Cth).

#### DPD Project - Commonwealth and NT coastal waters

DPD activities in NT Coastal Waters are being undertaken in compliance with the primary environmental approvals received from the Commonwealth and NT governments.

The Commonwealth Government approved the activities (-23km section) in Commonwealth Waters following assessment of an Environment Plan under the Offshore Petroleum and Greenhouse Gas Storage Act 2006 (Cth).

As noted above, the DPD Project was also assessed and approved by DCCEEW under the Environment Protection and Biodiversity Conservation Act 1999 (Cth).

The NT Government approved the activities in NT Coastal Waters (-8km section) following assessment of the Offshore Construction Environmental Management Plan (CEMP) under the Petroleum (Submerged Lands) Act 1981 (NT) and Environment Protection Act 2019 (NT).

The DPD EP can be viewed here while the CEMP is available here

#### Barossa Productions Operations -Commonwealth and NT waters

Santos submitted the Barossa Production Operations EP in Commonwealth waters (-285km section) to NOPSEMA in August 2024 for assessment. The link to the EP is accessible on the NOPSEMA website here

Following a request from NOPSEMA to provide further information, Santos re-submitted the EP with some amendments in December 2024 for further assessment.

The EP covers (among other things) the following activities:

- Arrival, hook-up and commissioning of the FPSO (section 2.5 of the EP);
- Initial start-up of production to a steady state (section 2.6 of the EP);
- Operation of the FPSO and subsea infrastructure and associated activities (section 2.7 of the EP); and
- Inspection, monitoring, maintenance and repair (IMMR) activities for subsea infrastructure and the Gas Export Pipeline (section 2.9 of the EP);

The EP also provides information on (among other thinas):

- the potential risks and impacts of these proposed activities; and
- the consultation undertaken by Santos on the proposed activities (section 4 of the EP).

If you have any queries about this EP, please contact us by phone or email (see details below).

Operations Environmental Management Plans (OEMPs) for Operational activities along the Barossa pipeline in NT coastal waters (-8km section) and NT internal waters (-92km section) will be submitted to the NT- DME for assessment and approval.

### Contact Us

Consultation for all Barossa regulatory approvals has been completed.

We are committed to ongoing engagement about our activities. If you require further information or would like to speak to us, please do not hesitate to contact us.

Phone: 1800 267 600 offshore.consultation@santos.com Email Website: santos.com/barossa

Santos Level 7, 100 St George's Terrace Perth WA 6000 Australia

GPO Box 5624 Perth WA 6831 1 +61 8 6218 7100 +61 8 6218 7200

# Appendix F2: Relevant Persons Advertising Material

# Table 4-8 (of the EP) Targeted advertising campaign

### **Preliminary consultation**

### March 2024 Social Media post

PROYEK GAS BAROSSA DI AUSTRALIA MENCARI ORANG TERKAIT UNTUK BERKONSULTASI

# Santos

#### **RENCANA LINGKUNGAN OPERASI PRODUKSI**

Proyek Gas Barossa yang dioperasikan Santos adalah proyek gas dan kondensat, yang melibatkan ekstraksi gas alam dari ladang gas Barossa, yang terletak di perairan Commonwealth Australia, sekitar 285-kilometer lepas pantai utara-barat laut Darwin. Gas alam kemudian akan disalurkan melalui pipa gas ke fasilitas gas alam cair (DLNG) Darwin yang ada. Kondensat akan diturunkan melalui kapal tanker untuk diangkut ke pelanggan di Asia.

Batas wilayah izin produksi kurang lebih 520 kilometer sebelah timur tenggara Dili dan kurang lebih 2.605 kilometer sebelah timur Jakarta, di Laut Arafura.

Sebagai bagian dari Proyek, Santos akan menyiapkan rencana lingkungan untuk proyek operasi produksi untuk diserahkan kepada Otoritas Pengelolaan Lingkungan dan Keselamatan Minyak Lepas Pantai Nasional Australia (NOSPEMA). Konsultasi dengan pihak-pihak terkait merupakan bagian penting dalam mempersiapkan rencana ini.

Santos berupaya mengidentifikasi dan berkonsultasi dengan orang-orang terkait yang fungsi, kepentingan, atau aktivitasnya mungkin terpengaruh oleh aktivitas yang diusulkan berdasarkan Rencana Lingkungan Operasi Produksi yang kami usulkan.

Jika anda menganggap bahwa anda mungkin orang yang relevan dan ingin diajak berkonsultasi mengenai rencana lingkungan ini, silakan hubungi kami paling lambat tanggal 15 Maret agar kami dapat berkonsultasi dengan anda. Konsultasi ditutup pada 9 April 2024.

Informasi lebih lanjut tersedia di santos.com/Barossa tentang siapa yang mungkin menjadi orang yang relevan, usulan kegiatan, lingkungan yang mungkin terkena dampak dari usulan kegiatan, potensi dampak dan risiko lingkungan, dan usulan tindakan pengendalian untuk mengurangi dampak dan risiko. ke tingkat yang serendah mungkin dan dapat dipraktikkan secara wajar.

#### Untuk informasi lebih lanjut:

Kunjungi santos.com/barossa Telepon 1800 267 600 Email ke luar offshore.consultation@santos.com PROJETU GAS BAROSA IHA AUSTRALIA

## Santos

Santos

## BUKA EMA RELEVANTE SIRA ATU HALO KONSULTASAUN

#### PLANU AMBIENTE BA OPERASAUN PRODUSAUN

Projetu Gas Barossa ne'ebé operadu husi Santos mak projetu gás no kondensadu ida, ne'ebé sei envolve hasai gás naturál husi kampu gás Barossa, ne'ebé lokaliza iha tasi Commonwealth Australia nian, maizumenus kilómetru 285 iha tasi laran parte norte noroeste husi Darwin. Tuirmai gás naturál sei lori liu husi kadoras gás ba instalasaun gás naturál ne'ebé iha Darwin (DLNG). Kondensadu sei deskarega liu husi tanke hodi transporta ba kilente sira iha Ásia.

Fronteira ba área lisensa produsaun nian maizumenus kilómetru 520 iha parte leste sudeste Dili no maizumenus kilómetru 2,605 iha parte leste Jakarta, iha Tasi Arafura.

Hanesan parte husi Projetu ne'e, Santos sei prepara planu ambientál ba operasaun produsaun Projetu refere hodi submete ba Autoridade Nasional Seguransa Petroleu no Jestaun Ambiental Australia (NOPSEMA) nian. Konsultasaun ho ema relevante sira sai parte importante ida hosi preparasaun planu hirak-ne'e.

Santos buka/identifika hela ema relevante sira atu konsulta ho sira ne'ebé nia funsaun, interese ka atividade sira bele afeta husi atividade sira ne'ebé propoin tuir ami-nia Planu Ambiente Operasaun Produsaun nian.

Se ita konsidera katak ita maka ema relevante no hakarak hetan konsulta ba planu ambiente ida-ne'e, favór ida kontaktu ami antes loron 15 fulan Marsu atu ami bele konsulta ho ita. Konsultasan sei taka iha 9 Abril 2024.

Informasaun klaru liu disponivel iha santos.com/Barossa kona-ba se mak bele sai ema relevante, atividade sira ne'ebé propoin, ambiente ne'ebé bele afeta husi atividade sira ne'ebé propoin, impaktu no risku ambientál potensiál sira, no medida kontrolu sira ne'ebé propoin atu buka atu hamenus impaktu no risku sira ba nivel ne'ebé ki'ik liu ho razoavel no aseitavel.

#### Atu hetan informasaun liután: Vizita santos.com/barossa Telefone 1800 267 600 Email offshore.consultation@santos.com

## BAROSSA GAS PROJECT SEEKING RELEVANT PERSONS FOR CONSULTATION

### PRODUCTION OPERATIONS ENVIRONMENT PLAN

The Santos-operated Barossa Gas Project is a gas and condensate project, which would involve extracting natural gas from the Barossa gas field, located in Australian Commonwealth waters, approximately 285 kilometres offshore north-north west from Darwin. Natural gas would then be transported via gas pipeline to the existing Darwin liquified natural gas (DLNG) facility. Condensate will be offloaded via tanker for transport to customers in Asia. As part of the Project, Santos will be preparing an environment plan for the Project's production operations submission to the Australian National Offshore Petroleum Safety and Environmental Management Authority (NOSPEMA). Consultation with relevant persons is an important part of preparing these plans.

Santos is seeking to identify and consult with relevant persons whose functions, interests or activities may be affected by the activities proposed under our proposed Production Operations Environment Plan.

If you consider that you may be a relevant person and would like to be consulted for this environment plan, please contact us to provide your feedback by 9 April 2024.

More information is available at santos.com/Barossa about who may be a relevant person, the proposed activities, the environment that may be affected by the proposed activities, potential environmental impacts and risks, and proposed control measures to seek to reduce impacts and risks to as low as reasonably practicable and an acceptable level.

#### For more information: Visit santos.com/barossa Phone 1800 267 600 Email offshore.consultation@santos.com

### Radio January- February - Darwin Hot 100

### Script A - Preliminary Consultation 8 Feb- 11 March 2024

Santos is seeking to consult with people whose functions, interests or activities may be affected by the proposed Production Operations Activity for the Barossa Gas Project. Including Santos's offshore production facility approximately 285 kilometres offshore from Darwin, and a Gas Export Pipeline. If you consider you may be affected, please contact Santos by 11 March 2024. For more visit <u>santos.com/barossa</u>,

Phone [1800 267 600] One Eight Hundred, Two Six Seven, Six Hundred or email: <u>offshore.consultation@santos.com</u>.

Saturday February 17, 2024 | NT News

NEWS 11

Iconic resort weighs up sale

Review to explore options

#### isa Allen, Ben Wilmot

<text><text><text><text><text><text><text><text><text><text><text>

and the

Santos

SEEKING RELEVANT PERSONS BAROSSA PRODUCTION OPERATIONS ACTIVITY

Santos is seeking to identify and consult with relevant persons whose functions, interests or activities may be affected by our proposed Barossa Production Operations Activity in Commonwealth and Northern Territory waters, north-northwest of Darwin

e. SDE was applied of the 1997 as therefore: The Early (MT) coastal-waters covered by the Antonium (Submanged Land) Acc) (MV PcT); and object accises of the GCP instead of PT options consend to the Arman Pareline Act ABI (MT).





- Los offices of contraction processes may need to be adapted to Bit offices research periods.



id. Although British

Thursday and they arrested the

29-year-old. He was charged with one He was charged with one count of failing to comply with a requirement of monitoring conditions. This offence carries a maximum penalty of five years jail and a \$93,900 fine. The man is next expected to appear in court on March 22.



have occurred before the spectacular collapse of her Modco company last August. Ms Lu, pictured left, did not appear at what was the first hearing of the charges, and neither did her lawyers. But they did contact the the firm had 29 homes unfinished on its books. The couple later issued an apology, months after the company's collapse, through a conformation spokesman.

"It is certainly reasonable to pursue an investigation on Ms before her scheduled court appearance, Ms Lu posted video of footage from an ap-parent rooftop party in Perth, that featured UFC champion Jon "Bones" Jones pouring out shots from a \$350-per-bottle of tequila.

### SEEKING RELEVANT PERSONS BAROSSA **PRODUCTION OPERATIONS ACTIVITY**

Santos is seeking to identify and consult with relevant persons whose functions, interests or activities may be affected by our proposed Barossa Production Operations Activity in Commonwealth and Northern Territory waters, north-northwest of Darwin

- and in contemp programs A Production Operations Environment Plan (EP) Initiating to the arrival and operations of the Flasting Production Storage and Officialing fields (PFSO), pervision of a allises production system and supporting subsex Infrastructure, and operation of a 205m section of the Gas Export Flashing (CD) located in Commonswish water where officienes patiolesen activities are regulated under the CO/theor Petroleum and Greenbase Obstances An Operations Environmental Management Plan (OEMP) which includes
- 8.26 km section of the GEP in Northern Territory (NT) coastal wat covered by the Petroleum (Submerged Lands Act) 1987 (NT); and
- -92km section of the GEP inihore of NT waters covered by the Energy Pipelines Act 1989 (NT).

#### Summary of activity

mmary or activity Statis-operate Bioresis Gar Project is an offshore gas and condense ed with the purpose of providing a new source of gas to the exiting an lightide natural gas (CLMS) heating at Wickem Provin in the NT-retended that initiary gas and condensate would be extracted from Gammark table, locating CLMS heating at Wickem Provide CLM at Gammark table, locating CLMS heating the work and condensate you of CLMS, so separate the maximum gas, wear and condensate extracted the CLMS heating the maximum gas. wear and condensate extracted the CLMS heating the maximum gas.



from the Baras is field. This dry natural gas would be transported through the GDP for anihone processing at the DLNC facility. The condensate manufactural experiment the FPSO facility to purpose-built tankers for the animate

International export. The estimated life of the Barossa Development is 25 years, and the Production Operations EP and the OEMP will be inviewed every five years. Productions Operations activity is expected to commence in 2025, subject to obtaining the respired approvals.

#### Activity location

Activity location The Reduction Operational Activity is confined to their operational areas: (Dependitional Areas I (OAI) and Operational Areas 2 (OA2). With the Barease field. This is the areas in which the IPSCD, uptice productions private and supporting updates and the Bareas with the confined to Commonwealth waters, approximately 202 km north-andth water confined to Commonwealth waters, approximately 202 km north-andth water I have a straight and the areas and approximately 202 km north-andth water Data (The Areas) and a straight and and approximately 202 km north-andth water DATs The 282 km waters of the GEP km coAl (be the Commonwealthher waters/ NT waters between the km 264 km section of the GEP shauster to NT coastie waters and the BaS km section of the GEP shauster built and and the Territorial See Baseline). The commonwealth waters (Section 200) km commonwealther built and the Territorial See Baseline).

The environment that may be affected by the

#### proposed activity

proposed activity Senta is assessing impacts and risks to the environment hom these emposed activities, inclusive of acosystems (including people and communities), making and physical resources, the qualities and places. This will include assessment of the social, economic and of places. This will include assessment of the social, economic and cultural features of the semiconneurs. This is USMA resources the gradest geographical estimat that couples articles by an inplanead functionate geographical estimat that couples articles by an inplanead function before amounts on the independent and the second parameters the greatest geographical estimat that couples articles by an inplanead function and before amounts on the independent ensempties that any the interact act in the

spin service, before ity impacts and values to reduce the impacts and risk of the activity. It is a requirement under relevant environmental legislatio that these impacts and risks are reduced to as low as reasonably practic (ALARP) and os an acceptation level.

#### Consultation

the Offshore Retroleum and Greenhouse Gas Storage Environmen tions Santos is required to consult with people and organisations, indicons, interests or activities that may be affected by the propose as Functione, interests or activities may include those ensing in

relation to spiritual or cultural connections to land and sea country in accordance with indigenous tradition; tourium, recreational and commercial fishing; other commercial or recreational activities and local commandian that might be affected by our pisposed activity (these are examples and not an exherative list).

#### Seeking information and what's next

Seeking information and what's next hyperprised and Erend OEM for examination to the regulators, a tillaholder mail consult with each hidewark person, including relevant Commonwaithy balar with Nethern Mereinon Department or expensions and persons (or organizations) whose functions, interests or activities may be affected by the activity proposate to be careful out under an EP. Your imput a important to Santas: • as that we can understand the environmental impact, and mails associated with the activity to inform development of the Prinductions Operations EP. (Commonwealth, valuers) and DEMP (VIT wares); • is inform to consultation personase may need to be adapted for different relevant persons; • is many that is activity to the

- to ensure that we provide information to people in an appropriate and accessible manner; and

 to assist with Santos' preparation of the EF. More information on the proposed activities is available on our website

#### Contact us

may be a relevant person, ple ntos to initiate consultation w If you consider you may be a relevant person, please contact us as soon or possible to allow Santoo to initiate consultation with you in relation to the proposed activity to you can tell us how you would like to be consulted. Consultation closes on **Tuesday 9 April 2024**.

Sectors is consistent on Travelog P April 2024. Sectors is constrained to underskip granular and meaningful consultation. We want to provide information for people to make informed assessmelts of the possible consequences of the perpended activity on them. Your input is important to us, and input will be considered in the development of the BP and QBHM.



## Santos

CF and a potential way to get around it.

CF is a chronic disease causes a build-up of mucus in the lungs and airways.

Researchershaveidentified an opportunity to reduce infections in people living with cystic fibrorsis. University of Queensland academics discovered a fault in the bacteria-killing function on on. inclident occurred many of the maximum cells in people with of many of the maximum cells in people with of many of the maximum cells in people with of many of the missiles, including the maximum cells in people with of the many of the missiles, including the maximum cells in people with of the maximum cells in people with the maximum cells in peopl

ing the 2023-24 summers son. The incident occurred about 2pm on Sunday, Janu-ary 2L and ratcheted up when in the five vouths off the premises. "They went over the fence

again, one was armed with a big metal bar," Mr Yan said. "I bunted them out again then they came around to the said. "I had a fairly big bruise on the property armed with sticks and rocks and it took a couple of weeks with a ticks and rocks and it took a couple of weeks the ankle bracelets are pre-began peliting them atus." Mr Yan was able to avoid mensiles, includes the other broken into the lower body. "I ducked and weaved out of the lower body. "I ducked and weaved out of the lower body. "I ducked and weaved out of the lower body. "I ducked and weaved out of the lower body. "I ducked and weaved out of the lower body. "I ducked and weaved out of the lower body. "I ducked and weaved out of the lower body. "I ducked and weaved out of the lower body. "I ducked and weaved out of the lower body. "I ducked and weaved out of the property armed the lower body. "I ducked and weaved out of the property armed the lower body. "I ducked and weaved out of the property armed the lower body. "I ducked and weaved out of the property armed the lower body. "I ducked and weaved out of the property armed the property armed the lower body. "I ducked and weaved out of the property armed the pr

Santos

### SEEKING RELEVANT PERSONS BAROSSA **PRODUCTION OPERATIONS ACTIVITY**

Santos is seeking to identify and consult with relevant persons whose functions, interests or activities may be affected by our proposed Barossa Production Operations Activity in Commonwealth and Northern Territory waters, north-northwest of Darwin

- non-in currently properting: A Production Dependence Environment Plan (EP) relating to the arrival and operations of the Planeting Production Storage and Officialing (EoDV (1993)), operation of a latebase production system and supporting subsise infrastructure, and operation of a 2 35th section of the data Export Paylence (EP) located in Commonshift water where officiane perceluum activities are required under the Officiane Abstractmand Generations and Storage Act 2006 (Coli). An Operations Environmental Management Pain (OEMP) which includes the
- 8.26 km section of the GEP in Northern Territory (NT) coastal enters covered by the Petroleum (Submerged Lanck Act) 7901 (NT), and
- -92km section of the GEP inshore of NT waters covered by the Energy Pipelines Act (98/ (NT).

#### Summary of activity

Summary of devices that the set of the set of the same set of the set of the



VI - NTHEOLZOIMA

from the Barossi field. The dry natural gas resuld be transported through the GSP for endone percensing at the DLKA facility. The condenses would be transported from the PPSD builty to suppose-built tarkers for international econom. The estimated list of the Barossi Development is 25 years, and the Productions Development and the DIPP will be reviewed every from years. Productions Development are expected to commence in 2025, subject to obtaining the required approvals.

Activity location

Activity location The Production Operations Activity is confined to two operations areas Operational Aura (Cold) and Operational Area 2 (Cold). GNL The Environment (Cold) and Operational Area 2 (Cold). The Environment (Cold) and Operational Area 2 (Cold). The Environment (Cold) and Operating Area (Cold) confined to Commonwealth wattrin, agrocommently (20 Km contin-orthwealt confined to Commonwealth wattrin, agrocommently (20 Km contin-orthwealt for Instand, at the Cold) and the Commonwealth (20 Km contin-wealt of the marking AT Cold) and a Cold (20 Km contin-wealth and the marking AT Cold). GNL The 20 Km section of the CBP form (A) to the Commonwealth (Area (Commonwealth) availing and an area (Commonwealth) (20 Km conti-tion) and an area (Commonwealth) waters/AT costal waters (b) Cold (20 Km contined (Sw Baseline).

#### The environment that may be affected by the proposed activity

proposed activity Simole accession impacts and risks to the environment from these prosenee instructs, inclusive or accessions: Ornhalmap seeple and characteristics of the environment of the social environment and cultural features of the environment. The map depicts of the environment that may be affected (EMBA) by the expected activities. The "EMBA represents the greatest real method accessions to reduce the risks are calculated (EMBA) by the expected activities. The "EMBA represents the greatest apertaphical environment that could be effected by an universe of the environment calculated on the environment to the risks are considered. Sentan proposed on implement measures to reduce the risks are considered.

Sentos proposes to implement maxiums to micros dre conserved. of the activity. It is a requirement under relevant environmental legislation that there impacts and rais are reduced to is low as reesonably practice (ALARP) and to an acceptatio leval.

#### Consultation

whether the Chilbrone Petroleum and Greenhouse Gas Storage Environme spulations Santos is required to consult with people and organisations we functions, interests or activities that may be affected by the proce thirties. Functions, laterests or activities may include these arising in

ation to solution) or cultural connections to land and sea resource to spiritual or contrast connectors to find and (aa country in accordance with indigenous tradition: business recreational and converse infi fishing; other commercial or increational activities and local communities that might be official to our proposed activity (these are examples and not an exhaustive (at).

#### Seeking information and what's next

Section 11 and 2014 for submission to the regulators, a tokinology in presenting and DEM for submission to the regulators, a tokinology Statis and herithem Territory Dopartments or agencies and pensions for organizational whose functions, retriests or activities may be affected by the activity preposed to be carried out under an ER.

- but input is important to Sentor: so that we can understand the encoursemental values in the OAs and the encoursent that may be effected, the encoursemental inputs and mixis speciated with the activity to inform development of the "induction Openations EP (Commonwealth waters) and OCHP (NT waters); to inform how consultation processes may need to be adapted for different relevant previous.
- to ensure that we provide information to people in an appropriate and accessible manner; and
- to assist with Santos' preparation of the EP.
   More information on the proposed activities is available on our wabsilts.

#### Contact us

Contract us If you consider you may be a relevant person, please contact us as soon a possible to allow Santos to initiate consultation with you in relation to the proposed activity so you can but as now you would like to be consulted. Consultation closes an Tuesday 9 April 2024.

Sertors is committed to undershifting parture and meminipul consultation. We want to provide information for people to make informed assessments of the possible consequences of the surposed activity on them. Your input is important to us, and input will be considered in the development of the EP and OEMS.



turned to their Coconut Grove home from a trip to Yarralian and she was convicted of man-1994, during which the court heard Ms Secretary's partner ad driven dangerously, inject-ed himself with amphetamines, and assaulted and threated and threaten to kill her. She was assaulted and threatened again at their Kula-

MS Secretary and ner part-court nearly a string of dom-said the party was producto and the party of the party and turned to their Cocout Grove MS Secretary by Mr Nelson, that it was well aware of the 30-



CLP Nightcliff candidate Helen Secretary was the accused in a court case where battered woman syndrome was first recognised as a defence to murder. Picture: Che Chorley

multiple rues therts, police say. The youths were allegedly involved in a Jeep Compass being stolen from a residence

being stolen from a residence in Rapid Creek. The four youths, aged 13 and 14, have been charged with multiple property offences and remain in custody. Investigations are ongoing to locate the six additional alleged offenders.

Santos

### SEEKING RELEVANT PERSONS BAROSSA **PRODUCTION OPERATIONS ACTIVITY**

#### Santos is seeking to identify and consult with relevant persons whose functions, interests or activities may be affected by our proposed Barossa Production Operations Activity in Commonwealth and Northern Territory waters, north-northwest of Darwin

#### Santos is currently preparing

- Santas is currently preparing: A Production Operations Environment Plan (EP) relating to the arrival and operations of the Floating Production Stonge and Offloading factifity (FPS), operation of a subsea production system and supporting subsea infrastructure, and operation of a 205km saction of the Gas Export Plegistic GEP) located in Commonwealth waters where offloares petroliuum activities are regulated under the Offloare Petroleum and Greenhouse Gas Stonge Act 2006 (Cbh). A no Operations Environmental Management Plan (CEMP) which includes the:

  - B.26 km section of the GEP in Northern Territory (NT) coastal waters covered by the Petroleum (Submerged Lands Act) 1987 (NT); and
  - -92km section of the GEP inshore of NT waters covered by the Energy Pipelines Act 1947 (NT).

#### Summary of activity

Summary or accurity The Santo-openetic Blazosa Gas Project II: an offshore gas and condensati project with the purpose of providing a new source of gas to the existing Darwin liquide analying as (DAC) focility at Widdhem Point in the NT. It is intended the natural gas and condensate would be extracted from the Blazosa field, located in Commonwealth waters approximately 288km offshore north-extra them Darwin. Initial processing would occur at the Point Darwin gas, water and condensate extracted



from the Barossa field. The dry natural gas would be transported through the GEP for on-hore processing at the DLNG facility. The condensate international sport. In the PLNG facility to purpose-built tarking for international sport. The entry of the processing the processing the The entremeter life of the Barcssa Development is 25 years, and the Production Operations EP and the GEPP will be revised every fire years. Production Operations activity is expected to commence in 2025, subject to obtaining the required approach.

#### Activity location

Activity focusion The Fraduction Operations Activity is confined to two operational areas: Operational Area 3 (OAT) and Operational Area 2 (OA2). OA2 The Burces and EdI. This is the area in which the FPSQ, tubuse production system, and supporting subses infrastructure will be used to process gas and confernaise estracted from the Burces will. The areas is confined to Commonsealth waters, approximately 208 inm north-orth-ese of Dawin (the Constiting and approximately 208 inm conth-orth-ese of Dawin (the Constiting and approximately 208 inm onth-orth-ese of Dawin (the Constiting and approximately 208 inm onth-orth-ese of Dawin (the Constiting and approximately 208 inm onth-orth-tes).

The transmis as the converse point (seeger samp). OA2: The 285 km section of the GEP from OA16 to the Commonwealth waters/NT waters boundary; and the 8.26 km section of the GEP situated in NT cosstal waters between the Commonwealth waters/NT cosstal waters boundary and the Territorial Sea Baseline). The environment that may be affected by the

The environment that may be arrected by the proposed activity. Earlies is assuing reparts and risks to the environment from these proposed activities, incluive of exceptions (including people and commutities), natural and physical resources, the qualities and characteristics of locations, places and areas and the heritage whus of places. This will include assummarie of the social, economic and cubinal features of the environment. The may displict GAN and GAN, and the environment the may be affected asymptotical extern that could be effected by an upplicate of the environment of the social economic and the asymptotical extern that could be effected by an upplicated worth case' of split scenario, before any measures to reduce the insist are considered.

Santos proposes to implement measures to reduce the impacts and risks of the activity. It is a requirement under relevant environmental legislation that these impacts and risks are reduced to as low as reasonably practice (ALARP) and to an acceptable level.

Consultation

Under the Offshore Petroleum and Greenhouse Gas Storage Environment Regulations Santos is required to consult with people and organisations who have functions, interests or activities that may be affected by the proposed activities. Functions, interests or activities may include those arising in

relation to spiritual or cultural connections to land and sea country in accordance with indigenous tradition; tourium; recreational and commercial finiting; other commercial or recreational activities and local communities that might be affected by our proposed activity (these are examples and not an exhaustive list).

#### Seeking information and what's next

Seeking information and what's next in proparing an IPM of DMP for submission to the regulators, a tilleholder muit consult with each 'relevant person', including relevant Commonwealth, Salar and Northem Tentizon Departments or agencies and persons (or organizational) whose functions, interests or activities may be affected by the activity proposed to be carried out under an ER. Your input is important to Santos:

 a of the ves or understand the environmental values in the GAs and the environment that may be affected, the environmental impacts and risks associated with the activity, to inform development of the Productions Operations EP (Commonwealth waters) and OEMP (NT waters); to inform how consultation processes may need to be adapted for different relevant persons;

to ensure that we provide information to people in an appropriate and accessible manner; and

to assist with Santos' preparation of the EP.
 More information on the proposed activities is available on our website.

#### Contact us

If you cantider you may be a relevant person, please contact us as soon as possible to allow Santos to initiate consultation with you in relation to the proposal activity on you can that just have you would like to be consulted. Consultation closes on **Teessing 7 April 1024.** Santos is correctined to undersking genuins and maningful consultation. We want to provide information for people to make informed assessments of the possible consequences of the proposal activity on them. Your input is important to an input will be considered in the development of the *P* and OLMP.



Vit tasto.com/barous/production-operations, and alfobbre.com/barous/production-operations, and alfobbre.com/barous/production-operations, bardial of the second sec

After The Missing Australia "Just know. I'vegota feeling unmarked grave was dug up from Darwin General Cem-daughter of the sportsman – "I'm hoping, even though it's and the sportsman – "I'm hoping, even though it's and the sportsman – "I'm hoping, even though it's and the sportsman – "I'm hoping, even though it's and the sportsman – "I'm hoping, even though it's and the sportsman – "I'm hoping, even though it's and the sportsman – "I'm hoping, even though it's and the sportsman – "I'm hoping, even though it's and the sportsman – "I'm hoping, even though it's and the sportsman – "I'm hoping, even though it's and the sportsman – "I'm hoping, even though it's and the sportsman – "I'm hoping, even though it's and the sportsman – "I'm hoping, even though it's and the sportsman – I'm hoping, even though it's and the sportsman – I'm hoping, even though it's and the sportsman – I'm hoping, even though it's and the sportsman – I'm hoping, even though it's and the sportsman – I'm hoping, even though it's and the sportsman – I'm hoping, even though it's and the sportsman – I'm hoping even though it's and the sportsman – I'm hoping, even though it's and the sportsman – I'm hoping even though it's and the sportsman – I'm hoping even though it's and the sportsman – I'm hoping even though it's and the sportsman – I'm hoping even though it's and the sportsman – I'm hoping even though it's and the sportsman – I'm hoping even though it's and the sportsman – I'm hoping even though it's and the sportsman – I'm hoping even though it's and the sportsman – I'm hoping even though it's and the sportsman – I'm hoping even though it's and the sportsman – I'm hoping even though it's and the sportsman – I'm hoping even though it's and the sportsman – I'm hoping even though it's and the sportsman – I'm hoping even though it's and the sportsman – I'm hoping even though the sportsman – I'

### SEEKING RELEVANT PERSONS BAROSSA **PRODUCTION OPERATIONS ACTIVITY**

Santos is seeking to identify and consult with relevant persons whose functions, interests or activities may be affected by our proposed Barossa Production Operations Activity in Commonwealth and Northern Territory waters, north-northwest of Darwin

- main is carriently preparing. A Production Operations Environment Plan (EP) relating to the arrival and operations of the Fredering Production Stronge and Offloading facility (FPG), operation of a subset production system and supporting subset infinitructure, and operation of a 205m section of the Gas Export Pipeline (EP) located in Commonwoith water where difficue patroleum activities are regulated under the Offshere Petroleum and Greenhoum Gas Stronge Act 2006 (Ch). An Operations Environmental Management Pion (CEMP) which includes the.
- 8.25 km section of the GEP in Northern Territory (NT) coastal waters covered by the Petroleum (Submerged Lands Act) 1981 (NT); and
- -92km section of the GEP inshare of NT waters covered by the Energy Pipelines Act 1961 (NT).

#### Summary of activity

Summery of activity The Santo-operated Baossa Gar Project is an offshore gas and condensa project with the purpose of providing a new source of gas to the existing Dawn logalificant total gas. (2014;5) facility at Wickhem Porint in the NT. It is interded that natural gas and condensate would be extracted from the Baosa Bield, located in Commonwealth waters approximately 2BBM offshore north-northwest from Dawnin. Initial processing would occur at the PSO family. To separate the natural gas, water and condensate startacted



from the Barossa field. The dry natural gas would be transported through the CBP for onabone processing at the DLNG facility. The condensate would be transferred from the FPSO facility to purpose-built trainers for international export. The reinimated iffe of the Barossa Development is 25 years, and the Production Operations EP and the OEMP will be eviewed every the years. Productions Operations extinty is expected to commence in 2025, subject to obtaining the required approvals.

#### Activity location

Activity location The Production Operations Area 2 (2A2) Dependent Area (2A4) and Operational Area 2 (2A2) DAT: The Operation DAT: A Constitute Area 2 (2A2) DAT: The Operation DAT: DATE 2 (2A2) DATE 2 (2A2)

The environment that may be affected by the

#### proposed activity

proposed activity Example is associated and table to the environment from these proposed activities, inclusive of ecosystems (including people and communities), naturel and physical resources, the quarkies and characteristics of locations, piaces and areas and the heritage value of giaces. This will include assessment of the social economic and cultural adjustment of the social economic and cultural the map depicts OAI and OA2, and the environment that may be affected geographical extent that could be affected by an unplaned "wortt case" of Dill science. Defore any measures to reduce the impactment of the activity. In a sequencement under relearced environmental legislation of the activity. In a sequencement under relearce the impacts and risks (ALARP) and to an acceptable level.

#### Consultation

Under the Offshore Petroleum and Greenhouse Gas Storage Environmer Regulations Santos is required to consult with people and organisations have functions, interests or activities that may be affected by the propo activities. Functions, interests or activities may include those arising in

Santos

In preparing an EP and CBMM for submission to the required as a hitkholder in preparing and EP and CBMM for submission to the required as a hitkholder State and Northern Territory Departments or agencies and persons (or organisation) whole functions, interests or activities may be affected by the activity proposed to be carried out under an EP. Your input is important to Santos: Your input is important to Santos: • so that we can understand the environmental values in the OAs and the environment that may be affected, the environmental impacts and risks associated with the activity, to inform development of the Productions Operations EP (Commonwealth waters) and OEMP (VIT waters); • to inform how consultation processes may need to be adapted for different relevant persons;

nearon to spirituit or cultural connections to land and sea country in accordance with indigenous tradition; toutism; recreational and commercial filling; other commercial or recreational activities and local communities that might be affected by our proposed activity (these are examples and not an exhaustive list).

Seeking information and what's next

- to ensure that we provide information to people in an appropriate and accessible manner, and
- · to assist with Santos' preparation of the EP More information on the proposed activities is available on our website

#### Contact us

Voi contider you may be a relevant person, please contact us as soon as possible to allow Santos to initiate consultation with you in relation to the perposed activity so you can tull us how you would like to be consulted. Consultation closes on **Tuesday 9 April 2024**.

Santos is committed to undertaking genuine and meaningful consultation. We want to provide information for people to make informed assessments of the possible consequences of the proposed activity on them. Your input is important to us, and input will be considered in the development of the EP and CEMP.



Visit santos.com/baressa/preduction-operations, email offshere.consultation is santos.com or call 1800 267 600 for more information, to self-identify as relevant person or to provide feedback.

NTNEOLZO1MA - V1

### 1 March 2024 The Australian

and federal Labor governments, thick process or comparison of the Alamon of the and the comparison of the and the foreign of the and the foreign of the and the foreign of the alabor the transmission of the alabor tra

#### SEEKING RELEVANT PERSONS

### BAROSSA **PRODUCTION OPERATIONS ACTIVITY**

Santos is seeking to identify and consult with relevant persons whose functions, interests or activities may be affected by our proposed Barossa Production Operations Activity in Commonwealth and Northern Territory waters, north-northwest of Darwin

#### Santos is currently preparing

- moto is currently presentions: Environment Plan (EP) relating to the arrival and operations of the Roading Production Storage and Offlooding facility (PESO), pamelino of a subsea production system and supporting subsea infrastructure, and operations of a 205km section of the Gas Export Replice (GEP) located is in Commonwealth waters where offshore petroleum activities are regulated under the Offshore Peordeum and Greenhouse Cas Storage Act 2006 (CIM).
- An Operations Environmental Management Plan (OEMP) which includes
- ac. E26 km section of the GEP in Northern Territory (NT) coastal waters covered by the Petroleum (Submerged Lands Act) 7591 (NT); and -92km section of the GEP inshore of NT waters covered by the Emergy Pipatinas Act 2081 (NT).

Summary of activity Summary or extent services as Project is an offshore gas and condensate project with the purpose of providing a new source of gas to the testing with the purpose of providing a new source of gas to the testing the instructed testing and and condensate would be extracted from the Berossa field, located in Commonwealth waters approximately 205km offshore north-north-network from Condensate would be extracted from the Berossa field, located in Commonwealth waters approximate extracted offshore north-north-network from Condensate would be testing of the PFSO facility, to separate the natural gas, water and condensate extracted from the Berossa field. The dry natural gas would be transported through



the GEP for onshore processing at the DLNG facility. The condensate would be transferred from the PPSO facility to purpose-built trainiers for international apport. The estimated life of the Barossa Development is 25 years, and the Production Development of the DEMM will be revised every five years. Productions Operations a 2014ty is expected to commence in 2025, subject to obtaining the regulard approvade.

#### Activity location

ACLIVITY IOCALION The Production Departions Activity is confined to two operational areas: Ceantinonal Area I (CAI) and Operational Area 2 (CA2). OAI: The Barcess field. This is the zeria in which the FPS, subsea production system, and supporting subsea infrastructure will be used to process gas and condensate entracted from the Barcess webs. The area is confined to Commonwealth webre, approximately 20 km north-north-west of the maintain NT Coastine, and approximately 20 km north-ort the the listwind at the closust point (special listwa).

Trivi travitation at the counter point (savigati trained). OA2: The 285 km section of the GEP from OA11 to the Commonwealth waters: NT waters between the GEP simo OA11 to the Commonwealth in NT coastal waters between the Commonwealth waters/NT coastal waters boundary and the Territorial Sas Baseline).

### The environment that may be affected by the

The environment that may be affected by the proposed activity. State of exceptions (relating people and proposed activity) and acception (relating people and characteristics). The state of the state of the state of proposed activities, incluing of exceptions (relating people and characteristics) of locations, places and yeas and the horitage value of places. This will include assessment of the social, acconomic and cutural distance of the semicorment. The TBMA represents the greatest geographical extent that could be affected by an undiamed word, case of all scenario, before any measures to reduce the impacts and risks of scenario the measures to reduce the impacts and risks and scenario, before any measure to reduce the impacts and risks that these impacts and risks are reduced to as it was reasonably practicable (ALARP) and to an acceptable level.

#### Consultation

Under the Offshore Petroleum and Greenhouse Gas Storage Environm Regulations Santos is required to consult with people and organisation have functions, interests or activities that may be affected by the prop-activities. Functions, interests or activities may include those arising in relation to spitical or cutural connections to brain and and sea country in

accordance with indigenous tradition; tourism; recreational and commercial fishing; other commercial or recreational activities and local communities that might be affected by our proposed activity (these are examples and not an exhaustive list).

#### Seeking information and what's next

Seeking information and what's next
 Imprepring an EP and ORMP for submission to the regulators, a blainbalar
 must cansul with each inferenze increasing inferenze Commonwealth.
 State and Northern Territory Departments or agarcsia and persons (or
 activity processed to be carried out under an EP.
 Vour input Is important to Santos:
 to sum of the environmental values in the OAs and the
 environment that may appreciate the environmental impacts and risks
 associated with the activity, to inform development of the Productions
 Operations IP (Commonwealth waters) and OEMP OF Waters);
 to inform how consultation processes may need to be adapted for
 different relevant persons:
 to ensure that we provide information to people in an appreprinte and
 association memory and
 association the persons of the EP.
 to ensure that we provide information to people in an appreprinte and
 association memory.

#### Contact us

If you consider you may be a relevant person, please contact us as soon as possible to allow Santas to initiate comutation with you in relation to the proposed activity so you can tell us how you would like to be consulted. Consultation closes on **Tuesday 9 April 2024**. Santos is committed to undertaining genuice and meaningful consultation. We want to provide information for people to make informed assussments of the possible consequences of the proposed activity on them. Your input is important to us, and input will be considered in the development of the ED and OEHP.



Voit santos.com/barossa/production-operations. email offshore.com/barossa/production-operations. email offshore.com/barossa/production-operations.



were finally able to identify Gus and organise his funeral. In his victim impact stateJackson Pannam, from Slater and Gordon lawyers, said the family's plight had

rrom when the law doesn't take into account someone in Felix's po-th had sition, it seems like changes

### SEEKING RELEVANT PERSONS BAROSSA **PRODUCTION OPERATIONS ACTIVITY**

Santos is seeking to identify and consult with relevant persons whose functions, interests or activities may be affected by our proposed Barossa Production Operations Activity in Commonwealth and Northern Territory waters, north-northwest of Darwin

#### as is currently pre-

- tota a currently preparing: A Poduction Operations Environment Plan (EP) relating to the arrival and operations of the Floating Production Storage and Offloading facility (FPS), operation of a subset production system and supporting subset infrastructure, and operation of a 285m section of the Gas Export Pipplere (EP) located in Commonweith waters where offshore petrodeum activities are regulated under the Offshore Petroleum and Greenhouse Gas Storage Act 2006 (Ctsh). nt Plan (OEMP) which inclu An Operat the: as Enviro ntal Mar
- 8.26 km section of covered by the Pet on of the GEP in Northern Territory (NT) coastal wat e Petroleum (Submerged Lands Act) 1981 (NT); and
- -92km section of the GEP inshore of NT waters covered by the Energy Pipelines Act 1981 (NT).

#### Summary of activity

Definition of the second secon



from the Barossa field. The dry natural gas would be transported through the GEP for enhance processing at the DLNG facitity. The condensate international export. The second barbon second the second the estimated life of the Barossa Development is 25 years, and the Production Operations IEP and the OEPM will be enviewed every five years. Productions Operations activity is expected to commence in 2025, subject to obtaining the required approvals.

#### Activity location

Accuracy to Control of the Production Operational Areas: Operational Area I (CAA) and Operational Area 2 (CA2). OAI: The Brossen field. This is the area in which the P260, Subsea production system, and supporting subsea infrastructure will be used to process gas and condensate astracted from the Barosta wells. The area is confided to Commonwealth waters, approximately 205 km north-north-well of Darwin (the closest mayor opoulded centre), approximately 206 km north-well of the maintain RT coastline, and approximately 310 km north-orth-well of the maintain RT coastline, and approximately 310 km north-oft helf the first of the closest point (Sequer) Listed).

Imm Issess is inc. BOBBER, politic (Seegul I Island). OA2: The 285 km section of the GEP from OA18 to the Commonwealth waters/ NT waters boundary; and the 8.26 km section of the GEP situated in NT coastal waters between the Commonwealth waters/NT coastal waters boundary and the Territorial Sea Baselink).

#### The environment that may be affected by the proposed activity

proposed activity Statistics inclusive of ecosystems (Including beaple and communities), inclusive of ecosystems (Including beaple and communities), inclusive of the social, ecosities and characteristics of locations, places and area and the heritage value of characteristics of locations, places and area and the heritage value of characteristics of locations, places and area and the heritage value of characteristics of locations, places and area and the heritage value of characteristics of locations. The the social, ecosimic and cultural failures of the environment. The time of the social, ecosimic and cultural geographical extent that could be affected by an unplanned twords case of plail scennice, before any measures to reduce the train are considered. Santos preposes to implement measures to reduce the impacts and risks that there impacts and risks are induced to as low as reasonably practicable (ALARP) and to an acceptable level.

#### Consultation

Under the Offshore Petroleum and Greenhouse Gas Storage Environm Regulations Santos is required to consult with people and organisation have functions, interests or activities that may be affected by the prop activities. Functions, interests or activities may include those arising in

÷

# Santos

relation to spiritual or cultural connections to land and sea country in accordance with indigenous tradition; tourism; recreational and commercial finding; other commercial or recreational activities and local communities that might be affected by our proposed activity (these are examples and not an exhaustive IIIO).

#### Seeking information and what's next

Secting innormation and while a next in preparing an EP and GEM for submission to the regulators, a titleholder must consult with each relevant person; including relevant Commonwealth, State and Norther Tentroty Departments or agencies and persons (or organisations) whose functions, interests or activities may be affected by the activity proposed to be carried out under an EP. Your input is important to Santos:

- so that we can understand the environmental values in the QAs and the environment that may be affected, the environmental impacts and risks associated with the activity, to inform development of the Productions Operations EP (Commonealth waters) and DEMP (NT waters);
- to inform how consultation processes may need to be adapted for different relevant persons;
- attreent reavant perions; to ensure that we provide information to people in an appropriate and accessible manner; and to assist with Santos' preparation of the EP.
- · More information on the proposed activities is available on our website

#### Contact us

If you cansider you may be a relevant person, please contact us as soon as possible to allow Santos to initiate consultation with you in relation to the proposed activity is o you can the low how you would like to be consulted. Consultation closes on **Tuesday 9 April 2024**. Santos is committed to undertaining genuine and meaningful consultation. We want to provide information for people to make informed assessments of the possible consequence of the proposed activity on them. Your input is important to us, and input will be considered in the development of the *E*<sup>-</sup> and OBMP.



Visit santos.com/barossa/production-operations. email effichere.censulfatilionissantos.cem era all 900 225 600 for ence information. to sef-identify as relevant person or to provide feedback

travel plans this year — a trip that's even more attractive now with these discounted airfares and other incentives," she said.

unchecked can lead to over tou-rism in an area," Mr Hall said. "What we want to see is proper

### SEEKING RELEVANT PERSONS BAROSSA **PRODUCTION OPERATIONS ACTIVITY**

Santos is seeking to identify and consult with relevant persons whose functions, interests or activities may be affected by our proposed Barossa Production Operations Activity in Commonwealth and Northern Territory waters, north-northwest of Darwin

- Into is currently preparing: A Production Operations Environment Plan (EP) relating to the arrival and operations of the Floating Production Storage and Offloading facility (FPSO), operation of a subses production system and supporting subsea inflastructure, and operation of a 258m section of the Gas Export Ppeline (GEP) located in Commonwealth waters where offshore petroleum activities are regulated under the Offshore Petroleum and Greenhouse Gas Storage Art 2005 (CIb). An Operations Environmental Nanagement Plan (DEMP) which includes the
- " 8.26 km section of the GEP in Northern Territory (NT) coastal wate covered by the Petroleum (Submerged Lands Act) 388T (NT); and -92km section of the GEP inshore of NT waters covered by the Energy Pipelines Act /981 (NT).

#### Summary of activity

mmary or activity Storbo-potential Baossa Gas Project is an offshore gas and condensate ext with the purpose of providing a new source of gas to the existing win liquified natural gas (DLNS) fourfliky at Wickham Point in the NT. Baossa field, Located in Commonwealth waters approximately 268im Baossa field, Locate in Commonwealth waters approximately 268im none north-northwest from Dawin. Initial processing would occur at the 0 Safity, to spassate the natural gas, water and condensate effective The Sa



from the Baresse field. The dry natural gas would be transported through the GEP for another processing at the DLAM facility. The condensate would be transformed from the FPSO facility to purpose-built tankars for international export. The estimated life of the Barossa Development is 25 years, and the Production Operations EP and the OEPM will be enviewed every five years. Productions Operations EV and the OEPM will be enviewed every five years. Productions Operations EV and the OEPM will be enviewed every five years.

#### Activity location

Activity location The Production Operations Activity is confined to two operational areas: Operational Areas I (OAA) and Operational Areas 2 (OA2). OAB: The Baross total, Takis the areas is which the FPSO, subsca production system, and supporting subsea interativities will be used to process gas and condensate estratistical form the Barossa wells. The areas is confined to Commonwealth subserve, approximately 2015 km north-north-well of Darwin (Mer downs point Caregan Likedon). Their Islands at the downs fount Caregan Likedon. OA2: The 3015 km section of the GEP miss DoAI section of the GEP alluaded In Mit Coatal alwars between the Commonwealth activity. Coastal waters boundary and the Territorial Sea Baseline).

#### The environment that may be affected by the proposed activity

#### Consultation

Under the Offshore Petroleum and Greenhouse Gas Storage Environm Regulations Santos is required to consult with people and organization have functions, interests or activities that may be affected by the prop activities, Functions, interests or activities may include those arising in

Santos

Seeking information and what's next Second information and What's next in propulsing an End ObeN or submission to the resplators, a titleholder must consult with each research person, including relevant Commonwalth. Stata and Horthma Mentory Departments or agencies and persons (or organisations) whose functions, interests or activities may be affected by the activity proposed to be carried out under an ER. Your input is important to Santos:

meanon to period or current commissions to and and see see accordance with Indigenous findition; tourism; recreational and commercial fishing; other commercial or recreational activities and local communities that might be affected by our proposed activity (these are examples and not an orihoustive list).

- Your input is important to Samos: so that we can sudentiated the environmental values in the OAs and the environment that may be affected, the environmental impacts and rules associated with the activity, in Inform development of the Productions Operations EP (Commonwealth waters) and OEMP (HT waters); to inform how consultation processes may need to be adapted for different relevant persons;
- to ensure that we provide information to people in an appropriate and accessible manner; and
- to assist with Santos' preparation of the EP.
   More information on the proposed activities ies is available on our website

#### Contact us

If you consider you may be a relevant person, please contact us as soon as possible to allow Santos to initiate consultation with you in relation to the proposed activity so you can tell us how you would like to be consulted. Consultation closes on **Teendry 9 April 2014**.

Sensitive of the sensity # April 2024. Sentos is committed to underskip genuine and meaningful consultation. We want to provide information for people to make informed assamments of the possible consequences of the proposed activity on them. Your input is important to us, and input will be considered in the development of the EP and OEMP.





Visit sardes con/baressa/production-operations. email antibure: consultation 8 sardes com or call topo 257 000 for more information. Its sard dettity as relevant person or to provide feedback.

acquisition would enable the group to work cruising into an integrated accommodation and hospitality offering in the Kimberley.

Australia, about 60 per cent of local government-operated regional air-ports operate at a loss due to ageing infrastructure, rising security and regulatory burdens and staffing costs.

across the board, and they can no longer afford to pick up the short-fall," she said. "Australia's regional airports provide a critical role in border protection, medivac, defence and

and our nation's health and secur-ity at risk." It comes as the Australian Feder-ation of Air Pilots went on strike this week with 230 pilots walking off the job and further action set

rely on tightly-run fly-in, fly-out schedules to get workers to site, forcing them to use other pro-viders, reschedule travel or find spare capacity on other Qan-tas flights.

Santos

### SEEKING RELEVANT PERSONS BAROSSA **PRODUCTION OPERATIONS ACTIVITY**

Santos is seeking to identify and consult with relevant persons whose functions, interests or activities may be affected by our proposed Barossa Production Operations Activity in Commonwealth and Northern Territory waters, north-northwest of Darwin

- Moto S currently preparing and operations three Totaling Production Stronge and Officiality and operations of the Floating Production stronge and Officiality ficibly (PRSO), operation of a autospectra of the stronge subset infrastructure, and operations of a 205m section of the Gas Expert Pipeline (CDP) located in Commonweilth weaks where offshore patients and stronge Art 2006 (CDI).
- An Operations Environmental Management Plan (OEMP) which includes e: 8.26 km section of the GEP in Northern Tenitory (NT) coastal waters covered by the Petroleam (Submerged Lands Act) 1981 (NT): and -92km section of the GEP inshore of NT waters covered by the Energy Pipekins Act 3981 (NT).

#### Summary of activity

Statistical backwards and a state of providing a new source of gas and condensate project with the purpose of providing a new source of gas to the existing parent linguined natural gas (DNA) focility at Witekam Port in the NT. It is intended that natural gas and condensate would be extracted from the atomas field, located in Commonwealth waters approximately 2020 the atomas field (Located in Commonwealth waters approximately 2020 the atomas field (Located in Commonwealth waters approximately 2020 the atomas field (Located in Commonwealth waters approximately 2020 the atomas field (Located in Commonwealth waters approximately 2020 the provided (Located in Commonwealth waters and condensate extracted from PSO facility, to separate the natural gas, water and condensate extracted in the common set of the se



- from the Barossa field. The dry natural gas would be transported throug the GEP for orshore processing at the DLNG facility. The condensate would be transferred from the FPSO facility to purpose-built tankers for international export.
- international export. The estimated life of the Barossa Development is 25 years, and the Production Operations EP and the OEMP will be reviewed every fix Productions Operations activity is expected to commence in 2025, to obtaining the required approals.

#### Activity location

The Production Operations Activity is confined to two opera Operational Area 1 (DA1) and Operational Area 2 (DA2). Coentinent Area 1 (OAI) and Operational Area 2 (OA2). OAE The Brease thich Teis take was in which the PEQS, subase production system, and supporting subase inheatructure will be used to process gas and condenside extractof from the Bacess wells. The area is confined to Commonwealth waters, approximately 285 km north-north-west of burns in the closest point coassilie, and approximately 130 km north-west of the manifest AT coassilie, and approximately 130 km north-west of the manifest AT coassilie, and approximately 130 km north-west of the manifest AT coassilie, and approximately 130 km north-west of the coassiliest Coassiliest Coassiliest ADA to the Commonwealth waters/ NT waters bloandary, and the 28 km soction of the GEP situated in NT coassilia waters between the Commonwealth waters/NT coastal waters boundary and the Interform 4 as Baselino.

#### The environment that may be affected by the proposed activity

proposed activity Satos is assessing impacts and risks to the environment from these proposed activities, including opposed and communities, natural and physical resources, the qualities and piece. This will include assessment of the social, economic and cultural features of the environment.

features of the environment. The may depict CAI and CA2, and the environment that may be affected (ENBA) by the proposed activities. The "EMBA" represents the greatest geographical activities. The "EMBA" represents the case of gala somarks, before any measures to reduce the risks are considered. Sandas proposes to implement measures to reduce the risks are considered. Sandas proposes to implement measures to reduce the lineacity and risks of the activity. It is a requirement under relevant environmental legislation that these represest and risks are educed to as low as reasonably practicable (ALARP) and to an acceptable level.

#### Consultation

Under the Ottshore Petroleum and Greenhouse Gas Storage Environment Regulations Santos is required to consult with people and organisations have functions, interests or activities that may be affected by the propos activities, Functions, interests or activities may include those arising in

\*\*\*extron to spinitual or cultural connections to land and sea country in accordance with indigenous tradition; tourism; excretational and commercial failing; other commercial or excretational activities and local communities that might be affected by our proposed activity (these are examples and not an exhaustive list).

#### Seeking information and what's next

Deteking information and what's next hip reparing on IP and DEMP terror burnising to the regulators, a tillsholder must consult with each 'relevant person', including relevant Commonselffu, Salar and Northem Tenting Departments or agencies and persons (or organizationi) whose functions, interests or activities may be affected by the activity propesate to be carried out under an EP. Your input is important to Santos: • so that we can understand the environmental values in the OAs and the environment that may be affected, the environmental impacts and risks associated with the activity; to inform development of the Poductions Operations EP (Commonwealth waters) and DEMP (NT weeks). • to Inform the consultation concesses may need to be available for

- to inform how consultation processes may need to be adapted for different network persons to ensure that we provide information to people in an appropriate and accessible manner, and
- to assist with Santos' preparation of the EP.
   More information on the proposed activities

#### Contact us

If you consider you may be a relevant person, please contact us as soon a possible to allow Sentos to initiate comutation with you in relation to the proposed activity so you can tell us how you would like to be consulted. Consultation closes on **Tweelay 9 April 2024**.

Construction cross on Teesday # April 2024. Stands is committed to underskip genuine and meaningful consultation. We want to provide information for people to make informed assessments of the possible consequences of the proposed activity on them. Your input is important to us, and input will be considered in the development of the EP and OEHP.



Viat sardos.com/barosas/production-operations. errel officions.com/barosas/production-operations.com or call 000 225 000 for more information. It suf-identify as relevant person or to provide feedback

o enable the city to thrive." the 2500sq m complex is set to \$600,000 in total, by the end fall and other liquids that gath-The major projects that are house a 50m, 25m and resort of the financial year, the eramong waste. The upgrade included the Improvements duplication of Lee Point Rd As part of a \$7m program,

### SEEKING RELEVANT PERSONS BAROSSA **PRODUCTION OPERATIONS ACTIVITY**

Santos is seeking to identify and consult with relevant persons whose functions, interests or activities may be affected by our proposed Barossa Production Operations Activity in Commonwealth and Northern Territory waters, north-northwest of Darwin

- real is careedly proparing A Production Operations Environment Plan (EP) relating to the annual and operations of the Floating Production Starspe and Officialling facility (FFRO), coverising of a sublem production system and supporting subset profiler (GFP) boated in Commonwealth waters where affinitione participant activities are regulated under the Other Annual Annual Generalized Stange Arz 2006 (CIN). An Operational Environmental Management Ran (OEMP) which includes the
- 8.36 km section of the GEV in Northern Territory (NT) coastal waters covered by the Petroleum (Submerged Lands Act)  $\rm MBP(\rm O(T))$  and
- -92em section of the GEP inshere of NT waters covered by the Energy Popelines Act 7991 (147).

#### Summary of activity

The limiton-spectral Biarcosis Gas Project is an offshore gas and sonderna anopet: with the purpose of providing a new source of gas to the existing Down in together annual gas COU-bio Nativity at Workshim Fraint in the NT. It is intended that natural gas and condensate would be extracted from the Biarcosis field locatied in Commonwaith weaters, approximately 205km offshores north-nerthweat them Davis. Initial processing would occur all PSO fability. Is largerate than acturating gas, well and condensate extinction



from the Baressa field. The dry natural gas would be transported to the GEP for coshore processing at the DLNG Incitity. The condensa would be transferred from the FPSO facility to purpose-built tanks interrutional separt.

Interferences (Ropert, The spinitural of file Statution Development in 25 years, and the Production Obserstoris XP and the ORMP will be reviewed every five years. The obserstories activity is expected to commerce in 2025, subject to obtaining the required activity is expected to commerce in 2025, subject

#### Activity location

Activity location The Production Operations Activity is confirmed to two operational areas: Development of Average 100AH and Operational Areas 2 (0A2). GNL The Barrase High This (the area is which the PFaC), so that provide the Barrase High This (the area is which the PFaC), and the provide the Confirmed Area the Area (the Area (the Area (the Confirmed High Confirmed Area (the Area (the Area (the Oranies) High Confirmed Area (the Area (the Area (the Oranies) High Confirmed Area (the Area (the Area (the Oranies) High Confirmed Area (the Area (the Area (the Oranies) High Confirmed Area (the Area (the Area (the Oranies) High Confirmed Area (the Area (the Area (the Oran (the Area (the

### The environment that may be affected by the

The environment that may be affected by the proposed activity. Summarizes the environment from these organed activities inclusion of economisms (Naturdia ganesias and hereactivities characteris of the environment from these organed activities, inclusion of economisms (Naturdia ganesias and hereactivities of characteris, size, and areas and the herebage value, of advances of the environment. The tensis economis and cultural herearce after encoded acculate the environment that may be effected (EMAA) by the encoded acculate the Mither environment that may be effected activities and the encoded acculate the Mither environment that may be effected advances after encoded acculate the Mither environment the ensure of the advance products and mither any environment theory environment advances products and mither any environment environment and the advances products and mither any environment environment and the advances products and mither any environment environment and the advances advances and the environment environment and the advances advances and the environment environment and the advances advances and the second activities and advances advances and the second activities and advances advances and the second activities advances and advances advances and the second activities advances advances and advances advances and the second activities advances advances advances and advances adv

#### Consultation

Under the Offshore Petroleum and Greenhouse Gas Storage Environmen Regulations Santas is required to consult with people and organisations have functions, interests or activities that may be affected by 5% propos offshotes. Environment and activities may include those arbitra in activities. Functions, interests or activities may include those arbitra.

## Santos

relation to spiritual or cultural connections to land and sea country in accordance with indigenous tradition; tourism; recreational and commercial fibring, other commercial or enventional activities and local communities that might be affected by nor proposed activity (these are exemples and not an exherptive list).

#### Seeking information and what's next

In preparing on EP and OEPP for submoslow to the regulators, a siteholder must consult with each relevant person, including relevant Commonwealth, Satu and Nothern Temforp Departments an agencia and persons (an organisations) whole functions, intelests or activates may be affected by the activity processed to be careted out under an EP. Your input is important to Sentos

- so thit we can understand the environmental values in the QAs and the environment that may be affected, the environmental impacts and risks associated with the activity, to inform development of the Productions Counsiliums (E. Commonwealth waters) and QEMS (MT waters),
- to inform how consultation processes may need to be adapted for different relevant persons. .
- to ensure that we provide information to people in an appropriate and accessible manner, and
- accession manner, and to adult with Santos' preparation of the EP. More information on the proposal activities is available on our website.

#### Contact us

Typic consider the intypic consideration of the selection of the second as possible to allow Sentos to initiate consultation with you in relation to the promoted activity so you can be full us how you would like to be consulted. Consultation closes on **Tuesday 5 April 2024**. Some constant of the state of the second state of the second state of the state of



Alter and a second barrane by reductive specializes, and affabre construction grants.com or cell 1900 327 680 for most information, to ad-ative science of priority or information, to ad-

NUMBER OF STREET

### Consultation

RadioMarch-April 2024Radio ad - Hit 101.3 BroomeMarch-April 2024Radio ad - Darwin Mix 1049March-April 2024Radio ad - Darwin Hot 100March-April 2024Radio ad - Pilbara and Kimberley Aboriginal Media Radio

### Script B - Consultation 11 March to 9 April

Santos is now consulting with people who maybe affected by the proposed Production Operations Activity for the Barossa Gas Project.

This includes Santos's offshore production facility, approximately 285 kilometres offshore from Darwin, and a Gas Export Pipeline.

If you may be affected by these activities, please contact Santos as soon as possible. Consultation closes on 9 April 2024

For more visit <u>santos.com/barossa</u>,

Phone [1800 267 600] One Eight Hundred, Two Six Seven, Six Hundred or email <u>offshore.consultation@santos.com</u>.

## Santos

## THE BAROSSA GAS PROJECT PRODUCTION OPERATIONS ACTIVITY CONSULTATION



#### Santos is now consulting with relevant persons for its proposed Production Operations activity and is currently preparing:

- A Production Operations Environment Plan (EP) relating to the arrival and operations of the Floating Production Storage and Offloading facility (FPSO), operation of a subsea production system and supporting subsea infrastructure, and operation of a 285km section of the Gas Export Pipeline (GEP) located in Commonwealth waters where offshore petroleum activities are regulated under the Offshore Petroleum and Greenhouse Gas Storage Act 2006 (Cth).
- An Operations Environmental Management Plan (OEMP) which
  includes the:
  - 8.26 km section of the GEP in Northern Territory (NT) coastal waters covered by the Petroleum (Submerged Lands Act) 1981 (NT); and
- 92km section of the GEP inshore of NT waters covered by the Energy Pipelines Act 1981 (NT).

More information is available at santos.com/Barossa about who may be a relevant person for consultation, the proposed activity, potential environmental impacts and risks, and proposed control measures to seek to reduce any impacts and risks to as low as reasonably practicable and an acceptable level.

We are asking relevant persons to provide feedback by 9 April 2024. Please contact us via any of the below channels to discuss consultation or provide feedback.

For more information: Visit santos.com/barossa Phone 1800 267 600 Email offshore.consultation@santos.com or scan the QR code



NTNE01201MA - VI

leagues at a bar in the nearby ed this. "Perenti categorically lice were responsible for the queries about missing items. 2020 and 2021. Santos Santos is now consulting with relevant persons for its proposed Production **Operations activity and is currently** preparing: **THE BAROSSA**  A Production Operations Environment Plan (EP) relating to the arrival and operations of the Floating Production Storage and Offloading facility (FPSO), operation of a subsea production system and supporting subsea infrastructure, and operation of a 285km section of the Gas Export Pipeline (GEP) located in **GAS PROJECT** Commonwealth waters where offshore petroleum activities are regulated under the Offshore Petroleum and Greenhouse Gas **PRODUCTION OPERATIONS** Storage Act 2006 (Cth). · An Operations Environmental Management Plan (OEMP) which **ACTIVITY CONSULTATION** includes the: 8.26 km section of the GEP in Northern Territory (NT) coastal waters covered by the Petroleum (Submerged Lands Act) 1981 (NT); and THEFT PROPERTY -92km section of the GEP inshore of NT waters covered by the Energy Pipelines Act 1981 (NT). More information is available at santos.com/Barossa about who may be a relevant person for consultation, the proposed activity, potential environmental impacts and risks, and proposed control measures to seek to reduce any impacts and risks to as low as reasonably practicable and an acceptable level. We are asking relevant persons to provide feedback by 9 April 2024. Please contact us via any of the below channels to discuss consultation or provide feedback. For more information: 回答 Visit santos.com/barossa Phone 1800 267 600 Email offshore.consultation@santos.com or scan the QR code

**THE BAROSSA** 

**GAS PROJECT** 

**PRODUCTION OPERATIONS** 

ACTIVITY CONSULTATION

Transmass 83.

Katherine CBD. Town Council CEO Ingrid Stonhill said a survey last

preparing:

The hope is to retain past food vendors and attract new ones to a "bigger and the vendors and attract new ones to a "bigger and the vendors and attract new ones to a "bigger and the vendors and the vendors are the vendors and the vendors and the vendors are the vendors and the vendors are the vendors are the vendors and the vendors are the vendors are the vendors and the vendors are the vendors and the vendors are the vendors are the vendors and the vendors are the vendors

### Santos is now consulting with relevant persons for its proposed Production **Operations activity and is currently**

- A Production Operations Environment Plan (EP) relating to the arrival and operations of the Floating Production Storage and Offloading facility (FPSO), operation of a subsea production System and supporting subsei infrastructure, and operation of a 285km section of the Gas Export Pipeline (GEP) located in Commonwealth waters where offshore petroleum activities are regulated under the Offshore Petroleum and Greenhouse Gas Storage Act 2006 (Cth).
  - · An Operations Environmental Management Plan (OEMP) which includes the:
    - 8.26 km section of the GEP in Northern Territory (NT) coastal ers covered by the Petroleum (Submerged Lands Act) 1981 (NT); and
    - -92km section of the GEP inshore of NT waters covered by the Energy Pipelines Act 1981 (NT).

More information is available at santos.com/Barossa about who may be a relevant person for consultation, the proposed activity, potential environmental impacts and risks, and proposed control measures to seek to reduce any impacts and risks to as low as reasonably practicable and an acceptable level.

We are asking relevant persons to provide feedback by 9 April 2024. Please contact us via any of the below channels to discuss consultation or provide feedback.

#### For more information:

Visit santos.com/barossa Phone 1800 267 600 Email offsh nsultation@santos.com or scan the QR code



NTREGLIZGINA - VI

ing sensors in schools, which Ms Thorpe said was "only one solution and a costly one at that". March 1. She said there were duced in parliament. There will "transitional arrangements" in be no penalties for people who place until July 1 to allow time for vape sponsors and distrib-vapes for personal use."



tered in May 2016 and jointly owned by Mr Edgington and Mr Newman until its deregis-tration in May last year.

"This afternoon I have noti-fied the Leader of the Oppo-sition that I have resigned from the role of shadow minister for the role of shadow minister for

Department of INDUSTRY, TOURISM AND TRADE

## **EXPRESSIONS OF INTEREST**

### CHAIRPERSON FOR McARTHUR RIVER MINE INDEPENDENT EXPERT MINE **CLOSURE PANEL**

The Minister for Mining is seeking expressions of interest from technical experts to be appointed as Chairperson for the independent panel of experts for McArthur River Mine focusing on Mine Closure.

A suitably credentialed individual will be appointed to deliver an independent technical review of action taken to achieve closure of the McArthur River Mine in line with approved plans.

The independent expert will have access to McArthur River Mine operations and performance information focusing on the three-year period following approval of the Overburden Management Project in November 2020.

The Chairperson will play an important role in demonstrating accountability of the mine operator to the wider community.

Prior to applying, please refer to the Terms of Reference and selection criteria at industry.nt.gov.au/mrm-panels

Written nominations should be emailed to Andria Handley, Director Mining Operations Policy and Support, andria.handley@nt.gov.au

Nominations close at midnight on Sunday 21 April 2024.

### Santos COMMUNITY CONSULTATION **DROP-IN SESSIONS** BAROSSA GAS PROJECT ACTIVITY

**YOUR VIEWS ARE** 

# **IMPORTANT TO US.**

Santos is preparing environment plans, as required by legislation, regarding activities for the Barossa Gas Project - a project that involves extracting natural gas from the Barossa field in Commonwealth waters approximately 285 kilometres offshore north-north west of Darwin and transporting it to the existing Darwin liquefied natural gas (DLNG) plant.

We are currently consulting with relevant persons whose functions, interests, or activities may be affected by the Production Operations Activity as we prepare a Production Operations Environment Plan and an Operations Environmental Management Plan. Detailed information about these proposed activities is available at Santos.com/barossa or by scanning the QR code below. Santos will hold community consultation drop-in sessions for relevant persons to obtain information about these proposed activities, provide feedback and ask any

DATES:

#### questions VENUE:

Santos Shopfront T14/15, 41 The Mall, Darwin

CONTACT US

For more information please scan QR code

TIME:

26<sup>TH</sup> March, Tuesday 27<sup>TH</sup> March, Wednesday 28<sup>TH</sup> March, Thursday 10.30am - 2.30pm





information, and said early evi-dence supported allegations that awards racist towards First Citing a need for the "serious Citing a need for the "serious

# Santos THE BAROSSA **GAS PROJECT PRODUCTION OPERATIONS ACTIVITY CONSULTATION**



#### Santos is now consulting with relevant persons for its proposed Production **Operations activity and is currently** preparing:

- · A Production Operations Environment Plan (EP) relating to the arrival and operations of the Floating Production Storage and Offloading facility (FPSO), operation of a subsea production system and supporting subsea infrastructure, and operation of a 285km section of the Gas Export Pipeline (GEP) located in Commonwealth waters where offshore petroleum activities are regulated under the Offshore Petroleum and Greenhouse Gas Storage Act 2006 (Cth).
- An Operations Environmental Management Plan (OEMP) which includes the:
  - 8.26 km section of the GEP in Northern Territory (NT) coastal waters covered by the Petroleum (Submerged Lands Act) 1981 (NT); and
- -92km section of the GEP inshore of NT waters covered by the Energy Pipelines Act 1981 (NT).

More information is available at santos.com/Barossa about who may be a relevant person for consultation, the proposed activity, potential environmental impacts and risks, and proposed control measures to seek to reduce any impacts and risks to as low as reasonably practicable and an acceptable level.

We are asking relevant persons to provide feedback by 9 April 2024. Please contact us via any of the below channels to discuss consultation or provide feedback.

#### For more information:

Visit santos.com/barossa Phone 1800 267 600 Email offshore.consultation@santos.com or scan the QR code



# Table 4-10 (of the EP) Advertising Tiwi sessions

- Social media notices
- Notice of Consultation emailed to several independent stakeholders for sharing across their direct networks, in person, and for posting on Tiwi Island notices boards

### March 2024

Santos Ltd Im · ● Details of our upcoming consultation sessions on the Tiwi Islands are available below. We look forward to chatting with you early next month. For more information, visit: www.santos.com/barossa

### Santos

NOTICE OF CONSULTATION WITH TIWI ISLANDS PEOPLE BAROSSA GAS PROJECT ACTIVITIES

### YOUR VIEWS ARE IMPORTANT TO US.

Santos is preparing Environment Plans (EPs) for the following proposed activities, as required by legislation:
Develop Explore Duplication (DPD) Environment Plans - ths Plase note that the in the final consultation session for the D

Darwis Pipeline Duplication (DPD) Environment Plan - this relates to the 35 km portion of the underwater DPD pipeline and supporting subsea infrastructure in Commonwealth waters 37 km south-vest of the Train stands.

Production Operations Environment Plan - this yealars to the annual and operations of the Floating Production Storage and Offloading (FFSO) facility, operation of a subsee production system and supporting subsea infrastructure, and operation of a 265 Am leader of the data Export Repeire (GEP) located in Commonwealth waters.

Operation Environmental Managemeet Plan (Production Operations): Two OMP relates to the 100 km portion of the 60 and supporting subset infrastructure located in Northern Territo Waters and laud, extending to the existing Darwin Liquefied Natural Gas (DLHG) facility

functions, interests or activities may be affected by activitie proposed under the EPs or the OERP listed above. Bared on input from Thei Islands people, we will consult with you at it sessions detailed in this notice through Clan group meetings with videos and visual aids available.

#### At the upcoming sessions we will: • Continue consulting with relevant persons at

- Continue consulting with relevant persons about our propos activities under the DPD EP, including. providing responses to information prohided and queries reaved about the DPD EP by Thirs tillands people, following
- previous Clan meetings.
   updating you about any measures we propose to adopt the DPD EP as a result of your information and comment

before it is submitted to the regulator for assessment. I dray input is outstanding, seeking your final input on possible consequences of the proposed DPO activity and any further measures you would like us to consider to reduprocessing the second persons about our grouposed activities, under the Moduction Operations (Ef and CBM) instance. The second CBM instance is the second test of test o

#### TIMING AND VENUES

Tuesday 5 Harch 2024, Hilikapiti, Sports & Recreation Centre 0.30an - Nenrikawyanga & Vimpinari Clans 100pm - Wullrankumu Clan Wedhesday 6 Harch 2024, Warrumiyanga, Hantiyyawi Motel

10.30am - Hardiyupwi Clan 100pm - Jikilarusu Clan Thuruday 7 March 2024, Warrumiyanga, Mantiyupwi Metel 10.30am - Wurakusu Clan 190pm - Halavu Clan

Friday 8 March 2024, Pirlangimpi, Sports & Social Clut 10.30am - Hunupi Clan



### April 2024



...

Details of our upcoming consultation sessions on the Tiwi Islands are available below. We look forward to chatting with you in April. For more information, visit www.santos.com/barossa



### May 2024



Top Contributor · 27m · 😁

Details of our upcoming consultation sessions on the Tiwi Islands are available below. We look forward to chatting with you in May. For more information, visit www.santos.com/barossa



...



Top contributor · Just now · 😁

Reminder today: Santos Munupi Clan Meeting at Pirlangimpi Sports Club at 10:30am.



## **NOTICE OF CONSULTATION WITH** TIWI ISLANDS PEOPLE BAROSSA GAS PROJECT ACTIVITIES

### YOUR VIEWS ARE IMPORTANT TO US.

Santos is preparing Environment Plans (EPs) for the following proposed activities, as required by legislation:

# Darwin Pipeline Duplication (DPO) Environment Plan - this relates to the 23 km sportion of the anderwarter DPD pipeline and supporting subses infestivatures in Commonwealth waters 27 km subth-wat of the Third Islands.

Production Operations Environment Plan - this relates to the Presultion Operational Environment Plan - this instance to the annuel and operations of the Floating Production Storage and Officialing (PPSO) facility, operation of a subsex production system and supporting subsex Intrastructure, and operation of a 205 km section of the Gas Export Pipeline (GEP) located in Commonwealth waters.

### Operation Environmental Management Plan (Production

Constituents - the OEMP relates to the IOO kin particip of the GEP and supporting subset infrastructure located in Northern Territory Waters and land, extending to the existing Darwin Liquefiel Network (DLNG) Tecility.

We are currently consulting with Tiwi Islands people whose We are currently consuming with the intends people whole functions, letterets or activities may be affected by activities proposed under the EPs or the OEMP listed above. Based on input from Tirk listed people, we will consult with you at the essions detailed in this notice through Clas group meetings with videos and visual also available.

#### At the upcoming sessions we will:

- Centinue consulting with relevant persons about our proposed activities under the DPD EP: including:
- providing separates to information provided and queries raised about the DPO EP by Tixl Islands people, following previous Clen meetings.
- updating you about any measures we propose to adopt in the DPD EP as a result of your information and comments before it is submitted to the regulator for assessment.
- If any input is outstanding, seeking your final input on possible consequences of the proposed DPD activity and any further measures you would like us to consider to reduce impacts and risks.

### CONTACT US

T: 1800 267 600 E: offshore.consultation@santos.com For more information please scan QR codes:



- Consult with relevant persons about our proposed activities under the Production Operations EP and OEMP, including: providing information and responding to questions about the proposed activities, potential impacts and risks and how we plan to reduce three to as low as reasonably practicable and to an acceptable level.
- Inviting you to consider the information given and tell us if you usek further or different information.

#### TIMING AND VENUES

Tuesday 5 March 2024, Hillkapiti, Sports & Recreation Centre

10.30am - Harricawayanga & Vinpinari Clane 100pm - Wullrankassa Clare

Wednesday 6 March 2024, Wurrumiyonga, Hantiyupwi Histol

10.30am - Monthyuperi Clan

100pm - Jikiimuwu Clan

Thursday 7 March 2024, Wumumiyanga, Hantiyupwi Hotel

10.30am - Wurankuwu Clan

100pm - Malawo Clan

Friday 8 Harch 2024, Pirlangimpi, Sports & Social Club

10.30am - Hunupi Clan



m the torn the intrapretern vitation of m vitation of m internet of an internet of an internet of an internet of an elign, when years in pro-ress in protch in the cou The Aasle Audit Office fait of the ALA fait, was inno supporting WI - an entropoli inhod and dro Warmanacha



ARIA roomer Warmendon Warmendon Warmendon The one-well twee flow from Ner and announced fact that the Lanskie too The come to a Q&A to Warmendon to a Q&A to Warmendon the second tables and these and the second the sec



erking to t 31.5 dites ing: off: its or SMS frau



## NOTICE OF **CONSULTATION WITH** TIWI ISLANDS PEOPLE BAROSSA GAS PROJECT ACTIVITIES

## YOUR VIEWS ARE IMPORTANT TO US.

Santos is preparing Environment Plans (EPs) for the following proposed activities, as required by legislation:

12

Darwis Pl a Duplication (DPD) Environment Plan - the valates to the 23 km portion of the underwater DHD pipeline and supporting subsea inhistructure in Commonwealth waters 27 km south-west of the Tier Islands.

Production Operations Environment Plas - this relates to the arriver and operations of the Risating Production Storage and Officialing (PPSO) facility, operation of a subsce production system and supporting subset infertils chure, and operation of a 285 km section of the Gai Esport Pipeline (SEP) located in Commonwealth waters.

**Operation Environmental Hanagement Han (Product)** Operations) - the OEMP relates to the IOO km portion of the GEP and supporting subses infrestructure located in Northern Territory Writers and land, extending to the existing Darwin Lageted Natural Gas (DLNG) facility

We are currently consulting with Tiwi Islands people whose functions, interests or activities may be affected by activities proposed under the EPs or the OEMP listed above. Based on input from Tiwi Islands people, we will consult with you at the sessions detailed in this notice through Clan group meetings with videos and visual aids available.

#### At the upcoming sessions we will:

- Continue consulting with relevant persons about our proposed. activities under the DPD EP, including,
- providing responses to information provided and quarks sised about the DPD EP by Trui Islands people, following previous Can meetings.
- updating you about any measures we propose to adopt in the DPD EP as a result of your information and comments. before it is submitted to the regulator for assessment.
- If any input is outstanding, sooking your final input on possible consequences of the proposed OPD activity and any further measures you would the us to consider to reduce spects and risks.

#### CONTACT US

Ti 1800 267 600 E offshore.consultation il santos.com For more information please scan GR codes: Please note that this is the final consultation session for the DFD

- 1.08 auit with relevant persons about our proposed activ under the Production Operations EP and OEMP, including:
- providing information and responding to questions about the proposed activities, potential impacts and risks and how we plan to reduce these to as low so reasonably practicable and to an acceptable level
- inviting you to consider the information given and tail us if you seek further or different information.

#### **TIMING AND VENUES**

Tuesday 5 March 2024, Hilliapill, Sports & Recreation Centre

10.30em - Mantkawayonga & Yingmari Clans

1.00pm - Wulirankuwu Clan Wednesday 6 Harch 2024, Wurrumlyinga,

**Mantiyupwi Notei** 10.30am - Hantiyupwi Clan

1.00pm - "Killenewu Class

Thursday 7 March 2024, Wurnamiyanga, Manilysperi Hotel

10.30am - Wurankums Clert

1.00pm - Halawa Can Friday 8 March 2024, Pirtangimpi, Sports & Social Club

10.30am - Munuel Clan

Research Total Designation Research Delevant Devan **D**i 2

Monday Februs

### With an

Ziei Avurill

If a bit offering 1 in a bit offering 1 in periorities to altiple root of NT would activate supp Chief Measure and end Opposition les nocthare have bet ted hore particision differing attackments by their soncement mary amimod dyin The NE and M

unity places in Aust out VAD laws, after old indexed gap



## NOTICE OF CONSULTATION WITH TIWI ISLANDS PEOPLE BAROSSA GAS PROJECT ACTIVITIES

## YOUR VIEWS ARE IMPORTANT TO US.

Santos is preparing Environment Plans (EPs) for the following proposed activities, as required by legislation:

Darwis Repetite Doptication (DRPD) Exvironment Plan - 1% values to the 21 km pontion of the underwater DRD pipoline and supporting subset inhibit uctuates in Commonwealth waters 27 km south-west of the Tee Islands.

Production Operations Environment Plas - this relates to the arrive and operations of the Rosting Production Storage and Officialing (PPSO) facility operation of a subca production system and supporting subces infestivicities, and operation of a JBS im section of the Gas Export Pipeline (GEP) located in Commonwealth weters.

Operation Environmental Hanagement Hen (Production Operations) - the OEMP relates to the IOO knt portion of the GEP and supporting subset infrastructure located in Northern Tarritory Writers and lists, deterding to the existing Derwin Legarbed Natural Gas (DLNG) Sectify

We are currently consulting with Tiwi Islands people whose functions, interests or activities may be affected by activities proposed under the EPs or the OEHP insted above. Based on input from Tiwi Islands people, we will consult with you at the sessions detailed in this notice through Clan group meetings with videos and visual aids available.

#### At the upcoming sessions we will:

 Continue consulting with relevant persons about our proposed activities under the DPD EE including.

- providing responses to information provided and quarter raised about the DPD EP by Trial Islands people, following previous Can meetings.
- updating you about any measures we propose to adopt in the DPO EP as a result of your information and comments before it is submitted to the regulator for assessment.
- If any input is substanding, socking your final input on possible consocuences of the personeet DRD activity and any further measures you would the us to consider to reduce impacts and risks.

Please note that this is the final consultation session for the DPD EX.

- Consult with relevant persons about our proposed activities under the Production Operations EP and OEMR including;
  - providing information and responding to questions about the processed activities, potential impacts and rises and how we plan to notice those to as low as measurably practicable and to an acceptable line(.)
- inviting you to consider the information given and bill us if you seek further or different information.

#### TIMING AND VENUES

Tuesday 5 March 2024, Hilliaphi, Sports & Recreation Centre

10.30em - Marrikawayanga & Yinginari Clans 1.00pm - Walirankawa Clan

Wednesday 6 Harch 2024, Wurrumiyanga, Mantiyupwi Hotel

10.30am - Mahtiyupwi Clan 1.00pm - Jikilaruwu Clan

Thursday 7 Harch 2024, Wurnamiyanga, Hantiyupwi Hotel

10.30am - Wurankums Clan 1.00pm - Holowy Clan

Friday 8 March 2024, Pirtangimpi, Sports & Social Club

10.30am - Munuci Cian



-

St

Monday Februs



Zizi Avurill

If a hill offering T the light is derived in performer to Riper cost of NT would activate suppchief Mainter T wolf Appendixet iss to the particle to the there particle to the there particle to allowing remote more than the support of the remot. The NT and M way places in AusDepartment of INDUSTRY, TOURISM AND TRADE

### EXPRESSIONS OF INTEREST

### CHAIRPERSON FOR McARTHUR RIVER MINE INDEPENDENT EXPERT MINE **CLOSURE PANEL**

The Minister for Mining is seeking expressions of interest from technical experts to be appointed as Chairperson for the independent panel of experts for McArthur River Mine focusing on Mine Closure.

A suitably credentialed individual will be appointed to deliver an independent technical review of action taken to achieve closure of the McArthur River Mine in line with approved plans.

The independent expert will have access to McArthur River Mine operations and performance information focusing on the three-year period following approval of the Overburden Management Project in November 2020.

The Chairperson will play an important role in demonstrating accountability of the mine operator to the wider community.

Prior to applying, please refer to the Terms of Reference and selection criteria at industry.nt.gov.au/mm-panels

Written nominations should be emailed to Andria Handley, Director Mining Operations Policy and Support, andria.handlev@nt.gov.au

Nominations close at midnight on Sunday 21 April 2024.

### Santos COMMUNITY CONSULTATION DROP-IN SESSIONS

BAROSSA GAS PROJECT ACTIVITY

## YOUR VIEWS ARE IMPORTANT TO US.

Santos is preparing environment plans, as required by legislation, regarding activities for the Barossa Gas Project - a project that involves extracting natural gas from the Barossa field in Commonwealth waters approximately 285 kilometres offshare north-narth west of Darwin and transporting it to the existing Darwin liquefied natural gas (DLNG) plant.

We are currently consulting with relevant persons whose functions, interests, or activities may be affected by the Production Operations Activity as we prepare a Production Operations Environment Plan and an Operations Environmental Management Plan. Detailed information about these proposed activities is available at Santos.com/barossa or by scanning the QR code below. Santos will hold community consultation drop-in sessions for relevant persons to obtain information about these proposed activities, provide feedback and ask any questions.

### VENUE: DATES:

Santos Shopfront 26<sup>TH</sup> March, Tuesday T14/15, 41 The Mall, Darwin 27<sup>TH</sup> March, Wednesday 28<sup>TH</sup> March, Thursday

For more information please scan QR codes

CONTACT US



TIME:

10.30am - 2.30pm



## NOTICE OF **CONSULTATION WITH** TIWI ISLANDS PEOPLE BAROSSA GAS PROJECT ACTIVITIES

## YOUR VIEWS ARE IMPORTANT TO US.

Santos is preparing Environment Plans (EPs) for the following proposed activities, as required by legislation:

as Production Operations Environment Plan (EP) - this relates to the arrival and operations of the Floating Production Storage and Offloading (FPSQ) facility, operation of a subset production system and supporting subset nfrastructure, and operation of a 285 km section of the Gas Export Pipeline (SEP) located in Commonwealth waters.

#### Baressa Gas Export Pipeline Operations Environ

Hanagement Plan (DEMP) - the DLMP relates to the IOD importion of the GEP and supporting subsea infrastructure located in Northern Territory waters, extending to the onshore termination point at the Darwin Liquefied Natural Get (DLNG) facility

We are currently consulting with Tiwi Islands people whose functions, interests or activities may be affected by activities proposed under the EP or the OEMP listed above. Based on input from Tiwi Islands people, we will conselt with you at the advertised sessions through Clan group meetings with videos and visual aids available.

#### At the upcoming sessions we will:

- · Consult with relevant persons about our proposed activities under the Production Operations EP and DEMP, including
- providing information and responding to questions about the proposed activities, potential impacts and risks and how we plan to reduce these to as low as reasonably practicable and to an acceptable level.
- insiting you to consider the information provided and tell us if you seek further or different information.

#### CONTACT US

#### T: 1800 267 600 E: offshore consultation il santos com

For more information please scan GR codes



Renting DRD Publics



Research Delegant Desard





## Honday 8 April 2024, Pirlangimpi, Sports & Social Club

10.30am - Munupi Clan Please note that in addition to consultation on the oduction Operations EP and OEMP. Santos will also be consulting with the Munupi Clan on the Darwin

Santos

Pipeline Duplication Project EP including:

- providing responses to information provided and queries raised about the DPD EP by Twi Islands people, following previous Clan meetings.
- updating you about any measures we propose to adopt in the DPD EP as a result of your information and comments before it is submitted to the egulator for assessment.
- If any input is outstanding, seeking your final input on possible consequences of the proposed OPD activity and any further measures you would like us to consider to reduce impacts and risks.

### Tuesday 9 April 2024, Milkapiti, Sports & Recreation Centre

10.30am - Martikawulyahga & Yingshari Clans 1.00pm - Wuliraniwwy Clan

#### Wednesday 10 April 2024, Wurrumiyanga.

**Mantiyupul Motel** 

10.30am - Mantiguerwi Clan 100pm - Jikilaruwu Clan

### Thursday 11 April 2024, Wurrumiyanga,

Mantlyupwi Hotel

10.30am - Wursnieuwu Cian 1.00pm - Maleveu Clan



Tuesday Mars



Alex Trency The Northern To ermannt checks mathine Alec Sp grid attaid an orige the signalar of re arriving benerfield Centre, a new repo The \$2.5m A future Crist proge the NT government of Shee, was law speece to inco-motion of survals entry in-shift has the word lift per / nd piss The project have

GROW TELL ' SERIC

Camden Sm Sierra Haigh

in Springsh the weath Schoviour" invo Tendham Wa Commander Br und Trensfands recorded death but held a but held core entry: A vitral life crypel the descert to sta ching water a vitrag thereas margent y any "force up and on the side."
## NOTICE OF CONSULTATION WITH TIWI ISLANDS PEOPLE BAROSSA GAS PROJECT ACTIVITIES

## YOUR VIEWS ARE IMPORTANT TO US.

### Santos is preparing Environment Plans (EPs) for the following proposed activities, as required by legislation:

Barossa Production Operations Environment Plan (EP) - this relates to the arrival and operations of the Floating Production Storage and Officiating (FPSO) facility, operation of a subsec production system and supporting subsec infrastructure, and operation of a 285 km section of the Gas Export Pipeline (SEP) located in Commonwealth waters.

#### Barossa Gas Export Pipeline Operations Environmental

Management Plan (CEMP) - the CEMP relates to the ICO km portion of the CEP and supporting subsets infristructure located in Northern Territory waters, extending to the onshore termination point at the Darwin Liquefied Natural See (DLNG) facility.

We are currently consulting with The Islands people whose functions, interests or activities may be affected by activities proposed under the EP or the OLMP listed above. Based on input from The Islands people, we will consult with you at the advecticed sessions through Clan group meetings with videos and visual aids available.

#### At the upcoming sessions we will:

- Consult with relevant persons about our proposed activities under the Production Operations EP and OEMP, Including;
  - providing information and responding to questions about the proposed activities, potential impacts and make and how we plan to reduce these to as low as reasonably practicable and to an acceptable level.
  - inviting you to consider the information provided and tell us if you seek further or different information.

### TIMING AND VENUES

Monday & April 2024, Pirlangimpi, Sports & Social Club

#### 10.30am - Munupi Clan

Please note that in addition to consultation on the Production Operations EP and OEMP, Sentos will also be consulting with the Planual Clan on the Derwin Pleasine Destantion Preset EP instantiat.

- providing responses to information provided and quaries roused about the DPD EP by Twi Islands people, following previous Clan meetings.
- updating you about any measures we propose to adopt in the DPD EP as a result of your information and comments before it is submitted to the regulator for assessment.
- If any input is outstanding, seeking your final input on possible conveguences of the proposed DPD activity and any further measures you would like us to consider to reduce impacts and mike.

#### Teesday 9 April 2024, Hilkspill, Sports & Recreation Centre

10.30am - Marrikowuyanga & Yimpinari Clami 1.00pm - Wulinenkuwu Clan

Wednesday 10 April 2024, Warrumiyanga, Mantiyupiwi Motel

10.30am - Mantiyupwi Clan 1.00pm - Jikiaruwu Clan

Thursday T April 2024, Wurramiyanga,

Mantiyupwi Hotel

10.30am - Wurznieuws Clan 1.00pm - Malaws Clan





Tuesday

May

Alice Spring errort says fambrig per of late result "just, a topp memory of to result in result in treach in the





Polico have white have built have yet program over commande stablere ven stablere v



### YOUR VIEWS ARE **IMPORTANT TO US.**

Santos is preparing Environment Plans (EPs) for the following proposed activities, as required by legislation:

## tion Operations Environment Plan (EP)

Barosa Production Operations Environment Plan (EP) - this initiates to the arrival ond operations of the Floating Production Storage and Offloading (FPSO) facility, operation of a subset production system and supporting subset infrastructure, and operation of a 285 km section of the Gas Export Popeline (GEP) located in Commonwealth waters.

#### Barossa Gas Export Pipeline Operations Environmental

Removal Gal Export Popular Operations Environmental Management Plan (OEMP) - the OEMP relates to the IOO km portion of the GEP and supporting subset inhestructure located in Northern Territory waters, extending to the onnhore termination point at the Dervin Liquefied Natural Gas (DLNG) facility.

Cas (CLNG) facility. We are currently consulting with Tiwi Islands people whose functions, interests or activities may be affected by activities proposed under the EP or the OCMP listed above. Based on input fram Tiwi Islands people, we will consult with you at the advertised sessions through Clan group meetings with videos and visual aids available.

### At the upcoming sessions we will:

- Consult with relevant persons about our proposed activities under the Production Operations EP and OEMP; including:
- providing information and responding to auestons about the proposed activities, potential impacts and risks and how we plan to induce these to as low as reasonably practicable and to an acceptable level.
- myting you to consider the information provided and tell us if you seek further or different information.

## T: 1800 267 600

E: offshore.consultation il santas.com For more information please scan QR codes:

### Honday & April 2024, Pirls Sports & Social Club 10.30am - Munupi Clan

TIMING AND VENUES

Please note that in addition to consultation on the Production Operations EP and OEMP, Santos will also be consulting with the Munusi Clan on the Clarwin Pipeline Duplication Project EP including:

- providing responses to information pro-
- providing responses to information possible and quarter stated about the DPD EP by Twe Itiands beoptic, following previous Clain meetings.
   updating you about any measures we propose to adopt in the DPD EP is a result of your information and comments before it is submitted to the regulator for assessment.
   if any input is outstanding, seeking your final input on possible consequences of the proposed DPD activity and any further measure you would like us for needuces to existent topost. to consider to reduce impacts and risks.
- Tuesday 9 April 2024, Milikapiti, Sports & Recreation Centre

10.30am - Marrikawuyunga & Vinpinari Clans 1.00gen - Wulirenkumu Cten

Wednesday 10 April 2024, Wurnumlyanga, Hantilyupai Notel

10.30am - Mantiyupwi Clan 1.00pm - Jikianuwu Clan

Thursday 11 April 2024, Wurrumiyonga, Hantiyopwi Hotel

10.30am - Wurznikowu Clan 1.00pm - Halawu Clan



Lav Fia Wal Chief Min sorten is usch paro recessioni disse and disse and Mis Las pert musi company stagram, saloringis moreduai lawi dha he anyo "The in an youth 48 BA SP ZIEA Isol Women's experience ansection with found thowen the experience solution as contracts a solution as perceived tomaty the activity on with found

Served



### YOUR VIEWS ARE IMPORTANT TO US.

Santos is preparing environment plans for the following proposed activities, as required by legislation:

Barossa Production Operations Environment Plan (EP) - this relates to the arrival and operations of the Floating Production Storage and Offloating (FPSO) fiscitly, operation of a subsets production system are supporting subset infrastructure, and operation of a 305 km section of the Gas Export Ppeline (GEP) located in Commonwealth waters.

Parossa Gas Export Pipeline Operations Environmental Hanagement Pian (OEMP) - the OEMP relates to the 100 km portion of the GEP and supporting subces infortituates located in Netterm Territory waters, extending to the ensitient termination point at the existing Derwin Louented Natural Gas (CLIAS) Fortility.

Barossa Darwin Pipeline Duplication (DPO) Emfronment Plan (EP) – this initiate to the 23 km partian of the underwater DPD pipeline and supporting systexis infrastructure in Commonwealth waters 27 km south-wast of the Tiwe Islands.

We are currently consulting with Tiwi Islands people whose functions, interests or activities may be affected by activities proposed under the EPs or the QEMP listed above.

### At all of the upcoming sessions we will:

- An an unit the upconting sessions we will: Continue consulting with relevant period about our proposed activities under the Poulocition Operations P and OEMP, including: providing responses to information provided and queries inside about the EP and OEMP by The island's people, following previous Can meetings. updating you about any measures we propose to adopt in the EP and OEMP as a result of your information and comments before it is submitted to the represent Production Operations activity and only further measures you would like us to consider to induce impacts and risks.

- and mixe. For the Munuk Clain on 17 May 2004: we will also continue consulting with relevant persons about our proposed activities under the DPO ER including: providing responses to information provided and queries reside about the DPD EP has performed in the properties of the term of te

T: 1800 267 600 E: offshore.consultation issantos.com For more information please scan QR codes:





If any input is outstanding, seeking your final input on possible consequences of the proposed DPD activity and any further measures you would like un to consider to reduce impact and risks.
 Based on input from Timi Islands people, we will consult with you at the advertised sessions through. Clan group meetings with videos and visual aids wallable.

#### TIMING AND VENUES

nday 13 May 2024 - Militapili, Sports & Recruition Centre

10.30am - Marikawuyanga & Yimpinari Clans 1.00pm - Wulitanikuwu Clan

Wednesday 15 May 2024 - Wumumiyanga, Manfusowi Motel

10.30am - Mantiyupwi Clan 100pm - Jikiknuwi Clan Pease note that these are the final consultation sessions for the Production Operations EP and OEMP

Thursday 16 Hay 2024 - Wumamiyanga, Mantiyuperi Notel

10.30am - Wurankuwu Clan 1.00pm - Malawu Clan

Friday 17 May 2024 - Finlangmpt Sports & Social Club 10.80am - Munupi Clan (This session has been reacheduled to May 17 at the request of the community. Note that this is the final Munupi Clan consultation session for the DPD EP.)

esday 21 May 2024 - Pirlangimal, Sports & Social Clut

10.30am - Munupi Clan

Wednesday 22 May 2024 - Wur

1.30am - Wurahkuwu Clan 1.00pm - Malseu Clan These essions have been rescheduled to May 21 and 22 at the request of the communities. Please note that these are the final consultation sessions for the Production Operations EP and OEMF



### YOUR VIEWS ARE IMPORTANT TO US.

Santos is preparing environment plans for the following proposed activities, as required by legislation:

Barossa Production Operations Environment Plan (EP) - this relates to the arrival and operations of the Floating Production Storage and Offloating (FPSO) fiscitly, operation of a subsets production system are supporting subset infrastructure, and operation of a 305 km section of the Gas Export Ppeline (GEP) located in Commonwealth waters.

Parossa Gas Export Pipeline Operations Environmental Hanagement Pian (OEMP) - the OEMP relates to the 100 km portion of the GEP and supporting subces infortituates located in Netterm Territory waters, extending to the ensitient termination point at the existing Derwin Louented Natural Gas (CLIAS) Fortility.

Barossa Darwin Pipeline Duplication (DPO) Emfronment Plan (EP) – this initiate to the 23 km partian of the underwater DPD pipeline and supporting systexis infrastructure in Commonwealth waters 27 km south-wast of the Tiwe Islands.

We are currently consulting with Tiwi Islands people whose functions, interests or activities may be affected by activities proposed under the EPs or the QEMP listed above.

### At all of the upcoming sessions we will:

- An an unit the upconting sessions we will: Continue consulting with relevant period about our proposed activities under the Poulocition Operations P and OEMP, including: providing responses to information provided and queries inside about the EP and OEMP by The island's people, following previous Can meetings. updating you about any measures we propose to adopt in the EP and OEMP as a result of your information and comments before it is submitted to the represent Production Operations activity and only further measures you would like us to consider to induce impacts and risks.

- and mixe. For the Munuk Clain on 17 May 2004: we will also continue consulting with relevant persons about our proposed activities under the DPO ER including: providing responses to information provided and queries reside about the DPD EP has performed in the properties of the term of te

T: 1800 267 600 E: offshore.consultation issantos.com For more information please scan QR codes:





If any input is outstanding, seeking your final input on possible consequences of the proposed DPD activity and any further measures you would like un to consider to reduce impact and risks.
 Based on input from Timi Islands people, we will consult with you at the advertised sessions through. Clan group meetings with videos and visual aids wallable.

#### TIMING AND VENUES

nday 13 May 2024 - Milkapill, Sports & Recreation Centre

10.30am - Marikawuyanga & Yimpinari Clans 1.00pm - Wulitanikuwu Clan

Wednesday 15 May 2024 - Wumumiyanga, Manfusowi Motel

10.30am - Mantigupwi Clan 100pm - Jikikruwi Clan Pease note that these are the final consultation sessions for the Production Operations EP and OEMP

Thursday 16 Hay 2024 - Wumamiyanga, Mantiyuperi Notel

10.30am - Wurankuwu Clan 1.00pm - Malawu Clan

Friday 17 May 2024 - Finlangmpt Sports & Social Club 10.80am - Munupi Clan (This session has been reacheduled to May 17 at the request of the community. Note that this is the final Munupi Clan consultation session for the DPD EP.)

esday 21 May 2024 - Pirlangimal, Sports & Social Clut

10.30am - Munupi Clan Wednesday 22 May 2024 - Wu

1.30am - Wurahkuwu Clan 1.00pm - Malseu Clan These essions have been rescheduled to May 21 and 22 at the request of the communities. Please note that these are the final consultation sessions for the Production Operations EP and OEMF



## **NOTICE OF CONSULTATION WITH** TIWI ISLANDS PEOPLE BAROSSA GAS PROJECT ACTIVITIES

### YOUR VIEWS ARE IMPORTANT TO US.

#### Santos is preparing environment plans for the following proposed activities, as required by legislation:

Barossa Production Operations Environment Plan (EP) – this relates to the anival and operations of the Flooting Production Storage and Offlooding (FPSG) facility, operation of a subsee production system and supporting subsea infrastructure, and operation of a 285 km section of the Gas Export Pipeline (GEP) located in Commonwealth waters.

Barossa Gas Export Pipeline Operations Environmental Management Plan (OEMP) - the OEMP relates to the IOD km portion of the GEP and supporting subset infrastructure located in Northern Territory waters, extending to the orshore termination point at the existing Darwin Liquefied Natural Gas (DLNG) facility.

Barossa Darwin Ploeline Duplication (DPD) Environment Plan (EP) – this instates to the 23 km particle of the underwater OPD pipeline and supporting subces infrastructure in Commonwealth waters 27 km south-west of the Twi Islands.

We are currently consulting with Tiwi Islands people whose function interests or activities may be affected by activities proposed under the EPs or the OEMP listed above.

### At all of the upcoming sessions we will:

- Continue consulting with relevant persons about our proposed activities under the Production Operations EP and CEMP, including: providing responses to Information provided and gavies asised about the EP and OEMP by Thei Islands people, following previous Clan
- meetings. · updating you about any measures we propose to adopt in the EP and OEMP as a result of your information and comments before it is submitted to the regulator for assessment.
- If any input is outstanding, seeking your final input on possible consequences of the proposed Production Operations activity and any further measures you would like us to consider to reduce impac-iend risks. ipacts
- For the Munupi Clan on 17 May 2024; we will also continue consulting with relevant persons about our proposed activities under the DPD EP including:
- retward periods about our proposed address under the DPD EP including; providing responses to information provided and quarties reside about the DPD EP by Timi Islands people, following previous Clan meetings: updating you about any measures we propose to adopt in the DPD EP as result of your information and comments before it is submitted to the regulator for assessment.

CONTACT US T: 1800 267 600 E: offshore.consultation @santos.com For more information please scan GR codes:





 If any input is outstanding, seeking your final input on possible consequences of the proposed DPD activity and any further measures you would like us to consider to reduce impacts and risks. Based on input from Tiwi Islands people, we will consult with you at the advertised sessions through Clan group meetings with videos and visual aids available.

Monday 13 May 2024 - Milliageti, Sports & Recreation Centre

Thursday 16 Hay 2024 - Womuniyanga, Mantiyupwi Hotal

Friday 17 May 2024 - Pirlangimpt, Sports & Social Club

Tuesday 21 May 2024 - Pelangmpi, Sports & Social Club

day 22 May 2024 + Wumum

10.30am - Munupi Clan (This session has been rescheduled to May 17 at the request of the community, Note that this is the final Munupi Clan consultation session for the DPD EP)

10.30am - Wurankuwu Clan 1.00pm - Malawu Clan These sessions have been rescheduled to May 21 and 22 at the researts of the communities. Please note that these are the final consultation sessions for the Production Operations EP and OEMP

imiyanga, Mantiyupiwi Motel

nga, Manto

TIMING AND VENUES

10.30am - Marrikawuyanga & Yimpinari Clans 1.00pm - Wulitankuwa Clan Wednesday 15 May 2024 - Wurn

10.80am - Wurankuwu Clan 1.00pm - Malawu Clan

10.30am - Munupi Clen



## Monday May 20, 2024 | N1 Watc Questions over ship lift project Camden Smith

Less than a fortnight after the NT government announced US company Pearloon would deliver key infrastructure on the Darwin ship. Bf develop-ment, a leading probity watch-dog has delivered a stinging report on the project's govern-ance.

they not interview inducts, and the project system reaction of the project system. The provide system and the project system is a system of the system is a system of the system is not system of the system of the system parcels from Papader, saw the government transfer IS.ha. of heat and seas to the company and lease in further land parcel at somial value in order to construct a provide manne follow to the com-pany.

cuty to be owned by the com-pany. The report, tabled in parlia-ment last week, identified that evidence to support key deci-sions made by the Department of Chief Minister and Cabinet



## Table 4-10 (of the EP) Advertising Larrakia

### sessions

23 March 2024 Press ad – NT News

Violence spokesman. Picture: Pema Tamang Pakhrin

tration in May last year.

the role of shadow minister for of offenders, representing one find there's more to the story."

Department of IERRITORY INDUSTRY, TOURISM AND TRADE

## **EXPRESSIONS OF INTEREST**

### CHAIRPERSON FOR McARTHUR RIVER MINE INDEPENDENT EXPERT MINE **CLOSURE PANEL**

The Minister for Mining is seeking expressions of interest from technical experts to be appointed as Chairperson for the independent panel of experts for McArthur River Mine focusing on Mine Closure.

A suitably credentialed individual will be appointed to deliver an independent technical review of action taken to achieve closure of the McArthur River Mine in line with approved plans.

The independent expert will have access to McArthur River Mine operations and performance information focusing on the three-year period following approval of the Overburden Management Project in November 2020.

The Chairperson will play an important role in demonstrating accountability of the mine operator to the wider community.

Prior to applying, please refer to the Terms of Reference and selection criteria at industry.nt.gov.au/mrm-panels

Written nominations should be emailed to Andria Handley. Director Mining Operations Policy and Support, andria.handlev@nt.gov.au

Nominations close at midnight on Sunday 21 April 2024.

Santos COMMUNITY CONSULTATION

**DROP-IN SESSIONS** BAROSSA GAS PROJECT ACTIVI

## YOUR VIEWS ARE IMPORTANT TO US.

Santos is preparing environment plans, as required by legislation, regarding activities for the Barossa Gas Project - a project that involves extracting natural gas from the Barossa field in Commonwealth waters approximately 285 kilometres offshore north-north west of Darwin and transporting it to the existing Darwin liquefied natural gas (DLNG) plant.

We are currently consulting with relevant persons whose functions, interests, or verare currently consulting with relevant persons whose functions, interests, or activities may be affected by the Production Operations Activity as we prepare Production Operations Environment Plan and an Operations Environmental Management Plan. Detailed information about these proposed activities is available at Santos.com/barossa or by scanning the QR code below. Santos will

hold community consultation drop-in sessions for relevant persons to obtain information about these proposed activities, provide feedback and ask any questions. DATES:

### VENUE:

CONTACT US

For more information please scan QR code:

Santos Shopfront 26<sup>114</sup> March, Tuesday T14/15, 41 The Mall, Darwin 27<sup>114</sup> March, Tuesday 28<sup>114</sup> March, Thursday



TIME:

10.30am - 2.30pm

April Notice of Consultation

- Emails to representative organisations for sharing across their direct networks.
- Emails and phone calls notifying individual Larrakia family representatives
- Promotion via Santos' Darwin shop front Targeted for Larrakia people

## NOTICE OF CONSULTATION WITH LARRAKIA PEOPLE

BAROSSA GAS PROJECT ACTIVITIES

### YOUR VIEWS ARE IMPORTANT TO US.

Santos is preparing environment plans for the following proposed activities, as required by legislation:

Berossa Production Operations Environment Plan (EP) - this relates to the arrival and operations of the Floating Production Storage and Offloading (FPSO) facility, operation of a subsea production system and supporting subsea infrastructure, and operation of a 285 km section of the Gas Export Pipeline (GEP) located in Commonwealth waters.

#### Barossa Gas Export Pipeline Operations Environmental Management Plan (OEMP) - the OEMP relates to the 100 km

Management Plan (OEMP) - the OEMP relates to the 100 km portion of the GEP and supporting subsea infrastructure located in Northern Territory waters, extending to the onshore termination point at the Darwin Liquefled Natural Gas (DLNG) facility.

We are currently consulting with Larrakia people whose functions, interests or activities may be affected by activities proposed under the EP or the OEMP listed above. Videos and visual alds will be available at the advertised sessions.

#### At the upcoming sessions we will:

 Consult with relevant persons about our proposed activities under the Production Operations EP and OEMP, including;
 – providing information and responding to questions about the proposed activities, potential impacts and risks and how we plan to reduce these to as low as reasonably practicable and to an acceptable level.
 – inviting you to consider the information provided and tell us if

 Inviting you to consider the information provided and tell us if you seek further or different information.

### TIMING AND VENUE

Tuesday 23 April Session 1: 9.00am - 11.00am Session 2: 5.30pm - 7.30pm

Hilton Hotel Darwin, 32 Mitchell Street, Darwin

#### CONTACT US

•

T: 1800 267 600 E: offshore.consultation@santos.com For more information please scan QR codes:





Santos

STATISTICS.



## YOUR VIEWS ARE IMPORTANT TO US.

Santos is preparing Environment Plans for the following proposed activities, as required by legislation: Barossa Production Operations Environment Plan (EP) Barossa Gas Export Pipeline Operations Environmental Management Plan (OEMP)

### TIMING AND VENUE Tuesday 23 April

Session 1: 9.00am - 11.00am Session 2: 5.30pm - 7.30pm

Hilton Hotel Darwin, 32 Mitchell Street, Darwin

# Santos

### CONTACT US

T: 1800 267 600 E: offshore.consultation@santos.com For more information please scan QR codes:







### June Notice of Consultation

- Emailed to representative organisations for sharing across their direct . networks.
- Promotion via Santos' Darwin shop front. •
- Emails and phone calls notifying individual Larrakia family representatives •

## Santos



### YOUR VIEWS ARE IMPORTANT TO US.

Santos is preparing environment plans for the following proposed activities, as required by legislation:

Barossa Production Operations Environment Plan (EP) - this relates to the arrival and operations of the Floating Production Storage and Offloading (FPSO) facility, operation of a subsea production system and supporting subsea infrastructure, and operation of a 285 km section of the Gas Export Pipeline (GEP) located in Commonwealth waters.

Barossa Gas Export Pipeline Operations Environmental Management Plan (OEMP) - the OEMP relates to the 100 km portion of the GEP and supporting subsea infrastructure located in Northern Territory waters, extending to the onshore termination point at the Darwin Liquefied Natural Gas (DLNG) facility.

Barossa Darwin Pipeline Duplication Project Environ Plan (EP) - this relates to the installation and precommissioning of the approximately 23 km long section of the Darwin Pipeline Duplication (DPD) and supporting subsea infrastructure, located in Commonwealth waters - 95km north-west of Darwin

Barossa Darwin Pipeline Duplication Project Offshore Construction Environmental Management Plan (EMP) this relates to the installation of approximately 8.26 km of the DPD pipeline in an area of Northern Territory (NT) coastal waters between the Commonwealth/NT coastal waters boundary and the Territorial Sea Baseline - 80km north-west of Darwin.

We are currently consulting with Larrakia people whose functions, interests or activities may be affected by activities proposed under the EPs, OEMP, and EMP listed above. Videos and visual aids will be available at the advertised session

#### At the upcoming sessions we will:

Continue consultation with Larrakia people about our proposed activities under the EPs, OEMP and EMP including: · providing responses to information requests and queries previously raised about proposed activities to be managed

- under the EPs, OEMP and EMP. updating you about any measures we propose to adopt in the EPs, OEMP, and EMP as a result of your information and
- comments before they are submitted to the regulator for assessment.
- if any input is outstanding, seeking your final input on possible consequences of the proposed activities under the EPs, OEMP, and EMP, and any further measures you would like us to consider to reduce impacts and risks.

TIMING AND VENUES

Wednesday 12 June Malak Community Centre 13 Malak Crescent, Malak

Session 1 - 9.30am - 11.30am Session 2 - 5.00pm - 7.00pm

### CONTACT US

T: 1800 267 600 offel iltation@santos.com ation please scan QR codes:









## NOTICE OF CONSULTATION WITH LARRAKIA PEOPLE

BAROSSA GAS PROJECT ACTIVITIES

### YOUR VIEWS ARE IMPORTANT TO US.

Santos is preparing environment plans for the following proposed activities, as required by legislation:

Barossa Production Operations Environment Plan (EP) - this relates to the arriva and operations of the Floating Production Storange and Officiality (PPSD) facility, operation of a subsea production system and supporting subsea infrastructure, and operation of a 285 km section of the Gas Export Pipeline (GEP) located in Commonwealth waters.

Barossa Gas Export Pipeline Operations Environmental Management Plan (OEMP) – the OEMP rotors to the 100 km portion of the GEP and supporting subses infrastructure located in Northern Territory waters, extending to the orshore termination point at the Darwin Liquefied Natural Gas OLNG facility. Baross Darwin Pipeline Duplication Project Environment Plan (EP) - this valates to the installation and precommissioning of the approximately 23 km long section at the Darwin Pipeline Duplication (DPD) and supporting subsea infrastructure. located in Commonwealth waters - 95km north-west of Darwin.

Barossa Darwin Pipeline Duplication Project Offshore Construction Environmental Management Pilan (EMP) – this relates to the installation of approximately 8.26 km of the DPD pipeline in an area of Northern Territory (NT) coastal waters between the Commonwealth/NT coastal waters boundary and the Territorial Sea Baseline - 80km north-wast of Darwin

We are currently consulting with Larrakia people whose functions, interests or activities may be affected by activities proposed under the EPs, OEMP, and EMP listed above. Videos and visual aids will be available at the advertised session.

#### At the upcoming sessions we will:

Continue consultation with Larrakia people about our proposed activities under the EPs, OEMP and EMP including

- providing responses to information requests and queries previously raised about proposed activities to be managed under the EPIs, OEMP and EMP.
- updating you about any measures we propose to adopt in the EPs. OEMP, and EMP as a result of your information and comments before they are submitted to the regulator for assessment.
- If any input is outstanding, seeking your final input on possible consecuences of the proposed activities under the EPs, OEMP, and EMP and any further measures you would like us to consider to reduce impacts and risks.

TIMING AND VENUES

Wednesday 12 June Malak Community Centre 13 Malak Crescent, Malak

Session 1 - 9.30am - 11.30am Session 2 - 5.00pm - 7.00pm









Saturday June 1, 2034 NY Nates

"There's so much to be annoyed about. I have a whole list of pet hates"

> have a new pet hate. Ob. I know, judging by your letters, you read this colorses and you thick what a resonable and level-broaded warmanTum. And unnetieses from

"Interface," and people two are present frame. Tames, and a very expedite of being autoreus, parts at me heatural, children, freeds mit to workens, but mody say bedend. Tam, for example, a faidy require sender of emails that say, "fail Hose porty-wolf Jost workering if you got my fait email' you show't replace, so fan just making saw it have't gooe mining?"



## NOTICE OF **CONSULTATION WITH** LARRAKIA PEOPLE

BAROSSA GAS PROJECT ACTIVITIES

### YOUR VIEWS ARE IMPORTANT TO US.

Santos is preparing environment plans for the following proposed activities, as required by legislation:

Barossa Production Operations Environ int Plan (EP) - this relates to the arrival and operations of the Ploating Production Storage and Offloading (FPSO) facility, operation of a subsea production system and supporting subsea infrastructure, and operation of a 285 km section of the Gas Export Pipeline (GEP) located in Commonwealth waters.

Barossa Gas Export Pipeline Operations Environmental Management Plan (OEMP) - the OEMP relates to the 100 km portion of the GEP and supporting subsea infrastructure located in Northern Territory waters. extending to the onshore termination point at the Darwin Liquefied Natural Gas (DLNG) facility

Barossa Darwin Pipeline Duplication Project Environ Plan (EP) - this relates to the installation and procommissioning of the approximately 23 km long section of the Barwin Pipeline Duplication (DPD) and supporting subsiss infractructure, located in Commonwealth waters - 95km north-west of Danwin

Barossa Darwin Pipeline Duplication Project Offshore Construction Environmental Management Plan (EMP) -this relates to the installation of approximately 8.26 km of the DPD pipeline in an area of Northern Territory (NT) coastal waters between the Commonwealth/NT coastal waters boundary and the Temborial Sea Baseline - BOkm north-west of Darwin.

We are currently consulting with Larrakia people whose functions, interests or activities may be affected by activities proposed under the EPs, OEMP, and ENP listed above. Videos and visual aids will be available at the advartised sassion.

#### At the upcoming sessions we will:

Continue consultation with Larrakia people about our proposed activities under the EPs, OEMP and EMP including · providing responses to information requests and queries previously raised about proposed activities to be managed

- under the EPs. CEMP and EMP.
- updating you about any measures we propose to adopt in the EPs. OEMP, and EMP as a result of your information and omments before they are submitted to the regulator for assessment.
- If any input is outstanding, seeking your final input on possible consequences of the proposed activities under the EPs, DEMP, and EMP and any further measures you would like us to consider to reduce impacts and roles.

Wednesday 12 June Session 1 - 9.30 nm - 11.30 nm TIMING AND VENUES Malek Community Centre 13 Malek Crescent, Malek Session 2 - 5:00pm - 7:00pm 1800 367 800 题 E off-hore consultation i santas con For more information please scan GR code 自注



Wadnesday June 5, 24



#### George Vankovich

Why would any one come (harwood? Arrest that gardies in ten anter the police locaring afferdability, accor-ing to resteriate cart acts so

biding about the output of the set from the second of the second of the second of the second of the tensor that the second of the second of the second of the second of the tensor best the second of the second of the second of the second of the tensor best the second of the second of the second of the second of the tensor best the second of the tensor tensor the second of the tensor tensor tensor tensor tensor of the second of the tensor tensor tensor tensor tensor of the tensor tensor of the tensor tensor tensor tensor tensor of the tensor tensor tensor of the tensor tensor tensor of the tensor tensor tensor of tensor tensor tensor tensor of tensor tens ing derived by the same record in factoric organizing builds to, skill obserbigger and to





## NOTICE OF CONSULTATION WITH LARRAKIA PEOPLE

BAROSSA GAS PROJECT ACTIVITIES

### YOUR VIEWS ARE IMPORTANT TO US.

Santos is preparing environment plans for the following proposed activities, as required by legislation:

Barossa Production Operations Environment Plan (EP) - this relates to the arrival and operations of the Floating Production Storage and Offloading (FPSQ) facility, operation of a subsea production system and supporting subsea infrastructure, and operation of a 285 km section of the Gas Export Pipeline (GEP) located in Commonwealth waters.

Bacossa Gas Export Pipeline Operations Environmental Management Plan (OEMP) - the OEMP relates to the 100 km portion of the GEP and supporting subsea infrastructure located in Northern Territory waters, extending to the onshore termination point at the Darwin Liquefied Natural Gas (DLNG) facility. Barossa Darwin Pipeline Duplication Project Environmen Plan (EP) - this relates to the installation and precommissioning of the approximately 23 km long section of the Darwin Pipeline Duplication (DPD) and supporting subsee infrastructure, located in Commonwealth waters - 95km north-west of Darwin.

Barossa Darwin Pipeline Duplication Project Offwhore Construction Environmental Management Plan (EMP) – this relates to the instaliation of approximately 8.26 km of the DPD pipeline in an area of Northern Territory (NT) coastal waters between the Commonwealth/NT coastal waters boundary and the Territorial Sea Baseline – 80km north-west of Darwin.

We are currently consulting with Larrakia people whose functions, interests or activities may be affected by activities proposed under the EPs, OEMP, and EMP listed above. Videos and visual aids will be available at the advertised session.

#### At the upcoming sessions we will:

- Continue consultation with Larrakia people about our proposed activities under the EPs, OEMP and EMP including:
- providing responses to information requests and queries previously raised about proposed activities to be managed under the EPs, DEMP and EMP
- updating you about any measures we propose to adopt in the EPs, OEMP, and EMP as a result of your information and comments before they are submitted to the regulator for assessment.
- If any input is outstanding, seeking your final input on possible consequences of the proposed activities under the EPs. OEMP, and EMP and any further measures you would like us to consider to reduce impacts and risks.

TIMING AND VENUES

Wednesday 12 June Malak Community Centre 13 Malak Crescent, Malak

Session 1 - 9.30am - 11.30am Session 2 - 5.00pm - 7.00pm



T: 1800 257 600 E: offshore consultation 2 senios.com For more information please scan GR cod







Santos is consulting with Larrakia People on Barossa Productions Operations and Darwin Pipeline Duplication (located ~80km north-west of Darwin) environment plans on Wednesday June 12 at the Malak Community Centre, 13 Malak Crescent, Malak



Santos is consulting with Larrakia People on Barossa Productions Operations and Darwin Pipeline Duplication (located ~80km north-west of Darwin) environment plans on Wednesday June 12 at the Malak Community Centre, 13 Malak Crescent, Malak



# Appendix E Santos' Environment Consequence Descriptors

Excerpt from Santos Environmental Hazard Identification and Assessment Guideline, Revision 5 (issued October 2020).

Consequence level		I	II	ш	IV	V	VI
	Acceptability	Acceptable	Acceptable	Unacceptable	Unacceptable	Unacceptable	Unacceptable
Severity description		Negligible No impact or negligible impact	Minor Detectable but insignificant change to local population, industry or ecosystem factors. Localised effect	Moderate Significant impact to local population, industry or ecosystem factors	Major Major long-term effect on local population, industry or ecosystem factors	Severe Complete loss of local population, industry or ecosystem factors AND/OR extensive regional impacts with slow recovery	Critical Irreversible impact to regional population, industry or ecosystem factors
	Fauna In particular, EPBC Act listed threatened/migratory fauna or WA <i>Biodiversity Conservation Act 2016</i> specially protected fauna	Short-term behavioural impacts only to small proportion of local population and not during critical lifecycle activity. No decrease in local population size. No reduction in area of occupancy of species. No loss/disruption of habitat critical to survival of a species. No disruption to the breeding cycle of any individual. No introduction of disease likely to cause a detectable population decline.	Detectable but insignificant decrease in local population size. Insignificant reduction in area of occupancy of species. Insignificant loss/disruption of habitat critical to survival of a species. Insignificant disruption to the breeding cycle of local population.	Significant decrease in local population size but no threat to overall population viability. Significant behavioural disruption to local population. Significant disruption to the breeding cycle of a local population. Significant reduction in area of occupancy of species. Significant loss of habitat critical to survival of a species. Modify, destroy, remove, isolate or decrease availability of quality of habitat to the extent that a significant decline in local population is likely. Introduce disease likely to cause a significant population decline.	Long-term decrease in local population size and threat to local population viability. Major disruption to the breeding cycle of local population. Major reduction in area of occupancy of species. Fragmentation of existing population. Major loss of habitat critical to survival of a species. Modify, destroy, remove, isolate or decrease availability of quality of habitat to the extent that a long-term decline in local population is likely. Introduce disease likely to cause a long- term population decline.	Complete loss of local population. Complete loss of habitat critical to survival of local population. Widespread (regional) decline in population size or habitat critical to regional population.	Complete loss of regional population. Complete loss of habitat critical to survival of regional population.
	Physical Environment/Habitat Includes: air quality; water quality; benthic habitat (biotic/abiotic), particularly habitats that are rare or unique; habitat that represents a Key Ecological Feature <sup>55</sup> ; habitat within a protected area; habitats that include benthic primary producers <sup>56</sup> and/or epi-fauna <sup>57</sup>	No or negligible reduction in physical environment/habitat area/function.	Detectable but localised and insignificant loss of area/function of physical environment/habitat. Rapid recovery evident within approximately two years (two season recovery).	Significant loss of area and/or function of local physical environment/habitat. Recovery over medium term (2–10 years).	Major, large-scale loss of area and/or function of physical environment/local habitat. Slow recovery over decades.	Extensive destruction of local physical environment/habitat with no recovery. Long-term (decades) and widespread loss of area or function of primary producers on a regional scale.	Complete destruction of regional physical environment/habitat with no recovery. Complete loss of area or function of primary producers on a regional scale.
Environmental Receptors	Threatened ecological communities (EPBC Act listed ecological communities)	No decline in threatened ecological community population size, diversity or function. No reduction in area of threatened ecological community. No introduction of disease likely to cause decline in threatened ecological community population size, diversity or function.	Detectable but insignificant decline in threatened ecological community population size, diversity or function; Insignificant reduction in area of threatened ecological community.	Significant decline in threatened ecological community population size, diversity or function. Significant reduction in area of threatened ecological community. Introduction of disease likely to cause significant decline in threatened ecological community population size, diversity or function.	<ul> <li>Major, long-term decline in threatened ecological community population size, diversity or function.</li> <li>Major reduction in area of threatened ecological community.</li> <li>Fragmentation of threatened ecological community.</li> <li>Introduce disease likely to cause long-term decline in threatened ecological community population size, diversity or function.</li> </ul>	Extensive, long-term decline in threatened ecological community population size, diversity or function. Complete loss of threatened ecological community.	Complete loss of threatened ecological community with no recovery.

## **Santos**

 $<sup>^{\</sup>rm 55}$  As defined by the Department of Agriculture, Water and Environment

<sup>&</sup>lt;sup>56</sup> Benthic photosynthetic organisms such as seagrass, algae, hard corals and mangroves

<sup>&</sup>lt;sup>57</sup> Fauna attached to the substrate including sponges, soft corals and crinoids.

Consequence level I II		II	ш	IV	V	VI	
Acceptability		Acceptable	Acceptable	Unacceptable	Unacceptable	Unacceptable	Unacceptable
	Severity description	Negligible No impact or negligible impact	Minor Detectable but insignificant change to local population, industry or ecosystem factors. Localised effect	Moderate Significant impact to local population, industry or ecosystem factors	Major Major long-term effect on local population, industry or ecosystem factors	Severe Complete loss of local population, industry or ecosystem factors AND/OR extensive regional impacts with slow recovery	Critical Irreversible impact to regional population, industry or ecosystem factors
	Protected Areas Includes: World Heritage Properties; Ramsar wetlands; Commonwealth/National Heritage Areas; Land/Marine Conservation Reserves.	No or negligible impact on protected area values. No decline in species population within protected area. No or negligible alteration, modification, obscuring or diminishing of protected area values.*	Detectable but insignificant impact on one of more of protected area's values. Detectable but insignificant decline in species population within protected area. Detectable but insignificant alteration, modification, obscuring or diminishing of protected area values.*	Significant impact on one of more of protected area's values. Significant decrease in population within protected area. Significant alteration, modification, obscuring or diminishing of protected area values.	Major long-term effect on one of more of protected area's values; Long-term decrease in species population contained within protected area and threat to that population's viability. Major alteration, modification, obscuring or diminishing of protected area values.	Extensive loss of one or more of protected area's values. Extensive loss of species population contained within protected area.	Complete loss of one or more of protected area's values with no recovery. Complete loss of species population contained within protected area with no recovery.
	Socio-economic receptors Includes: fisheries (commercial and recreational); tourism; oil and gas; defence; commercial shipping.	No or negligible loss of value of the local industry. No or negligible reduction in key natural features or populations supporting the activity.	Detectable but insignificant short- term loss of value of the local industry. Detectable but insignificant reduction in key natural features or population supporting the local activity.	Significant loss of value of the local industry. Significant medium-term reduction of key natural features or populations supporting the local activity.	Major long-term loss of value of the local industry and threat to viability. Major reduction of key natural features or populations supporting the local activity.	Shutdown of local industry or widespread major damage to regional industry. Extensive loss of key natural features or populations supporting the local industry.	Permanent shutdown of local or regional industry. Permanent loss of key natural features or populations supporting the local or regional industry.



## Appendix F

Spill Modelling Results (maximum values across all seasons and water depths)

## APPENDIX F - SPILL MODELLING RESULTS SUMMARY (MAXIMUM VALUES ACROSS ALL SEASONS AND WATER DEPTHS)

300 m<sup>3</sup> surface release of MDO from a vessel over 1 hour, as a result of external impact: Surface, dissolved and entrained impact oil spill modelling results (maximum values across all seasons and water depths)

		Probab	ility of exposure (p	percent)		Minimum time before exposure on the sea surface (hours)						
	Moderate exposure values			High exposure values		Moderate exposure values			High exposure values		Maximum	Maximum
Environmental Value Area	Surface hydrocarbons (≥10 g/m²)	Dissolved hydrocarbons (≥50 ppb)	Entrained hydrocarbons (≥100 ppb)	Surface hydrocarbons (≥50 g/m²)	Dissolved hydrocarbons (≥400 ppb)	Surface hydrocarbons (≥10 g/m²)	Dissolved hydrocarbons (≥50 ppb)	Entrained hydrocarbons (≥100 ppb)	Surface hydrocarbons (≥50 g/m²)	Dissolved hydrocarbons (≥400 ppb)	dissolved hydrocarbon exposure (ppb)	entrained hydrocarbon exposure (ppb)
Afghan Shoal	NC	-	5.33	NC	-	NC	-	48	NC	-	-	14
Beagle Gulf- Darwin Coast	-	-	NC	-	-	-	-	NC	-	-	-	3
Cobourg Peninsula- Nhulunbuy	-	-	NC	-	-	-	-	NC	-	-	-	<1
Djukbinj NP	-	-	NC	-	-	-	-	NC	-	-	-	2
Flat Top Bank	-	-	NC	-	-	-	-	NC	-	-	-	3
Hancox Shoal	-	-	NC	-	-	-	-	NC	-	-	-	3
Harris Reef	-	-	NC	-	-	-	-	NC	-	-	-	3
JBG East Coast	-	-	NC	-	-	-	-	NC	-	-	-	<1
Joseph Bonaparte Gulf AMP	-	-	NC	-	-	-	-	NC	-	-	-	<1
Lowry Shoal	-	-	NC	-	-	-	-	NC	-	-	-	3
Marsh Shoal	-	-	NC	-	-	-	-	NC	-	-	-	3
Moresby Shoals	-	-	NC	-	-	-	-	NC	-	-	-	3
Newby Shoal	-	-	NC	-	-	-	-	NC	-	-	-	<1
Shepparton Shoal	0.67	-	14.00	NC	-	15	-	15	NC	-	-	52
Skottowe Shoal	-	-	NC	-	-	-	-	NC	-	-	-	3
The Boxers Area	-	-	0.33	-	-	-	-	177	-	-	-	2
Tiwi Islands	-	-	0.67	-	-	-	-	66	-	-	-	5
Van Dieman Gulf Coast	-	-	NC	-	-	-	-	NC	-	-	-	<1
Van Diemen Gulf Shoals	-	-	NC	-	-	-	-	NC	-	-	-	2
Vernon Islands CR	-	-	NC	-	-	-	-	NC	-	-	-	4

NC: No contact to receptor predicted for specified threshold.

\*: This receptor is not an Environmental Value Area defined by Santos.

<: If exposure is predicted for a receptor at the low threshold but not at the moderate and/or high threshold, then the probability presented is <0.33%.

300 m<sup>3</sup> surface release of MDO from a vessel over 1 hour, as a result of external impact: Shoreline impact oil spill modelling results (maximum values across all seasons)

	Probability of	exposure (%)	Minimum time be the sea sur	before exposure on urface (days) Maximum length of shoreline impacted (km)				
Environmental Value Area	Moderate exposure values	High exposure values	Moderate exposure values	High exposure values	Moderate exposure values	High exposure values	Maximum accumulated concentration	Maximum accumulated volume along shoreline (m³) at ≥ 100g/m²
	Shoreline oil accumulation (≥ 100g/m²)	Shoreline oil accumulation (≥ 1000g/m²)	Shoreline oil concentration (≥ 100g/m²)	Shoreline oil concentration (≥ 1000g/m²)	Shoreline oil concentration (≥ 100g/m2)	Shoreline oil concentration (≥ 1000g/m²)	along shoreline (g/m²)	
Tiwi Islands	NC	NC	NC	NC	NC	NC	17	NC
Vernon Islands CR	NC	NC	NC	NC	NC	NC	19	NC

NC: No contact to receptor predicted for specified threshold.

<sup>\*:</sup> This receptor is not an Environmental Value Area defined by Santos.

<sup>&</sup>lt;: If exposure is predicted for a receptor at the low threshold but not at the moderate and/or high threshold, then the probability presented is <0.33%.

# Appendix G

## Santos Climate Change Policy

### <u>Climate</u>

Policy



3.1

### **Our Commitment**

Santos believes that access to reliable and affordable energy is critical to meeting sustainable development goals and improving living standards and economic prosperity in developed and developing nations. Santos recognises the scientific consensus of climate change assessed by the Intergovernmental Panel on Climate Change. We support the objective of the Paris Agreement to limit global temperature rise to less than 2 degrees Celsius and pursue efforts to limit the temperature rise to 1.5 degrees Celsius.

To achieve the goals of the Paris Agreement, we work with governments, customers and suppliers to minimise emissions from our operations and products whilst ensuring continued access to reliable and affordable energy to meet demand.

At Santos, we commit to continuing to take action to minimise the environmental impacts of our operations wherever practicable, including:

- Reducing our greenhouse gas emissions intensity whilst continuing to deliver critical, reliable and affordable fuels to meet demand.
- As a producer of energy, we are committed to achieving net-zero Scope 1 greenhouse gas
- emissions by 2040 and net-zero Scope 2 greenhouse gas emissions by 2050.
- Utilising carbon capture and storage, developing and trialling new emissions technologies and low carbon fuels as domestic and global markets evolve.
- Working with our customers to reduce their greenhouse gas emissions and sell the products we generate
  only to customers from countries that have a net-zero commitment or are signatories to the Paris
  Agreement.
- Providing our shareholders with an advisory vote, known as a 'Say on Climate', at regular intervals.

### **Our Actions**

The actions to achieve Our Commitment are detailed in our annual Sustainability and Climate Report and Climate Transition Action Plan, which details our decarbonisation projects and pathway to net-zero.

### Governance

The Safety and Sustainability Committee is responsible for reviewing the effectiveness of this policy.

This policy will be reviewed at appropriate intervals and revised when necessary to keep it current.

Kevin Gallagher Managing Director and CEO

Date Approved: 12 February 2025 Version: